Source: T1

Title: CR's to TS 34.123-1 v3.2.0 for approval

Agenda item: 6.1

Document for: Approval

This document contains 5 CRs to TS 34.123-1 v3.2.0. These CRs have been agreed by T1 and are put forward to TSG T for approval.

CRs with routine updates:

Spec	CR	Rev	Phase	Subject	Cat	Version- Current	Version -New	Doc-2nd- Level
34.123-1	048		R99	Idle mode test cases	F	3.2.0	3.3.0	T1-010076
34.123-1	049		R99	Updates to clause 8 of TS 34.123-1 version 3.2.0	F	3.2.0	3.3.0	T1-010106
34.123-1	050		R99	Update to GMM test case.	F	3.2.0	3.3.0	T1-010086
34.123-1	051		R99	Update to 16. SMS test specification	D	3.2.0	3.3.0	T1-010090
34.123-1	052		R99	Annex B: Update of versions of core specifications	F	3.2.0	3.3.0	T1-010091

3GPP TSG-T1 Meeting #10 Copenhagen, Denmark, 8-9 February, 2001

Tdoc T1-010076

3GPP TSG-T1/SIG Meeting #15 Copenhagen, Denmark, 5-7 February, 2001

Tdoc T1S-010004r1

CHANGE REQUEST						CR-Form-v3										
	3	4.123	-1 (CR (048		¥	rev	-	ж	Current	vers	ion:	3.2	2.0	*
For <u>HELP</u> or	า นร	sing this	form	, see	bottom	of this	s pag	ge or	look	at th	е рор-ир	text	over	the #	Ssyn	nbols.
Proposed chang	je a	affects:	ж	(U)S	IM	ME	/UE	X	Rad	io Ac	cess Ne	twork	(Cor	e Ne	twork
Title:	Ж	Idle m	ode t	est ca	ses											
Source:	Ж	Ericss	on													
Work item code:	: #	9									Date	e: #	200	01-02	-02	
Category:	Ж	F									Releas	e: #	R9	9		
		F (A (B (C (esser (corre (Addit (Func (Edito expla	ntial co sponds ion of t tional r rial mo anation		rrection ion of n) above	n in a featu	ıre)			Use <u>or</u> 2 e) R90 R90 R90 R50 RE0	6 7 8 9 L-4	(GSN (Rele (Rele (Rele (Rele (Rele	ollowing A Phase ease 19 ease 19 ease 19 ease 4 ease 5	se 2) 996) 997) 998) 999)	eases:

Reason for change: # 1

- Updated test cases to reflect recent changes in core specs: TS 25.304 V3.5.0 (2000-12), TS 23.122 V3.5.0 (2000-12), TS 25.133, V3.4.0 (2000-12)
- 2. The common requirements in the beginning of clause 6 have been updated
- 3. Requirements to receive a respons within a given time from a certain stimulus, have at large been removed as a general guard timer of 5 minutes exist. Timing requirements coming from core specifications are maintained.
- 4. Tables containing parameter test setups have been changed in order to show more clearly how parameters are modified during the test.
- 5. The following test cases have been removed:
 - "6.1.2.1 Cell Selection": Cell selection has been simplified and is now only a matter of finding a suitable cell and not necessarily the best ranked cell as in previous versions of TS 25.304.
 - "6.1.2.7 Cell reselection due to UE rejection "LA not allowed"": This test already exists in TS 34.123-1, 9.4.2.3 "Location updating / rejected / location area not allowed"
 - "6.1.2.8 Cell reselection due to UE rejection "Roaming not allowed in this LA"": This test already exists in TS 34.123-1, 9.4.2.4 "Location updating / rejected / roaming not allowed in this location area"
 - "6.1.2.10 Immediate cell evaluation": Corresponding spec in TS 25.304 has been removed.
- 6. New test cases:
 - "6.1.2.7 HCS Cell reselection using reselection timing parameters for the R criterion"

Summary of change	Updated test cases to reflect recent changes in core specs.				
Consequences if not approved:	Inconsistencies between T1 and RAN2 specifications				
Clauses affected:	₩ 6				
Other specs affected:	Other core specifications Test specifications O&M Specifications ** TS 34.123-2, Clause 4, Table 1 "Idle mode"				
Other comments:	x				

6 Idle mode operations

In the following paragraphs some explanatory text is given concerning the nature of the tests in this clause and the general behaviour of the SS is described.

Since the conformance requirements of most of the tests in this clause cannot be tested explicitly, testing is done implicitly by testing the UE behaviour from its responses to the SS. In some cases, a test is performed in multiple stages in order that the requirements can be tested within the above constraints.

For any UE all the carriers are in its supported band(s) of operation.

Unless otherwise stated in the method of test, in all of the tests of this clause:

- the SS is continuously paging the UE on all cells at the start of the test and does not respond to RACH requests from the UE. Where a test specifies that the UE is not paged in a particular cell, only idle paging is transmitted;
- the default values of the system information data fields given in TS 34.108 are used;
- the USIM is in the idle updated state in the default location area with a TMSI assigned at the beginning of each test:

It is a UE option whether to indicate access technologies to the user (TS 23.122, 4.4.3.1.2). Therefore, for combined UTRAN/GSM tests, it is indicated in parentheses which access technology shall be indicated to the user if the UE has this capability.

If a parameter is indicated with a *, it means that the parameter is calculated internally in the UE and is only shown for clarification of the test procedure.

The PLMN numbers indicated in Table 6.1.3.1 are used in test cases to associate a cell with an MCC and MNC for that cell. If no PLMN is explicitly specified, the default value is PLMN 1.

Table 6.1.3.1: Location Area Information (LAI) in System Information type 3 messages broadcast on the BCCH

PLMN	MCC1	MCC2	MCC3	MNC1	MNC2	MNC3	LAC
1	0	0	1	0	1	F	Х
2	0	0	2	1	F	F	X
3	0	0	4	2	F	F	X
4	0	0	5	3	F	F	X
5	0	0	6	4	F	F	Х
6	0	0	7	5	F	F	X
7	0	0	8	6	F	F	x
8	0	0	9	7	F	F	х
9	0	1	0	0	F	F	x
10	0	1	1	1	F	F	x
11	0	1	2	2	F	F	Х
12	0	1	3	3	F	F	Х

NOTE: 'x' denotes any value

References: TS 23.122, Annex A and TS 23.003, 2

The test channel numbers indicated in Table 6.1.3.2 and 6.1.3.3 are used in test cases to associate a cell with a frequency for that cell. The frequencies for GSM and DCS cells in Table 6.1.3.3 are identical to those used in GSM 11.10-1, clause 26.3.1. The RF signal levels for GSM cells are given in Table 6.1.3.3 and for UTRAN cells in TS 34.108, Table 6.1.1. If no channel is explicitly specified, the default value is Test Channel 1.

Table 6.1.3.2: UTRA test frequencies

		ITU region 2
Test Channel	UARFCN (uplink)	UARFCN (uplink)
1	9613	9263
2	9663	9313
3	9713	9363
4	9763	9413
5	9813	9463
6	9863	9513

References: TS 34.108, 5.1.1 and TS 34.121, 4

Table 6.1.3.3: GSM/DCS test frequencies and levels

	GS	SM 900	DCS 1 800			
Test	level	BCCH ARFCN	level	BCCH ARFCN		
Channel	dBμVemf()		dBμVemf()			
1	+65	1	+65	520		
2	+63	7	+63	580		
3	+61	39	+61	610		
4	+55	65	+55	702		
5	+59	66	+59	703		
6	+57	85	+57	830		
7	+55	97	+55	885		
8	+53	124				

	G	SM 450	DCS 480			
Test	level	BCCH ARFCN	level	BCCH ARFCN		
Channel	dBμVemf()		dBμVemf()			
1	+65	259	+65	306		
2	+63	261	+63	308		
3	+61	267	+61	314		
4	+55	268	+55	315		
5	+59	281	+59	328		
6	+57	288	+57	335		
7	+55	291	+55	338		
8	+53	293	+53	340		

	Multiba	nd 900/1800	
Test	level	BCCH ARFCN	
Channel	dBμVemf()		
1	+65	520	
2	+63	7	
3	+61	39	
4	+55	702	
5	+59	66	
6	+57	85	
7	+55	885	
8	+53	124	

	Multiba	and 450/900	Multiband 480/900		
Test	level	BCCH ARFCN	level	BCCH ARFCN	
Channel	dBμVemf()		dBμVemf()		
1	+65	1	+65	1	
2	+63	261	+63	308	
3	+61	267	+61	314	
4	+55	65	+55	65	
5	+59	281	+59	328	
6	+57	288	+57	335	
7	+55	124	+55	124	
8	+53	293	+53	340	

	Multiba	nd 450/1800	Multiband 480/1800		
Test	level	BCCH ARFCN	level	BCCH ARFCN	
Channel	dBμVemf()		dBμVemf()		
1	+65	520	+65	520	
2	+63	261	+63	308	
3	+61	267	+61	314	
4	+55	702	+55	702	
5	+59	281	+59	328	
6	+57	288	+57	335	
7	+55	885	+55	885	
8	+53	293	+53	340	

For testing an E-GSM Mobile station, the BCCH ARFCN of GSM Test Channel 7 at GSM 900 column shall be 985 (instead of 97). For testing an R-GSM Mobile station, the BCCH ARFCN of GSM Test Channel 7 at GSM 900 column shall be 965 (instead of 97).

6.1 In a pure 3GPP environment

6.1.1 PLMN selection and reselection

6.1.1.1 PLMN selection of RPLMN, HPLMN, UPLMN and OPLMN; Manual mode

6.1.1.1.1 Definition

Test to verify that the UE can present the available PLMNs in priority order to the user when asked to do so in manual mode and that the displayed PLMNs can be selected / reselected by the user. Forbidden PLMNs shall also by displayed in the list. If available, the RPLMN shall be selected at switch-on, otherwise the displayed list shall include in priority order HPLMN, User-PLMN and Operator-PLMN. The last priority in the list is "Other PLMN/access technology combinations" which is not included in this test.

Only UTRAN cells and a UE equipped with a USIM with Radio Access Technology fields set to UTRAN are considered.

6.1.1.1.2 Conformance requirement

 At switch on, the MS selects the registered PLMN (if it is available) using all access technologies that the MS is capable of and attempts to perform a Location Registration. The MS shall start its search using the access technology type stored in the RPLMN Last Used Access Technology data field on the SIM. If the RPLMN Last Used Access Technology is not available then an MS capable of GSM access technology shall start its search using GSM access technology.

On recovery from lack of coverage, the MS selects the registered PLMN (if it is available) using all access technologies that the MS is capable of and, if necessary attempts to perform a Location Registration.

If successful registration is achieved, the MS indicates the selected PLMN.

If there is no registered PLMN, or if registration is not possible due to the PLMN being unavailable or registration failure, the MS follows either Automatic or Manual Network Selection Mode Procedure depending on its operating mode.

2. Manual Network Selection Mode Procedure:

The MS indicates whether there are any PLMNs, which are available using all supported access technologies. This includes "Forbidden PLMNs" and PLMNs which only offer services not supported by the MS.

If displayed, PLMNs meeting the criteria above are presented in the following order:

- 2.1 HPLMN;
- 2.2 PLMNs contained in the "User Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order);
- 2.3 PLMNs contained in the "Operator Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order);
- 2.4 Other PLMN/access technology combinations with received high quality signal in random order;
- 2.5 Other PLMN/access technology combinations in order of decreasing signal quality.

The user may select his desired PLMN and the MS then initiates registration on this PLMN using the access technology chosen by the user for that PLMN or using the highest priority available access technology for that PLMN, if the associated access technologies have a priority order. (This may take place at any time during the presentation of PLMNs). For such a registration, the MS shall ignore the contents of the forbidden LAI and PLMN lists.

If the user does not select a PLMN, the selected PLMN shall be the one that was selected before the PLMN selection procedure started. If no such PLMN was selected or that PLMN is no longer available, then the MS shall attempt to camp on any acceptable cell and enter the limited service state.

3. If a "PLMN not allowed" message is received by an MS in response to an LR request from a VPLMN, that VPLMN is added to a list of "forbidden PLMNs" in the SIM and thereafter that VPLMN will not be accessed by the MS when in automatic mode. A PLMN is removed from the "forbidden" list if, after a subsequent manual selection of that PLMN, there is a successful LR. This list is retained when the MS is switched off or the SIM is removed. The HPLMN shall not be stored on the list of "forbidden PLMNs".

References

- 1. TS 23.122, 4.4.3.1
- 2. TS 23.122, 4.4.3.1.2
- 3. TS 23.122, 3.1

NOTE: TS 31.102 defines the USIM fields

6.1.1.1.3 Test purpose

- 1. To verify that if available, the RPLMN is selected at switch-on.
- 2. To verify that in Manual Network Selection Mode Procedure, the UE presents the HPLMN, UPLMN and OPLMN in a prioritized order.
- 3. To verify that forbidden PLMNs are also displayed in the list.

6.1.1.1.4 Method of test

Initial conditions

The UE is in manual PLMN selection mode.

All Radio Access Technology USIM fields and cells are UTRAN.

Cell	Test Channel	PLMN
Cell 1	1	PLMN 1
Cell 2	2	PLMN 2
Cell 3	3	PLMN 3
Cell 4	4	PLMN 4
Cell 5	5	PLMN 5
Cell 6	6	PLMN 6

The UE is equipped with a USIM containing default values except for those listed below.

USIM field	Priority	PLMN
EF _{LOCI}		PLMN 1
EF _{HPLMNwAcT}	1 st	PLMN 2
EF _{PLMNwACT}	1 st	PLMN 3
	2 ^{na}	PLMN 4
EFOPLMNWACT	1 st	PLMN 5
	2 ^{na}	PLMN 6
EF _{FPLMN}	PL	MN 3

NOTE: PLMN 3 is forbidden.

Test procedure

- a) The SS activates cells 1-6 and monitors the cells for random access requests from the UE
- b) The UE is switched on
- c) The SS waits for random access requests from the UE
- d) Cell 1 is switched off
- e) PLMN 4 shall be selected when the PLMN list is presented
- f) The SS waits for random access requests from the UE
- g) Cell 4 is switched off
- h) PLMN 3 shall be selected when the PLMN list is presented. The SS shall reject the Registration Request from the UE.
- i) PLMN 5 shall be selected (the list is already available)
- j) The SS waits for random access requests from the UE
- k) Cell 5 is switched off
- 1) PLMN 2 shall be selected when the PLMN list is presented
- m) The SS waits for random access requests from the UE
- n) Cell 2 is switched off
- o) PLMN 6 shall be selected when the PLMN list is presented
- p) The SS waits for random access requests from the UE
- q) Cell 6 is switched off

6.1.1.1.5 Test Requirements

- 1) In step c), the response from the UE shall be on Cell 1. The displayed PLMN shall be PLMN 1.
- 2) In step e), the list shall be presented. The priority shall be as follows: PLMN 2, PLMN 3, PLMN 4, PLMN 5, PLMN 6.
- 3) In step f), the response from the UE shall be on Cell 4. The displayed PLMN shall be PLMN 4.
- 4) In step h), the list shall be presented. The priority shall be as follows: PLMN 2, PLMN 3, PLMN 5, PLMN 6. After PLMN 3 has been selected, the list shall appear again as the UE cannot perform registration.
- 6) In step j), the response from the UE shall be on Cell 5. The displayed PLMN shall be PLMN 5.
- 7) In step l), the list shall be presented. The priority shall be as follows: PLMN 2, PLMN 3, PLMN 6.
- 8) In step m), the response from the UE shall be on Cell 2. The displayed PLMN shall be PLMN 2.
- 9) In step 0), the list shall be presented. The priority shall be as follows: PLMN 3, PLMN 6.
- 10) In step p), the response from the UE shall be on Cell 6. The displayed PLMN shall be PLMN 6.
- 11) After step q), the UE shall inform that only limited service is possible.

6.1.1.2 PLMN selection of "Other PLMN / access technology combinations"; Manual mode

6.1.1.2.1 Definition

Test to verify that the UE can present the available PLMNs in priority order to the user when asked to do so in manual mode and that the displayed PLMNs can be selected / reselected by the user. Forbidden PLMNs shall also by displayed in the list. In this test are only considered "Other PLMN/access technology combinations" in the priority list.

Only UTRAN cells and a UE equipped with a USIM with Radio Access Technology fields set to UTRAN are considered.

6.1.1.2.2 Conformance requirement

 At switch on, the MS selects the registered PLMN (if it is available) using all access technologies that the MS is capable of and attempts to perform a Location Registration. The MS shall start its search using the access technology type stored in the RPLMN Last Used Access Technology data field on the SIM. If the RPLMN Last Used Access Technology is not available then an MS capable of GSM access technology shall start its search using GSM access technology.

On recovery from lack of coverage, the MS selects the registered PLMN (if it is available) using all access technologies that the MS is capable of and, if necessary attempts to perform a Location Registration.

If successful registration is achieved, the MS indicates the selected PLMN.

If there is no registered PLMN, or if registration is not possible due to the PLMN being unavailable or registration failure, the MS follows either Automatic or Manual Network Selection Mode Procedure depending on its operating mode.

2. Manual Network Selection Mode Procedure:

The MS indicates whether there are any PLMNs, which are available using all supported access technologies. This includes "Forbidden PLMNs" and PLMNs which only offer services not supported by the MS.

If displayed, PLMNs meeting the criteria above are presented in the following order:

2.1 HPLMN;

- 2.2 PLMNs contained in the "User Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order);
- 2.3 PLMNs contained in the "Operator Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order);
- 2.4 Other PLMN/access technology combinations with received high quality signal in random order;
- 2.5 Other PLMN/access technology combinations in order of decreasing signal quality.

The user may select his desired PLMN and the MS then initiates registration on this PLMN using the access technology chosen by the user for that PLMN or using the highest priority available access technology for that PLMN, if the associated access technologies have a priority order. (This may take place at any time during the presentation of PLMNs). For such a registration, the MS shall ignore the contents of the forbidden LAI and PLMN lists.

If the user does not select a PLMN, the selected PLMN shall be the one that was selected before the PLMN selection procedure started. If no such PLMN was selected or that PLMN is no longer available, then the MS shall attempt to camp on any acceptable cell and enter the limited service state.

3. If a "PLMN not allowed" message is received by an MS in response to an LR request from a VPLMN, that VPLMN is added to a list of "forbidden PLMNs" in the SIM and thereafter that VPLMN will not be accessed by the MS when in automatic mode. A PLMN is removed from the "forbidden" list if, after a subsequent manual selection of that PLMN, there is a successful LR. This list is retained when the MS is switched off or the SIM is removed. The HPLMN shall not be stored on the list of "forbidden PLMNs".

References

- 1. TS 23.122, 4.4.3.1
- 2. TS 23.122, 4.4.3.1.2
- 3. TS 23.122, 3.1

NOTE: TS 31.102 defines the USIM fields

6.1.1.2.3 Test purpose

- 1. To verify that in Manual Network Selection Mode Procedure, the UE presents "Other PLMN/access technology combinations" in a prioritized order according to conformance requirement 2.4 and 2.5.
- 2. To verify that forbidden PLMNs are also displayed in the list.

6.1.1.2.4 Method of test

Initial conditions

The UE is in automatic mode.

All Radio Access Technology USIM fields and cells are UTRAN.

Cell	CPICH_Ec/lo [dB]	Test Channel	PLMN
Cell 1	-15	1	PLMN 6
Cell 2	-15	2	PLMN 7
Cell 3	-15	3	PLMN 8
Cell 4	-16	4	PLMN 9
Cell 5	-17	5	PLMN 10
Cell 6	-18	6	PLMN 11

NOTE: Cell 1 is OPLMN 2nd priority. HQ signal [FFS] is on Cell 2-3 but not on Cell 4-6.

The UE is equipped with a USIM containing default values except for those listed below.

USIM field	Priority	PLMN
EF _{LOCI}		PLMN 1
EF _{HPLMNwAcT}	1 st	PLMN 2
EF _{PLMNwACT}	1 st	PLMN 3
	2 ^{na}	PLMN 4
EFOPLMNWACT	1 st	PLMN 5
	2 nd	PLMN 6
EF _{FPLMN}	PLMN 10	

NOTE: PLMN 10 is forbidden.

Test procedure

- a) The SS activates cells 1-6 and monitors the cells for random access requests from the UE
- b) The UE is switched on
- c) PLMN 9 shall be selected when the PLMN list is presented
- d) The SS waits for random access requests from the UE
- e) Cell 4 is switched off
- f) PLMN 7 shall be selected when the PLMN list is presented
- g) The SS waits for random access requests from the UE
- h) Cell 2 is switched off
- i) PLMN 6 shall be selected when the PLMN list is presented
- j) The SS waits for random access requests from the UE
- k) Cell 1 is switched off
- 1) PLMN 11 shall be selected when the PLMN list is presented
- m) The SS waits for random access requests from the UE
- n) Cell 6 is switched off
- o) PLMN 10 shall be selected when the PLMN list is presented. The SS shall reject the Registration Request from the UE.
- p) Cell 5 is switched off
- q) PLMN 8 shall be selected (the list is already available)
- r) The SS waits for random access requests from the UE
- s) Cell 3 is switched off

6.1.1.2.5 Test Requirements

- 1) In step c), the list shall be presented. The priority shall be as follows: PLMN 6 followed by PLMN 7, PLMN 8 in random order, followed by PLMN 9, PLMN 10, PLMN 11.
- 2) In step d), the response from the UE shall be on Cell 4. The displayed PLMN shall be PLMN 9.
- 3) In step f), the list shall be presented. The priority shall be as follows: PLMN 6 followed by PLMN 7, PLMN 8 in random order, followed by PLMN 10, PLMN 11.
- 4) In step g), the response from the UE shall be on Cell 2. The displayed PLMN shall be PLMN 7.
- 5) In step i), the list shall be presented. The priority shall be as follows: PLMN 6, PLMN 8, PLMN 10, PLMN 11.
- 6) In step j), the response from the UE shall be on Cell 1. The displayed PLMN shall be PLMN 6.

- 7) In step l), the list shall be presented. The priority shall be as follows: PLMN 8, PLMN 10, PLMN 11.
- 8) In step m), the response from the UE shall be on Cell 6. The displayed PLMN shall be PLMN 11.
- 9) In step o), the list shall be presented. The priority shall be as follows: PLMN 8, PLMN 10. After PLMN 10 has been selected, the list shall appear again as the UE cannot perform registration.
- 10) In step q), the list shall be presented and shall only contain PLMN 8.
- 11) In step r), the UE shall respond on Cell 3. The displayed PLMN shall be PLMN 8.
- 12) After step s), the UE shall inform that no network is available.

6.1.1.3 PLMN selection / reselection; independence of RF level and preferred PLMN; Manual mode

6.1.1.3.1 Definition

Test to verify that in Manual Network Selection Mode, the UE is able to obtain normal service on a PLMN which is neither the better nor a preferred PLMN and that it tries to obtain service on a VPLMN if and only if the user selects it manually.

6.1.1.3.2 Conformance requirement

 At switch on, the MS selects the registered PLMN (if it is available) using all access technologies that the MS is capable of and attempts to perform a Location Registration. The MS shall start its search using the access technology type stored in the RPLMN Last Used Access Technology data field on the SIM. If the RPLMN Last Used Access Technology is not available then an MS capable of GSM access technology shall start its search using GSM access technology.

On recovery from lack of coverage, the MS selects the registered PLMN (if it is available) using all access technologies that the MS is capable of and, if necessary attempts to perform a Location Registration.

If successful registration is achieved, the MS indicates the selected PLMN.

If there is no registered PLMN, or if registration is not possible due to the PLMN being unavailable or registration failure, the MS follows either Automatic or Manual Network Selection Mode Procedure depending on its operating mode.

2. Manual mode - Here the MS indicates to the user which PLMNs are available. Only when the user makes a manual selection does the MS try to obtain normal service on the VPLMN.3. Manual Network Selection Mode Procedure:

The MS indicates whether there are any PLMNs, which are available using all supported access technologies. This includes "Forbidden PLMNs" and PLMNs which only offer services not supported by the MS.

If displayed, PLMNs meeting the criteria above are presented in the following order:

- 1.1 HPLMN;
- 1.2 PLMNs contained in the "User Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order);
- 1.3 PLMNs contained in the "Operator Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order);
- 1.4 Other PLMN/access technology combinations with received high quality signal in random order;
- 1.5 Other PLMN/access technology combinations in order of decreasing signal quality.

The user may select his desired PLMN and the MS then initiates registration on this PLMN using the access technology chosen by the user for that PLMN or using the highest priority available access technology for that PLMN, if the associated access technologies have a priority order. (This may take place at any time during the presentation of PLMNs). For such a registration, the MS shall ignore the contents of the forbidden LAI and PLMN lists.

References

- 1. TS 23.122, 4.4.3.1
- 2. TS 23.122, 3.1
- 3. TS 23.122, 4.4.3.1.2

NOTE: TS 31.102 defines the USIM fields

6.1.1.3.3 Test purpose

- 1. To verify that the selected PLMN at switch-on is the HPLMN
- 2. To verify that in Manual Network Selection Mode Procedure the UE tries to obtain service on a VPLMN if and only if the user selects it manually
- 3. To verify that the UE is able to obtain normal service on a PLMN which is neither the better nor a prefered PLMN.

6.1.1.3.4 Method of test

Initial conditions

The UE is in manual PLMN selection mode.

PLMN 1 is the HPLMN (i.e. IMSI). There is no RPLMN stored in the USIM (i.e. field EF_{BCCH}).

PLMN 2 is not contained in any preferred PLMN list on the USIM.

PLMN 3 is contained in the UPLMN selector list on the USIM and has a different MCC-MNC from PLMN 1.

The UE is equipped with a USIM containing default values.

Step a-d:

Parameter	Unit	Cell 1	Cell 2
Test Channel		1	2
\hat{I}_{or}/I_{oc}	dB	2.3	-4.1
CPICH_Ec/lo	dB	-12.0	-15.5
CPICH RSCP	dBm	-77.7	-84.1
Qqualmin	dB	-20	-20
Qrxlevmin	dBm	-100	-100
Squal*	dB	8.0	4.5
Srxlev*	dBm	22.3	15.9
PLMN		1	2

Step e-f:

CPICH_Ec/Io	ON -> OFF	-15.5

Step g-h:

CPICH_Ec/lo	OFF -> -12.0	-15.5
PLMN	1 -> 3	2

Step i-l:

CPICH_Ec/lo	-12	-15.5 -> OFF
-------------	-----	--------------

Test procedure

- a) The SS activates cells 1 and 2
- b) The UE is switched on.
- c) PLMN 1 is selected
- d) The SS waits for random access requests from the UE. A complete Location Update is done.
- e) Cell 1 is switched off
- f) The SS waits to see if there is any random access requests from the UE
- g) Cell 1 is switched on
- h) The SS waits to see if there is any random access requests from the UE
- i) PLMN 2 is selected manually
- j) The SS waits for random access requests from the UE. A complete Location Update is done.
- k) Cell 2 is switched off
- 1) The SS waits to see if there is any random access requests from the UE

6.1.1.3.5 Test Requirements

- 1) In step d), there shall be a response on Cell 1. The selected PLMN shall be PLMN 1.
- 2) In step f), there shall be no response from the UE.
- 3) In step h), there shall be no response from the UE.
- 4) In step j), there shall be a response on Cell 2. The selected PLMN shall be PLMN 2.
- 5) In step 1), there shall be no response from the UE.

6.1.1.4 PLMN selection of RPLMN, HPLMN, UPLMN and OPLMN; Automatic mode

6.1.1.4.1 Definition

Test to verify that in Automatic Network Selection Mode, the UE selects PLMNs in a prioritized order. Forbidden PLMNs shall not be selected. If available, the RPLMN shall be selected at switch-on, otherwise the list shall include in priority order HPLMN, User-PLMN and Operator-PLMN. The last priority in the list is "Other PLMN/access technology combinations" which is not included in this test.

Only UTRAN cells and a UE equipped with a USIM with Radio Access Technology fields set to UTRAN are considered.

6.1.1.4.2 Conformance requirement

 At switch on, the MS selects the registered PLMN (if it is available) using all access technologies that the MS is capable of and attempts to perform a Location Registration. The MS shall start its search using the access technology type stored in the RPLMN Last Used Access Technology data field on the SIM. If the RPLMN Last Used Access Technology is not available then an MS capable of GSM access technology shall start its search using GSM access technology.

On recovery from lack of coverage, the MS selects the registered PLMN (if it is available) using all access technologies that the MS is capable of and, if necessary attempts to perform a Location Registration.

If successful registration is achieved, the MS indicates the selected PLMN.

If there is no registered PLMN, or if registration is not possible due to the PLMN being unavailable or registration failure, the MS follows either Automatic or Manual Network Selection Mode Procedure depending on its operating mode.

2. Automatic Network Selection Mode Procedure:

The MS selects and attempts registration on other PLMNs, if available and allowable in the following order:

- 2.1 HPLMN (if not previously selected);
- 2.2 Each PLMN in the "User Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order)
- 2.3 Each PLMN in the "Operator Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order)
- 2.4 Other PLMN/access technology combinations with received high quality signal in random order
- 2.5 Other PLMN/access technology combinations in order of decreasing signal quality

If successful registration is achieved, the MS indicates the selected PLMN.

If registration cannot be achieved because no PLMNs are available and allowable, the MS indicates "no service" to the user, waits until a new PLMN is available and allowable and then repeats the procedure.

If there were one or more PLMNs which were available and allowable, but an LR failure made registration on those PLMNs unsuccessful or an entry in a forbidden LAI list prevented a registration attempt, the MS selects the first such PLMN again and enters a limited service state.

3. If a "PLMN not allowed" message is received by an MS in response to an LR request from a VPLMN, that VPLMN is added to a list of "forbidden PLMNs" in the SIM and thereafter that VPLMN will not be accessed by the MS when in automatic mode. A PLMN is removed from the "forbidden" list if, after a subsequent manual selection of that PLMN, there is a successful LR. This list is retained when the MS is switched off or the SIM is removed. The HPLMN shall not be stored on the list of "forbidden PLMNs".

References

- 1. TS 23.122, 4.4.3.1
- 2. TS 23.122, 4.4.3.1.1
- 3. TS 23.122, 3.1

NOTE: TS 31.102 defines the USIM fields

6.1.1.4.3 Test purpose

- 1. To verify that if available, the RPLMN is selected at switch-on.
- 2. To verify that in Automatic Network Selection Mode Procedure, the UE selects the RPLMN, HPLMN, UPLMN and OPLMN in a prioritized order.
- 3. To verify that forbidden PLMNs are not selected.

6.1.1.4.4 Method of test

Initial conditions

The UE is in automatic mode.

MCC of HPLMN shall be different from other PLMNs in the network to avoid periodic search for HPLMN.

All Radio Access Technology USIM fields and cells are UTRAN.

Cell	Test Channel	PLMN
Cell 1	1	PLMN 1
Cell 2	2	PLMN 2
Cell 3	3	PLMN 3
Cell 4	4	PLMN 4
Cell 5	5	PLMN 5
Cell 6	6	PLMN 6

The UE is equipped with a USIM containing default values except for those listed below.

USIM field	Priority	PLMN
EF _{LOCI}		PLMN 1
EF _{HPLMNwAcT}	1 st	PLMN 2
EF _{PLMNwACT}	1 st	PLMN 3
	2 nd	PLMN 4
EFOPLMNWACT	1 st	PLMN 5
	2 nd	PLMN 6
EF _{FPLMN}	PLMN 3	

NOTE: PLMN 3 is forbidden.

Test procedure

- a) The SS activates cells 1-6 and monitors the cells for random access requests from the UE
- b) The UE is switched on
- c) The SS waits for random access requests from the UE
- d) Cell 1 is switched off
- e) The SS waits for random access requests from the UE
- f) Cell 2 is switched off
- g) The SS waits for random access requests from the UE
- i) Cell 4 is switched off
- j) The SS waits for random access requests from the UE
- k) Cell 5 is switched off
- 1) The SS waits for random access requests from the UE
- m) Cell 6 is switched off

6.1.1.4.5 Test Requirements

- 1) In step c), the response from the UE shall be on Cell 1. The displayed PLMN shall be PLMN 1.
- 2) In step e), the response from the UE shall be on Cell 2. The displayed PLMN shall be PLMN 2.
- 3) In step g), the response from the UE shall be on Cell 4. The displayed PLMN shall be PLMN 4.
- 4) In step j), the response from the UE shall be on Cell 5. The displayed PLMN shall be PLMN 5.
- 5) In step l), the response from the UE shall be on Cell 6. The displayed PLMN shall be PLMN 6.
- 6) After step m), the UE shall inform that only limited service is possible

6.1.1.5 PLMN selection of "Other PLMN / access technology combinations"; Automatic mode

6.1.1.5.1 Definition

Test to verify that in Automatic Network Selection Mode, the UE selects PLMNs in a prioritized order. Forbidden PLMNs shall not be selected. In this test are only considered "Other PLMN/access technology combinations" in the priority list.

Only UTRAN cells and a UE equipped with a USIM with Radio Access Technology fields set to UTRAN are considered.

6.1.1.5.2 Conformance requirement

 At switch on, the MS selects the registered PLMN (if it is available) using all access technologies that the MS is capable of and attempts to perform a Location Registration. The MS shall start its search using the access technology type stored in the RPLMN Last Used Access Technology data field on the SIM. If the RPLMN Last Used Access Technology is not available then an MS capable of GSM access technology shall start its search using GSM access technology.

On recovery from lack of coverage, the MS selects the registered PLMN (if it is available) using all access technologies that the MS is capable of and, if necessary attempts to perform a Location Registration.

If successful registration is achieved, the MS indicates the selected PLMN.

If there is no registered PLMN, or if registration is not possible due to the PLMN being unavailable or registration failure, the MS follows either Automatic or Manual Network Selection Mode Procedure depending on its operating mode.

2. Automatic Network Selection Mode Procedure:

The MS selects and attempts registration on other PLMNs, if available and allowable in the following order:

- 2.1 HPLMN (if not previously selected);
- 2.2 Each PLMN in the "User Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order)
- 2.3 Each PLMN in the "Operator Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order)
- 2.4 Other PLMN/access technology combinations with received high quality signal in random order
- 2.5 Other PLMN/access technology combinations in order of decreasing signal quality

If successful registration is achieved, the MS indicates the selected PLMN.

If registration cannot be achieved because no PLMNs are available and allowable, the MS indicates "no service" to the user, waits until a new PLMN is available and allowable and then repeats the procedure.

If there were one or more PLMNs which were available and allowable, but an LR failure made registration on those PLMNs unsuccessful or an entry in a forbidden LAI list prevented a registration attempt, the MS selects the first such PLMN again and enters a limited service state.

3. If a "PLMN not allowed" message is received by an MS in response to an LR request from a VPLMN, that VPLMN is added to a list of "forbidden PLMNs" in the SIM and thereafter that VPLMN will not be accessed by the MS when in automatic mode. A PLMN is removed from the "forbidden" list if, after a subsequent manual selection of that PLMN, there is a successful LR. This list is retained when the MS is switched off or the SIM is removed. The HPLMN shall not be stored on the list of "forbidden PLMNs".

References

1. TS 23.122, 4.4.3.1

- 2. TS 23.122, 4.4.3.1.1
- 3. TS 23.122, 3.1

NOTE: TS 31.102 defines the USIM fields

6.1.1.5.3 Test purpose

- 1. To verify that in Automatic Network Selection Mode Procedure, the UE selects "Other PLMN/access technology combinations" in a prioritized order according to conformance requirement 2.4 and 2.5.
- 2. To verify that forbidden PLMNs are not selected.

6.1.1.5.4 Method of test

Initial conditions

The UE is in automatic mode.

MCC of HPLMN shall be different from other PLMNs in the network to avoid periodic search for HPLMN.

All Radio Access Technology USIM fields and cells are UTRAN.

Cell	CPICH_Ec/lo [dB]	Test Channel	PLMN
Cell 1	-15	1	PLMN 6
Cell 2	-15	2	PLMN 7
Cell 3	-15	3	PLMN 8
Cell 4	-16	4	PLMN 9
Cell 5	-17	5	PLMN 10
Cell 6	-18	6	PLMN 11

NOTE: Cell 1 is OPLMN 2nd priority. HQ signal [FFS] is on Cell 2-3 but not on Cell 4-6.

The UE is equipped with a USIM containing default values except for those listed below.

USIM field	Priority	PLMN
EF _{LOCI}		PLMN 1
EF _{HPLMNwAcT}	1 st	PLMN 2
EF _{PLMNwACT}	1 st	PLMN 3
	2 nd	PLMN 4
EFOPLMNWACT	1 st	PLMN 5
	2 nd	PLMN 6
EF _{FPLMN}	PLMN 10	

NOTE: PLMN 10 is forbidden.

Test procedure

- a) The SS activates cells 1-6 and monitors the cells for random access requests from the UE
- b) The UE is switched on
- c) The SS waits for random access requests from the UE
- d) Cell 1 is switched off
- e) The SS waits for random access requests from the UE
- f) The cell associated to the currently shown PLMN shall be switched off
- g) The SS waits for random access requests from the UE
- h) The cell associated to the currently shown PLMN shall be switched off
- i) The SS waits for random access requests from the UE

- j) Cell 4 is switched off
- k) The SS waits for random access requests from the UE
- 1) Cell 6 is switched off

6.1.1.5.5 Test Requirements

- 1) In step c), the response from the UE shall be on Cell 1. The displayed PLMN shall be be PLMN 6.
- 2) In step e), the response from the UE shall be on either Cell 2 or 3. The displayed PLMN shall be the one associated with the cell on which the response was received.
- 3) In step g), the response from the UE shall be on either Cell 2 or 3 (excluding the cell in step 2). The displayed PLMN shall be the one associated with the cell on which the response was received.
- 4) In step i), the response from the UE shall be on Cell 4. The displayed PLMN shall be PLMN 9 within.
- 5) In step k), the response from the UE shall be on Cell 6. The displayed PLMN shall be PLMN 11.
- 6) After step 1), the UE shall inform that only limited service is possible

6.1.1.6 UE will transmit only if PLMN available

6.1.1.6.1 Definition

Test to verify that the UE will not generate any RF output if no PLMN is available.

6.1.1.6.2 Conformance requirement

[FFS: Currently no requirements exist in core specs.]

6.1.1.6.3 Test purpose

- 1. To verify that the UE does not give any "Service indication" when no PLMN is available
- 2. To verify that the UE will not generate any RF output when no PLMN is available

6.1.1.6.4 Method of test

Initial conditions

Parameter	Unit	Cell 1	Cell 2	Cell 3
\hat{I}_{or}/I_{oc}	dB	14.6	12.1	10.6
CPICH_Ec/lo	dB	-13.0	-15.5	-17
CPICH RSCP	dBm	-65.4	-67.9	-69.4
Qqualmin	dB	-20	-20	-20
Qrxlevmin	dBm	-100	-100	-100
Squal*	dB	7	4.5	3
Srxlev*	dBm	34.6	32.1	30.6

Test procedure

- a) The SS activates the cells 1-3 and monitors them for random access requests from the UE
- b) The UE is switched on.
- c) The SS waits for random access request from the UE

- d) Cells 1-3 are switched off
- e) The SS shall wait 20 sec. to allow the UE to detect the loss of cells
- f) By MMI, an attempt to originate a call is made
- g) By MMI, an attempt to originate an emergency call is made (only if UE supports speech)

6.1.1.6.5 Test Requirements

- 1) In step c), there shall be a response on cell 1
- 2) In step f) and g), the UE shall not produce any RF output, neither give any "service indication".

6.1.2 Cell selection and reselection

6.1.2.1 Cell reselection

6.1.2.1.1 Definition

Test to verify that the UE performs the cell reselection correctly for intra/inter-frequency cells if the serving cell becomes barred or S<0.

6.1.2.1.2 Conformance requirement

- 1. The UE shall evaluate the cell reselection criteria based on radio measurements, and if a better cell is found that cell is selected, procedure *Cell reselection*. The change of cell may imply a change of radio access technology.
- 2. A "suitable cell" is a cell on which the UE may camp on to obtain normal service. Such a cell shall fulfil all the following requirements.
 - 2.1 The cell is part of the selected PLMN
 - 2.2 The cell is not barred
 - 2.3 The cell is not part of a forbidden registration area
 - 2.4 The cell selection criteria are fulfilled
 - 2.5 The SoLSA criteria are fulfilled [SoLSA support is not in the current release]
- 3. When camped normally, the UE shall execute the cell reselection evaluation process on the following occasions/triggers:
 - 3.1 UE internal triggers, so as to meet performance as specified in TS 25.133
 - 3.2 When information on the BCCH used for the cell reselection evaluation procedure has been modified
- 4. Cell Reselection Criteria:
 - 4.1 The UE shall perform ranking of all cells that fulfil the S criterion
 - 4.2 The cells shall be ranked according to the R criteria. The best ranked cell is the cell with the highest R value. If a FDD cell is ranked as the best cell, the UE shall perform cell re-selection to that FDD cell
 - 4.3 The UE shall reselect the new cell, if the cell reselection criteria are fulfilled during a time interval Treselection.
 - 4.4 The cell-ranking criterion R is derived from Q, Qhyst, Qoffset, TEMP_OFFSET and PENALTY_TIME. However, TEMP_OFFSET and PENALTY_TIME are only applicable when HCS is applied that is when serving cell belongs to a hierarchical structure.

5. Non-suitable cells (Squal > 0 and Srxlev > 0):

If the best cell according to cell reselection criteria does not fulfil all requirements for a suitable cell, that cell, together with all cells on that frequency shall be removed as candidate for cell re-selection

- 6. When cell status "barred" is indicated, the UE shall select another cell according to the following rule:
 - 6.1 If the "Intra-frequency cell re-selection indicator" IE in Cell Access Restriction IE is set to value "allowed", the UE may select another cell on the same frequency if selection/re-selection criteria are fulfilled.
 - 6.2 If the "Intra-frequency cell re-selection indicator" IE is set to "not allowed" the UE shall not re-select a cell on the same frequency as the barred cell. For emergency call, the Intra-frequency cell re-selection indicator IE" shall be ignored, i.e. even if it is set to "not allowed" the UE may select another intra-frequency cell.

References

- 1. TS 25.304, 5.2.2
- 2. TS 25.304, 4.3
- 3. TS 25.304, 5.2.5.1
- 4. TS 25.304, 5.2.6.1.4
- 5. TS 25.304, 5.2.6.1.3
- 6. TS 25.304, 5.3.1.1

6.1.2.1.3 Test purpose

- 1. To verify that the UE performs cell reselection on the following occasions:
 - 1.1 Serving cell becomes barred
 - 1.2 S<0 for serving cell
- 2. To verify conformance requirement 5 and 6

NOTE: Reselection triggered by the cell becoming a part of a forbidden registration area is tested in clause 9.4.2.3 "Location updating / rejected / location area not allowed" and 9.4.2.4 "Location updating / rejected / roaming not allowed in this LA"

6.1.2.1.4 Method of test

Initial conditions

Treselection, Qhyst, Qoffset, TEMP_OFFSET and PENALTY_TIME are not used, so the cell-ranking criterion R equals CPICH_Ec/Io.

Step a-c:

Parameter	Unit	Cell 1	Cell 2	Cell 3
Test Channel		1	1	2
\hat{I}_{or}/I_{oc}	dB	4.4	2.4	-6.0
CPICH_Ec/lo	dB	-13.0	-15.0	-17.0
CPICH RSCP	dBm	-75.6	-77.6	-86.0
Qqualmin	dB	-20	-20	-20
Qrxlevmin	dBm	-100	-100	-100
Squal*	dB	7.0	5.0	3.0
Srxlev*	dBm	24.4	22.4	14
Intra-frequency cell re-selection indicator		Not Allowed	Not Allowed	Not Allowed
CellBarred		0	0	0

Step d-f:

	CellBarred		0->1	0	0
--	------------	--	------	---	---

Step g-h:

Intra-frequency cell re-selection indicator	Allowed	Allowed	Allowed
CellBarred	0->1	0	0

Step i:

Qqualmin	dB	-20 -> -10	-20	-20
Squal*	dB	7.0 -> -3.0	5	3

Test procedure

- a) The SS activates Cell 1-3 and monitors them for random access requests from the UE.
- b) The UE is switched on.
- c) The SS waits for random access requests from the UE
- d) The SS sets Cell 1 to be barred
- e) The SS waits for random access requests from the UE
- f) The stored information cell selection list in the UE is deleted and the UE is switched off.
- g) Step a-e) is repeated except that "Intra-frequency cell re-selection indicator" is set to "Allowed"
- h) The stored information cell selection list in the UE is deleted and the UE is switched off.
- i) Step a-e) is repeated except that in step d), Qqualmin is increased to -10 dB, so S will become negative instead of the cell being barred while maintaining the same RF level.

6.1.2.1.5 Test requirements

- 1) In step c), the UE shall select a cell to camp on and eventually make a reselection to Cell 1
- 2) In step e), the UE shall respond on Cell 3
- 3) In step g), the UE shall respond on Cell 2
- 4) In step i), the UE shall respond on Cell 2

6.1.2.2 Cell reselection using Qhyst, Qoffset and Treselection

6.1.2.2.1 Definition

Test to verify that the UE performs the cell reselection correctly if system information parameters Qoffset, Qhyst and Treselection are applied for non-hierarchical cell structures. TEMP_OFFSET and PENALTY_TIME are only applicable when HCS is applied and are tested in clause 6.1.2.4 and 6.1.2.5.

6.1.2.2.2 Conformance requirement

- 1. When camped normally, the UE shall execute the cell reselection evaluation process on the following occasions/triggers:
 - 1.1 UE internal triggers, so as to meet performance as specified in TS 25.133
 - 1.2 When information on the BCCH used for the cell reselection evaluation procedure has been modified
- 2. When camped normally, the UE shall execute the cell reselection evaluation process on the following occasions/triggers:
 - 2.1 UE internal triggers, so as to meet performance as specified in TS 25.133
 - 2.2 When information on the BCCH used for the cell reselection evaluation procedure has been modified
- 3. Cell Reselection Criteria:
 - 3.1 The UE shall perform ranking of all cells that fulfil the S criterion
 - 3.2 The cells shall be ranked according to the R criteria. The best ranked cell is the cell with the highest R value. If a FDD cell is ranked as the best cell, the UE shall perform cell re-selection to that FDD cell
 - 3.3 The UE shall reselect the new cell, if the cell reselection criteria are fulfilled during a time interval Treselection.
 - 3.4 The cell-ranking criterion R is derived from Q, Qhyst, Qoffset, TEMP_OFFSET and PENALTY_TIME. However, TEMP_OFFSET and PENALTY_TIME are only applicable when HCS is applied that is when serving cell belongs to a hierarchical structure.

References

- 1. TS 25.304, 5.2.2
- 2. TS 25.304, 5.2.5.1
- 3. TS 25.304, 5.2.6.1.4

6.1.2.2.3 Test purpose

- 1. To verify that the UE calculates R from Qhyst and Qoffset and that the modification of these parameters on the BCCH triggers the cell reselection evaluation process. TEMP_OFFSET and PENALTY_TIME are not applied.
- 2. To verify that the UE reselects the new cell, if the cell reselection criteria are fulfilled during a time interval Treselection.

6.1.2.2.4 Method of test

Initial conditions

Step a-c:

Parameter	Unit	Cell 1	Cell 2
\hat{I}_{or}/I_{oc}	dB	8.6	5.1
CPICH_Ec/lo	dB	-12	-15.5
CPICH RSCP	dBm	-71.4	-74.9
Qqualmin	dB	-20	-20
Qrxlevmin	dBm	-100	-100
Squal*	dB	8	4.5
Srxlev*	dBm	28.6	25.1
Qhyst2 _s	dB	10	
R _s *	dB	-2	
R _n *	dB	-15.5	_

Step d-e:

\hat{I}_{or}/I_{oc}	dB	8.6 -> 5.1	5.1 -> 8.6
CPICH_Ec/lo	dB	-12 -> -15.5	-15.5 -> -12
R _s *	dB	-2 -> -5.5	
R _n *	dB	-15.5 -> -12	

Step f-g:

Qhyst2 _s	dB	10 -> 0	
R _s *	dB	-5.5 -> -15.5	
R _n *	dB	-12	

Step h-j:

\hat{I}_{or}/I_{oc}	dB	8.6	5.1
CPICH_Ec/lo	dB	-12	-15.5
Qoffset2 _{s,n}	dB	10	
R _s *	dB	-12	
R _n *	dB	-25.5	

Step k-1:

\hat{I}_{or}/I_{oc}	dB	8.6 -> 5.1	5.1 -> 8.6
CPICH_Ec/lo	dB	-12 -> -15.5	-15.5 -> -12
R _s *	dB	-12 -> -15.5	
R _n *	dB	-25.5 -> -22	

Step m-n:

Qoffset2 _{s,n}	dB	10 -> 0	
R _s *	dB	-15.5	
R _n *	dB	-22 -> -12	

Step o-p:

Treselections	S	30	

Test procedure

- a) The SS activates Cell 1 and 2 and monitors them for random access requests from the UE.
- b) The UE is switched on.
- c) The SS waits to see if there is any random access requests from the UE
- d) The SS changes the level of Cell 1 and 2
- e) The SS waits for random access requests from the UE
- f) The SS resets Qhyst for Cell 1
- g) The SS waits for random access requests from the UE

- h) The stored information cell selection list in the UE is deleted and the UE is switched off
- i) The UE is switched on.
- j) The SS waits to see if there is any random access requests from the UE
- k) The SS changes the level of Cell 1 and 2
- 1) The SS waits for random access requests from the UE
- m) The SS resets Qoffset for Cell 1
- n) The SS waits for random access requests from the UE
- o) Step h-n) is repeated except that Treselection is 30 sec.

6.1.2.2.5 Test Requirements

- 1) In step c), the UE shall select a cell to camp on and eventually make a reselection to Cell 1
- 2) In step e), the UE shall keep responding on Cell 1
- 3) In step g), the UE shall respond on Cell 2
- 4) In step j), the UE shall select a cell to camp on and eventually make a reselection to Cell 1
- 5) In step 1), the UE shall keep responding on Cell 1
- 6) In step n), the UE shall respond on Cell 2
- 7) In step o), the UE shall respond as in previous steps except that with the reselection to Cell 2, there shall be no response from the UE on Cell 2 within [FFS: Treselection taken into account] seconds of broadcasting Qoffset but the UE shall respond on Cell 2 within [FFS: Treselection taken into account] seconds

6.1.2.3 HCS Cell reselection

6.1.2.3.1 Definition

Test to verify that the UE performs the cell reselection correctly for hierarchical cell structures. This shall be done according to the HCS priority, the received signal quality value Q and the quality level threshold criterion H.

6.1.2.3.2 Conformance requirement

- 1. When camped normally, the UE shall execute the cell reselection evaluation process on the following occasions/triggers:
 - 1.1 UE internal triggers, so as to meet performance as specified in TS 25.133
 - 1.2 When information on the BCCH used for the cell reselection evaluation procedure has been modified
- 2. Cell Reselection Criteria for hierarchical cells:
 - 2.1 The quality level threshold criterion H for hierarchical cell structures is used to determine whether prioritised ranking according to hierarchical cell re-selection rules shall apply, and is calculated from the Q, Qhcs, TEMP_OFFSET and PENALTY_TIME parameters.
 - 2.2 The UE shall perform ranking of all cells that fulfil the S criterion among all cells that have the highest HCS PRIO among those cells that fulfil the criterion H >= 0.
 - 2.3 The cells shall be ranked according to the R criteria. The best ranked cell is the cell with the highest R value. If a FDD cell is ranked as the best cell, the UE shall perform cell re-selection to that FDD cell

- 2.4 The UE shall reselect the new cell, if the cell reselection criteria are fulfilled during a time interval Treselection.
- 2.5 The cell-ranking criterion R is derived from Q, Qhyst, Qoffset, TEMP_OFFSET, PENALTY_TIME.

References

- 1. TS 25.304, 5.2.2
- 2. TS 25.304, 5.2.6.1.4

6.1.2.3.3 Test purpose

- 1. Verify that the UE ignores cells with H<0 for reselection and that H is calculated from Qhcs. The modification of this parameter on the BCCH shall trigger the cell reselection evaluation process.
- 2. Verify that the UE ranks cells based on both HCS priority and R. Qhyst, Qoffset, TEMP_OFFSET, PENALTY_TIME and Treselection are not applied so R equals CPICH_Ec/Io

6.1.2.3.4 Method of test

Initial conditions

Step a-c:

Parameter	Unit	Cell 1	Cell 2	Cell 3
\hat{I}_{or}/I_{oc}	dB	14.6	12.1	10.6
CPICH_Ec/lo	dB	-13.0	-15.5	-17
CPICH RSCP	dBm	-65.4	-67.9	-69.4
Qqualmin	dB	-20	-20	-20
Qrxlevmin	dBm	-100	-100	-100
Squal*	dB	7	4.5	3
Srxlev*	dBm	34.6	32.1	30.6
HCS priority		6	7	7
Qhcs _s	dB	-30	-10	-10
H _s *	dB	17	-5.5	-7

Step d-e:

Qhcs _s	dB	-30	-10	-10 -> -30
H _s *	dB	17	-5.5	-7 -> 13

Step f-g:

Qhcss	dB	-30	-10 -> -30	-30
H _s *	dB	17	-5.5 -> 14.5	13

Test procedure

- a) The SS activates the cells 1-3 and monitors them for random access requests from the UE
- b) The UE is switched on.
- c) The SS waits for random access requests from the UE
- d) The SS changes Qhcs for Cell 3
- e) The SS waits for random access requests from the UE
- f) The SS changes Qhcs for Cell 2
- g) The SS waits for random access requests from the UE

6.1.2.3.5 Test requirements

- 1) In step c), the UE shall select a cell to camp on and eventually make a reselection to Cell 1
- 2) In step e), the UE shall respond on Cell 3
- 3) In step g), the UE shall respond on Cell 2

6.1.2.4 HCS Cell reselection using reselection timing parameters for the H criterion

6.1.2.4.1 Definition

Test to verify that the UE performs the cell reselection correctly for hierarchical cell structures using TEMP_OFFSET and PENALTY_TIME applied to the H criterion.

6.1.2.4.2 Conformance requirement

- 1. When camped normally, the UE shall execute the cell reselection evaluation process on the following occasions/triggers:
 - 1.1 UE internal triggers, so as to meet performance as specified in TS 25.133
 - 1.2 When information on the BCCH used for the cell reselection evaluation procedure has been modified
- 2. Cell Reselection Criteria for hierarchical cells:
 - 2.1 The quality level threshold criterion H for hierarchical cell structures is used to determine whether prioritised ranking according to hierarchical cell re-selection rules shall apply, and is calculated from the Q, Qhcs, TEMP_OFFSET and PENALTY_TIME parameters.
 - 2.2 The UE shall perform ranking of all cells that fulfil the S criterion among all cells that have the highest HCS_PRIO among those cells that fulfil the criterion H >= 0.
 - 2.3 The cells shall be ranked according to the R criteria. The best ranked cell is the cell with the highest R value. If a FDD cell is ranked as the best cell, the UE shall perform cell re-selection to that FDD cell
 - 2.4 The UE shall reselect the new cell, if the cell reselection criteria are fulfilled during a time interval Treselection.
 - 2.5 The cell-ranking criterion R is derived from Q, Qhyst, Qoffset, TEMP_OFFSET and PENALTY_TIME.
- 3. $TEMP_OFFSET_n$ applies an offset to the H criteria for the duration of $PENALTY_TIME_n$ after the timer T_n has started for that cell. T_n shall be started from zero when $Q_{meas_LEV,n} > Qhcs_n$. $TEMP_OFFSET$ is only applied to the H criteria if the cells have different HCS priorities.

References

- 1. TS 25.304, 5.2.2
- 2,3. TS 25.304, 5.2.6.1.4

6.1.2.4.3 Test purpose

1. Verify that TEMP_OFFSET is applied to the H criterion for a period of PENALTY_TIME and that the timer is started when $Q_{meas_LEV,n} > Qhcs_n$ if serving and neighbour cell have different HCS priorities.

6.1.2.4.4 Method of test

Initial conditions

Step a-c:

Parameter	Unit	Cell 1	Cell 2	Cell 3
\hat{I}_{or}/I_{oc}	dB	11.6	7.1	7.1
CPICH_Ec/Io	dB	-12.5	-17	-17
CPICH RSCP	dBm	-68.4	-72.9	-72.9
Qqualmin	dB	-20	-20	-20
Qrxlevmin	dBm	-100	-100	-100
Squal*	dB	7.5	3	3
Srxlev*	dBm	31.6	27.1	27.1
HCS priority		2	4	7
Qhcs _s	dB	-20		
Qhcs _{n=2}	dB	-10		
Qhcs _{n=3}	dB	-10		
TEMP_OFFSET2 _{n=2}	dB	10		
TEMP_OFFSET2 _{n=3}	dB	10		
H _s *	dB	7.5		
H _{n=2} *	dB	-7		
H _{n=3} *	dB	-7		
PENALTY_TIME _{n=2}	sec	40		
PENALTY_TIME _{n=3}	sec	60		

Step d-e:

Qhcss	dB	-20	
Qhcs _{n=2}	dB	-10 -> -20	
Qhcs _{n=3}	dB	-10 -> -20	
H _s *	dB	7.5	
H _{n=2} *	dB	-7 -> 3 (after 40 sec)	
H _{n=3} *	dB	-7 -> 3 (after 60 sec)	

Test procedure

- a) The SS activates the cells 1-3 and monitors them for random access requests from the UE
- b) The UE is switched on.
- c) The SS waits for random access requests from the UE
- d) The SS changes Qhcs for Cell 2 and 3
- e) The SS waits for random access requests from the UE

6.1.2.4.5 Test requirements

- 1) In step c), the UE shall select a cell to camp on and eventually make a reselection to Cell 1
- 2) In step e), there shall be no response from the UE on Cell 2 within 38 seconds of changing the parameters but the UE shall respond on Cell 2 within 61 seconds. The response on Cell 2 shall be before any response on Cell 3.

NOTE: Minimum time of sec. set by PENALTY_TIME (cell 2) - 2 sec. tolerance. Maximum time of 75 sec. set by PENALTY_TIME (cell 2) + 1280 msec. for DRX cycle + 1280 msec. for system info scheduling + 5 sec. actual reselection time + 1280 msec. for reading neighbour BCCH + 25%.

6.1.2.5 HCS Cell reselection using reselection timing parameters for the R criterion

6.1.2.5.1 Definition

Test to verify that the UE performs the cell reselection correctly for hierarchical cell structures using TEMP_OFFSET and PENALTY_TIME applied to the R criterion.

6.1.2.5.2 Conformance requirement

- 1. When camped normally, the UE shall execute the cell reselection evaluation process on the following occasions/triggers:
 - 1.1 UE internal triggers, so as to meet performance as specified in TS 25.133
 - 1.2 When information on the BCCH used for the cell reselection evaluation procedure has been modified
- 2. Cell Reselection Criteria for hierarchical cells:
 - 2.1 The quality level threshold criterion H for hierarchical cell structures is used to determine whether prioritised ranking according to hierarchical cell re-selection rules shall apply, and is calculated from the Q, Qhcs, TEMP_OFFSET and PENALTY_TIME parameters.
 - 2.2 The UE shall perform ranking of all cells that fulfil the S criterion among all cells, not considering HCS priority levels, if no cell fulfil the criterion H>=0.
 - 2.3 The cells shall be ranked according to the R criteria. The best ranked cell is the cell with the highest R value. If a FDD cell is ranked as the best cell, the UE shall perform cell re-selection to that FDD cell
 - 2.4 The UE shall reselect the new cell, if the cell reselection criteria are fulfilled during a time interval Treselection.
 - 2.5 The cell-ranking criterion R is derived from Q, Qhyst, Qoffset, TEMP_OFFSET, PENALTY_TIME.
- 3. TEMP_OFFSET_n applies an offset to the R criteria for the duration of PENALTY_TIME_n after the timer T_n has started for that cell. T_n shall be started from zero when $Q_{meas_LEV,n} > Q_{meas_LEV,s} + Qoffset2_{s,n}$. TEMP_OFFSET is only applied to the R criteria if the cells have identical priorities.

References

- 1. TS 25.304, 5.2.2
- 2,3. TS 25.304, 5.2.6.1.4

6.1.2.5.3 Test purpose

1. Verify that TEMP_OFFSET is applied to the R criterion for a period of PENALTY_TIME and that the timer is started when $Q_{meas_LEV,n} > Q_{meas_LEV,s} + Qoffset2_{s,n}$ if serving and neighbour cell have identical HCS priorities.

6.1.2.5.4 Method of test

Initial conditions

Step a-c:

Parameter	Unit	Cell 1	Cell 2	Cell 3
\hat{I}_{or}/I_{oc}	dB	11.6	7.1	7.1
CPICH_Ec/lo	dB	-12.5	-17	-17
CPICH RSCP	dBm	-68.4	-72.9	-72.9
Qqualmin	dB	-20	-20	-20
Qrxlevmin	dBm	-100	-100	-100
Squal*	dB	7.5	3	3
Srxlev*	dBm	31.6	27.1	27.1
HCS priority		1	1	1
H _s *	dB	-12.5		
H _{n=2} *	dB	-17		
H _{n=3} *	dB	-17		
R _s *	dB	-12.5		
R _{n=2} *	dB	-17		
R _{n=3} *	dB	-17		

Step d-e:

		ı	1	
Qoffset2 _{s,n=2}	dB	0 -> -10		
Qoffset2 _{s,n=3}	dB	0 -> -10		
TEMP_OFFSET2 _{n=2}	dB	10		
TEMP_OFFSET2 _{n=3}	dB	10		
PENALTY_TIME _{n=2}	sec	40		
PENALTY_TIME _{n=3}	sec	60		
R _s *	dB	-12.5		
R _{n=2} *	dB	-17 -> -7		
Kn=2	иь	(after 40 sec)		
D .*	dB	-17 -> -7		
$R_{n=3}^*$	иь	(after 60 sec)		

Test procedure

- a) The SS activates the cells 1-3 and monitors them for random access requests from the UE
- b) The UE is switched on.
- c) The SS waits for random access requests from the UE
- d) The SS broadcasts Qoffset, TEMP_OFFSET and PENALTY_TIME for Cell 2 and 3
- e) The SS waits for random access requests from the UE

6.1.2.5.5 Test requirements

- 1) In step c), the UE shall select a cell to camp on and eventually make a reselection to Cell 1
- 2) In step e), there shall be no response from the UE on Cell 2 within 38 seconds of changing the parameters but the UE shall respond on Cell 2 within 61 seconds. The response on Cell 2 shall be before any response on Cell 3.

NOTE: Minimum time of sec. set by PENALTY_TIME (cell 2) - 2 sec. tolerance. Maximum time of 75 sec. set by PENALTY_TIME (cell 2) + 1280 msec. for DRX cycle + 1280 msec. for system info scheduling + 5 sec. actual reselection time + 1280 msec. for reading neighbour BCCH + 25%.

6.1.2.6 Emergency calls

6.1.2.6.1 Definition

Test to verify that the UE shall be able to initiate emergency calls when no suitable cells of the selected PLMN are available, but at least one acceptable cell is available.

6.1.2.6.2 Conformance requirement

1. Acceptable cell:

An "acceptable cell" is a cell on which the UE may camp to obtain limited service (originate emergency calls). Such a cell shall fulfil the following requirements, which is the minimum set of requirements to initiate an emergency call in a UTRAN network:

- 1.1 The cell is not barred
- 1.2 The cell selection criteria are fulfilled
- 2. A "suitable cell" is a cell on which the UE may camp on to obtain normal service. Such a cell shall fulfil all the following requirements.
 - 2.1 The cell is part of the selected PLMN
 - 2.2 The cell is not barred
 - 2.3 The cell is not part of a forbidden registration area
 - 2.4 The cell selection criteria are fulfilled
 - 2.5 The SoLSA criteria are fulfilled [SoLSA support is not in the current release]
- 3. If no suitable cell is found, the UE shall attempt to find an acceptable cell of any PLMN, state *Any cell selection*. This state is also entered if a non-access stratum registration procedure is rejected, or if there is no USIM in the UE. If an acceptable cell is found, the UE shall camp on this cell and obtain limited service, state *Camped on any cell*. In this state, the UE shall behave as specified for state *Camped normally*, but typically with a different PLMN. Additionally, the UE shall regularly attempt to find a suitable cell using stored information, trying all radio access technologies that are supported by the UE. If a suitable cell is found, the PLMN is reselected.

When a cell reselection is triggered, the UE shall evaluate the cell reselection criteria based on radio measurements, and if a better cell is found that cell is selected. The change of cell may imply a change of radio access technology.

References

- 1. TS 25.304, 4.3
- 2. TS 25.304, 4.3
- 3. TS 25.304, 5.2.2

6.1.2.6.3 Test purpose

- 1. To verify that the UE shall be able to initiate emergency calls when no suitable cells of the selected PLMN are available, but at least one acceptable cell is available.
- 2. To verify that the UE selects a cell with S>0 (acceptable cell) and CellBarred = 0 when no suitable cells of the selected PLMN are available.
- 3. To verify that the UE ranks the acceptable cells according to the cell-ranking criterion R which in this test case equals Q as Qhyst, Qoffset, TEMP_OFFSET and PENALTY_TIME parameters are not used. Treselection is not used either.

6.1.2.6.4 Method of test

Initial conditions

In step a-d, Cell 1 and 2 are neither suitable nor acceptable cells. Cell 3 is an acceptable cell but not suitable.

In step e-f, both Cell 1 and 3 are acceptable cells.

Step a-d:

Parameter	Unit	Cell 1	Cell 2	Cell 3
\hat{I}_{or}/I_{oc}	dB	12.1	14.6	10.6
CPICH_Ec/Io	dB	-15.5	-13.0	-17.0
CPICH_RSCP	dBm	-67.9	-65.4	-69.4
Qqualmin	dB	-20	-10	-20
Qrxlevmin	dBm	-100	-100	-100
Squal*	dB	4.5	-3	3
Srxlev*	dBm	32.1	34.6	30.6
CellBarred		1	0	0
PLMN		forbidden	forbidden	forbidden

Step e-f:

	Celibarred		1 -> 0	U	Ü	
NOTE:	All the BCCH cells belong to	the same I	PLMN, which is	s not the UE's ho	me PLMN and is	in the USIM's
	forbidden PLMN's list.					

Test procedure

- a) The SS activates the cells. The SS monitors for RA attempts from the UE on cells 1, 2 and 3 for the duration of the test.
- b) The UE is switched on.
- c) 50 seconds after switch on, an emergency call is initiated on the UE.
- d) The SS waits for random access request from the UE.
- e) The SS changes the CellBarred of Cell 1 to 0.
- f) After 30 seconds an emergency call is initiated on the UE.
- g) The SS waits for random access request from the UE.

6.1.2.6.5 Test requirements

- 1) In step d), the first access from the UE shall be on Cell 3.
- 2) In step g), the first access from the UE shall be on Cell 1.

6.2 Multi-mode environment (2G/3G case)

[Editor's note: The setting of the RF level for GSM cells are FFS. In comparing GSM and UTRAN cells, mapping needs to be specified.]

6.2.1 PLMN and RAT selection and reselection

6.2.1.1 Selection of the correct combination of PLMN and associated RAT

6.2.1.1.1 Definition

Test to verify that the UE selects the correct combination of PLMN and associated access technology according to the fields on the USIM.

6.2.1.1.2 Conformance requirement

 At switch on, the MS selects the registered PLMN (if it is available) using all access technologies that the MS is capable of and attempts to perform a Location Registration. The MS shall start its search using the access technology type stored in the RPLMN Last Used Access Technology data field on the SIM. If the RPLMN Last Used Access Technology is not available then an MS capable of GSM access technology shall start its search using GSM access technology.

On recovery from lack of coverage, the MS selects the registered PLMN (if it is available) using all access technologies that the MS is capable of and, if necessary attempts to perform a Location Registration.

If successful registration is achieved, the MS indicates the selected PLMN.

If there is no registered PLMN, or if registration is not possible due to the PLMN being unavailable or registration failure, the MS follows either Automatic or Manual Network Selection Mode Procedure depending on its operating mode.

References

1. TS 23.122, 4.4.3.1

NOTE: TS 31.102 defines the USIM fields

6.2.1.1.3 Test purpose

1. To verify that the UE selects the correct combination of PLMN and associated access technology according to the fields on the USIM.

6.2.1.1.4 Method of test

Initial conditions

The UE is in manual PLMN selection mode.

Cell	Test	PLMN	Radio Access
	Channel		Technology
Cell 1	1	PLMN 1	GSM
Cell 2	1	PLMN 1	UTRAN
Cell 3	2	PLMN 2	UTRAN
Cell 4	2	PLMN 2	GSM

The UE is equipped with a USIM containing default values except for those listed below.

USIM A

USIM field	Priority	PLMN	Access Technology Identifier
EF _{LOCI} , EF _{RPLMNACT}		PLMN 1	GSM
EF _{HPLMNwAcT}	1 st	PLMN 1	UTRAN
	2 nd		

USIM B

USIM field	Priority	PLMN	Access Technology Identifier
EFLOCI, EFRPLMNACT		PLMN 2	UTRAN
EF _{HPLMNwAcT}	1 st	PLMN 2	GSM
	2 nd]	

Test procedure

- a) The SS activates cells 1-4 and monitors the cells for random access requests from the UE. The UE shall have a USIM with settings according to USIM A.
- b) The UE is switched on
- c) The SS waits for random access requests from the UE
- d) The UE is switched off and a USIM with settings according to USIM B is inserted
- e) The UE is switched on
- f) The SS waits for random access requests from the UE

6.2.1.1.5 Test Requirements

- 1) In step c), the response from the UE shall be on Cell 1. The displayed PLMN shall be PLMN1 (GSM).
- 2) In step f), the response from the UE shall be on Cell 3. The displayed PLMN shall be PLMN2 (UTRAN).

6.2.1.2 Selection of RAT for RPLMN

6.2.1.2.1 Definition

Test to verify that the UE selects the correct access technology for the registered PLMN at switch-on if a cell with the Last Used Access Technology is not available.

6.2.1.2.2 Conformance requirement

 At switch on, the MS selects the registered PLMN (if it is available) using all access technologies that the MS is capable of and attempts to perform a Location Registration. The MS shall start its search using the access technology type stored in the RPLMN Last Used Access Technology data field on the SIM. If the RPLMN Last Used Access Technology is not available then an MS capable of GSM access technology shall start its search using GSM access technology.

On recovery from lack of coverage, the MS selects the registered PLMN (if it is available) using all access technologies that the MS is capable of and, if necessary attempts to perform a Location Registration.

If successful registration is achieved, the MS indicates the selected PLMN.

If there is no registered PLMN, or if registration is not possible due to the PLMN being unavailable or registration failure, the MS follows either Automatic or Manual Network Selection Mode Procedure depending on its operating mode.

References

1. TS 23.122, 4.4.3.1

NOTE: TS 31.102 defines the USIM fields

6.2.1.2.3 Test purpose

1. To verify that at switch-on, if a cell with the RPLMN Last Used Access Technology is not available, the UE tries to obtain registration on the same PLMN using other UE-supported RATs.

6.2.1.2.4 Method of test

Initial conditions

The UE is in manual PLMN selection mode.

Cell	Test Channel	PLMN	Radio Access Technology
Cell 1	1	PLMN 1	UTRAN
Cell 2	2	PLMN 2	UTRAN
Cell 3	1	PLMN 2	GSM

The UE is equipped with a USIM containing default values except for those listed below.

USIM field	Priority	PLMN	Access Technology Identifier
EFLOCI, EFRPLMNACT		PLMN 1	GSM
EF _{HPLMNwAcT}	1 st	PLMN 2	UTRAN
	2 nd		

Test procedure

- a) The SS activates cells 1-3 and monitors the cells for random access requests from the UE
- b) The UE is switched on
- c) The SS waits for random access requests from the UE

6.2.1.2.5 Test Requirements

1) In step c), the response from the UE shall be on Cell 1. The displayed PLMN shall be PLMN1 (UTRAN). (The preferred PLMN1 network is not available on GSM so registration is attempted using other UE-supported RATs)

6.2.1.3 Selection of RAT for HPLMN; Manual mode

6.2.1.3.1 Definition

Test to verify that the UE selects the HPLMN RAT according to the HPLMN RAT priority list on the USIM. If no RAT on the list is available, the UE shall try to obtain registration on the same PLMN using other UE-supported RATs.

6.2.1.3.2 Conformance requirement

 To allow provision for multiple HPLMN codes, the HPLMN access technologies are stored on the SIM using the same format as the User Controlled PLMN Selector with Access Technology and Operator Controlled PLMN Selector with Access Technology data fields. It is assumed in this version of the specification that this HPLMN Selector with Access Technology data field should contain only one PLMN code identical to the HPLMN code included in the IMSI. Although this single code may be duplicated in the list if multiple access technologies with priority is defined.

2. Manual Network Selection Mode Procedure:

The MS indicates whether there are any PLMNs, which are available using all supported access technologies. This includes "Forbidden PLMNs" and PLMNs which only offer services not supported by the MS.

If displayed, PLMNs meeting the criteria above are presented in the following order:

- 2.1 HPLMN;
- 2.2 PLMNs contained in the "User Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order);
- 2.3 PLMNs contained in the "Operator Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order);
- 2.4 Other PLMN/access technology combinations with received high quality signal in random order;
- 2.5 Other PLMN/access technology combinations in order of decreasing signal quality.

The user may select his desired PLMN and the MS then initiates registration on this PLMN using the access technology chosen by the user for that PLMN or using the highest priority available access technology for that PLMN, if the associated access technologies have a priority order. (This may take place at any time during the presentation of PLMNs). For such a registration, the MS shall ignore the contents of the forbidden LAI and PLMN lists.

If the user does not select a PLMN, the selected PLMN shall be the one that was selected before the PLMN selection procedure started. If no such PLMN was selected or that PLMN is no longer available, then the MS shall attempt to camp on any acceptable cell and enter the limited service state.

NOTE: It is an MS implementation option whether to indicate access technologies to the user. If the MS does display access technologies, then the access technology used should be the access technology chosen by the user for that PLMN. If the MS does not display access technologies, then the access technology chosen for a particular PLMN should be the highest priority available access technology for that PLMN, if the associated access technologies have a priority order.

References

- 1. TS 23.122, 4.4.3 and 4.4.3.1.1 (f)
- 2. TS 23.122, 4.4.3.1.2

NOTE: TS 31.102 defines the USIM fields

6.2.1.3.3 Test purpose

- 1. To verify that,
 - 1.1 the UE searches for a HPLMN RAT according to the HPLMN Selector with Access Technology data field on the USIM in priority order
 - 1.2 If no RAT on the priority list is available, the UE tries to obtain registration on the same PLMN using other UE-supported RATs.

6.2.1.3.4 Method of test

Initial conditions

The UE is in manual PLMN selection mode.

Cell	Test Channel	PLMN	Radio Access Technology
Cell 1	1	PLMN 2	UTRAN
Cell 2	1	PLMN 2	GSM
Cell 3	2	PLMN 3	UTRAN
Cell 4	2	PLMN 3	GSM

The UE is equipped with a USIM containing default values except for those listed below.

USIM A

USIM field	Priority	PLMN	Access Technology Identifier
EFLOCI, EFRPLMNACT		PLMN 1	GSM
EF _{HPLMNwAcT}	1 st	PLMN 2	UTRAN
	2 nd		GSM

USIM B

USIM field	Priority	PLMN	Access Technology Identifier
EFLOCI, EFRPLMNACT		PLMN 1	GSM
EF _{HPLMNwAcT}	1 st	PLMN 2	UTRAN
	2 nd		

Test procedure

- a) The SS activates cells 1-4 and monitors the cells for random access requests from the UE. The UE shall have a USIM with settings according to USIM A.
- b) The UE is switched on
- c) PLMN2 (UTRAN) shall be selected when the PLMN list is presented
- d) The SS waits for random access requests from the UE
- e) Cell 1 is switched off
- f) PLMN2 (GSM) shall be selected when the PLMN list is presented
- g) The SS waits for random access requests from the UE
- h) The UE is switched off and a USIM with settings according to USIM B is inserted. All cells except Cell 1 are active.
- i) The UE is switched on
- j) PLMN2 (GSM) shall be selected when the PLMN list is presented
- k) The SS waits for random access requests from the UE

6.2.1.3.5 Test Requirements

- 1) In step c), the list shall be presented. It shall contain PLMN2 (UTRAN as number 1 on the list and GSM as number 2).
- 2) In step d), the response from the UE shall be on Cell 1 (1st priority RAT for EF_{HPLMNwAcT}). The displayed PLMN shall be PLMN2 (UTRAN).
- 3) In step f), the list shall be presented. It shall contain PLMN2 (GSM) as number 1.
- 4) In step g), the response from the UE shall be on Cell 2 (2nd priority RAT for EF_{HPLMNwAcT}). The displayed PLMN shall be PLMN2 (GSM).
- 5) In step j), the list shall be presented. It shall contain PLMN2 (GSM) as number 1.

6) In step k), the response from the UE shall be on Cell 2. The displayed PLMN shall be PLMN2 (GSM). (PLMN2 is not available on UTRAN so registration on the same PLMN is attempted using other UE-supported RATs).

6.2.1.4 Selection of RAT for UPLMN; Manual mode

6.2.1.4.1 Definition

Test to verify that the UE selects the UPLMN RAT according to the UPLMN RAT priority list on the USIM. If no RAT on the list is available, the UE shall not try to obtain registration on the same PLMN(s) with other RAT(s) but instead search for PLMNs in the OPLMN list.

6.2.1.4.2 Conformance requirement

1. Manual Network Selection Mode Procedure:

The MS indicates whether there are any PLMNs, which are available using all supported access technologies. This includes "Forbidden PLMNs" and PLMNs which only offer services not supported by the MS.

If displayed, PLMNs meeting the criteria above are presented in the following order:

- 1.1 HPLMN;
- 1.2 PLMNs contained in the "User Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order);
- 1.3 PLMNs contained in the "Operator Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order);
- 1.4 Other PLMN/access technology combinations with received high quality signal in random order;
- 1.5 Other PLMN/access technology combinations in order of decreasing signal quality.

The user may select his desired PLMN and the MS then initiates registration on this PLMN using the access technology chosen by the user for that PLMN or using the highest priority available access technology for that PLMN, if the associated access technologies have a priority order. (This may take place at any time during the presentation of PLMNs). For such a registration, the MS shall ignore the contents of the forbidden LAI and PLMN lists.

If the user does not select a PLMN, the selected PLMN shall be the one that was selected before the PLMN selection procedure started. If no such PLMN was selected or that PLMN is no longer available, then the MS shall attempt to camp on any acceptable cell and enter the limited service state.

NOTE: It is an MS implementation option whether to indicate access technologies to the user. If the MS does display access technologies, then the access technology used should be the access technology chosen by the user for that PLMN. If the MS does not display access technologies, then the access technology chosen for a particular PLMN should be the highest priority available access technology for that PLMN, if the associated access technologies have a priority order.

References

1. TS 23.122, 4.4.3.1.2

NOTE: TS 31.102 defines the USIM fields

6.2.1.4.3 Test purpose

- 1. To verify that,
 - 1.1 the UE selects the UPLMN RAT according to the UPLMN RAT priority list on the USIM.

1.2 If no RAT on the list is available, the UE does not try to obtain registration on the same PLMN with another RAT but instead searches for PLMNs in the OPLMN list.

6.2.1.4.4 Method of test

Initial conditions

The UE is in manual PLMN selection mode.

Cell	Test Channel	PLMN	Radio Access Technology
Cell 1	1	PLMN 3	UTRAN
Cell 2	1	PLMN 3	GSM
Cell 3	2	PLMN 4	UTRAN
Cell 4	2	PLMN 4	GSM
Cell 5	3	PLMN 5	UTRAN

The UE is equipped with a USIM containing default values except for those listed below.

USIM field	Priority	PLMN	Access Technology Identifier
EFLOCI, EFRPLMNACT		PLMN 1	GSM
EF _{HPLMNwAcT}	1 st	PLMN 2	UTRAN
	2 nd		GSM
EF _{PLMNwACT}	1 st	PLMN 3	UTRAN
	2 nd	PLMN 4	GSM
EFOPLMNWACT	1 st	PLMN 5	UTRAN
	2 nd	PLMN 6	GSM

Test procedure

- a) The SS activates cells 1-5 and monitors the cells for random access requests from the UE
- b) The UE is switched on
- c) PLMN3 (UTRAN) shall be selected when the PLMN list is presented
- d) The SS waits for random access requests from the UE
- e) Cell 1 is switched off
- f) PLMN4 (GSM) shall be selected when the PLMN list is presented
- g) The SS waits for random access requests from the UE
- h) Cell 4 is switched off
- i) PLMN5 (UTRAN) shall be selected when the PLMN list is presented
- j) The SS waits for random access requests from the UE

6.2.1.4.5 Test Requirements

- 1) In step c), the list shall be presented. It shall contain PLMN3 (UTRAN) as number 1 and PLMN4 (GSM) as number 2.
- 2) In step d), the response from the UE shall be on Cell 1 (1st priority RAT for EF_{PLMNwACT}). The displayed PLMN shall be PLMN3 (UTRAN).
- 3) In step f), the list shall be presented. It shall contain PLMN4 (GSM) as number 1 and PLMN5 (UTRAN) as number 2.
- 4) In step g), the response from the UE shall be on Cell 4 (2nd priority RAT for EF_{PLMNwACT}). The displayed PLMN shall be PLMN4 (GSM).

- 5) In step i), the list shall be presented. It shall contain PLMN5 (UTRAN) as number 1.
- 6) In step j), the response from the UE shall be on Cell 5 (1st priority RAT for EF_{OPLMNwACT}). The displayed PLMN shall be PLMN5 (UTRAN).

6.2.1.5 Selection of RAT for OPLMN; Manual mode

6.2.1.5.1 Definition

Test to verify that the UE selects the OPLMN RAT according to the OPLMN RAT priority list on the USIM. If no RAT on the list is available, the UE shall not try to obtain registration on the same PLMN(s) with other RAT(s) but instead search for other PLMN/access technology combinations with received high quality signal in random order.

6.2.1.5.2 Conformance requirement

1. Manual Network Selection Mode Procedure:

The MS indicates whether there are any PLMNs, which are available using all supported access technologies. This includes "Forbidden PLMNs" and PLMNs which only offer services not supported by the MS.

If displayed, PLMNs meeting the criteria above are presented in the following order:

- 1.1 HPLMN:
- 1.2 PLMNs contained in the "User Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order);
- 1.3 PLMNs contained in the "Operator Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order);
- 1.4 Other PLMN/access technology combinations with received high quality signal in random order;
- 1.5 Other PLMN/access technology combinations in order of decreasing signal quality.

The user may select his desired PLMN and the MS then initiates registration on this PLMN using the access technology chosen by the user for that PLMN or using the highest priority available access technology for that PLMN, if the associated access technologies have a priority order. (This may take place at any time during the presentation of PLMNs). For such a registration, the MS shall ignore the contents of the forbidden LAI and PLMN lists.

If the user does not select a PLMN, the selected PLMN shall be the one that was selected before the PLMN selection procedure started. If no such PLMN was selected or that PLMN is no longer available, then the MS shall attempt to camp on any acceptable cell and enter the limited service state.

NOTE: It is an MS implementation option whether to indicate access technologies to the user. If the MS does display access technologies, then the access technology used should be the access technology chosen by the user for that PLMN. If the MS does not display access technologies, then the access technology chosen for a particular PLMN should be the highest priority available access technology for that PLMN, if the associated access technologies have a priority order.

References

1. TS 23.122, 4.4.3.1.2

NOTE: TS 31.102 defines the USIM fields

6.2.1.5.3 Test purpose

- 1. To verify that,
 - 1.1 the UE selects the OPLMN RAT according to the OPLMN RAT priority list on the USIM.

1.2 If no RAT on the list is available, the UE does not try to obtain registration on the same PLMN(s) with other RAT(s) but instead searches for "other PLMN/access technology combinations with received high quality signal in random order".

6.2.1.5.4 Method of test

Initial conditions

The UE is in manual PLMN selection mode.

Cell	"High Quality signal"	Test Channel	PLMN	Radio Access Technology
Cell 1		1	PLMN 5	UTRAN
Cell 2	No	1	PLMN 5	GSM
Cell 3	No	2	PLMN 6	UTRAN
Cell 4		2	PLMN 6	GSM
Cell 5	Yes	3	PLMN 7	UTRAN

NOTE: "High Quality signal" has not yet been defined in 3GPP.

The UE is equipped with a USIM containing default values except for those listed below.

USIM field	Priority	PLMN	Access Technology Identifier
EFLOCI, EFRPLMNACT		PLMN 1	GSM
EF _{HPLMNwAcT}	1 st	PLMN 2	UTRAN
	2 nd		GSM
EF _{PLMNWACT}	1 st	PLMN 3	UTRAN
	2 nd	PLMN 4	GSM
EFOPLMNWACT	1 st	PLMN 5	UTRAN
	2 nd	PLMN 6	GSM

Test procedure

- a) The SS activates cells 1-5 and monitors the cells for random access requests from the UE
- b) The UE is switched on
- c) PLMN5 (UTRAN) shall be selected when the PLMN list is presented
- d) The SS waits for random access requests from the UE
- e) Cell 1 is switched off
- f) PLMN6 (GSM) shall be selected when the PLMN list is presented
- g) The SS waits for random access requests from the UE
- h) Cell 4 is switched off
- i) PLMN7 (UTRAN) shall be selected when the PLMN list is presented
- j) The SS waits for random access requests from the UE

6.2.1.5.5 Test Requirements

- 1) In step c), the list shall be presented. It shall contain PLMN5 (UTRAN) as number 1 and PLMN6 (GSM) as number 2.
- 2) In step d), the response from the UE shall be on Cell 1 (1st priority RAT for EF_{OPLMNwACT}). The displayed PLMN shall be PLMN5 (UTRAN).
- 3) In step f), the list shall be presented. It shall contain PLMN6 (GSM) as number 1 and PLMN7 (UTRAN) as number 2.

- 4) In step g), the response from the UE shall be on Cell 4 (2nd priority RAT for EF_{OPLMNwACT}). The displayed PLMN shall be PLMN6 (GSM).
- 5) In step i), the list shall be presented. It shall contain PLMN7 (UTRAN) as number 1.
- 6) In step j), the response from the UE shall be on Cell 5 (other PLMN/access technology combination). The displayed PLMN shall be PLMN7 (UTRAN).

6.2.1.6 Selection of "Other PLMN / access technology combinations"; Manual mode

6.2.1.6.1 Definition

Test to verify that if neither RPLMN, HPLMN, UPLMN nor OPLMN is available, the UE first tries to obtain registration on "Other PLMN/access technology combinations with received high quality signal in random order" and secondly on "Other PLMN/access technology combinations in order of decreasing signal quality".

6.2.1.6.2 Conformance requirement

1. Manual Network Selection Mode Procedure:

The MS indicates whether there are any PLMNs, which are available using all supported access technologies. This includes "Forbidden PLMNs" and PLMNs which only offer services not supported by the MS.

If displayed, PLMNs meeting the criteria above are presented in the following order:

- 1.1 HPLMN:
- 1.2 PLMNs contained in the "User Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order);
- 1.3 PLMNs contained in the "Operator Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order);
- 1.4 Other PLMN/access technology combinations with received high quality signal in random order;
- 1.5 Other PLMN/access technology combinations in order of decreasing signal quality.

The user may select his desired PLMN and the MS then initiates registration on this PLMN using the access technology chosen by the user for that PLMN or using the highest priority available access technology for that PLMN, if the associated access technologies have a priority order. (This may take place at any time during the presentation of PLMNs). For such a registration, the MS shall ignore the contents of the forbidden LAI and PLMN lists.

If the user does not select a PLMN, the selected PLMN shall be the one that was selected before the PLMN selection procedure started. If no such PLMN was selected or that PLMN is no longer available, then the MS shall attempt to camp on any acceptable cell and enter the limited service state.

NOTE: It is an MS implementation option whether to indicate access technologies to the user. If the MS does display access technologies, then the access technology used should be the access technology chosen by the user for that PLMN. If the MS does not display access technologies, then the access technology chosen for a particular PLMN should be the highest priority available access technology for that PLMN, if the associated access technologies have a priority order.

References

1. TS 23.122, 4.4.3.1.2

NOTE: TS 31.102 defines the USIM fields

6.2.1.6.3 Test purpose

- 1. To verify that,
 - 1.1 If neither RPLMN, HPLMN, UPLMN nor OPLMN is available, the UE tries to obtain registration on "Other PLMN/access technology combinations with received high quality signal in random order"
 - 1.2 If no PLMN is available in test purpose 1.1, the UE tries to obtain registration on "Other PLMN/access technology combinations in order of decreasing signal quality"
- 2. The "random order" in test purpose 1.1 is not verified

6.2.1.6.4 Method of test

Initial conditions

The UE is in manual PLMN selection mode.

Cell	Signal level [dBm]	"High Quality signal"	Test Channel	PLMN	Radio Access Technology
Cell 1	-80	Yes	1	PLMN 7	UTRAN
Cell 2	-80	Yes	1	PLMN 8	GSM
Cell 3	-85	No	2	PLMN 9	UTRAN
Cell 4	-90	No	2	PLMN 10	GSM
Cell 5	-95	No	3	PLMN 11	UTRAN
Cell 6	-100	No	3	PLMN 12	GSM

NOTE: "High Quality signal" has not yet been defined in 3GPP.

The UE is equipped with a USIM containing default values except for those listed below.

USIM field	Priority	PLMN	Access Technology Identifier	
EFLOCI, EFRPLMNACT		PLMN 1	GSM	
EF _{HPLMNwAcT}	1 st	PLMN 2	UTRAN	
	2 nd		GSM	
EF _{PLMNwACT}	1 st	PLMN 3	UTRAN	
	2 nd	PLMN 4	GSM	
EFOPLMNWACT	1 st	PLMN 5	UTRAN	
	2 nd	PLMN 6	GSM	
EF _{FPLMN}	PLMN 7			
	PLMN 12			

NOTE: PLMN 7 and PLMN 12 are forbidden.

Test procedure

- a) The SS activates cells 1-6 and monitors the cells for random access requests from the UE
- b) The UE is switched on
- c) PLMN11 shall be selected when the PLMN list is presented
- d) The SS waits for random access requests from the UE
- e) Cell 5 is switched off
- f) PLMN8 shall be selected when the PLMN list is presented
- g) The SS waits for random access requests from the UE
- h) Cell 2 is switched off
- i) PLMN10 shall be selected when the PLMN list is presented

- j) The SS waits for random access requests from the UE
- k) Cell 4 is switched off
- PLMN7 shall be selected when the PLMN list is presented. The SS shall reject the Registration Request from the UE.
- m) Cell 1 is switched off
- n) PLMN9 shall be selected when the PLMN list is presented
- o) The SS waits for random access requests from the UE
- p) Cell 3 is switched off
- q) PLMN12 shall be selected when the PLMN list is presented. The SS shall reject the Registration Request from the UE.
- r) Cell 6 is switched off

6.2.1.6.5 Test Requirements

- 1) In step c), the list shall be presented. The priority shall be as follows: PLMN7, PLMN8 in random order, followed by PLMN9, PLMN10, PLMN11, PLMN12.
- 2) In step d), the response from the UE shall be on Cell 5. The displayed PLMN shall be PLMN11.
- 3) In step f), the list shall be presented. The priority shall be as follows: PLMN7, PLMN8 in random order, followed by PLMN9, PLMN10, PLMN12.
- 4) In step g), the response from the UE shall be on Cell 2. The displayed PLMN shall be PLMN8.
- 5) In step i), the list shall be presented. The priority shall be as follows: PLMN7, PLMN9, PLMN10, PLMN12.
- 6) In step j), the response from the UE shall be on Cell 4. The displayed PLMN shall be PLMN10.
- 7) In step 1), the list shall be presented. The priority shall be as follows: PLMN7, PLMN9, PLMN12. After the PLMN has been selected, the list shall appear again as the UE cannot perform registration.
- 9) In step n), the list shall be presented. The priority shall be as follows: PLMN9, PLMN12.
- 10) In step o), the response from the UE shall be on Cell 3. The displayed PLMN shall be PLMN9.
- 11) In step q), the list shall be presented and shall only contain PLMN12. After the PLMN has been selected, the list shall appear again as the UE cannot perform registration.
- 13) After step r), the UE shall inform that no network is available

6.2.1.7 Selection of RAT for HPLMN; Automatic mode

6.2.1.7.1 Definition

Test to verify that the UE selects the HPLMN RAT according to the HPLMN RAT priority list on the USIM. If no RAT on the list is available, the UE shall try to obtain registration on the same PLMN using other UE-supported RATs.

6.2.1.7.2 Conformance requirement

 To allow provision for multiple HPLMN codes, the HPLMN access technologies are stored on the SIM using the same format as the User Controlled PLMN Selector with Access Technology and Operator Controlled PLMN Selector with Access Technology data fields. It is assumed in this version of the specification that this HPLMN Selector with Access Technology data field should contain only one PLMN code identical to the HPLMN code

included in the IMSI. Although this single code may be duplicated in the list if multiple access technologies with priority is defined.

NOTE: In selecting a HPLMN, the MS shall search for all access technologies it is capable of. The MS shall start its search using the access technologies stored in the HPLMN Selector with Access Technology data field on the SIM in priority order.

2. Automatic Network Selection Mode Procedure:

The MS selects and attempts registration on other PLMNs, if available and allowable in the following order:

- 2.1 HPLMN (if not previously selected);
- 2.2 Each PLMN in the "User Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order)
- 2.3 Each PLMN in the "Operator Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order)
- 2.4 Other PLMN/access technology combinations with received high quality signal in random order
- 2.5 Other PLMN/access technology combinations in order of decreasing signal quality

If successful registration is achieved, the MS indicates the selected PLMN.

If registration cannot be achieved because no PLMNs are available and allowable, the MS indicates "no service" to the user, waits until a new PLMN is available and allowable and then repeats the procedure.

If there were one or more PLMNs which were available and allowable, but an LR failure made registration on those PLMNs unsuccessful or an entry in a forbidden LAI list prevented a registration attempt, the MS selects the first such PLMN again and enters a limited service state.

References

- 1. TS 23.122, 4.4.3 and 4.4.3.1.1 (f)
- 2. TS 23.122, 4.4.3.1.1

NOTE: TS 31.102 defines the USIM fields

6.2.1.7.3 Test purpose

- 1. To verify that,
 - 1.1 the UE searches for a HPLMN RAT according to the HPLMN Selector with Access Technology data field on the USIM in priority order
 - 1.2 If no RAT on the priority list is available, the UE tries to obtain registration on the same PLMN using other UE-supported RATs.

6.2.1.7.4 Method of test

Initial conditions

The UE is in automatic mode PLMN selection.

MCC of HPLMN shall be different from other PLMNs in the network to avoid periodic search for HPLMN.

Cell	Test Channel	PLMN	Radio Access Technology
Cell 1	1	PLMN 2	UTRAN
Cell 2	1	PLMN 2	GSM
Cell 3	2	PLMN 3	UTRAN
Cell 4	2	PLMN 3	GSM

The UE is equipped with a USIM containing default values except for those listed below.

USIM A

USIM field	Priority	PLMN	Access Technology Identifier
EFLOCI, EFRPLMNACT		PLMN 1	GSM
EF _{HPLMNwAcT}	1 st	PLMN 2	UTRAN
	2 nd		GSM

USIM B

USIM field	Priority	PLMN	Access Technology Identifier
EFLOCI, EFRPLMNACT		PLMN 1	GSM
EF _{HPLMNwAcT}	1 st	PLMN 2	UTRAN
	2 nd		

Test procedure

- a) The SS activates cells 1-4 and monitors the cells for random access requests from the UE. The UE shall have a USIM with settings according to USIM A.
- b) The UE is switched on
- c) The SS waits for random access requests from the UE
- d) The UE is switched off and a USIM with settings according to USIM A is again inserted. All cells except Cell 1 are active.
- e) The SS waits for random access requests from the UE
- f) The UE is switched off and a USIM with settings according to USIM B is inserted. All cells except Cell 1 are active.
- g) The UE is switched on
- h) The SS waits for random access requests from the UE

6.2.1.7.5 Test Requirements

- 1) In step c), the response from the UE shall be on Cell 1 (1st priority RAT for EF_{HPLMNwAcT}). The displayed PLMN shall be PLMN2 (UTRAN).
- 2) In step e), the response from the UE shall be on Cell 2 (2nd priority RAT for EF_{HPLMNwAcT}). The displayed PLMN shall be PLMN2 (GSM).
- 3) In step h), the response from the UE shall be on Cell 2. The displayed PLMN shall be PLMN2 (GSM). (PLMN2 is not available on UTRAN so registration on the same PLMN is attempted using other UE-supported RATs).

6.2.1.8 Selection of RAT for UPLMN; Automatic mode

6.2.1.8.1 Definition

Test to verify that the UE selects the UPLMN RAT according to the UPLMN RAT priority list on the USIM. If no RAT on the list is available, the UE shall not try to obtain registration on the same PLMN(s) with other RAT(s) but instead search for PLMNs in the OPLMN list.

6.2.1.8.2 Conformance requirement

1. Automatic Network Selection Mode Procedure:

The MS selects and attempts registration on other PLMNs, if available and allowable in the following order:

- 1.1 HPLMN (if not previously selected);
- 1.2 Each PLMN in the "User Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order)
- 1.3 Each PLMN in the "Operator Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order)
- 1.4 Other PLMN/access technology combinations with received high quality signal in random order
- 1.5 Other PLMN/access technology combinations in order of decreasing signal quality

If successful registration is achieved, the MS indicates the selected PLMN.

If registration cannot be achieved because no PLMNs are available and allowable, the MS indicates "no service" to the user, waits until a new PLMN is available and allowable and then repeats the procedure.

If there were one or more PLMNs which were available and allowable, but an LR failure made registration on those PLMNs unsuccessful or an entry in a forbidden LAI list prevented a registration attempt, the MS selects the first such PLMN again and enters a limited service state.

References

1. TS 23.122, 4.4.3.1.1

NOTE: TS 31.102 defines the USIM fields

6.2.1.8.3 Test purpose

- 1. To verify that,
 - 1.1 the UE selects the UPLMN RAT according to the UPLMN RAT priority list on the USIM.
 - 1.2 If no RAT on the list is available, the UE does not try to obtain registration on the same PLMN with another RAT but instead searches for PLMNs in the OPLMN list.

6.2.1.8.4 Method of test

Initial conditions

The UE is in automatic mode PLMN selection.

MCC of HPLMN shall be different from other PLMNs in the network to avoid periodic search for HPLMN.

Cell	Test Channel	PLMN	Radio Access Technology
Cell 1	1	PLMN 3	UTRAN
Cell 2	1	PLMN 3	GSM
Cell 3	2	PLMN 4	UTRAN
Cell 4	2	PLMN 4	GSM
Cell 5	3	PLMN 5	UTRAN

The UE is equipped with a USIM containing default values except for those listed below.

USIM field	Priority	PLMN	Access Technology Identifier
EFLOCI, EFRPLMNACT		PLMN 1	GSM
EF _{HPLMNwAcT}	1 st	PLMN 2	UTRAN
	2 nd		GSM
EF _{PLMNwACT}	1 st	PLMN 3	UTRAN
	2 nd	PLMN 4	GSM
EFOPLMNWACT	1 st	PLMN 5	UTRAN
	2 nd	PLMN 6	GSM

Test procedure

- a) The SS activates cells 1-5 and monitors the cells for random access requests from the UE
- b) The UE is switched on
- c) The SS waits for random access requests from the UE
- d) Cell 1 is switched off
- e) The SS waits for random access requests from the UE
- f) Cell 4 is switched off
- g) The SS waits for random access requests from the UE

6.2.1.8.5 Test Requirements

- 1) In step c), the response from the UE shall be on Cell 1 (1st priority RAT for EF_{PLMNwACT}). The displayed PLMN shall be PLMN3 (UTRAN).
- 2) In step e), the response from the UE shall be on Cell 4 (2nd priority RAT for EF_{PLMNwACT}). The displayed PLMN shall be PLMN4 (GSM).
- 3) In step g), the response from the UE shall be on Cell 5 (1st priority RAT for EF_{OPLMNwACT}). The displayed PLMN shall be PLMN5 (UTRAN).

6.2.1.9 Selection of RAT for OPLMN; Automatic mode

6.2.1.9.1 Definition

Test to verify that the UE selects the OPLMN RAT according to the OPLMN RAT priority list on the USIM. If no RAT on the list is available, the UE shall not try to obtain registration on the same PLMN(s) with other RAT(s) but instead search for other PLMN/access technology combinations with received high quality signal in random order.

6.2.1.9.2 Conformance requirement

1. Automatic Network Selection Mode Procedure:

The MS selects and attempts registration on other PLMNs, if available and allowable in the following order:

- 1.1 HPLMN (if not previously selected);
- 1.2 Each PLMN in the "User Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order)
- 1.3 Each PLMN in the "Operator Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order)
- 1.4 Other PLMN/access technology combinations with received high quality signal in random order
- 1.5 Other PLMN/access technology combinations in order of decreasing signal quality

If successful registration is achieved, the MS indicates the selected PLMN.

If registration cannot be achieved because no PLMNs are available and allowable, the MS indicates "no service" to the user, waits until a new PLMN is available and allowable and then repeats the procedure.

If there were one or more PLMNs which were available and allowable, but an LR failure made registration on those PLMNs unsuccessful or an entry in a forbidden LAI list prevented a registration attempt, the MS selects the first such PLMN again and enters a limited service state.

References

1. TS 23.122, 4.4.3.1.1

NOTE: TS 31.102 defines the USIM fields

6.2.1.9.3 Test purpose

- 1. To verify that,
 - 1.1 the UE selects the OPLMN RAT according to the OPLMN RAT priority list on the USIM.
 - 1.2 If no RAT on the list is available, the UE does not try to obtain registration on the same PLMN(s) with other RAT(s) but instead searches for "other PLMN/access technology combinations with received high quality signal in random order".

6.2.1.9.4 Method of test

Initial conditions

The UE is in automatic mode PLMN selection.

MCC of HPLMN shall be different from other PLMNs in the network to avoid periodic search for HPLMN.

Cell	"High Quality	Test Channel	PLMN	Radio Access
	signal"			Technology
Cell 1		1	PLMN 5	UTRAN
Cell 2	No	1	PLMN 5	GSM
Cell 3	No	2	PLMN 6	UTRAN
Cell 4		2	PLMN 6	GSM
Cell 5	Yes	3	PLMN 7	UTRAN

NOTE: "High Quality signal" has not yet been defined in 3GPP.

The UE is equipped with a USIM containing default values except for those listed below.

USIM field	Priority	PLMN	Access Technology Identifier
EFLOCI, EFRPLMNACT		PLMN 1	GSM
EF _{HPLMNwAcT}	1 st	PLMN 2	UTRAN
	2 nd		GSM
EF _{PLMNwACT}	1 st	PLMN 3	UTRAN
	2 nd	PLMN 4	GSM
EFOPLMNWACT	1 st	PLMN 5	UTRAN
	2 nd	PLMN 6	GSM

Test procedure

- a) The SS activates cells 1-5 and monitors the cells for random access requests from the UE
- b) The UE is switched on
- c) The SS waits for random access requests from the UE
- d) Cell 1 is switched off
- e) The SS waits for random access requests from the UE
- f) Cell 4 is switched off
- g) The SS waits for random access requests from the UE

6.2.1.9.5 Test Requirements

- 1) In step c), the response from the UE shall be on Cell 1 (1st priority RAT for EF_{OPLMNwACT}). The displayed PLMN shall be PLMN5 (UTRAN).
- 2) In step e), the response from the UE shall be on Cell 4 (2nd priority RAT for EF_{OPLMNwACT}). The displayed PLMN shall be PLMN6 (GSM).
- 3) In step g), the response from the UE shall be on Cell 5 (other PLMN/access technology combination). The displayed PLMN shall be PLMN7 (UTRAN).

6.2.1.10 Selection of "Other PLMN / access technology combinations"; Automatic mode

6.2.1.10.1 Definition

Test to verify that if neither RPLMN, HPLMN, UPLMN nor OPLMN is available, the UE first tries to obtain registration on "Other PLMN/access technology combinations with received high quality signal in random order" and secondly on "Other PLMN/access technology combinations in order of decreasing signal quality".

6.2.1.10.2 Conformance requirement

1. Automatic Network Selection Mode Procedure:

The MS selects and attempts registration on other PLMNs, if available and allowable in the following order:

- 1.1 HPLMN (if not previously selected);
- 1.2 Each PLMN in the "User Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order)
- 1.3 Each PLMN in the "Operator Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order)
- 1.4 Other PLMN/access technology combinations with received high quality signal in random order

1.5 Other PLMN/access technology combinations in order of decreasing signal quality

If successful registration is achieved, the MS indicates the selected PLMN.

If registration cannot be achieved because no PLMNs are available and allowable, the MS indicates "no service" to the user, waits until a new PLMN is available and allowable and then repeats the procedure.

If there were one or more PLMNs which were available and allowable, but an LR failure made registration on those PLMNs unsuccessful or an entry in a forbidden LAI list prevented a registration attempt, the MS selects the first such PLMN again and enters a limited service state.

References

1. TS 23.122, 4.4.3.1.1

NOTE: TS 31.102 defines the USIM fields

6.2.1.10.3 Test purpose

- 1. To verify that,
 - 1.1 If neither RPLMN, HPLMN, UPLMN nor OPLMN is available, the UE tries to obtain registration on "Other PLMN/access technology combinations with received high quality signal in random order"
 - 1.2 If no PLMN is available in test purpose 1.1, the UE tries to obtain registration on "Other PLMN/access technology combinations in order of decreasing signal quality"
- 2. The "random order" in test purpose 1.1 is not verified

6.2.1.10.4 Method of test

Initial conditions

The UE is in automatic PLMN selection mode.

Cell	Signal level [dBm]	"High Quality signal"	Test Channel	PLMN	Radio Access Technology
Cell 1	-80	Yes	1	PLMN 7	UTRAN
Cell 2	-80	Yes	1	PLMN 8	GSM
Cell 3	-85	No	2	PLMN 9	UTRAN
Cell 4	-90	No	2	PLMN 10	GSM
Cell 5	-95	No	3	PLMN 11	UTRAN
Cell 6	-100	No	3	PLMN 12	GSM

NOTE: "High Quality signal" has not yet been defined in 3GPP.

The UE is equipped with a USIM containing default values except for those listed below.

USIM field	Priority	PLMN	Access Technology Identifier
EF _{LOCI} , EF _{RPLMNACT}		PLMN 1	GSM
EF _{HPLMNwAcT}	1 st	PLMN 2	UTRAN
	2 nd		GSM
EF _{PLMNwACT}	1 st	PLMN 3	UTRAN
	2 nd	PLMN 4	GSM
EF _{OPLMNwACT}	1 st	PLMN 5	UTRAN
	2 nd	PLMN 6	GSM
EF _{FPLMN}		PLMN 7	
		PLMN 12	

Test procedure

a) The SS activates cells 1-6 and monitors the cells for random access requests from the UE

- b) The UE is switched on
- c) The SS waits for random access requests from the UE
- d) Cell 2 is switched off
- e) The SS waits for random access requests from the UE
- f) Cell 3 is switched off
- g) The SS waits for random access requests from the UE
- h) Cell 4 is switched off
- i) The SS waits for random access requests from the UE
- i) Cell 5 is switched off

6.2.1.10.5 Test Requirements

- 1) In step c), the response from the UE shall be on Cell 2. The displayed PLMN8.
- 2) In step e), the response from the UE shall be on Cell 3. The displayed PLMN shall be PLMN9.
- 3) In step g), the response from the UE shall be on Cell 4. The displayed PLMN shall be PLMN10.
- 4) In step i), the response from the UE shall be on Cell 5. The displayed PLMN shall be PLMN11.
- 5) After step j), the UE shall inform that only limited service is possible.

6.2.2 Cell selection and reselection

6.2.2.1 Cell selection; UTRAN / GSM

6.2.2.1.1 Definition

Test to verify that the UE performs cell selection correctly when both a GSM and UTRA network is available.

6.2.2.1.2 Conformance requirement

- 1. Cell selection procedure to find a suitable cell to camp on:
 - 1.1 Create a candidate list of potential cells to camp on, using:
 - 1.1.1 Initial Cell Selection procedure; or
 - 1.1.2 Stored Information Cell Selection procedure
 - 1.2 For each cell on the candidate list, measure the quality value, $Q_{\text{meas,LEV}}$
 - 1.3 For each cell on the candidate list calculate the cell selection value, Squal and Srxlev
 - 1.4 Rank the cells and select the best cell
 - 1.5 Select the cell that fulfils the criteria $Q_{map,n} > Q_{map,s} + Q_{offsets,n}$ best. Check if the selected cell fulfils all requirements for a suitable cell. If so, choose this cell to camp on. If this cell does not fulfil all requirements for a suitable cell, this cell and all cells on the same frequency shall be removed as candidates for cell selection in case the barred cell does not accept intra-frequency cell selection and re-selection. On the other hand, in case the barred cell accepts intra-frequency cell selection and re-selection, only the barred cell shall be removed as candidate for cell selection, and step 1.4 shall be repeated for the remaining cells.

- 2. Different types of measurements are used in different radio access technologies and modes for the cell selection and reselection (CPICH Ec/N0 or CPICH SIR in UTRA FDD, P-CCPCH RSCP in UTRA TDD, RXLEV in GSM). Whenever a direct comparison of these measurements is required, mapping functions shall be applied. Mapping functions are used for mapping a certain range of measurement values Q_{meas_LEV} (CPICH_EC/N0, CPICH_RSCP_LEV, P-CCPCH_RSCP_LEV, RXLEV) to a representing quality value Q_{map} (0..99, step size 1).
- 3. In the *Initial cell selection* procedure, the UE shall select one radio access technology and search for a suitable cell. If no suitable cell is found, the UE shall select another radio access technology and search for a suitable cell, and so on. In the *Stored information cell selection* procedure, the UE may use stored information about the selected PLMN. The information may contain information from several radio access technologies.

References

- 1. TS 25.304, 5.2.2.1.1
- 2. TS 25.304, 7.1
- 3. TS 25.304, 5.2.1

6.2.2.1.3 Test purpose

To verify that

- 1. The UE meets conformance requirement 1.
- 2. The UE meets conformance requirement 2
- 3. The UE meets conformance requirement 3

6.2.2.1.4 Method of test

Initial conditions

The relative RF signal to total interference ratio at the UE (CPICH_Ec/lo) between the cells shall be:

T1: Cell 1 < Cell 2 < Cell 3 < Cell 4 < Cell 5 < Cell 6

T2: Cell 1 > Cell 2 > Cell 3 > Cell 4 > Cell 5 > Cell 6

Cell 2 and 5 have S<0, Cell 3 and 6 are barred.

Parameters changed from		Ce (UTF	II 1 RAN)		II 2 RAN)	Cel (UTR	-	
the default values in table 6.1.3.1.Param eter	Unit	T 1	Т2	T 1	Т2	T1	Т2	
Channel Number		UARFCN 1		UARFCN 1		UARFCN 2		
\hat{I}_{or}/I_{oc}	dBm	-5.71	3.06	-3.71	0.06	0.02	-7.25	
CPICH_Ec/lo	dB	-18	-13	-16	-16	-13	-18	
CPICH RSCP	dBm	-86	-77	-84	-80	-80	-87	
Qqualmin	dB	-20		-10				
Qrxlevmin	dBm	-100		-100 -100		-10	00	
Squal	dB	2	2 7		-6	7	2	
Srxlev	dBm	14	23	16	20	20	13	
CellBarred		(0		0		1	

Parameter	Unit	Cell 4 (GSM)			II 5 SM)	Cel (GS	-
		T1	T2	T1	T2	T1	T2
Channel Number		ARFCN 1		ARFCN 2		ARFCN 3	
RF Signal Level	dBm	-70	-95	-60	-100	-50	-105
RXLEV_ACCE SS_MIN	dBm	-100		-5	50	-11	0
C1	dBm	30	5	-10	-50	60	5
CellBarred		(0)	1	•

Test procedure

- a) The SS activates the cells 1-6 according to T1 and monitors cell 1, 2 and 3 for random access request from the UE
- b) The UE is switched on.
- c) The SS waits for random access request from the UE
- d) The UE is switched off.
- e) The SS monitors cells 4, 5 and 6 for random access requests from the UE.
- f) The UE is switched on.
- g) The SS waits to see if there is any random access request from the UE.
- h) The stored information cell selection list in the UE is deleted and the UE is switched off.
- i) Step a-g) is repeated except that the cells are set according to T2 and Cell 1 is set to another PLMN

6.2.2.1.5 Test Requirements

1) In step c), the first response from the UE shall be on Cell 1 within 33 seconds. (Initial cell selection)

[Editor's note: The 33 seconds is taken from GSM as there is no requirement in UMTS to the initial cell selection time. UMTS should not have worse performance than GSM]

2) In step g), there shall be no response from the UE on either Cell 4, 5 or 6 within 33 seconds. (Stored Information cell selection)

[Editor's note: The 33 seconds is taken from GSM as there is no requirement in UMTS to the stored cell selection time. UMTS should not have worse performance than GSM]

3) In step i), the first response from the UE shall be on Cell 4. (Initial cell selection) and no other responses

6.2.2.2 Cell reselection; UTRAN to GSM

6.2.2.2.1 Definition

Test to verify that the UE performs cell reselection correctly when both a GSM and UTRAN network is available and if the serving cell becomes barred or S<0.

6.2.2.2.2 Conformance requirement

- 1. The UE shall evaluate the cell reselection criteria based on radio measurements, and if a better cell is found that cell is selected, procedure *Cell reselection*. The change of cell may imply a change of radio access technology.
- 2. A "suitable cell" is a cell on which the UE may camp on to obtain normal service. Such a cell shall fulfil all the following requirements.

- 2.1 The cell is part of the selected PLMN
- 2.2 The cell is not barred
- 2.3 The cell is not part of a forbidden registration area
- 2.4 The cell selection criteria are fulfilled
- 2.5 The SoLSA criteria are fulfilled [SoLSA support is not in the current release]
- 3. When camped normally, the UE shall execute the cell reselection evaluation process on the following occasions/triggers:
 - 3.1 UE internal triggers, so as to meet performance as specified in TS 25.133
 - 3.2 When information on the BCCH used for the cell reselection evaluation procedure has been modified
- 4. Cell Reselection Criteria:
 - 4.1 The UE shall perform ranking of all cells that fulfil the S criterion
 - 4.2 The cells shall be ranked according to the R criteria. The best ranked cell is the cell with the highest R value. If a FDD cell is ranked as the best cell, the UE shall perform cell re-selection to that FDD cell
 - 4.3 The UE shall reselect the new cell, if the cell reselection criteria are fulfilled during a time interval Treselection.
 - 4.4 The cell-ranking criterion R is derived from Q, Qhyst, Qoffset, TEMP_OFFSET and PENALTY_TIME. However, TEMP_OFFSET and PENALTY_TIME are only applicable when HCS is applied that is when serving cell belongs to a hierarchical structure.
- 5. Different types of measurements are used in different radio access technologies and modes for the cell selection and reselection (CPICH Ec/N0 or CPICH RSCP in UTRA FDD, P-CCPCH RSCP in UTRA TDD, RXLEV in GSM). The use of mapping functions is indicated in system information. Mapping functions are used for mapping a certain range of measurement values Q_{meas_LEV} (CPICH_EC/N0, CPICH_RSCP_LEV, P-CCPCH_RSCP_LEV, RXLEV) to a representing quality value Q_{map} (0..99, step size 1).

References

- 1. TS 25.304, 5.2.2
- 2. TS 25.304, 4.3
- 3. TS 25.304, 5.2.5.1
- 4. TS 25.304, 5.2.6.1.4
- 5. TS 25.304, 7.1

6.1.2.2.3 Test purpose

- 1. To verify that the UE performs reselection on the following occasions:
 - 1.1 Serving cell becomes barred
 - 1.2 S<0 for serving cell

6.2.2.2.4 Method of test

Initial conditions

Step a-c:

Parameter	Unit	Cell 1 (UTRAN)
Test Channel		1
\hat{I}_{or}/I_{oc}	dB	5.87
CPICH_Ec/lo	dB	-11.0
CPICH RSCP	dBm	-74.1
Qqualmin	dB	-20
Qrxlevmin	dBm	-100
Squal*	dB	9.0
Srxlev*	dBm	25.9
CellBarred		0

Parameter	Unit	Cell 2 (GSM)	Cell 3 (GSM)
Test Channel		1	2
RF Signal Level	dBm	-80	-90
RXLEV_ACCESS _MIN	dBm	-100	-100
C1*	dBm	20	10

Step d-f:

Parameter	Unit	Cell 1 (UTRAN)
CellBarred		0 -> 1

Step g:

Parameter	Unit	Cell 1 (UTRAN)
Qqualmin	dB	-20 -> -5
Squal*	dB	9.0 -> -6

Test procedure

- a) The SS activates cells 1, 2, and 3. The SS monitors cells 1, 2 and 3 for random access requests from the UE.
- b) The UE is switched on.
- c) The SS waits for random access request from the UE
- d) The SS sets Cell 1 to be barred
- e) The SS waits for random access request from the UE
- f) The stored information cell selection list in the UE is deleted and the UE is switched off.
- g) Step a-e) is repeated except that in step d), Qqualmin is increased to -5 dB, so S will become negative instead of being barred

6.2.2.2.5 Test Requirements

- 1) In step c), the UE shall select a cell to camp on and eventually make a reselection to Cell 1
- 2) In step e), the UE shall respond on Cell 2
- 3) In step g), the UE shall respond on Cell 2 after Qualmin is increased to -5dB

6.2.2.3 Cell reselection timings; GSM to UTRAN

6.2.2.3.1 Definition

Test to verify that the UE meets the cell reselection timing requirements when both a GSM and UTRAN network is available.

6.2.2.3.2 Conformance requirement

- 1. If the broadcast neighbour cell list includes UTRAN cells or UTRAN frequencies (with or without scrambling code group information), the UE shall, at least every 5 seconds update the value RLA_C for the serving cell and each of the at least 6 strongest non serving GSM cells.
 - 1.1 The UE shall then reselect a suitable UTRAN cell if its measured RSCP value exceeds the value of RLA_C for the serving cell and all of the suitable non-serving GSM cells by the value XXX_Qoffset for a period of 5 seconds and, for FDD, the UTRAN cells measured Ec/No value is equal or greater than the value FDD Qmin.
 - Ec/No and RSCP are the measured quantities
 - FDD_Qmin and XXX_Qoffset are broadcast on BCCH of the serving cell. XXX indicates other radio access technology/mode.
 - 1.2 In case of a cell reselection occurring within the previous 15 seconds, XXX_Qoffset is increased by 5 dB.
 - 1.3 Cell reselection to UTRAN shall not occur within 5 seconds after the UE has reselected a GSM from an UTRAN cell if a suitable GSM cell can be found.
 - 1.4 If more than one UTRAN cell fulfils the above criteria, the UE shall select the cell with the greatest Qmeas value.
- 2. The UE shall be able to identify and select a new best UTRAN cell, which is part of the neighbour cell list, within 30 seconds after it has been activated under the condition that there is only one UTRAN frequency in the neighbour cell list. The allowed time is increased by 30 seconds for each additional UTRAN frequency in the neighbour cell list. However, multiple UTRAN cells on the same frequency in the neighbour cell list does not increase the allowed time.

NOTE: Definitions of measurements are in 3G TS 25.215 and 3G TS 25.101, 3.2 and GSM 05.08, 6.1.

References

- 1. GSM TS 05.08, 6.6.4
- 2. GSM TS 05.08, 6.6

6.1.2.3.3 Test purpose

- 1. To verify that
 - 1.1 The UE meets conformance requirement 1.1 and additionally, that no reselection is performed if the period is less than 5 sec.
 - 1.2 The UE meets conformance requirement 1.2
 - 1.3 The UE meets conformance requirement 1.3

6.2.2.3.4 Method of test

Initial conditions

Step a-e:

Parameter	Unit	Cell 1 (GSM)
Test Channel		1
RF Signal Level	dBm	-75
RXLEV_ACCESS_ MIN	dBm	-100
MS_TXPWR_MAX_ CCH	dBm	Max. output power of UE
FDD_Qmin	dBm	-20
XXX_Qoffset	dBm	5

Parameter	Unit	Cell 2 (UTRAN)
Test Channel		1
\hat{I}_{or}/I_{oc}	dB	-4.74
CPICH_Ec/lo	dB	-16
CPICH_RSCP	dBm	-85
Qqualmin	dB	-20
Qrxlevmin	dBm	-100
Squal*	dB	4
Srxlev*	dBm	15

Step f-g:

Parameter	Unit	Cell 1 (GSM)
RF Signal Level	dBm	-75 -> -95 (4sec) -> -75

Step h-i:

Parameter	Unit	Cell 1 (GSM)
RF Signal Level	dBm	-75 -> -95 ->
		-75

Step k:

Parameter	Unit	Cell 1 (GSM)
		-75 ->
RF Signal Level	dBm	-95 ->
-		-75

Test procedure

NOTE: Step a-c): Test purpose 1.3. Step d-g): test purpose 1.1. Step h-k): test purpose 1.2

- a) The SS activates the channels. The UE is not paged on any of the cells.
- b) The UE is switched on.
- c) After 50 seconds, the SS starts paging continuously on cells 1 and 2 for 20 seconds. The SS monitors cells 1 and 2 for random access requests from the UE.
- d) The SS stops paging on cells 1 and 2 and waits for 20 seconds. (The UE should revert to Cell 1 due to cell reselection).
- e) The SS starts paging continuously on Cell 2.
- f) The SS decreases the transmit level of Cell 1 to –95dBm for a period of 4 s (RSCP will then exceed RXLEV by more than XXX_Qoffset) and then changes the level back to the original value.
- g) The SS waits to see if there is any random access requests from the UE on Cell 2

- h) The SS decreases the transmit level of Cell 1 to –95dBm and waits for the UE to access on Cell 2. The SS records the time t from the decrease in the level of Cell 1 to the first response from the UE.
- i) The SS stops paging on Cell 2 and changes the transmit level of Cell 1 back to the original value.
- j) The SS waits 20 seconds. (The UE should revert to Cell 1 due to cell reselection).
- k) The SS decreases the transmit level of Cell 1 to -95dBm. After t+2 seconds, the SS starts paging continuously on Cell 1, changes the level of Cell 1 back to the original level and waits to see if there is any random access request on Cell 1.

6.2.2.3.5 Test Requirements

- 1) In step c), the UE shall select a cell to camp on and eventually make a reselection to Cell 1
- 2) In step g), there shall be no access on Cell 2 within 34 seconds of decreasing the level of Cell 1.
- 3) In step h), the UE shall respond on Cell 2.
- 4) In step k), there shall be no response on Cell 1 within 11 seconds after the level of Cell 1 is changed back to the original level.

NOTE: The 11 seconds is derived from (t+15) seconds minimum cell reselection timer minus (t+2) seconds from the start of step k) up to the increase of the level of Cell 1. A further 2 seconds are subtracted to cover for any uncertainty introduced by the random access process occurring after step g).

3GPP TSG-T1 Meeting #10 Copenhagen, Denmark, 8-9 February, 2001

Tdoc T1-010086

3GPP TSG-T WG1/SIG SWG Meeting #15 Copenhagen, Denmark, 5th-7th February 20001

Tdoc T1S-010024

	CRJ	Form-v3
	CHANGE REQUEST	01111-43
*	4.123-1 CR 050 # rev - # Current version: 3.2.0 #	
For <u>HELP</u> on u	ing this form, see bottom of this page or look at the pop-up text over the ₭ symbo	ls.
Proposed change	ffects: ### (U)SIM ME/UE Radio Access Network Core Netwo	ork
Title:	Update to GMM test case.	
Source: #	SONY	
Work item code: ₩	Date: 第 5 th Feb 2001	
Category: Ж	F Release: Release: Release: Release: Release: Releas	
	Use one of the following categories: F (essential correction) A (corresponds to a correction in an earlier release) B (Addition of feature), C (Functional modification of feature) D (Editorial modification) Detailed explanations of the above categories can be found in 3GPP TR 21.900. Use one of the following release 2 (GSM Phase 2) R96 (Release 1996) R97 (Release 1997) R98 (Release 1998) R99 (Release 1999) REL-4 (Release 4) REL-5 (Release 5)	es:
Reason for change	It is necessary to - modify the Expected Sequence for UMTS insert Specific message contents to keep consistency with other test items - correct some editorial mistakes.	
Summary of chang	1) Modification of Packet paging procedure in the Expected Sequence. 2) Modification of CS signaling connection setup procedure in the Expected Sequence. 3) Removal of the sub-clause 12.6.2 "Test of ciphering mode setting". 4) Modification of the "Network operation mode". 5) Removal of the "UE operation mode B". 6) Change the expression for the reference document. 7) Addition of the Specific message contents.	Э.
Consequences if not approved:	Inconsistencies with the core specification and other test items are left. Editorial mistakes are left.	
Clauses affected:	₩ Various	
Other specs affected:	# Other core specifications # Test specifications O&M Specifications	

Other comments:

How to create CRs using this form:

Comprehensive information and tips about how to create CRs can be found at: http://www.3gpp.org/3G Specs/CRs.htm. Below is a brief summary:

- 1) Fill out the above form. The symbols above marked # contain pop-up help information about the field that they are closest to.
- 2) Obtain the latest version for the release of the specification to which the change is proposed. Use the MS Word "revision marks" feature (also known as "track changes") when making the changes. All 3GPP specifications can be downloaded from the 3GPP server under ftp://www.3gpp.org/specs/ For the latest version, look for the directory name with the latest date e.g. 2000-09 contains the specifications resulting from the September 2000 TSG meetings.
- 3) With "track changes" disabled, paste the entire CR form (use CTRL-A to select it) into the specification just in front of the clause containing the first piece of changed text. Delete those parts of the specification which are not relevant to the change request.

12 Elementary procedure for Packet Switched Mobility Management

12.1 Applicability, default conditions and default messages

All test cases for PS mobility management apply for all PS mobiles unless otherwise stated in a specific test. Within each test case, the ICS statement indicates whether the test shall be performed for mobiles that can only operate in mode - class A, only in mode - class C, or in both mode - class A and C. For some procedures, the mobile class is of no importance.

Note that only the layer 3 messages are described in the document. The mapping of the layer 3 messages to lower layers and the use of logical channels is not described in this document.

The terms 'PS/CS mode of operation' and 'PS mode of operation' are not used in this specification with some exceptions. Instead the terms 'UE operation mode A' and 'UE operation mode C' are used.

The default conditions and default message contents not specified in this clause must be set as in "PS default conditions"

Below is a list of the RAI values and the corresponding RAC, LAC and MCC used in the test cases:

RAI-1: MCC1/MNC1/LAC1/RAC1 (Used if only one cell)

RAI-2: MCC2/MNC1/LAC1/RAC1

RAI-3: MCC1/MNC1/LAC2/RAC1

RAI-4: MCC1/MNC1/LAC1/RAC2

RAI-5: MCC1/MNC1/LAC1/RAC3

If the User Equipment initial condition specifies that the mobile has a valid IMSI but the initial condition does not mention P-TMSI, than that shall be interpreted as that the mobile has no valid P-TMSI.

The tests are based on 3G3GPP TS 24.008.

12.2 PS attach procedure

This procedure is used to indicate for the network that the IMSI is available for traffic by establishment of a GMM context.

12.2.1 Normal PS attach

The normal PS attach procedure is a GMM procedure used by PS UEs of UE operation mode A or C to IMSI attach for PS services only.

12.2.1.1 PS attach / accepted

12.2.1.1.1 Definition

12.2.1.1.2 Conformance requirement

- 1) If the network accepts the PS attach procedure (signalled by an IMSI) and allocates a P-TMSI, the UE shall acknowledge the P-TMSI and continue communication with the P-TMSI.
- 2) If the network accepts the PS attach procedure (signalled by P-TMSI) and reallocates a new P-TMSI, the UE shall acknowledge the new P-TMSI and continue communication with the new P-TMSI.

3) If the network accepts the PS attach procedure (signalled by a P-TMSI) from the UE without reallocation of the old P-TMSI, the UE shall continue communication with the old P-TMSI.

Reference

3G3GPP-TS 24.008 clause 4.7.3.1

12.2.1.1.3 Test purpose

To test the behaviour of the UE if the network accepts the PS attach procedure.

The following cases are identified:

- 1) P-TMSI / P-TMSI signature is allocated
- 2) P-TMSI / P-TMSI signature is reallocated
- 3) Old P-TMSI / P-TMSI signature is not changed

12.2.1.1.4 Method of test

Initial condition

System Simulator:

One cell operating in network operation mode **IIIII**.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No

UE operation mode A Yes/No UE operation mode C Yes/No Switch off on button Yes/No

Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

- 1) The UE sends an ATTACH REQUEST message with identity IMSI. The SS allocates a P-TMSI and returns ATTACH ACCEPT message with a P-TMSI. The UE acknowledge the P-TMSI by sending ATTACH COMPLETE message. Further communication UE SS is performed by the new P-TMSI.
- 2) The UE sends an ATTACH REQUEST message with identity P-TMSI. The SS reallocates a new P-TMSI and returns ATTACH ACCEPT message with the new P-TMSI. The UE acknowledge the P-TMSI by sending ATTACH COMPLETE message. Further communication UE SS is performed by the new P-TMSI. The UE will not answer signalling addressed to the old P-TMSI.
- 3) The UE sends an ATTACH REQUEST message with identity P-TMSI. The SS accepts the P-TMSI and returns ATTACH ACCEPT message without any P-TMSI. Further communication UE SS is performed by the old P-TMSI.

Expected Sequence

	Step	Direction UE SS	Message	Comments
-	1	UE		The UE is set in UE operation mode C (see ICS). If UE operation mode C not supported,
	2	UE		goto step 26. The UE is powered up or switched on and
	3	->	ATTACH REQUEST	initiates an attach (see ICS). Attach type = 'PS only attached' Mobile identity = IMSI
	4	<-	ATTACH ACCEPT	Attach result = 'PS attach' Mobile identity = P-TMSI-2 P-TMSI-2 signature
	5	->	ATTACH COMPLETE	Routing area identity = RAI-1
	6	<-	PACKET PAGING REQUESTPAGING TYPE1	Mobile identity = P-TMSI-2
	7	->	UPLINK RLC DATA BLOCKSERVICE REQUEST	LLC PDU implicitly indicating paging response. Service type = "paging response"
	8	UE		The UE is switched off or power is removed (see ICS).
	9	->	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, PS detach'
ľ	10	UE		The UE is powered up or switched on and
	4.4		ATTACH BEOLIEGE	initiates an attach (see ICS).
	11	->	ATTACH REQUEST	Attach type = 'PS attach' Mobile identity = P-TMSI-2
				P-TMSI-2 signature Routing area identity = RAI-1
	12	<-	ATTACH ACCEPT	Attach result = 'PS only attached'
		,		Mobile identity = P-TMSI-1
				P-TMSI-1 signature
	13	->	ATTACH COMPLETE	Routing area identity = RAI-1
	14	<-	GMM INFORMATION	Message sent with P-TMSI-1
	14b	->	GMM STATUS	Message sent in case the UE does not support
				reception of GMM information message Cause #97
	15	<-	PACKET PAGING REQUEST	Mobile identity = P-TMSI-2
			Or PAGING REQUEST TYPE	Both paging orders are for TBF establishment. PACKET PAGING REQUEST (used for NW
			4PAGING TYPE1	mode III)
				PAGING REQUEST TYPE 1PAGING TYPE1
	16	UE		(used for NW-mode II). No response from the UE to the request. This
				is checked for 10 seconds.
	17	UE		The UE is switched off or power is removed (see ICS).
	18	->	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, PS detach'
	19	UE		The UE is powered up or switched on and
	20	->	ATTACH REQUEST	initiates an attach (see ICS). Attach type = 'PS attach'
				Mobile identity = P-TMSI-1 P-TMSI-1 signature
				Routing area identity = RAI-1
	21	<-	ATTACH ACCEPT	No new mobile identity assigned.
				P-TMSI and P-TMSI signature not included. Routing area identity = RAI-1
				Attach result = 'PS only attached'
1		ı	1	1

22	<-	PACKET PACING PEOLIEST	Mobile identity = P-TMSI-1
		THORET HONO REGOLOT	Both paging orders are for TBF establishment.
		Or	. 0 0
		PAGING REQUEST TYPE	PACKET PAGING REQUEST (used for NW
		4PAGING TYPE1	mode III)
			PAGING REQUEST TYPE 1 (used for NW-
			mode II).
23	->	UPLINK RLC DATA	LLC PDU implicitly indicating paging
		BLOCKSERVICE REQUEST	response.Service type = "paging response"
24	UE		The UE is switched off or power is removed
			(see ICS).
25	->	DETACH REQUEST	Message not sent if power is removed.
			Detach type = 'power switched off, PS detach'
26	SS		The SS is set in network operation mode II.
27	UE		The UE is set in UE operation mode A (see
			ICS) and the test is repeated from step 2 to
			step 25.

Specific message contents

None.

12.2.1.1.5 Test requirements

UE shall:

- initiate the PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on
- perform the following actions depending on the Mobile identity in the ATTACH REQUEST message and on the Mobile identity in the ATTACH ACCEPT message.

Case 1) The Mobile identity in the ATTACH REQUEST is the IMSI and the Mobile identity in the ATTACH ACCEPT message is the P-TMSI.

UE shall:

- acknowledge the P-TMSI by sending ATTACH COMPLETE message. Further communication UE - SS is performed by the P-TMSI.

Case 2) The Mobile identity in the ATTACH REQUEST is the P-TMSI and the Mobile identity in the ATTACH ACCEPT message is the new P-TMSI.

UE shall:

- acknowledge the new P-TMSI by sending ATTACH COMPLETE message. Further communication UE - SS is performed by the other P-TMSI.

Case 3) The Mobile identity in the ATTACH REQUEST is the P-TMSI and the Mobile identity in the ATTACH ACCEPT message is the same P-TMSI.

UE shall:

- acknowledge the same P-TMSI by sending ATTACH COMPLETE message. Further communication UE - SS is performed by the same P-TMSI.

12.2.1.2 PS attach / rejected / IMSI invalid / illegal UE

12.2.1.2.1 Definition

12.2.1.2.2 Conformance requirement

- 1) If the network rejects a PS attach procedure from the User Equipment with the cause 'Illegal UE', the User Equipment shall consider USIM invalid for PS services until power is switched off or USIM is removed.
- 2) If the network rejects a PS attach procedure from the User Equipment with the cause 'Illegal UE' the User Equipment shall delete the stored RAI, PS-CKSN, P-TMSI and P-TMSI signature.

Reference

3G3GPP-TS 24.008 clause 4.7.3.1

12.2.1.2.3 Test purpose

To test the behaviour of the UE if the network rejects the PS attach procedure of the UE with the cause 'illegal UE'.

12.2.1.2.4 Method of test

Initial condition

System Simulator:

Three cells (not simultaneously activated), cell A with MCC1/MNC1/LAC1/RAC1, cell B in

MCC1/MNC1/LAC2/RAC1, cell C in MCC2/MNC1/LAC1/RAC1.

All three cells are operating in network operation mode II (in case of UE operation mode A), or in network

operation mode III (in case of UE operation mode C).

User Equipment:

The UE has a valid P-TMSI-1, P-TMSI-1 signature and RAI-1.

Related ICS/IXIT statements

Support of PS service Yes/No

UE operation mode C Yes/No

UE operation mode A Yes/No (only if mode C not supported)

Switch off on button Yes/No

Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The SS rejects a PS attach with the cause value 'Illegal UE'. The SS checks that the UE does not perform PS attach in the same or another PLMN.

Expected Sequence

Ī	Step	Direction	Message	Comments
	-	UE SS		
-		•		The following messages are sent and shall be
				received on cell A.
	1	UE		The UE is set in UE operation mode C or B
.				(see ICS).
	2	SS		The SS is set in network operation mode II or
	_			III and activates cell A.
	3	UE		The UE is powered up or switched on and
				initiates an attach (see ICS). Cell A is preferred
	4		ATTACLIBECT	by the UE.
	4	->	ATTACH REQUEST	Attach type = 'PS attach'
				Mobile identity = P-TMSI-1
				P-TMSI-1 signature Routing area identity = RAI-1
	5	_	ATTACH REJECT	GMM cause = 'Illegal UE'.
-	J	<-	ATTACTTRESECT	The following messages are sent and shall be
				received on cell B.
	6	SS		The SS deactivates cell A and activates cell B.
	7	UE		Cell B is preferred by the UE.
	8	ÜE		No ATTACH REQUEST sent to the SS
	-	-		(SS waits 30 seconds).
	9	UE		The UE initiates an attach by MMI or by AT
				command.
	10	UE		No ATTACH REQUEST sent to the SS
Ĺ				(SS waits 30 seconds).
				The following messages are sent and shall be
				received on cell C.
	11	SS		The SS deactivates cell B and activates cell C.
	12	UE		Cell C is preferred by the UE.
	13	UE		No ATTACH REQUEST sent to the SS
	14	UE		(SS waits 30 seconds). The UE initiates an attach by MMI or by AT
	17	OL		command.
	15	UE		No ATTACH REQUEST sent to the SS
	10	02		(SS waits 30 seconds).
	16	UE		If possible (see ICS) switch off is performed.
	-	-		Otherwise the power is removed.
ľ	17	UE		The UE is powered up or switched on and
				initiates an attach (see ICS).
	18	->	ATTACH REQUEST	Attach type = 'PS attach'
				Mobile identity = IMSI
	19	<-	ATTACH ACCEPT	Attach result = 'PS only attached'
				Mobile identity = P-TMSI-1
				P-TMSI-1 signature
	20		ATTACH COMPLETE	Routing area identity = RAI-2
	20	-> !!E	ATTACH COMPLETE	The LIE is switched off or newer is removed
	21	UE		The UE is switched off or power is removed
	22	->	DETACH REQUEST	(see ICS). Message not sent if power is removed.
	44	->	DETAGITREQUEST	Detach type = 'power switched off, PS detach'
. L				Detaon type - power switched on, F3 detach

Specific message contents

None.

12.2.1.2.5 Test requirements

UE shall:

- initiate a PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on
- check the GMM cause which is contained in the ATTACH REJECT message and stop timer T3310.
- set the PS update state to GU3 ROAMING NOT ALLOWD and delete stored P-TMSI, P-TMSI signature, RAI and PS ciphering key sequence number.
- not send the ATTACH REQUEST message to SS, even if there is an instruction of attach request from MMI or from AT command.

12.2.1.3 PS attach / rejected / IMSI invalid / PS services not allowed

12.2.1.3.1 Definition

12.2.1.3.2 Conformance requirement

- If the network rejects a PS attach procedure from the User Equipment with the cause 'PS services not allowed', the User Equipment shall consider USIM invalid for PS services until power is switched off or USIM is removed.
- 2) If the network rejects a PS attach procedure from the User Equipment with the cause 'PS services not allowed' the User Equipment shall delete the stored RAI, PS-CKSN, P-TMSI and P-TMSI signature.

Reference

3G-3GPP TS 24.008 clause 4.7.3.1

12.2.1.3.3 Test purpose

To test the behaviour of the UE if the network rejects the PS attach procedure of the UE with the cause 'PS services not allowed' (no valid PS-subscription for the IMSI).

12.2.1.3.4 Method of test

Initial condition

System Simulator:

Two cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1 (HPLMN) and cell B in MCC2/MNC1/LAC1/RAC1.

Both cells are operating in network operation mode II.

User Equipment:

The UE has a valid P-TMSI-1, P-TMSI-1 signature and RAI-1.

Related ICS/IXIT statements

Support of PS service Yes/No

UE operation mode C Yes/No

UE operation mode A Yes/No

USIM removal possible without powering down Yes/No

Switch off on button Yes/No

Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The SS rejects a normal attach with the cause value 'PS services not allowed'. The SS checks that the UE does not perform PS attach in another PLMN.

Expected Sequence

Step	Direction	Message	Comments
P	UE SS		
1 2	SS UE		The following messages are sent and shall be received on cell A. The SS activates cell A. The UE is set in UE operation mode C (see ICS). If UE operation mode C not supported,
3	UE		goto step 16. The UE is powered up or switched on and initiates an attach (see ICS). Cell A is preferred
4	->	ATTACH REQUEST	by the UE. Attach type = 'PS attach' Mobile identity = P-TMSI-1 P-TMSI-1 signature Routing area identity = RAI-1
5	<-	ATTACH REJECT	GMM cause = 'PS services not allowed'
6 7 8	SS UE UE		The following messages are sent and shall be received on cell B. The SS deactivates cell A and activates cell B. Cell B is preferred by the UE. No ATTACH REQUEST sent to the SS (SS waits 30 seconds).
9	UE		If possible (see ICS) USIM removal is performed. Otherwise if possible (see ICS) switch off is performed. Otherwise the power is removed.
10	UE		The UE gets the USIM replaced, is powered up or switched on and initiates an attach (see ICS).
11	->	ATTACH REQUEST	Attach type = 'PS attach' Mobile identity = IMSI
12	<-	ATTACH ACCEPT	Attach result = 'PS only attached' Mobile identity = P-TMSI-1 P-TMSI-1 signature Routing area identity = RAI-2
13 14	-> UE	ATTACH COMPLETE	The UE is switched off or power is removed (see ICS).
15	->	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, PS detach'
16 17	UE		The SS deactivates cell B and activates cell A. The UE is set in UE operation mode A(see ICS) and the test is repeated from step 3 to step 15.

Specific message contents

None.

12.2.1.3.5 Test requirements

UE shall:

- initiate a PS attach procedure with information elements specified in the above Expected Sequence when UE is powered up or switched on.
- check the GMM cause which is contained in the ATTACH REJECT message and stop timer T3310.

- set the PS update state to GU3 ROAMING NOT ALLOWD.
- delete stored P-TMSI, P-TMSI signature, RAI and PS ciphering key sequence number.

12.2.1.4 PS attach / rejected / PLMN not allowed

12.2.1.4.1 Definition

12.2.1.4.2 Conformance requirement

- 1) If the network rejects a PS attach procedure from the User Equipment with the cause 'PLMN not allowed' the User Equipment shall:
 - 1.1 not perform PS attach when switched on in the same routing area or location area.
 - 1.2 not perform PS attach when in the same PLMN and when that PLMN is not selected manually.
 - 1.3 delete the stored RAI, PS-CKSN, P-TMSI and P-TMSI signature.
 - 1.4 store the PLMN in the 'forbidden PLMN' list.
- 2) If the network rejects a PS attach procedure from the User Equipment with the cause 'PLMN not allowed' the User Equipment shall perform PS attach when a new PLMN is entered.
- 3) If the network rejects a PS attach procedure from the User Equipment with the cause 'PLMN not allowed' and if after that the PLMN from which this rejection was received, is manually selected, the User Equipment shall perform a PS attach procedure.

Reference

3G-3GPP TS 24.008 clause 4.7.3.1

12.2.1.4.3 Test purpose

To test the behaviour of the UE if the network rejects the PS attach procedure of the UE with the cause 'PLMN not allowed'.

12.2.1.4.4 Method of test

Initial condition

System Simulator:

Four cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC1, cell C in MCC1/MNC1/LAC2/RAC1 and cell D in MCC2/MNC1/LAC1/RAC1. All four cells are operating in network operation mode II (in case of UE operation mode A), or in network operation mode III (in case of UE operation mode C).

User Equipment:

The UE has a valid P-TMSI-1, P-TMSI-1 signature and RAI-1.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode C Yes/No
UE operation mode A Yes/No (only if mode C not supported)
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The SS rejects a PS attach with the cause value 'PLMN not allowed'. The SS checks that the UE does not perform PS attach if activated in the same routing area or location area and performs PS attach only when a new PLMN is entered.

Expected Sequence

The following messages are sent and shall be received on cell A. The UE is set in UE operation mode C or B (see ICS).
received on cell A. The UE is set in UE operation mode C or B
The SS is set in network operation mode II er
The UE is powered up or switched on and initiates an attach (see ICS). Cell A is preferred
by the UE. Attach type = 'PS attach' Mobile identity = P-TMSI-1 P-TMSI-1 signature
Routing area identity = RAI-1 GMM cause = 'PLMN not allowed' No ATTACH REQUEST sent to SS (SS waits 30 seconds).
The following messages are sent and shall be received on cell B. The SS deactivates cell A and activates cell B. Cell B is preferred by the UE. No ATTACH REQUEST sent to SS (SS waits 30 seconds).
The following messages are sent and shall be received on cell C. The SS deactivates cell B and activates cell C. Cell C is preferred by the UE. No ATTACH REQUEST sent to SS (SS waits 30 seconds).
The following messages are sent and shall be received on cell D. The SS deactivates cell C and activates cell D. Cell D is preferred by the UE. The UE initiates an attach automatically, by MMI or by AT command.
Attach type = 'PS attach' Mobile identity = IMSI
Attach result = 'PS only attached' Mobile identity = P-TMSI-1 P-TMSI-1 signature
Routing area identity = RAI-2 The UE is switched off or power is removed
(see ICS). Message not sent if power is removed. Detach type = 'power switched off, PS detach'

Specific message contents

None.

12.2.1.4.5 Test requirements

UE shall:

- initiate a PS attach procedure with information elements specified in the above Expected Sequence when UE is powered on or switched on.
- check the GMM cause which is contained in the ATTACH REJECT message and stop timer T3310.
- perform the following actions depending on the PLMN or the routing area or the location area

Case 1) UE is in the same routing area or location area when the power is switched on,

UE shall:

- not perform PS attach.
- delete the stored RAI, PS-CKSN, P-TMSI and P-TMSI signature.
- store the PLMN in the 'forbidden PLMN' list.

Case 2) UE is in the same PLMN, and this PLMN is not selected manually

UE shall:

- not perform PS attach.
- delete the stored RAI, PS-CKSN, P-TMSI and P-TMSI signature.
- store the PLMN in the 'forbidden PLMN' list.

Case3) UE is in a new location area.

UE shall:

- perform PS attach.

Case3) UE is in the new PLMN, and this PLMN is selected manually

UE shall

- perform PS attach.

12.2.1.5 PS attach / rejected / roaming not allowed in this location area

12.2.1.5.1 Definition

12.2.1.5.2 Conformance requirement

- 1) If the network rejects a PS attach procedure from the User Equipment with the cause 'roaming not allowed in this location area' the User Equipment shall:
 - 1.1 not perform PS attach when in the same location area.
 - 1.2 delete the stored RAI, PS-CKSN, P-TMSI and P-TMSI signature.
 - 1.3 store the LA in the 'forbidden location areas for roaming' list.
 - 1.4 perform PS attach when a new location area is entered.
 - 1.5 Periodically search for its HPLMN.
- 2) The User Equipment shall reset the list of 'Forbidden location areas for roaming' when switched off or when the USIM is removed.
- 3) The UE shall be capable of storing at least 6 entries in the list of 'Forbidden location areas for roaming'.

Reference

3G3GPP-TS 24.008 clause 4.7.3.1

12.2.1.5.3

Test purpose

Test purpose 1

To test that on receipt of a rejection using the 'roaming not allowed in this location area' cause code, the UE ceases trying to attach on that location area. Successful PS attach procedure is possible in other location areas.

Test purpose 2

To test that if the UE is switched off or the USIM is removed the list of 'forbidden location areas for roaming' is cleared.

Test purpose 3

To test that at least 6 entries can be held in the list of 'forbidden location areas for roaming' (the requirement in 3G3GPP TS 24.008 is to store at least 10 entries. This is not fully tested by the third procedure).

Test purpose 4

To test that if a cell of the Home PLMN is available then the UE returns to it in preference to any other available cell.

12.2.1.5.4 Method of test

12.2.1.5.4.1 Test procedure 1

Initial condition

System Simulator:

Three cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in

MCC1/MNC/LAC2/RAC1 and cell C in MCC1/MNC1/LAC1/RAC2.

All three cells are operating in network operation mode IIIII.

User Equipment:

The UE has a valid P-TMSI-1, P-TMSI-1 signature and RAI-1.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode C Yes/No
UE operation mode A Yes/No
Switch off on button Yes/No

Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The SS rejects a PS attach with the cause value 'Roaming not allowed in this area'. A new attempt for a PS attach is not possible. Successful PS attach / detach procedures are performed in another location area. A new attempt for a PS attach is performed in the 1^{st} location area. This attempt shall not succeed, as the LA is on the forbidden list.

Step	Direction	Message	Comments
	UE SS		
	SS		The following messages are sent and shall be
			received on cell A.
1	SS		The SS activates cell A.
2	UE		The UE is set in UE operation mode C (see
			ICS). If UE operation mode C not supported,
			goto step 19.
3	UE		The UE is powered up or switched on and
			initiates an attach (see ICS). Cell A is preferred by the UE.
4		ATTACH REQUEST	Attach type = 'PS attach'
4	->	ATTACH REQUEST	Mobile identity = P-TMSI-1
			P-TMSI-1 signature
			Routing area identity = RAI-1
5	<-	ATTACH REJECT	GMM cause = 'Roaming not allowed in this
		ATTAOTTRESECT	area'
6	UE		No ATTACH REQUEST sent to SS
			(SS waits 30 seconds).
			The following messages are sent and shall be
			received on cell B.
7	SS		The SS Deactivates cell A and activates cell B.
8	ÜE		Cell B is preferred by the UE.
9	ÜE		The UE initiates an attach automatically, by
			MMI or by AT command.
10	->	ATTACH REQUEST	Attach type = 'PS attach'
			Mobile identity = IMSI
11	<-	ATTACH ACCEPT	Attach result = 'PS only attached'
			Mobile identity = P-TMSI-1
			P-TMSI-1 signature
			Routing area identity = RAI-3
12	->	ATTACH COMPLETE	
13	UE		The UE initiates a PS detach (without power
		DETACH DECLIEST	off) by MMI or by AT command .
14 15	->	DETACH REQUEST	Detach type = 'normal detach, PS detach'
15	<-	DETACH ACCEPT	The following managers are cent and shall be
			The following messages are sent and shall be received on cell C.
16	SS		The SS deactivates cell B and activates cell C.
17	UE		Cell C is preferred by the UE.
18	UE		No ATTACH REQUEST sent to SS
10			(SS waits 30 seconds).
19	SS		The SS is set in network operation mode II.
20	UE		The UE is set in UE operation mode A (see
			ICS) and the test is repeated from step 3 to
			step 18.

12.2.1.5.4.2 Test procedure 2

Initial condition

System Simulator:

One cell operating in network operation mode II.

User Equipment:

The UE has a valid P-TMSI-1, P-TMSI-1 signature and RAI-1.

Related ICS/IXIT statements

Support of PS service Yes/No UE operation mode C Yes/No

UE operation mode A Yes/No (only if mode C not supported)

Switch off on button Yes/No

Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The SS rejects a PS attach updating with the cause value 'Roaming not allowed in this area'. The UE is switched off for 10 seconds and switched on again. The SS check that a PS attach is possible on the cell on which the PS attach had been rejected.

If USIM removal is possible without switching off: The SS rejects a PS attachupdating with the cause value 'Roaming not allowed in this area'. The USIM is removed and inserted in the UE. The SS check that a PS attach is possible on the cell on which the PS attach had been rejected.

Expected Sequence

Step	Direction	Message	Comments
	UE SS		
1	UE		The UE is set in UE operation mode C or B
			(see ICS).
2	UE		The UE is powered up or switched on and
3	_	ATTACH BEOLIEST	initiates an attach (see ICS).
3	->	ATTACH REQUEST	Attach type = 'PS attach' Mobile identity = P-TMSI-1
			P-TMSI-1 signature
			Routing area identity = RAI-1
4	<-	ATTACH REJECT	GMM cause = 'Roaming not allowed in this
			area'
5	UE		No ATTACH REQUEST sent to the SS
			(SS waits 30 seconds).
6	UE		If possible (see ICS) switch off is performed.
			Otherwise the power is removed.
7	UE		The UE is powered up or switched on and
•	0_		initiates an attach (see ICS).
8	->	ATTACH REQUEST	Attach type = 'PS attach'
			Mobile identity = IMSI
9	<-	ATTACH ACCEPT	Attach result = 'PS only attached'
			Mobile identity = P-TMSI-1
			P-TMSI-1 signature
10	->	ATTACH COMPLETE	Routing area identity = RAI-1
11	UÉ	THE SOUND LETE	The UE is switched off or power is removed
''	OL.		(see ICS).
12	->	DETACH REQUEST	Message not sent if power is removed.
			Detach type = 'power switched off, PS detach'

12.2.1.5.4.3 Test procedure 3

Initial condition

System Simulator:

Six cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC2/RAC1, cell C in MCC1/MNC1/LAC3/RAC1, cell D in MCC1/MNC1/LAC4/RAC1, cell E in MCC1/MNC1/LAC5/RAC1, cell F in MCC1/MNC1/LAC6/RAC1.

All six cells are operating in network operation mode II (in case of UE operation mode A), or in network operation mode III (in case of UE operation mode C).

User Equipment:

The UE has a valid P-TMSI-1, P-TMSI-1 signature and RAI-1.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode C Yes/No
UE operation mode A Yes/No (only if mode C not supported)
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The SS rejects a PS attach with the cause value 'Roaming not allowed in this area'. This is done for 6 different location areas. Then the SS checks that the UE does not attempt to perform an attach procedure on the non-allowed location areas.

Different types of UE may use different methods to periodically clear the list of forbidden areas (e.g. every day at 12am) for roaming. If the list is cleared while the test is being run, it may be necessary to re-run the test.

	Step	Direction UE SS	Message	Comments
F		SS		The following messages are sent and shall be
	1	SS		received on cell A. The SS is set in network operation mode II-er
	2	UE		III and activates cell A. The UE is set in UE operation mode C or B
	2	UE		(see ICS).
	3	UE		The UE is powered up or switched on and initiates an attach (see ICS). Cell A is preferred
				by the UE.
	4	->	ATTACH REQUEST	Attach type = 'PS attach' Mobile identity = P-TMSI-1
				P-TMSI-1 signature Routing area identity = RAI-1
	5	<-	ATTACH REJECT	GMM cause = 'Roaming not allowed in this
	6	UE		area' No ATTACH REQUEST sent to SS
				(SS waits 30 seconds)
				The following messages are sent and shall be received on cell B.
	7	SS		The SS deactivates cell A and activates cell B.
	8 9	UE UE		Cell B is preferred by the UE. The UE initiates an attach automatically, by
	· ·	0_		MMI or by AT command.
	10	->	ATTACH REQUEST	Attach type = 'PS attach' Mobile identity = IMSI
	11	<-	ATTACH REJECT	GMM cause = 'Roaming not allowed in this
	12	UE		area' No ATTACH REQUEST sent to SS
	12	UE		(SS waits 30 seconds).
Ī				The following messages are sent and shall be received on cell C.
	13	SS		The SS deactivates cell B and activates cell C.
	14	UE		Cell C is preferred by the UE.
	15	UE		The UE initiates an attach automatically, by MMI or by AT command.
	16	->	ATTACH REQUEST	Attach type = 'PS attach'
	47		ATTACH DE IECT	Mobile identity = IMSI
	17	<-	ATTACH REJECT	GMM cause = 'Roaming not allowed in this area'
	18	UE		No ATTACH REQUEST sent to SS (SS waits 30 seconds).
-				The following messages are sent and shall be
	19	SS		received on cell D. The SS deactivates cell C and activates cell D.
	20	UE		Cell D is preferred by the UE.
	21	UE		The UE initiates an attach automatically, by MMI or by AT command.
	22	->	ATTACH REQUEST	Attach type = 'PS attach'
	23	<-	ATTACH REJECT	Mobile identity = IMSI GMM cause = 'Roaming not allowed in this
	24	UE		area' No ATTACH REQUEST sent to SS
L		<u> </u>		(SS waits 30 seconds).
				The following messages are sent and shall be received on cell E.
	25	SS		The SS deactivates cell D and activates cell E.
	26 27	UE UE		Cell E is preferred by the UE.
	۷1	UE		The UE initiates an attach automatically, by MMI or by AT command.
	28	->	ATTACH REQUEST	Attach type = 'PS attach'
	29	<-	ATTACH REJECT	Mobile identity = IMSI GMM cause = 'Roaming not allowed in this
				area'

30	UE		No ATTACH REQUEST sent to SS (SS waits 30 seconds).
31 32 33	SS UE UE		The following messages are sent and shall be received on cell F. The SS deactivates cell E and activates cell F. Cell F is preferred by the UE. The UE initiates an attach automatically, by MMI or by AT command.
34	->	ATTACH REQUEST	Attach type = 'PS attach' Mobile identity = IMSI
35	<-	ATTACH REJECT	GMM cause = 'Roaming not allowed in this area'
36	UE		No ATTACH REQUEST sent to SS (SS waits 30 seconds)
37 38 39 40	SS SS UE UE		The following messages are sent and shall be received on cell E. The SS deactivates cell F and activates cell E. Cell E is preferred by the UE. The UE initiates an attach automatically, by MMI or by AT command. No ATTACH REQUEST sent to SS
41 42 43 44	SS SS UE UE		(SS waits 30 seconds). The following messages are sent and shall be received on cell C. The SS deactivates cell E and activates cell C. Cell C is preferred by the UE. The UE initiates an attach automatically, by MMI or by AT command. No ATTACH REQUEST sent to SS
15			(SS waits 30 seconds). The following messages are sent and shall be received on cell A.
45 46 47	SS SS UE		The SS deactivates cell C and activates cell A. Cell A will be preferred by the UE. The UE initiates an attach automatically, by
48	UE		MMI or by AT command. No ATTACH REQUEST sent to SS (SS waits 30 seconds).

12.2.1.5.4.4 Test procedure4

Initial condition

System Simulator:

Two cells, cell A in MCC2/MNC1/LAC1/RAC1 (not HPLMN) and cell B in MCC1/MNC1/LAC1/RAC1 (HPLMN).

Both cells are operating in network operation mode II (in case of UE operation mode A), or in network operation mode III (in case of UE operation mode C).

User Equipment:

The UE has a valid P-TMSI-1, P-TMSI-1 signature and RAI-2.

Related ICS/IXIT statements

Support of PS service Yes/No UE operation mode C Yes/No

UE operation mode A Yes/No (only if mode C not supported)

Switch off on button Yes/No

Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The SS rejects a PS attach with the cause value 'Roaming not allowed in this area. Two cells are then available. The cell with the weakest level belongs to the HPLMN. It is checked that the UE returns to its HPLMN.

Expected Sequence

Ī	Step	Direction	Message	Comments
	-	UE SS	_	
		SS		The following messages are sent and shall be
				received on cell A.
	1	UE		The UE is set in UE operation mode C or B
				(see ICS).
	2	SS		The SS is set in network operation mode II-or
				III and activates cell A.
	3	UE		The UE is powered up or switched on and
				initiates an attach (see ICS). Cell A is preferred
			4.TT4.011.DE01150T	by the UE.
	4	->	ATTACH REQUEST	Attach type = 'PS attach'
				Mobile identity = P-TMSI-1
				P-TMSI-1 signature
	_		ATTACLIBETECT	Routing area identity = RAI-2
	5	<-	ATTACH REJECT	GMM cause = 'Roaming not allowed in this area'
	6	UE		No ATTACH REQUEST sent to SS
	О	UE		(SS waits 30 seconds).
ŀ				
				The following messages are sent and shall be received on cell B.
	7	SS		Activate cell B with a lower signal strength than
	,	33		cell A.
	8	UE		The RF level of cell A is lowered until cell B is
	U	OL		preferred by the UE.
	9	UE		The UE initiates an attach automatically, by
	J	02		MMI or by AT command.
	10	->	ATTACH REQUEST	Attach type = 'PS attach'
				Mobile identity = IMSI
	11	<-	ATTACH ACCEPT	Attach result = 'PS only attached'
				Mobile identity = P-TMSI-1
				P-TMSI-1 signature
				Routing area identity = RAI-1
	12	->	ATTACH COMPLETE	
	13	UE		The UE is switched off or power is removed
				(see ICS).
	14	->	DETACH REQUEST	Message not sent if power is removed.
				Detach type = 'power switched off, PS detach'

Specific message contents

12.2.1.5.5 Test requirements

UE shall:

- initiate a PS attach procedure with information elements specified in the above Expected Sequence when UE is powered up or switched on.
- check the GMM cause which is contained in the ATTACH REJECT message and stop timer T3310.
- not perform PS attach when UE is in the same location area.
- delete the stored RAI, PS-CKSN, P-TMSI and P-TMSI signature.
- store the LA in the 'forbidden location areas for roaming' list.
- perform PS attach when a new location area is entered.

- search for its HPLMN periodically.

When Switched off or when the USIM is removed,

UE shall:

- reset the 'forbidden location areas for roaming' list.

12.2.1.6 PS attach / abnormal cases / access barred due to access class control

12.2.1.6.1 Definition

12.2.1.6.2 Conformance requirement

- 1) The UE shall not perform PS attach procedure, but stays in the current serving cell and applies normal cell reselection process.
- 2) The User Equipment shall perform the PS attach procedure when:
 - 2.1 Access is granted.
 - 2.2 Cell is changed.

Reference

3G-3GPP TS 24.008 clause 4.7.3.1

12.2.1.6.3

Test purpose

Test purpose1

To test the behaviour of the UE in case of access class control (access is granted).

Test purpose2

To test the behaviour of the UE in case of access class control (access is granted).

12.2.1.6.4 Method of test

12.2.1.6.4.1 Test procedure1

Initial condition

A random access class x (0-15) is selected. The USIM is programmed with this access class x. Communication with User Equipments using access class x is initially indicated to be barred.

System Simulator:

One cell operating in network operation mode IIIII. Access class x barred.

User Equipment:

The UE has a valid P-TMSI-1, P-TMSI-1 signature and RAI-1.

Related ICS/IXIT statements

Support of PS service Yes/No

UE operation mode C Yes/No

UE operation mode A Yes/No

Switch off on button Yes/No

Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The SS indicates access class x barred. A PS attach procedure is not performed.

The SS indicates that access class x is not barred. A PS attach procedure is performed.

Expected Sequence

Step	Direction	Message	Comments
	UE SS		
1	UE		The USIM is programmed with access class x.
2	UE		The UE is set in UE operation mode C (see ICS). If UE operation mode C not supported, goto step 12.
3	UE		The UE is powered up or switched on and attempts to initiate an attach (see ICS).
4	UE		No ATTACH REQUEST sent to SS, as access class X is barred (SS waits 30 seconds).
5	SS		The access class x is not barred anymore.
6	UE		The UE automatically initiates a PS attach.
7	->	ATTACH REQUEST	Attach type = 'PS attach' Mobile identity = P-TMSI-1
8	<-	ATTACH ACCEPT	P-TMSI-1 signature Routing area identity = RAI-1 Attach result = 'PS only attached' Mobile identity = P-TMSI-2
			P-TMSI-2 signature
9	->	ATTACH COMPLETE	Routing area identity = RAI-1
10	UE	THE TOTAL CONTRACT OF THE PARTY	The UE is switched off or power is removed
			(see ICS).
11	->	DETACH REQUEST	Message not sent if power is removed.
12	SS		Detach type = 'power switched off, PS detach' The SS is set in network operation mode II.
13	UE		The UE is set in UE operation mode A(see
13	UL.		ICS) and the test is repeated from step 2 to
			step 11.

12.2.1.6.4.2 Test procedure2

Initial condition

A random access class x (0-15) is selected. The USIM is programmed with this access class x. Communication with User Equipments using access class x is indicated to be barred on cell A.

System Simulator:

Two cells, cell A in MCC1/MNC1/LAC1/RAC1 has access class x barred, cell B in MCC1/MNC1/LAC1/RAC1 has access class x not barred.

Both cells are operating in network operation mode II (in case of UE operation mode A), or in network operation mode III (in case of UE operation mode C).

User Equipment:

The UE has a valid P-TMSI-1, P-TMSI-2 signature and RAI-1.

Related ICS/IXIT statements

Support of PS service Yes/No

UE operation mode C Yes/No

UE operation mode A Yes/No Switch off on button Yes/No

Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The SS indicates access class x barred. A PS attach procedure is not performed.

A cell change is performed into a cell where access class x is not barred. A PS attach procedure is performed.

Expected Sequence

Ī	Step	Direction	Message	Comments
	-	UE SS		
	1	UE SS		The USIM is programmed with access class x. The following messages are sent and shall be received on cell A.
	2	SS		The SS is set in network operation mode II-or III and activates cell A.
	3	UE		The UE is set in UE operation mode C or B (see ICS).
	4	UE		The UE is powered up or switched on and attempts to initiate an attach (see ICS).
	5	UE		No ATTACH REQUEST sent to SS, as access class X is barred (SS waits 30 seconds).
	6	SS		The following messages are sent and shall be received on cell B. Activate cell B with a lower signal strength than cell A The RF level of cell A is lowered until cell
	7	UE		B is preferred by the UE. The UE automatically initiates an attach.
	8	->	ATTACH REQUEST	Attach type = 'PS attach' Mobile identity = P-TMSI-2 P-TMSI-2 signature
	9	<-	ATTACH ACCEPT	Routing area identity = RAI-1 Attach result = 'PS only attached' Mobile identity = P-TMSI-1 P-TMSI-1 signature Routing area identity = RAI-1
	10 11	-> UF	ATTACH COMPLETE	The UE is switched off or power is removed
		OL		(see ICS).
	12	->	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, PS detach'

Specific message contents

None.

12.2.1.6.5 Test requirements

UE shall:

- perform the following actions depending on the UE access class X.

Case 1) The UE access class X is barred,

UE shall:

- not perform a PS attach procedure.
- stay in the current serving cell.
- apply normal cell reselection process.

Case 2) The UE access class X is granted or serving cell is changed,

UE shall:

- initiate PS attach procedure..

12.2.1.7 PS attach / abnormal cases / change of cell into new routing area

12.2.1.7.1 Definition

12.2.1.7.2 Conformance requirement

When a change of cell into a new routing area is performed before ATTACH ACCEPT message is received by the UE, the UE shall abort the PS attach procedure and re-initiate it immediately.

Reference

3G-3GPP TS 24.008 clause 4.7.3.1

12.2.1.7.3 Test purpose

To test the behaviour of the UE in case of procedure collision.

12.2.1.7.4 Method of test

Initial condition

System Simulator:

Two cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1 and cell B in MCC1/MNC1/LAC1/RAC2.

Both cells are operating in network operation mode II (in case of UE operation mode A), or in network operation mode III (in case of UE operation mode C).

User Equipment:

The UE has a valid P-TMSI-1, P-TMSI-1 signature and RAI-1.

Related ICS/IXIT statements

Support of PS service Yes/No

UE operation mode C Yes/No

UE operation mode A Yes/No (only if mode C not supported)

Switch off on button Yes/No

Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE initiates a PS attach procedure. The ATTACH ACCEPT message is delayed from the SS. The UE performs a cell reselection to a cell in a new routing area. The UE shall re-initiate a PS attach procedure in the new routing area.

Expected Sequence

Γ	Step	Direction	Message	Comments
	•	UE SS	· ·	
Ī		SS		The following messages are sent and shall be
.	_			received on cell A.
	1	UE		The UE is set in UE operation mode C or B
ı	2	SS		(see ICS).
	2	33		The SS is set in network operation mode II or III and activates cell A.
l	3	UE		The UE is powered up or switched on and
	O	OL.		initiates an attach (see ICS). Cell A is preferred
				by the UE.
	4	->	ATTACH REQUEST	Attach type = 'PS attach'
				Mobile identity = P-TMSI-1
				P-TMSI-1 signature
	_	00		Routing area identity = RAI-1
	5	SS		No response to the ATTACH REQUEST
-				message is given by the SS. The following messages are sent and shall be
				received on cell B.
	6	SS		The SS deactivates cell A and activates cell B.
	Ū			Cell B is preferred by the UE.
	7	UE		The UE automatically re-initiates the attach in
				the new cell.
	8	->	ATTACH REQUEST	Attach type = 'PS attach'
				Mobile identity = P-TMSI-1
				P-TMSI-1 signature
	9	<-	ATTACH ACCEPT	Routing area identity = RAI-1 No new mobile identity assigned.
	9	\- <u>-</u>	ATTACITACCETT	P-TMSI and P-TMSI signature not included.
				Attach result = 'PS only attached'
				Routing area identity = RAI-4
	10	UE		The UE is switched off or power is removed
				(see ICS).
	11	->	DETACH REQUEST	Message not sent if power is removed.
				Detach type = 'power switched off, PS detach'

Specific message contents

None.

12.2.1.7.5 Test requirements

UE shall:

- abort a PS attach procedure when a change of cell into a new routing area is performed before ATTACH ACCEPT or ATTACH REJECT message is received by the UE.
- re-initiate a PS attach procedure immediately with new information elements.

12.2.1.8 PS attach / abnormal cases / power off

12.2.1.8.1 Definition

12.2.1.8.2 Conformance requirement

When power is switched off before ATTACH ACCEPT message is received by the UE, the UE shall abort the PS attach procedure and perform a PS detach procedure.

Reference

```
3G-3GPP TS 24.008 clause 4.7.3.
```

12.2.1.8.3 Test purpose

To test the behaviour of the UE in case of procedure collision.

12.2.1.8.4 Method of test

Initial condition

System Simulator:

One cell operating in network operation mode IIIII.

User Equipment:

The UE has a valid P-TMSI-1, P-TMSI-1 signature and RAI-1.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode C Yes/No
UE operation mode A Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE is switched off after initiating an attach procedure. A PS detach is automatically performed by the UE before power is switched off.

Step	Direction	Message	Comments
	UE SS		
1	UE		The UE is set in UE operation mode C (see ICS). If UE operation mode C not supported,
2	UE		goto step 7. The UE is powered up or switched on and initiates an attach (see ICS).
3	->	ATTACH REQUEST	Attach type = 'PS attach' Mobile identity = P-TMSI-1 P-TMSI-1 signature
4	SS		Routing area identity = RAI-1 No response to the ATTACH REQUEST message is given by the SS.
5	UE		The UE is powered off and initiates a PS detach (with power off) by
6	->	DETACH REQUEST	Detach type = 'power switched off, PS detach'
7	SS		The SS is set in network operation mode II.
8	UE		The UE is set in UE operation mode A (see ICS) and the test is repeated from step 2 to step 6.

Specific message contents

None.

12.2.1.8.5 Test requirements

UE shall:

- initiate a PS attach procedure with information elements specified in the above Expected Sequence when UE is powered up or switched on.

When power is switched off before ATTACH ACCEPT message is received,

UE shall:

- abort the PS attach procedure and perform the PS detach procedure.

12.2.1.9 PS attach / abnormal cases / PS detach procedure collision

12.2.1.9.1 Definition

12.2.1.9.2 Conformance requirement

- 1) When a DETACH REQUEST message is received by the UE (any cause except re-attach) while waiting for an ATTACH ACCEPT message, the UE shall terminate the PS attach procedure and continue with the PS detach procedure.
- 2) When a DETACH REQUEST message is received by the UE (cause re-attach) while waiting for an ATTACH ACCEPT message, the UE shall ignore the PS detach procedure and continue with the PS attach procedure.

Reference

3G-3GPP TS 24.008 clause 4.7.3.1

12.2.1.9.3 Test purpose

To test the behaviour of the UE in case of procedure collision.

12.2.1.9.4 Method of test

Initial condition

System Simulator:

One cell operating in network operation mode II.

User Equipment:

The UE has a valid P-TMSI-1, P-TMSI-1 signature and RAI-1.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode C Yes/No
UE operation mode A Yes/No (only if mode C not supported)
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE initiates a PS attach procedure. The SS does not answer the PS attach procedure, but initiates a PS detach procedure (any cause except re-attach). The UE shall terminate the PS attach procedure and continue with the PS detach procedure.

The UE initiates a PS attach procedure. The SS does not answer the PS attach procedure, but initiates a PS detach procedure (cause re-attach). The UE shall ignore the PS detach procedure and continue with the PS attach.

Step	Direction	Message	Comments
	UE SS		
1	UE		The UE is set in UE operation mode C or B
			(see ICS).
2	UE		The UE is powered up or switched on and
_			initiates an attach (see ICS).
3	->	ATTACH REQUEST	Attach type = 'PS attach'
			Mobile identity = P-TMSI-1
			P-TMSI-1 signature
4	SS		Routing area identity = RAI-1 The SS ignores the ATTACH REQUEST
4	33		message and initiates a detach procedure.
5	<-	DETACH REQUEST	Detach type = 're-attach not required'
6	->	DETACH ACCEPT	Detach type = 1e-attach not required
7	UE	DE INTOTT NO CELL I	The UE initiates the attach procedure by MMI
	-		or AT command.
8	->	ATTACH REQUEST	Attach type = 'PS attach'
			Mobile identity = P-TMSI-1
			P-TMSI-1 signature
			Routing area identity = RAI-1
9	SS		The SS ignores the ATTACH REQUEST
			message and initiates a detach procedure.
10	<-	DETACH REQUEST	Detach type = 're-attach required'
11	UE		The UE ignores the DETACH REQUEST
			message and continue with the attach
12	<-	ATTACH ACCEPT	procedure. Attach result = 'PS only attached'
14	\-	ATTACITACCETT	Mobile identity = P-TMSI-2
			P-TMSI-2 signature
			Routing area identity = RAI-1
13	->	ATTACH COMPLETE	
14	UE		The UE is switched off or power is removed
			(see ICS).
15	->	DETACH REQUEST	Message not sent if power is removed.
			Detach type = 'power switched off, PS detach'

Specific message contents

None.

12.2.1.9.5 Test requirements

UE shall:

initiate a PS attach procedure with information elements specified in the above Expected Sequence when UE is powered up or switched on.

Case1) GMM cause is not re-attach

When a DETACH REQUEST message is received by the UE while waiting for an ATTACH ACCEPT message,

UE shall:

- terminate the PS attach procedure and continue with the PS detach procedure.

Case2) GMM cause is re-attach

When a DETACH REQUEST message is received by the UE while waiting for an ATTACH ACCEPT message,

UE shall:

- ignore the PS detach procedure and continue with the PS attach procedure.

12.2.2 Combined PS attach

12.2.2.1 Combined PS attach / PS and non-PS attach accepted

12.2.2.1.1 Definition

12.2.2.1.2 Conformance requirement

- 1) If the network accepts the combined PS attach procedure (signalled by an IMSI) and allocates a P-TMSI, the UE shall acknowledge the P-TMSI and continue communication with the P-TMSI.
- 2) If the network accepts the combined PS attach procedure (signalled by P-TMSI) and reallocates a new P-TMSI, the UE shall acknowledge the new P-TMSI and continue communication with the new P-TMSI.
- 3) If the network accepts the combined PS attach procedure (signalled by a P-TMSI) from the UE without reallocation of the previously used P-TMSI, the UE shall continue communication with the previously used P-TMSI.
- 4) If the network accepts the combined PS attach procedure and determines that IMSI shall be used in CS operations, the UE shall continue communication with the IMSI for CS operations.
- 5) If the network accepts the combined PS attach procedure and determines that a TMSI shall be used in CS operations, the UE shall continue communication with the TMSI for CS operations.

Reference

3G3GPP-TS 24.008 clause 4.7.3.2

12.2.2.1.3 Test purpose

To test the behaviour of the UE if the network accepts the PS attach procedure.

The following cases are identified:

- 1) P-TMSI / P-TMSI signature is allocated
- 2) P-TMSI / P-TMSI signature is reallocated
- 3) Old P-TMSI / P-TMSI signature is not changed
- 4) Mobile terminating CS call is allowed with IMSI
- 5) Mobile terminating CS call is not allowed with TMSI

12.2.2.1.4 Method of test

Initial condition

System Simulator:

One cell operating in network operation mode I.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

- 1) The UE sends an ATTACH REQUEST message with identity IMSI. The SS allocates a P-TMSI and returns ATTACH ACCEPT message with a P-TMSI. The UE acknowledge the P-TMSI by sending ATTACH COMPLETE message. Further communication UE SS is performed by the new P-TMSI. For CS calls, the IMSI is used.
- 2) The UE is CS paged in order to verify that the IMSI is used for CS calls.
- 3) The UE is PS paged in order to verify that the new P-TMSI is used for PS services.
- 4) The UE sends an ATTACH REQUEST message with identity P-TMSI. The SS allocates a new P-TMSI and returns ATTACH ACCEPT message with the new P-TMSI and a new TMSI. The UE acknowledge the P-TMSI and the TMSI by sending ATTACH COMPLETE message. Further communication UE SS is performed by the new P-TMSI. For CS calls, the new TMSI is used. The UE is CS paged in order to verify that the new TMSI is used for CS services.
- 5) The UE is PS paged in order to verify that the new P-TMSI is used for PS services. The UE will not answer signalling addressed to the old P-TMSI.
- 6) The UE sends an ATTACH REQUEST message with identity P-TMSI. The SS accepts the P-TMSI and returns ATTACH ACCEPT message without any P-TMSI. Further communication UE - SS is performed by the previously used P-TMSI.
- 7) The UE is PS paged in order to verify that the previously used P-TMSI is used for PS services.

	Step	Direction UE SS	Message	Comments
	1	UE		The UE is set in UE operation mode A (see ICS).
	2	UE		The UE is powered up or switched on and initiates an attach (see ICS).
	3	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach' Mobile identity =IMSI TMSI status = no valid TMSI available
	4	<-	ATTACH ACCEPT	Attach result = 'Combined PS / IMSI attached' Mobile identity = P-TMSI-1 P-TMSI-1 signature Mobile identity =IMSI Routing area identity = RAI-1
	5	->	ATTACH COMPLETE	
	6	<-	PAGING REQUEST TYPE	Mobile identity = IMSI
	7	->	4 <u>PAGING TYPE1</u> CHANNEL REQUEST RRC CONNECTION REQUEST	Paging order is for RRC-connection.
	8	<-	IMMEDIATE ASSIGNMENT RRC CONNECTION SETUP	
	<u>9</u>	<u>-></u>	RRC CONNECTION SETUP COMPLETE	
	9 10	->	PAGING RESPONSE	Mobile identity = IMSI
	1011 12	<- ->	CHANNEL RELEASE RRC CONNECTION RELEASE RRC CONNECTION RELEASE	After sending of this message, the SS waits for disconnection of the CS signalling link.
			COMPLETE	
	44 <u>13</u>	<-	PACKET PAGING REQUESTPAGING TYPE1	Mobile identity = P-TMSI-1 Paging order is for TBF establishment. Comment: A TBF will be established on lower
	12 14	->	UPLINK RLC DATA BLOCKSERVICE REQUEST	layers. LLC PDU implicitly indicating paging response.service type = "paging response" Comment: The TBF will be released on lower
	13 <u>15</u>	UE		Hayers. The UE is switched off or power is removed
	44 <u>16</u>	->	DETACH REQUEST	(see ICS). Message not sent if power is removed. Detach type = 'power switched off, combined PS / IMSI detach'
	15 17	UE		The UE is powered up or switched on and
1	16 18	->	ATTACH REQUEST	initiates an attach (see ICS). Attach type = 'Combined PS / IMSI attach' Mobile identity = P-TMSI-1 P-TMSI-1 signature
1	17 <u>19</u>	<-	ATTACH ACCEPT	TMSI status = no valid TMSI available Routing area identity = RAI-1 Attach result = 'Combined PS / IMSI attached' Mobile identity = P-TMSI-2 P-TMSI-2 signature Mobile identity = TMSI-1
1	18 20	->	ATTACH COMPLETE	Routing area identity = RAI-1
	19 21 19b 21b	<- ->	GMM INFORMATION GMM STATUS	Message sent with P-TMSI-2 Message sent in case the UE does not support reception of GMM information message Cause #97
	20 22	<-	PAGING REQUEST TYPE 4PAGING TYPE1	Mobile identity = TMSI-1 Paging order is for RRC-connection.
	21 23	->	CHANNEL REQUEST RRC	
	22 24	<-	CONNECTION REQUEST IMMEDIATE ASSIGNMENT RRC CONNECTION SETUP	

		Ī.,		
	<u>25</u>	<u>-></u>	RRC CONNECTION SETUP COMPLETE	
	23 26	->	PAGING RESPONSE	Mobile identity = TMSI-1
	24 27	<-	CHANNEL RELEASE RRC	After sending of this message, the SS waits for
			CONNECTION RELEASE	disconnection of the CS signalling link.
	28	<u>-></u>	RRC CONNECTION RELEASE	ů ů
	_	_	COMPLETE	
	25 29	<-	PACKET PAGING	Mobile identity = P-TMSI-2
			REQUESTPAGING TYPE1	Paging order is for TBF establishment.
	26 30	->	UPLINK RLC DATA	LLC PDU implicitly indicating paging response.
			BLOCKSERVICE REQUEST	service type = "paging response"
	27 31	<-	PACKET PAGING	Mobile identity = P-TMSI-1
			REQUESTPAGING TYPE1	Paging order is for TBF establishment.
	28 32	UE		No response from the UE to the request. This
				is checked for 10 seconds.
	29 33	UE		The UE is switched off or power is removed
				(see ICS).
	30 34	->	DETACH REQUEST	Message not sent if power is removed.
				Detach type = 'power switched off, combined
				PS / IMSI detach'
	31 35	UE		The UE is powered up or switched on and
ı			ATTA OLI DE OLIFOT	initiates an attach (see ICS).
1	32 36	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach'
				Mobile identity = P-TMSI-2
				P-TMSI-2 signature
				Routing area identity = RAI-1
1	33 37	_	ATTACH ACCEPT	TMSI status = valid TMSI available
I	33 31	<-	ATTACH ACCEPT	No new mobile identity assigned. TMSI and P-TMSI not included.
				Attach result = 'Combined PS / IMSI attached'
				P-TMSI-3 signature
				Routing area identity = RAI-1
	34 38	<-	PACKET PAGING	Mobile identity = P-TMSI-2
	0 1 <u>00</u>		REQUESTPAGING TYPE1	Paging order is for TBF establishment.
	35 39	->	UPLINK RLC DATA	LLC PDU implicitly indicating paging
	55 <u>55</u>		BLOCKSERVICE REQUEST	response.service type = "paging response"
	36 40	UE		The UE is switched off or power is removed
•		_		(see ICS).
	37 41	->	DETACH REQUEST	Message not sent if power is removed.
٠				Detach type = 'power switched off, combined
				PS / IMSI detach'
. '			•	

Specific message contents

None.

12.2.2.1.5 Test requirements

UE shall:

- initiate a PS attach procedure with information elements specified in the above Expected Sequence when UE is powered up or switched on.

Case 1) SS accept the combined PS attach procedure (signalled by an IMSI) and allocates a P-TMSI.

UE shall

- acknowledge the P-TMSI and continue communication with the P-TMSI.

Case 2) SS accepts the combined PS attach procedure (signalled by P-TMSI) and reallocates a new P-TMSI.

UE shall:

- acknowledge the new P-TMSI and continue communication with the new P-TMSI.

Case 3) SS accepts the combined PS attach procedure (signalled by a P-TMSI) from the UE without reallocation of the previously used P-TMSI.

UE shall:

- continue communication with the previously used P-TMSI.

Case 4) SS accepts the combined PS attach procedure and determines that IMSI shall be used in CS operations.

UE shall:

- continue communication with the IMSI for CS operations.

Case 5) SS accepts the combined PS attach procedure and determines that a TMSI shall be used in CS operations.

UE shall:

- continue communication with the TMSI for CS operations.

12.2.2.2 Combined PS attach / PS only attach accepted

12.2.2.2.1 Definition

12.2.2.2.2 Conformance requirement

- 1) If the network accepts the combined PS attach procedure, but GMM cause code 'IMSI unknown in HLR' is sent to the UE the User Equipment shall delete the stored TMSI, LAI and CKSN. The User Equipment shall consider USIM invalid for non-PS services until power is switched off or USIM is removed.
- 2) If the network accepts the combined PS attach procedure, but GMM cause code 'MSC temporarily not reachable', 'Network failure' or 'Congestion' is sent to the UE, an UE operation mode A UE may perform an MM IMSI attach procedure.

Reference

3G3GPP-TS 24.008 clause 4.7.3.2

12.2.2.2.3

Test propose1

To test the behaviour of the UE if the network accepts the PS attach procedure with indication PS only, GMM cause 'IMSI unknown in HLR'.

Test porpose2

To test the behaviour of the UE if the network accepts the PS attach procedure with indication PS only, GMM cause 'MSC temporarily not reachable', 'Network failure' or 'Congestion'.

12.2.2.2.4 Method of test

12.2.2.4.1 Test porpose1

Initial condition

System Simulator:

One cell operating in network operation mode I.

Test purpose

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE sends an ATTACH REQUEST message with identity IMSI. The SS allocates a P-TMSI and returns ATTACH ACCEPT message with a P-TMSI. GMM cause 'IMSI unknown in HLR' is indicated from SS. Further communication UE - SS is performed by the P-TMSI. CS services are not possible.

Expected Sequence

ſ	Step	Direction	Message	Comments
		UE SS		
	1	UE		The UE is set in UE operation mode A.
	2	UE		The UE is powered up or switched on and initiates an attach (see ICS).
	З	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach' Mobile identity =IMSI
	4	< -	ATTACH ACCEPT	TMSI status = no valid TMSI available Attach result = 'PS only attached' Mobile identity = P-TMSI-1 P-TMSI-1 signature GMM cause = 'IMSI unknown in HLR'
	5	->	ATTACH COMPLETE	
	6	<-	PACKET PAGING REQUESTPAGING TYPE1	Mobile identity = IMSI Paging order is for RRC-connection.
I	7	UE		The UE shall not initiate an RRC connection. This is checked during 3 seconds.
	8	UE		The UE is switched off or power is removed
	9	->	DETACH REQUEST	(see ICS). Message not sent if power is removed. Detach type = 'power switched off, PS detach'

12.2.2.4.2 Test porpose2

Initial condition

System Simulator:

One cell operating in network operation mode I.

User Equipment:

The UE has a valid TMSI, P-TMSI, P-TMSI signature and RAI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
Automatic MM IMSI attach procedure for UE operation mode A UE Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE sends an ATTACH REQUEST message. The SS allocates a P-TMSI and returns ATTACH ACCEPT message with a P-TMSI. GMM cause 'MSC temporarily not reachable', 'Network failure' or 'Congestion' is indicated from SS. The cause code is arbitrarily chosen. This procedure is repeated four times. An UE operation mode A UE may then perform an MM IMSI attach procedure (according to the ICS statement). Further communication UE - SS is performed by the P-TMSI. The existence of a signalling channel is verified by a request for mobile identity. CS services are not possible as an IMSI attach procedure is not performed.

Expected Sequence

Dependent whether the option 'Automatic MM IMSI attach procedure for UE operation mode A UE' is supported or not, the steps 1-22 or 23-53 apply depending on manufacturer (see ICS).

Step	Direction UE SS	Message	Comments
1	UE		The UE is set in UE operation mode A and no automatic MM IMSI attach procedure is indicated (see ICS).
2	UE		The UE is powered up or switched on and initiates an attach (see ICS).
3	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach' Mobile identity =P-TMSI-1 P-TMSI-1 signature
4	<-	ATTACH ACCEPT	Routing area identity = RAI-1 TMSI status = valid TMSI available Attach result = 'PS only attached' Mobile identity = P-TMSI-2 P-TMSI-2 signature
5		ATTACH COMPLETE	Routing area identity = RAI-1 GMM cause = 'MSC temporarily not reachable', 'Network failure' or 'Congestion' (arbitrarily chosen)
5 7	-> ->	ROUTING AREA UPDATE REQUEST	Update type = 'Combined RA / LA updating with IMSI attach' P-TMSI-2 signature Routing area identity = RAI-1
8	<-	ROUTING AREA UPDATE ACCEPT	TMSI status = no valid TMSI available No new mobile identity assigned. P-TMSI not included. Update result = 'RA updated' P-TMSI-3 signature
			Routing area identity = RAI-1 GMM cause = 'MSC temporarily not reachable', 'Network failure' or 'Congestion' (arbitrarily chosen)
10	->	ROUTING AREA UPDATE REQUEST	Update type = 'Combined RA / LA updating with IMSI attach' P-TMSI-3 signature Routing area identity = RAI-1
11	<-	ROUTING AREA UPDATE ACCEPT	TMSI status = no valid TMSI available No new mobile identity assigned. P-TMSI not included. Update result = 'RA updated' P-TMSI-4 signature
			Routing area identity = RAI-1 GMM cause = 'MSC temporarily not reachable', 'Network failure' or 'Congestion' (arbitrarily chosen)
12	SS		The SS verifies that the time between the requests are T3311
13	->	ROUTING AREA UPDATE REQUEST	Update type = 'Combined RA / LA updating with IMSI attach' P-TMSI-4 signature
14	<-	ROUTING AREA UPDATE ACCEPT	Routing area identity = RAI-1 TMSI status = no valid TMSI available No new mobile identity assigned. P-TMSI not included. Update result = 'RA updated'
		POLITING ASSAURS: T	P-TMSI-5 signature Routing area identity = RAI-1 GMM cause = 'MSC temporarily not reachable', 'Network failure' or 'Congestion' (arbitrarily chosen)
16	->	ROUTING AREA UPDATE REQUEST	Update type = 'Combined RA / LA updating with IMSI attach' P-TMSI-5 signature Routing area identity = RAI-1 TMSI status = no valid TMSI available

	17	<-	ROUTING AREA UPDATE ACCEPT	No new mobile identity assigned. P-TMSI not included. Update result = 'RA updated' P-TMSI-6 signature Routing area identity = RAI-1 GMM cause = 'MSC temporarily not reachable', 'Network failure' or 'Congestion' (arbitrarily
ĺ	19	<-	PACKET PAGING	chosen) Mobile identity = IMSI
į	20	UE	REQUESTPAGING TYPE1	Paging order is for RRC-connection. The UE shall not initiate an RRC connection.
	21	UE		This is checked during 3 seconds. The UE is switched off or power is removed
	22	->	DETACH REQUEST	(see ICS). Message not sent if power is removed. Detach type = 'power switched off, PS detach'. Stop the sequence.
-	23	UE		Automatic MM IMSI attach procedure is
	24	UE		indicated (see ICS). The UE is powered up or switched on and
	25	->	ATTACH REQUEST	initiates an attach (see ICS). Attach type = 'Combined PS / IMSI attach' Mobile identity = P-TMSI-1
	26	<-	ATTACH ACCEPT	P-TMSI-1 signature Routing area identity = RAI-1 TMSI status = valid TMSI available No new mobile identity assigned. P-TMSI not included. Attach result = 'PS only attached' P-TMSI-2 signature Routing area identity = RAI-1 GMM cause = 'MSC temporarily not reachable',
	28	->	ROUTING AREA UPDATE REQUEST	'Network failure' or 'Congestion' (arbitrarily chosen) Update type = 'Combined RA / LA updating with IMSI attach' P-TMSI-2 signature Routing area identity = RAI-1
	29	<-	ROUTING AREA UPDATE ACCEPT	TMSI status = no valid TMSI available No new mobile identity assigned. P-TMSI not included. Update result = 'RA updated' P-TMSI-3 signature Routing area identity = RAI-1 GMM cause = 'MSC temporarily not reachable',
	31	->	ROUTING AREA UPDATE REQUEST	'Network failure' or 'Congestion' (arbitrarily chosen) Update type = 'Combined RA / LA updating with IMSI attach' P-TMSI-3 signature Routing area identity = RAI-1
	32	<-	ROUTING AREA UPDATE ACCEPT	TMSI status = no valid TMSI available No new mobile identity assigned. P-TMSI not included. Update result = 'RA updated' P-TMSI-4 signature Routing area identity = RAI-1
	33	SS		GMM cause = 'MSC temporarily not reachable', 'Network failure' or 'Congestion' (arbitrarily chosen) The SS verifies that the time between the
	34	->	ROUTING AREA UPDATE REQUEST	requests are T3311 Update type = 'Combined RA / LA updating with IMSI attach' P-TMSI-4 signature Routing area identity = RAI-1 TMSI status = no valid TMSI available

35	<-	ROUTING AREA UPDATE ACCEPT	No new mobile identity assigned. P-TMSI not included. Update result = 'RA updated' P-TMSI-5 signature Routing area identity = RAI-1 GMM cause = 'MSC temporarily not reachable', 'Network failure' or 'Congestion' (arbitrarily
37	->	ROUTING AREA UPDATE REQUEST	chosen) Update type = 'Combined RA / LA updating with IMSI attach' P-TMSI-5 signature Routing area identity = RAI-1
38	<-	ROUTING AREA UPDATE ACCEPT	TMSI status = no valid TMSI available No new mobile identity assigned. P No new mobile identity assigned. P-TMSI not included. Update result = 'RA updated' P-TMSI-6 signature Routing area identity = RAI-1 GMM cause = 'MSC temporarily not reachable', 'Network failure' or 'Congestion' (arbitrarily chosen)
39	SS		The SS verifies that the time between the requests are T3311
40	UE		An automatic MM IMSI attach procedure is initiated.
41	->	CHANNEL REQUEST RRC CONNECTION REQUEST	
42	<-	IMMEDIATE ASSIGNMENT RRC CONNECTION SETUP	
<u>43</u>	<u>-></u>	RRC CONNECTION SETUP COMPLETE	
43 <u>44</u>	->	LOCATION UPDATING REQ	Location updating type = IMSI attach.
44 45	<-	LOCATION UPDATING ACC	The SS allocates a new TMSI.
45 46	->	TMSI REALLOCATION COMP	Location updating type = IMSI attach.
46 <u>47</u>	<-	CHANNEL RELEASE RRC CONNECTION RELEASE	After sending of this message, the SS waits for disconnection of the CS signalling link.
<u>48</u>	<u>-></u>	RRC CONNECTION RELEASE COMPLETE	
47 <u>49</u>	<-	PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = TMSI-1 Paging order is for RRC-connection.
48 <u>50</u>	->	CHANNEL REQUEST RRC CONNECTION REQUEST	
49 <u>51</u>	<-	IMMEDIATE ASSIGNMENT RRC CONNECTION SETUP	
<u>52</u>	<u>-></u>	RRC CONNECTION SETUP COMPLETE	
50 53	->	PAGING RESPONSE	Mobile identity = TMSI-1
51 <u>54</u>	<-	CHANNEL RELEASE RRC CONNECTION RELEASE	After sending of this message, the SS waits for disconnection of the CS signalling link.
<u>55</u>	<u>-></u>	RRC CONNECTION RELEASE COMPLETE	
52 <u>56</u>	UE		The UE is switched off or power is removed (see ICS).
53 <u>57</u>	->	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, PS detach'

Specific message contents

None.

12.2.2.5 Test requirements

UE shall:

- initiate a Combined PS attach procedure with information elements specified in the above Expected Sequence when UE is powered up or switched on.
- check the GMM cause which is contained in the ATTACH ACCEPT message received from SS.
- perform the following actions depending on the GMM cause.

Case1) GMM cause = 'IMSI unknown in HLR'

UE shall

- delete stored TMSI, LAI, and ciphering key sequence number.
- consider USIM invalid for non-PS service until power is switching off or USIM is removed.

Case2) GMM cause = 'MSC temporarily not reachable' or 'Network failure' or 'Congestion',

UE shall:

- stop the timer T3310(if running), and shall increment the routing area update attempt counter.
- perform the following actions depending on the conditions described below.

Case 2-1) the routing area updating attempt counter is less than 5 and the stored RAI is equal to the RAI of the current serving cell and the GMM update status is equal to GU1 UPDATED:

UE shall

- keep the GMM update status GU1 UPDATED.
- change state to GMM-REGISTERED.ATTEMPTING-TO-UPDATE-MM.
- start timer T3311. When timer T3311 expires, the combined routing area update procedure indicating "combined RA/LA updating with IMSI attach" is triggered again.

Case 2-2) the routing area updating attempt counter is greater than or equal to 5

UE shall

- start timer T3302 and change state to GMM-REGISTERED.ATTEMPTING-TO-UPDATE-MM

12.2.2.3 Combined PS attach / PS attach while IMSI attach

12.2.2.3.1 Definition

12.2.2.3.2 Conformance requirement

If the PS UE is already attached for non-PS services by the MM specific attach procedure, but wants to perform an attach for PS services, the combined PS attach procedure is performed.

Reference

3G-3GPP TS 24.008 clause 4.7.3.2

12.2.2.3.3 Test purpose

To test the behaviour of the UE if PS attach performed while IMSI attached.

12.2.2.3.4 Method of test

Initial condition

System Simulator:

One cell operating in network operation mode I. ATT flag is set.

User Equipment:

The UE has a valid TMSI-1, P-TMSI-1, P-TMSI signature and RAI-1.

Related ICS/IXIT statements

Support of PS service Yes/No

UE operation mode A Yes/No Switch off on button Yes/No Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE attach for non-PS services. The UE does not answer to paging orders for PS. The UE attach for PS services. Paging orders for PS are answered.

	Step	Direction	Message	Comments
		UE SS		
	1	ÜE		The UE is set in UE operation mode C (see ICS) and configured not to perform a PS attach.
	2	UE		The UE is powered up or switched on. No PS attach is performed (see ICS).
	3	->	CHANNEL REQUEST RRC CONNECTION REQUEST	
	4	<-	IMMEDIATE ASSIGNMENT RRC CONNECTION SETUP	
	<u>5</u>	<u>-></u>	RRC CONNECTION SETUP COMPLETE	
	5 6	->	LOCATION UPDATING REQ	Location updating type = IMSI attach.
	6 7	<-	LOCATION UPDATING ACC	The SS allocates a new TMSI.
	7 <u>8</u> 8 <u>9</u>	->	TMSI REALLOCATION COMP	Location updating type = IMSI attach.
		<-	CHANNEL RELEASE RRC CONNECTION RELEASE	After sending of this message, the SS waits for disconnection of the CS signalling link.
	<u>10</u>	<u>-></u>	RRC CONNECTION RELEASE COMPLETE	
	9 11	<-	PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = P-TMSI-1 Paging order is for TBF establishment.
	10 12	UE		No response from the UE to the request. This is checked for 10 seconds.
	44 <u>13</u>	UE		The UE is triggered to perform a PS attach (in combination with IMSI attach).
I	12 <u>14</u>	->	ATTACH REQUEST	Attach type = 'PS attach while IMSI attached' Mobile identity =P-TMSI-1
	13 15	<-	ATTACH ACCEPT	P-TMSI-1 signature Mobile identity = TMSI-1 Routing area identity = RAI-1 TMSI status = valid TMSI available Attach result = 'Combined PS / IMSI attached' No new mobile identity assigned. TMSI and P- TMSI not included P-TMSI-2 signature Routing area identity = RAI-1
	14 16	->	ATTACH COMPLETE	
	15 17	<-	PACKET PAGING REQUESTPAGING TYPE1	Mobile identity = P-TMSI-1 Paging order is for TBF establishment.
	16 18	->	UPLINK RLC DATA BLOCKSERVICE REQUEST	LLC PDU implicitly indicating paging response.service type = "paging response"
1	17 <u>19</u>	UE		The UE is switched off or power is removed (see ICS).
	18 <u>20</u>	->	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, combined PS / IMSI detach'

Specific message contents

None.

12.2.2.3.5 Test requirements

UE is already attached for non-PS service with the MM specific attach procedure.

UE shall:

- perform the combined PS attach procedure when UE want to attach for PS service.

12.2.2.4 Combined PS attach / rejected / IMSI invalid / illegal ME

12.2.2.4.1 Definition

12.2.2.4.2 Conformance requirement

- 1) If the network rejects a combined PS attach procedure from the User Equipment with the cause 'Illegal ME', the User Equipment shall consider USIM invalid for PS and non-PS services until power is switched off or USIM is removed.
- 2) If the network rejects a combined PS attach procedure from the User Equipment with the cause 'Illegal ME', the User Equipment shall delete the stored TMSI, LAI, CSKN, RAI, PS-CKSN, P-TMSI and P-TMSI signature.

Reference

3G3GPP-TS 24.008 clause 4.7.3.2

12.2.2.4.3 Test purpose

To test the behaviour of the UE if the network rejects the combined PS attach procedure of the UE with the cause 'Illegal ME'.

12.2.2.4.4 Method of test

Initial condition

System Simulator:

Three cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2, cell C in MCC2/MNC1/LAC1/RAC1.

All three cells are operating in network operation mode I.

User Equipment:

The UE has a valid TMSI, P-TMSI, P-TMSI signature and RAI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
USIM removal possible without powering down Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The SS rejects a PS attach with the cause value 'Illegal ME'. The SS checks that the UE does not perform PS attach in the same or another PLMN. CS services are not possible as the USIM is blocked for CS services. PS services are not possible as the USIM is blocked for PS services.

Ī	Step	Direction UE SS	Message	Comments
-	1 2	SS UE		The following messages are sent and shall be received on cell A. The SS activates cell A. The UE is set in UE operation mode A (see
	3	UE		ICS). The UE is powered up or switched on and initiates an attach (see ICS). Cell A is preferred
	4	->	ATTACH REQUEST	by the UE. Attach type = 'Combined PS / IMSI attach' Mobile identity = P-TMSI-1 P-TMSI-1 signature
	5 6	<- <-	ATTACH REJECT PAGING TYPE1PACKET PAGING REQUEST	Routing area identity = RAI-1 TMSI status = valid TMSI available GMM cause 'Illegal ME'. Mobile identity = IMSI Paging order is for RRC-connection.
1	7	UE	THOMS REGIST	The UE shall not initiate an RRC connection.
Ì	8	<-	PAGING TYPE1PACKET PAGING REQUEST	This is checked during 3 seconds. Mobile identity = P-TMSI-1 Paging order is for TBF establishment.
J	9	UE	THE REGISTRESS OF THE PROPERTY	No response from the UE to the request. This is checked for 10 seconds.
	10 11 12	SS UE UE		The following messages are sent and shall be received on cell B. The SS deactivates cell A and activates cell B. Cell B is preferred by the UE. No ATTACH REQUEST sent to the SS (SS waits 30 seconds).
	13	<-	PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = IMSI
ļ	14	UE	PAGING REQUEST	Paging order is for RRC-connection. The UE shall not initiate an RRC connection. This is checked during 3 seconds.
•	15 16 17	SS UE UE		The following messages are sent and shall be received on cell C. The SS deactivates cell B and activates cell C. Cell C is preferred by the UE. No ATTACH REQUEST sent to the SS
	18	<-	PAGING TYPE1PACKET PAGING REQUEST	(SS waits 30 seconds). Mobile identity = IMSI Paging order is for TBF establishment.
ı	19	UE	THOMAS NEGOEST	No response from the UE to the request. This
	20	UE		is checked for 10seconds. If possible (see ICS) USIM removal is performed. Otherwise if possible (see ICS) switch off is performed. Otherwise the power is removed.
•	21	UE		The UE gets the USIM replaced, is powered up or switched on and initiates an attach (see
	22	->	ATTACH REQUEST	ICS). Attach type = 'Combined PS / IMSI attach' Mobile identity = IMSI
	23	<-	ATTACH ACCEPT	TMSI status = no valid TMSI available Attach result = 'Combined PS / IMSI attached' Mobile identity = P-TMSI-1 P-TMSI-1 signature Mobile identity = TMSI-1 Routing area identity = RAI-2
	24 25	-> <-	ATTACH COMPLETE PAGING TYPE1PACKET	Mobile identity = TMSI-1
	26	->	PAGING REQUEST CHANNEL REQUEST RRC CONNECTION REQUEST	Paging order is for RRC-connection.

27	<-	IMMEDIATE ASSIGNMENT RRC	
<u>28</u>	<u>-></u>	CONNECTION SETUP RRC CONNECTION SETUP	
		COMPLETE	
28 29	->	PAGING RESPONSE	Mobile identity = TMSI-1
29 30	<-	CHANNEL RELEASE RRC	After sending of this message, the SS waits for
		CONNECTION RELEASE	disconnection of the CS signalling link.
<u>31</u>	<u>-></u>	RRC CONNECTION RELEASE	
		COMPLETE	
30 32	UE		The UE is switched off or power is removed
			(see ICS).
31 33	->	DETACH REQUEST	Message not sent if power is removed.
			Detach type = 'power switched off, combined
			PS / IMSI detach'

Specific message contents

None.

12.2.2.4.5 Test requirements

UE shall:

- initiate a combined PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.
- check the GMM cause which is contained in the ATTACH REJECT message and stop the timer T3310.
- set the PS update status to GU3 ROAMING NOT ALLOWED.
- delete the stored P-TMSI, P-TMSI signature, RAI and PS ciphering key sequence number.
- consider USIM invalid for PS and non-PS services until power is switched off or USIM is removed.

12.2.2.5 Combined PS attach / rejected / PS services and non-PS services not allowed

12.2.2.5.1 Definition

12.2.2.5.2 Conformance requirement

- 1) If the network rejects a combined PS attach procedure from the User Equipment with the cause 'PS services and non-PS services not allowed', the User Equipment shall consider USIM invalid for PS and non-PS services until power is switched off or USIM is removed.
- 2) If the network rejects a combined PS attach procedure from the User Equipment with the cause 'PS services and non-PS services not allowed', the User Equipment shall delete the stored TMSI, LAI, CSKN, RAI, PS-CKSN, P-TMSI and P-TMSI signature.

Reference

3G-3GPP TS 24.008 clause 4.7.3.2

12.2.2.5.3 Test purpose

To test the behaviour of the UE if the network rejects the combined PS attach procedure of the UE with the cause 'PS services and non-PS services not allowed'.

12.2.2.5.4 Method of test

Initial condition

System Simulator:

Two cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1 and cell B in MCC2/MNC1/LAC1/RAC1.

1011

Both cells are operating in network operation mode I.

User Equipment:

The UE has a valid TMSI, P-TMSI, P-TMSI signature and RAI.

Related ICS/IXIT statements

Support of PS service Yes/No UE operation mode A Yes/No Switch off on button Yes/No Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The SS rejects a PS attach with the cause value 'PS services and non-PS services not allowed'. The SS checks that the UE does not perform PS attach in the same or another PLMN. CS services are not possible as the USIM is blocked for CS services. PS services are not possible as the USIM is blocked for PS services.

	Step	Direction UE SS	Message	Comments
		02 00		The following messages are sent and shall be
	1	SS		received on cell A. The SS activates cell A.
	2	UE		The UE is set in UE operation mode A (see
				ICS).
	3	UE		The UE is powered up or switched on and initiates an attach (see ICS). Cell A is preferred
				by the UE.
	4	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach'
				Mobile identity =P-TMSI-1 P-TMSI-1 signature
				Routing area identity = RAI-1
	_			TMSI status = valid TMSI available
	5	<-	ATTACH REJECT	GMM cause 'PS services and non-PS services not allowed'
	6	UE		No LOCATION UPDATING REQ with type
				'IMSI attach' is sent to the SS
ı	7	<-	PAGING TYPE1 PACKET	(SS waits 30 seconds). Mobile identity = IMSI
	•		PAGING REQUEST	Paging order is for RRC-connection.
	8	UE		The UE shall not initiate an RRC connection.
ı	9	<-	PAGING TYPE1 PACKET	This is checked during 3 seconds. Mobile identity = P-TMSI-1
		•	PAGING REQUEST	Paging order is for TBF establishment.
	10	->		No response from the UE to the request. This is checked for 10 seconds
	11	UE		Cell B is preferred by the UE.
	12	UE		No ATTACH REQUEST sent to the SS
	13	UE		(SS waits 30 seconds). No LOCATION UPDATING REQ with type
	10	02		'IMSI attach' is sent to the SS
ı	14		DACING TYPE 1 DACKET	(SS waits 30 seconds).
	14	<-	PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = IMSI Paging order is for RRC-connection.
·	15	UE		The UE shall not initiate an RRC connection.
ı	16	<-	PAGING TYPE1 PACKET	This is checked during 3 seconds. Mobile identity = TMSI-1
	10		PAGING REQUEST	Paging order is for TBF establishment.
	17	UE		No response from the UE to the request. This is checked for 10seconds.
	18	UE		If possible (see ICS) switch off is performed.
				Otherwise the power is removed.
	19	UE		The UE is powered up or switched on and initiates an attach (see ICS).
	20	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach'
				Mobile identity = IMSI TMSI status = no valid TMSI available
	21	<-	ATTACH ACCEPT	Attach result = 'Combined PS / IMSI attached'
				Mobile identity = P-TMSI-1
				P-TMSI-1 signature Mobile identity = TMSI-1
				Routing area identity = RAI-2
	22	->	ATTACH COMPLETE	
	23	<-	PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = TMSI-1 Paging order is for RRC-connection.
	24	->	CHANNEL REQUEST RRC	3 5
	25	/ -	CONNECTION REQUEST IMMEDIATE ASSIGNMENT RRC	
	20	<-	CONNECTION SETUP	
	<u>26</u>	<u>-></u>	RRC CONNECTION SETUP	
	26 27	->	COMPLETE PAGING RESPONSE	Mobile identity = TMSI-1

27 28	<-	CHANNEL RELEASE RRC CONNECTION RELEASE	After sending of this message, the SS waits for
<u>29</u>	<u>-></u>	RRC CONNECTION RELEASE	disconnection of the CS signalling link.
28 30	<-	COMPLETE PACKET PAGING	Mobile identity = TMSI-1
		REQUESTPAGING TYPE1	Paging order is for TBF establishment.
29 31	->	UPLINK RLC DATA BLOCKSERVICE REQUEST	LLC PDU implicitly indicating paging response. Service type = "paging response"
30 <u>32</u>	UE	SECON <u>GERMICE REGIST</u>	The UE is switched off or power is removed (see ICS).
31 33	->	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, combined
			PS / IMSI detach'

Specific message contents

None.

12.2.2.5.5 Test requirements

UE shall:

- initiate a combined PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.
- check the GMM cause which is contained in the ATTACH REJECT message and stop the timer T3310.
- set the PS update status to GU3 ROAMING NOT ALLOWED.
- delete the stored P-TMSI, P-TMSI signature, RAI and PS ciphering key sequence number.
- consider USIM invalid for PS and non-PS services until power is switched off or USIM is removed.

12.2.2.6 Combined PS attach / rejected / PS services not allowed

12.2.2.6.1 Definition

12.2.2.6.2 Conformance requirement

- 1) If the network rejects a combined PS attach procedure from the User Equipment with the cause 'PS services not allowed', the User Equipment shall consider USIM invalid for PS services until power is switched off or USIM is removed.
- 2) If the network rejects a combined PS attach procedure from the User Equipment with the cause 'PS services not allowed' the User Equipment shall delete the stored RAI, PS-CKSN, P-TMSI and P-TMSI signature.
- 3) A PS class AUE shall perform an MM IMSI attach procedure.

Reference

3G3GPP-TS 24.008 clause 4.7.3.2

12.2.2.6.3 Test purpose

To test the behaviour of the UE if the network rejects the PS attach procedure of the UE with the cause 'PS services not allowed'.

12.2.2.6.4 Method of test

Initial condition

System Simulator:

Two cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1 and cell B in MCC2/MNC1/LAC1/RAC1.

Both cells are operating in network operation mode I.

User Equipment:

The UE has a valid TMSI, P-TMSI, P-TMSI signature and RAI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The SS rejects a normal attach with the cause value 'PS services not allowed'. The SS checks that the UE does not perform PS attach. PS services are not possible. An UE operation mode A UE shall perform an MM IMSI attach.

Step	Direction UE SS	Message	Comments
1 2 3	SS UE ->	ATTACH REQUEST	The following messages are sent and shall be received on cell A. The SS activates cell A. The UE is powered up or switched on and initiates an attach (see ICS). Attach type = 'Combined PS / IMSI attach' Mobile identity =P-TMSI-1 P-TMSI-1 signature Routing area identity = RAI-2 TMSI status = valid TMSI available
4 5	<- UE	ATTACH REJECT	GMM cause 'PS services not allowed' An automatic MM IMSI attach procedure is initiated.
6	->	CHANNEL REQUEST RRC CONNECTION REQUEST	
7	<-	IMMEDIATE ASSIGNMENT RRC CONNECTION SETUP	
<u>8</u>	<u>-></u>	RRC CONNECTION SETUP COMPLETE	
89 910 911 4112	-> <- -> <-	LOCATION UPDATING REQ LOCATION UPDATING ACC TMSI REALLOCATION COMP CHANNEL RELEASE RRC CONNECTION RELEASE RRC CONNECTION RELEASE	Location updating type = IMSI attach. The SS allocates TMSI-2. Location updating type = IMSI attach. After sending of this message, the SS waits for disconnection of the CS signalling link.
12 14	<-	COMPLETE PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = TMSI-2 Paging order is for RRC-connection.
13 15	->	CHANNEL REQUEST RRC CONNECTION REQUEST	raging order is for KKC-connection.
14 16	<-	IMMEDIATE ASSIGNMENT RRC CONNECTION SETUP	
<u>17</u>	<u>-></u>	RRC CONNECTION SETUP COMPLETE	
15 18 16 19	-> <-	PAGING RESPONSE CHANNEL RELEASE RRC CONNECTION RELEASE	Mobile identity = TMSI-2 After sending of this message, the SS waits for disconnection of the CS signalling ignaling link.
<u>20</u>	<u>-></u>	RRC CONNECTION RELEASE COMPLETE	
1721 1822 1923 2024	SS UE UE ->	CHANNEL REQUEST RRC CONNECTION REQUEST	The following messages are sent and shall be received on cell B. The SS deactivates cell A and activates cell B. Cell B is preferred by the UE. A location updating procedure is initiated.
21 25	<-	IMMEDIATE ASSIGNMENT RRC CONNECTION SETUP	
<u>26</u>	<u>-></u>	RRC CONNECTION SETUP COMPLETE	
2227 2328 2429 2530	-> <- -> <-	LOCATION UPDATING REQ LOCATION UPDATING ACC TMSI REALLOCATION COMP CHANNEL RELEASE RRC	Location updating type = normal. The SS allocates TMSI-1. After sending of this message, the SS waits for
<u>31</u>	<u>-></u>	CONNECTION RELEASE RRC CONNECTION RELEASE	disconnection of the CS signalling link.
26 32	<-	COMPLETE PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = TMSI-1 Paging order is for RRC-connection.
27 33	->	CHANNEL REQUEST RRC CONNECTION REQUEST	raging order is for KKC-connection.

	28 <u>34</u>	<-	IMMEDIATE ASSIGNMENT RRC	
	<u>35</u>	<u>-></u>	CONNECTION SETUP RRC CONNECTION SETUP	
	29 36	->	COMPLETE PAGING RESPONSE	Mobile identity = TMSI-1
	30 37	<-	CHANNEL RELEASE RRC CONNECTION RELEASE	After sending of this message, the SS waits for disconnection of the CS signalling link.
	<u>38</u>	<u>-></u>	RRC CONNECTION RELEASE COMPLETE	also medicinor are de signalling link.
	31 39	<-	PAGING TYPE1PACKET	Mobile identity = TMSI-1
	<u>3240</u>	UE	PAGING REQUEST	Paging order is for TBF establishment. No response from the UE to the request. This
1	33 <u>41</u>	UE		is checked for 10seconds. If possible (see ICS) switch off is performed. Otherwise the power is removed.
	<u>3442</u>	UE		The UE is powered up or switched on and
1	<u>3543</u>	->	ATTACH REQUEST	initiates an attach (see ICS). Attach type = 'Combined PS / IMSI attach' Mobile identity = IMSI
1	36<u>44</u>	<-	ATTACH ACCEPT	TMSI status = no valid TMSI available Attach result = 'Combined PS / IMSI attached' Mobile identity = P-TMSI-1 P-TMSI-1 signature Mobile identity = TMSI-2
ı	2745	_	ATTACH COMPLETE	Routing area identity = RAI-2
	37<u>45</u> 38<u>46</u>	-> <-	ATTACH COMPLETE PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = TMSI-2
	39<u>47</u>	->	CHANNEL REQUEST RRC	Paging order is for RRC-connection.
	40 <u>48</u>	<-	CONNECTION REQUEST IMMEDIATE ASSIGNMENT RRC	
	<u>49</u>	<u>-></u>	RC CONNECTION SETUP	
	4450	->	COMPLETE PAGING RESPONSE	Mobile identity = TMSI-2
	4 <u>2</u> 51	<-	CHANNEL RELEASE RRC CONNECTION RELEASE	After sending of this message, the SS waits for disconnection of the CS signalling link.
	<u>52</u>	<u>-></u>	RRC CONNECTION RELEASE COMPLETE	uisconnection of the Co signalling link.
	43 <u>53</u>	UE	SOMI LETE	The UE is switched off or power is removed (see ICS).
	44 <u>54</u>	- >	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, combined PS / IMSI detach'

Specific message contents

None.

12.2.2.6.5 Test requirements

UE shall:

- initiate a combined PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.
- check the GMM cause which is contained in the ATTACH REJECT message and stop the timer T3310.
- set the PS update status to GU3 ROAMING NOT ALLOWED.
- delete stored P-TMSI, P-TMSI signature, RAI and PS ciphering key sequence number.
- consider USIM invalid for PS services until power is switched off or USIM is removed.
- perform an MM IMSI attach procedure, if the UE is PS class A.

12.2.2.7 Combined PS attach / rejected / location area not allowed

12.2.2.7.1 Definition

12.2.2.7.2 Conformance requirement

- 1) If the network rejects a combined PS attach procedure from the User Equipment with the cause 'location area not allowed' the User Equipment shall:
 - 1.1 not perform combined PS attach when in the same location area.
 - 1.2 delete the stored LAI, CKSN, TMSI, RAI, PS-CKSN, P-TMSI and P-TMSI signature.
 - 1.3 store the LA in the 'forbidden location areas for regional provision of service'.
- 2) If the network rejects a combined PS attach procedure from the User Equipment with the cause 'location area not allowed' the User Equipment shall:
 - 2.1 perform combined PS attach when a new location area is entered.
 - 2.2 delete the list of forbidden LAs when power is switched off.

Reference

3G3GPP-TS 24.008 clauses 4.7.3.2

12.2.2.7.3 Test purpose

To test the behaviour of the UE if the network rejects the combined PS attach procedure with the cause 'Location Area not allowed'.

To test that the UE deletes the list of forbidden LAs when power is switched off.

12.2.2.7.4 Method of test

Initial condition

System Simulator:

Three cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2, cell C in MCC1/MNC1/LAC2/RAC1. All cells are operating in network operation mode I.

User Equipment:

The UE has a valid TMSI, P-TMSI, P-TMSI signature and RAI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The SS rejects a combined PS attach with the cause value 'Location Area not allowed'. The SS checks that the UE does not perform combined PS attach while in the location area, performs PS attach when a new location area is entered and

deletes the list of forbidden LAs when switched off. CS services are not possible unless an IMSI attach procedure is performed.

Different types of UE may use different methods to periodically clear the list of forbidden location areas (e.g. every day at 12am). If the list is cleared while the test is being run, it may be necessary to re-run the test.

	Step	Direction UE SS	Message	Comments
-		SS		The following messages are sent and shall be received on cell A.
	1	SS		The SS activates cell A.
	2	UE		The UE is set in UE operation mode A (see
	3	UE		ICS). The UE is powered up or switched on and
				initiates an attach (see ICS). Cell A is preferred
	4	->	ATTACH REQUEST	by the UE. Attach type = 'Combined PS / IMSI attach'
				Mobile identity = P-TMSI-1
				P-TMSI-1 signature Routing area identity = RAI-1
	_		ATT A OU DE 150T	TMSI status = valid TMSI available
	5 6	<- UE	ATTACH REJECT	GMM cause 'Location Area not allowed' No LOCATION UPDATING REQ with type
	Ü	02		'IMSI attach' is sent to the SS
	7	<-	PAGING TYPE1 PACKET	(SS waits 30 seconds). Mobile identity = TMSI
			PAGING REQUEST	Paging order is for RRC-connection.
	8	UE		The UE shall not initiate an RRC connection. This is checked during 3 seconds.
	9	<-	PAGING TYPE1PACKET	Mobile identity = P-TMSI-1
	10	->	PAGING REQUEST	Paging order is for TBF establishment. No response from the UE to the request.
	10	,		This is checked for 10 seconds
				The following messages are sent and shall be received on cell B.
	11	SS		The SS deactivates cell A and activates cell B.
	12 13	UE UE		Cell B is preferred by the UE. No ATTACH REQUEST sent to SS
				(SS waits 30 seconds)
	14	UE		No LOCATION UPDATING REQ with type 'IMSI attach' is sent to the SS
				(SS waits 30 seconds).
	15	<-	PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = IMSI Paging order is for RRC-connection.
1	16	UE		The UE shall not initiate an RRC connection.
	17	<-	PAGING TYPE1 PACKET	This is checked during 3 seconds. Mobile identity = TMSI-1
	40		PAGING REQUEST	Paging order is for TBF establishment.
	18	UE		No response from the UE to the request. This is checked for 10seconds.
	19	UE		The UE initiates an attach by MMI or AT command.
	20			No attach is performed by the UE. This is
-				checked for 10 seconds. The following messages are sent and shall be
				received on cell C.
	21 22	SS UE		The SS deactivates cell B and activates cell C. Cell C is preferred by the UE.
	23	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach'
				Mobile identity = IMSI TMSI status = no valid TMSI available
	24	<-	ATTACH ACCEPT	Attach result = 'Combined PS / IMSI attached'
				Mobile identity = P-TMSI1 P-TMSI-1 signature
				Mobile identity = TMSI-1
	25	->	ATTACH COMPLETE	Routing area identity = RAI-3
	26	<-	PAGING TYPE1PACKET	Mobile identity = TMSI-1
	27	->	PAGING REQUEST CHANNEL REQUEST RRC	Paging order is for RRC-connection.
			CONNECTION REQUEST	

11	28	<-	IMMEDIATE ASSIGNMENT RRC	1
			CONNECTION SETUP RRC CONNECTION SETUP	
	<u>29</u>	<u>-></u>	COMPLETE	
	29 <u>30</u> 30 <u>31</u>	-> <-	PAGING RESPONSE CHANNEL RELEASE RRC CONNECTION RELEASE	Mobile identity = TMSI-1 After sending of this message, the SS waits for disconnection of the CS signalling link.
	<u>32</u>	<u>-></u>	RRC CONNECTION RELEASE	disconnection of the GG signalling link.
	31 33	<-	COMPLETE PACKET PAGING	Mobile identity = TMSI-1
	32 <u>34</u>	->	REQUESTPAGING TYPE1 UPLINK RLC DATA	Paging order is for TBF establishment. LLC PDU implicitly indicating paging response.
	33 35	UE	BLOCKSERVICE REQUEST	Service type = "paging response" The UE is switched off or power is removed
	3 4 <u>36</u>	->	DETACH REQUEST	(see ICS). Message not sent if power is removed. Detach type = 'power switched off, combined PS / IMSI detach'
				The following messages are sent and shall be
l	35 37	UE		received on cell B. The SS deactivates cell C and activates cell B. Cell B is preferred by the UE.
	36 38	UE		The UE is powered up or switched on and
l	37 <u>39</u>	->	ATTACH REQUEST	initiates an attach (see ICS). Attach type = 'Combined PS / IMSI attach' Mobile identity = P-TMSI-1
1	38 <u>40</u>	<-	ATTACH ACCEPT	P-TMSI-1 signature Routing area identity = RAI-3 TMSI status = valid TMSI available Attach result = 'Combined PS / IMSI attached' Mobile identity = P-TMSI-2 P-TMSI-2 signature Mobile identity = TMSI-2 Routing area identity = RAI-4
1	39 41	->	ATTACH COMPLETE	Routing area identity = RAI-4
	40 <u>42</u>	<-	PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = TMSI-2 Paging order is for RRC-connection.
	41 <u>43</u>	->	CHANNEL REQUEST RRC CONNECTION REQUEST	raying order is for KKC-connection.
	4 <u>244</u>	<-	IMMEDIATE ASSIGNMENT RRC CONNECTION SETUP	
	<u>45</u>	<u>-></u>	RRC CONNECTION SETUP COMPLETE	
	43 <u>46</u>	->	PAGING RESPONSE	Mobile identity = TMSI-2
	44 <u>47</u>	<-	CHANNEL RELEASE RRC CONNECTION RELEASE	After sending of this message, the SS waits for disconnection of the CS signalling link.
	<u>48</u>	<u>-></u>	RRC CONNECTION RELEASE COMPLETE	
	45 49	<-	PACKET PAGING REQUESTPAGING TYPE1	Mobile identity = TMSI-1 Paging order is for TBF establishment.
	46 <u>50</u>	->	UPLINK RLC DATA BLOCKSERVICE REQUEST	LLC PDU implicitly indicating paging response.service type = "paging response"
	47 <u>51</u>	UE	BEOGROEKVICE REQUEST	The UE is switched off or power is removed
	48 <u>52</u>	->	DETACH REQUEST	(see ICS). Message not sent if power is removed. Detach type = 'power switched off, combined PS / IMSI detach'

Specific message contents

None.

12.2.2.7.5 Test requirements

UE shall:

- initiate a combined PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.
- perform the following action depending on UE location.

When in the same location area, UE shall

- check the CMM cause which is contained in the ATTACH REJECT message and stop timer T3310.
- delete any stored LAI, CKSN, TMSI, RAI, PS-CKSN, P-TMSI and P-TMSI signature.
- store the LAI or the PLMN identity in the appropriate forbidden list("forbidden location areas for regional provision of service").

When a new location area is entered, UE shall

- perform combined PS attach when UE entered a new location area.
- delete the list of forbidden LAs when power is switched off.

12.2.2.8 Combined PS attach / abnormal cases / attempt counter check / miscellaneous reject causes

12.2.2.8.1 Definition

12.2.2.8.2 Conformance requirement

- 1) When a combined PS attach procedure is rejected with the attempt counter less than five, the User Equipment shall repeat the combined PS attach procedure after T3311 timeout.
- 2) When a combined PS attach procedure is rejected with the attempt counter five, the User Equipment shall delete the stored TMSI, LAI, CKSN, P-TMSI, P-TMSI signature, PS CKSN and RAI and start T3302.
- 3) When the T3302 expire, a new combined PS attach procedure shall be initiated.

GMM cause codes that can be selected are:

'IMSI unknown in HLR'

'IMEI not accepted'

'Illegal ME'

'UE identity cannot be derived by the network'

'Network failure'

'Congestion'

'retry upon entry into a new cell'

'Semantically incorrect message'

'Invalid mandatory information'

'Message type non-existent or not implemented'

'Message type not compatible with the protocol state'

'Information element non-existent or not implemented'

'Conditional IE error'

'Message not compatible with the protocol state'

'Protocol error, unspecified'

Reference

3G-3GPP TS 24.008 clause 4.7.3.2

12.2.2.8.3 Test purpose

To test the behaviour of the UE with respect to the attempt counter.

12.2.2.8.4 Method of test

Initial condition

System Simulator:

One cell operating in network operation mode I.

User Equipment:

The UE has a valid TMSI, P-TMSI, P-TMSI signature and RAI.

Related ICS/IXIT statements

Support of PS service Yes/No

UE operation mode A Yes/No Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE initiates a combined PS attach procedure (attempt counter zero).

The SS rejects the attach with a random cause code.

The UE initiates a new combined PS attach procedure (attempt counter one) after T3311 expires.

The SS rejects the attach with a random cause code.

The UE initiates a new combined PS attach procedure (attempt counter two) after T3311 expires.

The SS rejects the attach with a random cause code.

The UE initiates a new combined PS attach procedure (attempt counter three) after T3311 expires.

The SS rejects the attach with a random cause code.

The UE initiates a new combined PS attach procedure (attempt counter four) after T3311 expires.

The SS rejects the attach with a random cause code.

The UE initiates a new combined PS attach procedure with attempt counter five (after T3311 expires).

The SS rejects the attach with a random cause code. The UE shall not perform a new successful attach procedure after 15 seconds.

The UE initiates a combined PS attach procedure with attempt counter zero after T3302 expires without P-TMSI, P-TMSI signature, PS CKSN and RAI.

T3302; set to 10 minutes.

T3311; 15 seconds.

Step	Direction UE SS	Message	Comments
1	UE		The UE is set in UE operation mode A (see
	U.E		ICS).
2	UE		The UE is powered up or switched on and initiates an attach (see ICS).
3	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach'
			Mobile identity =P-TMSI-1 P-TMSI-1 signature
			Routing area identity = RAI-1
			TMSI status = valid TMSI available
4 5	<- SS	ATTACH REJECT	Random GMM cause The SS verifies that the time between the
			attach requests is T3311
6	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach' Mobile identity = P-TMSI-1
			P-TMSI-1 signature
			Routing area identity = RAI-1
7	<-	ATTACH REJECT	TMSI status = valid TMSI available Random GMM cause
8	ss	, the trace of	The SS verifies that the time between the
		ATTACH DECLICA	attach requests is T3311
9	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach' Mobile identity = P-TMSI-1
			P-TMSI-1 signature
			Routing area identity = RAI-1 TMSI status = valid TMSI available
10	<-	ATTACH REJECT	Random GMM cause
11	SS		The SS verifies that the time between the
12	->	ATTACH REQUEST	attach requests is T3311 Attach type = 'Combined PS / IMSI attach'
			Mobile identity = P-TMSI-1
			P-TMSI-1 signature Routing area identity = RAI-1
			TMSI status = valid TMSI available
13	<-	ATTACH REJECT	Random GMM cause
14	SS		The SS verifies that the time between the attach requests is T3311
15	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach'
			Mobile identity =P-TMSI-1 P-TMSI-1 signature
			Routing area identity = RAI-1
40		ATTAOU DE ISOT	TMSI status = valid TMSI available
16 17	<- UE	ATTACH REJECT	Random GMM cause No LOCATION UPDATING REQ with type
	"-		'IMSI attach' is sent to the SS
18	<-	PAGING TYPE1 PACKET	(SS waits 30 seconds). Mobile identity = TMSI
10		PAGING REQUEST	Paging order is for RRC-connection.
19	UE		The UE shall not initiate an RRC connection.
20	<-	PAGING TYPE1 PACKET	This is checked during 3 seconds. Mobile identity = TMSI-1
		PAGING REQUEST	Paging order is for TBF establishment.
21	UE		No response from the UE to the request. This is checked for 10seconds.
22	SS		The SS verifies that the UE does not attempt to
- 00		ATTACH DECLECT	attach for T3302 .
23	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach' Mobile identity = IMSI
			TMSI status = no valid TMSI available
24	<-	ATTACH ACCEPT	Attach result = 'Combined PS / IMSI attached' Mobile identity P-TMSI-1
			P-TMSI signature
			Mobile identity = TMSI-1
	l	I	Routing area identity = RAI-1

25	->	ATTACH COMPLETE	
26	<-	PAGING TYPE1PACKET	Mobile identity = TMSI-1
		PAGING REQUEST	Paging order is for RRC-connection.
27	->	CHANNEL REQUEST RRC	
		CONNECTION REQUEST	
28	<-	IMMEDIATE ASSIGNMENT RRC	
		CONNECTION SETUP	
<u>29</u>	<u>-></u>	RRC CONNECTION SETUP	
2020	_	COMPLETE DACING DECRONGE	Mahila idantitu. TMCI 4
29 30	->	PAGING RESPONSE	Mobile identity = TMSI-1
30 31	<-	CHANNEL RELEASE RRC	After sending of this message, the SS waits for
		CONNECTION RELEASE	disconnection of the CS signalling link.
32	<u>-></u>	RRC CONNECTION RELEASE	
		COMPLETE	
31 33	<-	PACKET PAGING	Mobile identity = TMSI-1
		REQUESTPAGING TYPE1	Paging order is for TBF establishment.
32 34	->	UPLINK RLC DATA	LLC PDU implicitly indicating paging response.
		BLOCKSERVICE REQUEST	Service type = "paging response"
33 35	UE		The UE is switched off or power is removed
·			(see ICS).
34 36	->	DETACH REQUEST	Message not sent if power is removed.
· []			Detach type = 'power switched off, combined
			PS / IMSI detach'

Specific message contents

None.

12.2.2.8.5 Test requirements

UE shall:

- initiate a combined PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.
- perform the following actions depending on the conditions described below.

Case1) A combined PS attach procedure is rejected with the attempt counter less than five

UE shall:

- repeat the combine PS attach procedure after the timer T3311 timeout.

Case2) A combined PS attach procedure is rejected with the attempt counter five

UE shall:

- delete the stored TMSI, LAI, CKSN, P-TMSI, P-TMSI signature, PS CKSN and RAI and
- start the timer T3302.

Case3) The T3302 expires

UE shall:

- re-initiate a new combined PS attach procedure.

12.2.2.9 Combined PS attach / abnormal cases / PS detach procedure collision

12.2.2.9.1 Definition

12.2.2.9.2 Conformance requirement

- 1) When a DETACH REQUEST message is received by the UE (any cause except re-attach) while waiting for an ATTACH ACCEPT message or ATTACH REJECT message, the UE shall terminate the combined PS attach procedure and continue with the combined PS detach procedure.
- 2) When a DETACH REQUEST message is received by the UE (cause re-attach) while waiting for an ATTACH ACCEPT message or ATTACH REJECT message, the UE shall ignore the combined PS detach procedure and continue with the combined PS attach procedure.

Reference

3G3GPP-TS 24.008 clause 4.7.3.2

12.2.2.9.3 Test purpose

To test the behaviour of the UE in case of procedure collision.

12.2.2.9.4 Method of test

Initial condition

System Simulator:

One cell operating in network operation mode I.

User Equipment:

The UE has a valid TMSI, P-TMSI, P-TMSI signature and RAI.

Related ICS/IXIT statements

Support of PS service Yes/No UE operation mode A Yes/No Switch off on button Yes/No Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE initiates a combined PS attach procedure. The SS does not answer the combined PS attach procedure, but initiates a combined PS detach procedure (any cause except re-attach). The UE shall terminate the combined PS attach procedure and continue with the combined PS detach procedure. CS services are not possible as an IMSI attach procedure is not performed.

The UE initiates a combined PS attach procedure. The SS does not answer the combined PS attach procedure, but initiates a combined PS detach procedure (cause re-attach). The UE shall ignore the combined PS detach procedure and continue with the combined PS attach. CS services are also possible.

	Step	Direction	Message	Comments
L		UE SS		
	1	UE		The UE is set in UE operation mode B (see
	2	UE		ICS). The UE is powered up or switched on and
				initiates an attach (see ICS).
	3	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach'
				Mobile identity = P-TMSI-1
				P-TMSI-1 signature Routing area identity = RAI-1
				TMSI status = valid TMSI available
	4	SS		The SS ignores the ATTACH REQUEST
				message and initiates a detach procedure.
	5	<-	DETACH REQUEST	Detach type = 're-attach not required'
1	6 7	->	DETACH ACCEPT PAGING TYPE1PACKET	Mobile identity – IMSI
	,	<-	PAGING TIPET PAGET	Mobile identity = IMSI Paging order is for RRC-connection.
J	8	UE	THOMAS REGISTER	The UE shall not initiate an RRC connection.
		_		This is checked during 3 seconds.
	9	UE		The UE is attached by MMI or AT command
	10	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach'
				Mobile identity = P-TMSI-1 P-TMSI-1 signature
				Routing area identity = RAI-1
				TMSI status = valid TMSI available
	11	SS		The SS ignores the ATTACH REQUEST
	40		DETA OLI DEGLIEGE	message and initiates a detach procedure.
	12 13	<- UE	DETACH REQUEST	Detach type = 're-attach required' The UE ignores the DETACH REQUEST
	13	OL		message and continue with the attach
				procedure
	14	<-	ATTACH ACCEPT	Attach result = 'Combined PS / IMSI attached'
				Mobile identity = P-TMSI-2
				P-TMSI-2 signature Mobile identity = TMSI-2
				Routing area identity = RAI-1
				TMSI status = valid TMSI available
ı	15	->	ATTACH COMPLETE	MALE TAKE
	16	<-	PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = TMSI-2 Paging order is for RRC-connection.
	17	->	CHANNEL REQUEST RRC	aging order is for title-connection.
			CONNECTION REQUEST	
	18	<-	IMMEDIATE ASSIGNMENT RRC	
	10		CONNECTION SETUP RRC CONNECTION SETUP	
	<u>19</u>	<u>-></u>	COMPLETE	
	19 20	->	PAGING RESPONSE	Mobile identity = TMSI-2
	20 21	<-	CHANNEL RELEASE RRC	After sending of this message, the SS waits for
	22		CONNECTION RELEASE	disconnection of the CS signalling link.
	<u>22</u>	<u>-></u>	RRC CONNECTION RELEASE COMPLETE	
	21 23	<-	PACKET PAGING	Mobile identity = TMSI-2
			REQUESTPAGING TYPE1	Paging order is for TBF establishment.
	22 24	->	UPLINK RLC DATA	LLC PDU implicitly indicating paging response.
	23 25	UE	BLOCKSERVICE REQUEST	Service type = "paging response" The UE is switched off or power is removed
1	20 20	UE		(see ICS).
П	24 26	->	DETACH REQUEST	Message not sent if power is removed.
				Detach type = 'power switched off, combined
L				PS / IMSI detach'

Specific message contents

None.

12.2.2.9.5 Test requirements

UE shall:

- initiate a combined PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.
- perform the following actions depending on the Detach type described below.

Case1) Detach type is not re-attach

UE shall:

- terminate the combined PS attach procedure.
- continue with the combined PS detach procedure.

Case2) Detach type is re-attach

UE shall:

- ignore the combined PS detach procedure.
- continue with the combined PS attach procedure.

12.3 PS detach procedure

12.3.1 UE initiated PS detach procedure

12.3.1.1 PS detach / power off / accepted

12.3.1.1.1 Definition

12.3.1.1.2 Conformance requirement

The UE detaches the IMSI for PS services if the UE is switched off.

Reference

3G3GPP-TS 24.008 clause 4.7.4.1

12.3.1.1.3 Test purpose

To test the behaviour of the UE for the detach procedure.

12.3.1.1.4 Method of test

Initial condition

System Simulator:

One cell operating in network operation mode II.

User Equipment:

The UE has a valid P-TMSI-1, P-TMSI-1 signature and RAI-1.

Related ICS/IXIT statements

Support of PS service Yes/No UE operation mode C Yes/No UE operation mode A Yes/No Switch off on button Yes/No

Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE performs a PS attach procedure.

The UE sends a DETACH REQUEST message to the SS.

Expected Sequence

Step	Direction	Message	Comments
	UE SS		
1	UE		The UE is set in UE operation mode C (see
			ICS). If UE operation mode C not supported,
2	UE		goto step 8. The UE is powered up or switched on and
	OL		initiates an attach (see ICS).
3	->	ATTACH REQUEST	Attach type = 'PS attach'
			Mobile identity = P-TMSI-1
			P-TMSI-1 signature
		ATTACLLACOERT	Routing area identity = RAI-1
4	<-	ATTACH ACCEPT	Attach result = 'PS only attached' Mobile identity = P-TMSI-2
			P-TMSI-2 signature
			Routing area identity = RAI-1
5	->	ATTACH COMPLETE	
6	UE		The UE is switched off (see ICS).
7	->	DETACH REQUEST	Detach type = 'power switched off, PS detach'
8	UE		The UE is set in UE operation mode A(see
			ICS) and the test is repeated from step 2 to
			step 7.

Specific message contents

None.

12.3.1.1.5 Test requirements

UE shall:

- initiate a PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.

After the PS attach procedure is completed, UE shall:

- send the DETACH REQUEST message to SS with the Detach type = 'power switched off, PS detach'.

12.3.1.2 PS detach / accepted

12.3.1.2.1 Definition

12.3.1.2.2 Conformance requirement

The UE detaches the IMSI for PS services if the UE is ordered to do so with MMI or AT commands.

Reference

3G3GPP-TS 24.008 clause 4.7.4.1

12.3.1.2.3 Test purpose

To test the behaviour of the UE for the detach procedure.

12.3.1.2.4 Method of test

Initial condition

System Simulator:

One cell operating in network operation mode **!!!!!**.

User Equipment:

The UE has a valid P-TMSI-1, P-TMSI-1 signature and RAI-1.

Related ICS/IXIT statements

Support of PS service Yes/No UE operation mode C Yes/No UE operation mode A Yes/No Switch off on button Yes/No Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE performs a PS attach procedure and activates a PDP context.

The UE sends a DETACH REQUEST message to the SS.

The SS signal to the UE, but no response is received, as the signalling link is disconnected.

Step	Direction	Message	Comments
	UE SS		
1	UE		The UE is set in UE operation mode C (see ICS). If UE operation mode C not supported, goto step 11.
2	UE		The UE is powered up or switched on and
3	->	ATTACH REQUEST	initiates an attach (see ICS). Attach type = 'PS attach' Mobile identity = P-TMSI-1 P-TMSI-1 signature
4	<-	ATTACH ACCEPT	Routing area identity = RAI-1 No new mobile identity assigned. P-TMSI and P-TMSI signature not included. Attach result = 'PS only attached' Routing area identity = RAI-1
5	->	ATTACH COMPLETE	Trouting area identity = 17.11
6	UE		The UE initiates a PS detach (without power off) by MMI or AT command.
7	->	DETACH REQUEST	Detach type = 'normal detach, PS detach'
8	<-	DETACH ACCEPT	
9	<-	PAGING TYPE1PACKET	Mobile identity = P-TMSI-1
		PAGING REQUEST	Both paging orders are for TBF establishment.
		Or	PAGING TYPE1PACKET PAGING REQUEST
		PAGING REQUEST TYPE 1	(used for NW mode III)
			PAGING REQUEST TYPE 1 (used for NW-mode II).
10	UE		No response from the UE to the request. This
10			is checked for 10 seconds.
11	SS		The SS is set in network operation mode II.
12	UE		The UE is set in UE operation mode A (see
			ICS) and the test is repeated from step 2 to
			step 10.

Specific message contents

None.

12.3.1.2.5 Test requirements

UE shall:

- initiate a PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.

After the PS attach procedure is completed, UE shall:

- sends the DETACH REQUEST message(without power off) to SS.
- start timer T3321.

When UE receives the DETACH ACCEPT message from SS before the timer T3321 is not expired, UE shall:

- stop timer T3321.

12.3.1.3 PS detach / abnormal cases / attempt counter check / procedure timeout

12.3.1.3.1 Definition

12.3.1.3.2 Conformance requirement

- 1) When a T3321 timeout has occurred during a PS detach procedure with the attempt counter less than five, the User Equipment shall repeat the PS detach procedure.
- 2) When a T3321 timeout has occurred during a PS detach procedure with the attempt counter five, the User Equipment shall not repeat the procedure.

Reference

3G3GPP-TS 24.008 clause 4.7.4.1

12.3.1.3.3 Test purpose

To test the behaviour of the UE with respect to the attempt counter.

12.3.1.3.4 Method of test

Initial condition

System Simulator:

One cell operating in network operation mode IIII.

User Equipment:

The UE has a valid P-TMSI-1, P-TMSI-1 signature and RAI-1.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode C Yes/No
UE operation mode A Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE performs a PS attach procedure.

The UE initiates a PS detach procedure (attempt counter zero). The SS does not answer with DETACH ACCEPT message before T3321 timeout.

The UE initiates a new PS detach procedure (attempt counter one) after T3311 expires. The SS does not answer with DETACH ACCEPT message before T3321 timeout.

The UE initiates a new PS detach procedure (attempt counter two) after T3311 expires. The SS does not answer with DETACH ACCEPT message before T3321 timeout.

The UE initiates a new PS detach procedure (attempt counter three) after T3311 expires. The SS does not answer with DETACH ACCEPT message before T3321 timeout.

The UE initiates a new PS detach procedure (attempt counter four) after T3311 expires. The SS does not answer with DETACH ACCEPT message before T3321 timeout.

The UE initiates a new PS detach procedure with attempt counter five (after T3311 expires). The SS does not answer with DETACH ACCEPT message before T3321 timeout.

At T3321 timeout in the UE, the UE then deletes the logical link.

The UE performs a new PS attach procedure.

T3321; 15 seconds.

Expected Sequence

Step	Direction	Message	Comments
	UE SS		
1	UE		The UE is set in UE operation mode C (see ICS). If UE operation mode C not supported, goto step 22.
2	UE		The UE is powered up or switched on and initiates an attach (see ICS).
3	->	ATTACH REQUEST	Attach type = 'PS attach' Mobile identity = P-TMSI-1 P-TMSI-1 signature
4	<-	ATTACH ACCEPT	Routing area identity = RAI-1 No new mobile identity assigned. P-TMSI and P-TMSI signature not included. Attach result = 'PS only attached' Routing area identity = RAI-1
5	->	ATTACH COMPLETE	Routing area identity = RAI-1
6	UÉ	ATTACH COMPLETE	The UE initiates a PS detach (without power off) by MMI or AT command.
7 8 9	-> SS SS	DETACH REQUEST	Detach type = 'normal detach, PS detach' No response is given from the SS. The SS verifies that the time between the detach requests is 15 seconds
10 11 12	-> SS SS	DETACH REQUEST	Detach type = 'normal detach, PS detach' No response is given from the SS. The SS verifies that the time between the
13 14 15	-> SS SS	DETACH REQUEST	detach requests is 15 seconds Detach type = 'normal detach, PS detach' No response is given from the SS. The SS verifies that the time between the
16 17 18	-> SS SS	DETACH REQUEST	detach requests is 15 seconds Detach type = 'normal detach, PS detach' No response is given from the SS. The SS verifies that the time between the detach requests is 15 seconds
19 20	-> SS SS	DETACH REQUEST	Detach type = 'normal detach, PS detach' No response is given from the SS. The SS verifies that the time between the detach requests is 15 seconds
21 22	-> SS	DETACH REQUEST	Detach type = 'normal detach, PS detach' No response is given from the SS.
23 24	SS UE		The SS is set in network operation mode II. The UE is set in UE operation mode A (see ICS) and the test is repeated from step 2 to step 22.

Specific message contents

None.

12.3.1.3.5 Test requirements

UE shall:

- initiate a PS attaché procedure with the information elements specified in the above Expected Sequence when the UE is powered up or switched on.

After the PS attach procedure is completed, UE shall:

- initiate a PS detach procedure(without power off).
- start timer T3321.

When a T3221 expires with the attempt counter less than five, UE shall:

- initiate a new PS detach procedure.
- increment the attempt counter.
- re-start timer T3321.

When a T3221 expires with the attempt counter five, UE shall:

- not repeat the procedure.

12.3.1.4 PS detach / abnormal cases / GMM common procedure collision

12.3.1.4.1 Definition

12.3.1.4.2 Conformance requirement

When any of the GMM common messages P-TMSI REALLOCATION COMMAND, GMM STATUS or GMM INFORMATION is received by the UE while waiting for a DETACH ACCEPT message with detach cause different from "power off", the UE shall ignore the GMM common message.

Reference

3G3GPP-TS 24.008 clause 4.7.4.1

12.3.1.4.3 Test purpose

To test the behaviour of the UE in case of procedure collision.

12.3.1.4.4 Method of test

Initial condition

System Simulator:

One cell operating in network operation mode II.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode C Yes/No
UE operation mode A Yes/No (only if mode C not supported)
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE performs a PS attach.

The UE initiates a PS detach. The SS initiates a P-TMSI REALLOCATION COMMAND message, a GMM STATUS message and a GMM INFORMATION message. The UE shall ignore the GMM common messages and continue with the PS detach procedure.

The SS signal to the UE, but no response is received, as the signalling link is disconnected.

Expected Sequence

Ī	Step	Direction	Message	Comments
		UE SS		
	1	UE		The UE is set in UE operation mode C or B (see ICS).
	2	UE		The UE is powered up or switched on and initiates an attach (see ICS).
	3	->	ATTACH REQUEST	Attach type = 'PS attach' Mobile identity = IMSI
	4	<-	ATTACH ACCEPT	Attach result = 'PS only attached' Mobile identity = P-TMSI-1 P-TMSI-1 signature Routing area identity = RAI-1
	5	->	ATTACH COMPLETE	
	6	UE		The UE initiates a detach (without power off) by MMI or AT command.
	7	->	DETACH REQUEST	Detach type = 'normal detach, PS detach'
	8	SS		The SS sends a P-TMSI REALLOCATION COMMAND message
	9	<-	P-TMSI REALLOCATION COMMAND	Ç
	10	UE		The UE ignores the message.
	11	SS		The SS sends a GMM STATUS message
	12	<-	GMM STATUS	
	13	UE		The UE ignores the message.
	14	SS		The SS sends a GMM INFORMATION
				message
	15	<-	GMM INFORMATION	
	16	UE		The UE ignores the message.
	17	<-	DETACH ACCEPT	The SS responds to the DETACH REQUEST
	18	<-	PAGING TYPE1PAGING	Mobile identity = P-TMSI-1
			REQUEST TYPE 1	Paging order is for TBF establishment.
	19	UE		No response from the UE to the request. This is checked for 10 seconds.

Specific message contents

None.

12.3.1.4.5 Test requirements

UE shall:

- initiate a PS attaché procedure with the information elements specified in the above Expected Sequence when the UE is powered up or switched on.

After the PS attach procedure is completed, UE shall:

- initiate a PS detach procedure(without power off).

When any of the GMM common messages P-TMSI REALLOCATION COMMAND, GMM STATUS or GMM INFORMATION is received by the UE while waiting for a DETACH ACCEPT message with detach cause different from "power off, UE shall: .

- ignore any of the GMM common.

12.3.1.5 PS detach / power off / accepted

12.3.1.5.1 Definition

12.3.1.5.2 Conformance requirement

The UE detach the IMSI for PS and non-PS services.

Reference

3G3GPP-TS 24.008 clause 4.7.4.1

12.3.1.5.3 Test purpose

To test the behaviour of the UE for the detach procedure.

12.3.1.5.4 Method of test

Initial condition

System Simulator:

One cell operating in network operation mode I.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
Switch off on button Yes/No

Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE performs a combined PS attach procedure (for PS and non-PS services).

The UE sends a DETACH REQUEST message to the SS. The UE then deletes the logical link.

Step	Direction	Message	Comments
	UE SS		
1	UE		The UE is set in UE operation mode A (see ICS).
2	UE		The UE is powered up or switched on and initiates an attach (see ICS).
3	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach' Mobile identity = IMSI
4	<-	ATTACH ACCEPT	TMSI status = no valid TMSI available Attach result = 'Combined PS / IMSI attached' Mobile identity = P-TMSI-1 P-TMSI-1 signature Routing area identity = RAI-1
5	->	ATTACH COMPLETE	Trodding area identity = Trai-1
6	UE		The UE is switched off (see ICS).
7	->	DETACH REQUEST	Detach type = 'power switched off, combined PS / IMSI detach'

Specific message contents

None.

12.3.1.5.5 Test requirements

UE shall:

- initiate a combined PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.

After the PS attach procedure is completed, UE shall:

- send the DETACH REQUEST message to SS with the Detach type = 'power switched off, combined PS / IMSI detach' after the PS attach procedure is completed.

12.3.1.6 PS detach / accepted / PS/IMSI detach

12.3.1.6.1 Definition

12.3.1.6.2 Conformance requirement

The UE detach the IMSI for PS and non-PS services.

Reference

3G3GPP-TS 24.008 clause 4.7.4.1

12.3.1.6.3 Test purpose

To test the behaviour of the UE for the detach procedure.

12.3.1.6.4 Method of test

Initial condition

System Simulator:

One cell operating in network operation mode I.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE performs a combined PS attach procedure (for PS and non-PS services).

The UE sends a DETACH REQUEST message to the SS. When the UE receives the DETACH ACCEPT, the UE then deletes the logical link.

The SS signal to the UE, but no response is received, as the signalling link is disconnected.

Expected Sequence

Step	Direction	Message	Comments
	UE SS		
1	UE		The UE is set in UE operation mode A(see
2	UE		The UE is powered up or switched on and
3	->	ATTACH REQUEST	initiates an attach (see ICS). Attach type = 'Combined PS / IMSI attach' Mobile identity = IMSI
4	<-	ATTACH ACCEPT	TMSI status = no valid TMSI available Attach result = 'Combined PS / IMSI attached' Mobile identity = P-TMSI-1 P-TMSI-1 signature
			Mobile identity = TMSI-1 Routing area identity = RAI-1
5	->	ATTACH COMPLETE	l seeming area seeming
6	UE		The UE initiates a detach (without power off) by MMI or AT command.
7	->	DETACH REQUEST	Detach type = 'normal detach, combined PS / IMSI detach'
8	<-	DETACH ACCEPT	Imer datasi:
9	<-	PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = P-TMSI-1 Paging order is for TBF establishment.
10	UE		No response from the UE to the request. This is checked for 10 seconds.
11	<-	PAGING TYPE1PACKET	Mobile identity = IMSI Paging order is for RRC connection.
12	UE	TABLE NEW CEST	The UE shall not initiate an RRC connection. This is checked during 3 seconds.

Specific message contents

None.

12.3.1.6.5 Test requirements

UE shall:

- initiate a combined PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.

After the combined PS attach procedure is completed, UE shall:

- sends the DETACH REQUEST message(UE not switched off) to SS.
- shall start timer T3321.

When the UE receive the DETACH ACCEPT message from SS before the timer T3321 is not expired, the UE shall:

- stop timer T3321.

12.3.1.7 PS detach / accepted / IMSI detach

12.3.1.7.1 Definition

12.3.1.7.2 Conformance requirement

The UE shall detach for CS services.

Reference

3G3GPP-TS 24.008 clause 4.7.4.1

12.3.1.7.3 Test purpose

To test the behaviour of the UE for the detach procedure.

12.3.1.7.4 Method of test

Initial condition

System Simulator:

One cell operating in network operation mode I.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No
MMI controlled attach / detach procedures for non-PS services Yes/No

Test procedure

The UE performs a combined PS attach procedure (for PS and non-PS services).

The UE performs an PS detach (for non-PS services).

CS services are not possible.

The UE attach for non-PS services by a routing area update procedure and CS services are again possible.

	Step	Direction UE SS	Message	Comments
-	1	UE		The UE is set in UE operation mode A (see ICS).
	2	UE		The UE is powered up or switched on and initiates an attach (see ICS).
	3	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach' Mobile identity = IMSI
	4	<-	ATTACH ACCEPT	TMSI status = no valid TMSI available Attach result = 'Combined PS / IMSI attached' Mobile identity = P-TMSI-1 P-TMSI-1 signature Mobile identity = TMSI-1
	5 6	-> UE	ATTACH COMPLETE	Routing area identity = RAI-1 The UE initiates a detach for non-PS services (without power off) (age ICS)
	7 8	-> <-	DETACH REQUEST DETACH ACCEPT	(without power off) (see ICS). Detach type = 'normal detach, IMSI detach'
	9	<-	PACKET PAGING	Mobile identity = P-TMSI-1
	10	->	REQUESTPAGING TYPE1 UPLINK RLC DATA BLOCKSERVICE REQUEST	Paging order is for TBF establishment. LLC PDU implicitly indicating paging response.service type = "paging response"
	11	<-	PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = TMSI-1 Paging order is for RRC connection.
J	12	UE	THOMAS NEEDS	The UE shall not initiate an RRC connection. This is checked during 3 seconds.
	13	UE		The UE initiates an attach for non-PS services by a RA update procedure (see ICS).
	14	->	ROUTING AREA UPDATE REQUEST	Update type = 'Combined RA/LA updating'' P-TMSI-1 signature
	15	<-	ROUTING AREA UPDATE ACCEPT	Routing area identity = RAI-1 TMSI status = valid TMSI available Update result = 'Combined RA/LA updated' Mobile identity = P-TMSI-2 P-TMSI-2 signature Mobile identity = TMSI-1
	16	->	ROUTING AREA UPDATE COMPLETE	Routing area identity = RAI-1
	17	<-	PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = TMSI-1 Paging order is for RRC connection.
	18	->	CHANNEL REQUEST RRC CONNECTION REQUEST	3 - 5
	19	<-	IMMEDIATE ASSIGNMENT RRC CONNECTION SETUP	
	<u>20</u>	<u>-></u>	RRC CONNECTION SETUP COMPLETE	
	2021 2122	-> <-	PAGING RESPONSE CHANNEL RELEASE RRC CONNECTION RELEASE	Mobile identity = TMSI-1 After sending of this message, the SS waits for disconnection of the CS signalling link.
	<u>23</u>	<u>-></u>	RRC CONNECTION RELEASE COMPLETE	
	22 24	UE		The UE is switched off or power is removed (see ICS).
	23 25	->	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, combined PS / IMSI detach'

Specific message contents

None.

12.3.1.7.5 Test requirements

UE shall:

- initiate a combined PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.

After the combined PS attach procedure is completed, UE shall:

- sends the DETACH REQUEST message(UE not switched off) to SS.
- start timer T3321.

When the UE receives the DETACH ACCEPT message from SS before the timer T3321 is not expired, the UE shall:

- stop timer T3321.

12.3.1.8 PS detach / abnormal cases / change of cell into new routing area

12.3.1.8.1 Definition

12.3.1.8.2 Conformance requirement

When a change of cell into a new routing area is performed before DETACH ACCEPT message is received by the UE, the UE shall abort the PS detach procedure and re-initiate it after the routing area update procedure.

Reference

3G3GPP-TS 24.008 clause 4.7.4.1

12.3.1.8.3 Test purpose

To test the behaviour of the UE in case of procedure collision.

12.3.1.8.4 Method of test

Initial condition

System Simulator:

Two cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1 and cell B in MCC1/MNC1/LAC1/RAC2.

Both cells are operating in network operation mode I.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No UE operation mode A Yes/No

Switch off on button Yes/No Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE performs a combined PS attach procedure (for PS and non-PS services).

The UE initiates a PS detach procedure. The DETACH ACCEPT message is delayed from the SS.

The UE performs a cell update into a new routing area.

The Ms shall re-initiate a PS detach procedure when the routing area update procedure is finished.

The UE deletes the logical link.

Step	Direction	Message	Comments
-	UE SS]	
	SS		The following messages are sent and shall be
			received on cell A.
1	SS		The SS activates cell A.
2	UE		The UE is set in UE operation mode A (see
			ICS).
3	UE		The UE is powered up or switched on and
			initiates an attach (see ICS). Cell A is preferred
		ATTACLIBEOLIECT	by the UE.
4	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach'
			Mobile identity = IMSI TMSI status = no valid TMSI available
5	<-	ATTACH ACCEPT	Attach result = 'Combined PS / IMSI attached'
3	ζ-	ATTACITACCEFT	Mobile identity = P-TMSI-1
			P-TMSI-1 signature
			Mobile identity = TMSI-1
			Routing area identity = RAI-1
6	->	ATTACH COMPLETE	Trouming area recrimity
7	UE		The UE initiates a PS detach (without power
			off) by MMI or AT command.
8	->	DETACH REQUEST	Detach type = 'normal detach, combined PS /
			IMSI detach'
9	SS		No response to the DETACH REQUEST
			message is given by the SS
			The following messages are sent and shall be
40	SS		received on cell B.
10	55		The SS deactivates cell A and activates cell B. Cell B is preferred by the UE.
11	UE		The UE performs a RA update in the new cell.
12	->	ROUTING AREA UPDATE	Update type = 'Combined RA/LA updating'
		REQUEST	Mobile identity = P-TMSI-1
			P-TMSI-1 signature
			Routing area identity = RAI-1
			TMSI status = valid TMSI available
13	<-	ROUTING AREA UPDATE	Update result = 'Combined RA/LA updated'
		ACCEPT	
			Mobile identity = P-TMSI-2
			P-TMSI-2 signature
1		DOLLTING A DEALUSE A TE	Routing area identity = RAI-4
14	->	ROUTING AREA UPDATE	
15	1	COMPLETE	The detech is sutemptically to ottomated
15	->	DETACH REQUEST	The detach is automatically re-attempted. Detach type = 'normal detach, combined PS /
			IMSI detach'
16	->	DETACH ACCEPT	INIOI GELACII
10		D = 1/1011/100E1	

Specific message contents

None.

12.3.1.8.5 Test requirements

UE shall:

- initiate a combined PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.

After the combined PS attach procedure is completed, UE shall:

- initiate a PS detach.

When a change of cell into a new routing area is performed before DETACH ACCEPT message by the UE, UE shall:

- abort a PS detach procedure.
- re-initiate a PS detach procedure after successfully performing a routing area updating procedure.

12.3.1.9 PS detach / abnormal cases / PS detach procedure collision

12.3.1.9.1 Definition

12.3.1.9.2 Conformance requirement

When a DETACH REQUEST is received by the UE while waiting for a DETACH ACCEPT message, the UE shall answer the network initiated PS detach procedure.

Reference

3G3GPP-TS 24.008 clause 4.7.4.1

12.3.1.9.3 Test purpose

To test the behaviour of the UE in case of procedure collision.

12.3.1.9.4 Method of test

Initial condition

System Simulator:

One cell operating in network operation mode I.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No UE operation mode A Yes/No Switch off on button Yes/No Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE performs a combined PS attach procedure (for PS and non-PS services).

The UE initiates a PS detach. The SS does not answer the detach procedure, but initiates a detach procedure (cause reattach not required). The UE shall continue with the network initiated detach procedure.

The UE deletes the logical link.

PS and CS services are not possible.

Step	Direction	Message	Comments
	UE SS		
1	UE		The UE is set in UE operation mode A(see ICS).
2	UE		The UE is powered up or switched on and initiates an attach (see ICS).
3	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach' Mobile identity = IMSI TMSI status = no valid TMSI available
4	< -	ATTACH ACCEPT	Attach result = 'Combined PS / IMSI attached' Mobile identity = P-TMSI-1 P-TMSI-1 signature Mobile identity = TMSI-1 Routing area identity = RAI-1
5	->	ATTACH COMPLETE	Trouting area identity = Train
6	UÉ		The UE initiates a PS detach (without power off) by MMI or AT command.
7	->	DETACH REQUEST	Detach type = 'normal detach, combined PS / IMSI detach'
8	<-	DETACH REQUEST	Detach type = 're-attach not required'
9	->	DETACH ACCEPT	The UE answers the network initiated detach.
10	<-	DETACH ACCEPT	The SS answers the UE initiated detach.
11	<-	PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = P-TMSI-1 Paging order is for TBF establishment.
12	UE		No response from the UE to the request. This is checked for 10 seconds.
13	<-	PAGING TYPE1PAGING REQUEST TYPE 1	Mobile identity = TMSI-1 Paging order is for RRC connection.
14	UE		The UE shall not initiate an RRC connection. This is checked during 3 seconds.

Specific message contents

None.

12.3.1.9.5 Test requirements

UE shall:

- initiate a combined PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.

After the combined PS attach procedure is completed, UE shall:

- initiate a PS detach procedure.

When the UE receive DETACH REQUEST message from SS before UE initiated PS detach procedure has been completed, UE shall:

- send the DETACH ACCEPT message to SS

12.3.2 Network initiated PS detach procedure

12.3.2.1 PS detach / re-attach not required / accepted

12.3.2.1.1 Definition

12.3.2.1.2 Conformance requirement

The UE detach the IMSI for PS services.

Reference

3G3GPP-TS 24.008 clause 4.7.4.2

12.3.2.1.3 Test purpose

To test the behaviour of the UE for the detach procedure.

12.3.2.1.4 Method of test

Initial condition

System Simulator:

One cell operating in network operation mode II (in case of UE operation mode A), or in network operation mode III (in case of UE operation mode C).

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode C Yes/No
UE operation mode A Yes/No (only if mode C not supported)
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE performs a PS attach procedure.

The SS sends a DETACH REQUEST message to the UE. The UE then deletes the logical link.

The SS signal to the UE, but no response is received, as the signalling link is disconnected.

	Step	Direction	Message	Comments
		UE SS		
	1	SS		The SS is set in network operation mode II-or
				₩.
	2	UE		The UE is set in UE operation mode A or C
	_			(see ICS).
	3	UE		The UE is powered up or switched on and
			ATTACH DECLIEST	initiates an attach (see ICS).
	4	->	ATTACH REQUEST	Attach type = 'PS attach'
	5	_	ATTACH ACCEPT	Mobile identity = IMSI
	Э	<-	ATTACH ACCEPT	Attach result = 'PS only attached' Mobile identity = P-TMSI-1
				P-TMSI-1 signature
				Routing area identity = RAI-1
	6	->	ATTACH COMPLETE	Trouting aroa idonaty – Tota 1
	7	UÉ		The UE initiates a PS detach (without power
				off) by MMI or AT command.
	8	<-	DETACH REQUEST	Detach type = 're-attach not required'
	9	->	DETACH ACCEPT	
	10	<-	PAGING TYPE1PACKET	Mobile identity = P-TMSI-1
			PAGING REQUEST	Both paging orders are for TBF establishment.
			Or	PAGING TYPE1 PACKET PAGING REQUEST
			PAGING REQUEST TYPE 1	(used for NW mode III)
1				PAGING REQUEST TYPE 1 (used for NW-
	44			mode II).
	11	UE		No response from the UE to the request. This is checked for 10 seconds.
L				ris checked for to seconds.

Specific message contents

None.

12.3.2.1.5 Test requirements

UE shall:

- initiate PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.

When UE receives the DETACH REQUEST message from SS and the detach type IE indicates 're-attach not required', the UE shall

- deactivate the PDP context and the logical link(s).
- send DETACH ACCEPT message to SS.

12.3.2.2 PS detach / rejected / IMSI invalid / PS services not allowed

12.3.2.2.1 Definition

12.3.2.2.2 Conformance requirement

- 1) If the network performs a PS detach procedure with the cause 'PS services not allowed', the User Equipment shall consider USIM invalid for PS services until power is switched off or USIM is removed.
- 2) If the network performs a PS detach procedure with the cause 'PS services not allowed' the User Equipment shall delete the stored RAI, PS-CKSN, P-TMSI and P-TMSI signature.

Reference

3G3GPP-TS 24.008 clause 4.7.4.2

12.3.2.2.3 Test purpose

To test the behaviour of the UE if the network order a PS detach procedure with the cause 'PS services not allowed' (no valid PS-subscription for the IMSI).

12.3.2.2.4 Method of test

Initial condition

System Simulator:

Two cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1 (HPLMN) and cell B in MCC2/MNC1/LAC1/RAC1.

Both cells are operating in network operation mode II.

User Equipment:

The UE has a valid P-TMSI-1, P-TMSI-1 signature and RAI-1.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode C Yes/No
UE operation mode A Yes/No
USIM removal possible without powering down Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The SS performs a detach with the cause value 'PS services not allowed'. The SS checks that the UE does not perform PS attach in another PLMN.

Step	Direction	Message	Comments
	UE SS		
			The following messages are sent and shall be
	00		received on cell A.
1	SS		The SS activates cell A.
2	UE		The UE is set in UE operation mode C (see
			ICS). If UE operation mode C not supported,
			goto step 19.
3	UE		The UE is powered up or switched on and
			initiates an attach (see ICS). Cell A is preferred
4		ATTACH REQUEST	by the UE. Attach type = 'PS attach'
4	->	ATTACTIKEQUEST	Mobile identity = P-TMSI-1
			P-TMSI-1 signature
			Routing area identity = RAI-1
5	<-	ATTACH ACCEPT	Attach result = 'PS only attached'
		ATTACTTACCETT	Mobile identity = P-TMSI-2
			P-TMSI-2 signature
			Routing area identity = RAI-1
6	->	ATTACH COMPLETE	Rodding area identity = 10 ti 1
7	<-	DETACH REQUEST	Detach type = 're-attach not required'
		DE INCITILE QUEUT	Cause = 'PS services not allowed'
8	->	DETACH ACCEPT	Sauce = 1 5 cervices fiet allewed
			The following messages are sent and shall be
			received on cell B.
9	SS		The SS deactivates cell A and activates cell B.
10	UE		Cell B is preferred by the UE.
11	UE		No ATTACH REQUEST sent to the SS
			(SS waits 30 seconds).
12	UE		If possible (see ICS) USIM removal is
			performed. Otherwise if possible (see ICS)
			switch off is performed. Otherwise the power is
			removed.
13	UE		The UE gets the USIM replaced, is powered up
			or switched on and initiates an attach (see
			ICS).
14	->	ATTACH REQUEST	Attach type = 'PS attach'
			Mobile identity = IMSI
15	<-	ATTACH ACCEPT	Attach result = 'PS only attached'
			Mobile identity = P-TMSI-1
			P-TMSI-1 signature
40		ATTACH COMPLETE	Routing area identity = RAI-2
16	-> !!E	ATTACH COMPLETE	The LIE is quitehed off or newer is removed
17	UE		The UE is switched off or power is removed (see ICS).
18		DETACH REQUEST	Message not sent if power is removed.
10	->	DETACTIVEQUEST	Detach type = 'power switched off, PS detach'
19			The SS deactivates cell B and activates cell A.
20	UE		The UE is set in UE operation mode A (see
20			ICS) and the test is repeated from step 3 to
			step 18.
	1	i .	O.OP O.

Specific message contents

None.

12.3.2.2.5 Test requirements

UE shall:

- initiate a PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.

After the completion of the PS attach procedure, and when UE receives the DETACH REQUEST message (Detach type = 're-attach not required', Cause = 'PS services not allowed') from SS, UE shall:

- delete the stored P-TMSI, P-TMSI signature, RAI and PS ciphering key sequence number.
- consider the USIM as invalid for PS service until power is switched off or USIM is removed.

12.3.2.3 PS detach / IMSI detach / accepted

12.3.2.3.1 Definition

12.3.2.3.2 Conformance requirement

The UE detach the IMSI for PS services.

Reference

3G3GPP-TS 24.008 clause 4.7.4.2

12.3.2.3.3 Test purpose

To test the behaviour of the UE for the detach procedure.

12.3.2.3.4 Method of test

Initial condition

System Simulator:

One cell operating in network operation mode I.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No
MMI controlled attach / detach procedures for non-PS services Yes/No

Test procedure

The UE performs a combined PS attach procedure (for PS and non-PS services).

The SS sends a DETACH REQUEST message to the UE. The UE then performs an IMSI detach (detach for non-PS services) .

The SS signal to the UE, but no response is received, as the signalling link is disconnected.

The UE attach for non-PS services by a routing area update procedure. Both PS and CS services are possible.

	Step	Direction UE SS	Message	Comments
	1	UE		The UE is set in UE operation mode A (see
	2	UE		ICS). The UE is powered up or switched on and initiates an attach (see ICS).
	3	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach' Mobile identity = IMSI
	4	<-	ATTACH ACCEPT	TMSI status = no valid TMSI available Attach result = 'Combined PS / IMSI attached' Mobile identity = P-TMSI-1 P-TMSI-1 signature Mobile identity = IMSI
	5 6	-> SS	ATTACH COMPLETE	Routing area identity = RAI-1 The SS initiates a detach for non-PS services.
	7 8	<- ->	DETACH REQUEST DETACH ACCEPT	Detach type = 'IMSI detach'
	9	<-	PACKET PAGING	Mobile identity = P-TMSI-1
	10	->	REQUESTPAGING TYPE1 UPLINK RLC DATA BLOCKSERVICE REQUEST	Paging order is for TBF establishment. LLC PDU implicitly indicating paging response.service type = "paging response"
ĺ	11	<-	PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = IMSI Paging order is for RRC-connection.
Ī	12	UE	FAGING REQUEST	The UE shall not initiate an RRC connection. This is checked during 3 seconds.
	13	UE		The UE initiates an attach for non-PS services (see ICS).
	14	->	ROUTING AREA UPDATE REQUEST	Update type = 'Combined RA/LA updating with IMSI attach' P-TMSI-1 signature Routing area identity = RAI-1
	15	<-	ROUTING AREA UPDATE ACCEPT	TMSI status = no valid TMSI available Update result = 'Combined RA/LA updating' Mobile identity = P-TMSI-2 P-TMSI-2 signature Mobile identity = TMSI-1
	16	->	ROUTING AREA UPDATE COMPLETE	Routing area identity = RAI-1
	17	<-	PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = TMSI-1 Paging order is for RRC-connection.
	18	->	CHANNEL REQUEST RRC CONNECTION REQUEST	aging cracino for rate commodicini
	19	<-	IMMEDIATE ASSIGNMENT RRC CONNECTION SETUP	
	<u>20</u>	<u>-></u>	RRC CONNECTION SETUP COMPLETE	
	20 21 21 22	-> <-	PAGING RESPONSE CHANNEL RELEASE RRC CONNECTION RELEASE	Mobile identity = TMSI-1 After sending of this message, the SS waits for disconnection of the CS signalling link.
	<u>23</u>	<u>-></u>	RRC CONNECTION RELEASE COMPLETE	
	22 24	UE	<u> </u>	The UE is switched off or power is removed (see ICS).
1	23 25	->	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, combined PS / IMSI detach'

Specific message contents

None.

12.3.2.3.5 Test requirements

UE shall:

- initiate a combined PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.

After the completion of the PS attach procedure, UE shall:

- receive DETACH REQUEST message(Detach type = 'IMSI detach') from SS.
- not deactivate the PDP context.
- and send the DETACH ACCEPT message to SS.

12.3.2.4 PS detach / re-attach requested / accepted

12.3.2.4.1 Definition

12.3.2.4.2 Conformance requirement

The UE shall deactivate the logical link and re-activate it.

Reference

3G3GPP-TS 24.008 clause 4.7.4.2

12.3.2.4.3 Test purpose

To test the behaviour of the UE for the detach procedure in case automatic re-attach.

12.3.2.4.4 Method of test

Initial condition

System Simulator:

One cell in operating in network operation mode I.

User Equipment:

The UE has a valid TMSI, P-TMSI, P-TMSI signature and RAI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE performs a combined PS attach procedure (for PS and non-PS services).

The SS sends a DETACH REQUEST message to the UE with cause re-attach. The UE then detach for PS and non-PS services. The UE automatically performs a new combined PS attach procedure (for PS and non-PS services) and PS and CS services are again possible.

Ī	Step	Direction	Message	Comments
		UE SS		
	1	UE		The UE is set in UE operation mode A (see ICS).
	2	UE		The UE is powered up or switched on and
	3	->	ATTACH REQUEST	initiates an attach (see ICS). Attach type = 'Combined PS / IMSI attach' Mobile identity = P-TMSI-1 P-TMSI-1 signature
	4	<-	ATTACH ACCEPT	Routing area identity = RAI-1 TMSI status = valid TMSI available Attach result = 'Combined PS / IMSI attached' Mobile identity = TMSI-1 Routing area identity = RAI-1 No new P-TMSI and P-TMSI signature assigned
	5	->	ATTACH COMPLETE	assigned
	6 7 8	SS <- ->	DETACH REQUEST DETACH ACCEPT	The SS initiates a detach with re-attach. Detach type = 're-attach required'
	9	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach'
	10	<-	ATTACH ACCEPT	Mobile identity = P-TMSI-1 P-TMSI-1 signature Mobile identity = TMSI-1 Routing area identity = RAI-1 TMSI status = valid TMSI available Attach result = 'Combined PS / IMSI attached' Mobile identity = TMSI-1 Mobile identity = P-TMSI-2 P-TMSI-2 signature Routing area identity = RAI-1
.	11	->	ATTACH COMPLETE	
	12	<-	PACKET PAGING REQUESTPAGING TYPE1	Mobile identity = P-TMSI-2 Paging order is for TBF establishment.
	13	->	UPLINK RLC DATA BLOCKSERVICE REQUEST	LLC PDU implicitly indicating paging response.service type = "paging response"
	14	<-	PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = TMSI-1 Paging order is for RRC-connection.
	15	->	CHANNEL REQUEST RRC CONNECTION REQUEST	r aging cross to for thice commodition.
	16	<-	IMMEDIATE ASSIGNMENT RRC CONNECTION SETUP	
	<u>17</u>	<u>-></u>	RRC CONNECTION SETUP COMPLETE	
	17 18 18 19	-> <-	PAGING RESPONSE CHANNEL RELEASE RRC CONNECTION RELEASE	Mobile identity = TMSI-1 After sending of this message, the SS waits for disconnection of the CS signalling link.
	<u>20</u>	<u>-></u>	RRC CONNECTION RELEASE COMPLETE	
	19 21	UE		The UE is switched off or power is removed (see ICS).
	20 22	->	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, combined PS / IMSI detach'

Specific message contents

None.

12.3.2.4.5 Test requirements

UE shall:

- initiate a combined PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.

After the completion of the combined PS attach procedure, UE shall:

- deactivate the PDP context and the logical link(s).
- send DETACH ACCEPT message to SS.

After UE completed PS detach procedure, UE shall:

- initiate a combined PS attach procedure.

12.3.2.5 PS detach / rejected / location area not allowed

12.3.2.5.1 Definition

12.3.2.5.2 Conformance requirement

- 1) If the network performs a PS detach procedure with the cause 'location area not allowed' the User Equipment shall:
 - 1.1 not perform combined PS attach when in the same location area.
 - 1.2 delete the stored LAI, CKSN, TMSI, RAI, PS-CKSN, P-TMSI and P-TMSI signature.
 - 1.3 store the LA in the 'forbidden location areas for regional provision of service'.
- 2) If the network performs a PS detach procedure with the cause 'location area not allowed' the User Equipment shall:
 - 2.1 perform combined PS attach when a new location area is entered.
 - 2.2 delete the list of forbidden LAs when power is switched off.

Reference

3G3GPP-TS 24.008 clauses 4.7.4.2

12.3.2.5.3 Test purpose

To test the behaviour of the UE if the network orders the PS detach procedure with the cause 'Location Area not allowed'.

To test that the UE deletes the list of forbidden LAs when power is switched off.

12.3.2.5.4 Method of test

Initial condition

System Simulator:

Three cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2, cell C in MCC1/MNC1/LAC2/RAC1. All cells are operating in network operation mode I.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No UE operation mode A Yes/No Switch off on button Yes/No Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The SS orders a PS detach with the cause value 'Location Area not allowed'. The SS checks that the UE does not perform combined PS attach while in the location area, performs PS attach when a new location area is entered and deletes the list of forbidden LAs when switched off. CS services are not possible unless an IMSI attach procedure is performed.

Different types of UE may use different methods to periodically clear the list of forbidden location areas (e.g. every day at 12am). If the list is cleared while the test is being run, it may be necessary to re-run the test.

	Step	Direction UE SS	Message	Comments
=		SS		The following messages are sent and shall be
	1	SS		received on cell A. The SS activates cell A.
	2	UE		The UE is set in UE operation mode A (see
	3	UE		ICS). The UE is powered up or switched on and
				initiates an attach (see ICS). Cell A is preferred
	4	->	ATTACH REQUEST	by the UE. Attach type = 'Combined PS / IMSI attach'
				Mobile identity = IMSI TMSI status = no valid TMSI available
	5	<-	ATTACH ACCEPT	Attach result = 'Combined PS / IMSI attached'
				Mobile identity = P-TMSI-1 P-TMSI-1 signature
				Mobile identity = TMSI-1
	6	->	ATTACH COMPLETE	Routing area identity = RAI-1
	7	<-	DETACH REQUEST	Detach type = 're-attach not required'
	8	->	DETACH COMPLETE	Cause 'Location Area not allowed'
-	9	UE		No LOCATION UPDATING REQ with type
				'IMSI attach' is sent to the SS (SS waits 30 seconds).
	10	<-	PAGING TYPE1PACKET	Mobile identity = IMSI
ļ	11	UE	PAGING REQUEST	Paging order is for RRC-connection. The UE shall not initiate an RRC connection.
	40		DA OINIO TYPE ADA OICET	This is checked during 3 seconds.
	12	<-	PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = P-TMSI-1 Paging order is for TBF establishment.
	13	->		No response from the UE to the request. This is checked for 10 seconds
-				The following messages are sent and shall be
	14	SS		received on cell B. The SS deactivates cell A and activates cell B.
	15	UE		Cell B is preferred by the UE.
	16	UE		The UE initiates an attach automatically, by MMI or by AT command.
	17	UE		No ATTACH REQUEST sent to SS
	18	UE		(SS waits 30 seconds) No LOCATION UPDATING REQ with type
				'IMSI attach' is sent to the SS
	19	<-	PAGING TYPE1PACKET	(SS waits 30 seconds). Mobile identity = IMSI
	20	UE	PAGING REQUEST	Paging order is for RRC-connection. The UE shall not initiate an RRC connection.
	20	OE		This is checked during 3 seconds.
	21	<-	PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = P-TMSI-1 Paging order is for TBF establishment.
1	22		TARITO REGOLOT	No response from the UE to the request.
-				This is checked for 10 seconds The following messages are sent and shall be
	0.0	0.5		received on cell C.
	23 24	SS UE		The SS deactivates cell B and activates cell C. Cell C is preferred by the UE.
	25	UE		The UE initiates an attach automatically, by
	26	->	ATTACH REQUEST	MMI or by AT command. Attach type = 'Combined PS / IMSI attach'
	-			Mobile identity = IMSI
I		I		TMSI status = no valid TMSI available

	27	<-	ATTACH ACCEPT	Attach result = 'Combined PS / IMSI attached' Mobile identity = P-TMSI1 P-TMSI-1 signature Mobile identity = TMSI-1 Routing area identity = RAI-3
	28	->	ATTACH COMPLETE	
	29	<-	PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = TMSI-1 Paging order is for RRC-connection.
	30	->	CHANNEL REQUEST RRC CONNECTION REQUEST	aging order to the time common in
	31	<-	IMMEDIATE ASSIGNMENT RRC	
	<u>32</u>	<u>-></u>	CONNECTION SETUP RRC CONNECTION SETUP COMPLETE	
	32 33 33 34	-> <-	PAGING RESPONSE CHANNEL RELEASE RRC CONNECTION RELEASE	Mobile identity = TMSI-1 After sending of this message, the SS waits for disconnection of the CS signalling link.
	<u>35</u>	<u>-></u>	RRC CONNECTION RELEASE COMPLETE	disconnection of the CS signalling link.
	34 <u>36</u>	<-	PACKET PAGING	Mobile identity = P-TMSI-1
	35 <u>37</u>	->	REQUESTPAGING TYPE1 UPLINK RLC DATA BLOCKSERVICE REQUEST	Paging order is for TBF establishment. LLC PDU implicitly indicating paging response.service type = "paging response"
1	36 38	UE		The UE is switched off or power is removed
	37 <u>39</u>	->	DETACH REQUEST	(see ICS). Message not sent if power is removed. Detach type = 'power switched off, combined PS / IMSI detach'
				The following messages are sent and shall be received on cell B.
	38 <u>40</u>	UE		The SS deactivates cell C and activates cell B. Cell B is preferred by the UE.
	39 41	UE		The UE is powered up or switched on and
I	4 <u>042</u>	->	ATTACH REQUEST	initiates an attach (see ICS). Attach type = 'Combined PS / IMSI attach' Mobile identity = P-TMSI-1
ĺ	41<u>43</u>	<-	ATTACH ACCEPT	P-TMSI-1 signature Routing area identity = RAI-3 TMSI status = valid TMSI available Attach result = 'Combined PS / IMSI attached' Mobile identity = P-TMSI-2 P-TMSI-2 signature Mobile identity = TMSI-2 Routing area identity = RAI-4
	42 <u>44</u>	-> <-	ATTACH COMPLETE PAGING TYPE1PACKET	Mobile identity = TMSI-2
	43 <u>45</u> 4446		PAGING TIPE I FACKET PAGING REQUEST CHANNEL REQUEST RRC	Paging order is for RRC-connection.
	_	->	CONNECTION REQUEST	
	45 <u>47</u>	<-	IMMEDIATE ASSIGNMENT RRC CONNECTION SETUP	
	<u>48</u>	<u>-></u>	RRC CONNECTION SETUP COMPLETE	
	4 <u>649</u> 4 7 50	-> <-	PAGING RESPONSE CHANNEL RELEASE RRC	Mobile identity = TMSI-2 After sending of this message, the SS waits for
	51		CONNECTION RELEASE RRC CONNECTION RELEASE	disconnection of the CS signalling link.
		<u>^</u>	COMPLETE	Mahila idantina B TMOLO
	48 <u>52</u>	<-	PACKET PAGING REQUESTPAGING TYPE1	Mobile identity = P-TMSI-2 Paging order is for TBF establishment.
	49 53	->	UPLINK RLC DATA BLOCKSERVICE REQUEST	LLC PDU implicitly indicating paging response.service type = "paging response"
	50 <u>54</u>	UE		The UE is switched off or power is removed (see ICS).

51 55	->	DETACH REQUEST	Message not sent if power is removed.
			Detach type = 'power switched off, combined
			PS / IMSI detach'

Specific message contents

None.

12.3.2.5.5 Test requirements

UE shall:

- initiate a PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.

After the completion of the PS attach procedure, and when UE receive the DETACH REQUEST message (Detach type = 're-attach not required', Cause = 'Location Area not allowed') from SS, UE shall:.

- perform the following action depending on UE location.

When in the same location area, UE shall:

- not perform combined PS attach.
- delete the stored RAI or LAI, P-TMSI, P-TMSI signature and PS ciphering key sequence number
- store the LA in the 'forbidden location areas for regional provision of service'.

When a new location area is entered, UE shall:

- perform combined PS attach.
- delete the list of forbidden LAs when power is switched off.

12.4 Routing area updating procedure

This procedure is used to update the actual routing area of an UE in the network.

12.4.1 Normal routing area updating

The routing area updating procedure is a GMM procedure used by PS UEs of UE operation mode A or C that are IMSI attached for PS services only.

12.4.1.1 Routing area updating / accepted

12.4.1.1.1 Definition

12.4.1.1.2 Conformance requirement

- 1) If the network accepts the routing area updating procedure and reallocates a P-TMSI, the UE shall acknowledge the new P-TMSI and continue communication with the new P-TMSI.
- 2) If the network accepts the routing area updating procedure from the UE without reallocation of the old P-TMSI, the UE shall continue communication with the old P-TMSI.

Reference

3G3GPP-TS 24.008 clause 4.7.5.1

12.4.1.1.3 Test purpose

To test the behaviour of the UE if the network accepts the routing area updating procedure.

The following cases are identified:

- 1) P-TMSI / P-TMSI signature is reallocated
- 2) Old P-TMSI / P-TMSI signature is not changed

12.4.1.1.4 Method of test

Initial condition

System Simulator:

Two cells, cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2. Both cells are operating in network operation mode II.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
UE operation mode C Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

- 1) The UE sends a ROUTING AREA UPDATE REQUEST message. The SS reallocates the P-TMSI and returns ROUTING AREA UPDATE ACCEPT message with a new P-TMSI. The UE acknowledge the new P-TMSI by sending ROUTING AREA UPDATING COMPLETE message. Further communication UE SS is performed by the new P-TMSI. The UE will not answer signalling addressed to the old P-TMSI.
- 2) The UE sends a ROUTING AREA UPDATING REQUEST message. The SS accepts the P-TMSI and returns ROUTING AREA UPDATING ACCEPT message without any P-TMSI. Further communication UE SS is performed by the P-TMSI.

ſ	Step	Direction UE SS	Message	Comments
-		UE SS		The following messages are sent and shall be
	4	00		received on cell A.
	1 2	SS UE		The SS activates cell A. The UE is set in UE operation mode C (see
	2	OL		ICS). If UE operation mode C not supported,
	_			goto step 22.
	3	UE		The UE is powered up or switched on and initiates an attach (see ICS).
	4	->	ATTACH REQUEST	Attach type = 'PS attach'
	_			Mobile identity = IMSI
	5	<-	ATTACH ACCEPT	Attach result = 'PS only attached' Mobile identity = P-TMSI-2
				P-TMSI-2 signature
	_			Routing area identity = RAI-1
_	6	->	ATTACH COMPLETE	The following messages are sent and shall be
				received on cell B.
	7	SS		Activate cell B with a lower signal strength than
				cell A The RF level of cell A is lowered until cell B is preferred by the UE.
	8	->	ROUTING AREA UPDATING	Update type = 'RA updating'
			REQUEST	P-TMSI-2 signature
	9	<-	ROUTING AREA UPDATING	Routing area identity = RAI-1 Update result = 'RA updated'
	9	ζ-	ACCEPT	Mobile identity = P-TMSI-1
				P-TMSI-1 signature
	10	->	ROUTING AREA UPDATING	Routing area identity = RAI-4
	10	->	COMPLETE	
	11	<-	GMM INFORMATION	Message sent with P-TMSI-1
	11b	->	GMM STATUS	Message sent in case the UE does not support reception of GMM information message
				Cause #97
	12	<-	PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = P-TMSI-2
			Of	PAGING TYPE1 Both paging orders are for TBF establishment.
			PAGING REQUEST TYPE 1	PACKET PAGING REQUEST (used for NW
				mode III) PAGING REQUEST TYPE 1-(used for NW-
				mode II).
	13	UE		No response from the UE to the request. This
-				is checked for 10 seconds. The following messages are sent and shall be
				received on cell A.
	14	SS		Set the signal strength of cell A to a lower
				signal strength than cell B The RF level of cell B is lowered until cell A is preferred by the UE.
	15	UE		Cell A is preferred by the UE.
	16	->	ROUTING AREA UPDATING REQUEST	Update type = 'RA updating'
			KEQUEST	P-TMSI-1 signature Routing area identity = RAI-4
	17	<-	ROUTING AREA UPDATING	No new mobile identity assigned.
			ACCEPT	P-TMSI not included. Update result = 'RA updated'
				P-TMSI-2 signature
				Routing area identity = RAI-1
	18	<-	PACKET PAGING REQUEST or	Mobile identity = P-TMSI-1 Both paging orders are for TBF establishment.
			PAGING REQUEST TYPE	PACKET PAGING REQUEST (used for NW
			4PAGING TYPE1	mode III)
				PAGING REQUEST TYPE 1 (used for NW-mode II).
ı	l		I .	

19	->	UPLINK RLC DATA BLOCKSERVICE REQUEST	LLC PDU implicitly indicating paging response.service type = "paging response"
20	UE		The UE is switched off or power is removed (see ICS).
21	->	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, PS detach'
22	UE		The UE is set in UE operation mode A (see ICS) and the test is repeated from step 3 to
			step 21.

Specific message contents

None

12.4.1.1.5 Test requirements

UE shall:

- initiate a PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.

After completing the PS attach procedure, UE shall:,

- initiate a routing area updating procedure with the information elements specified in the above Expected Sequence when the RF level of the attached cell is lower than the RF level of the new cell.
- use the P-TMSI which is included in the ROUTING AREA UPDATING ACCEPT message.
- acknowledge the new P-TMSI and continue communication with the new P-TMSI.
- continue communication with the old P-TMSI.

12.4.1.2 Routing area updating / rejected / IMSI invalid / illegal ME

12.4.1.2.1 Definition

12.4.1.2.2 Conformance requirement

- If the network rejects a routing area updating procedure from the User Equipment with the cause 'Illegal ME',
 the User Equipment shall consider USIM invalid for PS services until power is switched off or USIM is
 removed.
- 2) If the network rejects a routing area updating procedure from the User Equipment with the cause 'Illegal ME', the User Equipment shall delete the stored RAI, PS-CKSN, P-TMSI and P-TMSI signature.

Reference

3G-3GPP TS 24.008 clause 4.7.5.1

12.4.1.2.3 Test purpose

To test the behaviour of the UE if the network rejects the routing area updating procedure of the UE with the cause 'Illegal ME'.

12.4.1.2.4 Method of test

Initial condition

System Simulator:

Three cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2, cell C in MCC2/MNC1/LAC1/RAC1.

All three cells are operating in network operation mode II (in case of UE operation mode A), or in network operation mode III (in case of UE operation mode C).

User Equipment:

The UE has a valid P-TMSI-1, P-TMSI-1 signature and RAI-1.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode C Yes/No
UE operation mode A Yes/No (only if mode C not supported)
USIM removal possible without powering down Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The SS rejects a routing area updating with the cause value 'Illegal ME'. The SS checks that the UE does not perform PS attach in the same or another PLMN.

Ī	Step	Direction	Message	Comments
ļ		UE SS		The fellowing property and should be
				The following messages are sent and shall be received on cell A.
ĺ	1	UE		The UE is set in UE operation mode C or B
				(see ICS).
	2	SS		The SS is set in network operation mode II-or
]	3	UE		## and activates cell A. The UE is powered up or switched on and
	3	OL		initiates an attach (see ICS). Cell A is preferred
				by the UE.
	4	->	ATTACH REQUEST	Attach type = 'PS attach'
				Mobile identity = P-TMSI-1 P-TMSI-1 signature
				Routing area identity = RAI-1
	5	<-	ATTACH ACCEPT	No new mobile identity assigned.P-TMSI and
				P-TMSI signature not included.Attach result =
				'PS only attached'
ŀ				Routing area identity = RAI-1 The following messages are sent and shall be
				received on cell B.
	6	SS		The SS deactivates cell A and activates cell B.
	7	UE	DOLITING ADEA LIDDATING	Cell B is preferred by the UE.
	8	->	ROUTING AREA UPDATING REQUEST	Update type = 'RA updating' P-TMSI-2 signature
			1120201	Routing area identity = RAI-1
	9	<-	ROUTING AREA UPDATING	GMM cause = 'Illegal UE'
ı	10		REJECT	Mahila idantity D TMCL 2
	10	<-	PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = P-TMSI-2 PAGING TYPE1 Both paging orders are for
			Or	TBF establishment.
			PAGING REQUEST TYPE 1	PACKET PAGING REQUEST (used for NW
				mode III) PAGING REQUEST TYPE 1 (used for NW-
1				mode II).
	11	UE		No response from the UE to the request. This
ŀ				is checked for 10 seconds. The following messages are sent and shall be
				received on cell C.
	12	SS		The SS deactivates cell B and activates cell C.
	13	UE		Cell C is preferred by the UE.
	14	UE		No ATTACH REQUEST sent to the SS (SS waits 30 seconds).
	15	UE		If possible (see ICS) USIM removal is
				performed. Otherwise if possible (see ICS)
				switch off is performed. Otherwise the power is
	16	UE		removed. The UE gets the USIM replaced, is powered up
	10	OL		or switched on and initiates an attach (see
				ICS).
	17	->	ATTACH REQUEST	Attach type = 'PS only attached'
	18	<-	ATTACH ACCEPT	Mobile identity = IMSI Attach result = 'PS attach'
	10	\-	ATTAOLI AOOLI T	Mobile identity = P-TMSI-1
				P-TMSI-1 signature
	40		ATTACLL COMPLETE	Routing area identity = RAI-2
	19 20	-> UE	ATTACH COMPLETE	The UE is switched off or power is removed
	20	OL.		(see ICS).
, [21	->	DETACH REQUEST	Message not sent if power is removed.

Specific message contents

None.

12.4.1.2.5 Test requirements

UE shall:

- initiate a PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.

After completing the PS attach procedure, UE shall:

- initiate a routing area updating procedure with the information elements specified in the above Expected Sequence.
- delete the stored P-TMSI, P-TMSI signature, RAI and PS ciphering key sequence number .
- consider the USIM as invalid for PS services until the UE is switched off or the USIM is removed.

12.4.1.3 Routing area updating / rejected / UE identity cannot be derived by the network

12.4.1.3.1 Definition

12.4.1.3.2 Conformance requirement

If the network rejects a routing area updating procedure from the User Equipment with the cause 'UE identity cannot be derived by the network', the User Equipment shall delete the stored RAI, PS-CKSN, P-TMSI and P-TMSI signature.

Depending on the manufacturer the UE may or may not perform a PS attach procedure.

Reference

3G3GPP -TS 24.008 clause 4.7.5.1

12.4.1.3.3 Test purpose

To test the behaviour of the UE if the network rejects the routing area updating procedure of the UE with the cause 'UE identity cannot be derived by the network'.

12.4.1.3.4 Method of test

Initial condition

System Simulator:

Two cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2.

Both cells are operating in network operation mode II (in case of UE operation mode A), or in network operation mode III (in case of UE operation mode C).

User Equipment:

The UE has a valid P-TMSI-1, P-TMSI-1 signature and RAI-1.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode C Yes/No
UE operation mode A Yes/No (only if mode C not supported)
Automatic attach procedure when UE identity cannot be derived by the network Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The SS rejects a normal routing area updating with the cause value 'UE identity cannot be derived by the network'. The UE detach locally. A new PS attach may be performed.

Ī	Step	Direction	Message	Comments
Ĺ		UE SS		
				The following messages are sent and shall be
.	4	00		received on cell A.
	1	SS		The SS is set in network operation mode II_or III_and activates cell A.
	2	UE		The UE is set in UE operation mode C or B
	2	OL		(see ICS).
	3	UE		The UE is powered up or switched on and
				initiates an attach (see ICS). Cell A is preferred
				by the UE.
	4	->	ATTACH REQUEST	Attach type = 'PS attach'
				Mobile identity =P-TMSI-1
				P-TMSI-1 signature Routing area identity = RAI-1
	5	<-	ATTACH ACCEPT	Attach result = 'PS only attached'
	3	~	ATTACITACCETT	Mobile identity = P-TMSI-2
				P-TMSI-2 signature
				Routing area identity = RAI-1
	6	->	ATTACH COMPLETE	· ·
				The following messages are sent and shall be
	_			received on cell B.
	7	SS		The SS deactivates cell A and activates cell B.
	8 9	UE	ROUTING AREA UPDATING	Cell B is preferred by the UE. Update type = 'RA updating'
	9	->	REQUEST	P-TMSI-2 signature
			TREGOLO!	Routing area identity = RAI-1
	10	<-	ROUTING AREA UPDATING	GMM cause = 'UE identity cannot be derived
			REJECT	by the network'
	11	UE		If an automatic attach procedure by the UE is
				not possible when the UE identity cannot be
	40			derived by the network (see ICS) goto step 19.
	12	UE		An Automatic PS attach procedure is initiated (see ICS).
	13	->	ATTACH REQUEST	Attach type = 'PS attach'
	10		MINOTINEGOLOT	Mobile identity = IMSI
	14	<-	ATTACH ACCEPT	Attach result = 'PS only attached'
				Mobile identity = P-TMSI-1
				P-TMSI-1 signature
	4		4.TT4.011.0014B1.ETE	Routing area identity = RAI-4
ŀ	15	->	ATTACH COMPLETE	The LIE is quitehed off or newer is removed
	16	UE		The UE is switched off or power is removed (see ICS).
	17	->	DETACH REQUEST	Message not sent if power is removed.
	.,		DETAGNIKEQUEUT	Detach type = 'power switched off, PS detach'
	18			Stop the sequence
lĺ	19	<-	PAGING TYPE1PACKET	Mobile identity = P-TMSI-2
			PAGING REQUEST	PAGING TYPE1 Both paging orders are for
			Or	TBF establishment.
			PAGING REQUEST TYPE 1	PACKET PAGING REQUEST (used for NW
				mode III) PAGING REQUEST TYPE 1-(used for NW-
				mode II).
	20	UE		No response from the UE to the request, as the
		0_		UE has detached locally. This is checked for 10
				seconds.

Specific message contents

None.

12.4.1.3.5 Test requirements

UE shall:

initiate a PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.

After completing the PS attach procedure, UE shall:

- initiate a routing area updating procedure with the information elements specified in the above Expected Sequence.
- delete stored P-TMSI, P-TMSI signature, RAI and PS ciphering key sequence number.

12.4.1.4 Routing area updating / rejected / location area not allowed

12.4.1.4.1 Definition

12.4.1.4.2 Conformance requirement

- 1) If the network rejects a routing area updating procedure from the User Equipment with the cause 'location area not allowed' the User Equipment shall:
 - 1.1 not perform PS attach when in the same location area.
 - 1.2 delete the stored RAI, PS-CKSN, P-TMSI and P-TMSI signature.
 - 1.3 store the LA in the 'forbidden location areas for regional provision of service'.
- 2) If the network rejects a routing area updating procedure from the User Equipment with the cause 'location area not allowed' the User Equipment shall:
 - 2.1 perform PS attach when a new location area is entered.
 - 2.2 delete the list of forbidden LAs after switch off (power off).

Reference

3G3GPP-TS 24.008 clauses 4.7.5.1

12.4.1.4.3 Test purpose

To test the behaviour of the UE if the network rejects the routing area updating procedure of the UE with the cause 'Location Area not allowed'.

To test that the UE deletes the list of forbidden LAs when power is switched off.

12.4.1.4.4 Method of test

Initial condition

System Simulator:

Three cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2, cell C in MCC1/MNC1/LAC2/RAC1.

All cells are operating in network operation mode IIII.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
UE operation mode C Yes/No
USIM removal possible without powering down Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The SS rejects a routing area updating with the cause value 'Location Area not allowed'. The SS checks that the UE does not perform PS attach while in the location area, performs PS attach when a new location area is entered and deletes the list of forbidden LAs when switched off.

Different types of UE may use different methods to periodically clear the list of forbidden location areas (e.g. every day at 12am). If the list is cleared while the test is being run, it may be necessary to re-run the test.

Ī	Step	Direction UE SS	Message	Comments
f		SS		The following messages are sent and shall be
				received on cell C.
	1	SS UE		The SS activates cell C.
	2	UE		The UE is set in UE operation mode C (see ICS). If UE operation mode C not supported,
				goto step 33.
	3	UE		The UE is powered up or switched on and
				initiates an attach (see ICS). Cell C is preferred
	4	_	ATTACH REQUEST	by the UE. Attach type = 'PS attach'
	4	->	ATTACTI REQUEST	Mobile identity = IMSI
	5	<-	ATTACH ACCEPT	Attach result = 'PS only attached'
				Mobile identity = P-TMSI-1
				P-TMSI-1 signature
	6	->	ATTACH COMPLETE	Routing area identity = RAI-3
F			ATTACK COM LETE	The following messages are sent and shall be
				received on cell B.
	7	00		The SS deactivates cell C and activates cell B.
	8 9	SS ->	 ROUTING AREA UPDATING	Cell B is preferred by the UE. Update type = 'RA updating'
	3		REQUEST	P-TMSI-1 signature
				Routing area identity = RAI-3
	10	<-	ROUTING AREA UPDATING	GMM cause = 'Location Area not allowed'
ı	11	<-	REJECT PAGING TYPE1PACKET	Mobile identity = P-TMSI-1
			PAGING REQUEST	PAGING TYPE1 Both paging orders are for
			Or	TBF establishment.
			PAGING REQUEST TYPE 1	PACKET PAGING REQUEST (used for NW
				mode III) PAGING REQUEST TYPE 1 (used for NW-
1				mode II).
	12	UE		No response from the UE to the request. This
-				is checked for 10 seconds. The following messages are sent and shall be
				received on cell A.
	13	SS		The SS deactivates cell B and activates cell A.
	14	UE		Cell A is preferred by the UE.
	15	UE		No ATTACH REQUEST sent to SS (SS waits 30 seconds)
-				The following messages are sent and shall be
				received on cell C.
	16	SS		The SS deactivates cell B and activates cell C.
	17 18	UE	ATTACH REQUEST	Cell C is preferred by the UE. Attach type = 'PS attach'
	10	->	ATTACHNEQUEST	Mobile identity = IMSI
	19	<-	ATTACH ACCEPT	Attach result = 'PS only attached'
				Mobile identity = P-TMSI-2
				P-TMSI-2 signature Routing area identity = RAI-3
	20	->	ATTACH COMPLETE	Trouting area lucititity = RAI-3
	21	UÉ		If possible (see ICS) USIM removal is
				performed. Otherwise if possible (see ICS)
				switch off is performed. Otherwise the power is
	22	->	DETACH REQUEST	removed. Message not sent if power is removed.
			DE MONTRE QUEUT	Detach type = 'power switched off, PS detach'
Ī	23	UE		The UE gets the USIM replaced, is powered up
				or switched on and initiates an attach (see
I			I	ICS).

24	->	ATTACH REQUEST	Attach type = 'PS attach' Mobile identity = P-TMSI-2
25	<-	ATTACH ACCEPT	P-TMSI-2 signature Routing area identity = RAI-3 Attach result = 'PS only attached' Mobile identity = P-TMSI-1 P-TMSI-1 signature Routing area identity = RAI-3
26	->	ATTACH COMPLETE	
	SS		The following messages are sent and shall be received on cell A.
27			The SS deactivates cell C and activates cell A.
28			Cell A is preferred by the UE.
29	->	ROUTING AREA UPDATING	Update type = 'RA updating'
		REQUEST	P-TMSI-1 signature
00		DOLITING ADEA LIBBATING	Routing area identity = RAI-3
30	<-	ROUTING AREA UPDATING ACCEPT	No new mobile identity assigned.P-TMSI and P-TMSI signature not included.Update result =
		7.0021 1	'RA updated'
			P-TMSI-2 signature
			Routing area identity = RAI-1
31	UE		The UE is switched off or power is removed
			(see ICS).
32	->	DETACH REQUEST	Message not sent if power is removed.
			Detach type = 'power switched off, PS detach'
33	SS		The SS is set in network operation mode II.
34	UE		The UE is set in UE operation mode A (see
			ICS) and the test is repeated from step 3 to
			step 32.

Specific message contents

None.

12.4.1.4.5 Test requirements

UE shall:

- initiate a PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.

After completing the PS attach procedure, UE shall:

- initiate a routing area updating procedure with the information elements specified in the above Expected Sequence.

When in the same location area, UE shall

- not perform PS attach..
- store the LA in the 'forbidden location areas for regional provision of service'.

When a new location area is entered, UE shall

- perform PS attach when a.
- delete the list of forbidden LAs when power is switched off.

12.4.1.5 Routing area updating / abnormal cases / attempt counter check / miscellaneous reject causes

12.4.1.5.1 Definition

12.4.1.5.2 Conformance requirement

When a routing area updating procedure is rejected with the attempt counter less than five, the UE shall repeat the routing area updating procedure after T3330 timeout.

When a T3330 timeout has occurred during a routing area updating procedure with the attempt counter five, the UE shall start timer T3302.

When the T3302 expire, a new routing area updating procedure shall be initiated.

GMM cause codes that can be selected are:

'IMSI unknown in HLR'

'IMEI not accepted'

'Illegal ME'

'UE identity cannot be derived by the network'

'Network failure'

'Congestion'

'retry upon entry into a new cell'

'Semantically incorrect message'

'Invalid mandatory information'

'Message type non-existent or not implemented'

'Message type not compatible with the protocol state'

'Information element non-existent or not implemented'

'Conditional IE error'

'Message not compatible with the protocol state'

'Protocol error, unspecified'

Reference

3G3GPP-TS 24.008 clause 4.7.5.1

12.4.1.5.3 Test purpose

To test the behaviour of the UE with respect to the attempt counter.

12.4.1.5.4 Method of test

Initial condition

System Simulator:

Two cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2.

Both cells are operating in network operation mode II (in case of UE operation mode A), or in network operation mode III (in case of UE operation mode C).

User Equipment:

The UE has a valid P-TMSI-1, P-TMSI-1 signature and RAI-1.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode C Yes/No
UE operation mode A Yes/No (only if mode C not supported)
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE initiates a routing area updating procedure (attempt counter zero).

The SS rejects the routing area updating procedure with a random cause code.

The UE initiates a new routing area updating procedure (attempt counter one) after T3311 expires.

The SS rejects the routing area updating procedure with a random cause code.

The UE initiates a new routing area updating procedure (attempt counter two) after T3311 expires.

The SS rejects the routing area updating procedure with a random cause code.

The UE initiates a new routing area updating procedure (attempt counter three) after T3311 expires.

The SS rejects the routing area updating procedure with a random cause code.

The UE initiates a new routing area updating procedure (attempt counter four) after T3311 expires.

The SS rejects the routing area updating procedure with a random cause code.

The UE initiates a new routing area updating procedure with attempt counter five (after T3311 expires).

The SS rejects the routing area updating procedure with a random cause code.

The UE shall not perform a new successful routing area updating procedure after T3311 seconds.

The UE initiates a routing area updating procedure with attempt counter zero after T3302 expires with the stored P-TMSI, P-TMSI signature, PS CKSN and RAI.

T3302; set to 10 minutes.

T3330; 15 seconds.

Ī	Step	Direction	Message	Comments
-		UE SS SS		The following messages are sent and shall be
		33		received on cell A.
	1	UE		The UE is set in UE operation mode C or B
1	2	SS		(see ICS). The SS is set in network operation mode II-or
	_			₩ and activates cell A.
	3	UE		The UE is powered up or switched on and initiates an attach (see ICS). Cell A is preferred
				by the UE.
	4	->	ATTACH REQUEST	Attach type = 'PS attach' Mobile identity = P-TMSI-1
				P-TMSI-1 signature
	5	<-	ATTACH ACCEPT	Routing area identity = RAI-1 No new mobile identity assigned.
	3	~	ATTACITACCETT	P-TMSI not included.
				Attach result = 'PS only attached'
				P-TMSI-2 signature Routing area identity = RAI-1
				The following messages are sent and shall be
	6	SS		received on cell B. The SS deactivates cell A and activates cell B.
	7	SS		Cell B is preferred by the UE.
	8	->	ROUTING AREA UPDATING REQUEST	Update type = 'RA updating' P-TMSI-2 signature
				Routing area identity = RAI-1
	9	<-	ROUTING AREA UPDATING REJECT	Random GMM cause
	10	SS		The SS verifies that the time between the
	11	->	ROUTING AREA UPDATING	routing area updating requests is 15 seconds Update type = 'RA updating'
			REQUEST	
				P-TMSI-2 signature Routing area identity = RAI-1
	12	<-	ROUTING AREA UPDATING	Random GMM cause
	13	SS	REJECT	The SS verifies that the time between the
	15	30		routing area updating requests is 15 seconds
	14	->	ROUTING AREA UPDATING REQUEST	Update type = 'RA updating'
			REGUEST	P-TMSI-2 signature
	15		ROUTING AREA UPDATING	Routing area identity = RAI-1 Random GMM cause
	15	<-	REJECT	Random Givilvi cause
	16	SS		The SS verifies that the time between the
	17	->	ROUTING AREA UPDATING	routing area updating requests is 15 seconds Update type = 'RA updating'
			REQUEST	P-TMSI-2 signature
	18	<-	ROUTING AREA UPDATING	Routing area identity = RAI-1 Random GMM cause
			REJECT	
	19	SS		The SS verifies that the time between the routing area updating requests is 15 seconds
	20	->	ROUTING AREA UPDATING	Update type = 'RA updating'
			REQUEST	P-TMSI-2 signature
				Routing area identity = RAI-1
	21	<-	ROUTING AREA UPDATING REJECT	Random GMM cause
	22	SS	1.20201	The SS verifies that the UE does not attempt to
-	22	SS		attach for 10 minutes . The SS shall release the PS signalling
	23	33		connection.

24	->	ROUTING AREA UPDATING REQUEST	Update type = 'RA updating'
25	<-	ROUTING AREA UPDATING ACCEPT	P-TMSI-2 signature Routing area identity = RAI-1 Update result = 'RA updated' Mobile identity = P-TMSI-2 P-TMSI-3 signature Routing area identity = RAI-4
26	->	ROUTING AREA UPDATING COMPLETE	l l l l l l l l l l l l l l l l l l l
27	UE		The UE is switched off or power is removed (see ICS).
28	->	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, PS detach'

Specific message contents

None.

12.4.1.5.5 Test requirements

UE shall:

- initiate a PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.

After completing the PS attach procedure, UE shall:

- perform the following actions depending on the conditions described below.

Case 1) A routing area updating procedure is rejected from SS with the attempt counter less than five

UE shall:

- repeat the routing area updating procedure after T3330 timeout

Case2) A timer T3330 timeout has occurred during a routing area updating procedure with the attempt counter five

UE shall:

- start timer T3302

Case3) The T3302 expires

UE shall:

- initiate a new routing area updating procedure

12.4.1.6 Routing area updating / abnormal cases / change of cell into new routing area

12.4.1.6.1 Definition

12.4.1.6.2 Conformance requirement

When a change of cell into a new routing area is performed before the routing area updating procedure is finished, the UE shall abort the routing area updating procedure and re-initiate it in the new routing area.

Reference

3G3GPP-TS 24.008 clause 4.7.5.1

12.4.1.6.3 Test purpose

To test the behaviour of the UE in case of procedure collision.

12.4.1.6.4 Method of test

Initial condition

System Simulator:

Three cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2 and cell C In MCC1/MNC1/LAC1/RAC3.

All cells are operating in network operation mode II (in case of UE operation mode A), or in network operation mode III (in case of UE operation mode C).

User Equipment:

The UE has a valid P-TMSI-1, P-TMSI-1 signature and RAI-1.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode C Yes/No
UE operation mode A Yes/No (only if mode C not supported)
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE initiates a routing area updating procedure. The ROUTING AREA UPDATE ACCEPT message is delayed from the SS. The UE performs a cell update into a new routing area. The UE shall re-initiate a routing area updating procedure in the new routing area.

Step	Direction	Message	Comments
	UE SS		The fellowing are seen and and about he
	SS		The following messages are sent and shall be received on cell A.
1	SS		The SS activates cell A.
2	UE		The UE is set in UE operation mode C (see
			ICS). If UE operation mode C not supported,
			goto step 18.
3	UE		The UE is powered up or switched on and
			initiates an attach (see ICS). Cell A is preferred
		ATTACH REQUEST	by the UE.
4	->	ATTACH REQUEST	Attach result = 'PS attach'
			Mobile identity = P-TMSI-1 P-TMSI-1 signature
			Routing area identity = RAI-1
5	<-	ATTACH ACCEPT	Attach result = 'PS only attached'
			Mobile identity = P-TMSI-2
			P-TMSI-2 signature
			Routing area identity = RAI-1
6	->	ATTACH COMPLETE	
			The following messages are sent and shall be
7	SS		received on cell B. The SS deactivates cell A and activates cell B.
8	SS		Cell B is preferred by the UE.
9	->	ROUTING AREA UPDATING	Update type = 'RA updating'
	,	REQUEST	P-TMSI-2 signature
			Routing area identity = RAI-1
10	SS		No response to the ROUTING AREA
			UPDATING REQUEST message is given by
			the SS
			The following messages are sent and shall be received on cell C.
11	SS		The SS deactivates cell B and activates cell C.
12	SS		Cell C is preferred by the UE.
13	->	ROUTING AREA UPDATING	Update type = 'RA updating'
		REQUEST	P-TMSI-2 signature
			Routing area identity = RAI-1
14	<-	ROUTING AREA UPDATING	Update result = 'RA updated'
1		ACCEPT	Mobile identity = P-TMSI-2
1			P-TMSI-3 signature Routing area identity = RAI-5
15	->	ROUTING AREA UPDATING	Routing area identity = RAI-3
'		COMPLETE	
16	UE		The UE is switched off or power is removed
1			(see ICS).
17	->	DETACH REQUEST	Message not sent if power is removed.
<u></u>	0.5		Detach type = 'power switched off, PS detach'
18	SS		The SS is set in network operation mode II.
19	UE		The UE is set in UE operation mode A(see ICS) and the test is repeated from step 3 to
1			step 17.
	l	<u> </u>	step 17.

Specific message contents

None.

12.4.1.6.5 Test requirements

UE shall:

- initiate a PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.

After completing the PS attach procedure, UE shall:

- initiate the routing area update procedure.

When change of cell into a new routing area is performed before the routing area updating procedure is finished, UE shall:

- abort the routing area updating procedure.
- re-initiate new routing area updating procedure in the new routing area.

12.4.1.7 Routing area updating / abnormal cases / change of cell during routing area updating procedure

12.4.1.7.1 Definition

12.4.1.7.2 Conformance requirement

When a change of cell within a new routing area is performed before the routing area updating procedure is finished, the UE shall perform the cell update before the routing area updating procedure is finished.

Reference

3G3GPP-TS 24.008 clause 4.7.5.1

12.4.1.7.3 Test purpose

To test the behaviour of the UE in case of procedure collision.

12.4.1.7.4 Method of test

Initial condition

System Simulator:

Three cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2 and cell C in MCC1/MNC1/LAC1/RAC2. All three cells are operating in network operation mode IIHH.

User Equipment:

The UE has a valid P-TMSI-1, P-TMSI-1 signature and RAI-1.

Related ICS/IXIT statements

Support of PS service Yes/No UE operation mode C Yes/No UE operation mode A Yes/No Switch off on button Yes/No

Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE initiates a routing area updating procedure. The ROUTING AREA UPDATE ACCEPT message is delayed from the SS. The UE performs a cell update within the routing area. The UE then waits for the ROUTING AREA UPDATE ACCEPT message.

Expected Sequence

Γ	Step	Direction		Message	Comments
	2.26	UE	SS		
F		SS			The following messages are sent and shall be
					received on cell A.
	1	UE	=		The UE is set in UE operation mode C or B
	•	-			(see ICS).
	2	SS	3		The SS is set in network operation mode II-or
					## and activates cell A.
	3	UE	Ē		The UE is powered up or switched on and
					initiates an attach (see ICS). Cell A is preferred
					by the UE.
	4	->	>	ATTACH REQUEST	Attach result = 'PS attach'
					Mobile identity = P-TMSI-1
					P-TMSI-1 signature
					Routing area identity = RAI-1
	5	<	-	ATTACH ACCEPT	No new mobile identity assigned.
					P-TMSI not included.
					Attach result = 'PS only attached'
					P-TMSI-2 signature
Ļ					Routing area identity = RAI-1
					The following messages are sent and shall be
		0.0			received on cell B.
	6	SS			The SS deactivates cell A and activates cell B.
	7	SS		DOLITING ADEA LIDDATING	Cell B is preferred by the UE.
	8	->	>	ROUTING AREA UPDATING REQUEST	Update type = 'RA updating'
				REQUEST	P-TMSI-2 signature Routing area identity = RAI-1
	9	SS	,		No response to the ROUTING AREA
	9	33	•		UPDATING REQUEST message is given by
					the SS
ŀ					The following messages are sent and shall be
					received on cell C.
	10	SS	3		The SS deactivates cell B and activates cell C.
	11	SS			Cell C is preferred by the UE.
	12	->	>	UPLINK RLC DATA BLOCK	LLC PDU implicitly indicating cell update.
					1 7 3 1
	13	<	-	ROUTING AREA UPDATING	Update result = 'RA updated'
				ACCEPT	Mobile identity = P-TMSI-2
					P-TMSI-3 signature
					Routing area identity = RAI-4
	14	->	>	ROUTING AREA UPDATING	
				COMPLETE	
	15	UE			The UE is switched off or power is removed
					(see ICS).
	16	->	>	DETACH REQUEST	Message not sent if power is removed.
L					Detach type = 'power switched off, PS detach'

Specific message contents

None.

12.4.1.7.5 Test requirements

UE shall:

- initiate a PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.

After completing the PS attach procedure, UE shall:

- initiate routing area update procedure.

When a change of cell within a new routing area is performed, UE shall:

- perform the cell update before the routing area updating procedure is finished.

12.4.1.8 Routing area updating / abnormal cases / P-TMSI reallocation procedure collision

12.4.1.8.1 Definition

12.4.1.8.2 Conformance requirement

When a P-TMSI REALLOCATION REQUEST message is received by the UE while waiting for a ROUTING AREA UPDATE ACCEPT message, the UE shall ignore the P-TMSI reallocation procedure and continue with the routing area updating procedure.

Reference

3G3GPP-TS 24.008 clause 4.7.5.1

12.4.1.8.3 Test purpose

To test the behaviour of the UE in case of procedure collision.

12.4.1.8.4 Method of test

Initial condition

System Simulator:

Two cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1 and cell B in MCC1/MNC1/LAC1/RAC2.

Both cells are operating in network operation mode II (in case of UE operation mode A), or in network operation mode III (in case of UE operation mode C).

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode C Yes/No
UE operation mode A Yes/No (only if mode C not supported)
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE initiates a routing area updating procedure. The SS does not answer the routing area updating procedure, but initiates a P-TMSI reallocation procedure and continue with the routing area updating procedure.

ſ	Step	Direction	Message	Comments
	-	UE SS	_	
		SS		The following messages are sent and shall be received on cell A.
	1	UE		The UE is set in UE operation mode C or B (see ICS).
	2	SS		The SS is set in network operation mode II-or III and activates cell A.
	3	UE		The UE is powered up or switched on and initiates an attach (see ICS). Cell A is preferred by the UE.
	4	->	ATTACH REQUEST	Attach result = 'PS attach' Mobile identity = IMSI
	5	<-	ATTACH ACCEPT	Attach result = 'PS only attached' Mobile identity = P-TMSI-1 P-TMSI-1 signature
	6	->	ATTACH COMPLETE	Routing area identity = RAI-1
				The following messages are sent and shall be received on cell B.
	7	SS		The SS deactivates cell A and activates cell B.
	8	SS		Cell B is preferred by the UE.
	9	->	ROUTING AREA UPDATING REQUEST	Update type = 'RA updating' P-TMSI-1 signature
	10	<-	P-TMSI REALLOCATION REQUEST	Routing area identity = RAI-1 Mobile identity = P-TMSI-1 P-TMSI-1 signature Routing area identity = RAI-1
	11	UE		The UE ignores the P-TMSI reallocation request.
	12	<-	ROUTING AREA UPDATING ACCEPT	Update result = 'RA updated' Mobile identity = P-TMSI-2 P-TMSI-2 signature Routing area identity = RAI-4
	13	->	ROUTING AREA UPDATING COMPLETE	Trouble a round individual and in a round in
	14	UE	· · · - - · · -	The UE is switched off or power is removed (see ICS).
	15	->	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, PS detach'

Specific message contents

None.

12.4.1.8.5 Test requirements

UE shall:

- initiate a PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.

After completing the PS attach procedure, UE shall:

- initiate the routing area updating procedure.

When a P-TMSI REALLOCATION REQUEST message is received from SS while waiting for a ROUTING AREA UPDATE ACCEPT message, UE shall:

- ignore the P-TMSI reallocation procedure

- continue with the routing area updating procedure.

12.4.2 Combined routing area updating

The combined routing area updating procedure is a GMM procedure used by PS UEs of UE operation mode A that are IMSI attached for PS and non-PS services. In order to use the combined routing area updating procedure, the network must operate in network operation mode I.

12.4.2.1 Combined routing area updating / combined RA/LA accepted

12.4.2.1.1 Definition

12.4.2.1.2 Conformance requirement

- 1) If the network accepts the combined routing area updating procedure and reallocates a P-TMSI, the UE shall acknowledge the new P-TMSI and continue communication with the new P-TMSI.
- 2) If the network accepts the combined routing area updating procedure from the UE without reallocation of the old P-TMSI, the UE shall continue communication with the old P-TMSI.

Reference

3G3GPP-TS 24.008 clause 4.7.5.2

12.4.2.1.3 Test purpose

To test the behaviour of the UE if the network accepts the combined routing area updating procedure.

The following cases are identified:

- 1) P-TMSI / P-TMSI signature is reallocated
- 2) Old P-TMSI / P-TMSI signature is not changed
- 3) Mobile terminating CS call is allowed with IMSI
- 4) Mobile terminating CS call is allowed with TMSI

12.4.2.1.4 Method of test

Initial condition

System Simulator:

Two cells, cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2. Both cells operating in network operation mode I.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

- 1) A combined PS attach procedure is performed. The UE sends a ROUTING AREA UPDATE REQUEST message. The SS reallocates the P-TMSI, unassigns the TMSI and returns ROUTING AREA UPDATE ACCEPT message with a new P-TMSI and IMSI. The UE acknowledge the new P-TMSI by sending ROUTING AREA UPDATING COMPLETE message. Further communication UE SS is performed by the new P-TMSI. For CS calls, the IMSI is used
- 2) The UE is CS paged in order to verify that the IMSI is used for CS calls.
- 3) A combined PS attach procedure is performed. The UE sends an ROUTING AREA UPDATING REQUEST message. The SS accepts the P-TMSI signature and returns ROUTING AREA UPDATING ACCEPT message without any P-TMSI and with a new TMSI. The UE acknowledge the new TMSI by sending ROUTING AREA UPDATING COMPLETE message. Further communication UE-SS is performed by the old P-TMSI. For CS calls, the new TMSI is used.
- 4) The UE is CS paged in order to verify that the TMSI is used for CS calls.

	Step	Direction UE SS	Message	Comments
-	1	UE		The UE is set in UE operation mode A (see
	2	UE		ICS). The UE is powered up or switched on and initiates an attach (see ICS).
	3	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach' Mobile identity =IMSI
	4	<-	ATTACH ACCEPT	TMSI status = no valid TMSI available Attach result = 'Combined PS / IMSI attached' Mobile identity = P-TMSI-2 P-TMSI-2 signature
	5	->	ATTACH COMPLETE	Routing area identity = RAI-1
•	6	SS		The following messages are sent and shall be received on cell B. Activate cell B with a lower signal strength than cell A The RF level of cell A is lowered until cell
	7	->	ROUTING AREA UPDATING REQUEST	B is preferred by the UE. Update type = 'Combined RA/LA updating' P-TMSI-2 signature Routing area identity = RAI-1
	8	<-	ROUTING AREA UPDATING ACCEPT	TMSI status = no valid TMSI available Update result = 'Combined RA/LA updated' Mobile identity = P-TMSI-1 P-TMSI-1 signature Mobile identity = IMSI Routing area identity = RAI-4
	9	->	ROUTING AREA UPDATING COMPLETE	routing area identity = ICAI-4
	10	<-	PACKET PAGING	Mobile identity = P-TMSI-1
	11	->	REQUESTPAGING TYPE1 UPLINK RLC DATA BLOCKSERVICE REQUEST	Paging order is for TBF establishment. LLC PDU implicitly indicating paging response.service type = "paging response"
Î	12	<-	PACKET PAGING REQUESTPAGING TYPE1	Mobile identity = IMSI Paging order is for RRC-connection.
	13	->	CHANNEL REQUEST RRC CONNECTION REQUEST	
	14	<-	IMMEDIATE ASSIGNMENT RRC CONNECTION SETUP	
	<u>15</u>	<u>-></u>	RRC CONNECTION SETUP COMPLETE	
	15 16 16 17	-> <-	PAGING RESPONSE CHANNEL RELEASE RRC CONNECTION RELEASE	Mobile identity = IMSI After sending of this message, the SS waits for disconnection of the CS signalling link.
	<u>18</u>	<u> </u>	RRC CONNECTION RELEASE COMPLETE	a.com.out.out.out.out.out.out.out.out.out.out
1	17 <u>19</u>	SS		The following messages are sent and shall be received on cell A. The RF level of cell A is increased and the RF level of cell B is lowered until cell A is preferred
	48 <u>20</u>	->	ROUTING AREA UPDATING REQUEST	by the UE. Update type = 'Combined RA/LA updating' P-TMSI-1 signature Routing area identity = RAI-4 TMSI status = no valid TMSI available
	19 21	<-	ROUTING AREA UPDATING ACCEPT	Update result = 'Combined RA/LA updated' Mobile identity = P-TMSI-2 P-TMSI-2 signature Mobile identity = TMSI-1 Routing area identity = RAI-1
	20 <u>22</u>	->	ROUTING AREA UPDATING COMPLETE	

21 23	<-	PACKET PAGING	Mobile identity = P-TMSI-2
22 24	->	REQUESTPAGING TYPE1 UPLINK RLC DATA	Paging order is for TBF establishment. LLC PDU implicitly indicating paging
		BLOCKSERVICE REQUEST	response.service type = "paging response"
23 25	<-	PAGING TYPE1 PACKET	Mobile identity = TMSI-1
		PAGING REQUEST	Paging order is for RRC-connection.
24 26	->	CHANNEL REQUEST RRC	
		CONNECTION REQUEST	
25 27	<-	IMMEDIATE ASSIGNMENT RRC	
		CONNECTION SETUP	
<u>28</u>	<u>-></u>	RRC CONNECTION SETUP	
		COMPLETE	
26 29	->	PAGING RESPONSE	Mobile identity = TMSI-1
27 30	<-	CHANNEL RELEASE RRC	After sending of this message, the SS waits for
		CONNECTION RELEASE	disconnection of the CS signalling link.
<u>31</u>	<u>-></u>	RRC CONNECTION RELEASE	
		COMPLETE	
28 32	UE		The UE is switched off or power is removed
. [(see ICS).
29 33	->	DETACH REQUEST	Message not sent if power is removed.
			Detach type = 'power switched off, combined
			PS / IMSI detach'

Specific message contents

None.

12.4.2.1.5 Test requirements

UE shall:

- initiate a combined PS attach procedure with information elements specified in the above Expected Sequence when UE is powered up or switched on.

After completing the PS attach procedure, UE shall:

- initiate a combined routing area update procedure(Update type = 'Combined RA/LA updating') with the information elements specified above Expected Sequence when RF level of the attached cell is lower than the RF level of the new cell.
- acknowledge the new P-TMSI
- continue communication with the new P-TMSI If SS reallocates a P-TMSI.
- continue communication with the old P-TMSI If SS does not reallocate the old P-TMSI.

12.4.2.2 Combined routing area updating / UE in CS operation at change of RA

12.4.2.2.1 Definition

12.4.2.2.2 Conformance requirement

PS UE that is in an ongoing CS transaction at change of routing area shall initiate the routing area updating procedure only after the CS transaction has been released.

Reference

3G3GPP-TS 24.008 clause 4.7.5.2

12.4.2.2.3 Test purpose

To test the behaviour of the UE when using the combined routing area updating procedure in cases where the UE is CS connected at change of RA.

12.4.2.2.4 Method of test

Initial condition

System Simulator:

Two cells, cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2. Both cells operating in network operation mode I.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No UE operation mode A Yes/No Switch off on button Yes/No Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

A combined PS attach procedure is performed. The UE initiates a CS call. The routing area change. UE will not send a ROUTING AREA UPDATE REQUEST message until the CS operation is terminated.

Ī	Step	Direction	Message	Comments
L		UE SS		
	1	UE		The UE is set in UE operation mode A (see ICS).
	2	UE		The UE is powered up or switched on and
	3	->	ATTACH REQUEST	initiates an attach (see ICS). Attach type = 'Combined PS / IMSI attach' Mobile identity =IMSI
	4	<-	ATTACH ACCEPT	TMSI status = no valid TMSI available Attach result = 'Combined PS / IMSI attached' Mobile identity = P-TMSI-2 P-TMSI-2 signature
	5	->	ATTACH COMPLETE	Routing area identity = RAI-1
L	6	UE		A CS call is initiated.
	7	SS		The following messages are sent and shall be received on cell B. Activate cell B with a lower signal strength than cell A The RF level of cell A is lowered until cell B is preferred by the UE.
	8	UE		No RA updating procedure is initiated. This is checked for 60 seconds.
	9	UE		The CS call is terminated
	10	->	ROUTING AREA UPDATING REQUEST	Update type = 'Combined RA/LA updating' P-TMSI-2 signature
	11	<-	ROUTING AREA UPDATING ACCEPT	Routing area identity = RAI-1 TMSI status = no valid TMSI available Update result = 'Combined RA/LA updated' Mobile identity = P-TMSI-1 P-TMSI-1 signature Mobile identity = IMSI Routing area identity = RAI-4
	12	->	ROUTING AREA UPDATING COMPLETE	j
	13	<-	PACKET PAGING REQUESTPAGING TYPE1	Mobile identity = P-TMSI-1 Paging order is for TBF establishment.
	14	->	UPLINK RLC DATA BLOCKSERVICE REQUEST	LLC PDU implicitly indicating paging response.service type = "paging response"
	15	<-	PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = IMSI
	16	->	CHANNEL REQUEST RRC CONNECTION REQUEST	Paging order is for RRC-connection.
	17	<-	IMMEDIATE ASSIGNMENT RRC CONNECTION SETUP	
	<u>18</u>	<u>-></u>	RRC CONNECTION SETUP COMPLETE	
$\ \ $	18 19	->	PAGING RESPONSE	Mobile identity = IMSI
	19 20	<-	CHANNEL RELEASE RRC CONNECTION RELEASE	After sending of this message, the SS waits for disconnection of the CS signalling link.
	<u>21</u>	<u>-></u>	RRC CONNECTION RELEASE COMPLETE	
	20 22	UE		The UE is switched off or power is removed (see ICS).
	21 23	->	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, combined PS / IMSI detach'

Specific message contents

None.

12.4.2.2.5 Test requirements

UE shall:

- initiate a combined PS attach procedure with information elements specified in the above Expected Sequence when UE is powered up or switched on.

After completing the PS attach procedure, UE shall:

- initiate a CS call. at change of routing area.
- initiate a routing area updating procedure only after the CS transaction has been released.

12.4.2.3 Combined routing area updating / RA only accepted

12.4.2.3.1 Definition

12.4.2.3.2 Conformance requirement

- 1) If the network accepts the combined PS attach procedure, but GMM cause code 'IMSI unknown in HLR' is sent to the UE the User Equipment shall delete the stored TMSI, LAI and CKSN. The User Equipment shall consider USIM invalid for non-PS services until power is switched off or USIM is removed.
- 2) If the network accepts the combined PS attach procedure, but GMM cause code 'MSC temporarily not reachable', 'Network failure' or 'Congestion' is sent to the UE, an UE operation mode A UE may perform an MM IMSI attach procedure.

Reference

3G3GPP-TS 24.008 clause 4.7.3.2

12.4.2.3.3

Test purpose

Test porpose1

To test the behaviour of the UE if the network accepts the routing area updating procedure with indication RA only, GMM cause 'IMSI unknown in HLR'.

Test porpose2

To test the behaviour of the UE if the network accepts the routing area updating procedure with indication RA only, GMM cause 'MSC temporarily not reachable', 'Network failure' or 'Congestion'.

12.4.2.3.4 Method of test

Test Procedure1

Initial condition

System Simulator:

Two cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2.

Both cells operating in network operation mode I.

User Equipment:

The UE has a valid ITMSI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

After attach, the UE sends an ROUTING AREA UPDATE REQUEST message. The SS allocates a P-TMSI and returns ROUTING AREA UPDATE ACCEPT message with a P-TMSI. GMM cause 'IMSI unknown in HLR' is indicated from SS. Further communication UE - SS is performed by the P-TMSI. CS services are not possible.

Expected Sequence

Step	Direction	Message	Comments
	UE SS	_	
1	UE		The UE is set in UE operation mode A (see
			ICS).
2	UE		The UE is powered up or switched on and
3	->	ATTACH REQUEST	initiates an attach (see ICS). Attach type = 'Combined PS / IMSI attach'
3	->	ATTACH REQUEST	Mobile identity =IMSI
			TMSI status = no valid TMSI available
4	<-	ATTACH ACCEPT	Attach result = 'Combined PS / IMSI attached'
			Mobile identity = P-TMSI-2
			P-TMSI-2 signature
_		ATTACH COMPLETE	Routing area identity = RAI-1
5	->	ATTACH COMPLETE	The following messages are sent and shall be
			received on cell B.
6	SS		The SS deactivates cell A and activates cell B.
7	->	ROUTING AREA UPDATING	Update type = 'Combined RA/LA updating'
		REQUEST	P-TMSI-2 signature
			Routing area identity = RAI-1
8		ROUTING AREA UPDATING	TMSI status = no valid TMSI available
0	<-	ACCEPT	Update result = 'RA updated' Mobile identity = P-TMSI-1
		1,0021 1	P-TMSI-1 signature
			Routing area identity = RAI-4
			GMM cause = 'IMSI unknown in HLR'
9	->	ROUTING AREA UPDATING	
10	<-	COMPLETE PACKET PAGING	Mobile identity = P-TMSI-1
10		REQUESTPAGING TYPE1	Paging order is for TBF establishment.
11	->	UPLINK RLC DATA	LLC PDU implicitly indicating paging
		BLOCKSERVICE REQUEST	response.service type = "paging response"
40		DACKET DACINO	Malailla identifica IMOI
12	<-	PACKET PAGING REQUESTPAGING REQUEST	Mobile identity = IMSI Paging order is for RRC-connection.
13	UE	NEGOCOTI NOING NEGOCOT	The UE shall not initiate an RRC connection.
			This is checked during 3 seconds.
14	UE		The UE is switched off or power is removed
			(see ICS).
15	->	DETACH REQUEST	Message not sent if power is removed.
			Detach type = 'power switched off, PS detach'

Test Procedure2

Initial condition

System Simulator:

Two cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2. Both cells operating in network operation mode I.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
Automatic MM IMSI attach procedure for UE operation mode A UE Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

After attach, the UE sends an ROUTING AREA UPDATE REQUEST message . The SS allocates a new P-TMSI signature and returns ROUTING AREA UPDATE ACCEPT message. GMM cause 'MSC temporarily not reachable', 'Network failure' or 'Congestion' is indicated from SS. The cause code is arbitrarily chosen. An UE operation mode A UE may perform an MM IMSI attach procedure (according to the ICS statement). Further communication UE - SS is performed by the P-TMSI. The existence of a signalling channel is verified by a request for mobile identity. CS services are not possible unless an IMSI attach procedure is performed.

Expected Sequence

Dependent whether the option 'Automatic MM IMSI attach procedure for UE operation mode A UE' is not supported or not, the steps 1-13 or 14-35 apply depending on manufacturer (see ICS).

Ī	Step	Direction UE SS	Message	Comments
-	1	UE		The following messages are sent and shall be received on cell A The UE is set in UE operation mode A and no automatic MM IMSI attach procedure is
	2	UE		indicated (see ICS). The UE is powered up or switched on and initiates an attach (see ICS).
	3	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach' Mobile identity =IMSI
	4	<-	ATTACH ACCEPT	TMSI status = no valid TMSI available Attach result = 'Combined PS / IMSI attached' Mobile identity = P-TMSI-2 P-TMSI-2 signature Routing area identity = RAI-1
	5	->	ATTACH COMPLETE	
-	6 7	SS ->	ROUTING AREA UPDATING REQUEST	The following messages are sent and shall be received on cell B. The SS deactivates cell A and activates cell B. Update type = 'Combined RA/LA updating' P-TMSI-2 signature Routing area identity = RAI-1
	8	<-	ROUTING AREA UPDATING ACCEPT	TMSI status = no valid TMSI available Update result = 'RA updated' Mobile identity = P-TMSI-1P-TMSI-1 signature Routing area identity = RAI-4 GMM cause = 'MSC temporarily not reachable', 'Network failure' or 'Congestion' (arbitrarily
	9	->	ROUTING AREA UPDATING COMPLETE	chosen)
	10	<-	PAGING TYPE1PACKET	Mobile identity = IMSI
l	11 12	UE UE	PAGING REQUEST	Paging order is for RRC-connection. The UE shall not initiate an RRC connection. This is checked during 3 seconds. The UE is switched off or power is removed
	13	->	DETACH REQUEST	(see ICS). Message not sent if power is removed. Detach type = 'power switched off, PS detach' Stop the sequence.
ļ	14	UE		The following messages are sent and shall be received on cell B Automatic MM IMSI attach procedure is
	15	UE		indicated (see ICS). The UE is powered up or switched on and
	16	->	ATTACH REQUEST	initiates an attach (see ICS). Attach type = 'Combined PS / IMSI attach' Mobile identity = IMSI
	17	<-	ATTACH ACCEPT	TMSI status = no valid TMSI available Attach result = 'Combined PS / IMSI attached' Mobile identity = P-TMSI-2 P-TMSI-2 signature Routing area identity = RAI-4
	18	->	ATTACH COMPLETE	
	19 20	SS ->	ROUTING AREA UPDATING REQUEST	The following messages are sent and shall be received on cell A. The SS deactivates cell B and activates cell A. Update type = 'Combined RA/LA updating' P-TMSI-2 signature Routing area identity = RAI-4 TMSI status = no valid TMSI available

21	<-	ROUTING AREA UPDATING ACCEPT	Update result = 'RA updated' Mobile identity = P-TMSI-1 P-TMSI-1 signature Routing area identity = RAI-1 GMM cause = 'MSC temporarily not reachable', 'Network failure' or 'Congestion' (arbitrarily chosen)
22	->	ROUTING AREA UPDATING COMPLETE	5.1655.17
23	->	CHANNEL REQUEST RRC CONNECTION REQUEST	
24	<-	IMMEDIATE ASSIGNMENT RRC CONNECTION SETUP	
<u>25</u>	<u>-></u>	RRC CONNECTION SETUP COMPLETE	
25 26	->	LOCATION UPDATING REQ	Location updating type = IMSI attach.
26 27	<-	LOCATION UPDATING ACC	The SS allocates a new TMSI.
27 28	->	TMSI REALLOCATION COMP	Location updating type = IMSI attach.
28 29	<-	CHANNEL RELEASE RRC	After sending of this message, the SS waits for
		CONNECTION RELEASE	disconnection of the CS signalling link.
<u>30</u>	<u>-></u>	RRC CONNECTION RELEASE	
		COMPLETE	
29 31	<-	PAGING TYPE1PACKET	Mobile identity = TMSI-1
		PAGING REQUEST	Paging order is for RRC-connection.
30 32	->	CHANNEL REQUEST RRC	
		CONNECTION REQUEST	
31 33	<-	IMMEDIATE ASSIGNMENT RRC	
		CONNECTION SETUP	
<u>34</u>	<u>-></u>	RRC CONNECTION SETUP	
		COMPLETE	
32 35	->	PAGING RESPONSE	Mobile identity = TMSI-1
33 36	<-	CHANNEL RELEASE RRC	After sending of this message, the SS waits for
		CONNECTION RELEASE	disconnection of the CS signalling link.
<u>37</u>	<u>-></u>	RRC CONNECTION RELEASE	
		COMPLETE	
34 38	UE		The UE is switched off or power is removed
			(see ICS).
35 39	->	DETACH REQUEST	Message not sent if power is removed.
			Detach type = 'power switched off, PS detach'

Specific message contents

None.

12.4.2.3.5 Test requirements

UE shall:

- initiate a combined PS attach procedure with information elements specified in the above Expected Sequence when UE is powered up or switched on.

After completing the PS attach procedure, UE shall:

- initiate a combined routing area updating procedure.
- perform the following actions depending on the GMM cause.

Case 1) GMM cause = 'IMSI unknown in HLR'.

UE shall:

- delete the stored TMSI, LAI and CKSN.

- consider USIM invalid for non-PS services until power is switched off or USIM is removed.

Case 2) GMM cause = 'MSC temporarily not reachable', 'Network failure' or 'Congestion'.

UE shall:

- perform an MM IMSI attach procedure. (only applied UE operation mode A)

12.4.2.4 Combined routing area updating / rejected / PLMN not allowed

12.4.2.4.1 Definition

12.4.2.4.2 Conformance requirement

- 1) If the network rejects a combined routing area updating procedure from the User Equipment with the cause 'PLMN not allowed' the User Equipment shall:
 - 1.1 not perform combined GPRA attach when switched on in the same location area or PLMN.
 - 1.2 delete the stored RAI, PS-CKSN, P-TMSI, P-TMSI signature, TMSI CKSN and LAI.
 - 1.3 store the PLMN in the 'forbidden PLMN list'.

Reference

3G3GPP-TS 24.008 clause 4.7.5.2

12.4.2.4.3 Test purpose

To test the behaviour of the UE if the network rejects the combined routing area updating procedure of the UE with the cause 'PLMN not allowed'.

12.4.2.4.4 Method of test

Initial condition

System Simulator:

Four cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2, cell C in MCC1/MNC1/LAC2/RAC1 and cell D in MCC2/MNC1/LAC1/RAC1. All four cells are operating in network operation mode I

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The SS rejects a combined routing area updating with the cause value 'PLMN not allowed'. The SS checks that the UE does not perform PS attach if activated in the same PLMN. The SS checks that the UE does not perform IMSI attach if activated in the same PLMN.

ſ	Step	Direction	Message	Comments
F		UE SS SS		The following messages are sent and shall be
		00		received on cell A.
	1	SS		The SS activates cell A.
	2	UE		The UE is powered up or switched on and initiates an attach (see ICS.
	3	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach'
				Mobile identity =IMSI
	4	<-	ATTACH ACCEPT	TMSI status = no valid TMSI available Attach result = 'Combined PS / IMSI attached'
	4	ζ-	ATTACITACCETT	Mobile identity = P-TMSI-2
				P-TMSI-2 signature
				Routing area identity = RAI-1 Mobile identity = TMSI-1
	5	->	ATTACH COMPLETE	Wobile Identity = TWSI-1
ŀ		·		The following messages are sent and shall be
	7	CC		received on cell B.
	7 8	SS UE		The SS deactivates cell A and activates cell B. Cell B is preferred by the UE.
	9	->	ROUTING AREA UPDATING	Update type = 'Combined RA/LA updating'
			REQUEST	P-TMSI-2 signature
				Routing area identity = RAI-1 TMSI status = valid TMSI available
	10	<-	ROUTING AREA UPDATING	GMM cause = 'PLMN not allowed'
			REJECT	
	11	UE		The UE initiates an attach by MMI or AT command.
	12	UE		No ATTACH REQUEST sent to SS
		-		(SS waits 30 seconds).
	13	<-	PAGING TYPE1PACKET	Mobile identity = P-TMSI-2
l	14	UE	PAGING REQUEST	Paging order is for TBF establishment. No response from the UE to the request. This
		V		is checked for 10 seconds.
				The following messages are sent and shall be
	15	SS		received on cell C. The SS deactivates cell B and activates cell C.
	16	UE		Cell C is preferred by the UE.
	17	UE		The UE initiates an attach by MMI or by AT
	18	UE		command. No ATTACH REQUEST sent to SS
	. •	0_		(SS waits 30 seconds).
	19	<-	PAGING TYPE1PACKET	Mobile identity = TMSI-1
1	20	UE	PAGING REQUEST	Paging order is for RRC-connection. The UE shall not initiate an RRC connection.
				This is checked during 3 seconds.
				The following messages are sent and shall be
	21	SS		received on cell A. The SS deactivates cell C and activates cell A.
	22	UE		Cell A is preferred by the UE.
	23	UE		The UE initiates an attach by MMI or by AT
	24	UE		command. No ATTACH REQUEST sent to SS
	_ T	01		(SS waits 30 seconds).
	25	<-	PAGING TYPE1PACKET	Mobile identity = P-TMSI-2
۱	26	UE	PAGING REQUEST	Paging order is for TBF establishment. No response from the UE to the request. This
	20	OL.		is checked for 10 seconds.
Ī				The following messages are sent and shall be
	27	SS		received on cell D. The SS deactivates cell A and activates cell D.
	27 28	SS UE		Cell D is preferred by the UE.
	29	ÜE		The UE initiates an attach automatically, by
I				MMI or by AT command.

->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach'
		Mobile identity =IMSI
		TMSI status = no valid TMSI available
<-	ATTACH ACCEPT	Attach result = 'Combined PS / IMSI attached'
-		Mobile identity = P-TMSI-1
		P-TMSI-1 signature
		Routing area identity = RAI-2
		Mobile identity = IMSI
->	ATTACH COMPLETE	,
-	7.1.7.6.1.66.1.66.1.1.	The UE is switched off or power is removed
0_		(see ICS).
->	DETACH REQUEST	Message not sent if power is removed.
	DE 17 (011 (12 Q 02 0 1	Detach type = 'power switched off, combined
		PS / IMSI detach'
	<-	-> ATTACH ACCEPT -> ATTACH COMPLETE UE

Specific message contents

None.

12.4.2.4.5 Test requirements

UE shall:

- initiate a combined PS attach procedure with information elements specified in the above Expected Sequence when UE is powered up or switched on.

After completing the PS attach procedure, UE shall:

- initiate a combined routing area update procedure(Update type = 'Combined RA/LA updating') with the information elements specified above Expected Sequence
- delete the stored P-TMSI, P-TMSI signature, TMSI, RAI, LAI, ciphering key sequence number PS ciphering key sequence number.
- reset the location update attempt counter when UE receive the ROUTING AREA UPDATING REJECT message(GMM cause = 'PLMN not allowed') from SS.
- store the PLMN identity in the 'forbidden PLMN list'.
- not perform combined PS attach procedure when the UE is switched on in the same PLMN.

12.4.2.5 Combined routing area updating / rejected / roaming not allowed in this location area

12.4.2.5.1 Definition

12.4.2.5.2 Conformance requirement

- 1) If the network rejects a combined routing area updating procedure from the User Equipment with the cause 'roaming not allowed in this location area' the User Equipment shall:
 - 1.1 not perform combined PS attach when in the same location area.
 - 1.2 delete the stored RAI, PS-CKSN, P-TMSI P-TMSI signature, TMSI, CKSN and LAI.
 - 1.3 store the LA in the 'forbidden location areas for roaming'.
 - 1.4 perform combined PS attach when a new location area is entered.

2) The User Equipment shall reset the list of 'Forbidden location areas for roaming' when switched off or when the USIM is removed.

Reference

3G3GPP-TS 24.008 clause 4.7.5.2

12.4.2.5.3

Test purpose

Test purpose1

To test that on receipt of a rejection using the 'Roaming not allowed in this area' cause code, the UE ceases trying a routing area updating procedure on that location area. Successful combined routing area updating procedure is possible in other location areas.

Test purpose2

To test that if the UE is switched off or the USIM is removed the list of 'forbidden location areas for roaming' is cleared.

12.4.2.5.4 Method of test

12.4.2.5.4.1 Test procesure1

Initial condition

System Simulator:

Two cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC/LAC2/RAC1.

Both cells are operating in network operation mode I.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The SS rejects a combined routing area updating with the cause value 'Roaming not allowed in this area'. A new attempt for a combined PS attach is not possible. Successful combined PS attach procedure is performed in another location area. The UE is mobed back to the 1st location area. A combined routing area updating shall not be performed, as the LA is on the forbidden list.

Ī	Step	Direction UE SS	Message	Comments
ŀ		SS		The following messages are sent and shall be
				received on cell A.
	1	SS UE		The SS activates cell A. The UE is powered up or switched on and
	2	OL		initiates an attach (see ICS.
	3	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach'
				Mobile identity =IMSI TMSI status = no valid TMSI available
	4	<-	ATTACH ACCEPT	Attach result = 'Combined PS / IMSI attached'
				Mobile identity = P-TMSI-2
				P-TMSI-2 signature Routing area identity = RAI-1
				Mobile identity = IMSI
-	5	->	ATTACH COMPLETE	T. ()
				The following messages are sent and shall be received on cell B.
	7	SS		The SS deactivates cell A and activates cell B.
	8	UE		Cell B is preferred by the UE.
	9	->	ROUTING AREA UPDATING REQUEST	Update type = 'Combined RA/LA updating' P-TMSI-2 signature
				Routing area identity = RAI-1
	10	_	ROUTING AREA UPDATING	TMSI status = no valid TMSI available GMM cause = 'Roaming not allowed in this
	10	<-	REJECT	area'
	11	UE		The UE initiates an attach by MMI or by AT
	12	UE		command. No ATTACH REQUEST sent to SS
	12	OL		(SS waits 30 seconds).
	13	<-	PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = P-TMSI-2
l	14	UE	PAGING REQUEST	Paging order is for TBF establishment. No response from the UE to the request. This
		0_		is checked for 10 seconds.
	15	<-	PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = IMSI Paging order is for RRC-connection.
ı	16	UE	FAGING REQUEST	The UE shall not initiate an RRC connection.
				This is checked during 3 seconds.
				The following messages are sent and shall be received on cell A.
	17	SS		The SS deactivates cell B and activates cell A.
	18	UE		Cell A is preferred by the UE.
	19	UE		The UE initiates an attach automatically, by MMI or by AT command.
	20	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach'
				Mobile identity =IMSI TMSI status = no valid TMSI available
	21	<-	ATTACH ACCEPT	Attach result = 'Combined PS / IMSI attached'
				Mobile identity = P-TMSI-1
				P-TMSI-1 signature Routing area identity = RAI-1
				Mobile identity = TMSI-1
ı	22 23	->	ATTACH COMPLETE	Mobile identity = TMSI-1
	۷3	<-	PAGING TYPE1PACKET PAGING REQUEST	Paging order is for RRC-connection.
	24	->	CHANNEL REQUEST RRC	
	25	<-	CONNECTION REQUEST IMMEDIATE ASSIGNMENT RRC	
	20		CONNECTION SETUP	
	<u>26</u>	<u>-></u>	RRC CONNECTION SETUP	
	26 27	->	COMPLETE PAGING RESPONSE	Mobile identity = TMSI-1
	27 28	<-	CHANNEL RELEASE RRC	After sending of this message, the SS waits for
			CONNECTION RELEASE	disconnection of the CS signalling link.

	<u>29</u>	<u>-></u>	RRC CONNECTION RELEASE COMPLETE	
	28 30	<-	PACKET PAGING	Mobile identity = P-TMSI-1
	29 31	->	REQUESTPAGING TYPE1 UPLINK RLC DATA BLOCKSERVICE REQUEST	Paging order is for TBF establishment. LLC PDU implicitly indicating paging response.service type = "paging response"
Ī	30 32	SS		The following messages are sent and shall be received on cell B. The SS deactivates cell A and activates cell B.
	31 33	UE		No ROUTING AREA UPDATING REQUEST sent to SS
1	<u>3234</u>	<-	PAGING TYPE1PACKET	(SS waits 30 seconds). Mobile identity = P-TMSI-2 Paging order is for TBF establishment.
	33 35	UE	THOMO NEGOLOT	No response from the UE to the request. This is checked for 10 seconds.
	34 <u>36</u>	<-	PAGING TYPE1PACKET	Mobile identity = IMSI
	35 <u>37</u>	UE	PAGING REQUEST	Paging order is for RRC-connection. The UE shall not initiate an RRC connection. This is checked during 3 seconds.

12.4.2.5.4.2 Test procesure2

Initial condition

System Simulator:

Two cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC/LAC2/RAC1.

Both cells are operating in network operation mode I.

User Equipment:

The UE has a valid P-TMSI-1, P-TMSI-1 signature and RAI-1.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
USIM removal possible without powering down Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The SS rejects a combined routing area updating with the cause value 'Roaming not allowed in this area'. The UE is switched off for 10 seconds and switched on again. The SS checks that a combined PS attach is possible on the cell on which the previous combined routing area updating had been rejected.

If USIM removal is possible without switching off:

The SS rejects a routing area updating with the cause value 'Roaming not allowed in this area'. The USIM is removed and inserted in the UE. The SS checks that a PS attach procedure and routing area updating procedure is possible on the cell on which the routing area updating had previously been rejected.

	Step	Direction UE SS	Message	Comments
-		SS		The following messages are sent and shall be
	1	SS		received on cell A. The SS activates cell A.
	2	UE		The UE is powered up or switched on and
	3		ATTACH BEOLIEST	initiates an attach (see ICS.
	3	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach' Mobile identity =IMSI
	_			TMSI status = no valid TMSI available
	4	<-	ATTACH ACCEPT	Attach result = 'Combined PS / IMSI attached' Mobile identity = P-TMSI-2
				P-TMSI-2 signature
				Routing area identity = RAI-1
	5	->	ATTACH COMPLETE	Mobile identity = IMSI
				The following messages are sent and shall be
	7	SS		received on cell B. The SS deactivates cell A and activates cell B.
	8	UE		Cell B is preferred by the UE.
	9	->	ROUTING AREA UPDATING REQUEST	Update type = 'Combined RA/LA updating' P-TMSI-2 signature
			REQUEST	Routing area identity = RAI-1
				TMSI status = no valid TMSI available
	10	<-	ROUTING AREA UPDATING REJECT	GMM cause = 'Roaming not allowed in this area'
	11	UE	NEUE O I	The UE initiates an attach by MMI or by AT
	12	UE		command. No ATTACH REQUEST sent to SS
	12	OL		(SS waits 30 seconds).
	13	<-	PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = P-TMSI-2
!	14	UE	PAGING REQUEST	Paging order is for TBF establishment. No response from the UE to the request. This
				is checked for 10 seconds.
	15	<-	PAGING TYPE1PACKET PAGING REQUEST	Mobile identity = IMSI Paging order is for RRC-connection.
!	16	UE	. 7.6.1.6 1.2.626.	The UE shall not initiate an RRC connection.
	17	UE		This is checked during 3 seconds. If possible (see ICS) USIM removal is
	.,	OL.		performed. Otherwise if possible (see ICS)
				switch off is performed. Otherwise the power is
	18	UE		removed. The UE gets the USIM replaced, is powered up
				or switched on and initiates an attach (see
	19	UE		ICS). The UE initiates an attach by MMI or AT
			ATTAQUE DE 01:	command.
	20	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach' Mobile identity =IMSI
				TMSI status = no valid TMSI available
	21	<-	ATTACH ACCEPT	Attach result = 'Combined PS / IMSI attached' Mobile identity = P-TMSI-1
				P-TMSI-1 signature
				Routing area identity = RAI-1
	22	->	ATTACH COMPLETE	Mobile identity = TMSI-1
	23	<-	PAGING TYPE1PACKET	Mobile identity = TMSI-1
	24	->	PAGING REQUEST CHANNEL REQUEST RRC	Paging order is for RRC-connection.
			CONNECTION REQUEST	
	25	<-	IMMEDIATE ASSIGNMENT RRC CONNECTION SETUP	
	<u>26</u>	<u>-></u>	RRC CONNECTION SETUP	
	26 27	->	COMPLETE PAGING RESPONSE	Mobile identity = TMSI-1
1 1	20 <u>21</u>	-/	I AGING KEGI GINGE	Mosile Identity = TWOI-1

27 28	<-	CHANNEL RELEASE RRC CONNECTION RELEASE	After sending of this message, the SS waits for disconnection of the CS signalling link.	
<u>29</u>	<u>-></u>	RRC CONNECTION RELEASE COMPLETE		
28 30	<-	PACKET PAGING	Mobile identity = P-TMSI-1	l
29 31	->	REQUEST PAGING REQUEST	Paging order is for TBF establishment. LLC PDU implicitly indicating paging	
20 01		BLOCKSERVICE REQUEST	response.service type = "paging response"	
30 <u>32</u>	UE		The UE is switched off or power is removed (see ICS).	
31 33	->	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, combined	
			PS/IMSI detach'	

Specific message contents

None.

12.4.2.5.5 Test requirements

UE shall:

- initiate a combined PS attach procedure with information elements specified in the above Expected Sequence when UE is powered up or switched on.

After completing the PS attach procedure, UE shall:

- initiate a combined routing area update procedure(Update type = 'Combined RA/LA updating') with the information elements specified above Expected Sequence
- delete the stored P-TMSI, P-TMSI signature, TMSI, RAI, LAI, ciphering key sequence number PS ciphering key sequence number.
- reset the location update attempt counter when UE receive the ROUTING AREA UPDATING REJECT message(GMM cause = 'Roaming not allowed in this area') from SS.
- store the LAI in the 'forbidden location areas for roaming'.
- not perform combined PS attach procedure when the UE is switched on in the same location area.
- perform combined PS attach procedure when a new location area is entered.

12.4.2.6 Combined routing area updating / abnormal cases / access barred due to access class control

12.4.2.6.1 Definition

12.4.2.6.2 Conformance requirement

- 1) The UE shall not perform combined routing area updating procedure, but stays in the current serving cell and applies normal cell reselection process.
- 2) The User Equipment shall perform the combined routing area updating procedure when:
 - 2.1 Access is granted.
 - 2.2 Cell is changed.

Reference

3G3GPP-TS 24.008 clause 4.7.5.2

12.4.2.6.3

Test purpose

Test porpose1

To test the behaviour of the UE in case of access class control (access is granted).

Test porpose2

To test the behaviour of the UE in case of access class control (cell is changed).

12.4.2.6.4 Method of test

12.4.2.6.4.1 Test procedure1

Initial condition

A random access class x (0-15) is selected. The USIM is programmed with this access class x. Communication with User Equipments using access class x is initially indicated to be barred.

System Simulator:

Two cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC/LAC1/RAC2.

Both cells are operating in network operation mode I.

Access class x barred.

User Equipment:

The UE has a valid P-TMSI, P-TMSI signature and RAI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

A PS attach procedure is performed. The routing area is changed. The SS indicates access class x barred. A routing area updating procedure is not performed.

The SS indicates that access class x is not barred. A routing area updating procedure is performed.

Step	Direction	Message	Comments
	UE SS		
	SS		The following messages are sent and shall be received on cell A.
1	SS		The SS activates cell A.
2	UE		The UE is powered up or switched on and
			initiates an attach (see ICS.
3	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach'
			Mobile identity =IMSI
			TMSI status = no valid TMSI available
4	<-	ATTACH ACCEPT	Attach result = 'Combined PS / IMSI attached'
			Mobile identity = P-TMSI-2
			P-TMSI-2 signature
			Routing area identity = RAI-1
			Mobile identity = IMSI
5	->	ATTACH COMPLETE	
			The following messages are sent and shall be
			received on cell B.
7	SS		The SS deactivates cell A and activates cell B.
8	UE		Cell B is preferred by the UE.
9	UE		No ROUTING AREA UPDATE REQUEST sent
			to SS, as access class X is barred
			(SS waits 30 seconds).
10	SS		The access class x is not barred anymore.
11	->	ROUTING AREA UPDATING	Update type = 'Combined RA/LA updating'
		REQUEST	P-TMSI-2 signature
			Routing area identity = RAI-1
12		BOLITING AREA LIRRATING	TMSI status = no valid TMSI available
12	<-	ROUTING AREA UPDATING	Update result = 'Combined RA/LA updated' Mobile identity = P-TMSI-1
		ACCEPT	P-TMSI-1 signature
			Mobile identity = TMSI-1
			Routing area identity = RAI-4
13	->	ROUTING AREA UPDATING	Routing area identity = RAI-4
13		COMPLETE	
		OOMI LETE	
14	UE		The UE is switched off or power is removed
			(see ICS).
15	->	DETACH REQUEST	Message not sent if power is removed.
			Detach type = 'power switched off, combined
			PS/IMSI detach'

12.4.2.6.4.2 Test procedure2

Initial condition

A random access class x (0-15) is selected. The USIM is programmed with this access class x. Communication with User Equipments using access class x is indicated to be barred on cell A.

System Simulator:

Three cells, cell A in MCC1/MNC1/LAC1/RAC1 has access class x not barred, cell B in MCC1/MNC1/LAC1/RAC2 has access class x barred, cell C in MCC1/MNC1/LAC1/RAC2 has access class x not barred.

All three cells are operating in network operation mode I.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No UE operation mode A Yes/No Switch off on button Yes/No Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

A PS attach procedure is performed. The routing area is changed. The SS indicates access class x barred. A routing area updating procedure is not performed.

A cell change is performed into a cell where access class x is not barred. A routing area updating procedure is performed.

Step	Direction	Message	Comments
	UE SS	3	
	SS		The following messages are sent and shall be received on cell A.
1 2	SS UE		The SS activates cell A. The UE is powered up or switched on and
3	->	ATTACH REQUEST	initiates an attach (see ICS. Attach type = 'Combined PS / IMSI attach' Mobile identity =IMSI
4	<-	ATTACH ACCEPT	TMSI status = no valid TMSI available Attach result = 'Combined PS / IMSI attached' Mobile identity = P-TMSI-2 P-TMSI-2 signature Routing area identity = RAI-1 Mobile identity = IMSI
5	->	ATTACH COMPLETE	Wobile Identity = IWSI
7 8 9	SS UE UE		The following messages are sent and shall be received on cell B. The SS deactivates cell A and activates cell B. Cell B is preferred by the UE. No ROUTING AREA UPDATING REQUEST sent to SS, as access class X is barred (SS waits 30 seconds).
10 11 12	SS UE ->	ROUTING AREA UPDATING REQUEST	The following messages are sent and shall be received on cell C. The SS deactivates cell B and activates cell C. Cell C is preferred by the UE. Update type = 'Combined RA/LA updating' P-TMSI-2 signature Routing area identity = RAI-1 TMSI status = no valid TMSI available
13	<-	ROUTING AREA UPDATING ACCEPT	Update result = 'Combined RA/LA updated' Mobile identity = P-TMSI-1 P-TMSI-1 signature Mobile identity = IMSI Routing area identity = RAI-4
14	->	ROUTING AREA UPDATING COMPLETE	Nouting area lucituity = NAI-4
15	UE	OOWII LETE	The UE is switched off or power is removed (see ICS).
16	->	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, combined PS/IMSI detach'

Specific message contents

None.

12.4.2.6.5 Test requirements

UE shall:

- initiate a combined PS attach procedure with information elements specified in the above Expected Sequence when UE is powered up or switched on.

After completing the PS attach procedure, UE shall:

- not perform the combined routing area updating procedure.
- stay in the current serving cell.

- apply the normal cell reselection process.(as access class X is barred)
- perform the combined routing area updating procedure when the barred state is removed or because of a cell change.

12.4.2.7 Combined routing area updating / abnormal cases / attempt counter check / procedure timeout

12.4.2.7.1 Definition

12.4.2.7.2 Conformance requirement

- 1) When a T3330 timeout has occurred during a routing area updating procedure with the attempt counter less than five, the User Equipment shall repeat the routing area updating procedure after T3330 timeout.
- 2) When a T3330 timeout has occurred during a routing area updating procedure with the attempt counter five, the User Equipment shall start timer T3302.
- 3) When the T3302 expire, a new routing area updating procedure shall be initiated.

Reference

3G3GPP-TS 24.008 clause 4.7.5.2

12.4.2.7.3 Test purpose

To test the behaviour of the UE with respect to the attempt counter.

12.4.2.7.4 Method of test

Initial condition

System Simulator:

Two cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC/LAC1/RAC2.

Both cells are operating in network operation mode I.

User Equipment:

The UE has a valid TMSI, P-TMSI, P-TMSI signature and RAI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
Switch off on button Yes/No

Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE initiates a routing area updating procedure (attempt counter zero). The SS does not answer with ROUTING AREA UPDATE ACCEPT message before T3330 timeout.

The UE initiates a new routing area updating procedure (attempt counter one) after T3311 expires. The SS does not answer with ROUTING AREA UPDATE ACCEPT message before T3330 timeout.

The UE initiates a new routing area updating procedure (attempt counter two) after T3311 expires. The SS does not answer with ROUTING AREA UPDATE ACCEPT message before T3330 timeout.

The UE initiates a new routing area updating procedure (attempt counter three) after T3311 expires. The SS does not answer with ROUTING AREA UPDATE ACCEPT message before T3330 timeout.

The UE initiates a new routing area updating procedure (attempt counter four) after T3311 expires. The SS does not answer with ROUTING AREA UPDATE ACCEPT message before T3330 timeout.

The UE initiates a new routing area updating procedure with attempt counter five (after T3311 expires). The SS does not answer with ATTACH ACCEPT message before T3330 timeout. The UE shall not perform a new successful routing area updating procedure after 15 seconds.

The UE initiates a routing area updating procedure with attempt counter zero after T3302 expires without P-TMSI, P-TMSI signature, PS CKSN and RAI.

T3302; set to 10 minutes

T3311; 15 seconds

T3330; 15 seconds

Step	Direction UE SS	Message	Comments
	SS		The following messages are sent and shall be
			received on cell A.
1 2	SS UE		The SS activates cell A. The UE is powered up or switched on and
	OL		initiates an attach (see ICS.
3	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach'
			Mobile identity =IMSI TMSI status = no valid TMSI available
4	<-	ATTACH ACCEPT	Attach result = 'Combined PS / IMSI attached'
			Mobile identity = P-TMSI-2
			P-TMSI-2 signature Routing area identity = RAI-1
			Mobile identity = IMSI
5	->	ATTACH COMPLETE	The following managers are cent and shall be
			The following messages are sent and shall be received on cell B.
7	SS		The SS deactivates cell A and activates cell B.
8 9	UE ->	ROUTING AREA UPDATING	Cell B is preferred by the UE. Update type = 'Combined RA/LA updating'
9	-	REQUEST	P-TMSI-2 signature
			Routing area identity = RAI-1
10	SS		TMSI status = no valid TMSI available No response is given from the SS.
11	SS		The SS verifies that the time between the RA
40		DOLITING AREA LIREATING	update requests is 15 seconds
12	->	ROUTING AREA UPDATING REQUEST	Update type = 'Combined RA/LA updating' P-TMSI-2 signature
			Routing area identity = RAI-1
13	SS		TMSI status = no valid TMSI available
14	SS		No response is given from the SS. The SS verifies that the time between the RA
			update requests is 15 seconds
15	->	ROUTING AREA UPDATING REQUEST	Update type = 'Combined RA/LA updating' P-TMSI-2 signature
		INLEGOLOT	Routing area identity = RAI-1
4.0	00		TMSI status = no valid TMSI available
16 17	SS SS		No response is given from the SS. The SS verifies that the time between the RA
''			update requests is 15 seconds
18	->	ROUTING AREA UPDATING REQUEST	Update type = 'Combined RA/LA updating'
		REQUEST	P-TMSI-2 signature Routing area identity = RAI-1
			TMSI status = no valid TMSI available
19 20	SS SS		No response is given from the SS. The SS verifies that the time between the RA
20	00		update requests is 15 seconds
21	->	ROUTING AREA UPDATING	Update type = 'Combined RA/LA updating'
		REQUEST	P-TMSI-2 signature Routing area identity = RAI-1
			TMSI status = no valid TMSI available
22	SS		No response is given from the SS.
23	SS		The SS verifies that the time between the RA update requests is T3302 seconds
24	->	ROUTING AREA UPDATING	Update type = 'Combined RA/LA updating'
		REQUEST	P-TMSI-2 signature Routing area identity = RAI-1
			TMSI status = no valid TMSI available
25	<-	ROUTING AREA UPDATING	Update result = 'Combined RA/LA updated'
		ACCEPT	Mobile identity = P-TMSI-1 P-TMSI-1 signature
			Mobile identity = IMSI
			Routing area identity = RAI-4

26	->	ROUTING AREA UPDATING COMPLETE	
27	UE		The UE is switched off or power is removed
			(see ICS).
28	->	DETACH REQUEST	Message not sent if power is removed.
			Detach type = 'power switched off, combined
			PS/IMSI detach'

Specific message contents

None.

12.4.2.7.5 Test requirements

UE shall:

- initiate a combined PS attach procedure with information elements specified in the above Expected Sequence when UE is powered up or switched on.

After completing the PS attach procedure, UE shall:

- initiate a combined routing area updating procedure with information elements specified in the above Expected Sequence when UE is powered up or switched on.
- perform the following actions depending on the conditions described below.

Case 1) A timer T3330 timeout has occurred during a combined routing area updating procedure with the attempt counter less than five.

UE shall:

- repeat the combined routing area updating procedure after the timer T3330 timeout

Case2) A timer T3330 timeout has occurred during a combined routing area updating procedure with attempt counter five

UE shall:

- start the timer T3302

Case3) The timer T3302 expires

UE shall:

- initiate a new routing area updating procedure

12.4.2.8 Combined routing area updating / abnormal cases / change of cell into new routing area

12.4.2.8.1 Definition

12.4.2.8.2 Conformance requirement

When a change of cell into a new routing area is performed before the routing area updating procedure is finished, the UE shall abort the routing area updating procedure and re-initiate it in the new routing area.

Reference

3G3GPP-TS 24.008 clause 4.7.5.2

12.4.2.8.3 Test purpose

To test the behaviour of the UE in case of procedure collision.

12.4.2.8.4 Method of test

Initial condition

System Simulator:

Three cells, cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2, cell C in MCC1/MNC1/LAC1/RAC3.

All three cells are operating in network operation mode I.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE initiates a routing area updating procedure. The ROUTING AREA UPDATE ACCEPT message is delayed from the SS. The UE performs a cell update into a new routing area. The Ms shall re-initiate a routing area updating procedure in the new routing area. The UE shall not increment the attempt counter.

Step	Direction	Message	Comments
	UE SS		
	SS		The following messages are sent and shall be received on cell A.
1 2	SS UE		The SS activates cell A. The UE is powered up or switched on and
3	->	ATTACH REQUEST	initiates an attach (see ICS. Attach type = 'Combined PS / IMSI attach' Mobile identity =IMSI TMSI attach'
4	<-	ATTACH ACCEPT	TMSI status = no valid TMSI available Attach result = 'Combined PS / IMSI attached' Mobile identity = P-TMSI-2 P-TMSI-2 signature
			Routing area identity = RAI-1 Mobile identity = IMSI
5	->	ATTACH COMPLETE	
6	SS UE		The following messages are sent and shall be received on cell B. The SS deactivates cell A and activates cell B. Cell B is preferred by the UE.
8	->	ROUTING AREA UPDATING REQUEST	Update type = 'Combined RA/LA updating' P-TMSI-2 signature Routing area identity = RAI-1 TMSI status = no valid TMSI available
9	SS		No response id given from the SS.
10	SS		The following messages are sent and shall be received on cell A. Activate cell C with a lower signal strength than
11	UE		cell B. The RF level of cell B is lowered, and the RF
			level of cell C is increased, until cell C is preferred by the UE.
12	->	ROUTING AREA UPDATING REQUEST	Update type = 'Combined RA/LA updating' P-TMSI-2 signature Routing area identity = RAI-1
13	<-	ROUTING AREA UPDATING ACCEPT	TMSI status = no valid TMSI available Update result = 'Combined RA/LA updated' Mobile identity = P-TMSI-1 P-TMSI-1 signature Mobile identity = IMSI
14	->	ROUTING AREA UPDATING COMPLETE	Routing area identity = RAI-5
15	UE		The UE is switched off or power is removed (see ICS).
16	->	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, combined PS/IMSI detach'

Specific message contents

None.

12.4.2.8.5 Test requirements

UE shall:

- initiate a combined PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.

After completing the PS attach procedure, UE shall:

- initiate the routing area update procedure.

When change of cell into new routing area is performed before the routing area updating procedure is finished, UE shall:

- abort the routing area updating procedure.
- re-initiate new routing area updating procedure in the new routing area.

12.4.2.9 Combined routing area updating / abnormal cases / change of cell during routing area updating procedure

12.4.2.9.1 Definition

12.4.2.9.2 Conformance requirement

When a change of cell within new routing area is performed before the routing area updating procedure is finished, the UE shall perform the cell update before the routing area updating procedure is finished.

Reference

3G3GPP-TS 24.008 clause 4.7.5.2

12.4.2.9.3 Test purpose

To test the behaviour of the UE in case of procedure collision.

12.4.2.9.4 Method of test

Initial condition

System Simulator:

Three cells, cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2, cell C in MCC1/MNC1/LAC1/RAC2.

All three cells are operating in network operation mode I.

User Equipment:

The UE has a valid P-TMSI, P-TMSI signature and RAI.

Related ICS/IXIT statements

Support of PS service Yes/No UE operation mode A Yes/No Switch off on button Yes/No

Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE initiates a routing area updating procedure. The ROUTING AREA UPDATE ACCEPT message is delayed from the SS. The UE performs a cell update within the routing area. The UE then waits for the ROUTING AREA UPDATE ACCEPT message.

Ī	Step	Direction	Message	Comments
L		UE SS		
		SS		The following messages are sent and shall be received on cell A.
	1	SS		The SS activates cell A.
	2	UE		The UE is powered up or switched on and
	_			initiates an attach (see ICS.
	3	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach'
				Mobile identity =IMSI
			ATTA OLI A OOEDT	TMSI status = no valid TMSI available
	4	<-	ATTACH ACCEPT	Attach result = 'Combined PS / IMSI attached'
				Mobile identity = P-TMSI-2 P-TMSI-2 signature
				Routing area identity = RAI-1
				Mobile identity = IMSI
	5	->	ATTACH COMPLETE	
				The following messages are sent and shall be
	6	SS		received on cell B. The SS deactivates cell A and activates cell B.
	6 7	UE		Cell B is preferred by the UE.
	8	->	ROUTING AREA UPDATING	Update type = 'Combined RA/LA updating'
	Ü		REQUEST	P-TMSI-2 signature
				Routing area identity = RAI-1
		0.0		TMSI status = no valid TMSI available
F	9	SS		No response id given from the SS. The following messages are sent and shall be
				received on cell C.
	10	SS		Activate cell C with a lower signal strength than
	. •			cell B.
	11	UE		The RF level of cell B is lowered until cell C is
				preferred by the UE.
	12	->	UPLINK RLC DATA BLOCK	LLC PDU implicitly indicating cell update.
	13	<-	ROUTING AREA UPDATING ACCEPT	Update result = 'Combined RA/LA updated' Mobile identity = P-TMSI-1
			1,0021 1	P-TMSI-1 signature
				Mobile identity = IMSI
				Routing area identity = RAI-4
	14	->	ROUTING AREA UPDATING COMPLETE	
	15	UE		The UE is switched off or power is removed
	4.5		DETA OLI DEGLIEGE	(see ICS).
	16	->	DETACH REQUEST	Message not sent if power is removed.
				Detach type = 'power switched off, combined PS/IMSI detach'
, L				I O/IIVIOI UCIAUII

Specific message contents

None.

12.4.2.9.5 Test requirements

UE shall:

- initiate a combined PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.

After completing the PS attach procedure, UE shall:

- initiate routing area update procedure.

When a change of cell within a new routing area is performed before the routing area updating procedure is finished, UE shall:

- perform the cell update.

12.4.2.10 Combined routing area updating / abnormal cases / PS detach procedure collision

12.4.2.10.1 Definition

12.4.2.10.2 Conformance requirement

- 1) When a detach request is received with cause 'PS detach' or 'combined PS/IMSI detach' by the UE while waiting for a ROUTING AREA UPDATE ACCEPT message, the UE shall terminate the routing area updating procedure and continue with the PS detach procedure.
- 2) When a detach request is received with cause 'IMSI detach' by the UE while waiting for a ROUTING AREA UPDATE ACCEPT message, the UE shall ignore the detach request and continue with the routing area updating procedure.

Reference

3G3GPP-TS 24.008 clause 4.7.5.2

12.4.2.10.3 Test purpose

To test the behaviour of the UE in case of procedure collision.

12.4.2.10.4 Method of test

12.4.2.10.4.1 Test procedure1

Initial condition

System Simulator:

Two cells, cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2. Both cells are operating in network operation mode I.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No UE operation mode A Yes/No Switch off on button Yes/No Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE initiates a routing area updating procedure. The SS does not answer the routing area updating procedure, but initiates a PS detach procedure with cause 'PS detach' or 'combined PS/IMSI detach'. The UE shall terminate the routing area updating procedure and continue with the PS detach procedure.

Step	Direction	Message	Comments
	UE SS		
	SS		The following messages are sent and shall be received on cell A.
1	SS		The SS activates cell A.
2	UE		The UE is powered up or switched on and
			initiates an attach (see ICS.
3	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach'
			Mobile identity =IMSI
			TMSI status = no valid TMSI available
4	<-	ATTACH ACCEPT	Attach result = 'Combined PS / IMSI attached'
			Mobile identity = P-TMSI-2
			P-TMSI-2 signature
			Routing area identity = RAI-1
5	->	ATTACH COMPLETE	Mobile identity = IMSI
		ATTACH COMITELIE	The following messages are sent and shall be
			received on cell B.
6	SS		The SS deactivates cell A and activates cell B.
7	ÜE		Cell B is preferred by the UE.
8	->	ROUTING AREA UPDATING	Update type = 'Combined RA/LA updating'
		REQUEST	P-TMSI-2 signature
			Routing area identity = RAI-1
			TMSI status = no valid TMSI available
9	SS		The SS ignores the ROUTING AREA
			UPDATING REQUEST message and initiates a
			detach procedure.
10	<-	DETACH REQUEST	Detach type = 're-attach not required'
11	->	DETACH ACCEPT	

12.4.2.10.4.2 Test procedure2

Initial condition

System Simulator:

Two cells, cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2. Both cells are operating in network operation mode I.

User Equipment:

The UE has a valid P-TMSI, P-TMSI signature and RAI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE initiates a routing area updating procedure. The SS does not answer the routing area updating procedure, but initiates a PS detach procedure with cause 'IMSI detach'. The UE shall ignore the detach procedure and continue with the routing area updating procedure.

Step	Direction	Message	Comments
	UE SS		
	SS		The following messages are sent and shall be
	00		received on cell A.
1 2	SS UE		The SS activates cell A.
2	UE UE		The UE is powered up or switched on and initiates an attach (see ICS.
3	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach'
		7.1.7.6.1.1.2.3.2.3.1	Mobile identity =IMSI
			TMSI status = no valid TMSI available
4	<-	ATTACH ACCEPT	Attach result = 'Combined PS / IMSI attached'
			Mobile identity = P-TMSI-2
			P-TMSI-2 signature
			Routing area identity = RAI-1 Mobile identity = IMSI
5	->	ATTACH COMPLETE	Mobile Identity = IMSI
		711710110011111111111111111111111111111	The following messages are sent and shall be
			received on cell B.
6	SS		The SS deactivates cell A and activates cell B.
7	UE		Cell B is preferred by the UE.
8	->	ROUTING AREA UPDATING	Update type = 'Combined RA/LA updating'
		REQUEST	P-TMSI-2 signature Routing area identity = RAI-1
			TMSI status = no valid TMSI available
9	SS		The SS ignores the ROUTING AREA
			UPDATING REQUEST message and initiates a
			detach procedure.
10	<-	DETACH REQUEST	Detach type = 'IMSI detach'
11	UE		The UE ignores the DETACH REQUEST
			message and continue the routing area updating procedure.
12	<-	ROUTING AREA UPDATING	Update result = 'Combined RA/LA updated'
'-		ACCEPT	Mobile identity = P-TMSI-1
			P-TMSI-1 signature
			Mobile identity = IMSI
100		DOLITING ADEA LIBRATING	Routing area identity = RAI-4
13	->	ROUTING AREA UPDATING COMPLETE	
14	UE	CONFLETE	The UE is switched off or power is removed
'"			(see ICS).
15	->	DETACH REQUEST	Message not sent if power is removed.
1			Detach type = 'power switched off, combined
			PS/IMSI detach'

Specific message contents

None.

12.4.2.10.5 Test requirements

UE shall:

- initiate a combined PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.

After completing the PS attach procedure, UE shall:

- initiate routing area update procedure.
- perform the follow actions depending on the conditions described below.

Case 1) UE receives a DETACH REQUEST message with cause 'PS detach' or 'combined PS/IMSI detach' from SS while waiting for a ROUTING AREA UPDATE ACCEPT message, UE shall:

- terminate the routing area updating procedure
- continue with the PS detach procedure.

Case 2) UE receives a DETACH REQUEST message with cause 'IMSI detach' from SS while waiting for a ROUTING AREA UPDATING ACCEPT message, UE shall:

- ignore the detach request.
- continue with the routing area updating procedure.

12.4.3 Periodic routing area updating

12.4.3.1 Periodic routing area updating / accepted

12.4.3.1.1 Definition

12.4.3.1.2 Conformance requirement

The User Equipment shall perform a periodic routing area update procedure after a T3312 timeout.

Reference

3G3GPP-TS 24.008 clause 4.7.2.2 and 4.7.5.1

12.4.3.1.3 Test purpose

To test the behaviour of the UE with respect to the periodic routing area updating procedure.

12.4.3.1.4 Method of test

Initial condition

System Simulator:

One cell operating in network operation mode II (in case of UE operation mode A), or in network operation mode III (in case of UE operation mode C).

User Equipment:

The UE has a valid P-TMSI-1, P-TMSI-1 signature and RAI-1.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode C Yes/No
UE operation mode A Yes/No
USIM removal possible without powering down Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE initiates a PS attach procedure with identity P-TMSI. The SS reallocates the P-TMSI and returns ATTACH ACCEPT message with a new P-TMSI and timer T3312. The UE acknowledge the new P-TMSI by sending ATTACH COMPLETE message. A routing area updating procedure is performed at T3312 timeout.

T3312; set to 6 minutes.

Expected Sequence

Step	Direction	Message	Comments
	UE SS		
1	SS		The UE is set in UE operation mode C (see ICS). If UE operation mode C not supported, goto step 11.
2	UE		The UE is powered up or switched on and initiates an attach (see ICS).
3	->	ATTACH REQUEST	Attach type = 'PS attach' Mobile identity = P-TMSI-1 P-TMSI-1 signature
4	<-	ATTACH ACCEPT	Routing area identity = RAI-1 Attach result = 'PS only attached' Mobile identity = P-TMSI-2 P-TMSI-2 signature Routing area identity = RAI-1 T3312 = 6 minutes
5	->	ATTACH COMPLETE	10012 = 0 minutes
6	->	ROUTING AREA UPDATING REQUEST	Update type = 'Periodic updating' P-TMSI-2 signature Routing area identity = RAI-1
7	SS		The SS verifies that the time between the attach and the periodic RA updating is T3312
8	<-	ROUTING AREA UPDATING ACCEPT	No new mobile identity assigned. P-TMSI not included. Update result = 'RA updated' P-TMSI-3 signature Routing area identity = RAI-1
9	UE		The UE is switched off or power is removed
10	->	DETACH REQUEST	(see ICS). Message not sent if power is removed. Detach type = 'power switched off, PS detach'
11 12	UE		The SS is set in network operation mode II. The UE is set in UE operation mode A(see ICS) and the test is repeated from step 3 to step 10.

Specific message contents

None.

12.4.3.1.5 Test requirements

UE shall:

- initiate a PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.
- set and start the timer T3312 when the ATTACH ACCEPT message and ROUTING AREA UPDATING
 ACCEPT message from SS. The value of the timer T3312 is sent by SS to UE in ATTACH ACCEPT message
 and ROUTING AREA UPDATING ACCEPT message.

- initiate a routing area updating procedure with Update type = 'Periodic updating' when the timer T3312 is expired.

12.4.3.2 Periodic routing area updating / accepted / T3312 default value

12.4.3.2.1 Definition

12.4.3.2.2 Conformance requirement

The User Equipment shall perform a periodic routing area update procedure after a T3312 timeout.

Reference

3G3GPP-TS 24.008 clause 4.7.2.2 and 4.7.5.2

12.4.3.2.3 Test purpose

To test the behaviour of the UE with respect to the periodic routing area updating procedure.

12.4.3.2.4 Method of test

Initial condition

System Simulator:

One cell operating in network operation mode I.

User Equipment:

The UE has a valid P-TMSI-1, P-TMSI-1 signature and RAI-1.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE initiates a combined PS attach procedure. The SS reallocates the P-TMSI and returns ATTACH ACCEPT message with a new P-TMSI and timer T3312 is omitted. The UE acknowledge the new P-TMSI by sending ATTACH COMPLETE message. After 54 minutes, a periodic routing area updating procedure is initiated by the UE.

T3312; default value 54 minutes.

Step	Direction	Message	Comments
	UE SS		
1 2	UE ->	ATTACH REQUEST	The UE is powered up or switched on and initiates an attach (see ICS). Attach type = 'Combined PS / IMSI attach'
			Mobile identity = P-TMSI-1 P-TMSI-1 signature Routing area identity = RAI-1
3	<-	ATTACH ACCEPT	TMSI status = no valid TMSI available Attach result = 'Combined PS /IMSI attached' Mobile identity = P-TMSI-2 P-TMSI-2 signature Mobile identity = TMSI-1
4		ATTACH COMPLETE	Routing area identity = RAI-1
4 5	-> ->	ROUTING AREA UPDATING	Update type = 'Periodic updating'
		REQUEST	P-TMSI-2 signature
			Routing area identity = RAI-1 TMSI status = valid TMSI available
6	SS		The SS verifies that the time between the attach request and the periodic RA updating is T3312
7	<-	ROUTING AREA UPDATING ACCEPT	No new mobile identity assigned. P-TMSI and TMSI not included. Update result = 'RAupdated' P-TMSI-3 signature
			Routing area identity = RAI-1
8	UE		The UE is switched off or power is removed (see ICS).
9	->	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, combined
			PS/IMSI detach'

Specific message contents

None.

12.4.3.2.5 Test requirements

UE shall:

- initiate a PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.
- set and start the timer T3312 when the ATTACH ACCEPT message and ROUTING AREA UPDATING ACCEPT message from SS.
- initiate a routing area updating procedure with Update type = 'Periodic updating' when the timer T3312 is expired.

12.4.3.3 Periodic routing area updating / no cell available / network mode I

12.4.3.3.1 Definition

12.4.3.3.2 Conformance requirement

If the UE is both IMSI attached for PS and non-PS services, and if the UE lost coverage of the registered PLMN and timer T3312 expires; if the UE returns to coverage in a cell that supports PS and the network is in network operation mode I, then the UE shall perform a combined routing area update procedure.

Reference

3G3GPP-TS 24.008 clause 4.7.2.2 and 4.7.5.1.

12.4.3.3.3 Test purpose

To test the behaviour of the UE with respect to the periodic routing area updating procedure.

12.4.3.3.4 Method of test

Initial condition

System Simulator:

Two cells, cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2.

Cell A is in are operating in network operation mode II and cell B is in network operation mode I. operating in network operation mode II.

User Equipment:

The UE has a valid P-TMSI-1, P-TMSI-1 signature and RAI-1.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE initiates a PS attach procedure and an IMSI attach procedure. The SS reallocates the P-TMSI and returns ATTACH ACCEPT message with a new P-TMSI and timer T3312. The UE acknowledge the new P-TMSI by sending ATTACH COMPLETE message. PS radio contact is distorted before T3312 timeout. PS radio contact is established again (after T3312 timeout), and a routing area updating procedure is performed immediately.

T3312; set to 6 minutes.

Ī	Step	Direction	Message	Comments
		UE SS		
		SS		The following messages are sent and shall be received on cell A.
	1	SS		The SS activates cell A.
	2	SS		The UE is set in UE operation mode A (see ICS).
	3	UE		The UE is powered up or switched on and initiates an attach (see ICS).
	4	->	ATTACH REQUEST	Attach type = 'PS attach' Mobile identity = P-TMSI-1 P-TMSI-1 signature
	5	<-	ATTACH ACCEPT	Routing area identity = RAI-1 Attach result = 'PS only attached' Mobile identity = P-TMSI-2 P-TMSI-2 signature Routing area identity = RAI-1 T3312 = 6 minutes
	6	-~	ATTACH COMPLETE	13312 = 0 minutes
	7	-> ->	CHANNEL REQUEST RRC	
	8	<-	CONNECTION REQUEST IMMEDIATE ASSIGNMENT RRC CONNECTION SETUP	
	<u>9</u>	<u>-></u>	RRC CONNECTION SETUP COMPLETE	
	9 10	->	LOCATION UPDATING REQ	Location updating type = IMSI attach.
	10 11	<-	LOCATION UPDATING ACC	The SS allocates a new TMSI.
	11 12	->	TMSI REALLOCATION COMP	Location updating type = IMSI attach.
	12 13	<-	CHANNEL RELEASE RRC CONNECTION RELEASE	After sending of this message, the SS waits for disconnection of the CS signalling link.
	<u>14</u>	<u>-></u>	RRC CONNECTION RELEASE COMPLETE	
	13 15	SS		After 5 minutes, the signal strength is lowered until the UE have lost contact with the SS.
	14 <u>16</u>	SS		Wait 2 minutes.
				The following messages are sent and shall be received on cell B.
Ī	15 17	SS		The SS deactivates cell A and activates cell B.
	16 18	UE		Cell B is preferred by the UE.
	17 19	UE		The UE immediately start a combined RA updating procedure
	<u> 1820</u>	->	ROUTING AREA UPDATING REQUEST	Update type = 'Combined RA/LA updating' P-TMSI-2 signature Routing area identity = RAI-1
,				TMSI status = valid TMSI available
1	19 <u>21</u>	<-	ROUTING AREA UPDATING ACCEPT	Update result = 'Combined RA/LA updated' Mobile identity = P-TMSI-3 P-TMSI-3 signature
				Mobile identity = TMSI-2 Routing area identity = RAI-4
	20 22	UE		The UE is switched off or power is removed (see ICS).
	21 <u>23</u>	->	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, combined
				PS / IMSI detach'

Specific message contents

None.

12.4.3.3.5 Test requirements

UE shall:

- initiate a PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.

When the UE is both IMSI attached for PS and non-PS service, and if the MS lost coverage of the reiterated PLMN and the timer T3312 expires, if the UE returns to coverage in a cell that supports PS and the network is in network oration mode I, UE shall:

- perform the combined routing area update procedure indicating "combined RA/LA updating with IMSI attach".

12.4.3.4 Combined periodic routing area updating / no cell available

12.4.3.4.1 Definition

12.4.3.4.2 Conformance requirement

If the UE is both IMSI attached for PS and non-PS services, and if the UE lost coverage of the registered PLMN and timer T3312 expires; if the UE returns to coverage in a cell that supports PS and the network is in network operation mode II, then the UE shall perform a periodic routing area update procedure and a periodic location update procedure.

Reference

3G3GPP-TS 24.008 clause 4.7.2.2 and 4.7.5.2

12.4.3.4.3 Test purpose

To test the behaviour of the UE with respect to the periodic routing area updating procedure.

12.4.3.4.4 Method of test

Initial condition

System Simulator:

One cell operating in network operation mode II.

User Equipment:

The UE has a valid P-TMSI-1, P-TMSI-1 signature and RAI-1.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The UE initiates a PS attach procedure and an IMSI attach procedure. The SS reallocates the P-TMSI and returns ATTACH ACCEPT message with a new P-TMSI and timer T3312. The UE acknowledge the new P-TMSI by sending ATTACH COMPLETE message. PS radio contact is distorted before T3312 timeout. PS radio contact is established again (after T3312 timeout), and a periodic routing area updating procedure and a periodic location update procedure is performed immediately.

T3312; set to 6 minutes.

Step	Direction UE SS	Message	Comments
1 2	UE ->	ATTACH REQUEST	The UE is powered up or switched on and initiates an attach (see ICS). Attach type = 'PS attach'
3	<-	ATTACH ACCEPT	Mobile identity = P-TMSI-1 P-TMSI-1 signature Routing area identity = RAI-1 Attach result = 'PS only attached' Mobile identity = P-TMSI-2 P-TMSI-2 signature Routing area identity = RAI-1 T3312 = 6 minutes
4	->	ATTACH COMPLETE	
5	->	CHANNEL REQUEST RRC	
6	<-	CONNECTION REQUEST IMMEDIATE ASSIGNMENT RRC CONNECTION SETUP	
<u>7</u>	<u>-></u>	RRC CONNECTION SETUP COMPLETE	
7 8	->	LOCATION UPDATING REQ	Location updating type = IMSI attach.
<u>89</u>	<-	LOCATION UPDATING ACC TMSI REALLOCATION COMP	The SS allocates a new TMSI. Location updating type = IMSI attach.
9 10 10 11	-> <-	CHANNEL RELEASE RRC CONNECTION RELEASE	After sending of this message, the SS waits for disconnection of the CS signalling link.
<u>12</u>	<u>-></u>	RRC CONNECTION RELEASE COMPLETE	
11 13	SS		After 5 minutes, the signal strength is lowered
12 <u>14</u>	SS		until the UE have lost contact with the SS. After 2 minutes, the signal strength is increased until the UE have got contact with the SS.
13 <u>15</u>	UE		The UE immediately start the periodic RA
14 <u>16</u>	->	ROUTING AREA UPDATING REQUEST	updating procedure Update type = 'Periodic updating' P-TMSI-2 signature Routing area identity = RAI-1
15 <u>17</u>	<-	ROUTING AREA UPDATING ACCEPT	No new mobile identity assigned. P-TMSI not included. Update result = 'RAupdated' P-TMSI-3 signature Routing area identity = RAI-1
16<u>18</u>	->	CHANNEL REQUEST RRC CONNECTION REQUEST	
17 <u>19</u>	<-	IMMEDIATE ASSIGNMENT RRC CONNECTION SETUP	
<u>20</u>	<u>-></u>	RRC CONNECTION SETUP COMPLETE	
18 21	->	LOCATION UPDATING REQ	Location updating type = Periodic LA updating.
19 22 20 23	<- <-	LOCATION UPDATING ACC CHANNEL RELEASE RRC CONNECTION RELEASE	After sending of this message, the SS waits for disconnection of the CS signalling link.
<u>24</u>	<u>-></u>	RRC CONNECTION RELEASE COMPLETE	
21 25	UE		The UE is switched off or power is removed (see ICS).
22 26	->	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, PS detach'

Specific message contents

None.

12.4.3.4.5 Test requirements

UE shall:

- initiate a combined PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.

When the UE is both IMSI attached for PS and non-PS service, and if the MS lost coverage of the reiterated PLMN and the timer T3312 expires, if the UE returns to coverage in a cell in the same RA that supports PS and that indicates that the network is in network operation mode II, UE shall:

- perform the periodic routing area updating procedure indicating "Periodic updating".
- perform the periodic location updating procedure.

12.5 P-TMSI reallocation

12.5.1 Definition

12.5.2 Conformance requirement

- 1) A User Equipment shall acknowledge a new P-TMSI when explicitly allocated.
- 2) The P-TMSI shall be updated on the USIM when the User Equipment is correctly deactivated in accordance with the manufacturer's instructions.
- 3) A User Equipment shall use the given P-TMSI in further communication with the network.

Reference

3G3GPP-TS 24.008 clause 4.7.6

12.5.3 Test purpose

To verify that the UE is able to receive and acknowledge a new P-TMSI by means of an explicit P-TMSI reallocation procedure.

To verify that the UE has stored the P-TMSI in a non-volatile memory.

The implicit reallocation procedure is tested in the attach procedure.

12.5.4 Method of test

Initial condition

System Simulator:

One cell operating in network operation mode II.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
UE operation mode C Yes/No (only if mode A not supported)
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

An explicit P-TMSI reallocation procedure is performed (P-TMSI reallocation command sent from the SS and acknowledged from the UE by P-TMSI reallocation complete). The UE is PS detached and switched off. Its power supply is interrupted for 10 seconds. The power supply is resumed and then the UE is switched on. A PS attach procedure is performed with the given P-TMSI as identity.

Expected Sequence

Step	Direction	Message	Comments
- 356	UE SS		
1	UE		The UE is set in UE operation mode A (see
			ICS). If UE operation mode A not supported set
			the UE in operation mode C.
2	UE		The UE is powered up or switched on and
			initiates an attach (see ICS).
3	->	ATTACH REQUEST	Attach type = 'PS attach'
4		ATTACHLACCEDT	Mobile identity = IMSI
4	<-	ATTACH ACCEPT	Attach result = 'PS only attached'
			Mobile identity = P-TMSI-1 P-TMSI-1 signature
			Routing area identity = RAI-1
5	->	ATTACH COMPLETE	Rodding area identity = IVAI-1
6	<-	P-TMSI REALLOCATION	Mobile identity = P-TMSI-2
		COMMAND	P-TMSI-2 signature
			Routing area identity = RAI-1
7	->	P-TMSI REALLOCATION	,
		COMPLETE	
8	UE		The UE is switched off or power is removed
			(see ICS).
9	->	DETACH REQUEST	Message not sent if power is removed.
			Detach type = 'power switched off, PS detach'
10	UE		The UE is powered up or switched on and
11		ATTACH DECLIEST	initiates an attach (see ICS).
11	->	ATTACH REQUEST	Attach type = 'PS attach' Mobile identity = P-TMSI-2
			P-TMSI-2 signature
			Routing area identity = RAI-1
12	<-	ATTACH ACCEPT	No new mobile identity assigned.
	,	/ · · · / · · · · · · · · · · · · · · ·	P-TMSI not included.
			Attach result = 'PS only attached'
			P-TMSI-3 signature
			Routing area identity = RAI-1
13	<-	PAGING REQUEST TYPE	Mobile identity = P-TMSI-2
		4PAGING TYPE1	Paging order is for TBF establishment.
14	->	UPLINK RLC DATA	LLC PDU implicitly indicating paging
		BLOCKSERVICE REQUEST	response.service type = "paging response"
15	UE		The UE is switched off or power is removed
10	J 0E		(see ICS).
16	->	DETACH REQUEST	Message not sent if power is removed.
'0		DE MONTRE QUEUT	Detach type = 'power switched off, PS detach'

Specific message contents

None

12.5.5 Test requirements

UE shall:

- initiate a PS attach procedure with the information elements specified in the above Expected Sequence when UE is powered up or switched on.

After completing the PS attach procedure, and when UE receive of the P-TMSI REALLOCATION COMMAND message, UE shall:

- store the allocated Routing Area Identifier(RAI) and the allocated P-TMSI.
- acknowledge a new P-TMSI.
- send the P-TMSI and a P-TMSI REALLOCATION COMPLETE message to SS.
- update P-TMSI on the USIM when UE is correctly deactivated in accordance with the manufacturer's instructions.
- use the given P-TMSI in further communication with SS.

12.6 PS authentication and ciphering

12.6.1 Test of authentication

The purpose of this procedure is to verify the user identity. A correct response is essential to guarantee the establishment of the connection. If not, the connection will drop.

12.6.1.1 Authentication accepted

12.6.1.1.1 Definition

12.6.1.1.2 Conformance requirement

A User Equipment shall correctly respond in an authentication and ciphering procedure by sending a response with the RES information field set to the same value as the one produced by the authentication and ciphering algorithm in the network.

Reference

3G3GPP-TS 24.008 clause 4.7.7

12.6.1.1.3 Test purpose

To test the behaviour of the UE if the network accepts the authentication and ciphering procedure.

12.6.1.1.4 Method of test

Initial condition

System Simulator:

Two cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2. Both cells are operating in network operation mode II.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
UE operation mode C Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

A PS attach is performed, and the SS initiates an authentication and ciphering procedure.

The SS checks the value RES sent by the UE in the AUTHENTICATION AND CIPHERING RESPONSE message.

The UE initiates a routing area updating procedure and the SS checks the value of the PS Ciphering Key Sequence Number sent by the UE in the ROUTING AREA REQUEST message.

Step	Direction	Message	Comments
_	UE SS	_	
1 2	SS UE		The following messages are sent and shall be received on cell A. The SS activates cell A. The UE is set in UE operation mode C (see ICS). If UE operation mode C not supported,
3	UE		goto step 17. The UE is powered up or switched on and initiates an attach (see ICS).
4	->	ATTACH REQUEST	Attach type = 'PS attach' Mobile identity = IMSI
5	<-	AUTHENTICATION AND CIPHERING REQUEST	Request authentication. Set PS-CKSN-1
6	->	AUTHENTICATION AND CIPHERING RESPONSE	RES
7	SS		The SS checks the RES value.
8	<-	ATTACH ACCEPT	Attach result = 'PS only attached' Mobile identity = P-TMSI-2 P-TMSI-2 signature
9	->	ATTACH COMPLETE	Routing area identity = RAI-1
			The following messages are sent and shall be
10	SS		received on cell B. Activate cell B with a lower signal strength than cell A The RF level of cell A is lowered until cell
11	->	ROUTING AREA UPDATING REQUEST	B is preferred by the UE. Update type = 'RA updating' P-TMSI-2 signature Routing area identity = RAI-1
			PS-CKSN-1
12	SS		The value of PS-CKSN is checked
13	<-	ROUTING AREA UPDATING	Update result = 'RA updated'
		ACCEPT	Mobile identity = P-TMSI-1
			P-TMSI-1 signature
14	->	ROUTING AREA UPDATING COMPLETE	Routing area identity = RAI-2
15	UE		The UE is switched off or power is removed
1 .			(see ICS).
16	->	DETACH REQUEST	Message not sent if power is removed.
17	UE		Detach type = 'power switched off, PS detach' The UE is set in UE operation mode A (see
''	UE		ICS) and the test is repeated from step 1 to
1			step 16.

Specific message contents

None.

12.6.1.1.5 Test requirements

UE shall:

- initiate a PS attach procedure with information elements specified in the above Expected Sequence when UE is powered on or switched on.

When the UE receives the AUTHENTICATION AND CIPHERING REQUEST message form SS, UE shall:

- send the AUTHENTICATION AND CIPHERING RESPONSE message with the RES information field set to the same value as the one produced by the authentication and ciphering algorithm in the network.

12.6.1.2 Authentication rejected by the network

12.6.1.2.1 Definition

12.6.1.2.2 Conformance requirement

- 1) After reception of an Authentication Reject message the UE shall:
 - 1.1 not perform normal routing area updating
 - 1.2 not perform periodic routing area updating
 - 1.3 not perform PS detach if switched off
- 2) The UE shall delete the stored RAI, PS-CKSN P-TMSI and P-TMSI signature. USIM shall be considered invalid until power is switched off or USIM is removed.

Reference

```
3G3GPP-TS 24.008 clauses 4.7.7
```

12.6.1.2.3 Test purpose

To test the behaviour of the UE if the network rejects the authentication and ciphering procedure.

12.6.1.2.4 Method of test

Initial condition

System Simulator:

Two cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2.

Both cells are operating in network operation mode II.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No UE operation mode A Yes/No

UE operation mode C Yes/No

Switch off on button Yes/No

Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

A PS attach is performed, and the SS rejects the authentication and ciphering procedure.

The SS checks that the UE does not perform normal routing area updating, does not perform periodic routing area updating and does not perform PS detach if switched off.

T3312; set to 10 minutes.

ſ	Step	Direction	Message	Comments
		UE SS	3	
-	1 2	SS UE UE		The following messages are sent and shall be received on cell A. The SS activates cell A. The UE is set in UE operation mode C (see ICS). If UE operation mode C not supported, goto step 14. The UE is powered up or switched on and
	4	->	ATTACH REQUEST	initiates an attach (see ICS). Attach type = 'PS attach' Mobile identity = IMSI
	5	<-	AUTHENTICATION AND CIPHERING REQUEST	Request authentication. Set PS-CKSN-1
	6	->	AUTHENTICATION AND CIPHERING RESPONSE	RES
	7	<-	AUTHENTICATION AND CIPHERING REJECT	
	8	<-	PAGING TYPE1PAGING REQUEST TYPE 1	Mobile identity = IMSI Paging order is for TBF establishment.
ļ	9	UE		No response from the UE to the request. This is checked for 10 seconds.
•	10	SS		The following messages are sent and shall be received on cell B. Activate cell B with a lower signal strength than cell A The RF level of cell A is lowered until cell
	11	UE		B is preferred by the UE. No ROUTING AREA UPDATING REQUEST sent to the SS
	12	UE		(SS waits 30 seconds). No periodic ROUTING AREA UPDATING REQUEST sent to the SS (SS waits T3310.
	13 14	UE SS		The UE is switched off (see ICS). No DETACH REQUEST sent to the SS (SS waits 30 seconds).
	15	UE		The UE is set in UE operation mode A (see ICS) and the test is repeated from step 1 to step 13.

Specific message contents

None.

12.6.1.2.5 Test requirements

UE shall:

- initiate a PS attach procedure with information elements specified in the above Expected Sequence when UE is powered on or switched on.

After UE receives the AUTHENTICATION AND CIPHERING REJECT message, UE shall:

- not perform normal routing area updating when the RF level of the attached cell is lower than the RF level of the new cell.
- not perform a periodic routing area updating when the timer T3312 expires.

- not perform PS detach when UE is switched off.
- delete the stored RAI, PS-CKSN, P-TMSI and P-TMSI signature.
- consider the USIM as invalid until power is switched off or USIM is removed.

12.6.1.3 Authentication rejected by the UE

12.6.1.3.1 GMM cause 'MAC failure'

12.6.1.3.1.1 Definition

12.6.1.3.1.2 Conformance requirement

If the UE considers the MAC code (supplied by the core network in the AUTN parameter) to be invalid, the UE shall send AUTHENTICATION AND CIPHERING FAILURE message with the reject cause 'MAC failure' to the System Simulator.

Reference

3G3GPP-TS 24.008 clause 4.7.7

12.6.1.3.1.3 Test purpose

To test the behaviors of the UE, when the UE considers the MAC code (supplied by the core network in the AUTN parameter) to be invalid.

12.6.1.3.1.4 Method of test

Initial condition

System Simulator:

Two cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2.

Both cells are operating in network operation mode II.

The MAC (Message Authentication Code) code, which is included in AUTHENTICATION AND CIPHERING REQUEST, is invalid value.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
UE operation mode C Yes/No
Switch off on button Yes/No

Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

A PS attach is performed, and the SS initiates an authentication and ciphering procedure.

The UE sends AUTHENTICATION AND CIPHERING FAILURE message with reject cause 'MAC failure' to the SS and starts timer T3214.

The SS initiates an identification procedure, upon receipt of a failure message with reject cause 'MAC failure'.

After the identification procedure is complete, the SS re-initiates an authentication and ciphering procedure.

T3360; set to 6 seconds.

T3318; set to 5 seconds.

Step	Direction	Message	Comments
	UE SS		
1 2	SS UE		The following messages are sent and shall be received on cell A. The SS activates cell A. The UE is set in UE operation mode C (see ICS). If UE operation mode C is not supported, goto step 25.
3 4	UE		The following messages are sent and shall be received on cell A.
5	UE		The UE is powered up or switched on and initiates an attach (see ICS).
6	->	ATTACH REQUEST	Attach type = 'PS attach' Mobility identity = IMSI
7	<-	AUTHENTICATION AND CIPHERING REQUEST	Request authentication. Invalid Message Authentication Code (MAC).
8 9	SS ->	AUTHENTICATION AND CIPHERING FAILURE	The SS starts the timer T3360 GMM cause='MAC failure'
10 11 12	<- -> SS	IDENTITY REQUEST IDENTITY RESPONSE	Identity type = IMSI Mobile identity = IMSI The SS checks that the P-TMSI originally used in the authentication challenge corresponded to the correct IMSI.
13	<-	AUTHENTICATION AND CIPHERING REQUEST	Request authentication.
14	->	AUTHENTICATION AND CIPHERING RESPONSE	RES
15 16	SS <-	ATTACH ACCEPT	The SS checks the RES value. Attach result = 'PS only attached' Mobile identity = P-TMSI-2 P-TMSI-2 signature
17	->	ATTACH COMPLETE	Routing area identity = RAI-1
18	SS		The following messages are sent and shall be received on cell B. Activate cell B with a lower signal strength than cell A. RF level of cell A is lowered until cell B is
19	->	ROUTING AREA UPDATING REQUEST	preferred by the UE. Update type = 'RA updating' P-TMSI-2 signature Routing area identity = RAI-1 PS-CKSN-1
20 21	SS <-	ROUTING AREA UPDATING ACCEPT	The value of PS-CKSN is checked Update result = 'RA updated' Mobile identity = P-TMSI-1 P-TMSI-1 signature Routing area identity = RAI-2
22	->	ROUTING AREA UPDATING COMPLETE	
23	UE		The UE is switched off or power is removed (see ICS).
24	->	DETACH REQUEST	Message is not sent if power is removed. Detach type = 'power switched off, PS detach'
25	UE		The UE is set in UE operation mode A (see ICS) and the test is repeated from step 1 to step 24.

Specific message contents

None.

12.6.1.3.1.5 Test requirements

UE shall:

- initiate a PS attach procedure with information element specified in the above Expected Sequence when UE is powered on or switched on.
- send an AUTHENTICATION AND CIPHERING FAILURE message with GMM cause 'MAC failure' to the SS
- start timer T3318.

After UE receives the IDENTITY REQUEST message from SS, UE shall:

- send the IDENTITY RESPONSE message to SS

After UE receives the second AUTHENTICATION AND CIPHERING REQUEST message (containing a valid MAC) from SS, UE shall:

- stop timer T3318, if running
- send the AUTHENTICATION AND CIPHERING RESPONSE message to SS

12.6.1.3.2 GMM cause 'Synch failure'

12.6.1.3.2.1 Definition

12.6.1.3.2.2 Conformance requirement

If the UE considers the SQN (supplied by the core network in the AUTN parameter) to be out of range, the UE shall send AUTHENTICATION AND CIPHERING FAILURE message with the reject cause 'Synch failure' to the System Simulator.

Reference

3G3GPP-TS 24.008 clause 4.7.7

12.6.1.3.2.3 Test purpose

To test the behaviors of the UE, when the UE considers the SQN (supplied by the core network in the AUTN parameter) to be out of range.

12.6.1.3.2.4 Method of test

Initial condition

System Simulator:

Two cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2.

Both cells are operating in network operation mode II.

. . .

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
UE operation mode C Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

A PS attach is performed, and the SS initiates an authentication and ciphering procedure.

UE sends AUTHENTICATION AND CIPHERING FAILURE message with reject cause 'synch failure' to the SS and starts timer T3214.

SS re-initiates an authentication and ciphering procedure.

T3360; set to 6 seconds.

T3320; set to 15 seconds.

Step	Direction	Message	Comments
	UE SS		
1 2	SS UE		The following messages are sent and shall be received on cell A. The SS activates cell A. The UE is set in UE operation mode C (see ICS). If UE operation mode C is not supported, goto step 21. The following messages are sent and shall be
3	UE		received on cell A. The UE is powered up or switched on and initiates an attach (see ICS).
4	->	ATTACH REQUEST	Attach type = 'PS attach' Mobility identity = IMSI
5	<-	AUTHENTICATION AND CIPHERING REQUEST	Request authentication. SQN is out of range.
6	SS		The SS starts the timer T3360
7	->	AUTHENTICATION AND CIPHERING FAILURE	GMM cause = 'Synch failure' AUTS parameter
8	SS		set new authentication vectors. (resynchronisation)
9	<-	AUTHENTICATION AND CIPHERING REQUEST	Request authentication.
10	->	AUTHENTICATION AND CIPHERING RESPONSE	RES
11	SS		The SS checks the RES value.
12	<-	ATTACH ACCEPT	Attach result = 'PS only attached' Mobile identity = P-TMSI-2 P-TMSI-2 signature
13	->	ATTACH COMPLETE	Routing area identity = RAI-1
14	SS		The following messages are sent and shall be received on cell B. Activate cell B with lower signal strength than cell A. RF level of cell A is lowered until cell B is
15	->	ROUTING AREA UPDATING REQUEST	preferred by the UE. Update type = 'RA updating' P-TMSI-2 signature Routing area identity = RAI-1 PS-CKSN-1
16 17	SS <-	ROUTING AREA UPDATING ACCEPT	The value of PS-CKSN is checked Update result = 'RA updated' Mobile identity = P-TMSI-1 P-TMSI-1 signature
18	->	ROUTING AREA UPDATING COMPLETE	Routing area identity = RAI-2
19	UE	John Leve	The UE is switched off or power is removed
20	->	DETACH REQUEST	(see ICS). Message is not sent if power is removed. Detach type = 'power switched off, PS detach'
21	UE		The UE is set in UE operation mode A (see ICS) and the test is repeated from step 1 to step 20.

Specific message contents

None.

12.6.1.3.2.5 Test requirements

UE shall:

- initiate a PS attach procedure with information element specified in the above Expected Sequence when UE is powered on or switched on.
- send AUTHENTICATION AND CIPHERING FAILURE message with GMM cause 'synch failure' to the SS
- start timer T3320.

After UE receives the second AUTHENTICATION AND CIPHERING REQUEST message from SS, UE shall:

- stop timer T3320, if running.
- send AUTHENTICATION AND CIPHERING RESPONSE message to SS.

12.6.2 Test of ciphering mode settingvoid

The purpose of this procedure is to let the network to trigger the start and stop of stream ciphering.

The SS shall start and synchronise ciphering and deciphering according to GSM 03.20. The bitstream shall be generated by algorithm GEA/1.

12.6.2.1 Ciphering mode / start ciphering

12.6.2.1.1 Definition

12.6.2.1.2 Conformance requirement

- 1) When the UE receives the AUTHENTICATION AND CIPHERING REQUEST message during the attach procedure, with Ciphering indicator information element set to 'ciphering mode off', the User Equipment shall:
 - 1.1 responds with an AUTHENTICATION AND CIPHERING RESPONSE message
 - 1.2 not start ciphering
- 2) When the UE receives the AUTHENTICATION AND CIPHERING REQUEST message during the routing area updating procedure, with Ciphering indicator information element set to 'ciphering mode on', the User Equipment shall:
 - 2.1 responds with an AUTHENTICATION AND CIPHERING RESPONSE message
 - 2.2 start ciphering and deciphering with the algorithm indicated by the Ciphering algorithm information element
 - 2.3 the ciphering uses the cipher key determined during the authentication procedure

Reference

3G TS 24.008 clause 4.7.7

12.6.2.1.3 Test purpose

To test the behaviour of the UE if the network accepts the authentication and ciphering procedure with ciphering.

12.6.2.1.4 Method of test

Initial condition

System Simulator:

 $\frac{Two\;cells\;(not\;simultaneously\;activated),\;cell\;A\;in\;MCC1/MNC1/LAC1/RAC1,\;cell\;B\;in\;MCC1/MNC1/LAC1/RAC2.}{}$

Both cells are operating in network operation mode II.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
UE operation mode C Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

A PS attach is performed. Authentication procedure without ciphering is performed.

The UE initiates a routing area updating procedure, and the SS initiates an authentication and ciphering procedure to start ciphering.

## US SS The following messages are sent an received on cell A. ## The SS activates cell A. ## The UE is set in UE operation mode C not surget step 27. ## UE is powered up or switched or initiates an attach (see ICS). ## ATTACH REQUEST Attach type = 'PS attach' Mobile identity = IMSI Message not ciphered Request authentication. ## CIPHERING REQUEST Ciphering off Message not ciphered RES CIPHERING RESPONSE ATTACH ACCEPT Attach result = 'PS only attached' Mobile identity = P TMSI-2 P T	C (see pported,
4 SS 2 UE 3 The UE is set in UE operation mode C not surption middle surption mode C not surption middle surption mode C not surption middle surption mode C not surpt	C (see pported,
The UE is set in UE operation mode ICS). If UE operation mode C not sugeto step 27. The UE is powered up or switched or initiates an attach (see ICS). ATTACH REQUEST ATTACH REQUEST AUTHENTICATION AND Request authentication. CIPHERING REQUEST CIPHERING RESPONSE AUTHENTICATION AND RESSAGE not ciphered RES CIPHERING RESPONSE ATTACH ACCEPT ATTACH ACCEPT ATTACH COMPLETE PAGING REQUEST TYPE 1 Mobile identity = P TMSI-2 Paging order is for TBF establishment Message not ciphered Message not ciphered Message not ciphered Message not ciphered ATTACH COMPLETE Mobile identity = P TMSI-2 Paging order is for TBF establishment Message not ciphered LLC PDU implicitly indicating paging Message not ciphered LLC PDU implicitly indicating paging Message not ciphered LLC PDU implicitly indicating paging Message not ciphered The following messages are sent an received on cell B. Activate cell B with a lower signal street cell A The RF level of cell A is lowerd B is preferred by the UE. LPTMSI-2 signature REQUEST REQUEST	oported,
ICS). If UE operation mode C not sugoto step 27. The UE is powered up or switched of initiates an attach (see ICS). Attach type = 'PS attach'. Mobile identity = IMSI Message not ciphered Request authentication. CIPHERING REQUEST Ciphering off Message not ciphered RESPONSE ATTACH ACCEPT	oported,
3 UE 4 ATTACH REQUEST ATTACH REQUEST ATTACH REQUEST AUTHENTICATION AND CIPHERING REQUEST CIPHERING RESPONSE AUTHENTICATION AND CIPHERING RESPONSE ATTACH ACCEPT ATTACH ACCEPT ATTACH ACCEPT ATTACH COMPLETE PAGING REQUEST TYPE 1 Mobile identity = P.TMSI-2 P.TMSI-2 signature Routing area identity = RAI-1 Message not ciphered Message not ciphered Message not ciphered ATTACH COMPLETE ATTACH COMPLETE ATTACH COMPLETE ATTACH COMPLETE Mobile identity = P.TMSI-2 P.TMSI-2 signature Routing area identity = P.TMSI-2 Paging order is for TBF establishmen Message not ciphered LLC PDU implicitly indicating paging Message not ciphered The following messages are sent an received on cell B. Activate cell B with a lower signal str cell A The RF level of cell A is lowere B is preferred by the UE. Update type = 'RA updating' P-TMSI-2 signature P-TMSI-2 signature ROUTING AREA UPDATING REQUEST	
The UE is powered up or switched of initiates an attach (see ICS). Attach type = 'PS attach' Mebile identity = IMSI Message not ciphered Request authentication. Ciphering off Message not ciphered Research (ciphered Research) Research (ciphered Re	n and
initiates an attach (see ICS). Attach type = 'PS attach' Mobile identity = IMSI Message not ciphered Request authentication. Ciphering off Message not ciphered Ressage not ciphered Attach result = 'PS only attached' Mobile identity = P-TMSI-2 P-TMSI-2 signature Routing area identity = RAI-1 Message not ciphered Message not ciphered Message not ciphered Message not ciphered Message not ciphered The following messages are sent an received on cell B. Activate cell B with a lower signal streel and the cell A is lower as is preferred by the UE. Update type = 'RA updating' P-TMSI-2 signature Request Request Request Initiates an attach (see ICS). Attach type = 'PS attach' Mobile identity = IMSI Message not ciphered The following messages are sent an received on cell B. Activate cell B with a lower signal streel and the cell A is lower as is preferred by the UE. Update type = 'RA updating' P-TMSI-2 signature	n and
4	
Mobile identity = IMSI Message not ciphered Request authentication. Ciphering off Message not ciphered Request authentication. Ciphering off Message not ciphered Response Res	
AUTHENTICATION AND CIPHERING REQUEST Siphering off Message not ciphered AUTHENTICATION AND CIPHERING RESPONSE AUTHENTICATION AND CIPHERING RESPONSE ATTACH ACCEPT Message not ciphered ATTACH ACCEPT ACCEPT Attach result = 'PS only attached' Mobile identity = P TMSI-2 P-TMSI-2 signature Routing area identity = RAI-1 Message not ciphered Message not ciphered ATTACH COMPLETE Message not ciphered Wessage not ciphered LLC PDU implicitly indicating paging Message not ciphered LLC PDU implicitly indicating paging Message not ciphered The following messages are sent an received on cell B. Activate cell B with a lower signal streel and the cell A The RF level of cell A is lower B is preferred by the UE. P-TMSI-2 signature ROUTING AREA UPDATING REQUEST P-TMSI-2 signature	
AUTHENTICATION AND CIPHERING REQUEST AUTHENTICATION AND CIPHERING RESPONSE AUTHENTICATION AND RES CIPHERING RESPONSE ATTACH ACCEPT ATTACH ACCEPT ATTACH ACCEPT ATTACH COMPLETE B ATTACH COMPLETE PAGING REQUEST TYPE 1 Wessage not ciphered Message not ciphered LLC PDU implicitly indicating paging Message not ciphered The following messages are sent an received on cell B. Activate cell B with a lower signal str cell A The RF level of cell A is lowerd B is preferred by the UE. Update type = 'RA updating' P TMSI-2 signature	
CIPHERING REQUEST Ciphering off Message not ciphered RES AUTHENTICATION AND CIPHERING RESPONSE ATTACH ACCEPT Attach result = 'PS only attached' Mebile identity = P-TMSI-2 P-TMSI-2 signature Routing area identity = RAI-1 Message not ciphered Message not ciphered Message not ciphered Message not ciphered UPLINK REC DATA BLOCK LEC PDU implicitly indicating paging Message not ciphered The following messages are sent an received on cell B. Activate cell B with a lower signal str cell A The RE level of cell A is lowered B is preferred by the UE. Update type = 'RA updating' P-TMSI-2 signature	
Message not ciphered RES AUTHENTICATION AND CIPHERING RESPONSE ATTACH ACCEPT ATTACH ACCEPT ATTACH ACCEPT AUTHORITICATION AND CIPHERING RESPONSE ATTACH ACCEPT AUTHORITICATION AND CIPHERING RESPONSE ATTACH ACCEPT Attach result = 'PS only attached' Mebile identity = P TMSI-2 P TMSI-2 signature Routing area identity = RAI-1 Message not ciphered Message not ciphered Message not ciphered LLC PDU implicitly indicating paging Message not ciphered The following messages are sent an received on cell B. Activate cell B with a lower signal str cell A The RF level of cell A is lower B is preferred by the UE. Liphate type = 'RA updating' P TMSI-2 signature Update type = 'RA updating' P TMSI-2 signature	
GIPHERING RESPONSE ATTACH ACCEPT ATTACH ACCEPT ATTACH ACCEPT ATTACH ACCEPT ATTACH ACCEPT	
GIPHERING RESPONSE ATTACH ACCEPT Message not ciphered Attach result = 'PS only attached' Mobile identity = P-TMSI-2 P-TMSI-2 signature Routing area identity = RAI-1 Message not ciphered Message not ciphered Message not ciphered Message not ciphered Message not ciphered Message not ciphered Message not ciphered Message not ciphered Message not ciphered Message not ciphered LLC PDU implicitly indicating paging Message not ciphered LLC PDU implicitly indicating paging Message not ciphered The following messages are sent an received on cell B. Activate cell B with a lower signal streed on cell B. Activate cell B with a lower signal streed on cell A The RE level of cell A is lower B is preferred by the UE. LUpdate type = 'RA updating' P-TMSI-2 signature	
ATTACH ACCEPT Attach result = 'PS only attached' Mobile identity = P-TMSI-2 P-TMSI-2 signature Routing area identity = RAI-1 Message not ciphered Message not ciphered Message not ciphered PAGING REQUEST TYPE-1 Mobile identity = P-TMSI-2 Paging order is for TBF establishmen Message not ciphered LLC PDU implicitly indicating paging Message not ciphered The following messages are sent an received on cell B. Activate cell B with a lower signal str cell A The RF level of cell A is lowered B is preferred by the UE. Update type = 'RA updating' P-TMSI-2 signature	
Mobile identity = P-TMSI-2 P-TMSI-2 signature Routing area identity = RAI-1 Message not ciphered 10 -> UPLINK RLC DATA BLOCK LLC PDU implicitly indicating paging Message not ciphered LLC PDU implicitly indicating paging Message not ciphered The following messages are sent an received on cell B. Activate cell B with a lower signal str cell A The RF level of cell A is lowere B is preferred by the UE. Update type = 'RA updating' P-TMSI-2 signature	
P-TMSI-2 signature Routing area identity = RAI-1 Message not ciphered Message not ciphered Message not ciphered Message not ciphered Message not ciphered Wessage not ciphered LLC PDU implicitly indicating paging Message not ciphered The following messages are sent an received on cell B. Activate cell B with a lower signal str cell A The RF level of cell A is lowere B is preferred by the UE. Wessage not ciphered The following messages are sent an received on cell B. Activate cell B with a lower signal str cell A The RF level of cell A is lowere B is preferred by the UE. Update type = 'RA updating' P-TMSI-2 signature	
Routing area identity = RAI-1 Message not ciphered Message not ciphered Message not ciphered PAGING REQUEST TYPE 1 Webile identity = P-TMSI-2 Paging order is for TBF establishmen Message not ciphered LLC PDU implicitly indicating paging Message not ciphered The following messages are sent an received on cell B. Activate cell B with a lower signal str cell A The RF level of cell A is lowere B is preferred by the UE. P-TMSI-2 signature	
8 ATTACH COMPLETE Message not ciphered Message not ciphered PAGING REQUEST TYPE 1 Mobile identity = P-TMSI-2 Paging order is for TBF establishmen Message not ciphered LLC PDU implicitly indicating paging Message not ciphered The following messages are sent an received on cell B. Activate cell B with a lower signal str cell A The RF level of cell A is lowered B is preferred by the UE. Update type = 'RA updating' P-TMSI-2 signature	
9 — PAGING REQUEST TYPE 1 Mobile identity = P-TMSI-2 Paging order is for TBF establishment Message not ciphered LLC PDU implicitly indicating paging Message not ciphered The following messages are sent and received on cell B. Activate cell B with a lower signal structure cell A The RF level of cell A is lowered B is preferred by the UE. Update type = 'RA updating' REQUEST Mobile identity = P-TMSI-2 Paging order is for TBF establishment Message not ciphered LLC PDU implicitly indicating paging Message not ciphered The following messages are sent and received on cell B. Activate cell B with a lower signal structure and paging Message not ciphered LLC PDU implicitly indicating paging Message not ciphered The following messages are sent and received on cell B. Activate cell B with a lower signal structure and paging Message not ciphered The following messages are sent and received on cell B. Activate cell B with a lower signal structure and paging Message not ciphered The following messages are sent and received on cell B. Activate cell B with a lower signal structure and received on cell B. Activate cell B with a lower signal structure and received on cell B. Activate cell B with a lower signal structure and received on cell B.	
Paging order is for TBF establishment Message not ciphered LLC PDU implicitly indicating paging Message not ciphered The following messages are sent and received on cell B. Activate cell B with a lower signal structure cell A The RF level of cell A is lowered B is preferred by the UE. ROUTING AREA UPDATING P-TMSI-2 signature	
Paging order is for TBF establishment Message not ciphered LLC PDU implicitly indicating paging Message not ciphered LLC PDU implicitly indicating paging Message not ciphered The following messages are sent and received on cell B. Activate cell B with a lower signal structure cell A The RF level of cell A is lowered B is preferred by the UE. Update type = 'RA updating' REQUEST P-TMSI-2 signature	
Paging order is for TBF establishment Message not ciphered LLC PDU implicitly indicating paging Message not ciphered The following messages are sent and received on cell B. Activate cell B with a lower signal structure cell A The RF level of cell A is lowered B is preferred by the UE. Update type = 'RA updating' REQUEST Paging order is for TBF establishment Message not ciphered LLC PDU implicitly indicating paging Message not ciphered The following messages are sent and received on cell B. Activate cell B with a lower signal structure by the UE. Update type = 'RA updating' P-TMSI-2 signature	
10 -> UPLINK RLC DATA BLOCK Message not ciphered LLC PDU implicitly indicating paging Message not ciphered The following messages are sent an received on cell B. Activate cell B with a lower signal str cell A The RF level of cell A is lower B is preferred by the UE. Update type = 'RA updating' REQUEST Hessage not ciphered LLC PDU implicitly indicating paging Message not ciphered The following messages are sent an received on cell B. Activate cell B with a lower signal str cell A The RF level of cell A is lower B is preferred by the UE. Update type = 'RA updating' P-TMSI-2 signature	
10 -> UPLINK RLC DATA BLOCK LLC PDU implicitly indicating paging Message not ciphered The following messages are sent an received on cell B. Activate cell B with a lower signal str cell A The RF level of cell A is lower B is preferred by the UE. 12 -> ROUTING AREA UPDATING REQUEST Update type = 'RA updating' P-TMSI-2 signature	it.
Message not ciphered The following messages are sent an received on cell B. Activate cell B with a lower signal str cell A The RF level of cell A is lowered B is preferred by the UE. B is preferred by the UE. Update type = 'RA updating' P-TMSI-2 signature	
The following messages are sent and received on cell B. 11 SS Activate cell B with a lower signal stricell A The RF level of cell A is lowered B is preferred by the UE. 12 -> ROUTING AREA UPDATING Update type = 'RA updating' P-TMSI-2 signature	response.
11 SS Activate cell B. Activate cell B with a lower signal str cell A The RF level of cell A is lowered B is preferred by the UE. 12 -> ROUTING AREA UPDATING Update type = 'RA updating' P-TMSI-2 signature	
11 SS Activate cell B with a lower signal str cell A The RF level of cell A is lowered B is preferred by the UE. 12 -> ROUTING AREA UPDATING Update type = 'RA updating' P-TMSI-2 signature	d shall be
cell A The RF level of cell A is lowere B is preferred by the UE. Update type = 'RA updating' REQUEST Cell A The RF level of cell A is lowere B is preferred by the UE. Update type = 'RA updating' P-TMSI-2 signature	anath than
B is preferred by the UE. +-> ROUTING AREA UPDATING REQUEST B is preferred by the UE. Update type = 'RA updating' P-TMSI-2 signature	
12 -> ROUTING AREA UPDATING Update type = 'RA updating' REQUEST P-TMSI-2 signature	Ju unin con
REQUEST P-TMSI-2 signature	
Routing area identity = RAI-1	
Message not ciphered	
13 — AUTHENTICATION AND Request authentication.	
CIPHERING REQUEST Ciphering on	
Message not ciphered	
14 -> AUTHENTICATION AND RES	
CIPHERING RESPONSE Message not ciphered	
45 ← ROUTING AREA UPDATING Update result = 'RA updated'	
ACCEPT Mobile identity = P-TMSI-1	
P-TMSI-1 signature	
Routing area identity = RAI-2	
Message ciphered	
16 -> ROUTING AREA UPDATING Message ciphered	
COMPLETE	
17 — PAGING REQUEST TYPE 1 Mobile identity = P-TMSI-1	
Paging order is for TBF establishmen	
Message not ciphered	at.
18 -> UPLINK RLC DATA BLOCK LLC PDU implicitly indicating paging	ìt.
Message ciphered	
19 — P-TMSI REALLOCATION Mobile identity = P-TMSI-2	
COMMAND P-TMSI-2 signature	
Routing area identity = RAI-2	
Message ciphered	
20 -> P-TMSI REALLOCATION Message ciphered	
COMPLETE	

21	-	IDENTITY REQUEST	Identity type = IMEI
			Message not ciphered
22	>	IDENTITY RESPONSE	Mobile identity = IMEI
			Message not ciphered
23	-	P-TMSI REALLOCATION	Mobile identity = P-TMSI-1
		COMMAND	P-TMSI-1 signature
			Routing area identity = RAI-2
			Message ciphered
24	>	P-TMSI REALLOCATION	Message ciphered
		COMPLETE	
25	UE		The UE is switched off or power is removed
			(see ICS).
26	->	DETACH REQUEST	Message not sent if power is removed.
			Detach type = 'power switched off, PS detach'
			Message ciphered
27	₩		The UE is set in UE operation mode A (see
			ICS) and the test is repeated from step 3 to
			step 26.

12.6.2.1.5 Test requirements

Case1)

UE shall:

— initiate a PS attach procedure with information elements specified in the above Expected Sequence when UE is powered on or switched on.

When the UE receives the AUTHENTICATION AND CIPHERING REQUEST message during the PS attach procedure, with Ciphering indicator information element set to 'ciphering mode off', UE shall:

- respond with an AUTHENTICATION AND CIPHERING RESPONSE message.
- not start ciphering.

Case2)

UE shall:

- initiate a routing area updating procedure with information elements specified in the above Expected Sequence.
- When the UE receives the AUTHENTICATION AND CIPHERING REQUEST message during the routing area updating procedure, with Ciphering indicator information element set to 'ciphering mode on', UE shall:
 - responds with an AUTHENTICATION AND CIPHERING RESPONSE message.
 - start ciphering and deciphering with the algorithm indicated by the Ciphering algorithm information element.
 - apply the ciphering that uses the cipher key determined during the authentication procedure.

12.6.2.2 Ciphering mode / stop ciphering

12.6.2.2.1 Definition

12.6.2.2.2 Conformance requirement

- 1) When the UE receives the AUTHENTICATION AND CIPHERING REQUEST message during the attach procedure, with Ciphering indicator information element set to 'ciphering mode on', the User Equipment shall:
 - 1.1 responds with an AUTHENTICATION AND CIPHERING RESPONSE message

- 1.2 start ciphering and deciphering with the algorithm indicated by the Ciphering algorithm information element
- 1.3 the ciphering uses the cipher key determined during the authentication procedure
- 2) When the UE receives the AUTHENTICATION AND CIPHERING REQUEST message during the routing area updating procedure, with Ciphering indicator information element set to 'ciphering mode on', the User Equipment shall:
 - 2.1. responds with an AUTHENTICATION AND CIPHERING RESPONSE message
 - 2.2. start ciphering and deciphering with the algorithm indicated by the Ciphering algorithm information element
 - 2.3. the ciphering uses the cipher key determined during the authentication procedure

Reference

3G TS 24.008 clause 4.7.7

12.6.2.2.3 Test purpose

To test the behaviour of the UE if the network accepts the authentication and ciphering procedure without ciphering.

12.6.2.2.4 Method of test

Initial condition

System Simulator:

Two cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2.

Both cells are operating in network operation mode II.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No UE operation mode A Yes/No UE operation mode C Yes/No

Switch off on button Yes/No

Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

A PS attach is performed, and the SS initiates an authentication and ciphering procedure to start ciphering.

The UE initiates a routing area updating procedure. A RA updating procedure is initiated, and authentication procedure without ciphering is performed. Ciphering is turned off.

Step	Direction	Message	Comments
	UE SS		The fellowing research and should be
			The following messages are sent and shall be received on cell A.
4	SS		The SS activates cell A.
2	HE		The UE is set in UE operation mode C (see
_	0 2		ICS). If UE operation mode C not supported,
			goto step 21.
3	UE		The UE is powered up or switched on and
			initiates an attach (see ICS).
4	>	ATTACH REQUEST	Attach type = 'PS attach'
			Mobile identity = IMSI
			Message not ciphered
5	-<-	AUTHENTICATION AND	Request authentication.
		CIPHERING REQUEST	Ciphering on
		ALITUENTICATION AND	Message not ciphered
6	>	AUTHENTICATION AND	RES
7		CIPHERING RESPONSE ATTACH ACCEPT	Message not ciphered Attach result = 'PS only attached'
+		ATTACH ACCEPT	Mobile identity = P-TMSI-2
			P-TMSI-2 signature
1			Routing area identity = RAI-1
			Message ciphered
8	>	ATTACH COMPLETE	Message ciphered
			3.17
9	-<-	PAGING REQUEST TYPE 1	Mobile identity = P-TMSI-2
			Paging order is for TBF establishment.
			Message not ciphered
10	->	UPLINK RLC DATA BLOCK	LLC PDU implicitly indicating paging response.
			Message ciphered
			The following messages are sent and shall be
11	SS		received on cell B. Activate cell B with a lower signal strength than
++	33		cell A The RF level of cell A is lowered until cell
			B is preferred by the UE.
12	>	ROUTING AREA UPDATING	Update type = 'RA updating'
		REQUEST	P-TMSI-2 signature
			Routing area identity = RAI-1
			Message not ciphered
13		AUTHENTICATION AND	Request authentication.
		CIPHERING REQUEST	Ciphering off
			Message not ciphered
14	>	AUTHENTICATION AND	RES
4.5		CIPHERING RESPONSE	Message not ciphered
15		ROUTING AREA UPDATING	Update result = 'RA updated'
1		ACCEPT	Mobile identity = P-TMSI-1 P-TMSI-1 signature
			Routing area identity = RAI-2
			Message not ciphered
16	>	ROUTING AREA UPDATING	Message not ciphered
		COMPLETE	
1			
1			
17	-<-	PAGING REQUEST TYPE 1	Mobile identity = P-TMSI-1
			Paging order is for TBF establishment.
18	>	UPLINK RLC DATA BLOCK	LLC PDU implicitly indicating paging response.
1			Message not ciphered
19	UE		The UE is switched off or power is removed
20	_	DETACH REQUEST	(see ICS).
20	>	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, PS detach'
			Message not ciphered

21	₩E		E operation mode A (see
			s repeated from step 1 to
		step 20.	

Note that due to the test of ciphering, it is in this test case indicated whether each message is ciphered or not.

12.6.2.2.5 Test requirements

Case1)

UE shall:

- initiate a PS attach procedure with information elements specified in the above Expected Sequence when UE is powered on or switched on.
- When the UE receive the AUTHENTICATION AND CIPHERING REQUEST message form SS during the PS attach procedure, with Ciphering indicator information element set to 'ciphering mode on', UE shall:
 - respond with an AUTHENTICATION AND CIPHERING RESPONSE message.
 - start ciphering and deciphering with the algorithm indicated by the Ciphering algorithm information element.
 - apply the ciphering that uses the cipher key determined during the authentication procedure.

Case2)

UE shall:

- initiate a routing area updating procedure with information elements specified in the above Expected Sequence.
- When the UE receives the AUTHENTICATION AND CIPHERING REQUEST message during the routing area updating procedure, with Ciphering indicator information element set to 'ciphering mode on', UE shall:
 - respond with an AUTHENTICATION AND CIPHERING RESPONSE message.
 - start ciphering and deciphering with the algorithm indicated by the Ciphering algorithm information element.
 - apply the ciphering that uses the cipher key determined during the authentication procedure.

12.6.2.3 Ciphering mode / IMEISV request

12.6.2.3.1 Definition

12.6.2.3.2 Conformance requirement

- 1) When the UE receives the AUTHENTICATION AND CIPHERING REQUEST message during the attach procedure, with Ciphering indicator information element set to 'ciphering mode on' and 'IMEISV requested', the User Equipment shall:
 - 1.1 responds with an AUTHENTICATION AND CIPHERING RESPONSE message
 - 1.2 include IMEISV
 - 1.3 start ciphering and deciphering with the algorithm indicated by the Ciphering algorithm information element
 - 1.4 the ciphering uses the cipher key determined during the authentication procedure
- 2) When the UE receives the AUTHENTICATION AND CIPHERING REQUEST message during the routing area updating procedure, with Ciphering indicator information element set to 'ciphering mode off' and 'IMEISV not requested', the User Equipment shall:

2.1 responds with an AUTHENTICATION AND CIPHERING RESPONSE message

2.2 not include IMEISV

2.3 not start ciphering

Reference

3G TS 24.008 clause 4.7.7

12.6.2.3.3 Test purpose

To test the behaviour of the UE with respect to return IMEISV on request only.

12.6.2.3.4 Method of test

Initial condition

System Simulator:

Two cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2.

Both cells are operating in network operation mode II.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No UE operation mode A Yes/No

UE operation mode C Yes/No

Switch off on button Yes/No

Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

A PS attach is performed, and the SS initiates an authentication and ciphering procedure. IMEISV is requested.

The UE initiates a routing area updating procedure, and the SS initiates a new authentication and ciphering procedure without requesting IMEISV.

Step	Direction	Message	Comments
	UE SS		
			The following messages are sent and shall be
_	00		received on cell A. The SS activates cell A.
4 2	SS LIE		The UE is set in UE operation mode C (see
-	UE		ICS). If UE operation mode C not supported,
			goto step 21.
3	UE		The UE is powered up or switched on and
			initiates an attach (see ICS).
4	>	ATTACH REQUEST	Attach type = 'PS attach'
			Mobile identity = IMSI
_		ALITUENTION TION AND	Message not ciphered
5	-<-	AUTHENTICATION AND	Request authentication.
		CIPHERING REQUEST	Ciphering on IMEISV requested
			Message not ciphered
6		AUTHENTICATION AND	RES
		CIPHERING RESPONSE	Mobile identity = IMEISV
			Message not ciphered
7		ATTACH ACCEPT	Attach result = 'PS attach'
			Mobile identity = P-TMSI-2
			P-TMSI-2 signature
			Routing area identity = RAI-1
			Message ciphered
8	>	ATTACH COMPLETE	Message ciphered
9		PAGING REQUEST TYPE 1	Mobile identity = P-TMSI-1
•		THE REGISTER OF THE T	Paging order is for TBF establishment.
			Message net ciphered
10	>	UPLINK RLC DATA BLOCK	LLC PDU implicitly indicating paging response.
			Message ciphered
			The following messages are sent and shall be
			received on cell B.
41	SS		Activate cell B with a lower signal strength than
			cell A The RF level of cell A is lowered until cell B is preferred by the UE.
12	>	ROUTING AREA UPDATING	Update type = 'RA updating'
+2		REQUEST	P-TMSI-2 signature
		REGOLOT	Routing area identity = RAI-1
			Message not ciphered
13		AUTHENTICATION AND	Request authentication.
		CIPHERING REQUEST	Ciphering off
			IMEISV not requested
		ALITHERITIC - TO COMPANY	Message not ciphered
14	>	AUTHENTICATION AND	RES
		CIPHERING RESPONSE	No IMEISV included
15		ROUTING AREA UPDATING	Message not ciphered Update result = 'RA updated'
10		ACCEPT	Mobile identity = P-TMSI-1
			P-TMSI-1 signature
			Routing area identity = RAI-2
			Message not ciphered
16	>	ROUTING AREA UPDATING	Message not ciphered
		COMPLETE	
17		PAGING REQUEST TYPE 1	Mobile identity = P-TMSI-1
++		FAGING REQUEST TYPE T	Paging order is for TBF establishment.
			Message not ciphered
18	>	UPLINK RLC DATA BLOCK	LLC PDU implicitly indicating paging response.
			Message not ciphered
19	UE		The UE is switched off or power is removed
			(see ICS).

20	>	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, PS detach'
			Message not ciphered
21	UE		The UE is set in UE operation mode A (see
			ICS) and the test is repeated from step 1 to
			step 20.

Note that due to the test of ciphering, it is in this test case indicated whether each message is ciphered or not.

12.6.2.3.5 Test requirements

Case1)

UE shall:

- initiate a PS attach procedure with information elements specified in the above Expected Sequence when UE is powered on or switched on.
- When the UE receives the AUTHENTICATION AND CIPHERING REQUEST message during the attach procedure, with Ciphering indicator information element set to 'ciphering mode on' and 'IMEISV requested', UE shall:
 - respond with the AUTHENTICATION AND CIPHERING RESPONSE message.
 - include the IMEISV in the AUTHENTICATION AND CIPHERING RESPONSE message.
 - start eiphering and deciphering with the algorithm indicated by the Ciphering algorithm information element.
 - apply the ciphering that uses the cipher key determined during the authentication procedure

Case2)

UE shall:

- initiate a routing area updating procedure with information elements specified in the above Expected Sequence.
- When the UE receives the AUTHENTICATION AND CIPHERING REQUEST message during the routing area updating procedure, with Ciphering indicator information element set to 'ciphering mode off' and 'IMEISV not requested', UE shall:
 - respond with the AUTHENTICATION AND CIPHERING RESPONSE message.
 - not include the IMEISV in the AUTHENTICATION AND CIPHERING RESPONSE message.
 - not start ciphering.

12.7 Identification procedure

The purpose of this procedure is to check that the UE gives its identity as requested by the network. If this procedure does not work, it will not be possible for the network to rely on the identity claimed by the UE.

12.7.1General Identification

12.7.1.1 Definition

12.7.1.2 Conformance requirement

1) When requested by the network the User Equipment shall send its IMSI.

- 2) When requested by the network the User Equipment shall send its IMEI as stored in the Mobile Equipment.
- 3) When requested by the network the User Equipment shall send its IMEISV as stored in the Mobile Equipment.

Reference

3G-3GPP TS 24.008 clauses 4.7.8

12.7.1.3 Test purpose

To verify that the UE sends identity information as requested by the system. The following identities can be requested: IMSI, IMEI and IMEISV.

12.7.1.4 Method of test

Initial condition

System Simulator:

Two cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC2. Both cells are operating in network operation mode II.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
UE operation mode C Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

The SS requests identity information from the UE:

- IMSI
- IMEI
- IMEISV

Step	Direction	Message	Comments
	UE SS		
1	SS		The UE is set in UE operation mode C (see ICS). If UE operation mode C not supported, goto step 14.
2	UE		The UE is powered up or switched on and initiates an attach (see ICS).
3	->	ATTACH REQUEST	Attach type = 'PS attach' Mobile identity = IMSI
4	<-	ATTACH ACCEPT	Attach result = 'PS only attached' Mobile identity = P-TMSI-1 P-TMSI-1 signature Routing area identity = RAI-1
5	->	ATTACH COMPLETE	
6	<-	IDENTITY REQUEST	Identity type = IMSI
7	->	IDENTITY RESPONSE	Mobile identity = IMSI
8	<-	IDENTITY REQUEST	Identity type = IMEI
9	->	IDENTITY RESPONSE	Mobile identity = IMEI
10	<-	IDENTITY REQUEST	Identity type = IMEISV
11	->	IDENTITY RESPONSE	Mobile identity = IMEISV
12	UE		The UE is switched off or power is removed (see ICS).
13	->	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, PS detach'
14	UE		The UE is set in UE operation mode A (see ICS) and the test is repeated from step 2 to step 13.

Specific message contents

None.

12.7.1.5 Test requirements

UE shall:

- initiate a PS attach procedure with information elements specified in the above Expected Sequence when UE is powered on or switched on.

When SS requests an IMSI with the IDENTITY REQUEST message, UE shall:

- send the IDENTITY RESPONSE message with the Mobile identity = IMSI.

When SS requests an IMEI with the IDENTITY REQUEST message, UE shall:

- send the IDENTITY RESPONSE message with the Mobile identity = IMEI.

When SS requests an IMEISV with the IDENTITY REQUEST message, UE shall:

- send the IDENTITY RESPONSE message with the Mobile identity = IMEISV.

12.8 GMM READY timer handling

The READY timer is not applicable for UMTS.

12.8.1 Definition

12.8.2 Conformance requirement

If a READY timer value is received by an UE capable of both UMTS and GSM in the ATTACH ACCEPT or the ROUTING AREA UPDATE ACCEPT messages, then the received value shall be stored by the UE in order to be used at an intersystem change from UMTS to GSM.

Reference

3G3GPP-TS 24.008 clause 4.7.2.1

12.8.3 Test purpose

To verify the functionality of the READY timer.

12.8.4 Method of test

12.8.4.1 Test procedure1

Initial condition

System Simulator:

Two cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1, cell B in MCC1/MNC1/LAC1/RAC1.

Both cells are operating in network operation mode II.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No
UE operation mode A Yes/No
Switch off on button Yes/No
Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

An attach is performed.

T3314; set to 60 seconds

Step	Direction	Message	Comments
	UE SS		
			The following messages are sent and shall be
			received on cell A.
1	SS		The SS activates cell A.
2	UE		The UE is set in UE operation mode A (see
			ICS). If UE operation mode A not supported set
			the UE in operation mode C.
			The UE is powered up or switched on and
_			initiates an attach (see ICS).
3	->	ATTACH REQUEST	Attach type = 'PS attach'
			Mobile identity = IMSI
4	<-	ATTACH ACCEPT	Attach result = 'PS only attached'
			Mobile identity = P-TMSI-2
			P-TMSI-2 signature
			Routing area identity = RAI-1
_		ATTACH COMPLETE	T3314 = 60 seconds
5	->	ATTACH COMPLETE	
6	-> UE		The UE is switched off or power is removed
	0		(see ICS).
7	->	DETACH REQUEST	Message not sent if power is removed.
'		DET. COLOT	Detach type = 'power switched off, PS detach'

Specific message contents

None.

12.8.5 Test requirements

When UE receives the ATTACH ACCEPT or the ROUTING AREA UPDATE ACCEPT messages, UE shall:

- store the received READY timer value.

3GPP TSG-T1 Meeting #10 Copenhagen, Denmark, 8-9 February, 2001 Tdoc T1-010090

3GPP TSG-T1/SIG SWG Meeting #15 Copenhagen, Denmark, 5-7 Feb 2001 TSG T1S-010???

	CHANGE REQUEST			
*	34.123-1 CR 051			
For <u>HELP</u> o	n using this form, see bottom of this page or look at the pop-up text over the ★ symbols.			
Proposed chang	ge affects: 第 (U)SIM ME/UE X Radio Access Network Core Network			
Title:	₩ Update to 16. SMS test specification			
Source:	₩ Denso			
Work item code	:# Date : # 2001-02-05			
Category:	Release: ■ R99			
	Use one of the following categories: F (essential correction) A (corresponds to a correction in an earlier release) B (Addition of feature), C (Functional modification of feature) D (Editorial modification) Detailed explanations of the above categories can be found in 3GPP TR 21.900. Use one of the following releases: 2 (GSM Phase 2) R96 (Release 1996) R97 (Release 1997) R98 (Release 1998) R99 (Release 1999) REL-4 (Release 4) REL-5 (Release 5)			
Reason for change: # This CR has been modified according to new release of core specifications (2000-12).				
Summary of cha	nnge: Modification of references and terminology, and correction of initial conditions.			
Consequences not approved:	if #			
Clauses affected	d: 第 16.1, 16.2, 16.3, 16.4			
Other specs affected:	# Other core specifications # Test specifications O&M Specifications			
Other comment	s: X			

How to create CRs using this form:

Comprehensive information and tips about how to create CRs can be found at: http://www.3gpp.org/3G Specs/CRs.htm. Below is a brief summary:

- 1) Fill out the above form. The symbols above marked \$\mathbb{X}\$ contain pop-up help information about the field that they are closest to.
- 2) Obtain the latest version for the release of the specification to which the change is proposed. Use the MS Word "revision marks" feature (also known as "track changes") when making the changes. All 3GPP specifications can be downloaded from the 3GPP server under ftp://www.3gpp.org/specs/ For the latest version, look for the directory name with the latest date e.g. 2000-09 contains the specifications resulting from the September 2000 TSG meetings.
 - 3) With "track changes" disabled, paste the entire CR form (use CTRL-A to select it) into the specification just in front of the clause containing the first piece of changed text. Delete those parts of the specification which are not relevant to the change request.

16 Short message service (SMS)

Ref.: <u>3GPP</u> TS 23.040, <u>3GPP</u> TS 24.011 (point to point)

3GPP TS 23.041, (cell broadcast)

General

The purpose of these tests is to verify that the UE can handle <u>UMTSIu mode</u> system functions when submitting or receiving Short Messages (SM) between UE and a short message service centre as described in <u>3GPP TS</u> 23.040.

The procedures are based upon services provided by the Mobility Management (MM) sublayer and GPRS Mobility Managemant(GMM) sublayer which are not tested in this case.

The SMS comprises three basic services; SMS point to point services on CS mode, on PS mode and SMS cell broadcast service. The SMS point to point services on CS mode shall work in an active UE at any time independent of whether or not there is a speech or data call in progress. The SMS point to point services on PS mode shall work in an active UE at any time independent of whether or not there is a PDP context in progress. The SMS cell broadcast service only works when the UE is in idle mode.

Since the timer TC1M currently is not standardized, the value of TC1M shall be declared by the manufacturer (to be used in sub-clauses 16.1.1 and 16.1.2).

The manufacturer shall declare whether SMS messages are stored in the USIM and/or the ME. This shall be referred to as the SMS message store in the following tests.

Unless otherwise stated default message contents from 3GPP TS 34.108 applies for following tests.

16.1 Short message service point to point on CS mode

All of test cases in this sub-clause are applied to UE supporting CS mode.

16.1.1 SMS mobile terminated

16.1.1.1 Definition

16.1.1.2 Conformance requirements

An active UE shall be able to receive short message TPDU (SMS-DELIVER) at any time, independently of whether or not there is a speech or data call in progress. A report will always be returned to the SC, confirming that the UE has received the short message.

Reference

3GPP TS 23.040, sub-clause 3.1.

16.1.1.3 Test purpose

To verify the ability of a UE to receive and decode the SMS where provided for the point to point service.

16.1.1.4 Method of test

Initial Conditions

- System simulator:
 - 1 cell, default parameters.

- User Equipment:
 - the UE shall be in "Idle, updated" state MM-state "Idle, updated";
 - the SMS message storage shall be empty.

Related ICS/IXIT Statements

Support for Short message MT/PP.

Description of the basic procedures to display a mobile terminated short message.

The value of timer TC1M.

Whether SMS messages are stored in the USIM and/or the ME.

Support for call control state U10.

Test procedure

a) Mobile terminates establishment of Radio Resource Connection. After the completion of RRC Connection SS authenticates UE.

After the SS receives SECURITY MODE COMPLETE, the SS sends a CP-DATA message. The information element of the CP-DATA message will be RP-DATA RPDU (SMS DELIVER TPDU).

- b) The SS waits a maximum of 25 seconds for the CP-ACK message and then a maximum of 60 seconds for the CP-DATA message containing the RP-ACK RPDU.
- c) The SS sends a CP-ACK to the UE within TC1M with no further CP-DATA messages and the SS initiates RRC Connection release.
- d) Steps a), b) and c) are repeated but the first CP-DATA message from the UE is not acknowledged. The second CP-DATA message from the UE is acknowledged by a CP-ACK within a time TC1M.
- e) Steps a) and b) are repeated. The SS is configured not to send CP-ACK. Then maximum 3 CP-DATA retransmissions may occur. After a duration of TC1M + 5 seconds after the last CP-DATA retransmission the SS then initiates the channel release. The 5 seconds is the appropriate time to wait to verify that the UE does not send more than the maximum allowed (3) CP-DATA retransmissions.
- f) The SMS message store shall be cleared manually by the operator.
- g) A data or speech call is established on a DTCH with the SS and the state U10 of call control is entered. The SS sends a PAGING TYPE 2.

The SS sends a CP-DATA message. The information element of the CP-DATA message will be RP-DATA RPDU (SMS DELIVER TPDU). The SS waits a maximum of 25 seconds for the CP-ACK message and then a maximum of 60 seconds for the CP-DATA message containing the RP-ACK RPDU.

- h) The SS sends a CP-ACK to the UE within TC1M with no further CP-DATA messages and the SS initiates RRC Connection release. The SMS message store shall be cleared manually by the operator.
- i) Steps g) and h) are repeated but the first CP-DATA message from the UE is not acknowledged. The second CP-DATA message from the UE is acknowledged by a CP-ACK within a time TC1M.
- j) Step g) is repeated. The SS is configured not to send CP-ACK. Then maximum 3 CP-DATA retransmissions may occur. After a duration of TC1M + 15 seconds after the last CP-DATA retransmission the SS initiates the channel release. The 15 seconds is the appropriate time to wait to verify that the UE does not send more than the maximum allowed (3) CP-DATA retransmissions (during a call in progress).

k) A data or speech call is established on a DTCH with the SS and the state U10 of call control shall be entered. DTCHThe speech call is cleared by the SS with a disconnect message. (The call clearing is continued on the DCCH in parallel to the following exchange of messages related to SMS).

The SS sends a CP-DATA RPDU (SMS DELIVER TPDU) message. The information element of the CP-DATA message is RP-DATA.

The SS waits a maximum of 25 seconds for the CP-ACK message and then a maximum of 60 seconds for the CP-DATA message containing the RP-ACK RPDU.

The SS sends a CP-ACK to the UE within TC1M with no further CP-DATA messages and the SS initiates channel release.

The SMS message store shall be cleared manually by the operator.

 A data or speech call is established with the SS and the state U10 of call control is entered. After the SS sends a PAGING TYPE 2, the speech call shall be cleared from the UE. (The call clearing is continued in parallel to the following exchange of messages related to SMS).

The SS sends a CP-DATA message. The information element of the CP-DATA message is RP-DATA RPDU (SMS DELIVER TPDU).

The SS waits a maximum of 25 seconds for the CP-ACK message and then a maximum of 60 seconds for the CP-DATA message containing the RP-ACK RPDU.

The SS sends a CP-ACK to the UE within TC1M with no further CP-DATA messages and the SS initiates RRC Connection release.

The SMS message store shall be cleared manually by the operator.

Expected sequence

Step	p Direction		Message	Comments
	UE	SS		
1		•	Mobile terminated establishment	See <u>3GPP</u> TS34.108
			of Radio Resource Connection	
2		·>	PAGING RESPONSE	
3	<		AUTHENTICATION REQUEST	
4		·>	AUTHENTICATION RESPONSE	
5	<		SECURITY MODE COMMAND	
6		·>	SECURITY MODE COMPLETE	
7			CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU)
8	S	S		Waits max 25 seconds for CP-ACK
9		·>	CP-ACK	
10	S	S		Waits max 60 seconds for RP-ACK RPDU
11		·>	CP-DATA	Contains RP-ACK RPDU
12			CP-ACK	
13	U	ΙE		There should be no further CP-DATA messages until the UF aborts the RRC connection.
14	U	F		The UE shall indicate that an SM has arrived. If the UE
1-7		_		provides the functionality to display MT messages, it is
				checked that the correct message is displayed
15			Mobile terminated establishment	See 3GPP TS34.108
10			of Radio Resource Connection	000 0011 100 11100
16		·>	PAGING RESPONSE	
17			AUTHENTICATION REQUEST	
18	1	·>	AUTHENTICATION RESPONSE	
19			SECURITY MODE COMMAND	
20	1	·>	SECURITY MODE COMPLETE	
21			CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU)
22	S	S		Waits max 25 seconds for CP-ACK
23	_	·>	CP-ACK	
24	S	S		Waits max 60 seconds for RP-ACK RPDU

01	Discotion	Manager	0
Step	Direction	Message	Comments
	UE SS		
25	>	CP-DATA	First CP-DATA from UE, contains RP-ACK RPDU
26	SS		First CP-DATA message not acknowledged by SS
27	>	CP-DATA	Retransmitted CP-DATA from UE within twice TC1M,
			after step 25, contains RP-ACK RPDU
28	<	CP-ACK	Second CP_DATA message is acknowledged
29	ÛE		There should be no further CP-DATA messages until the
	0_		UE aborts the RRC connection.
30	UE		The UE shall indicate that an SM has arrived. If the UE
	02		provides the functionality to display MT messages, it is
			checked that the correct message is displayed
31		Mobile terminated establishment	See 3GPP TS34.108
		of Radio Resource Connection	000 0011 1004.100
32	>	PAGING RESPONSE	
33	<	AUTHENTICATION REQUEST	
34	>	AUTHENTICATION RESPONSE	
35	<i>></i>	SECURITY MODE COMMAND	
36		SECURITY MODE COMPLETE	
37	>	CP-DATA	Contains DD DATA DDDLL (SMS DELIVED TDDLI)
38	< SS	OI PATA	Contains RP-DATA RPDU (SMS DELIVER TPDU) Waits max 25 seconds for CP-ACK
38		CP-ACK	vvails max 20 seconds for GF-AGN
40	> SS	OI -ACK	Waits max 60 seconds for RP-ACK RPDU
41		CD DATA	Contains RP-ACK RPDU
	> SS	CP-DATA	
42	55	CD DATA	First CP-DATA message not acknowledged by SS
43		CP-DATA	Retransmitted CP-DATA from UE within twice TC1M after
4.4	00		step 41, contains RP-ACK RPDU
44	SS		Retransmitted CP-DATA message not acknowledged by
45	UE		SS
45	UE		Depending upon the maximum number of CP-DATA
			retransmissions implemented, step 43 and 44 may be
40		DDG GONNEGTION DELEAGE	repeated.
46	<	RRC CONNECTION RELEASE	RRC connecton is released after a duration of TC1M + 5
4-		DDG GGNINEGTIGNI DEL EAGE	seconds after the last CP-DATA retransmission
47	>	RRC CONNECTION RELEASE	
40		COMPLETE	
48	UE		The UE shall indicate that an SM has arrived. If the UE
			provides the functionality to display MT messages, it is
4.0	00		checked that the correct message is displayed
49	SS		A data or speech call is established on a DTCH and the
			state U10 of call control is entered.
50		DAGING TYPE O	
		PAGING TYPE 2	
51	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU)
52	SS	00.4014	Waits max 25 seconds for CP-ACK
53	>	CP-ACK	W '' 00 1 / 00 10 / 00 / 00 / 00 / 00 / 0
54	SS	OD DATA	Waits max 60 seconds for RP-ACK RPDU
55	>	CP-DATA	Contains RP-ACK RPDU
56	<	CP-ACK	Di un u u
57	<	DISCONNECT	Disconnect the active call
58	>	RELEASE	
59	UE		The UE shall indicate that an SM has arrived. If the UE
			provides the functionality to display MT messages, it is
			checked that the correct message is displayed
60	UE		Clear the SMS message store
61	SS		A data or speech call is established on a DTCH and the
			state U10 of call control is entered.
62	<	PAGING TYPE 2	
63	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU)
64	SS		Waits max 25 seconds for CP-ACK
65	>	CP-ACK	
66	SS		Waits max 60 seconds for RP-ACK RPDU
67	>	CP-DATA	First CP-DATA from UE, contains RP-ACK RPDU
68	SS		First CP-DATA message not acknowledged by SS
69	>	CP-DATA	Retransmitted CP-DATA message within twice TC1M
			after step 67, contains RP-ACK RPDU

Step	Direction	Message	Comments
Otep	UE SS	- Wessage	Comments
70		CD ACK	Cocond CD DATA masses is sales
70	<	CP-ACK	Second CP-DATA message is acknowledged
71	<	DISCONNECT	Disconnect the active call
72	>	RELEASE	
74	UE		There should be no further CD DATA messages until the
74	UE UE		There should be no further CP-DATA messages until the UE aborts the RRC connection
75	UE		The UE shall indicate that an SM has arrived. If the UE
73			provides the functionality to display MT messages, it is
			checked that the correct message is displayed
76	UE		Clear the SMS message store
77	SS		A data or speech call is established on a DTCH and the
			state U10 of call control is entered.
78	<	PAGING TYPE 2	
	>		
79	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU)
80	SS		Waits max 25 seconds for CP-ACK
81	>	CP-ACK	
82	SS		Waits max 60 seconds for RP-ACK RPDU
83	>	CP-DATA	First CP-DATA from UE, contains RP-ACK RPDU
84	SS	05 5474	First CP-DATA message not acknowledged by SS
85	>	CP-DATA	Transmitted CP-DATA message within twice TC1M after
00	00		step 83, contains RP-ACK RPDU
86	SS		Retransmitted CP-DATA message not acknowledged by SS
87	UE		Depending on the maximum number of CP-DATA
01	UE UE		retransmissions implemented, step 85-86 may be
			repeated. The maximum number of retransmissions may
			however not exceed three.
88	<	RRC CONNECTION RELEASE	RRC CONNECTION is released after a duration of TC1M
	`	THE SOUND THE TREE TO SEE	+ 15 seconds after the last CP-DATA retransmission.
89	>	RRC CONNECTION RELEASE	
		COMPLETE	
90	UE		The UE shall indicate that an SM has arrived. If the UE
			provides the functionality to display MT messages, it is
			checked that the correct message is displayed
91	UE		Clear the SMS message store
92	SS		A data or speech call is established on a DTCH and the
		DA ON 10 TV /DE 0	state U10 of call control is entered.
93	<	PAGING TYPE 2	Sent on DCCH associated with the DTCH
0.4		DISCONNECT	The speech call is cleared by the SS. The call clearing is
94	<	DISCONNECT	continued in parallel to the following exchange of
			messages related to SMS.
95	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU)
96	SS	J. 5/1/1	Waits max 25 seconds for CP-ACK
97	>	CP-ACK	
98	SS	_	Waits max 60 seconds for RP-ACK RPDU
99	>	CP-DATA	Contains RP-ACK RPDU
100	<	CP-ACK	
101	UE		There should be no further CP-DATA messages until the
			UE aborts the RR connection.
102	UE		The UE shall indicate that an SM has arrived. If the UE
			provides the functionality to display MT messages, it is
400			checked that the correct message is displayed
103	UE		Clear the SMS message store
104	SS		A data or speech call is established on a DTCH and the state U10 of call control is entered.
105		PAGING TYPE 2	state o to or call control is entered.
105	< >	DISCONNECT	The speech call is cleared from the UE. The call clearing
100		DISCOULTED !	is continued in parallel to the following exchange of
			messages related to SMS.
107	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU)
108	<	RELEASE	This message is likely to be sent before all of the CP-
			DATA message has been sent on the DCCH.
			- -

Step	Direction	Message	Comments
	UE SS		
109	>	RELEASE COMPLETE	
110	>	CP-ACK	shall be sent before 25 seconds after the start of step 107
111	SS		Waits max 60 seconds for RP-ACK RPDU
112	>	CP-DATA	Contains RP-ACK RPDU
113	<	CP-ACK	
114	UE		There should be no further CP-DATA messages until the
			UE aborts the RRC connection.
115	UE		The UE shall indicate that an SM has arrived. If the UE
			provides the functionality to display MT messages, it is
			checked that the correct message is displayed
116	UE		Clear the SMS message store

NOTE: Time values for SS wait time are chosen sufficiently high to be sure that the UE has enough time to respond to the different messages.

Specific Message Contents

SMS DELIVER TPDU

Information element	Comment Value
TP-UDL	160
TP-UD (140 octets)	text of message (160 characters)

NOTE: The 160 characters in TP-UD shall include at least one occurrence of each character in the default alphabet (see <u>3GPP TS 23.038</u>, sub-clause 6.2.1).

16.1.1.5 Test requirements

After step 7 UE shall receive SMS-DELIVER TPDU and send CP-ACK within 25 seconds and CP-DATA containing RP-ACK within 60 seconds.

After step 25 UE shall retransmit CP-DATA containing RP-ACK within twice TC1M.

After step 30 UE shall indicate that an SM has arrived.

After step 43 UE shall repeat CP-DATA retransmissions as many times as the decided maximum number.

After step 48 UE shall indicate that an SM has arrived.

After step 51 UE shall receive SMS-DELIVER TPDU and send CP-ACK within 25 seconds and CP-DATA containing RP-ACK within 60 seconds.

After step 67 UE shall retransmit CP-DATA containing RP-ACK within twice TC1M.

After step 74 UE shall indicate that an SM has arrived.

After step 79 UE shall repeat CP-DATA retransmissions as many times as the decided maximum number.

After step 90 UE shall indicate that an SM has arrived.

After step 95 UE shall receive SMS-DELIVER TPDU and send CP-ACK within 25 seconds and CP-DATA containing RP-ACK within 60 seconds.

After step 107 UE shall receive SMS-DELIVER TPDU and send CP-ACK within 25 seconds and CP-DATA containing RP-ACK within 60 seconds.

16.1.2 SMS mobile originated

16.1.2.1 Definition

16.1.2.2 Conformance requirements

An active UE shall be able to submit short message TPDU (SMS-SUBMIT) at any time, independently of whether or not there is a speech or data call in progress.

Reference

3GPP TS 23.040, sub-clause 3.1.

16.1.2.3 Test purpose

To verify that the UE is able to correctly send a short message where the SMS is provided for the point to point service. The test also verifies that the UE is capable of simultaneously receiving a network originated SM whilst sending a mobile originated SM.

16.1.2.4 Method of test

Initial Conditions

- System simulator:
 - 1 cell, default parameters.
- User Equipment:
 - the UE shall be in "Idle, updated" state MM-state "Idle, updated";
 - the SMS message storage shall be empty.

Related ICS/IXIT Statements

Support for Short message MO/PP.

Description of the basic procedures to display a mobile originated short message.

Support for state U10 of call control.

The value of timer TC1M.

Whether SMS messages are stored in the USIM and/or the ME.

Maximum length (characters) of a mobile originated short message.

Test procedure

- a) The UE shall be set up to send a SM to the SS. The SS responds to RRC CONNECTION REQUEST by allocating aCCCH. The SS recievesRRC CONNECTION SETUP COMPLETE on DCCH and then performs the authentication.
- b) After receiving SECURITY MODE COMMAND UE shall send SECURITY COMMAND COMPLETE.
- c) The SS responds to the CP-DATA containing RP-DATA RPDU (SMS SUBMIT TPDU) from the UE with a CP-ACK message within TC1M followed by a CP-DATA message containing the correct RP-ACK RPDU. The SS waits a maximum of 25 seconds for the CP-ACK message.
- d) The SS sends a channel release message to the UE.

- e) Steps a) and b) are repeated. The SS is configured not to send the CP-ACK message. Then maximum 3 CP-DATA retransmissions may occur. After a duration of TC1M + 5 seconds after the last CP-DATA retransmission the SS initiates channel release. The 5 seconds is the appropriate time to wait to verify that the UE does not send more than the maximum CP-DATA retransmissions.
- f) Steps a) and b) are repeated. On receipt of the CP-DATA from the UE the SS sends a CP-ERROR message within TC1M containing a "Network Failure" cause. Then the SS initiates channel release.
- g) g) A data or speech call is established with the SS and the state U10 of call control is entered. The UE is setup to send an SM to the SS. After the reception of the CM SERVICE REQUEST, the SS sends a CM SERVICE ACCEPT message.
- h) h) The SS responds to the CP-DATA containing RP-DATA RPDU (SMS SUBMIT TPDU) from the UE with a CP-ACK message within TC1M followed by a CP-DATA message containing the correct RP-ACK RPDU. The SS waits a maximum of 25 seconds for the CP-ACK message. Then the SS sends a channel release message to the UE.
- i) Step g) is repeated. The SS is configured not to send the CP-ACK message. Then maximum 3 CP-DATA retransmissions may occur. After a duration of TC1M + 15 seconds after the last CP-DATA retransmission the SS initiates channel release. The 15 seconds is the appropriate time to wait to verify that the UE does not send more than the maximum CP-DATA retransmissions (during a call in progress).
- j) The SS is configured to receive a mobile originated SM. Steps a) and b) are repeated and, using the end of the CP-DATA message from the UE as a trigger, the SS sends a SM to the UE. In this case a new transaction identifier shall be used in the CP messages of SMS mobile terminated.
- k) The UE is set up to send an SM to the SS. On receipt of the CM SERVICE REQUEST the SS sends a CM SERVICE REJECT message with the reject cause set to "Service Option not supported" or "Service Option temporarily out of order". After 5 seconds the SS initiates channel release.

Expected sequence

Step	Direction	Message	Comments
	UE SS		
1	<	SYSTEM INFORMATION	BCCH
2	>	RRC CONNECTION REQUEST	CCCH
3	<	RRC CONNECTION SETUP	CCCH
4	>	RRC CONNECTION SETUP	DCCH
		COMPLETE	
5	>	CM SERVICE REQUEST	
6	<	AUTHENTICATION REQUEST	
7	>	AUTHENTICATION RESPONSE	
8	<	SECURITY MODE COMMAND	
9	>	SECURITY MODE COMPLETE	
10	>	CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU)
11	<	CP-ACK	Sent within TC1M after step 10
12	<	CP-DATA	Contains RP-ACK RPDU
13	SS		Waits max 25 seconds for CP-ACK
14	>	CP-ACK	
15	<	RRC CONNECTION RELEASE	RRC connection is released.
16	>	RRC CONNECTION RELEASE COMPLETE	
17	<	SYSTEM INFORMATION	BCCH
18	>	RRC CONNECTION REQUEST	CCCH
19	<	RRC CONNECTION SETUP	CCCH
20	>	RRC CONNECTION SETUP	DCCH
		COMPLETE	
21			
22	>	CM SERVICE REQUEST	
23	<	AUTHENTICATION REQUEST	
24	>	AUTHENTICATION RESPONSE	

Step	Direction	Message	Comments
Otop	UE SS	oodge	Comments
25	<	SECURITY MODE COMMAND	
26	>	SECURITY MODE COMPLETE	
27	>	CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU)
28	SS		SS configured not to send CP-ACK
29	>	CP-DATA	Retransmitted CP-DATA message within twice TC1M
			after step 27
30	UE		Depending on the maximum number of CP-DATA
			retransmissions implemented, step 29 may be repeated.
			The maximum number of retransmissions may however not exceed three.
31	<	RRC CONNECTION RELEASE	RRC CONNECTION is released after a duration of TC1M
31		RRC CONNECTION RELEASE	+ 5 seconds after the last CP-DATA retransmission.
32	>	RRC CONNECTION RELEASE	To decorde and the last of Britist of an anomicolosis.
"-		COMPLETE	
33	<	SYSTEM INFORMATION	ВССН
34	>	RRC CONNECTION REQUEST	CCCH
35	<	RRC CONNECTION SETUP	CCCH
36	>	RRC CONNECTION SETUP	DCCH
		COMPLETE	
		014 055) #05 550 #507	
37	>	CM SERVICE REQUEST AUTHENTICATION REQUEST	
38 39	<	AUTHENTICATION REQUEST	
40	> <	SECURITY MODE COMMAND	
41	>	SECURITY MODE COMPLETE	
42	>	CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU)
43	<	CP-ERROR	Sent within TC1M containing "Network Failure" cause.
44	<	RRC CONNECTION RELEASE	RRC CONNECTION is released.
45	>	RRC CONNECTION RELEASE	
		COMPLETE	
46	SS		A data or speech call is established on a DTCH and the
			state U10 of call control is entered.
47	UE	OM OFFINIOF PEOUFOT	The UE is set up to send an SM
48	>	CM SERVICE REQUEST	CM service type set to "short message "
49 50	<	CM SERVICE ACCEPT CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU)
51	> <	CP-ACK	Sent within TC1M after step 50
52	<	CP-DATA	Contains RP-ACK RPDU
53	ŝs	or shink	Waits max 25 seconds for CP-ACK
54	>	CP-ACK	
55	<	RRC CONNECTION RELEASE	RRC CONNECTION is released.
56	>	RRC CONNECTION RELEASE	
		CONPLETE	
57	SS		A data or speech call is established on a DTCH and the
	_	CM SERVICE REQUEST	state U10 of call control is entered.
58 50	>	CM SERVICE REQUEST CM SERVICE ACCEPT	CM service type set to "short message "
59 60	< >	CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU)
61	SS	OI DATA	SS configured not to send CP-ACK
62	>	CP-DATA	Transmitted CP-DATA message within twice TC1M after
"-		S. 27	step 60
63	UE		Depending on the maximum number of CP-DATA
			retransmissions implemented, step 62 may be repeated.
			The maximum number of retransmissions may however
			not exceed three.
64	<	RRC CONNECTION RELEASE	RRC CONNECTION is released after a duration of TC1m
65		DDC CONNECTION DELEASE	+ 15 seconds after the last CP-DATA retransmission.
65	>	RRC CONNECTION RELEASE COMPLETE	
66	<	SYSTEM INFORMATION	вссн
67	>	RRC CONNECTION REQUEST	CCCH
68	<	RRC CONNECTION SETUP	CCCH
69	>	RRC CONNECTION SETUP	DCCH
		COMPLETE	
•	•	•	•

Step	Direction	Message	Comments
-	UE SS	_	
70	>	CM SERVICE REQUEST	
71	<	AUTHENTICATION REQUEST	
72	>	AUTHENTICATION RESPONSE	
73	<	SECURITY MODE COMMAND	
74	>	SECURITY MODE COMPLETE	
75	>	CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU)
76	SS		The SS sends an SM to the UE triggered by the end of
		00.0474	the CP-DATA message from the UE
77	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU)
78	UE		The UE shall correctly receive the SM and indicate that a
			message has arrived. If the UE provides the functionality
			to display MT messages, it is checked that the correct
			message is displayed. In the MO case the UE shall send
			the CP-ACK message with transaction identifier assigned
			to this transfer. In the MT case the UE shall send a CP-
			ACK message and a CP-DATA message containing the
			RP-ACK RPDU. The transaction identifier shall be the
70		RRC CONNECTION REQUEST	same as chosen by the SS for the MT transfer.
79 80	>	RRC CONNECTION REQUEST	
81	<	RRC CONNECTION SETUP	
01	>	COMPLETE	
82	>	COMPLETE CM SERVICE REQUEST	. CM service type set to "short message transfer"
83	<i>></i>	CM SERVICE REQUEST	Reject cause set to "Service Option not supported" or
03	ζ	OW SERVICE REJ	"Service Option temporarily out of order"
84	UE		Oetvice Option temporality out of order
85	<	RRC CONNECTION RELEASE	Sent 5 seconds after CM SERVICE REJ
86	>	RRC CONNECTION RELEASE	Jeni J Seconds after CIVI SERVICE REJ
30	>		
		COMPLETE	

NOTE: Time values for SS wait times are chosen sufficiently high to be sure that the UE has enough time to respond to the different messages.

Specific Message Contents

SMS SUBMIT TPDU

Information element	Comment Value
TP-UDL	as applicable
TP-UD (140 octets max)	maximum number of characters (text of message) as
	defined by the manufacturer (see ICS/IXIT)

16.1.2.5 Test requirements

After step 9 UE shall send a CP-DATA containing RP-data. The RP-DATA shall contain SMS SUBMIT TPDU.

After step 27 UE shall retransmit a CP-DATA containing RP-data. The RP-DATA shall contain SMS SUBMIT TPDU.

After step 44 UE shall send the RRC CONNECTION RELEASE COMPLETE.

After step 49 UE shall send a CP-DATA containing RP-data. The RP-DATA shall contain SMS SUBMIT TPDU.

After step 60 UE shall repeat CP-DATA retransmissions as many times as the decided maximum number.

After step 77 UE shall correctly receive the SM and indicate that a message has arrived.

After step 83 UE shall not send CP-DATA.

16.1.3 Test of memory full condition and memory available notification:

The Memory Available Notification provides a means for the UE to notify the network that it has memory available to receive one or more short messages. The SMS status field in the USIM contains status information on the "memory available" notification flag.

16.1.3.1 Definition

16.1.3.2 Conformance requirement

- 1. When a mobile terminated message is Class 2, the UE shall ensure that the message has been transferred to the SMS data field in the USIM before sending an acknowledgement to the SC. The UE shall return a protocol error message if the short message cannot be stored in the USIM and there is other short message storage available in the UE. If all the short message storage in the UE is already in use, the UE shall return "memory capability exceeded".
- 2. When the UE rejects a short message due to lack of available memory capability the need to transfer notification shall be stored in the USIM.
- 3. If the memory capability becomes available because memory is cleared, the value of the memory capability exceeded notification flag in the USIM is read. If the flag is set, the UE notifies the network that memory capability is now available. After a positive acknowledgement from the network, the ME unsets the memory capability exceeded notification flag in the USIM.

References

- <u>3GPP</u> TS 23.040, sub-clause 9.2.3.10, <u>3GPP</u> TS 23.038, clause 4.
- 3GPP TS 23.040, sub-clause 10.3 (operation 14).
- <u>3GPP</u> TS 23.040, sub-clause 10.3 (operation 14).

16.1.3.3 Test purpose

- 1. To verify that the UE sends the correct acknowledgement when its memory in the USIM becomes full.
- 2. To verify that the UE sends the correct acknowledgement when its memory in the ME and the USIM becomes full, and sets the "memory exceeded" notification flag in the USIM.
- 3. To verify that the UE performs the "memory available" procedure when its message store becomes available for receiving short messages, and only at this moment.

16.1.3.4 Method of test

Initial conditions

- System Simulator:
 - 1 cell, default parameters.
- User Equipment:
 - the UE shall be in the idle updated state MM-state "Idle, updated";
 - the SMS message storage shall be empty;
 - the UE shall be connected to the USIM simulator. The following shall be present in the USIM simulator:
 - EF_{SMS} with at least one record;
 - EF_{SMSstatus}, with SMS "Memory Cap. Exceed" notification flag set to "memory available";

- Service no. 4 (SMS) in EF_{SST} set to allocated and activated.
- for storing of Class 1 Short Messages the UE shall be set up to store Short Messages in the ME memory (by way of MMI, as described in ICS/IXIT statement).

Related ICS/IXIT Statements

Support for Short message MT/PP.

Description of the basic procedures to display a mobile terminated short message.

Whether SMS messages are stored in the USIM and/or the ME.

The value of timer TC1M.

Test procedure

- a) step a) of sub-clause 16.1.5.3 (test of Class 2 Short Messages) is repeated until the UE sends a negative acknowledgement (RP-ERROR). The USIM simulator shall indicate if the "memory capability exceeded" notification flag has been set on the USIM.
- b) a Class 1 Short Message is sent to the UE.
- c) step b) is repeated until the UE sends a negative acknowledgement (RP-ERROR). The USIM simulator shall indicate if the "memory capability exceeded" notification flag has been set on the USIM.
- d) a Short Message is sent to the UE with the DCS field of the SMS-DELIVER TPDU set to 0.
- e) the SS prompts the operator to read a short message and to remove it from the message store of the UE.
- f) the SS waits for a RRC CONNECTION REQUEST from the UE, and sends a RRC CONNECTION SETUP.
- g) after the SS receives a RRC CONNECTION SETUP COMPLETE, the SS authenticates the UE and activates ciphering.
- h) the SS answers to the RP-SMMA from the UE with a CP-DATA containing a RP-ACK RPDU.
- i) after the UE has acknowledged the CP-DATA with a CP-ACK, the SS releases the RRC connection. The USIM simulator shall indicate if the "memory capability exceeded" notification flag has been unset on the USIM.
- j) step e) is repeated.

Step	Direction	Message	Comments
	UE SS		
1		Mobile terminated establishment	See <u>3GPP</u> TS34.108
		of Radio Resource Connection	
2	>	PAGING RESPONSE	
3	<	AUTHENTICATION REQUEST	
4	>	AUTHENTICATION RESPONSE	
5	<	SECURITY MODE COMMAND	
6	>	SECURITY MODE COMPLETE	
7	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU), Class
			2 Short Message
8	SS		Waits max 25 seconds for CP-ACK
9	>	CP-ACK	
10	SS		Waits max 60 seconds for RP-ACK RPDU
11	>	CP-DATA	Contains RP-ACK RPDU
12	<	CP-ACK	Within TC1M after step 11

	Step	Direction UE SS	Message	Comments
	13	<	RRC CONNECTION RELEASE RRC CONNECTION RELEASE	RRC connection is released. Step 1-13 is repeated until UE sends a negative acknowledgement (RP-ERROR) in step 16. The RP-ERROR RPDU cause field shall be "Protocol error, unspecified" if there is message capability in the ME, or "Memory capability exceeded" if there is no message capability in the ME. If the total memory store of the UE is full, the ME shall set the "memory capability exceeded" notification flag on the USIM.
ı		>	COMPLETE	
	15 16 17 18 19 20	> < > <	Mobile terminated establishment of Radio Resource Connection PAGING RESPONSE AUTHENTICATION REQUEST AUTHENTICATION RESPONSE SECURITY MODE COMMAND SECURITY MODE COMPLETE	See <u>3GPP</u> TS34.108
	21	< SS	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU), Class 1 Short Message Waits max 25 seconds for CP-ACK
	23	>	CP-ACK	
	24 25	SS >	CP-DATA	Waits max 60 seconds for RP-ACK RPDU Shall contain RP-ACK RPDU if there is memory capability in the ME. If not it shall contain RP-ERROR RPDU which cause field shall be "memory capability exceeded". If the total memory store of the UE now becomes full at this step, the ME shall set the "memory cap. exceed" notification flag on the USIM.
	26 27	< <	CP-ACK RRC CONNECTION RELEASE	Within TC1M after step 25 RRC connection is released. Step 15-27 is repeated until the UE sends an RP-ERROR. The USIM simulator shall indicate if the "memory capability exceeded" notification flag has been set on the USIM.
	28	>	RRC CONNECTION RELEASE COMPLETE	
	29		Mobile terminated establishment of Radio Resource Connection	See <u>3GPP</u> TS34.108
	30 31	> <	PAGING RESPONSE AUTHENTICATION REQUEST	
	32	>	AUTHENTICATION RESPONSE	
	33	<	SECURITY MODE COMMAND	
	34	>	SECURITY MODE COMPLETE	Contains DD DATA DDDLL (OMC DELIVED TDDLL) with
	35 36	< SS	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU) with TP-DCS set to 0 Waits max 25 seconds for CP-ACK
	37	>	CP-ACK	
	38 39	SS >	CP-DATA	Waits max 60 seconds for RP-ACK RPDU Shall contain RP-ERROR RPDU with error cause "memory capability exceeded".
	40	<	CP-ACK	Within TC1M after step 39
	41 42	< >	RRC CONNECTION RELEASE RRC CONNECTION RELEASE	RRC connection is released.
	43	SS	COMPLETE	Prompts the operator to remove one of the short
	44 45	<	SYSTEM INFORMATION RRC CONNECTION REQUEST	messages from the message store of the UE. BCCH CCCH
	45 46	> <	RRC CONNECTION REQUEST	CCCH
	47	>	RRC CONNECTION SETUP COMPLETE	DCCH
	48	>	CM SERVICE REQUEST	CM service type information element is set to "Short message transfer".
	49	<	CM SERVICE ACCEPT	moodgo transier .

3GPP TS 34.123-1 V3.2.0(2001-1)

1319

Step	Direction	Message	Comments
	UE SS		
50	>	CP-DATA	Contains RP-SMMA RPDU
51	<	CP-ACK	
52	<	CP-DATA	Contains RP-ACK RPDU
53	>	CP-ACK	Acknowledge of CP-DATA containing the RP-ACK
			RPDU. The ME shall unset the "memory capability
			exceeded" notification flag on the USIM.
54	<	RRC CONNECTION RELEASE	RRC connection is released. The USIM simulator shall
			indicate if the "memory capability exceeded" notification
			flag has been unset on the USIM.
55	>	RRC CONNECTION RELEASE	
		COMPLETE	
56	SS		Prompts the operator to remove one of the short
			messages from the message store of the UE.
57	UE		Shall not attempt to send a RP-SMMA RPDU. This is
			verified by checking that the UE does not send a
			CHANNEL REQUEST message with the establishment
			cause "Other services which can be completed with an
			SDCCH"

NOTE: Time values for SS wait time are chosen sufficiently high to be sure that the UE has enough time to respond to the different messages.

Specific Message Contents

SMS-DELIVER TPDU in step 7

Information element	Comment Value
TP-DCS	default alphabet, class 2 "11110010"B

SMS-DELIVER TPDU in step 21

DCS	default alphabet, class 1 "11110001"B
-----	---------------------------------------

SMS-DELIVER TPDU in step 35

TP-DCS	default alphabet "00000000"B

16.1.3.5 Test requirements

After UE sends a negative acknowledgement (RP-ERROR) in step 11, the USIM simulator shall indicate if the "memory capability exceeded" notification flag has been set on the USIM.

After UE sends a negative acknowledgement (RP-ERROR) in step 23, the USIM simulator shall indicate if the "memory capability exceeded" notification flag has been set on the USIM.

After step 53 the ME shall unset the "memory capability exceeded" notification flag on the USIM.

After step 56 UE shall not attempt to send a RP-SMMA RPDU.

16.1.4 Test of the status report capabilities and of SMS-COMMAND:

This test applies to UEs which support the status report capabilities.

16.1.4.1 Definition

16.1.4.2 Conformance requirement

The SMS offers the SC the capabilities of informing the UE of the status of a previously sent mobile originated short message. This is achieved by the SC returning a status report TPDU (SMS-STATUS-REPORT) to the originating UE.

SMS-COMMAND enables an UE to invoke an operation at the SC.

The UE shall increment TP-MR by 1 for each SMS-SUBMIT or SMS-COMMAND being submitted.

References

- <u>3GPP</u>TS 23.040, sub-clause 3.2.9.
- <u>3GPP</u>TS 23.040, sub-clause 9.2.3.6.

16.1.4.3 Test purpose

- 1) To verify that the UE is able to accept a SMS-STATUS-REPORT TPDU.
- 2) To verify that the UE is able to use the SMS-COMMAND functionality correctly and sends an SMS-COMMAND TPDU with the correct TP-Message-Reference.

16.1.4.4 Method of test

Initial conditions

- System Simulator:
 - 1 cell, default parameters.
- User Equipment:
 - the UE shall be in the idle updated state MM-state "Idle, updated".

Related ICS/IXIT Statements

Support of SMS MO/PP and MT/PP.

Test procedure

- a) The UE is made to send a Mobile Originated short message setting TP-SRR as in steps a) to d) of test 16.1.2 (SMS Mobile originated).
- b) The SS sends a CP-DATA message containing a RP-DATA RPDU itself containing an SMS-STATUS-REPORT TPDU.
- c) The SS sends a RRC CONNECTION RELEASE message.
- d) The UE is made to send an SMS-COMMAND message enquiring about the previously submitted short message.

e)

f) The SS acknowledges the CP-DATA message from the UE with a CP-ACK followed by a CP-DATA message containing an RP-ACK RPDU

- g) After receiving the CP-ACK from the UE, the SS releases the RRC connection by using a RRC CONNECTION RELEASE message.
- h) The UE is made to send an SMS-COMMAND message requiring to delete the previously submitted short message.
- i) steps e) to g) are repeated.

Step	Direction	Message	Comments
	UE SS		
1	<	SYSTEM INFORMATION	BCCH
2	>	RRC CONNECTION REQUEST	CCCH
3	<	RRC CONNECTION SETUP	CCCH
4	>	RRC CONNECTION SETUP	DCCH
		COMPLETE	
5	>	CM SERVICE REQUEST	
6	<	AUTHENTICATION REQUEST	
7	>	AUTHENTICATION RESPONSE	
8	<	SECURITY MODE COMMAND	
9	>	SECURITY MODE COMPLETE	
10	>	CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU)
11	<	CP-ACK	Sent within TC1M after step 10
12	<	CP-DATA	Contains RP-ACK RPDU
13	SS	CD ACK	Waits max 25 seconds for CP-ACK
14	>	CP-ACK	DDC compostion is released
15 16	<	RRC CONNECTION RELEASE	RRC connection is released.
10	>	RRC CONNECTION RELEASE COMPLETE	
17		Mobile terminated establishment	See 3GPP TS34.108
''		of Radio Resource Connection	366 <u>3611</u> 1334.100
18	>	PAGING RESPONSE	
19	<	AUTHENTICATION REQUEST	
20	>	AUTHENTICATION RESPONSE	
21	<	SECURITY MODE COMMAND	
22	>	SECURITY MODE COMPLETE	
23	<	CP-DATA	Contains RP-DATA RPDU (SMS-STATUS-REPORT
		or shint	TPDU)
24	>	CP-ACK	, and the second
25	>	CP-DATA	Contains RP-ACK RPDU
26	<	CP-ACK	
27	<	RRC CONNECTION RELEASE	RRC connection is released.
28	>	RRC CONNECTION RELEASE	
		COMPLETE	
29	UE		The UE is made to send an SMS-COMMAND message
			enquiring about the previously submitted SM
30	<	SYSTEM INFORMATION	BCCH
31	>	RRC CONNECTION REQUEST	CCCH
32	<	RRC CONNECTION SETUP	CCCH
33	>	RRC CONNECTION SETUP	DCCH
		COMPLETE	
34		CM SERVICE REQUEST	
35	> <	AUTHENTICATION REQUEST	
36	>	AUTHENTICATION RESPONSE	
37	<>	SECURITY MODE COMMAND	
38	>	SECURITY MODE COMMAND	
39	>	CP-DATA	Contains RP-DATA RPDU (SMS-COMMAND TPDU)
		J. 27.11.	which shall contain the correct TP-MR
40	<	CP-ACK	The second secon
41	<	CP-DATA	Contains RP-ACK RPDU
42	>	CP-ACK	

3GPP TS 34.123-1 V3.2.0(2001-1)

1322

Step	Direction	Message	Comments
	UE SS	_	
43	<	RRC CONNECTION RELEASE	RRC connection is released.
44	>	RRC CONNECTION RELEASE	
45	UE	COMPLETE The UE is made to send an SMS- COMMAND	message requiring to delete the previously submitted SM.
46	>	RRC CONNECTION REQUEST	CCCH
47	<	RRC CONNECTION SETUP	CCCH
48	>	RRC CONNECTION SETUP COMPLETE	DCCH
49	>	CM SERVICE REQUEST	
50	<	AUTHENTICATION REQUEST	
51	>	AUTHENTICATION RESPONSE	
52	<	SECURITY MODE COMMAND	
53	>	SECURITY MODE COMPLETE	
54	>	CP-DATA	Contains RP-DATA RPDU (SMS-COMMAND TPDU) which shall contain the correct TP-MR
55	<	CP-ACK	
56	<	CP-DATA	Contains RP-ACK RPDU
57	>	CP-ACK	
58	<	RRC CONNECTION RELEASE	RRC connection is released.
59	>	RRC CONNECTION RELEASE COMPLETE	

Specific Message Contents

SMS SUBMIT TPDU

Information element	Comment Value
TP-SRR	status report is requested "1"B

SMS-STATUS-REPORT TPDU (SS to UE in step 23):

Information element	Comment Value
TP-MR	same as previous SMS-SUBMIT
TP-MMS	no more messages "1"B
TP-SRQ	result of SMS-SUBMIT "0"B
TP-RA	same as the Destination address of the SMS-SUBMIT
TP-ST	SM received "00000000"B

first SMS-COMMAND TPDU (UE to SS in step 39)

Information element	Comment Value
TP-MR	TP-MR in previous SMS-SUBMIT plus "1"
TP-SRR	status report requested "1"B
TP-CT	Enquiry relating to previously submitted
	short message "00000000"B
TP-MN	not checked (TP-MR in previous SMS-SUBMIT)

second SMS-COMMAND TPDU (UE to SS in step 54)

Information element	Comment Value
TP-MR	TP-MR in previous SMS-COMMAND plus "1"
TP-CT	Delete previously submitted short message "00000010"B
TP-MN	not checked (TP-MR in previous SMS-SUBMIT)

16.1.4.5 Test requirements

After step 23 UE accept a SMS-STATUS-REPORT TPDU.

After step 38 UE shall send a SMS-COMMAND TPDU with the correct TP-Message-Reference.

After step 53 UE shall send a SMS-COMMAND TPDU with the correct TP-Message-Reference.

16.1.5 Test of message class 0 to 3

The tests under this sub-clause only apply to a UE capable of displaying short messages (see ICS/IXIT).

16.1.5.1 Short message class 0

16.1.5.1.1 Definition

16.1.5.1.2 Conformance requirement

When a mobile terminated message is class 0 and the UE has the capability of displaying short messages, the UE shall display the message immediately and send an acknowledgement to the SC when the message has successfully reached the UE irrespective of whether there is memory available in the USIM or ME. The message shall not be automatically stored in the USIM or ME.

References

3GPP TS 23.038, clause 4.

16.1.5.1.3 Test purpose

To verify that the UE will accept and display but not store a class 0 message, and that it will accept and display a class 0 message if its message store is full.

NOTE: failure of this test in a UE could cause it to reject a class 0 message when its SMS memory becomes full. This could lead to unwanted repetitions between the UE and the service centre.

16.1.5.1.4 Method of test

Initial conditions

- System Simulator:
 - 1 cell, default parameters.
- User Equipment:
 - the UE shall be in the idle updated state MM-state "Idle, updated";
 - the UE message store shall be empty.

Related ICS/IXIT Statements

Support for Short message MT/PP.

Description of the basic procedures to display a mobile terminated short message.

The value of timer TC1M.

Whether SMS messages are stored in the USIM and/or the ME.

Test procedure

- a) The SS sends a class 0 message by using the method described in step a) of sub-clause 16.1.1 but with the TPDU described in this sub-clause.
- b) The UE message store shall be filled (for example by using the method of sub-clause 16.1.3 test of the memory available notification) with the same SMS-DELIVER TPDU except that TP-DCS is set to class 1.
- c) The SS sends a class 0 message as in step a).

01	Discrition	M	0
Step	Direction	Message	Comments
	UE SS		
1		Mobile terminated establishment	See <u>3GPP</u> TS34.108
		of Radio Resource Connection	
2	>	PAGING RESPONSE	
3	<	AUTHENTICATION REQUEST	
4	>	AUTHENTICATION RESPONSE	
5	<	SECURITY MODE COMMAND	
6	>	SECURITY MODE COMPLETE	
7	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU), Class
			0 Short Message
8	>	CP-ACK	
9	>	CP-DATA	Contains RP-ACK RPDU.
10	<	CP-ACK	
11	<	RRC CONNECTION RELEASE	
12	>	RRC CONNECTION RELEASE	
		COMPLETE	
13	UE		The content of the short message shall be displayed by
			the ME. The UE shall not store the message. This can be
			checked by verifying that it is impossible to retrieve any
			short messages from the UE message store.
14	SS		The UE message store shall be filled (for example by
			using the method of 16.1.3) with Class 1 SMS-DELIVER
			TPDU.
15		Mobile terminated establishment	See <u>3GPP</u> TS34.108
		of Radio Resource Connection	
16	>	PAGING RESPONSE	
17	<	AUTHENTICATION REQUEST	
18	>	AUTHENTICATION RESPONSE	
19	<	SECURITY MODE COMMAND	
20	>	SECURITY MODE COMPLETE	
21	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU), Class
			0 Short Message
22	>	CP-ACK	D AN DRAGK BRRU
23	>	CP-DATA	Contains RP-ACK RPDU.
24	<	CP-ACK	
25	<	RRC CONNECTION RELEASE	
26	>	RRC CONNECTION RELEASE	
07		COMPLETE	The content of the chart magnetic to the West Content of
27	UE		The content of the short message shall be displayed by
			the ME.

Specific Message Contents

SMS-DELIVER TPDU (containing a class 0 message) (SS to UE)

Information element	Comment Value	
TP-DCS	default alphabet, class 0	"1111 0000"B

SMS-DELIVER TPDU (containing a class 1 message to fill the UE message store) (SS to UE)

Information element	Comment Val	ue
TP-DCS	default alphabet, class 1	"1111 0001"B

16.1.5.1.5 Test requirements

After step 7 UE shall accept and display but not store a class 0 message.

After step 21 UE shall accept and display a class 0 message.

16.1.5.2 Test of class 1 short messages

This test shall apply to UEs which support:

- storing of received Class 1 Short Messages; and
- displaying of stored Short Messages.

16.1.5.2.1 Definition

16.1.5.2.2 Conformance requirement

When a mobile terminated message is class 1, the UE shall send an acknowledgement to the SC when the message has successfully reached the UE and can be stored, either in the ME or in the USIM.

References

3GPP TS 23.038, clause 4.

16.1.5.2.3 Test purpose

This procedure verifies that the UE acts correctly on receiving a class 1 message, i.e. that it stores the message in the ME or USIM and sends an acknowledgement (at RP and CP-Layer).

16.1.5.2.4 Method of test

Initial conditions

- System Simulator:
 - 1 cell, default parameters.
- User Equipment:
 - the UE shall be in the idle updated state MM-state "Idle, updated";
 - the UE message store shall be empty;

- for storing of class 1 Short Messages, the UE shall be set up to store Short Messages in the ME memory (by way of MMI, as described in ICS/IXIT statement).

Related ICS/IXIT Statements

Support for Short message MT/PP.

Description of the basic procedures to display a mobile terminated short message.

The value of timer TC1M.

Whether SMS messages are stored in the USIM and/or the ME.

Test procedure

- a) The SS delivers a Short Message of class 1 to the UE as specified in sub-clause 16.1.1, step a).
- b) The Short Message is recalled (e.g. by means of the MMI).

Expected sequence

Step	Direction	Message	Comments
	UE SS		
1		Mobile terminated establishment	See <u>3GPP</u> TS34.108
		of Radio Resource Connection	
2	>	PAGING RESPONSE	
3	<	AUTHENTICATION REQUEST	
4	>	AUTHENTICATION RESPONSE	
5	<	SECURITY MODE COMMAND	
6	>	SECURITY MODE COMPLETE	
7	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU), Class
			1 Short Message
8	>	CP-ACK	
9	>	CP-DATA	Contains RP-ACK RPDU.
10	<	CP-ACK	
11	<	RRC CONNECTION RELEASE	
12	>	RRC CONNECTION RELEASE	
		COMPLETE	
13	UE		The short message shall be recalled and displayed at the
			UE.

Specific Message Contents

SMS-DELIVER TPDU (containing a class 1 message) (SS to UE)

Information element	Comment Valu	ue
TP-DCS	default alphabet, class 1	"1111 0001"B

16.1.5.2.5 Test requirements

After step 7 UE shall store the message in the ME or USIM and send an acknowledgement.

16.1.5.3 Test of class 2 short messages

16.1.5.3.1 Definition

Class 2 Short Messages are defined as USIM specific, and the UE shall ensure that a message of this class is stored on the USIM.

16.1.5.3.2 Conformance requirement

When a mobile terminated message is Class 2, the UE shall ensure that the message has been correctly transferred to the SMS data field in the USIM before sending an acknowledgement to the SC. The UE shall return a "protocol error, unspecified" error message if the short message cannot be stored in the USIM and there is other short message storage available at the UE. If all the short message storage at the UE is already in use, the UE shall return "memory capacity exceeded".

Reference(s)

3GPP TS 23.040, sub-clause 9.2.3.10; 3GPP TS 23.038, clause 4. 3GPP TS 34.108, sub-clause 6.11.3.2.27

16.1.5.3.3 Test purpose

This procedure verifies that the UE acts correctly on receiving a class 2 message, i.e. that it stores the message correctly in the USIM, and if this is not possible, returns a protocol error message, with the correct error cause, to the network.

There are 2 cases:

- 1) if the UE supports storing of short messages in the USIM and in the ME, and storage in the ME is not full, and the short message cannot be stored in the USIM, the error cause shall be "protocol error, unspecified";
- 2) if the UE supports storing of short messages in the USIM and not in the ME, and storage in the ME is not full, and the short message cannot be stored in the USIM, the error cause shall be "memory capacity exceeded".

NOTE: If the UE supports storing of short messages in the USIM and the ME, and storage in the ME is full, and the short message cannot be stored in the USIM, the error cause shall be "memory capacity exceeded". This case is not tested in this test.

16.1.5.3.4 Method of test

Initial conditions

- System Simulator:
 - 1 cell, default parameters.
- User Equipment:
 - the UE shall be in the idle updated state MM-state "Idle, updated";
 - the ME message store shall be empty;
 - the ME shall be connected to the USIM simulator. The following shall be present in the USIM simulator:
 - EF_{SMS} with at least two free records and one full record;
 - EF_{SMSstatus}, with SMS "Memory Cap. Exceed" notification flag set to "memory available";
 - Service no. 4 (SMS) in EF_{SST} set to allocated and activated;
 - for storing of Class 1 Short Messages the UE shall be set up to store Short Messages in the ME memory (by way of MMI, as described in ICS/IXIT statement).

Related ICS/IXIT Statements

Support for Short message MT/PP.

The value of timer TC1M.

Whether SMS messages are stored in the USIM and/or the ME.

Test procedure

- a) The SS delivers a Short Message of class 2 to the UE as specified in sub-clause 16.1.1, step b).
- b) Following an attempt by the ME to store the short message in a free record of EF_{SMS} in the USIM, the USIM simulator returns the status response "OK" ("90 00").
- c) Step a) is repeated.
- d) Following an attempt by the ME to store the short message in a free record of EF_{SMS} in the USIM, the USIM simulator returns the status response "memory problem" ("92 40").
- e) The USIM simulator indicates if an attempt was made in steps a) and c) to store the messages and if the messages are stored according to the requirement.

Step	Direction	Message	Comments
-	UE SS	1	
1	<	Mobile terminated establishment	See <u>3GPP</u> TS34.108
		of Radio Resource Connection	
2	>	PAGING RESPONSE	
3	<	AUTHENTICATION REQUEST	
4	>	AUTHENTICATION RESPONSE	
5 6	<	SECURITY MODE COMMAND SECURITY MODE COMPLETE	
7	<>	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU), Class
,	\	CF-DATA	2 Short Message
8	>	CP-ACK	2 Official wessage
9	MÉ	or non	The ME shall correctly store the short message in a free
			record of EFSMS in the USIM, i.ethe ME shall use a
			free record - the first byte of the record shall indicate
			"message received by UE from network"
			- the TS-Service-Centre-Address shall be correctly
			stored
			- the TPDU shall be identical to that sent by the SS
			- bytes following the TPDU shall be set to "FF"
10	USIM		The USIM simulator returns the status response "OK"
			("90 00"). The USIM simulator shall indicate if an attempt
			was made by the ME to store the short message in the
			USIM.
11	>	CP-DATA	Contains RP-ACK RPDU.
12	<	CP-ACK	
13	<	RRC CONNECTION RELEASE	
14	>	RRC CONNECTION RELEASE	
45		COMPLETE	Con 2CDD TC24 400
15		Mobile terminated establishment of Radio Resource Connection	See <u>3GPP</u> TS34.108
16	>	PAGING RESPONSE	
17	<	AUTHENTICATION REQUEST	
18	>	AUTHENTICATION RESPONSE	
19	<	SECURITY MODE COMMAND	
20	>	SECURITY MODE COMPLETE	

Step	Direction	Message	Comments
	UE SS		
21	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU), Class 2 Short Message
22	>	CP-ACK	, and the second
23	ME		The ME shall attempt to store the short message in a free record of EFSMS in the USIM.
24	USIM		The USIM simulator returns the status response "memory problem" ("92 40"). The USIM simulator shall indicate if an attempt was made by the ME to store the short message in the USIM.
25	>	CP-DATA	Contains RP-ERROR RPDU with error cause "protocol error, unspecified" if the UE supports storing of short messages in the ME, or error cause "memory capacity exceeded" if not.
26	<	CP-ACK	
27	<	RRC CONNECTION RELEASE	
28	>	RRC CONNECTION RELEASE COMPLETE	

Specific Message Contents

SMS-DELIVER TPDU (containing a class 2 message) (SS to UE)

Information element	Comment Value	
TP-DCS	default alphabet, class 2	"1111 0010"B

16.1.5.3.5 Test requirements

After step 10 UE shall confirm that the short message is stored in the USIM and send CP-DATA containing RP-ACK RPDU.

After step 24 UE shall confirm that the short message cannot be stored in the USIM and send CP-DATA containing RP-ERROR RPDU. If UE supports storing of short message in the ME, the error cause of RP-ERROR RPDU shall be "protocol error, unspecified", and if not the error cause of RP-ERROR RPDU shall be "memory capacity exceeded"

16.1.5.4 Test of class 3 short messages

For further study.

16.1.6 Test of short message type 0

For further study.

16.1.7 Test of the replace mechanism for SM type 1-7

16.1.7.1 Definition

16.1.7.2 Conformance requirement

On receipt of a short message, the UE shall check to see if the associated Protocol Identifier contains a Replace Short Message Type code. If such a code is present, then the UE will check the associated SC address (RP-OA) and originating address (TP-OA) and replace any existing stored message having the same Protocol Identifier code, SC address and originating address with the new short message.

Reference(s)

3GPP TS 23.040; sub-clause 9.2.3.9.

16.1.7.3 Test purpose

This procedure verifies the correct implementation of the replace mechanism for Replace Short Messages.

16.1.7.4 Method of test

Initial conditions

- System Simulator:
 - 1 cell, default parameters.
- User Equipment:
 - the UE shall be in the idle updated state MM-state "Idle, updated";
 - the UE message store shall be empty.

Related ICS/IXIT Statements

Support for Short message MT/PP.

Description of the basic procedures to display a mobile terminated short message.

The value of timer TC1M.

Test procedure

- a) Two different numbers n and m are drawn randomly between 1 and 7. Two different addresses for TP-Originating-Address (TPOA1 and TPOA2) are drawn. Two different addresses for RP-Originating-Address (RPOA1 and RPOA2) are drawn.
- b) The SS delivers a short message to the UE as specified in sub-clause 16.1.1 step a). In the SMS-DELIVER TPDU, the TP-Protocol-Identifier parameter is "Replace Short Message Type n", the TP-Originating-Address is TPOA1, and the RP-Originating-Address is RPOA1.
- c) Step b) is repeated but with a different TP-Originating-Address (TPOA2), and different contents of TP-User-Data in the SMS-DELIVER TPDU. The other parameters are the same as in step b).
- d) Step c) is repeated but with RPOA2 in the RP-Originated-Address, and contents of TP-User-Data different from the former two messages. The other parameters are the same as in step c).
- e) Step d) is repeated but with the TP-Protocol-Identifier equal to "Replace Short Message Type m", and contents of TP-User-Data different from the former three messages. The other parameters are the same as in step d).
- f) Step e) is repeated but the contents of TP-User-Data are different from that used in step e).
- g) The SS prompts the operator to display the Short Messages stored in the UE.

Ī	Step	Direction		Message	Comments
		UE	SS		
Ī	1			Mobile terminated establishment	See <u>3GPP</u> TS34.108
				of Radio Resource Connection	
	2		>	PAGING RESPONSE	
	3	<		AUTHENTICATION REQUEST	

Step	Direction	Message	Comments
4	UE SS	AUTHENTICATION RESPONSE	
5	<	SECURITY MODE COMMAND	
6	>	SECURITY MODE COMPLETE	
7	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU) TP-PID is "Replace Short Message Type n", TP-OA is TPOA1 and RP-OA is RPOA1
8	>	CP-ACK	Contains DR ACK DRDU
9 10	> <	CP-DATA CP-ACK	Contains RP-ACK RPDU.
11	<	RRC CONNECTION RELEASE	
12	>	RRC CONNECTION RELEASE COMPLETE	
13		Mobile terminated establishment of Radio Resource Connection	See <u>3GPP</u> TS34.108
14	>	PAGING RESPONSE	
15 16	< >	AUTHENTICATION REQUEST AUTHENTICATION RESPONSE	
17	<	SECURITY MODE COMMAND	
18	>	SECURITY MODE COMPLETE	
19	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU) TP-PID is "Replace Short Message Type n", TP-OA is TPOA2 and RP-OA is RPOA1, TP-UD different from step 7
20	>	CP-ACK	
21 22	> <	CP-DATA CP-ACK	Contains RP-ACK RPDU.
23	<	RRC CONNECTION RELEASE	
24	>	RRC CONNECTION RELEASE COMPLETE	
25		Mobile terminated establishment of Radio Resource Connection	See <u>3GPP</u> TS34.108
26	>	PAGING RESPONSE	
27	<	AUTHENTICATION REQUEST	
28	>	AUTHENTICATION RESPONSE SECURITY MODE COMMAND	
29 30	< >	SECURITY MODE COMMAND	
31	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU) TP-PID is "Replace Short Message Type n", TP-OA is TPOA2 and RP-OA is RPOA2, TP-UD different from step 7 and 19
32	>	CP-ACK	
33 34	> <	CP-DATA CP-ACK	Contains RP-ACK RPDU.
35 36	< >	RRC CONNECTION RELEASE RRC CONNECTION RELEASE	
37		COMPLETE Mobile terminated establishment	See <u>3GPP</u> TS34.108
38 39 40	> < >	of Radio Resource Connection PAGING RESPONSE AUTHENTICATION REQUEST AUTHENTICATION RESPONSE	
41	<	SECURITY MODE COMMAND	
42	>	SECURITY MODE COMPLETE	Contains DD DATA DDDU (CMC DELIVED TDDU) TD
43	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU) TP-PID is "Replace Short Message Type m", TP-OA is TPOA2 and RP-OA is RPOA2, TP-UD different from step 7, 19 and 31
44	>	CP-ACK	
45 46	>	CP-DATA	Contains RP-ACK RPDU.
46 47	< <	CP-ACK RRC CONNECTION RELEASE	
48	>	RRC CONNECTION RELEASE COMPLETE	

Step	Direction	Message	Comments
_	UE SS		
49		Mobile terminated establishment	See <u>3GPP</u> TS34.108
		of Radio Resource Connection	
50	>	PAGING RESPONSE	
51	<	AUTHENTICATION REQUEST	
52	>	AUTHENTICATION RESPONSE	
53	<	SECURITY MODE COMMAND	
54	>	SECURITY MODE COMPLETE	
55	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU) TP-
			PID is "Replace Short Message Type m", TP-OA is
			TPOA2 and RP-OA is RPOA2, TP-UD different from step
			43
56	>	CP-ACK	
57	>	CP-DATA	Contains RP-ACK RPDU.
58	<	CP-ACK	
59	<	RRC CONNECTION RELEASE	
60	>	RRC CONNECTION RELEASE	
		COMPLETE	
61	SS		Prompts the operator to display the Short Messages
			stored in the UE. Only the Short Messages delivered in
			step 7, 19, 31 and 55 shall be retrievable and displayed

Specific Message Contents

SMS-DELIVER TPDU

Information element	Comment Value
TP-MMS	no more messages are waiting in SC "1"B
TP-PID	binary 01000xxx, xxx represents n resp. m (see test
	method description)

16.1.7.5 Test requirements

After step 60 only the Short Messages delivered in step 7, 19, 31 and 55 shall be retrieved and displayed.

16.1.8 Test of the reply path scheme

16.1.8.1 Definition

16.1.8.2 Conformance requirement

When a replying UE receives an original mobile terminated short message it has:

- originating SME = TP-Originating Address in the SMS-DELIVER TPDU;
- original SC = RP-Originating Address in the RP-MT-DATA.

When submitting the reply mobile originated short message, the replying UE should use parameters as follows:

- TP-Destination Address in SMS-SUBMIT TPDU = originating SME;
- RP-Destination Address in RP-MO-DATA = original SC.

Reference(s)

3GPP TS 23.040 Annex D.5, D.6

16.1.8.3 Test purpose

This procedure verifies that the UE is able to send a Reply Short Message back to the correct originating SME even if in the meantime it receives another Short Message.

16.1.8.4 Method of test

Initial conditions

- System Simulator:
 - 1 cell, default parameters.
- User Equipment:
 - the UE shall be in the idle updated state MM-state "Idle, updated";
 - the UE message store shall be empty.

Related ICS/IXIT Statements

Support for Short message MT/PP.

Support for Short message MO/PP.

Description of the basic procedures to display a mobile terminated short message.

Description of the basic procedures to send a mobile originated short message.

The value of timer TC1M.

Test procedure

- a) The SS delivers a Short Message as specified in sub-clause 16.1.1, step b) with TP-Reply-Path set to 1.
- b) Step a) is repeated but with:
 - different TP-Originating-Address for the originating SME;
 - different RP-Originating-Address for the original SC; and
 - different message contents TP-User-Data.
- c) One of the two Short Messages is displayed (e.g. by means of the MMI) and the Reply Short Message is submitted (e.g. by means of the MMI).
- d) step c) is repeated for the other Short Message.

Step	Direc	ction	Message	Comments
	UE	SS		
1			Mobile terminated establishment	See <u>3GPP</u> TS34.108
			of Radio Resource Connection	
2		>	PAGING RESPONSE	
3	<		AUTHENTICATION REQUEST	
4		>	AUTHENTICATION RESPONSE	
5	<		SECURITY MODE COMMAND	
6		>	SECURITY MODE COMPLETE	
7	<		CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU) TP-
				RP set to 1
8		>	CP-ACK	Sent within TC1M after step 7

Step	Direction UE SS	Message	Comments
9	>	CP-DATA	Contains RP-ACK RPDU.
10	<	CP-ACK	
11	<	RRC CONNECTION RELEASE	
12	>	RRC CONNECTION RELEASE	
		COMPLETE	
13		Mobile terminated establishment of Radio Resource Connection	See <u>3GPP</u> TS34.108
14	>	PAGING RESPONSE	
15	<	AUTHENTICATION REQUEST	
16	>	AUTHENTICATION RESPONSE	
17	<	SECURITY MODE COMMAND	
18	>	SECURITY MODE COMPLETE	
19	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU) TP-OA, RP-OA and TP-UD different from step 7
20		CD ACK	
20	>	CP-ACK	Sent within TC1M after step 7
21	>	CP-DATA	Contains RP-ACK RPDU.
22	<	CP-ACK	
23	<	RRC CONNECTION RELEASE	
24	>	RRC CONNECTION RELEASE COMPLETE	
25	UE		One of the two Short Messages is displayed and the
			Reply Short Message is submitted.
26	<	SYSTEM INFORMATION	BCCH
27	>	RRC CONNECTION REQUEST	CCCH
28	<	RRC CONNECTION SETUP	СССН
29	>	RRC CONNECTION SETUP	DCCH
		COMPLETE	
30	>	CM SERVICE REQUEST	
31	<	AUTHENTICATION REQUEST	
32	>	AUTHENTICATION RESPONSE	
33	<	SECURITY MODE COMMAND	
34	>	SECURITY MODE COMPLETE	
35	>	CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU) RP-DA
		0. 27171	= RP-OA corresponding to the message displayed TP-
			DA = TP-OA corresponding to the message displayed
36	<	CP-ACK	Sent within TC1M after step 35
37	<	CP-DATA	Contains RP-ACK RPDU
38	SS	OI -DATA	Waits max 25 seconds for CP-ACK
39	>	CP-ACK	Walts max 25 seconds for Cr -ACR
1 1	· ·		RRC connection is released.
40	<	RRC CONNECTION RELEASE	RRC connection is released.
	>	RRC CONNECTION RELEASE COMPLETE	
44	UE	COMPLETE	The other Chart Massage is displayed and the Danky
41	l OE		The other Short Message is displayed and the Reply Short Message is submitted.
42	<	SYSTEM INFORMATION	BCCH
43	>	RRC CONNECTION REQUEST	CCCH
44	<	RRC CONNECTION SETUP	CCCH
45	>	RRC CONNECTION SETUP	DCCH
		COMPLETE	
46	>	CM SERVICE REQUEST	
47	<	AUTHENTICATION REQUEST	
48	>	AUTHENTICATION RESPONSE	
49	<	SECURITY MODE COMMAND	
50	>	SECURITY MODE COMPLETE	
50	>	CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU) RP-DA
31	>	OI POATA	= RP-OA corresponding to the Message displayed TP-
			DA = TP-OA corresponding to the message displayed
50		CB-VCK	
52	<	CP-ACK	Sent within TC1M after step 51
53	<	CP-DATA	Contains RP-ACK RPDU
54	SS	CD ACK	Waits max 25 seconds for CP-ACK
55 56	>	CP-ACK	DDC connection is released
56	<	RRC CONNECTION RELEASE	RRC connection is released.

Step	Direc	ction	Message	Comments
	UE	SS		
57		·>	RRC CONNECTION RELEASE	
			COMPLETE	

Specific Message Contents

SMS-DELIVER TPDU

Information element	Comment Value
TP-MMS	no more messages are waiting in SC "1"B
TP-RP	Reply Path exists "1"B

16.1.8.5 Test requirements

After step 34 UE shall send the Reply Short Message corresponding to one of two previously received short messages.

After step 50 UE shall send the Reply Short Message corresponding to the other of two previously received short messages.

16.1.9 Multiple SMS mobile originated

16.1.9.1 UE in idle mode

This test applies to UE supporting the ability of sending multiple short messages on the same RRC connection when there is no call in progress.

16.1.9.1.1 Definition

16.1.9.1.2 Conformance requirements

When the UE chooses to use the same RRC connection to send another short message or a memory available notification, then:

- the UE shall transmit a CM SERVICE REQUEST for the new CM connection before the final CP-ACK (e.g. the one that acknowledges the CP-DATA that carried the RP-ACK) for the old MM connection is transmitted;
- before transmission of the first CP-DATA on the new MM connection, the UE shall transmit the CP-ACK for the old MM connection;
- the Transaction Identifier used on the new MM connection shall be different to that used on the old MM connection; and
- the UE shall not initiate establishment of the new MM connection before the final CP-DATA (e.g. the one carrying the RP-ACK) has been received.

Reference

- <u>3GPP</u>TS 23.040; sub-clause 3.1.
- <u>3GPP</u> TS 24.011; sub-clause 5.4.

16.1.9.1.3 Test purpose

To verify that the UE is able to correctly send multiple short messages on the same RRC connection when using a DCCH.

16.1.9.1.4 Method of test

Initial conditions

- System simulator:
 - 1 cell, default parameters.
- User Equipment:
 - the UE shall be in "Idle, updated" state MM-state "Idle, updated";
 - the SMS message storage shall be empty.

Related ICS/IXIT statements

Support for multiple short message MO/PP on the same RRC connection.

Description of how to enter multiple SMS.

Description of the basic procedures to display a mobile originated short message.

Whether SMS messages are stored in the USIM and/or the ME.

Foreseen final state of UE

Idle, updated.

Test procedure

- a) The UE shall be set up to send 3 short messages as multiple SM to the SS. The SS answers correctly to RRC CONNECTION REQUEST on CCCH and then performs the authentication.
- b) After receiving SECURITY MODE COMMAND UE shall send SECURITY COMMAND COMPLETE.
- c) The SS responds to the CP-DATA containing RP-DATA RPDU (SMS SUBMIT TPDU) from the UE with a CP-ACK message followed by a CP-DATA message containing the correct RP-ACK RPDU. The Transaction Identifier used on this MM connection is 'x'.
- d) The UE shall transmit a CM SERVICE REQUEST for the new CM connection (for the second short message) before the final CP-ACK (the one that acknowledges the CP-DATA that carried the RP-ACK before) for the old MM connection is transmitted. The UE shall not initiate establishment of the new MM connection before the final CP-DATA (i.e. the one carrying the RP-ACK for the first short message) has been received. Before transmission of the first CP-DATA on the new MM connection, the UE shall transmit the CP-ACK for the old MM connection. The Transaction Identifier used on the new MM connection shall be y, where y <> x (see procedure c)).
- e) The SS waits a maximum of 5 seconds after receiving the CM SERVICE REQUEST for the CP-ACK message from the UE.
- f) The SS responds to the CP-DATA containing RP-DATA RPDU (SMS SUBMIT TPDU) from the UE with a CP-ACK message followed by a CP-DATA message containing the correct RP-ACK RPDU.
- g) The UE shall transmit a CM SERVICE REQUEST for the new CM connection (for the third short message) before the final CP-ACK (the one that acknowledges the CP-DATA that carried the RP-ACK before) for the old MM connection is transmitted. Before transmission of the first CP-DATA on the new MM connection, the UE shall transmit the CP-ACK for the old MM connection. The Transaction Identifier used on the new MM connection shall be z, where z <> y (see procedure d)). The UE shall not initiate establishment of the new MM connection before the final CP-DATA (i.e. the one carrying the RP-ACK for the second short message) has been received
- h) The SS waits a maximum of 5 seconds after receiving the CM SERVICE REQUEST for the CP-ACK message from the UE.

- i) The SS responds to the CP-DATA containing RP-DATA RPDU (SMS SUBMIT TPDU) from the UE with a CP-ACK message followed by a CP-DATA message containing the correct RP-ACK RPDU.
- j) The SS waits a maximum of 5 seconds after sending CP-DATA for the CP-ACK message from the UE.
- k) The SS sends a RRC CONNECTION RELEASE to the UE.

Expected sequence

Step	Direction	Message	Comments
-	UE SS		
1	<	SYSTEM INFORMATION	BCCH
2	>	RRC CONNECTION REQUEST	CCCH
3	<	RRC CONNECTION SETUP	CCCH
4	>	RRC CONNECTION SETUP	DCCH
		COMPLETE	
5	>	CM SERVICE REQUEST	
6	<	AUTHENTICATION REQUEST	
7	>	AUTHENTICATION RESPONSE	
8	<	SECURITY MODE COMMAND	
9	>	SECURITY MODE COMPLETE	O
10	>	CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU). The
			Transaction Identifier used in steps 10, 11, 12 and 14
11	<	CP-ACK	shall be x.
12	<	CP-DATA	Contains RP-ACK RPDU
13	>	CM SERVICE REQUEST	CM service type set to "Short message transfer".
14	>	CP-ACK	Shall be sent within 5 seconds of step 13
15	<	CM SERVICE ACCEPT	Official be sent within 5 seconds of step 15
16	>	CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU). The
10		or Britis	Transaction Identifier used in steps 16, 17, 18 and 20
			shall be y where y <> x (see step 10).
17	<	CP-ACK	(**************************************
18	<	CP-DATA	Contains RP-ACK RPDU
19	>	CM SERVICE REQUEST	CM service type set to "Short message transfer".
20	>	CP-ACK	Shall be sent within 5 seconds of step 19
21	<	CM SERVICE ACCEPT	·
22	>	CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU). The
			Transaction Identifier used in steps 22, 23, 24 and 25
			shall be z, where z <> y (see step 16).
23	<	CP-ACK	
24	<	CP-DATA	Contains RP-ACK RPDU
25	>	CP-ACK	Shall be sent within 5 seconds of step 24
26	<	RRC CONNECTION RELEASE	RRC connection is released.
27	>	RRC CONNECTION RELEASE	
		COMPLETE	

16.1.9.1.5 Test requirements

In step 12 the UE shall transmit a CM SERVICE REQUEST for the new CM connection (for the second short message) before the final CP-ACK for the old MM connection is transmitted.

In step 13 the UE shall transmit a CM SERVICE REQUEST for the new CM connection (for the third short message) before the final CP-ACK for the old MM connection is transmitted.

16.1.9.2 UE in active mode

This test applies to UE supporting the ability of sending concatenated multiple short messages when there is a call in progress.

16.1.9.2.1 Definition

16.1.9.2.2 Conformance requirements

When the UE chooses to use the same RRC connection to send another short message or a memory available notification, then:

- the UE shall transmit a CM SERVICE REQUEST for the new CM connection before the final CP-ACK (e.g. the one that acknowledges the CP-DATA that carried the RP-ACK) for the old MM connection is transmitted;
- before transmission of the first CP-DATA on the new MM connection, the UE shall transmit the CP-ACK for the old MM connection;
- the Transaction Identifier used on the new MM connection shall be different to that used on the old MM connection; and
- the UE shall not initiate establishment of the new MM connection before the final CP-DATA (e.g. the one carrying the RP-ACK) has been received.

Reference

- 3GPP TS 23.040; sub-clause 3.1.
- <u>3GPP</u>TS 24.011; sub-clause 5.4.

16.1.9.2.3 Test purpose

To verify that the UE is able to correctly concatenate multiple short messages on the same RRC connection when sent parallel to a call.

16.1.9.2.4 Method of test

Initial conditions

- System simulator:
 - 1 cell, default parameters.
- User Equipment:
 - the UE shall be in "Idle, updated" state MM-state "Idle, updated";
 - the SMS message storage shall be empty.

Related ICS/IXIT statements

Support for multiple short message MO/PP on the same RRC connection.

Description of how to enter multiple SMS.

Description of the basic procedures to display a mobile originated short message.

Support for state U10 of call control.

Whether SMS messages are stored in the USIM and/or the ME.

Foreseen final state of UE

Idle, updated.

Test procedure

- a) a) A data or speech call is established on a DTCH with the SS and the state U10 of call control is entered. The UE is set up to send 3 short messages as multiple SM to the SS. After the reception of the CM SERVICE REQUEST, the SS sends a CM SERVICE ACCEPT message.
- b) Steps c) to k) of the test procedure in sub-clause 16.1.9.1.4 are repeated.

Expected sequence

UE	Step	Direction	Message	Comments
state U10 of call control is entered. The UE is set up to send 3 short messages as multiple SM Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Contains RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 7, 8, 9 and 11 shall be x. CP-ACK CP-DATA CP-ACK CP-DATA CP-ACK CP-DATA COP-BATA CP-DATA CP-DATA CP-DATA COP-DATA COP-DATA COP-DATA COP-DATA COP-DATA COP-DATA COP-DATA CP-ACK CP-DATA CP-ACK CP-DATA COP-DATA COP-DATA COP-DATA CP-ACK CP-DATA COM SERVICE REQUEST Shall be sent within 5 seconds of step 10 Contains RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 13, 14, 15 and 17 shall be y where y <> x (see step 7). CONTAINS RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 16 Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 16 Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 16 Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 16 Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 16 Contains RP-ACK RPDU Shall be sent within 5 seconds of step 21 RRC connection is released.		UE SS	_	
The UE is set up to send 3 short messages as multiple SM Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" CM SERVICE ACCEPT CP-DATA Contains RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 7, 8, 9 and 11 shall be x. CP-ACK CP-DATA Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" CONTAIN REPUBLIES SET UP to send 3 short message as multiple SM Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" CONTAIN RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 13, 14, 15 and 17 shall be ywhere y <> x (see step 7). CP-ACK CP-DATA CONTAIN RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 13, 14, 15 and 17 shall be ywhere y <> x (see step 7). CP-ACK CP-DATA CONTAIN RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 16 CONTAIN RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 16 CONTAIN RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13). CONTAIN RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 16 CONTAIN RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 16 CONTAIN RP-ACK RPDU Shall be sent within 5 seconds of step 21 RRC connection is released.	1	SS		A data or speech call is established on a DTCH and the
SM Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" CP-DATA CP-ACK CP-ACK CP-ACK CP-ACK CP-ACK CP-DATA CONSERVICE REQUEST CP-DATA Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 10 Contains RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 13, 14, 15 and 17 shall be y where y <> x (see step 7). CP-ACK CP-DATA Contains RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 13, 14, 15 and 17 shall be ywhere y <> x (see step 7). CP-ACK CP-DATA Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 16 Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 16 Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Contains RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13). CONTAINS RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 16 Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 16 Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 16 Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 16 Contains RP-ACK RPDU Shall be sent within 5 seconds of step 21 RPC connection is released.				
3> CM SERVICE REQUEST 4 CM SERVICE ACCEPT 7> CP-DATA 8 CP-ACK 9 CP-ACK 11> CP-ACK 12 CP-DATA CP-DATA CP-DATA CP-DATA CP-DATA CP-ACK 15 CP-ACK 16 CP-ACK 17 CP-ACK 18 CP-ACK 18 CP-ACK 19 CP-ACK 10 CP-ACK 20 CP-ACK 21 CP-ACK 22 CP-ACK 21 CP-ACK 22 CP-ACK 22 CP-ACK 23 CP-ACK 24 CP-ACK 25 CP-ACK 26 CP-ACK 27 CP-ACK 28 CP-ACK 29 CP-ACK 20 CP-ACK 21 CP-ACK 22 CP-ACK 23 CP-ACK 24 CP-ACK 25 CP-ACK 26 CP-ACK 27 CP-ACK 28 CP-ACK 29 CP-ACK 29 CP-ACK 20 CP-ACK 20 CP-ACK 21 CP-ACK 22 CP-ACK 23 RRC CONNECTION RELEASE Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Contains RP-ACK RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <-> y (see step 13). Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 16 Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 16 Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 16 Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 21 RRC connection is released.	2	UE		· · · · · · · · · · · · · · · · · · ·
set to "short message transfer" Contains RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 7, 8, 9 and 11 shall be x. CP-ACK CP-DATA CONTAINS RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" CP-ACK CP-DATA CP-ACK CP-DATA CP-DATA CP-DATA CP-DATA CP-DATA CP-ACK CP-DATA CONTAINS RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 13, 14, 15 and 17 shall be y where y <> x (see step 7). CONTAINS RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" CONTAINS RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 13, 14, 15 and 17 shall be y where y <> x (see step 7). CONTAINS RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" CONTAINS RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" CONTAINS RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13). CONTAINS RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" CONTAINS RP-ACK RPDU Shall be zent within 5 seconds of step 21 RRC CONNECTION RELEASE RRC CONNECTION RELEASE REC CONNECTION RELEASE				9111
CP-DATA CP-DATA CP-DATA Contains RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 7, 8, 9 and 11 shall be x. CP-DATA CP-DATA CONTAINS RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 7, 8, 9 and 11 shall be x. CONTAINS RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 10 CONTAINS RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 13, 14, 15 and 17 shall be y where y <> x (see step 7). CP-ACK CP-DATA CONTAINS RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 16 CONTAINS RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 16 CONTAINS RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13). CONTAINS RP-ACK RPDU Shall be zent within 5 seconds of step 21 RRC connection is released.	3	>	CM SERVICE REQUEST	
Transaction Identifier used in steps 7, 8, 9 and 11 shall be x. CP-ACK CP-DATA Com SERVICE REQUEST Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 10 CONSERVICE ACCEPT CP-DATA Contains RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 13, 14, 15 and 17 shall be y where y <> x (see step 7). CP-ACK CP-DATA Contains RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 13, 14, 15 and 17 shall be y where y <> x (see step 7). CONTAINS RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be sent within 5 seconds of step 16 CONTAINS RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13). CONTAINS RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13). CONTAINS RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13). CONTAINS RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13). CONTAINS RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13). CONTAINS RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13).	4	<	CM SERVICE ACCEPT	-
be x. CP-ACK CP-DATA CONSERVICE REQUEST CP-ACK CP-DATA COM SERVICE ACCEPT CP-DATA CP-DATA CP-ACK CP-DATA COM SERVICE ACCEPT CP-DATA COntains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 10 Contains RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 13, 14, 15 and 17 shall be y where y <> x (see step 7). CP-ACK CP-DATA Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 16 Contains RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13). Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 16 Contains RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13). Contains RP-ACK RPDU Shall be sent within 5 seconds of step 21 RRC connection is released.	7	>	CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU). The
8				Transaction Identifier used in steps 7, 8, 9 and 11 shall
9 < CP-DATA 10> CM SERVICE REQUEST 11> CP-ACK 12 < CM SERVICE ACCEPT 13> CP-DATA 15> CP-ACK 16> CP-ACK 17> CP-ACK 18 < CP-DATA 18> CP-ACK 19> CP-ACK 19> CP-DATA 19> CP-ACK 20				be x.
CM SERVICE REQUEST Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 10		<		
set to "short message transfer" Shall be sent within 5 seconds of step 10 Contains RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 13, 14, 15 and 17 shall be y where y <> x (see step 7). Contains RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 13, 14, 15 and 17 shall be y where y <> x (see step 7). Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 16 Contains RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13). Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Contains RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13). Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Contains RP-ACK RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13). Contains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Contains RP-ACK RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13).	_	<		
11	10	>	CM SERVICE REQUEST	
12				
CP-DATA CP-DATA COntains RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 13, 14, 15 and 17 shall be y where y <> x (see step 7). CP-ACK CP-DATA COntains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 16 COntains RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 16 Contains RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13). CONTAINS RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13). CONTAINS RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Contains RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13). CONTAINS RP-ACK RPDU Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Contains RP-ACK RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13). CONTAINS RP-ACK RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13). CONTAINS RP-ACK RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13).		-		Shall be sent within 5 seconds of step 10
Transaction Identifier used in steps 13, 14, 15 and 17 shall be y where y <> x (see step 7). CP-ACK CP-DATA CM SERVICE REQUEST CP-ACK CP-ACK CP-DATA CM SERVICE ACCEPT CP-DATA CM SERVICE ACCEPT CP-DATA CP-DATA CP-DATA CP-DATA CP-ACK CP-DATA CP-ACK CP-DATA CP-ACK CP-DATA CP-ACK CP-DATA CP-ACK CP-DATA CP-ACK CP-CP-CP-CP-CP-CP-CP-CP-CP-CP-CP-CP-CP-C		<		
shall be y where y <> x (see step 7). CP-ACK CP-DATA CM SERVICE REQUEST CP-ACK CP-ACK CP-DATA CM SERVICE REQUEST CP-ACK CP-ACK CP-DATA CM SERVICE ACCEPT CP-DATA CM SERVICE ACCEPT CP-DATA CP-DATA CP-DATA CP-ACK CP-CP-ACK CP-CP-ACK CP-CP-CP-ACK CP-CP-CP-CP-CP-CP-CP-CP-CP-CP-CP-CP-CP-C	13	>	CP-DATA	
14				
15			00 4014	shall be y where y <> x (see step 7).
16> CM SERVICE REQUEST 17> CP-ACK 18 CM SERVICE ACCEPT 19> CP-DATA 20 CP-DATA 20 CP-DATA 21 CP-DATA 22 RRC CONNECTION RELEASE Sent in a layer 2 frame on the DCCH. CM service type set to "short message transfer" Shall be sent within 5 seconds of step 16 Contains RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13). Contains RP-ACK RPDU Shall be sent within 5 seconds of step 21 RRC connection is released.				Ocataina DD AOK DDDII
set to "short message transfer" 17		-		
17> CP-ACK 18 CM SERVICE ACCEPT 19> CP-DATA CP-DATA CP-DATA CP-ACK CM SERVICE ACCEPT CP-DATA COntains RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13). CP-ACK 21	16	>	CM SERVICE REQUEST	
18	47		OD AOK	
CP-DATA CP-DATA Contains RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13). CP-ACK CP-DATA CONTAINS RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13). Contains RP-ACK RPDU Shall be sent within 5 seconds of step 21 RRC CONNECTION RELEASE RRC connection is released.				Snall be sent within 5 seconds of step 16
Transaction Identifier used in steps 19, 20, 21 and 22 shall be z, where z <> y (see step 13). CP-ACK CP-DATA CP-ACK CP-ACK CP-ACK Shall be sent within 5 seconds of step 21 RRC CONNECTION RELEASE RRC connection is released.	_			Contains DD DATA DDDU (CMC CLIDMIT TDDU). The
shall be z, where z <> y (see step 13). CP-ACK CP-DATA CP-ACK CP-ACK Shall be z, where z <> y (see step 13). Contains RP-ACK RPDU Shall be sent within 5 seconds of step 21 RRC CONNECTION RELEASE RRC connection is released.	19	>	CP-DATA	
20 < CP-ACK 21 < CP-DATA Contains RP-ACK RPDU 22> CP-ACK Shall be sent within 5 seconds of step 21 23 < RRC CONNECTION RELEASE RRC connection is released.				
21	20		CB ACK	Strail be 2, where 2 <> y (see step 13).
22> CP-ACK Shall be sent within 5 seconds of step 21 RRC CONNECTION RELEASE RRC connection is released.	-	-		Contains BD ACK BDDU
23 < RRC CONNECTION RELEASE RRC connection is released.		-		
			I	· · · · · · · · · · · · · · · · · · ·
24 > KKU CUNNECTION KELEAGE		-		NNO connection is released.
COMPLETE	24	>		

16.1.9.2.5 Test requirements

In step 9 the UE shall transmit a CM SERVICE REQUEST for the new CM connection (for the second short message) before the final CP-ACK for the old MM connection is transmitted.

In step 15 the UE shall transmit a CM SERVICE REQUEST for the new CM connection (for the third short message) before the final CP-ACK for the old MM connection is transmitted.

16.2 Short message service point to point on PS mode

All of test cases in this sub-clause are applied to the UE supported PS mode.

16.2.1 SMS mobile terminated

16.2.1.1 Definition

16.2.1.2 Conformance requirements

An active UE shall be able to receive short message TPDU (SMS-DELIVER) at any time, independently of whether or not there is a PDP context in progress. A report will always be returned to the SC, confirming that the UE has received the short message.

Reference

3GPP TS 23.040, sub-clause 3.1.

16.2.1.3 Test purpose

To verify the ability of a UE to receive and decode the SMS where provided for the point to point service.

16.2.1.4 Method of test

Initial Conditions

- System simulator:
 - 1 cell, default parameters.
- User Equipment:
 - the UE shall be in "Idle, updated" state GMM-state "GMM-REGISTERED";
 - the SMS message storage shall be empty.

Related ICS/IXIT Statements

Support for Short message MT/PP.

Description of the basic procedures to display a mobile terminated short message.

The value of timer TC1M.

Whether SMS messages are stored in the USIM and/or the ME.

Support for session management state"PDP-ACTIVE".

Test procedure

a) Mobile terminates establishment of Radio Resource Connection. After the completion of RRC Connection the SS authenticates the UE and activates ciphering.

After the SS receives SECURITY MODE COMPLETE, the SS sends a CP-DATA message. The information element of the CP-DATA message will be RP-DATA RPDU (SMS DELIVER TPDU).

- b) The SS waits a maximum of 25 seconds for the CP-ACK message and then a maximum of 60 seconds for the CP-DATA message containing the RP-ACK RPDU.
- c) The SS sends a CP-ACK to the UE within TC1M with no further CP-DATA messages and the SS initiates channel release.
- d) Steps a), b) and c) are repeated but the first CP-DATA message from the UE is not acknowledged. The second CP-DATA message from the UE is acknowledged by a CP-ACK within a time TC1M.

- e) Steps a) and b) are repeated. The SS is configured not to send CP-ACK. Then maximum 3 CP-DATA retransmissions may occur. After a duration of TC1M + 5 seconds after the last CP-DATA retransmission the SS then initiates the channel release. The 5 seconds is the appropriate time to wait to verify that the UE does not send more than the maximum allowed (3) CP-DATA retransmissions.
- f) The SMS message store shall be cleared manually by the operator.
- g) A PDP context is established with the SS and the state PDP-ACTIVE of session management is entered. The SS sends a PAGING TYPE 2.

The SS sends a CP-DATA message. The information element of the CP-DATA message will be RP-DATA RPDU (SMS DELIVER TPDU). The SS waits a maximum of 25 seconds for the CP-ACK message and then a maximum of 60 seconds for the CP-DATA message containing the RP-ACK RPDU.

- h) The SS sends a CP-ACK to the UE within TC1M with no further CP-DATA messages and the SS initiates channel release. The SMS message store shall be cleared manually by the operator.
- i) Steps g) and h) are repeated but the first CP-DATA message from the UE is not acknowledged. The second CP-DATA message from the UE is acknowledged by a CP-ACK within a time TC1M.
- j) Step g) is repeated. The SS is configured not to send CP-ACK. Then maximum 3 CP-DATA retransmissions may occur. After a duration of TC1M + 15 seconds after the last CP-DATA retransmission the SS initiates the channel release. The 15 seconds is the appropriate time to wait to verify that the UE does not send more than the maximum allowed (3) CP-DATA retransmissions (during PDP context in progress).
- k) A PDP context is established with the SS and the state PDP-ACTIVE of session management is entered. The PDP context is cleared by the SS with a disconnect message. (The PDP context deactivating is continued in parallel to the following exchange of messages related to SMS).

The SS sends a CP-DATA RPDU (SMS DELIVER TPDU) message. The information element of the CP-DATA message is RP-DATA.

The SS waits a maximum of 25 seconds for the CP-ACK message and then a maximum of 60 seconds for the CP-DATA message containing the RP-ACK RPDU.

The SS sends a CP-ACK to the UE within TC1M with no further CP-DATA messages and the SS initiates channel release.

The SMS message store shall be cleared manually by the operator.

1) A PDP context is established with the SS and the state PDP-ACTIVE of session management is entered. After the SS sends a PAGING TYPE 2,the PDP context shall be cleared from the UE. (The PDP context deactivating is continued in parallel to the following exchange of messages related to SMS).

The SS sends a CP-DATA message. The information element of the CP-DATA message is RP-DATA RPDU (SMS DELIVER TPDU).

The SS waits a maximum of 25 seconds for the CP-ACK message and then a maximum of 60 seconds for the CP-DATA message containing the RP-ACK RPDU.

The SS sends a CP-ACK to the UE within TC1M with no further CP-DATA messages and the SS initiates channel release.

The SMS message store shall be cleared manually by the operator.

Step	Direction		Message	Comments
	UE	SS		
1			Mobile terminated establishment	See <u>3GPP</u> TS34.108
			of Radio Resource Connection	
2		>	SERVICE REQUEST	
3	<		AUTHENTICATION AND	
			CIPHERING REQUEST	

Step	Direction	Message	Comments
P	UE SS		2
4		AUTHENTICATION AND	
4	>		
5		CIPHERING RESPONSE SECURITY MODE COMMAND	
5 6	<		
	>	SECURITY MODE COMPLETE	Contains DD DATA DDDU (CMC DELIVED TDDU)
7	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU) Waits max 25 seconds for CP-ACK
8	SS	CD ACK	Walts max 25 seconds for CP-ACK
9	> SS	CP-ACK	Weite may 60 accords for DD ACK DDDLI
10		CD DATA	Waits max 60 seconds for RP-ACK RPDU Contains RP-ACK RPDU
11	>	CP-DATA	Contains RP-ACK RPDU
12 13	< UE	CP-ACK	There should be no further CD DATA magazage until the
13	UE		There should be no further CP-DATA messages until the
14	UE		UE aborts the RR connection (disconnection of layer 2). The UE shall indicate that an SM has arrived. If the UE provides the functionality to display MT messages, it is checked that the correct message is displayed
15		Mobile terminated establishment	See <u>3GPP</u> TS34.108
40		of Radio Resource Connection	
16	>	SERVICE REQUEST	
17	<	AUTHENTICATION AND	
18	>	CIPHERING REQUEST AUTHENTICATION AND CIPHERING RESPONSE	
19	<	SECURITY MODE COMMAND	
20	>	SECURITY MODE COMPLETE	
21	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU)
22	ŝs	OF BATTA	Waits max 25 seconds for CP-ACK
23	>	CP-ACK	Traile max 20 coolide for or rior
24	SS	or non	Waits max 60 seconds for RP-ACK RPDU
25	>	CP-DATA	First CP-DATA from UE, contains RP-ACK RPDU
26	SS	CF-DATA	
27		CP-DATA	First CP-DATA message not acknowledged by SS Retransmitted CP-DATA from UE within twice TC1M,
21	>	CF-DATA	after step 25, contains RP-ACK RPDU
28	_	CP-ACK	
29	< UE	CP-ACK	Second CP_DATA message is acknowledged
29	UE		There should be no further CP-DATA messages until the UE aborts the RRC connection
30	UE		The UE shall indicate that an SM has arrived. If the UE
30	OL		provides the functionality to display MT messages, it is
			checked that the correct message is displayed
31		Mobile terminated establishment	See <u>3GPP</u> TS34.108
31		of Radio Resource Connection	Gee <u>GGI 1 </u> 1004.100
32	>	SERVICE REQUEST	
33	<	AUTHENTICATION AND	
33		CIPHERING REQUEST	
34	>	AUTHENTICATION AND	
"		CIPHERING RESPONSE	
35	<	SECURITY MODE COMMAND	
36	>	SECURITY MODE COMPLETE	
37	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU)
38	SS	OI DATA	Waits max 25 seconds for CP-ACK
		CP-ACK	VVAILS IIIAA 25 SCOULUS IUL OF "MON
39 40	> SS	OF-ACK	Waita may 60 accords for DD ACK DDDU
-		CP-DATA	Waits max 60 seconds for RP-ACK RPDU Contains RP-ACK RPDU
41	>	OF-DATA	
42	SS	CD DATA	First CP-DATA message not acknowledged by SS
43		CP-DATA	Retransmitted CP-DATA from UE within twice TC1M after
44	SS		step 41, contains RP-ACK RPDU Retransmitted CP-DATA message not acknowledged by
44	33		SS
45	UE		Depending upon the maximum number of CP-DATA
45	UE		retransmissions implemented, step 43 and 44 may be
			repeated.
46	<	RRC CONNECTION RELEASE	RRC connecton is released after a duration of TC1M + 5
40	ζ	INTO CONNECTION RELEASE	seconds after the last CP-DATA retransmission.
47		RRC CONNECTION RELEASE	36001103 alter the last OF DATA lettalistilission.
41	>	COMPLETE	
I .	ļ	OOMFLETE	1

Step	Direction	Message	Comments
2.56	UE SS		
48	UE SS		The UE shall indicate that an SM has arrived. If the UE provides the functionality to display MT messages, it is checked that the correct message is displayed A PDP context is established with the SS and the state
50	00	PAGING TYPE 2	PDP-ACTIVE of session management is entered.
51	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU)
52	ŝs	OI BATA	Waits max 25 seconds for CP-ACK
53	>	CP-ACK	
54	SS		Waits max 60 seconds for RP-ACK RPDU
55	>	CP-DATA	Contains RP-ACK RPDU
56	<	CP-ACK	
57	<	DEACTIVATE PDP CONTEXT REQUEST	Deactivates an existing PDP context.
58	> UE	DEACTIVATE PDP CONTEXT ACCEPT	The UE shall indicate that an SM has arrived. If the UE
59 60	UE		provides the functionality to display MT messages, it is checked that the correct message is displayed Clear the SMS message store
61	SS		A PDP context is established with the SS and the state PDP-ACTIVE of session management is entered.
62	<	PAGING TYPE 2 CP-DATA	Contains DD DATA DDDLL/CMC DELIVED TDDLIV
63 64	< SS	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU) Waits max 25 seconds for CP-ACK
65	>	CP-ACK	Walls Hax 25 seconds for Cr -ACK
66	SS		Waits max 60 seconds for RP-ACK RPDU
67	>	CP-DATA	First CP-DATA from UE, contains RP-ACK RPDU
68	SS		First CP-DATA message not acknowledged by SS
69	>	CP-DATA	Retransmitted CP-DATA message within twice TC1M after step 67, contains RP-ACK RPDU
70 71	< <	CP-ACK DEACTIVATE PDP CONTEXT REQUEST	Second CP-DATA message is acknowledged Deactivates an existing PDP context.
72	>	DEACTIVATE PDP CONTEXT ACCEPT	
73	UE		There should be no further CP-DATA messages until the UE aborts the RRC connection
74	UE		The UE shall indicate that an SM has arrived. If the UE provides the functionality to display MT messages, it is checked that the correct message is displayed
75 76	UE SS		Clear the SMS message store A PDP context is established with the SS and the state
77	<	PAGING TYPE 2	PDP-ACTIVE of session management is entered.
78	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU)
79	ŝs		Waits max 25 seconds for CP-ACK
80	>	CP-ACK	
81	SS		Waits max 60 seconds for RP-ACK RPDU
82	>	CP-DATA	First CP-DATA from UE, contains RP-ACK RPDU
83 84	SS >	CP-DATA	First CP-DATA message not acknowledged by SS Transmitted CP-DATA message within twice TC1M after
85	SS	OI -DATA	step 82, contains RP-ACK RPDU Retransmitted CP-DATA message within twice TCTM after step 82, contains RP-ACK RPDU Retransmitted CP-DATA message not acknowledged by
	UE		SS
86	UE		Depending on the maximum number of CP-DATA retransmissions implemented, step 83-84 may be repeated. The maximum number of retransmissions may however not exceed three.
87	<	RRC CONNECTION RELEASE	RRC CONNECTION is released after a duration of TC1M + 15 seconds after the last CP-DATA retransmission.
88	>	RRC CONNECTION RELEASE COMPLETE	

Step	Direction	Message	Comments
	UE SS	1	
89	UE		The UE shall indicate that an SM has arrived. If the UE provides the functionality to display MT messages, it is checked that the correct message is displayed
90	UE		Clear the SMS message store
91	SS		A PDP context is established with the SS and the state PDP-ACTIVE of session management is entered.
92	<	PAGING TYPE 2	Sent on DCCH associated with the DTCH
93	<	DEACTIVATE PDP CONTEXT	The PDP context is deactivated by the SS. The PDP
		REQUEST	context deactivating is continued in parallel to the
94	<	CP-DATA	following exchange of messages related to SMS. Contains RP-DATA RPDU (SMS DELIVER TPDU)
95	SS	CF-DATA	Waits max 25 seconds for CP-ACK
96	>	CP-ACK	Walls max 20 seconds for or Norc
97	SS		Waits max 60 seconds for RP-ACK RPDU
98	>	CP-DATA	Contains RP-ACK RPDU
99	<	CP-ACK	
100	UE		There should be no further CP-DATA messages until the
404	UE		UE aborts the RR connection.
101	UE		The UE shall indicate that an SM has arrived. If the UE provides the functionality to display MT messages, it is checked that the correct message is displayed
102	UE		Clear the SMS message store
103	SS		A PDP context is established with the SS and the state
			PDP-ACTIVE of session management is entered.
104	<	PAGING TYPE 2	TI DDD
105	>	DEACTIVATE PDP CONTEXT REQUEST	The PDP context is deactivated by the UE. The PDP context deactivation is continued in parallel to the
		REQUEST	following
106	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU)
107	<	DEACTIVATE PDP CONTEXT	
		REQUEST COMPLETE	
108	>	CP-ACK	shall be sent before 25 seconds after the start of step 106
109	SS		Waits max 60 seconds for RP-ACK RPDU
110	>	CP-DATA	Contains RP-ACK RPDU
111 112	< UE	CP-ACK	There should be no further CP-DATA messages until the
''2	UE		UE aborts the RRC connection.
113	UE		The UE shall indicate that an SM has arrived. If the UE
			provides the functionality to display MT messages, it is
			checked that the correct message is displayed
114	UE		Clear the SMS message store

NOTE: Time values for SS wait time are chosen sufficiently high to be sure that the UE has enough time to respond to the different messages.

Specific Message Contents

SMS DELIVER TPDU

Information element	Comment Value
TP-UDL	160
TP-UD (140 octets)	text of message (160 characters)

NOTE: The 160 characters in TP-UD shall include at least one occurrence of each character in the default alphabet (see <u>3GPP TS</u> 23.038, sub-clause 6.2.1).

16.2.1.5 Test requirements

After step 7 UE shall receive SMS-DELIVER TPDU and send CP-ACK within 25 seconds and CP-DATA containing RP-ACK within 60 seconds.

3GPP TS 34.123-1 V3.2.0(2001-1)

1345

After step 25 UE shall retransmit CP-DATA containing RP-ACK within twice TC1M.

After step 30 UE shall indicate that an SM has arrived.

After step 43 UE shall repeat CP-DATA retransmissions as many times as the decided maximum number.

After step 48 UE shall indicate that an SM has arrived.

After step 51 UE shall receive SMS-DELIVER TPDU and send CP-ACK within 25 seconds and CP-DATA containing RP-ACK within 60 seconds.

After step 67 UE shall retransmit CP-DATA containing RP-ACK within twice TC1M.

After step 73 UE shall indicate that an SM has arrived.

After step 78 UE shall repeat CP-DATA retransmissions as many times as the decided maximum number.

After step 89 UE shall indicate that an SM has arrived.

After step 94 UE shall receive SMS-DELIVER TPDU and send CP-ACK within 25 seconds and CP-DATA containing RP-ACK within 60 seconds.

After step 106 UE shall receive SMS-DELIVER TPDU and send CP-ACK within 25 seconds and CP-DATA containing RP-ACK within 60 seconds.

16.2.2 SMS mobile originated

16.2.2.1 Definition

16.2.2.2 Conformance requirements

An active UE shall be able to submit short message TPDU (SMS-SUBMIT) at any time, independently of whether or not there is a PDP context in progress.

Reference

3GPP TS 23.040, sub-clause 3.1.

16.2.2.3 Test purpose

To verify that the UE is able to correctly send a short message where the SMS is provided for the point to point service. The test also verifies that the UE is capable of simultaneously receiving a network originated SM whilst sending a mobile originated SM.

16.2.2.4 Method of test

Initial Conditions

- System simulator:
 - 1 cell, default parameters.
- User Equipment:
 - the UE shall be in "Idle, updated" state GMM-state "GMM-REGISTERED";
 - the SMS message storage shall be empty.

Related ICS/IXIT Statements

Support for Short message MO/PP.

Description of the basic procedures to display a mobile originated short message.

Support for state PDP-ACTIVE of session management.

The value of timer TC1M.

Whether SMS messages are stored in the USIM and/or the ME.

Maximum length (characters) of a mobile originated short message.

Test procedure

- a) The UE shall be set up to send a SM to the SS. The SS responds to RRC CONNECTION REQUEST by allocating a CCCH. The SS receives RRC CONNECTION SETUP COMPLETE on DCCH and then performs the authentication.
- b) After receiving SECURITY MODE COMMAND UE shall send SECURITY COMMAND COMPLETE.
- c) The SS responds to the CP-DATA containing RP-DATA RPDU (SMS SUBMIT TPDU) from the UE with a CP-ACK message within TC1M followed by a CP-DATA message containing the correct RP-ACK RPDU. The SS waits a maximum of 25 seconds for the CP-ACK message.
- d) The SS sends a channel release message to the UE.
- e) Steps a) and b) are repeated. The SS is configured not to send the CP-ACK message. Then maximum 3 CP-DATA retransmissions may occur. After a duration of TC1M + 5 seconds after the last CP-DATA retransmission the SS initiates channel release. The 5 seconds is the appropriate time to wait to verify that the UE does not send more than the maximum CP-DATA retransmissions.
- f) Steps a) and b) are repeated. On receipt of the CP-DATA from the UE the SS sends a CP-ERROR message within TC1M containing a "Network Failure" cause. Then the SS initiates channel release.
- g) A PDP context is established with the SS and the state PDP-ACTIVE of session management is entered. The UE is setup to send an SM to the SS. After the reception of the SERVICE REQUEST, the SS sends a SERVICE ACCEPT message.
- h) The SS responds to the CP-DATA containing RP-DATA RPDU (SMS SUBMIT TPDU) from the UE with a CP-ACK message within TC1M followed by a CP-DATA message containing the correct RP-ACK RPDU. The SS waits a maximum of 25 seconds for the CP-ACK message. Then the SS sends a channel release message to the UE.
- i) Step g) is repeated. The SS is configured not to send the CP-ACK message. Then maximum 3 CP-DATA retransmissions may occur. After a duration of TC1M + 15 seconds after the last CP-DATA retransmission the SS initiates channel release. The 15 seconds is the appropriate time to wait to verify that the UE does not send more than the maximum CP-DATA retransmissions (during a PDP context in progress).
- j) The SS is configured to receive a mobile originated SM. Steps a) and b) are repeated and, using the end of the CP-DATA message from the UE as a trigger, the SS sends a SM to the UE. In this case a new transaction identifier shall be used in the CP messages of SMS mobile terminated.
- k) The UE is set up to send an SM to the SS. On receipt of the SERVICE REQUEST the SS sends a SERVICE REJECT message with the reject cause set to "GPRS services not allowed". After 5 seconds the SS initiates channel release.

Step	Direction	Message	Comments
Step	UE SS	Message	Comments
4		CVCTEM INICODMATION	DOCH
1	<	SYSTEM INFORMATION	BCCH
2	>	RRC CONNECTION REQUEST	CCCH
3	<	RRC CONNECTION SETUP	СССН
4	>	RRC CONNECTION SETUP	DCCH
		COMPLETE	
5	>	SERVICE REQUEST	
6	<	AUTHENTICATION AND	
		CIPHERING REQUEST	
7	>	AUTHENTICATION AND	
· ·		CIPHERING RESPONSE	
8	<	SECURITY MODE COMMAND	
9	>	SECURITY MODE COMPLETE	
10	>	CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU)
		_	
11	<	CP-ACK	Sent within TC1M after step 10
12	<	CP-DATA	Contains RP-ACK RPDU
13	SS		Waits max 25 seconds for CP-ACK
14	>	CP-ACK	
15	<	RRC CONNECTION RELEASE	RRC connection is released.
16	>	RRC CONNECTION RELEASE	
		COMPLETE	
17	<	SYSTEM INFORMATION	BCCH
18	>	RRC CONNECTION REQUEST	CCCH
19	<	RRC CONNECTION SETUP	CCCH
20	>	RRC CONNECTION SETUP	DCCH
		COMPLETE	
21	>	SERVICE REQUEST	
22	<	AUTHENTICATION AND	
	,	CIPHERING REQUEST	
23	>	AUTHENTICATION AND	
20		CIPHERING RESPONSE	
24	<	SECURITY MODE COMMAND	
25	>	SECURITY MODE COMPLETE	
26		CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU)
_	> SS	CF-DATA	
27		CD DATA	SS configured not to send CP-ACK
28	>	CP-DATA	Retransmitted CP-DATA message within twice TC1M
			after step 26
29	UE		Depending on the maximum number of CP-DATA
			retransmissions implemented, step 28 may be repeated.
			The maximum number of retransmissions may however
		BBO CONNECTION BELEACE	not exceed three.
30	<	RRC CONNECTION RELEASE	RRC CONNECTION is released after a duration of TC1M
a .		DDG GOVINESTICK STITLES	+ 5 seconds after the last CP-DATA retransmission.
31	>	RRC CONNECTION RELEASE	
		COMPLETE	
32	<	SYSTEM INFORMATION	BCCH
33	>	RRC CONNECTION REQUEST	CCCH
34	<	RRC CONNECTION SETUP	CCCH
35	>	RRC CONNECTION SETUP	DCCH
		COMPLETE	
36	>	SERVICE REQUEST	
37	<	AUTHENTICATION AND	
		CIPHERING REQUEST	
38	>	AUTHENTICATION AND	
		CIPHERING RESPONSE	
39	<	SECURITY MODE COMMAND	
40	>	SECURITY MODE COMPLETE	
41	>	CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU)
42	<	CP-ERROR	Sent within TC1M containing "Network Failure" cause.
43	<	RRC CONNECTION RELEASE	RRC CONNECTION is released.
44	>	RRC CONNECTION RELEASE	
''	_	COMPLETE	
I	ı	, · · · ·	1

Step	Direction	Message	Comments
	UE SS		
45 46	SS UE		A PDP context is established with the SS and the state PDP-ACTIVE of session management is entered. The UE is set up to send an SM
47	>	SERVICE REQUEST	·
48	<	SERVICE ACCEPT	
49	>	CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU)
50	<	CP-ACK	Sent within TC1M after step 49
51	<	CP-DATA	Contains RP-ACK RPDU
52	SS	J. 17.17.	Waits max 25 seconds for CP-ACK
53	>	CP-ACK	
54	<	RRC CONNECTION RELEASE	RRC CONNECTION is released.
55	>	RRC CONNECTION RELEASE CONPLETE	TARG GOTALESTION IS TO LOCALISE.
56	SS		A PDP context is established with the SS and the state PDP-ACTIVE of session management is entered.
57	>	SERVICE REQUEST	
58	<	SERVICE ACCEPT	
59	>	CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU)
60	SS		SS configured not to send CP-ACK
61	>	CP-DATA	Transmitted CP-DATA message within twice TC1M after step 59
62	UE		Depending on the maximum number of CP-DATA retransmissions implemented, step 61 may be repeated. The maximum number of retransmissions may however not exceed three.
63	<	RRC CONNECTION RELEASE	RRC CONNECTION is released after a duration of TC1m + 15 seconds after the last CP-DATA retransmission.
64	>	RRC CONNECTION RELEASE COMPLETE	
65	<	SYSTEM INFORMATION	BCCH
66	>	RRC CONNECTION REQUEST	CCCH
67	<	RRC CONNECTION SETUP	CCCH
68	>	RRC CONNECTION SETUP COMPLETE	DCCH
69	>	SERVICE REQUEST	
70	<	AUTHENTICATION AND CIPHERING REQUEST	
71	>	AUTHENTICATION AND CIPHERING RESPONSE	
72	<	SECURITY MODE COMMAND	
73	>	SECURITY MODE COMPLETE	
74	>	CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU)
75	SS		The SS sends an SM to the UE triggered by the end of
76	<	CP-DATA	the CP-DATA message from the UE Contains RP-DATA RPDU (SMS DELIVER TPDU)
76	VE	OI -DATA	The UE shall correctly receive the SM and indicate that a
''	UE		message has arrived. If the UE provides the functionality
			to display MT messages, it is checked that the correct
			message is displayed. In the MO case the UE shall send
			the CP-ACK message with transaction identifier assigned
			to this transfer. In the MT case the UE shall send a CP-
			ACK message and a CP-DATA message containing the
			RP-ACK RPDU. The transaction identifier shall be the
70		DDG GONNEGTICAL BEGLIEST	same as chosen by the SS for the MT transfer.
78	>	RRC CONNECTION REQUEST	initiate outgoing call
79	<	RRC CONNECTION SETUP	
80	>	RRC CONNECTION SETUP	
0.4		COMPLETE	
81	>	SERVICE REQUEST	Deient course and to IICDDO
82	<	SERVICE REJECT	Reject cause set to "GPRS services not allowed"
83	<	RRC CONNECTION RELEASE	Sent 5 seconds after SERVICE REJ
84	>	RRC CONNECTION RELEASE	
		COMPLETE	

NOTE: Time values for SS wait times are chosen sufficiently high to be sure that the UE has enough time to respond to the different messages.

Specific Message Contents

SMS SUBMIT TPDU

Information element	Comment Value
TP-UDL TP-UD (140 octets max)	as applicable maximum number of characters (text of message) as defined by the manufacturer (see ICS/IXIT)

16.2.2.5 Test requirements

After step 9 UE shall send a CP-DATA containing RP-data. The RP-DATA shall contain SMS SUBMIT TPDU.

After step 26 UE shall retransmit a CP-DATA containing RP-data. The RP-DATA shall contain SMS SUBMIT TPDU.

After step 45 UE shall send the RRC CONNECTION RELEASE COMPLETE.

After step 48 UE shall send a CP-DATA containing RP-data. The RP-DATA shall contain SMS SUBMIT TPDU.

After step 71 UE shall repeat CP-DATA retransmissions as many times as the decided maximum number.

After step 76 UE shall correctly receive the SM and indicate that a message has arrived.

After step 82 UE shall not send CP-DATA.

16.2.3 Test of memory full condition and memory available notification:

The Memory Available Notification provides a means for the UE to notify the network that it has memory available to receive one or more short messages. The SMS status field in the USIM contains status information on the "memory available" notification flag.

16.2.3.1 Definition

16.2.3.2 Conformance requirement

- When a mobile terminated message is Class 2, the UE shall ensure that the message has been transferred to the SMS data field in the USIM before sending an acknowledgement to the SC. The UE shall return a protocol error message if the short message cannot be stored in the USIM and there is other short message storage available in the UE. If all the short message storage in the UE is already in use, the UE shall return "memory capability exceeded".
- 2. When the UE rejects a short message due to lack of available memory capability the need to transfer notification shall be stored in the USIM.
- 3. If the memory capability becomes available because memory is cleared, the value of the memory capability exceeded notification flag in the USIM is read. If the flag is set, the UE notifies the network that memory capability is now available. After a positive acknowledgement from the network, the ME unsets the memory capability exceeded notification flag in the USIM.

References

- 3GPP TS 23.040, sub-clause 9.2.3.10, 3GPP TS 23.038, clause 4.
- 3GPP TS 23.040, sub-clause 10.3 (operation 14).
- <u>3GPP</u>TS 23.040, sub-clause 10.3 (operation 14).

16.2.3.3 Test purpose

- 1. To verify that the UE sends the correct acknowledgement when its memory in the USIM becomes full.
- 2. To verify that the UE sends the correct acknowledgement when its memory in the ME and the USIM becomes full, and sets the "memory exceeded" notification flag in the USIM.
- 3. To verify that the UE performs the "memory available" procedure when its message store becomes available for receiving short messages, and only at this moment.

16.2.3.4 Method of test

Initial conditions

- System Simulator:
 - 1 cell, default parameters.
- User Equipment:
 - the UE shall be in the idle updated state GMM-state "GMM-REGISTERED";
 - the SMS message storage shall be empty;
 - the UE shall be connected to the USIM simulator. The following shall be present in the USIM simulator:
 - EF_{SMS} with at least one record;
 - EF_{SMSstatus}, with SMS "Memory Cap. Exceed" notification flag set to "memory available";
 - Service no. 4 (SMS) in EF_{SST} set to allocated and activated.
 - for storing of Class 1 Short Messages the UE shall be set up to store Short Messages in the ME memory (by way of MMI, as described in ICS/IXIT statement).

Related ICS/IXIT Statements

Support for Short message MT/PP.

Description of the basic procedures to display a mobile terminated short message.

Whether SMS messages are stored in the USIM and/or the ME.

The value of timer TC1M.

Test procedure

- a) step a) of sub-clause 16.2.5.3 (test of Class 2 Short Messages) is repeated until the UE sends a negative acknowledgement (RP-ERROR). The USIM simulator shall indicate if the "memory capability exceeded" notification flag has been set on the USIM.
- b) a Class 1 Short Message is sent to the UE.
- c) step b) is repeated until the UE sends a negative acknowledgement (RP-ERROR). The USIM simulator shall indicate if the "memory capability exceeded" notification flag has been set on the USIM.
- d) a Short Message is sent to the UE with the DCS field of the SMS-DELIVER TPDU set to 0.
- e) the SS prompts the operator to read a short message and to remove it from the message store of the UE.
- f) the SS waits for a RRC CONNECTION REQUEST from the UE, and sends a RRC CONNECTION SETUP.
- g) after the SS receives a RRC CONNECTION SETUP COMPLETE, the SS authenticates the UE and activates ciphering.

- h) the SS answers to the RP-SMMA from the UE with a CP-DATA containing a RP-ACK RPDU.
- i) after the UE has acknowledged the CP-DATA with a CP-ACK, the SS releases the RRC connection. The USIM simulator shall indicate if the "memory capability exceeded" notification flag has been unset on the USIM.
- j) step e) is repeated.

Expected sequence

Step	Direction	on Message	Comments
Otop		S	Commonts
1	OL 3	Mobile terminated establishment	See 3GPP TS34.108
'		of Radio Resource Connection	000 <u>0011</u> 1004.100
2	>	SERVICE REQUEST	
3	<	AUTHENTICATION AND	
		CIPHERING REQUEST	
4	>	AUTHENTICATION AND	
4	>	CIPHERING RESPONSE	
5	<	SECURITY MODE COMMAND	
6	>	SECURITY MODE COMPLETE	
7	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU), Class
'		OI -DATA	2 Short Message
8	SS		Waits max 25 seconds for CP-ACK
9	>	CP-ACK	Walls Max 25 5000 lds for Or More
10	SS	or non	Waits max 60 seconds for RP-ACK RPDU
11	>	CP-DATA	Contains RP-ACK RPDU
12	<	CP-ACK	Within TC1M after step 11
13	<	RRC CONNECTION RELEASE	RRC connection is released. Step 1-18 is repeated until
	,	THE SOUND TON NEED NOT	UE sends a negative acknowledgement (RP-ERROR) in
			step 11. The RP-ERROR RPDU cause field shall be
			"Protocol error, unspecified" if there is message capability
			in the ME, or "Memory capability exceeded" if there is no
			message capability in the ME. If the total memory store of
			the UE is full, the ME shall set the "memory capability
			exceeded" notification flag on the USIM.
			-
14	>	RRC CONNECTION RELEASE	
4.5		COMPLETE	Can 2000 T024 400
15		Mobile terminated establishment of Radio Resource Connection	See <u>3GPP</u> TS34.108
16		SERVICE REQUEST	
17	> <	AUTHENTICATION AND	
17		CIPHERING REQUEST	
18	>	AUTHENTICATION AND	
'0		CIPHERING RESPONSE	
19	<	SECURITY MODE COMMAND	
20	>	SECURITY MODE COMPLETE	
21	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU), Class
			1 Short Message
22	SS		Waits max 25 seconds for CP-ACK
23	>	CP-ACK	
24	SS		Waits max 60 seconds for RP-ACK RPDU
25	>	CP-DATA	Shall contain RP-ACK RPDU if there is memory
			capability in the ME. If not it shall contain RP-ERROR
			RPDU which cause field shall be "memory capability
			exceeded". If the total memory store of the UE now
			becomes full at this step, the ME shall set the "memory
			cap. exceed" notification flag on the USIM.
26	<	CP-ACK	Within TC1M after step 25
27	<	RRC CONNECTION RELEASE	RRC connection is released. Step 19-36 is repeated until
			the UE sends an RP-ERROR. The USIM simulator shall
			indicate if the "memory capability exceeded" notification
			flag has been set on the USIM.
28	>	RRC CONNECTION RELEASE	
		COMPLETE	

Step		ction	Message	Comments
	UE	SS		
29			Mobile terminated establishment	See <u>3GPP</u> TS34.108
			of Radio Resource Connection	
30		>	SERVICE REQUEST	
31	<		AUTHENTICATION AND	
			CIPHERING REQUEST	
32		·>	AUTHENTICATION AND	
			CIPHERING RESPONSE	
33	<		SECURITY MODE COMMAND	
34		·>	SECURITY MODE COMPLETE	
35	<		CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU) with
				TP-DCS set to 0
36	S	S		Waits max 25 seconds for CP-ACK
37		>	CP-ACK	
38	S	S		Waits max 60 seconds for RP-ACK RPDU
39		>	CP-DATA	Shall contain RP-ERROR RPDU with error cause
				"memory capability exceeded".
40	<		CP-ACK	Within TC1M after step 39
41	<		RRC CONNECTION RELEASE	RRC connection is released.
42		>	RRC CONNECTION RELEASE	
			COMPLETE	
43	S	S		Prompts the operator to remove one of the short
				messages from the message store of the UE.
44	<		SYSTEM INFORMATION	BCCH
45		>	RRC CONNECTION REQUEST	CCCH
46			RRC CONNECTION SETUP	СССН
47		>	RRC CONNECTION SETUP	DCCH
			COMPLETE	
48		>	SERVICE REQUEST	
49			SERVICE ACCEPT	
50		>	CP-DATA	Contains RP-SMMA RPDU
51			CP-ACK	
52			CP-DATA	Contains RP-ACK RPDU
53		>	CP-ACK	Acknowledge of CP-DATA containing the RP-ACK
				RPDU. The ME shall unset the "memory capability
				exceeded" notification flag on the USIM.
54	<		RRC CONNECTION RELEASE	RRC connection is released. The USIM simulator shall
				indicate if the "memory capability exceeded" notification
				flag has been unset on the USIM.
55		>	RRC CONNECTION RELEASE	
	_	_	COMPLETE	
56	l S	S		Prompts the operator to remove one of the short
	l	_		messages from the message store of the UE.
57	l U	E		Shall not attempt to send a RP-SMMA RPDU. This is
				verified by checking that the UE does not send a
				CHANNEL REQUEST message with the establishment
				cause "Other services which can be completed with an
1				SDCCH"

NOTE: Time values for SS wait time are chosen sufficiently high to be sure that the UE has enough time to respond to the different messages.

Specific Message Contents

SMS-DELIVER TPDU in step 7

Information element	Comment Value
TP-DCS	default alphabet, class 2 "11110010"B

SMS-DELIVER TPDU in step 21

SMS-DELIVER TPDU in step 35

Γ	TP-DCS	default alphabet "00000000"B

16.2.3.5 Test requirements

After UE sends a negative acknowledgement (RP-ERROR) in step 11, the USIM simulator shall indicate if the "memory capability exceeded" notification flag has been set on the USIM.

After UE sends a negative acknowledgement (RP-ERROR) in step 23, the USIM simulator shall indicate if the "memory capability exceeded" notification flag has been set on the USIM.

After step 53 the ME shall unset the "memory capability exceeded" notification flag on the USIM.

After step 56 UE shall not attempt to send a RP-SMMA RPDU.

16.2.4 Test of the status report capabilities and of SMS-COMMAND:

This test applies to UEs which support the status report capabilities.

16.2.4.1 Definition

16.2.4.2 Conformance requirement

The SMS offers the SC the capabilities of informing the UE of the status of a previously sent mobile originated short message. This is achieved by the SC returning a status report TPDU (SMS-STATUS-REPORT) to the originating UE.

SMS-COMMAND enables an UE to invoke an operation at the SC.

The UE shall increment TP-MR by 1 for each SMS-SUBMIT or SMS-COMMAND being submitted.

References

- 3GPP TS 23.040, sub-clause 3.2.9.
- <u>3GPP</u>TS 23.040, sub-clause 9.2.3.6.

16.2.4.3 Test purpose

- 1) To verify that the UE is able to accept a SMS-STATUS-REPORT TPDU.
- 2) To verify that the UE is able to use the SMS-COMMAND functionality correctly and sends an SMS-COMMAND TPDU with the correct TP-Message-Reference.

16.2.4.4 Method of test

Initial conditions

- System Simulator:

- 1 cell, default parameters.
- User Equipment:
 - the UE shall be in the idle updated state GMM-state "GMM-REGISTERED".

Related ICS/IXIT Statements

Support of SMS MO/PP and MT/PP.

Test procedure

- a) The UE is made to send a Mobile Originated short message setting TP-SRR as in steps a) to d) of test 16.2.2 (SMS Mobile originated).
- b) The SS sends a CP-DATA message containing a RP-DATA RPDU itself containing an SMS-STATUS-REPORT TPDU.
- c) The SS sends a RRC CONNECTION RELEASE message.
- d) The UE is made to send an SMS-COMMAND message enquiring about the previously submitted short message.

e)

- f) The SS acknowledges the CP-DATA message from the UE with a CP-ACK followed by a CP-DATA message containing an RP-ACK RPDU
- g) After receiving the CP-ACK from the UE, the SS releases the RRC connection by using a RRC CONNECTION RELEASE message.
- h) The UE is made to send an SMS-COMMAND message requiring to delete the previously submitted short message.
- i) steps e) to g) are repeated.

Expected sequence

Step	Direction	Message	Comments
	UE SS		
1	<	SYSTEM INFORMATION	BCCH
2	>	RRC CONNECTION REQUEST	CCCH
3	<	RRC CONNECTION SETUP	CCCH
4	>	RRC CONNECTION SETUP	DCCH
		COMPLETE	
5	>	SERVICE REQUEST	
6	<	AUTHENTICATION AND	
		CIPHERING REQUEST	
7	>	AUTHENTICATION AND	
		CIPHERING RESPONSE	
8	<	SECURITY MODE COMMAND	
9	>	SECURITY MODE COMPLETE	
10	>	CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU)
11	<	CP-ACK	Sent within TC1M after step 10
12	<	CP-DATA	Contains RP-ACK RPDU
13	SS		Waits max 25 seconds for CP-ACK
14	>	CP-ACK	
15	<	RRC CONNECTION RELEASE	RRC connection is released.
16	>	RRC CONNECTION RELEASE	
		COMPLETE	
17		Mobile terminated establishment	See <u>3GPP</u> TS34.108
		of Radio Resource Connection	
18	>	SERVICE REQUEST	
19	<	AUTHENTICATION AND	
		CIPHERING REQUEST	

Step	Direction	Message	Comments
	UE SS		33
20	>	AUTHENTICATION AND	
		CIPHERING RESPONSE	
21	<	SECURITY MODE COMMAND	
22	>	SECURITY MODE COMPLETE	0 4 1 00 0474 00014 (040 074740 050007
23	<	CP-DATA	Contains RP-DATA RPDU (SMS-STATUS-REPORT
24	>	CP-ACK	TPDU)
25	>	CP-DATA	Contains RP-ACK RPDU
26	<	CP-ACK	Contains IXI -ACIX IXI DO
27	<	RRC CONNECTION RELEASE	
28	>	RRC CONNECTION RELEASE	
	-	COMPLETE	
29	UE		The UE is made to send an SMS-COMMAND message
			enquiring about the previously submitted SM
30	<	SYSTEM INFORMATION	BCCH
31	>	RRC CONNECTION REQUEST	CCCH
32	<	RRC CONNECTION SETUP	CCCH
33	>	RRC CONNECTION SETUP	DCCH
24		COMPLETE	
34	>	SERVICE REQUEST	
35	<	AUTHENTICATION AND CIPHERING REQUEST	
36	>	AUTHENTICATION AND	
00		CIPHERING RESPONSE	
37	<	SECURITY MODE COMMAND	
38	>	SECURITY MODE COMPLETE	
39	>	CP-DATA	Contains RP-DATA RPDU (SMS-COMMAND TPDU)
			which shall contain the correct TP-MR
40	<	CP-ACK	
41	<	CP-DATA	Contains RP-ACK RPDU
42	>	CP-ACK	
43	<	RRC CONNECTION RELEASE	
44	>	RRC CONNECTION RELEASE COMPLETE	
45	UE	The UE is made to send an SMS-	message requiring to delete the previously submitted
75	OL	COMMAND	SM.
46	>	RRC CONNECTION REQUEST	CCCH
47	<	RRC CONNECTION SETUP	CCCH
48	>	RRC CONNECTION SETUP	DCCH
		COMPLETE	
49	>	SERVICE REQUEST	
50	<	AUTHENTICATION AND	
		CIPHERING REQUEST	
51	>	AUTHENTICATION AND CIPHERING RESPONSE	
52	<	SECURITY MODE COMMAND	
52	>	SECURITY MODE COMMAND	
54	>	CP-DATA	Contains RP-DATA RPDU (SMS-COMMAND TPDU)
54			which shall contain the correct TP-MR
55	<	CP-ACK	2
56	<	CP-DATA	Contains RP-ACK RPDU
57	>	CP-ACK	
58	<	RRC CONNECTION RELEASE	
59	>	RRC CONNECTION RELEASE	
		COMPLETE	

Specific Message Contents

SMS SUBMIT TPDU

Information element	Comment Value	
TP-SRR	status report is requested "1"B	

SMS-STATUS-REPORT TPDU (SS to UE in step 23):

Information element	Comment Value
TP-MR	same as previous SMS-SUBMIT
TP-MMS	no more messages "1"B
TP-SRQ	result of SMS-SUBMIT "0"B
TP-RA	same as the Destination address of the SMS-SUBMIT
TP-ST	SM received "00000000"B

first SMS-COMMAND TPDU (UE to SS in step 39)

Information element	Comment Value
TP-MR	TP-MR in previous SMS-SUBMIT plus "1"
TP-SRR	status report requested "1"B
TP-CT	Enquiry relating to previously submitted
	short message "00000000"B
TP-MN	not checked (TP-MR in previous SMS-SUBMIT)

second SMS-COMMAND TPDU (UE to SS in step 54)

Information	n element	Comment Value
TP-MR		TP-MR in previous SMS-COMMAND plus "1"
TP-CT		Delete previously submitted short message "00000010"B
TP-MN		not checked (TP-MR in previous SMS-SUBMIT)

16.2.4.5 Test requirements

After step 23 UE accept a SMS-STATUS-REPORT TPDU.

After step 39 UE shall send a SMS-COMMAND TPDU with the correct TP-Message-Reference.

After step 54 UE shall send a SMS-COMMAND TPDU with the correct TP-Message-Reference.

16.2.5 Test of message class 0 to 3

The tests under this sub-clause only apply to a UE capable of displaying short messages (see ICS/IXIT).

16.2.5.1 Short message class 0

16.2.5.1.1 Definition

16.2.5.1.2 Conformance requirement

When a mobile terminated message is class 0 and the UE has the capability of displaying short messages, the UE shall display the message immediately and send an acknowledgement to the SC when the message has successfully reached the UE irrespective of whether there is memory available in the USIM or ME. The message shall not be automatically stored in the USIM or ME.

References

3GPP TS 23.038, clause 4.

16.2.5.1.3 Test purpose

To verify that the UE will accept and display but not store a class 0 message, and that it will accept and display a class 0 message if its message store is full.

NOTE: failure of this test in a UE could cause it to reject a class 0 message when its SMS memory becomes full. This could lead to unwanted repetitions between the UE and the service centre.

16.2.5.1.4 Method of test

Initial conditions

- System Simulator:
 - 1 cell, default parameters.
- User Equipment:
 - the UE shall be in the idle updated state GMM-state "GMM-REGISTERED";
 - the UE message store shall be empty.

Related ICS/IXIT Statements

Support for Short message MT/PP.

Description of the basic procedures to display a mobile terminated short message.

The value of timer TC1M.

Whether SMS messages are stored in the USIM and/or the ME.

Test procedure

- a) The SS sends a class 0 message by using the method described in step a) of sub-clause 16.2.1 but with the TPDU described in this sub-clause.
- b) The UE message store shall be filled (for example by using the method of sub-clause 16.2.3 test of the memory available notification) with the same SMS-DELIVER TPDU except that TP-DCS is set to class 1.
- c) The SS sends a class 0 message as in step a).

Expected sequence

01	D:		•
Step	Direction	Message	Comments
	UE SS		
1		Mobile terminated establishment	See <u>3GPP</u> TS34.108
		of Radio Resource Connection	
2	>	SERVICE REQUEST	
3	<	AUTHENTICATION AND	
		CIPHERING REQUEST	
4	>	AUTHENTICATION AND	
		CIPHERING RESPONSE	
5	<	SECURITY MODE COMMAND	
6	>	SECURITY MODE COMPLETE	
7	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU), Class
			0 Short Message
8	>	CP-ACK	
9	>	CP-DATA	Contains RP-ACK RPDU.
10	<	CP-ACK	
11	<	RRC CONNECTION RELEASE	
12	>	RRC CONNECTION RELEASE	
		COMPLETE	
13	UE		The content of the short message shall be displayed by
			the ME. The UE shall not store the message. This can be
			checked by verifying that it is impossible to retrieve any
			short messages from the UE message store.
14	SS		The UE message store shall be filled (for example by
			using the method of 16.2.3) with Class 1 SMS-DELIVER
			TPDU.
15		Mobile terminated establishment	See <u>3GPP</u> TS34.108
		of Radio Resource Connection	
16	>	SERVICE REQUEST	
17	<	AUTHENTICATION AND	
		CIPHERING REQUEST	
18	>	AUTHENTICATION AND	
		CIPHERING RESPONSE	
19	<	SECURITY MODE COMMAND	
20	>	SECURITY MODE COMPLETE	
21	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU), Class
			0 Short Message
22	>	CP-ACK	
23	>	CP-DATA	Contains RP-ACK RPDU.
24	<	CP-ACK	
25	<	RRC CONNECTION RELEASE	
26	>	RRC CONNECTION RELEASE	
		COMPLETE	
27	UE		The content of the short message shall be displayed by
			the ME.

Specific Message Contents

SMS-DELIVER TPDU (containing a class 0 message) (SS to UE)

Information element	Comment Value	
TP-DCS	default alphabet, class 0	"1111 0000"B

SMS-DELIVER TPDU (containing a class 1 message to fill the UE message store) (SS to UE)

Information element	Comment Value	
TP-DCS	default alphabet, class 1	"1111 0001"B

16.2.5.1.5 Test requirements

After step 7 UE shall accept and display but not store a class 0 message.

After step 21 UE shall accept and display a class 0 message.

16.2.5.2 Test of class 1 short messages

This test shall apply to UEs which support:

- storing of received Class 1 Short Messages; and
- displaying of stored Short Messages.

16.2.5.2.1 Definition

16.2.5.2.2 Conformance requirement

When a mobile terminated message is class 1, the UE shall send an acknowledgement to the SC when the message has successfully reached the UE and can be stored, either in the ME or in the USIM.

References

3GPP TS 23.038, clause 4.

16.2.5.2.3 Test purpose

This procedure verifies that the UE acts correctly on receiving a class 1 message, i.e. that it stores the message in the ME or USIM and sends an acknowledgement (at RP and CP-Layer).

16.2.5.2.4 Method of test

Initial conditions

- System Simulator:
 - 1 cell, default parameters.
- User Equipment:
 - the UE shall be in the idle updated state GMM-state "GMM-REGISTERED";
 - the UE message store shall be empty;
 - for storing of class 1 Short Messages, the UE shall be set up to store Short Messages in the ME memory (by way of MMI, as described in ICS/IXIT statement).

Related ICS/IXIT Statements

Support for Short message MT/PP.

Description of the basic procedures to display a mobile terminated short message.

The value of timer TC1M.

Whether SMS messages are stored in the USIM and/or the ME.

Test procedure

- a) The SS delivers a Short Message of class 1 to the UE as specified in sub-clause 16.2.1, step a).
- b) The Short Message is recalled (e.g. by means of the MMI).

Expected sequence

Step	Directio	n Message	Comments
	UE S	5	
1		Mobile terminated establishment	See <u>3GPP</u> TS34.108
		of Radio Resource Connection	
2	>	SERVICE REQUEST	
3	<	AUTHENTICATION AND	
		CIPHERING REQUEST	
4	>	AUTHENTICATION AND	
		CIPHERING RESPONSE	
5	<	SECURITY MODE COMMAND	
6	>	SECURITY MODE COMPLETE	
7	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU), Class
			1 Short Message
8	>	CP-ACK	
9	>	CP-DATA	Contains RP-ACK RPDU.
10	<	CP-ACK	
11	<	RRC CONNECTION RELEASE	
12	>	RRC CONNECTION RELEASE	
		COMPLETE	
13	UE		The short message shall be recalled and displayed at the
			UE.

Specific Message Contents

SMS-DELIVER TPDU (containing a class 1 message) (SS to UE)

Information element	Comment Value	
TP-DCS	default alphabet, class 1	"1111 0001"B

16.2.5.2.5 Test requirements

After step 7 UE shall store the message in the ME or USIM and send an acknowledgement.

16.2.5.3 Test of class 2 short messages

16.2.5.3.1 Definition

Class 2 Short Messages are defined as USIM specific, and the UE shall ensure that a message of this class is stored on the USIM.

16.2.5.3.2 Conformance requirement

When a mobile terminated message is Class 2, the UE shall ensure that the message has been correctly transferred to the SMS data field in the USIM before sending an acknowledgement to the SC. The UE shall return a "protocol error, unspecified" error message if the short message cannot be stored in the USIM and there is other short message storage available at the UE. If all the short message storage at the UE is already in use, the UE shall return "memory capacity exceeded".

Reference(s)

3GPP TS 23.040, sub-clause 9.2.3.10; 3GPP TS 23.038, clause 4. 3GPP TS 34.108, sub-clause 6.11.3.2.27

16.2.5.3.3 Test purpose

This procedure verifies that the UE acts correctly on receiving a class 2 message, i.e. that it stores the message correctly in the USIM, and if this is not possible, returns a protocol error message, with the correct error cause, to the network.

There are 2 cases:

- 1) if the UE supports storing of short messages in the USIM and in the ME, and storage in the ME is not full, and the short message cannot be stored in the USIM, the error cause shall be "protocol error, unspecified";
- 2) if the UE supports storing of short messages in the USIM and not in the ME, and storage in the ME is not full, and the short message cannot be stored in the USIM, the error cause shall be "memory capacity exceeded".

NOTE: If the UE supports storing of short messages in the USIM and the ME, and storage in the ME is full, and the short message cannot be stored in the USIM, the error cause shall be "memory capacity exceeded".

This case is not tested in this test.

16.2.5.3.4 Method of test

Initial conditions

- System Simulator:
 - 1 cell, default parameters.
- User Equipment:
 - the UE shall be in the idle updated state GMM-state "GMM-REGISTERED";
 - the ME message store shall be empty;
 - the ME shall be connected to the USIM simulator. The following shall be present in the USIM simulator:
 - EF_{SMS} with at least two free records and one full record;
 - EF_{SMSstatus}, with SMS "Memory Cap. Exceed" notification flag set to "memory available";
 - Service no. 4 (SMS) in EF_{SST} set to allocated and activated;
 - for storing of Class 1 Short Messages the UE shall be set up to store Short Messages in the ME memory (by way of MMI, as described in ICS/IXIT statement).

Related ICS/IXIT Statements

Support for Short message MT/PP.

The value of timer TC1M.

Whether SMS messages are stored in the USIM and/or the ME.

Test procedure

- a) The SS delivers a Short Message of class 2 to the UE as specified in sub-clause 16.2.1, step b).
- b) Following an attempt by the ME to store the short message in a free record of EF_{SMS} in the USIM, the USIM simulator returns the status response "OK" ("90 00").
- c) Step a) is repeated.
- d) Following an attempt by the ME to store the short message in a free record of EF_{SMS} in the USIM, the USIM simulator returns the status response "memory problem" ("92 40").
- e) The USIM simulator indicates if an attempt was made in steps a) and c) to store the messages and if the messages are stored according to the requirement.

Expected sequence

C4	Dinastis	Mag	Comp
Step	Direction UE SS	Message	Comments
1	UE 33	Mobile terminated establishment	See <u>3GPP</u> TS34.108
'		of Radio Resource Connection	See <u>SGFF</u>
2	>	SERVICE REQUEST	
3	<	AUTHENTICATION AND	
	,	CIPHERING REQUEST	
4	>	AUTHENTICATION AND	
		CIPHERING RESPONSE	
5	<	SECURITY MODE COMMAND	
6	>	SECURITY MODE COMPLETE	
7	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU), Class 2 Short Message
8	>	CP-ACK	
9	ME		The ME shall correctly store the short message in a free record of EFSMS in the USIM, i.ethe ME shall use a free record - the first byte of the record shall indicate "message received by UE from network" - the TS-Service-Centre-Address shall be correctly stored
			- the TPDU shall be identical to that sent by the SS
			- bytes following the TPDU shall be set to "FF"
10	USIM		The USIM simulator returns the status response "OK" ("90 00"). The USIM simulator shall indicate if an attempt was made by the ME to store the short message in the USIM.
11	>	CP-DATA	Contains RP-ACK RPDU.
12	<	CP-ACK	
13	<	RRC CONNECTION RELEASE	
14	>	RRC CONNECTION RELEASE	
15		COMPLETE Mobile terminated establishment	Son 2CDD TS24 109
15		Mobile terminated establishment of Radio Resource Connection	See <u>3GPP</u> TS34.108
16	>	SERVICE REQUEST	
17	<	AUTHENTICATION AND	
		CIPHERING REQUEST	
18	>	AUTHENTICATION AND	
		CIPHERING RESPONSE	
19	<	SECURITY MODE COMMAND	
20	>	SECURITY MODE COMPLETE	Contains DD DATA DDDLL/CMC DELIVED TDDLIV Class
21	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU), Class 2 Short Message
22	>	CP-ACK	2 OHOR Wessage
23	ME	or non	The ME shall attempt to store the short message in a free record of EFSMS in the USIM.

Step	Direction		Message	Comments
	UE	SS		
25	US 		CP-DATA	The USIM simulator returns the status response "memory problem" ("92 40"). The USIM simulator shall indicate if an attempt was made by the ME to store the short message in the USIM. Contains RP-ERROR RPDU with error cause "protocol error, unspecified" if the UE supports storing of short messages in the ME, or error cause "memory capacity exceeded" if not.
26	<-		CP-ACK	
27	<-		RRC CONNECTION RELEASE	
28		>	RRC CONNECTION RELEASE COMPLETE	

Specific Message Contents

SMS-DELIVER TPDU (containing a class 2 message) (SS to UE)

Information element	Comment Value		
TP-DCS	default alphabet, class 2	"1111 0010"B	

16.2.5.3.5 Test requirements

After step 10 UE shall confirm that the short message is stored in the USIM and send CP-DATA containing RP-ACK RPDU.

After step 24 UE shall confirm that the short message cannot be stored in the USIM and send CP-DATA containing RP-ERROR RPDU. If UE supports storing of short message in the ME, the error cause of RP-ERROR RPDU shall be "protocol error, unspecified", and if not the error cause of RP-ERROR RPDU shall be "memory capacity exceeded"

16.2.5.4 Test of class 3 short messages

For further study.

16.2.6 Test of short message type 0

For further study.

16.2.7 Test of the replace mechanism for SM type 1-7

16.2.7.1 Definition

16.2.7.2 Conformance requirement

On receipt of a short message, the UE shall check to see if the associated Protocol Identifier contains a Replace Short Message Type code. If such a code is present, then the UE will check the associated SC address (RP-OA) and originating address (TP-OA) and replace any existing stored message having the same Protocol Identifier code, SC address and originating address with the new short message.

Reference(s)

3GPP TS 23.040; sub-clause 9.2.3.9.

16.2.7.3 Test purpose

This procedure verifies the correct implementation of the replace mechanism for Replace Short Messages.

16.2.7.4 Method of test

Initial conditions

- System Simulator:
 - 1 cell, default parameters.
- User Equipment:
 - the UE shall be in the idle updated state GMM-state "GMM-REGISTERED";
 - the UE message store shall be empty.

Related ICS/IXIT Statements

Support for Short message MT/PP.

Description of the basic procedures to display a mobile terminated short message.

The value of timer TC1M.

Test procedure

- a) Two different numbers n and m are drawn randomly between 1 and 7. Two different addresses for TP-Originating-Address (TPOA1 and TPOA2) are drawn. Two different addresses for RP-Originating-Address (RPOA1 and RPOA2) are drawn.
- b) The SS delivers a short message to the UE as specified in sub-clause 16.2.1 step a). In the SMS-DELIVER TPDU, the TP-Protocol-Identifier parameter is "Replace Short Message Type n", the TP-Originating-Address is TPOA1, and the RP-Originating-Address is RPOA1.
- c) Step b) is repeated but with a different TP-Originating-Address (TPOA2), and different contents of TP-User-Data in the SMS-DELIVER TPDU. The other parameters are the same as in step b).
- d) Step c) is repeated but with RPOA2 in the RP-Originated-Address, and contents of TP-User-Data different from the former two messages. The other parameters are the same as in step c).
- e) Step d) is repeated but with the TP-Protocol-Identifier equal to "Replace Short Message Type m", and contents of TP-User-Data different from the former three messages. The other parameters are the same as in step d).
- f) Step e) is repeated but the contents of TP-User-Data are different from that used in step e).
- g) The SS prompts the operator to display the Short Messages stored in the UE.

Expected sequence

Step	Direction	Message	Comments
	UE SS		
1		Mobile terminated establishment	See <u>3GPP</u> TS34.108
		of Radio Resource Connection	
2	>	SERVICE REQUEST	
3	<	AUTHENTICATION AND	
		CIPHERING REQUEST	
4	>	AUTHENTICATION AND	
		CIPHERING RESPONSE	
5	<	SECURITY MODE COMMAND	
6	>	SECURITY MODE COMPLETE	
7	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU) TP-
			PID is "Replace Short Message Type n", TP-OA is
			TPOA1 and RP-OA is RPOA1
8	>	CP-ACK	
9	>	CP-DATA	Contains RP-ACK RPDU.
10	<	CP-ACK	

	Step	UE SS	Message	1
		UE 33		
	11	<	RRC CONNECTION RELEASE	
	12	>	RRC CONNECTION RELEASE	
			COMPLETE	
	13		Mobile terminated establishment	See <u>3GPP</u> TS34.108
	1.1	_	of Radio Resource Connection SERVICE REQUEST	
	14 15	> <	AUTHENTICATION AND	
	13	_	CIPHERING REQUEST	
	16	>	AUTHENTICATION AND	
			CIPHERING RESPONSE	
	17	<	SECURITY MODE COMMAND	
	18	>	SECURITY MODE COMPLETE	
	19	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU) TP-
				PID is "Replace Short Message Type n", TP-OA is TPOA2 and RP-OA is RPOA1, TP-UD different from step
				7
	20	>	CP-ACK	'
	21	>	CP-DATA	Contains RP-ACK RPDU.
	22	<	CP-ACK	
	23	<	RRC CONNECTION RELEASE	
	24	>	RRC CONNECTION RELEASE	
1	25		COMPLETE Mobile terminated establishment	See <u>3GPP</u> TS34.108
ı	23		of Radio Resource Connection	1004.100
	26	>	SERVICE REQUEST	
	27	<	AUTHENTICATION AND	
			CIPHERING REQUEST	
	28	>	AUTHENTICATION AND	
	29	_	CIPHERING RESPONSE SECURITY MODE COMMAND	
	30	< >	SECURITY MODE COMMAND SECURITY MODE COMPLETE	
	31	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU) TP-
				PID is "Replace Short Message Type n", TP-OA is
				TPOA2 and RP-OA is RPOA2, TP-UD different from step
				7 and 19
	32 33	>	CP-ACK CP-DATA	Contains RP-ACK RPDU.
	34	> <	CP-ACK	Contains RF-ACK RPDU.
	35	<	RRC CONNECTION RELEASE	
	36	>	RRC CONNECTION RELEASE	
			COMPLETE	
	37		Mobile terminated establishment	See <u>3GPP</u> TS34.108
	20		of Radio Resource Connection	
	38 39	> <	SERVICE REQUEST AUTHENTICATION AND	
	39	\	CIPHERING REQUEST	
	40	>	AUTHENTICATION AND	
			CIPHERING RESPONSE	
	41	<	SECURITY MODE COMMAND	
	42	>	SECURITY MODE COMPLETE	O
	43	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU) TP-PID is "Replace Short Message Type m", TP-OA is
				TPOA2 and RP-OA is RPOA2, TP-UD different from step
				7, 19 and 31
	44	>	CP-ACK	·
	45	>	CP-DATA	Contains RP-ACK RPDU.
	46	<	CP-ACK	
	47	<	RRC CONNECTION RELEASE	
	48	>	RRC CONNECTION RELEASE COMPLETE	
	49		Mobile terminated establishment	See <u>3GPP</u> TS34.108
'	-		of Radio Resource Connection	
	50	>	SERVICE REQUEST	

Step	Direction	Message	Comments
	UE SS		
51	<	AUTHENTICATION AND	
		CIPHERING REQUEST	
52	>	AUTHENTICATION AND	
		CIPHERING RESPONSE	
53	<	SECURITY MODE COMMAND	
54	>	SECURITY MODE COMPLETE	
55	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU) TP-
			PID is "Replace Short Message Type m", TP-OA is
			TPOA2 and RP-OA is RPOA2, TP-UD different from step
			43
56	>	CP-ACK	
57	>	CP-DATA	Contains RP-ACK RPDU.
58	<	CP-ACK	
59	<	RRC CONNECTION RELEASE	
60	>	RRC CONNECTION RELEASE	
		COMPLETE	
61	SS		Prompts the operator to display the Short Messages
			stored in the UE. Only the Short Messages delivered in
			step 7, 19, 31 and 55 shall be retrievable and displayed

Specific Message Contents

SMS-DELIVER TPDU

Information element	Comment Value
TP-MMS	no more messages are waiting in SC "1"B
TP-PID	binary 01000xxx, xxx represents n resp. m (see test
	method description)

16.2.7.5 Test requirements

After step 60 only the Short Messages delivered in step 7, 19, 31 and 55 shall be retrieved and displayed.

16.2.8 Test of the reply path scheme

16.2.8.1 Definition

16.2.8.2 Conformance requirement

When a replying UE receives an original mobile terminated short message it has:

- originating SME = TP-Originating Address in the SMS-DELIVER TPDU;
- original SC = RP-Originating Address in the RP-MT-DATA.

When submitting the reply mobile originated short message, the replying UE should use parameters as follows:

- TP-Destination Address in SMS-SUBMIT TPDU = originating SME;
- RP-Destination Address in RP-MO-DATA = original SC.

Reference(s)

3GPP TS 23.040 Annex D.5, D.6

16.2.8.3 Test purpose

This procedure verifies that the UE is able to send a Reply Short Message back to the correct originating SME even if in the meantime it receives another Short Message.

16.2.8.4 Method of test

Initial conditions

- System Simulator:
 - 1 cell, default parameters.
- User Equipment:
 - the UE shall be in the idle updated state GMM-state "GMM-REGISTERED";
 - the UE message store shall be empty.

Related ICS/IXIT Statements

Support for Short message MT/PP.

Support for Short message MO/PP.

Description of the basic procedures to display a mobile terminated short message.

Description of the basic procedures to send a mobile originated short message.

The value of timer TC1M.

Test procedure

- a) The SS delivers a Short Message as specified in sub-clause 16.2.1, step b) with TP-Reply-Path set to 1.
- b) Step a) is repeated but with:
 - different TP-Originating-Address for the originating SME;
 - different RP-Originating-Address for the original SC; and
 - different message contents TP-User-Data.
- c) One of the two Short Messages is displayed (e.g. by means of the MMI) and the Reply Short Message is submitted (e.g. by means of the MMI).
- d) step c) is repeated for the other Short Message.

Expected sequence

Step	Dire	ction	Message	Comments
	UE	SS		
1			Mobile terminated establishment	See <u>3GPP</u> TS34.108
			of Radio Resource Connection	
2		>	SERVICE REQUEST	
3	<		AUTHENTICATION AND	
			CIPHERING REQUEST	
4		>	AUTHENTICATION AND	
			CIPHERING RESPONSE	
5	<		SECURITY MODE COMMAND	
6		>	SECURITY MODE COMPLETE	
7	<		CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU) TP-
				RP set to 1
8		>	CP-ACK	Sent within TC1M after step 7

Step	Direction UE SS	Message	Comments
9	>	CP-DATA	Contains RP-ACK RPDU.
10	<	CP-ACK	
11	<	RRC CONNECTION RELEASE	
12	>	RRC CONNECTION RELEASE	
		COMPLETE	
13		Mobile terminated establishment	See <u>3GPP</u> TS34.108
		of Radio Resource Connection	
14	>	SERVICE REQUEST	
15	<	AUTHENTICATION AND	
40		CIPHERING REQUEST	
16	>	AUTHENTICATION AND	
17		CIPHERING RESPONSE SECURITY MODE COMMAND	
18	< >	SECURITY MODE COMPLETE	
19	<	CP-DATA	Contains RP-DATA RPDU (SMS DELIVER TPDU) TP-
13		OI -DATA	OA, RP-OA and TP-UD different from step 7
20	>	CP-ACK	Sent within TC1M after step 7
21	>	CP-DATA	Contains RP-ACK RPDU.
22	<	CP-ACK	
23	<	RRC CONNECTION RELEASE	
24	>	RRC CONNECTION RELEASE	
		COMPLETE	
25	UE		One of the two Short Messages is displayed and the
			Reply Short Message is submitted.
26	<	SYSTEM INFORMATION	BCCH
27	>	RRC CONNECTION REQUEST	CCCH
28	<	RRC CONNECTION SETUP	CCCH
29	>	RRC CONNECTION SETUP	DCCH
20		COMPLETE	
30 31	> <	SERVICE REQUEST AUTHENTICATION AND	
31	<	CIPHERING REQUEST	
32	>	AUTHENTICATION AND	
32		CIPHERING RESPONSE	
33	<	SECURITY MODE COMMAND	
34	>	SECURITY MODE COMPLETE	
35	>	CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU) RP-DA
			= RP-OA corresponding to the message displayed TP-
			DA = TP-OA corresponding to the message displayed
36	<	CP-ACK	Sent within TC1M after step 35
37	<	CP-DATA	Contains RP-ACK RPDU
38	SS	OD AOK	Waits max 25 seconds for CP-ACK
39	>	CP-ACK	DDC composition is relatived.
40 41	<	RRC CONNECTION RELEASE RRC CONNECTION RELEASE	RRC connection is released.
41	>	COMPLETE	
42	UE	OOWIF LETE	The other Short Message is displayed and the Reply
74			Short Message is submitted.
43	<	SYSTEM INFORMATION	BCCH
44	>	RRC CONNECTION REQUEST	CCCH
45	<	RRC CONNECTION SETUP	CCCH
46	>	RRC CONNECTION SETUP	DCCH
		COMPLETE	
47	>	SERVICE REQUEST	
48	<	AUTHENTICATION AND	
		CIPHERING REQUEST	
49	>	AUTHENTICATION AND	
E 0	_	CIPHERING RESPONSE	
50 51	<	SECURITY MODE COMPLETE	
51 52	> >	SECURITY MODE COMPLETE CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU) RP-DA
52		OI PAIA	= RP-OA corresponding to the Message displayed TP-
			DA = TP-OA corresponding to the message displayed
53	<	CP-ACK	Sent within TC1M after step 52
, 55	1	1	1

Step	Direction	Message	Comments
	UE SS		
54	<	CP-DATA	Contains RP-ACK RPDU
55	SS		Waits max 25 seconds for CP-ACK
56	>	CP-ACK	
57	<	RRC CONNECTION RELEASE	RRC connection is released.
58	>	RRC CONNECTION RELEASE	
		COMPLETE	

Specific Message Contents

SMS-DELIVER TPDU

Information element	Comment Value
TP-MMS	no more messages are waiting in SC "1"B
TP-RP	Reply Path exists "1"B

16.2.8.5 Test requirements

After step 34 UE shall send the Reply Short Message corresponding to one of two previously received short messages.

After step 51 UE shall send the Reply Short Message corresponding to the other of two previously received short messages.

16.2.9 Multiple SMS mobile originated

16.2.9.1 UE in idle mode

This test applies to UE supporting the ability of sending multiple short messages on the same RRC connection when there is no PDP context in progress.

16.2.9.1.1 Definition

16.2.9.1.2 Conformance requirements

When the UE chooses to use the same RRC connection to send another short message or a memory available notification, then:

- the UE shall transmit a SERVICE REQUEST for the new CM connection before the final CP-ACK (e.g. the one that acknowledges the CP-DATA that carried the RP-ACK) for the old GMM context is transmitted;
- before transmission of the first CP-DATA on the new GMM context, the UE shall transmit the CP-ACK for the old GMM context;
- the Transaction Identifier used on the new GMM context shall be different to that used on the old GMM context; and
- the UE shall not initiate establishment of the new GMM context before the final CP-DATA (e.g. the one carrying the RP-ACK) has been received.

Reference

- 3GPP TS 23.040; sub-clause 3.1.
- <u>3GPP</u>TS 24.011; sub-clause 5.4.

16.2.9.1.3 Test purpose

To verify that the UE is able to correctly send multiple short messages on the same RRC connection when using an DCCH.

16.2.9.1.4 Method of test

Initial conditions

- System simulator:
 - 1 cell, default parameters.
- User Equipment:
 - the UE shall be in "Idle, updated" state GMM-state "GMM-REGISTERED";
 - the SMS message storage shall be empty.

Related ICS/IXIT statements

Support for multiple short message MO/PP on the same RRC connection.

Description of how to enter multiple SMS.

Description of the basic procedures to display a mobile originated short message.

Whether SMS messages are stored in the USIM and/or the ME.

Test procedure

- a) The UE shall be set up to send 3 short messages as multiple SM to the SS. The SS answers correctly to RRC CONNECTION REQUEST on CCCH and then performs the authentication.
- b) After receiving SECURITY MODE COMMAND UE shall send SECURITY COMMAND COMPLETE.
- c) The SS responds to the CP-DATA containing RP-DATA RPDU (SMS SUBMIT TPDU) from the UE with a CP-ACK message followed by a CP-DATA message containing the correct RP-ACK RPDU. The Transaction Identifier used on this MM connection is 'x'.
- d) The UE shall transmit a SERVICE REQUEST for the new CM connection (for the second short message) before the final CP-ACK (the one that acknowledges the CP-DATA that carried the RP-ACK before) for the old GMM context is transmitted. The UE shall not initiate establishment of the new GMM context before the final CP-DATA (i.e. the one carrying the RP-ACK for the first short message) has been received. Before transmission of the first CP-DATA on the new GMM context, the UE shall transmit the CP-ACK for the old GMM context. The Transaction Identifier used on the new GMM context shall be y, where y <> x (see procedure c)).
- e) The SS waits a maximum of 5 seconds after receiving the SERVICE REQUEST for the CP-ACK message from the UE.
- f) The SS responds to the CP-DATA containing RP-DATA RPDU (SMS SUBMIT TPDU) from the UE with a CP-ACK message followed by a CP-DATA message containing the correct RP-ACK RPDU.
- g) The UE shall transmit a SERVICE REQUEST for the new CM connection (for the third short message) before the final CP-ACK (the one that acknowledges the CP-DATA that carried the RP-ACK before) for the old GMM context is transmitted. Before transmission of the first CP-DATA on the new GMM context, the UE shall transmit the CP-ACK for the old GMM context. The Transaction Identifier used on the new GMM context shall be z, where z <> y (see procedure d)). The UE shall not initiate establishment of the new GMM context before the final CP-DATA (i.e. the one carrying the RP-ACK for the second short message) has been received.
- h) The SS waits a maximum of 5 seconds after receiving the SERVICE REQUEST for the CP-ACK message from the UE.

- i) The SS responds to the CP-DATA containing RP-DATA RPDU (SMS SUBMIT TPDU) from the UE with a CP-ACK message followed by a CP-DATA message containing the correct RP-ACK RPDU.
- j) The SS waits a maximum of 5 seconds after sending CP-DATA for the CP-ACK message from the UE.
- k) The SS sends a RRC CONNECTION RELEASE to the UE.

Expected sequence

Step	Direction	Message	Comments
	UE SS		
1	<	SYSTEM INFORMATION	BCCH
2	>	RRC CONNECTION REQUEST	CCCH
3	<	RRC CONNECTION SETUP	CCCH
4	>	RRC CONNECTION SETUP	DCCH
		COMPLETE	
5	>	SERVICE REQUEST	
6	<	AUTHENTICATION AND	
		CIPHERING REQUEST	
7	>	AUTHENTICATION AND	
		CIPHERING RESPONSE	
8	<	SECURITY MODE COMMAND	
9	>	SECURITY MODE COMPLETE	
10	>	CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU). The
			Transaction Identifier used in steps 10, 11, 12 and 14
			shall be x.
11	<	CP-ACK	
12	<	CP-DATA	Contains RP-ACK RPDU
13	>	SERVICE REQUEST	
14	>	CP-ACK	Shall be sent within 5 seconds of step 13
15	<	SERVICE ACCEPT	
16	>	CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU). The
			Transaction Identifier used in steps 16, 17, 18 and 20
			shall be y where y <> x (see step 10).
17	<	CP-ACK	
18	<	CP-DATA	Contains RP-ACK RPDU
19	>	SERVICE REQUEST	
20	>	CP-ACK	Shall be sent within 5 seconds of step 19
21	<	SERVICE ACCEPT	O A STANDARD CHIRANT TODAY
22	>	CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU). The
			Transaction Identifier used in steps 22, 23, 24 and 25
22		CD ACK	shall be z, where z <> y (see step 16).
23	<	CP-ACK	Contains BD ACK BDDU
24	<	CP-DATA CP-ACK	Contains RP-ACK RPDU
25 26	>	RRC CONNECTION RELEASE	Shall be sent within 5 seconds of step 24 RRC connection is released.
26	<	RRC CONNECTION RELEASE	NNO CONTRECTION IS TELEASEU.
21	>		
		COMPLETE	

16.2.9.1.5 Test requirements

In step 12 the UE shall transmit a SERVICE REQUEST for the new CM connection (for the second short message) before the final CP-ACK for the old GMM context is transmitted.

In step 17 the UE shall transmit a SERVICE REQUEST for the new CM connection (for the third short message) before the final CP-ACK for the old GMM context is transmitted.

16.2.9.2 UE in active mode

This test applies to UE supporting the ability of sending concatenated multiple short messages when there is a PDP context in progress.

16.2.9.2.1 Definition

16.2.9.2.2 Conformance requirements

When the UE chooses to use the same RRC connection to send another short message or a memory available notification, then:

- the UE shall transmit a SERVICE REQUEST for the new CM connection before the final CP-ACK (e.g. the one that acknowledges the CP-DATA that carried the RP-ACK) for the old GMM context is transmitted;
- before transmission of the first CP-DATA on the new GMM context, the UE shall transmit the CP-ACK for the old GMM context;
- the Transaction Identifier used on the new GMM context shall be different to that used on the old GMM context; and
- the UE shall not initiate establishment of the new GMM context before the final CP-DATA (e.g. the one carrying the RP-ACK) has been received.

Reference

- 3GPP TS 23.040; sub-clause 3.1.
- <u>3GPP</u> TS 24.011; sub-clause 5.4.

16.2.9.2.3 Test purpose

To verify that the UE is able to correctly concatenate multiple short messages on the same RRC connection when sent parallel to a PDP context.

16.2.9.2.4 Method of test

Initial conditions

- System simulator:
 - 1 cell, default parameters.
- User Equipment:
 - the UE shall be in "Idle, updated" state GMM-state "GMM-REGISTERED";
 - the SMS message storage shall be empty.

Related ICS/IXIT statements

Support for multiple short message MO/PP on the same RRC connection.

Description of how to enter multiple SMS.

Description of the basic procedures to display a mobile originated short message.

Support for state PDP-ACTIVATE of session management.

Whether SMS messages are stored in the USIM and/or the ME.

Test procedure

- a) a)—A PDP context is established with the SS and the state PDP-ACTIVE of session management is entered. The UE is set up to send 3 short messages as multiple SM to the SS. After the reception of the SERVICE REQUEST, the SS sends a SERVICE ACCEPT message.
- b) Steps c) to k) of the test procedure in sub-clause 16.2.9.1.4 are repeated.

Expected sequence

Step	Direction	Message	Comments
	UE SS	_	
1 2	SS UE		A PDP context is established and the state PDP-ACTIVE of session management is entered. The UE is set up to send 3 short messages as multiple SM
3	>	SERVICE REQUEST	
4	<	SERVICE ACCEPT	
5	>	CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 5, 6, 7 and 9 shall be x.
6	<	CP-ACK	
7	<	CP-DATA	Contains RP-ACK RPDU
8	>	SERVICE REQUEST	
9	>	CP-ACK	Shall be sent within 5 seconds of step 8
10	<	SERVICE ACCEPT	
11	> <	CP-ACK	Contains RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 11, 12, 13 and 15 shall be y where y <> x (see step 5).
13	<	CP-DATA	Contains RP-ACK RPDU
14	>	SERVICE REQUEST	Contains IXI -ACIX IXI DO
15	>	CP-ACK	Shall be sent within 5 seconds of step 14
16	<	SERVICE ACCEPT	Chair so continuin o cocontac or ctop 1.
17	>	CP-DATA	Contains RP-DATA RPDU (SMS SUBMIT TPDU). The Transaction Identifier used in steps 17, 18, 19 and 20 shall be z, where z <> y (see step 11).
18	<	CP-ACK	
19	<	CP-DATA	Contains RP-ACK RPDU
20	>	CP-ACK	Shall be sent within 5 seconds of step 19
21	<	RRC CONNECTION RELEASE	RRC connection is released.
22	>	RRC CONNECTION RELEASE COMPLETE	

16.2.9.2.5 Test requirements

In step 7 the UE shall transmit a SERVICE REQUEST for the new CM connection (for the second short message) before the final CP-ACK for the old GMM context is transmitted.

In step 13 the UE shall transmit a SERVICE REQUEST for the new CM connection (for the third short message) before the final CP-ACK for the old GMM context is transmitted.

16.3 Short message service cell broadcast

16.3.1 Definition

16.3.2 Conformance requirements

In idle mode, the UE listens to the BCCH and to the paging sub-channel for the paging group it belongs to. The UE is required to receive and analyse the paging messages and immediate assignment messages sent on the paging subchannel corresponding to its paging subgroup.

Reference

- <u>3GPP</u> TS 23.041; clause 8.
- <u>3GPP</u>TS 25.324; clause 11

16.3.3 Test purpose

This test verifies that an UE supporting SMS-CB is able to receive SMS-CB messages.

16.3.4 Method of test

Initial conditions

- System Simulator:
 - 1 cell, default parameters;
 - the SS provides a BCCH/CCCH to support the UE in idle mode;
 - periodic location updating is disabled.
- User Equipment:
 - the UE shall be in the idle updated state.

Related ICS/IXIT Statements

Support for short message transmission cell broadcast.

Description of the basic procedures to display a cell broadcasted short message.

Test procedure

Three Cell Broadcast (CB) messages are sent by the SS on the CBCH with message codes 0,1,1 in serial number fields respectively.

The UE shall respond to the page.

Expected sequence

Since the SMS-CB messages are sent continuously, a table is not applicable in this test.

Specific Message Contents:

Cell broadcast test message content

Information element	Comment Value
Message Type	CBS Message "1"B (see <u>3GPP</u> TS 25.324, sub-clause
	11.1)
Message ID	
Serial Number	
- Geographical scope	"00"B
- Message code	see test procedure "000000000"B or "000000001"B
- Update number	as applicable
Data Coding Scheme	Default alphabet, English "00000001"B
CB Data	max 1246 octets

16.3.5 Test requirements

In consequence of test the UE shall ignore third message and store two messages.

16.4 Default message contents:

16.4.1 Default message contents for SM-CP protocol

CP-DATA

Protocol Discriminator	SMS messages ("1001"B)
Transaction Identifier	
TIO	any value from the set {0,, 6}
TI flag	0
Message type	00000001
CP-User data	
length indicator	
RPĎU	max 248 octets

CP-ACK

Protocol Discriminator	SMS messages ("1001"B)
Transaction Identifier	
TIO	
TI flag	
Message type	00000100

CP-ERROR

Protocol Discriminator	SMS messages ("1001"B)
Transaction Identifier	
TIO	
TI flag	
Message type	00010000
CP-Cause	
Cause value	see <u>3GPP</u> TS 24.011, sub-clause 8.1.4.2

16.4.2 Default message contents for SM-RP protocol

RP-DATA

Information element	Comment Value
RP-Message Type	"001"B (SS->UE) or "000"B(UE->SS)
RP-Message Reference	see <u>3GPP</u> TS 24.011, sub-clause 8.2.3
RP-Originator Address	see <u>3GPP</u> TS 24.011, sub-clause 8.2.5.1
RP-Destination Address	see <u>3GPP</u> TS 24.011, sub-clause 8.2.5.2
RP-User Data	see <u>3GPP</u> TS 24.011, sub-clause 8.2.5.3
Length indicator	
TP-ĎATA	max 233 octets

RP-ACK

Information element	Comment Value
RP-Message Type	"010"B (UE->SS) or "011"B(SS->UE)
RP-Message Reference	see <u>3GPP</u> TS 24.011, sub-clause 8.2.3
RP-User Data	see 3GPP TS 24.011, subclause 8.2.5.3 : optional, may
	be present or not
RP-User Data IEI	"1000001"B
Length indicator	
TP-Data	max 232 octets

RP-ERROR

Information element	Comment Value
RP-Message Type	"100"B (UE->SS) or "101"B(SS->UE)
RP-Message Reference	see 3GPP TS 24.011, sub-clause 8.2.3
RP-Cause	see <u>3GPP</u> TS 24.011, sub-clause 8.2.5.4
RP-User Data	see 3GPP TS 24.011, sub-clause 8.2.5.3: optional, may
	be present or not
RP-User Data IEI	"1000001"B
Length indicator	
TP-Data	max 232 octets

RP-SMMA UE->SS)

Information element	Comment Value					
RP-Message Type	"110"B (UE->SS)					
RP-Message Reference	see <u>3GPP</u> TS 24.011, sub-clause 8.2.3					

16.4.3 Default message contents for SM-TP protocol

SMS DELIVER TPDU

Information element	Comment Value
TP-MTI	SMS DELIVER "00"B
TP-MMS	more messages are waiting in SC "0"B
TP-RP	no reply path "0"B
TP-UDHI	TP-UD contains only the SM"0"B
TP-SRI	no status report returned"0"B
TP-OA	an international number coded E.164
TP-PID	default "00000000"B
TP-DCS	default alphabet "00000000"B
TP-SCTS	any legal value (cf. 3GPP TS 23.040)
TP-UDL	
TP-UD	max 140 octets

SMS SUBMIT TPDU

Information element	Comment Value
TP-MTI	SMS SUBMIT"01"B
TP-RD	SC shall accept same SMS-SUBMIT "0"B
TP-VPF	TP-VP field not present "00"B
TP-RP	no reply path "0"B
TP-UDHI	TP-UD contains only the SM "00"B
TP-SRR	no request of status report "00"B
TP-MR	
TP-DA	an international number coded E164
TP-PID	default "00000000"B
TP-DCS	default alphabet "00000000"B
TP-VP	·
TP-UDL	
TP-UD	max 140 octets

SMS COMMAND TPDU

Information element	Comment Value
TP-MTI	SMS-COMMAND "10"B
TP-UDHI	TP-UD contains only the SM "00"B
TP-SRR	status report not requested "0"B
TP-MR	
TP-PID	default "00000000"B
TP-CT	
TP-MN	
TP-DA	an international number coded E164
TP-CDL	
TP-CD	

SMS STATUS REPORT TPDU

Information element	Comment Value
TP-MTI	SMS-STATUS-REPORT"10"B
TP-MMS	no more messages "1"B
TP-SRQ	result of SMS-SUBMIT "0"B
TP-MR	
TP-RA	the destination address of the previous SM MO
TP-SCTS	any legal value (cf. <u>3GPP</u> TS 23.040, sub-clause 9.2.3.11)
TP-DT	any legal value (cf. <u>3GPP</u> TS 23.040, sub-clause 9.2.3.13)
TP-ST	see <u>3GPP</u> TS 23.040, sub-clause 9.2.3.15

3GPP TSG-T1 Meeting #10 Copenhagen, Denmark, 8-9 February, 2001

				(CHAN	NGE	R	EQ	UE	ST	•					CR-Form-v3
*	3	34.1	23-1	CR	052		ж	rev	-	ж	Currer	nt vers	sion:	3.2	.0	¥
For <u>HEL</u>	<u>.Р</u> on u	ısing	this for	m, see	e bottom	of this	s pag	ge or	look	at th	e pop-u	p text	over	the ¥	syr	nbols.
Proposed c	hange	affec	ts: ¥	(U)	SIM	ME	/UE	X	Rad	io Ac	cess N	etwor	k	Core	e Ne	etwork
Title:	ж	An	nex B:	Updat	te of vers	sions c	of co	re sp	ecific	ation	ns					
Source:	ж	T1/	'Sig													
Work item o	ode: ೫										Da	ıte: ૠ	02/	02/20	01	
Category:	ж	F									Relea	se: ೫	R99	9		
		Deta	F (ess A (cor B (Add C (Fur D (Edi iled exp	ential or respon dition or nctional torial m	owing cate correction, ds to a co f feature), I modifica nodificatio ons of the TR 21.900	orrection tion of an) above	n in a	ıre)		elease	2 e) R! R! R! R!	<u>one</u> of 96 97 98 99 EL-4 EL-5	(GSM (Rele (Rele (Rele (Rele	llowing A Phase Pase 19 Pase 19 Pase 19 Pase 4)	se 2) 996) 997) 998) 999)	eases:
					. 00											
Reason for	change	e: #	Upda	ate due	e to CRs	on tes	st ca	ise cl	ause	S						
Summary o	f chang	ge: ૠ			ersion nu of Annex				on his	story						
Consequent not approve		ж	Diffic	cult to t	track whi	ch cor	re sp	ecific	cation	vers	sions ar	e test	ed			
Clauses off	4 - d:	مه	Anne	v D												
Clauses affe	ectea:	Ж	Anne	X D												
Other specs affected:	5	ж	Te	est spe	ore speci ecification pecification	ns	ns	Ħ								
Other comn	nents:	ж														

How to create CRs using this form:

Comprehensive information and tips about how to create CRs can be found at: http://www.3gpp.org/3G Specs/CRs.htm.

Below is a brief summary:

- 1) Fill out the above form. The symbols above marked # contain pop-up help information about the field that they are closest to.
- 2) Obtain the latest version for the release of the specification to which the change is proposed. Use the MS Word "revision marks" feature (also known as "track changes") when making the changes. All 3GPP specifications can be downloaded from the 3GPP server under ftp://www.3gpp.org/specs/ For the latest version, look for the directory name with the latest date e.g. 2000-09 contains the specifications resulting from the September 2000 TSG meetings.
- 3) With "track changes" disabled, paste the entire CR form (use CTRL-A to select it) into the specification just in front of the clause containing the first piece of changed text. Delete those parts of the specification which are not relevant to the change request.

Annex B (informative): Core specification versions to which test cases relate

The table B/1 lists for each section of this specification the related core specification version on which the test cases were based. Where the test cases have been partially updated towards the next released version, but this work has not completed yet, each change request considered is listed in the final column.

Section	Section heading	Related core	Current version	Current change requests
number	_	specifications	supported	taken into account
6	Idle mode operations	25.304	3. <u>5</u> 3.0 (2000-06)	
		23.122	3. <u>5</u> 3.0 (2000-06)	
		31.102	3.3.0 (2000-10)	
		25.331	3.3.0 (2000-06)	
		25.133	3. <u>42</u> .0 (2000-06)	
		25.214	3.3.0 (2000-06)	
		25.101	3. <u>5</u> 3. <u>0</u> 4 (2000-	
			06)	
		GSM 05.08	8.5.0 (2000-07)	
7.1	MAC	25.321	3.2.0	
7.2	RLC	25.322	3.4.0	
7.3	PDCP	25.323	3.2.0	
7.4	BMC	25.324	3.1.0	
8	Radio Resource Control (RRC)	25.331	3. <u>5.0</u> 4.1	
		04.18	9.0.0	
9	Elementary procedures of mobility management	24.008		
10	Circuit Switched Call Control (CC)	24.008	3.5.0	
11	Session Management Procedures	24.008	3.5.0	
12	Elementary procedure for Packet Switched Mobility Management	24.008	3.5.0	
13	General Tests	24.008	3.3.0	
14	Radio Bearer Services	N/A		
15	Supplementary Services	N/A		
16	Short message service (SMS)	23.040	3.5.0	
		23.041	3.3.0	
		24.011	3. <u>5</u> 4.0	
17	User Equipment features (MMI, VHE, MexE, SAT)			

Annex B-C (informative): Change history

3GPP TSG-T1 Meeting #10 Copenhagen, Denmark, 8-9 February, 2001

3GPP TSG-T1/SIG Meeting #15 Copenhagen, Denmark, 5-7 February 2001

T1S-010019

	CHANGE REQUEST
3 2	# Current version: 3.2.0 #
For <u>HELP</u> on us	ing this form, see bottom of this page or look at the pop-up text over the \mathbb{K} symbols.
Proposed change a	ffects: # (U)SIM ME/UE X Radio Access Network Core Network
Title: 第	Updates to clause 8 of TS 34.123-1 version 3.2.0
Source: #	Matsushita Communication Industries (MCI)
Work item code: ₩	Date: # 5 February 2001
Category:	F Release: Release: Re
I k	Use one of the following categories: F (essential correction) A (corresponds to a correction in an earlier release) B (Addition of feature), C (Functional modification of feature) D (Editorial modification) Cetailed explanations of the above categories can be found in 3GPP TR 21.900. Use one of the following releases: 2 (GSM Phase 2) R96 (Release 1996) R97 (Release 1997) R98 (Release 1998) R99 (Release 1999) REL-4 (Release 4) REL-5 (Release 5)
Reason for change:	To update RRC test cases for terminal conformance testing, so as to be compatible with the latest version of RRC core specification (TS 25.331 version 3.5.0)
Summary of change	cell environment specified in TS 34.108 clause 6.1, tables are added into "test procedures" sub-clauses of measurement test cases to clearly state the downlink power of each cell at various stages of test execution. As such, the "method of test" and "test requirements" clauses found in measurement test cases are revised. This CR provides an update to the existing RRC test cases found in clause 8 of TS 34.123-1 version 3.2.0, in accordance to the following CRs to RRC core specification (TS 25.331 version 3.4.1) approved during RAN2 #16 and RAN2#17 meetings. 1. CR-544r1: IE "CFN-SFN observed time difference" in IE "Cell measured results" is renamed to "Cell synchronisation information", while the same IE in IE "Cell reporting quantities" is renamed to "Cell synchronisation information reporting indicator". At the same time, all IEs in IE "Cell reporting quantities" are appended with "reporting indicator" to explicitly show that these are flags
	for reporting quantities, rather than the reported quantities themselves. 2. CR-545r1: Sub-IEs in IE "Downlink information common for all radio links" are revised. IE "Timing Indication" is mandatory and IE "CFN-targetSFN frame offset" is absent if IE "Timing Indication" is set to "Maintain".

3. CR-548r1: Establishment cause and paging cause are revised.

- The enumeration of Paging cause is removed.
- The expression as "PS data call" or "CS data call" are revised. Clause 8.1.2.5, 8.1.2.7.
- Paging cause for SMS is revised as "Low Priority Signalling".
- 4. CR-549: A possible deadlock can occur when all cells in the active set are forbidden to affect the reporting range. It was decided in RAN2 that the UE shall continue to update the reporting range in this case. This scenario was not previously considered in TS 34.123-1. A new test case is added to check for the UE's conformance in this aspect. Clause 8.4.1.14 is added.
- 5.CR-571r3: RAN2 has specified some performance requirements for UE processing delay. The proposed changes related to CR-571r3 had to be discussed with T1-SIG members, as it was felt that the additional requirements for response time of uplink messages are highly performance related in nature.
- 6.5. CR-573r1: "Flow id" concept is removed. Message routing will be performed using the CN domain identity. The content of clause 8.1.9 Signaling Connection Release Request is revised. UPLINK DIRECT TRANSFER and INITIAL DIRECT TRANSFER messages have been revised. SIGNALLING CONNECTION RELEASE REQUEST message is moved from clause 9 in TS34.108 to clause 8 in TS34.123-1.
- 7.6. CR-587r2: A series of changes are carried out in accordance to the modifications introduced in CR-587r1. They are listed below:
 - IE "Measurement identity number" in MEASUREMENT CONTROL and MEASUREMENT REPORT messages is renamed to "Measurement identity" in all specific message contents of clause 8.4.X.
 - The MEASUREMENT CONTROL FAILURE messages in clauses 8.4.1.9 and 8.4.1.10 shall have identical values for IE "RRC transaction identifier", as that found in the respective MEASUREMENT CONTROL messages.
 - IE "Measured Results on RACH" is added as an optional IE to MEASUREMENT REPORT messages. The purpose of this addition is to allow the UE to append neighbouring cell measurements to uplink RACH messages, in the case of traffic volume measurement reporting in CELL_FACH state. Since traffic volume measurements are currently not being tested, this optional IE is set to "Not Present" in all MEASUREMENT REPORT messages.
 - The name of IE "Cell selection and re-selection info for SIB 11/12" is incorrect. It is corrected to "Cell selection and re-selection info".
 - IE "TGCFN" is added as a mandatory present IE into IEs "DPCH
 Compressed Mode Info" and "DPCH Compressed Mode Status Info". All
 affected messages in the specific message content sub-clauses are
 updated to reflect this addition.
 - IE "FACH measurement occasion info" has been re-structured. The corresponding IEs found in SIB type 11 or 12 in the specific message contents sub-clauses are updated.
 - Both "Inter-frequency cell info list" and "Intra-frequency cell info list" IEs
 are changed to allow the possibility of removing all previously stored interfrequency cells. The affected messages containing these IEs in the
 specific message content sub-clauses are updated.
 - It is proposed in RAN2 that event-triggered periodic reporting be disallowed for inter-frequency type measurements. Therefore, IEs "Amount of reporting" and "Reporting interval" are removed from IE "Inter-frequency reporting criteria". MEASUREMENT CONTROL messages

- containing theseis IEs are updated.
- IE "Inter-system measurement" and its sub-IEs found in specific message contents of clause 8.4.1.9.4 are revised.
- All IEs with name containing the phrase "inter-system ..." are renamed to "inter-RAT ..."
- In IE "Intra-frequency measurement reporting criteria", a new trigger condition (triggering condition 2) is defined on top of the existing condition (triggering condition 1). Affect messages are revised.
- It was decided in RAN2 that if "Measurement validity" IE is present in a
 message, the associated measurement task should be resumed in the
 state specified in "UE state" IE. Otherwise, if "Measurement validity" IE is
 absent in the message, the UE shall deleted the associated measurement
 task. Hence the sub-IE "resume/release" found in "Measurement validity"
 IE is no longer needed and deleted from all messages.
- IE "Reporting cell status" is rephrased and modified to include detected set. Affected messages containing this IE are revised.
- 8-7. CR-595r4: This CR states example of state-transition with RB control.

 According to this CR, two new taest_-cases, RB set up from CELL__DCH to CELL__PCH and RB setup from CELL__DCH to URA__PCH are introduced. Clause 8.2.2.25-19 and 8.2.2.26-20 are added. PHYSICAL_Physical Channel RECONFIGURATION Reconfiguration from CELL__FACH to CELL__PCH (clause 8.2.6.21), PHYSICAL_Physical Channel RECONFIGURATION Reconfiguration from CELL__FACH to URA__PCH (clause 8.2.6.22), Transport Channel rReconfiguration from CELL__FACH to CELL__PCH (clause 8.2.4.22), Transport Channel rReconfiguration from CELL__FACH to URA__PCH (clause 8.2.4.23) are removed.
- 9.8. CR-596r1: IE "References to other system information blocks" are removed from all instances of SIB messages in the specific message content subclauses. This is because RAN2 decided that only MIB and SB 1 or SB 2 contain system information scheduling information. Also, IE "SIB12 indicator" is inserted into SIB Type 11 messages. The value of this IE in SIB Type 11 is set to "FALSE" so that the UE will read only SIB Type 11 in connected state. IE "UE Timers and constants in connected mode" has been moved from SIB type 2 to SIB type 1.
- 10.9. CR-597r5: The following areas are revised due to this approved CR:
 - In RRC Connection Establishment procedure, clause 8.1.2.9 is added to test for physical channel failure condition.
 - All the definition and test cases in RRC connection re-establishment procedure are removed because these concept is moved to CELL UPDATE procedure.
 - Since RRC connection re-establishment procedure has been removed, test cases that use this procedure have been revised using cell-update procedure.
 - The possibility to assign PRACH and SCCPCH in UE dedicated messages for CELL_FACH, CELL_PCH and URA_PCH states has been removed since there is no lur support. A RB control message cannot assign IE Downlink information for each radio link and PRACH info for state transition to CELL FACH, CELL PCH, URA PCH. The test-cases with transition to CELL FACH, CELL PCH, URA PCH are revised. Clauses 8.2.1.8, 8.2.1.16, 8.2.2.8, 8.2.2.17, 8.2.3.7, 8.2.3.15, 8.2.4.7, 8.2.6.7 and 8.2.6.15.
 - IE "DRX Indicator" has been changed to IE "RRC State Indicator".
 - The variables of IE "RRC State Indicator" have been totally removedal

- and new variables have been added. "no DRX", "DRX with cell updating" and "DRX with URA updating" are removed. "CELL_DCH", "CELL_FACH", "CELL_PCH" and "URA_PCH" are added. All message content that contains IE "DRX Indicator" are revised.
- Clarifications of when procedures end, has caused many test cases to be corrected.
- Cell update due to RB control response has been removed, hence the related test case has been removed. Clause 8.3.1.14 is removed.
- Two new causes for cell update has been added. They are "radio link failure" and "RLC unrecoverable error". Therefore new test cases for these two causes have been added.
- T306 has been removed and all T306 timers have been replaced with T305 timers.
- In cell update, condition for invalid configuration has been added. Therefore, new clause 8.3.1.20 has been added.
- In active set update, the condition for unsupported configuration in the UE has been change and condition for invalid configuration has been added. Clause 8.3.4.4 has been change-revised to cater for this change.
- In active set update, the condition for subsequently received active set update messages and for incompatible simultaneous reconfiguration has been removed. Clause 8.3.4.6 has therefore been removed.
- IE "URA update cause" with value of "change of URA" has been changed to "URA reselection".
- T303 has been removed and all T303 timers have been replaced with T302 timers.
- V303 has been removed and all V303 timers constants have been replaced with V302 timers constant.
- N303 has been removed and all N303 timers constants have been replaced with N302 timers constants.
- In cell updating the UE shall responde to Cell Update ConfirmCELL
 <u>UPDATE CONFIRM</u> message on downlink CCCH with valid IE "U-RNTI"
 in the same way it respondes to message sent on downlink DCCH.

41.10. CR-598: The name of IE in "Paging record" is revised.

12.11. CR-600r3: The following areas are revised due to this approved CR:

- Invalid configuration condition has been added to all RB procedures, therefore new test procedures have been added to current test cases.
 Clauses 8.2.1.7, 8.2.1.15, 8.2.2.6, 8.2.2.15, 8.2.3.6, 8.2.3.14, 8.2.4.6, 8.2.4.15, 8.2.5.4, 8.2.6.6, 8.2.6.14 are affected.
- Cell re-selection condition has been added to all RB procedures. Test
 cases concerning failure to access the assigned physical channel during
 the transition from CELL_FACH to CELL_DCH have been revised.
 Clauses 8.2.1.9, 8.2.2.9, 8.2.2.18, 8.2.3.8, 8.2.4.9, 8.2.4.17, 8.2.6.8,
 8.2.6.16 are affected.
- Test cases concerning failure to revert to the old configuration after the
 detection of physical channel failure during the transition from CELL_DCH
 to CELL_FACH has been revised. Clauses 8.2.1.5, 8.2.1.13, 8.2.2.4,
 8.2.2.13, 8.2.3.4, 8.2.3.12, 8.2.4.4, 8.2.4.13, 8.2.6.4, 8.2.6.12 are affected.
- IE "suspend/resume" change to "stop/continue". Some changes in the word used are made. Test cases concerning the control of uplink transmission by RB reconfiguration messages have been revised.

Clauses 8.2.2.7, 8.2.2.16 are affected.

13.12. CR-602: Many IEs are revised.

- DRX indicator → RRC state indicator
- RRC transaction identifier is added.
- Critical spare values in downlink messages removed, conditions for creating invalid message reception test scenario have been revised.
 Clauses 8.1.1.7, 8.1.2.8, 8.1.3.5, 8.1.5.1, 8.1.5.4, 8.1.6.1, 8.1.6.2, 8.1.7.1, 8.1.7.2, 8.1.8.1, 8.1.8.2, 8.2.1.7, 8.2.1.15, 8.2.2.6, 8.2.2.15, 8.2.3.6, 8.2.3.14, 8.2.4.6, 8.2.4.15, 8.2.6.6, 8.2.6.14, 8.3.1.13 are affected.
- 14.13. CR-604r2: IE "FACH measurement occasion info" has been re-structured. The corresponding IEs found in SIB type 11 or 12 in the specific message contents sub-clauses are updated.
- 15.14. CR-641: Downlink outer loop control procedure has been removed. Hence, clause 8.2.9 has been removed.

The following errors are found and corrected:

- In many test cases, BCCH is stated as being broadcasted on primary CPICH, which is technically wrong. BCCH should be broadcast on primary CCPCH.
- In clause 8.3.1.X, the specific message contents of some CELL UPDATE CONFIRM messages contain an incorrect IE name - "Downlink DPCH info for one radio link". The correct name for this IE should be "Downlink information for each radio link".
- References for the initial condition in some of the test cases have been updated according to the new clause 7.4 in TS 34.108 version 3.2.0.
- All instances of "UL data transmission" for IE "Cell update cause" are renamed to "uplink data transmission". All instances of "Periodic cell update" for IE "Cell update cause" are renamed to "Periodical cell update".
- Clause 8.1.2.2 is revised such that two PRACHs are prepared in first step to let the UE randomly choose a PRACH.
- In the last meeting, it was decided that state 7 defined in clause 7.4 of TS 34.108 should be added in the initial condition of some test cases in RRC Connection Management.
- Clause 8.2.4.8 is an invalid test-case. PHYSICAL CHANNEL FAILURE can't be occurred when UE enter into CELL_-FACH state.
- In clause 8.1.1.1, UE is required to send RRC connection setup complete message to SS after step 5.
- Clause 8.4.1.10 The condition that triggers the UE to send MEASUREMENT CONTROL FAILURE message is revised. Specifically, an unknown arbitrary 32-bits string is append to MEASUREMENT CONTROL message in step 3.
- Routing info requires unque identity as suggested by ESTI. In clauses 8.1.8.1 and 8.1.8.2, RB identity has been revised.

Other corrections

(a) Due to Nokia's comments:

- "Step xx" is missing from some of the specific message content sections.
 For the purpose of clarity, they are added.
- Clause 8.4.1.2 Expected Sequence: An "arrow" is missing in step 4, RRC CONNECTION SETUP message.
- Pointed out by Nokia, test procedure in all the "Incompatible simultaneous reconfiguration" test cases should end with RB complete message. Clauses 8.2.1.6, 8.2.1.14, 8.2.2.5, 8.2.2.14, 8.2.3.5, 8.2.3.13, 8.2.4.5, 8.2.4.14, 8.2.6.5, 8.2.6.13 are affected.
- Following Nokia's comment, IE "Allowed Transport format combination" in TRANSPORT FORMAT COMBINATION CONTROL message has been revised. Clauses 8.2.5.1 and 8.2.5.2 are corrected.

This CR also provides an update to the existing RRC messages in Annex A of TS 34.123-1 version 3.2.0, in accordance to the following CRs to RRC core specification (TS 25.331 version 3.4.1) approved during RAN2 #16 and RAN2#17 meetings.

- 1. CR-542: "Logical channel max loss" IE is deleted.
- 2. CR-543r1: The missing IE "NAS Synchronisation Indicator" is added to IE "RAB info". The value of this IE is set to "Not Present".
- 3. CR-544r1: IE "CFN-SFN observed time difference" in IE "Cell measured results" is renamed to "Cell synchronisation information", while the same IE in IE "Cell reporting quantities" is renamed to "Cell synchronisation information reporting indicator". At the same time, all IEs in IE "Cell reporting quantities" are appended with "reporting indicator" to explicitly show that these are flags for reporting quantities, rather than the reported quantities themselves.
- 4. CR-545r1: IE "Timing indication" was previously absent in IE "Downlink DPCH info common for all RL", it is added and set to "Maintain". IE "CFN-targetSFN frame offset" is introduced as conditional (but Not Present) IE in IE "Downlink DPCH info common for all RL".
- 5. CR-548r1: The enumerated value "SMS" in IE "Paging cause" is changed to "Low Priority Signalling".
- 6. CR-554r1: The signalled range for IE "BLER Quality value" is changed. The current BLER quality value (0.00) is replaced by 10^{-6.3}.
- 7. CR-549: A "CHOICE report criteria" IE is added before "Periodic reporting criteria" IE in MEASUREMENT CONTROL message.
- 8. CR-587r1: A series of changes are carried out in accordance to the modifications introduced in CR-587r1. They are listed below:
 - IE "Measurement identity number" in MEASUREMENT CONTROL and MEASUREMENT REPORT messages is renamed to "Measurement identity".
 - "Inter-system measurement" IE is renamed to "Inter-RAT measurement".
 Also, "LCS measurement" IE is renamed to "UP measurement".
- 9. CR-596: IE "UE Timers and constants in connected mode" is added to UTRAN MOBILITY INFORMATION message.
- 10. CR-597r5: Several modifications are introduced as described below:

IE "DRX Indicator" was changed to IE "RRC State Indicator", with a set of different enumerated values.

Optional IEs "RB information to release list", "RB information to reconfigure list", "RB information to be affected list", "UL Transport channel information common for all transport channels", "Deleted UL TrCH information list", "Added or reconfigured TrCH UL information list", "DL Transport channel information common for all transport channels", "Deleted DL TrCH information list", "Added or reconfigured DL TrCH information list" and "Downlink information per radio link list" are added to CELL UPDATE CONFIRM message.

"Failure cause" and "RB timer indicator" IEs are added to CELL UPDATE message.

- RRC connection re-establishment procedure is deleted, hence all related messages are removed from Annex A.
- 11. CR-598r1: IE "Paging Record" in PAGING TYPE 1 messages is revised.
- 12. CR-602: The following changes are carried out according to this CR:

IE "RRC transaction identifier" is inserted into many messages

IE "DRx indicator" is replaced by IE "RRC state indicator" in many messages. The appropriate value for this IE is chosen, depending on the final state the UE will reside on.

13. CR-606: Several measures are approved by RAN2 in order to reduce the size of IEs "RLC info" and "RB mapping info" transmitted in the downlink. As a result of these optimisation efforts, the corresponding changes to the default message contents need to be carried out:

IE "Downlink RLC logical channel info" needs to be added to IE "RB mapping info". RADIO BEARER SETUP, RADIO BEARER RELEASE and RRC CONNECTION SETUP messages are affected.

IEs "RLC info" are modified to become IEs "CHOICE RLC info type" in RADIO BEARER SETUP message.

- 14. CR-609r1: Optional IE "COUNT-C activation time" is added to all uplink radio-bearer related complete messages (e.g. RADIO BEARER RECONFIGURATION COMPLETE, TRANSPORT CHANNEL RECONFIGURATION COMPLETE). The presence of this IE is conditional on whether the UE transits to CELL_DCH state and also existence of RLC-TM RB(s). Also, IE "Activation time for DPCH" in IE "Ciphering mode info" is renamed to "Ciphering activation time info".
- 15. CR-615r2: IE "CHOICE RLC size list" is added into IE "RB mapping info". All instances of this IE are set to "All" so that all the RLC sizes listed in the TFS can be used by the UE. Also, IE "Transport Format Set" is modified to include a list of logical channels so that configuration of RLC size for certain logical channels is made possible.
- 16. CR-625: An optional IE "RB with PDCP information list" is added into RADIO BEARER SETUP and RADIO BEARER RELEASE messages. These IEs are set to "Not Present" at the moment.
- 17. CR-629r1: An optional IE "Signalling Connection release indicator" is added into downlink RADIO BEARER RELEASE messages. This IE is set to "Not Present" as it's not desired to release a RB and a signalling connection simultaneously in the test cases.

Other error corrections include:

- 18. IEs "CN Domain Identity" and "NAS Information" are deleted from MEASUREMENT REPORT message. At the same time, IEs "Additional Measured Results" and "Event results" were mistakenly left out of the default MEASUREMENT REPORT message are now reinstated.
- 19. IE "CHOICE Measurement type" was mistakenly left out of the default MEASUREMENT CONTROL message. It is now reinserted, with all the sub-IEs set to "Not Present".
- 20.CR-XXX: IE "RRC transaction identity" is added into MEASUREMENT CONTROL and MEASUREMENT CONTROL FAILURE messages.

Technical error corrections:

- 21.20. RRC CONNECTION SETUP (Transition to CELL_FACH state): It was pointed out (by SAS on 10 Jan 01) that only RB#1 to RB#4 can be established by specifying IE "Signalling RB information to setup". Therefore, IE "Signalling RB information to setup" for SRBs on CCCH, BCCH and PCCH are deleted.
- 22.21. The values of Transport channel identity, Logical channel identity and Radio Bearer identity were changed because of the revision of the channel structure as proposed by ETSI in the draft TS 34.123-3 version 1.0.0 (T1S-01001).

Consequences if not approved:

Test specifications will not be compatible with core specifications.

Clauses affected: #

8.4.X, 8.4.1.14 (new), 8.4.1.9, 8.4.1.10

Other specs affected:

Cother core specifications

Test specifications

O&M Specifications

TS 34.108 version 3.2.0

Other comments: #

How to create CRs using this form:

Comprehensive information and tips about how to create CRs can be found at: http://www.3gpp.org/3G Specs/CRs.htm. Below is a brief summary:

- 1) Fill out the above form. The symbols above marked \(\mathcal{H} \) contain pop-up help information about the field that they are closest to.
- 2) Obtain the latest version for the release of the specification to which the change is proposed. Use the MS Word "revision marks" feature (also known as "track changes") when making the changes. All 3GPP specifications can be downloaded from the 3GPP server under ftp://www.3gpp.org/specs/ For the latest version, look for the directory name with the latest date e.g. 2000-09 contains the specifications resulting from the September 2000 TSG meetings.
- 3) With "track changes" disabled, paste the entire CR form (use CTRL-A to select it) into the specification just in front of the clause containing the first piece of changed text. Delete those parts of the specification which are not relevant to the change request.

8. Radio Resource Control RRC

8.1 RRC Connection Management Procedure

8.1.1 Paging

8.1.1.1 Paging for Connection in idle mode

8.1.1.1.1 Definition

8.1.1.1.2 Conformance requirement

In idle mode, UE monitors the paging occasions determined using parameters from SYSTEM INFORMATION BLOCK messages. When the UE receives a PAGING TYPE 1 message transmitted on PCCH during one of its assigned paging occasions, it should attempt to establish an RRC connection.

Reference

3GPP TS 25.331 clause 8.1.2, 3GPP TS 25.211 clause 5.3.3.7, 3GPP TS 25.304 clause 8.

8.1.1.1.3 Test purpose

To confirm that the UE establishes an RRC connection after it receives a PAGING TYPE 1 message which includes IE "Paging Record" (UE identity) set to the IMSI of the UE.

8.1.1.1.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: Idle state (state 2 or state 3 or state 7) as specified in clause 7.4 of TS 34.108-with a CN UE identity (set to IMSI), depending on the CN domain(s) supported by the UE.

Test Procedure

The SS transmits a PAGING TYPE 1 message, which includes an unmatched CN UE identity for the UE in the idle state. The UE shall not change its state. The SS transmits a PAGING TYPE 1 message, which includes a matched CN UE identity for the UE in the idle state. During transmission of PAGING TYPE 1 messages, SS selects the correct paging indicator on the PICH in order to allow the UE to respond to paging. Then the UE transmits an RRC CONNECTION REQUEST to the SS, the SS transmits an RRC CONNECTION SETUP to the UE. When the UE receives this message, the UE establishes an RRC connection and transmits an RRC CONNECTION SETUP COMPLETE message on the uplink DCCH.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1	*	-	SYSTEM INFORMATION BLOCK TYPE 13	Transmit these messages on the BCCH, in addition to the normal BCCH transmissions. See specific message contents.
2	•	-	PAGING TYPE 1	The SS transmits the message, which includes an unmatched identity (incorrect IMSI), and the UE does not change its state.
3	•	-	PAGING TYPE 1	The SS transmits the message, which includes a matched identity (test-SIM IMSI).
4	1)	RRC CONNECTION REQUEST	
5	•	_	RRC CONNECTION SETUP	SS assigns DPCH resources to allow UE to establish an RRC connection.
<u>6</u>	1	<u> </u>	RRC CONNECTION SETUP COMPLETE	

Specific Message Contents

SYSTEM INFORMATION TYPE 13

Information Element	Value/remark
CN domain system information list	Only 1 entry
CN domain system information	
- CN domain identity	Supported Domain (PS Domain or CS Domain)
- CHOICE CN Type	Supported CN type
 CN domain specific NAS system information 	Default
 CN domain specific DRX cycle length coefficient 	6
UE Timers and constants in idle mode	
- T300	6000 milliseconds
- N300	3
- T312	10 sec
- N312	200

PAGING TYPE 1 (Step 2)

Information Element	Value/remark
Paging record list	Only 1 entry
Paging record	
CHOICEUsed -Ppaging identityoriginator	CN <u>identity</u> originator
- Paging cause	Terminating Call with one of the supported services
	(Conversational Call, Streaming Call, Interactive Call,
	Background Call, SMS)
- CN domain identity	Supported Domain (PS Domain or CS Domain)
- CHOICE UE Identity	IMSI
- IMSI	Set to an arbitrary octet string of length 7 bytes which is
	different from the IMSI value stored in the USIM card.

PAGING TYPE 1 (Step 3)

Information Element	Value/remark
Paging record list	Only 1 entry
Paging record	
CHOICE <u>Used Ppaging identityoriginator</u>	CN identityoriginator
- Paging cause	Terminating Call with one of the supported services
	(Conversational Call, Streaming Call, Interactive Call,
	Background Call, SMS)
- CN domain identity	Supported Domain (PS Domain or CS Domain)
- CHOICE UE Identity	IMŚI
- IMSI	Set to the same octet string as in the IMSI stored in the
	USIM card

RRC CONNECTION REQUEST (Step 4)

Information Element	Value/remark
Initial UE identity	Same as the IMSI stored in the USIM card, or the
	registered TMSI or P-TMSI
Establishment Cause	Check to see if it is set to the same value as "Paging
	Cause" IE in the PAGING TYPE 1 message transmitted
Protocol Error Indicator	on step 3.
Measured results on RACH	Check to see if it is set to FALSE
	Not checked.

8.1.1.1.5 Test requirement

After step 2 the UE shall not transmit on the uplink CCCH in order to establish a RRC connection.

After step 5 the UE shall have an RRC connection based on dedicated physical channel resources <u>and transmit an RRC CONNECTION SETUP COMPLETE message</u> on the uplink <u>DCCH</u>.

8.1.1.2 Paging for Connection in connected mode (CELL_PCH)

8.1.1.2.1 Definition

8.1.1.2.2 Conformance requirement

In CELL_PCH state, a UE can respond to a paging request from UTRAN. In this case, the UTRAN has requested to establish a connection with the UE. The UE should then attempt to perform a cell update procedure and move to CELL_FACH state in order to respond to the paging using uplink CCCH.

Reference

3GPP TS 25.331 clause 8.1.2

8.1.1.2.3 Test purpose

To confirm that the UE enters the CELL_FACH state after it receives a PAGING TYPE 1 message which indicates that the paging has originated from UTRAN. To verify that the UE performs cell update procedure after entering the CELL_FACH state.

8.1.1.2.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CELL_PCH state (state 6-1512) as specified in clause 7.4 of TS 34.108 with a valid U-RNTI already assigned by the SS

Test Procedure

The SS transmits a PAGING TYPE 1 message, which includes an unmatched U-RNTI in CELL_PCH state. The UE does not change its state. Then SS transmits a PAGING TYPE 1 message with a matched identifier but originates from the CN instead of UTRAN. The UE should not change state after receiving this message. The SS transmits a PAGING TYPE 1 message, which includes a matched U-RNTI in the connected state. Then the UE enters the CELL_FACH state and performs the cell updating procedure.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	+	SYSTEM INFORMATION BLOCK TYPE 13	Transmit these messages on the BCCH, in addition to the normal BCCH transmissions. See specific message contents
2	+	PAGING TYPE 1	The SS transmits a message including an unmatched identifier. UE shall not respond to the paging.
3	+	PAGING TYPE 1	The SS transmits a message includes a matched identifier but with the originator being the CN, UE shall not respond to the paging.
4	+	PAGING TYPE 1	The SS transmits the message with the UTRAN being the originator and including the UE's assigned U-RNTI
5)	CELL UPDATE	The UE enters the CELL_FACH state. UE performs cell updating procedure. The CELL UPDATE message shall contain the value "Cell Update Cause" set to "paging response".
6	+	CELL UPDATE CONFIRM	Use the default message specified in Annex A.

Specific Message Contents

PAGING TYPE 1 (Step 2)

Information Element	Value/remark
Paging record list	Only 1 entry
Paging record	
 CHOICE <u>Used Ppaging identityoriginator</u> 	UTRAN <u>identity</u> originator
- U-RNTI	
- SRNC Identity	Set to an arbitrary 16-bit string which is different from
- S-RNTI	the SRNC identity assigned.
	Set to an arbitrary 20-bit string which is different from
	the S-RNTI assigned.

PAGING TYPE 1 (Step 3)

Same as the PAGING TYPE 1 message as in Clause 8.1.1.1.4, with the exception that the "BCCH modification info" IE should be omitted in the message.

PAGING TYPE 1 (Step 4)

Value/remark
Only 1 entry
UTRAN identityoriginator
Set to the same SRNC identity as previously assigned.
Set to the same S-RNTI as previously assigned.

SYSTEM INFORMATION BLOCK TYPE 13

Use the same SYSTEM INFORMATION BLOCK TYPE 13 message as specified in Clause 8.1.1.1.4.

8.1.1.2.5 Test requirement

After step 2 the UE shall not respond to the PAGING TYPE 1 message sent in step 2.

After step 3 the UE shall not respond to the PAGING TYPE 1 message sent in step 2.

After step 4 the UE shall enter the CELL FACH state and send a CELL UPDATE message with "Cell Update Cause" IE set to "paging response".

8.1.1.3 Paging for Connection in connected mode(URA_PCH)

8.1.1.3.1 Definition

8.1.1.3.2 Conformance requirement

This procedure is used to transmit a PAGING TYPE 1 message from the network to selected UEs in URA_PCH state using the paging control channel (PCCH). The UE listens to it and then enters the CELL_FACH state.

Reference

3GPP TS 25.331 clause 8.1.2

8.1.1.3.3 Test purpose

To confirm that the UE enters the CELL_FACH state after it receives a PAGING TYPE 1 message which includes IE "Paging Record" (U-RNTI) for the UE and which is set to "UTRAN originator" in IE "paging originator".

8.1.1.3.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: URA_PCH state (state 6-1613) as specified in clause 7.4 of TS 34.108 with a valid U-RNTI assigned by the SS.

Test Procedure

The SS transmits a PAGING TYPE 1 message, which includes an unmatched U-RNTI in URA_PCH state. The UE does not change its current state. The SS transmits a PAGING TYPE 1 message which includes a matched U-RNTI in the connected state. Then the UE listens to it and enters the CELL_FACH state.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1	*	-	PAGING TYPE 1	The SS transmits the message that includes an unmatched identifier, then the UE does not change its state.
2	+	-	PAGING TYPE 1	The SS transmits the message that includes a matched identifier.
3	7	•	CELL UPDATE	The UE enters the CELL_FACH state.

Specific Message Contents

PAGING TYPE 1 (Step 1)

Information Element	Value/remark
Paging record list	Only 1 entry
Paging record	
 CHOICE <u>PUsed paging identityoriginator</u> 	UTRAN identityoriginator
- U-RNTI	
- SRNC Identity	Set to an unused SRNC identity which is different from
·	the SRNC identity assigned.
- S-RNTI	Set to an arbitrary 20-bit string which is different from
	the S-RNTI assigned.
	, and the second

PAGING TYPE 1 (Step 2)

Information Element	Value/remark
Paging record list	Only 1 entry
Paging record	
 CHOICE <u>Used Ppaging identityoriginator</u> 	UTRAN identityoriginator
- U-RNTI	
- SRNC Identity	Set to the previously assigned SRNC identity
- S-RNTI	Set to previously assigned S-RNTI

8.1.1.3.5 Test requirement

After step 1 the UE shall not respond to the paging.

After step 2 the UE shall enter the CELL FACH state, and transmit CELL UPDATE message to initiate the cell updating procedure with the paging cause set to "paging response".

8.1.1.4 Paging for Notification in idle mode

8.1.1.4.1 Definition

8.1.1.4.2 Conformance requirement

When a system information block on the BCCH is modified, the PAGING TYPE 1 message can be sent on the PCCH to inform the UE about the changes, which are currently taking place in the idle mode. The PAGING TYPE 1 message includes the IE "BCCH Modification Information". Upon receiving this notification from the UTRAN, the UE shall read the relevant MIB and/or SIB(s) subsequently during idle mode.

Reference

3GPP TS 25.331 clause 8.1.1.2

8.1.1.4.3 Test purpose

To confirm that the UE checks the new value tag of the master information block and reads the updated SYSTEM INFORMATION BLOCK messages after it receives a PAGING TYPE 1 message which includes the IE "BCCH Modification Information".

8.1.1.4.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: Idle state (state 2 or state 3 or state 7) as specified in clause 7.4 of TS 34.108 with a CN UE identity, depending on the CN domain(s) supported by the UE.

Test Procedure

The UE is in the idle state before it starts to change the SYSTEM INFORMATION BLOCK TYPE 1 or 13 messages, depending on the CN type supported by the UE. The SS transmits a PAGING TYPE 1 message on the paging occasions assigned to the UE. The message shall include the IE "BCCH Modification Information" indicating the time when the first modified master information block is available. Before the starting time, SS continuously broadcast the original MASTER INFORMATION BLOCK and various types of SYSTEM INFORMATION BLOCK on the BCCH mapped to BCH transport channel. SS maintains this status until the SFN which corresponds to the starting time is reached. Then it transmits the new master information block followed by the new SYSTEM INFORMATION BLOCK TYPE 1 or 13 messages. In the new SIB TYPE 1 or 13 messages, the IE "DRX Cycle Length Coefficient" is altered when compared to the original SIB TYPE 1 or 13 messages. At the next paging occasion, SS transmits a new PAGING TYPE 1 message. The message addresses the UE using its IMSI and the "paging cause" IE set to a terminating call type that is supported by the UE. The UE shall react to the PAGING TYPE 1 message and then send a RRC CONNECTION REQUEST message to SS.

Notes: For UEs supporting GSM-MAP CN type only, SYSTEM INFORMATION TYPE 1 messages are to be sent by SS in this test case. On the other hand, SS transmits SYSTEM INFORMATION TYPE 13

messages if the UE under test supports only ANSI-41 CN type.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	÷	MASTER INFORMATION BLOCK SYSTEM INFORMATION BLOCK TYPE 1 or SYSTEM INFORMATION BLOCK TYPE 13	SS transmits information on BCCH in order for UE to listen to one of the S-CCPCH physical channel, which carries PAGING TYPE 1 or 13 messages on PCCH. Relevant paging parameters are also broadcasted.
2	+	PAGING TYPE 1	SS transmits the message includes the IE "BCCH Modification Information", with the "Value Tag" changed from the "MIB Value Tag" of the current Master Information Block. Also the modification time is set to 4088 radio frame from the current SFN. SS continuously broadcast the same MASTER INFORMATION BLOCK and various types of SYSTEM INFORMATION BLOCK on BCCH for a period stretching 4087 frames.
3	+	MASTER INFORMATION BLOCK	SS starts to transmit the MIB with the "MIB Value Tag" IE different from the original setting.
	←	SYSTEM INFORMATION BLOCK TYPE 1 or SYSTEM INFORMATION BLOCK TYPE 13	At the same time, SS starts to transmit the affected SIB TYPE 1 or TYPE 13 messages continuously. The value of IE "DR*X Cycle Length Coefficient" is changed in this message. SS starts to monitor the uplink RACH after approximately 4087 frames from step 2.
4	+	PAGING TYPE 1	SS starts to transmit this message continuously on the PCCH according to the new value of "DRX* Cycle Length Coefficient", at the next paging occasion immediately following step 3.
5	→	RRC CONNECTION REQUEST	UE transmits a request due to answer to the PAGING TYPE 1 received in step 4. The IE "Establishment Cause" should be set to "Terminating Call" supported by the UE and the "Initial UE Identity" set to UE's IMSI.
6	+	RRC CONNECTION REJECT	UE shall return to idle mode after receiving this message

Specific Message Contents

SYSTEM INFORMATION BLOCK TYPE 1 (Step 1) – for UEs supporting GSM-MAP core networks

Information Element	Value/remark
References to other system information blocks	Not Present
CN common GSM-MAP NAS system information	Location Area Information (LAI)
CN domain system information list	Supports both CS and PS domains
- CN domain system information	
- CN domain identity	CS domain
- CHOICE CN Type	GSM-MAP
- CN domain specific NAS system information	Set to the same octet string as in IE "CN common GSM MAP-NAS system information"
- CN domain specific DRX cycle length coefficient - CN domain system information	12
- CN domain identity	PS domain
- CHOICE CN Type	GSM-MAP
- CN domain specific NAS system information	Set to the same octet string as in IE "CN common GSM MAP-NAS system information"
 CN domain specific DRX cycle length coefficient 	12
UE Timers and constants in idle mode	
- T300	400 milliseconds
- N300	7
- T312	10 seconds
- N312	200
UE Timers and constants in connected mode	Not Present

SYSTEM INFORMATION BLOCK TYPE 13 (Step 1) – for UEs supporting ANSI-41 core networks

Information Element	Value/remark
References to other system information blocks	Not Present
CN domain system information list	Supports both CS and PS domains
CN domain system information	
- CN domain identity	CS
- CHOICE CN Type	ANSI-41
 CN domain specific NAS system information 	
- ANSI-41 NAS System Information	
- ANSI-41 NAS Parameter	Set to an ANSI-41 user zone information
 CN domain specific DRX cycle length coefficient 	12
- CN domain identity	PS
- CHOICE CN Type	ANSI-41
- CN domain specific NAS system information	
- ANSI-41 NAS System Information	
- ANSI-41 NAS Parameter	Set to an ANSI-41 user zone information
 CN domain specific DRX cycle length coefficient 	12
UE Capability update requirement	Not Present

PAGING TYPE 1 (Step 2)

Information Element	Value/remark
Paging record list	Only 1 entry
Paging record	
- CHOICE <u>Used Ppaging identityoriginator</u>	CN identityoriginator
- Paging Cause	Terminating Call with one of the supported services
	(Conversation Call, Streaming Call, Interactive Call,
	Background Call, SMS)
- CN Domain Identity	CS Domain
- CHOICE UE Identity	IMSI
- IMSI	Set to the same octet string as in the IMSI value stored
	in the USIM card
BCCH modification info	
MIB Value Tag	2
BCCH Modification time	4088

MASTER INFORMATION BLOCK (Step 3)

Information Element	Value/remark
MIB Value tag	2

SYSTEM INFORMATION BLOCK TYPE 1 (Step 3) – for UEs supporting GSM-MAP core networks

Information Element	Value/remark
References to other system information blocks	Not Present
CN common GSM-MAP NAS system information	Location Area Information (LAI)
CN domain system information list	Supports both CS and PS domains
 CN domain system information 	
 CN domain identity 	CS domain
- CHOICE CN Type	GSM-MAP
- CN domain specific NAS system information	Set to the same octet string as in IE "CN common GSM MAP-NAS system information"
 CN domain specific DRX cycle length coefficient 	6
- CN domain system information	
 CN domain identity 	PS domain
- CHOICE CN Type	GSM-MAP
 CN domain specific NAS system information 	Set to the same octet string as in IE "CN common GSM
	MAP-NAS system information"
 CN domain specific DRX cycle length coefficient 	6
UE Timers and constants in idle mode	
- T300	400 milliseconds
- N300	7
- T312	10 seconds
- N312	200
UE Timers and constants in connected mode	Not Present

SYSTEM INFORMATION BLOCK TYPE 13 (Step 3) – for UEs supporting ANSI-41 core networks

Information Element	Value/remark
References to other system information blocks	Not Present
CN domain system information list	Supports both CS and PS domains
CN domain system information	
- CN domain identity	CS domain
- CHOICE CN Type	ANSI-41
 CN domain specific NAS system information 	
- ANSI-41 NAS system information	
- ANSI-41 NAS Parameter	Set to an ANSI-41 user zone information
 CN domain specific DRX cycle length coefficient 	6
- CN domain identity	PS domain
- CHOICE CN Type	ANSI-41
 CN domain specific NAS system information 	
- ANSI-41 NAS system information	
- ANSI-41 NAS Parameter	Set to an ANSI-41 user zone information
 CN domain specific DRX cycle length coefficient 	6

PAGING TYPE 1 (Step 4)

Information Element	Value/remark
Paging record list	Only 1 entry
Paging record	
 CHOICE <u>Used Ppaging identityoriginator</u> 	CN <u>identity</u> originator
- Paging Cause	Terminating Call with one of the supported service
	(Conversation Call, Streaming Call, Interactive Call,
	Background Call, SMS)
- CN Domain Identity	CS Domain
- CHOICE UE Identity	IMSI
- IMSI	Set to the same octet string as in the IMSI value stored
	in the USIM card

RRC CONNECTION REJECT (Step 6)

Information Element	Value/remark
Initial UE identity	
- CHOICE UE id type	IMSI
- IMSI	Set to the same octet string as in the IMSI value stored
Rejection cause	in the USIM card
Wait time	Unspecified
Redirection info	0
	Not Present

8.1.1.4.5 Test requirement

After step 5 the UE shall transmit RRC CONNECTION REQUEST message in response to the PAGING TYPE 1 messages sent in step 4.

8.1.1.5 Paging for Notification in connected mode (CELL_PCH)

8.1.1.5.1 Definition

8.1.1.5.2 Conformance requirement

When a system information block on the BCCH is modified, the message PAGING TYPE 1 can be sent on the PCCH to inform the UE about this change in the CELL_PCH state. This message includes the IE "BCCH Modification Information". Upon receiving this notification from the UTRAN, the UE shall read the relevant MIB and/or SIB(s) subsequently while in CELL_PCH state.

Reference

3GPP TS 25.331 clause 8.1.1.2

8.1.1.5.3 Test purpose

To confirm that the UE enters the CELL_FACH state, checks the new value tag of the master information block, and read the SYSTEM INFORMATION messages after it receives a PAGING TYPE 1 message which includes the IE "BCCH Modification Information"

8.1.1.5.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CELL_PCH state (state 6-1512) as specified in clause 7.4 of TS 34.108 with valid a U-RNTI assigned to it.

Test Procedure

Identical test steps 1 to 5 in Clause 8.1.1.4 are applied to this test. However, the PAGING TYPE 1 messages used in step 2 and step 4 are altered. The changes are indicated in the specific message content paragraph under this clause. At step 6, UE shall send the CELL UPDATE message indicating the "cell update cause" to be "paging response". SS then replies with a CELL UPDATE CONFIRM message to allow the UE to transit to CELL_FACH state.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	←	MASTER INFORMATION BLOCK SYSTEM INFORMATION BLOCK TYPE 1 or SYSTEM INFORMATION BLOCK TYPE 13	SS transmits information on BCCH in order for UE to listen to one of the S-CCPCH physical channel, which carries PAGING TYPE 1 or 13 messages on PCCH. Relevant paging parameters are also broadcasted.
2	\	PAGING TYPE 1	SS transmits the paging message which comprises IE "BCCH Modification Information", with the "Value Tag" changed from the "MIB Value Tag" of the current Master Information Block. Also the modification time is set to 4088 radio frame from the current SFN. SS continuously broadcast the same MASTER INFORMATION BLOCK and various types of SYSTEM INFORMATION BLOCK on BCCH for a period stretching 4087 frames.
3	+	MASTER INFORMATION BLOCK	SS starts to transmit the MIB with the "MIB Value Tag" IE different from the original setting.
	←	SYSTEM INFORMATION BLOCK TYPE 1 or SYSTEM INFORMATION BLOCK TYPE 13	At the same time, SS starts to transmit the affected SIB TYPE 1 or TYPE 13 messages continuously. The value of IE "DRX* Cycle Length Coefficient" is changed in this message. SS starts to monitor the uplink RACH after approximately 4087 SFN from step 2.
4	+	PAGING TYPE 1	SS transmits this message continuously on the PCCH according to the new value of "DRX* Cycle Length Coefficient", at the next paging occasion immediately following step 3. This message shall page the UE with its U-RNTI and setting the UTRAN as the paging originator.
5	→	CELL UPDATE	The IE "Cell Update Cause" should be set to "Paging Response" and the IE "U-RNTI" shall be similar to the UE's U-RNTI value. The "Protocol Error Indicator" IE shall be set to FALSE.
6	+	CELL UPDATE CONFIRM	UE shall transit to CELL_FACH state after receiving this message.

Specific Message Contents

SYSTEM INFORMATION BLOCK TYPE 1 or TYPE 13 (Step 1)

The content of this message is the same in the message used in step 1 specified in Clause 8.1.1.4.3.

PAGING TYPE 1 (Step 2)

Information Element	Value/remark
Paging record list	Only 1 entry
Paging record	
- CHOICE Used Ppaging identityoriginator	UTRAN identityoriginator
- U-RNTI	Equal to the U-RNTI assigned earlier.
- SRNC Identity	
- S-RNTI	
BCCH modification info	
- MIB Value Tag	2
- BCCH Modification time	4088

MASTER INFORMATION BLOCK (Step 3) and

SYSTEM INFORMATION BLOCK TYPE 1 or TYPE 13 (Step 3)

The content of these messages is the same in the message used in step 3 specified in Clause 8.1.1.4.4.

PAGING TYPE 1 (Step 4)

Information Element	Value/remark
Paging record list	Only 1 entry
Paging record	
 CHOICE <u>Used</u> <u>Ppaging identityoriginator</u> 	UTRAN identityoriginator
- U-RNTI	Equal to the U-RNTI assigned earlier.
- SRNC Identity	Same as the current SRNC allocated
- S-RNTI	Same as the current S-RNTI allocated
BCCH modification info	Not Present

CELL UPDATE CONFIRM (Step 6)

Information Element	Value/remark
DRX Indicator RRC State Indicator	DRx with Cell UpdatingCELL_FACH

8.1.1.5.5 Test requirement

After step 5 the UE shall transmit a CELL UPDATE message with "cell update cause" IE set to "paging response". Upon receiving CELL UPDATE CONFIRM message, the UE shall enter the CELL_FACH state.

8.1.1.6 Paging for Notification in connected mode (URA_PCH)

8.1.1.6.1 Definition

8.1.1.6.2 Conformance requirement

When a system information block on the BCCH is modified, the UTRAN can send a PAGING TYPE 1 message on the PCCH to inform UE about the changes while the UE is in the URA_PCH state. This message includes the IE "BCCH Modification Information". When receiving this message in URA_PCH state, the UE shall read the relevant MIB and/or SIB(s).

Reference

3GPP TS 25.331 clause 8.1.1.2

8.1.1.6.3 Test purpose

To confirm that the UE enters the CELL_FACH state, checks the included new value tag of the master information block and reads the relevant SYSTEM INFORMATION block(s) after it receives a PAGING TYPE 1 message which includes the IE "BCCH Modification Information".

8.1.1.6.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: URA_PCH state (state 6-1613) as specified in clause 7.4 of TS 34.108 with a valid U-RNTI assigned.

Test Procedure

The UE is in the URA_PCH state before the SS starts changing SYSTEM INFORMATION BLOCK messages. SS modifies its SYSTEM INFORMATION and updates the "value tag" of both the SYSTEM INFORMATION BLOCK TYPE 1 and of the MASTER INFORMATION BLOCK. After a while, the SS transmits a PAGING TYPE 1 message, which includes the IE "BCCH Modification Information". The UE enters the CELL_FACH state and reads the modified SYSTEM INFORMATION BLOCK. The UE shall act according to the modified message. In this test case, the UE shall adjust it paging occasions and read the new PCCH blocks newly assigned to it under DRX mode.

Expected sequence

Step	Dire	ction	Message	Comment
	UE	SS		
1		←	MASTER INFORMATION BLOCK SYSTEM INFORMATION	The SS changes the SYSTEM INFORMATION when the UE is in the connected state (URA_PCH).
2		←	PAGING TYPE 1	SS transmits the message includes the IE "BCCH Modification Information".
3				The UE enters the CELL_FACH state and reads the SYSTEM INFORMATION and then the UE follows this message.

Specific Message Contents

None

8.1.1.6.5 Test requirement

After step 2 the UE shall enter the CELL_FACH state and read the SYSTEM INFORMATION message and follow it.

8.1.1.7 Paging for Connection in connected mode (CELL_DCH)

8.1.1.7.1 Definition

8.1.1.7.2 Conformance requirement

This procedure is used to transmit a PAGING TYPE 2 message from the network to selected UEs in CELL_DCH state using the dedicated control channel (DCCH). The UE listens to it and responds to this message accordingly.

Reference

3GPP TS 25.331 clause 8.1.11

8.1.1.7.3 Test purpose

To confirm that the UE responds this message after it receives a PAGING TYPE2 message which includes IE "Paging Record Type Identifier" for the UE.

8.1.1.7.4 Method of test

Initial Condition

System Simulator: 1 cell.

UE: CELL_DCH state (state 6-9 or state 6-10) as specified in clause 7.4 of TS 34.108, depending on the CN domain(s) supported by the UE, after executing a location registration or attach procedure followed by the release of the TMSI of P-TMSI allocated.

Test Procedure

The SS transmits a PAGING TYPE 2 message which includes an unmatched Paging Record Type Identifier in CELL_DCH state. The UE shall not respond to this message. SS pages the UE again, this time with a matched Paging Record Type Identifier but with the IE "paging cause" set to an invalid value which is not defined one of the spare values. UE shall respond by transmitting a RRC STATUS message on the DCCH using RLC-AM mode. Finally, SS transmits a PAGING TYPE 2 message, which includes a matched Paging Record Type Identifier. Then the UE shall responds to this message by the transmission of an upper layer message.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	\	PAGING TYPE 2	The SS transmits the message includes an unmatched identifier, then the UE does not respond.
2	+	PAGING TYPE 2	The SS transmits the message includes a matched identifier. In the paging message, IE "paging cause" is set to invalid value one of the spare values.
3	\rightarrow	RRC STATUS	The UE shall respond by reporting the protocol error to the SS.
4	+	PAGING TYPE 2	SS pages the UE with a matched identifier and with a valid "paging cause" IE.
5	\rightarrow	UPLINK DIRECT TRANSFER	The UE shall respond to the paging message sent in step 4.

Specific Message Contents

PAGING TYPE 2 (Step 1)

Information Element	Value/remark
Paging cause	Terminating Call supported by the UE
CN domain identity	Domain supported by the UE
Paging Record record Type type Identifier identifier	Set to "TMSI" or "P-TMSI" allocated during the
	execution of location registration or attach procedure
	respectively

PAGING TYPE 2 (Step 2)

Information Element	Value/remark
Paging cause	invalid value which is not defined Use one of the spare
CN domain identity	values
Paging Record Type Identifier	Domain supported by the UE
	Set to "IMSI (GSM-MAP)" for UEs supporting GSM-
	MAP core network type or "IMSI (DS-41)" for UEs
	supporting ANSI-41 core network type.

RRC STATUS (Step 3)

Information Element	Value/remark	
Protocol error information	Checked to see if set to "Information element value not comprehended"	

PAGING TYPE 2 (Step 4)

Information Element	Values/Remarks
Paging cause	Terminating Call supported by the UE
CN domain identity	Domain supported by the UE
Paging Record record Type type Identifier Identifier	Set to "IMSI (GSM-MAP)" for UEs supporting GSM-MAP
	core network type or "IMSI (DS-41)" for UEs supporting
	ANSI-41 core network type.

UPLINK DIRECT TRANSFER (Step 5)

Only the message type IE for this message is checked.

8.1.1.7.5 Test requirement

After step 1 the UE shall not respond to the paging message on the DCCH.

After step 2 the UE shall respond to the paging message by transmitting RRC STATUS on the DCCH, stating the protocol error as "Information element value not comprehended".

After step 4 the UE shall respond to the paging message by transmitting an UPLINK DIRECT TRANSFER message on the uplink DCCH.

8.1.1.8 Paging for Connection in connected mode (CELL_FACH)

8.1.1.8.1 Definition

8.1.1.8.2 Conformance requirement

This procedure is used to transmit a PAGING TYPE 2 message from the network to selected UEs in CELL_FACH state using the dedicated control channel (DCCH). The UE shall listen to it and responds to this message accordingly.

Reference

3GPP TS 25.331 clause 8.1.11

8.1.1.8.3 Test purpose

To confirm that the UE responds to a PAGING TYPE 2 message, which includes a matching value for IE "Paging Record Type Identifier".

8.1.1.8.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CELL_FACH state (state 6-11) as specified in clause 7.4 of TS 34.108.

Test Procedure

The SS transmits a PAGING TYPE 2 message, which includes an unmatched Paging Record Type Identifier in CELL_FACH state. The UE shall not respond to this message. The SS transmits a PAGING TYPE 2 message, which includes a matched Paging Record Type Identifier. Then the UE shall respond by transmitting an upper layer message to answer this page.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1	+		PAGING TYPE 2	The SS transmits the message includes an unmatched identifier, then the UE does not respond.
2	+		PAGING TYPE 2	The SS transmits the message includes a matched identifier.
3	→		UPLINK DIRECT TRANSFER	The UE responds by sending an upper layer message.

Specific Message Content

PAGING TYPE 2 (Step 1)

Use the same message content as in step 1 from 8.1.1.7.4

PAGING TYPE 2 (Step 2)

Use the same message content as in step 4 from 8.1.1.7.4

8.1.1.8.5 Test requirement

After step 1 the UE shall not respond.

After step 2 the UE shall respond to the second PAGING TYPE 2 message by transmitting an UPLINK DIRECT TRANSFER message on the uplink DCCH.

8.1.2 RRC Connection Establishment

8.1.2.1 RRC Connection Establishment in CELL_DCH state: Success

8.1.2.1.1 Definition

8.1.2.1.2 Conformance requirement

- 1. The RRC connection establishment is initiated by the UE, which leaves the idle mode and transmits an RRC CONNECTION REQUEST message. This message shall include the IE "Initial UE identity" and is to be transmitted on the uplink CCCH.
- 2. After the UE receives an RRC CONNECTION SETUP message which includes the same value of the IE "initial UE identity", radio resource parameters (i.e. Signalling link type and multiplexing info) and U-RNTI, UE then configures

the layer 2 and layer 1 processing so as to support the DCCH according to the radio resource parameters specified. The procedure successfully ends when the network receives an RRC CONNECTION SETUP COMPLETE message on the uplink DCCH.

Reference

3GPP TS 25.331 clause 8.1.3

8.1.2.1.3 Test purpose

To confirm that the UE leaves the Idle Mode and correctly establishes a signalling link on the DCCH.

8.1.2.1.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: Idle state (state 2 or state 3 or state 7) as specified in clause 7.4 of TS 34.108 depending on the CN domain(s) supported by the UE

Test Procedure

The UE transmits an RRC CONNECTION REQUEST message to the SS on the uplink CCCH by attempting to make an outgoing call. After SS receives this message, it assigns the necessary radio resources and U-RNTI to be used by the UE. SS then transmits an RRC CONNECTION SETUP message containing an IE "Initial UE Identity" that does not match the IE "Initial UE Identity" in the most recent RRC CONNECTION REQUEST message sent by the UE. UE receives the RRC CONNECTION SETUP message within timer T300 but discards it due to the IE "Initial UE Identity" mismatch. UE shall wait for timer T300 to time out before re-transmitting a RRC CONNECTION REQUEST message to the SS. SS again assigns the necessary radio resources and U-RNTI. SS then follows by transmitting a RRC CONNECTION SETUP message containing an IE "Initial UE Identity" that matches the IE "Initial UE Identity" in the most recent RRC CONNECTION REQUEST sent by the UE. SS then waits for the UE to transmit an RRC CONNECTION SETUP COMPLETE message on the DCCH.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	\rightarrow	RRC CONNECTION REQUEST	By outgoing call operation
2	+	RRC CONNECTION SETUP	This message is not addressed to the UE.
3	÷	RRC CONNECTION REQUEST	UE shall re-transmit the request message again after a time out of T300 from step 1.
4	+	RRC CONNECTION SETUP	
5			The UE configures the layer 2 and layer 1.
6	→	RRC CONNECTION SETUP COMPLETE	

Specific Message Content

RRC CONNECTION SETUP (Step 2)

Information Element	Value/remark
Initial UE Identity	
CHOICE UE id type	IMSI
IMSI	Set to an arbitrary octet string of length 7 which different
	from the IMSI value stored in the USIM card.

8.1.2.1.5 Test requirement

After step 2 the UE shall re-transmit the RRC CONNECTION REQUEST message again in order to continue the RRC connection establishment procedure.

After step 6 the UE shall establish an RRC connection and continue the procedure of the outgoing call on the DCCH.

8.1.2.2 RRC Connection Establishment: Success after T300 timeout

8.1.2.2.1 Definition

8.1.2.2.2 Conformance requirement

1. The RRC connection establishment is initiated by the UE, which leaves the idle mode. The UE shall transmit an RRC CONNECTION REQUEST message which includes the IE "Initial UE identity". This message shall be sent on the uplink CCCH.

When there are more than one PRACHs available, the UE shall select one PRACH randomly and transmit an RRC CONNECTION REQUEST message by use of selected PRACH.

2. In the case of a failure to establish the RRC connection at the expiry of timer T300, the UE retries to establish the RRC connection until V300 is greater than N300

When the UE receives a RRC CONNECTION SETUP message, which contains a protocol error and causing the internal variable PROTOCOL_ERROR_REJECT set to TRUE, it shall perform the appropriate error handling procedure.

Reference

3GPP TS 25.331 clause 8.1.3

8.1.2.2.3 Test purpose

To confirm that the UE retries to establish the RRC connection until V300 is greater than N300 after the expiry of timer T300 when the SS transmits no response for an RRC CONNECTION REQUEST message.

8.1.2.2.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: Idle state (state 2 or state 3 or state 7) as specified in clause 7.4 of TS 34.108 depending on the CN domain(s) supported by the UE

Test Procedure

Before the test starts, an internal counter K in SS is initialized to a value = 1. Following this, the UE shall transmit an RRC CONNECTION REQUEST message to the SS on the uplink CCCH by use of selected PRACH from the available PRACH No.1 and PRACH No.2, after the operator attempts to make an outgoing call. SS ignores this message, increments K every time such a message is received and waits for T300 timer to expire. This cycle is repeated until K reaches N300. When K is equal to N300, the SS transmits the RRC CONNECTION SETUP message specified in step 5 6 to the UE and wait until T300 expires. The UE shall send another RRC CONNECTION REQUEST message on the uplink CCCH. SS verifies that the UE does not access the radio resource allocated in step 5 6. After confirming this restriction is observed, SS replies with a valid RRC CONNECTION SETUP message. The UE shall then acknowledge the establishment of RRC connection by sending the RRC CONNECTION SETUP COMPLETE message on uplink DCCH.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
<u>1</u>	<u></u>	SYSTEM INFORMATION BLOCK TYPE 5	Transmit these messages on the BCCH. See specific message contents.
<u> </u>			SS initializes counter K to 1. Operator is asked to make an outgoing call and SS starts to wait for RRC CONNECTION REQUEST on uplink CCCH.
2 3	\rightarrow	RRC CONNECTION REQUEST	
<u>34</u>			SS checks to see if K is equal to N300. If so, goes to step <u>56</u> . Else, continues to execute step 45.
4 <u>5</u>			SS increments K. The next step is step <u>23</u> .
<u>56</u>	+	RRC CONNECTION SETUP	The message contains a protocol error, see specific message content. SS waits for T300 to expire again.
6 7	→	RRC CONNECTION REQUEST	UE shall not access the radio resource indicated in RRC CONNECTION SETUP message sent in step 56.
7 <u>8</u>	+	RRC CONNECTION SETUP	This is a legal message. See the clause 6.19 in TS 34.108 on default message content for RRC.
<u>89</u>			The UE configures the <u>layer</u> 1 and layer 2.
9 10	\rightarrow	RRC CONNECTION SETUP COMPLETE	-

- Assigned Sub-channel Number

Available signature Start IndexAvailable signature End Index

SYSTEM INFORMATION TYPE 5 (Step 1)

STSTEM INFORMATION TTPE 5 (Step 1)	
- PRACH system information	2PRACHs
- PRACH info (PRACH No.1)	500
- CHOICE mode	FDD (0000 1444 1444)
- Available Signature - Available SF	(0000 0000 1111 1111'B Reference to clause 6.10 Parameter Set
- Preamble scrambling code number	0
- Puncturing Limit	Reference to clause 6.10 Parameter Set
- Available Sub Channel number	'1111 1111 1111'B
- Transport Channel Identity	<u>15</u>
- RACH TFS	
- CHOICE Transport channel type	Common transport channels
Dynamic Transport format information	(This IE is repeated for TFI number)
- RLC size	Reference to clause 6.10 Parameter Set
- Number of TB and TTI List - Number of Tranport blocks	Reference to clause 6.10 Parameter Set Reference to clause 6.10 Parameter Set
- CHOICE Mode	FDD
- CHOICE Logical Channel List	ALL
- Semi-static Transport Format information	
- Transmission time interval	Reference to clause 6.10 Parameter Set
- Type of channel coding	Reference to clause 6.10 Parameter Set
- Coding Rate	Reference to clause 6.10 Parameter Set
- Rate matching attribute	Reference to clause 6.10 Parameter Set
- CRC size	Reference to clause 6.10 Parameter Set
- RACH TFCS - Normal	(This IE is repeated for TFC number.)
- TFCI Field 1 information	
- CHOICE TFCS representation	Addition
- TFCS addition information	- Name:
- CHOICE CTFC Size	Number of bits used must be enough to cover all
	combinations of CTFC from clause 6.10.
- CTFC information	Refer to clause 6.10 Parameter Set
- Power offset information	0
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor ßc - Gain factor ßd	$\begin{bmatrix} \underline{0} \\ 0 \end{bmatrix}$
- Reference TFC ID	Not Present
- Power offset Pp-m	0dB
- PRACH partitioning	
- Access Service Class	
ASC Settings	
- Available signature Start Index	0 (ASC#0)
- Available signature End Index	7 (ASC#0)
- Assigned Sub-channel Number	<u>'1111'B</u>
- Available signature Start Index	0 (ASC#1)
- Available signature End Index	7 (ASC#1)
- Assigned Sub-channel Number	<u>'1111'B</u>
- Available signature Start Index	0 (ASC#2)
- Available signature End Index	7 (ASC#2)
- Assigned Sub-channel Number	<u>'1111'B</u>
- Available signature Start Index	0 (ASC#3)
- Available signature End Index	7 (ASC#3)
Assigned Sub-channel Number	<u>'1111'B</u>
- Available signature Start Index	0 (ASC#4)
- Available signature End Index	7 (ASC#4)
- Assigned Sub-channel Number	<u>'1111'B</u>
- Available signature Start Index	0 (ASC#5)
- Available signature End Index	7 (ASC#5)

<u>'1111'B</u>

0 (ASC#6) 7 (ASC#6)

Accionad Cub abanas Number	Lagrania
- Assigned Sub-channel Number	<u>'1111'B</u>
- Available signature Start Index	<u>0 (ASC#7)</u>
 Available signature End Index 	7 (ASC#7)
- Assigned Sub-channel Number	<u>'1111'B</u>
- Persistence scaling factor	
- Persistence scaling factor	0.9 (for ASC#2)
- Persistence scaling factor	0.9 (for ASC#3)
 Persistence scaling factor 	0.9 (for ASC#4)
- Persistence scaling factor	0.9 (for ASC#5)
- Persistence scaling factor	0.9 (for ASC#6)
- Persistence scaling factor	0.9 (for ASC#7)
- AC-to-ASC mapping table	6 (4000)
- AC-to-ASC mapping - AC-to-ASC mapping	6 (AC0-9) 5 (AC10)
- AC-to-ASC mapping	4 (AC11)
- AC-to-ASC mapping	3 (AC12)
- AC-to-ASC mapping	2 (AC13)
- AC-to-ASC mapping	1 (AC14)
- AC-to-ASC mapping	0 (AC15)
- Primary CPICH DL TX power	Reference to clause 6.10 Parameter Set
- Constant value	Reference to clause 6.10 Parameter Set
- PRACH power offset	
Power Ramp Step	<u>3dB</u>
- Preamble Retrans Max	2
- RACH transmission parameters	
- Mmax	2
- NB01min	3 slot 10 slot
- NB01max - AICH info	10 Slot
- Channelisation code	SF-1(SF is reference to clause 6.10 Parameter Set)
- STTD indicator	FALSE
- AICH transmission timing	0
- PRACH info (PRACH No.2)	_
- CHOICE mode	<u>FDD</u>
- Available Signature	<u>'0000 0000 1111 1111'B</u>
- Available SF	Reference to clause 6.10 Parameter Set
- Preamble scrambling code number	1
- Puncturing Limit	Reference to clause 6.10 Parameter Set
- Available Sub Channel number	<u>'1111 1111 1111'B</u>
- Transport Channel Identity	<u>16</u>
- RACH TFS - CHOICE Transport channel type	Common transport channels
- Dynamic Transport format information	(This IE is repeated for TFI number)
- RLC size	Reference to clause 6.10 Parameter Set
- Number of TB and TTI List	Reference to clause 6.10 Parameter Set
- Number of Tranport blocks	Reference to clause 6.10 Parameter Set
- CHOICE Mode	<u>FDD</u>
- CHOICE Logical Channel List	ALL
- Semi-static Transport Format information	
- Transmission time interval	Reference to clause 6.10 Parameter Set
 Type of channel coding 	Reference to clause 6.10 Parameter Set
- Coding Rate	Reference to clause 6.10 Parameter Set
- Rate matching attribute	Reference to clause 6.10 Parameter Set
- CRC size	Reference to clause 6.10 Parameter Set
- RACH TFCS	(This IE is repeated for TFC number.)
- Normal - TFCI Field 1 information	
- CHOICE TFCS representation	Addition
- TFCS addition information	/ Maniori
- CHOICE CTFC Size	Number of bits used must be enough to cover all
	combinations of CTFC from clause 6.10.
- CTFC information	Refer to clause 6.10 Parameter Set
- Power offset information	
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor ßc	0
- Gain factor ßd	0 Net Broomt
- Reference TFC ID	Not Present
- Power offset Pp-m	0dB

11	,
- PRACH partitioning	
- Access Service Class	
- ASC Settings	
- Available signature Start Index	0 (ASC#0)
- Available signature End Index	7 (ASC#0)
- Assigned Sub-channel Number	<u>'1111'B</u>
- Available signature Start Index	0 (ASC#1)
- Available signature End Index	7 (ASC#1)
- Assigned Sub-channel Number	1111'B
	
- Available signature Start Index	0 (ASC#2)
 Available signature End Index 	7 (ASC#2)
- Assigned Sub-channel Number	<u>'1111'B</u>
- Available signature Start Index	0 (ASC#3)
- Available signature End Index	7 (ASC#3)
- Assigned Sub-channel Number	<u>'1111'B</u>
- Available signature Start Index	0 (ASC#4)
- Available signature End Index	7 (ASC#4)
- Assigned Sub-channel Number	<u>'1111'B</u>
:1	
- Available signature Start Index - Available signature End Index	0 (ASC#5) 7 (ASC#5)
- Assigned Sub-channel Number	<u>'1111'B</u>
- Available signature Start Index	0 (ASC#6)
- Available signature End Index	7 (ASC#6)
- Assigned Sub-channel Number	'1111'B
- Available signature Start Index	0 (ASC#7)
- Available signature End Index	7 (ASC#7)
- Assigned Sub-channel Number	<u>'1111'B</u>
- Persistence scaling factor	
- Persistence scaling factor	0.9 (for ASC#2)
- Persistence scaling factor	0.9 (for ASC#3)
- Persistence scaling factor	0.9 (for ASC#4)
- Persistence scaling factor	0.9 (for ASC#5)
- Persistence scaling factor	0.9 (for ASC#6)
- Persistence scaling factor	0.9 (for ASC#7)
- AC-to-ASC mapping table	0.9 (101 A30#1)
- AC-to-ASC mapping table	6 (400 0)
	6 (AC0-9)
- AC-to-ASC mapping	5 (AC10)
- AC-to-ASC mapping	4 (AC11)
- AC-to-ASC mapping	3 (AC12)
- AC-to-ASC mapping	2 (AC13)
- AC-to-ASC mapping	1 (AC14)
- AC-to-ASC mapping	<u>0 (AC15)</u>
- Primary CPICH DL TX power	Reference to clause 6.10 Parameter Set
- Constant value	Reference to clause 6.10 Parameter Set
- PRACH power offset	
- Power Ramp Step	3dB
- Preamble Retrans Max	<u>2</u>
- RACH transmission parameters	
- Mmax	<u>2</u>
- NB01min	$\frac{3}{3}$ slot
- NB01max	10 slot
- AICH info	
- Channelisation code	SF-1(SF is reference to clause 6.10 Parameter Set)
- STTD indicator	FALSE
- AICH transmission timing	0
<u> Alon transmission uning</u>	I ≌

RRC CONNECTION SETUP (Step 56)

Information Element	Value/remark
RRC State Indicator	Not Present
Uplink Radio Resource	Neither PRACH Info (for RACH) nor Uplink DPCH info
CHOICE channel requirement	is chosen. One of the spare value is used.
· ·	·

8.1.2.2.5 Test requirement

After step 2 the UE shall select either PRACH No.1 or PRACH No.2 and transmit an RRC CONNECTION REQUEST message.

After step $5-\underline{6}$ the UE shall re-send another RRC CONNECTION REQUEST message and not access any radio resources specified in RRC CONNECTION SETUP message sent in step $5\underline{6}$.

After step <u>8-9</u> the UE shall transmit an RRC CONNECTION SETUP COMPLETE message and establish an RRC connection on the DCCH logical channel.

8.1.2.3 RRC Connection Establishment: Failure (V300 is greater than N300)

8.1.2.3.1 Definition

8.1.2.3.2 Conformance requirement

- 1. The RRC connection establishment is initiated by the UE, which leaves the idle mode and transmits an RRC CONNECTION REQUEST message. This message shall include the IE "Initial UE identity" on the uplink CCCH.
- 2. In the case of a failure to establish the RRC connection after (N300+1) attempts, the UE goes back to idle mode.

Reference

3GPP TS 25.331 clause 8.1.3

8.1.2.3.3 Test purpose

To confirm that the UE stops retrying to establish the RRC connection if V300 is greater than N300 and goes back to idle mode.

8.1.2.3.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: Idle state (state 2 or state 3 or state 7) as specified in clause 7.4 of TS 34.108 depending on the CN domain(s) supported by the UE

Test Procedure

Before the test starts, SS initializes an internal counter K to 1. The UE transmits an RRC CONNECTION REQUEST message to the SS on the uplink CCCH by an outgoing call operation. SS shall not respond to any RRC CONNECTION REQUEST message, instead the counter K is increased by 1 every time such a message is received. To arrive at the verdict, the SS checks that a total of (N300+1) such messages are received.

Expected sequence

Step	Dire	ction	Message	Comment
	UE	SS		
1				SS initializes counter K to 1and then prompts the operator to make an outgoing call.
2		\rightarrow	RRC CONNECTION REQUEST	
3				SS increments K by 1.
4				If K is greater than N300, goes to step 5 else proceed to step 2.
5				SS monitor the uplink CCCH for a time period enough for UE to goes back to normal service. The exact amount of time to wait shall be derived from TS related to cell selection. If any uplink transmission is detected, the test fails.

Specific Message Contents

None

8.1.2.3.5 Test requirement

After step 5, counter K shall be equals to (N300+1) and there shall be no uplink transmission in the monitoring period specified in step 5.

8.1.2.4 RRC Connection Establishment: Reject ("wait time" is not equal to 0)

8.1.2.4.1 Definition

8.1.2.4.2 Conformance requirement

- 1. The RRC connection establishment is initiated by the UE, which leaves the idle mode and transmits an RRC CONNECTION REQUEST message . This message shall include the IE "Initial UE identity" and is to be sent on the uplink CCCH.
- 2. After the UE receives an RRC CONNECTION REJECT message which includes IE "wait time" not set to 0, and neither IE "frequency info" nor IE "system info" is present, the UE shall wait for a period specified in the IE "wait time". Thereafter re-transmit an RRC CONNECTION REQUEST message to attempt to establish the RRC connection again. However, either IE "frequency info" or IE "system info" is available in the message, the UE shall attempt to perform cell reselection using these information.

Reference

3GPP TS 25.331 clause 8.1.3

8.1.2.4.3 Test purpose

To confirm that the UE retries to establish the RRC connection after the "wait time" if the UE receives an RRC CONNECTION REJECT message which includes the IE "wait time" not set to 0.

To confirm that the UE perform a cell reselection when receiving an RRC CONNECTION REJECT message, containing relevant frequency information of the target cell to be re-selected.

8.1.2.4.4 Method of test

Initial Condition

System Simulator: 2 cells – both cell 1 and cell 2 are active and suitable for camping, but cell 1 is transmitted using a larger power. Cell 1 and cell 2 are being transmitted from different 2 UARFCNs.

UE: Idle state (state 2 or state 3 or state 7) as specified in clause 7.4 of TS 34.108 depending on the CN domain(s) supported by the UE

Test Procedure

The UE transmits an RRC CONNECTION REQUEST message to the SS on the uplink CCCH by an outgoing call operation in cell 1. SS rejects the first request by transmitting an RRC CONNECTION REJECT message which indicates a non-zero wait time. In this message, frequency information for cell 2 is available. SS then waits for RRC CONNECTION REQUEST message on the uplink CCCH of cell 2. SS will also monitor the uplink of cell 1 simultaneously to ensure that all transmission activities from cell 1 have ceased. When the UE has successfully camp onto cell 2, it shall send an RRC CONNECTION REQUEST with the same establishment cause as its previous attempt in cell 1. SS responds with an RRC CONNECTION REJECT message, indicating a non-zero "wait time" and omitting the IE "Redirection Info". The UE shall observe the wait time period indicated. After the wait time has elapsed, the UE shall re-transmit RRC CONNECTION REEQUEST again. Finally, SS transmits an RRC CONNECTION SETUP message to establish an RRC connection with the UE, and the UE replies with an RRC CONNECTION SETUP COMPLETE message and enters CELL_DCH state.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	\rightarrow	RRC CONNECTION REQUEST	SS prompts the operator to make an outgoing call in cell 1.
2	←	RRC CONNECTION REJECT	This message shall includes the IE "wait time" set to 15 seconds and IE "frequency info" set to the UARFCN of cell 2.
3			SS waits for a period of time sufficient for UE to reselect to cell 2. At the same time, it monitors the uplink of cell 1 to make sure that all transmissions have ceased.
4	→	RRC CONNECTION REQUEST	UE shall attempt to re-start an RRC connection establishment procedure in cell 2. The establishment cause shall remain unchanged.
5	+	RRC CONNECTION REJECT	This message shall includes the IE "wait time" set to 15 seconds, but with IE "Redirection Info" absent.
6	→	RRC CONNECTION REQUEST	SS waits until the duration specified in IE "wait time" has elapsed and then listens to the uplink CCCH for a second RRC CONNECTION REQUEST message.
7	+	RRC CONNECTION SETUP	SS sends the message to UE, to setup an RRC connection with the UE.
8			The UE shall configure the layer 2 and layer 1 in order to access the uplink and downlink DCCH assigned.
9	→	RRC CONNECTION SETUP COMPLETE	<u> </u>

Specific Message Contents

RRC CONNECTION REQUEST (Step 1)

Information Element	Value/remark
Initial UE Identity Initial UE Capability Establishment Cause	Must be equal to U-RNTI assigned previously Must be compatible with UE settings in TS25.926 Must be "Originating Call"
LStabilistifferit Gause	Must be Originating Call

RRC CONNECTION REJECT (Step 2)

Value/remark
15 seconds
Set to a different UARFCN from uplink carrier of cell 1
Not present – assuming a duplex distance of 190MHz.

RRC CONNECTION REQUEST (Step 4 and step 6)

Same requirement as in step 1.

RRC CONNECTION REJECT (Step 5)

Information Element	Value/remark
Wait time	15 seconds
Redirection Info	Not present

8.1.2.4.5 Test requirement

After step 3 the UE shall have successfully re-selected to cell 2, using information transmitted in IE "frequency info" of RRC CONNECTION REJECT message. UE shall trigger the start of RRC connection establishment by transmitting RRC CONNECTION REQUEST. The establishment cause shall be similar to the message sent in step 1.

After step 5 the UE shall observe the period specified in IE "wait time" of an RRC CONNECTION REJECT message and not transmit an RRC CONNECTION REQUEST message in this period.

After step 7 the UE shall transmit an RRC CONNECTION SETUP COMPLETE message to SS on uplink DCCH and then establish an RRC connection.

8.1.2.5 RRC Connection Establishment: Reject ("wait time" is not equal to 0 and V300 is greater than N300)

8.1.2.5.1 Definition

8.1.2.5.2 Conformance requirement

The RRC connection establishment is initiated by the UE, which leaves the idle mode and transmits an RRC CONNECTION REQUEST message . This message shall include the IE "Initial UE identity" and is to be sent on the uplink CCCH.

After the UE receives an RRC CONNECTION REJECT message which includes IE "wait time" not set to 0, and neither IE "frequency info" nor IE "system info" is present, the UE shall wait for a period specified in the IE "wait time". Thereafter it re-transmits an RRC CONNECTION REQUEST message to attempt to establish the RRC connection again. In the case of a failure to establish the RRC connection after (N300+1) attempts, the UE goes back to idle mode.

Reference

3GPP TS 25.331 clause 8.1.3

8.1.2.5.3 Test purpose

To confirm that the UE retries to establish the RRC connection after the "wait time" if the UE receives an RRC CONNECTION REJECT message which specifies a non-zero IE "wait time".

To confirm that the UE stops retrying to establish the RRC connection if V300 is greater than N300 and goes back to idle mode.

8.1.2.5.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: Idle state (state 2 or state 3 or state 7) as specified in clause 7.4 of TS 34.108 depending on the CN domain(s) supported by the UE

Test Procedure

Before the test starts, SS initializes an internal counter K to 0. The UE transmits an RRC CONNECTION REQUEST message to the SS on the uplink CCCH, triggered by an outgoing data call operation. SS rejects all requests by transmitting an RRC CONNECTION REJECT message which indicates a non-zero wait time and the counter K is increased by 1 every time such a message is received. To arrive at the verdict, the SS checks that a total of (N300+1) such messages are received.

Expected sequence

Step	Direc	tion	Message	Comment
	UE	SS	_	
1				SS initializes counter K to 0 and then prompts the operator to make an outgoing data call.
2)	RRC CONNECTION REQUEST	Shall be sent on CCCH and contain the correct establishment cause.
3	,	(RRC CONNECTION REJECT	This message includes the IE "wait time" set to 15 seconds.
4				SS increments K by 1.
5				If K is greater than N300, goes to step 6. Else SS waits for 15 sec before proceeding to step 2.
6				SS monitor the uplink CCCH for a time period enough for UE to goes back to normal service. The exact amount of time to wait shall be derived from TS related to cell selection. If any uplink transmission is detected, the test fails.

Specific Message Contents

RRC CONNECTION REQUEST (Step 2)

Information Element	Value/remark
Initial UE Identity	Must be equal to U-RNTI assigned previously
Initial UE Capability	Must be compatible with UE settings in TR25.926
Establishment Cause	Must be "Originating CS Data Call" or "Originating PS
	Data Call"

RRC CONNECTION REJECT (Step 3)

Information Element	Value/remark
Wait time	15 seconds

8.1.2.5.5 Test requirement

After step 6, counter K shall be equals to (N300+1) and there shall be no uplink transmission in the monitoring period specified in step 6.

8.1.2.6 RRC Connection Establishment: Reject ("wait time" is set to 0)

8.1.2.6.1 Definition

8.1.2.6.2 Conformance requirement

- 1. The RRC connection establishment is initiated by the UE, which leaves the idle mode and transmits an RRC CONNECTION REQUEST message on the uplink CCCH.
- 2. In the case of a failure to establish the RRC connection by the reception of a RRC CONNECTION REJECT message which contains IE "wait time" equals to 0, the UE shall go back to idle mode immediately.

Reference

3GPP TS 25.331 clause 8.1.3

8.1.2.6.3 Test purpose

To confirm that the UE goes back to idle mode, if the SS transmits an RRC CONNECTION REJECT message which includes IE "wait time" set to 0. To confirm that the UE ignores an RRC CONNECT REJECT message not addressed to it. To confirm that the UE is capable of handling an erroneous RRC CONNECTION REJECT message correctly.

8.1.2.6.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: Idle state (state 2 or state 3 or state 7) as specified in clause 7.4 of TS 34.108 depending on the CN domain(s) supported by the UE

Test Procedure

The UE transmits an RRC CONNECTION REQUEST message to the SS on the uplink CCCH by making an outgoing call. After the SS receives this message, it transmits an RRC CONNECTION REJECT message which is not addressed to the UE. The UE shall disregard this message and proceed to re-transmit RRC CONNECTION REQUES message upon T300 timer expiry. SS answers the second RRC CONNECTION REQUEST message by transmitting an RRC CONNECTION REJECT message with IE "wait time" set to 15 seconds, but without the mandatory IE "rejection cause". The UE shall continue to send the third RRC CONNECTION REQUEST message after a 15 second lapse. Next, the SS sends a legal RRC CONNECTION REJECT message which is expected to cause the UE to move to idle mode spontaneously. To confirm that finally the UE goes back to idle mode immediately after receiving the reject message, SS shall monitor the uplink CCCH for the next 60 second and verify that there is no further transmission in the uplink direction.

Expected sequence

Step	Directi	ion	Message	Comment
	UE	SS		
1	\rightarrow		RRC CONNECTION REQUEST	Test operator is prompted to make
2	+		RRC CONNECTION REJECT	an out-going call, IE "Initial UE identity" contains an identity different from any of the UE identities available.
3	→		RRC CONNECTION REQUEST	UE shall continue to send this message after T300 time-out.
4	+		RRC CONNECTION REJECT	IE "Reject Cause" is omitted, IE "wait time" is set to 15 seconds (maximum).
5	→		RRC CONNECTION REQUEST	UE shall continue to send this message after the expiry of "wait time" IE indicated in RRC CONNECTION REJECT message in step 4.
6	+		RRC CONNECTION REJECT	IE "wait time" is set to 0.
7				The UE goes back to idle mode

Specific Message Contents

RRC CONNECTION REQUEST (Step 1)

Information Element	Value/remark
Initial UE Identity	Checked to see if it is set to IMSI stored in the test
	USIM card.
Initial UE Capability	Checked to see if it is compatible to ICS/IXIT
	statements.
Establishment Cause	Checked to see if set to one of the supported originating
	call types
Protocol Error Indicator	Checked to see if set to "FALSE"
Measured Results on RACH	Checked to see if it is absent

RRC CONNECTION REJECT (Step 2)

Information Element	Value/remark
Initial UE Identity	
IMSI	Set to an arbitrary octet string of length 7 bytes, which is different from the IMSI stored in USIM.
Wait time	15 seconds
Redirection Info	Not present

RRC CONNECTION REQUEST (Step 3)

Information Element	Value/remark
Initial UE Identity	Checked to see if it is set to IMSI stored in the test
	USIM card.
Initial UE Capability	Checked to see if it is compatible to ICS/IXIT
	statements.
Establishment Cause	Checked to see if set to one of the supported originating
	call types
Protocol Error Indicator	Checked to see if set to "TRUE"
Measured Results on RACH	Checked to see if it is absent

RRC CONNECTION REJECT (Step 4)

Information Element	Value/remark
Initial UE Identity	
IMSI	Set to the identical octet string as the IMSI stored in the
Reject Cause	USIM card.
Wait time	Not Present
Redirection Info	15 seconds (Maximum)
	Not Present

RRC CONNECTION REQUEST (Step 5)

The contents of this message must be identical to those specified for the message in step 3.

RRC CONNECTION REJECT (Step 6)

Information Element	Value/remark
Initial UE Identity	
IMSI	Set to the identical octet string as the IMSI stored in the USIM card.
Reject Cause	Congestion
Wait time	0 second
Redirection Info	Not present

Note: T300 is set to 5 seconds and N300 is arbitrarily selected from 4 to 8 in SYSTEM INFORMATION

BLOCK TYPE 1 message on BCCH.

8.1.2.6.5 Test requirement

After step 2 the UE shall transmit an RRC CONNECTION REQUEST message on uplink CCCH.

After step 4 the UE shall re-transmit an RRC CONNECTION REQUEST message on the uplink CCCH 15 seconds after the transmission of the second downlink RRC CONNECTION REJECT message. In this message, the "protocol error indicator" IE shall be set to "TRUE".

After step 6 the UE shall stop sending an RRC CONNECTION REQUEST message, go back to idle mode immediately and not transmit in the uplink direction again.

8.1.2.7 RRC Connection Establishment in CELL FACH state: Success

8.1.2.7.1 Definition

8.1.2.7.2 Conformance requirement

During the RRC connection establishment, the UTRAN might assign common physical resource to the UE using an RRC CONNECTION SETUP message. When no information about the physical channels accessible is available from

the message, the UE shall utilize the PRACH and S-CCPCH information transmitted on the BCCH and then enter the CELL_FACH. Subsequently, the UE shall establish the required signalling links with the UTRAN using common physical resources.

Reference

3GPP TS 25.331 clause 8.1.3

8.1.2.7.3 Test Purpose

To confirm that the UE is able to enter CELL_FACH state and setup signalling links using common physical channels.

8.1.2.7.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: Idle state (state 2 or state 3 or state 7) as specified in clause 7.4 of TS 34.108 depending on the CN domain(s) supported by the UE

Test Procedure

The UE transmits an RRC CONNECTION REQUEST message to the SS on the uplink CCCH by attempting to make an outgoing call. After the SS receives this message, it assigns the necessary radio resources and U-RNTI to be used by the UE, and then transmits an RRC CONNECTION SETUP message to the UE within timer T300. SS then waits for the UE to transmit an RRC CONNECTION SETUP COMPLETE message on the DCCH.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1		\rightarrow	RRC CONNECTION REQUEST	Test operator is requested to make an outgoing call. The UE shall transmit this message, indicating the correct establishment cause.
2		←	RRC CONNECTION SETUP	SS omits both IE "Uplink DPCH Info" and IE "Downlink DPCH Info" from the message.
3				The UE shall configure the layer 2 and layer 1.
4		\rightarrow	RRC CONNECTION SETUP COMPLETE	UE shall send this message on the DCCH, carried by the assigned PRACH resources.

Specific Message Content

RRC CONNECTION REQUEST

Information Element	Value/remark
Establishment Cause	Originating PS Data <u>Interactive</u> Call

RRC CONNECTION SETUP

For this message, the contents of the message to be used are basically identical to the message sub-type entitled "RRC CONNECTION SETUP message (Transition to CELL_FACH)" found in the default message content part. The following exceptions are applicable in this test:

Information Element	Value/remark
Uplink DPCH Info	Not Present
Downlink information common for all radio links	Not Present
Downlink information per radio link list	Not Present

RRC CONNECTION SETUP COMPLETE

Information Element	Value/remark
UE Radio Access Capability	Checked to see if compatible with the stated capability
	in PIXIT/PICS statements.

8.1.2.7.5 Test requirements

After step 3 the UE shall establish the RRC connection, and transmit RRC CONNECTION SETUP COMPLETE message on the DCCH using PRACH physical resource specified in system information block messages.

8.1.2.8 RRC Connection Establishment: Invalid system information message reception

8.1.2.8.1 Definition

8.1.2.8.2 Conformance requirement

The UE shall ignore the message and shall not select the cell, if the associated a SYSTEM INFORMATION message on the BCCH which includes thean unknown invalid value in the mandatory information element which eriticality is set to "reject" in the master information block is broadcasting.

Reference

3GPP TS 25.331 clause 8.1.1 clause 16

8.1.2.8.3 Test purpose

To confirm that the UE does not select the cell if the transmitted SYSTEM INFORMATION message on the BCCH which includes the an unknown invalid value in the mandatory-information element whose criticality is set to "reject" value in the master information block is broadcasting

8.1.2.8.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: Power off (state 1) as specified in clause 7.4 of TS 34.108

Test Procedure

The SS broadcasts the SYSEM INFORMATION message on the BCCH which includes the PLMN Type information element having "spare value" an invalid value in the master information block. When the UE is supplied the power, it finds that the SYSTEM INFORMATION message on the BCCH includes the unknown value in the mandatory information element and the UE shall ignore this message. When an outgoing call is attempted, the test operator shall be informed that the UE is in a "No Service" state. The UE shall not transmit an RRC CONNECTION REQUEST message on the uplink CCCH.

Step	Dire	ction	Message	Comment
	UE	SS		
1		(SYSTEM INFORMATION	The SS broadcasts the SYSTEM INFORMATION message on the BCCH which includes an invalid valuean unknown value in the mandatory information element whose criticality is set to "reject" in the master information block.
2				The UE is supplied the power.
3				SS waits for 1 minute and then asks the test operator to attempt to make an outgoing call.
4				SS checks that no uplink transmission on CCCH is detected.

Specific Message Contents

SYSTEM INFORMATION (master information block)

The contents of a SYSTEM INFORMATION message in this test case is identical to the corresponding message found in the default contents of layer 3 messages for RRC tests, with the following exceptions:

Information Element	Value/remark
Supported PLMN Types	invalid value which is not defined spare value

8.1.2.8.5 Test requirement

After step 3 the UE shall not transmit an RRC CONNECTION REQUEST message on the uplink CCCH.

8.1.2.9 RRC Connection Establishment: Success after Physical channel failure

8.1.2.9.1 Definition

8.1.2.9.2 Conformance requirement

- 1. The RRC connection establishment is initiated by the UE, which leaves the idle mode. The UE shall transmit an RRC CONNECTION REQUEST message which includes the IE "Initial UE identity". This message shall be sent on the uplink CCCH.
- 2. In the case of a failure to establish the RRC connection at the physical channel failure after the UE receives an RRC CONNECTION SETUP message, the UE retries to establish the RRC connection until V300 is greater than N300

Reference

3GPP TS 25.331 clause 8.1.3

8.1.2.9.3 Test purpose

To confirm that the UE retries to establish the RRC connection until V300 is greater than N300 for the physical channel failure as the SS does not configure the physical channel which is specified in the transmitted RRC CONNECTION SETUP message.

8.1.2.9.4 Method of test

Initial Condition

System Simulator: 1 cell

<u>UE</u>: Idle state (state 2 or state 3 or state 7) -as specified in clause 7.4 of TS 34.108 ,depending on the CN domain(s) supported by the <u>UE</u>

Test Procedure

Before the test starts, an internal counter K in SS is initialised to a value = 1. Following this, the UE shall transmit an RRC CONNECTION REQUEST message to the SS on the uplink CCCH, after the operator attempts to make an outgoing call. SS transmits an RRC CONNECTION SETUP message to make the UE configure the physical channel in order to communicate on the DCCH but SS does not configure the physical channel. Then the UE detects the physical channel failure and transmits an RRC CONNECTION REQUEST message. SS increments K every time such a message is received. This cycle is repeated until K reaches N300. When K is equal to N300, the SS transmits the RRC CONNECTION SETUP message and configures the physical channel. The UE shall detect "in-sync" from physical layer and then acknowledge the establishment of RRC connection by sending the RRC CONNECTION SETUP COMPLETE message on uplink DCCH.

Expected sequence

Step	<u>Direction</u>	<u>Message</u>	<u>Comment</u>
	UE SS		
<u>1</u>			SS initialises counter K to 1. Operator is asked to make an outgoing call and SS starts to wait for RRC CONNECTION REQUEST on uplink CCCH.
2	\rightarrow	RRC CONNECTION REQUEST	<u> </u>
<u>2</u> <u>3</u>	_		SS checks to see if K is equal to N300+1. If so, goes to step 6. Else, continues to execute step 4.
4	€	RRC CONNECTION SETUP	See the clause 9 in TS 34.108 on default message content for RRC. But SS does not configure the physical channel.
<u>5</u>			SS increments K. The next step is step 2.
<u>6</u>	<u> </u>	RRC CONNECTION SETUP	See the clause 9 in TS 34.108 on default message content for RRC. SS configures the physical channel.
<u>7</u>			The UE configures the layer 1 and layer 2.
<u>8</u>	<u> </u>	RRC CONNECTION SETUP COMPLETE	

Specific Message Contents

None

8.1.2.9.5 Test requirement

After step 4 the UE shall re-send RRC CONNECTION REQUEST message.

After step 8 the UE shall transmit an RRC CONNECTION SETUP COMPLETE message and establish an RRC connection.

8.1.3 RRC Connection Release

8.1.3.1 RRC Connection Release in CELL DCH state: Success

8.1.3.1.1 Definition

8.1.3.1.2 Conformance requirement

In case of an RRC connection release from CELL_DCH state, the UTRAN transmits an RRC CONNECTION RELEASE message to the UE using unacknowledged mode on the DCCH. The UE then responds by transmitting an RRC CONNECTION RELEASE COMPLETE message using unacknowledged mode to UTRAN for N308 times, each time at the expiry of T308 timer. Then the UE leaves the RRC connected mode and initiates release of the layer 2 signalling link. The RRC Connection Release procedure ends when all UE dedicated resources (such as radio resources and radio access bearers) associated with the RRC connection are released and the UE returns to idle mode.

Reference

3GPP TS 25.331 clause 8.1.4

8.1.3.1.3 Test purpose

To confirm that the UE releases the L2 signalling link and dedicated resources and goes back to the idle state after it receives an RRC CONNECTION RELEASE message from the SS and transmits an RRC CONNECTON RELEASE COMPLETE message to the SS for N308 times at the interval specified by the value of T308 timer.

8.1.3.1.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CELL_DCH state (state 6-1 or state 6-3) as specified in clause 7.4 of TS 34.108 depending on the CN domain(s) supported by the UE

Test Procedure

The UE is brought to the CELL_DCH state by prompting the operator to initiate an outgoing call. After the DCCH is established, SS transmits an RRC CONNECTION RELEASE message to the UE to disconnect the connection. SS then waits for the UE to transmit an RRC CONNECTION RELEASE COMPLETE message using unacknowledged mode. SS checks to see if P such messages has been received at each expiry of T308 timer. P is equal to the value of IE "Number of RRC Message Transmissions" in an RRC CONNECTION RELEASE message.

Step	Direction		Message	Comment
	UE	SS		
1				The UE is in the CELL_DCH state after a successful RRC connection establishment by virtue of the operator making an outgoing call.
2		(RRC CONNECTION RELEASE	SS disconnect the connection established. The value in IE "Number of RRC Message Transmissions" is arbitrarily chosen from 4 to 8 and denoted by P.
3	-	→	RRC CONNECTION RELEASE COMPLETE	SS waits for the arrival of N308 such message at the expiry of each T308 timer, using unacknowledged mode.
4				The UE releases L2 signalling link and dedicated resources. Then the UE goes to idle mode.

Specific Message Content

RRC CONNECTION RELEASE (Step 2)

Information Element	Value/remark	
Number of RRC Message Transmission	Arbitrarily chosen between 4 and 8	

8.1.3.1.5 Test requirement

After step 2 the UE shall start to transmit P times RRC CONNECTION RELEASE COMPLETE messages at the expiry of each T308 timer.

After step 3 the UE shall initiate the release L2 signalling link and dedicated resources, then it shall go to idle mode.

8.1.3.2 RRC Connection Release using on DCCH in CELL FACH state: Success

8.1.3.2.1 Definition

8.1.3.2.2 Conformance requirement

In CELL_FACH state, the RRC layer entity in the network may issue an RRC CONNECTION RELEASE message using unacknowledged mode on the DCCH. Upon the reception of this message, the UE transmits an RRC CONNECTION RELEASE COMPLETE message using acknowledged mode to UTRAN on the DCCH and goes back to idle mode after it receives an RLC confirmation from the UTRAN.

Reference

3GPP TS 25.331 clause 8.1.4

8.1.3.2.3 Test purpose

To confirm that the UE releases the L2 signalling link and resources and goes back to the idle state after it receives an RRC CONNECTION RELEASE message on downlink DCCH from the SS. It shall transmit an RRC CONNECTON RELEASE COMPLETE message using acknowledged mode on uplink DCCH to the SS.

8.1.3.2.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CELL_FACH state (state 6-2 or state 6-4) as specified in clause 7.4 of TS 34.108 depending on the CN domain(s) supported by the UE

Test Procedure

The UE is brought to an initial state of CELL_FACH. After the successful establishment of the RRC connection, the SS transmits an RRC CONNECTION RELEASE message to the UE to disconnect the radio link. When the UE receives this message the UE transmits an RRC CONNECTION RELEASE COMPLETE message using acknowledged mode to the SS. Finally, SS checks that the UE performs proper release of all radio resources and then goes back to idle mode.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1				The UE is brought to the
				CELL_FACH state.
2	•	-	RRC CONNECTION RELEASE	SS sends this message using
				unacknowledged mode RLC
				operations on the uplink DCCH.
3		\rightarrow	RRC CONNECTION RELEASE	The UE transmits this message
			COMPLETE	using acknowledged mode.
4				The UE releases L2 signalling
				link and radio resources. Then
				the UE goes to idle mode.

Specific Message Contents

None.

8.1.3.2.5 Test requirement

After step 2 the UE shall transmit an RRC CONNECTION RELEASE COMPLETE message using acknowledged mode then it shall receive a response for this message from the SS-RLC.

After step 3 the UE shall release its L2 signalling link and radio resources, then it shall go back to idle mode.

8.1.3.3 RRC Connection Release using on CCCH in CELL_FACH state: Success

8.1.3.3.1 Definition

8.1.3.3.2 Conformance requirement

In CELL_FACH state, the RRC layer entity in the network may issue an RRC CONNECTION RELEASE message using unacknowledged mode on the CCCH. Upon the reception of this message, the UE shall release the RRC connection immediately, without replying with a RRC CONNECTION RELEASE COMPLETE message on the uplink.

Reference

3GPP TS 25.331 clause 8.1.4

8.1.3.3.3 Test purpose

To confirm that the UE releases all its radio resources upon the reception of a RRC CONNECTION RELEASE message on the downlink CCCH, without transmitting RRC CONNECTION RELEASE COMPLETE message on the uplink.

8.1.3.3.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CELL_FACH state (state 6-2 or state 6-4) as specified in clause 7.4 of TS 34.108, depending on the CN domain(s) supported by the UE

Test Procedure

The UE is brought to an initial state of CELL_FACH. After the successful establishment of the RRC connection, SS transmits RRC CONNECTION RELEASE message on the downlink CCCH. The UE shall terminate the RRC connection and release all radio resources allocated to it. SS monitors the uplink DCCH and CCCH to verify that no transmission is detected.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1				The UE is brought to the
				CELL_FACH state.
2	+	-	RRC CONNECTION RELEASE	SS transmits this message with
				the contents identical to that
				found in TS 34.108 clause 9 on
				downlink CCCH.
3				SS waits for a period equivalent
				to (N308+1) times T308 timer
				expiry. The UE shall not send
				any response message on uplink
				direction during this period. It
				shall release the radio resources
				allocated and return to idle
				mode.

Specific Message Contents

None.

8.1.3.3.5 Test requirement

After step 2 the UE shall release all its radio resources, return to idle mode, without transmitting RRC CONNECTION RELEASE COMPLETE message on the uplink direction.

8.1.3.4 RRC Connection Release in CELL FACH state: Failure

8.1.3.4.1 Definition

8.1.3.4.2 Conformance requirement

In case of RRC connection release from CELL_FACH state, the RRC layer entity in the network issues an RRC CONNECTION RELEASE message using unacknowledged mode on the DCCH. When the UE does not succeed to transmit the RRC CONNECTION RELEASE COMPLETE message using acknowledged mode, it shall release all its radio resources, enter idle mode and the procedure ends on the UE side.

Reference

3GPP TS 25.331 clause 8.1.4

8.1.3.4.3 Test purpose

To confirm that the UE releases all its radio resources and enters idle mode when the UE does not succeed in transmitting the RRC CONNECTION RELEASE COMPLETE message using acknowledged mode to the SS (i.e. the UE-RLC cannot receive acknowledgement for the transmission of the RRC CONNECTION RELEASE COMPLETE message from SS.).

8.1.3.4.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CELL_FACH state (state 6-2 or state 6-4) as specified in clause 7.4 of TS 34.108 depending on the CN domain(s) supported by the UE

Test Procedure

At the start of the test, the UE is brought to CELL_FACH state. When the RRC connection has been established, the SS transmits an RRC CONNECTION RELEASE message to the UE to disconnect the radio link. When the UE receives this message the UE transmits an RRC CONNECTION RELEASE COMPLETE message using acknowledged mode to the SS. The SS ignores the message and does not transmit a STATUS PDU of RLC for this message. SS checks to see that UE continues to release all its radio resources and then enters idle mode.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1				The UE is brought into CELL_FACH state by asking the operator to perform an outgoing call attempt. clause
2	+	-	RRC CONNECTION RELEASE	SS ask to disconnect the radio link
3	-	>	RRC CONNECTION RELEASE COMPLETE	The UE transmits this message using acknowledged mode. The SS ignores this message and shall not transmit a STATUS PDU of RLC for this message.
4				SS checks to make sure that UE releases its all radio resources and enter idle mode.

Specific Message Contents

None

8.1.3.4.5 Test requirement

After step 3 the UE shall release its L2 signalling link and radio resources then it shall go to idle mode.

8.1.3.5 RRC Connection Release in CELL_FACH state: Invalid message

8.1.3.5.1 Definition

8.1.3.5.2 Conformance requirement

In CELL_FACH state, the RRC layer entity in UTRAN may issue an RRC CONNECTION RELEASE message using unacknowledged mode on the DCCH. If an invalid RRC CONNECTION RELEASE message is received by the UE, the UE shall activate the appropriate error-handling mechanism and report the error to the UTRAN. After this, the UE shall release the RRC connection.

Reference

3GPP TS 25.331 clause 8.1.4

8.1.3.5.3 Test purpose

When the UE receives an invalid RRC CONNECTION RELEASE message on the downlink DCCH, it shall transmit an RRC CONNECTON RELEASE COMPLETE message that includes the appropriate error cause on the uplink DCCH. Thereafter, it shall release the RRC connection.

8.1.3.5.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CELL_FACH state (state 6-2 or state 6-4) as specified in clause 7.4 of TS 34.108, depending on the CN domain(s) supported by the UE

Test Procedure

The UE is brought to an initial state of CELL_FACH. SS transmits an RRC CONNECTION RELEASE message on the DCCH to request to disconnect the RRC connection. However, the message contains an spare invalid value in the IE "Release cause". As a result, the UE shall transmit an RRC CONNECTION RELEASE COMPLETE message on the uplink DCCH, which includes the IE "Error indication". This IE shall contain "Failure cause" IE which is set to "Protocol error" and "Protocol error information" IE which is set to "Information element value not comprehended". The UE shall release the RRC connection and go back to idle mode after transmitting the RRC CONNECTION RELEASE COMPLETE message.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1				The UE is brought to the
				CELL_FACH state.
2	+	•	RRC CONNECTION RELEASE	See specific message contents
				for this message
3	\rightarrow		RRC CONNECTION RELEASE	The IE "Protocol error cause"
			COMPLETE	found in IE "Protocol error
				information" shall be set to
				"Information element value not
				comprehended".
4				The UE shall release the
				signalling link and radio
				resources, and then return to idle
				mode.

Specific Message Contents

RRC CONNECTION RELEASE (Step 2)

Information Element	Value/remark
Release cause	Contains a spare value in this leinvalid value

RRC CONNECTION RELEASE COMPLETE

Information Element	Value/remark
Error Indication	
Failure cause	Protocol error
Protocol error information	
Protocol error cause	Information element value not comprehended

8.1.3.5.5 Test requirement

After step2 the UE shall transmit an RRC CONNECTION RELEASE COMPLETE message which includes the appropriate cause values in IE "Error Indication".

After step3 the UE shall release its L2 signalling link and radio resources, then it shall go back to idle mode.

8.1.4 <u>VoidRRC Connection Re-Establishment</u>

8.1.4.1 RRC Connection Re-Establishment: Success

8.1.4.1.1 Definition

8.1.4.1.2 Conformance requirement

RRC connection re establishment is needed, when a UE loses the radio connection due to e.g. radio link failure in CELL_DCH state. After a successful cell re-selection and transiting to CELL_FACH state, the UE transmits an RRC CONNECTION RE ESTABLISHMENT REQUEST message which includes the IE "U RNTI" on the uplink CCCH to request for the re-establishment of an RRC connection. The UTRAN transmits an RRC CONNECTION RE-ESTABLISHMENT message on the downlink DCCH, which includes the new configuration information. Then the UE transmits an RRC CONNECTION RE-ESTABLISHMENT COMPLETE message on the uplink DCCH using AM RLC.

Reference

3GPP TS 25.331 clause 8.1.5

8.1.4.1.3 Test purpose

To confirm that the UE tries to find a new cell by transiting to CELL_FACH state, after detecting that a radio link failure has occurred. The UE shall then attempt to establish a new RRC connection using an RRC CONNNECTION RE ESTABLISHMENT REQUEST message which includes the IE "U RNTI". Finally when the SS allocates a new configuration to the UE using RRC CONNECTION ESTABLISHMENT message, the UE shall establish a new RRC connection link.

8.1.4.1.4 Method of test

Initial Condition

System Simulator: 2 cells (Cell 1 is active, Cell 2 is inactive)

UE: CELL_DCH state (state 6 9 or state 6 10) as specified in clause 7.4 of TS 34.10 in cell 1

Test Procedure

The UE is brought to CELL_DCH state in a cell 1 after a successful outgoing call attempt. After the call has been established, the SS stops transmitting and receiving in a cell No.1 and begins to broadcast the BCCH in cell 2, the UE should detect a radio link failure in cell 1 and attempts to re select to cell 2. It should then enter CELL_FACH state and transmits an RRC CONNECTION RE ESTABLISHMENT REQUEST message which includes the IE "U RNTI" on the uplink CCCH to SS. The SS transmits an RRC CONNECTION RE ESTABLISHMENT message which includes a new configuration information. After the UE receives an RRC CONNECTION RE ESTABLISHMENT message, it reconfigures the new radio connection and transmits an RRC CONNECTION RE ESTABLISHMENT COMPLETE message on the uplink DCCH using AM RLC.

Expected sequence

Step	Direc	ction	Message Message	Comment
-	UE	SS		
4				The UE is brought to
				CELL_DCH state in a cell
				1, after making a
				successful outgoing call.
2		(BCCH	The SS starts transmitting
				the BCCH in a cell 2 using
				the same contents
				(except for cell identity
				which is set to "0000 0000
				0000 0010") for system
				information sent on cell 1.
				SS starts to listen to the
				uplink CCCH of cell 2.
3				The SS stops transmitting
				and receiving in a cell .1.
4		\rightarrow	RRC CONNECTION	The UE should find a new
			RE-ESTABLISHMENT REQUEST	cell 2 and then transmits
				this message which
				includes the IE "U-RNTI"
				assigned previously.
5	- ←		RRC CONNECTION	Including the new
			RE-ESTABLISHMENT	configuration information.
6		}	RRC CONNECTION	
			RE-ESTABLISHMENT COMPLETE	

Specific Message Contents

Cell 1: SYSTEM INFORMATION TYPE 2

Information Element	Value/remark
UE Timers and constants in connected mode	
T301	8 seconds
T313	15 seconds
T314	20 seconds
T315	30 seconds
N313	50

8.1.4.1.5 Test requirement

After step 3, the UE shall detect the presence of cell 2 and attempts to re established the lost RRC connection by transmitting RRC CONNECTION RE ESTABLISHMENT REQUEST in the new cell.

After step 5, the UE shall re establish an RRC connection and observe the new configuration information specified in an RRC CONNECTION RE ESTABLISHMENT message.

8.1.4.2 RRC Connection Re-Establishment: Success after T301 timeout (T314 and T315 are running)

8.1.4.2.1 Definition

8.1.4.2.2 Conformance requirement

- 1. RRC connection re establishment is needed when a UE loses the radio connection due to e.g. radio link failure in CELL_DCH state. After a cell re-selection to a new cell, the UE transits to CELL_FACH state. In this state, the UE transmits an RRC CONNECTION RE ESTABLISHMENT REQUEST message which includes the IE "U-RNTI" on the uplink CCCH to request the re establishment of an RRC connection.
- 2. In the case of a failure to re establish the RRC connection after the expiry of timer T301, the UE retries to reestablish the RRC connection. Finally when the UTRAN allocates a new configuration to the UE using RRC CONNECTION ESTABLISHMENT message, the UE shall establish a new RRC connection link.

Reference

3GPP TS 25.331 clause 8.1.5

8.1.4.2.3 Test purpose

To confirm that the UE retries to re-establish a new RRC connection after the expiry of timer T301. This occurs after the UE loses the radio connection and starts to transmit an RRC CONNECTION RE ESTABLISHMENT REQUEST message.

8.1.4.2.4 Method of test

Initial Condition

System Simulator: 2 cells (Cell 1 is active, Cell 2 is inactive)

UE: CELL_DCH state (state 6 9 or state 6 10) as specified in clause 7.4 of TS 34.108 in cell 1

Test Procedure

The UE is in the CELL_DCH state in cell 1, after making a successful outgoing call attempt. SS begins to broadcast the BCCH in cell 2 and then stops transmitting and receiving in cell 1. The UE shall detect a radio link failure and the presence of cell 2. It then enters CELL_FACH state and transmits to SS an RRC CONNECTION RE-ESTABLISHMENT REQUEST message which includes the IE "U RNTI" on the uplink CCCH. The SS does not transmit an RRC CONNECTION RE ESTABLISHMENT to answer to the request. This causes T301 timer to expire and the UE re transmits an RRC CONNECTION RE ESTABLISHMENT REQUEST message again. Then the SS transmits an RRC CONNECTION RE ESTABLISHMENT message and the UE reconfigures the new radio connection and transmits an RRC CONNECTION RE ESTABLISHMENT COMPLETE message on the uplink DCCH using AM RLC.

Step	Direc	tion	Message	Comment
	UE	SS	_	
4				SS prompts the operator to
				make an outgoing call. The
				UE is in the CELL_DCH
				state in cell 1.
2		(BCCH	The SS starts to transmit
				BCCH in a cell 2, with the
				SYSTEM INFORMATION
				message similar (with the
				exception of cell identity
				IE) to that transmitted in
				cell 1.
3				The SS stops transmitting
				and receiving in cell 1.
4		\rightarrow	RRC CONNECTION	The UE finds cell 2 and
			RE-ESTABLISHMENT REQUEST	transmits this message
				which includes the IE "U-
				RNTI".
5				The SS does not transmit
				a response.
6		\rightarrow	RRC CONNECTION	After the expiry of T301
			RE-ESTABLISHMENT REQUEST	the UE re-transmits this
				message.
7		←	RRC CONNECTION	Including the new
			RE-ESTABLISHMENT	configuration information.
8		→	RRC CONNECTION	
			RE-ESTABLISHMENT COMPLETE	

Specific Message Contents

None

8.1.4.2.5 Test requirement

After step 3, the UE shall detect presence of cell 2 and starts to transmit an RRC CONNECTION RE-ESTABLISHMENT REQUEST message to setup the radio link again.

After step 5, the UE shall re-transmit an RRC CONNECTION RE-ESTABLISHMENT REQUEST message on the uplink CCCH after the expiry of timer T301.

After step 7 the UE shall re establish an RRC connection in cell 2, using the new configuration commanded by the SS.

8.1.4.3 RRC Connection Re-Establishment: Success after reception of invalid message (V301 is not greater than N301)

8.1.4.3.1 Definition

8.1.4.3.2 Conformance requirement

- 1. RRC connection re establishment is needed when a UE loses the radio connection due to e.g. radio link failure in CELL_DCH state. After a cell re selection to a new cell, the UE transits to CELL_FACH state. In this state, the UE transmits an RRC CONNECTION RE ESTABLISHMENT REQUEST message which includes the IE "U-RNTI" on the uplink CCCH to request the re establishment of an RRC connection.
- 2. In the case of a reception of an invalid RRC CONNECTION RE ESTABLISHMENT message, the UE retries to re-establish the RRC connection until up to a maximum of N301 attempts.

Reference

3GPP TS 25.331 clause 8.1.5

8.1.4.3.3 Test purpose

To confirm that the UE retries to re establish a new RRC connection until its internal variable V301 is greater than N301 when the UE receives an invalid RRC CONNECTION RE ESTABLISHMENT message.

8.1.4.3.4 Method of test

Initial Condition

System Simulator: 2 cells (Cell 1 is active, Cell 2 is inactive)

UE: CELL_DCH state (state 6-9 or state 6-10) as specified in clause 7.4 of TS 34.108 in cell 1

Test Procedure

Before the test starts, an SS internal counter K is initialized to 1. The UE is in the CELL_DCH state in cell 1, after making a successful outgoing call attempt. SS begins to broadcast the BCCH in cell 2 and then stops transmitting and receiving in cell 1. The UE shall detect a radio link failure and the presence of cell 2. It then enters CELL_FACH state and transmits to SS an RRC CONNECTION RE ESTABLISHMENT REQUEST message which includes the IE "URNTI" on the uplink CCCH. The SS transmits an RRC CONNECTION RE ESTABLISHMENT message to answer to the request which includes the "DRX Indicator" IE set to "spare value". SS increments internal counter K. After receiving this invalid message, the UE shall re transmit an RRC CONNECTION RE ESTABLISHMENT REQUEST message again. After the reception of each RRC CONNECTION RE ESTABLISHMENT REQUEST message on the uplink CCCH, SS checks the value of K. If K is less than (N301+1), SS re-sends the same invalid RRC CONNECTION RE ESTABLISHMENT message again on the downlink DCCH and then increments the internal counter K. When K is equal to N301+1 then the SS transmits a valid RRC CONNECTION RE ESTABLISHMENT message and the UE shall reconfigure the new radio connection and transmits an RRC CONNECTION RE ESTABLISHMENT COMPLETE message on the uplink DCCH using AM RLC.

Step	Direc	tion	Message	Comment	
	UE	SS			
4				SS sets internal counter K	
				to 1 and prompts the	
				operator to make an	
				outgoing call. The UE is in	
				the CELL_DCH state in	
				cell 1.	
2	€	-	BCCH	The SS starts to transmit	
				BCCH in a cell 2, with the	
				SYSTEM INFORMATION	
				message similar (with the	
				exception of cell identity	
				IE) to that transmitted in	
				cell 1.	
3				The SS stops transmitting	
				and receiving in cell 1.	
4		>	RRC CONNECTION	The UE finds cell 2 and	
			RE-ESTABLISHMENT REQUEST	transmits this message	
				which includes the IE "U-	
				RNTI".	
5		(RRC CONNECTION	The SS transmits a	
			RE-ESTABLISHMENT	response which includes	
				the "DRX indicator" IE set	
				to "spare value" and	
				increments K.	
6		>	RRC CONNECTION	The UE transmits this	
			RE-ESTABLISHMENT REQUEST	message which includes	
				the IE "U-RNTI".	
7				If K is less than or equal to	
				N301 then jump back to	
				Step 5, otherwise proceed	
		,	DDG GGVINEGTIGN	with step 8.	
8		-	RRC CONNECTION	Including the new	
			RE-ESTABLISHMENT	configuration information.	
9)	RRC CONNECTION		
			RE-ESTABLISHMENT COMPLETE		

Specific Message Contents

RRC CONNECTION RE-ESTABLISHMENT REQUEST (Step 4)

The content of this message shall be identical to the expected default message specified in Annex A.

RRC CONNECTION RE-ESTABLISHMENT (Step 5)

The content of this message is identical to the message sub-type titled "Speech in CS", specified in Annex A with the following exceptions.

Information Element	Value/remark
"DRX Indicator"	'spare value'

RRC CONNECTION RE-ESTABLISHMENT REQUEST (Step 6)

Information Element	Value/remark
"Protocol error indicator"	TRUE
"Protocol error information"	'Information element value not comprehended'

RRC CONNECTION RE-ESTABLISHMENT (Step 8)

The content of this message is identical to the message sub-type titled "Speech in CS", specified in Annex A.

8.1.4.3.5 Test requirement

After step 3, the UE shall detect presence of cell 2 and starts to transmit RRC CONNECTION RE ESTABLISHMENT REQUEST message to setup the radio link again.

After step 5, the UE shall re transmit an RRC CONNECTION RE ESTABLISHMENT REQUEST message on the uplink CCCH. The total number of RRC CONNECTION RE ESTABLISHMENT messages received by SS shall be equal to N301+1.

After step 8 the UE shall re-establish an RRC connection in cell 2, using the new configuration commanded by the SS.

8.1.4.4 RRC Connection Re-Establishment: Failure after reception of invalid message (V301 is greater than N301)

8.1.4.4.1 Definition

8.1.4.4.2 Conformance requirement

- 1. RRC connection re establishment is needed when a UE loses the radio connection due to e.g. radio link failure in CELL_DCH state. After cell re-selection and then transiting to CELL_FACH state, the UE RRC transmits an RRC CONNECTION RE ESTABLISHMENT REQUEST message which includes the IE "U RNTI" on the uplink CCCH to request for re establishment of an RRC connection.
- 2. In the case of a reception of an invalid RRC CONNECTION RE ESTABLISHMENT message, if the UE retries to re establish the RRC connection until up to a maximum of N301 attempts and receives the invalid message, the UE goes back to idle mode.

Reference

3GPP TS 25.331 clause 8.1.5

8.1.4.4.3 Test purpose

To confirm that the UE stops retrying to re establish the RRC connection if its internal counter V301 is greater than N301 and then goes back to idle state.

8.1.4.4.4 Method of test

Initial Condition

System Simulator: 2 cells (Cell 1 is active, Cell 2 is inactive)

UE: CELL_DCH state (state 6 9 or state 6 10) as specified in clause 7.4 of TS 34.108 in cell 1

Test Procedure

The UE is in the CELL_DCH state in a cell 1. SS begins to broadcast the BCCH in cell 2, and then stops transmitting and receiving in cell 1. The UE shall detect a radio link failure and the presence of cell 2. It then enters CELL_FACH state and transmits to SS an RRC CONNECTION RE ESTABLISHMENT REQUEST message which includes the IE "U-RNTI" on the uplink CCCH. The SS transmits an RRC CONNECTION RE-ESTABLISHMENT message to answer to the request which includes the "DRX Indicator" IE set to "spare value", resulting in a protocol error and the UE shall re-transmit an RRC CONNECTION RE-ESTABLISHMENT REQUEST message again. The SS keeps a count of the number of RRC CONNECTION RE-ESTABLISHMENT REQUEST messages received. The UE shall go back to idle mode after making exactly N301 re-transmission attempts.

Step	Direction		Message	Comment
	UE	SS]	
4				SS set counter K to 1. The UE is in CELL_DCH state in a cell 1 after establishing an outgoing call.
2		(BCCH	The SS starts to transmit BCCH in a cell 2.
3				The SS stops transmitting and receiving in a cell 1.
4	_	→	RRC CONNECTION RE-ESTABLISHMENT REQUEST	The UE finds new cell 2 and transmits this message which includes the IE "U-RNTI".
5	+	(-	RRC CONNECTION RE-ESTABLISHMENT	The SS transmits a response which includes the "DRX indicator" IE set to "spare value" and increments K.
6	_	>	RRC CONNECTION RE-ESTABLISHMENT REQUEST	The UE transmits this message which includes the IE "U-RNTI".
7				If K is less than or equal to N301 then jump back to Step 5, otherwise proceed with step 8.
8	*	<u> </u>	RRC CONNECTION RE-ESTABLISHMENT	The SS transmits a response which includes the "DRX Indicator" IE set to "spare value"
9				SS waits for a period equivalent to time T301 time-out value. The UE shall not transmit RRC CONNECTION RE- ESTABLISHMENT REQUEST message on the uplink CCCH during this period. The UE shall go back to idle mode.

Specific Message Contents

RRC CONNECTION RE-ESTABLISHMENT REQUEST (Step 4)

The content of this message shall be identical to the expected default message specified in Annex A.

RRC CONNECTION RE-ESTABLISHMENT (Steps 5 and 8)

Information Element	<u>Value/remark</u>
DRX Indicator	spare value

RRC CONNECTION RE-ESTABLISHMENT REQUEST (Step 6)

Information Element	Value/remark
Protocol error indicator	TRUE
Protocol error information	Information element value not comprehended

8.1.4.4.5 Test requirement

After step 3 the UE shall find the presence of cell 2, enters CELL_FACH state, and starts to transmit RRC CONNECTION RE-ESTABLISHMENT REQUEST message to setup the radio link again.

After step 5 the UE shall transmit an RRC CONNECTION RE ESTABLISHMENT REQUEST message on the uplink CCCH until V301 is greater than N301.

After step 8, when V301 is greater than N301 the UE shall stop sending RRC CONNECTION RE ESTABLISHMENT REQUEST on the uplink CCCH and then go back to idle mode.

8.1.4.5 RRC Connection Re-Establishment: Failure (Release)

8.1.4.5.1 Definition

8.1.4.5.2 Conformance requirement

RRC connection re establishment is needed, when a UE loses the radio connection due to e.g. radio link failure in CELL_DCH state. After the completion of cell re selection and transiting to CELL_FACH state, the UE transmits an RRC CONNECTION RE ESTABLISHMENT REQUEST message which includes the IE "U RNTI" on the uplink CCCH to request for re establishment of an RRC connection. The UTRAN transmits an RRC CONNECTION RELEASE message signifying that it is not able to accept the request, so the UE shall go back to idle mode.

Reference

3GPP TS 25.331 clause 8.1.5

8.1.4.5.3 Test purpose

To confirm that the UE transmits an RRC CONNECTION RELEASE COMPLETE message and goes back to idle mode after its request to re-establish an RRC connection was rejected by the SS in the new cell. The request to re-establish should be made with an RRC CONNECTION RE-ESTABLISHMENT REQUEST message which includes the IE "U-RNTI".

8.1.4.5.4 Method of test

Initial Condition

System Simulator: 2 cells Cell 1 is active, Cell 2 is inactive

UE: CELL DCH state (state 6 9 or state 6 10) as specified in clause 7.4 of TS 34.108 in cell 1

Test Procedure

The UE starts from CELL_DCH state in cell 1 after making a successful outgoing call. When the SS stops transmitting and receiving in cell 1 and begins to broadcast the BCCH in cell 2, the UE should detect radio link failure and the presence of cell 2. It then enters CELL_FACH state and transmits an RRC CONNECTION RE ESTABLISHMENT REQUEST message, which includes the IE "U RNTI" on the uplink CCCH. The SS replies with a RRC CONNECTION RELEASE message on the downlink CCCH using transport mode operation. Finally, it goes back to idle mode.

Step	Direc	ction	Message	Comment
	UE	SS		
4				The UE is in the CELL_DCH state in cell 1 after completing a successful outgoing call setup.
2	_	-	BCCH	The SS transmits BCCH in cell 2
3				The SS stops transmitting and receiving in cell 1.
4		>	RRC CONNECTION RE-ESTABLISHMENT REQUEST	The UE finds new cell 2 and transmits this message which includes the IE "U-RNTI".
5		(RRC CONNECTION RELEASE	SS rejects the re- establishment request. Sent on downlink CCCH using RLC-TM.
6				The UE goes back to idle.

Specific Message Contents

RRC CONNECTION RELEASE: TM

Information Element	Value/remark
Message Type	
U-RNTI	Not Present
- SRNC Identity	Set to the same SRNC identity as previously assigned
- S-RNTI	Set to the same S-RNTI as previously assigned
Integrity check info	Not Checked
Number of RRC Message Transmissions	Not Present
Release cause	Check to see if set to "Re-establishment reject"

8.1.4.5.5 Test requirement

After step 3 the UE shall find the new cell 2 and transmits RRC CONNECTION RE ESTABLISHMENT REQUEST to ask for re-connection of the radio link.

After step 5 the UE shall go back to idle mode.

8.1.4.6 RRC Connection Re-Establishment: Failure(T315=0,T314=0)

8.1.4.6.1 Definition

8.1.4.6.2 Conformance requirement

RRC connection re establishment is needed, when a UE loses the radio connection due to e.g. radio link failure in CELL_DCH state. However the UE must enter to idle mode when T314 is set to 0 in RADIO BEARER SETUP message and also T315 is set to 0 in SYSTEM INFORMATION message.

Reference

3GPP TS 25.331 clause 8.1.5

8.1.4.6.3 Test purpose

To confirm that the UE enters to idle mode, after detecting that a radio link failure has occurred.

8.1.4.6.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: idle state (state 2 or state 3) as specified in clause 7.4 of TS 34.108

Test Procedure

The MASTER INFORMATION BLOCK and SYSTEM INFORMATION BLOCK TYPE 2 messages on the downlink BCCH are modified with respect to the default settings. The UE is brought to CELL_DCH state, after successfully executing the mobile originated RRC connection establishment procedure (i.e. P3+P7 or P2+P5) as outlined in clause 7.4 in TS 34.108. Next, SS transmits RADIO BEARER SETUP message on the downlink DCCH in order to establish radio bearers for user data on DTCH. In this message, the re establishment timers T314 and T315 are both set to 0. The UE shall respond by sending a RADIO BEARER SETUP COMPLETE message on the uplink DCCH. After the DTCH has been established, the SS stops transmitting and receiving on the radio link except P CCPCH, the UE should detect a radio link failure. Then the UE shall enter to idle mode as both T314 and T315 are set to 0. The SS transmits a PAGING TYPE 1 message on the PCCH to confirm that the UE is in idle state., the UE transmits an RRC CONNECTION REQUEST message on the uplink CCCH. SS replies with an RRC CONNECTION SETUP message and allocates dedicated channels to the UE. Then, the UE shall complete this test by transmitting RRC CONNECTION SETUP COMPLETE message on the uplink DCCH.

Step	Direction	Message	Comment
-	UE SS	1	
4	- \	MASTER INFORMATION BLOCK SYSTEM INFORMATION BLOCK TYPE 2	T315=0
2			RRC connection establishment procedure (i.e. P3+P7 or P5+P9) as outlined in clause 7.4 in TS 34.108.
3	- →	RADIO BEARER SETUP	T314=0
4	_→	RADIO BEARER SETUP COMPLETE	
5			The UE is brought to CELL_DCH state, after making a successful outgoing call.
6			The SS stops transmitting and receiving on the radio link except the PCCPCH.
7			The UE should enter to idle mode after it detects the radio link failure and SS confirms that the UE does not transmit any RRC CONNECTION RESTABLISHMENT REQUEST messages for 20 seconds.
8	- +	PAGING TYPE 1	SS resumes transmission and reception on radio links suspended in step 4. SS pages the UE using a matched identity (test- SIM IMSI).
9	>	RRC CONNECTION REQUEST	SS checks the following IEs: "Initial UE identity" — must be the same as test-SIM IMSI. "Establishment Cause — must "Terminating Call" "Protocol Indicator" — must be FALSE
10	+	RRC CONNECTION SETUP	SS assigns DPCH resources to allow UE to establish a RRC connection.
11	->	RRC CONNECTION SETUP COMPLETE	UE shall acknowledge the completion of RRC connection establishment procedure.

Specific Message Contents

MASTER INFORMATION BLOCK (Step 1)

Use the same message type in clause 6.1 of TS 34.108, with the following exception:

Information Element	Value/remark
MIB Value Tag	2

SYSTEM INFORMATION BLOCK TYPE 2 (Step 1)

Use the same message type in clause 6.1 of TS 34.108, with the following exceptions:

Information Element	Value/remark
UE Timers and constants in connected mode	
T301	3 seconds
T313	15 seconds
T314	20 seconds
T315	0 seconds
N313	50

RADIO BEARER SETUP (Step 3)

For circuit switch only UEs that do not support speech service, use the same message sub-type titled "Non-speech in CS" in Annex A, with the following exceptions:

For circuit switch only UEs that support speech service, use the same message sub-type titled "Speech in CS" in clause 9 of TS 34.108, with the following exceptions:

For all other types of UEs, use the same message sub-type titled "Packet to CELL_DCH from CELL_DCH in PS" in Annex A, with the following exceptions:

Information Element	Value/remark
RAB information to setup list	
— - RAB information for setup	
——- RAB info	
- RAB identity	4
	One of the CN domains supported by the UE
	T314
	θ

PAGING TYPE 1 (Step 8)

Information Element	Value/remark
Paging record list	Only 1 entry
Paging record	
CHOICE Paging originator	CN originator
Paging cause	Terminating Call with one of the supported services
	(Conversational Call, Streaming Call, Interactive Call,
	Background Call, SMS)
- CN domain identity	Supported Domain (PS Domain or CS Domain)
- CHOICE UE Identity	IMSi
- IMSI	Set to the same octet string as in the IMSI stored in the
	USIM card

8.1.4.6.5 Test requirement

After step 6 the UE shall not transmits an RRC CONNECTION RE ESTABLISHMENT REQUEST message.

After step 10 the UE shall have an RRC connection based on dedicated physical channel resources. SS verifies that the UE transmits RRC CONNECTION SETUP COMPLETE on the uplink DCCH.

8.1.4.7 RRC Connection Re-Establishment: Success(T314=0, T315>0 and radio link failure)

8.1.4.7.1 Definition

8.1.4.7.2 Conformance requirement

RRC connection re establishment is needed, when a UE loses the radio connection due to e.g. radio link failure in CELL_DCH state. But the UE must release the radio bearer which is associated with T314 if T314 is set to 0. After a successful cell re selection and a subsequent transition to CELL_FACH state, the UE transmits an RRC CONNECTION RE ESTABLISHMENT REQUEST message which includes the IE "U RNTI" on the uplink CCCH to request for the re establishment of an RRC connection. The UTRAN transmits an RRC CONNECTION RE ESTABLISHMENT message on the downlink DCCH, which includes the new configuration information assigned. Then the UE transmits an RRC CONNECTION RE ESTABLISHMENT COMPLETE message on the uplink DCCH using AM RLC.

Reference

3GPP TS 25.331 clause 8.1.5

8.1.4.7.3 Test purpose

To confirm that the UE indicates to the non-access stratum the release of radio access bearer which is associated with T314 and tries to find a new cell by transiting to CELL_FACH state, after detecting that a radio link failure has occurred. The UE shall then attempt to establish a new RRC connection using an RRC CONNECTION RE-ESTABLISHMENT REQUEST message which includes the IE "U RNTI". Finally when the UTRAN allocates a new configuration to the UE using RRC CONNECTION ESTABLISHMENT message, the UE shall establish a new RRC connection link.

8.1.4.7.4 Method of test

Initial Condition

System Simulator: 2 cells (Cell 1 is active, Cell 2 is inactive)

UE: CELL_DCH state (state 6 5 or state 6 7) as specified in clause 7.4 of TS 34.108 in cell 1

Test Procedure

The UE is brought to CELL_DCH state in a cell 1 after making an successful outgoing call attempt. After the call has been established, SS begins to broadcast the BCCH in cell 2, and then stops transmitting and receiving in cell 1. The UE shall detect a radio link failure in cell 1 and indicate to the non access stratum the release of the radio bearer which is associated with T314. It shall attempt to re select to cell 2. It should then enter CELL_FACH state and transmits an RRC CONNECTION RE ESTABLISHMENT REQUEST message which includes the IE "U RNTI" on the uplink CCCH to SS. The SS transmits an RRC CONNECTION RE ESTABLISHMENT message which includes a new configuration. After the UE receives an RRC CONNECTION RE ESTABLISHMENT message, it reconfigures the new radio connection and transmits an RRC CONNECTION RE ESTABLISHMENT COMPLETE message on the uplink DCCH using AM RLC.

Step	Direction		Message	Comment
	UE	SS		
4	_	-	RADIO BEARER SETUP	T314=0
2		>	RADIO BEARER SETUP	
			COMPLETE	
3				The UE is brought to
				CELL_DCH state in a cell
				1, after making a
				successful outgoing call.
4		-	BCCH	The SS starts
				transmitting the BCCH in
				a cell 2 using the same
				contents (except for cell
				identity which is set to
				"0000 0000 0000 0010")
				for system information
				sent on cell 1. SS starts
				to listen to the uplink
				CCCH of cell 2.
5				The SS stops transmitting
				and receiving in a cell .1.
6				The UE detects the radio
				link failure which is
				associated with T314.
				The UE indicates to the
				non-access stratum the
				release of the affected
				radio bearer .
7		>	RRC CONNECTION	The UE should find a new
			RE-ESTABLISHMENT REQUEST	cell 2 and then transmits
				this message which
				includes the IE "U-RNTI"
				assigned previously.
8	←	-	RRC CONNECTION	Including the new
			RE-ESTABLISHMENT	configuration information.
9		}	RRC CONNECTION	
			RE-ESTABLISHMENT COMPLETE	

Specific Message Contents

RADIO BEARER SETUP

For circuit switched only UEs that do not support speech service, use the same message sub-type titled "Non-speech in CS" in Annex A, with the following exceptions:

For circuit switched only UEs that support speech service, use the same message sub-type titled "Speech in CS" in clause 9 of TS 34.108, with the following exceptions:

For all other types of UEs, use the same message sub-type titled "Packet to CELL_DCH from CELL_DCH in PS" in Annex A, with the following exceptions:

Information Element	Value/remark
RAB information to setup list	
- RAB information to setup	
- T314	0

8.1.4.7.5 Test requirement

After step 5, the UE shall indicate to the non access stratum the release of the radio bearer which is associated with T314. SS checks that no further data transmission from the affected radio bearer is received from the UE.

After step 6, the UE shall detect the presence of cell 2 and attempts to re established the lost RRC connection by transmitting RRC CONNECTION RE ESTABLISHMENT REQUEST in the new cell.

After step 8, the UE shall re establish an RRC connection.

8.1.4.8 RRC Connection Re-Establishment: Success(T314>0, T315=0 and radio link failure)

8.1.4.8.1 Definition

8.1.4.8.2 Conformance requirement

RRC connection re-establishment is needed, when a UE loses the radio connection due to e.g. radio link failure in CELL_DCH state. But the UE must release the radio bearer which is associated with T315 if T315 is set to 0. After a successful cell re-selection and subsequent transition to CELL_FACH state, the UE transmits an RRC CONNECTION RE-ESTABLISHMENT REQUEST message which includes the IE "U-RNTI" on the uplink CCCH to request for the re-establishment of an RRC connection. The UTRAN transmits an RRC CONNECTION RE-ESTABLISHMENT message on the downlink DCCH, which includes the new configuration information assigned. Then the UE transmits an RRC CONNECTION RE-ESTABLISHMENT COMPLETE message on the uplink DCCH using AM-RLC.

Reference

3GPP TS 25.331 clause 8.1.5

8.1.4.8.3 Test purpose

To confirm that the UE shall indicate to the non-access stratum the release of radio access bearer which is associated with T315 and try to find a new cell by transiting to CELL_FACH state, after detecting that a radio link failure has occurred. The UE shall then attempt to establish a new RRC connection using an RRC CONNNECTION RE-ESTABLISHMENT REQUEST message which includes the IE "U-RNTI". Finally when the SS allocates a new configuration to the UE using RRC CONNECTION ESTABLISHMENT message, the UE shall establish a new RRC connection link.

8 1 4 8 4 Method of test

Initial Condition

System Simulator: 2 cells (Cell 1 is active, Cell 2 is inactive)

UE: CELL_DCH state (state 6-5 or state 6-7) as specified in clause 7.4 of TS 34.108 in cell 1

Test Procedure

The UE is brought to CELL_DCH state in a cell 1 after making an successful outgoing call attempt. After the call has been established, SS begins to broadcast the BCCH in cell 2, and then stops transmitting and receiving in cell 1. The UE shall detect a radio link failure in cell 1 and indicate to the non access stratum the release of the radio bearer which is associated with T315. Then it shall attempt to re-select to cell 2. After that, it should then enter CELL_FACH state and transmits an RRC CONNECTION RE-ESTABLISHMENT REQUEST message which includes the IE "U-RNTI" on the uplink CCCH to SS. The SS transmits an RRC CONNECTION RE-ESTABLISHMENT message which includes a new configuration information. After the UE receives an RRC CONNECTION RE-ESTABLISHMENT message, it reconfigures the new radio connection and transmits an RRC CONNECTION RE-ESTABLISHMENT COMPLETE message on the uplink DCCH using AM RLC.

Step	Direction		Message	Comment
	UE	SS		
4	_ ←		RADIO BEARER SETUP	T315=0
2	_ →		RADIO BEARER SETUP	
			COMPLETE	
3				The UE is brought to
				CELL_DCH state in a cell
				1, after making a
				successful outgoing call.
4		÷	BCCH	The SS starts
				transmitting the BCCH in
				a cell 2 using the same
				contents (except for cell
				identity which is set to
				"0000 0000 0000 0010")
				for system information
				sent on cell 1. SS starts
				to listen to the uplink
				CCCH of cell 2.
5				The SS stops transmitting
				and receiving in a cell .1.
6				The UE detects the radio
				link failure which is
				associated with T315.
				The UE indicates to the
				non-access stratum the
				release of the radio
				bearer.
7	_→		RRC CONNECTION	The UE should find a new
			RE-ESTABLISHMENT REQUEST	cell 2 and then transmits
				this message which
				includes the IE "U-RNTI"
				assigned previously.
8	-←		RRC CONNECTION	Including the new
			RE-ESTABLISHMENT	configuration information
				channel.
9	_ →		RRC CONNECTION	
			RE-ESTABLISHMENT COMPLETE	

Specific Message Contents

RADIO BEARER SETUP

For circuit switched only UEs that do not support speech service, use the same message sub-type titled "Non-speech in CS" in Annex A, with the following exceptions:

For circuit switched only UEs that support speech service, use the same message sub-type titled "Speech in CS" in clause 9 of TS 34.108, with the following exceptions:

For all other types of UEs, use the same message sub-type titled "Packet to CELL_DCH from CELL_DCH in PS" in Annex A, with the following exceptions:

Information Element	Value/remark
RAB information to setup list	
—— - T314	θ

8.1.4.8.5 Test requirement

After step 5, the UE shall indicate to the non access stratum the release of the radio bearer which is associated with T314.

After step 6, the UE shall detect the presence of cell 2 and attempts to re established the lost RRC connection by transmitting RRC CONNECTION RE ESTABLISHMENT REQUEST to the new cell.

After step 8, the UE shall re establish an RRC connection.

8.1.4.9 RRC Connection Re-Establishment: Failure (T314 is timeout and T315=0)

8.1.4.9.1 Definition

8.1.4.9.2 Conformance requirement

RRC connection re establishment is needed, when a UE loses the radio connection due to e.g. radio link failure in CELL_DCH state. When the UE could not find a new cell before timer T314 expires, the UE shall enter idle mode.

Reference

3GPP TS 25.331 clause 8.1.5

8.1.4.9.3 Test purpose

To confirm that the UE enters idle mode, after T314 timeout following a radio link failure.

8.1.4.9.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CELL_DCH state (state 6 5 or state 6 7) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is brought to CELL_DCH state after a successful RRC connection establishment. After the DTCH has been established using radio bearer establishment procedure, the SS stops transmitting and receiving the DPCCH and DPDCH on the radio link. The UE shall detect a radio link failure. Before T314 is timeout, the UE shall continue to search for a new cell. During this period, SS transmits a PAGING TYPE 1 message on the PCCH to confirm that the UE has not entered the idle state. This is confirmed when no response from UE is received before T314 timeout. After T314 timeout, the UE shall enter idle mode. The SS transmits a PAGING TYPE 1 message on the PCCH again to confirm that the UE is in idle state. The UE shall respond by transmitting an RRC CONNECTION REQUEST message on the uplink CCCH.

Step	Direction		tep Direction		Message	Comment
	UE	SS				
4	_ - ←		RADIO BEARER SETUP	T314=20 seconds		
2	_ →		RADIO BEARER SETUP			
			COMPLETE			
3				The UE is brought to		
				CELL_DCH state, after		
				making a successful		
				outgoing call.		
4				The SS stops transmitting		
				and receiving the DPCCH		
				and DPDCH on the radio		
				link.		
5	-	(PAGING TYPE 1	The SS transmits this		
				message, which includes		
				a matched identity (set to		
				the test-USIM IMSI)		
				before T314 timeout.		
6				The SS confirms that the		
				UE does not transmit a		
				response in the next 20		
				seconds.		
7		\leftarrow	PAGING TYPE 1	The SS transmits the		
				message, which includes		
				a matched identity (set to		
				the test-USIM IMSI).		
8	-	\rightarrow	RRC CONNECTION REQUEST	SS checks the following		
				I Es:		
				"Initial UE identity" -		
				must be the same as		
				test-SIM IMSI.		
				"Establishment Cause		
				- must "Terminating		
				Call"		
				"Protocol Indicator" -		
				must be FALSE		
9	≺	:	RRC CONNECTION SETUP	SS assigns DPCH		
				resources to allow UE to		
				establish a RRC		
				connection.		

Specific Message Contents

SYSTEM INFORMATION TYPE 2

Information Element	Value/remark
UE Timers and constants in connected mode	
T301	3 seconds
T313	15 seconds
T314	20 seconds
T315	0 seconds
N313	50

RADIO BEARER SETUP (Step 1)

The contents of RADIO BEARER SETUP message in this test case is identical to message which is defined in clause 9 of TS 34.108 in case of speech and in Annex .A in case of the other with the following exceptions:

Information Element	Value/remark
RAB information to setup	
- RAB identity	0000-0001B
- CN domain identity	CS domain
Re-establishment timer	
 T314	20 seconds
- RB information to setup	
	5
- PDCP info	Not Present
- Downlink RLC mode	TM RLC
- In-sequence delivery	TRUE
- RB mapping info	
- Information for each multiplexing option	
- Number of RLC logical channels	1
- Uplink transport channel type	DCH
- Transport channel identity	2
- MAC logical channel priority	4
- Number of RLC logical channels	1
- Downlink transport channel type	DCH
- Transport channel identity	2
- Logical channel identity	1

PAGING TYPE 1 (Step 5 and 7)

Information Element	Value/remark
Paging record list	Only 1 entry
Paging record	
CHOICE Paging originator	CN originator
- Paging cause	Terminating Call with one of the supported services
	(Conversational Call, Streaming Call, Interactive Call,
	Background Call, SMS)
CN domain identity	Supported Domain (PS Domain or CS Domain)
CHOICE UE Identity	IMSI
 IMSI	Set to the same octet string as in the IMSI stored in the
	USIM card

RRC CONNECTION SETUP (Step 9)

Use the default message found in clause 9 of TS 34.108.

8.1.4.9.5 Test requirement

After step 5 the UE shall not transmits an RRC CONNECTION REQUEST message.

After step 9 the UE shall have an RRC connection based on dedicated physical channel resources.

8.1.4.10 RRC Connection Re-Establishment: Failure (T315 is timeout and T314=0)

8.1.4.10.1 Definition

8.1.4.10.2 Conformance requirement

RRC connection re establishment is needed, when a UE loses the radio connection due to e.g. radio link failure in CELL_DCH state. When the UE could not find a new cell before timer T315 is expires, the UE shall enter idle mode.

Reference

3GPP TS 25.331 clause 8.1.5

8.1.4.10.3 Test purpose

To confirm that the UE enters idle mode, after T315 is expires following a radio link failure.

8.1.4.10.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CELL_DCH state (state 6 5 or state 6 7) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is brought to CELL_DCH state after a successful RRC connection establishment. After the DTCH has been established using the radio bearer establishment procedure, the SS stops transmitting and receiving the DPCCH and DPDCH on the radio link. The UE shall detect a radio link failure. Before T315 is timeout, the UE shall continue to search for a new cell. During this period, SS transmits a PAGING TYPE 1 message on the PCCH to confirm that the UE has not entered the idle state. This is confirmed when no response from the UE is received before T315 timeout. After T315 is timeout, the UE shall enter idle mode. The SS transmits a PAGING TYPE 1 message on the PCCH again to confirm that the UE is in idle state. The UE shall respond by transmitting an RRC CONNECTION REQUEST message on the uplink CCCH.

Step	Direction		Step Direction		Message	Comment
	UE	SS				
4	_ - ←		RADIO BEARER SETUP	T315=30 seconds		
2	_ →		RADIO BEARER SETUP			
			COMPLETE			
3				The UE is brought to		
				CELL_DCH state, after		
				making a successful		
				outgoing call.		
4				The SS stops transmitting		
				and receiving the DPCCH		
				and DPDCH on the radio		
				link.		
5		(PAGING TYPE 1	The SS transmits this		
				message, which includes		
				a matched identity (set to		
				the test-USIM IMSI)		
				before T315 timeout.		
6				The SS confirms that the		
				UE does not transmit a		
				response in the next 30		
				seconds.		
7		\leftarrow	PAGING TYPE 1	The SS transmits the		
				message, which includes		
				a matched identity (set to		
				the test-USIM IMSI).		
8		\rightarrow	RRC CONNECTION REQUEST	SS checks the following		
				IEs:		
				"Initial UE identity" —		
				must be the same as		
				test-SIM IMSI.		
				<u>"Establishment Cause</u>		
				 must "Terminating 		
				Call"		
				"Protocol Indicator" -		
				must be FALSE		
9	←	-	RRC CONNECTION SETUP	SS assigns DPCH		
				resources to allow UE to		
				establish a RRC		
				connection.		

Specific Message Contents

SYSTEM INFORMATION TYPE 2

Information Element	Value/remark
UE Timers and constants in connected mode	
T301	3 seconds
T313	15 seconds
T314	0 seconds
T315	30 seconds
N313	50

RADIO BEARER SETUP (Step 1)

The contents of RADIO BEARER SETUP message in this test case is identical to message which is defined in clause 9 of TS 34.108 in case of speech and in Annex .A in case of the other of speech with the following exceptions:

Information Element	Value/remark
RAB information to setup	
	0000 0001B
- CN domain identity	CS domain
Re-establishment timer	
 T315	30 seconds
- RB information to setup	
	5
- PDCP info	Not Present
Downlink RLC mode	TM RLC
- In-sequence delivery	TRUE
- Information for each multiplexing option	
- Number of RLC logical channels	4
- Uplink transport channel type	DCH
- Transport channel identity	2
- MAC logical channel priority	1
- Number of RLC logical channels	1
- Downlink transport channel type	DCH
- Transport channel identity	2
- Logical channel identity	1

PAGING TYPE 1 (Step 5 and 7)

Information Element	Value/remark
Paging record list	Only 1 entry
Paging record	
CHOICE Paging originator	CN originator
- Paging cause	Terminating Call with one of the supported services
	(Conversational Call, Streaming Call, Interactive Call,
	Background Call, SMS)
CN domain identity	Supported Domain (PS Domain or CS Domain)
—- CHOICE UE Identity	IMSI
—- IMSI	Set to the same octet string as in the IMSI stored in the
	USIM card

RRC CONNECTION SETUP (Step 9)

Use the default message found in clause 9 of TS 34.108.

8.1.4.10.5 Test requirement

After step 5 the UE shall not transmits an RRC CONNECTION REQUEST message.

After step 9 the UE shall have an RRC connection based on dedicated physical channel resources.

8.1.4.11 RRC Connection Re-Establishment: Success (Unrecoverable error in RLC)

8.1.4.11.1 Definition

8.1.4.11.2 Conformance requirement

RRC connection re establishment is needed, when a UE loses the radio connection due to e.g. detection of RLC unrecoverable error (amount of the retransmission of RESET_PDU reaches the value of Max_DAT and receives no ACK) in CELL_DCH state. After a successful cell re selection and transition to CELL_FACH state, the UE transmits an RRC CONNECTION RE ESTABLISHMENT REQUEST message which includes the IE "U RNTI" on the uplink CCCH to request for the re establishment of an RRC connection. The UTRAN transmits an RRC CONNECTION RE-ESTABLISHMENT message on the downlink DCCH, which includes the new configuration information assigned. Then the UE transmits an RRC CONNECTION RE-ESTABLISHMENT COMPLETE message on the uplink DCCH using AM RLC.

Reference

3GPP TS 25.331 clause 8.1.5

8.1.4.11.3 Test purpose

To confirm that the UE tries to find a new cell by transiting to CELL_FACH state, after detecting that a RLC unrecoverable error has occurred. The UE shall then attempt to establish a new RRC connection using an RRC CONNNECTION RE ESTABLISHMENT REQUEST message which includes the IE "U RNTI". Finally when the SS allocates a new configuration to the UE using RRC CONNECTION ESTABLISHMENT message, the UE shall establish a new RRC connection link.

8.1.4.11.4 Method of test

Initial Condition

System Simulator: 2 cells (Cell 1 is active, Cell 2 is inactive)

UE: CELL_DCH state (state 6 9 or state 6 10) as specified in clause 7.4 of TS 34.108 in cell 1

Test Procedure

The UE is brought to CELL_DCH state in a cell 1 after a successful outgoing call attempt. After the call has been established, the SS transmits a PAGING TYPE 2 message on the downlink DCCH. Then the UE transmits an UPLINK DIRECT TRANSFER message on the uplink using AM RLC for the response and the SS does not transmit a STATUS PDU for the response to AM RLC PDU. SS begins to broadcast the BCCH in cell 2. The UE should detect an unrecoverable error in cell 1 and attempts to re select to cell 2. It should then enter CELL_FACH state and transmits an RRC CONNECTION RE ESTABLISHMENT REQUEST message which includes the IE "U RNTI" on the uplink CCCH to SS. The SS transmits an RRC CONNECTION RE ESTABLISHMENT message which includes "TRUE" in RLC reset indicator(for C plane) IE and a new TFCS setting according to the new transport channel. After the UE receives an RRC CONNECTION RE ESTABLISHMENT message, it reconfigures the RLC and the new radio connection and transmits an RRC CONNECTION RE ESTABLISHMENT COMPLETE message on the uplink DCCH using AM RLC.

Step	Direction		Message Message	Comment
	UE	SS		
4				The UE is brought to
				CELL_DCH state in a cell
				1, after making a
				successful outgoing call.
2	→	(PAGING TYPE 2	The SS transmits a
				PAGING TYPE 2
				message to the UE on the
				downlink DCCH in cell 1.
3	_	\rightarrow	UPLINK DIRECT TRANSFER	The UE responds to the
				PAGING TYPE 2
				message using AM-RLC
				but the SS does not
				transmit a STATUS PDU
				as an acknowledgement.
4		←	BCCH	The SS starts transmitting
				the BCCH in a cell 2 using
				the same contents
				(except for cell identity
				which is set to "0000 0000
				0000 0010") for system
				information sent on cell 1.
				SS starts to listen to the
				uplink CCCH of cell 2.
5				The UE detects an
				unrecoverable error in the
			DDO COMMECTION	RLC level.
6		\rightarrow	RRC CONNECTION	The UE should find a new
			RE-ESTABLISHMENT REQUEST	cell 2 and then transmits
				this message which includes the IE "U-RNTI"
7			RRC CONNECTION	assigned previously.
<i>+</i>		_		Including the new
<u> </u>			RE-ESTABLISHMENT	configuration information.
8	-	7	RRC CONNECTION RE-ESTABLISHMENT COMPLETE	
			KE-E9 I ABLISHMEN I COMPLETE	

Specific Message Contents

Cell 1: SYSTEM INFORMATION TYPE 2

Information Element	Value/remark
UE Timers and constants in connected mode	
T301	8-seconds
T313	15 seconds
T314	20 seconds
T315	30 seconds
N313	20

8.1.4.11.5 Test requirement

After step 5, the UE shall detect the presence of cell 2 and attempt to re established the RRC connection by transmitting RRC CONNECTION RE ESTABLISHMENT REQUEST in the new cell.

After step 7, the UE shall re establish an RRC connection.

8.1.5 UE capability

8.1.5.1 UE Capability in CELL_DCH state: Success

8.1.5.1.1 Definition

8.1.5.1.2 Conformance requirement

1. The UE CAPABILITY ENQUIRY message is sent by the UTRAN to request the UE to transmit its capability information related to any radio access network that is supported by the UE or if the UTRAN needs an update of the UE's UMTS capability information or of its inter-system classmark.

When the UE receives a UE CAPABILITY ENQUIRY message, the UE transmits a UE CAPABILITY INFORMATION message on the uplink DCCH. Then the UTRAN transmits a UE CAPABILITY INFORMATION CONFIRM message.

If during the execution of UE capability update procedure, an invalid UE CAPABILITY INFORMATION CONFIRM is received, the UE shall respond with RRC STATUS message and decide whether to re-transmit UE CAPABILITY INFORMATION message by comparing its internal counter against N304.

Reference

3GPP TS 25.331 clause 8.1.6, 8.1.7

8.1.5.1.3 Test purpose

To confirm that the UE transmits a UE CAPABILITY INFORMATION message after it receives a UE CAPABILITY ENQUIRY message from the SS. To confirm that the UE indicate an invalid message reception when erroneous downlink UE CAPABILITY ENQUIRY and UE CAPABILITY INFORMATION CONFIRM messages are received. The UE shall transmit RRC STATUS message with the correct error cause value to SS.

8.1.5.1.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CELL_DCH state (state 6-1 or state 6-3) as specified in clause 7.4 of TS 34.108 depending on the CN domain(s) supported by the UE

Test Procedure

The UE is brought to the CELL_DCH state after a successful outgoing call attempt. The SS transmits an erroneous UE CAPABILITY ENQUIRY message containing spare-invalid value in the IE "Capability update requirement" as a a message. After receiving such a message, the UE should report the error using RRC STATUS message with the appropriate error cause specified. Then SS transmits a UE CAPABILITY ENQUIRY message which includes the IE "Capability update requirement", the UE receives this message and transmits a UE CAPABILITY INFORMATION message on the uplink DCCH which includes the "Inter-system message" IE. The SS transmits a UE CAPABILITY INFORMATION CONFIRM message to the UE to complete the test. Then SS initiates another UE capability update procedure by transmitting the same UE CAPABILITY ENQUIRY using as in step 4. The UE shall reply with a UE CAPABILITY INFORMATION message on the uplink DCCH. When SS receives this message, it transmit an erroneous UE CAPABILITY INFORMATION CONFIRM message. The content of this message is lack of a mandatory IEa concatenation of the message type IE and an arbitrary 32 bits string. The UE shall detect a protocol error and send RRC STATUS message to report this event. After receiving RLC acknowledgement for this message, the UE shall re-transmit UE CAPABILITY INFORMATION message on the uplink DCCH. SS completes this test by an error-free UE CAPABILITY INFORMATION CONFIRM message similar to the message sent in step 6.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1			The UE is brought to CELL_DCH state after an outgoing call has been established successfully.
2	+	UE CAPABILITY ENQUIRY	See specific message contents for this message
3)	RRC STATUS	The IE "Protocol error cause" found in IE "Protocol error information" should be set to "Information element value not comprehended"
4	+	UE CAPABILITY ENQUIRY	Use default message.
5	→	UE CAPABILITY INFORMATION	The message shall includethe IE "Intersystem message", which carries the GSM classmark information requested.
6	+	UE CAPABILITY INFORMATION CONFIRM	Use default message.
7	+	UE CAPABILITY ENQUIRY	Same as in step 4.
8	→	UE CAPABILITY INFORMATION	Shall be the same message content as in step 5.
9	+	UE CAPABILITY INFORMATION CONFIRM	See specific message contents for this messageThis message contains an arbitrary 32- bits patterns, following the IE "Message Type"
10	→	RRC STATUS	UE shall detect an error and then transmit this message.
11)	UE CAPABILITY INFORMATION	UE shall re-transmit this message after receiving acknowledgement from the SS for RRC STATUS message.
12	+	UE CAPABILITY INFORMATION CONFIRM	SS sends an error-free message to acknowledge the receipt of the uplink message.

Specific Message Contents

UE CAPABILITY ENQUIRY (Step 2)

Information Element	Value/remark
Capability update requirement	
- UE radio access FDD capability update	TRUE
requirement	
- UE radio access FDD capability update	FALSE
requirement	
 System specific capability update requirement list 	invalid value
 System specific capability update 	TRUE
requirementCapability update requirement	
UE radio access capability update requirement	Contains a spare value in this IE
- System specific capability update requirement list	•
- System specific capability update requirement	

RRC STATUS (Step 3)

Information Element	Value/remark
Protocol Error Information - Protocol Error Cause	Checked to see if set to "Information element not
- Protocol Error Cause	comprehended"

UE CAPABILITY INFORMATION CONFIRM (Step 9)

Information Element	Value/remark
RRC transaction identifier Unknown bit pattern	Not PresentAn arbitrary 32 bits

RRC STATUS (Step 10)

Information Element	Value/remark
Protocol Error Information	
- Protocol Error Cause	Checked to see if set to "ASN.1 violation or encoding
	error" or "conditional information element error"

8.1.5.1.5 Test requirement

After step 2, the UE shall transmit a RRC STATUS message on the uplink DCCH, reporting the error with protocol error cause set to "Information element value not comprehended".

After step 4 the UE shall transmit a UE CAPABILITY INFORMATION message on the uplink DCCH to respond to the UE CAPABILITY ENQUIRY message.

After step 9, the UE shall transmit a RRC STATUS message on the uplink DCCH. The protocol error cause shall be set to "ASN.1 violation or encoding error" or "conditional information element error".

After step 10, the UE shall re-transmit the UE CAPABILITY INFORMATION message with a similar content as in step 8.

8.1.5.2 UE Capability in CELL_DCH state: Success after T304 timeout

8.1.5.2.1 Definition

8.1.5.2.2 Conformance requirement

- 1. The UE CAPABILITY ENQUIRY message is sent by the UTRAN to request the UE to transmit its capability information related to any radio access network that is supported by the UE, if the UTRAN needs an update of the UE's UMTS capability information or of its inter-system classmark.
- 2. After the UE receives a UE CAPABILITY ENQUIRY message, it transmits a UE CAPABILITY INFORMATION message on the uplink DCCH. If it fails to receive a UE CAPABILITY INFORMATION CONFIRM message, the UE re-transmits another UE CAPABILITY INFORMATION message until V304 is greater than N304.

Reference

3GPP TS 25.331 clause 8.1.6, 7

8.1.5.2.3 Test purpose

To confirm that the UE re-transmits a UE CAPABILITY INFORMATION message until V304 is greater than N304, after the expiry of timer T304 when the UE cannot receive a UE CAPABILITY INFORMATION CONFIRM message in response to a UE CAPABILITY INFORMATION message.

8.1.5.2.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CELL_DCH state (state 6-1 or state 6-3) as specified in clause 7.4 of TS 34.108 depending on the CN domain(s) supported by the UE

Test Procedure

The UE is brought to CELL_DCH state. When the SS transmits a UE CACAPABILITY ENQUIRY message which includes the "Capability update requirement" IE, the UE shall reply with a UE CAPABILITY INFORMATION message on the uplink DCCH which includes the "Inter-system message" IE. The SS does not transmits a UE CAPABILITY INFORMATION CONFIRM message to the UE, resulting in the T304 timer to expire. SS shall observe that the UE attempts to transmit a UE CAPABILITY INFORMATION message again. The UE shall re-transmit N304 times, and SS transmits a UE CAPABILITY INFORMATION CONFIRM message to answer the last request and completes this test procedure.

Expected sequence

Step	Direc	tion	Message	Comment
	UE	SS		
1				The UE is brought to
				CELL_DCH state.
				SS sets internal counter K
				=1
2		÷	UE CAPABILITY ENQUIRY	Including the "Capability
				update requirement"IE.
3		\rightarrow	UE CAPABILITY INFORMATION	Including the "Inter-system
				message" IE, which
				indicated the radio access
				network supported by the
				UE.
4				If K is greater to N304,
				then proceed to step 6.
5				The SS does not transmit
				a response and wait for
				T304 timer to expire.
				K=K+1 and goes to step 3.
6	•	(UE CAPABILITY INFORMATION CONFIRM	Use default message
				contents

Specific Message Contents

None

8.1.5.2.5 Test requirement

After step 3 the UE shall re-transmits a UE CAPABILITY INFORMATION message on the uplink DCCH, after each expiry of timer T304. The UE CAPABILITY INFORMATION message shall contain IE "UE radio access capability" indicating the settings found in PIC/PIXIT statements. IE "UE system specific capability" shall carry relevant GSM classmark information. After (N304+1) re-transmissions, the UE shall receive a UE CAPABILITY INFORMATION CONFIRM message.

8.1.5.3 UE Capability in CELL_DCH state: Failure (After N304 re-transmissions)

8.1.5.3.1 Definition

8.1.5.3.2 Conformance requirement

- 1. The UE CAPABILITY ENQUIRY message is sent by the UTRAN to request the UE to transmit its capability information related to any radio access network that is supported by the UE if the UTRAN needs an update of the UE's UMTS capability information or of its inter-system classmark.
- 2. In the case of a failure to transmit a UE CAPABILITY INFORMATION in excess of N304 times, the UE initiates the RRC connection re-establishment procedure.

Reference

3GPP TS 25.331 clause 8.1.6, 8.1.7

8.1.5.3.3 Test purpose

To confirm that the UE stops retrying to transmit a UE CAPABILITY INFORMATION message if V304 is greater than N304. It then initiates the RRC re-establishment procedure.

8.1.5.3.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CELL_DCH state (state 6-1 or state 6-3) as specified in clause 7.4 of TS 34.108 depending on the CN domain(s) supported by the UE

Test Procedure

The UE is brought to CELL_DCH state. When the SS transmits a UE CAPABILITY ENQUIRY message which includes the "Capability update requirement" IE, the UE receives this message and transmits a UE CAPABILITY INFORMATION message on the uplink DCCH which includes the "Inter-system message" IE. The SS does not respond with a UE CAPABILITY INFORMATION CONFIRM message but keeps a count on the number of messages received When the T304 timer expires, the UE shall transmit a UE CAPABILITY INFORMATION message again. After sending (N304+1) messages, the UE shall stop sending UE CAPABILITY INFORMATION messages and initiates the RRC connection re-establishment procedure. This is verified in SS by the reception of RRC CONNECTION RE-ESTABLISHMENT REQUEST. SS allows UE to return to "connected state" by issuing RRC CONNECTION RE-ESTABLISHMENT message on the downlink CCCH.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1			The UE starts from CELL_DCH state. SS sets counter K to 1
2	+	UE CAPABILITY ENQUIRY	Including the "Capability update requirement" IE.
3	\rightarrow	UE CAPABILITY INFORMATION	Including the "Intersystem message" IE.
4			The SS does not transmit a response and allows T304 timer to expire. SS increments counter K If K is greater than N304, proceeds to step 5 else returns to 3.
5	→	RRC CONNECTION RE-ESTABLISHMENT REQUEST	The UE assumes that radio link failure has occurred and transmits this message which includes the IE "U-RNTI" containing the U-RNTI allocated to the UE earlier.
6	+	RRC CONNECTION RE-ESTABLISHMENT	Including the new TFCS according to the new transport channel.
7	→	RRC CONNECTION RE-ESTABLISHMENT COMPLETE	

Specific Message Contents

None

8.1.5.3.5 Test requirement

After step 3 the UE shall transmit a UE CAPABILITY INFORMATION message on the uplink DCCH. The UE shall re-transmit this message for N304 times. Thereafter, the UE shall initiate the RRC re-establishment procedure by sending the message RRC CONNECTION RE-ESTALISHMENT REQUEST.

After step 7 the UE shall have a new RRC connection, using the new transport format dictated in the RRC CONNECTION RE-ESTABLISHMENT message.

8.1.5.4 UE Capability in CELL_FACH state: Success

8.1.5.4.1 Definition

8.1.5.4.2 Conformance requirement

- 1. The UE CAPABILITY ENQUIRY message is sent by the UTRAN to request the UE to transmit its capability information related to any radio access network(s) supported by the UE. UTRAN initiates this procedure when it needs an update of the UE's UMTS capability information or of its inter-system classmark.
- 2. When the UE receives a UE CAPABILITY ENQUIRY message, the UE shall transmit a UE CAPABILITY INFORMATION message on the uplink DCCH.

3. If during the execution of UE capability update procedure, an invalid UE CAPABILITY INFORMATION CONFIRM is received, the UE shall respond with RRC STATUS message and decide whether to re-transmit UE CAPABILITY INFORMATION message by comparing its internal counter against N304.

Reference

3GPP TS 25.331 clause 8.1.6, 8.1.7

8.1.5.4.3 Test purpose

To confirm that the UE transmits an UE CAPABILITY INFORMATION message after it receives a UE CAPABILITY ENQUIRY message from the SS. To confirm that the UE indicates an invalid message reception when erroneous downlink UE CAPABILITY ENQUIRY and UE CAPABILITY INFORMATION CONFIRM messages are received. The UE shall transmit RRC STATUS message with the correct error cause value to SS.

8.1.5.4.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CELL_FACH state (state 6-2 or state 6-4) as specified in clause 7.4 of TS 34.108 depending on the CN domain(s) supported by the UE

Test Procedure

The UE is brought to the CELL_FACH state after a successful outgoing call attempt. The SS transmits an erroneous UE CAPABILITY ENQUIRY message containing spare-invalid value in the IE "Capability update requirement". After receiving such a message, the UE shall report an error using RRC STATUS message with the appropriate error cause specified. Then SS transmits a UE CAPABILITY ENQUIRY message which includes the IE "Capability update requirement" as a correct message, the UE receives this message and transmits a UE CAPABILITY INFORMATION message on the uplink DCCH, which includes the IE "Inter-system message". The SS transmits a UE CAPABILITY INFORMATION CONFIRM message to the UE to complete the UE capability enquiry procedure. Then SS initiates another UE capability enquiry procedure by transmitting the same UE CAPABILITY ENQUIRY message as in step 4. The UE shall reply with a UE CAPABILITY INFORMATION message on the uplink DCCH. When SS receives this message, it transmits an erroneous UE CAPABILITY INFORMATION CONFIRM message. The content of this message is lack of a mandatory IEa concatenation of the message type IE and an arbitrary 32 bits string. The UE shall detect a protocol error and send RRC STATUS message to report this event. After receiving the RLC layer acknowledgement PDU for this message, the UE shall re-transmit UE CAPABILITY INFORMATION CONFIRM message similar to the message sent in step 6.

Expected sequence

Step	Dire	ction	Message	Comment
	UE	SS		
1				The UE is brought to CELL_FACH state after an outgoing call has been established successfully.
2		←	UE CAPABILITY ENQUIRY	See specific message contents for this message
3		→	RRC STATUS	The IE "Protocol error cause" found in IE "Protocol error information" should be set to "Information element value not comprehended"
4		\leftarrow	UE CAPABILITY ENQUIRY	Use default message.
5		\rightarrow	UE CAPABILITY INFORMATION	The message shall include the IE "Intersystem message", which carries the GSM classmark information requested.
6		(UE CAPABILITY INFORMATION CONFIRM	Use default message.
7		-	UE CAPABILITY ENQUIRY	Same as in step 4.
8		\rightarrow	UE CAPABILITY INFORMATION	The message content shall be the same as in step 5.
9		(UE CAPABILITY INFORMATION CONFIRM	See specific message contents for this message contains an arbitrary 32-bits patterns, following the IE "Message Type"
10		\rightarrow	RRC STATUS	UE shall detect an error and then transmit this message on uplink DCCH.
11		→	UE CAPABILITY INFORMATION	UE shall re-transmit this message after receiving the RLC acknowledgement PDU for RRC STATUS message from SS.
12		←	UE CAPABILITY INFORMATION CONFIRM	SS sends an error-free message to acknowledge the receipt of the uplink message.

Specific Message Contents

UE CAPABILITY ENQUIRY (Step 2-and 7)

Information Element	Value/remark
Capability update requirement	
 UE radio access <u>FDD</u> capability update 	TRUE
requirement	
 UE radio access FDD capability update 	<u>FALSE</u>
requirement	
 System specific capability update requirement list 	Contains a spare value in this leinvalid value
 System specific capability update requirement 	

RRC STATUS (Step 3)

Information Element	Value/remark
Protocol Error Information	
- Protocol Error Cause	Checked to see if set to "Information element not
	comprehended"

UE CAPABILITY INFORMATION CONFIRM (Step 9)

Information Element	Value/remark
RRC transaction identifier Unknown bit pattern	Not PresentAn arbitrary 32 bits

RRC STATUS (Step 3)

Information Element	Value/remark
Protocol Error Information	
- Protocol Error Cause	Checked to see if set to "ASN.1 violation or encoding
	error" or "conditional information element error"

8.1.5.4.5 Test requirement

After step 2, the UE shall transmit a RRC STATUS message on the uplink DCCH, reporting the error with protocol error cause set to "Information element value not comprehended".

After step 4 the UE shall transmit a UE CAPABILITY INFORMATION message on the uplink DCCH to respond to the downlink UE CAPABILITY ENQUIRY message.

After step 9, the UE shall transmit a RRC STATUS message on the uplink DCCH. The protocol error cause shall be set to "ASN.1 violation or encoding error" or "conditional information element error".

After step 10, the UE shall re-transmit the UE CAPABILITY INFORMATION message with a similar content as in step 8.

8.1.5.5 UE Capability in CELL_FACH state: Success after T304 timeout

8.1.5.5.1 Definition

8.1.5.5.2 Conformance requirement

- 1. The UE CAPABILITY ENQUIRY message is sent by the UTRAN to request the UE to transmit its capability information related to any radio access network(s) supported by the UE. UTRAN initiates this action when it needs an update of the UE's UMTS capability information or of its inter-system classmark.
- 2. After the UE receives a UE CAPABILITY ENQUIRY message, it transmits a UE CAPABILITY INFORMATION message on the uplink DCCH. If it fails to receive a UE CAPABILITY INFORMATION CONFIRM message, the UE re-transmits another UE CAPABILITY INFORMATION message until its internal counter V304 is greater than N304.

Reference

3GPP TS 25.331 clause 8.1.6, 7

8.1.5.5.3 Test purpose

To confirm that the UE re-transmits a UE CAPABILITY INFORMATION message until V304 is greater than N304, after the expiry of timer T304 when it fail to receive a downlink UE CAPABILITY INFORMATION CONFIRM message in response to the uplink UE CAPABILITY INFORMATION message sent.

8.1.5.5.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CELL_FACH state (state 6-2 or state 6-4) as specified in clause 7.4 of TS 34.108 depending on the CN domain(s) supported by the UE

Test Procedure

The UE is brought to CELL_FACH state. When the SS transmits a UE CAPABILITY ENQUIRY message which includes the IE "Capability update requirement", the UE shall reply with a UE CAPABILITY INFORMATION message on the uplink DCCH that contains the IE "Inter-system message". The SS waits and does not transmit a UE CAPABILITY INFORMATION CONFIRM message to the UE, resulting in the T304 timer to expire. SS shall observe that the UE attempts to transmit a UE CAPABILITY INFORMATION message again. The UE shall re-transmit N304 times, and SS transmits a UE CAPABILITY INFORMATION CONFIRM message to answer the last request and completes this test procedure.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1				The UE is brought to CELL_FACH state. SS sets internal counter K =1
2		←	UE CAPABILITY ENQUIRY	Including the IE "Capability update requirement".
3		→	UE CAPABILITY INFORMATION	Including the IE "Inter- system capability", which indicated the radio access network supported by the UE.
4				If K is greater to N304,then proceeds to step 6. Else, continue with step 5.
5				The SS does not transmit a response and wait for T304 timer to expire. K=K+1 and goes to step 3.
6	•	(UE CAPABILITY INFORMATION CONFIRM	Use default message contents

Specific Message Contents

None

8.1.5.5.5 Test requirement

After step 3 the UE shall re-transmit a UE CAPABILITY INFORMATION message on the uplink DCCH, after each expiry of timer T304. The UE CAPABILITY INFORMATION message shall contain IE "UE radio access capability" with the value matching those stated in the ICS/IXIT statements. In the same message, IE "UE system specific capability" shall be present and it carries relevant GSM classmark information. After (N304+1) re-transmissions, the UE shall receive a UE CAPABILITY INFORMATION CONFIRM message.

8.1.6 Direct Transfer

8.1.6.1 Direct Transfer in CELL DCH state (invalid message reception)

8.1.6.1.1 Definition

8.1.6.1.2 Conformance requirement

The UE shall transmit an RRC STATUS message stating the reason "protocol error" in IE "failure cause" and also set value "Information element value not comprehended" in IE "Protocol error cause" when the UE receives a DOWNLINK DIRECT TRANFER message, which does not includes a spare value for the mandatory IE "CN domain identity" having criticality defined as "Reject" NAS message".

Reference

3GPP TS 25.331 clause 8.1.9

8.1.6.1.3 Test purpose

To confirm that the UE transmits an RRC STATUS message on the DCCH using AM RLC if it receives a DOWNLINK DIRECT TRANSFER message which does not include the specifying a spare value in the mandatory-IE "NAS message CN domain identity"

8.1.6.1.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CELL_DCH (state 6-1 or state 6-3) as specified in clause 7.4 of TS 34.108 depending on the CN domain(s) supported by the UE

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a DOWNLINK DIRECT TRANSFER message to the UE and does not includes the spare value in the mandatory IE "NAS message CN domain identity". The UE shall transmit an RRC STATUS message on the DCCH using AM RLC, setting the value "protocol error" in IE "failure cause". The error type "Information element value not comprehended" shall also be indicated in IE "Protocol error cause".

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1	+		DOWNLINK DIRECT TRANSFER	
2	\rightarrow		RRC STATUS	

Specific Message Contents

DOWNLINK DIRECT TRANSFER

The contents of DOWNLINK DIRECT TRANSFER message in this test case is identical to those in default contents of layer 3 messages for RRC tests with the following exceptions:.

Information Element	Value/remark
NAS messageCN domain identity	Spare valueNot Present

RRC STATUS

Information Element	Value/remark
Message Type	
Protocol error information	Information element value not comprehended
Other information element	Not checked

8.1.6.1.5 Test requirement

After step 1 the UE shall transmit a n RRC STATUS message on the DCCH using AM RLC setting "protocol error" in IE "failure cause" and setting "Information element value not comprehended" in IE "Protocol error cause".

8.1.6.2 Direct Transfer in CELL FACH state (invalid message reception)

8.1.6.2.1 Definition

8.1.6.2.2 Conformance requirement

The UE shall transmit an RRC STATUS message stating the reason "protocol error" in IE "failure cause" and also set value "Information element value not comprehended" in IE "Protocol error cause" when the UE receives a DOWNLINK DIRECT TRANFER message, which does not includes a spare value for the mandatory IE "NAS message CN domain identity" having criticality defined as "Reject".

Reference

3GPP TS 25.331 clause 8.1.9

8.1.6.2.3 Test purpose

To confirm that the UE transmits an RRC STATUS message on the DCCH using AM RLC if it receives a DOWNLINK DIRECT TRANSFER message which does not include specifying a spare value in the mandatory-IE "NAS message CN domain identity".

8.1.6.2.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CELL_FACH (state 6-2 or state 6-4) as specified in clause 7.4 of TS 34.108 depending on the CN domain(s) supported by the UE

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a DOWNLINK DIRECT TRANSFER message to the UE and does not include includes the spare value in the mandatory-IE "NAS message CN domain identity". The UE shall transmit an RRC STATUS message on the DCCH using AM RLC, setting the value "protocol error" in IE "failure cause". The error type "Information element value not comprehended" shall also be indicated in IE "Protocol error cause".

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1		←	DOWNLINK DIRECT TRANSFER	
2		\rightarrow	RRC STATUS	

Specific Message Contents

DOWNLINK DIRECT TRANSFER

The contents of DOWNLINK DIRECT TRANSFER message in this test case is identical to those in default contents of layer 3 messages for RRC tests with the following exceptions:.

Information Element	Value/remark
NAS messageCN domain identity	Spare valueNot Present

RRC STATUS

Information Element	Value/remark
Message Type	
Protocol error information	Information element value not comprehended
Other information element	Not checked

8.1.6.2.5 Test requirement

After step 1 the UE shall transmit a n RRC STATUS message on the DCCH using AM RLC setting "protocol error" in IE "failure cause" and setting "Information element value not comprehended" in IE "Protocol error cause".

8.1.7 Security mode control

8.1.7.1 Security mode control in CELL_DCH state

8.1.7.1.1 Definition

8.1.7.1.2 Conformance requirement

- 1. This procedure is used to trigger the stop or start of ciphering or to command the restart of ciphering with the new ciphering configuration. It is also used to start integrity protection or modify integrity protection configuration, both for the signalling links and any of radio bearers.
- 2. When the UE receives a SECURITY MODE COMMAND message from the UTRAN, which indicates the downlink activation time and new integrity protection configuration, the UE shall apply the old ciphering configuration before the stated downlink activation time. It shall start to decipher using the new ciphering configuration at the downlink activation time.
- 3. After the UE transmit the SECURITY MODE COMPLETE message using the new integrity protection configuration which includes uplink activation time, it starts to cipher transmission in the uplink using the new configuration at the uplink activation time.

Reference

3GPP TS 25.331 clause 8.1.12

8.1.7.1.3 Test purpose

To confirm that the UE correctly communicates to the UTRAN and activates the new ciphering configurations after the stated activation time. To confirm that after the UE receives a SECURITY MODE COMMAND message, it transmits a SECURITY MODE COMPLETE message to the UTRAN using the old ciphering configuration together with the application of the new integrity protection configuration.

8.1.7.1.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CELL_DCH (state 6-1 or state 6-3) as specified in clause 7.4 of TS 34.108 ,depending on the CN domain(s) supported by the UE

Test Procedure

The UE is in the RRC connected state CELL_DCH. The SS transmits a SECURITY MODE COMMAND message in which ciphering is requested to be activated, but the IE "Ciphering algorithm_capability" is set to an unknown value. The UE shall not trigger any ciphering algorithm and it shall respond by sending SECURITY MODE FAILURE message on the DCCH. Next, SS transmits a valid SECURITY MODE COMMAND message which includes the "Downlink activation time" IE for RB2 and "Integrity check info" IE. Then the UE shall check the integrity check info and shall start to configure ciphering in downlink and transmits a SECURITY MODE COMPLETE message which contains the uplink activation time for RB1,RB2,RB 3 and RB 4 and also "Integrity check info" IE. SS records the uplink ciphering activation time for RB 2. Next, SS transmits COUNTER CHECK message repeated on the downlink DCCH using RLC-AM mode. The UE shall respond to each downlink message with a COUNTER CHECK RESPONSE message on the uplink DCCH using RLC-AM. This cycle repeats itself until both the uplink and downlink ciphering activation time for RB 2 has elapsed. SS checks all uplink COUNTER CHECK RESPONSE messages are integrity-protected by UIA algorithm, and that the messages contain the correct values for "Integrity mode info" IE. After both the uplink and downlink ciphering activation time for RB 2 has passed, the UE shall be able to communicate with the SS. This can be verified in SS through the reception of a correctly ciphering and integrity-protected COUNTER CHECK RESPONSE message.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1				RRC connected state on DCCH
2	+		SECURITY MODE COMMAND	IE "Ciphering Algorithm capability" is set to an invalid value
3	→		SECURITY MODE FAILURE	IE "Failure Cause" should be set to "Protocol Error" and IE "Protocol Error Information" should be set to "Information element value not comprehended".
4		←	SECURITY MODE COMMAND	See specific message contents.
5		\rightarrow	SECURITY MODE COMPLETE	SS verifies that this message is sent unciphered. SS records the uplink ciphering activation time for RB 2.
6		←	COUNTER CHECK	SS repeats step 6 and step 7 until its internal uplink and downlink RLC SN have both surpassed the uplink and downlink ciphering activation time specified for RB2. This message is sent on the downlink DCCH using RLC-AM. See specific message content.
7		\rightarrow	COUNTER CHECK RESPONSE	UE shall send this message on the uplink DCCH using RLC-AM. See specific message content.
8				SS verifies that the last COUNTER CHECK RESPONSE message is both integrity-protected and ciphered correctly.

Specific Message Contents

SECURITY MODE COMMAND (Step 2)

See notes below for the value of Y.

Information Element	Value/remark
RRC transaction identifier	<u>0</u>
Integrity check info	
Message authentication code	Calculated result in SS
RRC Message sequence number	0
Security Capability	
Ciphering algorithm capability	"11111111111111"BSpare value
Integrity protection algorithm capability	"11111111111111"B Spare value
Ciphering mode info	
Ciphering mode command	Start
Activation time for DPCH	Not Present
Radio bearer downlink ciphering activation time info	
RB Identity	2
RLC sequence number	Current RLC SN + Y
Integrity protection mode info	
Integrity protection mode command	Start
Downlink integrity protection activation info	Not Present
Integrity protection algorithm	If integrity is indicated to be active on IXIT statements in
	TS 34.123-2, use one of the supported integrity
	algorithms
Integrity protection initialisation number	0000 0000 0000 0000 H (FRESH)
CN domain identity	Supported domain

SECURITY MODE COMMAND (Step 4)

See notes below for the value of Y.

Information Element	Value/remark
RRC transaction identifier	<u>0</u>
Integrity check info	
Message authentication code	Calculated result in SS
RRC Message sequence number	0
Security Capability	
Ciphering algorithm capability	If ciphering is indicated to be active on IXIT statements
	in TS 34.123-2, use one of the supported ciphering
	algorithms
Integrity protection algorithm capability	00000000000010B(UIA1)
Ciphering mode info	Start
Ciphering mode command	Not Present
Activation time for DPCH	
Radio bearer downlink ciphering activation time info	2
RB Identity	Current RLC SN + Y
RLC sequence number	Start
Integrity protection mode info	Not Present
Integrity protection mode command	If integrity is indicated to be active on IXIT statements in
Downlink integrity protection activation info	TS 34.123-2, use one of the supported integrity
Integrity protection algorithm	algorithms
Integrity protection initialisation number	0000 0000 0000 0000 H (FRESH)
CN domain identity	Supported domain

SECURITY MODE COMPLETE (Step 5)

Information Element	Value/remark
RRC transaction identifier	<u>0</u>
Integrity check info	
- Message Authentication code	Checked to see if present
 RRC Message sequence number 	Checked to see if present
Uplink integrity protection activation info	
- RRC message sequence number list	Check to see if it the RRC SN for RB 0 to RB 4 are present
Radio bearer uplink ciphering activation info	
- RB Identity	2
- RLC sequence number	SS records this value. See step 8 in 'expected sequence'

COUNTER CHECK (Step 6)

Information Element	Value/remark
RRC transaction identifier	<u>0</u>
Integrity check info	Calculated value
RB COUNT-C MSB information	
- RB identity	2
- COUNT-C MSB uplink	Current COUNT-C MSB for RB#2 in uplink
- COUNT-C MSB downlink	Current COUNT-C MSB for RB#2 in downlink

COUNTER CHECK RESPONSE (Step 7)

Information Element	Value/remark
RRC transaction identifier	<u>0</u>
Integrity check info	Checked to see if the MAC code match
RB COUNT-C information	Check to if this IE is absent
- RB identity	
- COUNT-C uplink	
- COUNT-C downlink	

Note: Y = 2 * (size of COUNTER CHECK message, after PER encoding). The unit of Y is the number of RLC-AM PDU.

8.1.7.1.5 Test requirement

After step 2 the UE shall transmit a SECURITY MODE FAILURE message to report the protocol error detected in the first SECURITY MODE COMMAND message. The UE shall be able to communicate normally with the SS, with all control data on the signalling radio bearers unciphered.

After step 4 the UE shall RLC-acknowledge the receipt of the SECRUITY MODE COMMAND message using unciphered mode and which includes calculated integrity check info. SS checks that the SECURITY MODE COMPLETE message is received unciphered and that the calculated "integrity check info" IE is correct.

After step 5 SS verifies that all uplink signalling messages on RB1, RB2, RB3 and RB4 are integrity protected with UIA1 algorithm.

After step 7 SS verifies that the last COUNTER CHECK RESPONSE message received is integrity protected with UIA algorithm and ciphered with the algorithm indicated in the second SECURITY MODE COMMAND (Step 4) message.

8.1.7.2 Security mode control in CELL_FACH state

8.1.7.2.1 Definition

8.1.7.2.2 Conformance requirement

- 1. This procedure is used to trigger the stop or start of ciphering, or to command the restart of ciphering with the new ciphering configuration. It is also used to start integrity protection or modify integrity protection configuration, both for signalling link(s) and any radio access bearer(s).
- 2. When the UE receives a SECURITY MODE COMMAND message from the UTRAN, which indicates the downlink activation time and new integrity protection configuration, the UE shall apply the old ciphering configuration before the stated downlink activation time. It shall start to decipher using the new ciphering configuration at the downlink activation time.
- 3. The UE shall transmit SECURITY MODE COMPLETE message using the new integrity protection configuration stated in the received SECURITY MODE COMMAND message. The SECURITY MODE COMPLETE message shall include the ciphering uplink activation time. The UE shall start to apply the new ciphering configuration on the uplink direction, after the uplink activation time has elapsed.

Reference

3GPP TS 25.331 clause 8.1.12

8.1.7.2.3 Test purpose

To confirm that after the UE receives a SECURITY MODE COMMAND message, it transmits a SECURITY MODE COMPLETE message to the UTRAN using the old ciphering configuration together with the application of the new integrity protection configuration. To confirm that the UE applies the old ciphering configuration in the downlink prior to the activation time; and uses the new ciphering configuration on and after the activation time. To confirm that the UE starts to cipher its uplink transmissions after the uplink activation time stated in SECURITY MODE COMPLETE message is reached.

8.1.7.2.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CELL_FACH (state 6-2 or state 6-4) as specified in clause 7.4 of TS 34.108 depending on the CN domain(s) supported by the UE

Test Procedure

The UE is in the RRC connected state CELL FACH. The SS transmits a SECURITY MODE COMMAND message in which ciphering is requested to be activated, but the IE "Ciphering algorithm capability" is set to an unknown value. The UE shall not trigger any ciphering algorithm and it shall respond by sending SECURITY MODE FAILURE message on the DCCH. Next, SS transmits a valid SECURITY MODE COMMAND message which includes IE "Downlink activation time" for RB2 and IE "Integrity check info". The UE shall check the integrity check info. It shall start to configure ciphering in downlink and transmit a SECURITY MODE COMPLETE message, which contains the uplink activation time for RB2. This message shall contain the IE "Integrity check info". SS records the uplink ciphering activation time for RB 2. Next, SS transmits COUNTER CHECK message repeated on the downlink DCCH using RLC-AM mode. The UE shall respond to each downlink message with a COUNTER CHECK RESPONSE message on the uplink DCCH using RLC-AM. SS confirms that the uplink COUNTER CHECK RESPONSE messages are not ciphered. SS also checks all uplink messages are integrity-protected by UIA1 algorithm, and that the messages contain the correct values for "Integrity mode info" IE. This cycle repeats itself until both the uplink and downlink ciphering activation time for RB 2 have elapsed. After both the uplink and downlink ciphering activation time for RB 2 have passed, the UE shall be able to communicate with the SS using the new ciphering configurations. This can be verified in SS through the reception of a correctly ciphered and integrity-protected COUNTER CHECK RESPONSE message.

Expected sequence

Step		ction	Message	Comment
	UE	SS		
1				UE is initially in CELL_FACH
				state.
2		\leftarrow	SECURITY MODE COMMAND	IE "Ciphering Algorithm
				capability" is set to an invalid
				value
3		\rightarrow	SECURITY MODE FAILURE	IE "Failure Cause" shall be set
				to "Protocol Error" and IE
				"Protocol Error Information"
				shall be set to "Information
				element value not
4		-	SECURITY MODE COMMAND	comprehended". See specific message
4			SECURITY WIDDE COMMINIAND	contents.
5		\rightarrow	SECURITY MODE COMPLETE	SS verifies that this message
3		,	SECONTT WODE COM LETE	is sent unciphered. SS records
				the uplink ciphering activation
				time for RB 2.
6		-	COUNTER CHECK	SS repeats step 6 and step 7
				until its internal uplink and
				downlink RLC sequence
				numbers have both surpassed
				the uplink and downlink
				ciphering activation times
				specified for RB2. This
				message is sent on the
				downlink DCCH using RLC-
				AM. See specific message
7		\rightarrow	COUNTED CHECK DECDONICE	content.
'		フ	COUNTER CHECK RESPONSE	UE shall send this message
				on the uplink DCCH using RLC-AM. See specific
				message content.
8				SS verifies that the last
				COUNTER CHECK
				RESPONSE message is both
				integrity-protected and
				ciphered correctly.

Specific Message Contents

SECURITY MODE COMMAND (Step 2)

See notes below for the value of Y.

Information Element	Value/remark
RRC transaction identifier	<u>0</u>
Integrity check info	
Message authentication code	Calculated result in SS
RRC Message sequence number	0
Security Capability	
Ciphering algorithm capability	"11111111111111"B Spare value
Integrity protection algorithm capability	"11111111111111"BSpare value
Ciphering mode info	
Ciphering mode command	Start
Activation time for DPCH	Not Present
Radio bearer downlink ciphering activation time info	
RB Identity	2
RLC sequence number	Current RLC SN + Y
Integrity protection mode info	
Integrity protection mode command	Start
Downlink integrity protection activation info	Not Present
Integrity protection algorithm	If integrity is indicated to be active on IXIT statements in
	TS 34.123-2, use one of the supported integrity
	algorithms
Integrity protection initialisation number	0000 0000 0000 0000 H (FRESH)
CN domain identity	Supported domain

SECURITY MODE COMMAND (Step 4)

See notes below for the value of Y.

Information Element	Value/remark
RRC transaction identifier	<u>0</u>
Integrity check info	
Message authentication code	Calculated result in SS
RRC Message sequence number	0
Security Capability	
Ciphering algorithm capability	If ciphering is indicated to be active on IXIT statements
	in TS 34.123-2, use one of the supported ciphering
	algorithms
Integrity protection algorithm capability	00000000000010B(UIA1)
Ciphering mode info	Start
Ciphering mode command	Not Present
Activation time for DPCH	
Radio bearer downlink ciphering activation time info	2
RB Identity	Current RLC SN + Y
RLC sequence number	Start
Integrity protection mode info	Not Present
Integrity protection mode command	If integrity is indicated to be active on IXIT statements in
Downlink integrity protection activation info	TS 34.123-2, use one of the supported integrity
Integrity protection algorithm	algorithms
Integrity protection initialisation number	0000 0000 0000 0000 H (FRESH)
CN domain identity	Supported domain

SECURITY MODE COMPLETE (Step 5)

Information Element	Value/remark
RRC transaction identifier	<u>0</u>
Integrity check info	
- Message Authentication code	Checked to see if present
- RRC Message sequence number	Checked to see if present
Uplink integrity protection activation info	
 RRC message sequence number list 	Check to see if it the RRC SN for RB 0 to RB 4 are
	present
Radio bearer uplink ciphering activation info	
- RB Identity	2
- RLC sequence number	SS records this value. See step 8 in 'expected sequence'

COUNTER CHECK (Step 6)

Information Element	Value/remark
RRC transaction identifier	<u>0</u>
Integrity check info	Calculated value
RB COUNT-C MSB information	
- RB identity	2
- COUNT-C MSB uplink	Current COUNT-C MSB for RB#2 in uplink
- COUNT-C MSB downlink	Current COUNT-C MSB for RB#2 in downlink

COUNTER CHECK RESPONSE (Step 7)

Information Element	Value/remark
RRC transaction identifier	<u>0</u>
Integrity check info	Checked to see if the MAC code match
RB COUNT-C information	Check to if this IE is absent
- RB identity	
- COUNT-C uplink	
- COUNT-C downlink	

Note: Y = 2 * (size of COUNTER CHECK message, after PER encoding). The unit of Y is the number of RLC-AM PDU.

8.1.7.2.5 Test requirement

After step 2 the UE shall transmit a SECURITY MODE FAILURE message to report the protocol error detected in the first SECURITY MODE COMMAND message. The UE shall be able to communicate normally with the SS, with all control data on the signalling radio bearers unciphered.

After step 4 the UE shall RLC-acknowledge the receipt of the SECRUITY MODE COMMAND message using unciphered mode and which includes calculated integrity check info. SS checks that the SECURITY MODE COMPLETE message is received unciphered and that the calculated MAC-I values in "integrity check info" IE is correct.

After step 5 SS verifies that all uplink signalling messages on RB1, RB2, RB3 and RB4 are integrity protected with UIA1 algorithm.

After step 7 SS verifies that the last COUNTER CHECK RESPONSE message received is integrity protected with UIA1 algorithm and ciphered with the algorithm indicated in the second SECURITY MODE COMMAND (Step 4) message.

8.1.8 Counter check

8.1.8.1 Counter check in CELL DCH state

8.1.8.1.1 Definition

8.1.8.1.2 Conformance requirement

When the UE receives a COUNTER CHECK message that includes matched COUNT-C MSB values, the UE shall transmit a COUNTER CHECK RESPONSE message on the uplink DCCH and omitting "RB COUNT-C information" IE in this message.

When the UE receives a COUNTER CHECK message that includes any mismatched COUNT-C MSB values, the UE shall transmit a COUNTER CHECK RESPONSE message on the uplink DCCH and specifies the current COUNT-C information of the RAB(s) with mismatched values.

Reference

3GPP TS 25.331 clause 8.1.15

8.1.8.1.3 Test purpose

To confirm that the UE transmits a COUNTER CHECK RESPONSE message after it receives a COUNTER CHECK message from the SS. To confirm that the UE responds to the reception of an invalid downlink COUNTER CHECK message by transmitting a RRC STATUS message on the uplink DCCH, stating the correct error cause value in message.

8.1.8.1.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CELL_DCH state (state 6-9 or state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is brought to the CELL_DCH state after a successful outgoing call attempt. The SS transmits an erroneous COUNTER CHECK message. The content of this message is <u>lack of a mandatory IEa concatenation of the message type IE and an arbitrary 32 bits string</u>. The UE shall detect a protocol error and send RRC STATUS message to report this event. Next, the SS transmits a COUNTER CHECK message that includes the current COUNT-C MSB information in each radio access bearer. The UE shall react by sending a COUNTER CHECK RESPONSE message on the uplink DCCH, which does not include "RB COUNT-C information" IE. The SS transmits a COUNTER CHECK message which includes the current COUNT-C MSB information reversed all the bits in each radio bearer. The UE shall send a COUNTER CHECK RESPONSE message on the uplink DCCH, specifying the current COUNT-C information for each radio access bearer established.

Expected sequence

Step	Direc	tion	Message	Comment
	UE	SS		
1				The UE is brought to CELL_DCH state after an outgoing call has been established successfully.
2	*	.	COUNTER CHECK	See specific message contents for this messageThis message contains an arbitrary 32- bits pattern, following the IE "Message Type"
3	=)	RRC STATUS	UE shall detect a protocol error and then transmit this message.
4	+	-	COUNTER CHECK	See specific message content.
5	_	>	COUNTER CHECK RESPONSE	The message shall not include the IE "RB COUNT-C information".
6	•	-	COUNTER CHECK	See specific message content.
7	-	>	COUNTER CHECK RESPONSE	The message shall include the IE "RB COUNT-C information".

Specific Message Contents

COUNTER CHECK (Step 2)

Information Element	Value/remark
RRC transaction identifier Unknown bit pattern	Not PresentAn arbitrary 32-bits string

RRC STATUS (Step 3)

Information Element	Value/remark
Protocol Error Information	
- Protocol Error Cause	Checked to see if set to "ASN.1 violation or encoding
	error" or "conditional information element error"

COUNTER CHECK (Step 4)

Information Element	Value/remark
RRC transaction identifier	0
Integrity check info	Calculated value
RB COUNT-C MSB information	
- RB identity	<u>520</u>
- COUNT-C MSB uplink	Current COUNT-C MSB for RB#5-20 in uplink
- COUNT-C MSB downlink	Current COUNT-C MSB for RB#5-20 in downlink

COUNTER CHECK RESPONSE (Step 5)

Information Element	Value/remark
RRC transaction identifier	<u>0</u>
Integrity check info	Not checked
RB COUNT-C information	Check to if this IE is absent
- RB identity	
- COUNT-C uplink	
- COUNT-C downlink	

COUNTER CHECK (Step 6)

Information Element	Value/remark
RRC transaction identifier	<u>0</u>
Integrity check info	<u>Calculated value</u>
RB COUNT-C MSB information	
- RB identity	Check to see if set to 420
- COUNT-C MSB uplink	Toggle all bits of the current COUNT-C MSB in uplink
	for RB# 5 20
- COUNT-C MSB downlink	Toggle all bits of the current COUNT-C MSB in downlink
	for RB#520

COUNTER CHECK RESPONSE (Step 7)

Information Element	Value/remark
RRC transaction identifier	<u>0</u>
Integrity check info	Not checked
RB COUNT-C information	
- RB identity	Check to see if set to 520
- COUNT-C uplink	Check to see if set to Current COUNT-C for RB#5-20 in
	uplink
- COUNT-C downlink	Check to see if set to COUNT-C for RB#5-20 in
	downlink

8.1.8.1.5 Test requirement

After step 2, the UE shall transmit a RRC STATUS message on the uplink DCCH. The protocol error cause shall be set to "ASN.1 violation or encoding" or "conditional information element error".

After step 4 the UE shall transmit a COUNTER CHECK RESPONSE message which does not includes the IE "RB COUNT-C information" to indicates that a matched comparison result is obtained.

After step 6, the UE shall transmit a COUNTER CHECK RESPONSE message which includes the IE "RB COUNT-C information" to report that a mismatch in COUNT-C value is detected in RB#520.

8.1.8.2 Counter check in CELL FACH state

8.1.8.2.1 Definition

8.1.8.2.2 Conformance requirement

When the UE receives a COUNTER CHECK message that includes matched COUNT-C MSB values, the UE shall transmit a COUNTER CHECK RESPONSE message on the uplink DCCH and omitting "RB COUNT-C information" IE in this message.

When the UE receives a COUNTER CHECK message that includes any mismatched COUNT-C MSB values, the UE shall transmit a COUNTER CHECK RESPONSE message on the uplink DCCH and specifies the current COUNT-C information of the RAB(s) with mismatched values.

Reference

3GPP TS 25.331 clause 8.1.15

8.1.8.2.3 Test purpose

To confirm that the UE transmits a COUNTER CHECK RESPONSE message after it receives a COUNTER CHECK message from the SS. To confirm that the UE responds to the reception of an invalid downlink COUNTER CHECK message by transmitting a RRC STATUS message on the uplink DCCH, stating the correct error cause value in message.

8.1.8.2.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CELL_FACH state (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is brought to the CELL_FACH state after a successful outgoing call attempt. The SS transmits an erroneous COUNTER CHECK message. The content of this message is <u>lack of a mandatory IEa concatenation of the message type IE and an arbitrary 32 bits string</u>. The UE shall detect a protocol error and send RRC STATUS message to report this event. Next, the SS transmits a COUNTER CHECK message that includes the current COUNT-C MSB information in each radio access bearer. The UE shall react by sending a COUNTER CHECK RESPONSE message on the uplink DCCH, which does not include "RB COUNT-C information" IE. The SS transmits a COUNTER CHECK message, which includes the current COUNT-C MSB information for each radio bearer but with all the bits reversed. The UE shall send a COUNTER CHECK RESPONSE message on the uplink DCCH, specifying the current COUNT-C information for each radio access bearer established.

Expected sequence

Step	Dire	ction	Message	Comment
	UE	SS		
1				The UE is brought to CELL_FACH state after an outgoing call has been established successfully.
2		(COUNTER CHECK	See specific message contents for this messageThis message contains an arbitrary 32- bits pattern, following the IE "Message Type"
3		→	RRC STATUS	UE shall detect a protocol error and then transmit this message.
4	•	(COUNTER CHECK	See specific message content.
5)	COUNTER CHECK RESPONSE	The message shall not include the IE "RB COUNT-C information".
6	•	(COUNTER CHECK	See specific message content.
7		→	COUNTER CHECK RESPONSE	The message shall include the IE "RB COUNT-C information".

Specific Message Contents

COUNTER CHECK (Step 2)

	Information Element	Value/remark
l	RRC transaction identifier Unknown bit pattern	Not PresentAn arbitrary 32-bits string

RRC STATUS (Step 3)

Information Element	Value/remark
Protocol Error Information - Protocol Error Cause	Checked to see if set to "ASN.1 violation or encoding" or
	"conditional information element error"

COUNTER CHECK (Step 4)

	Information Element	Value/remark
	RRC transaction identifier	<u>0</u>
•	Integrity check info	Calculated value
	RB COUNT-C MSB information	
	- RB identity	5 20
	- COUNT-C MSB uplink	Current COUNT-C MSB for RB#5-20 in uplink
	- COUNT-C MSB downlink	Current COUNT-C MSB for RB#5-20 in downlink

COUNTER CHECK RESPONSE (Step 5)

Information Element	Value/remark
RRC transaction identifier	<u>0</u>
Integrity check info	Not checked
RB COUNT-C information	Check to if this IE is absent
- RB identity	
- COUNT-C uplink	
- COUNT-C downlink	

COUNTER CHECK (Step 6)

Information Element	Value/remark
RRC transaction identifier	<u>0</u>
Integrity check info	<u>Calculated value</u>
RB COUNT-C MSB information	
- RB identity	Check to see if set to 420
- COUNT-C MSB uplink	Toggle all bits of the current COUNT-C MSB in uplink
	for RB# <u>520</u>
- COUNT-C MSB downlink	Toggle all bits of the current COUNT-C MSB in downlink
	for RB# <u>520</u>

COUNTER CHECK RESPONSE (Step 7)

Information Element	Value/remark
RRC transaction identifier	<u>0</u>
Integrity check info	Not checked
RB COUNT-C information	
- RB identity	Check to see if set to 520
- COUNT-C uplink	Check to see if set to Current COUNT-C for RB#5-20 in uplink
- COUNT-C downlink	Check to see if set to COUNT-C for RB# <u>5</u> _20_in downlink

8.1.8.2.5 Test requirement

After step 2, the UE shall transmit a RRC STATUS message on the uplink DCCH. The protocol error cause shall be set to "ASN.1 violation or encoding" or "conditional information element error".

After step 4 the UE shall transmit a COUNTER CHECK RESPONSE message which does not includes the IE "RB COUNT-C information" to indicates that a matched comparison result is obtained.

After step 6, the UE shall transmit a COUNTER CHECK RESPONSE message which includes the IE "RB COUNT-C information" to report that a mismatch in COUNT-C value is detected in RB#520.

8.1.9 Signalling Connection Release Request

8.1.9.1 Definition

8.1.9.2 Conformance requirement

The UE shall initiate the signalling connection release procedure when the higher layer entities in the UE request to release one or more signalling session (one example of such case is location update failure). In this case, the UE shall transmit a SIGNALLING CONNECTION RELEASE REQUEST message, which includes the <u>CN domain identity flow identifier</u> of the <u>signalling connection</u> flow to be released.

Reference

3GPP TS 25.331 clause 8.1.14

8.1.9.3 Test purpose

To confirm that the UE transmits a SIGNALLING CONNECTION RELEASE REQUEST message after it fails to receive a response for the LOCATION UPDATING REQUEST message.

8.1.9.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: Switched off (state 1) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is turned on and it shall find a suitable cell to camp on. The UE shall initiate a location updating procedure. The UE shall establish an RRC connection and transmits a LOCATION UPDATING REQUEST message using the INITIAL DIRECT TRANSFER message. The SS does not respond to this message, and the UE shall send a SIGNALLING CONNECTION RELEASE REQUEST message which includes the <a href="Months In Initial Content of the Initial Con

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1			The UE is powered on.
2	→	RRC CONNECTION REQUEST	UE shall initiate the location updating procedure.
3	-	RRC CONNECTION SETUP	apading procedure.
4			The UE configures the layer 2 and layer 1.
5	\rightarrow	RRC CONNECTION SETUP COMPLETE	
6	\rightarrow	INITIAL DIRECT TRANSFER (LOCATION UPDATING REQUEST)	LOCATION UPDATE REQUEST is embedded in this message transmission.
7			The SS does not respond and waits until the timer for location update procedure expires.
8	→	SIGNALLING CONNECTION RELEASE REQUEST	

Specific Message Content

SIGNALLING CONNECTION RELEASE REQUEST (Step 8)

Information Element	Value/remark
CN domain identityFlow Identifier	Check to see if this value is the as same as in the uplink INITIAL DIRECT TRANSFER message.

8.1.9.5 Test requirement

After step 1 the UE shall initiate the LOCATION UPDATING procedure and establish an RRC connection.

After step 7 the UE shall transmit a SIGNALLING CONNECTION RELEASE REQUEST message which includes the same <u>CN domain identity</u> flow identifier as that found in the INITIAL DIRECT TRANSFER message.

8.2 Radio Bearer control procedure

8.2.1 Radio Bearer Establishment

8.2.1.1 Radio Bearer Establishment for transition from CELL_DCH to CELL_DCH: Success (Data integrity protection algorithm is not applied)

8.2.1.1.1 Definition

8.2.1.1.2 Conformance requirement

The UE shall correctly set up a radio bearer according to a RADIO BEARER SETUP message and communicate with the UTRAN for its implementation.

(This is the case where data integrity protection algorithm is not applied.)

Reference

3GPP TS 25.331 clause 8.2.1

8.2.1.1.3 Test purpose

To confirm that the UE establishes a new radio bearer according to a RADIO BEARER SETUP message received from the SS.

8.2.1.1.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CS-DCCH_DCH (state 6-5) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state, after the test operator is prompted to make an out-going call. Before step 1, only signalling radio bearers have been established. The SS transmits a RADIO BEARER SETUP message to the UE after it sets up L1 including the start of tx/rx. This message requests the establishment of RABs for carrying the traffic of the speech call. After the UE receives this message, it configures them and establishes a radio bearer. Finally the UE transmits a RADIO BEARER SETUP COMPLETE message using AM RLC. Then the UE and the SS enters the communicating state.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1	1 ←		RADIO BEARER SETUP	This message do not contain IE
				"integrity check info" and
				"integrity protection mode info"
2	2 →		RADIO BEARER SETUP	This message do not contain
			COMPLETE	"integrity check info" and "Uplink
				integrity activation info
3	3			To confirm the communication.

Specific Message Contents

For RADIO BEARER SETUP in step 1, using the message sub-type indicated as "Speech in CS" or "Non-speech in CS" found in default message content clause 9 of TS 34.108 and Annex A of TS 34.123-1, respectively.

8.2.1.1.5 Test requirement

After step 2 the UE shall communicate with the SS on the radio bearer for its implementation.

8.2.1.2 Radio Bearer Establishment for transition from CELL_DCH to CELL_DCH: Success (Effected Data integrity protection algorithm)

8.2.1.2.1 Definition

8.2.1.2.2 Conformance requirement

The UE shall correctly set up a radio bearer according to a RADIO BEARER SETUP message, which applies data integrity function, and then communicate with the UTRAN for its implementation.

Reference

3GPP TS 25.331 clause 8.2.1, 8.5.11.

8.2.1.2.3 Test purpose

To confirm that the UE establishes a new radio bearer according to a RADIO BEARER SETUP message which contains IE "Integrity check info" and IE "Integrity protection mode info" received from the SS.

8.2.1.2.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CS-DCCH_DCH (state 6-5) as specified in clause 7.4 of TS 34.108 and data integrity algorithm is not applied

Test Procedure

The UE is in the CELL_DCH state, after the test operator is prompted to make an out-going data call. Before step 1, only signalling radio bearers have been established. The SS transmits a RADIO BEARER SETUP message which is including IE "integrity check info" and "integrity protection mode info" to the UE after it sets up L1 including the start of tx/rx. This message requests the establishment of RABs for carrying the traffic of the call. After the UE receives this message, it configures them and establishes a radio bearer. Finally the UE transmits a RADIO BEARER SETUP COMPLETE message which is including IE "integrity check info" using AM RLC. Then the UE and the SS enters the communicating state.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1	•	+	RADIO BEARER SETUP	This message contain IE "integrity check info" and "integrity protection mode info"
2		\rightarrow	RADIO BEARER SETUP COMPLETE	This message contain "integrity check info"
3				To confirm the communication.

Specific Message Contents

For RADIO BEARER SETUP in step 1, using the message sub-types indicated as "Speech in CS" or "Non-spech in CS" as found in default message content clause 9 of TS 34.108 and Annex A, respectively.

RADIO BEARER SETUP

The contents of RADIO BEARER SETUP message in this test case is identical with the following exceptions:

Information Element	Value/remark
Integrity check info	
- Message authentication code	Bit string (32) MAC-I See TS 33.102
- RRC Message sequence number	0
Integrity protection mode info	Present
- Integrity protection mode command	"Start"
- Downlink integrity protection activation info	Not present (It is needed only when the IE "Integrity
	protection mode command" has the value "modify".)
- integrity protection algorithm	UIA1
- integrity protection initialisation number	Bit string (32) FRESH See TS 33.102

RADIO BEARER SETUP COMPLETE

Information Element	Value/remark
Integrity check info	
- Message authentication code	Not checked(MAC-I See TS 33.102)
- RRC Message sequence number	Not checked
Uplink Integrity protection activation info	Not checked
Hyper Frame Number	Not checked

8.2.1.2.5 Test requirement

After step 2 the UE shall communicate with the SS on the radio bearer for its implementation. This can be verified by the correct reproduction of the u-plane data transmitted and received between the test operator and SS.

8.2.1.3 Radio Bearer Establishment for transition from CELL_DCH to CELL_DCH: Failure (Unsupported configuration)

8.2.1.3.1 Definition

8.2.1.3.2 Conformance requirement

The UE shall keep its old configuration when the UE receives a RADIO BEARER SETUP message which includes unsupported configuration parameters and transmit a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC which is set to "configuration unsupported" in IE "failure cause".

Reference

3GPP TS 25.331 clause 8.2.1

8.2.1.3.3 Test purpose

To confirm that the UE keeps its configuration and transmits a RADIO BEARER SETUP FAILURE message in case of receiving a RADIO BEARER SETUP message which includes parameters of its unsupported configuration.

8.2.1.3.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CS-DCCH_DCH (state 6-5) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a RADIO BEARER SETUP message as the frequency cannot be supported by the UE. After the UE receives this message, it transmits a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC which is set to "configuration unsupported" in IE "failure cause".

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1		-	RADIO BEARER SETUP	Including the unsupported configuration for the UE.
2		\rightarrow	RADIO BEARER SETUP FAILURE	The UE does not change the configuration.

Specific Message Contents

RADIO BEARER SETUP

The contents of RADIO BEARER SETUP message in this test case is indicated as "Speech in CS" or "Non-speech in CS" as found in default message content clause 9 of TS 34.108 and Annex A respectively with the following exceptions:

Information Element	Value/remark
Frequency info	
- UARFCN uplink(Nu)	16383.
- UARFCN downlink(Nd)	Not Present.

RADIO BEARER SETUP FAILURE

Information Element	Value/remark
Message Type	
Failure cause	Configuration unsopportedunsupported
Other information element	Not checked

8.2.1.3.5 Test requirement

After step 1 the UE shall keep its configuration and transmit a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC which is set to "configuration unsupported" in IE "failure cause".

8.2.1.4 Radio Bearer Establishment for transition from CELL_DCH to CELL_DCH: Failure (Physical channel Failure and successful reversion to old configuration)

8.2.1.4.1 Definition

8.2.1.4.2 Conformance requirement

The UE shall revert to the old configuration when the UE fails to configure the new radio bearer by the T312 expiry and transmit a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC which is set to "physical channel failure" in IE "failure cause".

Reference

3GPP TS 25.331 clause 8.2.1

8.2.1.4.3 Test purpose

To confirm that the UE reverts to the old configuration and transmits a RADIO BEARER SETUP FAILURE message when the UE fails to configure the new radio bearer following detection of physical channel failure after T312 expiry.

8.2.1.4.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CS-DCCH_DCH (state 6-5) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a RADIO BEARER SETUP message to the UE and does not configure the new radio bearer. Then after T312 expiry, the UE reverts to the old configuration and transmits a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC which is set to "physical channel failure" in IE "failure cause".

Expected sequence

Step	Dire	ction	Message	Comment
	UE	SS		
1		←	RADIO BEARER SETUP	The SS does not configure the new radio bearer stated in the message.
2				The UE does not configure the new radio bearer and reverts to the old configuration.
3		→	RADIO BEARER SETUP FAILURE	UE shall transmit this message using the old RRC signalling bearer operating in RLC-AM mode.

Specific Message Contents

RADIO BEARER SETUP

The contents of RADIO BEARER SETUP message in this test case is indicated as "Speech in CS" or "Non-speech in CS" as found in default message content clause 9 of TS 34.108 and Annex A, respectively.

RADIO BEARER SETUP FAILURE

Information Element	Value/remark
Message Type	
Failure cause	Physical channel failure
Other information element	Not checked

8.2.1.4.5 Test requirement

After step 2 the UE shall revert to the old configuration and transmit a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC which is set to "physical channel failure" in IE "failure cause".

8.2.1.5 Radio Bearer Establishment for transition from CELL_DCH to CELL_DCH: Failure (Physical channel Failure and reversion failure)

8.2.1.5.1 Definition

8.2.1.5.2 Conformance requirement

The UE shall perform a <u>cell update n RRC connection re establishment-procedure</u> when the UE fails to revert to the old configuration after the detection of physical channel failure in the radio bearer establishment procedure. <u>After the UE complete cell update procedure</u>, the UE transmit RADIO BEARER SETUP FAILURE message on the DCCH using <u>AM RLC</u> which is set to "physical channel failure" in IE "failure cause".

Reference

3GPP TS 25.331 clause 8.2.1

8.2.1.5.3 Test purpose

To confirm that <u>UE transmits RADIO BEARER SETUP FAILURE message after the UEit completes initiates</u> an <u>cell update RRC connection re establishment</u> procedure when the UE cannot establish the new radio bearer for the L1 configuration and cannot revert to the old configuration.

8.2.1.5.4 Method of test

Initial Condition

System Simulator: 2 cells- Cell 1 is active, Cell 2 is inactive

UE: CS-DCCH DCH (state 6-5) as specified in clause 7.4 of TS 34.108 in cell No.1

Test Procedure

The UE is in the CELL_DCH state in cell 1. The SS begins to broadcast the BCCH in cell 2, with a power level suitable for camping but lower than that of cell 1. Next, SS transmits a RADIO BEARER SETUP message to the UE. The SS transmits a RADIO BEARER SETUP message to the UE. After transmitting the RADIO BEARER SETUP message, the SS shall not configure L1 in accordance to the settings in the message. At the same time, SS begins to broadcast the BCCH in a cell 2. The UE recognize that it cannot configure the new radio bearer and wants to revert to the old configuration, but the UE cannot revert to the old configuration. The UE shall select Cell 2 by performing cell reselection and enter CELL FACH state. The UE transmit CELL UPDATE message on uplink CCCH with IE "Cell update cause" set to "radio link failure". The SS shall transmit CELL UPDATE CONFIRM message on downlink CCCH after receiving CELL UPDATE message. The UE transmits UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC and subsequently transmits RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC, setting the value of IE "failure cause" to "physical channel failure".

As the SS does not configure the new radio bearer and deletes the old configuration. The UE fails to configure L1 and fails to revert to the old configurationThen the UE shall find cell 2 and transmits to it an RRC CONNECTION RE ESTABLISHMENT REQUEST message, which includes the IE "U RNTI" on the uplink CCCH. The SS transmits an RRC CONNECTION RE ESTABLISHMENT message, which includes a new TFCS according to the new transport channel. After the UE receives an RRC CONNECTION RE ESTABLISHMENT message, it reconfigures the new radio connection and transmits an RRC CONNECTION RE ESTABLISHMENT COMPLETE message on the uplink DCCH using AM RLC.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
4		BCCH	The SS transmits a BCCH in cell 2 but with a lower power level than in cell 1.
<u>21</u>	+	RADIO BEARER SETUP	The SS does not configure the new radio bearer and delete the old configuration.
<u>2</u>	<u>←</u>	BCCH	The SS does not configure the new radio bearer and starts to transmit the BCCH in cell 2
3			The UE shall detect a failure to configure the new radio bearers and also subsequently, cannot revert to old configuration.
4			The UE select the cell 2 and enter CELL FACH state.
4 <u>5</u>	→	CELL UPDATERRC CONNECTION RE-ESTABLISHMENT REQUEST	The value "radio link failure" shall be set in IE "Cell update cause". The UE shall detect the presence of cell 2 and transmits this message which includes the IE "U-RNTI" it has been assigned to.
5 6	+	CELL UPDATE CONFIRMRRC CONNECTION RE-ESTABLISHMENT	This message include IE "new U-RNTI" and IE "new C-RNTI" Including the new TFCS according to the new transport channel.
<u>67</u>	→	UTRAN MOBILITY INFORMATION CONFIRMRRC CONNECTION RE-ESTABLISHMENT COMPLETE	This message include IE "new U-RNTI" and IE "new C-RNTI"
<u>8</u>		RADIO BEARER SETUP FAILURE	The IE "failure cause" shall be set to "physical channel failure"

Specific Message Contents

RADIO BEARER SETUP (Step 1)

The contents of RADIO BEARER SETUP message in this test case is indicated as "Speech in CS" or "Non-speech in CS" as found in default message content clause 9 of TS 34.108 and Annex A

CELL UPDATE (Step 5)

The contents of CELL UPDATE message is identical as "Contents of CELL UPDATE message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	
- SRNC Identity	Assigned previously in cell 1
- S-RNTI	Assigned previously in cell 1
Cell Update Cause	"radio link failure"

CELL UPDATE CONFIRM (Step 7)

The contents of CELL UPDATE CONFIRM message is identical as "CELL UPDATE CONFIRM message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	Same as CELL UPDATE message in step 7
New U-RNTI	
- SRNC Identity	<u>'0000 0000 0000 0001'</u>
- S-RNTI	Different from previous S-RNTI
New C-RNTI	Different from previous C-RNTI

RADIO BEARER RECONFIGURATION FAILURE (Step 8)

Information Element	<u>Value/remark</u>
Message Type	"RADIO BEARER RECONFIGURATION FAILURE"
Failure cause	"physical channel failure"
Other information element	Not checked

8.2.1.5.5 Test requirement

After step 3 the UE shall find a new cell 2, enter CELL_FACH state...

After step 8 the UE shall transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

After step 9 the UE shall transmit RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC, setting the IE "failure cause" to "physical channel failure".

and transmit RRC CONNECTION RE ESTABLISHMENT REQUEST message.

After step 6 the UE shall re establish an RRC connection in cell 2.

8.2.1.6 Radio Bearer Establishment for transition from CELL_DCH to CELL_DCH: Failure (Incompatible simultaneous reconfiguration)

8.2.1.6.1 Definition

8.2.1.6.2 Conformance requirement

If the UE receives a RADIO BEARER SETUP message whilst reconfiguring due to a radio bearer message other than RADIO BEARER SETUP, then it shall keep its configuration as if the RADIO BEARER SETUP message had not been received.

If the UE is reconfiguring itself after receiving a radio bearer message other than RADIO BEARER SETUP message, and when the UE subsequently receive a RADIO BEARER SETUP message, the UE shall transmit an RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC which is set to "incompatible simultaneous reconfiguration" in IE "failure case".

Reference

3GPP TS 25.331 clause 8.2.1

8.2.1.6.3 Test purpose

To confirm that if the UE receives a RADIO BEARER SETUP message whilst reconfiguring due to a radio bearer message other than RADIO BEARER SETUP, it shall keep its configuration as if the RADIO BEARER SETUP message had not been received and complete the reconfiguration according to the previously received message.

To confirm that the UE keeps its configuration and transmits an RADIO BEARER SETUP FAILURE message when the UE receives a RADIO BEARER SETUP message before the UE configures the radio bearer according to a message.

8.2.1.6.4 Method of test

Initial Condition

System Simulator: 1 cell

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a RADIO BEARER RECONFIGURATION message to the UE. The SS transmits a RADIO BEARER SETUP message before the "activation time" indicated in the RADIO BEARER RECONFIGURATION message expires. When the UE receives the RADIO BEARER SETUP message, the UE shall keep the configuration as if it had not received the RADIO BEARER SETUP message and shall transmit a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC with IE "failure cause" set to "incompatible simultaneous reconfiguration". After the SS receives the RADIO BEARER SETUP FAILURE message, the UE reconfigures the new physical channel parameters and transmits a RADIO BEARER RECONFIGURATION COMPLETE message on DCCH using AM RLC. The UE is in the CELL_DCH state. When the SS transmits a RADIO BEARER SETUP message to the UE before the UE configures the radio bearer the UE keeps the old configuration and transmits a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC which is set to "incompatible simultaneous reconfiguration in IE "failure cause".

Expected sequence

Step	Direction		Message	Comment
	UE	SS	-	
1	+		RADIO BEARER RECONFIGURATION	Including IE "Uplink DPCH info"
2	←		RADIO BEARER SETUP	The SS send this message before the expiry of activation time specified in the message of step 1.
3	→		RADIO BEARER SETUP FAILURE	The UE does not change the configuration because of receiving the RADIO BEARER SETUP message.
<u>4</u>	<u>→</u>		RADIO BEARER RECONFIGURATION COMPLETE	This message is on DCCH using AM RLC.

Specific Message Contents

For RADIO BEARER RECONFIGURATION in step 1, use the message sub-type indicated as "Speech in CS" found in Annex A, with the exception of the following Information Elements:

RADIO BEARER RECONFIGURATION (Step 1)

Information Element	Value/remark	
Activation Time	[256+Current CFN-[current CFN mod 8 + 8]]MOD 256	

RADIO BEARER SETUP (Step 2)

The contents of RADIO BEARER SETUP message in this test case is indicated as "Speech in CS" or "Non-speech in CS" as found in default message content clause 9 of TS 34.108 and Annex A respectively with the following exceptions:

Information Element	Value/remark
Activation Time	Not Present

RADIO BEARER SETUP FAILURE

Information Element	Value/remark
Message Type	
Failure case	Incompatible simultaneous reconfiguration
Other information element	Not checked

8.2.1.6.5 Test requirement

After step 1 The SS transmits a RADIO BEARER SETUP message before the expiry of the activation time specified in the message of step 1.

After step 2 the UE shall keep its configuration as if the UE had not received the RADIO BEARER SETUP message and shall transmit a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC with IE "failure cause" set to "Incompatible simultaneous reconfiguration".

After step 4 the UE communicates with the SS on the DCCH and DTCH using the new physical channel parameters reconfigured as a result of the RADIO BEARER RECONFIGURATION message.

After step 2 the UE shall keep its configuration and transmit a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC which is set to "Incompatible simultaneous reconfiguration" in IE "failure cause".

8.2.1.7 Radio Bearer Establishment for transition from CELL_DCH to CELL_DCH: Failure (Invalid message reception and Invalid configuration)

8.2.1.7.1 Definition

8.2.1.7.2 Conformance requirement

The UE shall keep its old configuration when the UE receives a RADIO BEARER SETUP message, which includes the spare-undefined value in the mandatory IE "undefined and criticality is defined as "Reject". Then it transmits a RADIO BEARER SETUP FAILURE message which is set to "protocol error" in IE "failure cause" and is set to "Information element value not comprehended" in IE "Protocol error cause". The UE shall keep existing configuration before reception of a RADIO BEARER SETUP message which includes some IEs set to invalid value, and then the UE shall transmit RADIO BEARER SETUP FAILURE including IE "failure cause" set to "invalid configuration".

Reference

3GPP TS 25.331 clause 8.2.1

8.2.1.7.3 Test purpose

To confirm that the UE transmits a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC if it receives a RADIO BEARER SETUP message which indicates the spare-undefined value in the mandatory IE "uTRANDRX.cycle.length.coefficientDRX.indicator" whose criticality is defined as "Reject".

To confirm that the UE transmits a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC if it receives a RADIO BEARER SETUP message including some IEs set to invalid value.

8.2.1.7.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CS-DCCH DCH (state 6-5) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a RADIO BEARER SETUP message to the UE which indicates a spare-undefined value in the mandatory IE "UTRAN DRX cycle length coefficientDRX indicator" whose criticality is defined as "Reject". The UE keeps the old configuration and transmits a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC which is set to "protocol error" in IE "failure cause", and is set to "Information element value not comprehended" in IE "Protocol error cause". The UE keeps initial configuration when SS transmits RADIO

BEARER SETUP message including some IEs set to invalid value. The UE transmit RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC, setting the value "invalid configuration" to IE "failure cause".

Expected sequence

Step	Direction		Message	Comment
-	UE	SS	_	
1	-		RADIO BEARER SETUP	
2	→		RADIO BEARER SETUP FAILURE	The UE does not change the configuration.
<u>3</u>	<u>←</u>		RADIO BEARER SETUP	This message includes IE set to invalid value.
<u>4</u>				The UE does not change the configuration.
<u>5</u>	<u>→</u>		RADIO BEARER SETUP FAILURE	The IE "failure cause" shall be set to "invalid configuration"

Specific Message Contents

RADIO BEARER SETUP (Step 1)

The contents of RADIO BEARER SETUP message in this test case is identical as "Speech in CS" or "Non-speech in CS" as found in default message content clause 9 of TS 34.108 and Annex A respectively with the following exceptions:

Information Element	Value/remark	
UTRAN DRX cycle length coefficientDRX indicator	spare Undefined value	

RADIO BEARER SETUP FAILURE (Step 2)

Information Element	Value/remark	
Message Type		
Failure cause		
- Failure cause	Protocol error	
- Protocol error information		
- Protocol error cause	Information element value not comprehended	
Other information element	Not checked	

RADIO BEARER SETUP (Step 3)

The contents of RADIO BEARER SETUP message in this test case is identical as "Speech in CS" or "Non-speech in CS" as found in default message content clause 9 of TS 34.108 and Annex A respectively with the following exceptions:

Information Element	<u>Value/remark</u>
Added or Reconfigured UL TrCH information	
 Transport channel identity 	<u>1</u>
TFS	
 Dynamic Transport format information 	(This IE is repeated for TFI number)
- RLC size	Reference to TS34.108 clause 6.10 Parameter Set
- CHOICE Logical Channel list	
- Explicit List	
- RB identity	<u>2</u>

RADIO BEARER SETUP FAILURE (Step 5)

Information Element	<u>Value/remark</u>
Message Type	
Failure cause	Invalid configuration
Other information element	Not checked

8.2.1.7.5 Test requirement

After step 1 the UE shall keep its old configuration and transmit a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC which is set to "protocol error" in IE "failure cause" and set to "Information element value not comprehended" in IE "Protocol error cause".

After step 3 the UE shall keep its old configuration

After step 4 the UE shall transmit RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC, setting the value "invalid configuration" to IE "failure cause".

8.2.1.8 Radio Bearer Establishment for transition from CELL_DCH to CELL_FACH: Success

8.2.1.8.1 Definition

8.2.1.8.2 Conformance requirement

The UE shall correctly set up a radio bearer according to a RADIO BEARER SETUP message and communicate with the UTRAN for its implementation.

Reference

3GPP TS 25.331 clause 8.2.1

8.2.1.8.3 Test purpose

To confirm that the UE establishes a new radio bearer according to a RADIO BEARER SETUP message received from the SS.

8.2.1.8.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH_DCH (state 6-7) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state, after the test operator is asked to initiate a packet-switched data call. The SS transmits a RADIO BEARER SETUP message to the UE after it sets up L1. After the UE receives this message, it configures them and establishes a radio bearer. Finally the UE transmits a RADIO BEARER SETUP COMPLETE message using AM RLC. Then the UE and the SS enters the communicating state.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	\	RADIO BEARER SETUP	SS requests test operator to make an outgoing packet-switched data call.
<u>2</u>			The UE select PRACH and S- CCPCH using SIB5 or SIB6 after entering CELL FACH state.
2 3	\rightarrow	RADIO BEARER SETUP COMPLETE	
<u>34</u>			To confirm the communication between UE and SS, based on the exchange of packets.

Specific Message Contents

For RADIO BEARER SETUP message in step 1, use the message sub-type indicated as "Packet to CELL_FACH from CELL_DCH in PS" found in Annex A.

8.2.1.8.5 Test requirement

After step 2-3 the UE shall communicate with the SS on the radio bearer for its implementation.

8.2.1.9 Radio Bearer Establishment for transition from CELL_DCH to CELL_FACH: Failure (Cell re-selection Physical channel Failure)

8.2.1.9.1 Definition

8.2.1.9.2 Conformance requirement

The UE shall initiate the cell update procedure when the UE failed to access the assigned physical channel according to a RADIO BEARER SETUP message when transiting from CELL_DCH state to CELL_FACH state. After the UE completes cell update procedure, the UE transmit RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC which sets IE "failure cause" to "cell reselection".

Reference

3GPP TS 25.331 clause 8.2.1

8.2.1.9.3 Test purpose

To confirm that the UE <u>transmit RADIO BEARER SETUP FAILURE message after it completes</u> perform a cell update procedure when the UE cannot use the assigned physical channel according to a RADIO BEARER SETUP message as it transits from CELL_DCH to CELL_FACH.

8.2.1.9.4 Method of test

Initial Condition

System Simulator: 2 cells- Cell 1 is active, Cell 2 is inactive

UE: PS-DCCH_DCH (state 6-7) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state in cell 1. The SS begins to broadcast the BCCH in cell 2 and transmits a RADIO BEARER SETUP message as the transition occurs from CELL_DCH to CELL_FACH and then broadcast BCCH in cell 2. The UE cannot access the assigned physical channel, as the SS does not transmit any data on the downlink common channel in cell 1. Then the UE shall initiate the cell update procedure in cell 2. The UE transmit CELL UPDATE message on uplink CCCH with IE "Cell update cause" set to "cell reselection". The SS shall transmit CELL UPDATE CONFIRM message on downlink CCCH after receiving CELL UPDATE message. The UE transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC and subsequently transmit RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC, setting the value of IE "failure cause" to "cell reselection".

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	+	BCCH	The SS broadcasts BCCH in cell
			No. 2.
2	←	RADIO BEARER SETUP	Assigned the transition from
			CELL_DCH to CELL_FACH
3			The UE cannot use the
			assigned the physical channel.
4	\rightarrow	CELL UPDATE	The UE finds a new cell 2 and
			enter CELL FACH state in cell 2
			to begins a cell update
			procedure. The value "cell
			reselection" set in IE "Cell
			update cause".
5	←	CELL UPDATE CONFIRM	This message include IE "new
			U-RNTI" and IE "new C-RNTI"".
			In the CELL_FACH stateIn the
			CELL_FACH state
<u>6</u>	\rightarrow	UTRAN MOBILITY INFORMATION	
		CONFIRM	
<u>7</u>	<u>→</u>	RADIO BEARER SETUP FAILURE	The IE "failure cause" shall be
			set to "cell reselection"

Specific Message Contents

RADIO BEARER SETUP (Step 1)

The contents of RADIO BEARER SETUP message in this test case is identical as "Packet to CELL_FACH from CELL_DCH in PS" as found in Annex A.

CELL UPDATE (Step 4)

The contents of CELL UPDATE message is identical as "Contents of CELL UPDATE message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	
- SRNC Identity	Assigned previously in cell 1
<u>- S-RNTI</u>	Assigned previously in cell 1
Cell Update Cause	<u>"radio link failure"</u>

CELL UPDATE CONFIRM (Step 5)

The contents of CELL UPDATE CONFIRM message is identical as "CELL UPDATE CONFIRM message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	Same as CELL UPDATE message in step 7
New U-RNTI	
- SRNC Identity	<u>'0000 0000 0000 0001'</u>
<u>- S-RNTI</u>	<u>Different from previous S-RNTI</u>
New C-RNTI	Different from previous C-RNTI

RADIO BEARER SETUP FAILURE (Step 7)

Information Element	<u>Value/remark</u>
Message Type	"RADIO BEARER RECONFIGURATION FAILURE"
Failure cause	"cell reselection"
Other information element	Not checked

8.2.1.9.5 Test requirement

After step_3 the UE shall find a new cell No.2 and enter to CELL_FACH state. Then the UE shall initiate a cell update procedure in cell No.2.

After step 5 transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

After step 7 RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC, setting the value of IE "failure cause" to "cell reselection".

8.2.1.10 Radio Bearer Establishment for transition from CELL_FACH to CELL_DCH: Success

8.2.1.10.1 Definition

8.2.1.10.2 Conformance requirement

The UE shall correctly set up a radio bearer according to a RADIO BEARER SETUP message and communicate with the UTRAN for its implementation.

Reference

3GPP TS 25.331 clause 8.2.1

8.2.1.10.3 Test purpose

To confirm that the UE establishes a new radio bearer according to a RADIO BEARER SETUP message received from the SS.

8.2.1.10.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH_FACH (state 6-8) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state, after SS prompts the test operator to initiate a packet-switched data call. The SS transmits a RADIO BEARER SETUP message to the UE after it sets up L1 including the start of tx/rx. After the UE receives this message, it configures them and establishes the required radio bearers. Finally the UE transmits a RADIO BEARER SETUP COMPLETE message using AM RLC. Then the UE and the SS enters the communicating state.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1	+		RADIO BEARER SETUP	
2	\rightarrow		RADIO BEARER SETUP COMPLETE	
3				To confirm the communication

Specific Message Contents

RADIO BEARER SETUP

The contents of RADIO BEARER SETUP message in this test case is identical the message sub-type indicated by "Packet to CELL DCH from CELL FACH in PS" in Annex A.

8.2.1.10.5 Test requirement

After step2 the UE shall communicate with the SS using the radio bearer indicated in RADIO BEARER SETUP message. Particularly, SS should be able to receive packet data using a terminal equipment (TE) attached to the UE.

8.2.1.11 Radio Bearer Establishment for transition from CELL_FACH to CELL_DCH: Failure (Unsupported configuration)

8.2.1.11.1 Definition

8.2.1.11.2 Conformance requirement

The UE shall keep its old configuration when the UE receives a RADIO BEARER SETUP message which includes an unsupported configuration and then transmit a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC, which sets value "configuration unsupported" in IE "failure cause".

Reference

3GPP TS 25.331 clause 8.2.1

8.2.1.11.3 Test purpose

To confirm that the UE keeps its configuration and transmits a RADIO BEARER SETUP FAILURE message in case of it receiving a RADIO BEARER SETUP message, which includes parameters of an unsupported configuration.

8.2.1.11.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CS-DCCH_FACH (state 6-86) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a RADIO BEARER SETUP message with a stated frequency that cannot be supported by the UE. After the UE receives this message, it shall transmit a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC setting value "configuration unsupported" in IE "failure cause".

Expected sequence

Step	Direction		tion Message	Comment	
	UE	SS	_		
1	+		RADIO BEARER SETUP	This message includes an unsupported configuration for the UE.	
2	→		RADIO BEARER SETUP FAILURE	The UE shall transmit this message using RLC-AM mode and do not change the current configuration.	

Specific Message Contents

RADIO BEARER SETUP

The contents of RADIO BEARER SETUP message in this test case is identical as "Packet to CELL_DCH from CELL_FACH in PS" as found in Annex A:

Information Element	Value/remark
Frequency info	
- UARFCN uplink(Nu)	0
- UARFCN downlink(Nd)	Not Present

RADIO BEARER SETUP FAILURE

Information Element	Value/remark
Message Type	
Failure cause	Configuration unsupported
Other information element	Not checked

8.2.1.11.5 Test requirement

After step 1 the UE shall keep its configuration and transmit a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC which is set to "configuration unsupported" in IE "failure cause".

8.2.1.12 Radio Bearer Establishment for transition from CELL_FACH to CELL_DCH: Failure (Physical channel Failure and successful reversion to old configuration)

8.2.1.12.1 Definition

8.2.1.12.2 Conformance requirement

The UE shall attempt to revert to the old configuration when the UE fails to configure the new radio bearer by the T312 expiry. It shall transmit a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC containing value "physical channel failure" in IE "failure cause".

Reference

3GPP TS 25.331 clause 8.2.1

8.2.1.12.3 Test purpose

To confirm that the UE reverts to the old configuration and transmits a RADIO BEARER SETUP FAILURE message when the UE fails to configure the new radio bearer when it detects physical channel failure, followed by the T312 expiry.

8.2.1.12.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH_FACH (state 6-8) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a RADIO BEARER SETUP message to the UE and does not configure the new radio bearer. After T312 expiry, the UE shall revert to the old configuration and transmit a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC. The content of the message shall indicate "physical channel failure" in IE "failure cause".

Expected sequence

Step	Dire	ction	Message	Comment
	UE	SS		
1		←	RADIO BEARER SETUP	The SS does not configure a new radio bearer.
2				The UE does not configure a new radio bearer but reverts to the old configuration.
3		\rightarrow	RADIO BEARER SETUP FAILURE	

Specific Message Contents

RADIO BEARER SETUP

The contents of RADIO BEARER SETUP message in this test case is identical the message sub-type indicated by "Packet to CELL_DCH from CELL_FACH in PS" in Annex A.

RADIO BEARER SETUP FAILURE

Information Element	Value/remark
Message Type	
Failure cause	Physical channel failure
Other information element	Not checked

8.2.1.12.5 Test requirement

After step 2 the UE shall revert to the old configuration and transmit a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC which is set to "physical channel failure" in IE "failure cause".

8.2.1.13 Radio Bearer Establishment for transition from CELL_FACH to CELL_DCH: Failure (Physical channel Failure and reversion failure)

8.2.1.13.1 Definition

8.2.1.13.2 Conformance requirement

The UE shall perform a <u>cell update n RRC connection re establishment</u>-procedure when the UE fails to revert to the old configuration after the detection of physical channel failure in the radio bearer establishment procedure. <u>After the UE completes</u> cell update procedure, the UE transmits RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC which set value to IE "failure cause" to "physical channel failure".

Reference

3GPP TS 25.331 clause 8.2.1

8.2.1.13.3 Test purpose

To confirm that the UE <u>transmit RADIO BEARER SETUP FAILURE message after it completes</u> <u>initiates a cell update</u> <u>n RRC connection re establishment procedure when the UE cannot establish the new radio bearer for the L1 configuration and subsequently fail to revert to the old configuration.</u>

8.2.1.13.4 Method of test

Initial Condition

System Simulator: 2 cells- Cell 1 is active, Cell 2 is inactive

UE: PS-DCCH_FACH(state 6-8) as specified in clause 7.4 of TS 34.108 in cell 1

Test Procedure

The UE is in the CELL_FACH state in cell 1. The SS begins to broadcast the BCCH in cell 2 but with a transmission level lower than that for cell 1. It thenSS transmits a RADIO BEARER SETUP message to the UE. After transmitting the RADIO BEARER SETUP message, the SS shall not configure L1 in accordance to the settings in the message. At the same time, SS begins to broadcast the BCCH in a cell 2. The UE recognize that it cannot configure the new radio bearer and wants to revert to the old configuration, but the UE cannot revert to the old configuration. The UE shall select Cell 2 by performing cell re-selection and enter CELL FACH state. The UE transmit CELL UPDATE message on uplink CCCH with IE "Cell update cause" set to "radio link failure". The SS shall transmit CELL UPDATE CONFIRM message on downlink CCCH after receiving CELL UPDATE message. The UE transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC and subsequently transmit RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC, setting the value "physical channel failure" to IE "failure cause" from cell 1. Then the SS deletes the old downlink channel configuration after sending this message. This causes the UE to fail to configure L1 and could not revert to the old configuration. The UE shall find the presence of cell 2 and transmits an RRC CONNECTION RE ESTABLISHMENT REQUEST message. The message shall include the IE "U RNTI" and sent on the uplink CCCH. The SS responds with an RRC CONNECTION RE-ESTABLISHMENT message, which includes a new TFCS according to the new transport channel. After the UE receives an RRC CONNECTION RE ESTABLISHMENT message, it reconfigures the new radio connection and transmits an RRC CONNECTION RE ESTABLISHMENT COMPLETE message on the uplink DCCH using AM RLC.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
4	_ -	BCCH	The SS transmits the BCCH in cell 2, but with a power level lower that that in cell 1.
<u>21</u>	+	RADIO BEARER SETUP	The SS does not configure the new radio bearer and instead delete the old configuration.
<u>2</u>		BCCH	The SS does not configure the new radio bearer and starts to transmit the BCCH in cell 2
3			The UE cannot configure a new radio bearer and cannot revert to old configuration.
4			The UE select the cell 2.
4 <u>5</u>	→	CELL UPDATERRC CONNECTION RE-ESTABLISHMENT REQUEST	The value "radio link failure" shall be set in IE "Cell update cause". The UE shall find the presence of cell2lt shall then transmit this message including its assigned U-RNTI in the IE "U-RNTI".
<u>56</u>	←	CELL UPDATE CONFIRMRRC CONNECTION RE-ESTABLISHMENT	This message include IE "new U-RNTI" and IE "new C-RNTI"". Including the new TFCS according to the new transport channel.
6 <u>7</u>	\rightarrow	UTRAN MOBILITY INFORMATION CONFIRMREC CONNECTION RE-ESTABLISHMENT COMPLETE	
<u>8</u>		RADIO BEARER SETUP FAILURE	The IE "failure cause" shall be set to "physical channel failure"

Specific Message Contents

RADIO BEARER SETUP (Step 1)

The contents of RADIO BEARER SETUP message in this test case is identical to the message sub-type "Packet to CELL_DCH from CELL_FACH in PS" found in Annex A.

CELL UPDATE (Step 5)

The contents of CELL UPDATE message is identical as "Contents of CELL UPDATE message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	
- SRNC Identity	Assigned previously in cell 1
<u>- S-RNTI</u>	Assigned previously in cell 1
Cell Update Cause	<u>"radio link failure"</u>

CELL UPDATE CONFIRM (Step 7)

The contents of CELL UPDATE CONFIRM message is identical as "CELL UPDATE CONFIRM message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	Same as CELL UPDATE message in step 7
New U-RNTI	
- SRNC Identity	<u>'0000 0000 0000 0001'</u>
- S-RNTI	<u>Different from previous S-RNTI</u>
New C-RNTI	Different from previous C-RNTI

RADIO BEARER SETUP FAILURE (Step 8)

Information Element	<u>Value/remark</u>
Message Type	"RADIO BEARER RECONFIGURATION FAILURE"
Failure cause	"physical channel failure"
Other information element	Not checked

RRC CONNECTION RE-ESTABLISHMENT REQUEST

Information Element	Value/remark
U-RNTI	Set to assigned UE U-RNTI in RRC CONNECTION
	SETUP message.
Protocol error indication	FALSE

RRC CONNECTION RE-ESTABLISHMENT COMPLETE

Information Element	Value/remark
Radio bearer uplink ciphering activation time info	Not Present
RB with PDCP information list	Not Present
- RB with PDCP information list	

8.2.1.13.5 Test requirement

After step 3 the UE shall find the presence of cell 2, which is suitable for camping.

-After step 8 the UE shall transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

After step 9 the UE shall transmit RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC, setting the IE "failure cause" to "physical channel failure".

It shall then enter CELL_FACH state and transmit a RRC CONNECTION RE ESTABALISHMENT REQUEST message to prevent the RRC connection from being lost.

After step 6 the UE shall re establish an RRC connection, using the new TFCS settings specified in RRC CONNECTION RE ESTABLISHMENT message

8.2.1.14 Radio Bearer Establishment for transition from CELL_FACH to CELL_DCH: Failure (Incompatible simultaneous reconfiguration)

8.2.1.14.1 Definition

8.2.1.14.2 Conformance requirement

If the UE receives a RADIO BEARER SETUP message whilst reconfiguring due to a radio bearer message other than RADIO BEARER SETUP, it shall keep its configuration as if the RADIO BEARER SETUP message had not been received. If the UE is reconfiguring itself after receiving a radio bearer message other than RADIO BEARER SETUP message, and when the UE subsequently receive a RADIO BEARER SETUP message, the UE shall transmit a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC with value "Incompatible simultaneous reconfiguration" set in IE "failure cause".

Reference

3GPP TS 25.331 clause 8.2.1

8.2.1.14.3 Test purpose

To confirm that if the UE receives a RADIO BEARER SETUP message whilst reconfiguring due to a radio bearer message other than RADIO BEARER SETUP, it shall keep its configuration as if the RADIO BEARER SETUP message had not been received and complete the reconfiguration according to the previously received message.

To confirm that the UE keeps its configuration and transmits a RADIO BEARER SETUP FAILURE message when the UE receives a RADIO BEARER SETUP message, before the UE configures the radio bearer according to an earlier message.

8.2.1.14.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCT+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a RADIO BEARER RECONFIGURATION message to the UE. The SS transmits a RADIO BEARER SETUP message before the "activation time" indicated in the RADIO BEARER RECONFIGURATION message expires. When the UE receives the RADIO BEARER SETUP message, the UE shall keep the configuration as if it had not received the RADIO BEARER SETUP message and shall transmit a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC with IE "failure cause" set to "incompatible simultaneous reconfiguration". After the SS receives the RADIO BEARER SETUP FAILURE message, the UE reconfigures the new physical channel parameters and transmits a RADIO BEARER RECONFIGURATION COMPLETE message on DCCH using AM RLC.

The UE is in the CELL_FACH state. The SS transmits a RADIO BEARER RECONFIGURATION message. The activation time of this event is specified to be 255 frames from the SS's current CFN. However, SS sends a RADIO BEARER SETUP message before 255 frames has passed. The UE shall then abandon its current reconfiguration operation, and keep the old configuration. Finally, the UE transmits a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC. This message shall contain the value "Incompatible simultaneous reconfiguration" in IE "failure cause".

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	+	RADIO BEARER RECONFIGURATION	Including IE "Uplink DPCH info"
2	\	RADIO BEARER SETUP	The SS send this message before the expiry of activation time specified in the message of step 1.
3	→	RADIO BEAER SETUP FAILURE	The UE does not change the configuration because of receiving the RADIO BEARER SETUP message, and transmit this message on its uplink DCCH using the same RLC-AM mode radio bearer before step 1.
4	≥	RADIO BEARER RECONFIGURATION COMPLETE	This message is on DCCH using AM RLC.

Specific Message Contents

RADIO BEARER RECONFIGURATION (Step 1)

The contents of RADIO BEARER RECONFIGURATION message in this test case are identical as "Packet to CELL_DCH from CELL_FACH in PS" as found in Annex A with the following exceptions:

Information Element	Value/remark	
Activation Time	Current CFN-[current CFN mod 8 + 8]	

RADIO BEARER SETUP (for Step 2)

For this message, use the message sub-type entitled "Packet to CELL_DCH from CELL_FACH in PS" in the default message content. Information element(s) to be changed are listed below:

Information Element	Value/remark
Activation Time	Not present

RADIO BEARER SETUP FAILURE

Information Element	Value/remark
Message Type	
Failure cause	Incompatible simultaneous reconfiguration
Other information element	Not checked

8.2.1.14.5 Test requirement

After step 1 The SS transmits a RADIO BEARER SETUP message before the expiry of the activation time specified in the message of step 1.

After step 2 the UE shall keep its configuration as if the UE had not received the RADIO BEARER SETUP message and shall transmit a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC with IE "failure cause" set to "Incompatible simultaneous reconfiguration".

After step 4 the UE communicates with the SS on the DCCH and DTCH using the new physical channel parameters reconfigured as a result of the RADIO BEARER RECONFIGURATION message.

After step 2 the UE shall keep its configuration and transmit a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC which is set to "Incompatible simultaneous reconfiguration" in IE "failure cause".

8.2.1.15 Radio Bearer Establishment for transition from CELL_FACH to CELL_DCH: Failure (Invalid message reception and Invalid configuration)

8.2.1.15.1 Definition

8.2.1.15.2 Conformance requirement

The UE shall keep its old configuration when the UE receives a RADIO BEARER SETUP message, which includes the spare undefined value in the mandatory IE "UTRAN DRX cycle length coefficient DRX indicator" having criticality defined as "Reject". It shall transmit a RADIO BEARER SETUP FAILURE message which set value "protocol error" in IE "failure cause" and also value "Information element value not comprehended" in IE "Protocol error cause". The UE shall keep existing configuration before reception of a RADIO BEARER SETUP message when the RADIO BEARER SETUP message include some IEs set to invalid value, and then the UE shall transmit RADIO BEARER SETUP FAILURE including IE "failure cause" set to "invalid configuration".

Reference

3GPP TS 25.331 clause 8.2.1

8.2.1.15.3 Test purpose

To confirm that the UE transmits a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC if it receives a RADIO BEARER SETUP message, with an spare-undefined value in the mandatory IE "UTRAN DRX cycle length coefficientDRX indicator" and having criticality defined as "Reject".

To confirm that the UE transmits a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC if it receives a RADIO BEARER SETUP message including some IEs set to invalid value.

8.2.1.15.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH_FACH (state 6-8) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in CELL_FACH state. The SS transmits a RADIO BEARER SETUP message to the UE which includes an spare undefined value in the mandatory IE "UTRAN DRX cycle length coefficient DRX indicator" with criticality defined as "Reject". The UE keeps the old configuration and transmits a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC. This message shall specify "protocol error" in IE "failure cause" and also set the value "Information element value not comprehended" in IE "Protocol error cause". The UE keeps initial configuration when SS transmits RADIO BEARER SETUP message including some IEs set to invalid value. The UE transmit RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC, setting the value "invalid configuration" to IE "failure cause".

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	+	RADIO BEARER SETUP	
2	→	RADIO BEARER SETUP FAILURE	The UE does not change the configuration.
<u>3</u>	<u>←</u>	RADIO BEARER SETUP	This message includes IE set to invalid value.
<u>4</u>			The UE does not change the configuration.
<u>5</u>	<u>→</u>	RADIO BEARER SETUP FAILURE	The IE "failure cause" shall be set to "invalid configuration

Specific Message Contents

RADIO BEARER SETUP (Step 1)

The contents of RADIO BEARER SETUP message in this test case is identical as "Packet to CELL_DCH from CELL_FACH in PS" as found in Annex with the following exceptions:

Information Element	Value/remark
UTRAN DRX cycle length coefficientDRX indicator	spare-Undefined value

RADIO BEARER SETUP FAILURE (Step 2)

Information Element	Value/remark	
Message Type		
Failure cause		
- Failure cause	Protocol error	
- Protocol error information		
- Protocol error cause	Information element value not comprehended	
Other information element	Not checked	

RADIO BEARER SETUP (Step 3)

The contents of RADIO BEARER SETUP message in this test case is identical as "Packet to CELL DCH from CELL FACH in PS" as found in Annex with the following exceptions:

Added or Reconfigured UL TrCH information	
- Transport channel identity	<u>1</u>
TFS	
 - Dynamic Transport format information 	(This IE is repeated for TFI number)
- RLC size	Reference to TS34.108 clause 6.10 Parameter Set
- CHOICE Logical Channel list	
- Explicit List	
- RB identity	<u>2</u>

RADIO BEARER SETUP FAILURE (Step 5)

Information Element	<u>Value/remark</u>
Message Type	
Failure cause	Invalid configuration
Other information element	Not checked

8.2.1.15.5 Test requirement

After step 1 the UE shall keep its old configuration and transmits a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC. The message shall indicate the reason of failure as "protocol error" in IE "failure cause" and set the value "Information element value not comprehended" in IE "Protocol error cause".

After step 3 the UE shall keep its old configuration

After step 4 the UE shall transmit RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC, setting the value of IE "failure cause" to "invalid configuration".

8.2.1.16 Radio Bearer Establishment for transition from CELL_FACH to CELL_FACH: Success

8.2.1.16.1 Definition

8.2.1.16.2 Conformance requirement

The UE shall correctly set up a radio bearer according to a RADIO BEARER SETUP message and communicate with the UTRAN for its implementation.

Reference

3GPP TS 25.331 clause 8.2.1

8.2.1.16.3 Test purpose

To confirm that the UE establishes a new radio bearer according to a RADIO BEARER SETUP message received from the SS.

8.2.1.16.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH_FACH (state 6-8) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state, after the test operator is being prompted to make an outgoing packet-switched call. The SS transmits a RADIO BEARER SETUP message to the UE after it sets up L1 including the start of tx/rx. After the UE receives this message, it configures them and establishes a radio bearer. Finally the UE transmits a RADIO BEARER SETUP COMPLETE message using AM RLC. Then the UE and the SS enters the communicating state.

Expected sequence

Step	Direction		Message	Comment
	UE	SS	-	
1			RADIO BEARER SETUP	
2				The UE select PRACH and S-CCPCH using SIB5 or SIB6.
2 3	→		RADIO BEARER SETUP COMPLETE	
3 4				To confirm the proper establishment of the new radio bearer by checking the packet data exchanged between the SS and a TE attached to the UE.

Specific Message Contents

RADIO BEARER SETUP

For this message, use the message sub-type entitled "Packet to CELL_FACH from CELL_FACH in PS" in Annex A.

8.2.1.16.5 Test requirement

After step 2–3 the UE shall communicate with the SS using the new radio bearer, this can be confirmed by the exchange of packet data between a terminal equipment (TE) attached to the UE and the SS.

8.2.1.17 Radio Bearer Establishment for transition from CELL_DCH to CELL_DCH: success (Subsequently received)

8.2.1.17.1 Definition

8.2.1.17.2 Conformance requirement

If the UE receives a RADIO BEARER SETUP message before the UE configures the radio bearer according to the previous RADIO BEARER SETUP message, the UE shall ignore the new RADIO BEARER SETUP message and configure according to the previous RADIO BEARER SETUP message. Finally, the UE shall transmit RADIO BEARER SETUP COMPLETE message on the DCCH using AM RLC.

Reference

3GPP TS 25.331 clause 8.2.1

8.2.1.17.3 Test purpose

To confirm that if the UE receives a new RADIO BEARER SETUP message before the UE configures the radio bearer according to a previous RADIO BEARER SETUP message, it ignore the new RADIO BEARER SETUP message and configures according to the previous RADIO BEARER SETUP message received.

8.2.1.17.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CS-DCCH_DCH (state 6-5) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. SS transmits a RADIO BEARER SETUP message to the UE before the UE configures the radio bearer according to the RADIO BEARER SETUP message prior to this new message. The UE ignores the new RADIO BEARER SETUP message and configures according to the former RADIO BEARER SETUP message. On completion of radio bearer configuration, the UE shall transmit RADIO BEARER SETUP COMPLETE message on the DCCH using AM RLC.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	←	RADIO BEARER SETUP	Including IE "Uplink DPCH info"
2	+	RADIO BEARER SETUP	SS send this message before the expiry of activation time specified in RADIO BEARER SETUP message of step 1.
3	→	RADIO BEARER SETUP COMPLETE	The UE ignores the RADIO BEARER SETUP message in step 2 and confirms configuration according to the RADIO BEARER SETUP message in step 1.

Specific Message Contents

For RADIO BEARER SETUP in step 1, use the message sub-type indicated as "The other of speech in CS" found in Annex A, with the exception of the following Information Elements:

RADIO BEARER SETUP (Step 1)

Information Element	Value/remark
Activation Time	[256+Current CFN-[current CFN mod 8 + 8]]MOD 256

RADIO BEARER SETUP (Step 2)

For RADIO BEARER SETUP in step 2, use the message sub-type indicated as "The other of speech in CS" found in Annex A, with the exception of the following

Information Element	Value/remark
Activation Time	Not Present

8.2.1.17.5 Test requirement

After step 3 the UE shall communicate with the SS on the radio bearer specified in the RADIO BEARER SETUP message in step 1.

8.2.1.18 Radio Bearer Establishment for transition from CELL_FACH to CELL_DCH: Success (Subsequently received)

8.2.1.18.1 Definition

8.2.1.18.2 Conformance requirement

If the UE receives a RADIO BEARER SETUP message before the UE configures the radio bearer according to the previous RADIO BEARER SETUP message, the UE shall ignore the new RADIO BEARER SETUP message and configure according to the previous RADIO BEARER SETUP message. Finally, the UE shall transmit RADIO BEARER SETUP COMPLETE message on the DCCH using AM RLC.

Reference

3GPP TS 25.331 clause 8.2.1

8.2.1.18.3 Test purpose

To confirm that if the UE receives a new RADIO BEARER SETUP message before the UE configures the radio bearer according to a previous RADIO BEARER SETUP message, it ignore the new RADIO BEARER SETUP message and configures according to the previous RADIO BEARER SETUP message received.

8.2.1.18.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH FACH (state 6-8) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a RADIO BEARER SETUP message, requesting the UE to setup radio bearers using DPCH physical channels. The activation time of this event is specified to be 255 frames from the SS's current CFN. However, SS sends another RADIO BEARER SETUP message before 255 frames has lapsed. The UE ignores the new RADIO BEARER SETUP message and configures according to the former RADIO BEARER SETUP message. On completion of radio bearer configuration, the UE shall transmit RADIO BEARER SETUP COMPLETE message on the DCCH using AM RLC.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1	-		RADIO BEARER SETUP	Including IE "Uplink DPCH info"
2	←		RADIO BEARER SETUP	SS send this message before the expiry of activation time specified in RADIO BEARER SETUP message of step 1.
3		→	RADIO BEAER SETUP COMPLETE	The UE ignores the RADIO BEARER SETUP message in step 2 and confirms configuration according to the RADIO BEARER SETUP message in step 1.

Specific Message Contents

RADIO BEARER SETUP (Step 1)

For this message, use the message sub-type entitled "Packet to CELL_DCH from CELL_FACH in PS" in Annex A

RADIO BEARER SETUP (for Step 2)

For this message, use the message sub-type entitled "Packet to CELL_DCH from CELL_FACH in PS" in Annex A. Information element(s) to be changed are listed below:

Information Element	Value/remark
Activation Time	[256+Current CFN-[current CFN mod 8 + 8]]MOD 256

8.2.1.18.5 Test requirement

After step 2 the UE shall keep its configuration and transmit a RADIO BEARER SETUP COMPLETE message on the DCCH using AM RLC.

After step 3 the UE shall communicate with the SS on the radio bearer specified in the RADIO BEARER SETUP message in step 1.

8.2.1.19 Radio Bearer Establishment from CELL DCH to CELL PCH: Success

8.2.1.19.1 Definition

8.2.1.19.2 Conformance requirement

The UE shall configure radio bearers and transition from CELL_DCH state to CELL_PCH state according to the receiveding RADIO BEARER SETUP message.

Reference

3GPP TS 25.331 clause 8.2.2

8.2.1.19.3 Test purpose

<u>To conform that the UE transmits a RADIO BEARER SETUP COMPLETE message and enters CELL PCH state after</u> it received a RADIO BEARER SETUP message from SS and configured new radio bearers.

8.2.1.19.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_DCH(state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL DCH state. The SS transmits a RADIO BEARER SETUP message. The UE transmits RADIO BEARER SETUP COMPLETE message to the UE using AM RLC and enters into CELL_PCH state. The SS transmits a PAGING TYPE 1 message, to enter-causing the UE to enter in CELL FACH state and the UE shall transmit CELL UPDATE message on uplink CCCH with IE "Cell update cause" set to "paging response".

Expected sequence

Step	Direction	<u>Message</u>	Comment
	UE SS		
<u>1</u>	<u></u>	RADIO BEARER SETUP-message	
<u>2</u>	<u>→</u>	RADIO BEARER SETUP COMPLETE	The UE sends this message
			before state transition.
<u>3</u>			Configuration of Radio Bearer
			after state transition.
4	<u></u>	PAGING TYPE 1	The SS transmits this message
			included a matched identity.
<u>5</u>	<u>→</u>	CELL UPDATE	The UE is in CELL_FACH state.

Specific Message Contents

RADIO BEARER SETUP (Step 1)

<u>Use the same message sub-type titled "Packet to CELL_FACH from CELL_DCH in PS" in Annex A with following exceptions:</u>

Information Element	<u>Value/remark</u>
RRC State Indicator	CELL PCH

PAGING TYPE 1 (Step 4)

Use the same message sub-type titled "TM (SMS in PS)" in Annex A with following exceptions:

Information Element	<u>Value/remark</u>
Paging record list	
Paging record	
- CHOICE Paging originator	UTRAN originator
<u>- U-RNTI</u>	
- SRNC Identity	Previously assigned SRNC identity in Initial Condition
- S-RNTI	Previously assigned S-RNTI in Initial Condition

8.2.1.19.5 Test requirement

After step 1, the UE transmits RADIO BEARER RECONFIGURATIONSETUP COMPLETE message to the UE on uplink DCCH using AM RLC.

After step 34, the UE shall transit from CELL_DCH to CELL_PCH transmit CELL UPDATE message on the CCCH.

8.2.1.20 Radio Bearer Establishment from CELL DCH to URA PCH: Success

8.2.1.20.1 Definition

8.2.1.20.2 Conformance requirement

The UE shall configure radio bearers and transition from CELL DCH state to URA PCH state according to receiving RADIO BEARER SETUP message.

Reference

3GPP TS 25.331 clause 8.2.2

8.2.1.20.3 Test purpose

To conform that the UE transmit a RADIO **BEARER** SETUP COMPLETE message and enters URA_PCH state after it received a RADIO BEARER SETUP message from SS and configured the new radio bearers.

8.2.1.20.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH DCH(state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL DCH state. The SS transmits a RADIO BEARER SETUP message. The UE transmits RADIO BEARER SETUP COMPLETE message to the UE using AM RLC and enters into-URA PCH state. The SS transmits a PAGING TYPE 1 message, causing to enter the UE to enter in CELL_FACH state and the UE shall transmit CELL UPDATE message on uplink CCCH with IE "Cell update cause" set to "paging response".

Expected sequence

Step	Direction	<u>Message</u>	Comment
	UE SS		
1	<u></u>	RADIO BEARER SETUP-message	
2	<u> </u>	RADIO BEARER SETUP COMPLETE	The UE sends this message before state transition.
<u>3</u>			Configuration of Radio Bearer after state transition.
4	<u></u>	PAGING TYPE 1	The SS transmits this message included a matched identity.
<u>5</u>	<u>→</u>	CELL UPDATE	The UE is in CELL_FACH state.

Specific Message Contents

RADIO BEARER SETUP (Step 1)

<u>Use the same message sub-type titled "Packet to CELL_FACH from CELL_DCH in PS" in Annex A with following exceptions:</u>

Information Element	<u>Value/remark</u>
RRC State Indicator	<u>URA PCH</u>

PAGING TYPE 1 (Step 4)

Use the same message sub-type titled "TM (SMS in PS)" in Annex A with following exceptions:

Information Element	<u>Value/remark</u>
Paging record list	
Paging record	
- CHOICE Paging originator	UTRAN originator
U-RNTI	
- SRNC Identity	Previously assigned SRNC identity in Initial Condition
- S-RNTI	Previously assigned S-RNTI in Initial Condition

8.2.1.20.5 Test requirement

After step 1, the UE transmits RADIO BEARER RECONFIGURATION-SETUP COMPLETE message to the UE on uplink DCCH using AM RLC.

After step 3, the UE shall transmit CELL UPDATE message on the CCCHtransit from CELL DCH to URA PCH.

8.2.2 Radio Bearer Reconfiguration

8.2.2.1 Radio Bearer Reconfiguration (Hard handover) from CELL_DCH to CELL_DCH: Success

8.2.2.1.1 Definition

8.2.2.1.2 Conformance requirement

The UE shall correctly reconfigure a radio bearer and L1 according to the RADIO BEARER RECONFIGURATION message, which specifies a hard handover to another radio frequency. After executing the reconfiguration, the UE shall be able to communicate with the UTRAN on the newly configured radio bearer.

Reference

3GPP TS 25.331 clause 8.2.2

8.2.2.1.3 Test purpose

To confirm that the UE reconfigures a new radio bearer by following a RADIO BEARER RECONFIGURATION message, which indicates a hard handover to another radio frequency.

8.2.2.1.4 Method of test

Initial Condition

System Simulator: 2 cells – cell 1 and cell 6 are active. The CPICH_Ec/No and CPICH RSCP of cell 4 are improved to -15 dB and -70 dBm respectively. The $Q_{rxlevmin}$ and $Q_{rxqualmin}$ values in SYSTEM INFORMATION BLOCK TYPE 3 and 4 messages of cell 6 is changed to -90 dBm and -20 dB respectively

UE: PS-DCCH+DTCH_DCH (state 6-1110) as specified in clause 7.4 of TS 34.108 in cell 1

Test Procedure

The UE is in the CELL_DCH state in cell 1. The SS transmits a RADIO BEARER RECONFIGURATION message to the UE, which commands that hard handover to cell 6 be performed. The UE reconfigures the new physical channel parameters and transmits a RADIO BEARER RECONFIGURATION COMPLETE message using AM RLC.

Expected sequence

Step	Dire	ction	Message	Comment
	UE	SS		
1		←	RADIO BEARER RECONFIGURATION	Hard handover to cell 6,
2				UE shall suspend stop all uplink transmissions to cell 1 and shall commence the reconfiguration of the affected physical channel parameters to that of cell 6.
3		\rightarrow	RADIO BEARER RECONFIGURATION COMPLETE	

Specific Message Contents

RADIO BEARER RECONFIGURATION

The contents of RADIO BEARER RECONFIGURATION message in this test case is identical to the message sub-type titled "Packet to CELL_DCH from CELL_DCH in PS" in Annex A, with the following exceptions:

Information Element	Value/remark
Frequency info	
- UARFCN uplink(Nu)	Same uplink UARFCN as used for cell 6
- UARFCN downlink(Nd)	Same downlink UARFCN as used for cell 6
Downlink information for each radio links	Same downlink UARFCN as used for cell 6
- Primary CPICH info	
- Primary Scrambling Code	350
Downlink information common for all radio links	
- Downlink DPCH info common for all RL	
- Timing Indicator	<u>Initialise</u>

8.2.2.1.5 Test requirement

After step 1 the UE shall reconfigure the radio links with the SS.

After step 3 the UE shall change its physical channel configuration and communicate with the SS on the DCCH and DTCH of cell 6.

8.2.2.2 Radio Bearer Reconfiguration from CELL_DCH to CELL_DCH: Failure (Unsupported configuration)

8.2.2.2.1 Definition

8.2.2.2.2 Conformance requirement

The UE shall keep its old configuration when the UE receives a RADIO BEARER RECONFIGURATION message which includes unsupported configuration parameters and transmit a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC which is set to "configuration unsupported" in IE "failure cause

Reference

3GPP TS 25.331 clause 8.2.2

8.2.2.2.3 Test purpose

To confirm that the UE transmits a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC if the received RADIO BEARER RECONFIGURATION message includes unsupported configuration parameters.

8.2.2.2.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_DCH (state 6-710) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a RADIO BEARER RECONFIGURATION message to the UE, which includes unsupported configuration parameters for the UE. The UE shall transmit a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC which is set to "configuration unsupported" in IE "failure cause".

Expected sequence

Step	Dire	ction	Message	Comment
	UE	SS		
1			RADIO BEARER	Including unsupported
			RECONFIGURATION	configuration by the UE
2		\rightarrow	RADIO BEARER	The UE does not change the
			RECONFIGURATION FAILURE	radio bearer.

Specific Message Contents

RADIO BEARER RECONFIGURATION

The contents of RADIO BEARER RECONFIGURATION message in this test case is identical as "Packet to CELL_DCH from CELL_DCH in PS" as found in Annex with the following exceptions:

Information Element	Value/remark
Frequency info	
- UARFCN uplink(Nu)	63984.
- UARFCN downlink(Nd)	Not Present

RADIO BEARER RECONFIGURATION FAILURE

Information Element	Value/remark
Message Type	"RADIO BEARER RECONFIGURATION FAILURE"
Failure cause	Configuration unsupported
Other information element	Not checked

8.2.2.2.5 Test requirement

After step 2 the UE shall keep its old configuration and transmit a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC with the value "configuration unsupported" set in IE "failure cause".

8.2.2.3 Radio Bearer Reconfiguration from CELL_DCH to CELL_DCH: Failure (Physical channel failure and reversion to old configuration)

8.2.2.3.1 Definition

8.2.2.3.2 Conformance requirement

The UE shall revert to the old configuration when the UE fails to reconfigure the new physical channel by received RADIO BEARER RECONFIGURATION message and transmit a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC which is set to "physical channel failure" in IE "failure cause".

Reference

3GPP TS 25.331 clause 8.2.2

8.2.2.3.3 Test purpose

To confirm that the UE reverts to the old configuration and transmits a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC if the UE fails to reconfigure the new radio bearer according to the RADIO BEARER RECONFIGURATION message received previously.

8.2.2.3.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_DCH (state 6-710) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a RADIO BEARER RECONFIGURATION message to the UE which includes the new radio bearer parameters but it does not reconfigure L1 according to the settings found in the message. The UE shall revert to the old configuration. Then the UE shall transmit a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting value "physical channel failure" in IE "failure cause".

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	←	RADIO BEARER RECONFIGURATION	
2			SS does not reconfigure L1 parameters to reflect the radio bearer reconfigurations specified in the message.
3	→	RADIO BEARER RECONFIGURATION FAILURE	The UE shall detect a failure to reconfigure the new radio bearer, and send this message using the old radio bearer configuration.

Specific Message Contents

RADIO BEARER RECONFIGURATION

The contents of RADIO BEARER RECONFIGURATION message in this test case is identical as "Packet to CELL_DCH from CELL_DCH in PS" as found in Annex A.

RADIO BEARER RECONFIGURATION FAILURE

Information Element	Value/remark
Message Type	"RADIO BEARER RECONFIGURATION FAILURE"
Failure cause	Physical channel failure
Other information element	Not checked

8.2.2.3.5 Test requirement

After step 2 the UE shall revert to the old configuration and transmit a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC setting value "physical channel failure" in IE "failure cause".

8.2.2.4 Radio Bearer Reconfiguration from CELL_DCH to CELL_DCH: Failure (Physical channel failure and reversion failure)

8.2.2.4.1 Definition

8.2.2.4.2 Conformance requirement

The UE shall perform an RRC connection re establishment procedure cell update when the UE fails to revert to the old configuration after the detection of physical channel failure in the radio bearer reconfiguration procedure. After the UE completes cell update procedure, the UE transmit RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC which is set IE "failure cause" to "physical channel failure".

Reference

3GPP TS 25.331 clause 8.2.2

8.2.2.4.3 Test purpose

To confirm that the UE <u>transmits RADIO BEARER SETUP FAILURE message after it completes initiates</u> an RRC connection re establishmentcell update procedure when the UE cannot reconfigure the new radio bearer, and a subsequent failure to revert to the old configuration.

8.2.2.4.4 Method of test

Initial Condition

System Simulator: 2 cells - Cell 1 is active, Cell 2 is inactive

UE: PS-DCCH+DTCH_DCH (state 6-710) as specified in clause 7.4 of TS 34.108 in cell 1

Test Procedure

The UE is in the CELL_DCH state in a cell 1. The SS transmits a RADIO BEARER RECONFIGURATION message to the UE, which includes the new radio bearer parameters. <u>After transmitting the RADIO BEARER</u> <u>RECONFIGURATION message, t</u>The SS <u>does-shall</u> not reconfigure L1 in accordance to the settings in the message and delete all radio bearer related contexts in cell 1. At the same time, SS begins to broadcast the BCCH in a cell 2. The UE discovers that it cannot reconfigure the new radio bearer and wants to revert to the old configuration, <u>but the UE cannot revert to the old configuration</u>. The UE shall select Cell 2 by performing cell re-selection and enter CELL

FACH state. The UE transmits CELL UPDATE message on uplink CCCH with IE "Cell update cause" set to "radio link failure". The SS shall transmit CELL UPDATE CONFIRM message on downlink CCCH after receiving CELL UPDATE message. The UE transmits UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC and subsequently transmits RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting the value "physical channel failure" to IE "failure cause". The UE shall find cell 2 and transmit to it an RRC CONNECTION RE ESTABLISHMENT REQUEST message which includes the IE "URNTI" on the uplink CCCH. The SS transmits an RRC CONNECTION RE ESTABLISHMENT message which specifies a new TFCS according to the new transport channel allocated. After the UE receives an RRC CONNECTION RE ESTABLISHMENT message, it reconfigures the new radio connection and transmits an RRC CONNECTION RE ESTABLISHMENT COMPLETE message on the uplink DCCH using AM RLC.

Expected sequence

Step	Dire	ction	Message	Comment
	UE	SS	_	
1		(RADIO BEARER RECONFIGURATION	
2				The SS does not reconfigure the
				new radio bearer and delete the
				old configuration in cell 1.
<u>3</u>		<u>É</u>	<u>BCCH</u>	The SS transmit the BCCH in
				the cell 2.
<u>34</u>				The UE fails to reconfigure a
				new radio bearer.
<u>5</u>				The SS shall not revert old
-				configuration.
<u>6</u>				The UE select the cell 2 and
47			CELL LIDDATERCOLI	enter CELL FACH state. The value "radio link failure"
4 <u>7</u>		~	CELL UPDATEBCCH	shall be set in IE "Cell update
				cause". The SS transmits a
				BCCH in a cell 2 and delete the
				old radio bearer.
5 8		\rightarrow	CELL UPDATE CONFIRM RRC	This message include IE "new
			CONNECTION	U-RNTI" and IE "new C-
			RE-ESTABLISHMENT REQUEST	RNTI""The UE finds cell 2 and
				transmits this message which
				includes the IE "U-RNTI" it was
				previously assigned.
6 9] =	<u>←</u>	UTRAN MOBILITY INFORMATION	This message includes a new
			CONFIRMRRC CONNECTION	TFCS according to the new
			RE-ESTABLISHMENT	transport channel indicated in
				RRC CONNECTION RE-
10			DADIO DEADED DECONEICUDATION	ESTABLISHMENT message.
<u>10</u>	=	<u>></u>	RADIO BEARER RECONFIGURATION	The IE "failure cause" shall be
			<u>FAILURE</u>	set to "physical channel failure"
7	_	→	RRC CONNECTION	
<u>'</u>		-	RE-ESTABLISHMENT COMPLETE	

Specific Message Contents

RADIO BEARER RECONFIGURATION message (Step 1)

The contents of RADIO BEARER RECONFIGURATION message in this test case is identical as "Packet to CELL_DCH from CELL_DCH in PS" as found in Annex.

CELL UPDATE (Step 7)

The contents of CELL UPDATE message is identical as "Contents of CELL UPDATE message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	
- SRNC Identity	Assigned previously in cell 1
<u>- S-RNTI</u>	Assigned previously in cell 1
Cell Update Cause	"radio link failure"

CELL UPDATE CONFIRM (Step 8)

The contents of CELL UPDATE CONFIRM message is identical as "CELL UPDATE CONFIRM message" as found in Annex with the following exceptions:

Information Element	Value/remark
<u>U-RNTI</u>	Same as CELL UPDATE message in step 7
New U-RNTI	
- SRNC Identity	<u>'0000 0000 0000 0001'</u>
- S-RNTI	Different from previous S-RNTI
New C-RNTI	Different from previous C-RNTI

RADIO BEARER RECONFIGURATION FAILURE (Step 9)

Information Element	<u>Value/remark</u>
Message Type	"RADIO BEARER RECONFIGURATION FAILURE"
Failure cause	"physical channel failure"
Other information element	Not checked

8.2.2.4.5 Test requirement

After step 4-5 the UE shall detect the presence of cell 2, enters CELL_FACH state, and transmit RRC CONNECTION RE ESTABLISHMENT to attempt to keep the current RRC connection.

After step 7–8 the UE transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

After step 9 the UE shall transmit RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting the IE "failure cause" to "physical channel failure".shall successfully re establish an RRC connection.

8.2.2.5 Radio Bearer Reconfiguration from CELL_DCH to CELL_DCH: Failure (Incompatible simultaneous reconfiguration)

8.2.2.5.1 Definition

8.2.2.5.2 Conformance requirement

If the UE receives a RADIO BEARER RECONFIGURATION message whilst reconfiguring due to a radio bearer message other than RADIO BEARER RECONFIGURATION, it shall keep its configuration as if the RADIO RECONFIGURATION SETUP message had not been received. If the UE is configuring itself after receiving a radio bearer message other than RADIO BEARER RECONFIGURATION message, and when the UE subsequently receive a RADIO BEARER RECONFIGURATION message, the UE shall transmit a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC which is set to "Incompatible simultaneous reconfiguration" in IE 'failure cause'.

Reference

3GPP TS 25.331 clause 8.2.2

8.2.2.5.3 Test purpose

To confirm that if the UE receives a RADIO BEARER RECONFIGURATION message whilst reconfiguring due to a radio bearer message other than RADIO BEARER RECONFIGURATION, it shall keep its configuration as if the RADIO BEARER RECONFIGURATION message had not been received and complete the reconfiguration according to the previously received message. To confirm that the UE keeps its configuration and transmits an RADIO BEARER

RECONFIGURATION FAILURE message when the UE receives a RADIO BEARER RECONFIGURATION message before the UE can complete the configuration of the radio bearer indicated in an earlier message.

8.2.2.5.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_DCH (state 6-710) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL DCH state. The SS transmits a RADIO BEARER SETUP message to the UE. The SS transmits a RADIO BEARER RECONFIGURATION message before the "activation time" indicated in the RADIO BEARER SETUP message expires. When the UE receives the RADIO BEARER RECONFIGURATION message, the UE shall keep the configuration as if it had not received the RADIO BEARER RECONFIGURATION message and shall transmit a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC with IE "failure cause" set to "incompatible simultaneous reconfiguration". After the SS receives the RADIO BEARER SETUP FAILURE message, the UE reconfigures the new physical channel parameters and transmits a RADIO BEARER SETUP COMPLETE message on DCCH using AM RLC. The UE is in the CELL_DCH state. When the SS transmits a RADIO BEARER RECONFIGURATION message to the UE before the UE configures the requested radio bearer(s), the UE keeps the old configuration and transmits an RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC. This message shall set value "Incompatible simultaneous reconfiguration" in IE "failure eause".

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	+	RADIO BEARER SETUP	Including IE "Uplink DPCH info"
2	+	RADIO BEARER RECONFIGURATION	Sent before the "activation time" in step 1 has elapsed
3	\rightarrow	RADIO BEARER RECONFIGURATION FAILURE	The UE does not change the configuration because of receiving the RADIO BEARER RECONFIGURATION message
<u>4</u>	<u></u>	RADIO BEARER SETUP COMPLETE	This message is on DCCH using AM RLC.

Specific Message Contents

RADIO BEARER SETUP (Step 1)

For RADIO BEARER SETUP in step 1, use the message sub-type indicated as "Packet to CELL_DCH from CELL_DCH in PS" found in Annex A.

RADIO BEARER RECONFIGURATION (Step 2)

The contents of RADIO BEARER RECONFIGURATION message in this test case is identical to those in the default contents of layer 3 messages for RRC tests with the following exceptions as "Packet to CELL_DCH from CELL_DCH in PS" found in Annex A with following exceptions:

Information Element	Value/remark
Activation Time	Not Present.

RADIO BEARER RECONFIGURATION FAILURE

Information Element	Value/remark
Message Type	"RADIO BEARER RECONFIGURATION FAILURE"
Failure cause	Incompatible simultaneous reconfiguration
Other information element	Not checked

8.2.2.5.5 Test requirement

After step 1 The SS transmits a RADIO BEARER RECONFIGURATION message before the expiry of the activation time specified in the message of step 1.

After step 2 the UE shall keep its configuration as if the UE had not received the RADIO BEARER SETUP message and shall transmit a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC with IE "failure cause" set to "Incompatible simultaneous reconfiguration".

After step 4 the UE communicates with the SS on the DCCH and DTCH using the new physical channel parameters configured as a result of the RADIO BEARER SETUP message After step 2 the UE shall keep its configuration and transmit an RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC which contains the value "Incompatible simultaneous reconfiguration" set in IE "failure cause".

8.2.2.6 Radio Bearer Reconfiguration from CELL_DCH to CELL_DCH: Failure (Invalid message reception_and Invalid configuration)

8.2.2.6.1 Definition

8.2.2.6.2 Conformance requirement

The UE shall keep its old configuration when the UE receives a RADIO BEARER RECONFIGURATION message, which includes the spare-undefined value in the mandatory IE "UTRAN DRX cycle length coefficient DRX indicator" having criticality defined as "Reject". The UE shall transmit a RADIO BEARER RECONFIGURATION FAILURE message which is set to "protocol error" in IE "failure cause" and is set to "Information element value not comprehended" in IE "Protocol error cause". The UE shall keep existing configuration before reception of a RADIO BEARER RECONFIGURATION message when the RADIO BEARER RECONFIGURATION message include some IEs set to invalid value, and then the UE shall transmit RADIO BEARER RECONFIGURATION FAILURE including IE "failure cause" set to "invalid configuration".

Reference

3GPP TS 25.331 clause 8.2.2

8.2.2.6.3 Test purpose

To confirm that the UE transmits a RADIO BEARER RECONFIGRATION FAILURE message on the DCCH using AM RLC, if it receives a RADIO BEARER RECONFIGURATION message containing a <u>spare-undefined</u> value in the mandatory IE "<u>UTRAN DRX cycle length coefficientDRX indicator</u>" with criticality defined as "Reject".

To confirm that the UE transmits a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC if it receives a RADIO BEARER RECONFIGURATION message including some IEs set to invalid value.

8.2.2.6.4 Method of test

Initial Condition

System Simulator: 1 cell

Test Procedure

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	+	RADIO BEARER RECONFIGURATION	The message contains an illegal error in a mandatory IE.
2	\rightarrow	RADIO BEARER RECONFIGRATION FAILURE	The UE does not change the configuration.
<u>3</u>	<u>←</u>	RADIO BEARER RECONFIGURATION	This message includes IE set to invalid value
4			The UE does not change the configuration.
<u>5</u>	<u></u>	RADIO BEARER RECONFIGRATION FAILURE	The IE "failure cause" shall be set to "invalid configuration

Specific Message Contents

RADIO BEARER RECONFIGURATION (Step 1)

The contents of RADIO BEARER RECONFIGRATION message in this test case is identical as "Packet to CELL_DCH from CELL_DCH in PS" found in Annex A with following exceptions, with the following exceptions:

Information Element	Value/remark
UTRAN DRX cycle length coefficientDRX indicator	spare Undefined value

RADIO BEARER RECONFIGURATION FAILURE (Step 2)

Information Element	Value/remark
Message Type	"RADIO BEARER RECONFIGURATION FAILURE"
Failure cause	
- Failure cause	Protocol error
- Protocol error information	
- Protocol error cause	Information element value not comprehended
Other information element	Not checked

RADIO BEARER RECONFIGURATION (Step 3)

The contents of RADIO BEARER RECONFIGRATION message in this test case is identical as "Packet to CELL DCH from CELL DCH in PS" found in Annex A with following exceptions, with the following exceptions:

Added or Reconfigured UL TrCH information	
- Transport channel identity	1
<u>TFS</u>	
 Dynamic Transport format information 	(This IE is repeated for TFI number)
- RLC size	Reference to TS34.108 clause 6.10 Parameter Set
- CHOICE Logical Channel list	
- Explicit List	
- RB identity	2

RADIO BEARER RECONFIGURATION FAILURE (Step 5)

Information Element	<u>Value/remark</u>	
Message Type		
Failure cause	Invalid configuration	
Other information element	Not checked	

8.2.2.6.5 Test requirement

After step 1 the UE shall keep its old configuration and transmit a RADIO BEARER RECONFIGRATION FAILURE message on the DCCH using AM RLC stating the reason "protocol error" in IE "failure cause". The message shall contain the value "Information element value not comprehended" in IE "Protocol error cause".

After step 3 the UE shall keep its old configuration

After step 4 the UE shall transmit RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting the value "invalid configuration" to IE "failure cause".

8.2.2.7 Radio Bearer Reconfiguration from CELL_DCH to CELL_DCH: <u>Success</u> (Continue and stopFailure (Suspension of signalling bearer)

8.2.2.7.1 Definition

8.2.2.7.2 Conformance requirement

The UE shall continue or stop the uplink transmission when the UTRAN indicate stop or continue uplink transmission in radio bearer reconfiguration procedure.revert to the old configuration prior to the reception of RADIO BEARER RECONFIGURATION message, which includes IE "RB suspend/resume" specified as "Suspend", and transmit a RADIO BEARER RECONFIGURATION FAILURE message to UTRAN.

Reference

3GPP TS 25.331 clause 8.2.2

8.2.2.7.3 Test purpose

To confirm that the UE <u>reconfigures</u> new radio bearer and have the uplink transmission according to a RADIO BEARER RECONFIGURATIO message which indicates that uplink transmission is continued.

To confirm that the UE reconfigures new radio bearer and don't transmit data according to a RADIO BEARER RECONFIGURATION message which indicates that uplink transmission is stopped reverts to the old configuration and transmits a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC if the UE fails to reconfigure the new radio bearer according to a RADIO BEARER RECONFIGURATION message.

8.2.2.7.4 Method of test

Initial Condition

System Simulator: 1 cell

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a RADIO BEARER RECONFIGURATION message including IE" RB stop/continue" set to "continue". The UE reconfigures new radio bearer and transmit RADIO BEARER RECONFIGURATION COMPLETE message on the DCCH using AM RLC. The UE communicate with the SS after transmission the RADIO BEARER RECONFIGURATION COMPLETE message. Then, SS transmit a RADIO BEARER RECONFIGURATION message including IE" RB stop/continue" set to "stop". The UE reconfigures new radio bearer and transmit RADIO BEARER RECONFIGURATION COMPLETE message on the DCCH using AM RLC. The UE don't transmit any uplink data without Signalling message after transmission the RADIO BEARER RECONFIGURATION COMPLETE message. to the UE which specifies the suspension of a signalling radio bearer to be reconfigured. The UE shall revert to the old configuration. Then the UE transmits a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC with the IE "failure cause" set to "configuration unsupported".

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	←	RADIO BEARER RECONFIGURATION	This message include IE" RB stop/continue ".Specifies a signalling radio bearer to be suspended.
2	\rightarrow	RADIO BEARER RECONFIGURATION COMPLETE FAILURE	The UE shall revert to the old configuration
<u>3</u>			The SS Shall communicate with the UE.
<u>4</u>		RADIO BEARER RECONFIGURATION	This message include IE" RB stop/continue ".
<u>5</u>		RADIO BEARER RECONFIGURATION COMPLETE	
<u>6</u>			The SS shall not receive any data from the UE without Signalling message.

Specific Message Contents

RADIO BEARER RECONFIGURATION (Step 1)

The contents of RADIO BEARER RECONFIGURATION message in this test case is identical as "Packet to CELL_DCH from CELL_DCH in PS" found in Annex A with the following exceptions:

Information Element	Value/remark
RB information to reconfigure list	
RB information to reconfigure I	
-RB identity	<u>35</u>
-RB stop/continuesuspend/resume	"continueSuspend"

RADIO BEARER RECONFIGURATION FAILURE (Step 4)

The contents of CELL UPDATE message in this test case is identical as "Packet to CELL_DCH from CELL_DCH in PS" as found in Annex with the following exceptions:

Information Element	Value/remark
RB information to reconfigure list	
RB information to reconfigure	
RB identity	<u>5</u>
-RB stop/continue	<u>"stop"</u>
Message Type	"RADIO BEARER RECONFIGURATION FAILURE"
Failure cause	Configuration unsupported
Other information element	Not checked

8.2.2.7.5 Test requirement

After step 2 the UE shall communicate with the SS using new configuration.

After step 5 the UE shall communicate with the SS using new configuration, but shall not transmit any data to the SS without signalling message revert to the old configuration and transmit a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC specifying the reason "configuration unsupported" in IE "failure cause".

8.2.2.8 Radio Bearer Reconfiguration from CELL_DCH to CELL_FACH: Success

8.2.2.8.1 Definition

8.2.2.8.2 Conformance requirement

The UE shall correctly reconfigure a radio bearer according to a RADIO BEARER RECONFIGURATION message which is communicate with the UTRAN on the new radio bearer in case of a transition from CELL_DCH to CELL_FACH in the same cell.

Reference

3GPP TS 25.331 clause 8.2.2

8.2.2.8.3 Test purpose

To confirm that the UE establishes the reconfigured radio bearer(s) using common physical channel, after a RADIO BEARER RECONFIGURATION message has been received from the SS.

8.2.2.8.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a RADIO BEARER RECONFIGURATION message to the UE, which includes the new radio bearer parameters and sets up L1 including the start of tx/rx. The UE reconfigures the new radio bearer and transmits a RADIO BEARER RECONFIGURATION COMPLETE message using AM RLC.

Expected sequence

Step	Direction		Direction Message	Comment
	UE	SS		
1	+		RADIO BEARER RECONFIGURATION	
2				The UE select PRACH and S-CCPCH using SIB5 and SIB6 after entering CELL FACH state. Reconfiguration of radio bearer
3	→		RADIO BEARER RECONFIGURATION COMPLETE	

Specific Message Contents

RADIO BEARER RECONFIGURATION (Step 1)

Use the same message sub-type titled "Packet to CELL_DCH from CELL_FACH in PS" in Annex A.

8.2.2.8.5 Test requirement

After step 1 the UE shall reconfigure the radio links with the SS.

After step 3 the UE shall change its radio bearer configuration and communicate with the SS on the DCCH and DTCH, using the common physical channel-allocated in RADIO BEARER RECONFIGURTION message.

8.2.2.9 Radio Bearer Reconfiguration from CELL_DCH to CELL_FACH: Failure (<u>Cell re-selectionPhysical channel failure</u>)

8.2.2.9.1 Definition

8.2.2.9.2 Conformance requirement

The UE shall initiate the cell reselection update procedure when the UE cannot use the assigned physical channel according to a RADIO BEARER CHANNEL RECONFIGURATION message in current cell, during the transition from CELL_DCH to CELL_FACH. After the UE completes cell update procedure, the UE transmits RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC which set IE "failure cause" to "cell reselection".

Reference

3GPP TS 25.331 clause 8.2.2

8.2.2.9.3 Test purpose

To confirm that the UE transmits RADIO BEARER SETUP FAILURE message after it completes initiate a cell reselection after encountering a physical channel failure, when the UE cannot use the assigned physical channel according to a RADIO BEARER RECONFIGURATION message as it transits from CELL_DCH to CELL_FACH state during the execution of radio bearer reconfiguration procedure. The UE is first instructed to select the cell implied in the RADIO BEARER RECONFIGURATION message, but discover a failure as the indicated cell is not present.

8.2.2.9.4 Method of test

Initial Condition

System Simulator: 2 cells – Cell 1 is active and cell 2 is inactive

UE: PS-DCCH+DTCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state in cell 1. SS starts to transmit a RADIO BEARER RECONFIGURATION message to the UE, which includes the new radio bearer parameters. After transmitting the RADIO BEARER RECONFIGURATION message, the SS shall not reconfigure L1 in accordance to the settings in the message. At the same time, SS begins to broadcast the BCCH in a cell 2. The UE recognize that it cannot reconfigure the new radio bearer and wants to revert to the old configuration, but the UE cannot revert to the old configuration. The UE shall select cell 2 by performing cell re-selection and enter CELL FACH state. The UE transmits CELL UPDATE message on uplink CCCH with IE "Cell update cause" set to "cell reselection". The SS shall transmit CELL UPDATE CONFIRM message on downlink CCCH after receiving CELL UPDATE message. The UE transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC and transmit RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting the value "cell reselection" to IE "failure cause". the BCCH for cell 2, but with a power level lower that of cell 1. Following this, the SS send a RADIO BEARER RECONFIGURATION message to the UE, which includes relevant information about the target cell's P CPICH. SS selects an unused primary scrambling code for P CPICH, which neither matches the code used by cell 1 nor cell 2. This should lead to the detection of a physical channel failure in the UE. The UE shall trigger a cell reselection and initiate a cell update procedure if it is able to access cell 2. When SS receives a CELL UPDATE message in cell 2 on uplink CCCH, it replies with CELL UPDATE CONFIRM message.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1				SS begin to broadcast the BCCH
				data in cell 2, but with a power
				level lower than in cell 1.
2 1	•	-	RADIO BEARER	The message includes IE
			RECONFIGURATION	"Primary CPICH Info" and set the
				IE "Primary Scrambling Code" to
				an unknown value (not used by
				cell 1 or cell 2).
<u>32</u>				The SS does not reconfigure the
				new radio bearer in cell 1.UE
				shall detect a "physical channel
				failure" condition and then trigger
				a cell reselection procedure.
<u>3</u>	<u> </u>	<u>←</u>	BCCH	The SS transmit the BCCH in the
				cell2.
<u>4</u>				The UE fails to reconfigure a
				new radio bearer.
<u>5</u>				The SS shall not revert old
				configuration.
<u>6</u>				The UE select the cell 2 and
				enter CELL FACH state.
4 <u>7</u>	-	\rightarrow	CELL UPDATE	The value "cell reselection" shall
				be set in IE "Cell update
				cause". After successfully
				camping onto cell 2, UE shall
				initiate a cell update procedure.
				The updating cause shall be set
		,	OF LUDDATE CONFIDM	to "cell reselection"
5 8	1	←	CELL UPDATE CONFIRM	This message include IE "new U-
				RNTI" and IE "new C-
		`	LITE AND MODILITY INCORDANTION	RNTI"".Contains cell
<u>9</u>	-	\rightarrow	UTRAN MOBILITY INFORMATION	
10			CONFIRM	The IT "feiture equal" shall be
<u>10</u>	-	<u>→</u>	RADIO BEARER RECONFIGURATION	The IE "failure cause" shall be
			<u>FAILURE</u>	set to "cell reselection"

Specific Message Contents

RADIO BEARER RECONFIGURATION

The contents of RADIO BEARER RECONFIGURATION message in this test case is identical as "Packet to CELL_FACH from CELL_DCH in PS" found in Annex A-with the following exceptions:

Information Element	Value/remark
Downlink information for each radio link	
- Primary CPICH Info	
— - Primary Scrambling Code	Set to an unused code different from the 3 cells defined.

CELL UPDATE (Step 7)

The contents of CELL UPDATE message is identical as "Contents of CELL UPDATE message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	
- SRNC Identity	Assigned previously in cell 1
- S-RNTI	Assigned previously in cell 1
Cell Update Cause	"radio link failure"

CELL UPDATE CONFIRM (Step 9)

The contents of CELL UPDATE CONFIRM message is identical as "CELL UPDATE CONFIRM message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	Same as CELL UPDATE message in step 7
New U-RNTI	
- SRNC Identity	<u>'0000 0000 0000 0001'</u>
- S-RNTI	Different from previous S-RNTI
New C-RNTI	Different from previous C-RNTI

RADIO BEARER RECONFIGURATION FAILURE (Step 10)

Information Element	<u>Value/remark</u>
Message Type	"RADIO BEARER RECONFIGURATION FAILURE"
Failure cause	<u>"cell reselection"</u>
Other information element	Not checked

CELL UPDATE

Information Element	Value/remark
U-RNTI	The assigned U-RNTI indicated In RRC CONNECTION
	SETUP message
Cell Update Cause	Cell Reselection

8.2.2.9.5 Test requirement

After step <u>35</u>, the UE shall discover the presence of cell 2, enter CELL_FACH and then perform <u>a-cell reselection</u> <u>updating procedure</u> by the transmission of CELL UPDATE using RLC-TM mode on CCCH. <u>The UE shall transit to CELL_PCH state to monitor the PCCH channel after the reception of CELL UPDATE CONFIRM message.</u>

After step 8 UE transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

After step 9, the UE shall transmit RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC, setting IE "failure cause" to "cell reselection".

8.2.2.10 Radio Bearer Reconfiguration: from CELL_FACH to CELL_DCH: Success

8.2.2.10.1 Definition

8.2.2.10.2 Conformance requirement

The UE shall correctly reconfigure a radio bearer according to a RADIO BEARER RECONFIGURATION message which is communicate with the UTRAN on the new radio bearer in case of a transition from CELL_FACH to CELL_DCH in the same cell.

Reference

3GPP TS 25.331 clause 8.2.2

8.2.2.10.3 Test purpose

To confirm that the UE establishes a new radio bearer by following a RADIO BEARER RECONFIGURATION message received from the SS.

8.2.2.10.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a RADIO BEARER RECONFIGURATION message to the UE which includes the new radio bearer parameters and sets up L1 including the start of tx/rx. The UE reconfigures the new radio bearer and transmits a RADIO BEARER RECONFIGURATION COMPLETE message using AM RLC.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1	•	(RADIO BEARER	This message includes IE
			RECONFIGURATION	"Uplink DPCH Info"
2				Reconfiguration of radio bearer
3		\rightarrow	RADIO BEARER	
			RECONFIGURATION COMPLETE	

Specific Message Contents

RADIO BEARER RECONFIGURATION (Step 1)

Use the same message sub-type titled "Packet to CELL_DCH from CELL_FACH in PS" in Annex A.

8.2.2.10.5 Test requirement

After step 2 the UE shall change its radio bearer configuration and communicate with the SS on the DCCH and DTCH which are being carried by the DPCH physical channel resources.

8.2.2.11 Radio Bearer Reconfiguration from CELL_FACH to CELL_DCH: Failure (Unsupported configuration)

8.2.2.11.1 Definition

8.2.2.11.2 Conformance requirement

The UE shall keep its old configuration when the UE receives a RADIO BEARER RECONFIGURATION message which includes unsupported configuration parameters and transmit a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC which is set to "configuration unsupported" in IE "failure cause

Reference

3GPP TS 25.331 clause 8.2.2

8.2.2.11.3 Test purpose

To confirm that the UE transmits a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC if the RADIO BEARER RECONFIGURATION message received includes unsupported configuration parameters.

8.2.2.15.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a RADIO BEARER RECONFIGURATION message to the UE which includes unsupported configuration parameters of the UE. The UE shall transmit a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC and set "configuration unsupported" in IE "failure cause".

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1	*	(-	RADIO BEARER RECONFIGURATION	The message includes an unsupported configuration for the UE
2	_	>	RADIO BEARER RECONFIGURATION FAILURE	The UE does not change the radio bearer.

Specific Message Contents

RADIO BEARER RECONFIGURATION

The contents of RADIO BEARER RECONFIGURATION message in this test case is identical as "Packet to CELL_DCH from CELL_FACH in PS" found in Annex A with the following exceptions:

Information Element	Value/remark
Frequency info	
- UARFCN uplink(Nu)	63984
- UARFCN downlink(Nd)	Not Present

RADIO BEARER RECONFIGURATION FAILURE

Information Element	Value/remark
Message Type	
Failure cause	Configuration unsupported
Other information element	Not checked

8.2.2.11.5 Test requirement

After step1 the UE shall keep its old configuration and transmit a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC stating "configuration unsupported" in IE "failure cause".

8.2.2.12 Radio Bearer Reconfiguration from CELL_FACH to CELL_DCH: Failure (Physical channel failure and reversion to old configuration)

8.2.2.12.1 Definition

8.2.2.12.2 Conformance requirement

The UE shall revert to the old configuration when the UE fails to reconfigure the new physical channel by received RADIO BEARER RECONFIGURATION message and transmit a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC which is set to "physical channel failure" in IE "failure cause".

Reference

3GPP TS 25.331 clause 8.2.2

8.2.2.12.3 Test purpose

To confirm that the UE reverts to the old configuration and transmits a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC if the UE fails to reconfigure the new radio bearer according to a RADIO BEARER RECONFIGURATION message.

8.2.2.12.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a RADIO BEARER RECONFIGURATION message to the UE, which includes the new radio bearer parameters and does not reconfigure L1. Therefore, the UE cannot reconfigure the new radio bearer and shall attempt to revert to the old configuration. Then the UE shall transmit a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC, with the value "physical channel failure" in IE "failure cause".

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1	+		RADIO BEARER	
			RECONFIGURATION	
2				The SS does not reconfigures L1
				including the start of tx/rx
3	\rightarrow		RADIO BEARER	The UE fails to reconfigure a
			RECONFIGURATION FAILURE	new radio bearer.

Specific Message Contents

RADIO BEARER RECONFIGURATION

Use the same message sub-type titled "Packet to CELL_DCH from CELL_FACH in PS" in Annex A.

RADIO BEARER RECONFIGURATION FAILURE

Information Element	Value/remark
Message Type	
Failure cause	Physical channel failure
Other information element	Not checked

8.2.2.12.5 Test requirement

After step 2 the UE shall revert to the old configuration and transmit a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting "physical channel failure" in IE "failure cause".

8.2.2.13 Radio Bearer Reconfiguration from CELL_FACH to CELL_DCH: Failure (Physical channel failure and reversion failure)

8.2.2.13.1 Definition

8.2.2.13.2 Conformance requirement

The UE shall perform a <u>cell update n RRC connection re establishment</u> procedure when the UE fails to revert to the old configuration after the detection of physical channel failure in the radio bearer reconfiguration procedure. <u>After the UE completes</u> cell update procedure, the UE <u>transmits</u> RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC which set IE "failure cause" to "physical channel failure".

Reference

3GPP TS 25.331 clause 8.2.2

8.2.2.13.3 Test purpose

To confirm that the UE <u>transmits RADIO BEARER RECONFIGURATION FAILURE message after it completes initiates a cell update n RRC connection re establishment procedure, when the UE cannot reconfigure the new radio bearer following a failure to configure L1 parameters for the new radio bearer settings and also a reversion failure to the old configuration.</u>

8.2.2.13.4 Method of test

Initial Condition

System Simulator: 2 cells - Cell 1 is active, Cell 2 is inactive

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108 in cell 1

Test Procedure

The UE is in the CELL_FACH state in cell 1. The SS transmits a RADIO BEARER RECONFIGURATION message to the UE, which includes the new radio bearer parameters but <u>SS</u> does not reconfigure L1 <u>such as to-catered</u> to the new radio bearer settings. The UE cannot reconfigure the new radio bearer and wants to revert to the old configuration. But the SS does not configure the old radio bearer and begins to broadcast the BCCH in a cell 2 and stops broadcasting the BCCH in cell 1. Then the UE finds a new cell 2 and <u>transmits</u> CELL UPDATE message on uplink CCCH with IE "Cell update cause" set to "radio link failure". The SS shall transmit CELL UPDATE CONFIRM message on downlink CCCH after receiving CELL UPDATE message. The UE transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC and subsequently transmit RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting the value "physical channel failure" to IE "failure cause" transmits to it an RRC CONNECTION RE ESTABLISHMENT REQUEST message, which includes the IE "U RNTI" on the uplink CCCH. The SS replies with an RRC CONNECTION RE ESTABLISHMENT message, specifying a new TFCS according to the new transport channel. After the UE receives an RRC CONNECTION RE ESTABLISHMENT message, it reconfigures the new radio connection and transmits an RRC CONNECTION RE ESTABLISHMENT COMPLETE message on the uplink DCCH using AM RLC.

Expected sequence

Step	Direction	Message	Comment
	UE SS	1	
1	←	RADIO BEARER RECONFIGURATION	
2			The SS does not reconfigure the new radio bearer and delete the old configuration.
α			The UE shall detect a failure to reconfigure a new radio bearer, and start to revert to old configuration.
4	←	ВССН	The SS starts to transmit the BCCH in cell 2 and delete the old radio bearer settings in cell 1.
<u>5</u>			The SS shall not revert old configuration.
<u>6</u>			The UE enter CELL FACH state and select the cell2.
<u>7</u>	<u>→</u>	CELL UPDATE	The value "radio link failure" shall be set in IE "Cell update cause".
<u>8</u>	<u>←</u>	CELL UPDATE CONFIRM	This message include IE "new U-RNTI" and IE "new C-RNTI".
9	<u>→</u>	UTRAN MOBILITY INFORMATION CONFIRM	
<u>10</u>		RADIO BEARER RECONFIGURATION FAILURE	The IE "failure cause" shall be set to "physical channel failure"
5	->	RRC CONNECTION RE-ESTABLISHMENT REQUEST	The UE shall find cell 2 and transmits this message which includes the IE "U-RNTI" it possesses.
6	+	RRC CONNECTION RE-ESTABLISHMENT	Including the new TFCS according to the new transport channel.
7	→	RRC CONNECTION RE-ESTABLISHMENT COMPLETE	

Specific Message Contents

RADIO BEARER RECONFIGURATION (Step 1)

Use the same message sub-type titled "Packet to CELL_DCH from CELL_FACH in PS" in Annex A.

CELL UPDATE (Step 7)

The contents of CELL UPDATE message is identical as "Contents of CELL UPDATE message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	
- SRNC Identity	Assigned previously in cell 1
<u>- S-RNTI</u>	Assigned previously in cell 1
Cell Update Cause	<u>"radio link failure"</u>

CELL UPDATE CONFIRM (Step 9)

The contents of CELL UPDATE CONFIRM message is identical as "CELL UPDATE CONFIRM message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	Same as CELL UPDATE message in step 7
New U-RNTI	
- SRNC Identity	<u>'0000 0000 0000 0001'</u>
<u>- S-RNTI</u>	<u>Different from previous S-RNTI</u>
New C-RNTI	Different from previous C-RNTI

RADIO BEARER RECONFIGURATION FAILURE (Step 9)

Information Element	<u>Value/remark</u>
Message Type	"RADIO BEARER RECONFIGURATION FAILURE"
Failure cause	"physical channel failure"
Other information element	Not checked

8.2.2.13.5 Test requirement

After step 4-5 the UE shall detect the presence of cell 2, enter CELL_FACH state., and attempt to perform RRC connection re-establishment procedure.

After step 7 the UE shall successfully re-establish an RRC connection, and be able to access the dedicated physical resources in cell 2.

After step 8 the UE shall transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

After step 9 the UE shall transmit RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting the IE "failure cause" to "physical channel failure".

8.2.2.14 Radio Bearer Reconfigure from CELL_FACH to CELL_DCH: Failure (Incompatible simultaneous reconfiguration)

8.2.2.14.1 Definition

8.2.2.14.2 Conformance requirement

If the UE receives a RADIO BEARER RECONFIGURATION message whilst reconfiguring due to a radio bearer message other than RADIO BEARER RECONFIGURATION, it shall keep its configuration as if the RADIO BEARER RECONFIGURATION message had not been received. If the UE is configuring itself after receiving a radio bearer message other than RADIO BEARER RECONFIGURATION message, and when the UE subsequently receive a RADIO BEARER RECONFIGURATION message, the UE shall transmit a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC which is set to "Incompatible simultaneous reconfiguration" in IE "failure cause".

Reference

3GPP TS 25.331 clause 8.2.2

8.2.2.14.3 Test purpose

To confirm that if the UE receives a RADIO BEARER RECONFIGURATION message whilst reconfiguring due to a radio bearer message other than RADIO BEARER RECONFIGURATION, it shall keep its configuration as if the RADIO BEARER RECONFIGURATION message had not been received and complete the reconfiguration according to the previously received message.

To confirm that the UE keeps its configuration and transmits a RADIO BEARER RECONFIGURATION FAILURE message when the UE receives a RADIO BEARER RECONFIGURATION message before the UE configures the radio bearer according to a previous message.

8.2.2.14.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL DCH state. The SS transmits a RADIO BEARER SETUP message to the UE. The SS transmits a RADIO BEARER RECONFIGURATION message before the "activation time" indicated in the RADIO BEARER SETUP message expires. When the UE receives the RADIO BEARER RECONFIGURATION message, the UE shall keep the configuration as if it had not received the RADIO BEARER RECONFIGURATION message and shall transmit a RADIO RECONFIGURATION SETUP FAILURE message on the DCCH using AM RLC with IE "failure cause" set to "incompatible simultaneous reconfiguration". After the SS receives the RADIO BEARER SETUP FAILURE message, the UE reconfigures the new physical channel parameters and transmits a RADIO BEARER SETUP COMPLETE message on DCCH using AM RLC. The UE is in the CELL_FACH state. When the SS transmits a RADIO BEARER RECONFIGURATION message to the UE before the completion of a radio bearer reconfiguration commanded earlier, the UE keeps the old configuration. The UE shall also transmit a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC which is set to "incompatible simultaneous reconfiguration" in IE "failure cause".

Expected sequence

Step	Direc	tion	Message	Comment
	UE	SS		
1	+	-	RADIO BEARER SETUP	Including IE "Uplink DPCH info"
2	+	-	RADIO BEARER RECONFIGURATION	Sent before the elapse of the "Activation Time" indicated in the previous message.
3	1	>	RADIO BEARER RECONFIGURATION FAILURE	The UE does not change the configuration because of receiving the RADIO BEARER RECONFIGURATION message.
4	7]	<u>></u>	RADIO BEARER SETUP COMPLETE	This message is on DCCH using AM RLC

Specific Message Contents

RADIO BEARER SETUP (Step 1)

For RADIO BEARER SETUP in step 1, use the message sub-type indicated as "Packet to CELL_DCH from CELL_FACH in PS" found in Annex A.

RADIO BEARER RECONFIGURATION (Step 2)

The contents of RADIO BEARER RECONFIGURATION message in this test case is identical as "Packet to CELL_DCH from CELL_FACH in PS" found in Annex A with the following exceptions:

Information Element	Value/remark
Activation Time	Current CFN-[current CFN mod 8 + 8]

RADIO BEARER RECONFIGURATION FAILURE

Information Element	Value/remark
Message Type	"RADIO BEARER RECONFIGURATION FAILURE"
Failure cause	Incompatible simultaneous reconfiguration
Other information element	Not checked

8.2.2.14.5 Test requirement

After step 1, SS transmits a RADIO BEARER RECONFIGURATION message before the expiry of the activation time specified in the message of step 1.

After step 2 the UE shall keep its configuration as if the UE had not received the RADIO BEARER SETUP message and shall transmit a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC with IE "failure cause" set to "Incompatible simultaneous reconfiguration".

After step 4 the UE communicates with the SS on the DCCH and DTCH using the new physical channel parameters reconfigured as a result of the RADIO BEARER SETUP message.

After step 2 the UE shall keep its configuration and transmit a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC which is set to "incompatible simultaneous reconfiguration" in IE "failure cause".

8.2.2.15 Radio Bearer Reconfiguration from CELL_FACH to CELL_DCH: Failure (Invalid message reception and Invalid configuration)

8.2.2.15.1 Definition

8.2.2.15.2 Conformance requirement

The UE shall keep its old configuration when the UE receives a RADIO BEARER RECONFIGURATION message, which includes the spare undefined value in the mandatory IE "UTRAN DRX cycle length coefficient DRX indicator" with criticality defined as "Reject". Then it shall transmit a RADIO BEARER RECONFIGURATION FAILURE message setting "protocol error" in IE "failure cause" and also setting "Information element value not comprehended" in IE "Protocol error cause". The UE shall keep existing configuration before reception of a RADIO BEARER RECONFIGURATION message when the RADIO BEARER RECONFIGURATION message include some IEs set to invalid value, and then the UE shall transmit RADIO BEARER RECONFIGURATION FAILURE including IE "failure cause" set to "invalid configuration".

Reference

3GPP TS 25.331 clause 8.2.2

8.2.2.15.3 Test purpose

To confirm that the UE transmits a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC if it receives a RADIO BEARER RECONFIGURATION message which includes the spare undefined value in the mandatory IE "UTRAN DRX cycle length coefficient DRX indicator", with criticality defined as "Reject".

To confirm that the UE transmits a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC if it receives a RADIO BEARER RECONFIGURATION message including some IEs set to invalid value.

8.2.2.15.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a RADIO BEARER RECONFIGURATION message to the UE which includes the spare undefined value in the mandatory IE "UTRAN DRX cycle length coefficient DRX indicator". The UE shall keep the old configuration and transmit a RADIO BEARER RECONFIGURATION FAILURE message

on the DCCH using AM RLC indicating "protocol error" in IE "failure cause" and also set "Information element value not comprehended" in IE "Protocol error cause". The UE keeps initial configuration when SS transmits RADIO BEARER RECONFIGURATION message including some IEs set to invalid value. The UE transmit RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting the value "invalid configuration" to IE "failure cause".

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	+	RADIO BEARER RECONFIGURATION	
2	\rightarrow	RADIO BEARER RECONFIGRATION FAILURE	The UE does not change the configuration.
<u>3</u>	<u>←</u>	RADIO BEARER RECONFIGURATION	This message includes IE set to invalid value
4			The UE does not change the configuration
<u>5</u>	\rightarrow	RADIO BEARER RECONFIGRATION FAILURE	The IE "failure cause" shall be set to "invalid configuration

Specific Message Contents

RADIO BEARER RECONFIGURATION (Step 1)

The contents of RADIO BEARER RECONFIGRATION message in this test case is identical as "Packet to CELL_DCH from CELL_FACH in PS" found in Annex A with the following exceptions:

Information Element	Value/remark
UTRAN DRX cycle length coefficientDRX indicator	<u>Undefined spare-</u> value

RADIO BEARER RECONFIGURATION FAILURE (Step 2)

Information Element	Value/remark
Message Type	"RADIO BEARER RECONFIGURATION FAILURE"
Failure cause	
- Failure cause	Protocol error
- Protocol error information	
- Protocol error cause	Information element value not comprehended
Other information element	Not checked

RADIO BEARER RECONFIGURATION (Step 3)

The contents of RADIO BEARER RECONFIGRATION message in this test case is identical as "Packet to CELL DCH from CELL FACH in PS" found in Annex A with the following exceptions:

Added or Reconfigured UL TrCH information	
- Transport channel identity	<u>2</u>
TFS	
 Dynamic Transport format information 	(This IE is repeated for TFI number)
- RLC size	Reference to TS34.108 clause 6.10 Parameter Set
- CHOICE Logical Channel list	
- Explicit List	
- RB identity	2

RADIO BEARER RECONFIGURATION FAILURE (Step 5)

Information Element	<u>Value/remark</u>
Message Type	
Failure cause	Invalid configuration
Other information element	Not checked

8.2.2.15.5 Test requirement

After step1 the UE shall keep its old configuration and transmit a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC, which contain the cause "protocol error" in IE "failure cause" and "Information element value not comprehended" in IE "Protocol error cause".

After step 3 the UE shall keep its old configuration.

After step 4 the UE shall transmit RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting the value "invalid configuration" to IE "failure cause".

8.2.2.16 Radio Bearer Reconfiguration from CELL_FACH to CELL_<u>FACH</u>: Success (Continue and StopDCH: Failure (Suspension of signalling bearer)

8.2.2.16.1 Definition

8.2.2.16.2 Conformance requirement

The UE shall continue or stop the uplink transmission when the UTRAN indicate stop or continue uplink transmission in radio bearer reconfiguration procedure revert to the configuration prior to the reception of the RADIO BEARER RECONFIRGURATION message which specifies the suspension of an existing signalling radio bearer. It shall then transmit a RADIO BEARER RECONFIGURATION FAILURE message to UTRAN.

Reference

3GPP TS 25.331 clause 8.2.2

8.2.2.16.3 Test purpose

To confirm that the UE <u>reconfigures new radio bearer and have the uplink transmission according to a RADIO BEARER RECONFIGURATIO message which indicates that uplink transmission is continued.</u>

To confirm that the UE reconfigures new radio bearer and don't transmit data according to a RADIO BEARER RECONFIGURATIO message which indicates that uplink transmission is stopped. reverts to the old configuration and transmits a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC if the UE fails to reconfigure the new radio bearer due to an inappropriate suspension request in RADIO BEARER RECONFIGURATION message.

8.2.2.16.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a RADIO BEARER RECONFIGURATION message to the UE, which includes IE" RB stop/continue" set to "continue". The UE reconfigures new radio bearer and transmit RADIO BEARER RECONFIGURATION COMPLETE message on the DCCH using AM RLC. The UE communicates with the SS after transmission the RADIO BEARER RECONFIGURATION COMPLETE message. Then, SS transmits a RADIO BEARER RECONFIGURATION message including IE" RB stop/continue" set to "stop". The UE reconfigures new radio bearer and transmits RADIO BEARER RECONFIGURATION COMPLETE message on the DCCH using AM RLC. The UE shall not transmit any uplink data without Signalling message after transmission the RADIO BEARER RECONFIGURATION COMPLETE message.includes the IE "RB suspend/resume" set to "Suspend" for one existing signalling radio bearer. The UE shall revert to the old configuration. Then the UE transmits a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC indicating "configuration unsupported" in IE "failure cause".

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1	•	(RADIO BEARER RECONFIGURATION	The message includes IE "RB stop/continuesuspend/resume" for one of the signalling radio bearer.
2				The UE select PRACH and S-CCPCH, using SIB5 or SIB6.
2 3	-	→	RADIO BEARER RECONFIGURATION COMPLETE FAILURE	The UE shall revert to the old configuration
4				The SS Shall communicate with the UE.
<u>5</u>	_	€	RADIO BEARER RECONFIGURATION	This message include IE" RB stop/continue ".
<u>6</u>				The UE select PRACH and S-CCPCH, using SIB5 or SIB6.
7	=	<u>></u>	RADIO BEARER RECONFIGURATION COMPLETE	
8				The SS shall not receive any data from the UE without Signalling message.

Specific Message Contents

RADIO BEARER RECONFIGURATION (Step 1)

The contents of RADIO BEARER RECONFIGURATION message in this test case is identical as "Packet to CELL_DCH from CELL_FACH in PS" found in Annex A with the following exceptions:

Information Element	Value/remark
RB information to reconfigure list	
RB information to reconfigure I	
-RB identity	<u>25</u>
-RB stop/continuesuspend/resume	Set to "continueSuspend"

RADIO BEARER RECONFIGURATION FAILURE

Information Element	Value/remark	
Message Type	"RADIO BEARER RECONFIGURATION FAILURE"	
Failure cause	configuration unsupported	
Other information element	Not checked	
RB information to reconfigure list		
RB information to reconfigure	<u>5</u>	
-RB identity		
-RB stop/continue	Set to "continue"	

8.2.2.16.5 Test requirement

After step 2-3 the UE shall communicate with the SS using new configuration.

After step 7 the UE shall communicate with the SS using new configuration, but shall not transmit any data to the SS without signalling message.revert to the old configuration and transmit a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC. This message shall contain the value "configuration unsupported" in IE "failure cause".

8.2.2.17 Radio Bearer Reconfiguration from CELL_FACH to CELL_FACH: Success

8.2.2.17.1 Definition

8.2.2.17.2 Conformance requirement

The UE shall correctly reconfigure a radio bearer according to a RADIO BEARER RECONFIGURATION message which is communicate with the UTRAN on the new radio bearer and a transition from CELL_FACH to CELL_FACH in the another cell.

Reference

3GPP TS 25.331 clause 8.2.2

8.2.2.17.3 Test purpose

To confirm that the UE establishes a new radio bearer by following a RADIO BEARER RECONFIGURATION message received from the SS.

8.2.2.17.4 Method of test

Initial Condition

System Simulator: 2-1 cells. Cell 1 is active, Cell 2 is inactive

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. SS begins to broadcast the BCCH of cell 2. The SS transmits a RADIO BEARER RECONFIGURATION message to the UE which includes the new transport channel parameter reconfigure for transit. The UE reconfigures the new transport cannel and transmits a RADIO BEARER RECONFIGURATION COMPLETE message using AM RLC.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
4	1	-	BCCH	The SS starts to broadcast the
				BCCH in cell 2.
2 1	←		RADIO BEARER	Including Primary CPICH Info for
			RECONFIGURATION	cell 2.
<u>2</u>				The UE select PRACH and S-
				CCPCH using SIB5 or SIB6.
3				Reconfiguration of
				Transport channel
4 <u>3</u>	-	→	RADIO BEARER	Received on the uplink DCCH of
			RECONFIGURATION COMPLETE	cell 2.

Specific Message Contents

RADIO BEARER RECONFIGURATION (Step 2)

Use the same message sub-type titled "Packet to CELL_FACH from CELL_FACH in PS" in Annex A.

8.2.2.17.5 Test requirement

After step 3 the UE shall <u>change its radio bearer configuration and be in transit from CELL_FACH to CELL_FACH in the another cell and communicate with the SS on the DCCH and DTCH using the modified transport channel configurations</u>

After step 4 the UE shall communicate with the SS on the DCCH and DTCH, using the common physical channel.

8.2.2.18 Radio Bearer Reconfiguration from CELL_FACH to CELL_FACH: Failure (Cell re-selectionPhysical channel failure)

8.2.2.18.1 Definition

8.2.2.18.2 Conformance requirement

The UE shall initiate the cell reselection procedure when the UE cannot use the assigned physical channel according to a RADIO BEARER RECONFIGURATION message, during the transition from CELL_FACH to CELL_FACH—in the another cell. After the UE completes cell update procedure, the UE transmits RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC which is set to "cell reselection" in IE "failure cause".

Reference

3GPP TS 25.331 clause 8.2.2

8.2.2.18.3 Test purpose

To confirm that the UE initiates a cell update because it cannot access the assigned physical channel in current cell, when attempting to transit from CELL FACH to CELL FACH state during the execution of radio bearer reconfiguration procedure.

To confirm that the UE transmit RADIO BEARER SETUP FAILURE message after complete a cell update.

To confirm that the UE reverts to the old configuration and transmits a RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC if the UE fails to reconfigure the new radio bearer according to the RADIO BEARER RECONFIGURATION message received previously.

8.2.2.18.4 Method of test

Initial Condition

System Simulator: 3-2 cells Cell 1 is active, Cell 2 is inactive, Cell 3 is inactive

UE: PS-DCCH+DTCH FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state in cell 1. The SS transmits a RADIO BEARER RECONFIGURATION message to the UE, which includes the new radio bearer parameters. After transmitting the RADIO BEARER RECONFIGURATION message, the SS shall not reconfigure L1 in accordance to the settings in the message. At the same time, SS begins to broadcast the BCCH in a cell 2. The UE recognizes that it cannot reconfigure the new radio bearer and wants to revert to the old configuration, but the UE cannot revert to the old configuration. The UE shall select cell 2 by performing cell re-selection. The UE shall move to CELL FACH and transmit CELL UPDATE message on uplink CCCH with IE "Cell update cause" set to "cell reselection". The SS shall transmit CELL UPDATE CONFIRM message on downlink CCCH after receiving CELL UPDATE message. The UE transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC and transmit RADIO BEARER RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting the value "cell reselection" to IE "failure cause". The SS begins to broadcast the BCCH in cell 2 and then transmits a RADIO BEARER RECONFIGURATION message, resulting in the UE to attempt to transit from CELL_FACH in cell 1 to CELL_FACH in cell 3. The UE is expected to fail to access the assigned physical channel as the SS ceases downlink

transmission in cell 3. After the UE successfully camp onto cell 2, it shall initiate the cell update procedure in cell 2. The procedure completes when the SS confirms UE's cell update request.

Expected sequence

Step	Direction	Message	Comment
_	UE SS	_	
1	+	BCCH	The SS starts to broadcast the
			BCCH in cell 2.
2	←	RADIO BEARER RECONFIGURATION	The message assigns common
			resources so that the UE is
			expected to execute a transition
			from CELL_FACH in cell 1 to
			CELL_FACH in cell 3
<u>3</u>			The SS does not reconfigure the
4		DOOL	new radio bearer in cell 1.
<u>4</u>	_ ←	<u>BCCH</u>	The SS transmit the BCCH in
0.5			the cell 2.
<u>35</u>			The UE shall detect a failure to
			use the assigned common
			physical channel and try to
46	→	CELL UPDATE	perform a cell reselection.
4 <u>6</u>	7	CELL UPDATE	The value "cell reselection" shall be set in IE "cell update
			cause". The UE discover the
			presence of cell 2 and triggers a
			cell update procedure.
5 7	←	CELL UPDATE CONFIRM	This message include IE "new
<u> </u>	`	OLLE OF BATE CONTINU	U-RNTI" and IE "new C-
			RNTI"". SS sends this message
			so that the UE moves to
			CELL_PCH state.
8	<u></u>	UTRAN MOBILITY INFORMATION	
		CONFIRM	
9	<u> </u>	RADIO BEARER RECONFIGURATION	The IE "failure cause" shall be
		<u>FILURE</u>	set to "cell reselection"

Specific Message Contents

Use the same message sub-type titled "Packet to CELL_FACH from CELL_FACH in PS" in Annex A, with the following exceptions:

Information Element	Value/remark
Downlink information for each radio links	
- Primary CPICH Info	
- Primary scrambling code	200

CELL UPDATE (Step 6)

The contents of CELL UPDATE message is identical as "Contents of CELL UPDATE message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	
- SRNC Identity	Assigned previously in cell 1
<u>- S-RNTI</u>	Assigned previously in cell 1
Cell Update Cause	<u>"radio link failure"</u>

CELL UPDATE CONFIRM (Step 8)

The contents of CELL UPDATE CONFIRM message is identical as "CELL UPDATE CONFIRM message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	Same as CELL UPDATE message in step 7
New U-RNTI	
- SRNC Identity	<u>'0000 0000 0000 0001'</u>
<u>- S-RNTI</u>	<u>Different from previous S-RNTI</u>
New C-RNTI	Different from previous C-RNTI

RADIO BEARER RECONFIGURATION FAILURE (Step 9)

Information Element	<u>Value/remark</u>	
Message Type	"RADIO BEARER RECONFIGURATION FAILURE"	
Failure cause	<u>"cell reselection"</u>	
Other information element	Not checked	

8.2.18.5 Test requirement

After step3-step 4the UE shall discover the presence of cell 2, enter CELL_FACH state, and initiate a cell update procedure in that cell.

After step 7 UE transmits UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

After step 8 UE transmits RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC, setting the value "cell reselection" to IE "failure cause".

8.2.2.19 Radio Bearer Reconfiguration from CELL_DCH to CELL_DCH: Success (Subsequently received)

8.2.2.19.1 Definition

8.2.2.19.2 Conformance requirement

If the UE receives a RADIO BEARER RECONFIGURATION message before the UE configures the radio bearer according to the previous RADIO BEARER RECONFIGURATION message, the UE shall ignore the new RADIO BEARER RECONFIGURATION message and configure according to the previous RADIO BEARER RECONFIGURATION message. Finally, the UE shall transmit RADIO BEARER RECONFIGURATION COMPLETE message on the DCCH using AM RLC.

Reference

3GPP TS 25.331 clause 8.2.2

8.2.2.19.3 Test purpose

If the UE receives another RADIO BEARER RECONFIGURATION message before the UE configures the radio bearer according to a previous RADIO BEARER RECONFIGURATION message, the UE shall ignore the new RADIO BEARER RECONFIGURATION message and configure according to the previous RADIO BEARER RECONFIGURATION message. Finally, the UE shall transmit RADIO BEARER RECONFIGURATION COMPLETE message on the DCCH using AM RLC.

8.2.2.19.4 Method of test

Initial Condition

System Simulator: 1 cell

Test Procedure

The UE is in the CELL_DCH state. SS transmits a RADIO BEARER RECONFIGURATION message to the UE before the UE configures the radio bearer according to the RADIO BEARER RECONFIGURATION message prior to this new message. The UE ignores the new RADIO BEARER RECONFIGURATION message and configures according to the former RADIO BEARER RECONFIGURATION message. On completion of radio bearer configuration, the UE shall transmit RADIO BEARER RECONFIGURATION COMPLETE message on the DCCH using AM RLC.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	+	RADIO BEARER RECONFIGURATION	Including IE "Uplink DPCH info"
2	←	RADIO BEARER RECONFIGURATION	Sent before the "activation time" in step 1 has elapsed
3	÷	RADIO BEARER RECONFIGURATION COMPLETE	The UE ignores the RADIO BEARER RECONFIGURATION message in step 2 and confirms configuration according to the RADIO BEARER RECONFIGURATION message in step 1.

Specific Message Contents

RADIO BEARER RECONFIGURATION (Step 1)

The contents of RADIO BEARER RECONFIGURATION message in this test case is identical as "Packet to CELL_DCH from CELL_DCH in PS" found in Annex A with the following exceptions:

Information Element	Value/remark	
Activation Time	[256+Current CFN-[current CFN mod 8 + 8]]MOD 256	

8.2.2.19.5 Test requirement

After step 3 the UE shall communicate with the SS on the radio bearer specified in the RADIO BEARER RECONFIGURATION message in step 1.

8.2.2.20 Radio Bearer Reconfigure from CELL_FACH to CELL_DCH: Success (Subsequently received)

8.2.2.20.1 Definition

8.2.2.20.2 Conformance requirement

If the UE receives a RADIO BEARER RECONFIGURATION message before the UE configures the radio bearer according to the previous RADIO BEARER RECONFIGURATION message, the UE shall ignore the new RADIO BEARER RECONFIGURATION message and configure according to the previous RADIO BEARER RECONFIGURATION message. Finally, the UE shall transmit RADIO BEARER RECONFIGURATION COMPLETE message on the DCCH using AM RLC.

Reference

3GPP TS 25.331 clause 8.2.2

8.2.2.20.3 Test purpose

To confirm that if the UE receives another RADIO BEARER RECONFIGURATION message before the UE configures the radio bearer according to a previous RADIO BEARER RECONFIGURATION message, the UE shall ignore the new RADIO BEARER RECONFIGURATION message and configure according to the previous RADIO BEARER RECONFIGURATION message. Finally, the UE shall transmit RADIO BEARER RECONFIGURATION COMPLETE message on the DCCH using AM RLC.

8.2.2.20.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_FACH(state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. SS transmits a RADIO BEARER RECONFIGURATION message to the UE before the UE configures the radio bearer according to the RADIO BEARER RECONFIGURATION message prior to this new message. The UE ignores the new RADIO BEARER RECONFIGURATION message and configures according to the former RADIO BEARER RECONFIGURATION message. On completion of radio bearer configuration, the UE shall transmit RADIO BEARER RECONFIGURATION COMPLETE message on the DCCH using AM RLC.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	+	RADIO BEARER RECONFIGURATION	Including IE "Uplink DPCH info"
2	+	RADIO BEARER RECONFIGURATION	SS send this message before the expiry of activation time specified in RADIO BEARER SETUP message of step 1.
3	→	RADIO BEARER RECONFIGURATION COMPLETE	The UE ignores the RADIO BEARER RECONFIGURATION message in step 2 and confirms configuration according to the RADIO BEARER RECONFIGURATION message in step 1.

Specific Message Contents

The contents of RADIO BEARER RECONFIGURATION message in this test case is identical as "Packet to CELL_FACH from CELL_DCH in PS" found in Annex A with the following exceptions:

RADIO BEARER RECONFIGURATION (step 1)

Information Element	Value/remark
Activation Time	[256+Current CFN-[current CFN mod 8 + 8]]MOD 256

8.2.2.20.5 Test requirement

After step 3 the UE shall communicate with the SS on the radio bearer specified in the RADIO BEARER RECONFIGURATION message in step 1.

8.2.2.3121 Radio Bearer Reconfiguration from CELL_DCH to CELL_PCH: Success

8.2.2.3121.1 Definition

8.2.2.3121.2 Conformance requirement

The UE shall transmit RADIO BEARER RECONFIGURATION COMPLETE message and transit from CELL_DCH to CELL_PCH when receives a RADIO BEARER RECONFIGURATION message. And then, the UE shall reconfigure radio bearers according to the RADIO BEARER RECONFIGURATION message.

Reference

3GPP TS 25.331 clause 8.2.2

8.2.2.3121.3 Test purpose

To confirm that the UE transmit RADIO BEARER RECONFIGURATION COMPLETE before entering CELL_PCH state after it received a RADIO BEARER RECONFIGURATION message and reconfigured its radio bearers. The UE is in CELL PCH state of the same cell.

8.2.2.3121.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_DCH(state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a RADIO BEARER RECONFIGURATION message. The UE transmits RADIO BEARER RECONFIGURATION COMPLETE message to the UE using AM RLC and enters into CELL_PCH state. The SS transmits a PAGING TYPE 1 message and the UE accepts it and enters the CELL_FACH state.

Expected sequence

Step	Direc	tion	Message	Comment
	UE	SS		
1	•	(RADIO BEARER RECONFIGURATION	
2	-	\rightarrow	RADIO BEARER RECONFIGURATION COMPLETE	The UE sends this message before state transition.
3				Reconfiguration of Radio Bearer after state transition.
4	•	(PAGING TYPE 1	The SS transmits this message included a matched identity.
5	-	\rightarrow	CELL UPDATE	The UE is in CELL_FACH state.

Specific Message Contents

RADIO BEARER RECONFIGURATION (Step 1)

Use the same message sub-type titled "Packet to CELL_FACH from CELL_DCH in PS" in Annex A with following exceptions:

Information Element	Value/remark
DRX indicatorRRC State Indicator	DRX with CELL updatingCELL_PCH

PAGING TYPE 1 (Step 4)

Use the same message sub-type titled "TM (SMS in PS)" in Annex A with following exceptions:

Information Element	Value/remark
Paging record list	
Paging record	
- CHOICE Paging originator	UTRAN originator
- U-RNTI	-
- SRNC Identity	Previously assigned SRNC identity
- S-RNTI	Previously assigned S-RNTI

8.2.2.3121.5 Test requirement

After step 1 the UE transmits RADIO BEARER RECONFIGURATION COMPLETE message to the UE on uplink DCCH using AM RLC.

After step 3 the UE shall transit from CELL_DCH to CELL_PCH.

8.2.2.3222 Radio Bearer Reconfiguration from CELL_DCH to URA_PCH: Success

8.2.2.3222.1 Definition

8.2.2.3222.2 Conformance requirement

The UE shall transmit RADIO BEARER RECONFIGURATION COMPLETE message and transit from CELL_DCH to URA_PCH when receives a RADIO BEARER RECONFIGURATION message. And then, the UE shall reconfigure a radio bearer according to the RADIO BEARER RECONFIGURATION message.

Reference

3GPP TS 25.331 clause 8.2.2

8.2.2.3222.3 Test purpose

To confirm that the UE transmit RADIO BEARER RECONFIGURATION COMPLETE before entering URA_PCH state after it received a RADIO BEARER RECONFIGURATION message and reconfigured its radio bearers. The UE is in URA_PCH state of the same cell.

8.2.2.3222.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_DCH(state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the URA_DCH state. The SS transmits a RADIO BEARER RECONFIGURATION message. The UE transmits RADIO BEARER RECONFIGURATION COMPLETE message to the UE using AM RLC and enters into URA_PCH state. The SS transmits a PAGING TYPE 1 message and the UE accepts it and enters the CELL_FACH state.

Expected sequence

Step	Dire	ction	Message	Comment
	UE	SS		
1		←	RADIO BEARER	
			RECONFIGURATION	
2		\rightarrow	RADIO BEARER	The UE sends this message
			RECONFIGURATION COMPLETE	before state transition.
3				Reconfiguration of
				Radio Bearer after state
				transition.
4		+	PAGING TYPE 1	The SS transmits this message
				included a matched identity.
5		\rightarrow	CELL UPDATE	The UE is in CELL_FACH state.

Specific Message Contents

RADIO BEARER RECONFIGURATION (Step 1)

Use the same message sub-type titled "Packet to CELL_FACH from CELL_DCH in PS" in Annex A with following exceptions:

Information Element	Value/remark
DRX indicator RRC State Indicator	DRX with URA updatingURA_PCH

PAGING TYPE 1 (Step 4)

Use the same message sub-type titled "TM (SMS in PS)" in Annex A with following exceptions:

Information Element	Value/remark
Paging record list	
Paging record	
- CHOICE Paging originator	UTRAN originator
- U-RNTI	
- SRNC Identity	Previously assigned SRNC identity
- S-RNTI	Previously assigned S-RNTI

8.2.2.3222.5 Test requirement

After step 1 the UE transmits RADIO BEARER RECONFIGURATION COMPLETE message to the UE on uplink DCCH using AM RLC.

After step 3 the UE shall transits from CELL_DCH to URA_PCH.

8.2.2.3323 Radio Bearer Reconfiguration from CELL_FACH to CELL_PCH: Success

8.2.2.3323.1 Definition

8.2.2.3323.2 Conformance requirement

The UE shall transmit RADIO BEARER RECONFIGURATION COMPLETE message and transit from CELL_FACH to CELL_PCH when receive a RADIO BEARER RECONFIGURATION message. And then, the UE shall reconfigure radio bearers according to the RADIO BEARER RECONFIGURATION message.

Reference

3GPP TS 25.331 clause 8.2.2

8.2.2.3323.3 Test purpose

To confirm that the UE transmits RADIO BEARER RECONFIGURATION COMPLETE before entering CELL_PCH state after it received a RADIO BEARER RECONFIGURATION message and reconfigured its radio bearers. The UE is in CELL_PCH state of the same cell.

8.2.2.33.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_FACH(state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a RADIO BEARER RECONFIGURATION message. The UE transmits RADIO BEARER RECONFIGURATION COMPLETE message to the UE using AM RLC and enters into CELL_PCH state. The SS transmits a PAGING TYPE 1 message and the UE accepts it and enters the CELL_FACH state again.

Expected sequence

Step	Dire	ction	Message	Comment
	UE	SS		
1		+	RADIO BEARER	
			RECONFIGURATION	
2		\rightarrow	RADIO BEARER	The UE sends this message
			RECONFIGURATION COMPLETE	before state transition.
3				Reconfiguration of Radio Bearer
				after state transition.
4		+	PAGING TYPE 1	The SS transmits this message
				included a matched identity.
5		\rightarrow	CELL UPDATE	The UE is in CELL_FACH state.

Specific Message Contents

RADIO BEARER RECONFIGURATION (Step 1)

Use the same message sub-type titled "Packet to CELL_FACH from CELL_FACH in PS" in Annex A with following exceptions:

Information Element	Value/remark	
DRX indicator RRC State Indicator	DRX with CELL updatingCELL PCH	

PAGING TYPE 1 (Step 4)

Use the same message sub-type titled "TM (SMS in PS)" in Annex A with following exceptions:

Information Element	Value/remark
Paging record list	
Paging record	
- CHOICE Paging originator	UTRAN originator
- U-RNTI	
- SRNC Identity	Previously assigned SRNC identity
- S-RNTI	Previously assigned S-RNTI

8.2.2.3323.5 Test requirement

After step 1 the UE transmits RADIO BEARER RECONFIGURATION COMPLETE message to the UE on uplink DCCH using AM RLC.

After step 3 the UE shall transit from CELL_DCH to CELL_PCH.

8.2.2.3424 Radio Bearer Reconfiguration from CELL_FACH to URA_PCH: Success

8.2.2.3424.1 Definition

8.2.2.3424.2 Conformance requirement

The UE shall transmit RADIO BEARER RECONFIGURATION COMPLETE message and transit from CELL_FACH to URA_PCH when receive a RADIO BEARER RECONFIGURATION message. And the UE shall reconfigure radio bearers according to the RADIO BEARER RECONFIGURATION message.

Reference

3GPP TS 25.331 clause 8.2.2

8.2.2.34<u>24</u>.3 Test purpose

To confirm that the UE transmits RADIO BEARER RECONFIGURATION COMPLETE before entering URA_PCH state after it received a RADIO BEARER RECONFIGURATION message and reconfigured its radio bearers. The UE is in URA_PCH state in the same cell.

8.2.2.3424.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH FACH(state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a RADIO BEARER RECONFIGURATION message. The UE transmits RADIO BEARER RECONFIGURATION COMPLETE message to the UE using AM RLC and enters into URA_PCH state. The SS transmits a PAGING TYPE 1 message and the UE accepts it and enters the CELL_FACH state again.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	←	RADIO BEARER	
		RECONFIGURATION	
2	\rightarrow	RADIO BEARER	The UE sends this message
		RECONFIGURATION COMPLETE	before state transition.
3			Reconfiguration of Radio Bearer
			after state transition.
4	+	PAGING TYPE 1	The SS transmits this message
			included a matched identity.
5	\rightarrow	CELL UPDATE	The UE is in CELL_FACH state.

Specific Message Contents

RADIO BEARER RECONFIGURATION (Step 1)

Use the same message sub-type titled "Packet to CELL_FACH from CELL_FACH in PS" in Annex A with following exceptions:

Information Element	Value/remark	
DRX indicatorRRC State Indicator	DRX with URA updatingURA_PCH	

PAGING TYPE 1 (Step 4)

Use the same message sub-type titled "TM (SMS in PS)" in Annex A with following exceptions:

Information Element	Value/remark
Paging record list	
Paging record	
- CHOICE Paging originator	UTRAN originator
- U-RNTI	
- SRNC Identity	Previously assigned SRNC identity
- S-RNTI	Previously assigned S-RNTI

8.2.2.3424.5 Test requirement

After step 1 the UE transmits RADIO BEARER RECONFIGURATION COMPLETE message to the UE on uplink DCCH using AM RLC.

After step 3 the UE shall transit from CELL_FACH to URA_PCH.

8.2.3 Radio Bearer Release

8.2.3.1 Radio Bearer Release for transition from CELL_DCH to CELL_DCH: Success

8.2.3.1.1 Definition

8.2.3.1.2 Conformance requirement

The UE shall correctly release a radio bearer according to a RADIO BEARER RELEASE message.

Reference

3GPP TS 25.331 clause 8.2.3

8.2.3.1.3 Test purpose

To confirm that the UE release the existing radio bearer according to a RADIO BEARER RELEASE message received from the SS.

8.2.3.1.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CS-DCCH+DTCH_DCH (state 6-9) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a RADIO BEARER RELEASE message to the UE. The UE release the radio bearer and transmits a RADIO BEARER RELEASE COMPLETE message on the uplink DCCH using AM RLC.

Expected sequence

Step Direction		ction	Message	Comment
	UE	SS		
1	•	(RADIO BEARER RELEASE	
2				Release the radio bearer
3	-)	RADIO BEARER RELEASE COMPLETE	

Specific Message Contents

RADIO BEARER RELEASE

The contents of RADIO BEARER RELEASE message are indicated as "Speech in CS" found in default message content clause 9 of TS 34.108.

8.2.3.1.5 Test requirement

After step 1 the UE shall release its radio bearers.

After step 3 the UE shall stop communicating on the released radio bearers, no uplink transmission shall be observed originating from the released link. The remaining radio bearers shall continue to be operational.

8.2.3.2 Radio Bearer Release for transition from CELL_DCH to CELL_DCH: Failure (Unsupported configuration)

8.2.3.2.1 Definition

8.2.3.2.2 Conformance requirement

The UE shall keep its old configuration when the UE receives a RADIO BEARER RELEASE message which includes unsupported configuration parameters and transmit a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC, setting value "configuration unsupported" in IE "failure cause".

Reference

3GPP TS 25.331 clause 8.2.3

8.2.3.2.3 Test purpose

To confirm that the UE keeps its current configuration and transmits a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC, if the received RADIO BEARER RELEASE message indicates an unsupported configuration parameters for the UE.

8.2.3.2.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CS-DCCH+DTCH DCH (state 6-9) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a RADIO BEARER RELAESE message to the UE specifying a frequency which is not supported by the UE. The UE transmits a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC indicating "configuration unsupported" in IE "failure cause".

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1			RADIO BEARER RELEASE	Including unsupported configuration by the UE
2		\rightarrow	RADIO BEARER RELAESE FAILURE	The UE does not change the radio bearer.

Specific Message Contents

RADIO BEARER RELEASE

The contents of RADIO BEARER RELEASE message in this test case is identical as "Speech in CS" found in default message content clause 9 of TS 34.108 with the following exceptions:

Information Element	Value/remark
Frequency info	
- UARFCN uplink(Nu)	63984
- UARFCN downlink(Nd)	Not Present

RADIO BEARER RELEASE FAILURE

Information Element	Value/remark
Message Type	
Failure cause	Configuration unsupported
Other information element	Not checked

8.2.3.2.5 Test requirement

After step 1 the UE shall keep its old configuration and transmit a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC with the IE "failure cause" set to "configuration unsupported". The UE shall able to continue receiving and sending user data.

8.2.3.3 Radio Bearer Release for transition from CELL_DCH to CELL_DCH: Failure (Physical channel failure and reversion to old configuration)

8.2.3.3.1 Definition

8.2.3.3.2 Conformance requirement

The UE shall revert to the old configuration when the UE fails to configure the new radio bearer by timer T312 expiry and transmit a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC which is set to "physical channel failure" in IE "failure cause".

Reference

3GPP TS 25.331 clause 8.2.3

8.2.3.3.3 Test purpose

To confirm that the UE reverts to the old configuration and transmits a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC if the UE fails to release the radio bearer according to a RADIO BEARER RELEASE message by timer T312 expiry.

8.2.3.3.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CS-DCCH+DTCH_DCH (state 6-9) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a RADIO BEARER RELEASE message but it does not configure L1 correspondingly. This causes the UE to fail to release the radio bearer, and after T312 expiry the UE reverts to the old configuration. The UE then transmits a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC which specifies "physical channel failure" in IE "failure cause".

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	+	RADIO BEARER RELEASE	
2			The SS does not configure L1 to reflect the release of the indicated bearer.
3	→	RADIO BEARER RELEASE FAILURE	After T312 expiry, the UE finds that it fails to release a radio bearer and reverts to the old configuration.

Specific Message Contents

RADIO BEARER RELEASE

The contents of RADIO BEARER RELEASE message in this test case is identical as "Speech in CS" found in default message content clause 9 of TS 34.108

RADIO BEARER RELEASE FAILURE

Information Element	Value/remark
Message Type	
Failure cause	Physical channel failure
Other information element	Not checked

8.2.3.3.5 Test requirement

After step 2 the UE shall revert to the old configuration and transmit a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC which includes the value "physical channel failure" in IE "failure cause".

8.2.3.4 Radio Bearer Release for transition from CELL_DCH to CELL_DCH: Failure(Physical channel failure and reversion failure)

8.2.3.4.1 Definition

8.2.3.4.2 Conformance requirement

The UE shall perform a <u>cell update n RRC connection re establishment</u>-procedure when the UE fails to revert to the old configuration after the detection of physical channel failure in the radio bearer release procedure. <u>After the UE completes cell update procedure</u>, the UE transmits RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC which set IE "failure cause" to "physical channel failure".

Reference

3GPP TS 25.331 clause 8.2.3

8.2.3.4.3 Test purpose

To confirm that the UE <u>transmits RADIO BEARER RELEASE FAILURE message after completes initiates an cell update RRC connection re-establishment procedure when the UE cannot revert to the old configuration after encountering a physical channel failure during the execution of a radio bearer release procedure.</u>

8.2.3.4.4 Method of test

Initial Condition

System Simulator: 2 cells - Cell 1 is active, Cell.2 is inactive

UE: CS-DCCH+DTCH_DCH (state 6-9) as specified in clause 7.4 of TS 34.108 in cell 1

Test Procedure

The UE is in the CELL_DCH state in cell 1. The SS begins to broadcast the BCCH in cell 2 with a lower power level than for cell 1. At the same time, it transmits a RADIO BEARER RELAESE message to the UE but does not configure L1. As a result, the UE fails to release the radio bearer properly and tries to revert to the old configuration—after T312 expiry. But the SS deletes the old radio bearer so the UE shall find the availability of cell 2 and transmits CELL UPDATE message on uplink CCCH with IE "Cell update cause" set to "radio link failure". The SS shall transmit CELL UPDATE CONFIRM message on downlink CCCH after receiving CELL UPDATE message. The UE transmits UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC and subsequently transmits RADIO RELEASE FAILURE message on the DCCH using AM RLC, setting IE "failure cause" to "physical channel failure". to it an RRC CONNECTION RE-ESTABLISHMENT REQUEST message, which includes the IE "URNTI" and sent on the uplink CCCH. The SS transmits an RRC CONNECTION RE-ESTABLISHMENT message, which includes a new TFCS according to the new transport channel. After the UE receives an RRC CONNECTION RE-ESTABLISHMENT message, it reconfigures the new radio connection and transmits an RRC CONNECTION RE-ESTABLISHMENT message on the uplink DCCH using AM RLC.

Expected sequence

Step	Dire	ction	Message	Comment
	UE	SS]	
1	•	(BCCH	The SS starts to transmit the BCCH for cell 2 but at a lower power level than cell 1.
2	•	(RADIO BEARER RELEASE	
3				The SS does not configure L1 and deletes the old configuration. This is expected to cause the UE to fail in the release the radio bearer and can not revert to the old configuration.
4	-	→	CELL UPDATERRC CONNECTION RE-ESTABLISHMENT REQUEST	The UE shall find cell 2 and enter CELL FACH state. This message include the value "radio link failure" set in IE "Cell update cause". then transmit this message which includes the IE "U-RNTI" set to its assigned connected mode identity.
5	•	(-	CELL UPDATE CONFIRMRRC CONNECTION RE-ESTABLISHMENT	This message include IE "new U-RNTI" and IE "new C-RNTI". Including the new TFCS according to the new transport channel.
6	-)	UTRAN MOBILITY INFORMATION CONFIRMREC CONNECTION RE-ESTABLISHMENT COMPLETE	
7	=	<u> ></u>	RADIO BEARER RELEASE FAILURE	The IE "failure cause" shall be set to "physical channel failure"

Specific Message Contents

RADIO BEARER RELEASE

The contents of RADIO BEARER RELEASE message in this test case are identical as "Speech in CS" found in default message content clause 9 of TS 34.108.

CELL UPDATE (Step 4)

The contents of CELL UPDATE message is identical as "Contents of CELL UPDATE message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	
- SRNC Identity	Assigned previously in cell 1
<u>- S-RNTI</u>	Assigned previously in cell 1
Cell Update Cause	<u>"radio link failure"</u>

CELL UPDATE CONFIRM (Step 5)

The contents of CELL UPDATE CONFIRM message is identical as "CELL UPDATE CONFIRM message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	Same as CELL UPDATE message in step 7
New U-RNTI	
- SRNC Identity	<u>'0000 0000 0000 0001'</u>
- S-RNTI	Different from previous S-RNTI
New C-RNTI	Different from previous C-RNTI

RADIO BEARER RELEASE FAILURE (Step 7)

Information Element	<u>Value/remark</u>	
Message Type	"RADIO BEARER RELEASE FAILURE"	
Failure cause	"physical channel failure"	
Other information element	Not checked	

8.2.3.4.5 Test requirement

After step 3 the UE shall find the presence of cell 2, enter CELL_FACH state, and transmits <u>CELL UPDATE RRC CONNECTION RE ESTABLISHMENT REQUEST</u> message on the uplink CCCH-of cell 2.

After step 5 the UE shall transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

After step 6 the UE shall transmit RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC, setting the IE "failure cause" to "physical channel failure".

After step 6 the UE shall successfully re establish an RRC connection, after sending RRC CONNECTION RE-RESTABLISHMENT COMPLETE message.

8.2.3.5 Radio Bearer Release for transition from CELL_DCH to CELL_DCH: Failure (Incompatible simultaneous reconfiguration)

8.2.3.5.1 Definition

8.2.3.5.2 Conformance requirement

If the UE receives a RADIO BEARER RELAESE message whilst reconfiguring due to a radio bearer message other than RADIO RELEASE SETUP, it shall keep its configuration as if the RADIO BEARER SETUP message had not been received. If the UE is configuring itself after receiving a message and when the UE subsequently receive a RADIO BEARER RELEASE messagethe

Reference

3GPP TS 25.331 clause 8.2.3

8.2.3.5.3 Test purpose

To confirm that if the UE receives a RADIO BEARER RELEASE message whilst reconfiguring due to a radio bearer message other than RADIO BEARER RELEASE, it shall keep its configuration as if the RADIO BEARER RELEASE message had not been received and complete the reconfiguration according to the previously received message. To confirm that the UE keeps its configuration when the UE receives a RADIO BEARER RELEASE message before the UE releases the radio bearer according to the earlier message.

8.2.3.5.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH DCH (state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL DCH state. The SS transmits a RADIO BEARER SETUP message to the UE. The SS transmits a RADIO BEARER SETUP message before the "activation time" indicated in the RADIO BEARER SETUP message expires. When the UE receives the RADIO BEARER SETUP message, the UE shall keep the configuration as if it had not received the RADIO BEARER RELEASE message and shall transmit a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC with IE "failure cause" set to "incompatible simultaneous reconfiguration". After the SS receives the RADIO BEARER RELEASE FAILURE message, the UE reconfigures the new physical channel parameters and transmits a RADIO BEARER SETUP COMPLETE message on DCCH using AM RLC.

The UE is in the CELL_DCH state. SS transmits a RADIO BEARER RELEASE message on the downlink DCCH, requesting the UE to release all radio bearers mapped to dedicated transport channels. Next, the SS transmits another RADIO BEARER RELEASE message to the UE before the UE is able to release the radio bearers requested in the earlier message. The UE shall keep the old configuration. It shall transmit a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC which is set to "incompatible simultaneous reconfiguration" in IE "failure cause".

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	←	RADIO BEARER SETUP	
2	+	RADIO BEARER RELEASE	Message sent before the "Activation time" indicated in the message of step 1 has elapsed.
3	→	RADIO BEARER RELEASE FAILURE	The UE does not change the configuration due to the reception of RADIO BEARER RELEASE message.
<u>4</u>	<u></u>	RADIO BEARER SETUP COMPLETE	This message is on DCCH using AM RLC.

Specific Message Contents

RADIO BEARER SETUP (Step 1)

The contents of RADIO SETUP RELEASE message in this test case is identical to the message sub-type title "Packet to CELL_DCH from CELL_DCH in PS" found in Annex A, with the following exceptions:

Information Element	Value/remark	
Activation Time	Current CFN-[current CFN mod 8 + 8]	

RADIO BEARER RELEASE (Step 2)

The contents of RADIO BEARER RELEASE message in this test case is identical to the message sub-type title "Packet to CELL_DCH from CELL_DCH in PS" found in Annex A. Information element(s) to be changed are listed below:

RADIO BEARER RELEASE FAILURE

Information Element	Value/remark	
Message Type		
Failure cause	Incompatible simultaneous reconfiguration	
Other information element	Not checked	

8.2.3.5.5 Test requirement

After step 1, SS transmits a RADIO BEARER RELEASE message before the expiry of the activation time specified in the message of step 1.

After step 2 the UE shall keep its configuration as if the UE had not received the RADIO BEARER RELEASE message and shall transmit a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC with IE "failure cause" set to "Incompatible simultaneous reconfiguration".

After step 4 the UE communicates with the SS on the DCCH and DTCH using the new physical channel parameters reconfigured as a result of the RADIO BEARER SETUP message. After step 2 the UE shall keep its old configuration and transmit a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC which is set to "incompatible simultaneous reconfiguration" in IE "failure cause".

8.2.3.6 Radio Bearer Release for transition from CELL_DCH to CELL_DCH: Failure (Invalid message reception and Invalid configuration)

8.2.3.6.1 Definition

8.2.3.6.2 Conformance requirement

The UE shall keep its old configuration when the UE receives a RADIO BEARER RELEASE message which includes the spare undefined value in the mandatory IE "UTRAN DRX cycle length coefficient DRX indicator". It shall transmit a RADIO BEARER RELEASE FAILURE message which contains value "protocol error" in IE "failure cause" and value "Information element value not comprehended" in IE "Protocol error cause". The UE shall keep existing configuration before reception of a RADIO BEARER RELEASE message when the RADIO BEARER RELEASE message include some IEs set to invalid value, and then the UE shall transmit RADIO BEARER RELEASE FAILURE including IE "failure cause" set to "invalid configuration".

Reference

3GPP TS 25.331 clause 8.2.3

8.2.3.6.3 Test purpose

To confirm that the UE transmits a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC if it receives a RADIO BEARER RELEASE message, which uses a <u>spare-undefined</u> value in the mandatory IE "<u>UTRAN</u> <u>DRX cycle length coefficient-DRX indicator</u>".

To confirm that the UE transmits a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC if it receives a RADIO BEARER RELEASE message including some IEs set to invalid value.

8.2.3.6.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CS-DCCH+DTCH_DCH (state 6-9) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a RADIO BEARER RELAESE message to the UE which includes the spare undefined value in the mandatory IE "UTRAN DRX cycle length coefficient DRX indicator". The UE keeps the old configuration and transmits a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC. This message shall indicate "protocol error" in IE "failure cause" and also "Information element value not comprehended" in IE "Protocol error cause". The UE keeps initial configuration and SS transmits RADIO BEARER RELEASE message including some IEs set to invalid value. The UE transmits RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC, setting IE "failure cause" to "invalid configuration".

Expected sequence

Step	Direction	on	Message	Comment
	UE S	SS		
1	+		RADIO BEARER RELEASE	
2	\rightarrow		RADIO BEARER RELEASE FAILURE	The UE shall not change the configuration.
<u>3</u>	<u></u>		RADIO BEARER RELEASE	This message includes IE set to invalid value
4				The UE does not change the configuration
<u>5</u>	<u></u>		RADIO BEARER RELEASE FAILURE	The IE "failure cause" shall be set to "invalid configuration

Specific Message Contents

RADIO BEARER RELEASE (Step1)

The contents of RADIO BEARER RELEASE message in this test case is identical as "Speech in CS" found in default message content clause 9 of TS 34.108 with the following exceptions:

Information Element	Value/remark	
UTRAN DRX cycle length coefficientDRX indicator	Spare-Undefined value	

RADIO BEARER RELEASE FAILURE (Step 2)

Information Element	Value/remark	
Message Type		
Failure cause		
- Failure cause	Protocol error	
- Protocol error information		
- Protocol error cause	Information element value not comprehended	
Other information element	Not checked	

RADIO BEARER RELEASE (Step 3)

The contents of RADIO BEARER RELEASE message in this test case is identical as "Speech in CS" found in default message content clause 9 of TS 34.108 with the following exceptions:

Added or Reconfigured UL TrCH information	
- Transport channel identity	1
- TFS	_
- Dynamic Transport format information	(This IE is repeated for TFI number)
- RLC size	Reference to TS34.108 clause 6.10 Parameter Set
- CHOICE Logical Channel list	
- Explicit List	
- RB identity	<u>2</u>

RADIO BEARER RELEASE FAILURE (Step 5)

Information Element	<u>Value/remark</u>
Message Type	
Failure cause	Invalid configuration
Other information element	Not checked

8.2.3.6.5 Test requirement

After step 1 the UE shall keep its old configuration and transmit a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC, which is set to "protocol error" in IE "failure cause" and is set to "Information element value not comprehended" in IE "Protocol error cause".

After step 3 the UE shall keep its old configuration.

After step 4 the UE shall transmit RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC, setting the value "invalid configuration" to IE "failure cause".

8.2.3.7 Radio Bearer Release for transition from CELL_DCH to CELL_FACH: Success

8.2.3.7.1 Definition

8.2.3.7.2 Conformance requirement

The UE shall correctly release a radio bearer according to a RADIO BEARER RELEASE message, when the common physical channel are requested to be used for the remaining radio bearers.

Reference

3GPP TS 25.331 clause 8.2.3

8.2.3.7.3 Test purpose

To confirm that the UE release the existing the radio bearer according to a RADIO BEARER RELEASE message received from the SS.

8.2.3.7.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DTCH+DCCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a RADIO BEARER RELEASE message to the UE. The UE release the radio bearer and transmits a RADIO BEARER RELEASE COMPLETE message on the uplink DCCH using AM RLC.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	+	RADIO BEARER RELEASE	SS releases the radio bearer in the fashion specified in the message and allocate common channel resources to carry the remaining radio bearers.
2			The UE select PRACH and S-CCPCH using SIB5 and SIB6 after entering CELL FACH state. The UE shall release radio bearers on dedicated transport channels, and reconfigure the remaining radio bearers using the given-selected common control channel.
3	→	RADIO BEARER RELEASE COMPLETE	UE shall be able to continue communication over the remaining radio bearers using the allocated common control channels.

1

Specific Message Contents

RADIO BEARER RELEASE (Step 1)

Use the same message sub-type titled "Packet to CELL_FACH from CELL_DCH in PS" in Annex A.

8.2.3.7.5 Test requirement

After step 3 the UE shall release the specified radio bearer(s) and cease any further uplink transmission from these radio bearer(s). At the same time, it shall resume the transmission or reception of data for the previously suspended radio bearers.

8.2.3.8 Radio Bearer Release for transition from CELL_DCH to CELL_FACH: Failure (Cell re-selectionPhysical channel failure)

8.2.3.8.1 Definition

8.2.3.8.2 Conformance requirement

The UE shall initiate the cell update procedure when the UE cannot use the assigned physical channel according to a RADIO BEARER RELEASE message as the transition from CELL_DCH to CELL_FACH. After the UE completes cell update procedure, the UE transmits RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC which set IE "failure cause" to "cell reselection".

Reference

3GPP TS 25.331 clause 8.2.3

8.2.3.8.3 Test purpose

To confirm that the UE transmits RADIO BEARER RELEASE FAILURE message after the UE perform completes a cell update procedure when the UE cannot use the assigned physical channel according to a RADIO BEARER RELEASE message as the transition from CELL DCH to CELL FACH.

8.2.3.8.4 Method of test

Initial Condition

System Simulator: 2 cells. No.1 is active, No.2 is inactive.

UE: PS-DCCH+DTCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state in cell No.1. The SS begins to broadcast the BCCH in cell No.2 and transmits a RADIO BEARER RELEASE message as the transition from CELL_DCH to CELL_FACH. The UE cannot use the assigned physical channel as the SS does not transmit any data on the DL common channel in cell No.1. Then the UE reselects cell 2 and initiates the cell update procedure in cell No.2. The UE transmits CELL UPDATE message on uplink CCCH with IE "Cell update cause" set to "cell reselection". The SS shall transmit CELL UPDATE CONFIRM message on downlink CCCH after receiving CELL UPDATE message. The UE transmits UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC and transmits RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC, setting IE "failure cause" to "cell reselection".

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1	←		ВССН	The SS broadcasts BCCH in cell No.2.
2	+		RADIO BEARER RELEASE	Assigned the transition from CELL_DCH to CELL_FACH
3				The UE cannot use the assigned the physical channel.
4		>	CELL UPDATE	The UE finds a new cell □No.2 □ and begins a cell update procedure. The value "cell reselection" shall be set in IE "cell update cause".
5	+	<u>-</u>	CELL UPDATE CONFIRM	This message include IE "new U-RNTI" and IE "new C-RNTI". In the CELL_FACH state
<u>6</u>	_	<u>></u>	UTRAN MOBILITY INFORMATION CONFIRM	
7	_	<u>></u>	RADIO BEARER RELEASE FAILURE	The IE "failure cause" shall be set to "cell reselection"

Specific Message Contents

RADIO BEARER RELEASE

Use the same message sub-type titled "Packet to CELL_FACH from CELL_DCH in PS" in Annex A.

CELL UPDATE (Step 4)

The contents of CELL UPDATE message is identical as "Contents of CELL UPDATE message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	
- SRNC Identity	Assigned previously in cell 1
<u>- S-RNTI</u>	Assigned previously in cell 1
Cell Update Cause	"radio link failure"

CELL UPDATE CONFIRM (Step 5)

The contents of CELL UPDATE CONFIRM message is identical as "CELL UPDATE CONFIRM message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	Same as CELL UPDATE message in step 7
New U-RNTI	
- SRNC Identity	<u>'0000 0000 0000 0001'</u>
<u>- S-RNTI</u>	Different from previous S-RNTI
New C-RNTI	Different from previous C-RNTI

RADIO BEARER RELEAS FAILURE (Step 6)

Information Element	<u>Value/remark</u>
Message Type	"RADIO BEARER RECONFIGURATION FAILURE"
Failure cause	<u>"cell reselection"</u>
Other information element	Not checked

8.2.3.8.5 Test requirement

After step 3 the UE shall find a new cell No.2 and enter to CELL_FACH state. Then the UE shall initiate a cell update procedure in cell No.2.

After step 4 UE shall transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

After step 5 UE shall transmit RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC, setting the value "cell reselection" to IE "failure cause".

8.2.3.9 Radio Bearer Release for transition from CELL_FACH to CELL_DCH: Success

8.2.3.9.1 Definition

8.2.3.9.2 Conformance requirement

The UE shall correctly release a radio bearer according to a RADIO BEARER RELEASE message.

Reference

3GPP TS 25.331 clause 8.2.3

8.2.3.9.3 Test purpose

To confirm that an UE, in state CELL_FACH, releases the radio access bearers on RACH and FACH transport channels. After the release, it shall access the affected radio bearers on the newly allocated DCH transport channel.

8.2.3.9.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_FACH (state 6-1011) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a RADIO BEARER RELEASE message to the UE. In this message, SS commands the UE to release radio bearers on RACH and FACH. At the same time, SS allocates DCH to support the affected radio bearers. The UE shall release the indicated radio bearer and transmits a RADIO BEARER RELEASE COMPLETE message on the uplink DCCH using AM RLC.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	-	RADIO BEARER RELEASE	
2			UE shall release the radio access bearers carried by RACH and FACH transport channels.
3	\rightarrow	RADIO BEARER RELEASE COMPLETE	

Specific Message Contents

RADIO BEARER RELEASE (Step 1)

Use the same message sub-type titled "Packet to CELL_DCH from CELL_FACH in PS" in Annex A.

8.2.3.9.5 Test requirement

After step3 the UE shall stop communicating on the released radio bearers, and resume all <u>suspended stopped radio</u> bearer using the dedicated physical channel allocated.

8.2.3.10 Radio Bearer Release for transition from CELL_FACH to CELL_DCH: Failure (Unsupported configuration)

8.2.3.10.1 Definition

8.2.3.10.2 Conformance requirement

The UE shall keep its old configuration when it receives a RADIO BEARER RELEASE message which specifies unsupported configuration parameters for the UE. Then the UE shall transmit a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC which, setting value "configuration unsupported" in IE "failure cause".

Reference

3GPP TS 25.331 clause 8.2.3

8.2.3.10.3 Test purpose

To confirm that the UE keeps its configuration and transmits a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC if the received RADIO BEARER RELEASE message requests for unsupported configuration parameters for the UE.

8.2.3.10.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a RADIO BEARER RELAESE message to the UE, referring to a frequency which cannot be supported by the UE. The UE shall transmit a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC and set "configuration unsupported" in IE "failure cause".

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1		←	RADIO BEARER RELEASE	The message contains a configuration not supported by the UE
2		→	RADIO BEARER RELAESE FAILURE	The UE shall not change the radio bearer configuration.

Specific Message Contents

RADIO BEARER RELEASE

The contents of RADIO BEARER RELEASE message in this test case is identical to the message sub-type title "Packet to CELL_DCH from CELL_FACH in PS" found in Annex A with the following exceptions:

Information Element	Value/remark
Frequency info	
- UARFCN uplink(Nu)	63984
- UARFCN downlink(Nd)	Not Present

RADIO BEARER RELEASE FAILURE

Information Element	Value/remark
Message Type	
Failure cause	Configuration unsupported
Other information element	Not checked

8.2.3.10.5 Test requirement

After step 2 the UE shall keep its old configuration and transmit a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC, stating the reason "configuration unsupported" in IE "failure cause".

8.2.3.11 Radio Bearer Release for transition from CELL_FACH to CELL_DCH: Failure (Physical channel failure and reversion to old configuration)

8.2.3.11.1 Definition

8.2.3.11.2 Conformance requirement

The UE shall revert to the old configuration when the UE fails to configure the new radio bearer before T312 timer expiry. Then it shall transmit a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC, setting the value "physical channel failure" in IE "failure cause".

Reference

3GPP TS 25.331 clause 8.2.3

8.2.3.11.3 Test purpose

To confirm that the UE reverts to the old configuration and transmits a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC if the UE fails to release the radio bearer in accordance the specified settings in RADIO BEARER RELEASE message by T312 timer expiry.

8.2.3.11.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a RADIO BEARER RELEASE message and does not configure L1. The UE is expected to encounter a failure while releasing the radio bearer. After T312 timer expiry, the UE shall revert to the old radio bearer configuration, so the UE transmits a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC which is set to "physical channel failure" in IE "failure cause".

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1	+		RADIO BEARER RELEASE	
2				The SS does not configure L1.
3	•	→	RADIO BEARER RELEASE FAILURE	After T312 expiry the UE fails to release a radio bearer and reverts to the old configuration.

Specific Message Contents

RADIO BEARER RELEASE

Use the message sub-type titled "Packet to CELL_DCH from CELL_FACH in PS" in Annex A.

RADIO BEARER RELEASE FAILURE

Information Element	Value/remark
Message Type	
Failure cause	Physical channel failure
Other information element	Not checked

8.2.3.11.5 Test requirement

After step 2 the UE shall revert to the old configuration and transmit a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC, setting the value "physical channel failure" in IE "failure cause".

8.2.3.12 Radio Bearer Release for transition from CELL_FACH to CELL_DCH: Failure (Physical channel failure and reversion failure)

8.2.3.12.1 Definition

8.2.3.12.2 Conformance requirement

The UE shall perform an RRC connection re establishmentcell update procedure when the UE fails to revert to the old configuration after the detection of physical channel failure in the radio bearer release procedure. After the UE completes cell update procedure, the UE transmits RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC which set IE "failure cause" to "physical channel failure".

Reference

3GPP TS 25.331 clause 8.2.3

8.2.3.12.3 Test purpose

To confirm that the UE <u>transmits RADIO BEARER RELEASE FAILURE message after it completes initiates an cell update RRC connection re establishment procedure when the UE cannot revert to the old configuration, following a physical channel failure during the radio bearer release.</u>

8.2.3.12.4 Method of test

Initial Condition

System Simulator: 2 cells - Cell.1 is active, Cell 2 is inactive

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108 in cell No.1

Test Procedure

The UE is in the CELL_FACH state in cell 1. The SS begins to broadcast the BCCH in cell 2 and then transmits a RADIO BEARER RELAESE message to the UE, but it does not configure L1. This is expected to cause the UE to experience a failure to release the radio bearer and it subsequently tries to revert to the old configuration after T312 expiry. The SS deletes the old radio bearer, so the UE shall find cell 2 and transmits CELL UPDATE message on uplink CCCH with IE "Cell update cause" set to "radio link failure". The SS shall transmit CELL UPDATE CONFIRM message on downlink CCCH after receiving CELL UPDATE message. The UE transmits UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC and subsequently transmits RADIO RELEASE FAILURE message on the DCCH using AM RLC, setting IE "failure cause" to "physical channel failure". to it an RRC CONNECTION RE ESTABLISHMENT REQUEST message, which includes the IE "U RNTI" on the uplink CCCH. After receiving this message, the SS transmits an RRC CONNECTION RE ESTABLISHMENT message which includes a new TFCS according to the new transport channel. After the UE receives an RRC CONNECTION RE ESTABLISHMENT message, it reconfigures the new radio connection and transmits an RRC CONNECTION RE ESTABLISHMENT message on the uplink DCCH using AM RLC.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1	+		ВССН	The SS starts to transmit the BCCH in a cell 2.
2	+		RADIO BEARER RELEASE	
3				The SS does not configure L1 and deletes the old configuration so the UE cannot release the radio bearer and also fails revert to the old configuration.
4)	•	CELL UPDATERRC CONNECTION RE-ESTABLISHMENT REQUEST	The UE finds a new cell 2 and enter CELL_FACH state. This message include the value "radio link failure" set in IE "Cell update cause".and then transmits this message which includes the IE "U-RNTI" it has been assigned to.
5	+		CELL UPDATE CONFIRMRRC CONNECTION RE-ESTABLISHMENT	This message include IE "new U-RNTI" and IE "new C-RNTI". Including the new TFCS according to the new transport channel.
6	→	•	UTRAN MOBILITY INFORMATION CONFIRMREC CONNECTION RE-ESTABLISHMENT COMPLETE	
<u>7</u>	<u> </u>		RADIO BEARER RELEASE FAILURE	The IE "failure cause" shall be set to "physical channel failure"

Specific Message Contents

RADIO BEARER RELEASE

Use the message sub-type titled "Packet to CELL_DCH from CELL_FACH in PS" in Annex A.

CELL UPDATE (Step 4)

The contents of CELL UPDATE message is identical as "Contents of CELL UPDATE message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	
- SRNC Identity	Assigned previously in cell 1
<u>- S-RNTI</u>	Assigned previously in cell 1
Cell Update Cause	"radio link failure"

CELL UPDATE CONFIRM (Step 5)

The contents of CELL UPDATE CONFIRM message is identical as "CELL UPDATE CONFIRM message" as found in Annex with the following exceptions:

Information Element	Value/remark
<u>U-RNTI</u>	Same as CELL UPDATE message in step 7
New U-RNTI	
- SRNC Identity	<u>'0000 0000 0000 0001'</u>
- S-RNTI	Different from previous S-RNTI
New C-RNTI	Different from previous C-RNTI

RADIO BEARER RELEASE FAILURE (Step 7)

Information Element	<u>Value/remark</u>	
Message Type	"RADIO BEARER RELEASE FAILURE"	
Failure cause	"physical channel failure"	
Other information element	Not checked	

8.2.3.12.5 Test requirement

After step 3 the UE shall find the presence of cell 2, enter CELL_FACH state, and transmit a <u>CELL UPDATE message RRC CONNECTION RE ESTABLISHMENT REQUEST</u> on the uplink CCCH.

After step 6-5 the UE shall transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

After step 6 the UE shall transmit RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC, setting the IE "failure cause" to "physical channel failure".

the UE shall successfully re establish an RRC connection and access the dedicated physical channels assigned.

8.2.3.13 Radio Bearer Release for transition from CELL_FACH to CELL_DCH: Failure (Incompatible simultaneous reconfiguration)

8.2.3.13.1 Definition

8.2.3.13.2 Conformance requirement

If the UE receives a RADIO BEARER RELEASE message whilst reconfiguring due to a radio bearer message other than RADIO BEARER RELEASE, it shall keep its configuration as if the RADIO BEARER RELEASE message had not been received. If the UE is configuring itself after receiving a radio bearer message other than RADIO BEARER RELEASE message, and when the UE subsequently receive a RADIO BEARER RELEASE message the UE shall transmit a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC which is set to "Incompatible simultaneous reconfiguration" in IE "failure cause".

Reference

3GPP TS 25.331 clause 8.2.3

8.2.3.13.3 Test purpose

To confirm that if the UE receives a RADIO BEARER RELEASE message whilst reconfiguring due to a radio bearer message other than RADIO BEARER RELEASE, it shall keep its configuration as if the RADIO BEARER RELEASE message had not been received and complete the reconfiguration according to the previously received message. To

confirm that the UE keeps its old configuration when it receives a RADIO BEARER RELEASE message before it releases the radio bearer according to an earlier message.

8.2.3.13.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL DCH state. The SS transmits a RADIO BEARER SETUP message to the UE. The SS transmits a RADIO BEARER RELEASE message before the "activation time" indicated in the RADIO BEARER SETUP message expires. When the UE receives the RADIO BEARER SETUP message, the UE shall keep the configuration as if it had not received the RADIO BEARER SETUP message and shall transmit a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC with IE "failure cause" set to "incompatible simultaneous reconfiguration". After the SS receives the RADIO BEARER SETUP FAILURE message, the UE reconfigures the new physical channel parameters and transmits a RADIO BEARER SETUP COMPLETE message on DCCH using AM RLC. The UE is in the CELL_FACH state. When the SS transmits a RADIO BEARER RELEASE message to the UE before the UE completes the release of the radio bearer, it shall keep the old configuration and transmit an RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC. This message shall specify the reason "incompatible simultaneous reconfiguration" in IE "failure cause".

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	+	RADIO BEARER SETUP	The UE receive any message other than RADIO BEARER RELEASE. (e.g. RADIO BEARER SETUP)
2	+	RADIO BEARER SETUP	Sent before the expiry stated in IE "Activation Time" of message in step 1.
3	→	RADIO BEARER RELEASE FAILURE	The UE does not change the configuration due to the reception of RADIO BEARER SETUP message
4	<u>→</u>	RADIO BEARER SETUP COMPLETE	This message is on DCCH using AM RLC.

Specific Message Contents

RADIO BEARER SETUP (Step 1)

The contents of RADIO BEARER SETUP message in this test case is identical to the message sub-type title "Packet to CELL_DCH from CELL_FACH in PS" found in Annex A. Information element(s) to be changed are listed below:

Information Element	Value/remark
Activation Time	Current CFN-[current CFN mod 8 + 8]

RADIO BEARER RELEASE (Step 2)

The contents of RADIO BEARER RELEASE message in this test case is identical to the message sub-type title "Packet to CELL_DCH from CELL_FACH in PS" found in Annex A with the following exceptions:

Information Element	Value/remark	
Activation Time Info	Current CFN-[current CFN mod 8 + 8]	

RADIO BEARER RELEASE FAILURE

Information Element	Value/remark
Message Type	
Failure cause	Incompatible simultaneous reconfiguration
Other information element	Not checked

8.2.3.13.5 Test requirement

After step 1, SS transmits a RADIO BEARER RELEASE message before the expiry of the activation time specified in the message of step 1.

After step 2 the UE shall keep its configuration as if the UE had not received the RADIO BEARER RELEASE message and shall transmit a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC with IE "failure cause" set to "Incompatible simultaneous reconfiguration".

After step 4 the UE communicates with the SS on the DCCH and DTCH using the new physical channel parameters reconfigured as a result of the RADIO SETUP message. After step 2 the UE shall keep its old configuration and transmit an RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC, and set the value "incompatible simultaneous reconfiguration" in IE "failure cause".

8.2.3.14 Radio Bearer Release for transition from CELL_FACH to CELL_DCH: Failure (Invalid message reception and Invalid configuration)

8.2.3.14.1 Definition

8.2.3.14.2 Conformance requirement

The UE shall keep its old configuration when the UE receives a RADIO BEARER RELEASE message which uses a spare undefined value in the mandatory IE "UTRAN DRX cycle length coefficient DRX indicator". It shall transmit a RADIO BEARER RELEASE FAILURE message which indicate the value "protocol error" in IE "failure cause" and setting "Information element value not comprehended" in IE "Protocol error cause". The UE shall keep existing configuration before reception of a RADIO BEARER RELEASE message when the RADIO BEARER RELEASE message include some IEs set to invalid value, and then the UE shall transmit RADIO BEARER RELEASE FAILURE including IE "failure cause" set to "invalid configuration".

Reference

3GPP TS 25.331 clause 8.2.3

8.2.3.14.3 Test purpose

To confirm that the UE transmits RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC if it receives a RADIO BEARER RELEASE message which uses a spare-undefined value in the mandatory IE "<u>UTRAN</u> DRX cycle length coefficient DRX indicator".

To confirm that the UE transmits a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC if it receives a RADIO BEARER RELEASE message including some IEs set to invalid value.

8.2.3.14.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS_DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a RADIO BEARER RELAESE message to the UE containing a spare undefined value in the mandatory IE "UTRAN DRX cycle length coefficient DRX indicator". The UE keeps the old configuration and transmits a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC, which shall indicate the reason "protocol error" in IE "failure cause" and also "Information element value not comprehended" in IE "Protocol error cause". The UE keeps initial configuration and SS transmits RADIO BEARER RELEASE message including some IEs set to invalid value. The UE transmit RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC, setting the value " invalid configuration" to IE "failure cause".

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1	+	-	RADIO BEARER RELEASE	
2		>	RADIO BEARER RELEASE FAILURE	The UE shall not change its current configuration.
3 ←		_	RADIO BEARER RELEASE	This message includes IE set to invalid value
4				The UE does not change the configuration
<u>5</u>	≥		RADIO BEARER RELEASE FAILURE	The IE "failure cause" shall be set to "invalid configuration

Specific Message Contents

RADIO BEARER RELEASE (Step 3)

The contents of RADIO BEARER RELEASE message in this test case is identical to the message sub-type title "Packet to CELL_DCH from CELL_FACH in PS" found in Annex A with the following exceptions:

Information Element	Value/remark
UTRAN DRX Indicator DRX indicator	spare Undefined value

RADIO BEARER RELEASE FAILURE (Step 2)

Information Element	Value/remark	
Message Type		
Failure cause		
- Failure cause	Protocol error	
- Protocol error information		
- Protocol error cause	Information element value not comprehended	
Other information element	Not checked	

RADIO BEARER RELEASE (Step 3)

The contents of RADIO BEARER RELEASE message in this test case is identical to the message sub-type title "Packet to CELL DCH from CELL FACH in PS" found in Annex A with the following exceptions:

Added or Reconfigured UL TrCH information	
- Transport channel identity	<u>1</u>
TFS	
 Dynamic Transport format information 	(This IE is repeated for TFI number)
- RLC size	Reference to TS34.108 clause 6.10 Parameter Set
- CHOICE Logical Channel list	
- Explicit List	
- RB identity	<u>4</u>

RADIO BEARER RELEASE FAILURE (Step 5)

Information Element	<u>Value/remark</u>
Message Type	
Failure cause	Invalid configuration
Other information element	Not checked

8.2.3.14.5 Test requirement

After step_1 the UE shall keep its old configuration and transmit a RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC, setting "protocol error" in IE "failure cause" and also indicating "Information element value not comprehended" in IE "Protocol error cause".

After step 3 the UE shall keep its old configuration.

After step 4 the UE shall transmit RADIO BEARER RELEASE FAILURE message on the DCCH using AM RLC, setting IE "failure cause" to "invalid configuration".

8.2.3.15 Radio Bearer Release for transition from CELL_FACH to CELL_FACH: Success

8.2.3.15.1 Definition

8.2.3.15.2 Conformance requirement

The UE shall correctly release a radio bearer according to the RADIO BEARER RELEASE message received.

Reference

3GPP TS 25.331 clause 8.2.3

8.2.3.15.3 Test purpose

To confirm that the UE release the existing the radio bearer(s) according to the RADIO BEARER RELEASE message received from the SS.

8.2.3.15.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a RADIO BEARER RELEASE message to the UE. The UE release the radio bearer and transmits a RADIO BEARER RELEASE COMPLETE message on the uplink DCCH using AM RLC.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1	+	-	RADIO BEARER RELEASE	
2				The UE select PRACH and S- CCPCH using SIB5 and SIB6. The UE shall release the requested radio bearer(s), and stop transmitting using these radio bearer(s).
3	-	>	RADIO BEARER RELEASE COMPLETE	

Specific Message Contents

RADIO BEARER RELEASE

Use the same message sub-type titled "Packet to CELL_FACH from CELL_FACH in PS" in Annex A.

8.2.3.15.5 Test requirement

After step 1 the UE shall cease the transmission and reception of the affected radio bearers.

After step 3 the UE shall stop communicating on radio bearers to be released.

8.2.3.16 Radio Bearer Release for transition from CELL_DCH to CELL_DCH: Success (Subsequently received)

8.2.3.16.1 Definition

8.2.3.16.2 Conformance requirement

If the UE receives a RADIO BEARER RELEASE message before the UE releases the radio bearer according to the previous RADIO BEARER RELEASE message, the UE shall ignore the new RADIO BEARER RELEASE message and releases according to the previous RADIO BEARER RELEASE message. Finally, the UE shall transmit RADIO BEARER RELEASE COMPLETE message on the DCCH using AM RLC.

Reference

3GPP TS 25.331 clause 8.2.3

8.2.3.16.3 Test purpose

To confirm that if the UE receives a new RADIO BEARER RELEASE message before the UE releases the radio bearer according to a previous RADIO BEARER RELEASE message it ignore the new RADIO BEARER RELEASE message and configures according to the previous RADIO BEARER RELEASE message received.

8.2.3.16.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. When the SS transmits a RADIO BEARER RELEASE message to the UE before the UE releases the radio bearer, the UE ignores the second RADIO BEARER RELEASE message and releases according to the previous RADIO BEARER RELEASE message received. Finally, the UE shall transmit RADIO BEARER RELEASE COMPLETE message on the DCCH using AM RLC.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1		-	RADIO BEARER RELEASE	
2		←	RADIO BEARER RELEASE	Message sent before, the expiry of "activation time" specified in message of step 1.
3	→		RADIO BEARER RELEASE COMPLETE	The UE ignores the RADIO BEARER RELEASE message in step 2 and confirms release according to the RADIO BEARER RELEASE message in step 1.

Specific Message Contents

RADIO BEARER RELEASE (Step 1)

The contents of RADIO BEARER RELEASE message in this test case is identical to the message sub-type title "Packet to CELL_DCH from CELL_DCH in PS" found in Annex A, with the following exceptions:

Information Element	Value/remark
Activation Time	[256+Current CFN-[current CFN mod 8 + 8]]MOD 256

RADIO BEARER RELEASE (Step 2)

The contents of RADIO BEARER RELEASE message in this test case is identical to the message sub-type title "Packet to CELL_DCH from CELL_DCH in PS" found in Annex A.

8.2.3.16.5 Test requirement

After step 2 the UE shall releases the radio bearer specified in the first RADIO BEARER RELEASE message and transmit an RADIO BEARER RELEASE COMPLETE message on the DCCH using AM RLC.

8.2.3.17 Radio Bearer Release for transition from CELL_FACH to CELL_DCH: Success (Subsequently received)

8.2.3.17.1 Definition

8.2.3.17.2 Conformance requirement

If the UE receives a RADIO BEARER RELEASE message before the UE releases the radio bearer according to the previous RADIO BEARER RELEASE message, the UE shall ignore the new RADIO BEARER RELEASE message and releases according to the previous RADIO BEARER RELEASE message. Finally, the UE shall transmit RADIO BEARER RELEASE COMPLETE message on the DCCH using AM RLC.

Reference

3GPP TS 25.331 clause 8.2.3

8.2.3.17.3 Test purpose

To confirm that if the UE receives a new RADIO BEARER RELEASE message before the UE releases the radio bearer according to a previous RADIO BEARER RELEASE message it ignore the new RADIO BEARER RELEASE message and configures according to the previous RADIO BEARER RELEASE message received.

8.2.3.17.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The UE is in the CELL_DCH state. When the SS transmits a RADIO BEARER RELEASE message to the UE before the UE releases the radio bearer, the UE ignores the second RADIO BEARER RELEASE message and releases according to the previous RADIO BEARER RELEASE message received. Finally, the UE shall transmit RADIO BEARER RELEASE COMPLETE message on the DCCH using AM RLC.

Expected sequence

Step	Dire	ction	Message	Comment
	UE	SS		
1		(RADIO BEARER RELEASE	
2		←	RADIO BEARER RELEASE	Sent before the expiry stated in IE "Activation Time" of RADIO BEARER RELEASE message in step 1.
3		→	RADIO BEARER RELEASE COMPLETE	The UE ignores the RADIO BEARER RELEASE message in step 2 and confirms release according to the RADIO BEARER RELEASE message in step 1.

Specific Message Contents

RADIO BEARER RELEASE

The contents of RADIO BEARER RELEASE message in this test case is identical to the message sub-type title "Packet to CELL_DCH from CELL_FACH in PS" found in Annex A with the following exceptions:

(Step 1)

Information Element	Value/remark	
Activation Time Info	[256+Current CFN-[current CFN mod 8 + 8]]MOD 256	

8.2.3.17.5 Test requirement

After step 2 the UE shall releases the radio bearer specified in the first RADIO BEARER RELEASE message and transmit an RADIO BEARER RELEASE COMPLETE message on the DCCH using AM RLC.

8.2.3.2618 Radio Bearer Release from CELL_DCH to CELL_PCH: Success

8.2.3.2618.1 Definition

8.2.3.2618.2 Conformance requirement

The UE shall transmit RADIO BEARER RELEASE COMPLETE message before completes transition from CELL_DCH to CELL_PCH when receives a RADIO BEARER RELEASE message. And then, the UE shall release radio bearers according to the RADIO BEARER Release message.

Reference

3GPP TS 25.331 clause 8.2.2

8.2.3.2618.3 Test purpose

To confirm that the UE transmits RADIO BEARER RELEASE COMPLETE before entering CELL_PCH state after it received a RADIO BEARER RELEASE message and released its radio bearers. The UE is in CELL_PCH state of the same cell.

8.2.3.2618.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a RADIO BEARER RELEASE message. The UE transmits RADIO BEARER RELEASE COMPLETE message to the UE using AM RLC and enters into CELL_PCH state. The SS transmits a PAGING TYPE 1 message and the UE accepts it and enters the CELL_FACH state.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1		+	RADIO BEARER	
			RELEASE	
2		\rightarrow	RADIO BEARER	The UE sends this message
			RELEASE COMPLETE	before completes state
				transition.
3		+	PAGING TYPE 1	The SS transmits this message
				included a matched identity.
4		\rightarrow	CELL UPDATE	The UE is in CELL_FACH state.

Specific Message Contents

RADIO BEARER RELEASE (Step 1)

Use the same message sub-type titled "Packet to CELL_FACH from CELL_DCH in PS" in Annex A with following exceptions:

Information Element	Value/remark
DRX indicatorRRC State Indicator	DRX with CELL updatingCELL PCH

PAGING TYPE 1 (Step 3)

Use the same message sub-type titled "TM (SMS in PS)" in Annex A with following exceptions:

Information Element	Value/remark
Paging record list	
Paging record	
- CHOICE Paging originator	UTRAN originator
- U-RNTI	
- SRNC Identity	Previously assigned SRNC identity
- S-RNTI	Previously assigned S-RNTI

8.2.3.2618.5 Test requirement

After step 1 the UE transmits RADIO BEARER RELEASE COMPLETE message to the UE on uplink DCCH using AM RLC before completes state transition.

8.2.3.2719 Radio Bearer Release from CELL_DCH to URA_PCH: Success

8.2.3.2719.1 Definition

8.2.3.2719.2 Conformance requirement

The UE shall transmit RADIO BEARER RELEASE COMPLETE message before completes transition from CELL_DCH to CELL_PCH when receives a RADIO BEARER RELEASE message. And then, the UE shall release radio bearers according to the RADIO BEARER Release message.

Reference

3GPP TS 25.331 clause 8.2.2

8.2.3.2719.3 Test purpose

To confirm that the UE transmits RADIO BEARER RELEASE COMPLETE before entering CELL_PCH state after it received a RADIO BEARER RELEASE message and released its radio bearers. The UE is in CELL_PCH state of the same cell.

8.2.3.2719.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_DCH(state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a RADIO BEARER RELEASE message. The UE transmit RADIO BEARER RELEASE COMPLETE message to the UE using AM RLC and enters into CELL_PCH state. The SS transmits a PAGING TYPE 1 message and the UE accepts it and enters the CELL_FACH state.

Expected sequence

Step	Direction	Message	Comment
	UE S	8	
1	+	RADIO BEARER RELEASE	
2	\rightarrow	RADIO BEARER RELEASE COMPLETE	The UE sends this message before completes state transition.
3	+	PAGING TYPE 1	The SS transmits this message included a matched identity.
4	\rightarrow	CELL UPDATE	The UE is in CELL_FACH state.

Specific Message Contents

RADIO BEARER RELEASE (Step 1)

Use the same message sub-type titled "Packet to CELL_FACH from CELL_DCH in PS" in Annex A with following exceptions:

Information Element	Value/remark
DRX indicatorRRC State Indicator	DRX with URA updating URA_PCH

PAGING TYPE 1 (Step 3)

Use the same message sub-type titled "TM (SMS in PS)" in Annex A with following exceptions:

Information Element	Value/remark
Paging record list	
Paging record	
- CHOICE Paging originator	UTRAN originator
- U-RNTI	
- SRNC Identity	Previously assigned SRNC identity
- S-RNTI	Previously assigned S-RNTI

8.2.3.2719.5 Test requirement

After step 1 the UE transmits RADIO BEARER RELEASE COMPLETE message to the UE on uplink DCCH using AM RLC before completes state transition.

8.2.4 Transport channel reconfiguration

8.2.4.1 Transport channel reconfiguration from CELL_DCH to CELL_DCH (Hard handover to same radio frequency): Success with no transport channel type switching

8.2.4.1.1 Definition

8.2.4.1.2 Conformance requirement

The UE shall correctly reconfigure a radio bearer according to the TRANSPORT CHANNEL RECONFIGURATION message, which specifies a hard handover to another cell. After the completion of this procedure, the UE shall be able to communicate with the UTRAN on the new transport channel.

Reference

3GPP TS 25.331 clause 8.2.4

8.2.4.1.3 Test purpose

To confirm that the UE reconfigures a new transport channel according to a TRANSPORT CHANNEL RECONFIGURATION message, which also specifies that a hard handover to another cell be performed simultaneously.

8.2.4.1.4 Method of test

Initial Condition

System Simulator: 2 cells – cell 1 and cell 2 are both active

UE: PS-DCCH+DTCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108 in cell 1

Test Procedure

The UE is in the CELL_DCH state in cell 1. The SS transmits a TRANSPORT CHANNEL RECONFIGURATION message to the UE, which includes new transport channel parameters to be applied in cell 2. The UE shall reconfigure the new transport channel and then transmit a TRANSPORT CHANNEL RECONFIGURATION COMPLETE message on the uplink DCCH of cell 2 using AM RLC.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	+	TRANSPORT CHANNEL RECONGURATION	Hard handover to cell 2. Including UE information elements("TFS"I)
2			UE shall suspend stop all uplink transmissions and reconfigure itself to use the new transport channel parameters
3	\rightarrow	TRANSPORT CHANNEL RECONFIGURATION COMPLETE	

Specific Message Contents

TRANSPORT CHANNEL RECONFIGURATION

The contents of TRANSPORT CHANNEL RECONFIGURATION message in this test case is identical to the message sub-type titled "Packet to CELL_DCH from CELL_DCH in PS" in Annex A, with the following exceptions:

Information Element	Value/remark
TrCH Information Elements	
-Uplink transport Channels	
-Added or Reconfigured TrCH information list	Number of Transport blocks = 2
-Downlink transport Channels	
-Added or Reconfigured TrCH information list	
	Number of Transport blocks = 2
Downlink information for each radio links	Same downlink UARFCN as used for cell 2
- Primary CPICH info	
- Primary Scrambling Code	150
Downlink information common for all radio links	
 Downlink DPCH info common for all RL 	
- Timing Indicator	<u>Initialise</u>

8.2.4.1.5 Test requirement

After step 1 the UE shall reconfigure the radio links affected by the changes for uplink and downlink DCH. The UE shall stop transmitting on the uplink of cell 1.

After step 3 the UE shall continue to communicate with the SS on the DCCH of cell 2, using the new Transport Format Set (TFS) applicable on the existing transport channel.

8.2.4.2 Transport channel reconfiguration from CELL_DCH to CELL_DCH: Failure (Unsupported configuration)

8.2.4.2.1 Definition

8.2.4.2.2 Conformance requirement

The UE shall keep its old configuration when the UE receives a TRANSPORT CHANNEL RECONFIGURATION message which includes unsupported configuration parameters and transmit a TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC which is set to "configuration unsupported" in IE "failure cause

Reference

3GPP TS 25.331 clause 8.2.4

8.2.4.2.3 Test purpose

To confirm that the UE transmits a TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC if the received TRANSPORT CHANNEL RECONFIGURATION message specifies unsupported configuration parameters.

8.2.4.2.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a TRANSPORT CHANNEL RECONFIGURATION message to the UE, which includes unsupported configuration parameters of the UE. The UE transmits a TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, reporting the event "configuration unsupported" in IE "failure cause".

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1	•	(TRANSPORT CHANNEL	Including unsupported
			RECONGURATION	configuration by the UE
2	-	>	TRANSPORT CHANNEL	The UE shall not change the
			RECONFIGURATION FAILURE	settings used by the transport
				channel.

Specific Message Contents

TRANSPORT CHANNEL RECONFIGURATION

The contents of TRANSPORT CHANNEL RECONFIGURATION message in this test case is identical to the message sub-type title "Packet to CELL_DCH from CELL_DCH in PS" found in Annex A with the following exceptions:

Information Element	Value/remark
TrCH Information Elements	
-Uplink transport Channels	
-Added or Reconfigured TrCH information list	
	Number of Transport blocks = 4096
-Downlink transport Channels	·
-Added or Reconfigured TrCH information list	Selected value as the UE can not support.
	Number of Transport blocks = 4096

TRANSPORT CHANNEL RECONFIGURATION FAILURE

Information Element	Value/remark
Message Type	"TRANSPORT CHANNEL RECONFIGURATION
	FAILURE"
Failure cause	Configuration unsupported
Other information element	Not checked

8.2.4.2.5 Test requirement

After step 1 the UE shall keep its old configuration and transmit a TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, indicating "configuration unsupported" in IE "failure cause".

8.2.4.3 Transport channel reconfiguration from CELL_DCH to CELL_DCH: Failure (Physical channel failure and reversion to old configuration)

8.2.4.3.1 Definition

8.2.4.3.2 Conformance requirement

The UE shall revert to the old configuration when the UE fails to reconfigure the new physical channel by received TRANSPORT CHANNEL RECONFIGURATION message and transmit a TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC which is set to "physical channel failure" in IE "failure cause".

Reference

3GPP TS 25.331 clause 8.2.4

8.2.4.3.3 Test purpose

To confirm that the UE reverts to the old configuration and transmits a TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, if the UE fails to reconfigure the new transport channel according to a TRANSPORT CHANNEL RECONFIGURATION message.

8.2.4.3.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a TRANSPORT CHANNEL RECONFIGURATION message to the UE, which includes new transport channel parameters but it does not reconfigure the new transport channel. Therefore, the UE cannot reconfigure them and have to revert to the old configuration. Then the UE shall transmit a

TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting "physical channel failure" in IE "failure cause".

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1	•	=	TRANSPORT CHANNEL RECONGURATION	Specifies a change in the TFS of the dedicated transport channel used.
2				The SS does not reconfigure the transport channel, leading to the UE unable to reconfigure the new transport channel.
3		>	TRANSPORT CHANNEL RECONFIGURATION FAILURE	The UE reverts to the old configuration and transmits this message.

Specific Message Contents

TRANSPORT CHANNEL RECONFIGURATION

Use the message sub-type titled "Packet to CELL_DCH from CELL_DCH in PS" in Annex A.

TRANSPORT CHANNEL RECONFIGURATION FAILURE

Information Element	Value/remark
Message Type	"TRANSPORT CHANNEL RECONFIGURATION
	FAILURE"
Failure cause	Physical channel failure
Other information element	Not checked

8.2.4.3.5 Test requirement

After step 2 the UE shall revert to the old configuration and transmit a TRANPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, and it shall set the value "physical channel failure" in IE "failure cause".

8.2.4.4 Transport channel reconfiguration from CELL_DCH to CELL_DCH: Failure (Physical channel failure and reversion failure)

8.2.4.4.1 Definition

8.2.4.4.2 Conformance requirement

The UE shall perform a cell update This procedure is used to upon failure of reconfiguration for a transport channel because of physical channel failure and reversion failure for the transition from CELL_FACH to CELL_FACH in the same cell. After the UE completes cell update procedure, the UE transmits TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC which set IE "failure cause" to "physical channel failure".

Reference

3GPP TS 25.331 clause 8.2.4

8.2.4.4.3 Test purpose

To confirm that the UE transmits RADIO TRANSPORT CHANNEL RECONFIGURATION FAILURE message after it completes initiates an RRC connection re establishmentcell update procedure when the UE cannot reconfigure the new transport channel due to a failure of L1 configuration, and subsequently fail to revert to the old configuration after T312 expiry.

8.2.4.4.4 Method of test

Initial Condition

System Simulator: 2 cells – Cell 1 is active, Cell 2 is inactive

UE: PS-DCCH+DTCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108 in cell 1

Test Procedure

The UE is in the CELL_DCH state in cell 1. SS begins to broadcast the BCCH of cell 2 using a power level lower than that applied for cell 1. The SS transmits a TRANSPORT CHANNEL RECONFIGURATION message to the UE. The message specifies a new set of transport channel parameters but the SS does not reconfigure L1 correspondingly. At the same time, SS deletes its current contexts for cell 1. As a result, the UE cannot reconfigure the new transport channel and shall attempt to revert to the old configuration. The UE shall find the presence of cell 2 and then transmit a a CELL UPDATE message on uplink CCCH with IE "Cell update cause" set to "radio link failure". The SS shall transmit CELL UPDATE CONFIRM message on downlink CCCH after receiving CELL UPDATE message. The UE transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC and subsequently transmit TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting IE "failure cause" to "physical channel failure".

n RRC CONNECTION RE ESTABLISHMENT REQUEST message which includes the IE "U RNTI" on the uplink CCCH. The SS transmits an RRC CONNECTION RE ESTABLISHMENT message, which includes a new TFCS according to the new transport channel assigned. After the UE receives an RRC CONNECTION RE ESTABLISHMENT message, it reconfigures the new radio connection and transmits an RRC CONNECTION RE ESTABLISHMENT COMPLETE message on the uplink DCCH using AM RLC.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	+	ВССН	SS begins to transmit the BCCH for cell 2.
2	+	TRANSPORT CHANNEL RECONGURATION	Specifies the use of a new setting for transport channel.
3			The SS does not reconfigure L1 and deletes the old configuration present in cell 1.
4			The UE fails to reconfigure a new transport channel.
5)	CELL UPDATERRC CONNECTION RE-ESTABLISHMENT REQUEST	The UE detects the presence of cell.2, and enters CELL FACH state. This message includes the value "radio link failure" set in IE "Cell update cause". then transmits this message which includes the IE "U-RNTI".
6	←	CELL UPDATE CONFIRMRRC CONNECTION RE-ESTABLISHMENT	This message includes IE "new U-RNTI" and IE "new C-RNTI"". Including the new TFCS according to the new transport channel.
7	→	UTRAN MOBILITY INFORMATION CONFIRMERC CONNECTION RE-ESTABLISHMENT COMPLETE	
<u>8</u>	<u></u>	TRANSPORT CHANNEL RECONGURATION FAILURE	The IE "failure cause" shall be set to "physical channel failure"

Specific Message Contents

TRANSPORT CHANNEL RECONFIGURATION (Step 1)

Use the message sub-type titled "Packet to CELL_DCH from CELL_DCH in PS" in Annex A.

CELL UPDATE (Step 5)

The contents of CELL UPDATE message is identical as "Contents of CELL UPDATE message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	
- SRNC Identity	Assigned previously in cell 1
- S-RNTI	Assigned previously in cell 1
Cell Update Cause	<u>"radio link failure"</u>

CELL UPDATE CONFIRM (Step 6)

The contents of CELL UPDATE CONFIRM message is identical as "CELL UPDATE CONFIRM message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	Same as CELL UPDATE message in step 7
New U-RNTI	
- SRNC Identity	<u>'0000 0000 0000 0001'</u>
- S-RNTI	Different from previous S-RNTI
New C-RNTI	Different from previous C-RNTI

TRANSPORT CHANNELRECONGURATION FAILURE (Step 8)

Information Element	<u>Value/remark</u>
Message Type	"TRANSPORT CHANNEL RECONGURATION"
Failure cause	"physical channel failure"
Other information element	Not checked

8.2.4.4.5 Test requirement

After step 4 the UE shall discover cell 2, enter CELL_FACH state, and then transmit a RRC CONNECTION RE-ESTABLISHMENT REQUESTCELL UPDATE message on the uplink CCCH.

After step 6 the UE shall transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

After step 7 the UE shall transmit TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting the IE "failure cause" to "physical channel failure".

After step 7 the UE shall successfully re establish an RRC connection in cell 2.

8.2.4.5 Transport Channel Reconfiguration from CELL_DCH to CELL_DCH: Failure (Incompatible simultaneous reconfiguration)

8.2.4.5.1 Definition

8.2.4.5.2 Conformance requirement

If the UE receives a TRANSPORT CHANNEL RECONFIGURATION message whilst reconfiguring due to a radio bearer message other than TRANSPORT CHANNEL RECONFIGURATION, it shall keep its configuration as if the TRANSPORT CHANNEL RECONFIGURATION message had not been received. If the UE is reconfiguring itself after receiving a radio bearer message other than TRANSPORT CHANNEL RECONFIGURATION, and when the UE subsequently receive a TRANSPORT CHANNEL RECONFIGURATION message and UE shall transmit a TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC which is set to "Incompatible simultaneous reconfiguration" in IE "failure cause".

Reference

3GPP TS 25.331 clause 8.2.4

8.2.4.5.3 Test purpose

To confirm that if the UE receives a TRANSPORT CHANNEL RECONFIGURATION message whilst reconfiguring due to a radio bearer message other than TRANSPORT CHANNEL RECONFIGURATION, it shall keep its configuration as if the TRANSPORT CHANNEL RECONFIGURATION message had not been received and complete the reconfiguration according to the previously received message. To confirm that the UE keeps its current configuration and transmits a TRANSPORT CHANNEL RECONFIGURATION FAILURE message when it receives a TRANSPORT CHANNEL RECONFIGURATION message before the completion of an earlier reconfiguration request.

8.2.4.5.4 Method of test

Initial Condition

System Simulator: 1 cell

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a RADIO BEARER RECONFIGURATION message to the UE. The SS transmits a TRANSPORT CHANNEL RECONFIGURATION message before the "activation time" indicated in the RADIO BEARER RECONFIGURATION message expires. When the UE receives the TRANSPORT CHANNEL RECONFIGURATION message, the UE shall keep the configuration as if it had not received the TRANSPORT CHANNEL RECONFIGURATION message and shall transmit a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC with IE "failure cause" set to "incompatible simultaneous reconfiguration". After the SS receives the TRANSPORT CHANNEL RECONFIGURATION FAILURE message, the UE reconfigures the new physical channel parameters and transmits a RADIO BEARER RECONFIGURATION COMPLETE message on DCCH using AM RLC.

The UE is in the CELL_DCH state. When the SS transmits a TRANSPORT CHANNEL RECONFIGURATION message to the UE before the UE has successfully configured the transport channel specified in a previous message, the UE keeps the old configuration. After that, the UE shall transmit a TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, indicating "incompatible simultaneous reconfiguration" in IE "failure cause":

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	+	RADIO BEARER RECONFIGURATION	Including IE "Uplink DPCH info"
2	+	TRANSPORT CHANNEL RECONFIGURATION	Sent before the time specified in IE "Activation Time Info" of message in step 1 has elapsed.
3	→	TRANSPORT CHANNEL RECONFIGURATION FAILURE	The UE shall not change the configuration due to the reception of TRANSPORT CHANNEL RECONFIGURATION message.
4	<u> </u>	RADIO BEARER RECONFIGURATION COMPLETE	This message is on DCCH using AM RLC.

Specific Message Contents

RADIO BEARER RECONFIGURATION (Step 1)

For RADIO BEARER RECONFIGURATION in step 1, use the message sub-type indicated as "Packet to CELL_DCH from CELL_DCH in PS" found in Annex A.

TRANSPORT CHANNEL RECONFIGURATION (Step 2)

The contents of TRANSPORT CHANNEL RECONFIGURATION message in this test case is identical to the corresponding message found in Annex A with the following exceptions:

Information Element	Value/remark
Activation Time	Current CFN-[current CFN mod 8 + 8]

TRANSPORT CHANNEL RECONFIGURATION FAILURE

Information Element	Value/remark
Message Type	"TRANSPORT CHANNEL RECONFIGURATION
	FAILURE"
Failure cause	Incompatible simultaneous reconfiguration
Other information element	Not checked

8.2.4.5.5 Test requirement

After step 1, SS transmits a TRANSPORT CHANNEL RECONFIGURATION message before the expiry of the activation time specified in the message of step 1.

After step 2 the UE shall keep its configuration as if the UE had not received the RADIO BEARER SETUP message and shall transmit a TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC with IE "failure cause" set to "Incompatible simultaneous reconfiguration".

After step 4 the UE communicates with the SS on the DCCH and DTCH using the new physical channel parameters reconfigured as a result of the RADIO BEARER RECONFIGURATION message. After step2 the UE shall keep its configuration and transmit an TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, stating the reason "incompatible simultaneous reconfiguration" in IE "failure cause".

8.2.4.6 Transport channel reconfiguration from CELL_DCH to CELL_DCH: Failure (Invalid message reception and Invalid configuration)

8.2.4.6.1 Definition

8.2.4.6.2 Conformance requirement

The UE shall keep its old configuration when the UE receives a TRANSPORT CHANNEL RECONFIGURATION message which makes use of a spare-undefined value in the mandatory IE "UTRAN DRX cycle length coefficientDRX indicator". Then it shall transmit a TRANSPORT CHANNEL RECONFIGURATION FAILURE message with the value "protocol error" set in IE "failure cause" and also "Information element value not comprehended" in IE "Protocol error cause". The UE shall keep existing configuration before reception of a TRANSPORT CHANNEL RECONFIGURATION message when the TRANSPORT CHANNEL RECONFIGURATION message include some IEs set to invalid value, and then the UE shall transmit TRANSPORT CHANNEL RECONFIGURATION FAILURE including IE "failure cause" set to "invalid configuration".

Reference

3GPP TS 25.331 clause 8.2.4

8.2.4.6.3 Test purpose

To confirm that the UE transmits TRANSPORT CHANNEL RECONFIGRATION FAILURE message on the DCCH using AM RLC, if the received TRANSPORT CHANNEL RECONFIGURATION message comprises an spare undefined value in the mandatory IE "UTRAN DRX cycle length coefficient DRX indicator".

To confirm that the UE transmits a TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC if it receives a TRANSPORT CHANNEL RECONFIGURATION message including some IEs set to invalid value.

8.2.4.6.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a TRANSPORT CHANNEL RECONFIGURATION message to the UE, which includes a <u>spare-undefined</u> value in the mandatory IE "<u>UTRAN DRX cycle length coefficientDRX indicator</u>". The UE shall keep the old configuration and transmit a TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, specifying "protocol error" in IE "failure cause" and also indicating

"Information element value not comprehended" in IE "Protocol error cause". The UE keeps initial configuration and SS transmits TRANSPORT CHANNEL RECONFIGURATION message including some IEs set to invalid value. The UE transmits TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting the value "invalid configuration" to IE "failure cause".

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1	+	-	TRANSPORT CHANNEL	Contains an illegal value for a
			RECONFIGURATION	mandatory IE
2	\rightarrow		TRANSPORT CHANNEL	The UE does not change the
			RECONFIGRATION FAILURE	configuration.
<u>3</u>	<u> </u>	_	TRANSPORT CHANNEL	This message includes IE set to
			RECONFIGURATION	invalid value
<u>4</u>				The UE does not change the
				<u>configuration</u>
<u>5</u>		<u>></u>	TRANSPORT CHANNEL	The IE "failure cause" shall be set
			RECONFIGRATION FAILURE	to "invalid configuration

Specific Message Contents

TRANSPORT CHANNEL RECONFIGURATION (Step 1)

The contents of TRANSPORT CHANNEL RECONFIGURATION message in this test case is identical in Annex A for RRC tests with the following exceptions:

Information Element	Value/remark
UTRAN DRX cycle length coefficientDRX indicator	spare Undefined value

TRANSPORT CHANNEL RECONFIGURATION FAILURE (Step 2)

Information Element	Value/remark
Message Type	"TRANSPORT CHANNEL RECONFIGURATION FAILURE"
Failure couce	FAILURE
Failure cause - Failure cause	Protocol error
	Protocol error
- Protocol error information	
- Protocol error cause	Information element value not comprehended
Other information element	Not checked

TRANSPORT CHANNEL RECONFIGURATION (Step 3)

The contents of TRANSPORT CHANNEL RECONFIGURATION message in this test case is identical in Annex A for RRC tests with the following exceptions:

Added or Reconfigured UL TrCH information	
- Transport channel identity	1
	<u> </u>
- TFS	
 Dynamic Transport format information 	(This IE is repeated for TFI number)
- RLC size	Reference to TS34.108 clause 6.10 Parameter Set
- CHOICE Logical Channel list	
- Explicit List	
- RB identity	2

TRANSPORT CHANNEL RECONFIGURATION FAILURE (Step 5)

Information Element	<u>Value/remark</u>	
Message Type		
Failure cause	Invalid configuration	
Other information element	Not checked	

8.2.4.6.5 Test requirement

After step 1 the UE shall keep its old configuration and transmit a TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC. The message shall specify "protocol error" in IE "failure cause" and set value "Information element value not comprehended" in IE "Protocol error cause".

After step 3 the UE shall keep its old configuration.

After step 4 the UE shall transmit TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting IE "failure cause" to "invalid configuration".

8.2.4.7 Transport channel reconfiguration from CELL_DCH to CELL_FACH: Success

8.2.4.7.1 Definition

8.2.4.7.2 Conformance requirement

The UE shall correctly reconfigure the transport channels according to TRANSPORT CHANNEL RECONFIGURATION message, after it is requested to perform a transition from CELL_DCH to CELL_FACH in the same cell in conjunction with the transport channel reconfiguration.

Reference

3GPP TS 25.331 clause 8.2.4

8.2.4.7.3 Test purpose

To confirm that the UE reconfigures a new Transport channel according to a TRANSPORT CHANNEL RECONFIGURATION message received from the SS.

8.2.4.7.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits TRANSPORT CHANNEL RECONFIGURATION message to the UE and the UE which omits IE "Uplink DPCH info" and IE "Downlink DPCH info". This should cause the UE to performs a state transition from CELL_DCH to CELL_FACH in the same cell. The UE then reconfigures the new transport channel according to this message and reconfigure the new physical channel according to the system information messages. Finally, the UE shall transmit a TRANSPORT CHANNEL RECONFIGURATION COMPLETE message using AM RLC.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1		←	TRANSPORT CHANNEL RECONFIGURATION	IE "Uplink DPCH Info" and IE "Downlink DPCH Info" are not specified.
2				UE shall perform the reconfiguration of transport channel
3		\rightarrow	TRANSPORT CHANNEL RECONFIGURATION COMPLETE	

Specific Message Contents

TRANSPORT CHANNEL RECONFIGURATION

Use the message sub-type titled "Packet to CELL_FACH from CELL_DCH in PS" in Annex A.

8.2.4.7.5 Test requirement

After step 3 the UE shall transit from CELL_DCH to CELL_FACH in the same cell, and then continue to communicate with SS on the new transport channel and common physical channels.

8.2.4.8 Transport channel reconfiguration from CELL_DCH to CELL_FACH: Failure (Physical channel failure and reversion to old configuration)

8.2.4.8.1 Definition

8.2.4.8.2 Conformance requirement

The UE shall revert the old configuration when it fails to reconfigure the new transport channel requested. When reversion is successful, it shall transmit a TRANSPORT CHANNEL RECONFIGURATION FAILURE message to UTRAN.

Reference

3GPP TS 25.331 clause 8.2.4

8.2.4.8.3 Test purpose

To confirm that the UE reverts to the old configuration and transmits a TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, if the UE fails to reconfigure the new transport channel according to the TRANSPORT CHANNEL RECONFIGURATION message received.

8.2.4.8.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS DCCH+DTCH_DCH (state 6 10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a TRANSPORT CHANNEL RECONFIGURATION message to the UE which includes the new transport channel parameters. However, it does not reconfigure the new transport channel to reflect change. As a result, the UE should encounter difficulties in the reconfigure process and after T312 expiry, it shall revert to its stored old configuration. When the UE manages to return to the old configuration, it transmits a TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC and set the reason "physical channel failure" in IE "failure cause".

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
4			TRANSPORT CHANNEL -RECONGURATION	
2				The SS does not reconfigure the transport channel hence the UE could not access the new transport channel.
3		>	TRANSPORT CHANNEL -RECONFIGURATION FAILURE	After T312 expiry, the UE reverts to the old configuration and transmits this message using the original transport channel settings.

Specific Message Contents

TRANSPORT CHANNEL RECONFIGURATION

Use the message sub-type titled "Packet to CELL_FACH from CELL_DCH in PS" in Annex A.

TRANSPORT CHANNEL RECONFIGURATION FAILURE

Information Element	Value/remark
Message Type	TRANSPORT CHANNEL RECONFIGURATION
	FAILURE
Failure cause	Physical channel failure
Other information element	Not checked

8.2.4.8.5 Test requirement

After step2 the UE shall attempt revert to the old configuration and transmit a TRANPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC. The failure message shall contain the reason "physical channel failure" in IE "failure cause". Void

8.2.4.9 Transport channel reconfiguration from CELL_DCH to CELL_FACH: Failure (Cell re-selection Physical channel failure and reversion failure)

8.2.4.9.1 Definition

8.2.4.9.2 Conformance requirement

The UE shall initiate a RRC connection re establishmentcell update procedure when the UE cannot use the assigned physical channel according to a TRANSPORT CHANNEL RECONFIGURATION message as the transition from CELL DCH to CELL FACH. After the UE completes cell update procedure, the UE transmits TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC which is set to "cell reselection" in IE "failure cause". , after it encounters a physical channel failure followed by an inability to revert to the old transport channel configuration.

Reference

3GPP TS 25.331 clause 8.2.4

8.2.4.9.3 Test purpose

To confirm that the UE initiates an RRC connection re establishmentcell update procedure when the UE fails successively in the following actions: (a) configure the new transport channel according to TRANSPORT CHANNEL

RECONFIGGURATION message received and (b) revert to the old channel configuration after timer T312 expiry. The UE transmits TRANSPORT CHANNEL RECONFIGURATION FAILURE message after it completes a cell update.

8.2.4.10.4 Method of test

Initial Condition

System Simulator: 2 cells - Cell 1 is active, Cell 2 is inactive

UE: PS-DCCH+DTCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108 in cell 1

Test Procedure

The UE is in the CELL_DCH state in cell 1. SS begins to transmit the BCCH for cell 2. The SS then transmits a TRANSPORT CHANNEL RECONFIGURATION message to the UE, which specifies the new transport channel parameters on common physical channel. But the SS does not reconfigure the L1 parameters and the new transport channel to reflect this change. Consequently, the UE discovers that it cannot reconfigure the new transport channel and try to revert to the old configuration. At this time, SS deletes all context related to the old configuration in cell 1. The UE shall select cell 2 by performing cell re-selection and enter CELL FACH state. The UE transmits CELL UPDATE message on uplink CCCH with IE "Cell update cause" set to "cell reselection". The SS shall transmit CELL UPDATE CONFIRM message on downlink CCCH after receiving CELL UPDATE message. The UE transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC and transmit TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting IE "failure cause" to "cell reselection". Then the UE should find cell 2 and transmits an RRC CONNECTION RE-ESTABLISHMENT REQUEST message which includes the IE "U RNTI" on the uplink CCCH. After the reception of such a message, the SS transmits an RRC CONNECTION RE-ESTABLISHMENT message and includes a new TFCS for the new transport channel in cell 2. After receiving this message, the UE reconfigures the new radio connection and transmits an RRC CONNECTION RE-ESTABLISHMENT COMPLETE message on the uplink DCCH using AM RLC.

Expected sequence

Step	Direction		tep Direction		Direction Message	Comment
-	UE	SS	1			
1	+		BCCH	The SS starts to transmit the BCCH cell 2.		
2		←	TRANSPORT CHANNEL RECONGURATION			
3				The SS does not reconfigure L1. At the same time, it deletes the old channel configuration.		
4				The UE fails to reconfigure a new transport channel.		
5		→	CELL UPDATERRC CONNECTION RE-ESTABLISHMENT REQUEST	The UE shall discover the presence of cell 2 and transmits this message, which includes the IE "U-RNTI". The value "cell reselection" shall be set in IE "Cell update cause".		
6		←	CELL UPDATE CONFIRMRRC CONNECTION RE-ESTABLISHMENT	This message include IE "new U-RNTI" and IE "new C-RNTI". A new TFCS is commanded according to the new transport channel.		
7	→		UTRAN MOBILITY INFORMATION CONFIRMRRC CONNECTION RE-ESTABLISHMENT COMPLETE			
<u>8</u>	<u></u>		TRANSPORT CHANNEL RECONFIGURATION FAILURE	The IE "failure cause" shall be set to "cell reselection"		

Specific Message Contents

TRANSPORT CHANNEL RECONFIGURATION (Step 1)

Use the message sub-type titled "Packet to CELL_FACH from CELL_DCH in PS" in Annex A.

CELL UPDATE (Step 5)

The contents of CELL UPDATE message is identical as "Contents of CELL UPDATE message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	
- SRNC Identity	Assigned previously in cell 1
<u>- S-RNTI</u>	Assigned previously in cell 1
Cell Update Cause	"radio link failure"

CELL UPDATE CONFIRM (Step 6)

The contents of CELL UPDATE CONFIRM message is identical as "CELL UPDATE CONFIRM message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	Same as CELL UPDATE message in step 7
New U-RNTI	
- SRNC Identity	<u>'0000 0000 0000 0001'</u>
<u>- S-RNTI</u>	Different from previous S-RNTI
New C-RNTI	Different from previous C-RNTI

TRANSPORT CHANNEL RECONFIGURATION FAILURE (Step 8)

Information Element	<u>Value/remark</u>
Message Type	"RADIO BEARER RECONFIGURATION FAILURE"
Failure cause	"cell reselection"
Other information element	Not checked

8.2.4.9.5 Test requirement

After step 4 the UE shall find cell 2, enter CELL_FACH state and then initiate <u>cell update RRC connection reestablishment-procedure</u> in cell 2.

After step 6 **UE shall** transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

After step 7 UE shall transmit TRANSPORT CHANNEL FAILURE message on the DCCH using AM RLC, setting the value "cell reselection" to IE "failure cause".

After step 7 the UE shall successfully re establish an RRC connection in cell 2.

8.2.4.10 Transport channel reconfiguration from CELL_FACH to CELL_DCH: Success

8.2.4.10.1 Definition

8.2.4.10.2 Conformance requirement

The UE shall correctly reconfigure the transport channels according to TRANSPORT CHANNEL RECONFIGURATION message, which trigger a state transition from CELL_FACH to CELL_DCH in the same cell.

Reference

3GPP TS 25.331 clause 8.2.4

8.2.4.10.3 Test purpose

To confirm that the UE reconfigures a new transport channel using dedicated physical channel according to a TRANSPORT CHANNEL RECONFIGURATION message received from the SS.

8.2.4.10.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a TRANSPORT CHANNEL RECONFIGURATION message to the UE, which includes IE "Uplink DPCH info" and IE "Downlink DPCH info" leading to a state transition from CELL_FACH to CELL_DCH in the same cell. The UE shall reconfigure the new transport channel according to this message and then reconfigure the new physical channel according to the system information message. Finally, the UE transmits a TRANSPORT CHANNEL RECONFIGURATION COMPLETE message using AM RLC.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1	←		TRANSPORT CHANNEL	Includes both IE "Uplink DPCH
			RECONFIGURATION	Info" and IE "Downlink DPCH
				Info" in the message.
2				Reconfiguration of transport
				channel
3		\rightarrow	TRANSPORT CHANNEL	
			RECONFIGURATION COMPLETE	

Specific Message Contents

TRANSPORT CHANNEL RECONFIGURATION

Use the message sub-type titled "Packet to CELL_DCH from CELL_FACH in PS" in Annex A.

8.2.4.10.5 Test requirement

After step 3 the UE shall transit from CELL_FACH to CELL_DCH in the same cell, and continue to communicate with SS using the new transport channel configuration based on DPCH physical channels.

8.2.4.11 Transport channel reconfiguration from CELL_FACH to CELL_DCH: Failure (Unsupported configuration)

8.2.4.11.1 Definition

8.2.4.11.2 Conformance requirement

The UE shall keep its old configuration when the UE receives a TRANSPORT CHANNEL RECONFIGURATION message which includes unsupported configuration parameters and transmit a TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC which is set to "configuration unsupported" in IE "failure cause

Reference

3GPP TS 25.331 clause 8.2.4

8.2.4.11.3 Test purpose

To confirm that the UE transmits a TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC when it receives a TRANSPORT CHANNEL RECONFIGURATION message which includes unsupported configuration parameters.

8.2.4.14.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a TRANSPORT CHANNEL RECONFIGURATION message to the UE, which includes unsupported configuration parameters for the UE. The UE shall transmit a TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting "configuration unsupported" in IE "failure cause".

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1		←	TRANSPORT CHANNEL RECONGURATION	The message includes unsupported configuration by the UE
2		\rightarrow	TRANSPORT CHANNEL RECONFIGURATION FAILURE	The UE shall not change the transport channel.

Specific Message Contents

TRANSPORT CHANNEL RECONFIGURATION

The contents of TRANSPORT CHANNEL RECONFIGURATION message in this test case is identical to the message sub-type title "Packet to CELL_DCH from CELL_FACH in PS" found in Annex A with the following exceptions:

Information Element	Value/remark
TrCH Information Elements	
-Uplink transport Channels	
-Added or Reconfigured TrCH information list	
	Number of transport channels = 4096
-Downlink transport Channels	·
-Added or Reconfigured TrCH information list	
	Number of transport channels = 4096

TRANSPORT CHANNEL RECONFIGURATION FAILURE

Information Element	Value/remark
Message Type	"TRANSPORT CHANNEL RECONFIGURATION
	FAILURE"
Failure cause	Configuration unsupported
Other information element	Not checked

8.2.4.11.5 Test requirement

After step1 the UE shall keep its old configuration and transmit a TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC. The UE shall set "configuration unsupported" in IE "failure cause" of the message.

8.2.4.12 Transport channel reconfiguration from CELL_FACH to CELL_DCH: Failure (Physical channel failure and reversion to old channel)

8.2.4.12.1 Definition

8.2.4.12.2 Conformance requirement

The UE shall revert to the old configuration when the UE has failed to reconfigure the new transport channel requested, and then transmit a TRANSPORT CHANNEL RECONFIGURATION FAILURE message to UTRAN.

Reference

3GPP TS 25.331 clause 8.2.4

8.2.4.12.3 Test purpose

To confirm that the UE reverts to the old configuration and transmits a TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, if the UE fails to reconfigure the new transport channel according to a TRANSPORT CHANNEL RECONFIGURATION message.

8.2.4.12.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. SS transmits a TRANSPORT CHANNEL RECONFIGURATION message to the UE, which includes the new transport channel parameters. However, SS does not reconfigure the new transport channel accordingly. Hence, the UE shall experience a failure in the reconfiguration process. After T312 expiry, the UE shall revert to the old channel configuration. Then the UE transmits a TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, stating the reason "physical channel failure" in IE "failure cause".

Expected sequence

Step	Direction		Message	Comment	
	UE	SS			
1		+	TRANSPORT CHANNEL RECONGURATION	Message includes IE "Downlink DPCH Info" and IE "Uplink DPCH Info"	
2				SS does not reconfigure the transport channel causing the UE to detect a physical channel failure.	
3		\rightarrow	TRANSPORT CHANNEL RECONFIGURATION FAILURE	After T312 expiry the UE shall revert to the old configuration and transmit this message.	

Specific Message Contents

TRANSPORT CHANNEL RECONFIGURATION

Use the message sub-type titled "Packet to CELL_DCH from CELL_FACH in PS" in Annex A.

TRANSPORT CHANNEL RECONFIGURATION FAILURE

Information Element	Value/remark
Message Type	TRANSPORT CHANNEL RECONFIGURATION
	FAILURE
Failure cause	Physical channel failure
Other information element	Not checked

8.2.4.12.5 Test requirement

After step 2 the UE shall revert to the old configuration and transmit a TRANPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting the value "physical channel failure" in IE "failure cause".

8.2.4.13 Transport channel reconfiguration from CELL_FACH to CELL_DCH: Failure (Physical channel failure and reversion failure)

8.2.4.13.1 Definition

8.2.4.13.2 Conformance requirement

The UE shall initiate a <u>cell update RRC re establishment</u> procedure when it fails to revert to the old channel configuration, following a physical channel failure in the transport channel reconfiguration procedure. <u>After the UE completes cell update procedure, the UE transmit TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC which is set to "physical channel failure" in IE "failure cause".</u>

Reference

3GPP TS 25.331 clause 8.2.4

8.2.4.13.3 Test purpose

To confirm that the UE transmits RADIO TRANSPORT CHANNEL RECONFIGURATION FAILURE message after it completes initiates an cell update RRC connection re establishment procedure, when the UE cannot reconfigure the new transport channel for the failure of L1 configuration and subsequently fails to revert to the old configuration after T312 expiry.

8.2.4.13.4 Method of test

Initial Condition

System Simulator: 2 cells - Cell 1 is active, Cell 2 is inactive

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108 in cell 1

Test Procedure

The UE is in the CELL_FACH state in a cell 1. SS begins to broadcast the BCCH of cell 2 at a power level lower than in cell 1, and then it transmits a TRANSPORT CHANNEL RECONFIGURATION message to the UE in cell 1. The message includes a new set of transport channel parameters. However, the SS does not reconfigure L1 and the new

transport channel accordingly. At the same time, it deletes the current channel configurations in cell 1. As a result, the UE cannot reconfigure the new transport channel and an attempt to revert to the old configuration fails. The UE shall then find that cell 2 is available, camp onto it, and transmits CELL UPDATE message on uplink CCCH with IE "Cell update cause" set to "radio link failure". The SS shall transmit CELL UPDATE CONFIRM message on downlink CCCH after receiving CELL UPDATE message. The UE transmits UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC and subsequently transmits TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting the value "physical channel failure" to IE "failure cause".

an RRC CONNECTION RE-ESTABLISHMENT REQUEST message. This message shall include the IE "U-RNTI" on the uplink CCCH. When the SS receives this message, it transmits an RRC CONNECTION RE ESTABLISHMENT message, which specifies a new TFCS according to the new transport channel used. After the UE receives this message, it reconfigures the new radio connection and transmits an RRC CONNECTION RE ESTABLISHMENT COMPLETE message on the uplink DCCH using AM RLC.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	+	BCCH	The SS starts to transmit BCCH in cell 2
2	+	TRANSPORT CHANNEL RECONGURATION	
3			The SS does not reconfigure L1 and transport channel, it also deletes its current contexts for cell 1.
4			The UE shall experience a failure to reconfigure a new transport channel and also fails to revert to old configuration.
5	→	CELL UPDATERRC CONNECTION RE-ESTABLISHMENT REQUEST	The UE shall find cell 2, camp onto it, and and be in CELL_FACH state. This message include the value "radio link failure" set in IE "Cell update cause".then transmits this message which includes the IE "U-RNTI".
6	+	CELL UPDATE CONFIRMRRC CONNECTION RE-ESTABLISHMENT	This message include IE "new U-RNTI" and IE "new C-RNTI". Including the new TFCS according to the new transport channel.
7	→	UTRAN MOBILITY INFORMATION CONFIRMREC CONNECTION RE-ESTABLISHMENT COMPLETE	
<u>8</u>	<u>→</u>	TRANSPORT CHANNEL RECONGURATION FAILURE	The IE "failure cause" shall be set to "physical channel failure"

Specific Message Contents

TRANSPORT CHANNEL RECONFIGURATION (Step 1)

Use the message sub-type titled "Packet to CELL_DCH from CELL_FACH in PS" in Annex A.

CELL UPDATE (Step 5)

The contents of CELL UPDATE message is identical as "Contents of CELL UPDATE message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	
- SRNC Identity	Assigned previously in cell 1
- S-RNTI	Assigned previously in cell 1
Cell Update Cause	"radio link failure"

CELL UPDATE CONFIRM (Step 6)

The contents of CELL UPDATE CONFIRM message is identical as "CELL UPDATE CONFIRM message" as found in Annex with the following exceptions:

Information Element	Value/remark
<u>U-RNTI</u>	Same as CELL UPDATE message in step 7
New U-RNTI	
- SRNC Identity	<u>'0000 0000 0000 0001'</u>
- S-RNTI	Different from previous S-RNTI
New C-RNTI	Different from previous C-RNTI

TRANSPORT CHANNELRECONGURATION FAILURE (Step 8)

Information Element	<u>Value/remark</u>
Message Type	"TRANSPORT CHANNEL RECONGURATION"
Failure cause	<u>"physical channel failure"</u>
Other information element	Not checked

8.2.4.13.5 Test requirement

After step 4 the UE shall find the presence of cell 2, enter CELL_FACH state, and transmit <u>CELL UPDATE message</u> on the uplink <u>CCCH RRC RE ESTABLISHMENT REQUEST message to attempt to setup the RRC connection in cell 2</u>

After step 7–6 the UE shall transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

After step 7 the UE shall transmit TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting the IE "failure cause" to "physical channel failure". the UE shall successfully re establish the RRC connection in cell 2.

8.2.4.14 Transport Channel Reconfiguration from CELL_FACH to CELL_DCH: Failure (Incompatible simultaneous reconfiguration)

8.2.4.14.1 Definition

8.2.4.14.2 Conformance requirement

If the UE receives a TRANSPORT CHANNEL RECONFIGURATION message whilst reconfiguring due to a radio bearer message other than TRANSPORT CHANNEL RECONFIGURATION, it shall keep its configuration as if the TRANSPORT CHANNEL RECONFIGURATION message had not been received.

If the UE is reconfiguring itself after receiving a radio bearer message other than TRANSPORT CHANNEL RECONFIGURATION, and when the UE subsequently receive a TRANSPORT CHANNEL RECONFIGURATION message and UE shall transmit a TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC which is set to "Incompatible simultaneous reconfiguration" in IE "failure cause".

Reference

3GPP TS 25.331 clause 8.2.4

8.2.4.14.3 Test purpose

To confirm that if the UE receives a TRANSPORT CHANNEL RECONFIGURATION message whilst reconfiguring due to a radio bearer message other than TRANSPORT CHANNEL RECONFIGURATION, it shall keep its

configuration as if the TRANSPORT CHANNEL RECONFIGURATION message had not been received and complete the reconfiguration according to the previously received message. To confirm that the UE keeps its configuration and transmits a TRANSPORT CHANNEL RECONFIGURATION FAILURE message when it receives a TRANSPORT CHANNEL RECONFIGURATION message before the completion of configuration from an earlier reques.

8.2.4.14.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a RADIO BEARER RECONFIGURATION message to the UE. The SS transmits a TRANSPORT CHANNEL RECONFIGURATION message before the "activation time" indicated in the RADIO BEARER RECONFIGURATION message expires. When the UE receives the TRANSPORT CHANNEL RECONFIGURATION message, the UE shall keep the configuration as if it had not received the TRANSPORT CHANNEL RECONFIGURATION message and shall transmit a RADIO BEARER SETUP FAILURE message on the DCCH using AM RLC with IE "failure cause" set to "incompatible simultaneous reconfiguration". After the SS receives the RADIO TRANSPORT CHANNEL RECONFIGURATION FAILURE message, the UE reconfigures the new physical channel parameters and transmits a RADIO BEARER RECONFIGURATION COMPLETE message on DCCH using AM RLC. The UE is in the CELL_FACH state. When the SS transmits a TRANSPORT CHANNEL RECONFIGURATION message to the UE before it has configured based on an earlier request. It shall keep the old configuration and then transmitting a TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC. The content of the failure message shall indicate "incompatible simultaneous reconfiguration" in IE "failure cause".

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	+	RADIO BEARER RECONFIGURATION	Includes the IE "Uplink DPCH info"
2	+	TRANSPORT CHANNEL RECONFIGURATION	Sent before the elapse of the Activation time specified in step 1.
3	→	TRANSPORT CHANNEL RECONFIGURATION FAILURE	The UE does not change the configuration due to the reception of RADIO BEARER SETUP message
4	<u> </u>	RADIO BEARER RECONFIGURATION FAILURE	This message is on DCCH using AM RLC.

Specific Message Contents

RADIO BEARER RECONFIGURATION (Step 1)

For RADIO BEARER RECONFIGURATION in step 1, use the message sub-type indicated as "Packet to CELL_DCH from CELL_FACH in PS" found in Annex A.

TRANSPORT CHANNEL RECONFIGURATION (Step 2)

For TRANSPORT CHANNEL RECONFIGURATION in step 2, use the message sub-type indicated as "Packet to CELL_DCH from CELL_FACH in PS" found in Annex A.

Information Element	Value/remark
Activation Time Info	Current CFN-[current CFN mod 8 + 8]

TRANSPORT CHANNEL RECONFIGURATION FAILURE

Information Element	Value/remark
Message Type	"TRANSPORT CHANNEL RECONFIGURATION
	FAILURE"
Failure cause	Incompatible simultaneous reconfiguration
Other information element	Not checked

8.2.4.14.5 Test requirement

After step 1, SS transmits a TRANSPORT CHANNEL RECONFIGURATION message before the expiry of the activation time specified in the message of step 1.

After step 2 the UE shall keep its configuration as if the UE had not received the RADIO BEARER SETUP message and shall transmit a TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC with IE "failure cause" set to "Incompatible simultaneous reconfiguration".

After step 4 the UE communicates with the SS on the DCCH and DTCH using the new physical channel parameters reconfigured as a result of the RADIO BEARER RECONFIGURATION message. After step 2 the UE shall keep its configuration and transmit an TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting the value to "incompatible simultaneous reconfiguration" in IE "failure cause".

8.2.4.15 Transport channel reconfiguration from CELL_FACH to CELL_DCH:_Failure (Invalid message reception and Invalid configuration)

8.2.4.15.1 Definition

8.2.4.15.2 Conformance requirement

The UE shall keep its old configuration when the UE receives a TRANSPORT CHANNEL RECONFIGURATION message which includes a spare-undefined value in the mandatory IE "UTRAN DRX cycle length coefficient DRX indicator". The UE shall then transmit a TRANSPORT CHANNEL RECONFIGURATION FAILURE message, specifying "protocol error" in IE "failure cause" and also "Information element value not comprehended" in IE "Protocol error cause". The UE shall keep existing configuration before reception of a TRANSPORT CHANNEL RECONFIGURATION message when the TRANSPORT CHANNEL RECONFIGURATION message include some IEs set to invalid value, and then the UE shall transmit TRANSPORT CHANNEL RECONFIGURATION FAILURE including IE "failure cause" set to "invalid configuration".

Reference

3GPP TS 25.331 clause 8.2.4

8.2.4.15.3 Test purpose

To confirm that the UE transmits TRANSPORT CHANNEL RECONFIGRATION FAILURE message on the DCCH using AM RLC, if it receives a TRANSPORT CHANNEL RECONFIGURATION message which uses a spare undefined value in the mandatory IE "UTRAN DRX cycle length coefficient DRX indicator".

To confirm that the UE transmits a TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC if it receives a TRANSPORT CHANNEL RECONFIGURATION message including some IEs set to invalid value.

8.2.4.15.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a TRANSPORT CHANNEL RECONFIGURATION message to the UE, which includes a spare-undefined value in the mandatory IE "UTRAN DRX cycle-length coefficientDRX indicator". The UE shall keep the old configuration and then transmit a TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC. This message shall contain the value "protocol error" in IE "failure cause" and also "Information element value not comprehended" in IE "Protocol error cause". The UE keeps initial configuration and SS transmits TRANSPORT CHANNEL RECONFIGURATION message including some IEs set to invalid value. The UE transmit TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting IE "failure cause" to "invalid configuration".

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	+	TRANSPORT CHANNEL RECONFIGURATION	Contains an error in one of the mandatory IE in the message.
2	\rightarrow	TRANSPORT CHANNEL RECONFIGRATION FAILURE	The UE does not change the configuration.
<u>3</u>	<u>←</u>	TRANSPORT CHANNEL RECONFIGURATION	This message includes IE set to invalid value
<u>4</u>	4 The UE does r configuration		The UE does not change the configuration
<u>5</u>	<u>→</u>	TRANSPORT CHANNEL RECONFIGRATION FAILURE	The IE "failure cause" shall be set to "invalid configuration

Specific Message Contents

TRANSPORT CHANNEL RECONFIGURATION (Step 1)

The contents of TRANSPORT CHANNEL RECONFIGURATION message in this test case is identical to the message sub-type title "Packet to CELL_DCH from CELL_FACH in PS" found in Annex A with the following exceptions:

Information Element	Value/remark
UTRAN DRX cycle length coefficientDRX indicator	spare-Undefined value

TRANSPORT CHANNEL RECONFIGURATION FAILURE (Step 2)

Information Element	Value/remark
Message Type	"TRANSPORT CHANNEL RECONFIGURATION FAILURE"
Failure cause	
- Failure cause	Protocol error
- Protocol error information	
- Protocol error cause	Information element value not comprehended
Other information element	Not checked

TRANSPORT CHANNEL RECONFIGURATION (Step 3)

The contents of TRANSPORT CHANNEL RECONFIGURATION message in this test case is identical to the message sub-type title "Packet to CELL DCH from CELL FACH in PS" found in Annex A with the following exceptions:

Added or Reconfigured UL TrCH information	
- Transport channel identity	1
TFS	
 Dynamic Transport format information 	(This IE is repeated for TFI number)
- RLC size	Reference to TS34.108 clause 6.10 Parameter Set
- CHOICE Logical Channel list	
- Explicit List	
- RB identity	<u>2</u>

TRANSPORT CHANNEL RECONFIGURATION FAILURE (Step 5)

Information Element	<u>Value/remark</u>	
Message Type		
Failure cause	Invalid configuration	
Other information element	Not checked	

8.2.4.15.5 Test requirement

After step_1 the UE shall keep its old configuration and transmit a TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC. The content of the message shall specify "protocol error" in IE "failure cause" and also "Information element value not comprehended" in IE "Protocol error cause".

After step 3 the UE shall keep its old configuration.

After step 4 the UE shall transmit TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting IE "failure cause" to "invalid configuration".

8.2.4.16 Transport channel reconfiguration from CELL_FACH to CELL_FACH: Success with no transport channel type switching

8.2.4.16.1 Definition

8.2.4.16.2 Conformance requirement

The UE shall remain in CELL_FACH state and transition from CELL_FACH to CELL_FACH in the another cell requested in the received TRANSPORT CHANNEL RECONFIGURATION message.

Reference

3GPP TS 25.331 clause 8.2.4

8.2.4.16.3 Test purpose

To confirm that the UE reconfigures a new transport channel according to a TRANSPORT CHANNEL RECONFIGURATION message received from the SS.

8.2.4.16.4 Method of test

Initial Condition

System Simulator: 2-1 cells Cell 1 is active, Cell 2 is inactive

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a TRANSPORT CHANNEL RECONFIGURATION message to the UE, which includes new transport channel parameters for Hand Over. The UE reconfigures the new transport

channel and the new physical channel according to the system information messages. The UE transmits a TRANSPORT CHANNEL RECONFIGURATION COMPLETE message using AM RLC.

Expected sequence

Step	Direction		Message	Comment
-	UE	SS]	
4	- ←		BCCH	The SS starts to broadcast the BCCH in cell 2.
<u>21</u>	+		TRANSPORT CHANNEL RECONGURATION	Including Primary CPICH Info
<u>32</u>				Reconfiguration of a new transport channel
4 <u>3</u>	\rightarrow		TRANSPORT CHANNEL RECONFIGURATION COMPLETE	

Specific Message Contents

TRANSPORT CHANNEL RECONFIGURATION

Use the message sub-type titled "Packet to CELL_FACH from CELL_FACH in PS" in Annex A.

8.2.4.16.5 Test requirement

After step3 the UE shall transit from CELL_FACH to CELL_FACH in the another cell and continue to communicate with the SS on the DCCH using the existing transport channel.

8.2.4.17 Transport channel reconfiguration from CELL_FACH to CELL_FACH: Failure (Cell re-selection Physical channel failure)

8.2.4.17.1 Definition

8.2.4.17.2 Conformance requirement

The UE shall initiate the cell <u>reselection_update</u> procedure when the UE cannot use the assigned physical channel according to a TRANSPORT CHANNEL RECONFIGURATION message, during the transition from CELL_FACH to CELL_FACH in the another cell. <u>After the UE complete cell update procedure, the UE transmit TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC which <u>set IE "failure cause" to "invalid configuration"</u>.</u>

Reference

3GPP TS 25.331 clause 8.2.4

8.2.4.17.3 Test purpose

To confirm that the UE perform a cell reselection and cell update procedure when the it fail to access the assigned physical channel, according to a TRANSPORT CHANNEL RECONFIGURATION message, during the transition from CELL_FACH to CELL_FACH in the another cell.

To confirm that the UE transmit TRANSPORT CHANNEL RECONFIGURATION FAILURE message after UE completes a cell update.

8.2.4.2117.4

Method of test

Initial Condition

System Simulator: 3-2 cells Cell 1 is active, Cell 2 is inactive, Cell 3 is inactive

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state in cell 1. The SS begins to broadcast the BCCH in cell 2 and then transmits a TRANSPORT CHANNEL RECONFIGURATION message, resulting in the UE to attempt to transit from CELL_FACH in cell 31. The UE is expected to fail to access the assigned physical channel as the SS ceases downlink transmission in cell 31. After the UE successfully camp onto cell 2, it shall initiate the cell update procedure in cell 2. The UE transmit CELL UPDATE message on uplink CCCH with IE "Cell update cause" set to "cell reselection". The SS shall transmit CELL UPDATE CONFIRM message on downlink CCCH after receiving CELL UPDATE message. The UE transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC and transmit TRANSPORT CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting the value "cell reselection" to IE "failure cause". The procedure completes when the SS confirms UE's cell update request.

Expected sequence

Step	Directio	n Message	Comment	
	UE S	S		
1	+	BCCH	The SS starts to broadcast the BCCH in cell 2.	
2	←	TRANSPORT CHANNEL RECONFIGURATION	The message assigns common resources so that the UE is expected to execute a transition from CELL_FACH in cell 1-to CELL_FACH in cell 31	
3			The UE shall detect a failure to use the assigned common physical channel and try to perform a cell reselection.	
4	→	CELL UPDATE	The UE discover the presence of cell 2 and triggers a cell update procedure. The value "cell reselection" shall be set in IE "Cell update cause".	
5	+	CELL UPDATE CONFIRM	This message include IE "new U-RNTI" and IE "new C-RNTI". SS sends this message so that the UE moves to CELL_PCH state.	
<u>6</u>	<u>→</u>	UTRAN MOBILITY INFORMATION CONFIRM		
7	\rightarrow	TRANSPORT CHANNELRECONFIGURATION FAILURE	The IE "failure cause" shall be set to "cell reselection"	

Specific Message Contents

TRANSPORT CHANNEL RECONFIGURATION (Step 1)

Use the message sub-type titled "Packet to CELL_FACH from CELL_FACH in PS" in Annex A.

CELL UPDATE (Step 4)

The contents of CELL UPDATE message is identical as "Contents of CELL UPDATE message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	
- SRNC Identity	Assigned previously in cell 1
<u>- S-RNTI</u>	Assigned previously in cell 1
Cell Update Cause	"radio link failure"

CELL UPDATE CONFIRM (Step 5)

The contents of CELL UPDATE CONFIRM message is identical as "CELL UPDATE CONFIRM message" as found in Annex with the following exceptions:

Information Element	Value/remark
<u>U-RNTI</u>	Same as CELL UPDATE message in step 7
New U-RNTI	
- SRNC Identity	<u>'0000 0000 0000 0001'</u>
- S-RNTI	Different from previous S-RNTI
New C-RNTI	Different from previous C-RNTI

TRANSPORT CHANNEL RECONFIGURATION FAILURE (Step 7)

Information Element	<u>Value/remark</u>
Message Type	"RADIO BEARER RECONFIGURATION FAILURE"
Failure cause	<u>"cell reselection"</u>
Other information element	Not checked

8.2.4.17.5 Test requirement

After step 3 the UE shall discover the presence of cell 2, enter CELL_FACH state, and initiate a cell update procedure in that cell.

After step 5 UE shall transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

After step 6 UE shall transmit TRANSPORT CHANNEL FAILURE message on the DCCH using AM RLC, setting IE "failure cause" to "cell reselection".

8.2.4.18 Transport Channel Reconfiguration from CELL_DCH to CELL_DCH: Success (Subsequently received)

8.2.4.18.1 Definition

8.2.4.18.2 Conformance requirement

If the UE receives a TRANSPORT CHANNEL RECONFIGURATION message before the UE configures the radio bearer according to the previous TRANSPORT CHANNEL RECONFIGURATION message, the UE shall ignore the new TRANSPORT CHANNEL RECONFIGURATION message and configure according to the first TRANSPORT CHANNEL RECONFIGURATION message received. Finally, the UE shall transmit TRANSPORT CHANNEL RECONFIGURATION COMPLETE message on the DCCH using AM RLC.

Reference

3GPP TS 25.331 clause 8.2.4

8.2.4.18.3 Test purpose

To confirm that if the UE receives a TRANSPORT CHANNEL RECONFIGURATION message before the UE configures the radio bearer according to the previous TRANSPORT CHANNEL RECONFIGURATION message it ignores the second TRANSPORT CHANNEL RECONFIGURATION message and configures according to the previous TRANSPORT CHANNEL RECONFIGURATION message.

8.2.4.18.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. When the SS transmits a TRANSPORT CHANNEL RECONFIGURATION message to the UE before the UE configures the radio bearer, the UE ignores the new TRANSPORT CHANNEL RECONFIGURATION message and configures according to the previous TRANSPORT CHANNEL RECONFIGURATION message. Finally, the UE shall transmit TRANSPORT CHANNEL RECONFIGURATION COMPLETE message on the DCCH using AM RLC.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1		←	TRANSPORT CHANNEL RECONFIGURATION	Including IE "Uplink DPCH info"
2		←	TRANSPORT CHANNEL RECONFIGURATION	SS send this message before the expiry of "activation time" specified in TRANSPORT CHANNEL SETUP message of step 1.
3		→	TRANSPORT CHANNEL RECONFIGURATION COMPLETE	The UE ignores the TRANSPORT CHANNEL RECONFIGURATION message in step 2 and confirms configuration according to the TRANSPORT CHANNEL RECONFIGURATION message in step 1.

Specific Message Contents

The contents of TRANSPORT CHANNEL RECONFIGURATION messages in this test case is identical to the message sub-type title "Packet to CELL_DCH from CELL_DCH in PS" found in Annex A with the following exceptions:

TRANSPORT CHANNEL RECONFIGURATION (Step 1)

Information Element	Value/remark
Activation Time	[256+Current CFN-[current CFN mod 8 + 8]]MOD 256

8.2.4.18.5 Test requirement

After step 3 the UE shall communicate with the SS on the radio bearer specified in the TRANSPORT CHANNEL RECONFIGURATION message in step 1.

8.2.4.19 Transport Channel Reconfiguration from CELL_FACH to CELL_DCH: Success (Subsequently received)

8.2.4.19.1 Definition

8.2.4.19.2 Conformance requirement

If the UE receives a TRANSPORT CHANNEL RECONFIGURATION message before the UE configures the radio bearer according to the previous TRANSPORT CHANNEL RECONFIGURATION message, the UE shall ignore the new TRANSPORT CHANNEL RECONFIGURATION message and configure according to the first TRANSPORT CHANNEL RECONFIGURATION message received. Finally, the UE shall transmit TRANSPORT CHANNEL RECONFIGURATION COMPLETE message on the DCCH using AM RLC.

Reference

3GPP TS 25.331 clause 8.2.4

8.2.4.19.3 Test purpose

To confirm that if the UE receives a TRANSPORT CHANNEL RECONFIGURATION message before the UE configures the radio bearer according to the previous TRANSPORT CHANNEL RECONFIGURATION message it ignores the second TRANSPORT CHANNEL RECONFIGURATION message and configures according to the previous TRANSPORT CHANNEL RECONFIGURATION message.

8.2.4.19.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. When the SS transmits a TRANSPORT CHANNEL RECONFIGURATION message to the UE before the UE configures the radio bearer, the UE ignores the new TRANSPORT CHANNEL RECONFIGURATION message and configures according to the previous TRANSPORT CHANNEL RECONFIGURATION message. Finally, the UE shall transmit TRANSPORT CHANNEL RECONFIGURATION COMPLETE message on the DCCH using AM RLC.

Expected sequence

Step	Dire	ction	Message	Comment
	UE	SS		
1		←	TRANSPORT CHANNEL RECONFIGURATION	Includes the IE "Uplink DPCH info"
2		←	TRANSPORT CHANNEL RECONFIGURATION	Sent before the elapse of the activation time specified in step 1.
3		→	TRANSPORT CHANNEL RECONFIGURATION COMPLETE	The UE ignores the TRANSPORT CHANNEL RECONFIGURATION message in step 2 and confirms configuration according to the TRANSPORT CHANNEL RECONFIGURATION message in step 1.

Specific Message Contents

TRANSPORT CHANNEL RECONFIGURATION (Step 1)

The contents of TRANSPORT CHANNEL RECONFIGURATION message in this test case is identical to the message sub-type title "Packet to CELL_DCH from CELL_FACH in PS" found in Annex A with the following exceptions:

Information Element	Value/remark
Activation Time Info	[256+Current CFN-[current CFN mod 8 + 8]]MOD 256

8.2.4.19.5 Test requirement

After step 3 the UE shall communicate with the SS on the radio bearer specified in the TRANSPORT CHANNEL RECONFIGURATION message in step 1.

8.2.4.2820 Transport Channel Reconfiguration from CELL_DCH to CELL_PCH: Success

8.2.4.2820.1 Definition

8.2.4.2820.2 Conformance requirement

The UE shall transmits TRANSPORT CHANNEL RECONFIGURATION COMPLETE message using and transits from CELL_DCH to CELL_PCH when receives a TRANSPORT CHANNEL RECONFIGURATION message. And then, the UE shall reconfigure a radio bearer according to the TRANSPORT CHANNEL RECONFIGURATION message.

Reference

3GPP TS 25.331 clause 8.2.2

8.2.4.<u>2820</u>.3 Test purpose

To confirm that the UE transmits TRANSPORT CHANNEL RECONFIGURATION COMPLETE message before entering CELL_PCH state after it received a TRANSPORT CHANNEL RECONFIGURATION message and reconfigured its radio bearers. The UE is in CELL_PCH state of the same cell.

8.2.4.2820.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_DCH(state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a TRANSPORT CHANNEL RECONFIGURATION message. The UE transmits TRANSPORT CHANNEL RECONFIGURATION COMPLETE message to the UE using AM RLC and enters into CELL_PCH state. The SS transmits a PAGING TYPE 1 message and the UE accepts it and enters the CELL_FACH state.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1		←	TRANSPORT CHANNEL RECONFIGURATION	
2	→		TRANSPORT CHANNEL RECONFIGURATION COMPLETE	The UE sends this message before start state transition.
3				Reconfiguration of Transport channel after state transition.
4		-	PAGING TYPE 1	The SS transmits this message included a matched identity.
5	\rightarrow		CELL UPDATE	The UE is in CELL_FACH state.

Specific Message Contents

TRANSPORT CHANNEL RECONFIGURATION (Step 1)

Use the same message sub-type titled "Packet to CELL_FACH from CELL_DCH in PS" in Annex A with following exceptions:

Information Element	Value/remark
DRX indicatorRRC State Indicator	DRX with CELL updatingCELL PCH

PAGING TYPE 1 (Step 4)

Use the same message sub-type titled "TM (SMS in PS)" in Annex A with following exceptions:

Information Element	Value/remark
Paging record list	
Paging record	
- CHOICE Paging originator	UTRAN originator
- U-RNTI	
- SRNC Identity	Previously assigned SRNC identity
- S-RNTI	Previously assigned S-RNTI

8.2.4.28<u>20</u>.5 Test requirement

After step 1 the UE transmits TRANSPORT CHANNEL RECONFIGURATION COMPLETE message to the UE on uplink DCCH using AM RLC.

After step 3 the UE shall transit from CELL_DCH to CELL_PCH.

8.2.4.2921 Transport Channel Reconfiguration from CELL_DCH to URA_PCH: Success

8.2.4.2921.1 Definition

8.2.4.2921.2 Conformance requirement

The UE shall transmits TRANSPORT CHANNEL RECONFIGURATION COMPLETE message using and transits from CELL_DCH to URA_PCH when receives a TRANSPORT CHANNEL RECONFIGURATION message. And then, the UE shall reconfigure radio bearers according to the TRANSPORT CHANNEL RECONFIGURATION message.

Reference

3GPP TS 25.331 clause 8.2.2

8.2.4.2921.3 Test purpose

To confirm that the UE transmits TRANSPORT CHANNEL RECONFIGURATION COMPLETE message before entering URA_PCH state after it received a TRANSPORT CHANNEL RECONFIGURATION message and reconfigured its radio bearers. The UE is in URA_PCH state of the same cell.

8.2.4.2921.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_DCH(state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a TRANSPORT CHANNEL RECONFIGURATION message. The UE transmits TRANSPORT CHANNEL RECONFIGURATION COMPLETE message to the UE using AM RLC and enters into URA_PCH state. The SS transmits a PAGING TYPE 1 message and the UE accepts it and enters the CELL_FACH state.

Expected sequence

Step	Dire	ction	Message	Comment
	UE	SS		
1		←	TRANSPORT CHANNEL RECONFIGURATION	
2		\rightarrow	TRANSPORT CHANNEL	The UE sends this message
			RECONFIGURATION COMPLETE	before start state transition.
3				Reconfiguration of
				Transport channel after state
				transition.
4		-	PAGING TYPE 1	The SS transmits this message
				included a matched identity.
5		\rightarrow	Cell UPDATE	The UE is in CELL_FACH state.

Specific Message Contents

TRANSPORT CHANNEL RECONFIGURATION (Step 1)

Use the same message sub-type titled "Packet to CELL_FACH from CELL_DCH in PS" in Annex A with following exceptions:

Information Element	Value/remark
DRX indicatorRRC State Indicator	DRX with URA updatingURA PCH

PAGING TYPE 1 (Step 4)

Use the same message sub-type titled "TM (SMS in PS)" in Annex A with following exceptions:

Information Element	Value/remark
Paging record list	
Paging record	
- CHOICE Paging originator	UTRAN originator
- U-RNTI	
- SRNC Identity	Previously assigned SRNC identity
- S-RNTI	Previously assigned S-RNTI

8.2.4.2921.5 Test requirement

After step 1 the UE transmits TRANSPORT CHANNEL RECONFIGURATION COMPLETE message to the UE on uplink DCCH using AM RLC.

After step 3 the UE shall transit from CELL_DCH to URA_PCH.

8.2.4.3022 <u>Void</u>Transport Channel Reconfiguration from CELL_FACH to CELL_PCH: Success

8.2.4.30.1 Definition

8.2.4.30.2 Conformance requirement

The UE shall transmits TRANSPORT CHANNEL RECONFIGURATION COMPLETE message using and transits from CELL_FACH to CELL_PCH when receives a TRANSPORT CHANNEL RECONFIGURATION message. And

then, the UE shall reconfigure radio bearers according to the TRANSPORT CHANNEL RECONFIGURATION message.

Reference

3GPP TS 25.331 clause 8.2.2

8.2.4.30.3 Test purpose

To confirm that the UE transmits TRANSPORT CHANNEL RECONFIGURATION COMPLETE message before entering CELL_PCH state after it received a TRANSPORT CHANNEL RECONFIGURATION message and reconfigured its radio bearers. The UE is in CELL_PCH state of the same cell.

8.2.4.30.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS DCCH+DTCH_FACH(state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a TRANSPORT CHANNEL RECONFIGURATION message. The UE transmits TRANSPORT CHANNEL RECONFIGURATION COMPLETE message to the UE using AM RLC and enters into CELL_PCH state. The SS transmits a PAGING TYPE 1 message and the UE accepts it and enters the CELL_FACH state again.

Expected sequence

Step	Dire	ction	Message Message	Comment
	UE	SS		
1	_		TRANSPORT CHANNEL	
			RECONFIGURATION	
2	_	\rightarrow	TRANSPORT CHANNEL	The UE sends this message
			RECONFIGURATION COMPLETE	before start state transition.
3				Reconfiguration of Transport
				Channel after state transition.
4	_	(PAGING TYPE 1	The SS transmits this message
				included a matched identity.
5	_	\rightarrow	CELL UPDATE	The UE is in CELL_FACH state.

Specific Message Contents

TRANSPORT CHANNEL RECONFIGURATION (Step 1)

Use the same message sub-type titled "Packet to CELL_FACH from CELL_FACH in PS" in Annex A with following exceptions:

Information Element	Value/remark
DRX indicator	DRX with CELL updating

PAGING TYPE 1 (Step 4)

Use the same message sub-type titled "TM (SMS in PS)" in Annex A with following exceptions:

Information Element	Value/remark
Paging record list	
Paging record	
- CHOICE Paging originator	UTRAN originator
— - U-RNTI	
	Previously assigned SRNC identity
	Previously assigned S-RNTI

8.2.4.30.5 Test requirement

After step 1 the UE transmit TRANSPORT CHANNEL RECONFIGURATION COMPLETE message to the UE on uplink DCCH using AM RLC.

After step 3 the UE shall transit from CELL FACH to CELL PCH.

8.2.4.3123 Transport Channel Reconfiguration from CELL_FACH to URA_PCH: Success

8.2.4.31.1 Definition

8.2.4.31.2 Conformance requirement

The UE shall transmits TRANSPORT CHANNEL RECONFIGURATION COMPLETE message using and transits from CELL_FACH to URA_PCH when receives a TRANSPORT CHANNEL RECONFIGURATION message. And then, the UE shall reconfigure radio bearers according to the TRANSPORT CHANNEL RECONFIGURATION message.

Reference

3GPP TS 25.331 clause 8.2.2

8.2.4.31.3 Test purpose

To confirm that the UE transmits TRANSPORT CHANNEL RECONFIGURATION COMPLETE message before entering URA_PCH state after it received a TRANSPORT CHANNEL RECONFIGURATION message and reconfigured its radio bearers. The UE is in URA_PCH state.

8.2.4.31.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS DCCH+DTCH_FACH(state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a TRANSPORT CHANNEL RECONFIGURATION message. The UE transmits TRANSPORT CHANNEL RECONFIGURATION COMPLETE message to the UE using AM RLC and enters into URA_PCH state. The SS transmits a PAGING TYPE 1 message and the UE accepts it and enters the CELL_FACH state again.

Expected sequence

Step	Dire	ction	Message Message	Comment
	UE	SS		
4		(TRANSPORT CHANNEL RECONFIGURATION	
2		>	TRANSPORT CHANNEL RECONFIGURATION COMPLETE	The UE sends this message before start state transition.
3				Reconfiguration of Transport Channel after state transition.
4		(PAGING TYPE 1	The SS transmits this message included a matched identity.
5	_	\rightarrow	CELL UPDATE	The UE is in CELL_FACH state.

Specific Message Contents

TRANSPORT CHANNEL RECONFIGURATION (Step 1)

Use the same message sub-type titled "Packet to CELL_FACH from CELL_FACH in PS" in Annex A with following exceptions:

Information Element	Value/remark
DRX indicator	DRX with URA updating

PAGING TYPE 1 (Step 4)

Use the same message sub-type titled "TM (SMS in PS)" in Annex A with following exceptions:

Information Element	Value/remark
Paging record list	
Paging record	
CHOICE Paging originator	UTRAN originator
— - U-RNTI	
	Previously assigned SRNC identity
S-RNTI	Previously assigned S-RNTI

8.2.4.31.5 Test requirement

After step 1 the UE transmit TRANSPORT CHANNEL RECONFIGURATION COMPLETE message to the UE on uplink DCCH using AM RLC.

After step 3 the UE shall transit from CELL_FACH to URA_PCH. Void

8.2.5 Transport format combination control

8.2.5.1 Transport format combination control in CELL_DCH: restriction

8.2.5.1.1 Definition

8.2.5.1.2 Conformance requirement

The UE shall change the subset of allowed transport format combination of uplink when the UE receives TRANSPORT FORMAT COMBINATION CONTROL message.

Reference

3GPP TS 25.331 clause 8.2.5

8.2.5.1.3 Test purpose

To confirm that the UE do not transmit data on the DTCH in the uplink direction, following the reception of TRANSPORT FORMAT COMBINATION CONTROL message sent from the SS, which is set to the value "zero" in IE "<a href="https://www.minimum.allowed_Allowed_Transport format combination index".

8.2.5.1.4 Method of test

Initial Condition

System Simulator: 1cell

UE: DCCH+DTCH_DCH (state 6-9 or state 6-10) as specified in clause 7.4 of TS 34.108. The exact procedure to apply dependings on the CN domain(s) supported by the UE.

Test Procedure

The UE establishes a radio access bearer on the DCH for a communication. The SS transmits a TRANSPORT FORMAT COMBINATION CONTROL message, which indicates that only TFC₀ is allowed on the uplink for DCH transport channel. The UE shall reconfigure the TFCS, stop any transmission on DTCH logical channel and then continues the communication on DCCH only.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1				UE s in CELL_DCH state with a DTCH logical channel allocated for communication between UE and SS
2		+	TRANSPORT FORMAT COMBINATION CONTROL	The UE shall use the TFC Subset as defined in value IE "Allowed Set to "zero" in IE "Minimum allowed Transport format combination index".
3				The UE shall not transmit any data on the DTCH.

Specific Message Contents

TRANSPORT FORMAT COMBINATION CONTROL

Information Element	Value/remark
TrCH information elements	
-Minimum a Allowed Transport format combination	Set to "zero"
indexlist	
- Allowed transport format combination	0 and 3(If initial state is "state 6-9")
- Allowed transport format combination	0 and 5(If initial state is "state 6-10")

8.2.5.1.5 Test requirement

After step 2 the UE shall stop transmitting data on the DTCH in the uplink.

8.2.5.2 Transport format combination control in CELL_DCH: release a restriction

8.2.5.2.1 Definition

8.2.5.2.2 Conformance requirement

The UE shall change the subset of allowed transport format combination of uplink when it receives TRANSPORT FORMAT COMBINATION CONTROL message, specifying that an existing restriction for the usage of TFCS be removed.

Reference

3GPP TS 25.331 clause 8.2.5

8.2.5.2.3 Test purpose

To confirm that the UE resume transmission of data on the DTCH on the uplink, following the reception of TRANSPORT FORMAT COMBINATION CONTROL message which <u>include</u> is set to "one" as the prior used transport format combination number in IE "Minimum allowed Transport format combination indexset".

8.2.5.2.4 Method of test

Initial Condition

System Simulator: 1cell

UE: DCCH+DTCH_DCH (state 6-9 or state 6-10) as specified in clause 7.4 of TS 34.108. The exact procedure to apply dependings on the CN domain(s) supported by the UE.

Test Procedure

The UE is in CELL_DCH state with DTCH allocated but fully restricted. The UE cannot transmit the data on the DTCH, as a result of the restriction on the transport format combination. Next, the SS transmits a TRANSPORT FORMAT COMBINATION CONTROL message which is set to "one" in IEinclude "Minimum allowed Transport transport format combination indexset"

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1				No data transmission on the DTCH with a restriction in the uplink direction, following the execution of test 84.2.5.1.
2		←	TRANSPORT FORMAT COMBINATION CONTROL	Use the TFCS according to Set to "one" IE "Minimum allowed Transport format combination index".
3				The UE begins to transmit the data on the DTCH.

Specific Message Contents

TRANSPORT FORMAT COMBINATION CONTROL

Information Element	Value/remark
TrCH information elements	
-Minimum allowed t∓ransport format combination	5(If initial state is "state 6-9")Set to "one"
indexset	
-Minimum allowed transport format combination set	6(If initial state is "state 6-10")

8.2.5.2.5 Test requirement

After step 2 the UE shall begin to transmit the data on the DTCH in the uplink.

8.2.5.3 Transport format combination control in CELL_DCH: Failure (Incompatible simultaneous reconfiguration)

8.2.5.3.1 Definition

8.2.5.3.2 Conformance requirement

The UE shall keep its old configuration when the UE receives another TRANSPORT FORMAT COMBINATION CONTROL message before the UE reconfigures the transport channel completely according to a similar message received earlier. The UE shall transmit a TRANSPORT FORMAT COMBINATION CONTROL FAILURE message on the DCCH using AM RLC indicating "incompatible simultaneous reconfiguration" in IE "failure cause".

Reference

3GPP TS 25.331 clause 8.2.5

8.2.5.3.3 Test purpose

To confirm that after the UE receives TRANSPORT FORMAT COMBINATION CONTROL message, it transmits TRANSPORT FORMAT COMBINATION CONTROL FAILURE message and keeps the TFC subset as before the TRANSPORT FORMAT COMBINATION CONTROL message is received.

8.2.5.3.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE establishes a radio access bearer on the DCH for to be used for user-data exchange. SS sends a TRANSPORT CHANNEL RECONFIGURATION message on the downlink DCCH, to request that the channel coding scheme for a DCH be changed. After this message has been acknowledged by the UE RLC-AM entity, the SS transmits a TRANSPORT FORMAT COMBINATION CONTROL message, which includes a full restriction of the TFCS used in the uplink. The UE shall detect a failure to reconfigure the TFCS, then it transmits TRANSPORT FORMAT COMBINATION CONTROL FAILURE message on the uplink DCCH. After the activation time specified in the TRANSPORT CHANNEL RECONFIGURATION message has elapsed, the UE shall send TRANSPORT CHANNEL RECONFIGURATION COMPLETE message on the uplink DCCH. SS verifies that reconfiguration is completed by checking that the user-data exchange is resumed on DTCH.

Expected sequence

Step	Dire	ction	Message	Comment
	UE	SS		
1				UE is in CELL_DCH connected state, with a DTCH logical channel for user-data communication
2		←	TRANSPORT CHANNEL RECONFIGURATION	Requesting for a change in semi-static transport format for DCH carrying the DTCH. The dynamic part remains unchanged.
3		←	TRANSPORT FORMAT COMBINATION CONTROL	Requesting for a full restriction on TFCS for the DCH carrying DTCH.
4		→	TRANSPORT FORMAT COMBINATION CONTROL FAILURE	The UE shall keep the TFC subset as before the TRANSPORT FORMAT COMBINATION CONTROL message was received
5				The UE does not change the configuration of TFC and the UE continues reconfigure the affected transport channel.
6			TRANSPORT CHANNEL RECONFIGURATION COMPLETE	UE shall resume exchange of data over the DTCH logical channel.

Specific Message Contents

TRANSPORT CHANNEL RECONFIGURATION

The contents of TRANSPORT CHANNEL RECONFIGURATION message in this test case is identical to the message sub-type title "Packet to CELL_DCH from CELL_DCH in PS" found in Annex A with the following exceptions:

Information Element	Value/remark
TrCH Information Elements	
- Uplink transport channels	
 Added or reconfigured TrCH information list 	
- Transport channel identity	2
- Semi-Static Transport Format Information	
- Type of channel coding	Select a different coding scheme from default message
	content

TRANSPORT FORMAT COMBINATION CONTROL

Information Element	Value/remark
DPCH TFCS in Uplink	Restricted TrCH information
- Subset Representation	Not Present (All TFCs are restricted)
- Allowed TFIs	

TRANSPORT FORMAT COMBINATION CONTROL FAILURE

Information Element	Value/remark	
Message Type	"TRNSPORT FORMAT COMBINATION CNTROL	
	FAILURE"	
RRC transaction identifier	<u>0</u>	
Failure cause	Incompatible simultaneous reconfiguration	
Other information element	Not checked	

8.2.5.3.5 Test requirement

After step 3 the UE continue the transport channel reconfiguration as if no TRANSPORT FORMAT COMBINATION CONTROL message was received. Then it shall transmit a TRANSPORT FORMAT COMBINATION CONTROL FAILURE message on the DCCH using AM RLC, stating the reason "Incompatible simultaneous reconfiguration" in IE "Failure cause".

After step 6 the UE shall resume communication with SS on DTCH using the requested channel coding scheme on the transport blocks.

8.2.5.4 Transport format combination control in CELL_DCH: Failure (Invalid message reception and Invalid configuration)

8.2.5.4.1 Definition

8.2.5.4.2 Conformance requirement

The UE shall keep old configuration when it receives a TRANSPORT FORMAT COMBINATION CONTROL message which using a spare_undefined value in the mandatory IE "Minimum allowed Transport format combination index". It shall then transmits a TRANSPORT FORMAT COMBINATION CONTROL FAILURE message, indicating "protocol error" in IE "failure cause" and "Information element value not comprehended" in IE "Protocol error cause". The UE shall keep existing configuration before reception of a TRANSPORT FORMAT COMBINATION CONTROL message when the TRANSPORT CHANNEL RECONFIGURATION message include some IEs set to invalid value, and then the UE shall transmit TRANSPORT CHANNEL RECONFIGURATION FAILURE including IE "failure cause" set to "invalid configuration".

Reference

3GPP TS 25.331 clause 8.2.5

8.2.5.4.3 Test purpose

To confirm after the UE receives an invalid TRANSPORT FORMAT COMBINATION CONTROL message, it transmits TRANSPORT FORMAT COMBINATION CONTROL FAILURE message and keep the TFC subset as if no TRANSPORT FORMAT COMBINATION CONTROL message has been received.

To confirm that the UE transmits a TRANSPORT FORMAT COMBINATION CONTROL FAILURE message on the DCCH using AM RLC if it receives a TRANSPORT FORMAT COMBINATION CONTROL message including some IEs set to invalid value.

8.2.5.4.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: DCCH+DTCH_DCH (state 6-9 or state 6-10) as specified in clause 7.4 of TS 34.108. The exact procedure to apply dependings on the CN domain(s) supported by the UE.

Test Procedure

The UE establishes a radio access bearer on the DCH for a communication. The SS transmits a TRANSPORT FORMAT COMBINATION CONTROL message, which uses a spare-undefined value in the mandatory IE "Minimum allowed Transport format combination index". The UE shall then transmit TRANSPORT FORMAT COMBINATION CONTROL FAILURE message and continues the communication using the radio access bearer. The UE keeps initial configuration and SS transmits TRANSPORT FORMAT COMBINATION CONTROL message including some IEs set

Expected sequence

Step	Dire	ction	Message	Comment
	UE	SS		
1				RRC connected state on the DTCH for a communication
2	+		TRANSPORT FORMAT COMBINATION CONTROL	Contains an error in a mandatory IE.
3		\rightarrow	TRANSPORT FORMAT COMBINATION CONTROL FAILURE	The UE shall not change the configuration
<u>4</u>	←		TRANSPORT FORMAT COMBINATION CONTROL	This message includes IE set to invalid value
<u>5</u>				The UE does not change the configuration
<u>6</u>	<u></u>		TRANSPORT FORMAT COMBINATION CONTROL FAILURE	The IE "failure cause" shall be set to "invalid configuration

Specific Message Contents

TRANSPORT FORMAT COMBINATION CONTROL (Step 2)

Information Element	Value/remark
DPCH TFCS in uplink	Set to the value "MaxTFCValue"
- Minimum allowed Transport format combination index	

TRANSPORT FORMAT COMBINATION CONTROL FAILURE (Step 3)

Information Element	Value/remark	
Message Type	"TRNSPORT FORMAT COMBINATION CNTROL	
	FAILURE"	
Failure cause	"protocol error"	
Protocol error information		
-Protocol error case	Information element value not comprehended	
Other information element	Not checked	

TRANSPORT FORMAT COMBINATION CONTROL(Step 4)

Information Element	<u>Value/remark</u>
TrCH information elements	
 Allowed Transport format combination list 	
- Allowed transport format combination	10

TRANSPORT FORMAT COMBINATION CONTROL FAILURE (Step 6)

Information Element	<u>Value/remark</u>
Message Type	
Failure cause	Invalid configuration
Other information element	Not checked

8.2.5.4.5 Test requirement

After step 3 the UE shall keep its configuration before the TRANSPORT FORMAT COMBINATION CONTROL message was received and transmit a TRANSPORT FORMAT COMBINATION CONTROL FAILURE message on the DCCH using AM RLC. The UE shall set the value "protocol error" in IE "Failure cause" and the value "information element not comprehended" in IE "protocol error information". The UE shall continue communicate with SS using the radio access bearer.

After step 4 the UE shall keep its old configuration.

After step 5 the UE shall transmit TRANSPORT FORMAT COMBINATION CONTROL FAILURE message on the DCCH using AM RLC, setting IE "failure cause" to "invalid configuration".

8.2.6 Physical channel reconfiguration

8.2.6.1 Physical channel reconfiguration for transition from CELL_DCH to CELL_DCH (Hard handover to another frequency): Success

8.2.6.1.1 Definition

8.2.6.1.2 Conformance requirement

The UE shall correctly reconfigure a physical channel according to the PHYSICAL CHANNEL RECONFIGURATION message received, which is used for hard handover purposes. It shall be able to communicate with the UTRAN on the new frequency subsequently.

Reference

3GPP TS 25.331 clause 8.2.6

8.2.6.1.3 Test purpose

To confirm that the UE reconfigures the physical channel parameters according to a PHYSICAL CHANNEL RECONFIGURATION message received from the SS. After the reconfiguration, the UE shall resume normal transmission and reception operations.

8.2.6.1.4 Method of test

Initial Condition

System Simulator: 2 cells - cell 1 and cell 6 are active. The $Q_{rxlevmin}$ and $Q_{rxqualmin}$ values in SYSTEM INFORMATION BLOCK TYPE 3 and 4 messages of cell 6 is changed to -90dBm and -20dB respectively

UE: PS-DCCH+DTCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108 in cell 1

Test Procedure

The UE is in the CELL_DCH state in cell 1. The SS transmits a PHYSICAL CHANNEL RECONFIGURATION message to the UE, which includes a new physical channel parameter specified in the "Frequency Info" IE. The UE shall reconfigure itself and tune to the new physical channel and transmits a PHYSICAL CHANNEL RECONFIGURATION COMPLETE message on the uplink DCCH of cell 6 using AM RLC.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	+	PHYSICAL CHANNEL	Including new frequency
		RECONFIGURATION	information.
2			UE shall suspend stop uplink activities to cell 1 and begin to reconfigure the physical channel parameters.
3	\rightarrow	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	

Specific Message Contents

The contents of PHYSICAL CHANNEL RECONFIGURATION message in this test case is identical to the message sub-type titled "Packet to CELL_DCH from CELL_DCH in PS" in Annex A, with the following exceptions:

Information Element	Value/remark
Frequency info	
- UARFCN uplink(Nu)	Same uplink UARFCN as used for cell 6
- UARFCN downlink(Nd)	Same downlink UARFCN as used for cell 6
Downlink information for each radio links	Same downlink UARFCN as used for cell 6
- Primary CPICH info	
- Primary Scrambling Code	350
Downlink information common for all radio links	
- Downlink DPCH info common for all RL	
- Timing Indicator	Initialise

8.2.6.1.5 Test requirement

After step 2 the UE shall send PHYSICAL CHANNEL RECONFIGURATION COMPLETE message on the uplink DCCH of cell 6, and then continue to communicate with SS on the new physical channel.

8.2.6.2 Physical channel reconfiguration for transition from CELL_DCH to CELL_DCH (Hard handover to another frequency): Failure (Unsupported configuration)

8.2.6.2.1 Definition

8.2.6.2.2 Conformance requirement

The UE shall keep its old configuration when the UE receives a PHYSICAL CHANNEL RECONFIGURATION message which includes an unsupported configuration and transmit a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, with the reason "configuration unsupported" in IE "failure cause".

Reference

3GPP TS 25.331 clause 8.2.6

8.2.6.2.3 Test purpose

To confirm that the UE keeps its configuration and transmits a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC if the received PHYSICAL CHANNEL RECONFIGURATION message includes unsupported configuration parameters for the UE.

8.2.6.2.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a PHYSICAL CHANNEL RECONFIGURATION message to the UE which includes unsupported configuration parameters as the frequency cannot be supported by the UE. The UE transmits a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC which is set to "configuration unsupported" in IE "failure cause".

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1		←	PHYSICAL CHANNEL RECONFIGURATION	Includes an unsupported configuration as the frequency cannot be supported by the UE
2		→	PHYISICAL CHANNEL RECONFIGURATION FAILURE	The UE shall not change the physical channel and continue to communicate using the old configuration.

Specific Message Contents

PHYSICAL CHANNEL RECONFIGURATION

The contents of PHYSICAL CHANNEL RECONFIGURATION message in this test case is identical to the message sub-type title "Packet to CELL_DCH from CELL_DCH in PS" found in Annex A with the following exceptions:

Information Element	Value/remark
Frequency info	
- UARFCN uplink(Nu)	63984
- UARFCN downlink(Nd)	Not Present

PHYSICAL CHANNEL RECONFIGURATION FAILURE

Information Element	Value/remark
Message Type	
Failure cause	Configuration unsupported
Other information element	Not checked

8.2.6.2.5 Test requirement

After step 1 the UE shall keep its old configuration and transmit a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC and set "configuration unsupported" in IE "failure cause".

8.2.6.3 Physical channel reconfiguration for transition from CELL_DCH to CELL_DCH (Hard handover to another frequency): Failure (Physical channel failure and reversion to old channel)

8.2.6.3.1 Definition

8.2.6.3.2 Conformance requirement

The UE shall revert to the old configuration when the UE fails to reconfigure the new physical channel by the expiry of timer T312, and then transmit a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC indicating "physical channel failure" in IE "failure cause".

Reference

3GPP TS 25.331 clause 8.2.6

8.2.6.3.3 Test purpose

To confirm that the UE reverts to the old configuration and transmits a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC if the UE fails to reconfigure the new physical channel according to the received PHYSICAL CHANNEL RECONFIGURATION message by timer T312 expiry.

8.2.6.3.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH DCH (state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a PHYSICAL CHANNEL RECONFIGURATION message to the UE which includes new frequency parameters. However, the SS does not reconfigure the new physical channel. The UE is expected to encounter a failure to reconfigure the new physical channel and after T312 timer expiry the UE shall revert to the old configuration. Finally, the UE transmits a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC specifies "physical channel failure" in IE "failure cause".

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1		←	PHYSICAL CHANNEL	Including a new frequency
			RECONFIGURATION	information
2				The SS does not reconfigure the physical channel so that the UE fails to reconfigure to the new physical channel.
3		\rightarrow	PHYSICAL CHANNEL RECONFIGURATION FAILURE	After T312 expiry, the UE shall revert to the old configuration and transmits this message.

Specific Message Contents

PHYSICAL CHANNEL RECONFIGURATION

Use the message sub-type titled "Packet to CELL_DCH from CELL_DCH in PS" in Annex A.

PHYSICAL CHANNEL RECONFIGURATION FAILURE

Information Element	Value/remark
Message Type	
Failure cause	Physical channel failure
Other information element	Not checked

8.2.6.3.5 Test requirement

After step 2 the UE shall revert to the old configuration and transmit a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, with the value "physical channel failure" in IE "failure cause".

8.2.6.4 Physical channel reconfiguration for transition from CELL_DCH to CELL_DCH (Hard handover to another frequency): Failure (Physical channel failure and reversion failure)

8.2.6.4.1 Definition

8.2.6.4.2 Conformance requirement

The UE shall perform an RRC connection re establishment cell update procedure when the UE fails to revert to the old configuration, after the detection of physical channel failure during the course of executing a physical channel reconfiguration procedure. After the UE completes cell update procedure, the UE transmits PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC which set IE "failure cause" to "physical channel failure".

Reference

3GPP TS 25.331 clause 8.2.6

8.2.6.4.3 Test purpose

To confirm that the UE <u>transmits PHYSICAL CHANNEL RECONFIGURATION FAILURE message after UE completes initiates an a cell update RRC connection re establishment procedure when the UE cannot reconfigure the new physical channel for the failure of L1 configuration and for the failure of the reversion to the old configuration.</u>

8.2.6.4.4 Method of test

Initial Condition

System Simulator: 2 cells- Cell 1 is active, Cell 2 is inactive

UE: PS-DCCH+DTCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108 in cell 1

Test Procedure

The UE is in the CELL_DCH state in cell 1. The SS begins to broadcast the BCCH in cell 2 and then transmits a PHYSICAL CHANNEL RECONFIGURATION message to the UE, which includes new uplink and downlink frequency parameters and does not reconfigure L1. The UE is expected to fail to reconfigure the new physical channel and tries to revert to the old configuration—after T312 expiry. The SS then deletes the old physical channel so that the UE would perform cell reselection and finds cell 2. It shall then transmit and CELL UPDATE message on uplink CCCH with IE "Cell update cause" set to "radio link failure". The SS shall transmit CELL UPDATE CONFIRM message on downlink CCCH after receiving CELL UPDATE message. The UE transmits UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC and subsequently transmits PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting the value "physical channel failure" to IE "failure cause".

RRC CONNECTION RE ESTABLISHMENT REQUEST message which includes the IE "U RNTI" on the uplink CCCH. The SS transmits an RRC CONNECTION RE ESTABLISHMENT message which includes a new TFCS according to the new transport channel. After the UE receives an RRC CONNECTION RE-ESTABLISHMENT message, it reconfigures the new radio connection and transmits an RRC CONNECTION RE-ESTABLISHMENT COMPLETE message on the uplink DCCH using AM RLC.

Expected sequence

Step	Dire	ction	Message	Comment
	UE	SS		
1		←	ВССН	The SS transmits the BCCH in cell 2.
2		←	PHYSICAL CHANNEL RECONFIGURATION	The message includes new frequency information
3				SS does not reconfigure the physical channel to reflect the changes in step 2, at the same time, it deletes the old configuration so the UE cannot reconfigure the new physical channel and cannot revert to the old configuration.
4		\rightarrow	CELL UPDATERRC CONNECTION RE-ESTABLISHMENT REQUEST	The UE finds a new cell.2 and enters CELL FACH state. This message includes the value "radio link failure" set in IE "Cell update cause".then transmits this message which includes the IE "U-RNTI".
5		+	CELL UPDATE CONFIRMRRC CONNECTION RE-ESTABLISHMENT	This message includes IE "new U-RNTI" and IE "new C-RNTI". This message indicates a new TFCS according to the new transport channel.
6		\rightarrow	UTRAN MOBILITY INFORMATION CONFIRMREC CONNECTION RE-ESTABLISHMENT COMPLETE	
<u>7</u>	- 1	<u>→</u>	PHYSICAL CHANNEL RECONGURATION FAILURE	The IE "failure cause" shall be set to "physical channel failure"

Specific Message Contents

PHYSICAL CHANNEL RECONFIGURATION (Step 1)

Use the message sub-type titled "Packet to CELL_DCH from CELL_DCH in PS" in Annex A.

CELL UPDATE (Step 4)

The contents of CELL UPDATE message is identical as "Contents of CELL UPDATE message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	
- SRNC Identity	Assigned previously in cell 1
- S-RNTI	Assigned previously in cell 1
Cell Update Cause	"radio link failure"

CELL UPDATE CONFIRM (Step 5)

The contents of CELL UPDATE CONFIRM message is identical as "CELL UPDATE CONFIRM message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	Same as CELL UPDATE message in step 4
New U-RNTI	
- SRNC Identity	<u>'0000 0000 0000 0001'</u>
- S-RNTI	Different from previous S-RNTI
New C-RNTI	Different from previous C-RNTI

PHYSICAL CHANNEL RECONGURATION FAILURE (Step 7)

Information Element	<u>Value/remark</u>
Message Type	"PHYSICAL CHANNEL RECONGURATION FAILURE"
Failure cause	"physical channel failure"
Other information element	Not checked

8.2.6.4.5 Test requirement

After step 3 the UE shall find the availability of cell 2, enter CELL_FACH state, and transmits RRC CONNECTION RE ESTABLISHMENTCELL UPDATE message using RLC-TM mode on the uplink DCCHCCCH.

After step 6 the UE shall transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

After step 7 the UE shall transmit PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting the IE "failure cause" to "physical channel failure". the UE shall be able to successfully reestablish an RRC connection in cell 2.

8.2.6.5 Physical channel reconfiguration for transition from CELL_DCH to CELL_DCH (Hard handover to another frequency): Failure (Incompatible simultaneous reconfiguration)

8.2.6.5.1 Definition

8.2.6.5.2 Conformance requirement

If the UE receives a PHYSICAL CHANNEL RECONFIGURATION message whilst reconfiguring due to a radio bearer message other than PHYSICAL CHANNEL RECONFIGURATION SETUP, it shall keep its configuration as if the PHYSICAL CHANNEL RECONFIGURATION SETUP message had not been received. If the UE is reconfiguring itself after receiving a radio bearer message other than PHYSICAL CHANNEL RECONFIGURATION message, and when the UE subsequently receive a PHYSICAL CHANNEL RECONFIGURATION message and the UE shall transmit a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC which is set to "Incompatible simultaneous reconfiguration" in IE "failure cause".

Reference

3GPP TS 25.331 clause 8.2.6

8.2.6.5.3 Test purpose

To confirm that if the UE receives a PHYSICAL CHANNEL RECONFIGURATION message whilst reconfiguring due to a radio bearer message other than PHYSICAL CHANNEL RECONFIGURATION, it shall keep its configuration as if the PHYSICAL CHANNEL RECONFIGURATION message had not been received and complete the reconfiguration according to the previously received message.

To confirm that the UE keeps its current configuration when it receives a PHYSICAL CHANNEL RECONFIGURATION message before the completion of an existing configuration .

8.2.6.5.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL DCH state. The SS transmits a RADIO BEARER RECONFIGURATION message to the UE. The SS transmits a PHYSICAL CHANNEL RECONFIGURATION message before the "activation time" indicated in the RADIO BEARER RECONFIGURATION message expires. When the UE receives the PHYSICAL CHANNEL RECONFIGURATION message, the UE shall keep the configuration as if it had not received the PHYSICAL CHANNEL RECONFIGURATION message and shall transmit a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC with IE "failure cause" set to "incompatible simultaneous reconfiguration". After the SS receives the PHYSICAL CHANNEL RECONFIGURATION FAILURE message, the UE reconfigures the new physical channel parameters and transmits a RADIO BEARER RECONFIGURATION COMPLETE message on DCCH using AM RLC.

The UE is in the CELL_DCH state. When the SS transmits a PHYSICAL CHANNEL RECONFIGURATION message to the UE before it reconfigures the physical channel, the UE keeps the old configuration and transmits an PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC. This message shall indicate "incompatible simultaneous reconfiguration" in IE "failure cause".

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	←	RADIO BEARER RECONFIGURATION	
2	+	PHYSICAL CHANNEL RECONFIGURATION	Sent before the "Activation Time Info" specified in the message in step 1 has elapsed.
3	→	PHYSICAL CHANNEL RECONFIGURATION FAILURE	The UE does not change the configuration due to the reception of PHYSICAL CHANNEL RECONFIGURATION FAILURE message.
<u>4</u>	<u>→</u>	RADIO BEARER RECONFIGURATION COMPLETE	This message is on DCCH using AM RLC.

Specific Message Contents

RADIO BEARER RECONFIGURATION (Step 1)

For RADIO BEARER RECONFIGURATION in step 1, use the message sub-type indicated as "Packet to CELL_DCH from CELL_DCH in PS" found in Annex A.

PHYSICAL CHANNEL RECONFIGURATION (Step 2)

The contents of PHYSICAL CHANNEL RECONFIGURATION message in this test case is identical to the message sub-type title "Packet to CELL_DCH from CELL_DCH in PS" found in Annex A with the following exceptions:

Information Element	Value/remark
Activation Time Info	Current CFN-[current CFN mod 8 + 8]

PHYSICAL CHANNEL RECONFIGURATION FAILURE (step 3)

Information Element	Value/remark
Message Type	
Failure cause	Incompatible simultaneous reconfiguration
Other information element	Not checked

8.2.6.5.5 Test requirement

After step 1, SS transmits a PHYSICAL CHANNEL RECONFIGURATION message before the expiry of the activation time specified in the message of step 1.

After step 2 the UE shall keep its configuration as if the UE had not received the PHYSICAL CHANNEL RECONFIGURATION message and shall transmit a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC with IE "failure cause" set to "Incompatible simultaneous reconfiguration".

After step 4 the UE communicates with the SS on the DCCH and DTCH using the new physical channel parameters reconfigured as a result of the RADIO BEARER RECONFIGURATION message. After step 2 the UE shall keep its old configuration and transmit a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, which shall be set to "incompatible simultaneous reconfiguration" in IE "failure cause".

8.2.6.6 Physical channel reconfiguration for transition from CELL_DCH to CELL_DCH (Hard handover to another frequency): Failure (Invalid message reception and Invalid configuration)

8.2.6.6.1 Definition

8.2.6.6.2 Conformance requirement

The UE shall keep its old configuration when the UE receives a PHYSICAL CHANNEL RECONFIGURATION message, which includes the spare-undefined value in the mandatory IE "UTRAN DRX cycle length coefficient DRX indicator". It shall then transmits a PHYSICAL CHANNEL RECONFIGURATION FAILURE message which contains the value "protocol error" in IE "failure cause" and also "Information element value not comprehended" in IE "Protocol error cause". The UE shall keep existing configuration before reception of a TRANSPORT CHANNEL RECONFIGURATION message when the TRANSPORT CHANNEL RECONFIGURATION message include some IEs set to invalid value, and then the UE shall transmit TRANSPORT CHANNEL RECONFIGURATION FAILURE including IE "failure cause" set to "invalid configuration".

Reference

3GPP TS 25.331 clause 8.2.6

8.2.6.6.3 Test purpose

To confirm that the UE transmits a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC if it receives a PHYSICAL CHANNEL RECONFIGURATION message which uses a spare undefined value in the mandatory IE "UTRAN DRX cycle length coefficientDRX indicator".

To confirm that the UE transmits a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC if it receives a PHYSICAL CHANNEL RECONFIGURATION message including some IEs set to invalid value.

8.2.6.6.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a PHYSICAL CHANNEL RECONFIGURATION message to the UE, with a spare-undefined value in the mandatory IE "UTRAN DRX cycle length coefficientDRX indicator". The UE keeps the old configuration and transmits a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, with a value "protocol error" in IE "failure cause" and also a value "Information element value not comprehended" in IE "Protocol error cause". The UE keeps initial configuration and SS transmits PHYSICAL CHANNEL RECONFIGURATION message including some IEs set to invalid value. The UE transmit PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting the value " invalid configuration" to IE "failure cause".

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1		-	PHYSICAL CHANNEL	This message contains an error in
			RECONFIGURATION	one of the mandatory IEs.
2		\rightarrow	PHYSICAL CHANNEL	The UE does not change the
			RECONFIGURATION FAILURE	configuration.
<u>4</u>	<u> </u>	<u>É</u>	PHYSICAL CHANNEL	This message includes IE set to
			RECONFIGURATION	invalid value
<u>5</u>				The UE does not change the
				<u>configuration</u>
<u>6</u>	_	<u>></u>	PHYSICAL CHANNEL	The IE "failure cause" shall be set
			RECONFIGURATION FAILURE	to "invalid configuration

Specific Message Contents

PHYSICAL CHANNEL RECONFIGURATION (Step 1)

The contents of PHYSICAL CHANNEL RECONFIGURATION message in this test case is identical to the message sub-type title "Packet to CELL_DCH from CELL_DCH in PS" found in Annex A with the following exceptions:

Information Element	Value/remark
UTRAN DRX cycle length coefficientDRX indicator	spare <u>Undefined</u> value

PHYSICAL CHANNEL RECONFIGURATION FAILURE (Step 2)

Information Element	Value/remark
Message Type	
Failure cause	
- Failure cause	Protocol error
- Protocol error information	
- Protocol error cause	Information element value not comprehended
Other information element	Not checked

PHYSICAL CHANNEL RECONFIGURATION (Step 3)

The contents of PHYSICAL CHANNEL RECONFIGURATION message in this test case is identical to the message sub-type title "Packet to CELL_FACH from CELL_DCH in PS" found in Annex A with the following exceptions:

Information Element	<u>Value/remark</u>
Downlink information for each radio links	
- Primary CPICH info	
- Primary scrambling code	<u>100</u>
- PDSCH with SHO DCH info	Not Present
- PDSCH code mapping	Not Present
- Downlink DPCH info for each RL	B: OBIOLI I
- Primary CPICH usage for channel estimation	Primary CPICH may be used
- DPCH frame offset	0 chips
- DL channelisation code	
- Secondary scrambling code	1
- Spreading factor	Reference to TS34.108 clause 6.10 Parameter Set
- Code number	SF-1(SF is reference to TS34.108 clause 6.10 Parameter
	Set)
- Scrambling code change	No change
- TPC combination index	<u>0</u>
- SSDT Cell Identity	<u>-a</u>
- Closed loop timing adjustment mode	Not Present
- Secondary CCPCH info	Not Present
	Not Present
- FACH/PCH information	Not Present
- References to system information blocks	Not Present
 Scheduling information 	

PHYSICAL CHANNEL RECONFIGURATION FAILURE (Step 5)

Information Element	<u>Value/remark</u>
Message Type	
Failure cause	Invalid configuration
Other information element	Not checked

8.2.6.6.5 Test requirement

After step 1 the UE shall keep its old configuration and transmit a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting value "protocol error" in IE "failure cause" and also setting value "Information element value not comprehended" in IE "Protocol error cause".

After step 3 the UE shall keep its old configuration.

After step 4 the UE shall transmit PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting IE "failure cause" to "invalid configuration".

8.2.6.7 Physical channel reconfiguration for transition from CELL_DCH to CELL_FACH: Success

8.2.6.7.1 Definition

8.2.6.7.2 Conformance requirement

The UE shall correctly reconfigure a physical channel according to a PHYSICAL CHANNEL RECONFIGURATION message when asked to perform a transition from CELL_DCH to CELL_FACH.

Reference

3GPP TS 25.331 clause 8.2.6

8.2.6.7.3 Test purpose

To confirm that the UE reconfigures a new physical channel according to the PHYSICAL CHANNEL RECONFIGURATION message received from the SS.

8.2.6.7.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a PHYSICAL CHANNEL RECONFIGURATION message to the UE. The UE shall then reconfigure the new physical channel according to this message and the system information messages. Following this, it shall transmit a PHYSICAL CHANNEL RECONFIGURATION COMPLETE message using AM RLC on the RACH.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1		-	PHYSICAL CHANNEL RECONFIGURATION	
2				Reconfiguration of physical channel
3		\rightarrow	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	

Specific Message Contents

PHYSICAL CHANNEL RECONFIGURATION

Use the message sub-type titled "Packet to CELL_FACH from CELL_DCH in PS" in Annex A.

8.2.6.7.5 Test requirement

After step 3 the UE shall transit from CELL_DCH to CELL_FACH and continue to communicate with SS on the common physical channel.

8.2.6.8 Physical channel reconfiguration for transition from CELL_DCH to CELL_FACH: Failure (Cell re-selectionPhysical channel failure)

8.2.6.8.1 Definition

8.2.6.8.2 Conformance requirement

The UE shall initiate the cell <u>reselection update</u> procedure when the UE cannot use the assigned physical channel according to a PHYSICAL CHANNEL RECONFIGURATION message, during the transition from CELL_DCH to CELL_FACH. After the UE <u>completes</u> cell update procedure, the UE <u>transmits</u> PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC which is set to "cell reselection" in IE "failure cause".

Reference

3GPP TS 25.331 clause 8.2.6

8.2.6.8.3 Test purpose

To confirm that the UE transmits PHYSICAL CHANNEL RECONFIGURATION FAILURE message after the UE perform completes a cell reselection and cell update procedure when the it fail to access the assigned physical channel, according to a PHYSICAL CHANNEL RECONFIGURATION message, during the transition from CELL_DCH to CELL_FACH.

8.2.6.8.4 Method of test

Initial Condition

System Simulator: 2 cells Cell 1 is active, Cell 2 is inactive

UE: PS-DCCH+DTCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state in cell 1. The SS begins to broadcast the BCCH in cell 2 and then transmits a PHYSICAL CHANNEL RECONFIGURATION message, resulting in the UE to attempt to transit from CELL_DCH to CELL_FACH. The UE is expected to fail to access the assigned physical channel as the SS ceases downlink transmission in cell 1. After the UE successfully camp onto cell 2, it shall initiate the cell update procedure in cell 2. The UE transmits CELL UPDATE message on uplink CCCH with IE "Cell update cause" set to "cell reselection". The SS shall transmit CELL UPDATE CONFIRM message on downlink CCCH after receiving CELL UPDATE message. The UE transmits UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC and transmits PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting the value "cell reselection" to IE "failure cause". The procedure completes when the SS confirms UE's cell update request.

Expected sequence

Step	Direction	Message	Comment
-	UE SS	7	
1	-	BCCH	The SS starts to broadcast the
			BCCH in cell 2.
2	+	PHYSICAL CHANNEL	The message assigns common
		RECONFIGURATION	resources so that the UE is
			expected to execute a transition
			from CELL_DCH to
			CELL_FACH
3			The UE shall detect a failure to
			use the assigned common
			physical channel and try to
			perform a cell reselection.
4	\rightarrow	CELL UPDATE	The UE discover the presence
			of cell 2 and triggers a cell
			update procedure. The value
			"cell reselection" shall be set in
			IE "Cell update cause".
5	←	CELL UPDATE CONFIRM	This message include IE "new
			U-RNTI" and IE "new C-RNTI".
			SS sends this message so that
			the UE moves to CELL_PCH
			state.
<u>6</u>	<u>→</u>	UTRAN MOBILITY INFORMATION	
		CONFIRM	
<u>7</u>	<u> </u>	PHYSICAL CHANNEL	The IE "failure cause" shall be
		RECONFIGURATION FAILURE	set to "cell reselection"

Specific Message Contents

PHYSICAL CHANNEL RECONFIGURATION

Use the message sub-type titled "Packet to CELL_FACH from CELL_DCH in PS" in Annex A.

CELL UPDATE (Step 4)

The contents of CELL UPDATE message is identical as "Contents of CELL UPDATE message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	
- SRNC Identity	Assigned previously in cell 1
<u>- S-RNTI</u>	Assigned previously in cell 1
Cell Update Cause	"radio link failure"

CELL UPDATE CONFIRM (Step 5)

The contents of CELL UPDATE CONFIRM message is identical as "CELL UPDATE CONFIRM message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>	
<u>U-RNTI</u>	Same as CELL UPDATE message in step 7	
New U-RNTI		
- SRNC Identity	<u>'0000 0000 0000 0001'</u>	
<u>- S-RNTI</u>	Different from previous S-RNTI	
New C-RNTI	Different from previous C-RNTI	

PHYSICAL CHANNEL RECONFIGURATION FAILURE (Step 7)

Information Element	<u>Value/remark</u>
Message Type	"RADIO BEARER RECONFIGURATION FAILURE"
Failure cause	<u>"cell reselection"</u>
Other information element	Not checked

8.2.6.8.5 Test requirement

After step 3 the UE shall discover the presence of cell 2, enter CELL_FACH state, and initiate a cell update procedure in that cell.

After step 5 UE shall transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

After step 6 UE shall transmit PHYSICAL CHANNEL FAILURE message on the DCCH using AM RLC, setting IE "failure cause" to "cell reselection".

8.2.6.9 Physical channel reconfiguration for transition from CELL_FACH to CELL_DCH: Success

8.2.6.9.1 Definition

8.2.6.9.2 Conformance requirement

The UE shall correctly reconfigure a physical channel according to a PHYSICAL CHANNEL RECONFIGURATION message, which triggers a transition from CELL_FACH to CELL_DCH.

Reference

3GPP TS 25.331 clause 8.2.6

8.2.6.9.3 Test purpose

To confirm that the UE reconfigures a new physical channel according to a PHYSICAL CHANNEL RECONFIGURATION message received from the UTRAN, in the case of an assignment of dedicated physical resource from the common physical channels used previously by the UE.

8.2.6.9.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a PHYSICAL CHANNEL RECONFIGURATION message to the UE to start a transition from CELL_FACH to CELL_DCH. The UE shall reconfigure the new physical channel correctly according to this message. To complete this procedure, the UE shall transmit a PHYSICAL CHANNEL RECONFIGURATION COMPLETE message using AM RLC.

Expected sequence

Step	Dire	ction	Message	Comment
	UE	SS		
1		+	PHYSICAL CHANNEL RECONFIGURATION	
2				The UE shall reconfigure the physical channel in order to start using the dedicated channels allocated.
3		\rightarrow	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	

Specific Message Contents

PHYSICAL CHANNEL RECONFIGURATION

Use the message sub-type titled "Packet to CELL_DCH from CELL_FACH in PS" in Annex A.

8.2.6.9.5 Test requirement

After step 3 the UE shall transit from CELL_FACH to CELL_DCH and continue to communicate with SS on the dedicated physical channel.

8.2.6.10 Physical channel reconfiguration for transition from CELL_FACH to CELL_DCH: Failure (Unsupported configuration)

8.2.6.10.1 Definition

8.2.6.10.2 Conformance requirement

The UE shall keep its old configuration when the it receives a PHYSICAL CHANNEL RECONFIGURATION message, which specifies unsupported configuration parameters for the UE. It shall then transmit a PHYSICAL

CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, reporting the cause "configuration unsupported" in IE "failure cause".

Reference

3GPP TS 25.331 clause 8.2.6

8.2.6.10.3 Test purpose

To confirm that the UE keeps its configuration and transmits a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, if the received PHYSICAL CHANNEL RECONFIGURATION message includes unsupported configuration parameters.

8.2.6.10.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a PHYSICAL CHANNEL RECONFIGURATION message to the UE, which includes unsupported frequencies for the UE. The PHYSICAL CHANNEL RECONFIGURATION is structured in such a manner as to trigger a transition from CELL_FACH to CELL_DCH in the UE. The UE shall responds with a PHYSICAL CHANNEL RECONFIGURATION FAILURE message sent on the DCCH using AM RLC, setting "configuration unsupported" in IE "failure cause".

Expected sequence

Step	Dire	ction	Message	Comment
	UE	SS		
1		-	PHYSICAL CHANNEL RECONFIGURATION	Includes unsupported frequencies for the UE
2		→	PHYISICAL CHANNEL RECONFIGURATION FAILURE	The UE shall not change the physical channel configuration, this message shall be sent using the original allocated physical
				resource.

Specific Message Contents

PHYSICAL CHANNEL RECONFIGURATION

The contents of PHYSICAL CHANNEL RECONFIGURATION message in this test case is identical to the message sub-type title "Packet to CELL_DCH from CELL_FACH in PS" found in Annex A with the following exceptions:

Information Element	Value/remark
Frequency info	
- UARFCN uplink(Nu)	63984
- UARFCN downlink(Nd)	Not Present

PHYSICAL CHANNEL RECONFIGURATION FAILURE

Information Element	Value/remark
Message Type	
Failure cause	Configuration unsupported
Other information element	Not checked

8.2.6.10.5 Test requirement

After step 1 the UE shall keep its old configuration and transmit a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, the IE "failure cause" shall be set to "configuration unsupported".

8.2.6.11 Physical channel reconfiguration for transition from CELL_FACH to CELL_DCH: Failure (Physical channel failure and reversion to old configuration)

8.2.6.11.1 Definition

8.2.6.11.2 Conformance requirement

The UE shall revert to the old configuration when the UE fails to reconfigure the new physical channel by timer T312 expiry. It shall report the failure by transmitting a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, indicating "physical channel failure" in IE "failure cause".

Reference

3GPP TS 25.331 clause 8.2.6

8.2.6.11.3 Test purpose

To confirm that the UE reverts to the old configuration and transmits a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC if the UE fails to reconfigure the new physical channel according to a PHYSICAL CHANNEL RECONFIGURATION message by the T312 expiry.

8.2.6.11.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a PHYSICAL CHANNEL RECONFIGURATION message to the UE, requesting it to transit from CELL_FACH to CELL_DCH due to a switch in physical resource reallocation. However, it does not reconfigure the new physical channel accordingly but continue to use the old configuration. Consequently, the UE shall fail to reconfigure the new physical channel, and after T312 expiry the UE attempt to revert to the old configuration. Then the UE shall transmit a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC which reports "physical channel failure" in IE "failure cause".

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1		←	PHYSICAL CHANNEL RECONFIGURATION	
2				The SS does not reconfigure the physical channel, hence the UE shall detect a failure to reconfigure to the new physical channel.
3		→	PHYSICAL CHANNEL RECONFIGURATION FAILURE	After T312 expiry the UE reverts to the old configuration and transmits this message.

Specific Message Contents

PHYSICAL CHANNEL RECONFIGURATION

Use the message sub-type titled "Packet to CELL_DCH from CELL_FACH in PS" in Annex A.

PHYSICAL CHANNEL RECONFIGURATION FAILURE

Information Element	Value/remark
Message Type	
Failure cause	Physical channel failure
Other information element	Not checked

8.2.6.11.5 Test requirement

After step 2 the UE shall revert to the old configuration and transmit a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, specifying "physical channel failure" in IE "failure cause".

8.2.6.12 Physical channel reconfiguration for transition from CELL_FACH to CELL_DCH: Failure (Physical channel failure and reversion failure)

8.2.6.12.1 Definition

8.2.6.12.2 Conformance requirement

The UE shall perform an RRC connection re establishment cell update procedure when the UE fails to revert to the old configuration after the detection of physical channel failure in the physical channel reconfiguration procedure. After the UE completes cell update procedure, the UE transmit PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC which set IE "failure cause" to "physical channel failure".

Reference

3GPP TS 25.331 clause 8.2.6

8.2.6.12.3 Test purpose

To confirm that the UE initiates an RRC connection re establishment cell update procedure after it fails to reconfigure the new physical channel and experiences a subsequent failure to revert to the old configuration.

<u>To confirm that UE transmits PHYSICAL CHANNEL RECONFIGURATION FAILURE message after UE completes cell update procedure.</u>

8.2.6.12.4 Method of test

Initial Condition

System Simulator: 2 cells- Cell 1 is active, Cell 2 is inactive

UE: PS-DCCH+DTCH FACH (state 6-11) as specified in clause 7.4 of TS 34.108 in cell 1

Test Procedure

The UE is in the CELL_FACH state in cell 1. The SS begins to broadcast the BCCH in cell 2 and transmits a PHYSICAL CHANNEL RECONFIGURATION message to the UE, but it does not reconfigure L1 accordingly. This is expected to cause the UE to fail to reconfigure to the new physical channel. As a result, the UE shall try to revert to the old configuration—after timer T312 expiry. However, the SS deletes the old physical channel—before timer T312 has

expired. Therefore, UE shall reselect to cell 2 and sends an CELL UPDATE message on uplink CCCH with IE "Cell update cause" set to "radio link failure". The SS shall transmit CELL UPDATE CONFIRM message on downlink CCCH after receiving CELL UPDATE message. The UE transmits UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC and subsequently transmits PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting IE "failure cause" to "physical channel failure".

RRC CONNECTION RE ESTABLISHMENT REQUEST message, which includes its previously assigned U RNTI on the uplink CCCH. The SS transmits an RRC CONNECTION RE ESTABLISHMENT message, which includes a new TFCS according to the new transport channel used. After the UE receives this message, it reconfigures the new radio connection and transmits an RRC CONNECTION RE ESTABLISHMENT COMPLETE message on the uplink DCCH using AM RLC.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	←	BCCH	The SS begins to transmit the
			BCCH in a cell 2.
2	←	PHYSICAL CHANNEL	
		RECONFIGURATION	
3			The SS does not reconfigure the
			physical channel and deletes
			the old configuration. As a
			result, UE should encounter a
			failure to reconfigure the new
			physical channel, then attempt
	_		to revert to the old configuration.
4	\rightarrow	CELL UPDATERRC CONNECTION	The UE shall find cell 2 and be
		RE-ESTABLISHMENT REQUEST	in CELL_FACH state.
			This message includes the
			value "radio link failure" set in IE
			"Cell update cause".then
			transmit this message with its U-
	,	OF LUIDDATE CONFIDADO	RNTI included.
5	←	CELL UPDATE CONFIRMRRC	This message includes IE "new
		CONNECTION	U-RNTI" and IE "new C-
		RE-ESTABLISHMENT	RNTI"".This message specifies
			a new TFCS according to the
		LITE AND MODIFIED AND THOM	new transport channel.
6	\rightarrow	UTRAN MOBILITY INFORMATION	UE shall send this message in
		CONFIRMRRC CONNECTION	the cell 2.
		RE-ESTABLISHMENT COMPLETE	
<u>7</u>		PHYSICAL CHANNEL	The IE "failure cause" shall be
		RECONGURATION FAILURE	set to "physical channel failure"

Specific Message Contents

PHYSICAL CHANNEL RECONFIGURATION (Step 1)

Use the message sub-type titled "Packet to CELL_DCH from CELL_FACH in PS" in Annex A.

CELL UPDATE (Step 4)

The contents of CELL UPDATE message is identical as "Contents of CELL UPDATE message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	
- SRNC Identity	Assigned previously in cell 1
<u>- S-RNTI</u>	Assigned previously in cell 1
Cell Update Cause	<u>"radio link failure"</u>

CELL UPDATE CONFIRM (Step 5)

The contents of CELL UPDATE CONFIRM message is identical as "CELL UPDATE CONFIRM message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	Same as CELL UPDATE message in step 4
New U-RNTI	
- SRNC Identity	<u>'0000 0000 0000 0001'</u>
<u>- S-RNTI</u>	Different from previous S-RNTI
New C-RNTI	Different from previous C-RNTI

PHYSICAL CHANNEL RECONGURATION FAILURE (Step 7)

Information Element	<u>Value/remark</u>
Message Type	"PHYSICAL CHANNEL RECONGURATION FAILURE"
Failure cause	"physical channel failure"
Other information element	Not checked

8.2.6.12.5 Test requirement

After step 3 the UE shall detect the presence of cell 2, camp onto it, and transmit a <u>CELL UPDATE message using RLC-TM mode on the uplink CCCH.RRC CONNECTION RE ESTABLISHMENT REQUEST message.</u>

After step 6 the UE shall transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

After step 7 the UE shall transmit PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting the IE "failure cause" to "physical channel failure". the UE shall successfully re establish the RRC connection in cell 2.

8.2.6.13 Physical channel reconfiguration for transition from CELL_FACH to CELL_DCH: Failure (Incompatible simultaneous reconfiguration)

8.2.6.13.1 Definition

8.2.6.13.2 Conformance requirement

If the UE receives a PHYSICAL CHANNEL RECONFIGURATION message whilst reconfiguring due to a radio bearer message other than PHYSICAL CHANNEL RECONFIGURATION, it shall keep its configuration as if the PHYSICAL CHANNEL RECONFIGURATION message had not been received. If the UE is reconfiguring itself after receiving a radio bearer message other than PHYSICAL CHANNEL RECONFIGURATION message, and when the UE subsequently receive a PHYSICAL CHANNEL RECONFIGURATION message and the UE shall transmit a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC which is set to "Incompatible simultaneous reconfiguration" in IE "failure cause".

Reference

3GPP TS 25.331 clause 8.2.6

8.2.6.13.3 Test purpose

To confirm that if the UE receives a PHYSICAL CHANNEL RECONFIGURATION message whilst reconfiguring due to a radio bearer message other than PHYSICAL CHANNEL RECONFIGURATION, it shall keep its configuration as if the PHYSICAL CHANNEL RECONFIGURATION message had not been received and complete the reconfiguration according to the previously received message. To confirm that the UE keeps its old configuration, if receives a PHYSICAL CHANNEL RECONFIGURATION message before it manages to complete configuring according to an earlier message.

8.2.6.13.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL DCH state. The SS transmits a RADIO BEARER RECONFIGURATION message to the UE. The SS transmits a PHYSICAL CHANNEL RECONFIGURATION message before the "activation time" indicated in the RADIO BEARER RECONFIGURATION message expires. When the UE receives the PHYSICAL CHANNEL RECONFIGURATION message, the UE shall keep the configuration as if it had not received the PHYSICAL CHANNEL RECONFIGURATION message and shall transmit a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC with IE "failure cause" set to "incompatible simultaneous reconfiguration". After the SS receives the PHYSICAL CHANNEL RECONFIGURATION FAILURE message, the UE reconfigures the new physical channel parameters and transmits a RADIO BEARER RECONFIGURATION COMPLETE message on DCCH using AM RLC. The UE is in the CELL_FACH state. The SS transmits a message to the UE. Before the UE completes configuration due to the latter message, SS sends a PHYSICAL CHANNEL RECONFIGURATION message. The UE shall ignore the new message, maintains the old configuration, and transmits a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC indicating "incompatible simultaneous reconfiguration" in IE "failure cause".

Expected sequence

Step	Direction		Direction Message	Comment	
	UE	SS			
1		-	RADIO BEARER RECONFIGURATION		
2		←	PHYSICAL CHANNEL RECONFIGURATION	Sent before the elapse of the frame number specified in IE "Activation time info" of the message dispatched in step 1.	
3		\rightarrow	PHYSICAL CHANNEL RECONFIGURATION FAILURE	The UE does not change the configuration due to the reception of PHYSICAL CHANNEL RECONFIGURATION message.	
4	-	<u></u>	RADIO BEARER RECONFIGURATION COMPLETE	This message is on DCCH using AM RLC.	

Specific Message Contents

RADIO BEARER RECONFIGURATION (Step 1)

For RADIO BEARER RECONFIGURATION in step 1, use the message sub-type indicated as "Packet to CELL_DCH from CELL_FACH in PS" found in Annex A.

PHYSICAL CHANNEL RECONFIGURATION (Step 2)

For PHYSICAL CHANNEL RECONFIGURATION in step 2, use the message sub-type indicated as "Packet to CELL_DCH from CELL_FACH in PS" found in Annex A with the following exceptions:

Information Element	Value/remark
Activation Time Info	Current CFN-[current CFN mod 8 + 8]

PHYSICAL CHANNEL RECONFIGURATION FAILURE (Step 3)

Information Element	Value/remark
Message Type	
Failure cause	Incompatible simultaneous reconfiguration
Other information element	Not checked

8.2.6.13.5 Test requirement

After step 1, SS transmits a PHYSICAL CHANNEL RECONFIGURATION message before the expiry of the activation time specified in the message of step 1.

After step 2 the UE shall keep its configuration as if the UE had not received the PHYSICAL CHANNEL RECONFIGURATION message and shall transmit a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC with IE "failure cause" set to "Incompatible simultaneous reconfiguration".

After step 4 the UE communicates with the SS on the DCCH and DTCH using the new physical channel parameters reconfigured as a result of the RADIO BEARER RECONFIGURATION message.

After step 2 the UE shall keep its old configuration and transmit an PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting the value "incompatible simultaneous reconfiguration" in IE "failure cause":

8.2.6.14 Physical channel reconfiguration for transition from CELL_FACH to CELL_DCH: Failure (Invalid message reception and Invalid configuration)

8.2.6.14.1 Definition

8.2.6.14.2 Conformance requirement

The UE shall keep its old configuration when the UE receives a PHYSICAL CHANNEL RECONFIGURATION message containing a spare undefined value in the mandatory IE "UTRAN DRX cycle length coefficientDRX indicator". It shall then transmit a PHYSICAL CHANNEL RECONFIGURATION FAILURE message, set "protocol error" in IE "failure cause" and also set "Information element value not comprehended" in IE "Protocol error cause". The UE shall keep existing configuration before reception of a TRANSPORT CHANNEL RECONFIGURATION message when the TRANSPORT CHANNEL RECONFIGURATION message include some IEs set to invalid value, and then the UE shall transmit TRANSPORT CHANNEL RECONFIGURATION FAILURE including IE "failure cause" set to "invalid configuration"

Reference

3GPP TS 25.331 clause 8.2.6

8.2.6.14.3 Test purpose

To confirm that the UE transmits a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC if the received message uses a <u>spare undefined</u> value in the mandatory IE "<u>UTRAN DRX cycle length</u> <u>coefficientDRX indicator</u>".

To confirm that the UE transmits a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC if it receives a PHYSICAL CHANNEL RECONFIGURATION message including some IEs set to invalid value.

8.2.6.14.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a PHYSICAL CHANNEL RECONFIGURATION message to the UE which comprises a spare defined value in the mandatory IE "UTRAN DRX cycle length coefficientDRX indicator". The UE keeps the old configuration and transmits a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting "protocol error" in IE "failure cause" and also setting "Information element value not comprehended" in IE "Protocol error cause". The UE keeps initial configuration and SS transmits PHYSICAL CHANNEL RECONFIGURATION message including some IEs set to invalid value. The UE transmit PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting IE "failure cause" to "invalid configuration".

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1		+	PHYSICAL CHANNEL RECONFIGURATION	Contains a mandatory error due to illegal use of spare undefined value
2		\rightarrow	PHYSICAL CHANNEL RECONFIGURATION FAILURE	The UE does not change the configuration.
<u>3</u>	4	<u>É</u>	PHYSICAL CHANNEL RECONFIGURATION	This message includes IE set to invalid value
<u>4</u>				The UE does not change the configuration
<u>5</u>	<u>→</u>		PHYSICAL CHANNEL RECONFIGURATION FAILURE	The IE "failure cause" shall be set to "invalid configuration

Specific Message Contents

PHYSICAL CHANNEL RECONFIGURATION (Step 1)

The contents of PHYSICAL CHANNEL RECONFIGURATION message in this test case is identical to the message sub-type title "Packet to CELL_DCH from CELL_FACH in PS" found in Annex A with the following exceptions:

Information Element	Value/remark	
UTRAN DRX cycle length coefficientDRX indicator	spare-Undefined value	

PHYSICAL CHANNEL RECONFIGURATION FAILURE (Step 2)

Information Element	Value/remark	
Message Type		
Failure cause		
- Failure cause	Protocol error	
- Protocol error information		
- Protocol error cause	Information element value not comprehended	
Other information element	Not checked	

PHYSICAL CHANNEL RECONFIGURATION (Step 3)

The contents of PHYSICAL CHANNEL RECONFIGURATION message in this test case is identical to the message sub-type title "Packet to CELL DCH from CELL FACH in PS" found in Annex A with the following exceptions:

Information Element	<u>Value/remark</u>
Uplink DPCH info	Not present

PHYSICAL CHANNEL RECONFIGURATION FAILURE (Step 5)

Information Element	<u>Value/remark</u>
Message Type	
Failure cause	Invalid configuration
Other information element	Not checked

8.2.6.14.5 Test requirement

After step 1 the UE shall keep its old configuration, transmit a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC with "protocol error" in IE "failure cause" and also "Information element value not comprehended" in IE "Protocol error cause".

After step 3 the UE shall keep its old configuration.

After step 4 the UE shall transmit PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting IE "failure cause" to "invalid configuration".

8.2.6.15 Physical channel reconfiguration for transition from CELL_FACH to CELL_FACH (Hard handover to another frequency): Success

8.2.6.15.1 Definition

8.2.6.15.2 Conformance requirement

The UE shall correctly reconfigure a physical channel according to a PHYSICAL CHANNEL RECONFIGURATION message when asked to perform a transition from CELL_FACH to CELL_FACH—in the another cell.

Reference

3GPP TS 25.331 clause 8.2.6

8.2.6.15.3 Test purpose

To confirm that the UE reconfigures a new physical channel according to the PHYSICAL CHANNEL RECONFIGURATION message received from the SS.

8.2.6.15.4 Method of test

Initial Condition

System Simulator: 2-1 cells Cell 1 is active, Cell 2 is inactive

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS starts to broadcast the BCCH in cell 2 and transmits a PHYSICAL CHANNEL RECONFIGURATION message to the UE. The UE shall then reconfigure the new physical channel according to this message and the system information messages. Following this, it shall transmit a PHYSICAL CHANNEL RECONFIGURATION COMPLETE message using AM RLC on the RACH.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
4	- ←		BCCH	The SS starts to broadcast the BCCH in cell 2.
2 1		-	PHYSICAL CHANNEL RECONFIGURATION	
3 2				Reconfiguration of physical channel
4 <u>3</u>	<u>3</u> →		PHYSICAL CHANNEL RECONFIGURATION COMPLETE	

Specific Message Contents

PHYSICAL CHANNEL RECONFIGURATION

Use the message sub-type titled "Packet to CELL_FACH from CELL_FACH in PS" in Annex A.

8.2.6.15.5 Test requirement

After step 3 the UE shall transit from be in CELL_FACH to CELL_FACH in the another cell state and continue to communicate with SS on the common physical channel.

8.2.6.16 Physical channel reconfiguration for transition from CELL_FACH to CELL_FACH: Failure (Cell re-selection Physical channel failure)

8.2.6.16.1 Definition

8.2.6.16.2 Conformance requirement

The UE shall initiate the cell reselection procedure when the UE cannot use the assigned physical channel according to a PHYSICAL CHANNEL RECONFIGURATION message, during the transition from CELL_FACH to CELL_FACH in the another cell. After the UE completes cell update procedure, the UE transmits PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC which set IE "failure cause" to "cell reselection".

Reference

3GPP TS 25.331 clause 8.2.6

8.2.6.16.3 Test purpose

To confirm that the UE transmits PHYSICAL CHANNEL RECONFIGURATION FAILURE message after the UE perform completes a cell reselection and cell update procedure when the it fail to access the assigned physical channel, according to a PHYSICAL CHANNEL RECONFIGURATION message, during the transition from CELL_FACH to CELL_FACH in the another cell.

8.2.6.16.4 Method of test

Initial Condition

System Simulator: 3-2 cells Cell 1 is active, Cell 2 is inactive, Cell 3 is inactive

UE: PS-DCCH+DTCH FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state in cell 1. The SS begins to broadcast the BCCH in cell 2 and then transmits a PHYSICAL CHANNEL RECONFIGURATION message, resulting in the UE to attempt to transit from CELL_FACH in cell 1 to CELL_FACH-in cell 3. The UE is expected to fail to access the assigned physical channel as the SS ceases downlink transmission in cell 3. After the UE successfully camp onto cell 2, it shall initiate the cell update procedure in cell 2. The UE transmits CELL UPDATE message on uplink CCCH with IE "Cell update cause" set to "cell reselection". The SS shall transmit CELL UPDATE CONFIRM message on downlink CCCH after receiving CELL UPDATE message. The UE transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC and transmit PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC, setting IE "failure cause" to "cell reselection". The procedure completes when the SS confirms UE's cell update request.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1		←	ВССН	The SS starts to broadcast the BCCH in cell 2.
2		←	PHYSICAL CHANNEL RECONFIGURATION	The message assigns common resources so that the UE is expected to execute a transition from CELL_FACH in cell 1 to CELL_FACH in cell 3
3				The UE shall detect a failure to use the assigned common physical channel and try to perform a cell reselection.
4		→	CELL UPDATE	The UE discover the presence of cell 2 and triggers a cell update procedure. The value "cell reselection" shall be set in IE "Cell update cause".
5		+	CELL UPDATE CONFIRM	This message include IE "new U-RNTI" and IE "new C-RNTI". SS sends this message so that the UE moves to CELL_FACH state.
<u>6</u>	=	<u>></u>	UTRAN MOBILITY INFORMATION CONFIRM	
7		<u>></u>	PHYSICAL CHANNELRECONFIGURATION FAILURE	The IE "failure cause" shall be set to "cell reselection"

Specific Message Contents

PHYSICAL CHANNEL RECONFIGURATION

Use the message sub-type titled "Packet to CELL_FACH from CELL_FACH in PS" in Annex A.

CELL UPDATE (Step 4)

The contents of CELL UPDATE message is identical as "Contents of CELL UPDATE message" as found in Annex with the following exceptions:

İ	Information Element	<u>Value/remark</u>
	<u>U-RNTI</u>	
	- SRNC Identity	Assigned previously in cell 1
	<u>- S-RNTI</u>	Assigned previously in cell 1
	Cell Update Cause	"radio link failure"

CELL UPDATE CONFIRM (Step 5)

The contents of CELL UPDATE CONFIRM message is identical as "CELL UPDATE CONFIRM message" as found in Annex with the following exceptions:

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	Same as CELL UPDATE message in step 7
New U-RNTI	
- SRNC Identity	<u>'0000 0000 0000 0001'</u>
- S-RNTI	Different from previous S-RNTI
New C-RNTI	Different from previous C-RNTI

PHYSICAL CHANNEL RECONFIGURATION FAILURE (Step 7)

Information Element	<u>Value/remark</u>	
Message Type	"RADIO BEARER RECONFIGURATION FAILURE"	
Failure cause	"cell reselection"	
Other information element	Not checked	

8.2.6.16.5 Test requirement

After step 3 the UE shall discover the presence of cell 2, enter CELL_FACH state, and initiate a cell update procedure in that cell.

After step 5 the UE shall transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

After step 6 the UE shall transmit PHYSICAL CHANNEL FAILURE message on the DCCH using AM RLC, setting IE "failure cause" to "cell reselection".

8.2.6.17 Physical channel reconfiguration for transition from CELL_DCH to CELL_DCH (Hard handover to another frequency): Success (Subsequently received)

8.2.6.17.1 Definition

8.2.6.17.2 Conformance requirement

If the UE receives a PHYSICAL CHANNEL RECONFIGURATION message before the UE reconfigures the radio bearer according to the previous PHYSICAL CHANNEL RECONFIGURATION message, the UE shall ignore the new PHYSICAL CHANNEL RECONFIGURATION message and reconfigure according to the previous PHYSICAL CHANNEL RECONFIGURATION message received. Finally, the UE shall transmit PHYSICAL CHANNEL RECONFIGURATION COMPLETE message on the DCCH using AM RLC.

Reference

3GPP TS 25.331 clause 8.2.6

8.2.6.17.3 Test purpose

To confirm that if the UE receives a PHYSICAL CHANNEL RECONFIGURATION message before the UE reconfigures the radio bearer according to the previous PHYSICAL CHANNEL RECONFIGURATION message it ignores the new PHYSICAL CHANNEL RECONFIGURATION message and reconfigures according to the previous PHYSICAL CHANNEL RECONFIGURATION message.

8.2.6.17.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS=DCCH+DTCH_DCH(state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. When the SS transmits a PHYSICAL CHANNEL RECONFIGURATION message to the UE before the UE reconfigures the radio bearer, the UE ignores the new PHYSICAL CHANNEL RECONFIGURATION message and reconfigures according to the previous PHYSICAL CHANNEL RECONFIGURATION message received. Finally, the UE shall transmit PHYSICAL CHANNEL RECONFIGURATION COMPLETE message on the DCCH using AM RLC.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1		←	PHYSICAL CHANNEL RECONFIGURATION	
2		+	PHYSICAL CHANNEL RECONFIGURATION	SS send this message before the expiry of "activation time" specified in PHYSICAL CHANNEL RECONFIGURATION message of step 1.
3		→	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	The UE ignores the PHYSICAL CHANNEL RECONFIGURATION message in step 2 and confirms configuration according to the PHYSICAL CHANNEL RECONFIGURATION message in step 1.

Specific Message Contents

PHYSICAL CHANNEL RECONFIGURATION

The contents of PHYSICAL CHANNEL RECONFIGURATION message in this test case is identical to the message sub-type title "Packet to CELL_DCH from CELL_DCH in PS" found in Annex A with the following exceptions:

Information Element	Value/remark	
Activation Time Info	[256+Current CFN-[current CFN mod 8 + 8]]MOD 256	

8.2.6.17.5 Test requirement

After step 3 the UE shall communicate with the SS on the radio bearer specified in the PHYSICAL CHANNEL RECONFIGURATION message in step 1.

8.2.6.18 Physical channel reconfiguration for transition from CELL_FACH to CELL_DCH: Success (Subsequently received)

8.2.6.18.1 Definition

8.2.6.18.2 Conformance requirement

If the UE receives a PHYSICAL CHANNEL RECONFIGURATION message before the UE reconfigures the radio bearer according to the previous PHYSICAL CHANNEL RECONFIGURATION message, the UE shall ignore the new PHYSICAL CHANNEL RECONFIGURATION message and reconfigure according to the previous PHYSICAL

CHANNEL RECONFIGURATION message received. Finally, the UE shall transmit PHYSICAL CHANNEL RECONFIGURATION COMPLETE message on the DCCH using AM RLC.

Reference

3GPP TS 25.331 clause 8.2.6

8.2.6.18.3 Test purpose

To confirm that if the UE receives a PHYSICAL CHANNEL RECONFIGURATION message before the UE reconfigures the radio bearer according to the previous PHYSICAL CHANNEL RECONFIGURATION message it ignores the new PHYSICAL CHANNEL RECONFIGURATION message and reconfigures according to the previous PHYSICAL CHANNEL RECONFIGURATION message.

8.2.6.18.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. When the SS transmits a PHYSICAL CHANNEL RECONFIGURATION message to the UE before the UE reconfigures the radio bearer, the UE ignores the new PHYSICAL CHANNEL RECONFIGURATION message and reconfigures according to the previous PHYSICAL CHANNEL RECONFIGURATION message received. Finally, the UE shall transmit PHYSICAL CHANNEL RECONFIGURATION COMPLETE message on the DCCH using AM RLC.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	+	PHYSICAL CHANNEL	
		RECONFIGURATION	
2	←	PHYSICAL CHANNEL RECONFIGURATION	SS send this message before the expiry of "activation time" specified in PHYSICAL CHANNEL RECONFIGURATION message of step 1.
3	→	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	The UE ignores the PHYSICAL CHANNEL RECONFIGURATION message in step 2 and confirms configuration according to the PHYSICAL CHANNEL RECONFIGURATION message in step 1.

Specific Message Contents

PHYSICAL CHANNEL RECONFIGURATION

The contents of PHYSICAL CHANNEL RECONFIGURATION message in this test case is identical to the message sub-type title "Packet to CELL_DCH from CELL_FACH in PS" found in Annex A with the following exceptions:

Information Element	Value/remark	
Activation Time Info	[256+Current CFN-[current CFN mod 8 + 8]]MOD 256	

8.2.6.18.5 Test requirement

After step 3 the UE shall communicate with the SS on the radio bearer specified in the PHYSICAL CHANNEL RECONFIGURATION message in step 1.

8.2.6.2719 Physical Channel Reconfiguration from CELL_DCH to CELL_PCH: Success

8.2.6.2719.1 Definition

8.2.6.2719.2 Conformance requirement

The UE shall transmits PHYSICAL CHANNEL RECONFIGURATION COMPLETE message using and transits from CELL_DCH to CELL_PCH when receives a PHYSICAL CHANNEL RECONFIGURATION message. And then, the UE shall reconfigure radio bearers according to the PHYSICAL CHANNEL RECONFIGURATION message.

Reference

3GPP TS 25.331 clause 8.2.2

8.2.6.2719.3 Test purpose

To confirm that the UE transmits PHYSICAL CHANNEL RECONFIGURATION COMPLETE message before entering CELL_PCH state after it received a PHYSICAL CHANNEL RECONFIGURATION message and reconfigured its radio bearers. The UE is in CELL_PCH state in the same cell.

8.2.6.2719.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH_DCH(state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a PHYSICAL CHANNEL RECONFIGURATION message. The UE transmits PHYSICAL CHANNEL RECONFIGURATION COMPLETE message to the UE using AM RLC and enters into CELL_PCH state. The SS transmits a PAGING TYPE 1 message and the UE accepts it and enters the CELL_FACH state.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	+	PHYSICAL CHANNEL	
		RECONFIGURATION	
2	\rightarrow	PHYSICAL CHANNEL	The UE sends this message
		RECONFIGURATION COMPLETE	before start state transition.
3			Reconfiguration of Physical
			Channel after state transition.
4	←	PAGING TYPE 1	The SS transmits this message
			included a matched identity.
5	\rightarrow	CELL UPDATE	The UE is in CELL_FACH state.

Specific Message Contents

PHYSICAL CHANNEL RECONFIGURATION (Step 1)

Use the same message sub-type titled "Packet to CELL_FACH from CELL_DCH in PS" in Annex A with following exceptions:

Information Element	Value/remark
DRX indicatorRRC State Indicator	DRX with CELL updatingCELL PCH

PAGING TYPE 1 (Step 4)

Use the same message sub-type titled "TM (SMS in PS)" in Annex A with following exceptions:

Information Element	Value/remark
Paging record list	
Paging record	
- CHOICE Paging originator	UTRAN originator
- U-RNTI	
- SRNC Identity	Previously assigned SRNC identity
- S-RNTI	Previously assigned S-RNTI

8.2.6.2719.5 Test requirement

After step 1 the UE transmits PHYSICAL CHANNEL RECONFIGURATION COMPLETE message to the UE on uplink DCCH using AM RLC.

After step 3 the UE shall transit from CELL_DCH to CELL_PCH.

8.2.6.2820 Physical Channel Reconfiguration from CELL_DCH to URA_PCH: Success

8.2.6.2820.1 Definition

8.2.6.2820.2 Conformance requirement

The UE shall transmits PHYSICAL CHANNEL RECONFIGURATION COMPLETE message using and transits from CELL_DCH to URA_PCH when receives a PHYSICAL CHANNEL RECONFIGURATION message. And then, the UE shall reconfigure radio bearers according to the PHYSICAL CHANNEL RECONFIGURATION message.

Reference

3GPP TS 25.331 clause 8.2.2

8.2.6.<u>2820</u>.3 Test purpose

To confirm that the UE transmits PHYSICAL CHANNEL RECONFIGURATION COMPLETE message before entering URA_PCH state after it received a PHYSICAL CHANNEL RECONFIGURATION message and reconfigured its radio bearers. The UE is in CELL_PCH state.

8.2.6.2820.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS-DCCH+DTCH DCH(state 6-10) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a PHYSICAL CHANNEL RECONFIGURATION message. The UE transmits PHYSICAL CHANNEL RECONFIGURATION COMPLETE message to the UE using AM RLC and enters into URA_PCH state. The SS transmits a PAGING TYPE 1 message and the UE accepts it and enters the CELL_FACH state.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1		-	PHYSICAL CHANNEL	
			RECONFIGURATION	
2		\rightarrow	PHYSICAL CHANNEL	The UE sends this message
			RECONFIGURATION COMPLETE	before start state transition.
3				Reconfiguration of Physical
				Channel after state transition.
4		+	PAGING TYPE 1	The SS transmits this message
				included a matched identity.
5		\rightarrow	CELL UPDATE	The UE is in CELL_FACH state.

Specific Message Contents

PHYSICAL CHANNEL RECONFIGURATION (Step 1)

Use the same message sub-type titled "Packet to CELL_FACH from CELL_DCH in PS" in Annex A with following exceptions:

Information Element	Value/remark	
DRX indicatorRRC State Indicator	DRX with URA updatingURA_PCH	

PAGING TYPE 1 (Step 4)

Use the same message sub-type titled "TM (SMS in PS)" in Annex A with following exceptions:

Information Element	Value/remark
Paging record list	
Paging record	
- CHOICE Paging originator	UTRAN originator
- U-RNTI	
- SRNC Identity	Previously assigned SRNC identity
- S-RNTI	Previously assigned S-RNTI

8.2.6.2820.5 Test requirement

After step 1 the UE transmits PHYSICAL CHANNEL RECONFIGURATION COMPLETE message to the UE on uplink DCCH using AM RLC.

After step 3 the UE shall transit from CELL_DCH to URA_PCH.

8.2.6.2921 <u>VoidPhysical Channel Reconfiguration from CELL_FACH to CELL_PCH:</u> Success

8.2.6.29.1 Definition

8.2.6.29.2 Conformance requirement

The UE shall transmits PHYSICAL CHANNEL RECONFIGURATION COMPLETE message using and transits from CELL_DCH to CELL_PCH when receives a PHYSICAL CHANNEL RECONFIGURATION message. And then, the UE shall reconfigure radio bearers according to the PHYSICAL CHANNEL RECONFIGURATION message.

Reference

3GPP TS 25.331 clause 8.2.2

8.2.6.29.3 Test purpose

To confirm that the UE transmits PHYSICAL CHANNEL RECONFIGURATION COMPLETE message before entering CELL_PCH state after it received a PHYSICAL CHANNEL RECONFIGURATION message and reconfigured its radio bearers. The UE is in CELL_PCH state.

8.2.6.29.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS DCCH+DTCH_FACH(state 6 11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a PHYSICAL CHANNEL RECONFIGURATION message. The UE transmits PHYSICAL CHANNEL RECONFIGURATION COMPLETE message to the UE using AM RLC and enters into URA_PCH state. The SS transmits a PAGING TYPE 1 message and the UE accepts it and enters the CELL_FACH state.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
4		(PHYSICAL CHANNEL RECONFIGURATION	
2	_ →		PHYSICAL CHANNEL	The UE sends this message
			RECONFIGURATION COMPLETE	before start state transition.
3				Reconfiguration of Physical
				Channel after state transition.
4		\leftarrow	PAGING TYPE 1	The SS transmits this message
				included a matched identity.
5	_	\rightarrow	CELL UPDATE	The UE is in CELL_FACH state.

Specific Message Contents

PHYSICAL CHANNEL RECONFIGURATION (Step 1)

Use the same message sub-type titled "Packet to CELL_FACH from CELL_DCH in PS" in Annex A with following exceptions:

Information Element	Value/remark
DRX indicator	DRX with CELL updating

PAGING TYPE 1 (Step 4)

Use the same message sub-type titled "TM (SMS in PS)" in Annex A with following exceptions:

Information Element	Value/remark
Paging record list	
Paging record	
- CHOICE Paging originator	UTRAN originator
— - U-RNTI	
- SRNC Identity	Previously assigned SRNC identity
	Previously assigned S-RNTI

8.2.6.29.5 Test requirement

After step 1 the UE transmits PHYSICAL CHANNEL RECONFIGURATION COMPLETE message to the UE on uplink DCCH using AM RLC.

After step 3 the UE shall transit from CELL_DCH to CELL_PCH.

8.2.6.3022 <u>VoidPhysical Channel Reconfiguration from CELL_FACH to URA_PCH:</u> Success

8.2.6.30.1 Definition

8.2.6.30.2 Conformance requirement

The UE shall transmits PHYSICAL CHANNEL RECONFIGURATION COMPLETE message using and transits from CELL_FACH to URA_PCH when receives a PHYSICAL CHANNEL RECONFIGURATION message. And then, the UE shall reconfigure radio bearers according to the PHYSICAL CHANNEL RECONFIGURATION message.

Reference

3GPP TS 25.331 clause 8.2.2

8.2.6.30.3 Test purpose

To confirm that the UE transmits PHYSICAL CHANNEL RECONFIGURATION COMPLETE message before entering URA_PCH state after it received a PHYSICAL CHANNEL RECONFIGURATION message and reconfigured its radio bearers. The UE is in CELL_PCH state.

8.2.6.30.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: PS DCCH+DTCH_FACH(state 6 11) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_FACH state. The SS transmits a PHYSICAL CHANNEL RECONFIGURATION message. The UE transmits PHYSICAL CHANNEL RECONFIGURATION COMPLETE message to the UE using AM RLC and enters into URA_PCH state. The SS transmits a PAGING TYPE 1 message and the UE accepts it and enters the CELL_FACH state.

Expected sequence

Step	Direction		Message Message	Comment
	UE	SS		
4			PHYSICAL CHANNEL	
			RECONFIGURATION	
2	_→		PHYSICAL CHANNEL	The UE sends this message
			RECONFIGURATION COMPLETE	before start state transition.
3				Reconfiguration of Physical
				Channel after state transition.
4			PAGING TYPE 1	The SS transmits this message
				included a matched identity.
5	_	\rightarrow	CELL UPDATE	The UE is in CELL_FACH state.

Specific Message Contents

PHYSICAL CHANNEL RECONFIGURATION (Step 1)

Use the same message sub-type titled "Packet to CELL_FACH from CELL_FACH in PS" in Annex A with following exceptions:

Information Element	Value/remark
DRX indicator	DRX with URA updating

PAGING TYPE 1 (Step 4)

Use the same message sub-type titled "TM (SMS in PS)" in Annex A with following exceptions:

Information Element	Value/remark
Paging record list	
Paging record	
— CHOICE Paging originator	UTRAN originator
U-RNTI	
	Previously assigned SRNC identity
— - S-RNTI	Previously assigned S-RNTI

8.2.6.30.5 Test requirement

After step 1 the UE transmits PHYSICAL CHANNEL RECONFIGURATION COMPLETE message to the UE on uplink DCCH using AM RLC.

After step 3 the UE shall transit from CELL_FACH to URA_PCH.

8.2.7 Physical Shared Channel Allocation [TDD only]

[Editor's note: This message is not included in Release99 so this is FFS.]

8.2.8 PUSCH capacity request [TDD only]

[Editor's note: This message is not included in Release99 so this is FFS.]

8.2.9 <u>Void Downlink outer loop control</u>

For all following sub-clauses, all references to RRC CONNECTION SETUP messages shall use the default settings provided in clause 9 of TS 34.108, unless otherwise stated.

8.2.9.1 Downlink outer loop control: Increase is Disallowed

8.2.9.1.1 Definition

8.2.9.1.2 Conformance requirement

The UE shall maintain its existing internal target SIR value after it has received DOWNLINK OUTER LOOP CONTROL message which forbids further adjustment of SIR target. This restriction shall be observed even if the UE has detect that the current SIR target is not sufficient to achieve the downlink quality demanded by the UTRAN.

Reference

3GPP TS 25.331 clause 8.2.9

8.2.9.1.3 Test purpose

To confirm that the UE keeps its target SIR value after it receives a DOWNLINK OUTER LOOP CONTROL message with the IE "Downlink Outer Loop Control" set to "TRUE".

8.2.9.1.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: DCCH+DTCH_DCH (state 6 9 or state 6 10) as specified in clause 7.4 of TS 34.108 state with the no restriction by the DL outer loop power control

Test Procedure

The UE is in the CELL_DCH state and the SS adjusts its transmission power as the current SIR estimation by the UE has reached the same target SIR value determined autonomously by the UE. This condition can be observed when the TPC command bits on the DPCCH physical channel has stabilized (consistently alternating between values 0 and 1). The SS transmits the DOWNLINK OUTER LOOP CONTROL message, which is set to "TRUE" in IE "Downlink Outer Loop Control". Then the SS generates erroneous framessuch that the measured BLER in the UE falls below the value specified in IE "BLER Quality Value" of the RRC CONNECTION SETUP message received during the mobile-terminated RRC connection establishment procedure (described in clause 7.1.2 of TS 34.108). The UE shall detect this change but it maintains the current SIR target value and does not request that SS increases downlink transmission power by setting TPC in DPCCH to "1".

Expected sequence

Step	Dire	ction	Message Message	Comment
-	UE	SS		
4				The SS adjusts its transmission
				power as such that the
				measured value of SIR in the
				UE is the same as the SIR
				target value determined by the
				UE. The target SIR value is
				chosen by the UE in order that
				the target BLER value
				requested by SS is achieved.
2	_	\leftarrow	DOWNLINK OUTER LOOP CONTROL	Set value "TRUE" in IE
				"Downlink Outer Loop Control".
3				The SS starts to generate some
				error frames. The current SIR
				target in the UE cannot
				guarantee target BLER value
				specified earlier.
4				The UE shall keeps the existing
				SIR target value and it shall not
				request that the DL transmission
				power be increased using TPC
				bits in DPCCH physical channel.

Specific Message Content

DOWNLINK OUTER POWER CONTROL

Information Element	Value/remark
Integrity check info	Not Present
Downlink Outer Loop Control	
— - DL Outer loop control	Increased not allowed
Downlink DPCH power control information	
——- DPC Mode	Single

8.2.9.1.5 Test requirement

After step 3 the SS shall keep the target SIR value as same as step 1 and it shall not request for the downlink transmission power be increased by setting TPC bits to "1".

8.2.9.2 Downlink outer loop control: Increase is Allowed

8.2.9.2.1 Definition

8.2.9.2.2 Conformance requirement

The downlink power control procedure is used to control the downlink outer loop power control running in the UE. If the UE receives the DOWNLINK OUTER LOOP CONTROL message with value "FALSE" in IE "Downlink Outer Loop Control", it shall remove any existing restriction of DL outer loop power control. The UE shall begin to increase its DL target SIR value and make the DL transmission power increase until the desired BLER value for the DPCH transport channel has been attained.

Reference

3GPP TS 25.331 clause 8.2.9

8.1.2.9.2.3 Test purpose

To confirm that the UE begins to exercise DL outer loop power control, in order to increase the target SIR value, after it had received a DOWNLINK OUTER LOOP CONTROL message specifying "FALSE" in IE "Downlink Outer Loop Control".

8.2.9.2.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: DCCH+DTCH_DCH state (state 6 9 or state 6 10) as specified in clause 7.4 of TS 34.108 with the restriction by the DL outer loop power control

Test Procedure

The UE is in the CELL_DCH state with the restriction by the DL outer loop power control. The internal SIR target value determined by the UE is expected to be insufficient to allow the UE to achieve target BLER figure. The SS transmits the DOWNLINK OUTER LOOP CONTROL message, setting the value "FALSE" in IE "Downlink Outer Loop Control". Then the UE shall update its SIR target value and make the DL transmission power increase.

Expected sequence

Step	Direction		Message Message	Comment
	UE	SS		
4				The UE is brought to the same
				state as the end of step 4 in
				Clause 8.1.2.9.1.3.
2	_	(DOWNLINK OUTER LOOP CONTROL	Set the value "FALSE" in IE
				"Downlink Outer Loop Control".
3				The UE shall begin to request
				that the DL transmission power
				be increased. This is done by
				setting the TPC field to "1" in
				DPCČH.

8.2.9.2.5 Test requirement

After step 3 the SS shall increase the DL transmission power more than step 1.

8.2.9.3 Downlink outer loop control: Failure (Invalid message reception)

8.2.9.3.1 Definition

8.2.9.3.2 Conformance requirement

The UE shall correctly handle an erroneous DOWNLINK OUTER LOOP CONTROL message, and perform the appropriate error handling procedure. Specifically, it shall not alter the downlink power control mechanism, which is currently being applied. It shall also transmit a RRC STATUS message using RLC AM mode on the DCCH channel, state the appropriate failure cause.

Reference

3GPP TS 25.331 clause 8.2.9

8.2.9.3.3 Test purpose

To confirm that the UE transmits an RRC STATUS message on the DCCH using AM RLC when it receives a DOWNLINK OUTER LOOP CONTROL message, containing an invalid conditional IE with respect to the state of the UE.

8.2.9.3.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: DCCH+DTCH_DCH (state 6-9 or state 6-10) as specified in clause 7.4 of TS-34.108 and the integrity protection algorithm is not applied.

Test Procedure

The UE is in CELL_DCH state. Test steps 1 to 3 described in clause 8.2.9.1.4 are repeated. Next, the SS transmits another DOWNLINK OUTER LOOP CONTROL message, removing the power increase restriction, and at the same time includes the conditional IE "Integrity Check Info". The UE shall not request for an increase in the downlink transmission power through the TPC command field in DPCCH. It shall transmit RRC STATUS message stating the value "Message extension not comprehended" in IE "protocol cause".

Expected sequence

Step	Dire	ction	Message	Comment
	UE	SS		
4				Executes step 1 to step 3 in
				clause 8. 2.9.1.4.
2	_		DOWNLINK OUTER LOOP CONTROL	Contains an unexpected
				information element "Integrity
				Check Info" in the message.
				See specific message content.
3	_	\rightarrow	RRC STATUS	The UE shall not request for an
				increase in downlink
				transmission power. SS verifies
				this by checking the TPC
				command field in uplink
				DPCCH. The RRC STATUS
				message shall specify cause
				"Message extension is not
				comprehended" in IE "Protocol
				Error Information"

Specific Message Content

DOWNLINK OUTER LOOP CONTROL (Step 2)

The contents of DOWNLINK OUTER LOOP CONTROL message in this test case is identical to those of the default contents of layer 3 messages for RRC tests with the following exceptions:

Information Element	Value/remark
Integrity Check Info	Present
- Message authentication code	An arbitrary 32-bits string
- RRC Message Sequence	An arbitrarily chosen integer from range (015)

RRC STATUS

Information Element	Value/remark
Message Type	
Protocol error information	Information element value not comprehended
Other information element	Not-checked

8.2.9.3.5 Test requirement

After step 3 the UE shall keep its restriction for increase in downlink transmission power and not set TPC command fields in uplink DPCCH to "1". It shall transmit an RRC STATUS message on the DCCH using AM RLC, indicating "Information element value not comprehended" in IE "Protocol error cause".

8.3 RRC connection mobility procedure

8.3.1 Cell Update

8.3.1.1 Cell Update: cell reselection in CELL_FACH

8.3.1.1.1 Definition

8.3.1.1.2 Conformance requirement

This procedure is used to update UTRAN with the current cell of the UE after it has performed a cell reselection in CELL_FACH state.

Reference

3GPP TS 25.331 clause 8.3.1

8.3.1.1.3 Test purpose

To confirm that the UE executes a cell update procedure after the successful reselection of another UTRA cell. To confirm that the UE send the correct uplink respond message when executing cell update procedure due to cell reselection.

8.3.1.1.4 Method of test

Initial Condition

System Simulator: 2 cells - Cell 1 is active, with the downlink transmission power shown in column marked "T0" in Table 8.3.1.1-1, while Cell 2 is inactive

UE: CS-CELL_FACH_Initial (state 6-2) or PS-CELL_FACH_Initial (state 6-4) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE

Test Procedure

<u>Parameter</u>	<u>Unit</u>		Cell 1			Cell 2	
		<u>T0</u>	<u>T1</u>	<u>T2</u>	<u>T0</u>	<u>T1</u>	<u>T2</u>
UTRA RF Channel Number			<u>Ch. 1</u>			<u>Ch. 1</u>	
\hat{I}_{or}/I_{oc}	<u>dB</u>	<u>7.3</u>	<u>1.2</u>	<u>7.3</u>	off	<u>7.3</u>	<u>1.2</u>
CPICH Ec/lo	<u>dB</u>	<u>-10.7</u>	<u>-17.7</u>	<u>-11.6</u>	Cell 2 is	<u>-11.6</u>	<u>-17.7</u>
CPICH RSCP	<u>dBm</u>	<u>-72.7</u>	<u>-78.8</u>	<u>-72.7</u>	Switche	<u>-72.7</u>	<u>-78.8</u>

Table 8.3.1.1-1

Table 8.3.1.1-1 illustrates the downlink power to be applied for the 2 cells at various time instants of the test execution. Columns marked "T0" denote the initial conditions, while columns marked "T1" and "T2" are to be applied subsequently. SS switches the power settings repeatedly between columns "T1" and "T2", whenever the description below specifies that the transmission power settings for cell 1 and cell 2 are reversed.

The UE is in the CELL FACH state, camping onto cell 1. SS configures its downlink transmission power settings according to columns "T1" in Table 8.3.1.1-1. The SS starts to broadcast BCCH on the primary CCPICH in cell 2-with a power level greater than that in cell 1. The UE shall find cell 2 to be more suitable for service and hence perform a cell reselection. After the completion of cell reselection, the UE shall transmits a CELL UPDATE message to the SS on the uplink CCCH of cell 2 and set IE "Cell update cause" to "Cell Reselection". After the SS receives this message, it transmits a CELL UPDATE CONFIRM message, which includes the IE "DRX Indicator RRC State Indicator" set to "DRX with Cell updatingCELL_PCH", to the UE on the downlink DCCH. SS verifies that the UE does not send any response to this message. UE shall move to CELL_PCH state. SS then reverses the transmission power of cell 1 and cell 2. The UE shall send a CELL UPDATE message on the uplink CCCH of cell 1. SS replies with CELL UPDATE CONFIRM message and allocates new C-RNTI and U-RNTI identities to the UE. The IE "DRX Indicator RRC State Indicator" is set to "No DRXCELL FACH" in this message. The UE shall transmit UTRAN MOBILITY INFORMATION CONFIRM PHYSICAL CHANNEL RECONFIGURATION COMPLETE-message. Following this, SS reverses the transmission power settings for cell 1 and cell 2 again. The UE shall initiate a cell update procedure by transmitting a CELL UPDATE message and stating the cause as 'cell re-selection'. SS replies with a CELL UPDATE CONFIRM message which contains IE "Physical channel information elements" IE "PRACH Info" and IE "Secondary CCPCH Info". The UE shall send PHYSICAL CHANNEL RECONFIGURATION COMPLETE message to acknowledge the assignment of the newchange in physical resources. Then, SS reverses the transmission power settings for cell 1 and cell 2 again. The UE shall send a CELL UPDATE message on the uplink CCCH of cell 1. SS replies with a CELL UPDATE CONFIRM message which contains IE "Transport channel information elements". The UE shall send TRANSPORT CHANNEL RECONFIGURATION COMPLETE message. Following this, SS reverses the transmission power settings for cell 1 and cell 2 again. The UE shall send a CELL UPDATE message on the uplink CCCH of cell 2. SS replies with a CELL UPDATE CONFIRM message which contains IE "RB information to be affected list". The UE shall send RADIO BEARER RECONFIGURATION COMPLETE message. Finally, SS reverses the transmission power settings for cell 1 and cell 2 once again. The UE shall send a CELL UPDATE message on the uplink CCCH of cell 1. SS replies with a CELL UPDATE CONFIRM message which contains IE "RB information to release list". The UE shall send RADIO BEARER RELEASE COMPLETE message.

Expected sequence

Step	Direction UE SS	Message	Comment
1	02 00		The UE is in the CELL_FACH state in cell 1
2	←	ВССН	SS applies the downlink transmission power settings, according to the values in columns "T1" of Table 8.3.1.1-1. The SS starts to broadcast BCCH on the primary CCP4CH in cell 2 with a power level that is higher than that in cell 1. The UE shall find that the cell 2 is better for service and perform a reselection. SS waits for the maximum duration required for the UE to camp to cell 2.
3	→	CELL UPDATE	Value "cell reselection" should be indicated in IE " Cell update cause"
4	+	CELL UPDATE CONFIRM	IE "DRX IndicatorRRC State Indicator" is set to "DRX with cell updatingCELL_PCH". SS set k=0.
5			SS checks the uplink PRACH channel to verify that no response is sent by UE. SS reverses the transmission power level of cell 1 and cell 2.
<u>6</u>			SS reverses the transmission power level of cell 1 and cell 2.
<u>67</u>	→	CELL UPDATE	UE shall revert to normal service in cell 1 and set the cause to "cell reselection".
7 <u>8</u>	←	CELL UPDATE CONFIRM	IE "DRX IndicatorRRC State Indicator" is set to "Ne DRXCELL FACH". If k ≥ 0, New new C-RNTI and U-RNTI identities are assigned to the UE. If k>1, IE "Physical channel information elements" is included in this message. If k>2, IE "Transport channel information elements" is included in this message. If k>3, IE "RB information to be affected list" is included in this message. If k>4, IE "RB information to release list" is included in this message. If k>4, IE "RB information to release list" is included in this message. Increment k by 1.
<u>89</u>	→	UTRAN MOBILITY INFORMATION CONFIRMPHYSICAL CHANNEL RECONFIGURATION COMPLETE	If k=1 when SS received this message, go to step 6. Else test fails. If this message is not received, proceed to next step. UE moves to CELL_FACH state and sends this message.
9			SS reverses the transmission power level of cell 1 and cell 2.
10	_ →	CELL UPDATE	UE shall detect that cell 2 has become stronger.
11	-	CELL UPDATE CONFIRM	IE "DRX Indicator" is set to "No DRX". At the same time, IE "PRACH Info" and IE "Secondary CCPCH Info" are also indicated.

1			
12 10)	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	If k=2 when SS received this message, go to step 6. Else test fails. If this message is not received, proceed to next step. UE moves to CELL_FACH state and sends this message on the new PRACH assigned in step 11.
11	<u>→</u>	TRANSPORT CHANNEL RECONFIGURATION COMPLETE	If k=3 when SS received this message, go to step 6. Else test fails. If this message is not received, proceed to next step.
12	<u>→</u>	RADIO BEARER RECONFIGURATION COMPLETE	If k=4 when SS received this message, go to step 6. Else test fails. If this message is not received, proceed to next step.
<u>13</u>	<u>→</u>	RADIO BEARER RELEASE COMPLETE	If k=5 when SS received this message, test ends. Else test fails. If this message is not received, test fails.

Specific Message Contents

CELL UPDATE (Steps 3 and 67)

Use t<u>T</u>he same message sub type found in Annex A shall be transmitted by the UE on the uplink CCCH, with the exception of the following IEs:

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0000 0001'
- S-RNTI	Check to see if set to '0000 0000 0000 0000 0000 0001'
	when k<1. Check to see if set to same string in IE "S-
	RNTI" in IE "New U-RNTI" of CELL UPDATE CONFIRM
	message in previous step 8 when k<0.
Integrity check info	Not checked
AM_RLC error indicator (for C-plane)	Check to see if set to 'FALSE'
AM_RLC error indicator (for U-plane)	Check to see if set to 'FALSE'
START List	Checked to see if the 'CN domain identity' and 'START'
	IEs are present for all CN domains supported by the UE
Cell Update Cause	Check to see if set to 'Cell Re-selection'
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Measured results on RACH	Check to see if it is absent
Protocol error information	Check to see if it is absent

CELL UPDATE CONFIRM (Step 4)

Use the same message sub-type found in Annex A, with the following exceptions:

Information Element	Value/remark
DRX IndicatorRRC State Indicator	DRX with cell updatingCELL_PCH

CELL UPDATE CONFIRM (Step 78 and k = 0)

Use the same message sub-type found in Annex Astep 4, with the following exceptions:

Information Element	Value/remark
New U-RNTI	
- SRNC Identity	'0000 0000 0000- 0001'
- S-RNTI	An arbitrary 20-bits string which is different from original S-RNTI
New C-RNTI	An arbitrary 16-bits string which is different from original C-RNTI assigned in RRC connection establishment procedure.

CELL UPDATE (Step 10)

Use the same message sub-type found in Annex A.

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0000 0001'
- S-RNTI	Check to see if set to same string in IE "S-RNTI" in IE
	"New U-RNTI" of CELL UPDATE CONFIRM message in
	step 7.
Integrity check info	Not checked
AM_RLC error indicator (for C-plane)	Check to see if set to 'FALSE'
AM_RLC error indicator (for U-plane)	Check to see if set to 'FALSE'
START List	Checked to see if the 'CN domain identity' and 'START'
	IEs are present for all CN domains supported by the UE
Cell Update Cause	Check to see if set to 'Cell Re-selection'
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Measured results on RACH	Check to see if it is absent
Protocol error information	Check to see if it is absent

CELL UPDATE CONFIRM (Step 448 and k=1)

Use the same message sub-type found in Annex Astep 8 and k=0, with the following exceptions:

Information Element	Value/remark
Maximum allowed uplink TX power	3 dB below the follow value:
	Minimum of { 33 dBm, maximum uplink power allowed
	under the UE power class }
PRACH Info (for RACH)	'000000000000011'B:Signature 0&1
- Available Signature	Refer to the parameter set for TS 34.108
— - Available SF	0
- Preamble scrambling code number	Refer to the parameter set for TS 34.108
— - Puncturing Limit	'00000000011'B:SubChNumber 0&1
— - Available Sub-Channel number	
Downlink DPCH info for one radio link	
— - Primary CPICH Info	Set to the scrambling code for cell 1's P-CPICH.
- Primary Scrambling Code	Not Present.
PDSCH with SHO DCH Info	Not Present.
— - PDSCH code mapping	
Secondary CCPCH Info	TRUE
— - Primary CPICH usage for channel estimation	
	Not Present. Use default (Scrambling code of P-CPICH)
	1
	Not Present. Use default (Scrambling code of P-CPICH)
	FALSE
	256
	255
——- Code number	TRUE
- Pilot symbol existence	TRUE
	Flexible
	0 chips
— - Timing Offset	

CELL UPDATE CONFIRM (Step 8 and k=2)

Use the same message sub-type found in step 8 and k=1, with the following exceptions:

Added or Reconfigured uplink TrCH information	
-Transport channel identity	1
-TFS	
-Dynamic Transport format information	(This IE is repeated for TFI number)
-Number of Transport blocks	Reference to TS34.108 clause 6.10 Parameter Set
-RLC size	Reference to TS34.108 clause 6.10 Parameter Set
-Semi-static Transport Format information	
-Transmission time interval	Reference to TS34.108 clause 6.10 Parameter Set
-Type of channel coding	Reference to TS34.108 clause 6.10 Parameter Set
-Coding Rate	Reference to TS34.108 clause 6.10 Parameter Set
-Rate matching attribute	Reference to TS34.108 clause 6.10 Parameter Set
-CRC size	Reference to TS34.108 clause 6.10 Parameter Set

CELL UPDATE CONFIRM (Step 8 and k=3)

Use the same message sub-type found in step 8 and k=2, with the following exceptions:

RB information to be affected	(UM DCCH for RRC)
-RB identity	$ \frac{1}{2} $
-RB mapping info	
 -Information for each multiplexing option 	
 -Number of uplink RLC logical channels 	<u>1</u>
 -Uplink transport channel type 	RACH
 -uplink Transport channel identity 	<u>1</u>
 Logical channel identity 	<u>1</u>
-CHOICE RLC size list	
-MAC logical channel priority	<u>2</u>
 -Number of downlink RLC logical channels 	<u>1</u>
 -Downlink transport channel type 	<u>FACH</u>
-DL Transport channel identity	<u> 1</u>
-Logical channel identity	<u>1</u>

CELL UPDATE CONFIRM (Step 8 and k=4)

Use the same message sub-type found in step 8 and k=3, with the following exceptions:

RB information to release	
-RB identity	<u>4</u>

8.3.1.1.5 Test requirement

After step 2 the UE shall reselect to cell 2 and then it shall transmit a CELL UPDATE message which, sets the value "cell reselection" in IE "Cell update cause".

After step 4 the UE shall not transmit any uplink message in response to the CELL UPDATE CONFIRMATION message received in step 4.

After step <u>5-6</u> the UE shall sent CELL UPDATE message to cell <u>with stronger transmitting power</u>1, in order to indicate that a cell reselection has taken place.

After step 7-8, if k=1, the UE shall transmit <u>UTRAN MOBILITY INFORMATION CONFIRM PHYSICAL</u> <u>CHANNEL RECONFIGURATION COMPLETE</u> message to acknowledge that it has started to use the new RNTI identities allocated.

After step 9 the UE shall sent CELL UPDATE message to cell 2, in order to indicate that a cell reselection has taken place.

After step 11 If k=2, the UE shall transmit PHYSICAL CHANNEL RECONFIGURATION COMPLETE message on the new PRACH physical channel assigned previously in CELL UPDATE CONFIRM message sent in step 11.

<u>If k=3, the UE shall transmit TRANSPORT CHANNEL RECONFIGURATION COMPLETE message to acknowledge that it has reconfigured the transport channels.</u>

If k=4, the UE shall transmit RADIO BEARER RECONFIGURATION COMPLETE message to acknowledge that it has reconfigured the radio bearers.

If k=5, the UE shall transmit RADIO BEARER RELEASE COMPLETE message to acknowledge that it has release its radio bearers.

8.3.1.2 Cell Update: cell reselection in CELL_PCH

8.3.1.2.1 Definition

8.3.1.2.2 Conformance requirement

This procedure is to update UTRAN with information of the current cell, after a cell reselection has occurred in CELL_PCH state.

Reference

3GPP TS 25.331 clause 8.3.1

8.3.1.2.3 Test purpose

To confirm that the UE, in CELL_PCH state, executes a cell update procedure after the successful reselection of another UTRA cell. To confirm that the UE replies with an appropriate uplink message after receiving CELL UPDATE CONFIRM message during the cell update procedure.

8.3.1.2.4 Method of test

Initial Condition

System Simulator: 2 cells – Cell 1 is active with the downlink transmission power shown in column marked "T0" in Table 8.3.1.1-1, while cell , Cell 2 is inactive

UE: CELL_PCH (state 6-1512) in cell 1 as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is brought to CELL PCH state and is camped onto cell 1. The SS starts to broadcast system information on the BCCH on the primary CCPICH in cell 2. SS configures its downlink transmission power settings according to columns "T1" in Table 8.3.1.1-1. The transmission level of cell 2 is raised to be higher than in cell 1. When the UE detects the presence of cell 2, it moves to CELL FACH state and transmits a CELL UPDATE message on the uplink CCCH. The value "cell reselection" shall be set in IE "Cell update cause" in CELL UPDATE message. Upon reception of CELL_UPDATE message, the SS transmits a CELL UPDATE CONFIRM message which includes the Ies "PRACH info", "Secondary CCPCH info" to the UE on the downlink DCCH. In this message, the IE "DRX Indicator" is set to "No DRX". Then the UE shall transmit a PHYSICAL CHANNEL RECONFIGURATION COMPLETE message on the uplink DCCH and enters CELL_FACH state. SS exchanges the transmission power of cell 1 and cell 2, so that cell 1 becomes the stronger cell. UE shall initiate a cell update procedure again by transmitting CELL UPDATE message. SS replies with a CELL UPDATE CONFIRM message with the IE "DRX Indicator RRC State Indicator" set to "DRX with cell updatingCELL PCH". After receiving this message, the UE returns to CELL_PCH state without transmitting any uplink message. FinallyNext, SS reverses the transmission strengths of cell 1 and cell 2 again. This will cause the UE to send CELL UPDATE message on the uplink PRACH_CCCH of cell 21. SS then sends CELL UPDATE CONFIRM message with the assignment of new C-RNTI and U-RNTI identities. The UE shall transmit PHYSICAL CHANNEL RECONFIGURATION COMPLETEUTRAN MOBILITY INFORMATION CONFIRM message as a response. Following this, SS reverses the transmission power settings for cell 1 and cell 2 again. The UE shall initiate a cell update procedure by transmitting a CELL UPDATE message on the uplink CCCH of cell 2 and stating the cause as 'cell re-selection'. SS replies with a CELL UPDATE CONFIRM message which contains "Physical channel information elements". The UE shall send PHYSICAL CHANNEL RECONFIGURATION COMPLETE message to acknowledge the change in physical resources. Then, SS reverses the transmission power settings for cell 1 and cell 2 again. The UE shall send a CELL UPDATE message on the uplink CCCH of cell 1. SS replies with a CELL UPDATE CONFIRM message which contains "Physical channel information elements" and "Transport channel information elements". The UE shall send TRANSPORT CHANNEL RECONFIGURATION COMPLETE message. Following

this, SS reverses the transmission power settings for cell 1 and cell 2 again. The UE shall send a CELL UPDATE message on the uplink CCCH of cell 2. SS replies with a CELL UPDATE CONFIRM message which contains "Physical channel information elements", "Transport channel information elements" and IE "RB information to be affected list". The UE shall send RADIO BEARER RECONFIGURATION COMPLETE message. Finally, SS reverses the transmission power settings for cell 1 and cell 2 once again. The UE shall send a CELL UPDATE message on the uplink CCCH of cell 1. SS replies with a CELL UPDATE CONFIRM message which contains "Physical channel information elements", "Transport channel information elements", IE "RB information to reconfigure list" and IE "RB information to release list". The UE shall send RADIO BEARER RELEASE COMPLETE message.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1			The UE is brought to CELL_PCH state in cell 1
2	+	BCCH	The SS starts to broadcast system information message on BCCH on the primary CCP+CH from cell 2. SS reverse Tthe transmission
			level of cell 2 is set to be higher than that inand cell 1. The UE shall find that the cell 2 is better and attempt to perform a cell reselection.
3	→	CELL UPDATE	The UE moves to CELL_FACH state and transmits this message with the IE " Cell update cause" set to "cell reselection"
4	+	CELL UPDATE CONFIRM	IE "RRC State Indicator" is set to "CELL PCH". The message includes les "PRACH info", "Secondary CCPCH info" and set IE "DRX Indicator" to "No DRX". SS set k=0.
5	-→	PHYSICAL CHANNEL RECONFIGURATION CONFIRM	UE stays in CELL_FACH state.
<u>65</u>			SS swaps the transmission power of cell 1 and cell 2, making cell 1 the stronger cell.
7 6	\rightarrow	CELL UPDATE	UE move from CELL PCH to CELL_FACH to transmit this message.
8 <u>7</u>	←	CELL UPDATE CONFIRM	IE "RRC State Indicator" is set to "CELL FACH". If k ≥ 0, new C-RNTI and U-RNTI identities are assigned to the UE. If k>1, IE "Physical channel information elements" is included in this message. If k>2, IE "Transport channel information elements" is included in this message. If k>3, IE "RB information to be affected list" is included in this message. If k>4, IE "RB information to release list" is included in this message. If k>4, IE "RB information to release list" is included in this message. Increment k by 1. IE "DRX Indicator" is set to "DRX with cell updating".
<u>8</u>	<u>→</u>	UTRAN MOBILITY INFORMATION CONFIRM	If k=1 when SS received this message, go to step 5. Else test fails. If this message is not received, proceed to next step.
9			UE moves to CELL_PCH state. SS exchanges the transmission power of cell 1 and cell 2.
10	- →	CELL UPDATE	UE shall re-select to cell 2 and transmit this message.
11	- +	CELL UPDATE CONFIRM	Includes IE "new U-RNTI" and IE "new C-RNTI"

12 9	→	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	If k=2 when SS received this message, go to step 6. Else test fails. If this message is not received, proceed to next step.
<u>10</u>	<u> </u>	TRANSPORT CHANNEL RECONFIGURATION COMPLETE	If k=3 when SS received this message, go to step 6. Else test fails. If this message is not received, proceed to next step.
11	<u>→</u>	RADIO BEARER RECONFIGURATION COMPLETE	If k=4 when SS received this message, go to step 6. Else test fails. If this message is not received, proceed to next step.
<u>12</u>	<u> </u>	RADIO BEARER RELEASE COMPLETE	If k=5 when SS received this message, test ends. Else test fails. If this message is not received, test fails.

CELL UPDATE (Steps 3, 7 and 10 and 6)

Use the same message sub-type found in Annex A, with the following exception.

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0000 0001'
- S-RNTI	Check to see if set to '0000 0000 0000 0000 0000 0001'
	when k<1. Check to see if set to same string in IE "S-
	RNTI" in IE "New U-RNTI" of CELL UPDATE CONFIRM
	message in previous step 7 when k<0.
Integrity check info	Net checked
AM_RLC error indicator (for C-plane)	Check to see if set to 'FALSE'
AM_RLC error indicator (for U-plane)	Check to see if set to 'FALSE'
START List	Checked to see if the 'CN domain identity' and 'START'
	IEs are present for all CN domains supported by the UE
Cell Update Cause	Check to see if set to 'Cell Re-selection'
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Measured results on RACH	Check to see if it is absent
Protocol error information	Check to see if it is absent

CELL UPDATE CONFIRM (Step 4)

Use the same message found in step 11 of Clause 8.3.1.1.3

CELL UPDATE CONFIRM (Step 84)

Use the same message sub-type found in Annex A, with the following exceptions:

Information Element	Value/remark	
DRX IndicatorRRC State Indicator	DRX with cell updatingCELL_PCH	

CELL UPDATE CONFIRM (Step 117 and k=0)

Use the same message sub-type found in $\frac{Annex Astep 4}{A}$, with the following exceptions:

Information Element	Value/remark	
New U-RNTI		
- SRNC Identity	'0000 0000 0000 0001'	
- S-RNTI	An arbitrary 20-bits string which is different from original	
	S-RNTI	
New C-RNTI	An arbitrary 16-bits string which is different from original	
	C-RNTI.	
New C-RNTI	An arbitrary 16-bits string which is different from original	

CELL UPDATE CONFIRM (Step 7 and k=1)

Use the same message sub-type found in step 7 with k=0, with the following exceptions:

Information Element	<u>Value/remark</u>
Maximum allowed uplink TX power	3 dB below the follow value:
	Minimum of { 33 dBm, maximum uplink power allowed
	under the UE power class }

CELL UPDATE CONFIRM (Step 7 and k=2)

Use the same message sub-type found in step 7 with k=1, with the following exceptions:

Added or Reconfigured uplink TrCH information	1,
-Transport channel identity	1
-TFS	(This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set
-Dynamic Transport format information	Reference to TS34.108 clause 6.10 Parameter Set
-Number of Transport blocks	Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set
-RLC size	Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set
-Semi-static Transport Format information	Reference to TS34.108 clause 6.10 Parameter Set
-Transmission time interval	
-Type of channel coding	
-Coding Rate	
-Rate matching attribute	
-CRC size	

CELL UPDATE CONFIRM (Step 7 and k=3)

Use the same message sub-type found in step 7 with k=2, with the following exceptions:

RB information to be affected	(UM DCCH for RRC)
-RB identity	
-RB mapping info	1
-Information for each multiplexing option	RACH 1
-Number of uplink RLC logical channels	$\overline{\underline{1}}$
-Uplink transport channel type	<u>2</u> 1
-uplink Transport channel identity	<u>FACH</u>
-Logical channel identity	$\frac{1}{1}$
-CHOICE RLC size list	
-MAC logical channel priority	
-Number of downlink RLC logical channels	

-Downlink transport channel type	
-DL Transport channel identity	
-Logical channel identity	

CELL UPDATE CONFIRM (Step 7 and k=4)

Use the same message sub-type found in step 7 with k=3, with the following exceptions:

RB information to release		
RB identity	4	

8.3.1.2.5 Test requirement

After step 2 the UE shall reselect to cell 2 and transmit a CELL UPDATE message, containing the IE "Cell update cause" set to "cell reselection".

After step 4 the UE shall transmit a PHYSICAL CHANNEL RECONFIGURATION COMPLETE message on the uplink DCCH.

After step 5 the UE shall reselect to stronger transmitting cell and transmit a CELL UPDATE message, containing the IE "Cell update cause" set to "cell reselection".

After step 7, if k=1, the UE shall transmit UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH.

After step 9 the UE shall detect that cell 2 has become the stronger cell. It shall send a CELL UPDATE message on cell 2' PRACH channel with the cause set to "cell reselection".

After step 11 If k=2, the UE shall transmit PHYSICAL CHANNEL RECONFIGURATION COMPLETE message on the uplink DCCH.

 $\underline{If \ k=3, \ the \ UE \ shall \ transmit \ TRANSPORT \ CHANNEL \ RECONFIGURATION \ COMPLETE \ message \ to \ acknowledge \ that \ it \ has \ reconfigured \ the \ transport \ channels.}$

<u>If k=4</u>, the <u>UE shall transmit RADIO BEARER RECONFIGURATION COMPLETE message to acknowledge that it has reconfigured the radio bearers.</u>

If k=5, the UE shall transmit RADIO BEARER RELEASE COMPLETE message to acknowledge that it has release its radio bearers.

8.3.1.3 Cell Update: periodical cell update in CELL_FACH

8.3.1.3.1 Definition

8.3.1.3.2 Conformance requirement

This procedure is to update UTRAN with the current cell information, after the UE has remained in the service area in the CELL_FACH state for a period exceeding the timer value T305.

Reference

3GPP TS 25.331 clause 8.3.1

8.3.1.3.3 Test purpose

To confirm that the UE executes a periodic cell update procedure following the expiry of timer T305. To confirm that the UE sends a correct response to the CELL UPDATE CONFIRM message. To confirm that the UE listens to the system information messages and then responds to a change in the setting for timer T305.

8.3.1.3.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CS-CELL_FACH_Initial (state 6-2) or PS-CELL_FACH_Initial (state 6-4) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE

Test Procedure

The UE is in CELL FACH state. When the UE detects the expiry of timer T305 according to the settings in system information, the UE transmits a CELL UPDATE message to the SS on the uplink CCCH with a cause indicating periodic cell updating. SS replies with a CELL UPDATE CONFIRM message, omitting the IEs "new C RNTI", "new U RNTI", "PRACH Info" and "Secondary CCPCH Info". The and IE "DRX Indicator RRC State Indicator", however, is specified and set to "No DRXCELL FACH". SS verifies that the UE does not transmit any uplink message. SS then waits for T305 to expire again. The UE shall send another CELL UPDATE message to report periodic cell updating. After the SS receives this message, it transmits a CELL UPDATE CONFIRM message which includes the IEs "new C-RNTI", "new U-RNTI" to the UE on the downlink DCCH. Then the UE shall transmit an UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH to acknowledge the receipt of the new UE identities. SS once again allows T305 timer to expire, the UE shall transmit CELL UPDATE message for the third time. The new U-RNTI shall be indicated in this message together with the correct updating cause. In this sequence, SS replies with a CELL UPDATE CONFIRM message containing new channel parameters in IEs "PRACH Info" and "Secondary CCPCH Info". The UE shall then send a PHYSICAL CHANNEL RECONFIGURTION COMPLETE message by using the newly allocated PRACH resources. FinallyNext, the content of the SYSTEM INFORMATION BLOCK TYPE 2-1 is changed to disable periodic cell updating. SS then monitors the uplink DCCH CCCH for a period up to the maximum possible value for timer T305 (720minutes) and verifies that no CELL_UPDATE message is received. After this, the SS changes the timer T305 value to 5 minutes. UE shall resume periodic cell updating procedure and transmit CELL UPDATE message 5 minutes after this modification.

Expected sequence

Cton	Direction	Manage	Commont
Step	Direction	Message	Comment
	UE SS		
1			The UE is in the CELL_FACH
			state. SS waits until T305 has
			expired.
2	\rightarrow	CELL UPDATE	IE "Cell update cause" shall be
			set to "Periodic cell updating"
3	←	CELL UPDATE CONFIRM	No RNTI identities are given.
			No information on PRACH and
			S-CCPCH are provided.
4			SS verifies that no uplink
			message is received from UE.
			SS waits for another period to
			allow T305 to expire.
5	\rightarrow	CELL UPDATE	Which is set to "periodical cell
			update" in IE "Cell update
			cause" for upon the expiry of
			timer T305.
6	←	CELL UPDATE CONFIRM	Including IEs "new C-RNTI",
	`	OLLE OF BATTLE OOTH IT III	"new U-RNTI" and IE "DRX
			IndicatorRRC State Indicator"
			is set to "Ne
			DRXCELL FACH"
7	\rightarrow	UTRAN MOBILITY INFORMATION	DIGIOLE I MOLI
'		CONFIRM	
1	1	OOM MAN	

	8			SS waits for a duration to allow timer T305 in the UE to expire.
Ì	9	_→	CELL UPDATE	·
	10	_ +	CELL UPDATE CONFIRM	IEs "PRACH Info" and "Secondary CCPCH Info" are included.
	11	→	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	
	12 8	+	MASTER INFORMATION BLOCK	SS changes the contents of
			SYSTEM INFORMATION BLOCK TYPE 21	MASTER INFORMATION BLOCK and SYSTEM INFORMATION BLOCK (see specific message contents). It waits for 720 minutes and checks that no CELL UPDATE message is transmitted on uplink PRACH channel.
	13 9	+	MASTER INFORMATION BLOCK SYSTEM INFORMATION BLOCK TYPE 2	SS modified the contents of MASTER INFORMATION BLOCK and SYSTEM INFORMATION BLOCK (see
				specific message contents) again.
	14 10	→	CELL UPDATE	UE shall transmit this message 5 minutes after step 139, with "cell update cause" set to "periodic cell updating"
1	15 11	\	CELL UPDATE CONFIRM	

CELL UPDATE (Step 2 and 5)

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to '0000 0000 0000 0000 0000 0001'
Integrity check info	Not checked
AM_RLC error indicator (for C-plane)	Check to see if set to 'FALSE'
AM_RLC error indicator (for U-plane)	Check to see if set to 'FALSE'
START List	Checked to see if the 'CN domain identity' and 'START'
	IEs are present for all CN domains supported by the UE
Cell Update Cause	Check to see if set to 'Periodic cell updating'
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Measured results on RACH	Check to see if it is absent
Protocol error information	Check to see if it is absent

CELL UPDATE CONFIRM (Step 3)

Use the same message sub-type found in Annex A.

CELL UPDATE CONFIRM (Step 6 and 4511)

Use the same message sub-type found in Annex A, with the following exceptions:

Information Element	Value/remark
New U-RNTI	
- SRNC Identity	Set to '0000 0000 0001'
- S-RNTI	Set to an arbitrary string different from '0000 0000 0000
	0000 0001'

CELL UPDATE (Step 9 and 14)

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0000 0001'
- S-RNTI	Check to see if set to same bit string as in IE "S-RNTI"
	in IE "U-RNTI" of the CELL UPDATE CONFIRM
	message sent in step 6.
Integrity check info	Not checked
AM_RLC error indicator (for C-plane)	Check to see if set to 'FALSE'
AM_RLC error indicator (for U-plane)	Check to see if set to 'FALSE'
START List	Checked to see if the 'CN domain identity' and 'START'
	IEs are present for all CN domains supported by the UE
Cell Update Cause	Check to see if set to 'Periodic cell updating'
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Measured results on RACH	Check to see if it is absent
Protocol error information	Check to see if it is absent

CELL UPDATE CONFIRM (Step 10)

Use the same message sub-type found in Annex A, with the following exceptions:

Information Element	Value/remark
PRACH Info (for RACH)	
- Available Signature	'000000000000011'B:Signature 0&1
— - Available SF	Refer to the parameter set for TS 34.108
- Preamble scrambling code number	0
Puncturing Limit	Refer to the parameter set for TS 34.108
Available Sub-Channel number	'000000000011'B:SubChNumber 0&1
Downlink DPCH info for one radio link	
Primary CPICH Info	
- Primary Scrambling Code	Set to the scrambling code for cell 1's P-CPICH.
PDSCH with SHO DCH Info	Not Present.
— - PDSCH code mapping	Not Present.
- Secondary CCPCH Info	
- Primary CPICH usage for channel estimation	TRUE
- Secondary CPICH Info	
	Not Present. Use default (Scrambling code of P-CPICH)
- channelization code	4
- Secondary scrambling code	Not Present. Use default (Scrambling code of P-CPICH)
	FALSE
Spreading factor	256
	255
	TRUE
- TFCI existence	TRUE
- Fixed or Flexible Position	Flexible
- Timing Offset	0 chips

MASTER INFORMATION BLOCK (Step 128)

Information Element	Value/remark
MIB Tag	2

SYSTEM INFORMATION BLOCK TYPE 2-1 (Step 428)

Information Element	Value/remark
UE Timers and constants in connected mode	No update
T305	Infinity

MASTER INFORMATION BLOCK (Step 139)

Information Element	Value/remark
MIB Tag	1

SYSTEM INFORMATION BLOCK TYPE 2-1 (Step 139)

Information Element	Value/remark
UE Timers and constants in connected mode	
T305	5 minutes

8.3.1.3.5 Test requirement

After step 1 the UE shall detect the expiry of timer T305 then transmits a CELL UPDATE message setting value "periodical cell update" into IE "Cell update cause".

After step 3 the UE shall not send any uplink message as a response to CELL UPDATE CONFIRM message sent in step 3.

After step 4 the UE shall send CELL UPDATE message, specifying the cell updating cause to be "periodic<u>al</u> cell update".

After step 6 the UE shall transmit an UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH.

After step 8 the UE shall send CELL UPDATE message, specifying the cell updating cause to be "periodic cell update".

After step 10 the UE shall transmit PHYSICAL CHANNEL RECONFIGURATION COMPLETE message using the PRACH radio resources allocated.

Between step $\frac{12-8}{8}$ and step $\frac{13-9}{8}$ the UE shall cease periodic cell updating activity and not transmit any CELL UPDATE messages.

After step <u>13-9</u> the UE shall transmit a CELL UPDATE message stating the cell update cause to be periodic updating, 5 minutes after the SS has modified the BCCH data.

8.3.1.4 Cell Update: periodical cell update in CELL_PCH

8.3.1.4.1 Definition

8.3.1.4.2 Conformance requirement

This procedure is to update UTRAN with the information of the current cell when the UE detects that it is still in the service area, while residing in the CELL PCH state, after the expiry of timer T305.

Reference

3GPP TS 25.331 clause 8.3.1

8.3.1.4.3 Test purpose

To confirm that the UE, in CELL_PCH state, executes a cell update procedure after the expiry of timer T305. To confirm that the UE sends an appropriate response message after receiving the CELL UPDATE CONFIRM message.

8.3.1.4.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CELL_PCH (state 6-1512) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE starts from CELL_PCH state. When the UE detects the expiry of periodic cell updating timer T305 according to the system information, the UE moves to CELL_FACH state. It shall transmit a CELL UPDATE message on the uplink CCCH and set the value "periodical cell update" into IE "Cell update cause". SS answers with a CELL UPDATE CONFIRM message, with IE "DRX Indicator RRC State Indicator" set to "No DRX CELL FACH". IEs related to RNTI, PRACH and S CCPCH are omitted from this message. SS checks to confirm that the UE does not send a response. SS sends a UTRAN MOBILITY INFORMATION message to the UE, stating the new C-RNTI identity to be used and also setting IE "DRX IndicatorRRC State Indicator" to "DRX with cell updatingCELL PCH". The UE shall reply with UTRAN MOBILITY INFORMATION CONFIRM message and transit to CELL_PCH state. Next, SS stays idle until timer T305 is once again expired. The UE shall transmit CELL UPDATE message in order to initiate cell updating procedure. After the SS receives this message, it transmits a CELL UPDATE CONFIRM message which includes the IEs "new C-RNTI" and "new U-RNTI". Then the UE shall transmit an UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH and before entering CELL_PCH state. Next, the content of the SYSTEM INFORMATION BLOCK TYPE 1 is changed to disable periodic cell updating. SS then monitors the uplink CCCH for a period up to the maximum possible value for timer T305 (720minutes) and verifies that no CELL UPDATE message is received. After this, the SS changes the timer T305 value to 5 minutes. UE shall resume periodic cell updating procedure and transmit CELL UPDATE message 5 minutes after this modification. In the next sequence, SS assigns a new U RNTI identity to the UE by transmitting UTRAN MOBILITY INFORMATION message again. Once again, the IE "DRX Indicator" is set to "DRX with cell updating". After reception of this message, the UE moves to CELL_PCH state and start to listen to the paging sub-channels derived based on the value of new U RNTI given. Finally, SS waits until T305 has expired once more, the UE shall send CELL UPDATE message again on uplink CCCH. SS replies with CELL UPDATE CONFIRM which includes IEs "PRACH Info" and "Secondary CCPCH Info". The UE shall transmit PHYSICAL CHANNEL RECONFIGURATION COMPLETE message on the uplink DCCH using PRACH indicated in CELL UPDATE CONFIRM message.

Expected sequence

Step	Direction	Message	Comment
1	UE SS		The UE is in the CELL PCH
'			state. SS waits until T305 has
			expired.
2	\rightarrow	CELL UPDATE	The UE shall move to
			CELL_FACH state and
			transmits this message with
			the IE " Cell update cause" set
			to "periodic <u>al</u> cell update".
3	←	CELL UPDATE CONFIRM	Does not include IEs "PRACH
			Info", "Secondary CCPCH Info", "new C-RNTI" or "new U-
			RNTI".
4			SS verifies that no response
			message is received.
5	+	UTRAN MOBILITY INFORMATION	Allocates a new C-RNTI
			identity to UE and set IE "DRX
			IndicatorRRC State Indicator"
			to "DRX with Cell
	→	UTRAN MOBILITY INFORMATION	updatingCELL PCH". UE shall move to CELL PCH
6	7	CONFIRM	state
7		OOM INW	SS remains idle for a period
'			sufficient for T305 timer to
			expire.
8	\rightarrow	CELL UPDATE	This message shall contain
			the new U-RNTI identity
			assigned in step 5. <u>UE moves</u>
			t o CELL FACH to transmit
9	←	CELL UPDATE CONFIRM	this message. New C-RNTI and U-RNTI
9	`	CELL OF DATE CONFIRM	identities are assigned.
10	\rightarrow	UTRAN MOBILITY INFORMATION	UE moves back to CELL_PCH
		CONFIRM	after transmitting this
			message.
11	_←	UTRAN MOBILITY INFORMATION	SS assigns a new U-RNTI
			identity to the UE. At the same
			time, it sets IE "DRX Indicator" to "DRX with cell updating".
12	_→	UTRAN MOBILITY INFORMATION	UE shall move to CELL PCH
'-	_	CONFIRM	state after sending this
			message.
13			SS waits for T305 to expire.
14		CELL UPDATE	
15		CELL UPDATE CONFIRM	Allocates new common
			physical resources by
			specifying IEs "PRACH Info" and "Secondary CCPCH Info".
16		PHYSICAL CHANNEL	and coordary cor crimits.
		RECONFIGURATION COMPLETE	
<u>11</u>	<u></u>	MASTER INFORMATION BLOCK	SS changes the contents of
		OVOTEM INFORMATION DI COM TYPE :	MA OTED IN ECONAL TICK
		SYSTEM INFORMATION BLOCK TYPE 1	MASTER INFORMATION
			BLOCK and SYSTEM INFORMATION BLOCK (see
			specific message contents). It
			waits for 720 minutes and
			checks that no CELL UPDATE
			message is transmitted on
			uplink PRACH channel.
		<u>l</u>	

<u>12</u>		MASTER INFORMATION BLOCK	SS modified the contents of
		SYSTEM INFORMATION BLOCK TYPE 1	MASTER INFORMATION BLOCK and SYSTEM INFORMATION BLOCK (see specific message contents) again.
13	<u>→</u>	CELL UPDATE	UE shall transmit this message 5 minutes after step 12, with "cell update cause" set to "periodic cell updating"
<u>14</u>	<u>←</u>	CELL UPDATE CONFIRM	

CELL UPDATE (Step 2)

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to '0000 0000 0000 0000 0000 0001'
Integrity check info	Not checked
AM_RLC error indicator (for C-plane)	Check to see if set to 'FALSE'
AM_RLC error indicator (for U-plane)	Check to see if set to 'FALSE'
START List	Checked to see if the 'CN domain identity' and 'START'
	IEs are present for all CN domains supported by the UE
Cell Update Cause	Check to see if set to 'Periodic cell updating'
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Measured results on RACH	Check to see if it is absent
Protocol error information	Check to see if it is absent

CELL UPDATE CONFIRM (Step 3 and 13)

Use the same message sub-type found in Annex A.

UTRAN MOBILITY INFORMATION (Step 5-and step 11)

Use the same message sub-type found in Annex A, with the following exceptions:

Information Element	Value/remark
New U-RNTI	
- SRNC Identity	Set to '0000 0000 0001'
- S-RNTI	Set to '0000 0000 0000 0000 1111'
New C-RNTI	Set to '0000 0000 0000 1111'
DRX IndicatorRRC State Indicator	DRX with cell updatingCELL_PCH

CELL UPDATE (Step 8)

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to '0000 0000 0000 0000 1111'
Integrity check info	Not checked
AM_RLC error indicator (for C-plane)	Check to see if set to 'FALSE'
AM_RLC error indicator (for U-plane)	Check to see if set to 'FALSE'
START List	Checked to see if the 'CN domain identity' and 'START'
	IEs are present for all CN domains supported by the UE
Cell Update Cause	Check to see if set to 'Periodic cell updating'
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Measured results on RACH	Check to see if it is absent
Protocol error information	Check to see if it is absent

CELL UPDATE CONFIRM (Step 9)

Use the same message sub-type found in Annex A, with the following exceptions:

Information Element	Value/remark
New U-RNTI	
- SRNC Identity	Set to '0000 0000 0001'
- S-RNTI	Set to '0000 0000 0000 1010'
New C-RNTI	Set to '0000 0000 0000 0101'

CELL UPDATE (Step 14)

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
-S-RNTI	Check to see if set to '0000 0000 0000 0000 1111'
Integrity check info	Not checked
AM_RLC error indicator (for C-plane)	Check to see if set to 'FALSE'
AM_RLC error indicator (for U-plane)	Check to see if set to 'FALSE'
START List	Checked to see if the 'CN domain identity' and 'START'
	IEs are present for all CN domains supported by the UE
Cell Update Cause	Check to see if set to 'Periodic cell updating'
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Measured results on RACH	Check to see if it is absent
Protocol error information	Check to see if it is absent

CELL UPDATE CONFIRM (Step 15)

Use the same message sub-type found in Annex A, with the following exceptions:

Information Element	Value/remark
PRACH Info (for RACH)	
Available Signature	'0000000000000011'B:Signature 0&1
- Available SF	Refer to the parameter set for TS 34.108
- Preamble scrambling code number	0
Puncturing Limit	Refer to the parameter set for TS 34.108
- Available Sub-Channel number	'00000000011'B:SubChNumber 0&1
Downlink DPCH info for one radio link	
- Primary CPICH Info	
- Primary Scrambling Code	Set to the scrambling code for cell 1's P-CPICH.
PDSCH with SHO DCH Info	Not Present.
	Not Present.
Secondary CCPCH Info	
- Primary CPICH usage for channel estimation	TRUE
- Secondary CPICH Info	
	Not Present. Use default (Scrambling code of P-CPICH)
- Channelization code	4
	Not Present. Use default (Scrambling code of P-CPICH)
	FALSE
- Spreading factor	256
— - Code number	255
- Pilot symbol existence	TRUE
- TFCI existence	TRUE
	Flexible
- Timing Offset	0 chips

MASTER INFORMATION BLOCK (Step 11)

Information Element	<u>Value/remark</u>	
MIB Tag	2	

SYSTEM INFORMATION BLOCK TYPE 1 (Step 11)

Information Element	<u>Value/remark</u>
UE Timers and constants in connected mode	No update
<u>T305</u>	Infinity

MASTER INFORMATION BLOCK (Step 12)

Information Element	<u>Value/remark</u>	
MIB Tag	1	

SYSTEM INFORMATION BLOCK TYPE 1 (Step 12)

Information Element	<u>Value/remark</u>
UE Timers and constants in connected mode	
<u>T305</u>	5 minutes

8.3.1.4.5 Test requirement

After step 2 the UE shall detect the expiry of timer T305, it shall then move to CELL_FACH state and transmits a CELL UPDATE message with the IE "Cell update cause" set to "periodical cell update".

After step 5 the UE shall reply with UTRAN MOBILITY INFORMATION CONFIRM message. It shall subsequently move to CELL_PCH state.

After step 7 the UE shall initiate a cell updating procedure by the transmission of a CELL UPDATE message. In this message, it shall indicate the new U-RNTI value assigned in step 5 and also set IE "Cell Updating Cause" to "Periodic<u>al</u> Cell Updating".

Between step 11 and step 12 the UE shall cease periodic cell updating activity and not transmit any CELL UPDATE messages.

After step 12 the UE shall transmit a CELL UPDATE message stating the cell update cause to be periodic updating, 5 minutes after the SS has modified the BCCH data.

After step 11 the UE shall reply with UTRAN MOBILITY INFORMATION CONFIRM message. It shall subsequently move to CELL PCH state.

After step 13 the UE shall send a CELL UPDATE message. After step 15 the UE shall transmit PHYSICAL CHANNEL RECONFIGURATION COMPLETE message on the new PRACH channel assigned by the received CELL UPDATE CONFIRM message in step 15.

8.3.1.5 Cell Update: UL data transmission in URA PCH

8.3.1.5.1 Definition

8.3.1.5.2 Conformance requirement

This procedure is to update UTRAN with the current cell information if the UE wants to transmit ULuplink data while in URA_PCH state.

Reference

3GPP TS 25.331 clause 8.3.1

8.3.1.5.3 Test purpose

To confirm that the UE executes a cell update procedure when the UE transmits <u>ULuplink</u> data if the UE is in URA_PCH state. To confirm that the UE sends the correct response to CELL UPDATE CONFIRM message, after it has taken into consideration the current TFS and/or TFCS settings.

8.3.1.5.4 Method of test

Initial Condition

System Simulator: 1cell

UE: URA_PCH (state 6-1613) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE starts from URA_PCH state, after the operator initiates an outgoing packet data transmission. The UE then moves to CELL_FACH state and transmits a CELL UPDATE message to the SS on the uplink CCCH, with the IE "Cell update cause" set to value "Uuplink -data transmission". After receiving such a message, SS transmits CELL UPDATE CONFIRM message without specifying IE "new C-RNTI" or IE "new U-RNTI" or "CN information elements" or " Physical channel information elements" or "Transport channel information elements" or RB information elements".any of the following IEs: "new C RNTI", "new U RNTI", "PRACH Info" and "Secondary CCPCH Info". The UE shall stay in CELL_FACH state and transmit PHYSICAL CHANNEL RECONFIGURATION COMPLETEno response message on the DCCH. SS waits until uplink data transmission is completed and sends a UTRAN MOBILITY INFORMATION message. The UE shall reply with UTRAN MOBILITY INFORMATION CONFIRM message. Since the IE "DRX IndicatorRRC State Indicator" is set to "DRX with URA updatingURA_PCH" in the downlink message, the UE shall move to URA_PCH state. Then SS prompts the test operator to transmit packet data. The UE shall send CELL UPDATE message once more. After the SS receives this message, it replies with a CELL UPDATE CONFIRM message including "Physical channel information elements". The IE "RRC State Indicator" is set to "CELL_FACH" in this message. The UE shall transmit PHYSICAL CHANNEL RECONFIGURATION COMPLETE message. Then the UE shall enter CELL FACH state and proceed to transmit packet data. Next SS modifies the contents of SYSTEM INFORMATION BLOCK TYPE 5 and 6 messages. See specific message contents for further details. Finally, SS then prompts the test operator to transmit packet data once again. The UE shall send CELL UPDATE message and specifies the cause to be "Uluplink data transmission". SS replies with the default CELL UPDATE CONFIRM message defined in TS 34.108 including "Transport channel information elements". After receiving this message, the UE shall transmit TRANSPORT CHANNEL RECONFIGURATION COMPLETE message and remains in CELL_FACH state. SS again waits for the transmission of user packet data to complete and then sends UTRAN MOBILITY INFORMATION message on the downlink DCCH. The IE "DRX Indicator RRC State Indicator" is set to "DRX with URA updatingURA_PCH". The UE shall move to URA_PCH state and after it acknowledges that it starts using the new

RNTI identity provided. Finally, SS prompts the test operator to transmit packet data. The UE shall send CELL UPDATE message once more. After the SS receives this message, it replies with a CELL UPDATE CONFIRM message including the IEs "PRACH info" and "Secondary CCPCH info". The IE "DRX Indicator" is set to "No DRX" in this message. The UE shall transmit PHYSICAL CHANNEL RECONFIGURATION COMPLETE message on the uplink PRACH specified in the CELL UPDATE CONFIRM message. Then the UE shall enter CELL_FACH state and proceed to transmit packet data.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1			The UE is brought to URA_PCH state. SS prompts the test operator to begin a packet data transmission.
2	\rightarrow	CELL UPDATE	The UE shall move to CELL FACH state with the message set to "ULuplink-data transmission" in IE "Cell update cause".
3	+	CELL UPDATE CONFIRM	Use default message content.
4	- →	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	SS check that UE does not send response message.
5			SS waits until transmission of uplink data has been completed.
6	+	UTRAN MOBILITY INFORMATION	IE "DRX IndicatorRRC State Indicator" set to "DRX with URA update"
7	\rightarrow	UTRAN MOBILITY INFORMATION CONFIRM	UE moves to URA_PCH state
<u>8</u>			SS prompts test operator to initiate a packet data transmission.
<u>9</u>	<u> </u>	CELL UPDATE	Should be same as in step 2
<u>10</u>		CELL UPDATE CONFIRM	Including "Physical channel information elements".
<u>11</u>	<u> </u>	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	inemater elemente :
<u>12</u>			SS waits until transmission of uplink data has been completed.
<u>13</u>	<u></u>	UTRAN MOBILITY INFORMATION	IE "RRC State Indicator" set to "DRX with URA update"
<u>14</u>	<u> </u>	UTRAN MOBILITY INFORMATION CONFIRM	
8 15			SS modifies MASTER INFORMATION BLOCK, SYSTEM INFORMATION BLOCK TYPE 5 and 6 messages. SS prompts test operator to initiate a packet data transmission.
9 16	\rightarrow	CELL UPDATE	Should be same as in step 2
10 17	<u> </u>	CELL UPDATE CONFIRM	Use default message content.
11 18	→	TRANSPORT CHANNEL RECONFIGURATION COMPLETE	
12			SS waits until transmission of uplink data has been completed.
13	- ←	UTRAN MOBILITY INFORMATION	IE "DRX Indicator" set to "DRX with URA update"
14	- →	UTRAN MOBILITY INFORMATION CONFIRM	UE moves to URA_PCH state
15			SS prompts test operator to initiate a packet data transmission.
16	_→	CELL UPDATE	Should be same as in step 2
17	- ←	CELL UPDATE CONFIRM	Including the IEs" PRACH info"," Secondary CCPCH"

18	_ →	PHYSICAL CHANNEL	SS verifies that this message
		RECONFIGURATION COMPLETE	is received from PRACH
			physical channel using the
			configurations set in IE
			"PRACH Info" in CELL
			UPDATE CONFIRM message.

CELL UPDATE (Step 2, 9 and 16)

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to '0000 0000 0000 0000 1111'
Integrity check info	Not checked
AM_RLC error indicator (for C-plane)	Check to see if set to 'FALSE'
AM_RLC error indicator (for U-plane)	Check to see if set to 'FALSE'
START List	Checked to see if the 'CN domain identity' and 'START'
	IEs are present for all CN domains supported by the UE
Cell Update Cause	Check to see if set to 'ULuplink- dData tTransmission'
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Measured results on RACH	Check to see if it is absent
Protocol error information	Check to see if it is absent

CELL UPDATE CONFIRM (Step 3-and 10)

Use the same message sub-type found in Annex A.

CELL UPDATE CONFIRM (Step 10)

Use the same message sub-type found in step 3, with the following exceptions:

Information Element	<u>Value/remark</u>
Maximum allowed uplink TX power	3 dB below the follow value:
	Minimum of { 33 dBm, maximum uplink power allowed
	under the UE power class }

CELL UPDATE CONFIRM (Step 17)

Use the same message sub-type found in Annex Astep 10, with the following exceptions:

Added or Reconfigured uplink TrCH information	
-Transport channel identity	1
-TFS	(This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set
-Dynamic Transport format information	Reference to TS34.108 clause 6.10 Parameter Set
-Number of Transport blocks	Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set
-RLC size	Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set
-Semi-static Transport Format information	Reference to TS34.108 clause 6.10 Parameter Set
-Transmission time interval	
-Type of channel coding	
-Coding Rate	
-Rate matching attribute	
-CRC size	

Information Element	Value/remark	
PRACH Info (for RACH)		
Available Signature	'0000000000000001'B:Signature 0	
- Available SF	Refer to the parameter set for TS 34.108	
- Preamble scrambling code number	0	
Puncturing Limit	Refer to the parameter set for TS 34.108	
- Available Sub-Channel number	<u>'000000000001'B:SubChNumber 0</u>	
Downlink DPCH info for one radio link		
Primary CPICH Info		
- Primary Scrambling Code	Set to the scrambling code for cell 1's P-CPICH.	
	Not Present.	
PDSCH code mapping	Not Present.	
— - Secondary CCPCH Info		
- Primary CPICH usage for channel estimation	TRUE	
	Not Present. Use default (Scrambling code of P-CPICH)	
	4	
	Not Present. Use default (Scrambling code of P-CPICH)	
	FALSE	
	256	
— - Code number	255	
- Pilot symbol existence	TRUE	
- TFCI existence	TRUE	
	Flexible	
— - Timing Offset	0 chips	

UTRAN MOBILITY INFORMATION (Step 6)

Use the same message sub-type found in Annex A, with the following exceptions:

Information Element	Value/remark
New C-RNTI	'0000 0000 0000 1111'
DRX IndicatorRRC State Indicator	DRX with URA updating URA_PCH

UTRAN MOBILITY INFORMATION (Step 13)

Use the same message sub-type found in Annex A, with the following exceptions:

Information Element	Value/remark
New C-RNTI	'0000 0000 1111 1111'
DRX Indicator RRC State Indicator	DRX with URA updatingURA_PCH

PHYSICAL CHANNEL RECONFIGURATION COMPLETE (Step 4 and 1811)

Only the message type IE in this message will be checked.

TRANSPORT CHANNEL RECONFIGURATION COMPLETE (Step 4218)

Only the message type IE in this message will be checked.

MASTER INFORMATION BLOCK (Step 8)

Use the same message sub-type found in Clause 6.1 of TS 34.108, with the following exceptions:

Information Element	Value/remark
MIB Tag	2

SYSTEM INFORMATION BLOCK TYPE 5 and TYPE 6 (Step 8)

Use the same message sub-type found in Clause 6.1 of TS34.108, with the following exceptions:

Information Element	Value/remark
PRACH system information	
	Common transport channels
- Dynamic Transport Format Information	·
- Number of Transport blocks	1
	296 bits
- Semi-static Transport Format	
- Transmission time interval	80 msec
- Type of channel coding	No coding
	No Present
	4
	16 bits

8.3.1.5.5 Test requirement

After step 1 the UE shall move to CELL_FACH state to initiate a cell update procedure and transmits a CELL UPDATE message which is set to "ULuplink-data transmission" in IE "Cell update cause".

After step 3 the UE shall transmit PHYSICAL CHANNEL RECONFIGURATION COMPLETE message on the uplink DCCH.

After step 8 the UE shall initiate cell update procedure and transmit CELL UPDATE message on the uplink CCCH. The IE "Cell update cause" shall be set to "ULuplink-data transmission".

After step 10 the UE shall transmit TRANSPORT-PHYSICAL CHANNEL RECONFIGURATION COMPLETE message on the uplink DCCH.

After step 15 the UE shall initiate cell update procedure and transmit CELL UPDATE message on the uplink CCCH. The IE "Cell update cause" shall be set to "ULuplink data transmission".

After step 17 the UE shall transmit PHYSICAL TRANSPORT CHANNEL RECONFIGURATION COMPLETE message on the uplink DCCH using the PRACH resources assigned in CELL UPDATE CONFIRM, which is sent in step 17.

8.3.1.6 Cell Update: UL data transmission in CELL PCH

8.3.1.6.1 Definition

8.3.1.6.2 Conformance requirement

This procedure is to update UTRAN with the current cell of the UE if the UE wants to transmit ULuplink data when the UE is in CELL_PCH state.

Reference

3GPP TS 25.331 clause 8.3.1

8.3.1.6.3 Test purpose

To confirm that the UE executes a cell update procedure when the UE transmits <u>ULuplink</u> data if the UE is in CELL_PCH state. To confirm that the UE sends the correct response to CELL UPDATE CONFIRM message, after it has taken into consideration the current TFS and/or TFCS settings.

8.3.1.6.4 Method of test

Initial Condition

System Simulator: 1cell

UE: CELL_PCH (state 6-1512) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_PCH state. SS asks the test operator to send some packet data. The UE moves to CELL_FACH state and transmits a CELL UPDATE message to the SS on the uplink CCCH, which shall indicate "ULuplink data transmission" in IE "Cell update cause". After receiving such a message, SS transmits default CELL UPDATE CONFIRM message. The UE shall stay in CELL_FACH state and NOT transmit PHYSICAL CHANNEL RECONFIGURATION COMPLETE response message on the DCCH. SS waits until uplink data transmission is completed and sends a UTRAN MOBILITY INFORMATION message. The UE shall reply with UTRAN MOBILITY INFORMATION CONFIRM message. Since the IE "DRX Indicator RRC State Indicator" is set to "DRX with cell updatingCELL_PCH" in the downlink message, the UE shall move to CELL_PCH state. Next SS modifies the contents of SYSTEM INFORMATION BLOCK TYPE 5 and 6 messages. See specific message contents for further details. SS then prompts the test operator to transmit packet data. The UE shall send CELL UPDATE message and specifies the cause to be "ULuplink data transmission". SS replies with a CELL UPDATE CONFIRM message which assigns a new C-RNTI to the UE. After receiving this message, the UE shall transmit UTRAN MOBILITY INFORMATION CONFIRM TRANSPORT CHANNEL RECONFIGURATION COMPLETE message and remains in CELL_FACH state. SS again waits for the transmission of user packet data to complete and then sends UTRAN MOBILITY INFORMATION message on the downlink DCCH. The IE "DRX Indicator RRC State Indicator" is set to "DRX with cell updating CELL PCH". The UE shall move to CELL_PCH state and acknowledges that it starts using the new RNTI identity provided. Finally, SS prompts the test operator to transmit packet data. The UE shall send CELL UPDATE message once more. After the SS receives this message, it transmits a CELL UPDATE CONFIRM message which includes the "Physical channel information elements" IEs "PRACH info" and set IE "DRX Indicator RRC State Indicator" to "No DRXCELL_FACH". Then the UE shall enter to the CELL_FACH state after sending PHYSICAL CHANNEL RECONFIGURATION COMPLETE message.

	Step	Direction UE SS	Message	Comment
-	1	<u> </u>		The UE is in the CELL_PCH state. SS prompts the test operator to initiate a packet data call.
	2	→	CELL UPDATE	The UE moves to CELL FACH state and transmit this message which is set to "ULuplink data transmission" in IE "Cell update cause".
Ī	3	←	CELL UPDATE CONFIRM	Use default message content.
	4	>	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	
	5			SS waits until transmission of uplink data has been completed.
	6	+	UTRAN MOBILITY INFORMATION	IE "DRX IndicatorRRC State Indicator" set to "DRX with cell update"
_	7	\rightarrow	UTRAN MOBILITY INFORMATION CONFIRM	UE moves to CELL_PCH state.
	8			SS modifies MASTER INFORMATION BLOCK, SYSTEM INFORMATION BLOCK TYPE 5 and 6 messages. SS prompts test operator to initiate a packet data transmission.
	9	\rightarrow	CELL UPDATE	Should be same as in step 2
	10	+	CELL UPDATE CONFIRM	Including the IE "new C- RNTI". Use default message content.
	11	→	UTRAN MOBILITY INFORMATION CONFIRMTRANSPORT CHANNEL RECONFIGURATION COMPLETE	
	12			SS waits until transmission of uplink data has been completed.
	13	+	UTRAN MOBILITY INFORMATION	IE "DRX IndicatorRRC State Indicator" set to "DRX with cell update"
	14	\rightarrow	UTRAN MOBILITY INFORMATION CONFIRM	UE moves to CELL_PCH state.
	15			SS prompts test operator to initiate a packet data transmission.
	16	\rightarrow	CELL UPDATE	Should be same as in step 2
	17	+	CELL UPDATE CONFIRM	Including the "Physical channel information elements" IE "PRACH info" and set IE "DRX Indicator" to "No DRX".
	18	→	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	SS verifies that this message is received from PRACH physical channel using the configurations set in IE "PRACH Info" in CELL UPDATE CONFIRM message.

CELL UPDATE (Step 2, 9 and 16)

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to '0000 0000 0000 0000 1111'
Integrity check info	Not checked
AM_RLC error indicator (for C-plane)	Check to see if set to 'FALSE'
AM_RLC error indicator (for U-plane)	Check to see if set to 'FALSE'
START List	Checked to see if the 'CN domain identity' and 'START'
	IEs are present for all CN domains supported by the UE
Cell Update Cause	Check to see if set to 'ULuplink Ddata Ttransmission'
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Measured results on RACH	Check to see if it is absent
Protocol error information	Check to see if it is absent

CELL UPDATE CONFIRM (Step 3 and 10)

Use the same message sub-type found in Annex A.

CELL UPDATE CONFIRM (Step 10)

Use the same message sub-type found in step 3, with the following exceptions:

Information Element	<u>Value/remark</u>
New C-RNTI	<u>'0000 0000 1111 0000'</u>

CELL UPDATE CONFIRM (Step 17)

Use the same message sub-type found in Annex Astep 3, with the following exceptions:

Maximum allowed uplink TX power	3 dB below the follow value:
	Minimum of {33 dBm, maximum uplink power allowed
	under the UE power class }

Information Element	Value/remark
PRACH Info (for RACH)	
Available Signature	'000000000000001'B:Signature 0
Available SF	Refer to the parameter set for TS 34.108
Preamble scrambling code number	0
	Refer to the parameter set for TS 34.108
- Available Sub-Channel number	<u>'000000000001'B:SubChNumber 0</u>
Downlink DPCH info for one radio link	
Primary CPICH Info	
- Primary Scrambling Code	Set to the scrambling code for cell 1's P-CPICH.
PDSCH with SHO DCH Info	Not Present.
— - PDSCH code mapping	Not Present.
— - Secondary CCPCH Info	
 Primary CPICH usage for channel estimation 	TRUE
	Not Present. Use default (Scrambling code of P-CPICH)
	4
— - Secondary scrambling code	Not Present. Use default (Scrambling code of P-CPICH)
——- STTD indicator	FALSE
— - Spreading factor	256
— - Code number	255
- Pilot symbol existence	TRUE
— - TFCI existence	TRUE
	Flexible
— - Timing Offset	0 chips

UTRAN MOBILITY INFORMATION (Step 6)

Use the same message sub-type found in Annex A, with the following exceptions:

Information Element	Value/remark
New C-RNTI	'0000 0000 0000 1111'
DRX IndicatorRRC State Indicator	DRX with cell updatingCELL_PCH

UTRAN MOBILITY INFORMATION (Step 13)

Use the same message sub-type found in Annex A, with the following exceptions:

Information Element	Value/remark
New C-RNTI	'0000 0000 1111 1111'
DRX Indicator RRC State Indicator	DRX with cell updatingCELL_PCH

PHYSICAL CHANNEL RECONFIGURATION COMPLETE (Step 4-and-18)

Only the message type IE in this message will be checked.

TRANSPORT CHANNEL RECONFIGURATION COMPLETE (Step 12)

Only the message type IE in this message will be checked.

MASTER INFORMATION BLOCK (Step 8)

Use the same message sub-type found in Clause 6.1 of TS 34.108, with the following exceptions:

Information Element	Value/remark
MIB Tag	2

SYSTEM INFORMATION BLOCK TYPE 5 and TYPE 6 (Step 8)

Use the same message sub-type found in Clause 6.1 of TS 34.108, with the following exceptions:

Information Element	Value/remark
PRACH system information	
— - RACH TES	
	Common transport channels
Dynamic Transport Format Information	·
- Number of Transport blocks	1
	296 bits
- Semi-static Transport Format	
- Transmission time interval	80 msec
- Type of channel coding	No coding
	No Present
- Rate matching attribute	4
	16 bits

8.3.1.6.5 Test requirement

After step 1 the UE shall move to CELL_FACH state, initiate a cell update procedure for the UL data transmission, and transmit a CELL UPDATE message which is set to "ULuplink data transmission" in IE "Cell update cause".

After step 3 the UE shall transmit PHYSICAL CHANNEL RECONFIGURATION COMPLETE message on the uplink DCCH.

After step 8 the UE shall initiate cell update procedure and transmit CELL UPDATE message on the uplink CCCH. The IE "Cell update cause" shall be set to "ULuplink data transmission".

After step 10 the UE shall transmit <u>UTRAN MOBILITY INFORMATION CONFIRMTRANSPORT CHANNEL RECONFIGURATION COMPLETE</u> message on the uplink DCCH.

After step 15 the UE shall initiate cell update procedure and transmit CELL UPDATE message on the uplink CCCH. The IE "Cell update cause" shall be set to "ULuplink data transmission".

After step 17 the UE shall transmit PHYSICAL CHANNEL RECONFIGURATION COMPLETE message on the uplink DCCH-using the PRACH resources assigned in CELL UPDATE CONFIRM, which is sent in step 17.

8.3.1.7 Cell Update: paging response in URA PCH

8.3.1.7.1 Definition

8.3.1.7.2 Conformance requirement

This procedure is to update UTRAN with the current cell of the UE after it receives a PAGING TYPE 1 message addressed to itself while it is in URA_PCH state.

Reference

3GPP TS 25.331 clause 8.3.1

8.3.1.7.3 Test purpose

To confirm that the UE executes a cell update procedure when it receives a PAGING TYPE 1 message while operating in URA_PCH state. To confirm that the UE responds with an appropriate uplink message after receiving a CELL UPDATE CONFIRM message during cell updating procedure triggered by paging.

8.3.1.7.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: URA_PCH (state 6-1613) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is brought to URA_PCH state. SS transmits a PAGING TYPE 1 message to page for the UE, setting IE "paging originator" to "UTRAN Originator". The UE shall move to CELL_FACH state and transmits a CELL UPDATE message to the SS on the uplink CCCH. This message shall set IE "Cell update cause" to "Paging Response". After the SS receives this message, it transmits the default CELL UPDATE CONFIRM message. The UE shall stay in CELL_FACH state and not transmit PHYSICAL CHANNEL RECONFIGURATION COMPLETEresponse message on the DCCH. SS then sends a UTRAN MOBILITY INFORMATION message. The UE shall reply with UTRAN MOBILITY INFORMATION CONFIRM message. In this message, the IE "DRX Indicator RRC State Indicator" is set to "DRX with URA updating URA_PCH". As a result, the UE shall reply with UTRAN MOBILITY INFORMATION and move to URA PCH state. Next SS modifies the contents of SYSTEM INFORMATION BLOCK TYPE 5 and 6 messages. See specific message contents for further details. SS pages the UE again using PAGING TYPE 1 message. The UE shall send CELL UPDATE message once more. SS transmits a CELL UPDATE CONFIRM message which includes the IEs "new C-RNTI", "new U-RNTI" on the downlink DCCH. The UE shall send UTRAN MOBILITY INFORMATION CONFIRM message. In the next sequence, SS transmits UTRAN MOBILITY INFORMATION message and assigns a new C-RNTI identity to the UE. In this message, the IE "RRC State Indicator" is set to "URA PCH". The UE shall reply with UTRAN MOBILITY INFORMATION CONFIRM message and move to URA PCH state. SS then pages the UE once more using PAGING TYPE 1 message. The UE shall initial the cell update procedure by dispatching a CELL UPDATE message on the uplink CCCH. SS replies with a CELL UPDATE CONFIRM message which includes "Physical channel information elements". The UE shall respond with a PHYSICAL CHANNEL RECONFIGURATION COMPLETE. SS sends another UTRAN MOBILITY INFORMATION message on the downlink DCCH, with the IE"RRC State Indicator" set to "URA_PCH". The UE shall move to URA PCH state and acknowledges that it starts using the new C-RNTI identity provided by transmitting UTRAN MOBILITY INFORMATION CONFIRM message. SS pages the UE again using the U-RNTI identity. The UE shall send CELL UPDATE message and specify the cause to be "Paging Response". SS replies with the default CELL UPDATE CONFIRM message which includes "Transport channel information elements" again. After receiving this message, the UE shall transmit TRANSPORT CHANNEL RECONFIGURATION COMPLETE message and remains in CELL_FACH state. SS sends another UTRAN MOBILITY INFORMATION message on the downlink DCCH, with the IE "DRX Indicator" set to "DRX with URA updating". The UE shall move to URA PCH state and acknowledges that it starts using the new C RNTI identity provided by transmitting UTRAN MOBILITY INFORMATION CONFIRM message. SS pages the UE again using PAGING TYPE 1 message. The UE shall send CELL UPDATE message once more. SS transmits a CELL UPDATE CONFIRM message which includes the IEs "new C RNTI", "new U RNTI" on the downlink DCCH. The UE shall send PHYSICAL CHANNEL RECONFIGURATION COMPLETE message. In the next sequence, SS transmits UTRAN MOBILITY INFORMATION message and assigns a new C RNTI identity to the UE. The UE shall reply with UTRAN MOBILITY INFORMATION CONFIRM message and move to URA_PCH state. SS then pages the UE once more using PAGING TYPE 1 message. The UE shall initial the cell update procedure by dispatching a CELL UPDATE message on the uplink CCCH. SS replies with a CELL UPDATE CONFIRM message which includes information of the PRACH channel to be used. The UE shall respond with a PHYSICAL CHANNEL RECONFIGURATION COMPLETE, sent on the DCCH supported by the PRACH indicated in CELL UPDATE CONFIRM message.

Expected sequence

Step	Direction UE SS	Message	Comment
1	1		The UE is first brought to URA_PCH state.
2	+	PAGING TYPE 1	SS transmits a PAGING TYPE 1 message to the UE which includes the UE's assigned U- RNTI with the IE "paging originator" set to "UTRAN originator".
3	→	CELL UPDATE	The UE shall move to CELL_FACH state and transmits this message with the IE "Cell update cause" set to "paging response".
4	+	CELL UPDATE CONFIRM	Use the default message from TS 34.108 Clause 8.
5	->	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	The UE shall move to CELL_FACH state and not transmits this response message.
6	+	UTRAN MOBILITY INFORMATION	Allocates a new C-RNTI and set IE "DRX IndicatorRRC State Indicator" to "DRX with URA updatingURA PCH".
7	\rightarrow	UTRAN MOBILITY INFORMATION CONFIRM	UE moves to URA_PCH state.
8			SS modifies MASTER INFORMATION BLOCK, SYSTEM INFORMATION BLOCK TYPE 5 and 6 messages. The TFS of the PRACH is changed.
9	+	PAGING TYPE 1	SS pages the UE again
10	\rightarrow	CELL UPDATE	IE "Cell update cause" shall be set to "paging response".
11	←	CELL UPDATE CONFIRM	Includes IE "new C-RNTI" and "new U-RNTI". Use the default message from TS 34.108 Clause 8.
12	→	UTRAN MOBILITY INFORMATION CONFIRMTRANSPORT CHANNEL RECONFIGURATION COMPLETE	The UE shall move to CELL_FACH state and transmits this message.
13	←	UTRAN MOBILITY INFORMATION	Allocates a new C-RNTI and set IE "DRX IndicatorRRC State Indicator" to "DRX with URA updatingURA PCH".
14	\rightarrow	UTRAN MOBILITY INFORMATION CONFIRM	UE moves to URA_PCH state.
15	+	PAGING TYPE 1	SS pages the UE again
16	→	CELL UPDATE	IE "Cell update cause" shall be set to "paging response".
17	+	CELL UPDATE CONFIRM	"Physical channel information elements" IE "new C-RNTI" is included in this message
18	→	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	
19	+	UTRAN MOBILITY INFORMATION	Allocates a new C-RNTI and set IE "DRX IndicatorRRC State Indicator" to "DRX with URA updatingURA PCH".
20	→	UTRAN MOBILITY INFORMATION CONFIRM	UE moves to URA_PCH state.

21	+	PAGING TYPE 1	UE paged using U-RNTI identity.
22	→	CELL UPDATE	IE "Cell Update Cause" shall be set to "Paging Response"
23	+	CELL UPDATE CONFIRM	"Transport channel information elements" IE "PRACH Info" is included.
24	→	PHYSICAL_TRANSPORT_CHANNEL RECONFIGURATION COMPLETE	This message shall be sent on the PRACH resource allocated in step 23.

PAGING TYPE 1 (Step 2, 9, 15 and 219)

Information Element	Value/remark
CHOICE Paging Originator	UTRAN Originator
- U-RNTI	
- SRNC Identity	'0000 0000 0001'
- S-RNTI	'0000 0000 0000 0000 0001'

PAGING TYPE 1 (Step 15 and 21)

Information Element	<u>Value/remark</u>
CHOICE Paging Originator	UTRAN Originator
U-RNTI	-
- SRNC Identity	<u>'0000 0000 0001'</u>
<u>- S-RNTI</u>	Check to see if set to value provided in step 11

CELL UPDATE (Step 3, 10, 16 and 2210)

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to '0000 0000 0000 0000 0001'
Integrity check info	Not checked
AM_RLC error indicator (for C-plane)	Check to see if set to 'FALSE'
AM_RLC error indicator (for U-plane)	Check to see if set to 'FALSE'
START List	Checked to see if the 'CN domain identity' and 'START'
	IEs are present for all CN domains supported by the UE
Cell Update Cause	Check to see if set to 'Paging Response'
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Measured results on RACH	Check to see if it is absent
Protocol error information	Check to see if it is absent

CELL UPDATE (Step 16 and 22)

Information Element	Value/remark
<u>U-RNTI</u>	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to value provided in step 11
START List	Checked to see if the 'CN domain identity' and 'START'
	IEs are present for all CN domains supported by the UE
Cell Update Cause	Check to see if set to 'Paging Response'

CELL UPDATE CONFIRM (Step 4-and 11)

Use the same message sub-type found in Annex A.

CELL UPDATE CONFIRM (Step 11)

Use the same message sub-type found in step 4, with the following exceptions:

Information Element	<u>Value/remark</u>
New U-RNTI	
 SRNC Identity 	<u>'0000 0000 0000 0001'</u>
- S-RNTI	An arbitrary 20-bits string which is different from original
	S-RNTI
New C-RNTI	An arbitrary 16-bits string which is different from original
	C-RNTI.

PHYSICAL CHANNEL RECONFIGURATION (Step 5, 18 and 24)

Only the message type for this message is checked.

MASTER INFORMATION BLOCK (Step 8)

Use the same message sub-type found in Clause 6.1 of TS 34.108, with the following exceptions:

Information Element	Value/remark
MIB Tag	2

SYSTEM INFORMATION BLOCK TYPE 5 and TYPE 6 (Step 8)

Use the same message sub-type found in Clause 6.1 of TS 34.108, with the following exceptions:

Value/remark
Common transport channels
1
296 bits
80 msec
No coding
No Present
1
16 bits

UTRAN MOBILITY INFORMATION (Step 6, 13 and 19)

Use the same message sub-type found in Annex A, with the following exceptions:

Information Element	Value/remark
New C-RNTI	Selects any arbitrary unused 16-bits string
DRX IndicatorRRC State Indicator	DRX with URA updatingURA_PCH

TRANSPORT CHANNEL RECONFIGURATION COMPLETE(Step 1224)

Only the message type for this message is checked.

CELL UPDATE CONFIRM (Step 17)

Information Element	Value/remark
New C-RNTI	Selects any arbitrary unused 16-bits string

Use the same message sub-type found in step 11, with the following exceptions:

Maximum allowed uplink TX power	3 dB below the follow value:
	Minimum of {33 dBm, maximum uplink power allowed
	under the UE power class }

CELL UPDATE CONFIRM (Step 23)

Use the same message sub-type found in Annex Astep 17, with the following exceptions:

Information Element	Value/remark
Added or Reconfigured UL TrCH information	
-Transport channel identity	1
-TFS	(This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set
-Dynamic Transport format information	Reference to TS34.108 clause 6.10 Parameter Set
-Number of Transport blocks	Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set
RLC size	Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set
-Semi-static Transport Format information	Reference to TS34.108 clause 6.10 Parameter Set
-Transmission time interval	Refer to the parameter set for TS 34.108
-Type of channel coding	Refer to the parameter set for TS 34.108
-Coding Rate	
-Rate matching attribute	Set to the scrambling code for cell 1's P-CPICH.
-CRC size	Not Present. Not Present.
PRACH Info (for RACH) - Available Signature - Available SF - Preamble scrambling code number - Puncturing Limit - Available Sub-Channel number Downlink DPCH info for one radio link - Primary CPICH Info - Primary Scrambling Code - PDSCH with SHO DCH Info - PDSCH code mapping - Secondary CCPCH Info - Primary CPICH usage for channel estimation - Secondary CPICH Info - Secondary Scrambling code - Channelization code - Secondary scrambling code - Shannelization code - Secondary Scrambling code - Shannelization code - Secondary Scrambling code - TTD indicator - Spreading factor - Code number - Pilot symbol existence - TFCI existence - Fixed or Flexible Position - Timing Offset	TRUE Not Present. Use default (Scrambling code of P-CPICH) 1 Not Present. Use default (Scrambling code of P-CPICH) FALSE 256 255 TRUE TRUE Flexible 0-chips

8.3.1.7.5 Test requirement

After step 2 the UE shall answer to the paging message then moves to CELL_FACH state and transmit a CELL UPDATE message. This message shall set the value "paging response" into IE "Cell update cause".

After step 4 the UE shall transmit PHYSICAL CHANNEL RECONFIGURTION COMPLETE message on the uplink DCCH and enters CELL_FACH state.

After step 9 the UE shall respond the paging by replying with a CELL UPDATE message. IE "Cell Update Cause" shall be set to "Paging Response" in this message.

After step 11 the UE shall transmit <u>UTRAN MOBILITY INFORMATION CONFIRMTRANSPORT CHANNEL RECONFIGURATION COMPLETE</u> message on the uplink DCCH.

After step 15 the UE shall answer the PAGING TYPE 1 message by sending a CELL UPDATE message. The IE "Cell Update Cause" shall have a value equals to "Paging Response".

After step 17 the UE shall transmit PHYSICAL CHANNEL RECONFIGURATION COMPLETE message on the uplink DCCH.

After step 21 the UE shall respond to the PAGING TYPE 1 message addressed to itself and initiate a cell update procedure. It shall transmit CELL UPDATE message with the IE "Cell Update Cause" set to "Paging Response" in this message.

After step 23 the UE shall send <u>TRANSPORT PHYSICAL</u>-CHANNEL RECONFIGURATION COMPLETE message on the uplink DCCH-using the PRACH channel indicated in CELL UPDATE CONFIRM message sent in step 23.

8.3.1.8 Cell Update: paging response in CELL_PCH

8.3.1.8.1 Definition

8.3.1.8.2 Conformance requirement

This procedure is to update UTRAN with the current cell when the UE receives a PAGING TYPE 1 message addressed to it while in CELL PCH state.

Reference

3GPP TS 25.331 clause 8.3.1

8.3.1.8.3 Test purpose

To confirm that the UE executes a cell update procedure when the UE receives a PAGING TYPE 1 message while in CELL_PCH state. To confirm that the UE sends an appropriate uplink message after receiving a CELL UPDATE CONFIRM message during cell updating procedure due to paging.

8.3.1.8.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CELL_PCH (state 6-1512) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the CELL_PCH state. The SS transmits a PAGING TYPE 1 message to the UE on the downlink PCCH which includes the connected mode identity of the UE and set value "UTRAN originator" into IE "paging originator". The UE shall respond to this message. Then the UE shall move to CELL_FACH state and transmits a CELL UPDATE message to the SS on the uplink CCCH which and set the value "Paging Response" into IE "Cell update cause". After the SS receives this message, it transmits the default CELL UPDATE CONFIRM message. The UE shall stay in CELL_FACH state and not transmit PHYSICAL CHANNEL RECONFIGURATION COMPLETE response message on the DCCH. Next SS modifies the contents of SYSTEM INFORMATION BLOCK TYPE 5 and 6 messages. See specific message contents for further details. SS then sends a UTRAN MOBILITY INFORMATION message on the downlink DCCH, with the IE "DRX IndicatorRRC State Indicator" set to "DRX with cell updatingCELL_PCH". The UE shall move to CELL_PCH state and acknowledges that it starts using the new C-RNTI identity provided by transmitting UTRAN MOBILITY INFORMATION CONFIRM message and moves to CELL_PCH state. SS pages the UE again using PAGING TYPE 1 message. The UE shall send CELL UPDATE message once more. SS replies with a

CELL UPDATE CONFIRM message which includes the IE "new C-RNTI" to the UE on the downlink DCCH. Then the UE shall transmit an UTRAN MOBILITY INFORMATION CONFIRM TRANSPORT CHANNEL RECONFIGURTION COMPLETE-message on the uplink DCCH and enters the CELL_FACH state. In the final sequence, SS transmits UTRAN MOBILITY INFORMATION message and assigns a new C-RNTI identity to the UE. The UE shall reply with UTRAN MOBILITY INFORMATION CONFIRM message and move to CELL_PCH state. SS then pages the UE once more using PAGING TYPE 1 message. The UE shall initial the cell update procedure by dispatching a CELL UPDATE message on the uplink CCCH. SS replies with a CELL UPDATE CONFIRM message which includes the IE-"Physical channel information elements PRACH Info". The UE shall respond with a PHYSICAL CHANNEL RECONFIGURATION COMPLETE, sent on the DCCH-supported by the PRACH indicated in CELL UPDATE CONFIRM message. Next, SS sends UTRAN MOBILITY INFORMATION message to UE with the "DRX Indicator RRC State Indicator" IE set to "DRX with cell updating CELL PCH". The UE shall respond by transmitting UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH and then move to CELL_PCH state. SS pages the UE again using connected mode identity at the paging occasions assigned to the UE. The UE shall answer to the page and sent CELL UPDATE message on the uplink CCCH. SS responds to the reception of this message by transmitting a CELL UPDATE CONFIRM message on the downlink DCCH. In this message, "Transport channel information elements" a set of new PRACH resources is assigned. The UE shall acknowledge the receipt of this message and send PHYSICAL TRANSPORT CHANNEL RECONFIGURATION COMPLETE on the uplink DCCH-supported by the new PRACH channel.

Expected sequence

Step	Direction UE SS	Message	Comment
1	02 00		The UE is brought to CELL_PCH state.
2	+	PAGING TYPE 1	The SS transmits a PAGING TYPE 1 message addressing the UE with its connected mode identity and set IE "paging originator" to "UTRAN originator".
3	→	CELL UPDATE	The UE shall move to CELL_FACH state and transmits this message with the IE "Cell update cause" set to "paging response".
4	+	CELL UPDATE CONFIRM	Use the default message from TS 34.108 Clause 8.
5	→	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	The UE shall move to CELL_FACH state and transmits this response message.
6			SS modifies the contents of MASTER INFORMATION BLOCK, SYSTEM INFORMATION BLOCK TYPE 5 and 6 messages.
7	+	UTRAN MOBILITY INFORMATION	Allocates a new C-RNTI and set IE "DRX IndicatorRRC State Indicator" to "DRX with cell updatingCELL PCH"
8	→	UTRAN MOBILITY INFORMATION CONFIRM	UE moves to CELL_PCH state
9	←	PAGING TYPE 1	SS pages the UE again
10	→	CELL UPDATE	IE "Cell update cause" shall be set to "paging response"
11	←	CELL UPDATE CONFIRM	Contains the IE "new C-RNTI"
12	→	UTRAN MOBILITY INFORMATION CONFIRMTRANSPORT CHANNEL RECONFIGURATION COMPLETE	The UE shall move to CELL_FACH state and transmit this message.
13	←	UTRAN MOBILITY INFORMATION	Allocates a new C-RNTI and set IE "DRX IndicatorRRC State Indicator" to "DRX with cell updatingCELL PCH".
14	\rightarrow	UTRAN MOBILITY INFORMATION CONFIRM	UE moves to CELL_PCH state.
15	+	PAGING TYPE 1	SS pages the UE again
16	→	CELL UPDATE	IE "Cell update cause" shall be set to "paging response".
17	←	CELL UPDATE CONFIRM	Including "Physical channel information elements" IE "PRACH Info" in this message.
18	→	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	To be sent on the new PRACH channel.
19	+	UTRAN MOBILITY INFORMATION	Allocates a new C-RNTI and set IE "DRX Indicator RRC State Indicator" to "DRX with
20	→	UTRAN MOBILITY INFORMATION CONFIRM	UE moves to CELL_PCH state.
21	+	PAGING TYPE 1	UE paged using U-RNTI identity.
22	→	CELL UPDATE	IE "Cell Update Cause" shall be set to "Paging Response"

23	+	CELL UPDATE CONFIRM	"Transport channel information elements" E "PRACH Info" is included.
24	→	PHYSICAL_TRANSPORT_CHANNEL RECONFIGURATION COMPLETE	This message shall be sent on the PRACH resource allocated in step 23.

PAGING TYPE 1

Information Element	Value/remark
CHOICE Paging Originator	UTRAN Originator
- U-RNTI	
- SRNC Identity	'0000 0000 0001'
- S-RNTI	'0000 0000 0000 0000 0001'

CELL UPDATE (Step 3, 10 and 16)

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to '0000 0000 0000 0000 0001'
Integrity check info	Not checked
AM_RLC error indicator (for C-plane)	Check to see if set to 'FALSE'
AM_RLC error indicator (for U-plane)	Check to see if set to 'FALSE'
START List	Checked to see if the 'CN domain identity' and 'START'
	IEs are present for all CN domains supported by the UE
Cell Update Cause	Check to see if set to 'Paging Response'
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Measured results on RACH	Check to see if it is absent
Protocol error information	Check to see if it is absent

CELL UPDATE CONFIRM (Step 4)

Use the same message sub-type found Annex A.

CELL UPDATE CONFIRM (Step 11)

Use the same message sub-type found in step 4, with the following exceptions:

Information Element	<u>Value/remark</u>
New C-RNTI	An arbitrary 16-bits string which is different from original
	C-RNTI.

PHYSICAL CHANNEL RECONFIGURATION COMPLETE (Step 5 and 18)

Only the message type for this message is checked.

MASTER INFORMATION BLOCK (Step 6)

Use the same message sub-type found in Clause 6.1 of TS 34.108, with the following exceptions:

Information Element	Value/remark
MIB Tag	2

SYSTEM INFORMATION BLOCK TYPE 5 and TYPE 6 (Step 6)

Use the same message sub-type found in Clause 6.1 of TS34.108, with the following exceptions:

Information Element	Value/remark
PRACH system information	
- RACH TFS	
- CHOICE Transport channel type	Common transport channels
- Dynamic Transport Format Information	
- Number of Transport blocks	1
- RLC Size	296 bits
- Semi-static Transport Format	
- Transmission time interval	80 msec
- Type of channel coding	No coding
- Coding Rate	No Present
- Rate matching attribute	1
- CRC Size	16 bits

UTRAN MOBILITY INFORMATION (Step 7,13 and 19)

Use the same message sub-type found in Annex A, with the following exceptions:

Information Element	Value/remark
New C-RNTI	Selects any arbitrary unused 16-bits string
DRX IndicatorRRC State Indicator	DRX with cell updatingCELL_PCH

TRANSPORT CHANNEL RECONFIGURATION COMPLETE (Step 1224)

Only the message type for this message is checked.

CELL UPDATE CONFIRM (Step 17)

Use the same message sub-type found in Annex Astep 11, with the following exceptions:

Information Element	Value/remark
PRACH Info (for RACH)	
- Available Signature	'000000000000100'B:Signature 2
Available SF	Refer to the parameter set for TS 34.108
- Preamble scrambling code number	θ .
- Puncturing Limit	Refer to the parameter set for TS 34.108
Available Sub-Channel number	'00000000100'B:SubChNumber 2
Downlink DPCH info for one radio link	
Primary CPICH Info	
	Set to the scrambling code for cell 1's P-CPICH.
- PDSCH with SHO DCH Info	Not Present.
PDSCH code mapping	Not Present.
- Secondary CCPCH Info	
- Primary CPICH usage for channel estimation	TRUE
Secondary CPICH Info	
- Secondary scrambling code	Not Present. Use default (Scrambling code of P-CPICH)
	4
	Not Present. Use default (Scrambling code of P-CPICH)
	FALSE
	256
	255
— - Pilot symbol existence	TRUE
- TFCI existence	TRUE
- Fixed or Flexible Position	Flexible
	0 chips
Maximum allowed uplink TX power	3 dB below the follow value:
	Minimum of {33 dBm, maximum uplink power allowed
	under the UE power class }

CELL UPDATE CONFIRM (Step 23)

Use the same message sub-type found in Annex Astep 17, with the following exceptions:

Information Element	Value/remark
Added or Reconfigured UL TrCH information	
Transport shapped identity	1 1
-Transport channel identity	(This IC is reported for TCI number)
-TFS	(This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
-Dynamic Transport format information	Teresones to 100 in 100 states on 101 and motor con
Number of Transport blocks	Reference to TS34.108 clause 6.10 Parameter Set
-Number of Hansport blocks	Reference to TS34.108 clause 6.10 Parameter Set
-RLC size	Reference to TS34.108 clause 6.10 Parameter Set
0 1 1 7 1 1 1 1 1 1 1 1	Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set
-Semi-static Transport Format information	'0000000000000010'B:Signature 1
-Transmission time interval	Refer to the parameter set for TS 34.108
	0
-Type of channel coding	Refer to the parameter set for TS 34.108
Coding Rate	'00000000010'B:SubChNumber 1
-Coung Kate	
-Rate matching attribute	Set to the scrambling code for cell 1's P-CPICH.
	Not Present.
-CRC size	Not Present.
PRACH Info (for RACH)	
- Available Signature	TRUE
- Available SF	Not Present Lies default (Carambling ands of D. CDICLI)
Preamble scrambling code number	Not Present. Use default (Scrambling code of P-CPICH) 1
Puncturing Limit	Not Present. Use default (Scrambling code of P-CPICH)
Available Sub-Channel number Downlink DPCH info for one radio link	FALSE
- Primary CPICH Info	256
- Primary Scrambling Code	255
- PDSCH with SHO DCH Info	TRUE
PDSCH code mapping	TRUE Elexible
Secondary CCPCH Info	0 chips
- Primary CPICH usage for channel estimation	o ompo
- Secondary scrambling code - Channelization code	
- Secondary scrambling code	
- STTD indicator	
— Spreading factor	
- Pilot symbol existence	
iig Onoot	

8.3.1.8.5 Test requirement

After step 2 the UE shall answer to the paging message, moves to CELL_FACH state, and then transmits a CELL UPDATE message setting "paging response" into IE "Cell update cause".

After step 4 the UE shall transmit a PHYSICAL CHANNEL RECONFIGURATION COMPLETE message on the uplink DCCH and then enter CELL_FACH state.

After step 9 the UE shall respond to the paging again by sending CELL UPDATE message, with the IE "Cell update cause" set to "Paging response".

After step 11 the UE shall transmit <u>UTRAN MOBILITY INFORMATION CONFIRMTRANSPORT CHANNEL RECONFIGURATION COMPLETE</u> message to end the cell updating procedure.

After step 15 the UE shall answers the paging message by sending a CELL UPDATE message, with the IE "Cell update cause" set to "Paging response".

After step 17 the UE shall transmit PHYSICAL CHANNEL RECONFIGURATION COMPLETE message on the uplink DCCH-carried by the PRACH indicated in the CELL UPDATE CONFIRM message sent in step 17.

After step 21 the UE shall respond to the paging and initiate a cell update procedure. It shall transmit CELL UPDATE message with the IE "Cell Update Cause" set to "Paging Response" in this message.

After step 23 the UE shall send <u>PHYSICAL_TRANSPORT</u> CHANNEL RECONFIGURATION COMPLETE message on the uplink DCCH-using the PRACH channel indicated in CELL UPDATE CONFIRM message sent in step 23.

8.3.1.9 Cell Update: re-entering of service area after T305 expiry and being out of service area

8.3.1.9.1 Definition

8.3.1.9.2 Conformance requirement

When a UE detects that it's out of service area after experiencing a T305 timer expiry, it shall try to search for a suitable cell to camp on. At the same time, it shall start timer T307. If the UE subsequently re-enters the service area of a cell before T307 expires, it shall perform a cell update procedure.

Reference

3GPP TS 25.331 clause 8.3.1

8.3.1.9.3 Test purpose

To confirm that the UE performs a cell search after experiencing an "out of service area" condition following the expiry of timer T305. To confirm that the UE initiates cell updating procedure if it manages to re-enter the service area.

8.3.1.9.4 Method of test

Initial Condition

System Simulator: $\frac{2 \text{ cells}}{2 \text{ cell 1}}$ is active with the CPICH Ec/No set to $\frac{80 \text{dB}}{2 \text{ is inactive.}}$ and the $\frac{Q_{qualmin}}{2 \text{ cell.}}$ value is at $\frac{90 \text{dBm.}}{2 \text{ cell.}}$

UE: CS-CELL_FACH_Initial (state 6-2) or PS-CELL_FACH_Initial (state 6-4) in cell 1- as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE

Test Procedure

The UE is in the CELL_FACH state. SS decreases the transmission power of cell 1 until the cell selection parameter S<0 (decrease transmission power of cell 1's CPICH by 15 dBm). Following the expiry of periodic cell updating timer T305 according to the system information, the UE shall detect that it is out of service area. Within the time interval equivalent to T307 timer value, the SS restores the transmission power of cell 1. The UE shall find that it is back in service area, and transmits a CELL UPDATE message to the SS on the uplink CCCH. In this message, the IE "Cell update cause" shall be set to "re-entered service area". After the SS receives this message, it transmits a CELL UPDATE CONFIRM message with the IE "DRX Indicator RRC State Indicator" set "No DRXCELL PCH" on the downlink DCCH. The UE shall enter CELL_FACH-PCH state. SS decreases the transmission power of cell 1 until the cell selection parameter S<0. Following the expiry of periodic cell updating timer T305 according to the system information, the UE shall detect that it is out of service area. Within the time interval equivalent to T307 timer value, the SS restores the transmission power of cell 1. The UE shall find that it is back in service area, move to CELL_FACH and transmits a CELL UPDATE message to the SS on the uplink CCCH. In this message, the IE "Cell update cause" shall be set to "re-entered service area". After the SS receives this message, it transmits a CELL UPDATE CONFIRM message on the downlink DCCH.Next, cell 1 is switched off. SS waits until T305 timer has expired and then turns on cell 2. The transmission level for cell 2 is such that the CPICH Ec/No value is estimated at 70dB. Qaualmin of cell 2 is

identical to the value used previously by cell 1. The UE shall discover that cell 1 is no longer suitable for camping and initiate a cell search. It shall detect the presence of cell 2 and reselects to this cell. Following this, it shall transmit CELL UPDATE message with the cause set to "Cell Reselection". SS ends this test by sending a CELL UPDATE CONFIRM message.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1				The UE is in the CELL_FACH state of cell 1.
2				SS decreases the
				transmission power of cell 1 so
				that its S value falls below 0.
3				The UE shall detect a "out of
3				service" condition upon expiry
				of timer T305 and it shall
				search for other cells to camp
				on(T307 timer starts)
4				SS restores cell 1's original
·				power level before T307 timer
				expires.
5		\rightarrow	CELL UPDATE	The value "re-entered service
		-		area" should be found in IE
				"Cell update cause" in this
				message
6		-	CELL UPDATE CONFIRM	"DRX IndicatorRRC State
				Indicator" is set to "No
				DRXCELL_PCH"
7				SS decreases the
				transmission power of cell 1 so
				that its S value falls below
				Oswitches off cell 1 and wait
				until T305 has expired.
8				SS restores cell 1's original
				power level before T307 timer
				expires.turns on cell 2 and set
				the transmission strength of
				such that CPICH Ec/No is at -
		_		70dB.
9	_	<u>→</u>	CELL UPDATE	UE shall detect the presence
				of cell 2 and re-select to
				#move to CELL FACH. It shall
				transmit this message with
				cause set to "Cell
				Reselectionre-entered service
10		-	CELL UPDATE CONFIRM	area"
10	_		GELL OFDATE CONFIRM	

Specific Message Contents

CELL UPDATE (Step 5 and 9)

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to '0000 0000 0000 0000 1111'
Integrity check info	Not checked
AM_RLC error indicator (for C-plane)	Check to see if set to 'FALSE'
AM_RLC error indicator (for U-plane)	Check to see if set to 'FALSE'
START List	Checked to see if the 'CN domain identity' and 'START'
	IEs are present for all CN domains supported by the UE
Cell Update Cause	Check to see if set to 're-entered service area'
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Measured results on RACH	Check to see if it is absent
Protocol error information	Check to see if it is absent

CELL UPDATE CONFIRM (Step 6 and 10)

Use the same message sub-type found in Annex A, with the following exception.

Information Element	<u>Value/remark</u>
RRC State Indicator	CELL PCH

CELL UPDATE (Step 9)

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to '0000 0000 0000 0000 1111'
Integrity check info	Not checked
AM_RLC error indicator (for C-plane)	Check to see if set to 'FALSE'
AM_RLC error indicator (for U-plane)	Check to see if set to 'FALSE'
START List	Checked to see if the 'CN domain identity' and 'START'
	IEs are present for all CN domains supported by the UE
Cell Update Cause	Check to see if set to 'Cell Reselection'
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Measured results on RACH	Check to see if it is absent
Protocol error information	Check to see if it is absent

8.3.1.9.5 Test requirement

After step 4 the UE shall transmit a CELL UPDATE message in which the IE "Cell update cause" is set to the value "re-entered service area".

After step 8 the UE shall reselect to cell 2move to CELL FACH and then transmit a CELL UPDATE message, with the IE "Cell Update Cause" set to "Cell Reselectionre-entered service area".

8.3.1.10 Cell Update: expiry of T307 after T305 expiry and being out of service area

8.3.1.10.1 Definition

8.3.1.10.2 Conformance requirement

This procedure is required to cater for the case of a failure to update UTRAN with the current cell, after the expiry of T307. In this case, the UE shall return to idle mode and perform cell reselection if possible.

Reference

3GPP TS 25.331 clause 8.3.1

8.3.1.10.3 Test purpose

To confirm that the UE moves to idle mode after the expiry of T307, indicating that it is out of service area when attempting to perform a periodic cell updating procedure.

8.3.1.10.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CELL_PCH (state 6-1512) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in CELL_PCH state at the start of the test. Before the expiry of periodic cell updating timer T305, SS starts to decrease the downlink transmission power such that the UE discovers that the cell is no longer suitable for camping and this results in a "out of service area" condition. The SS continues to listen to the uplink channel to detect possible attempts to perform a cell updating procedure. The UE shall not send CELL UPDATE message on the uplink DCCH, instead it triggers timer T307. After the expiry of timer T307 the UE shall enter idle state. This is confirmed by the SS, when it sends a PAGING TYPE 1 message to the UE using its U-RNTI identity, and the UE does not respond to the page. SS then attempts to page for the UE again, this time using PAGING TYPE 2 message sent on downlink DCCH. Likewise, the UE shall not respond to this page.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1				The UE is brought to
				CELL_PCH state.
2				SS starts to decrease the
				transmission power until the
				cell is no longer suitable for
				camping. The UE shall detect
				that it is out of service area
				and refrains from transmitting
				CELL UPDATE message due
				to periodic cell updating.
3				The UE detects the expiry of timer T305 and it searches for
				other cells to camp on. After the expiry of timer T307, the
				UE shall enter idle mode.
4	+	<u> </u>	PAGING TYPE 1	SS pages the UE at its
7	`	•	17.0	assigned paging occasion
				using the allocated U-RNTI
				value. The UE shall not
				respond to this page as it has
				already entered the idle mode.
5	+	_	PAGING TYPE 2	SS pages the UE on the
				downlink DCCH. The UE shall
				not respond to this page.

Specific Message Contents

PAGING TYPE 1 (Step 4)

Information Element	Value/remark
Page record list	
- Paging record	
- CHOICE Paging originator	UTRAN Originator
- U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to '0000 0000 0000 0000 1111'
BCCH Modification info	Not Present.

PAGING TYPE 2 (Step 5)

Information Element	Value/remark
Integrity check info	Not Present
Paging cause	Set to a cause corresponding to one radio access
	bearer services supported by the UE.
CN domain identity	CS-Domain
Paging Record Type Identifier	IMSI

8.3.1.10.5 Test requirement

After step 4 the UE shall remain in the idle mode and not respond to the paging message sent on PCCH.

After step 5 the UE shall remain in the idle mode and not respond to the paging message addressed to it on the DCCH.

8.3.1.11 Cell Update: Success after T302 time-out

8.3.1.11.1 Definition

8.3.1.11.2 Conformance requirement

The UE transmits a CELL UPDATE message to the UTRAN when it needs to update the UTRAN with the current cell of the UE. When the UE does not receive a CELL UPDATE CONFIRM message upon expiry of timer T302, the UE transmits a CELL UPDATE message repeatedly until its internal counter V302 counter is greater than N302.

Reference

3GPP TS 25.331 clause 8.3.1

8.3.1.11.3 Test purpose

To confirm that the UE repeats the transmission of CELL UPDATE message upon the expiry of timer T302, after failing to receive any response from the SS during T302 timer period.

8.3.1.11.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CS-CELL_FACH_Initial (state 6-2) or PS-CELL_FACH_Initial (state 6-4) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE

Test Procedure

At the start of the test, the UE is brought to CELL_FACH state. When the UE detects the expiry of periodic cell updating timer T305 according to the system information, the UE transmits a CELL UPDATE message to the SS on the uplink CCCH. The IE "Cell update cause" in this message shall be set to "periodical cell update". SS ignores this message, and the UE shall then re-transmit a CELL UPDATE message after the expiry of timer T302. When the SS has received (N302+1) such messages, it transmits a CELL UPDATE CONFIRM message with new values for "C-RNTI" to the UE. Finally, the UE shall transmit an UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH.

Expected sequence

Step	Direction		tep Direction	Message	Comment
-	UE	SS			
1				The UE starts from CELL_FACH state. SS initializes its internal counter K to 0 and wait until the expiry of T302 timer.	
2		→	CELL UPDATE	The value "periodical cell update" shall be set in IE "Cell update cause" after the expiry of timer T305.	
3				If K is equal to N302+2 then proceeds to step 5.	
4				SS increments counter K, transmits no response to the UE and waits for an additional period equals to the value of timer T302. The next step is step 2.	
5		+	CELL UPDATE CONFIRM	The message includes IEs "new C-RNTI". The IE "DRX IndicatorRRC State Indicator" is set to "Ne DRXCELL FACH".	
6		\rightarrow	UTRAN MOBILITY INFORMATION CONFIRM		

Specific Message Contents

CELL UPDATE (Step 2)

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to '0000 0000 0000 0000 0001'
Integrity check info	Not checked
AM_RLC error indicator (for C-plane)	Check to see if set to 'FALSE'
AM_RLC error indicator (for U-plane)	Check to see if set to 'FALSE'
START List	Checked to see if the 'CN domain identity' and 'START'
	IEs are present for all CN domains supported by the UE
Cell Update Cause	Check to see if set to 'Periodic cell updating'
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Measured results on RACH	Check to see if it is absent
Protocol error information	Check to see if it is absent

CELL UPDATE CONFIRM (Step 5)

Use the same message sub-type found in Annex A, with the following exception:

Information Element	Value/remark
New C-RNTI	Set to an arbitrary string different from '0000 0000 0000
	0001'

8.3.1.11.5 Test requirement

After step 1 the UE shall detect the expiry of timer T305 then transmit a CELL UPDATE message on the uplink CCCH, setting "periodical cell update" into IE "Cell update cause".

After step 2 the UE shall re-transmits a CELL UPDATE message after the expiry of timer T302. A total of (N302+42) transmissions shall be detected in SS.

After step 5 the UE shall transmit an UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH and stays at CELL_FACH state.

8.3.1.12 Cell Update: Failure (After Maximum Re-transmissions)

8.3.1.12.1 Definition

8.3.1.12.2 Conformance requirement

The UE transmits a CELL UPDATE message to the UTRAN when it needs to update UTRAN with information on the current cell of the UE. If the UE fails to receive a CELL UPDATE CONFIRM message, it re-transmits a CELL UPDATE message repeatedly upon the expiry of timer T302 until the value of V302 counter is greater than N302. If V302 is greater than N302, the UE stop the re-transmission and enters idle state.

Reference

3GPP TS 25.331 clause 8.3.1

8.3.1.12.3 Test purpose

To confirm that the UE repeats the cell update procedure at the expiry of timer T302 and moves to idle state when its internal counter V302 is greater than N302.

8.3.1.12.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CS-CELL_FACH_Initial (state 6-2) or PS-CELL_FACH_Initial (state 6-4) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE

Test Procedure

The UE is initially in CELL_FACH state. When the UE detects the expiry of periodic cell updating timer T305, the UE transmits a CELL UPDATE message to the SS on the uplink CCCH to perform a periodic cell updating procedure. The SS ignores this message, and the UE shall attempt to re-transmit a CELL UPDATE message up to a maximum of (N302+1) times after the expiry of timer T302. After (N302+1) attempts of retransmission, the UE shall return to idle state.

Expected sequence

Step	Direction		Direction Message	Comment
-	UE	SS	_	
1				The UE is in the CELL_FACH state. SS sets its internal counter K=1-0 and waits for a period equals to timer value T302. If CELL UPDATE message is received upon timer expiry, proceeds to step 2. Else goes to step 4.
2		\rightarrow	CELL UPDATE	The value "periodical cell update" should be set in IE "Cell update cause" and this message should be sent for each expiry of timer T302.
3				SS transmits no response to the UE and increments counter K. SS waits for an additional period equals to T302 timer.
4				SS waits for an additional period equals to T302 timer. If CELL UPDATE message is received, proceed to step 2. Otherwise, terminates the test. If K is not equal to N302+2, the test should be considered as a failure.

Specific Message Contents

CELL UPDATE (Step 2)

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to '0000 0000 0000 0000 0001'
Integrity check info	Not checked
AM_RLC error indicator (for C-plane)	Check to see if set to 'FALSE'
AM_RLC error indicator (for U-plane)	Check to see if set to 'FALSE'
START List	Checked to see if the 'CN domain identity' and 'START'
	IEs are present for all CN domains supported by the UE
Cell Update Cause	Check to see if set to 'Periodic cell updating'
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Measured results on RACH	Check to see if it is absent
Protocol error information	Check to see if it is absent

8.3.1.12.5 Test requirement

After step 1 the UE shall transmit a CELL UPDATE message on the uplink CCCH and set value "periodical cell update" into IE "Cell update cause".

After step 4 the counter K in SS shall be equal to N302+2.

8.3.1.13 Cell Update: Reception of Invalid CELL UPDATE CONFIRM Message

8.3.1.13.1 Definition

8.3.1.13.2 Conformance Requirement

If the UE encounters an invalid CELL UPDATE CONFIRM message while executing a cell update procedure, it shall check the current value of its internal counter V302. If V302 is not greater than N302, the UE shall set contexts pertaining to protocol error, re-transmits CELL UPDATE message on uplink CCCH, restart T302 timer and increments V302. It shall use the same "Cell Update Cause" as before receiving the invalid downlink message. On the other hand, if V302 is greater than N302, the UE shall abandon cell update procedure and enters idle mode.

8.3.1.13.3 Test Purpose

To confirm that the UE retransmits CELL UPDATE message when it receives an erroneous CELL UPDATE CONFIRM message, if the number of retransmissions is not the maximum allowed value. To confirm that the UE returns to idle mode after sending maximum allowed number of CELL UPDATE messages without receiving a valid CELL UPDATE CONFIRM message.

8.3.1.13.4 Method of Test

Initial Condition

System Simulator: 1 cell

UE: CELL_PCH (state 6-1512) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is brought to CELL_PCH state at the beginning of the test. SS pages the UE by sending PAGING TYPE 1 message using the U-RNTI identity assigned during RRC connection establishment procedure. The UE shall start to transmit CELL UPDATE message on the uplink CCCH. Upon receiving such a message, the SS replies with a CELL UPDATE CONFIRM message containing a protocol error in IE "DRX IndicatorRRC transaction identifier". The UE shall detect the protocol error and re-transmit CELL UPDATE message up to a maximum of N302+1 times. The time interval between the transmissions shall be approximately equal to T302. SS verifies that it receives a total of (N302+12) identical CELL UPDATE messages. The UE shall return to idle mode after all uplink transmissions have finished. SS verifies this by paging the UE using the U-RNTI identity. The UE shall not respond to this page.

Expected Sequence

Step	Direction	Message	Comment
	UE SS		
1	+	PAGING TYPE 1	The UE is in the CELL_PCH state. SS sets its internal counter K=0-and waits for a period equals to timer value T302. SS pages for the UE using the allocated connected mode identity (U-RNTI)If CELL UPDATE message is received upon timer expiry, proceeds to step 2. Else goes to step 5.
2	→	CELL UPDATE	If CELL UPDATE message is received, check that the The value "paging response" should be is set in IE "Cell update cause". Else goes to step 6.
3	+	CELL UPDATE CONFIRM	SS transmits an invalid message. SS increments K.
4	→	CELL UPDATE	SS waits for T302 timer to expire. The UE shall send CELL UPDATE message.
5			If a CELL UPDATE message is received in step 4, SS increments K, restart T302 timer _and returns to execute step 43. Else, SS proceeds to step 6.
6			SS verifies that $K = (N302+42)$ and proceeds to the next step. Else, the test fails.
7	+	PAGING TYPE 1	SS pages the UE.
8			UE shall not respond.

Specific Message Content

CELL UPDATE (Step 2)

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to '0000 0000 0000 0000 0001'
Integrity check info	Not checked
AM_RLC error indicator (for C-plane)	Check to see if set to 'FALSE'
AM_RLC error indicator (for U-plane)	Check to see if set to 'FALSE'
START List	Checked to see if the 'CN domain identity' and 'START'
	IEs are present for all CN domains supported by the UE
Cell Update Cause	Check to see if set to 'Paging Response'
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Measured results on RACH	Check to see if it is absent
Protocol error information	Check to see if it is absent

CELL UPDATE CONFIRM (Step 3)

Use the same message sub-type found in Annex A, with the following exception:

Information Element	Value/remark
DRX Indicator RRC transaction identifier	Use one of the spareInvalid values.

CELL UPDATE (Step 4)

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to '0000 0000 0000 0000 0001'
Integrity check info	Not checked
AM_RLC error indicator (for C-plane)	Check to see if set to 'FALSE'
AM_RLC error indicator (for U-plane)	Check to see if set to 'FALSE'
START List	Checked to see if the 'CN domain identity' and 'START'
	IEs are present for all CN domains supported by the UE
Cell Update Cause	Check to see if set to 'Paging Response'
Protocol error indicator	Check to see if it is set to 'TRUE
Measured results on RACH	Check to see if it is absent
Failure cause	Check to see if it is set to 'protocol error'
Protocol error information	Check to see if it is set to 'Information element value not
	comprehended'

PAGING TYPE 1 (Step 1 and 7)

Information Element	Value/remark
Page record list	
- Paging record	
 CHOICE Paging originator 	UTRAN Originator
- U-RNTI	
- SRNC Identity	'0000 0000 0001'
- S-RNTI	'0000 0000 0000 0001'
BCCH Modification info	Not Present.

8.3.1.13.5 Test Requirement

After step 3 the UE shall continue to transmit CELL UPDATE message for N302+1 times.

At step 6 the counter K should be equal to (N302+12).

After step 7 the UE shall return to idle mode and not respond the PAGING TYPE 1 message sent by the SS.

8.3.1.14 Cell Update: Radio Bearer Control for Transition from CELL_DCH to CELL_FACH

8.3.1.14.1 Definition

8.3.1.14.2 Conformance Requirement

During a transition from CELL_DCH state to CELL_FACH state arising from the execution of radio bearer control procedure, the UE might be requested to re select to an unknown cell. The UE shall select a cell and perform cell updating procedure. In order to distinguish the 2 cases of cell updating: (i) due to UE mobility and (ii) due to radio bearer control procedure, the update cause in CELL UPDATE message shall be different for these 2 cases. When the UTRAN receives this message, it is then able to decide whether to initiate RNC reallocation and the establishment of new configuration in the target RNC.

8.3.1.14.3 Test Purpose

To confirm that the UE perform a cell update procedure after being instructed to move from CELL_DCH to CELL_FACH state as a result of radio bearer control procedure. To confirm that the UE indicates cause "RB Control" when transmitting a CELL UPDATE message to the selected cell. To confirm that the UE transmit a compatible response to conclude the radio bearer control procedure.

8.3.1.14.4 Method of Test

Initial Condition

System Simulator: 2 cells—both cell 1 and cell 2 are active and suitable for camping, but with cell 1 having a stronger transmission power.

UE: CS CELL_DCH_Initial (state 6 1) or PS CELL_DCH_Initial (state 6 3) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE

Test Procedure

The UE is brought to CELL_DCH state in cell 1, after the UE has successfully performed the RRC connection establishment procedure and was allocated dedicated physical resources. Next SS sends RADIO BEARER SETUP message to the UE on downlink DCCH. In this message, a DTCH channel is assigned to the UE. The UE shall reconfigure its channel resources and then return a RADIO BEARER SETUP COMPLETE message. Following this sequence, the SS sends RADIO BEARER RELEASE message to request that all radio bearers carried on dedicated physical channel to be released. However, this message does not contain information about the target cell to select when UE transits to CELL_FACH state. The UE shall perform cell reselection and it shall detect the presence of cell 2. The UE shall send a CELL UPDATE message with cause set to "RB Control" on the uplink CCCH carried by PRACH physical channel, specified in cell 2's system information message. SS replies with CELL UPDATE CONFIRM message, specifying the IE "PRACH Info" and "Secondary CCPCH Info". To complete this procedure, the UE shall transmit RADIO BEARER RELEASE COMPLETE message on the DCCH, which is carried on the PRACH physical channel specified in system information messages broadcasted in cell 2.

Expected Test Sequence

Step	Dire	ction	Message	Comment
-	UE	SS		
4				The UE is in the CELL_DCH
				state after completing a
				successful RRC connection
				establishment procedure.
2	_	(RADIO BEARER SETUP	Establishes DTCH logical
				channel.
3		\rightarrow	RADIO BEARER SETUP COMPLETE	
4	_		RADIO BEARER RELEASE	Information on target cell is not
				specified.
5	_	\rightarrow	CELL UPDATE	UE shall send this message
				on the uplink CCCH of cell 2.
				IE "Cell update cause" shall be
				set to "RB Control".
6		(CELL UPDATE CONFIRM	les "PRACH Info" and IE
				"Secondary CCPCH Info" are
				included in this message. IE
				"DRX Indicator" set to "No
				DRX".
7	_	\rightarrow	RADIO BEARER RELEASE COMPLETE	Sent on the PRACH given in
				system information messages.

Specific Message Contents

RADIO BEARER SETUP (Step 2)

Use the same message sub-type entitled "Packet to CELL_DCH from CELL_DCH in PS" found in Annex A, with the following exceptions:

Information Element	Value/remark
RAB information to setup list	
- RAB information for setup	
- RAB Info	
- RAB Identity	4
	PS
- Re-establishment timer	
- T315	1800 seconds
- RB Info to setup list	
- RB Info to setup	
- RB Identity	5
- PDCP Info	Not Present.
	Use the same RLC configuration as in TS 34.108.
- RB Mapping Info	Use the same RLC multiplexing scheme as in TS 34.108.
RB to be affected list	Not Present.
UL Transport Channel information common to all	
transport channels	
TFC Subset	Not Present – use default value, all TFCs are allowed.
	Not Present
	Not Present
- PRACH TECS	EDD
- UL DCH TFCS	
- CHOICE TFCI Signalling	Normal
- TFCI Field 1 Information	HOIMAT
- CHOICE TFCS representation	Addition
- TFCS addition information	Addition
- CHOICE CTFC Size	Number of hits used must be enough to sever all
- CHUICE CIPC SIZE	Number of bits used must be enough to cover all combinations of CTFC from clause 6.10 of TS 34.108.
- CTFC Information	Refer to TS 34.108 – This IE is repeated for the maximum number TFC to be added as specified in TS 34.108 for a selected transport channel
- Power offset information	
- CHOICE Gain Factor	Signalled Gain Factors
- Gain factor B _c	θ
- Gain factor β _d	0
- Reference TFC ID	Not Present
- Power offset Pp-m	Δ
Deleted TrCH information list (uplink)	Not Present
— - Deleted UL TrCH information	THOU TOOGH
Added or Reconfigured TrCH information list	
- Added or Reconfigured UL TrCH information	
<u> </u>	2
Transport channel identity	-
CHOICE Transport shapped type	Dedicated transport channels
	Больаю напорон онаннов
- Dynamic Transport Format Information	1
- Number of Transport blocks	1 128 hito
	128 bits
- Semi-static Transport Format	Use the same settings as in TS 34.108
- Transmission time interval	
- Type of channel coding	
- Coding Rate	
- Rate matching attribute	
	
DL Transport Channel information common to all	Not Present
transport channels	
Deleted TrCH information list (downlink)	Not Present
Added or Reconfigured TrCH information list	Not Present
Added of Accordinguica Front Information list	Not Present – use the existing frequency information
Frequency info	
•	Not Present – use the allowable UL TX power as
Frequency info	
Frequency info Maximum allow UL TX power	Not Present – use the allowable UL TX power as
Frequency info	Not Present – use the allowable UL TX power as specified by the UE's RF power class.

RADIO BEARER SETUP COMPLETE (Step 3)

Only the message type is checked for this message.

RADIO BEARER RELEASE (Step 4)

Use the same message sub-type entitled "Packet to CELL_FACH from CELL_DCH in PS" found in Annex A, with the following exceptions:

Information Element	Value/remark
RAB information to release list	
	5
RB information to be affected list	
- RB information to be affected	CCCH for RRC (TM)
	θ
 Information for each multiplexing option 	
	4
- RLC logical channel mapping indicator	Not Present
 Uplink transport channel type 	RACH
- UL Transport channel identity	Not Present
	5
- MAC logical channel priority	4
- Logical channel max loss	Not Present
- Number of RLC logical channels	4
 Downlink transport channel type 	FACH
- DL Transport channel identity	Not Present
	€
- RB information to be affected	DCCH for RRC (UM)
	4
- RB mapping info	
 Information for each multiplexing option 	
- Number of RLC logical channels	4
 Uplink transport channel type 	RACH
- Logical channel identity	4
- MAC logical channel priority	2
- Logical channel max loss	Not Present
- Number of RLC logical channels	4
 Downlink transport channel type 	FACH
- Logical channel identity	4
- RB information to be affected	DCCH for RRC (AM)
- RB identity	2
- RB mapping info	
 Information for each multiplexing option 	
- Number of RLC logical channels	4
- Uplink transport channel type	RACH
- Logical channel identity	2
- MAC logical channel priority	3
- Logical channel max loss	Not Present
- Number of RLC logical channels	4 5401
- Downlink transport channel type	FACH
- Logical channel identity	2
	DCCH for NAS_DT-AM High Priority
- RB identity	3
- RB mapping info	
- Information for each multiplexing option	
- Number of RLC logical channels	4
- Uplink transport channel type	RACH
- Logical channel identity	3
	4 Not Brooms
- Logical channel max loss	Not Present
- Number of RLC logical channels	1 FACH
- Downlink transport channel type	FACH
- Logical channel identity - RB information to be affected	3 DCCH for NAS DT AM Low Priority
	DCCH for NAS_DT-AM Low Priority
- RB identity	4
- RB mapping info - Information for each multiplexing option	
- Number of RLC logical channels	4
- Uplink transport channel type	RACH
- Logical channel identity	4
- MAC logical channel priority	5
- Logical channel max loss	Not Present
Logical orialino max 1000	1
- Number of RI C logical channels	
- Number of RLC logical channels	
- Number of RLC logical channels - Downlink transport channel type - Logical channel identity	FACH 4

	6
- RB mapping info	o de la companya de l
- Information for each multiplexing option	
- Number of RLC logical channels	4
- Downlink transport channel type	FACH
- DL Transport channel identity	Not Present
- Logical channel identity	5
- RB information to be affected	PCCH for RRC (TM)
- RB identity	Z
- RB mapping info	•
- Information for each multiplexing option	
- Number of RLC logical channels	4
- Downlink transport channel type	PCH
- DL Transport channel identity	Not Present
- Logical channel identity	4
UL Transport Channel information common to all	
transport channels	
- TFC Subset	Not Present – use default value, all TFCs are allowed.
	FDD
UL DCH TFCS	Use the same TFCS as in IE "RACH TFCS" found in
	system information block type 5 messages of cell 2.
Deleted TrCH information list (uplink)	Not Present.
Added or Reconfigured TrCH information list	
- Added or Reconfigured UL TrCH information	
- Transport channel identity	4
TFS	Use the same TFS as in IE "RACH TFS" found in
	system information block type 5 messages of cell 2.
DL Transport Channel information common to all	
transport channels	
	Use the same TFCS as in IE "TFCS" (for FACH) in IE
	"Secondary CCPCH system information" found in
	system information block type 5 messages of cell 2.
— - DL DCH TFCS	Not Present.
Deleted TrCH information list (downlink)	Not Present
Added or Reconfigured TrCH information list	Not Present.
Frequency info	Not Present – use the existing frequency information
Maximum allow UL TX power	Not Present – use the allowable UL TX power as
	specified by the UE's RF power class.
CHOICE Channel requirement	Not Present.
Downlink information common for all radio links	Not Present.
Downlink information per radio link list	Not Present.

CELL UPDATE (Step 5)

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to '0000 0000 0000 0000 0001'
Integrity check info	Not checked
AM_RLC error indicator (for C-plane)	Check to see if set to 'FALSE'
AM_RLC error indicator (for U-plane)	Check to see if set to 'FALSE'
START List	Checked to see if the 'CN domain identity' and 'START'
	IEs are present for all CN domains supported by the UE
Cell Update Cause	Check to see if set to 'RB Control'
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Measured results on RACH	Check to see if it is absent
Protocol error information	Check to see if it is absent

CELL UPDATE CONFIRM (Step 6)

Use the same message sub-type found in Annex A.

RADIO BEARER RELEASE COMPLETE (Step 7)

Only the message type is checked in this message.

8.3.1.14.5 Test Requirement

After step 4 the UE shall reselect to cell 2, perform a cell update procedure by transmitting CELL UPDATE message. In this message, the IE "Cell Update Cause" shall be set to "RB Control".

After step 6 the UE shall send RADIO BEARER RELEASE COMPLETE message on the DCCH carried by the PRACH channel. The applicable parameters of the PRACH resources are broadcasted on the system information messages of cell 2. Void

8.3.1.15 Cell Update: Acknowledged Mode RLC Reset

8.3.1.15.1 Definition

8.3.1.15.2 Conformance Requirement

In CELL_FACH, the UE shall ensure that all AM RLC entities (both signalling and u-plane links) are operational. In the event that an unrecoverable error has occurred, the UE shall trigger cell update procedure to report this event. The UE shall send CELL UPDATE message on the uplink CCCH and set the appropriate AM_RLC error indicator IE(s) to TRUE. After receiving the CELL UPDATE CONFIRM message, the UE shall reset the affected AM RLC entities and then resume transmission and reception activities.

8.3.1.15.3 Test Purpose

To confirm that the UE reports the occurrence of an unrecoverable error in a C-plane AM RLC entity by initiating cell update procedure. To confirm that the UE is able to resume normal C-plane data transmission and reception after the completion of cell update procedure.

8.3.1.15.4 Method of Test

Initial Condition

System Simulator: 1 cell

UE: CS-CELL_FACH_Initial (state 6-2) or PS-CELL_FACH_Initial (state 6-4) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE

Test Procedure

The UE is initially in CELL_FACH state. SS sends RADIO BEARER SETUP message on the DCCH using AM mode to establish a DTCH logical channel for u-plane packet data transfer. The UE shall reply with a RADIO BEARER SETUP COMPLETE message, sent using AM RLC on the DCCH. Then it activates the associated DTCH logical channel for user data transmission and reception. SS does not acknowledge the RADIO BEARER SETUP COMPLETE message. The UE shall continue to transmit the AM PDU carrying RADIO BEARER COMPLETE message until the maximum re-transmission count is reached. Thereafter, the UE shall start sending RESET PDUs to request that the AM RLC entity for RRC signalling be re-initialized. SS ignores the requests and wait for a duration equivalent to (MAX_RST+1) times expiry of Timer_RST. This figure is specified in IE "RLC info" of RADIO BEARER SETUP message in step 6. At this point, the UE shall initiate a cell update procedure by transmitting CELL UPDATE message on the uplink CCCH. The CELL UPDATE message shall specify the value "TRUE" in IE "AM_RLC error indicator (for C-plane)". SS replies with CELL UPDATE CONFIRM message using the default message content. SS then attempts to perform a local authentication by transmitting a COUNTER CHECK message using AM RLC on DCCH. The UE shall respond by sending a COUNTER CHECK RESPONSE message on the uplink DCCH, verifying that the AM RLC entity for RRC signalling was successfully reset.

Expected Sequence

Step	Direction	Message	Comment
-	UE SS		
1			The UE is initially in CELL_FACH state.
2	+	RADIO BEARER SETUP	Establishes a DTCH logical channel operating in AM mode.
3	→	RADIO BEARER SETUP COMPLETE	UE shall stay in CELL_FACH state. SS does not acknowledge this AM PDU. The UE shall re-transmit this AM PDU until the maximum number has been reached.
4			UE shall start to transmit RESET PDU using AM RLC on the DCCH. SS does not respond to any PDU frames originating from the UE, and it waits for a period equivalent to (MAX_RST+1) times expiry of Timer_RST. This figure is specified in IE "RLC info" of RADIO BEARER SETUP message in step 6.
5	→	CELL UPDATE	UE shall send this message on CCCH. IE "AM_RLC Error Indication (for C-plane)" shall be set to 'TRUE'
6	+	CELL UPDATE CONFIRM	"DRX Indicator RRC State Indicator" set to "Ne DRXCELL FACH". UE shall transit to CELL_FACH state.
7	+	COUNTER CHECK	SS requests for a local authentication of the amount of data sent/received during the lifetime of the RRC connection.
8	→	COUNTER CHECK RESPONSE	This message shall be transmitted using AM RLC for RRC signalling on the uplink DCCH.

Specific Message Contents

RADIO BEARER SETUP (Step 2)

Use the same message sub-type entitled "Packet to CELL_FACH from CELL_FACH in PS" found in Annex A.

RADIO BEARER SETUP COMPLETE (Step 3)

Only the message type IE is checked for this message.

CELL UPDATE (Step 5)

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to '0000 0000 0000 0000 0001'
Integrity check info	Not checked
AM_RLC error indicator (for C-plane)	Check to see if set to 'FALSE'
AM_RLC error indicator (for U-plane)	Check to see if set to 'TRUE'
START List	Checked to see if the 'CN domain identity' and 'START'
	IEs are present for all CN domains supported by the UE
Cell Update Cause	Not checked
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Measured results on RACH	Check to see if it is absent
Protocol error information	Check to see if it is absent

CELL UPDATE CONFIRM (Step 6)

Use the same message sub-type found in Annex A.

COUNTER CHECK (Step 7)

Information Element	Values/Remarks
Integrity check info	Not present
RB COUNT-C MSB Information	
- RB Identity	5
- COUNT-C-MSB-uplink	Set to an arbitrary integer equals to the 25 MSBs from
	COUNT-C for RB#5
- COUNT-C-MSB-downlink	Set to an arbitrary integer equals to the 25 MSBs from
	COUNT-C for RB#5

COUNTER CHECK RESPONSE (Step 8)

Information Element	Values/Remarks
Integrity check info	Not checked
RB COUNT-C Information	Not checked
- RB Identity	
- COUNT-C-uplink	
- COUNT-C-downlink	

8.3.1.15.5 Test Requirement

After step 4 the UE shall transmit a CELL UPDATE message on the uplink CCCH to report the occurrence of an unrecoverable error in AM RLC entity for C-plane data.

After step 7 the UE shall send a COUNTER CHECK RESPONSE message on the uplink DCCH. This message shall be sent using the AM RLC entity for RRC signalling.

8.3.1.16 Cell Update: cell reselection in CELL_FACH

8.3.1.16.1 Definition

8.3.1.16.2 Conformance requirement

This procedure is used to update UTRAN with the current cell of the UE after it has perform a cell reselection in CELL_FACH state. UE shall receive acknowledgement from UTRAN on downlink CCCH.

Reference

3GPP TS 25.331 clause 8.3.1

8.3.1.16.3 Test purpose

To confirm that the UE executes a cell update procedure after the successful reselection of another UTRA cell. To confirm that the UE send the correct uplink respond message when executing cell update procedure due to cell reselection. To confirm cell update procedure completes after UE receives CELL UPDATE CONFIRM on downlink CCCH from UTRAN.

8.3.1.16.4 Method of test

Initial Condition

System Simulator: 2 cells - Cell 1 is active, with the downlink transmission power shown in column marked "T0" in Table 8.3.1.1-1, while cell Cell 2 is inactive

UE: <u>CS-CELL_FACH_Initial</u> (state 6-2) or <u>PS-CELL_FACH_Initial</u> (state 6-4) <u>CELL_FACH</u> in cell <u>las specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE, ciphering in both UL and DL are disabled during RRC connection establishment.</u>

Test Procedure

The UE is in the CELL_FACH state, camping onto cell 1. <u>SS configures its downlink transmission power settings according to columns "T1" in Table 8.3.1.1-1. The SS starts to broadcast BCCH on the primary CPICH in cell 2 with a power level greater than that in cell 1. The UE shall find cell 2 to be more suitable for service and hence perform a cell reselection. After the completion of cell reselection, the UE shall transmits a CELL UPDATE message to the SS on the uplink CCCH of cell 2 and set IE "Cell update cause" to "Cell Reselection". After the SS receives this message, it transmits a CELL UPDATE CONFIRM message, which includes the IE "DRX Indicator RRC State Indicator" set to "DRX with Cell updating CELL PCH", IE "U-RNTI" and an IE "New U-RNTI" to the UE on the downlink CCCH. <u>UE shall response with UTRAN MOBILITY INFORMATION CONFIRM message.</u> SS verifies that the UE does not send any response to this message. UE shall move to CELL_PCH state.</u>

Expected sequence

Step	Direc	tion	Message	Comment
-	UE	SS		
1				The UE is in the CELL_FACH state in cell 1
2	•		BCCH	SS applies the downlink transmission power settings, according to the values in columns "T1" of Table 8.3.1.1- 1. The SS starts to broadcast BCCH on the primary CPICH in cell 2 with a power level that is higher than that in cell 1. The UE shall find that the cell 2 is better for service and perform a reselection. SS waits for the maximum duration required for the UE to camp to cell 2.
3	7	>	CELL UPDATE	Value "cell reselection" should be indicated in IE " Cell update cause"
5	<u> </u>	-	UTRAN MOBILITY INFORMATION CONFIRM	SS wait for T302+α. If SS receives CELL UPDATE message from the UE, this test fails.

Specific Message Contents

CELL UPDATE (Steps 3)

Use the same message sub-type found in Clause 9 of TS34.108.

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to '0000 0000 0000 0000 0001'
AM_RLC error indicator (for C-plane)	Check to see if set to 'FALSE'
AM_RLC error indicator (for U-plane)	Check to see if set to 'FALSE'
Cell Update Cause	Check to see if set to 'Cell Re-selection'
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Measured results on RACH	Check to see if it is absent
Protocol error information	Check to see if it is absent

CELL UPDATE CONFIRM (Step 4)

Use the same message sub-type found in Clause 9 of TS34.108 Annex A, with the following exceptions:

Information Element	Value/remark
<u>U-RNTI</u>	
 SRNC Identity 	<u>'0000 0000 0001'</u>
<u>- S-RNTI</u>	<u>'0000 0000 0000 0000 0001'</u>
DRX Indicator RRC State Indicator	DRX with cell updatingCELL_PCH
New U-RNTI	
- SRNC Identity	<u>'0000 0000 0000 0001'</u>
- S-RNTI	An arbitrary 20-bits string which is different from original
	<u>S-RNTI</u>

8.3.1.16.5 Test requirement

After step 2 the UE shall reselect to cell 2 and then it shall transmit a CELL UPDATE message which, sets the value "cell reselection" in IE "Cell update cause".

After step 4 the UE shall <u>transmit UTRAN MOBILITY INFORMATION CONFIRM message</u>.not transmit any uplink message in response to the CELL UPDATE CONFIRM message received in step 4. Test fails if UE does.

8.3.1.17 Cell Update: Failure (UTRAN initiate an RRC connection release procedure on DCCH)

8.3.1.17.1 Definition

8.3.1.17.2 Conformance requirement

The UE transmits a CELL UPDATE message to the UTRAN when it needs to update UTRAN with information on the current cell of the UE. If the UE receives a RRC CONNECTION RELEASE message on DCCH, it shall transmit an RRC CONNECTION RELEASE COMPLETE message using AM RLC on the DCCH to the UTRAN.

Reference

3GPP TS 25.331 clause 8.3.1

8.3.1.17.3 Test purpose

To confirm that the UE moves to idle state after sending RRC CONNECTION RELEASE COMPLETE message to UTRAN upon the reception of RRC CONNECTION RELEASE message on DCCH.

8.3.1.17.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CS-CELL_FACH_Initial (state 6-2) or PS-CELL_FACH_Initial (state 6-4) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE CELL_FACH

Test Procedure

The UE is initially in CELL_FACH state. When the UE detects the expiry of periodic cell updating timer T305, the UE transmits a CELL UPDATE message to the SS on the uplink CCCH to perform a periodic cell updating procedure. The SS transmits RRC CONNECTION RELEASE message on downlink DCCH. The UE shall transmit RRC CONNECTION RELEASE COMPLETE message using AM RLC on the DCCH and return to idle mode after release of all current signalling flows and radio access bearers.

Expected sequence

Step	Dire	ction	Message	Comment
	UE	SS		
1		→	CELL UPDATE	The value "periodic <u>al</u> cell update" should be set in IE " Cell update cause" and this message should be sent upon expiry of timer T302.
2	+	<u> </u>	RRC CONNECTION RELEASE	SS transmits RRC CONNECTION RELEASE message to the UE.
3	_	→	RRC CONNECTION RELEASE COMPLETE	The UE transmits this message using acknowledged mode. The UE releases L2 signalling link and radio resources then the UE goes to idle mode.

Specific Message Contents

CELL UPDATE (Step 1)

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to '0000 0000 0000 0000 0001'
Cell Update Cause	Check to see if set to 'Periodic cell updating'
·	

RRC CONNECTION RELEASE (Step 2)

Only the message type is checked for this message.

RRC CONNECTION RELEASE COMPLETE (Step 3)

Only the message type is checked for this message.

8.3.1.17.5 Test requirement

After step 1 the UE shall transmit a CELL UPDATE message on the uplink CCCH and set value "periodical cell update" into IE "Cell update cause".

After step 2 the UE shall transmit a RRC CONNECTION RELEASE COMPLETE message on the uplink DCCH and return to idle mode.

8.3.1.18 Cell Update: Radio Link Failure (T314>0, T315=0)

8.3.1.18.1 Definition

8.3.1.18.2 Conformance requirement

When a UE loses the radio connection due to e.g. radio link failure in CELL_DCH state. UE must release the radio bearer which is associated with T315 if T315 is set to 0. After a successful cell re-selection and subsequent transition to CELL FACH state, the UE transmits CELL UPDATE message on the uplink CCCH.

Reference

3GPP TS 25.331 clause 8.3.1

8.3.1.18.3 Test purpose

To confirm that the UE shall indicate to the non-access stratum the release of radio access bearer which is associated with T315 and try to find a new cell after detecting that a radio link failure has occurred.

8.3.1.18.4 Method of test

Initial Condition

System Simulator: 2 cells ☐ Cell 1 is active, Cell 2 is inactive ☐

UE: CS DCCH DCH (state 6-5) or PS DCCH DCH (state 6.7) in cell 1, after executing a generic call set-up procedure for mobile originating CS or PS call (clauses 7.2.3.2 and 7.2.4.2 in TS 34.108). The exact procedure to apply dependings on the CN domain(s) supported by the UE.

Test Procedure

The UE is brought to CELL_DCH state in a cell 1 after making an successful outgoing call attempt. After the call has been established, SS begins to broadcast the BCCH in cell 2, and then stops transmitting and receiving in cell 1. The UE shall detect a radio link failure in cell 1 and indicate to the non-access stratum the release of the radio bearer which is associated with T315. Then it shall attempt to re-select to cell 2. After that, it should then enter CELL_FACH state and transmits CELL UPDATE on the uplink CCCH to SS. The SS transmits CELL UPDATE CONFIRM message which includes IE "new C-RNTI". UE shall transmit an UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

Expected sequence

Step	Direction	<u>Message</u>	Comment
	UE SS		
<u>1</u>	<u></u>	RADIO BEARER SETUP	<u>T315=0</u>
2	<u>→</u>	RADIO BEARER SETUP COMPLETE	
<u>3</u>			The UE is brought to CELL DCH state in a cell 1, after making a successful outgoing call.
4	<u>+</u>	BCCH	The SS starts transmitting the BCCH in a cell 2 using the same contents (except for cell identity which is set to "0000 0000 0000 0010") for system information sent on cell 1. SS starts to listen to the uplink CCCH of cell 2.
<u>5</u>			The SS stops transmitting and receiving in a cell .1.
<u>6</u>			The UE detects the radio link failure which is associated with T315. The UE indicates to the non-access stratum the release of the radio bearer.
<u>7</u>	<u> →</u>	CELL UPDATE	The UE should find a new cell 2 and the value "radio link failure" should be set in IE "Cell update cause".
8	<u>+</u>	CELL UPDATE CONFIRM	Including IE "new U- RNTI" and IE "new C- RNTI"
<u>9</u>	<u>→</u>	UTRAN MOBILITY INFORMATION CONFIRM	

Specific Message Contents

RADIO BEARER SETUP

The contents of RADIO BEARER SETUP message in this test case is identical to those in default contents of layer 3 messages for RRC tests with the following exceptions:

Information Element	<u>Value/remark</u>
RAB information to setup list	
- RAB information to setup	
- RAB info	
<u>- T315</u>	<u>0</u>

CELL UPDATE (Step 7)

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	
-SRNC Identity	Check to see if set to value assigned previously in cell
- S-RNTI	1. Check to see if set to value assigned previously in cell 1.
Cell Update Cause	Check to see if set to 'radio link failure'

CELL UPDATE CONFIRM (Step 8)

Use the same message sub-type found in step 4, with the following exceptions:

Information Element	<u>Value/remark</u>
New U-RNTI	
- SRNC Identity	<u>'0000 0000 0000 0001'</u>
- S-RNTI	An arbitrary 20-bits string which is different from original
	S-RNTI
New C-RNTI	An arbitrary 16-bits string which is different from original
	C-RNTI.

8.3.1.18.5 Test requirement

After step 5, the UE shall indicate to the non-access stratum the release of the radio bearer which is associated with T315.

After step 6, the UE shall detect the presence of cell 2, perform cell re-selection and transmit CELL UPDATE message.

After step 8, the UE shall transmit UTRAN MOBILITY INFORMATION CONFIRM to SS.

8.3.1.19 Cell Update: Unrecoverble error in RLC

8.3.1.19.1 Definition

8.3.1.19.2 <u>Conformance requirement</u>

When a UE loses the radio connection due to e.g. detection of RLC unrecoverble error (amount of the retransmission of RESET_PDU reaches the value of Max_DAT and receives no ACK) in CELL_DCH state. After a successful cell reselection and transition to CELL_FACH state, the UE shall transmit CELL UPDATE message which includes the value "RLC unrecoverable error" in the IE "cell update cause" on the uplink CCCH.

Reference

3GPP TS 25.331 clause 8.3.1

8.3.1.19.3 Test purpose

<u>To confirm that the UE tries to find a new cell, after detecting that a RLC unrecoverble error has occurred. The UE shall move to CELL FACH state and transmit CELL UPDATE message to SS.</u>

8.3.1.19.4 Method of test

Initial Condition

System Simulator: 2 cells ☐ Cell 1 is active, with the downlink transmission power shown in column marked "T0" in Table 8.3.1.1-1, while cell 2 is inactive ☐

<u>UE: CS_DCCH_DCH</u> (state 6-5) or <u>PS_DCCH_DCH</u> (state 6-7) as defined in clause 7.4 of TS 34.108 in cell 1, The exact procedure to apply dependings on the <u>CN</u> domain(s) supported by the <u>UE</u>.

Test Procedure

The UE is brought to CELL DCH state in a cell 1 after a successful outgoing call attempt. After the call has been established, the SS transmits a PAGING TYPE 2 message on the downlink DCCH. Then the UE transmits an UPLINK DIRECT TRANSFER message on the uplink using AM-RLC for the response and the SS does not transmit a STATUS PDU for the response to AM-RLC PDU and begins to broadcast the BCCH in cell 2. The UE should detect an unrecoverble error in cell 1 and attempts to re-select to cell 2. It should then enter CELL FACH state and transmits CELL UPDATE message which includes the value "RLC unrecoverable error" in IE "cell update cause" on the uplink CCCH to SS. The SS transmits CELL UPDATE CONFIRM message which includes "TRUE" in RLC reset indicator(for C-plane) IE and a new TFCS setting according to the new transport channel. After this, UE shall reconfigure the

 $\frac{RLC \ and \ the \ new \ radio \ connection \ and \ transmits \ TRANSPORT \ CHANNEL \ RECONFIGURATION \ COMPLETE \ \underline{message} \ on \ the \ uplink \ DCCH \ using \ AM \ RLC.}$

Expected sequence

<u>Step</u>	Direction	<u>Message</u>	Comment
	UE SS		
1			The UE is brought to CELL_DCH state in a cell 1, after making a successful outgoing call.
2	<u>←</u>	PAGING TYPE2	The SS transmits a PAGING TYPE 2 message to the UE on the downlink DCCH in cell 1.
3	<u> </u>	UPLINK DIRECT TRANSFER	The UE responds to the PAGING TYPE 2 message using AM-RLC but the SS does not transmit a STATUS PDU as an acknowledgement.
4	<u></u>	BCCH	The SS starts transmitting the BCCH in a cell 2 using the same contents (except for cell identity which is set to "0000 0000 0000 0010") for system information sent on cell 1. SS starts to listen to the uplink CCCH of cell 2.
<u>5</u>			The UE detects an unrecoverble error in the RLC level.
<u>6</u>	<u>→</u>	CELL UPDATE	The UE should find a new cell 2 and then transmits this message which includes the value "RLC unrecoverable error" in the IE "cell update cause".
7	<u></u>	CELL UPDATE CONFIRM	Including the new configuration information.
<u>8</u>	<u>→</u>	TRANSPORT CHANNEL RECONFIGURATION COMPLETE	

Specific Message Contents

Cell 1: SYSTEM INFORMATION TYPE 1

Information Element	<u>Value/remark</u>
UE Timers and constants in connected mode	
<u>T301</u>	8 seconds
<u>T313</u>	15 seconds
<u>T314</u>	320 seconds
<u>T315</u>	30 seconds
T301 T313 T314 T315 N313	16 200

CELL UPDATE (Step 6)

Information Element	<u>Value/remark</u>
<u>U-RNTI</u>	
-SRNC Identity	Check to see if set to value assigned previously in cell
<u>- S-RNTI</u>	1. Check to see if set to value assigned previously in cell 1.
Cell Update Cause	Check to see if set to 'RLC unrecoverable error'

8.3.1.19.5 Test requirement

After step 5, the UE shall detect the presence of cell 2 and move to CELL_FACH in cell 2 to transmit CELL UPDATE message to SS.

After step 7, the UE shall transmit TRANSPORT CHANNEL RECONFIGURATION COMPLETE message to SS.

8.3.1.20 Cell Update: Reception of CELL UPDATE CONFIRM Message that causes invalid configuration

8.3.1.20.1 Definition

8.3.1.20.2 Conformance Requirement

Invalid Configuration
INVALID CONFIGURATION to TRUE while executing a cell update procedure, it shall check the current value of its internal counter V302. If V302 is not greater than N302, the UE shall set IE "failure cause" to "invalid configuration", re-transmits CELL UPDATE message on uplink CCCH, restart T302 timer and increments V302. It shall use the same "Cell Update Cause" as before receiving the invalid downlink message. On the other hand, if V302 is greater than N302, the UE shall abandon cell update procedure and enters idle mode.

8.3.1.20.3 Test Purpose

To confirm that the UE retransmits CELL UPDATE message when it receives a CELL UPDATE CONFIRM message that will trigger an invalid configuration in the UE, if the number of retransmissions ishas not reached the maximum allowed value. To confirm that the UE returns to idle mode after sending maximum allowed number of CELL UPDATE messages without receiving a valid CELL UPDATE CONFIRM message.

8.3.1.20.4 Method of Test

Initial Condition

System Simulator: 1 cell

UE: CELL PCH (state 6-12) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is brought to CELL_PCH state at the beginning of the test. SS pages the UE by sending PAGING TYPE 1 message using the U-RNTI identity assigned during RRC connection establishment procedure. The UE shall transmit CELL UPDATE message on the uplink CCCH. Upon receiving such a message, the SS replies with a CELL UPDATE CONFIRM message with IE "RRC State Indicator" set to "CELL_DCH". The UE shall detect its variable "invalid configuration" is set and re-transmit CELL UPDATE message up to a maximum of N302+1 times. SS verifies that it receives a total of (N302+2) identical CELL UPDATE messages. The UE shall return to idle mode after all uplink transmissions have finished. SS verifies this by paging the UE using the U-RNTI identity. The UE shall not respond to this page.

Expected Sequence

Step	<u>Direction</u>	<u>Message</u>	Comment
	UE SS		
<u>1</u>	_←	PAGING TYPE 1	The UE is in the CELL PCH state. SS sets its internal counter K=0. SS pages for the
			UE using the allocated connected mode identity (U-RNTI).
<u>2</u>	<u>→</u>	CELL UPDATE	If CELL UPDATE message is received, check that the value "paging response" is set in IE "Cell update cause". Else goes to step 6.
<u>3</u>	<u></u>	CELL UPDATE CONFIRM	SS transmits an invalid message. SS increments K.
4	<u> </u>	CELL UPDATE	
<u>4</u> <u>5</u>			If a CELL UPDATE message is received in step 4, SS increments K and returns to step 3. Else, SS proceeds to step 6.
<u>6</u>			SS verifies that K = (N302+2) and proceeds to the next step. Else, the test fails.
<u>7</u>		PAGING TYPE 1	SS pages the UE.
8			UE shall not respond.

Specific Message Content

CELL UPDATE (Step 2)

Value/remark	
Check to see if set to '0000 0000 0001'	
Check to see if set to '0000 0000 0000 0000 0001'	
Checked to see if the 'CN domain identity' and 'START'	
IEs are present for all CN domains supported by the UE	
Check to see if set to 'Paging Response'	

CELL UPDATE CONFIRM (Step 3)

Use the same message sub-type found in Annex A, with the following exception:

Information Element	<u>Value/remark</u>
RRC State Indicator	CELL DCH
<u>Uplink DPCH info</u>	Not Present

CELL UPDATE (Step 4)

Information Element	<u>Value/remark</u>
U-RNTI - SRNC Identity - S-RNTI START List	Check to see if set to '0000 0000 0001' Check to see if set to '0000 0000 0000 0000 0001' Checked to see if the 'CN domain identity' and 'START'
Cell Update Cause Failure cause	IEs are present for all CN domains supported by the UE Check to see if set to 'Paging Response' Check to see if it is set to 'invalid configuration'

PAGING TYPE 1 (Step 1 and 7)

Information Element	<u>Value/remark</u>
Page record list	
- Paging record	
 CHOICE Paging originator 	UTRAN Originator
<u>- U-RNTI</u>	
- SRNC Identity	<u>'0000 0000 0001'</u>
- S-RNTI	<u>'0000 0000 0000 0000 0001'</u>

8.3.1.20.5 Test Requirement

After step 3 the UE shall continue to transmit CELL UPDATE message for N302+1 times.

At step 6 the counter K should be equal to (N302+2).

After step 7 the UE shall return to idle mode and not respond the PAGING TYPE 1 message sent by the SS.

8.3.2 URA Update

8.3.2.1 URA Update: URA reselection

8.3.2.1.1 Definition

8.3.2.1.2 Conformance requirement

This procedure is to update UTRAN with the current URA of the UE after a URA reselection has occurred in URA_PCH state. It may also be used for supervision of the RRC connection, even if no URA reselection takes place.

Reference

3GPP TS 25.331 clause 8.3.<u>12</u>

8.3.2.1.3 Test purpose

To confirm that the UE executes an URA update procedure after the successful URA reselection.

8.3.2.1.4 Method of test

Initial Condition

System Simulator: 2 cells - Cell 1 is active with URA-ID 1 and the downlink transmission power shown in column marked "T0" in Table 8.3.1.1-1, Cell-while cell 2 is inactive with URA-ID 2

UE: URA_PCH (state 6-1613) as specified in clause 7.4 of TS 34.108, with URA-ID 1 from the list of URA-ID in cell 1

Test Procedure

The UE is in the URA_PCH state and assigned with only 1 URA identity in cell 1: URA-ID 1. The SS starts to broadcast BCCH in cell 2 with URA-ID 2 and stop transmitting BCCH in cell 1. This is expected to cause the UE to perform a cell reselection to cell 2. When the UE finds that its current URA-ID 1 is not in the new broadcasted list of URA-IDs, it moves to CELL_FACH state and transmits a URA UPDATE message on the uplink CCCH. After the SS receives this message, it transmits URA UPDATE CONFIRM message which includes the IEs "DRX Indicator RRC State Indicator" and "URA-ID" to the UE on the downlink DCCH. The "DRX Indicator RRC State Indicator" is set to "DRX with URA updating URA PCH". Finally, the UE returns to URA_PCH state in cell 2 without sending a uplink response message.

Expected sequence

Step	Direction		Message	Comment
-	UE	SS]	
1				The UE is updated with only 1 URA identity carried currently by cell 1. The starting state of the UE is URA_PCH
2		←	ВССН	SS starts sending BCCH for cell 2 with URA-ID 2 and ceases to transmit BCCH with URA-ID 1 carried by cell 1.
3		→	URA UPDATE	The UE shall perform a cell reselection first and when it finds that its current URA-ID 1 is not in the new broadcasted list of URA-IDs, it shall then transmit this message and set value "URA reselection" into IE "URA update cause".
4		←	URA UPDATE CONFIRM	Message comprises IE "DRX Indicator RRC State Indicator" set "DRX with URA updating URA_PCH", and also IE "URA Identity" equals to "URA-ID 2".

Specific Message Contents

URA UPDATE (Step 3)

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to '0000 0000 0000 0000 0001'
Integrity check info	Not checked
AM_RLC error indicator	Check to see if set to 'FALSE'
URA Update Cause	Check to see if set to 'Change of URA reselection'
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Protocol error information	Check to see if it is absent

URA UPDATE CONFIRM (Step 4)

Use the same message sub-type found in Annex A, with the following exceptions:

Information Element	Value/remark
DRX Indicator RRC State Indicator	DRX with URA updatingURA_PCH

8.3.2.1.5 Test requirement

After step 2 the UE shall find that URA-ID 2 is not in its maintained list of URA-IDs. After cell reselection, the UE shall move to CELL_FACH state and transmit URA UPDATE message setting value "Change of URA reselection" into IE "URA update cause".

8.3.2.2 URA Update: periodical URA update

8.3.2.2.1 Definition

8.3.2.2.2 Conformance requirement

This procedure is to update UTRAN with the current URA of the UE when the UE detects that it is still within the service area after the expiry of periodic URA updating timer <u>T306T305</u>.

Reference

3GPP TS 25.331 clause 8.3.21

8.3.2.2.3 Test purpose

To confirm that the UE executes a URA update procedure after the expiry of timer <u>T306T305</u>. To verify that the UE handles an invalid URA UPDATE CONFIRM message correctly when executing the URA update procedure.

8.3.2.2.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: URA_PCH (state 6-1613) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the URA_PCH state. When the UE detects the expiry of timer \$\frac{T306}{T305}\$, set according to the value specified in system information, the UE moves to CELL_FACH state and transmits a URA UPDATE message to the SS on the uplink CCCH. The message shall indicate the cause to be "periodic URA update" in IE "URA update cause". SS replies with an illegal URA UPDATE CONFIRM message sent on downlink CCCH, and check to see if the UE handles this event properly. The UE shall attempt to retransmit the identical URA UPDATE message. After the SS receives the second URA UPDATE message, it transmits a correct URA UPDATE CONFIRM message, which includes the IE "new U-RNTI", to the UE on the downlink DCCH. Then the UE shall then transmits an UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH. The UE returns to CELL_FACH state.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1				The UE is in the URA_PCH state. SS wait until T306T305 timer has expired.
2		→	URA UPDATE	UE shall transmit this message and set value "periodic URA update" into IE "URA update cause".
3		(URA UPDATE CONFIRM	SS sends an illegal message.
4		→	URA UPDATE	UE shall not return to idle mode immediately, but attempts to re-transmit this message.
5		+	URA UPDATE CONFIRM	Including IEs "new C-RNTI", and "new U-RNTI"
6		\rightarrow	UTRAN MOBILITY INFORMATION CONFIRM	

Specific Message Contents

URA UPDATE (Step 2 and 4)

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to '0000 0000 0000 0000 0001'
Integrity check info	Not checked
AM_RLC error indicator	Check to see if set to 'FALSE'
URA Update Cause	Check to see if set to 'Periodic URA update'
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Protocol error information	Check to see if it is absent

URA UPDATE CONFIRM (Step 3)

Use the same message sub-type found in Annex A, with the following exceptions:

Information Element	Value/remark
U-RNTI	Not Present

URA UPDATE CONFIRM (Step 5)

Use the same message sub-type found in Annex A, with the following exceptions:

Information Element	Value/remark
New U-RNTI	
SRNC Identity	'0000 0000 0001'
S-RNTI	'0000 0000 0000 0000 1111'

UTRAN MOBILITY INFORMATION CONFIRM (Step 6)

Only the message type IE of this message is checked.

8.3.2.2.5 Test requirement

After step <u>2-1</u> the UE shall detect the expiry of timer <u>T306T305</u>, move to CELL_FACH state, and transmit a URA UPDATE message which is set the value "periodic<u>al</u> cell update" into IE "URA update cause".

After step 3 the UE shall re-transmit URA UPDATE message.

After step 5 the UE shall transmit an UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH and returns to the CELL FACH state.

8.3.2.3 URA Update: re-entering of service area after T306T305 expiry

8.3.2.3.1 Definition

8.3.2.3.2 Conformance requirement

This procedure is to update UTRAN with the current URA of the UE if the UE detects that it is out of service area after the expiry of timer <u>T306T305</u>, and then subsequently re-enters the service area before the expiry of T307.

Reference

3GPP TS 25.331 clause 8.3.21

8.3.2.3.3 Test purpose

To confirm that the UE executes a URA update procedure when the UE re-enters the service area before the expiry of timer T307, after being out of service area at the expiry of timer T306T305.

8.3.2.3.4 Method of test

Initial Condition

System Simulator: 2 cells - Cell 1 is active with URA-ID 1 and the downlink transmission power shown in column marked "T0" in Table 8.3.1.1-1, while Ccell 2 is inactive with URA-ID 2

UE: URA_PCH (state 6-1613) as specified in clause 7.4 of TS 34.108, with URA-ID 1 in the list of URA-ID from cell 1

Test Procedure

The UE is initially in URA_PCH state. SS decrease the transmission power of cell such that cell selection figure of merit S<0. When the UE detects the expiry of timer T306T305 according to the system information, the UE moves to CELL_FACH state and finds that it is out of service area. The UE is expected to search for cell to camp. Then SS increases the transmission power so that the UE detects that it returns to normal service within T307. The UE shall move to CELL_FACH state and starts transmitting a URA UPDATE message which contains the value "re-entered service area" in IE "URA update cause" to the SS on the uplink CCCH. After the SS receives this message, it transmits a URA UPDATE CONFIRM message which includes the IE "new C-RNTI", and "new U-RNTI" to the UE on the downlink DCCH. Then the UE shall transmit an UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH. Next, cell 1 is switched off. SS waits until T305 timer has expired and then turns on cell 2. The UE shall discover that cell 1 is no longer suitable for camping and initiate a cell search. It shall detect the presence of cell 2 and reselects to this cell. When the UE finds that URA-ID 2 is not in its current list of URA-IDs, it moves to CELL_FACH state and transmits a URA UPDATE message on the uplink CCCH.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1				The UE starts operating from
				URA_PCH state.
2				SS decreases the
				transmission power such that
				the cell 1 is no longer suitable
				for camping i.e. S<0.
3				The UE shall attempt to
				perform a URA update upon
				the expiry of timer T306 <u>T305</u> .
				It shall discover that it is out of
				service and starts searching
				for cell to camp.(T307 timer
				starts)
4				SS increases the transmission
				power to the original level
				before T307 expires.
5		\rightarrow	URA UPDATE	Value "re-entered service
				area" shall be set in IE "URA
				update cause"
6			URA UPDATE CONFIRM	The message includes IEs
				"new C-RNTI", and "new U-
				RNTI"
7		\rightarrow	UTRAN MOBILITY INFORMATION	
'			CONFIRM	
8				SS ceases to transmit BCCH
				with URA-ID 1 carried by cell
				1and wait until T306 T305
				expires.
9				SS starts sending BCCH for
				cell 2 with URA-ID 2 and.
10		\rightarrow	URA UPDATE	UE shall detect the presence
				of cell 2 and re-select to it. It
				shall transmit this message
				with cause set to "change of
				URA reselection"
				Cra t resciention

Specific Message Contents

Use the same message sub-type found in Annex A, with the following exceptions:

URA UPDATE (Step 5)

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to '0000 0000 0000 0000 0001'
AM_RLC error indicator	Check to see if set to 'FALSE'
URA Update Cause	Check to see if set to 're-entered service area'
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Protocol error information	Check to see if it is absent

URA UPDATE CONFIRM (Step 6)

Information Element	Value/remark	
New U-RNTI - SRNC Identity	'0000 0000 0001'	
- S-RNTI New C-RNTI	'0000 0000 0000 1111 1111' Arbitrary 16-bit string which is different the assigned C-RNTI in RRC CONNECTION SETUP message.	

URA UPDATE (Step 10)

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to '0000 0000 0000 1111 1111'
AM_RLC error indicator	Check to see if set to 'FALSE'
URA Update Cause	Check to see if set to 'Change of URA reselection'
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Protocol error information	Check to see if it is absent

8.3.2.3.5 Test requirement

After step 2 the UE shall detect that it is out of service area and shall not send a URA UPDATE on the uplink CCCH channel.

After step 4 the UE shall transmit a URA UPDATE message which sets value "re-entered service area" into IE "URA update cause", before the expiry of timer T307.

After step 6 the UE shall transmit an UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH.

After step <u>10-9</u> the UE shall transmit a URA UPDATE message which sets value "change of URA reselection" into IE "URA update cause".

8.3.2.4 URA Update: loss of service after expiry of timers T307 and T306T305

8.3.2.4.1 Definition

8.3.2.4.2 Conformance requirement

This procedure is required to handle the case when the UE fails to update UTRAN with the current URA of after expiry of timers T307 and T306T305 consecutively. The UE shall move to idle mode subsequently.

Reference

3GPP TS 25.331 clause 8.3.21

8.3.2.4.3 Test purpose

To confirm that the UE moves to idle mode after the expiry of timer T307, following an expiry of timer $T306\underline{T305}$ when it discovers that it is out of service area.

8.3.2.4.4 Method of test

Initial Condition

System Simulator: 1cell

Test Procedure

The UE is in URA_PCH state. SS stops the downlink transmissions of cell 1. When the UE detects the expiry of periodic URA updating timer <u>T306T305</u> according to the system information, the UE moves to CELL_FACH state and detects that it is out of service area. After the expiry of timer T307, the UE moves to the idle state and start to perform cell reselection.

Expected sequence

Step	Direction		Message	Comment
	UE	SS	-	
1				Initially, the UE is in the
				URA_PCH state.
2				SS switched off the downlink
				transmission of cell 1 so that
				the UE detects that it is out of
				service area.
3				Upon the expiry of timer
				T306T305, the UE shall
				search for cell to camp and
				triggers T307 timer. SS listens
				to the uplink CCCH to verify
				that URA UPDATE message
				is not transmitted.
4				After the expiry of timer T307,
				the UE enters idle state.

Specific Message Contents

None

8.3.2.4.5 Test requirement

After step 2 the UE shall detect the expiry of timer <u>T306T305</u>, not transmit URA UPDATE message on the uplink CCCH, move to CELL_FACH state, and start timer T307.

8.3.2.5 URA Update: Success after Confirmation error of URA-ID list

8.3.2.5.1 Definition

8.3.2.5.2 Conformance requirement

UE transmits a URA UPDATE message to the UTRAN when it needs to update UTRAN with the current URA of the UE. UTRAN should respond to the URA UPDATE message by sending a URA UPDATE CONFIRM message. When the indicated URA-ID in the received URA UPDATE CONFIRM message is not found in the list of URA-IDs that is broadcasted in system information block type 2, the UE transmits a URA UPDATE message repeatedly until its internal counter <u>V303-V302</u> is greater than <u>N303N302</u>.

Reference

3GPP TS 25.331 clause 8.3.21

8.3.2.5.3 Test purpose

To confirm that the UE retries to perform the URA update procedure following a confirmation error of URA-ID list.

8.3.2.5.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: URA_PCH (state 6-1613) as specified in clause 7.4 of TS 34.108

Test Procedure

At the start of this test, the UE is brought to URA_PCH state and assigned a URA with URA-ID 1. When the UE detects the expiry of timer \$\frac{T306}{T305}\$ according to the system information, the UE moves to CELL_FACH state and transmits a URA UPDATE message to the SS on the uplink CCCH. The reason for performing URA updating shall be set to "periodic URA update" in IE "URA update cause". After the SS receives this message, it transmits a URA UPDATE CONFIRM message which includes the IE "new C-RNTI", "new U-RNTI" and "URA-ID 2" to the UE on the downlink DCCH. The UE finds that the indicated URA-ID is not included in the list of URA-IDs broadcasted in system information block type 2, then the UE shall retry to transmit a URA UPDATE message for a confirmation error of URA-ID list. SS continue to send the same URA UPDATE CONFIRM message until \$\frac{N303}{N302+1}\$ URA UPDATE messages have been received. Then SS transmits a URA UPDATE CONFIRM message to the UE which includes IE "URA Identity" set to "URA-ID 1". The UE shall find this URA-ID in its URA-ID list and transmits an UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH.

Expected sequence

Step	Direction	Message	Comment
-	UE SS	_	
1			The UE is URA_PCH state. SS initializes counter K to 0
2	→	URA UPDATE	This message shall contain value "periodic URA update" set in IE URA update cause" after expiry of timer T306T305.
3			SS increments K by 1.
4	+	URA UPDATE CONFIRM	SS transmits this message, setting the value "URA-ID 2" to IE "URA Identity". If K is not greater than N303N302+1, SS waits for T303-T302 to expires and then returns to step 2. If K is greater than N303N302+1, SS proceeds to step 5.
5	+	URA UPDATE CONFIRM	SS transmits this message, setting IE "URA Identity" to "URA-ID 1". This message also comprises IE "New U- RNTI".
6	→	UTRAN MOBILITY INFORMATION CONFIRM	

Specific Message Contents

URA UPDATE (Step 2)

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to '0000 0000 0000 0000 0001'
Integrity check info	Not checked
AM_RLC error indicator	Check to see if set to 'FALSE'
URA Update Cause	Check to see if set to 'Periodic URA update'
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Protocol error information	Check to see if it is absent

URA UPDATE CONFIRM (Step 4)

Use the same message sub-type as specified in Annex A, with the following exceptions:

Information Element	Value/remark
DRX Indicator RRC State Indicator	DRX with URA updatingURA_PCH
URA Identity	2

URA UPDATE CONFIRM (Step 5)

Use the same message sub-type as specified in Annex A, with the following exceptions:

Information Element	Value/remark
New U-RNTI	
SRNC Identity	'0000 0000 0001'
S-RNTI	'0000 0000 0000 0101 0101'
URA Identity	1

UTRAN MOBILITY INFORMATION CONFIRM (Step 6)

Only the message type IE in this message is checked.

8.3.2.5.5 Test requirement

After step 1 the UE shall detect the expiry of timer T306T305, move to CELL_FACH state, transmit a URA UPDATE message on the uplink CCCH and set value "periodic URA update" into IE "URA update cause".

After step 2 the UE shall repeatedly re-transmit a URA UPDATE message after it detects a confirmation error of URA-ID list for the URA-ID indicated in the URA UPDATE CONFIRM message. A total of (N303N302+12) URA UPDATE messages shall be received by the SS.

After step 5 the UE shall transmit an UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH.

8.3.2.6 URA Update: Failure (V303-<u>V302</u> is greater than N303<u>N302</u>: Confirmation error of URA-ID list)

8.3.2.6.1 Definition

8.3.2.6.2 Conformance requirement

UE transmits a URA UPDATE message to the UTRAN when it needs to update UTRAN with the current URA of the UE. When the indicated URA-ID in the received URA UPDATE CONFIRM message is not in the list of URA-IDs that

is broadcasted in system information block type 2, the UE transmits URA UPDATE messages repeatedly until its internal counter <u>V303-V302</u> is greater than <u>N303-N302</u>. If <u>V303-V302</u> is greater than <u>N303-N302</u> then the UE enters idle state.

Reference

3GPP TS 25.331 clause 8.3.21

8.3.2.6.3 Test purpose

To confirm that the UE make repeated attempts to perform the URA update procedure following a detection of a confirmation error of URA-ID list. It then moves to idle state when internal counter <u>V303-V302</u> is greater than <u>N303N302</u>.

8.3.2.6.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: URA_PCH (state 6-1613) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is originally in the URA_PCH state updated with URA-ID 1. When the UE detects the expiry of timer \$\frac{T306}{T305}\$ according to the system information, the UE shall move to CELL_FACH state and transmit a URA UPDATE message to the SS on the uplink CCCH. In this message, the value "periodic URA update" shall be set in IE "URA update cause". After the SS receives this message, it transmits a URA UPDATE CONFIRM message which includes the IE "new C-RNTI", "new U-RNTI" and indicating the IE "URA Identity" to be "URA-ID 2" to the UE on the downlink DCCH. The UE finds that the indicated URA-ID is not included in the list of URA-IDs broadcasted, the UE shall retry to transmit a URA UPDATE message for \$\frac{N303}{N302+1}\$ times. After that, the UE shall enter idle state.

Expected sequence

Step	Dire	ction	Message	Comment
	UE	SS		
1				The UE is in URA_PCH state at the start of the test. SS sets internal counter K to 40.
2		→	URA UPDATE	The message shall indicate "periodic URA update" in IE "URA update cause". This message is sent following the expiry of timer T306T305. SS increments counter K by 1.
3		+	URA UPDATE CONFIRM	The SS transmit this message and set IE "URA Identity" to "URA-ID 2". When K greater than N303-N302+2 proceeds to step 4, else SS waits for T303-T302 to expires and executes step 2.
4				SS waits for a T306T305 to verify that no further URA UPDATE messages are transmitted by UE. The counter K shall be equal to (N303N302+12). The UE shall

enter idle state.

Specific Message Contents

URA UPDATE CONFIRM (Step 4)

Use the same message sub-type defined in Annex A, with the following exceptions:

Information Element	Value/remark
URA Identity	2

8.3.2.6.5 Test requirement

After step 1 the UE shall detect the expiry of timer T306T305, then it shall move to CELL_FACH state and transmit a URA UPDATE message on the uplink CCCH, setting value "periodic URA update" into IE "URA update cause".

After step 2 the UE shall retry to transmit a URA UPDATE message after it detects the confirmation error of URA-ID list for the URA-ID included in the URA UPDATE CONFIRM message.

After step 3 the UE shall stop transmitting URA UPDATE message and then enters idle state. The counter K shall be equal to (N303N302+12).

8.3.2.7 URA Update: Success after T303 T302 timeout

8.3.2.7.1 Definition

8.3.2.7.2 Conformance requirement

The UE transmits an URA UPDATE message to the UTRAN when it needs to update UTRAN with the current URA identity stored the UE. When the UE fails to receive any URA UPDATE CONFIRM message after T303-T302 timer expiry, it transmits a URA UPDATE message repeatedly at an interval of T303-T302 timer value until its internal counter V303-V302 is greater than N303N302.

Reference

3GPP TS 25.331 clause 8.3.21

8.3.2.7.3 Test purpose

To confirm that the UE attempts to repeat the URA update procedure upon the expiry of timer <u>T303T302</u>. To confirm that a maximum of <u>N303-N302+1</u> re-transmission is performed.

8.3.2.7.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: URA_PCH (state 6-1613) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the URA_PCH. When the UE detects the expiry of timer \$\frac{T306}{T305}\$ according to the system information, the UE moves to CELL_FACH state and transmits a URA UPDATE message to the SS on the uplink CCCH, setting value "periodic URA update" into IE "URA update cause". The SS ignores this message, the UE shall then retry to transmit a URA UPDATE message after the expiry of timer \$\frac{T303}{T302}\$. SS continues to ignore further URA UPDATE message until it receives (\$\frac{N303}{N302} + 1\$) such messages. Then it transmits a URA UPDATE CONFIRM message to the UE which includes IEs "new C-RNTI", "new U-RNTI". The UE shall then transmit an UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH.

Expected sequence

Step	Direc	ction	Message	Comment
	UE	SS		
1				The UE is in URA_PCH state at the beginning of test. SS sets counter K to 40. SS waits for T306T305 to expire.
2		→	URA UPDATE	This message shall contain value "periodic URA update" in IE "URA update cause" sent upon the expiry of timer T306T305.
3				SS increments K by 1.
4				If K is not greater than N303N302+1, SS transmits no response to the UE, waits for an additional period equals to T303-T302 timer and returns to step 2. Else, SS executes step 5.
5		(URA UPDATE CONFIRM	This message includes IEs" new C-RNTI", "new U-RNTI"
6		→	UTRAN MOBILITY INFORMATION CONFIRM	

Specific Message Contents

URA UPDATE CONFIRM (Step 5)

Use the same message sub-type as in Annex A, with the following exceptions:

Information Element	Value/remark
New U-RNTI	
SRNC Identity	'0000 0000 0001'
	Arbitrary 20-bit string which is different from S-RNTI
S-RNTI	field in IE "U-RNTI"
New C-RNTI	Arbitrary 16-bit string which is different the assigned C-
	RNTI in RRC CONNECTION SETUP message.

8.3.2.7.5 Test requirement

After step 1 the UE shall detect the expiry of timer <u>T306T305</u>, then it shall move to CELL_FACH state and transmit a URA UPDATE message on the uplink CCCH. The updating cause shall be set to "periodic URA update" in IE "URA update cause".

After step 2 the UE shall retry to transmit a URA UPDATE message at each expiry of timer $\frac{\text{T303}\text{T302}}{\text{LMS03}\text{-}\text{N302}+1}$ UE shall attempt to re-transmit $\frac{\text{N303}\text{-}\text{N302}+1}{\text{LMS03}\text{-}\text{N302}+1}$ URA UPDATE messages.

After step 5 the UE shall transmit an UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH.

8.3.2.8 URA Update: Failure (V303-<u>V302</u> is greater than N303<u>N302</u>:T303-<u>T302</u> timeout)

8.3.2.8.1 Definition

8.3.2.8.2 Conformance requirement

The UE transmits a URA UPDATE message to the UTRAN when it needs to update the UTRAN with the current URA of the UE. When the UE fails to receive the URA UPDATE CONFIRM message, the UE transmits a URA UPDATE message repeatedly after every expiry of <u>T303-T302</u> until its internal counter <u>V303-V302</u> is greater than <u>N303N302</u>. If <u>V303-V302</u> is greater than <u>N303N302</u>, UE stops sending URA UPDATE message and then enters idle state.

Reference

3GPP TS 25.331 clause 8.3.21

8.3.2.8.3 Test purpose

To confirm that the UE retries to perform the URA update procedure upon expiry of timer <u>T303-T302</u> and moves to idle state after retrying for <u>N303-N302+1</u> times.

8.3.2.8.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: URA_PCH (state 6-1613) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the URA_PCH state. When the UE detects the expiry of timer \$\frac{T306}{T305}\$ according to the system information, the UE moves to CELL_FACH state and transmits a URA UPDATE message to the SS on the uplink CCCH. This message shall contain value "periodical URA update" in IE "URA update cause". SS ignores this message, the UE shall continue to transmit URA UPDATE messages for \$\frac{N303}{N302+1}\$ times after the expiry of timer \$\frac{T303}{T302}\$. After \$\frac{N303}{N302+1}\$ re-transmissions, the UE shall enter idle state.

Expected sequence

Step	Dire	ction	Message	Comment
	UE	SS		
1				The UE is in URA_PCH state and SS sets counter K=0. SS wait until T303 expires.
2		\rightarrow	URA UPDATE	The value "periodic URA update" shall be set in IE "URA update cause".
3				SS ignores the message, waits for T303-T302 timer to expire and increments K by 1. If a message is received after T303 expiry, return to step 2. Else, go to step 4.
4				SS checks that K is equal to (N303N302+21).
5				The UE shall enter idle state.

Specific Message Contents

None

8.3.2.8.5 Test requirement

After step 1 the UE shall detect the expiry of timer T306T305, then it shall move to CELL_FACH state and transmit a URA UPDATE message on the uplink CCCH, setting "periodical URA update" into IE "URA update cause".

After step 2 the UE shall retry to transmit a URA UPDATE message after the expiry of timer <u>T303T302</u>. SS shall receive (<u>N303N302+12</u>) CELL UPDATE message. After this, the UE shall enter idle state.

8.3.2.9 URA Update: Failure (UTRAN initiate an RRC connection release procedure on DCCH)

8.3.2.9.1 Definition

8.3.2.9.2 Conformance requirement

The UE transmits a URA UPDATE message to the UTRAN when it needs to update UTRAN with information on the current URA of the UE. If the UE receives a RRC CONNECTION RELEASE message on downlink DCCH, it shall enter idle state after sending a RRC CONNECTION RELEASE COMPLETE message to UTRAN.

Reference

3GPP TS 25.331 clause 8.3.21

8.3.2.9.3 Test purpose

To confirm that the UE moves to idle state after sending RRC CONNECTION RELEASE COMPLETE message to UTRAN upon the reception of RRC CONNECTION RELEASE message on downlink DCCH.

8.3.2.9.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: URA_PCH (state 6-13) as specified in clause 7.4 of TS 34.108

Test Procedure

The UE is in the URA_PCH state. When the UE detects the expiry of periodic URA updating timer \$\frac{T306}{T305}\$, the UE moves to CELL_FACH state and transmits a URA UPDATE message to the SS on the uplink CCCH. The message shall indicate the cause to be "periodic URA update" in IE "URA update cause". The SS transmits RRC CONNECTION RELEASE message on downlink DCCH. The UE shall transmit RRC CONNECTION RELEASE COMPLETE message using AM RLC on the DCCH and return to idle mode after release of all current signalling flows and radio access bearers.

Expected sequence

Step	Direc	ction	Message	Comment
	UE	SS	_	
1				The UE is in the URA_PCH state. SS wait until T306 T305 timer has expired.
2		\rightarrow	URA UPDATE	UE shall transmit this message and set value "periodic URA update" into IE " URA update cause".
3	*	-	RRC CONNECTION RELEASE	SS transmits RRC CONNECTION RELEASE message to the UE on the downlink CCCH.
4	-	>	RRC CONNECTION RELEASE COMPLETE	The UE transmits this message using acknowledged mode. The UE releases L2 signalling link and radio resources then the UE goes to idle mode.

Specific Message Contents

URA UPDATE (Step 2)

Information Element	Value/remark
U-RNTI - SRNC Identity - S-RNTI URA Update Cause	Check to see if set to '0000 0000 0001' Check to see if set to '0000 0000 0000 0000 0001' Check to see if set to 'Periodic URA update'

RRC CONNECTION RELEASE (Step 3)

Only the message type is checked for this message.

RRC CONNECTION RELEASE COMPLETE (Step 4)

Only the message type is checked for this message.

8.3.2.9.5 Test requirement

After step 1 the UE shall transmit a URA UPDATE message on the uplink CCCH and set value "periodic URA update" into IE "Cell update cause".

After step 3 the UE shall transmit a RRC CONNECTION RELEASE COMPLETE message on the uplink DCCH and return to idle mode.

8.3.3. UTRAN Mobility Information

8.3.3.1 UTRAN Mobility Information: Success

8.3.3.1.1 Definition

8.3.3.1.2 Conformance requirement

This procedure is used by the network to assign a new RNTI identity to the UE. It is initiated by the UTRAN when it sends an UTRAN MOBILITY INFORMATION message, which includes a new C-RNTI and/or U-RNTI on the downlink DCCH. The UE starts to use the new identities and transmits an UTRAN MOBILITY INFORMATION CONFIRM message to the UTRAN on the uplink DCCH.

Reference

3GPP TS 25.331 clause 8.3.3

8.3.3.1.3 Test purpose

To confirm that the UE starts to use the new identities after it receives an UTRAN MOBILITY INFORMATION message from the SS. To confirm that the UE use the new U-RNTI identity to calculate the applicable paging occasions.

[Editor's note] In this test case, it is assumed that the paging occasion during connected states is determined using U-RNTI. From TS 25.304 Clause 8, this assumption cannot be confirmed. Further clarification is required regarding this issue.

8.3.3.1.4 Method of test

Initial Condition

System Simulator: 1cell

UE: CS-CELL_FACH_Initial (state 6-2) or PS-CELL_FACH_Initial (state 6-4) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE

Test Procedure

Initially, the UE is in the CELL_FACH state and it has been assigned a C-RNTI and U-RNTI. The SS transmits an UTRAN MOBILITY INFORMATION message which includes new C-RNTI and U-RNTI to the UE. Then the UE shall transmit an UTRAN MOBILITY INFORMATION CONFIRM message as confirmation and transits to CELL_PCH state. SS pages the UE by sending a PAGING TYPE 1 message and specifying the newly assigned U-RNTI identity in this message. The UE shall transmit a CELL UPDATE message to the SS on the uplink CCCH which includes the U-RNTI identical to that found in UTRAN MOBILITY INFORMATION message received in step 2. The CELL UPDATE message shall also contain IE "Cell update cause" with this IE set to "paging response". After the SS receives this message, it transmits a CELL UPDATE CONFIRM message which includes IE "DRX Indicator RRC State Indicator" set to value "No DRXCELL FACH" to the UE on the downlink DCCH. The UE shall return to CELL_FACH state.

Expected sequence

Step	Dire	ction	Message	Comment
	UE	SS		
1				The initial state of the UE is CELL_FACH state. UE has been allocated both C-RNTI and U-RNTI during RRC connection establishment phase.
2		←	UTRAN MOBILITY INFORMATION	Contains new C-RNTI and U-RNTI identities.
3		\rightarrow	UTRAN MOBILITY INFORMATION CONFIRM	
4		←	PAGING TYPE 1	SS pages the UE using the new U-RNTI allocated in step 2.
5		→	CELL UPDATE	UE shall trigger cell updating. The message shall indicate the same U-RNTI assigned in the UTRAN MOBILITY INFORMATION message in step 2.
6		←	CELL UPDATE CONFIRM	IE "DRX indicatorRRC State Indicator" is set to "No DRXCELL_FACH".

Specific Message Content

UTRAN MOBILITY INFORMATION (Step 2)

Use the same message sub-type as in Annex A, with the following exceptions:

Information Element	Value/remark
New U-RNTI	
- SRNC Identity	'0000 0000 0001'
- S-RNTI	'0101 0101 0101 0101 [']
New C-RNTI	'1010 1010 1010 1010'
DRX Indicator RRC State Indicator	DRX with cell updatingCELL_PCH

UTRAN MOBILITY INFORMATION CONFIRM (Step 3)

Only the message type IE is checked in this message.

PAGING TYPE 1 (Step 4)

Use the same message sub-type as in TS 34.108 Clause 9, with the following exceptions:

Information Element	Value/remark
Page Record List	
- Paging record	
- CHOICE Paging originator	UTRAN originator
- U-RNTI	_
- SRNC Identity	'0000 0000 0001'
- S-RNTI	'0101 0101 0101 0101 o101'
BCCH modification info	Not Present

CELL UPDATE (Step 5)

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to '0101 0101 0101 0101'
Integrity check info	Not checked
AM_RLC error indicator (for C-plane)	Check to see if set to 'FALSE'
AM_RLC error indicator (for U-plane)	Check to see if set to 'FALSE'
START List	Checked to see if the 'CN domain identity' and 'START'
	IEs are present for all CN domains supported by the UE
Cell Update Cause	Check to see if set to 'Paging Response'
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Measured results on RACH	Check to see if it is absent
Protocol error information	Check to see if it is absent

CELL UPDATE CONFIRM (Step 6)

Use the same message sub-type as in Annex A.

8.3.3.1.5 Test requirement

After step 2 the UE shall transmit an UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH. The MAC PDU carrying this message shall comprise either the new C-RNTI or U-RNTI allocated in the "UE-id" field of the MAC header.

After step 4 the UE shall transmit a CELL UPDATE message on the uplink CCCH with IE "Cell update cause" set to "paging response". The IE "U-RNTI" shall be identical to the IE "New RNTI" found in UTRAN MOBILITY INFORMATION message sent by the SS in step 2.

8.3.3.2 UTRAN Mobility Information: Failure (Invalid message reception)

8.3.3.2.1 Definition

8.3.3.2.2 Conformance Requirements

When the UE receives an UTRAN MOBILITY INFORMATION message, which contains an error in one of the mandatory IE, it shall transmit a UTRAN MOBILITY INFORMATION FAILURE message on the DCCH using AM RLC and set the value "protocol error" in the IE "failure cause". The IE "protocol error information" in this message shall also be set to an appropriate value. The UE shall not utilize any identities relayed in the erroneous message, and it shall resume normal operations.

8.3.3.2.3 Test Purpose

To confirm that the UE ignore the new connected mode identities conveyed in an erroneous UTRAN MOBILITY INFORMATION message. To confirm that the UE report this event to the UTRAN by sending UTRAN MOBILITY INFORMATION FAILURE message, stating the appropriate failure cause and information.

8.3.3.2.4 Method of test

Initial Conditions

System Simulator: 1 cell

UE: CS-CELL_FACH_Initial (state 6-2) or PS-CELL_FACH_Initial (state 6-4) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE

Test Procedure

The UE is brought to CELL_FACH state. SS transmits a UTRAN MOBILITY INFORMATION message to the UE on the DCCH using UM-RLC mode. In this message, the IE "DRx IndicatorRRC State Indicator" is set to one of the spare values. A new U-RNTI identity is also present in this message. The UE shall respond by transmitting the UTRAN MOBILITY INFORMATION FAILURE message, indicating "protocol error" in IE "failure cause" and also "Information element not comprehended" in IE "Protocol error information". After receiving the UTRAN MOBILITY INFORMATION FAILURE message, SS waits for a duration to allow T305 to expire. The UE shall transmit CELL UDPATE uppate message with the original U-RNTI identity assigned. SS complete this test by sending CELL UPDATE CONFIRM message to the UE on the downlink DCCH.

Expected Sequence

Step	Direction		Message	Comment
	UE	SS		
1				The initial state of the UE is CELL_FACH state.
2		←	UTRAN MOBILITY INFORMATION	Contains a new U-RNTI identity, but a spare value is used in the IE "DRx indicator RRC State Indicator"
3		→	UTRAN MOBILITY INFORMATION FAILURE	UE shall transmit this message to report the error in UTRAN MOBILITY INFORMATION message. It shall include the appropriate cause in the message.
4				SS waits for a period up to timer T305 to allow the UE to start performing a cell updating procedure.
5		→	CELL UPDATE	UE shall trigger periodic cell updating. The message shall not contain the U-RNTI given in the UTRAN MOBILITY INFORMATION message in step 2.
6		←	CELL UPDATE CONFIRM	

Specific Message Content

UTRAN MOBILITY INFORMATION (Step 2)

Use the same message sub-type as in Annex A, with the following exceptions:

Value/remark
0000 0000 0001B
0000 0000 0000 0000 00011B
Set to one of the spare value

UTRAN MOBILITY INFORMATION FAILURE (Step 3)

Information Element	Value/remark
Failure Cause	Check to see if set to 'Protocol error'
Protocol Error Information	Check to see if set to 'Information Element not
	comprehended'

CELL UPDATE (Step 5)

Information Element	Value/remark
U-RNTI	Shall be the same as the original U-RNTI allocated
- SRNC Identity	Check to see if set to '0000 0000 0001'B
- S-RNTI	Check to see if set to '0000 0000 0000 0000 0001'B
Integrity check info	Not checked
AM_RLC error indicator (for C-plane)	Check to see if set to 'FALSE'
AM_RLC error indicator (for U-plane)	Check to see if set to 'FALSE'
START List	Checked to see if the 'CN domain identity' and 'START'
	IEs are present for all CN domains supported by the UE
Cell update cause	Check to see if set to 'Periodic Cell Updating'
Protocol error indicator	Check to see if it is absent or set to 'FALSE'
Measured results on RACH	Check to see if it is absent
Protocol error information	Check to see if it is absent

CELL UPDATE CONFIRM (Step 6)

Use the same message sub-type as in Annex A.

8.3.3.2.5 Test Requirement

After step 2 the UE shall transmit UTRAN MOBILITY INFORMATION FAILURE message, indicating the value "protocol error" in IE "failure cause" and also "information element not comprehended" in IE "protocol error information".

After step 4 the UE shall initiate a periodic cell updating procedure by transmitting CELL UPDATE message on the CCCH. In this message, the U-RNTI identity shall be set to the same value as assigned during the RRC connection establishment procedure.

8.3.4 Active set update in soft handover

8.3.4.1 Active set update in soft handover: Radio Link addition

8.3.4.1.1 Definition

8.3.4.1.2 Conformance requirement

Radio link addition is triggered in the network's RRC layer. The RRC entity in the network first configures the new radio link. Transmission and reception then begin immediately. This procedure is to update the active set of the connection between the UE and UTRAN. The UTRAN then transmits an ACTIVE SET UPDATE message to the UE. The UE configures layer 1 to begin reception for the additional radio link. After the UE receives confirmation from the physical layer in the UE, an ACTIVE SET UPDATE COMPLETE message is sent to the UTRAN.

Reference

3GPP TS 25.331 clause 8.3.4

8.3.4.1.3 Test purpose

To confirm that the UE continues to communicate with the SS on both the additional radio link and an already existing radio link after the radio link addition.

8.3.4.1.4 Method of test

Initial Condition

System Simulator: 2cells - Cell 1 is active, Cell 2 is active

UE: CS-CELL_DCH_Initial (state 6-1) or PS-CELL_DCH_Initial (state 6-3) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE

Test Procedure

Initially, the UE establishes a radio access bearer in the CELL_DCH state in cell 1. The SS begins to configure the new radio link to be added from cell 2. Then the SS transmits to the UE an ACTIVE SET UPDATE message in cell 1 on DCCH using AM RLC which includes the IE "Radio Link Addition Information" (e.g. Downlink DPCH information and other optional parameters relevant for the additional radio links with Primary CPICH info used for the reference ID). When the UE receives this message, the UE shall configure layer 1 to begin reception without affecting the current uplink and downlink activities of existing radio links. After the UE confirms the synchronization with the new radio link from cell 2, the UE shall transmit an ACTIVE SET UPDATE COMPLETE message to the SS on the uplink DCCH using AM RLC. The UE continues to communicate with the SS on the both radio links. To test this condition, SS ceases the operations of all uplink and downlink DPCH from cell 1. SS shall observe that the data communication for both DCCH and DTCH channels continue as per normal using cell 2, as if cell 1 is still operational.

Expected sequence

Step	Dire	ction	Message	Comment
_	UE	SS	_	
1				The UE is brought to CELL_DCH state in cell 1, after the successful establishment of a radio access bearer service.
2				The SS configures an additional radio link in the downlink direction from cell 2.
3		(ACTIVE SET UPDATE	SS transmits this message in cell 1 on downlink DCCH using AM RLC. The message includes IE "Radio Link Addition Information". (e.g. Downlink DPCH information and other optional parameters relevant for the additional radio links with Primary CPICH info used for the reference ID in cell 2)
4		→	ACTIVE SET UPDATE COMPLETE	The UE shall configure a new radio link to cell 2, without interfering with existing connections on the radio link in cell 1. SS ceases all Tx and Rx activities in cell 1. But it shall be able to communicate with UE through cell 2.

Specific Message Content

ACTIVE SET UPDATE

The message to be used in this test is defined in the default message content clause, with the following exceptions:

Information Element	Value/remark
Radio link addition information	
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as assigned for cell 2
- Downlink DPCH info for each RL	
 Primary CPICH usage for channel estimation 	P-CPICH can be used.
- DPCH frame offset	0 chips
- Secondary CPICH info	Not Present
- DL channelisation code	This IE is repeated for all existing downlink DPCHs
	allocated to the UE
- Secondary scrambling code	Not Present
- CHOICE Spreading factor	512
- Code Number	For each DPCH, assign the same code number in the
	current code given in cell 1.
- Scrambling code change	Not Present
- TPC Combination Index	Not Present
- SSDT Cell Identity	Not Present
 Close loop timing adjustment mode 	Not Present
- TFCI Combining Indicator	Not Present
- SCCPCH information for FACH	Not Present
Radio link removal information	Not Present

8.3.4.1.5 Test requirement

After step 3 the UE shall configures a new radio link to cell.2, with the connection on the old radio link in cell 1 remaining operational and unaffected. It shall transmit an ACTIVE SET UPDATE COMPLETE message on the uplink DCCH using AM RLC to acknowledge the completion of the active set additional procedure.

After step 4 the SS shall continue to communicate with the UE using the radio links added to the UE from cell 2.

8.3.4.2 Active set update in soft handover: Radio Link removal

8.3.4.2.1 Definition

8.3.4.2.2 Conformance requirement

This procedure is to update the active set of the connection between the UE and the UTRAN after the UTRAN has commanded a removal of a radio link from the current active set. The UTRAN RRC transmits an ACTIVE SET UPDATE message to the UE RRC. The UE RRC requests UE L1 to terminate transmission and reception of the radio link to be removed. The UE shall continue to communicate normally with the UTRAN using the new active set, without losing the connection link. After this the UE acknowledges the radio link removal by sending an ACTIVE SET UPDATE COMPLETE message to the UTRAN on DCCH using AM RLC.

Reference

3GPP TS 25.331 clause 8.3.4

8.3.4.2.3 Test purpose

To confirm that the UE continues to communicate with the SS on the remaining radio link after radio link removal on the active set.

8.3.4.2.4 Method of test

Initial Condition

System Simulator: 2 cells - both Cell 1 and Cell 2 are active

UE: CS-CELL_DCH_Initial (state 6-1) or PS-CELL_DCH_Initial (state 6-3) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE

Test Procedure

At the start of the test, the UE establishes a radio access bearer service in the CELL_DCH state in cell 1. This is followed by a radio link addition procedure in cell 2. SS then transmits an ACTIVE SET UPDATE message, which includes IE "Radio Link Removal Information" and specifying the P-CPICH information of the cell to be removed. When the UE receives this message, the UE RRC entity shall request UE L1 entity to terminate transmission and reception of the radio link from cell 1. Then the UE transmits an ACTIVE SET UPDATE COMPLETE message to the SS on the uplink DCCH using AM RLC. The UE shall continue to communicate with the SS on the remained radio link in cell 2.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1			The UE is in the CELL DCH state in cell 1. SS executes test 8. 3.4.1, and the UE shall update the active set to contain cell 1 and cell 2 after the radio link addition procedure.
2	←	ACTIVE SET UPDATE	The SS transmits this message on downlink DCCH using AM RLC which includes IE "Radio Link Removal Information".
3	→	ACTIVE SET UPDATE COMPLETE	The UE shall remove the radio link associated with cell 1.
4			The SS stops transmission on the downlink direction from cell 1 and the UE shall continue to communicate on the remaining radio link in cell 2.

Specific Message Contents

ACTIVE SET UPDATE

The message to be used in this test is the same as the message sub-type found in Annex A, with the following exceptions:

Information Element	Value/remark
Radio link addition information	Not Present
Radio link removal information - Primary CPICH info	1 radio link to be removed
- Primary scrambling code	Set to the same P-CPICH scrambling code assigned for cell 1

8.3.4.2.5 Test requirement

After step 2 the UE shall remove the radio link from cell 1 and it shall transmit an ACTIVE SET UPDATE COMPLETE message on the uplink DCCH using AM RLC.

After step 3 the UE shall continue to communicate on the remaining radio link from cell 2.

8.3.4.3 Active set update in soft handover: Combined radio link addition and removal (active set is not full)

8.3.4.3.1 Definition

8.3.4.3.2 Conformance requirement

When radio links are to be replaced, the UTRAN RRC first configures the UTRAN L1 to activate the radio link(s) that are being added. The UTRAN RRC then transmits an ACTIVE SET UPDATE message to the UE RRC, which shall configure the UE L1 to terminate transmission and reception on the removed radio link(s) and begin transmission and reception on the added radio link(s). At the completion of the reconfiguration of radio links, the UE shall acknowledge the replacement with an ACTIVE SET UPDATE COMPLETE message.

Reference

3GPP TS 25.331 clause 8.3.4

8.3.4.3.3 Test purpose

To confirm that the UE continues to communicate with the SS on the added radio link and removes radio link which exists prior to the execution of active set update procedure.

8.3.4.3.4 Method of test

Initial Condition

System Simulator: 2 cells- Both Cell 1 and Cell 2 are active

UE: CS-CELL_DCH_Initial (state 6-1) or PS-CELL_DCH_Initial (state 6-3) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE [Active set is not full.]

Test Procedure

The UE establishes a radio access bearer in the CELL_DCH state in cell 1. SS begin to configure the new radio link in cell 2. Then the SS transmits to the UE an ACTIVE SET UPDATE message in cell 1 on DCCH using AM RLC. The message includes IE "Radio Link Addition Information" and IE "Radio Link Removal Information", indicating the removal of cell 1 and addition of cell 2 into the active set. When the UE receives this message, the UE RRC shall terminate the transmission and reception of the removed radio link in cell 1 and then configures layer 1 to begin transmission and reception in cell 2. After the UE received confirmations from the physical layer regarding the update of active set, it transmits an ACTIVE SET UPDATE COMPLETE message on the uplink DCCH to the SS. The UE shall continue to communicate with the SS on the added radio link in cell 2. When SS receives ACTIVE SET UPDATE COMPLETE message, it verifies that the UE has ceased any uplink transmission in cell 1.

Expected sequence

Step	Dire	ction	Message	Comment
	UE	SS		
1				The UE is in the CELL_DCH state in cell 1
2				The SS configures an additional radio link in cell 2, starting the transmission and reception of data in cell 2. clause
3		+	ACTIVE SET UPDATE	The SS transmit this message on downlink DCCH using AM RLC which includes IE "Radio Link Addition Information" for cell 2 and IE "Radio Link Removal Information" for cell 1.
4		→	ACTIVE SET UPDATE COMPLETE	The UE shall configure a new radio link in cell 2 and removes the old radio link in cell 1.
5				The SS removes the radio link from cell 1 and the UE shall continue to communicate on the added radio link in cell 2, and not transmit any data in cell 1.

Specific Message Content

ACTIVE SET UPDATE

The message to be used in this test is defined in the default message content clause, with the following exceptions:

Information Element	Value/remark
Radio link addition information	
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as assigned for cell 2
- Downlink DPCH info for each RL	
 Primary CPICH usage for channel estimation 	P-CPICH can be used.
- DPCH frame offset	0 chips
- Secondary CPICH info	Not Present
- DL channelisation code	This IE is repeated for all existing downlink DPCHs
	allocated to the UE
- Secondary scrambling code	Not Present
- CHOICE Spreading factor	512
- Code Number	For each DPCH, assign the same code number in the
	current code given in cell 2.
- Scrambling code change	Not Present
- TPC Combination Index	Not Present
- SSDT Cell Identity	Not Present
 Close loop timing adjustment mode 	Not Present
- TFCI Combining Indicator	Not Present
- SCCPCH information for FACH	Not Present
Radio link removal information	
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code assigned as for cell 1

8.3.4.3.5 Test requirement

After step 3 the UE shall remove the radio link in cell 1 and add the radio link in cell 2. Then the UE shall transmit an ACTIVE SET UPDATE COMPLETE message the uplink DCCH.

After step 4 the UE shall continue to communicate on the added radio link in cell 2. SS monitors the uplink direction to confirm that no data are designated for reception in cell 1.

8.3.4.4 Active set update in soft handover: Unsupported Invalid Configuration in the UE

8.3.4.4.1 Definition

8.3.4.4.2 Conformance requirement

If the UTRAN attempts to remove a radio link that is not currently present in the UE's active set, the UE transmits an ACTIVE SET UPDATE FAILURE message on the DCCH using AM RLC and maintain its current communication status with the radio links.

Reference

3GPP TS 25.331 clause 8.3.4

8.3.4.4.3 Test purpose

To confirm that the UE transmits an ACTIVE SET UPDATE FAILURE message on the DCCH using AM RLC, following the reception of a message specifying the removal of a radio link unknown to the UE.

8.3.4.4.4 Method of test

Initial Condition

System Simulator: 2 cells - Cell 1 is active, Cell 2 is active.

UE: CS-CELL_DCH_Initial (state 6-1) or PS-CELL_DCH_Initial (state 6-3) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE.

Test Procedure

The UE establishes a radio access bearer in the CELL_DCH state in cell 1. SS requests for a radio link addition by executing the steps described in test case 8.3.4.1. The UE shall then include cell 2 into its active set and establish the transmission and reception capabilities related to cell 2. SS then transmits an ACTIVE SET UPDATE message in cell 1 on DCCH using AM RLC which includes IE "Radio Link Removal Information" This IE indicates that a cell with unknown P-CPICH scrambling code be removed from the active set. When the UE receives this message, it transmits an ACTIVE SET UPDATE FAILURE message which is set to "Invalid configuration unacceptable" in IE "failure cause" on the uplink DCCH using AM RLC to the SS, and continues to communicate on the existing radio links in cell 1 and cell 2.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1				The UE is in the CELL_DCH state in cell 1.
2				SS commands the UE to perform a radio link addition procedure by executing the steps in test case 8.3.4.1. The UE shall respond accordingly. Both cell 1 and cell 2 should be found in the active set maintained by the UE.
3		+	ACTIVE SET UPDATE	The SS transmits this message on downlink DCCH using AM RLC which includes IE "Radio Link Removal Information". This content of this IE indicates an unknown cell.
4		→	ACTIVE SET UPDATE FAILURE	The message shall state "Invalid configuration unacceptable" in IE "failure cause". UE shall continue to communicate normally with both cells

Specific Message Contents

ACTIVE SET UPDATE (Step 3)

The message to be used in this test is defined in the default message content clause, with the following exceptions:

Information Element	Value/remark
Radio link addition information	Not Present
Radio link removal information - Primary CPICH info	1 radio link to be removed
- Primary scrambling code	Set to an unknown scrambling code not assigned to any cells.

ACTIVE SET UPDATE FAILURE (Step 4)

Information Element	Value/remark
Integrity check info	Not Checked
Failure cause	Check to see if it's set to 'Invalid configuration
	unacceptable'

8.3.4.4.5 Test requirement

After step 3 the UE shall transmit an ACTIVE SET UPDATE FAILURE message, setting "<u>Invalid</u> configuration unacceptable" in IE "failure cause" and sent on the uplink DCCH using AM RLC.

After step 4 the UE shall continue to communicate on the radio links for both cell 1 and cell 2.

8.3.4.5 Active set update in soft handover: Combined radio link addition and removal (active set is full)

8.3.4.5.1 Definition

8.3.4.5.2 Conformance requirement

When the UE active set is full, the UE shall first remove the old radio link and then add the new radio link, after it receives an ACTIVE SET UPDATE message for the combined radio link addition and removal.

Reference

3GPP TS 25.331 clause 8.3.4

8.3.4.5.3 Test purpose

To confirm that the UE removes one of existing radio links, which is indicated in an ACTIVE SET UPDATE message and continues to communicate on the added radio link.

8.3.4.5.4 Method of test

Initial Condition

System Simulator: 3 cells - Cell 1, Cell 2, and Cell 3 are all active

UE: CS-CELL_DCH_Initial (state 6-1) or PS-CELL_DCH_Initial (state 6-3) in cell 1 and cell 2 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE (The assumed maximum number for active set is 2.)

[Editor's Note] The maximum number of radio link (i.e. MaxRL) specified in CR328 of TS 25.331 is 8. However, if the UE capability is more inferior in this aspect, can the assumption above still stands?

Test Procedure

The UE establishes a radio access bearer in the CELL_DCH state in cell 1 and cell 2. The SS configures the new radio link in cell 3 and sends an ACTIVE SET UPDATE message on DCCH using AM. This message includes IE "Radio Link Addition Information" indicating cell 3 to be added into the active set, and IE "Radio Link Removal Information" indicating the removal of cell 1 from the active set. When the UE receives this message, it shall not report a failure but firstly removes the indicated radio link and then adds the new radio link. Then the UE transmits an ACTIVE SET UPDATE COMPLETE message on the DCCH using AM RLC to the SS and continues to communicate with the SS on the added radio link and the remaining old radio link.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1				The UE is in the CELL_DCH state in cell 1 and cell 2.
2				The SS configures an additional radio link in for cell 3, and starts reception and transmission using cell 3.
3		←	ACTIVE SET UPDATE	The SS transmit this message on downlink DCCH using AM RLC which includes IE "Radio Link Addition Information" and IE "Radio Link Removal Information". The contents of the IE dictate the addition of cell 3 into the active set and removal of cell 1 from it.
4		→	ACTIVE SET UPDATE COMPLETE	The UE shall configure a new radio link in cell 3 and removes the old radio link in cell 1.
5				The SS removes the radio link in cell 1. The UE shall continue to communicate on the added radio link in cell 3 and also the existing radio link in cell 2.

Specific Message Content

ACTIVE SET UPDATE

The message to be used in this test case isidentical to the same message sub-type found in Annex A, with the following exceptions:

Information Element	Value/remark	
Radio link addition information		
- Primary CPICH Info		
- Primary Scrambling Code	Set to same code as assigned for cell 3	
- Downlink DPCH info for each RL		
 Primary CPICH usage for channel estimation 	P-CPICH can be used.	
- DPCH frame offset	0 chips	
- Secondary CPICH info	Not Present	
- DL channelisation code	This IE is repeated for all existing downlink DPCHs	
	allocated to the UE	
 Secondary scrambling code 	Not Present	
- CHOICE Spreading factor	512	
- Code Number	For each DPCH, assign the same code number in the	
	current code given in cell3.	
- Scrambling code change	Not Present	
- TPC Combination Index	Not Present	
- SSDT Cell Identity	Not Present	
- Close loop timing adjustment mode	Not Present	
- TFCI Combining Indicator	Not Present	
- SCCPCH information for FACH	Not Present	
Radio link removal information		
- Primary CPICH Info		
- Primary Scrambling Code	Set to same code assigned as for cell 1	

8.3.4.5.5 Test requirement

After step 3 the UE shall remove the radio link in cell 1 and add the radio link in cell 3. Then the UE shall transmit an ACTIVE SET UPDATE COMPLETE message on the uplink DCCH.

After step 4 the UE shall continue to communicate on the added radio link in cell 3 and on the existing old radio link in cell 2. It shall cease all transmission to cell 1.

8.3.4.6 Active set update in soft handover: Subsequent reception of ACTIVE SET UPDATE message / Incompatible simultaneous reconfiguration

8.3.4.6.1 Definition

8.3.4.6.2 Conformance Requirements

The UE shall ignore a subsequent ACTIVE SET UPDATE message, while it is still processing an existing active set update procedure. It shall continue to configure itself in accordance to the first ACTIVE SET UPDATE message received. When encountering a "simultaneous reconfiguration" situation, the UE shall transmit a ACTIVE SET FAILURE message on the DCCH using AM RLC with value "incompatible simultaneous reconfiguration" set in IE "failure cause". Then the UE shall continue to execute the ordered reconfiguration (for example due to a radio bearer reconfiguration) as if the ACTIVE SET UPDATE message has not been received.

Reference

3GPP TS 25.331 clause 8.3.4

8.3.4.6.3 Test Purpose

To confirm that the UE continues to execute the prior active set update request, when it receives a subsequent ACTIVE SET UPDATE messages before it has completely executed the first update procedure. To confirm that the UE transmits an ACTIVE SET UPDATE FAILURE message to report the detection of a "incompatible simultaneous reconfiguration" condition, when an ACTIVE SET UPDATE message was received before the UE can complete an ongoing radio bearer reconfiguration procedure.

8.3.4.6.4 Method of test

Initial Condition

System Simulator: 2 cells - both cell 1 and cell 2 are active

UE: CS CELL_DCH_Initial (state 6 1) or PS CELL_DCH_Initial (state 6 3) in cell 1 and cell 2 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE

Test Procedure

The UE establishes a radio access bearer in CELL_DCH state in cell 1 and cell 2. SS transmits an ACTIVE SET UPDATE message to request for the removal of cell 1 from the active list. When the UE sends an acknowledgement for this message from its RLC entity, SS immediately transmits a second ACTIVE SET UPDATE message, which specifies cell 1 to be added into the active list. SS verifies that the UE ceases transmission on the radio link associated with cell 1 when the activation time indicated in the first ACTIVE SET UPDATE message is reached. Next, SS sends a RADIO BEARER RELEASE message using AM RLC on the DCCH. In this message, SS requests the release of the radio access bearer. When RLC acknowledgement has been received from the UE, SS immediately sends an ACTIVE SET UPDATE message. In this message, SS commands the UE to add cell 1 into its active list with the activation time set to "now". The UE shall react by transmitting ACTIVE SET UPDATE FAILURE message on the DCCH using AM RLC mode. In this message, the IE "failure cause" shall be set to "incompatible simultaneous reconfiguration". When the activation time stated in RADIO BEARER RELEASE message has elapsed, the UE shall transmit the RADIO BEARER RELEASE complete message to inform that the assigned radio access bearer is release. When SS receives this message, it verifies that UE continues to communicate with the SS on the radio link associated with cell 2 only, and that no user data are exchanged on the previously available radio access bearer.

Expected Sequence

Γ	Step	Direction	Message	Comment
		UE SS		
	4			The UE is CELL_DCH state in both cell 1 and cell 2.
	2	- ←	ACTIVE SET UPDATE	The SS transmits this
				message on downlink DCCH
				using AM RLC, requesting for
				cell 1 to be removed from the
-	3	_ ←	ACTIVE SET UPDATE	active set. Immediately after UE
	÷	_~	NOTIVE SET UPDATE 	acknowledges the message in
				step 2, SS transmits this
				message. This message
				specifies that cell 1 be added
				into the active set.
	4			SS waits until the activation
				time stated in step 2 has
				elapsed, and verifies that the UE stops transmitting on the
				radio link of cell 1.
				radio in ik of cen 1.
ŀ	5	_ →	ACTIVE SET UPDATE COMPLETE	UE shall transmit this
	Ū	-	7.6	message to signal the
				completion of the active set
				update procedure triggered in
				step 2.
-	6	_←	RADIO BEARER RELEASE	SS checks that UE stops all
				uplink activities on the radio
				link associated with cell 1. SS
				requests that the radio access bearer allocated to the UE be
				released.
				Tolousou.
	7	- ←	ACTIVE SET UPDATE	SS indicates that the UE shall
				reinstate cell 1 into its active
				set, immediately after RLC
				acknowledgement is received
-			ACTIVE CET LIDDATE FAILURE	for the message sent in step 6.
	8	_→	ACTIVE SET UPDATE FAILURE	In IE "failure cause", the reason "incompatible
				simultaneous reconfiguration"
				shall be stated.
	9	— >	RADIO BEARER RELEASE COMPLETE	The UE shall send this
				message when the activation
				time specified in step 6 is reached. Upon reception of
				this message, SS verifies that
				there is no more uplink user
				traffic on the radio access
				bearer.
<u></u>				·

Specific Message Contents

ACTIVE SET UPDATE (Step 2)

Use the default message for this type found in Annex A, with the following exceptions:

Information Element	Value/remark
Radio link addition information	Not Present.
Radio link removal information	
- Primary CPICH info	
- Primary scrambling code	Set to the P-CPICH scrambling code assigned to cell 1.

ACTIVE SET UPDATE (Step 3)

Use the default message for this type found in Annex A, with the following exceptions

Information Element	Value/remark
Radio link addition information	
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as assigned for cell 1
- Downlink DPCH info for each RL	
	P-CPICH can be used.
	0 chips
- Secondary CPICH info	Not Present
	This IE is repeated for all existing downlink DPCHs
	allocated to the UE
- Secondary scrambling code	Not Present
	512
- Code Number	For each DPCH, assign the same code number in the
	current code given in cell 1.
- Scrambling code change	Not Present
- TPC Combination Index	Not Present
- SSDT Cell Identity	Not Present
- Close loop timing adjustment mode	Not Present
TFCI Combining Indicator	Not Present
- SCCPCH information for FACH	Not Present
Radio link removal information	Not Present

ACTIVE SET UPDATE COMPLETE (Step 5)

Only the message type for this message is checked.

RADIO BEARER RELEASE (Step 6)

Use the same message sub-type titled "AM or UM (The others of speech in CS)" found in Annex A.

ACTIVE SET UPDATE (Step 7)

Use the same message as in that for step 3, with the following exception:

Information Element	Value/remark
Activation Time	Not Present – use default

ACTIVE SET UPDATE FAILURE (Step 8)

Information Element	Value/remark	
Failure Cause	Check to see if set to "Incompatible simultaneous	
	reconfiguration"	

RADIO BEARER RELEASE COMPLETE (Step 9)

Information Element	Value/remark
Integrity check Info	Not checked
Uplink integrity protection activation info	Not checked
Radio bearer uplink ciphering activation time info	Not checked

8.3.4.6.5 Test Requirement

After step 5 the UE shall ignore the second ACTIVE SET UPDATE message received, terminate the radio link in relation to cell 1, and transmit ACTIVE SET UPDATE COMPLETE using AM RLC on the uplink DCCH of cell 2.

After step 7 the UE shall report the "incompatible simultaneous reconfiguration" error by transmitting ACTIVE SET UPDATE FAILURE message on the DCCH.

After step 8 the UE shall send RADIO BEARER RELEASE COMPLETE message to cell 2 on the uplink DCCH, using AM RLC mode. The UE shall stop all transmissions of user traffic on the radio access bearer assigned, Void

8.3.4.7 Active set update in soft handover: Invalid Message Reception

8.3.4.7.1 Definition

8.3.4.7.2 Conformance Requirement

The UE shall keep its old configuration when the UE receives an ACTIVE SET UPDATE message, which omits a conditional IE. It shall transmit a ACTIVE SET UPDATE FAILURE message which set value "protocol error" in IE "failure cause" and also value "Conditional information element error" in IE "Protocol error cause".

Reference

3GPP TS 25.331 clause 8.3.4

8.3.4.7.3 Test Purpose

To confirm that the UE retains its active set list when it receives an ACTIVE SET UPDATE message, with a conditional IE missing in the message.

8.3.4.7.4 Method of test

Initial Condition

System Simulator: 2 cells – both cell 1 and cell 2 are active.

UE: CS-CELL_DCH_Initial (state 6-1) or PS-CELL_DCH_Initial (state 6-3) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE (Integrity protection algorithm is not applied at the start of test)

Test Procedure

The UE establishes a radio access bearer in CELL_DCH in cell 1. SS requests that cell 2 be added into the active set by performing the steps described in test cases 8.3.4.1. The UE shall react accordingly and incorporate cell 2 into its active set. SS transmits an ACTIVE SET UPDATE message , with both IE "Integrity check info" and IE "Integrity protection mode info" present in the message. This message also commands the starting of integrity mode protection. However, the IE "integrity protection initialisation number" is omitted. The UE shall detect that it has received an invalid message. It shall then send an ACTIVE SET UPDATE FAILURE message, stating the reason "Conditional information element error" in the IE "Protocol error information". The UE shall not remove cell 1 from its current active set.

Expected Sequence

Step	Direction		Message	Comment
_	UE	SS	_	
1				The UE is CELL_DCH state in cell 1.
2				SS executes the steps in test case 8.3.4.1. The UE shall add cell 2 into its active set.
3		←	ACTIVE SET UPDATE	The SS transmits this message on downlink DCCH using AM RLC which includes IE "Integrity check info" and IE "Integrity protection mode info". This message indicates that integrity mode protection be started but omit the IE "integrity protection initialisation number". The message also specifies that cell 1 be removed from the active set.
4		→	ACTIVE SET UPDATE FAILURE	The message shall state "conditional information element error" in IE "protocol error information". UE shall continue to communicate normally with both cells.

Specific Message Contents

ACTIVE SET UPDATE (Step 3)

The message to be used in this test is defined in the default message content clause, with the following exceptions:

Information Element	Value/remark
Integrity Check Info	
- Message authentication code	Set to an arbitrary 32-bits string
- RRC Message sequence number	Set to an arbitrary integer between 0 and 15
Integrity Protection Mode Info	
- Integrity protection mode command	Start
- Downlink integration protection activation info	Not Present
- Integrity protection algorithm	Standard UMTS Integrity Algorithm UIA1
- Integrity protection initialisation number	Not Present
Radio link addition information	Not Present
Radio link removal information	
- Primary CPICH info	
- Primary scrambling code	Set to the P-CPICH scrambling code assigned to cell 1.

ACTIVE SET UPDATE FAILURE (Step 4)

Information Element	Value/remark
Protocol Error Information	
- Protocol Error Cause	Check to see if it's set to 'Conditional information
	element error'

8.3.4.7.5 Test Requirement

After step 3 the UE shall report a protocol error by transmitting the ACTIVE SET UPDATE FAILURE message on the DCCH. In this message, the value "Conditional information element error" shall be set in IE "Protocol Error Information". The UE shall continue to communicate normally with the SS using cell 1 and cell 2.

8.3.5 Hard Handover

[Editor's note: This test is included in the "Physical channel reconfiguration", "Radio bearer establishment", "Radio bearer reconfiguration", "Radio bearer release" and "Transport channel reconfiguration".]

8.3.6 Inter-system hard handover from GSM to UTRAN

Clauses 8.3.6 contains test procedures to be used for executing Inter-system Handover from GSM to UTRAN tests. Table 8.3.6-1 contains a summary of the different combinations of parameters being tested, together with a reference to the appropriate generic test procedure. If a test uses a parameter which the UE under test does not support, the test shall be skipped. Test cases in this clause are applicable only to the UE supporting both UTRAN and GSM. The test USIM shall support service 27 to carry out these test cases.

Table 8.3.6-1

From	То	State of call	Ref. clause	Exec counter	Remark
GSM FR	UTRAN AMR (conversational/speech/ ULuplink:12.2 DL:12.2 kbps/CS RAB + ULuplink:3.4 DL3.4 kbps SRBS)	U10	8.3.6.1	1	call active state
GSM EFR	UTRAN AMR (conversational/speech/ ULuplink:12.2 DL:12.2 kbps/CS RAB + ULuplink:3.4 DL3.4 kbps SRBS)	U10	8.3.6.1	2	call active state
GSM AMR	UTRAN AMR (conversational/speech/ ULuplink:12.2 DL:12.2 kbps/CS RAB + ULuplink:3.4 DL3.4 kbps SRBS)	U10	8.3.6.1	3	call active state
GSM HR	UTRAN AMR (conversational/speech/ ULuplink:12.2 DL:12.2 kbps/CS RAB + ULuplink:3.4 DL3.4 kbps SRBS)	U10	8.3.6.1	4	call active state
GSM 14.4 kbps CS data	UTRAN (Streaming/unknown/ ULuplink:14.4 DL:14.4 kbps/CS RAB + ULuplink:3.4 DL:3.4 kbps SRBS)	U10	8.3.6.2	1	same data rate
GSM 28.8 kbps CS data	UTRAN (Streaming/unknown/ ULuplink:28.8 DL:28.8 kbps/CS RAB + ULuplink:3.4 DL:3.4 kbps SRBS)	U10	8.3.6.2	2	same data rate
GSM 57.6 kbps CS data	UTRAN (Streaming/unknown/ ULuplink:57.6 DL:57.6 kbps/CS RAB + ULuplink:3.4 DL:3.4 kbps SRBS)	U10	8.3.6.2	3	same data rate
GSM 14.4 kbps CS data	UTRAN (Streaming/unknown/ ULuplink:28.8 DL:28.8 kbps/CS RAB + ULuplink:3.4 DL:3.4 kbps SRBS)	U10	8.3.6.3	1	data rate upgrading
GSM 14.4 kbps CS data	UTRAN (Streaming/unknown/ ULuplink:57.6 DL:57.6 kbps/CS RAB + ULuplink:3.4 DL:3.4 kbps SRBS)	U10	8.3.6.3	2	data rate upgrading
GSM 28.8 kbps CS data	UTRAN (Streaming/unknown/ ULuplink:57.6 DL:57.6 kbps/CS RAB + ULuplink:3.4 DL:3.4 kbps SRBS)	U10	8.3.6.3	3	data rate upgrading
GSM FR	UTRAN AMR (conversational/speech/ ULuplink:12.2 DL:12.2 kbps/CS RAB + ULuplink:3.4 DL3.4 kbps SRBS)	U1	8.3.6.4	1	during call establishment
GSM FR	UTRAN AMR (conversational/speech/ ULuplink:12.2 DL:12.2 kbps/CS RAB + ULuplink:3.4 DL3.4 kbps SRBS)	U10	8.3.6.5	1	blind handover
GSM FR	UTRAN AMR (conversational/speech/ ULuplink:12.2 DL:12.2 kbps/CS RAB +	U10	8.3.6.6	1	failure case

ULuplink:3.4 DL3.4 kbps SRBS)		

8.3.6.1 Inter system handover to UTRAN/From GSM/Speech/Success

8.3.6.1.1 Definition

8.3.6.1.2 Conformance requirement

The UE shall be able to receive a HANDOVER TO UTRAN COMMAND message from GSM and perform an intersystem handover, even if no prior UE measurements have been performed on the target UTRAN cell and/or frequency.

If the UE succeeds to establish the connection to UTRAN, it shall transmit a HANDOVER TO UTRAN COMPLETE message on the uplink DCCH.

Reference(s)

TS 25.331 Clause 8.3.6.

TS 04.18 Clause 3.4.4a.

8.3.6.1.3 Test purpose

To test that UE supporting both GSM and UTRAN handovers to the indicated channel in the UTRAN target cell when it is in the speech call active state in the GSM serving cell and receives a HANDOVER TO UTRAN COMMAND.

8.3.6.1.4 Method of test

Initial conditions

System Simulator: 2 cells - Cell 1 is GSM, Cell 2 is UTRAN. GSM 11.10-1 section 26.6.5.1 shall be referenced for the default parameters of cell 1.

UE: CC State U10 in cell 1

Related ICS/IXIT statement(s)

UE supports both GSM and UTRAN Radio Access Technologies,

UE supports UTRAN AMR,

UE supports GSM ARM,

UE supports GSM EFR,

UE supports GSM HR,

UE supports GSM-P, GSM-E, GSM-DCS, GSM-450, GSM-480.

Foreseen final state of the UE

The UE is in CC state U10 on cell 2.

Test Procedure

The SS starts the GSM cell and UTRAN cell with cell selection conditions in favour of GSM cell, the UE selects the GSM cell for camping on. In UTRAN cell SIB16 is broadcast and contains the pre-configuration for conversationa/speech/ $\frac{ULuplink}{ULuplink}$:12.2 DL:12.2 kbps/CS RAB + $\frac{ULuplink}{ULuplink}$:3.4 DL3.4 kbps SRBS. After UE received and stored the SIB16, the SS brings the UE into the call active state (CC state U10) with FR speech call (for execution counter M = 1). The SS configures the dedicated channel corresponding to the pre-configuration in UTRAN cell, then sends HANDOVER TO UTRAN COMMAND indicating the dedicated channel of the target cell to the UE through the GSM serving cell. After the UE receives the command it shall configure itself accordingly and switch to the dedicated

channel of UTRAN cell. The SS checks whether the handover is performed by checking that the UE transmits HANDOVER TO UTRAN COMPLETE to the SS through DCCH of the UTRAN cell.

Depending on the PIXIT parameters, the above procedure is executed maximum four times, each time for different initial conditions:

- If the UE supports GSM FR, the procedure is executed for execution counter M = 1;
 - If the UE supports GSM EFR, the procedure is executed for execution counter M = 2;
 - If the UE supports GSM AMR, the procedure is executed for execution counter M = 3;
 - If the UE supports GSM HR, the procedure is executed for execution counter M = 4.

Expected sequence

This sequence is performed for a maximum execution counter M = 1, 2, 3, 4, depending on the PIXIT parameters.

Step	Direction	Message	Comments
	UE SS		
1	SS		The SS configures GSM and UTRAN cells, UE camps on GSM cell and received SIB16 from UTRAN cell.
2	UE		The SS bring the UE into GSM U10 state in cell 1 and for M = 1: the UE is in GSM FR speech call; for M = 2: the UE is in GSM EFR speech call; for M = 3: the UE is in GSM AMR speech call; for M = 4: the UE is in GSM HR speech call.
3	SS		The SS configures the dedicated channel with the configuration: conversational/speech/ <u>ULuplink</u> :12.2 DL:12.2 kbps/CS RAB + <u>ULuplink</u> :3.4 DL3.4 kbps SRBs in UTRAN cell.
4	+	HANDOVER TO UTRAN COMMAND	Send on cell 1 (GSM cell)
5	UE		The UE accepts the handover command and configures its lower layers using the parameters contained in the HANDOVER TO UTRAN COMMAND
6	SS		The SS waits for uplink physical channel in synchronization
7	→	HANDOVER TO UTRAN COMPLETE	The SS receives this message on DCCH of cell 2 (UTRAN cell). It implies that the down link physical channel has synchronised with UTRAN.

Content of "SysInfoType16"

Information Element	Value/remark
re-EstablishmentTimer T315	30 s
PredefinedRB_Configuration	
- SRB information list	
- RB identity	1
- CHOICE RLC info choice	RLC info
- CHOICE ULuplink RLC mode	UM RLC mode
- Transmission RLC discard	MaxDAT retransmissions
- Max_DAT	4
- Timer_MRW	100
- MaxMRW	4
- CHOICE DL RLC mode	UM RLC mode
- RB mapping info	
 CHOICE ULuplink logical channel mappings 	One logical channel
 - ULuplink transport channel type 	DCH
 Transport channel identity 	1
 Logical channel identity 	1
 MAC logical channel priority 	1
 Logical channel max loss 	0
 DL logical channel mapping 	
 DL transport channel type 	DCH
 Transport channel identity 	1
 Logical channel identity 	1
- RB identity	2
- CHOICE RLC info choice	RLC info
 CHOICE ULuplink RLC mode 	AM RLC mode
 Transmission RLC discard 	MaxDAT retransmissions
- Max_DAT	4
- Timer_MRW	100
- MaxMRW	4
 transmission window size 	8
- Timer_RST	500
- MAX_RST	4
- Polling info	
- Timer_poll_prohibit	200
- Timer_poll	200
- Poll_SDU	1
- Last transmission PU poll	TRUE
- Last retransmission PU poll	TRUE
- Poll windows	99
- CHOICE DL RLC mode	AM RLC mode
- In-sequence delivery	TRUE
- Receiving window size	8
- Downlink RLC status info	200
- Timer_status_prohibit	200 200
- Timer_EPC- Missing PU indicator	TRUE
- RB mapping info	INOL
- RB mapping into - CHOICE UL uplink logical channel mappings	One logical channel
- ULuplink transport channel type	DCH
- Transport channel identity	1
- Logical channel identity	2
MAC logical channel priority	2
- Logical channel max loss	
- DL logical channel mapping	
- DL transport channel type	DCH
- Transport channel identity	1
- Logical channel identity	2
- RB identity	3
- CHOICE RLC info choice	RLC info
	I AM RLC mode
- CHOICE ULuplink RLC mode	AM RLC mode MaxDAT retransmissions
	MaxDAT retransmissions 4

- MaxMRW	4
- transmission window size	8
- Timer RST	500
- MAX_RST	4
- Polling info	
- Timer_poll_prohibit	200
- Timer_poll	200
- Poll_SDU	1
- Last transmission PU poll	TRUE
- Last retransmission PU poll	TRUE
- Poll windows	99
- CHOICE DL RLC mode	AM RLC mode
- In-sequence delivery	TRUE
- Receiving window size	8
- Downlink RLC status info	
- Timer_status_prohibit	200
- Timer_EPC	200
- Missing PU indicator	TRUE
- RB mapping info	
- CHOICE ULuplink logical channel mappings	One logical channel
- ULuplink transport channel type	DCH
- Transport channel identity	1
- Logical channel identity	3
- MAC logical channel priority	3
- Logical channel max loss	0
- DL logical channel mapping	
- DL transport channel type	DCH
- Transport channel identity	1
 Logical channel identity 	3
- RB identity	4
- CHOICE RLC info choice	RLC info
- CHOICE UL uplink RLC mode	AM RLC mode
- Transmission RLC discard	MaxDAT retransmissions
- Max_DAT	4
- Timer_MRW	100
- MaxMRW	4
- transmission window size	8
- Timer_RST	500
- MAX_RST	4
- Polling info	000
- Timer_poll_prohibit	200
- Timer_poll	200
- Poll_SDU	1 TRUE
- Last transmission PU poll	TRUE
- Last retransmission PU poll	TRUE
- Poll windows - CHOICE DL RLC mode	99 AM RLC mode
- In-sequence delivery	TRUE
- Receiving window size	8
- Downlink RLC status info	
- Timer_status_prohibit	200
- Timer_EPC	200
- Missing PU indicator	TRUE
- RB mapping info	
- CHOICE ULuplink logical channel mappings	One logical channel
- ULuplink transport channel type	DCH
- Transport channel identity	1
- Logical channel identity	4
- MAC logical channel priority	4
- Logical channel max loss	0
- DL logical channel mapping	
- DL transport channel type	DCH
- Transport channel identity	1
- Logical channel identity	4
- RB information list	
- RB information to setup	
- RB identity	10
- PDCP info	Not Present
- RLC info	

- CHOICE Uplink RLC mode TM RLC mode - Transmission RLC discard Not Present - Segmentation indication **TRUE** TM RLC mode - CHOICE Downlink RLC mode **TRUE** - Segmentation indication - RB mapping info - CHOICE ULuplink logical channel mappings One logical channel - Uplink transport channel type **DCH** - Transport channel identity 2 - Logical channel identity 1 - MAC logical channel priority 1 - Logical channel max loss 0 - DL logical channel mapping - Downlink transport channel type **DCH** - Transport channel identity 2 - Logical channel identity 1 - RB information to setup - RB identity - PDCP info Not Present - RLC info - CHOICE Uplink RLC mode TM RLC mode - Transmission RLC discard Not Present - Segmentation indication **TRUE** - CHÖICE Downlink RLC mode TM RLC mode - Segmentation indication **TRUE** - RB mapping info - CHOICE ULuplink logical channel mappings One logical channel - Uplink transport channel type DCH - Transport channel identity 3 - Logical channel identity 1 - MAC logical channel priority 1 - Logical channel max loss 0 - DL logical channel mapping - Downlink transport channel type **DCH** - Transport channel identity 3 - Logical channel identity 1 - RB information to setup - RB identity - PDCP info Not Present - RLC info - CHOICE Uplink RLC mode TM RLC mode - Transmission RLC discard Not Present - Segmentation indication **TRUE** - CHOICE Downlink RLC mode TM RLC mode - Segmentation indication **TRUE** - RB mapping info - CHOICE ULuplink logical channel mappings One logical channel - Uplink transport channel type DCH - Transport channel identity 4 - Logical channel identity 1 - MAC logical channel priority 1 - Logical channel max loss 0 - DL logical channel mapping 1 - Downlink transport channel type DCH - Transport channel identity 4 - Logical channel identity 1 preDefTransChConfiguration - ULuplink Common Transport channel Info - TFC subset - Allowed Transport Format combination 0, 1, 2, 3, 4, 5 - CHOICE Mode specific info **FDD** - CHOICE ULuplink DCH TFCS Normal signalling - CHOICE Normal signalling Addition - CHOICE CTFC Size 6 bits - CTFC information - Power offset information - CHOICE Gain Factors Signalled Gain Factor - Gain factor ßc

- Gain factor ßd	0
- Reference TFC ID	Not Present
- Power offset Pp-m	0dB
- CTFC information	1
- Power offset information	
- CHOICE Gain Factors	Signallad Cain Factor
	Signalled Gain Factor
- Gain factor ßc	0
- Gain factor ßd	0
- Reference TFC ID	Not Present
 Power offset Pp-m 	0dB
 CTFC information 	11
 Power offset information 	
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor ßc	
- Gain factor ßd	0
- Reference TFC ID	Not Present
- Power offset Pp-m	0dB
- CTFC information	12
- Power offset information	12
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor &c	
	0
- Gain factor ßd	0
- Reference TFC ID	Not Present
- Power offset Pp-m	0dB
- CTFC information	13
 Power offset information 	
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor ßc	0
- Gain factor ßd	0
- Reference TFC ID	Not Present
- Power offset Pp-m	0dB
- CTFC information	23
- Power offset information	20
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor &c	Signalled Gain Factor
- Gain factor ßd	0
- Reference TFC ID	Not Present
- Power offset Pp-m	0dB
- ULuplink Add or reconfigured transport channel	
info list	
- Transport channel identity	2
- TFS	
- Dynamic Transport format information	
 Number of Transport blocks 	0
- RLC size	81
- Dynamic Transport format information	
- Number of Transport blocks	1
- RLC size	39
- Dynamic Transport format information	
- Number of Transport blocks	1
- RLC size	81
- Semi-static Transport Format information	
- Transmission time interval	20 ms
- Type of channel coding	convolutional
- Coding Rate	1/3
- Rate matching attribute	180-220
- CRC size	12 bits
- Transport channel identity	3
- TFS	
 Dynamic Transport format information 	
- Number of Transport blocks	0
- RLC size	103
- Dynamic Transport format information	
- Number of Transport blocks	1
- RLC size	103
- Semi-static Transport Format information	
- Transmission time interval	20 ms
- Type of channel coding	convolutional
- Coding Rate	1/3
- Ouliiu ivale	1 1/J
5	1 ''-

- Rate matching attribute	170-210
- CRC size	N/A
- Transport channel identity	4
- TFS	
- Dynamic Transport format information	
- Number of Transport blocks	0
- RLC size	60
- Dynamic Transport format information	
 Number of Transport blocks 	1
- RLC size	60
 Semi-static Transport Format information 	
 Transmission time interval 	20 ms
 Type of channel coding 	convolutional
- Coding Rate	1/2
- Rate matching attribute	215-256
- CRC size	N/A
- Transport channel identity	1
- TFS	
- Dynamic Transport format information	
- Number of Transport blocks	0
- RLC size	148
- Dynamic Transport format information	
- Dynamic Transport format information - Number of Transport blocks	1
	1
- RLC size	148
- Semi-static Transport Format information	40
- Transmission time interval	40 ms
- Type of channel coding	convolutional
- Coding Rate	1/3
- Rate matching attribute	155-165
- CRC size	16 bits
- DL Common transport channel info	
- SCCPCH TFCS	Not Present
- CHOICE Mode specific info	FDD
- CHOICE TFCS Signalling mode	Same as UL uplink
 Added or reconfigured DL TrCH info list 	
 Transport channel identity 	2
 CHOICE TFS signalling mode 	SameAsUL
- ULuplink TrCH Identity	2
- DCH quality target	
- BLER Quality value	0
- Transparent mode signalling info	Not Present
- Addide or reconfigured DL TrCH info list	
- Transport channel identity	3
- CHOICE TFS signalling mode	SameAsUL
- ULuplink TrCH Identity	3
- DCH quality target	
- BLER Quality value	0
- Transparent mode signalling info	Not Present
- Added or reconfigured DL TrCH info list	THOU TOOUTE
- Transport channel identity	4
- CHOICE TFS signalling mode	SameAsUL
	4
- ULuplink TrCH Identity	7
- DCH quality target	
- BLER Quality value	0 Not Broomt
- Transparent mode signalling info	Not Present
PreDefPhyChConfiguration	
- ULuplink_DPCH_InfoPredef	
- ULuplink DPCH Info	
- Uplink DPCH power control info	
- DPCCH power offset	-6dB
- PC Preamble	15 slots
- CHOICE Mode specific info	FDD
- TFCI existence	TRUE
- Puncturing Limit	0.88
- DL_ CommonInformationPredef	
- DL_DPCH_InfoCommon	
- Timing indication	initialize
- CHOICE Mode specific info	FDD
- Spreading factor and pilot	128
	•

 Number of bits for Pilot bits 	4 bits
 Fixed or Flexible Position 	Fixed
- TFCI existence	FALSE
- Downlink DPCH Offset Value	0

HANDOVER TO UTRAN COMMAND

Information Element	Value/remark
RR management Protocol Discriminator	'0110'B
Skip Indicator	'0000'B
Inter System to UTRAN Handover Command	'01100011'B
Message Type	
Handover to UTRAN Command IEI	TBD
Length of Handover to UTRAN Command contents	Octet length of the "Handover to UTRAN Command
	value part"
Handover to UTRAN Command value part	PER encoded ASN.1 value of type
	"HandoverToUTRANCommand-v1-IEs", the content is
	presented in the next table.

Content of "HandoverToUTRANCommand-v1-IEs"

Information Element	Value/remark
New U-RNTI	
- SRNC Identiy	'00000000001'B
- S-RNTI-2	1
Activation time	now
Ciphering algorithm	Standard UMTS Encryption Algorithm UEA1
RAB Info	j.
- RAB identity	
- GSM-MAP RAB identity	'0000001'B
- CN domain identity	CS domain
CHOICE Specification mode	Preconfiguration
- Predefined configuration identity	1
- CHOICE Mode specific info	FDD
- UL uplink DPCH info	
- ULuplink DPCH power control info	
- Power control algorithm1	1
- Scrambling code type	long
- Reduced scrambling code number	0
- Spreading factor	128
 DL common information post 	
- DL DPCH info common	
- Timing indication	initialize
 CFN target SFN frame offset 	Not present
- DL DPCH power control info	
- CHOICE Mode specific info	FDD
- DPC mode	Single TPC
- DL information perRL list	
- Premary CPICH info	
- Primary scrambling code	100
- DL DPCH info perRL	
- pCPICH usage for channelEst	May be used
- DL channelisation code	
- Secondary scrambling code	1
- SF and code number	SF = 128, code number = 127
- Scrambling code change	No code change
- TPC combination index	0
- Frequency info	
- UARFCN uplink(Nu)	See PIXIT
- UARFCN downlink(Nd)	See PIXIT
Maximum allowed ULuplink TX power	33dBm

CR page 399

8.3.6.1.5 Test requirement

After step 7 the ongoing call shall be continued on UTRAN cell.

8.3.6.2 Inter system handover to UTRAN/From GSM/Data/Same data rate/Success

8.3.6.2.1 Definition

8.3.6.2.2 Conformance requirement

The UE shall be able to receive a HANDOVER TO UTRAN COMMAND message from GSM and perform an intersystem handover, even if no prior UE measurements have been performed on the target UTRAN cell and/or frequency.

If the UE succeeds to establish the connection to UTRAN, it shall transmit a HANDOVER TO UTRAN COMPLETE message on the uplink DCCH.

Reference(s)

TS 25.331 Clause 8.3.6.

TS 04.18 Clause 3.4.4a.

8.3.6.2.3 Test purpose

To test that the UE handovers to the indicated UTRAN target cell and the data rate of the target channel is the same as the old channel when it is in the data call active state in the GSM serving cell and receives a HANDOVER TO UTRAN COMMAND.

8.3.6.2.4 Method of test

Initial conditions

System Simulator: 2 cells - Cell 1 is GSM, Cell 2 is UTRAN. GSM 11.10-1 section 26.6.5.1 or section 26.13.1.3 (for HSCSD) shall be referenced for the default parameters of cell 1.

UE: CC State U10 in cell 1

Related ICS/IXIT statement(s)

UE supports both GSM and UTRAN Radio Access Technologies,

UE supports UTRAN Streaming/unknown/ULuplink:14.4 DL:14.4 kbps/CS RAB + ULuplink:3.4 DL:3.4 kbps SRBs,

UE supports UTRAN Streaming/unknown/ULuplink:28.8 DL:28.8 kbps/CS RAB + ULuplink:3.4 DL:3.4 kbps SRBs,

UE supports UTRAN Streaming/unknown/ULuplink:57.6 DL:57.6 kbps/CS RAB + ULuplink:3.4 DL:3.4 kbps SRBs,

UE supports GSM 14.4 kbps data (HSCSD or full rate traffic channel for 14.4 kbit/s user data (TCH/F14.4)),

UE supports GSM 28.8 kbps data (HSCSD or full rate traffic channel for 28.8 kbit/s user data (E-TCH/F28.8)),

UE supports GSM 57.6 kbps data,

UE supports GSM-P, GSM-E, GSM-DCS, GSM-450, GSM-480.

Foreseen final state of the UE

The UE is in CC state U10 on cell 2.

Test Procedure

The SS starts the GSM cell and the UTRAN cell, the cell selection conditions of these two cells are in favour of GSM cell. The UE selects the GSM cell for camping on. After the UE receives and stores pre-configuration information from SIB16 broadcast in the UTRAN cell, the SS brings the UE into the call active state (CC state U10) with 14.4 kbps CS data call (for execution counter M=1). The SS configures a dedicated channel corresponding to the pre-configuration (streaming/unknown/ $\frac{U}{U}$ the pre-configuration (streaming/unknown/ $\frac{U}{U}$ then sends HANDOVER TO UTRAN COMMAND indicating the dedicated channel of the target cell to the UE through the GSM serving cell. After the UE receives the command it shall configure itself accordingly and switch to the dedicated channel of the UTRAN cell. The SS checks whether the handover is performed by checking that the UE transmits HANDOVER TO UTRAN COMPLETE to the SS through DCCH of the UTRAN cell.

Depending on the PIXIT parameters, the above procedure is executed maximum three times, each time for different initial conditions:

- If the UE supports GSM 14.4 kbps CS data and UTRAN streaming/unknown/ULuplink:14.4 DL:14.4 kbps/CS RAB + ULuplink:3.4 DL3.4 kbps SRBs, the procedure is executed for execution counter M = 1;
- If the UE supports GSM 28.8 kbps CS data and UTRAN streaming/unknown/ULuplink:28.8 DL:28.8 kbps/CS RAB + ULuplink:3.4 DL3.4 kbps SRBs, the procedure is executed for execution counter M = 2;
- If the UE supports GSM 57.6 kbps CS data and UTRAN streaming/unknown/ULuplink:57.6 DL:57.6 kbps/CS RAB + ULuplink:3.4 DL3.4 kbps SRBs, the procedure is executed for execution counter M = 3;

Expected sequence

This sequence is performed for a maximum execution counter M = 1, 2, 3, depending on the PIXIT parameters.

Step	Direction	Message	Comments
	UE SS		
1	SS		The SS configures GSM and UTRAN cells, the UTRAN cell broadcasts SIB16 containing pre-configuration information: For M = 1: (streaming/unknown/\textsup_Luplink:14.4 DL:14.4 kbps/CS RAB + \textsup_Luplink:3.4 DL3.4 kbps SRBs); For M = 2: (streaming/unknown/\textsup_Luplink:28.8 DL:28.8 kbps/CS RAB + \textsup_Luplink:3.4 DL3.4 kbps SRBs); For M = 3: (streaming/unknown/\textsup_Luplink:57.6 DL:57.6 kbps/CS RAB + \textsup_Luplink:3.4 DL3.4 kbps SRBs). UE camps on GSM cell and received SIB16 from UTRAN cell.
2	UE		The SS bring the UE into GSM U10 state in cell 1 and for M = 1: the UE is in GSM 14.4 kbps CS data call; for M = 2: the UE is in GSM 28.8 kbps CS data call; for M = 3: the UE is in GSM 57.6 kbps CS data call;
3	SS		The SS configures a dedicated channel in the UTRAN cell with the configuration: For M = 1: (streaming/unknown/\textsupplink:14.4 DL:14.4 kbps/CS RAB + \textsupplink:3.4 DL3.4 kbps SRBs); For M = 2: (streaming/unknown/\textsupplink:28.8 DL:28.8 kbps/CS RAB + \textsupplink:3.4 DL3.4 kbps SRBs); For M = 3: (streaming/unknown/\textsupplink:57.6 DL:57.6 kbps/CS RAB + \textsupplink:3.4 DL3.4 kbps SRBs)
4	+	HANDOVER TO UTRAN COMMAND	Send on cell 1 (GSM cell)
5	UE		The UE accepts the handover command and configures its lower layers using the parameters contained in the HANDOVER TO UTRAN COMMAND
6	SS		The SS waits for uplink physical channel in synchronization
7	→	HANDOVER TO UTRAN COMPLETE	The SS receives this message on DCCH of cell 2 (UTRAN cell). It implies that the down link physical channel has synchronised with UTRAN.

For execution 1 (M = 1):

Information Element	Value/remark
re-EstablishmentTimer T315	30 s
PredefinedRB_Configuration	
- SRB information list	
- RB identity	1
- CHOICE RLC info choice	RLC info
- CHOICE UL uplink RLC mode	UM RLC mode
- Transmission RLC discard	MaxDAT retransmissions
- Max_DAT	4
- Timer_MRW	100
- MaxMRW	4
- CHOICE DL RLC mode	UM RLC mode
- RB mapping info	
 CHOICE ULuplink logical channel mappings 	One logical channel
 - ULuplink transport channel type 	DCH
- Transport channel identity	1
- Logical channel identity	1
- MAC logical channel priority	1
- Logical channel max loss	0
- DL logical channel mapping	POLL
- DL transport channel type	DCH
- Transport channel identity	1
- Logical channel identity - RB identity	1 2
	RLC info
- CHOICE RLC info choice - CHOICE UL uplink RLC mode	AM RLC mode
- Transmission RLC discard	MaxDAT retransmissions
- Max_DAT	4
- Timer_MRW	100
- MaxMRW	4
- transmission window size	8
- Timer_RST	500
- MAX_RST	4
- Polling info	·
- Timer_poll_prohibit	200
- Timer_poll	200
- Poll_SDU	1
 Last transmission PU poll 	TRUE
 Last retransmission PU poll 	TRUE
- Poll windows	99
- CHOICE DL RLC mode	AM RLC mode
- In-sequence delivery	TRUE
- Receiving window size	8
- Downlink RLC status info	000
- Timer_status_prohibit	200
- Timer_EPC	200
- Missing PU indicator	TRUE
- RB mapping info - CHOICE UL uplink logical channel mappings	One logical channel
- CHOICE <u>DEuplink</u> logical channel mappings - <u>ULuplink</u> transport channel type	DCH
- Transport channel identity	1
- Logical channel identity	
- MAC logical channel priority	2
- Logical channel max loss	
- DL logical channel mapping	
- DL transport channel type	DCH
- Transport channel identity	1
- Logical channel identity	2
- RB identity	3
- CHOICE RLC info choice	RLC info
- CHOICE UL uplink RLC mode	AM RLC mode
- Transmission RLC discard	MaxDAT retransmissions
- Max_DAT	4
- Timer_MRW	100
- MaxMRW	4
- transmission window size	8

- Timer_RST	500
- MAX_RST	4
- Polling info	
- Timer_poll_prohibit	200
- Timer_poll	200
- Poll_SDU	1
- Last transmission PU poll	TRUE
	TRUE
- Last retransmission PU poll	
- Poll windows - CHOICE DL RLC mode	99 AM RLC mode
	TRUE
- In-sequence delivery	
- Receiving window size	8
- Downlink RLC status info	200
- Timer_status_prohibit	200
- Timer_EPC	
- Missing PU indicator	TRUE
- RB mapping info	One legical abound
- CHOICE <u>ULuplink</u> logical channel mappings	One logical channel
- ULuplink transport channel type	DCH
- Transport channel identity	1
- Logical channel identity	3
- MAC logical channel priority	3
- Logical channel max loss	0
- DL logical channel mapping	DCH
- DL transport channel type	
- Transport channel identity	1 2
- Logical channel identity	3
- RB identity	4 DIC info
- CHOICE RLC info choice	RLC info
- CHOICE ULuplink RLC mode	AM RLC mode
- Transmission RLC discard	MaxDAT retransmissions
- Max_DAT	4
- Timer_MRW	100
- MaxMRW	4
- transmission window size	8
- Timer_RST	500
- MAX_RST	4
- Polling info	000
- Timer_poll_prohibit	200
- Timer_poll	200
- Poll_SDU	1
- Last transmission PU poll	TRUE
- Last retransmission PU poll - Poll windows	TRUE
	99
- CHOICE DL RLC mode	AM RLC mode
- In-sequence delivery	TRUE
- Receiving window size	8
- Downlink RLC status info	200
- Timer_status_prohibit	200
- Timer_EPC	200 TDUE
- Missing PU indicator	TRUE
- RB mapping info	One legical channel
- CHOICE UL uplink logical channel mappings - UL uplink transport channel type	One logical channel DCH
- Transport channel identity	
	1
Logical channel identity MAC logical channel priority	4 4
- Logical channel max loss	0
- Logical channel max loss - DL logical channel mapping	U
- DL logical channel mapping - DL transport channel type	DCH
- Transport channel identity	1 1
- Logical channel identity	4
- RB information list	
- RB information to setup	10
- RB identity	10 Not Present
- PDCP info	Not Present
- RLC info	
- CHOICE Unlink DLC mode	I IM DI C mode
- CHOICE Uplink RLC mode - Transmission RLC discard	TM RLC mode Not Present

- Segmentation indication	TRUE
	-
- CHOICE Downlink RLC mode	TM RLC mode
- Segmentation indication	TRUE
- RB mapping info	
- CHOICE UL uplink logical channel mappings	One logical channel
	DCH
- Uplink transport channel type	-
- Transport channel identity	2
 Logical channel identity 	1
- MAC logical channel priority	1
- Logical channel max loss	0
- DL logical channel mapping	1
- Downlink transport channel type	DCH
- Transport channel identity	2
 Logical channel identity 	1
preDefTransChConfiguration	
- ULuplink Common Transport channel Info	
- TFC subset	
	0.4.0.0
 Allowed Transport Format combination 	0, 1, 2, 3
- CHOICE Mode specific info	FDD
- CHOICE UL uplink DCH TFCS	Normal signalling
- CHOICE Normal signalling	Addition
- CHOICE CTFC Size	2 bits
- CTFC information	
	0
- Power offset information	
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor ßc	0
- Gain factor ßd	0
- Reference TFC ID	Not Present
- Power offset Pp-m	0dB
- CTFC information	1
- Power offset information	
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor ßc	0
- Gain factor ßd	0
- Reference TFC ID	Not Present
- Power offset Pp-m	0dB
- CTFC information	2
- Power offset information	
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor ßc	0
- Gain factor ßd	0
	-
- Reference TFC ID	Not Present
- Power offset Pp-m	0dB
- CTFC information	3
 Power offset information 	
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor ßc	0
- Gain factor ßd	0
- Reference TFC ID	Not Present
- Power offset Pp-m	0dB
- ULuplink Add or reconfigured transport channel	
info list	
- Transport channel identity	2
- TFS	
- Dynamic Transport format information	
- Number of Transport blocks	0
- RLC size	576
	310
- Dynamic Transport format information	
- Number of Transport blocks	1
- RLC size	576
- Semi-static Transport Format information	
- Transmission time interval	40 ms
- Type of channel coding	Turbo coding
	145-185
- Rate matching attribute	
- CRC size	16 bits
- Transport channel identity	1
- TFS	
110	
=	
Dynamic Transport format information Number of Transport blocks	0

- RLC size	148
- Dynamic Transport format information	140
- Number of Transport blocks	1
- RLC size	148
- Semi-static Transport Format information	140
- Transmission time interval	40 ms
- Type of channel coding	convolutional
- Coding Rate	1/3
- Rate matching attribute	155-165
- CRC size	16 bits
- DL Common transport channel info	10 bits
- SCCPCH TFCS	Not Present
- CHOICE Mode specific info	FDD
- CHOICE TFCS Signalling mode	Same as ULuplink
- Added or reconfigured DL TrCH info list	Carrio de CL <u>apriiri.</u>
- Transport channel identity	2
- CHOICE TFS signalling mode	SameAsUL
- UL uplink TrCH Identity	2
- DCH quality target	
- BLER Quality value	0
 Transparent mode signalling info 	Not Present
PreDefPhyChConfiguration	
- ULuplink_ DPCH_InfoPredef	
- ULuplink DPCH Info	
 Uplink DPCH power control info 	
 DPCCH power offset 	-6dB
- PC Preamble	15 slots
 CHOICE Mode specific info 	FDD
- TFCI existence	TRUE
- Puncturing Limit	1
 DL_ CommonInformationPredef 	
- DL_DPCH_InfoCommon	
- Timing indication	initialize
- CHOICE Mode specific info	FDD
- Spreading factor and pilot	128
- Number of bits for Pilot bits	8 bits
- Fixed or Flexible Position	Fixed
- TFCI existence	FALSE
 Downlink DPCH Offset Value 	0

HANDOVER TO UTRAN COMMAND

Information Element	Value/remark
RR management Protocol Discriminator	'0110'B
Skip Indicator	'0000'B
Inter System to UTRAN Handover Command	'01100011'B
Message Type	
Handover to UTRAN Command IEI	TBD
Length of Handover to UTRAN Command contents	Octet length of the "Handover to UTRAN Command
	value part"
Handover to UTRAN Command value part	PER encoded ASN.1 value of type
	"HandoverToUTRANCommand-v1-IEs", the content is
	presented in the next table.

Content of "HandoverToUTRANCommand-v1-IEs"

Information Element	Value/remark
New U-RNTI	
- SRNC Identiy	'00000000001'B
- S-RNTI-2	1
Activation time	now
Ciphering algorithm	Standard UMTS Encryption Algorithm UEA1
RAB Info	71 0
- RAB identity	
- GSM-MAP RAB identity	'0000001'B
- CN domain identity	CS domain
CHOICE Specification mode	Preconfiguration
 Predefined configuration identity 	1
- CHOICE Mode specific info	FDD
- ULuplink DPCH info	
 - ULuplink DPCH power control info 	
 Power control algorithm1 	1
 Scrambling code type 	long
 Reduced scrambling code number 	0
- Spreading factor	64
 DL common information post 	
 DL DPCH info common 	
 Timing indication 	initialize
 CFN target SFN frame offset 	Not present
 DL DPCH power control info 	
 CHOICE Mode specific info 	FDD
- DPC mode	Single TPC
 DL information perRL list 	
- Premary CPICH info	
 Primary scrambling code 	100
- DL DPCH info perRL	
 pCPICH usage for channelEst 	May be used
- DL channelisation code	
 Secondary scrambling code 	1
- SF and code number	SF = 128, code number = 127
- Scrambling code change	No code change
- TPC combination index	0
- Frequency info	2 20/2
- UARFCN uplink(Nu)	See PIXIT
- UARFCN downlink(Nd)	See PIXIT
Maximum allowed ULuplink TX power	33dBm

For execution 2 (M = 2):

preDeTransChConfiguration - LHuglink (Common Transport channel Info - TFC subset - Allowed Transport Format combination - CHOICE Mode specific info - CHOICE Mode specific info - CHOICE CTFC Size - CHOICE CTFC Size - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset pp-m - CTFC information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset pp-m - CTFC information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset pp-m - CTFC information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset pp-m - CTFC information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset pp-m - CTFC information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset pp-m - CTFC information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Reference	Information Element	Value/remark
- Hupplink Common Transport channel Info - TFC subset - Allowed Transport Format combination - CHOICE Muse specific info - CHOICE Was Mode Specific info - CHOICE CAP Mode Specific info - CHOICE Commation - CHOICE Gain Factors - Gain factor & Gain factor	preDefTransChConfiguration	- 3.35. 5110111
- TFC subset - Allowed Transport Format combination - CHOICE Mode specific into - CHOICE Mode specific into - CHOICE GUBURIDING DOLT TFCS - CHOICE CTFC Size - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor 8d - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor 8d - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor 8d - Gain factor 8d - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor 8d - Reference TFC ID - Power offset pp-m - CTFC information - CHOICE Gain Factors - Gain factor 8d - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor 8d - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor 8d - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor 8d - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor 8d - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor 8d - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor 8d - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor 8d - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor 8d - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor 8d - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor 8d - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor 8d - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor 8d - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor 8d - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor 8d - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor 8d - Reference TFC ID - Not Present Odb Other Cortect Members Other		
- CHOICE Musplink DCH TFCS - CHOICE Supplink DCH TFCS - CHOICE CTFC Size - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor & C - Gain factor &	- TFC subset	
- CHOICE Supuling Sympalling - CHOICE Normal signalling - CHOICE Sort Sympalling - CHOICE Gain Factors - Gain factor & C - CHOICE Gain Factors - CHOICE Gain Factors - Cain factor & C - Gain fa		
C-HOICE CTFC Size C-TFC information Power offset information C-HOICE Gain Factors Gain factor & 0 Cain factor		
- CHOICE CTFC Size - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor ßd - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Gain factor ßd - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Gain factor ßd - CHOICE Gain Factors - Gain factor ßd - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Information - Power offset Pp-m - CTFC information - Power offset Pp-m - CTFC information - Power offset Information - Power offset Pp-m - CTFC information - Power		
- CTFC information - Power offset information - CHOICE Gain Factors - Gain factor foc - Gain factor foc - Gain factor foc - Gain factor foc - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor foc - CHOICE Gain Factors - Gain factor foc - Gain factor fo		
- Power offset information - CHOICE Gain Factor 8c - Gain factor 8d - Gain factor 8d - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factor 8c - Gain factor 8d - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor 8d - Power offset information - CHOICE Gain Factors - Gain factor 8d - Reference TFC ID - Power offset Information - CHOICE Gain Factors - Gain factor 8d - Gain factor 8d - CHOICE Gain Factors - Gain factor 8d - Gain factor 8d - Gain factor 8d - Reference TFC ID - Power offset Information - CHOICE Gain Factors - Gain factor 8d - Gain factor 8d - Reference TFC ID - Power offset Information - CHOICE Gain Factors - Gain factor 8d - Reference TFC ID - Power offset Information - CHOICE Gain Factors - Gain factor 8d - Reference TFC ID - Power offset Information - CHOICE Gain Factors - Gain factor 8d - Reference TFC ID - Power offset Information - CHOICE Gain Factors - Gain factor 8d - Gain factor 8d - Gain factor 8d - Reference TFC ID - Power offset Information - CHOICE Gain Factors - Gain factor 8d - Reference TFC ID - Power offset Information - CHOICE Gain Factors - Gain factor 8d - Reference TFC ID - Power offset Information - Power offset Information - Power offset Pp-m - CTFC i		
- CHOICE Gain Factors - Gain factor fbc - Gain factor fbc - Gain factor fbc - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor fbc - Gain factor fbc - CHOICE Gain Factors - Gain factor fbc - Gain factor fbc - Gain factor fbc - CHOICE Gain Factors - Gain factor fbc - Gain factor fbc - Gain factor fbc - CHOICE Gain Factors - Gain factor fbc - CHOICE Gain Factors - Gain factor fbc -		
- Gain factor & Cain factor &		Signalled Gain Factor
Reference TFC ID Power offset Pp-m CTFC Information - Power offset Information CHOICE Gain Factors Gain factor ßd Reference TFC ID Power offset Information - CHOICE Gain Factors Gain factor ßd Reference TFC ID Power offset Information - CHOICE Gain Factors Gain factor ßd Reference TFC ID Power offset Information - CHOICE Gain Factors Gain factor ßd Reference TFC ID Power offset Information - CHOICE Gain Factors Gain factor ßd Reference TFC ID Power offset Information - CHOICE Gain Factors Gain factor ßd Reference TFC ID Power offset Information - CHOICE Gain Factors Gain factor ßd Reference TFC ID Power offset Information - CHOICE Gain Factors Gain factor ßd Reference TFC ID Power offset Information - CHOICE Gain Factors Gain factor ßd Reference TFC ID Power offset Information - CHOICE Gain Factors Gain factor ßd Reference TFC ID Power offset Information - CHOICE Gain Factors Gain factor ßd Reference TFC ID Power offset Information - CHOICE Gain Factors Gain factor ßd Reference TFC ID Power offset Information - CHOICE Gain Factors Gain factor ßd Reference TFC ID Power offset Information - CHOICE Gain Factors Gain factor ßd Reference TFC ID Power offset Information - CHOICE Gain Factors Gain factor ßd Reference TFC ID Power offset Information - CHOICE Gain Factors Gain factor ßd Reference TFC ID Power offset Information - CHOICE Gain Factors Gain factor ßd Reference TFC ID Power offset Information - CHOICE Gain Factors Gain factor ßd Reference TFC ID Power offset Information - CHOICE Gain Factors Gain factor ßd Reference TFC ID Power offset Information - CHOICE Gain Factors Gain factor ßd Reference TFC ID Power offset Information - CHOICE Gain Factors Gain factor ßd Reference TFC ID Reference TF		0
- Power offset Information - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor f&c - Gain factor f&c - Gain factor f&c - Reference TFC ID - Power offset Information - CHOICE Gain Factors - Gain factor f&c - CTFC information - CHOICE Gain Factors - Gain factor f&c - Gain		
- CTFC information - Power offset information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor fid - Power offset information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor fid - Reference TFC ID - Power offset information - Number of Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Semi-static Transport Format information		
- Power offset information - CHOICE Gain Factors - Gain factor f&C - Gain factor f&C - Gain factor f&C - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor f&C - CHOICE Gain Factors - Gain factor f&C - Gain fact		
- CHOICE Gain Factors - Gain factor ßc - Gain factor ßc - Gain factor ßc - Reference TFC ID - Power offset Pp-m - CTFC information - Power offset Information - CHOICE Gain Factors - Gain factor ßc - Gain factor ßc - Gain factor ßc - Gain factor ßc - Power offset Information - CHOICE Gain Factors - Gain factor ßc - Gain factor ß		1
- Gain factor ß d - Reference TFC ID - Power offset Pp-m - CTFC information - Power offset Rorm - CHOICE Gain Factors - Gain factor ß d - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ß d - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor ß d - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor ß d - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor ß d - Power offset information - CHOICE Gain Factors - Gain factor ß d - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor ß d - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor ß d - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor ß d - Reference TFC ID - Power offset pp-m - CTFC information - CHOICE Gain Factors - Gain factor ß d - Reference TFC ID - Power offset pp-m - ULuplink Add or reconfigured transport channel info list - Transport channel identity - TFS - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Semi-static Transport Format information - Number of Transport Format information - Number of Transport blocks - RLC size - Semi-static Transport Format information		Signalled Gain Factor
- Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor ßc - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - Power offset Pp-m - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor ßc - Gain f		
Reference TFC ID Power offset Pp-m CTFC information Power offset information CHOICE Gain Factors Gain factor & Gain factor & Gain factor Bd Reference TFC ID Power offset information CHOICE Gain Factors Gain factor & CTFC information CHOICE Gain Factors Gain factor & CTFC information CHOICE Gain Factors Gain factor & Gain factor & Gain factor & CTFC information CHOICE Gain Factors Gain factor & Gain factor		
- CTFC information - Power offset information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset p-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Power offset information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor ßd - Power offset information - CHOICE Gain Factors - Gain factor ßd - Power offset information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - ULuplink Add or reconfigured transport channel info list - Transport channel identity - TFS - Dynamic Transport format information - Number of Transport format information	- Reference TFC ID	
- CTFC information - Power offset information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset p-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Power offset information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor ßd - Power offset information - CHOICE Gain Factors - Gain factor ßd - Power offset information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - ULuplink Add or reconfigured transport channel info list - Transport channel identity - TFS - Dynamic Transport format information - Number of Transport format information		
- CHOICE Gain Factors - Gain factor ßc - Gain factor ßc - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßc - CTFC information - CTFC information - CTFC information - CTFC information - CHOICE Gain Factors - Gain factor ßc - CHOICE Gain Factors - Gain factor ßc - Gain factor ßc - Gain factor ßc - CHOICE Gain Factors - Gain factor ßc -		2
- Gain factor ßc - Gain factor ßd - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor ßc - CHOICE Gain Factors - Gain factor ßc - Gain factor ßc - CHOICE Gain Factors - Gain factor ßc - CHOICE Gain Factors - Gain factor ßc - Gain factor ßc - CHOICE Gain Factors - Gain factor ßc - CHOICE Gain Factors - Gain factor ßc - Gain factor ßc - CHOICE Gain Factors - Gain factor ßc - Gain fac		Signallad Cain Factor
- Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset information - CTFC information - CHOICE Gain Factors - Gain factor ßd - Power offset information - CHOICE Gain Factors - Gain factor ßd - Gain factor ßd - Reference TFC ID - Power offset pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset information - ULLuplink Add or reconfigured transport channel info list - Transport channel identity - TFS - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport format information - Number of Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport format information		9
- Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - Utuplink Add or reconfigured transport channel info list - Transport channel identity - TFS - Dynamic Transport format information - Number of Transport format information		
- Power offset Pp-m - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor ßc - Gain factor ßc - Reference TFC ID - Power offset pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßc - CTFC information - CHOICE Gain Factors - Gain factor ßc - CHOICE Gain Factors - Gain factor ßc - Gain factor		
- CTFC information - Power offset information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - ULuplink Add or reconfigured transport channel info list - Transport channel identity - TFS - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport format information		
- Power offset information - CHOICE Gain Factors - Gain factor ßc - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßc - Gain factor ßc - Gain factor ßc - Gain factor ßd - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor ßc -		
- Gain factor ßc - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - Cain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset pp-m - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - Uluplink Add or reconfigured transport channel info list - Transport channel identity - TFS - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Semi-static Transport Format information	 Power offset information 	
- Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor ßc - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - Power offset Pp-m - CTFC information - Power offset Pp-m - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - ULuplink Add or reconfigured transport channel info list - Transport channel identity - TFS - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport bromat information - Number of Transport format information - Number of Transport blocks - RLC size - Semi-static Transport Format information		=
- Reference TFC ID - Power offset Pp-m - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor ßc - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßc - Gain factor ßd - Reference TFC ID - Power offset Pp-m - ULuplink Add or reconfigured transport channel info list - Transport channel identity - TFS - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Semi-static Transport Format information		
- Power offset Pp-m - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor ßc - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Power offset Pp-m - CTFC information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset Pp-m - ULuplink Add or reconfigured transport channel info list - Transport channel identity - TFS - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport format information		-
- CTFC information - Power offset information - CHOICE Gain Factors - Gain factor ßc - Gain factor ßd - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor ßd - Reference TFC ID - Power offset information - CHOICE Gain Factors - Gain factor ßc - Gain factor ßc - Gain factor ßd - Reference TFC ID - Power offset Pp-m - ULuplink Add or reconfigured transport channel info list - Transport channel identity - TFS - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport format information - Semi-static Transport Format information		
- Power offset information - CHOICE Gain Factors - Gain factor ßc - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor ßc - Gain factor ßc - Gain factor ßc - Gain factor ßd - Reference TFC ID - Power offset Pp-m - ULuplink Add or reconfigured transport channel info list - Transport channel identity - TFS - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport format information		
- CHOICE Gain Factors - Gain factor ßc - Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor ßc - Power offset Pp-m - ULuplink Add or reconfigured transport channel info list - Transport channel identity - TFS - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport format information		
- Gain factor ßd - Reference TFC ID - Power offset Pp-m - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor ßc - Gain factor ßd - Reference TFC ID - Power offset Pp-m - ULuplink Add or reconfigured transport channel info list - Transport channel identity - TFS - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport format information	- CHOICE Gain Factors	-
- Reference TFC ID - Power offset Pp-m - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor ßc - Gain factor ßd - Reference TFC ID - Power offset Pp-m - ULuplink Add or reconfigured transport channel info list - Transport channel identity - TFS - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Semi-static Transport Format information		
- Power offset Pp-m - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor ßc - Gain factor ßd - Reference TFC ID - Power offset Pp-m - ULuplink Add or reconfigured transport channel info list - Transport channel identity - TFS - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Semi-static Transport Format information		
- CTFC information - Power offset information - CHOICE Gain Factors - Gain factor ßc - Gain factor ßd - Reference TFC ID - Power offset Pp-m - ULuplink Add or reconfigured transport channel info list - Transport channel identity - TFS - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Semi-static Transport Format information		
- Power offset information - CHOICE Gain Factors - Gain factor ßc - Gain factor ßd - Reference TFC ID - Power offset Pp-m - ULuplink Add or reconfigured transport channel info list - Transport channel identity - TFS - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport format information - Number of Transport blocks - RLC size - Semi-static Transport Format information		
- CHOICE Gain Factors - Gain factor ßc - Gain factor ßd - Reference TFC ID - Power offset Pp-m - ULuplink Add or reconfigured transport channel info list - Transport channel identity - TFS - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport format information - Number of Transport format information - Number of Transport blocks - RLC size - Semi-static Transport Format information		-
- Gain factor ßc - Gain factor ßd - Reference TFC ID - Power offset Pp-m - ULuplink Add or reconfigured transport channel info list - Transport channel identity - TFS - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Semi-static Transport Format information		Signalled Gain Factor
- Reference TFC ID - Power offset Pp-m - ULuplink Add or reconfigured transport channel info list - Transport channel identity - TFS - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport format information - Number of Transport blocks - RLC size - Semi-static Transport Format information	- Gain factor ßc	_ ~
- Power offset Pp-m - ULuplink Add or reconfigured transport channel info list - Transport channel identity - TFS - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport format information - Number of Transport format information - Number of Transport blocks - RLC size - Semi-static Transport Format information		
- ULuplink Add or reconfigured transport channel info list - Transport channel identity - TFS - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport format information - Number of Transport format information - Number of Transport blocks - RLC size - Semi-static Transport Format information		
info list - Transport channel identity - TFS - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport format information - Number of Transport format information - Number of Transport blocks - RLC size - Semi-static Transport Format information		UUB
- Transport channel identity - TFS - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport format information - Number of Transport blocks - RLC size - Semi-static Transport Format information		
- TFS - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport format information - Number of Transport blocks - RLC size - Semi-static Transport Format information		2
- Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - RLC size - Semi-static Transport Format information	- TFS	
- Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - RLC size - Semi-static Transport Format information		
- Dynamic Transport format information - Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Semi-static Transport Format information	 Number of Transport blocks 	
- Number of Transport blocks - RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Semi-static Transport Format information		5/6
- RLC size - Dynamic Transport format information - Number of Transport blocks - RLC size - Semi-static Transport Format information		
- Dynamic Transport format information - Number of Transport blocks - RLC size - Semi-static Transport Format information		
- Number of Transport blocks - RLC size - Semi-static Transport Format information		0.0
- RLC size - Semi-static Transport Format information		2
	- RLC size	
- Transmission time interval 40 ms		
	- Transmission time interval	40 ms

- Type of channel coding	Turbo coding
- Rate matching attribute	135-175
- CRC size	16 bits
PreDefPhyChConfiguration	
- UL uplink DPCH InfoPredef	
- UL uplink DPCH Info	
- Uplink DPCH power control info	
- DPCCH power offset	-6dB
- PC Preamble	15 slots
- CHOICE Mode specific info	FDD
- TFCI existence	TRUE
- Puncturing Limit	1
- DL CommonInformationPredef	
- DL DPCH InfoCommon	
- Timing indication	initialize
- CHOICE Mode specific info	FDD
- Spreading factor and pilot	64
- Number of bits for Pilot bits	8 bits
- Fixed or Flexible Position	Fixed
- Fixed or Flexible Position - TFCI existence - Downlink DPCH Offset Value	FALSE 0

HANDOVER TO UTRAN COMMAND

Information Element	Value/remark
RR management Protocol Discriminator	'0110'B
Skip Indicator	'0000'B
Inter System to UTRAN Handover Command	'01100011'B
Message Type	
Handover to UTRAN Command IEI	TBD
Length of Handover to UTRAN Command contents	Octet length of the "Handover to UTRAN Command
	value part"
Handover to UTRAN Command value part	PER encoded ASN.1 value of type
	"HandoverToUTRANCommand-v1-IEs", the content is
	presented in the next table.

Content of "HandoverToUtranCommand-v1-IEs"

Information Element	Value/remark
New U-RNTI	
- SRNC Identiy	'00000000001'B
- S-RNTI-2	1
Activation time	now
Ciphering algorithm	Standard UMTS Encryption Algorithm UEA1
RAB Info	, , , , , , , , , , , , , , , , , , ,
- RAB identity	
- GSM-MAP RAB identity	'0000001'B
- CN domain identity	CS domain
CHOICE Specification mode	Preconfiguration
 Predefined configuration identity 	1
 CHOICE Mode specific info 	FDD
- ULuplink DPCH info	
 - ULuplink DPCH power control info 	
 Power control algorithm1 	1
 Scrambling code type 	long
 Reduced scrambling code number 	0
 Spreading factor 	32
 DL common information post 	
- DL DPCH info common	
 Timing indication 	initialize
 CFN target SFN frame offset 	Not present
 DL DPCH power control info 	
- CHOICE Mode specific info	FDD
- DPC mode	Single TPC
- DL information perRL list	
- Premary CPICH info	
- Primary scrambling code	100
- DL DPCH info perRL	
- pCPICH usage for channelEst	May be used
- DL channelisation code	,
- Secondary scrambling code	1
- SF and code number	SF = 64, code number = 63
- Scrambling code change	No code change
- TPC combination index	0
- Frequency info	See PIXIT
- UARFCN uplink(Nu)	
- UARFCN downlink(Nd)	See PIXIT
Maximum allowed ULuplink TX power	33dBm

For execution 3 (M = 3):

Information Element	Value/remark
preDefTransChConfiguration	value/reiliark
- ULuplink Common Transport channel Info	
- DE uplink Common Transport channel into	
- Allowed Transport Format combination	0, 1, 2, 3, 4, 5, 6, 7, 8, 9
- CHOICE Mode specific info	FDD
- CHOICE ULuplink DCH TFCS	Normal signalling
- CHOICE Normal signalling	Addition
- CHOICE CTFC Size	4 bits
- CTFC information	0
 Power offset information 	
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor ßc	0
- Gain factor ßd	0
- Reference TFC ID	Not Present
 Power offset Pp-m 	0dB
 CTFC information 	1
- Power offset information	
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor ßc	0
- Gain factor ßd	0
- Reference TFC ID	Not Present
- Power offset Pp-m	OdB
 CTFC information Power offset information 	2
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor &c	0
- Gain factor &d	0
- Reference TFC ID	Not Present
- Power offset Pp-m	0dB
- CTFC information	3
- Power offset information	
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor ßc	0
- Gain factor ßd	0
- Reference TFC ID	Not Present
- Power offset Pp-m	0dB
 CTFC information 	4
 Power offset information 	
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor ßc	0
- Gain factor ßd	0
- Reference TFC ID	Not Present
- Power offset Pp-m	OdB
CTFC informationPower offset information	5
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor &c	0
- Gain factor &d	0
- Reference TFC ID	Not Present
- Power offset Pp-m	0dB
- CTFC information	6
- Power offset information	
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor ßc	0
- Gain factor ßd	0
- Reference TFC ID	Not Present
- Power offset Pp-m	0dB
- CTFC information	7
- Power offset information	
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor ßc	0
- Gain factor ßd	0 Not Droppet
- Reference TFC ID	Not Present
- Power offset Pp-m	OdB
- CTFC information	8

- Power offset information	
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor &c	0
- Gain factor ßd	
- Reference TFC ID	Not Present
- Power offset Pp-m	0dB
- CTFC information	9
- Power offset information	
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor &c	0
- Gain factor &d	0
- Reference TFC ID	Not Present
	OdB
- Power offset Pp-m	UQB
- ULuplink Add or reconfigured transport channel	
info list	
- Transport channel identity	2
- TFS	
- Dynamic Transport format information	
- Number of Transport blocks	0
- RLC size	576
- Dynamic Transport format information	
- Number of Transport blocks	1
- RLC size	576
- Dynamic Transport format information	
- Number of Transport blocks	2
- RLC size	576
- Dynamic Transport format information	
- Number of Transport blocks	3
- RLC size	576
- Dynamic Transport format information	
- Number of Transport blocks	4
- RLC size	576
- Semi-static Transport Format information	40
- Transmission time interval	40 ms
- Type of channel coding	Turbo coding
- Rate matching attribute	125-165
- CRC size	16 bits
PreDefPhyChConfiguration	
- ULuplink_DPCH_Info Predef	
- ULuplink DPCH Info	
- Uplink DPCH power control info	CAD
- DPCCH power offset	-6dB
- PC Preamble	15 slots
- CHOICE Mode specific info	FDD
- TFCI existence	TRUE
- Puncturing Limit	1
- DL_ CommonInformationPredef	
- DL_DPCH_InfoCommon	initialina
- Timing indication	initialize
- CHOICE Mode specific info	FDD
- Spreading factor and pilot	32 8 hita
- Number of bits for Pilot bits	8 bits
- Fixed or Flexible Position	Fixed
- TFCI existence	FALSE
- Downlink DPCH Offset Value	0

HANDOVER TO UTRAN COMMAND

Information Element	Value/remark
RR management Protocol Discriminator	'0110'B
Skip Indicator	'0000'B
Inter System to UTRAN Handover Command	'01100011'B
Message Type	
Handover to UTRAN Command IEI	TBD
Length of Handover to UTRAN Command contents	Octet length of the "Handover to UTRAN Command
	value part"
Handover to UTRAN Command value part	PER encoded ASN.1 value of type
	"HandoverToUTRANCommand-v1-IEs", the content is
	presented in the next table.

Content of "HandoverToUTRANCommand-v1-IEs"

Information Element	Value/remark
New U-RNTI	
- SRNC Identiy	'00000000001'B
- S-RNTI-2	1
Activation time	now
Ciphering algorithm	Standard UMTS Encryption Algorithm UEA1
RAB Info	, °
- RAB identity	
- GSM-MAP RAB identity	'0000001'B
- CN domain identity	CS domain
CHOICE Specification mode	Preconfiguration
- Predefined configuration identity	1
- CHOICE Mode specific info	FDD
- UL uplink DPCH info	
- ULuplink DPCH power control info	
- Power control algorithm1	1
- Scrambling code type	long
 Reduced scrambling code number 	0
- Spreading factor	16
- DL common information post	
- DL DPCH info common	
- Timing indication	initialize
 CFN target SFN frame offset 	Not present
- DL DPCH power control info	
- CHOICE Mode specific info	FDD
- DPC mode	Single TPC
- DL information perRL list	
- Premary CPICH info	
- Primary scrambling code	100
- DL DPCH info perRL	
 pCPICH usage for channelEst 	May be used
- DL channelisation code	
- Secondary scrambling code	1
- SF and code number	SF = 32, code number = 31
- Scrambling code change	No code change
- TPC combination index	0
- Frequency info	
- UARFCN uplink(Nu)	See PIXIT
- UARFCN downlink(Nd)	See PIXIT
Maximum allowed ULuplink TX power	33dBm

8.3.6.2.5 Test requirement

After step 7 the ongoing call shall be continued on UTRAN cell.

8.3.6.3 Inter system handover to UTRAN/From GSM/Data/Data rate upgrading/Success

8.3.6.3.1 Definition

8.3.6.3.2 Conformance requirement

The UE shall be able to receive a HANDOVER TO UTRAN COMMAND message from GSM and perform an intersystem handover, even if no prior UE measurements have been performed on the target UTRAN cell and/or frequency.

If the UE succeeds to establish the connection to UTRAN, it shall transmit a HANDOVER TO UTRAN COMPLETE message on the uplink DCCH.

Reference(s)

TS 25.331 Clause 8.3.6.

TS 04.18 Clause 3.4.4a.

8.3.6.3.3 Test purpose

To test that the UE being in the data call active state handovers from the GSM serving cell to the indicated channel of a higher data rate in the UTRAN target cell after it receives a HANDOVER TO UTRAN COMMAND.

8.3.6.3.4 Method of test

Initial conditions

System Simulator: 2 cells - Cell 1 is GSM, Cell 2 is UTRAN. GSM 11.10-1 section 26.6.5.1 or section 26.13.1.3(for HSCSD) shall be referenced for the default parameters of cell 1.

UE: CC State U10 in cell 1

Related ICS/IXIT statement(s)

UE supports both GSM and UTRAN Radio Access Technologies,

UE supports UTRAN Streaming/unknown/ULuplink:28.8 DL:28.8 kbps/CS RAB + ULuplink:3.4 DL:3.4 kbps SRBs,

UE supports UTRAN Streaming/unknown/ULuplink:57.6 DL:57.6 kbps/CS RAB + ULuplink:3.4 DL:3.4 kbps SRBs,

UE supports GSM 14.4 kbps data (HSCSD or full rate traffic channel for 14.4 kbit/s user data (TCH/F14.4)),

UE supports GSM 28.8 kbps data (HSCSD or full rate traffic channel for 28.8 kbit/s user data (E-TCH/F28.8)),

UE supports GSM-P, GSM-E, GSM-DCS, GSM-450, GSM-480.

Foreseen final state of the UE

The UE is in CC state U10 on cell 2.

Test Procedure

The SS starts the GSM cell and the UTRAN cell with cell selection conditions in favour of GSM cell. In UTRAN cell SIB16 is broadcast. The UE selects the GSM cell and received the pre-configuration information from SIB16. Then the SS brings the UE into the call active state (CC state U10) with 14.4 kbps CS data call (for execution counter M = 1). The SS configures a dedicated channel corresponding to the pre-configuration (streaming/unknown/\text{ULuplink}:28.8 DL:28.8 kbps/CS RAB + \text{ULuplink}:3.4 DL3.4 kbps SRBS for M = 1), then sends HANDOVER TO UTRAN COMMAND indicating the dedicated channel of the target cell to the UE through the GSM serving cell. After the UE receives the command it shall configure itself accordingly and switch to the new channel of the UTRAN cell. The SS checks whether the handover is performed by checking that the UE transmits HANDOVER TO UTRAN COMPLETE to the SS through DCCH of the UTRAN cell.

Depending on the PIXIT parameters, the above procedure is executed maximum three times, each time for different conditions:

- If the UE supports GSM 14.4 kbps CS data and UTRAN streaming/unknown/ULuplink:28.8 DL:28.8 kbps/CS RAB + ULuplink:3.4 DL3.4 kbps SRBs, the procedure is executed for execution counter M = 1;
- If the UE supports GSM 14.4 kbps CS data and UTRAN streaming/unknown/ULuplink:57.6 DL:57.6 kbps/CS RAB + ULuplink:3.4 DL3.4 kbps SRBs, the procedure is executed for execution counter M = 2;
- If the UE supports GSM 28.8 kbps CS data and UTRAN streaming/unknown/ULuplink:57.6 DL:57.6 kbps/CS RAB + ULuplink:3.4 DL3.4 kbps SRBs, the procedure is executed for execution counter M = 3;

Expected sequence

This sequence is performed for a maximum execution counter M = 1, 2, 3, depending on the PIXIT parameters.

Step	Dire	ction	Message	Comments
	UE	SS]	
1	S	S		The SS configures GSM and UTRAN cells, the UTRAN cell broadcasts SIB16 containing pre-configuration information: For M = 1: (streaming/unknown/\foundarrow\text{ULuplink}:28.8 DL:28.8 kbps/CS RAB + \foundarrow\text{ULuplink}:3.4 DL3.4 kbps SRBs); For M = 2: (streaming/unknown/\foundarrow\text{ULuplink}:57.6 DL:57.6 kbps/CS RAB + \foundarrow\text{ULuplink}:3.4 DL3.4 kbps SRBs); For M = 3: (streaming/unknown/\foundarrow\text{ULuplink}:57.6 DL:57.6 kbps/CS RAB + \foundarrow\text{ULuplink}:3.4 DL3.4 kbps SRBs). UE camps on GSM cell and received SIB16 from UTRAN cell.
2	U	E		The SS bring the UE into GSM U10 state in cell 1 and for M = 1: the UE is in GSM 14.4 kbps CS data call; for M = 2: the UE is in GSM 14.4 kbps CS data call; for M = 3: the UE is in GSM 28.8 kbps CS data call;
3	S	S		The SS configures a dedicated channel in the UTRAN cell with the configuration: For M = 1: (streaming/unknown/\foundarrow\text{ULuplink}:28.8 DL:28.8 kbps/CS RAB + \foundarrow\text{ULuplink}:3.4 DL3.4 kbps SRBs); For M = 2: (streaming/unknown/\foundarrow\text{ULuplink}:57.6 DL:57.6 kbps/CS RAB + \foundarrow\text{ULuplink}:3.4 DL3.4 kbps SRBs); For M = 3: (streaming/unknown/\foundarrow\text{ULuplink}:57.6 DL:57.6 kbps/CS RAB + \foundarrow\text{ULuplink}:3.4 DL3.4 kbps SRBs)
4	•		HANDOVER TO UTRAN COMMAND	Send on cell 1 (GSM cell)
5	U	E		The UE accepts the handover command and configures its lower layers using the parameters contained in the HANDOVER TO UTRAN COMMAND
6	S	S		The SS waits for uplink physical channel in synchronization
7	_	>	HANDOVER TO UTRAN COMPLETE	The SS receives this message on DCCH of cell 2 (UTRAN cell). It implies that the down link physical channel has synchronised with UTRAN.

Specific message contents

For execution 1 (M = 1):

Same as the default message contents in clause 8.3.6.2 for M=2.

For execution 2 (M = 2):

Same as the default message contents in clause 8.3.6.2 for M = 3.

For execution 3 (M = 3):

Same as the default message contents in clause 8.3.6.2 for M = 3.

8.3.6.3.5 Test requirement

After step 7 the ongoing call shall be continued on UTRAN cell.

8.3.6.4 Inter system handover to UTRAN/From GSM/Speech/Establishment/Success

8.3.6.4.1 Definition

8.3.6.4.2 Conformance requirement

The UE shall be able to receive a HANDOVER TO UTRAN COMMAND message from GSM and perform an intersystem handover, even if no prior UE measurements have been performed on the target UTRAN cell and/or frequency.

If the UE succeeds to establish the connection to UTRAN, it shall transmit a HANDOVER TO UTRAN COMPLETE message on the uplink DCCH.

Reference(s)

TS 25.331 Clause 8.3.6.

TS 04.18 Clause 3.4.4a.

8.3.6.4.3 Test purpose

To test that the UE supporting both GSM and UTRAN handovers from the GSM serving cell to the indicated channel in UTRAN target cell when the UE is in the call establishment phase and receives a HANDOVER TO UTRAN COMMAND.

8.3.6.4.4 Method of test

Initial conditions

System Simulator: 2 cells - Cell 1 is GSM, Cell 2 is UTRAN. GSM 11.10-1 section 26.6.5.1 shall be referenced for the default parameters of cell 1.

UE: CC State U1 in cell 1

Related ICS/IXIT statement(s)

UE supports both GSM and UTRAN Radio Access Technologies,

UE supports UTRAN AMR,

UE supports GSM FR,

UE supports GSM-P, GSM-E, GSM-DCS, GSM-450, GSM-480.

Foreseen final state of the UE

The UE is in CC state U1 on cell 2.

Test Procedure

The SS starts GSM cell and UTRAN cell with the cell selection conditions in favour of GSM cell. The UE selects the GSM cell. After the UE camps on the GSM cell and received SIB16 broadcast in the UTRAN cell, the UE is triggered to make an MO speech call. After the SS received SETUP message it configures a dedicated channel corresponding to the predefined configuration (conversationa/speech/ULuplink:12.2 DL:12.2 kbps/CS RAB + ULuplink:3.4 DL3.4 kbps SRBS) described by SIB16, then the SS sends HANDOVER TO UTRAN COMMAND indicating the dedicated channel to the UE through the GSM serving cell. After the UE receives the command and it shall configure itself accordingly and switch to the new channel of UTRAN cell. The SS checks whether the handover is performed by checking that the UE transmits HANDOVER TO UTRAN COMPLETE to the SS through DCCH of the UTRAN cell.

Expected sequence

Step	Direction	Message	Comments
	UE SS	1	
1	SS		The SS configures GSM and UTRAN cells, UE camps on GSM cell and received SIB16 from UTRAN cell.
2	UE		To trigger UE to make an MO call
3	\rightarrow	CHANNEL REQUEST	initiate outgoing call
4	+	IMMEDIATE ASSIGNMENT	SDCCH, U0
5	\rightarrow	CM SERVICE REQUEST	U0.1
6	\rightarrow	SETUP	U1
7	SS		The SS configures a dedicated channel with the configuration: conversational/speech/ ULuplink :12.2 DL:12.2 kbps/CS RAB + ULuplink :3.4 DL3.4 kbps SRBs in UTRAN cell.
8	+	HANDOVER TO UTRAN COMMAND	Send on cell 1 (GSM cell)
9	UE		The UE accepts the handover command and configures its lower layers using the parameters contained in the HANDOVER TO UTRAN COMMAND
10	SS		The SS waits for uplink physical channel in synchronization
11	→	HANDOVER TO UTRAN COMPLETE	The SS receives this message on DCCH of cell 2 (UTRAN cell). It implies that the down link physical channel has synchronised with UTRAN.

Specific message contents

Same as the default message contents in clause 8.3.6.1 for M = 1.

8.3.6.4.5 Test requirement

After step 11 the ongoing call shall be continued on UTRAN cell.

8.3.6.5 Inter system handover to UTRAN/From GSM/Speech/Blind HO/Success

8.3.6.5.1 Definition

8.3.6.5.2 Conformance requirement

The UE shall be able to receive a HANDOVER TO UTRAN COMMAND message from GSM and perform an intersystem handover, even if no prior UE measurements have been performed on the target UTRAN cell and/or frequency.

If the UE succeeds to establish the connection to UTRAN, it shall transmit a HANDOVER TO UTRAN COMPLETE message on the uplink DCCH.

Reference(s)

TS 25.331 Clause 8.3.6.

TS 04.18 Clause 3.4.4a.

8.3.6.5.3 Test purpose

To test that the UE handovers from the GSM serving cell to the indicated channel of UTRAN target cell when it is in the speech call active state without any knowledge of the target system (blind handover) and receives a HANDOVER TO UTRAN COMMAND.

8.3.6.5.4 Method of test

Initial conditions

System Simulator: 2 cells - Cell 1 is GSM, Cell 2 is UTRAN. GSM 11.10-1 section 26.6.5.1 shall be referenced for the default parameters of cell 1.

UE: CC State U10 in cell 1

Related ICS/IXIT statement(s)

UE supports both GSM and UTRAN Radio Access Technologies,

UE supports UTRAN AMR,

UE supports GSM FR,

UE supports GSM-P, GSM-E, GSM-DCS, GSM-450, GSM-480.

Foreseen final state of the UE

The UE is in CC state U10 on cell 2.

Test Procedure

The SS starts the GSM cell and the UTRAN cell with cell selection conditions in favour of GSM cell, SIB16 is not broadcast in the UTRAN cell and the UE has no any predefined configuration stored. The UE selects the GSM cell. Then the SS brings the UE into the call active state (CC state U10) with FR speech. The SS configures a dedicated channel (conversationa/speech/ULuplink:12.2 DL:12.2 kbps/CS RAB + ULuplink:3.4 DL3.4 kbps SRBS), then sends HANDOVER TO UTRAN COMMAND indicating the dedicated channel of the target cell to the UE through the GSM serving cell. After the UE receives the command it shall configure itself accordingly and switch to the dedicated channel of UTRAN cell. The SS checks whether the handover is performed by checking that the UE transmits HANDOVER TO UTRAN COMPLETE to the SS through DCCH of the UTRAN cell.

Expected sequence

Step	Direc	ction	Message	Comments
	UE	SS		
1	U	E		The SS bring the UE into GSM U10 state in cell 1 and the UE has no any pre-configuration information stored
2	S	S		The SS configures dedicated channel with the configuration: conversational/speech/ ULuplink :12.2 DL:12.2 kbps/CS RAB + ULuplink :3.4 DL3.4 kbps SRBs in UTRAN cell.
3	+	-	HANDOVER TO UTRAN COMMAND	Send on cell 1 (GSM cell)
4	U	E		The UE accepts the handover command and configures its lower layers using the parameters contained in the HANDOVER TO UTRAN COMMAND
5	S	S		The SS waits for uplink physical channel in synchronization
6	=	>	HANDOVER TO UTRAN COMPLETE	The SS receives this message on DCCH of cell 2 (UTRAN cell). It implies that the down link physical channel has synchronised with UTRAN.

Specific message contents

HANDOVER TO UTRAN COMMAND

Information Element	Value/remark
RR management Protocol Discriminator	'0110'B
Skip Indicator	'0000'B
Inter System to UTRAN Handover Command Message Type	'01100011'B
Handover to UTRAN Command IEI	TBD
Length of Handover to UTRAN Command contents	Octet length of the "Handover to UTRAN Command value part"
Handover to UTRAN Command value part	PER encoded ASN.1 value of type "HandoverToUTRANCommand-v1-IEs", content is presented in the next table.

Information Element	Value/remark	
New U-RNTI		
- SRNC Identiy	'00000000001'B	
- S-RNTI-2	1	
Activation time	NOW	
Ciphering algorithm RAB Info	Standard UMTS Encryption Algorithm UEA1	
- RAB identity		
- GSM-MAP RAB identity	'0000001'B	
- CN domain identity	CS domain	
CHOICE Specification mode	complete	
- re-Establishment timer	1800 seconds	
- SRB information setup list	4	
- RB identity - CHOICE RLC info choice	1 RLC info	
- CHOICE WLuplink RLC mode	UM RLC mode	
- Transmission RLC discard	MaxDAT retransmissions	
- Max_DAT	4	
- Timer_MRW	100	
- MaxMRW	4	
- CHOICE DL RLC mode	UM RLC mode	
- RB mapping info - CHOICE ULuplink logical channel mappings	One logical channel	
- ULuplink transport channel type	DCH	
- Transport channel identity	1	
- Logical channel identity	1	
 MAC logical channel priority 	1	
- Logical channel max loss	0	
- DL logical channel mapping	BOLL	
DL transport channel type Transport channel identity	DCH 1	
- Transport channel identity - Logical channel identity		
- RB identity	2	
- CHOICE RLC info choice	RLC info	
- CHOICE UL uplink RLC mode	AM RLC mode	
- Transmission RLC discard	MaxDAT retransmissions	
- Max_DAT	4	
- Timer_MRW - MaxMRW	100	
- transmission window size	8	
- Timer_RST	500	
- MAX_RST	4	
- Polling info		
- Timer_poll_prohibit	200	
- Timer_poll	200	
- Poll_SDU - Last transmission PU poll	1 TRUE	
- Last retransmission PU poll	TRUE	
- Poll windows	99	
- CHOICE DL RLC mode	AM RLC mode	
- In-sequence delivery	TRUE	
- Receiving window size	8	
- Downlink RLC status info	200	
- Timer_status_prohibit - Timer_EPC	200 200	
- Missing PU indicator	TRUE	
- RB mapping info		
 CHOICE UL<u>uplink</u> logical channel mappings 	One logical channel	
 - ULuplink transport channel type 	DCH	
- Transport channel identity	1	
- Logical channel identity	2	
- MAC logical channel priority - Logical channel max loss	2 0	
- Logical channel max loss - DL logical channel mapping		
- DL transport channel type	DCH	
- Transport channel identity	1	

- Logical channel identity	2
- RB identity	3
- CHOICE RLC info choice	RLC info
- CHOICE ULuplink RLC mode	AM RLC mode
- Transmission RLC discard	MaxDAT retransmissions
- Max_DAT	4
- Timer_MRW	100
- MaxMRW	4
 transmission window size 	8
- Timer_RST	500
- MAX_RST	4
- Polling info	
- Timer_poll_prohibit	200
- Timer_poll	200
- Poll_SDU	1
 Last transmission PU poll 	TRUE
- Last retransmission PU poll	TRUE
- Poll windows	99
- CHOICE DL RLC mode	AM RLC mode
- In-sequence delivery	TRUE
 Receiving window size 	8
 Downlink RLC status info 	
- Timer_status_prohibit	200
- Timer EPC	200
	TRUE
- Missing PU indicator	IRUE
- RB mapping info	
 CHOICE ULuplink logical channel mappings 	One logical channel
- ULuplink transport channel type	DCH
- Transport channel identity	1
- Logical channel identity	3
 MAC logical channel priority 	3
 Logical channel max loss 	0
- DL logical channel mapping	
- DL transport channel type	DCH
- Transport channel identity	1
- Logical channel identity	3
- RB identity	4
- CHOICE RLC info choice	RLC info
- CHOICE UL uplink RLC mode	AM RLC mode
- Transmission RLC discard	MaxDAT retransmissions
- Max_DAT	4
- Timer_MRW	100
- MaxMRW	4
 transmission window size 	8
- Timer_RST	500
- MAX_RST	4
	4
- Polling info	
Timer_poll_prohibit	200
- Timer_poll	200
- Poll_SDU	1
- Last transmission PU poll	TRUE
- Last retransmission PU poll	TRUE
- Poll windows	99
- CHOICE DL RLC mode	AM RLC mode
- In-sequence delivery	TRUE
- Receiving window size	8
- Downlink RLC status info	
	200
- Timer_status_prohibit	200
- Timer_EPC	200
 Missing PU indicator 	TRUE
- RB mapping info	
- CHOICE ULuplink logical channel mappings	One logical channel
- UL uplink transport channel type	DCH
- Transport channel identity	1
 Logical channel identity 	4
 MAC logical channel priority 	4
- Logical channel max loss	0
- DL logical channel mapping	
- DL transport channel type	DCH
DE transport charmer type	DOM

- Transport channel identity	1
- Logical channel identity	4
- RB information setup list	
- RB information to setup	
- RB identity	10
- PDCP info	Not Present
- RLC info	
- CHOICE Uplink RLC mode	TM RLC mode
- Transmission RLC discard	Not Present
- Segmentation indication	TRUE
- CHOICE Downlink RLC mode	TM RLC mode
- Segmentation indication	TRUE
- RB mapping info	
- CHOICE <u>ULuplink</u> logical channel mappings	One logical channel
- Uplink transport channel type	DCH
- Transport channel identity	2
 Logical channel identity MAC logical channel priority 	1 1
- Logical channel max loss	0
- DL logical channel mapping	1
- Downlink transport channel type	DCH
- Transport channel identity	2
- Logical channel identity	1
- RB information to setup	
- RB identity	11
- PDCP info	Not Present
- RLC info	
- CHOICE Uplink RLC mode	TM RLC mode
- Transmission RLC discard	Not Present
- Segmentation indication	TRUE
- CHOICE Downlink RLC mode	TM RLC mode
- Segmentation indication	TRUE
- RB mapping info	
 CHOICE UL<u>uplink</u> logical channel mappings 	One logical channel
 Uplink transport channel type 	DCH
- Transport channel identity	3
- Logical channel identity	1
- MAC logical channel priority	1
- Logical channel max loss	0 1
DL logical channel mapping Downlink transport channel type	DCH
- Transport channel identity	3
- Logical channel identity	1
- RB information to setup	
- RB identity	12
- PDCP info	Not Present
- RLC info	
- CHOICE Uplink RLC mode	TM RLC mode
- Transmission RLC discard	Not Present
- Segmentation indication	TRUE
- CHOICE Downlink RLC mode	TM RLC mode
- Segmentation indication	TRUE
- RB mapping info	
- CHOICE UL uplink logical channel mappings	One logical channel
 Uplink transport channel type 	DCH
- Transport channel identity	4
- Logical channel identity	1
- MAC logical channel priority	1
- Logical channel max loss	0
- DL logical channel mapping	1
- Downlink transport channel type	DCH
- Transport channel identity	4
- Logical channel identity	1
- ULuplink Common Transport channel Info	
- TFC subset	0 1 2 3 4 5
 Allowed Transport Format combination CHOICE Mode specific info 	0, 1, 2, 3, 4, 5
- CHOICE IVIOUE SPECIFIC ITIO	FDD
	Normal signalling
- CHOICE ULuplink DCH TFCS - CHOICE Normal signalling	Normal signalling Addition

- CHOICE CTFC Size	6 bits
- CTFC information	0
- Power offset information	
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor &c	0
- Gain factor ßd	0
- Reference TFC ID	Not Present
- Power offset Pp-m	0dB
- CTFC information	1
- Power offset information	
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor ßc	0
- Gain factor ßd	0
- Reference TFC ID	Not Present
 Power offset Pp-m 	0dB
 CTFC information 	11
 Power offset information 	
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor ßc	0
- Gain factor ßd	0
- Reference TFC ID	Not Present
- Power offset Pp-m	0dB
- CTFC information	12
- Power offset information	0: " 10 : 5 /
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor &c	0
- Gain factor ßd	0 Not Drogent
- Reference TFC ID	Not Present 0dB
Power offset Pp-mCTFC information	13
- Power offset information	10
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor &c	
- Gain factor ßd	0
- Reference TFC ID	Not Present
- Power offset Pp-m	0dB
- CTFC information	23
 Power offset information 	
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor ßc	0
- Gain factor ßd	0
- Reference TFC ID	Not Present
- Power offset Pp-m	0dB
 - ULuplink Add or reconfigured transport channel info list 	
- Transport channel identity	2
- TFS	
- Dynamic Transport format information	
- Number of Transport blocks	0
- RLC size	81
- Dynamic Transport format information	
- Number of Transport blocks	1
- RLC size	39
 Dynamic Transport format information 	
 Number of Transport blocks 	1
- RLC size	81
 Semi-static Transport Format information 	
- Transmission time interval	20 ms
- Type of channel coding	convolutional
- Coding Rate	1/3
- Rate matching attribute	180-220
- CRC size	12 bits
 Transport channel identity TFS 	3
- TFS - Dynamic Transport format information	
- Number of Transport blocks	0
- RLC size	103
- Dynamic Transport format information	1
- Number of Transport blocks	1
·	

l pro:	1.400
- RLC size	103
- Semi-static Transport Format information	
- Transmission time interval	20 ms
- Type of channel coding	convolutional
- Coding Rate	1/3
- Rate matching attribute	170-210
- CRC size	N/A
- Transport channel identity	4
- TFS	
- Dynamic Transport format information	
- Number of Transport blocks	0
- RLC size	60
 Dynamic Transport format information 	
 Number of Transport blocks 	1
- RLC size	60
 Semi-static Transport Format information 	
- Transmission time interval	20 ms
- Type of channel coding	convolutional
- Coding Rate	1/2
- Rate matching attribute	215-256
- CRC size	N/A
- Transport channel identity	1
- TFS	
- Dynamic Transport format information	
- Number of Transport blocks	0
- RLC size	148
- Dynamic Transport format information	
- Number of Transport blocks	1
- RLC size	148
- Semi-static Transport Format information	
- Transmission time interval	40 ms
- Type of channel coding	convolutional
- Coding Rate	1/3
- Rate matching attribute	155-165
- CRC size	16 bits
- DL Common transport channel info	10 bits
- SCCPCH TFCS	Not Present
- CHOICE Mode specific info	FDD
- CHOICE Widde specific fillo - CHOICE TFCS Signalling mode	Same as UL uplink
- Added or reconfigured DL TrCH info list	Same as OL upinik
- Transport channel identity	2
- CHOICE TFS signalling mode	SameAsUL
	2
- ULuplink TrCH Identity - DCH quality target	2
- BLER Quality value	0
	Not Present
- Transparent mode signalling info	Not Fresent
- Addide or reconfigured DL TrCH info list	
- Transport channel identity	3
- CHOICE TFS signalling mode	SameAsUL
- ULuplink TrCH Identity	3
- DCH quality target	
- BLER Quality value	0 Not Broomt
- Transparent mode signalling info	Not Present
- Added or reconfigured DL TrCH info list	
- Transport channel identity	4
- CHOICE TFS signalling mode	SameAsUL
- ULuplink TrCH Identity	4
- DCH quality target	
- BLER Quality value	0
BLER Quality valueTransparent mode signalling info	0 Not Present
 BLER Quality value Transparent mode signalling info ULuplink DPCH Info 	
 BLER Quality value Transparent mode signalling info <u>ULuplink</u> DPCH Info Uplink DPCH power control info 	Not Present
 BLER Quality value Transparent mode signalling info <u>ULuplink</u> DPCH Info Uplink DPCH power control info DPCCH power offset 	Not Present -6dB
 BLER Quality value Transparent mode signalling info ULuplink DPCH Info Uplink DPCH power control info DPCCH power offset PC Preamble 	Not Present -6dB 15 slots
- BLER Quality value - Transparent mode signalling info - ULuplink DPCH Info - Uplink DPCH power control info - DPCCH power offset - PC Preamble - Power Control Algorithm	Not Present -6dB 15 slots Algorithm1
- BLER Quality value - Transparent mode signalling info - ULuplink DPCH Info - Uplink DPCH power control info - DPCCH power offset - PC Preamble - Power Control Algorithm - TPC step size	Not Present -6dB 15 slots Algorithm1 1dB
- BLER Quality value - Transparent mode signalling info - ULuplink DPCH Info - Uplink DPCH power control info - DPCCH power offset - PC Preamble - Power Control Algorithm - TPC step size - CHOICE Mode specific info	-6dB 15 slots Algorithm1 1dB FDD
- BLER Quality value - Transparent mode signalling info - ULuplink DPCH Info - Uplink DPCH power control info - DPCCH power offset - PC Preamble - Power Control Algorithm - TPC step size - CHOICE Mode specific info - Scrambling code type	-6dB 15 slots Algorithm1 1dB FDD Long
- BLER Quality value - Transparent mode signalling info - ULuplink DPCH Info - Uplink DPCH power control info - DPCCH power offset - PC Preamble - Power Control Algorithm - TPC step size - CHOICE Mode specific info	-6dB 15 slots Algorithm1 1dB FDD

- Number of DPDCH	Not Present(1)
- spreading factor	64
- TFCI existence	TRUE
- Number of FBI bit	Not Present(0)
	0.88
- Puncturing Limit	FDD
- CHOICE Mode specific info	'
- DL PDSCH information	Not present
- CPCH SET info	Not present
- Downlink information common for all radio links	
 Downlink DPCH info common for all RL 	
- Timing indication	initialize
 CFN target SFN frame offset 	Not present
- CHOICE Mode specific info	FDD
 Downlink DPCH power control information 	
- DPC mode	0 (single)
- Spreading factor and pilot	128
- Number of bits for Pilot bits	4 bits
- Fixed or Flexible Position	Fixed
- TFCI existence	FALSE
- Downlink DPCH Offset Value	0
	O .
- DPCH compressed mode info	4
-TGPSI	1
-TGPS Status Flag	Inactive
- Transmission gap pattern sequence	
configuration parameters	
- TGMP	FDD Measurement
- TGPRC	62
- TGCFN	(Current CFN + (256 – TTI/10msec)) mod 256
- TGSN	8
- TGL1	10
- TGL2	5
- TGD	15
- TGPL1	35
- TGPL2	35
- RPP	Mode 1
- ITP	Mode 1
- UL uplink/DL Mode	DL
- Downlink frame type	A
- DeltaSIR1	2.0
- DeltaSIRafter1	1.0
- DeltaSIR2	Not Present
- DeltaSIRafter2	Not Present
- TX Diversity mode - SSDT information	None Not Propert
	Not Present
- S field	
- DL information perRL list	
- Premary CPICH info	100
- Primary scrambling code	100
- DL DPCH info perRL	l
 pCPICH usage for channelEst 	May be used
- DL channelisation code	
- Secondary scrambling code	1
- SF and code number	SF = 128, code number = 127
- Scrambling code change	No code change
- TPC combination index	0
- Frequency info	
- UARFCN uplink	See PIXIT
- UARFCN downlink	See PIXIT
- Maximum allowed ULuplink transmission power	33 dbm

8.3.6.5.5 Test requirement

At step 6 the HANDOVER TO UTRAN COMPLETE shall be received on UTRAN cell.

8.3.6.6 Inter system handover to UTRAN/From GSM/Speech/Failure

8.3.6.6.1 Definition

8.3.6.6.2 Conformance requirement

The UE shall be able to receive a HANDOVER TO UTRAN COMMAND message from GSM and perform an intersystem handover, even if no prior UE measurements have been performed on the target UTRAN cell and/or frequency.

If the UE can not establish the connection to UTRAN, it shall reactivate the old channel and transmit a HANDOVER FAILURE message on the old channel.

Reference(s)

TS 25.331 Clause 8.3.6.

TS 04.18 Clause 3.4.4a.

8.3.6.6.3 Test purpose

To test that the UE reactivates the old channel and transmits HANDOVER FAILURE message to the network on the old channel in the GSM cell when it received HANDOVER TO UTRAN COMMAND and the handover to UTRAN failed.

8.3.6.6.4 Method of test

Initial conditions

System Simulator: 2 cells - Cell 1 is GSM, Cell 2 is UTRAN. GSM 11.10-1 section 26.6.5.1 shall be referenced for the default parameters of cell 1.

UE: CC State U10 in cell 1

Related ICS/IXIT statement(s)

UE supports both GSM and UTRAN Radio Access Technologies,

UE supports UTRAN AMR,

UE supports GSM RF,

UE supports GSM-P, GSM-E, GSM-DCS, GSM-450, GSM-480.

Foreseen final state of the UE

The UE is in CC state U10 on cell 1.

Test Procedure

The SS starts the GSM cell and the UTRAN cell with cell selection conditions in favour of GSM cell. SIB16 is broadcast in UTRAN cell. The UE selects the GSM cell, and received the SIB16. Then the SS brings the UE into the call active state (CC state U10) with FR speech call. The SS does not configure the dedicated channel corresponding to the predefined configuration described in SIB16 (conversationa/speech/ULuplink:12.2 DL:12.2 kbps/CS RAB + ULuplink:3.4 DL3.4 kbps SRBS), then sends HANDOVER TO UTRAN COMMAND indicating the dedicated channel of the target cell to the UE through the GSM serving cell. The UE will not be able to establish the connection to UTRAN. The SS checks that the handover is failed by checking that the UE returns to the old channel and transmits HANDOVER FAILURE to the SS through the old channel.

Expected sequence

Step	Direction	Message	Comments
	UE SS		
1	SS		The SS starts GSM and UTRAN cells, SIB16 is broadcast in the UTRAN cell. The UE camps on GSM cell and received SIB16.
2	UE		The SS bring the UE into GSM U10 state in cell 1
3	SS		There is no dedicated channel with the configuration: conversational/speech/ ULuplink :12.2 DL:12.2 kbps/CS RAB + ULuplink :3.4 DL3.4 kbps SRBs in UTRAN cell.
4	←	HANDOVER TO UTRAN COMMAND	Send on cell 1 (GSM cell)
5	UE		The UE accepts the handover command and configures its lower layers using the parameters contained in the HANDOVER TO UTRAN COMMAND
6	UE		The UE fails to establish a connection to UTRAN cell
7		HANDOVER FAILURE	The SS receives this message on DCCH of cell 1 (old channel in GSM cell)

Specific message contents

Same as the specific message contents in clause 8.3.6.1 for M = 1.

8.3.6.6.5 Test requirement

At step 7 the HANDOVER FAILURE shall be received on GSM cell.

8.3.7 Inter-system hard handover from UTRAN to GSM

Clauses 8.3.7 contains test procedures to be used for executing Inter-system Handover from UTRAN to GSM tests. Table 8.3.7-1 contains a summary of the different combinations of parameters being tested, together with a reference to the appropriate generic test procedure. If a test uses a parameter which the UE under test does not support, the test shall be skipped. Test cases in this clause are applicable only to the UE supporting both UTRAN and GSM. The test USIM shall support service 27 to carry out these test cases.

Table 8.3.7-1

From	То	State of call	Ref. clause	Exec counter	Remark
UTRAN AMR (conversational/speech/ ULuplink:12.2 DL:12.2 kbps/CS RAB + ULuplink:3.4 DL3.4 kbps SRBS)	GSM AMR	U10	8.3.7.1	1	call active state
UTRAN AMR (conversational/speech/ ULuplink:12.2 DL:12.2 kbps/CS RAB + ULuplink:3.4 DL3.4 kbps SRBS)	GSM EFR	U10	8.3.7.1	2	call active state
UTRAN AMR (conversational/speech/ ULuplink:12.2 DL:12.2 kbps/CS RAB + ULuplink:3.4 DL3.4 kbps SRBS)	GSM FR	U10	8.3.7.1	3	call active state
UTRAN AMR (conversational/speech/ ULuplink:12.2 DL:12.2 kbps/CS RAB + ULuplink:3.4 DL3.4 kbps SRBS)	GSM HR	U10	8.3.7.1	4	call active state
UTRAN (Streaming/unknown/ ULuplink:14.4 DL:14.4 kbps/CS RAB + ULuplink:3.4 DL:3.4 kbps SRBS)	GSM 14.4 kbps CS data	U10	8.3.7.2	1	Same data rate
UTRAN (Streaming/unknown/ ULuplink:28.8 DL:28.8 kbps/CS RAB + ULuplink:3.4 DL:3.4 kbps SRBS)	GSM 28.8 kbps CS data	U10	8.3.7.2	2	Same data rate
UTRAN (Streaming/unknown/ ULuplink:57.6 DL:57.6 kbps/CS RAB + ULuplink:3.4 DL:3.4 kbps SRBS)	GSM 57.6 kbps CS data	U10	8.3.7.2	3	Same data rate
UTRAN (Streaming/unknown/ ULuplink:28.8 DL:28.8 kbps/CS RAB + ULuplink:3.4 DL:3.4 kbps SRBS)	GSM 14.4 kbps CS data	U10	8.3.7.3	1	Data rate down grading
UTRAN (Streaming/unknown/ ULuplink:57.6 DL:57.6 kbps/CS RAB + ULuplink:3.4 DL:3.4 kbps SRBS)	GSM 14.4 kbps CS data	U10	8.3.7.3	2	Data rate down grading
UTRAN (Streaming/unknown/ ULuplink:57.6 DL:57.6 kbps/CS RAB + ULuplink:3.4 DL:3.4 kbps SRBS)	GSM 28.8 kbps CS data	U10	8.3.7.3	3	Data rate down grading
UTRAN AMR (conversational/speech/ ULuplink:12.2 DL:12.2 kbps/CS RAB + ULuplink:3.4 DL3.4 kbps SRBS)	GSM FR	U1	8.3.7.4	1	During call establishment
UTRAN AMR (conversational/speech/ ULuplink:12.2 DL:12.2 kbps/CS RAB + ULuplink:3.4 DL3.4 kbps SRBS)	GSM FR	U10	8.3.7.5	1	failure case

8.3.7.1 Inter system handover from UTRAN/To GSM/Speech/Success

8.3.7.1.1 Definition

8.3.7.1.2 Conformance requirement

When the UE receives an INTER-SYSTEM HANDOVER COMMAND message from UTRAN the UE shall take the following actions:

- Establish the connection to the other radio access system, by using the contents of the IE "Inter system message".
 This IE contains candidate/ target cell identifier(s) and radio parameters relevant for the other radio access system.
- For each IE "Remaining radio access bearer", associate the radio access bearer given by the IE "RAB info" to the radio resources in the target system given by the IE "Inter system message". Other information for making the association may be included in the IE "Inter system message" and requirements may be stated in the specifications relevant for the target system [FFS].
- Switch the current connection to the other radio access system.
- NOTE 1: Requirements concerning the establishment of the radio connection towards the other radio access system and the signalling procedure are outside the scope of this specification.
- NOTE 2: The release of the UMTS radio resources is initiated by the other system.
- NOTE 3: Currently only one radio access bearer can be associated with the IE "Inter-system message", and this association is limited to the radio access bearers in the CS domain. It is assumed that all the radio access bearers in the PS domain, if any, remain after the handover.

Reference(s)

TS 25.331 Clause 8.3.7.3.

8.3.7.1.3 Test purpose

To test that the UE supporting both GSM and UTRAN handovers from a UTRAN serving cell to the indicated channel of GSM target cell when the UE is in the speech call active state and receives an INTER-SYSTEM HANDOVER COMMAND.

8.3.7.1.4 Method of test

Initial conditions

System Simulator: 2 cells - Cell 1 is UTRAN, Cell 2 is GSM. GSM 11.10-1 section 26.6.5.1 shall be referenced for the default parameters of cell 2.

UE: CC State U10 in cell 1

Related ICS/IXIT statement(s)

UE supports both GSM and UTRAN Radio Access Technologies,

UE supports GSM ARM,

UE supports GSM EFR,

UE supports GSM HR,

UE supports GSM-P, GSM-E, GSM-DCS, GSM-450, GSM-480.

Foreseen final state of the UE

The UE is in CC state U10 on cell 2.

Test Procedure

The SS starts the UTRAN cell and brings the UE into call active state (CC state U10) with AMR. The SS starts GSM cell and configures a traffic channel, then sends INTER-SYSTEM HANDOVER COMMAND indicating the traffic channel of the target GSM cell to the UE through DCCH of the serving UTRAN cell. After the UE receives the command it shall configure itself accordingly and switch to the new channel of the UTRAN cell. The SS checks whether the handover is performed by checking that the UE transmits the HANDOVER COMPLETE message to the SS through GSM cell.

Depending on the PIXIT parameters the above procedure is executed maximum four times, each time with different target channel in the GSM cell.

Expected sequence

This sequence is performed for a maximum execution counter M = 1, 2, 3, 4, depending on the PIXIT parameters.

Step	Direction	Message	Comments
	UE SS		
1	UE		The SS bring the UE into UTRAN U10 state in cell 1
2	SS		The SS configures cell 2 as a GSM cell with a traffic
			channel:
			for GSM AMR $(M = 1)$; or
			for GSM EFR (M = 2); or
			for GSM FR $(M = 3)$; or
			for GSM HR $(M = 4)$.
3	←	InterSystemHandoverCommand-	Send on cell 1 (UTRAN cell) and the message indicates:
		GSM	the target channel for GSM AMR (M = 1); or
			the target channel for GSM EFR (M = 2); or
			the target channel for GSM FR (M = 3); or
			the target channel for GSM HR (M = 4).
4	UE		The UE accepts the handover command and switches to
			the GSM traffic channel specified in the
			InterSystemHandoverCommand-GSM
5	\rightarrow	HANDOVER ACCESS	The SS receives this burst on the traffic channel of cell 2
			(GSM cell) It implies that the UE has switched to GSM
			cell.
6	\rightarrow	HANDOVER ACCESS	
7	\rightarrow	HANDOVER ACCESS	
8	\rightarrow	HANDOVER ACCESS	
9	+	PHYSICAL INFORMATION	
10	\rightarrow	SABM	
11	←	UA	
12	→	HANDOVER COMPLETE	The SS receives the message on the traffic channel of GSM cell.

Specific message contents

For execution:

InterSystemHandoverCommand-GSM

Information Element	Value/remark
Message Type	
Integrity check info	
 Message authentication code 	
 Message sequence number 	
Activation time	now
RAB Info	Not present
Inter-system message	
- System type	GSM
- CHOICE system	GSM
- Message	GSM HANDOVER COMMAND formatted as BIT
	STRING(1512). The contents of the HANDOVER
	COMMAND see next table.

HANDOVER COMMAND

Same as the HANDOVER COMMAND for M = 2 in clause 26.6.5.1 of GSM 11.10-1, Release 1999, except that the CHANNEL MODE IE is included with value = speech full rate or half rate version 3

For execution 2:

InterSystemHandoverCommand-GSM

Information Element	Value/remark
Message Type	
Integrity check info	
- Message authentication code	
- Message sequence number	
Activation time	now
RAB Info	Not present
Inter-system message	
- System type	GSM
- CHOICE system	GSM
- Message	GSM HANDOVER COMMAND formatted as BIT
	STRING(1512). The contents of the HANDOVER
	COMMAND see next table.

HANDOVER COMMAND

Same as the HANDOVER COMMAND for M = 2 in clause 26.6.5.1 of GSM 11.10-1, Release 1999, except that the CHANNEL MODE IE is included with value = speech full rate or half rate version 2

For execution 3:

InterSystemHandoverCommand-GSM

Information Element	Value/remark
Message Type	
Integrity check info	
- Message authentication code	
- Message sequence number	
Activation time	now
RAB Info	Not present
Inter-system message	
- System type	GSM
- CHOICE system	GSM
- Message	GSM HANDOVER COMMAND formatted as BIT
, and the second	STRING(1512). The contents of the HANDOVER
	COMMAND see next table.

HANDOVER COMMAND

Same as the HANDOVER COMMAND for M = 2 in clause 26.6.5.1 of GSM 11.10-1, Release 1999, except that the CHANNEL MODE IE is included with value = speech full rate or half rate version 1

For execution 4:

InterSystemHandoverCommand-GSM

Information Element	Value/remark
Message Type	
Integrity check info	
- Message authentication code	
- Message sequence number	
Activation time	now
RAB Info	Not present
Inter-system message	
- System type	GSM
- CHOICE system	GSM
- Message	GSM HANDOVER COMMAND formatted as BIT
	STRING(1512). The contents of the HANDOVER
	COMMAND see next table.

HANDOVER COMMAND

Same as the HANDOVER COMMAND for M = 4 in clause 26.6.5.1 of GSM 11.10-1 version 8.2.0 Release 1999, except that the CHANNEL MODE IE is included with value = speech full rate or half rate version 1

8.3.7.1.5 Test requirement

After step 12 the ongoing call shall be continued on the GSM cell.

8.3.7.2 Inter system handover from UTRAN/To GSM/Data/Same data rate/Success

8.3.7.2.1 Definition

8.3.7.2.2 Conformance requirement

When the UE receives an INTER-SYSTEM HANDOVER COMMAND message from UTRAN the UE shall take the following actions:

- Establish the connection to the other radio access system, by using the contents of the IE "Inter system message".
 This IE contains candidate/ target cell identifier(s) and radio parameters relevant for the other radio access system.
- For each IE "Remaining radio access bearer", associate the radio access bearer given by the IE "RAB info" to the radio resources in the target system given by the IE "Inter system message". Other information for making the association may be included in the IE "Inter system message" and requirements may be stated in the specifications relevant for the target system [FFS].
- Switch the current connection to the other radio access system.
- NOTE 1: Requirements concerning the establishment of the radio connection towards the other radio access system and the signalling procedure are outside the scope of this specification.
- NOTE 2: The release of the UMTS radio resources is initiated by the other system.
- NOTE 3: Currently only one radio access bearer can be associated with the IE "Inter-system message", and this association is limited to the radio access bearers in the CS domain. It is assumed that all the radio access bearers in the PS domain, if any, remain after the handover.

Reference(s)

TS 25.331 Clause 8.3.7.3.

8.3.7.2.3 Test purpose

To test that the UE handovers to the indicated channel of same data rate in the GSM target cell when it is in the data call active state in the UTRAN serving cell and receives an INTER-SYSTEM HANDOVER COMMAND.

8.3.7.2.4 Method of test

Initial conditions

System Simulator: 2 cells - Cell 1 is UTRAN, Cell 2 is GSM. GSM 11.10-1 section 26.6.5.1 or section 26.13.1.3 (for HSCSD) shall be referenced for the default parameters of cell 2.

UE: CC State U10 in cell 1

Related ICS/IXIT statement(s)

UE supports both GSM and UTRAN Radio Access Technologies,

UE supports UTRAN Streaming/unknown/ULuplink:14.4 DL:14.4 kbps/CS RAB + ULuplink:3.4 DL:3.4 kbps SRBs,

UE supports UTRAN Streaming/unknown/ULuplink:28.8 DL:28.8 kbps/CS RAB + ULuplink:3.4 DL:3.4 kbps SRBs,

UE supports UTRAN Streaming/unknown/ULuplink:57.6 DL:57.6 kbps/CS RAB + ULuplink:3.4 DL:3.4 kbps SRBs,

UE supports GSM 14.4 kbps data (HSCSD or full rate traffic channel for 14.4 kbit/s user data (TCH/F14.4)),

UE supports GSM 28.8 kbps data (HSCSD or enhanced circuit switched full rate traffic channel for 28.8 kbit/s user date (E-TCH/F28.8)),

UE supports GSM 57.6 kbps data,

UE supports GSM-P, GSM-E, GSM-DCS, GSM-450, GSM-480.

Foreseen final state of the UE

The UE is in CC state U10 on cell 2.

Test Procedure

The SS starts the UTRAN cell and brings the UE into data call active state (CC state U10) with a suitable configuration (e.g. Streaming/unknown/ $\frac{ULuplink}{14.4}$ DL:14.4 kbps/CS RAB + $\frac{ULuplink}{14.4}$ DL:3.4 kbps SRBs for M = 1). The SS starts GSM cell and configures a traffic channel (e.g. 14.4 kbps data channel for M = 1), then sends INTER-SYSTEM HANDOVER COMMAND indicating the traffic channel of the target GSM cell to the UE through DCCH of the serving UTRAN cell. After the UE receives the command it shall configure itself accordingly and switch to the new channel of the GSM cell. The SS checks whether the handover is performed by checking that the UE transmits the HANDOVER COMPLETE message to the SS in GSM cell.

Depending on the PIXIT parameters the above procedure is executed maximum three times, each time with different target channel in the GSM cell.

Expected sequence

This sequence is performed for a maximum execution counter M = 1, 2, 3, depending on the PIXIT parameters.

Step	Direction		Message	Comments
	UE	SS		
1	UE	Ξ		The SS bring the UE into UTRAN U10 state in cell 1, the configuration is: Streaming/unknown/ULuplink:14.4 DL:14.4 kbps/CS RAB + ULuplink:3.4 DL:3.4 kbps SRBs (for M = 1); Streaming/unknown/ULuplink:28.8 DL:28.8 kbps/CS RAB + ULuplink:3.4 DL:3.4 kbps SRBs (for M = 2); Streaming/unknown/ULuplink:57.6 DL:57.6 kbps/CS RAB + ULuplink:3.4 DL:3.4 kbps SRBs (for M = 3).
2	SS	8		The SS configures cell 2 as a GSM cell with a traffic channel: for GSM 14.4 kbps data (M = 1); or for GSM 28.8 kbps data (M = 2); or for GSM 57.6 kbps data (M = 3).
3	+	-	InterSystemHandoverCommand-GSM	Send on cell 1 (UTRAN cell) and the message indicates: the target channel for GSM 14.4 kbps data (M = 1); or for GSM 28.8 kbps data (M = 2); or for GSM 57.6 kbps data (M = 3).
4	UE			The UE accepts the handover command and switches to the GSM traffic channel specified in the InterSystemHandoverCommand-GSM
5)	•	HANDOVER ACCESS	The SS receives this burst on the traffic channel of cell 2 (GSM cell) It implies that the UE has switched to GSM cell.
6	\rightarrow	,	HANDOVER ACCESS	
7	\rightarrow	•	HANDOVER ACCESS	
8	\rightarrow	•	HANDOVER ACCESS	
9	+		PHYSICAL INFORMATION	
10	\rightarrow		SABM	
11	+		UA	
12	→	•	HANDOVER COMPLETE	The SS receives the message on the traffic channel of GSM cell.

Specific message contents

For execution:

InterSystemHandoverCommand-GSM

Information Element	Value/remark
Message Type Integrity check info - Message authentication code - Message sequence number Activation time RAB Info Inter-system message - System type - CHOICE system - Message	now Not present GSM GSM GSM HANDOVER COMMAND formatted as BIT STRING(1512). The contents of the HANDOVER COMMAND see next table.

If the UE supports 14.4 kbps single slot:

HANDOVER COMMAND

Same as the HANDOVER COMMAND for M = 2 in clause 26.6.5.1 of GSM 11.10-1, Release 1999, except that the CHANNEL MODE IE is included with value = data, 14.5 kbit/s radio interface rate (14.4 kbit/s user data (TCH/F14.4))

If the UE supports HSCSD:

HANDOVER COMMAND

Same as the HANDOVER COMMAND in clause 26.13.3.1 of GSM 11.10-1, Release 1999, except that the Description of a multislot configuration supporting 14.4 kbps user data.

For execution 2:

InterSystemHandoverCommand-GSM

Value/remark
now
Not present
GSM
GSM
GSM HANDOVER COMMAND formatted as BIT
STRING(1512). The contents of the HANDOVER
COMMAND see next table.
•

If the UE supports enhanced circuit switched full rate traffic channel for 28.8 kbps user data:

HANDOVER COMMAND

Same as the HANDOVER COMMAND for M = 2 in clause 26.6.5.1 of GSM 11.10-1, Release 1999, except that the CHANNEL MODE IE is included with value = data, 29.0 kbit/s radio interface rate (28.8 kbit/s user data (E-TCH/F28.8))

If the UE supports HSCSD:

HANDOVER COMMAND

Same as the HANDOVER COMMAND in clause 26.13.3.1 of GSM 11.10-1, Release 1999, except that the Description of a multislot configuration supporting 28.8 kbps user data.

For execution 3:

InterSystemHandoverCommand-GSM

Value/remark
now Not present GSM GSM GSM GSM HANDOVER COMMAND formatted as BIT
STRING(1512). The contents of the HANDOVER COMMAND see next table.

HANDOVER COMMAND

Same as the HANDOVER COMMAND in clause 26.13.3.1 of GSM 11.10-1, Release 1999, except that the Description of a multislot configuration supporting 57.6 kbps user data.

8.3.7.2.5 Test requirement

After step 12 the ongoing call shall be continued on the GSM cell.

8.3.7.3 Inter system handover from UTRAN/To GSM/Data/Data rate down grading/Success

8.3.7.3.1 Definition

8.3.7.3.2 Conformance requirement

When the UE receives an INTER-SYSTEM HANDOVER COMMAND message from UTRAN the UE shall take the following actions:

- Establish the connection to the other radio access system, by using the contents of the IE "Inter system message".
 This IE contains candidate/ target cell identifier(s) and radio parameters relevant for the other radio access system.
- For each IE "Remaining radio access bearer", associate the radio access bearer given by the IE "RAB info" to the radio resources in the target system given by the IE "Inter system message". Other information for making the association may be included in the IE "Inter system message" and requirements may be stated in the specifications relevant for the target system [FFS].
- Switch the current connection to the other radio access system.

- NOTE 1: Requirements concerning the establishment of the radio connection towards the other radio access system and the signalling procedure are outside the scope of this specification.
- NOTE 2: The release of the UMTS radio resources is initiated by the other system.
- NOTE 3: Currently only one radio access bearer can be associated with the IE "Inter-system message", and this association is limited to the radio access bearers in the CS domain. It is assumed that all the radio access bearers in the PS domain, if any, remain after the handover.

Reference(s)

TS 25.331 Clause 8.3.7.3.

8.3.7.3.3 Test purpose

To test that the UE handovers to the indicated channel of lower data rate in the GSM target cell when it is in the data call active state in the UTRAN serving cell and receives an INTER-SYSTEM HANDOVER COMMAND.

8.3.7.3.4 Method of test

Initial conditions

System Simulator: 2 cells - Cell 1 is UTRAN, Cell 2 is GSM. GSM 11.10-1 section 26.6.5.1 or section 26.13.1.3 (for HSCSD) shall be referenced for the default parameters of cell 2.

UE: CC State U10 in cell 1

Related ICS/IXIT statement(s)

UE supports both GSM and UTRAN Radio Access Technologies,

UE supports UTRAN Streaming/unknown/ULuplink:28.8 DL:28.8 kbps/CS RAB + ULuplink:3.4 DL:3.4 kbps SRBs,

UE supports UTRAN Streaming/unknown/ULuplink:57.6 DL:57.6 kbps/CS RAB + ULuplink:3.4 DL:3.4 kbps SRBs,

UE supports GSM 14.4 kbps data (HSCSD or full rate traffic channel for 14.4 kbit/s user data (TCH/F14.4)),

UE supports GSM 28.8 kbps data (HSCSD or enhanced circuit switched full rate traffic channel for 28.8 kbit/s user date (E-TCH/F28.8)),

UE supports GSM-P, GSM-E, GSM-DCS, GSM-450, GSM-480.

Foreseen final state of the UE

The UE is in CC state U10 on cell 2.

Test Procedure

The SS starts the UTRAN cell and brings the UE into data call active state (CC state U10) with a suitable configuration (e.g. Streaming/unknown/ $\frac{ULuplink}{28.8}$ DL:28.8 kbps/CS RAB + $\frac{ULuplink}{28.4}$ DL:3.4 kbps SRBs for M = 1). The SS starts GSM cell and configures a traffic channel (e.g. 14.4 kbps data channel for M = 1), then sends INTER-SYSTEM HANDOVER COMMAND indicating the traffic channel of the target GSM cell to the UE through DCCH of the serving UTRAN cell. After the UE receives the command it shall configure itself accordingly and switch to the new channel of the GSM cell. The SS checks whether the handover is performed by checking that the UE transmits the HANDOVER COMPLETE message to the SS in GSM cell.

Depending on the PIXIT parameters the above procedure is executed maximum three times, each time with different target channel in the GSM cell.

Expected sequence

This sequence is performed for a maximum execution counter M = 1, 2, 3, depending on the PIXIT parameters.

Step	Direction	Message	Comments		
	UE SS				
1	ÜĒ		The SS bring the UE into UTRAN U10 state in cell 1, the configuration is: Streaming/unknown/ ULuplink :28.8 DL:28.8 kbps/CS RAB + ULuplink :3.4 DL:3.4 kbps SRBs (for M = 1); Streaming/unknown/ ULuplink :57.6 DL:57.6 kbps/CS RAB + ULuplink :3.4 DL:3.4 kbps SRBs (for M = 2 and 3).		
2	SS		The SS configures cell 2 as a GSM cell with a traffic channel: for GSM 14.4 kbps data (M = 1 and 2); or for GSM 28.8 kbps data (M = 3).		
3	+	InterSystemHandoverCommand-GSM	Send on cell 1 (UTRAN cell) and the message indicates: the target channel for GSM 14.4 kbps data (M = 1 and 2); or for GSM 28.8 kbps data (M = 3).		
4	UE		The UE accepts the handover command and switches to the GSM traffic channel specified in the InterSystemHandoverCommand-GSM		
5	→	HANDOVER ACCESS	The SS receives this burst on the traffic channel of cell 2 (GSM cell) It implies that the UE has switched to GSM cell.		
6	→	HANDOVER ACCESS			
7	\rightarrow	HANDOVER ACCESS			
8	\rightarrow	HANDOVER ACCESS			
9	+	PHYSICAL INFORMATION			
10	\rightarrow	SABM			
11	←	UA			
12	\rightarrow	HANDOVER COMPLETE	The SS receives the message on the traffic channel of GSM cell.		

Specific message contents

For execution 1:

Same as the message contents of clause 8.3.7.2 for M = 1.

For execution 2:

Same as the message contents of clause 8.3.7.2 for M = 1.

For execution 3:

Same as the message contents of clause 8.3.7.2 for M=2.

8.3.7.3.5 Test requirement

After step 12 the ongoing call shall be continued on the GSM cell.

8.3.7.4 Inter system handover from UTRAN/To GSM/Speech/Establishment/Success

8.3.7.4.1 Definition

8.3.7.4.2 Conformance requirement

When the UE receives an INTER-SYSTEM HANDOVER COMMAND message from UTRAN the UE shall take the following actions:

- Establish the connection to the other radio access system, by using the contents of the IE "Inter system message". This IE contains candidate/ target cell identifier(s) and radio parameters relevant for the other radio access system.
- For each IE "Remaining radio access bearer", associate the radio access bearer given by the IE "RAB info" to the radio resources in the target system given by the IE "Inter system message". Other information for making the association may be included in the IE "Inter system message" and requirements may be stated in the specifications relevant for the target system [FFS].
- Switch the current connection to the other radio access system.
- NOTE 1: Requirements concerning the establishment of the radio connection towards the other radio access system and the signalling procedure are outside the scope of this specification.
- NOTE 2: The release of the UMTS radio resources is initiated by the other system.
- NOTE 3: Currently only one radio access bearer can be associated with the IE "Inter-system message", and this association is limited to the radio access bearers in the CS domain. It is assumed that all the radio access bearers in the PS domain, if any, remain after the handover.

Reference(s)

TS 25.331 Clause 8.3.7.3.

8.3.7.4.3 Test purpose

To test that the UE handovers to the indicated channel in the GSM target cell when it is in the call establishment phase in the UTRAN serving cell and receives an INTER-SYSTEM HANDOVER COMMAND.

8.3.7.4.4 Method of test

Initial conditions

System Simulator : 2 cells - Cell 1 is UTRAN, Cell 2 is GSM. GSM 11.10-1 section 26.6.5.1 shall be referenced for the default parameters of cell 2.

UE: CC State U1 in cell 1

Related ICS/IXIT statement(s)

UE supports both GSM and UTRAN Radio Access Technologies,

UE supports UTRAN AMR,

UE supports GSM FR,

UE supports GSM-P, GSM-E, GSM-DCS, GSM-450, GSM-480.

Foreseen final state of the UE

The UE is in CC state U1 on cell 2.

Test Procedure

The SS starts the UTRAN cell and the UE is triggered to initialise an MO speech call. During the call establishment phase, after the SS receives SETUP message the SS starts GSM cell and configures a dedicated channel, then sends the UE an INTER-SYSTEM HANDOVER COMMAND indicating the dedicated channel in the target GSM cell. After the UE receives the command it shall configure itself accordingly and switch to the new channel of the GSM cell. The SS checks whether the handover is performed by checking that the UE transmits the HANDOVER COMPLETE message to the SS in GSM cell.

Expected sequence

Step	Direction	Message	Comments	
	UE SS			
1	UE		To trigger the UE to initialise an MO call	
2	\rightarrow	SETUP	U1	
3	SS		The SS starts the GSM cell and configure a dedicated channel SDCCH.	
4	+	InterSystemHandoverCommand-GSM	Send on cell 1 (UTRAN cell) and the message indicates: the dedicated channel SDCCH.	
5	UE		The UE accepts the handover command and switches to the GSM dedicated channel specified in the InterSystemHandoverCommand-GSM	
6	\rightarrow	HANDOVER ACCESS	The SS receives this burst on the dedicated channel of cell 2 (GSM cell) It implies that the UE has switched to GSM cell.	
7	\rightarrow	HANDOVER ACCESS		
8	\rightarrow	HANDOVER ACCESS		
9	\rightarrow	HANDOVER ACCESS		
10	+	PHYSICAL INFORMATION		
11	\rightarrow	SABM		
12	+	UA		
13	\rightarrow	HANDOVER COMPLETE	The SS receives the message on the dedicated channel of GSM cell.	

Specific message contents

InterSystemHandoverCommand-GSM

Information Element	Value/remark
Message Type	
Integrity check info	
 Message authentication code 	
- Message sequence number	
Activation time	now
RAB Info	Not present
Inter-system message	
- System type	GSM
- CHOICE system	GSM
- Message	GSM HANDOVER COMMAND formatted as BIT
	STRING(1512). The contents of the HANDOVER
	COMMAND see next table.

HANDOVER COMMAND

Same as the HANDOVER COMMAND for M = 4 in clause 26.6.5.2 of GSM 11.10-1 version 8.2.0 Release 1999

8.3.7.4.5 Test requirement

At step 13 the SS shall receive HANDOVER COMPLETE message on the dedicated channel of the GSM cell.

8.3.7.5 Inter system handover from UTRAN/To GSM/Speech/Failure

8.3.7.5.1 Definition

8.3.7.5.2 Conformance requirement

If the UE does not succeed to establish the connection to the other radio access technology, it shall

- resume the connection to UTRAN using the resources used before receiving the INTER-SYSTEM HANDOVER COMMAND message; and
- transmit the INTER-SYSTEM HANDOVER FAILURE message on uplink DCCH using AM RLC. When the successful delivery of the INTER-SYSTEM FAILURE message has been confirmed by RLC, the procedure ends.

Reference(s)

TS 25.331 Clause 8.3.7.5.

8.3.7.5.3 Test purpose

To test that the UE reactivates the old channel and transmits INTER-SYSTEM HANDOVER FAILURE message to the network on the old channel in UTRAN cell when it receives an INTER-SYSTEM HANDOVER COMMAND and the connection to GSM for handover can not be established.

8.3.7.5.4 Method of test

Initial conditions

System Simulator: 2 cells - Cell 1 is UTRAN, Cell 2 is GSM. GSM 11.10-1 section 26.6.5.1 shall be referenced for the default parameters of cell 2.

UE: CC State U10 in cell 1

Related ICS/IXIT statement(s)

UE supports both GSM and UTRAN Radio Access Technologies,

UE supports GSM FR,

UE supports UTRAN AMR,

UE supports GSM-P, GSM-E, GSM-DCS, GSM-450, GSM-480.

Foreseen final state of the UE

The UE is in CC state U10 on cell 1.

Test Procedure

The SS starts the UTRAN cell and brings the UE into call active state (CC state U10) with AMR. The SS starts GSM cell without activating any dedicated channel in the cell, then sends INTER-SYSTEM HANDOVER COMMAND indicating a dedicated channel of the target GSM cell to the UE through DCCH of the serving UTRAN cell. The UE receives the command and configures itself accordingly but can not complete the handover. The SS checks that the handover is failed by checking that the UE transmits the INTER-SYSTEM HANDOVER FAILURE message to the SS in UTRAN cell.

Expected sequence

Step	Direction	Message	Comments	
	UE SS			
1	UE		The SS bring the UE into UTRAN U10 state in cell 1	
2	SS		The SS configures cell 2 as a GSM cell but without any traffic channel.	
3	←	InterSystemHandoverCommand- GSM	Send on cell 1 (UTRAN cell) and the message indicates: the target channel for GSM FR which does not exist in the GSM cell.	
4	UE		The UE accepts the handover command and switches to the GSM traffic channel specified in the InterSystemHandoverCommand-GSM	
5	→	HANDOVER ACCESS	The SS receives this burst on the traffic channel of cell 2 (GSM cell) It implies that the UE has switched to GSM cell.	
6	\rightarrow	HANDOVER ACCESS		
7	\rightarrow	HANDOVER ACCESS		
8	\rightarrow	HANDOVER ACCESS		
n	\rightarrow	HANDOVER ACCESS	The last handover access burst before T3124 times out.	
n+1	→	InterSystemHandoverFailure	The SS receives the message on the old channel of UTRAN cell.	

Specific message contents

Same as the message contents of clause 8.3.7.1 for M = 3.

8.3.7.5.5 Test requirement

After step n+1 the SS shall receive INTER-SYSTEM HANDOVER FAILURE message on the old channel of the UTRAN cell.

8.3.8 Inter system cell reselection to UTRAN

[Editor's note: This test is FFS until R2000 core specification will be defined.]

8.3.9 Inter system cell reselection from UTRAN

[Editor's note: This test is FFS until R2000 core specification will be defined.]

8.4 Measurement procedure

8.4.1 Measurement Control and Report

8.4.1.1 Measurement Control and Report: Intra-frequency measurement for transition from idle mode to CELL_DCH state

8.4.1.1.1 Definition

8.4.1.1.2 Conformance requirement

After a state transition from idle mode to CELL_DCH state, the UE shall continue to monitor the list of neighbouring cells which is specified in the SYSTEM INFORMATION BLOCK TYPESystem Information Block type 11 (or 12) messages on BCCH. When entering CELL_DCH state, The UE shall send a MEASUREMENT REPORT message when reporting criteria are satisfied. During CELL_DCH state, if the UE receives a MEASUREMENT CONTROL message, it shall terminate existing monitoring activities for the neighbouring cells previously known from SYSTEM INFORMATION BLOCK TYPESystem Information Block type 11 (or 12) messages. It shall perform the measurement and reporting tasks based on the latest MEASUREMENT CONTROL message received.

Reference

3GPP TS 25.331 clause 8.4.1.8.19

8.4.1.1.3 Test Purpose

To confirm that the UE continue to monitor <u>CPICH RSCP-intra-frequency</u> measurement quantity of the neighbour cells <u>listed in System Information Block type 11 or 12 messages</u>, after it has entered the CELL_DCH state from idle mode. When the reporting criteria specified in <u>SYSTEM INFORMATION BLOCK TYPESystem Information Block type 11</u> (or 12) <u>messages</u> have been met, it shall report the measurements to the <u>SS</u> using MEASUREMENT REPORT messages.—To confirm that in <u>CELL_DCH state</u>, the <u>UE respond to a modification of measurement criteria and adjust its measurement and reporting mechanism accordingly</u>. To confirm that the UE terminates monitoring <u>and measurement</u> activities for the neighbour cells found in <u>SYSTEM INFORMATION BLOCK TYPESystem Information Block type 11</u> (or 12) messages, after it has received a MEASUREMENT CONTROL message <u>which that specifies</u> the measurement type to be "intra-frequency measurement". To confirm that the UE reconfigures the monitoring and reporting activities based on the last MEASUREMENT CONTROL message received.

8.4.1.1.4 Method of test

Initial Condition

System Simulator: 2 cells – Cell 1 is active and the downlink P CPICH has a transmission level (RSCP) of 40 dBm, cell 2 is also active but with a P-CPICH transmission power 10dB below cell 1. The initial configurations of the 2 cells in the SS should follow the values indicated in the column marked "T0" in table 8.4.1.1-1. The table is found in "Test Procedure" sub-clause.

UE: CS-registered idle mode (state 2) or PS-registered idle mode (state 3) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE

Test Procedure

Table 8.4.1.1-1 illustrates the downlink power to be applied for the 2 cells at various time instants of the test execution. Column marked "T0" denotes the initial conditions, while columns marked "T1 are to be applied subsequently. The exact instants on which these values shall be applied are described in the texts in this sub-clause.

|--|

<u>Parameter</u>	<u>Unit</u>	<u>Cell 1</u>		Cell 2	
		<u>T0</u>	<u>T1</u>	<u>T0</u>	<u>T1</u>
UTRA RF		<u>Ch. 1</u>		<u>Ch. 1</u>	
Channel Number					
\hat{I}_{or}/I_{oc}	<u>dB</u>	<u>14.3</u>	<u>5.3</u>	<u>-11.3</u>	<u>-1.3</u>
CPICH Ec/lo	<u>dB</u>	<u>-11.9</u>	<u>-11.8</u>	<u>-14.9</u>	<u>-18.4</u>
CPICH RSCP	<u>dBm</u>	<u>-65.7</u>	<u>-74.4</u>	<u>-68.7</u>	<u>-81.3</u>

Table 8.4.1.1-1

The UE is initially at idle mode and has selected cell 1 for camping. The <u>SYSTEM INFORMATION BLOCK</u> <u>TYPESystem Information Block type</u> 11 messages is are modified with respect to the default settings to prevent reporting of CFN-SFN difference and also to include cell 2 into the monitored neighbour cell list. The key measurement parameters are as follow: measurement type = "intra-frequency measurement", measurement quantity = "CPICH RSCP", report criteria = "periodic reporting criteria", reporting interval = "12 seconds".

SS prompts the operator to make an outgoing call of a supported traffic class. The UE shall transmit a RRC CONNECTION REQUEST message on the CCCH, SS replies with RRC CONNECTION SETUP message and allocates uplink and downlink DPCH physical resources to the UE. UE shall then transmit RRC CONNECTION SETUP COMPLETE message and move to CELL_DCH state. After approximately 12 seconds, the UE shall transmit a MEASUREMENT REPORT message with measurement readings from cell 2. SS waits for 25 seconds after the sending of RRC CONNECTION SETUP message to verify that 2 consecutive MEASUREMENT REPORT messages are received.

SS sends a MEASUREMENT CONTROL message on the downlink DCCH. In this message, SS assigns an intra-frequency measurement type with the measurement quantity based on cell 2's CPICH RSCP value. Parameters used in this message are: measurement identity = "2", report criteria = "event-trigger", event identity = "1f", reporting threshold = "-675 dBm". After receiving this message, the UE shall delete the existing measurement and reporting contexts captured from System Information Block type 11 messages. SS checks to see that no MEASUREMENT REPORT messages are sent within the next 12 seconds (which is due to periodic reporting). SS reconfigures the downlink transmission power settings according to values in column "T1" in Table 8.4.1.1-1 gradually reduces the transmission power of P-CPICH in cell 2 until its RSCP value reaches -70dBm. The UE shall transmit MEASUREMENT REPORT messages when it detects that the CPICH RSCP of cell 2 has reached the threshold value specified.

Expected Sequence

Step	Direction		Message	Comment	
-	UE	SS			
1	(SYSTEM INFORMATION BLOCK TYPE 11System Information Block type 11	The UE is idle mode and camped onto cell 1. Gell 2 is also active but the P-CPICH has a RSCP 10dB below that of cell 1. SYSTEM INFORMATION TYPE 12The System Information Block type 11 messages to be transmitted isare different from the default settings (see specific message contents)	
2				SS prompts the operator to make an outgoing call.	
3		\rightarrow	RRC CONNECTION REQUEST		
4		←	RRC CONNECTION SETUP	SS allocates DPCH physical channels to UE	
5		→	RRC CONNECTION COMPLETE	UE transmits this message to acknowledge the RRC CONNECTION SETUP message and moves to CELL_DCH state.	
6		→	MEASUREMENT REPORT	SS waits for 25 seconds. It shall receive 2 consecutive MEASUREMENT REPORT messages	
7		+	MEASUREMENT CONTROL	A new measurement identity is assigned, with the reporting quantity changed to cell 2's P-CPICH RSCP. See specific message content for the rest of the message.	
8				SS waits for approximately 15 seconds and verifies that no further MEASUREMENT REPORT messages are detected on the uplink DCCH.	

CR page 444

9			SS gradually reduces the transmission power of cell 2 until its CPICH RSCP value drops to 70dBm.re-adjusts the downlink transmission power settings according to columns "T1" in Table 8.4.1.1-1.
10	→	MEASUREMENT REPORT	SS verifies that UE transmits 2 MEASUREMENT REPORT messages at 500 msec interval to report the P-CPICH RSCP value of cell 2.

Specific Message Contents

All messages indicated below shall use the same content as described in default message content, with the following exceptions:

SYSTEM INFORMATION BLOCK TYPESystem Information Block type 11 (Step 1)

Information Element	Value/Remark
Reference to other system information blocks	Not Present
SIB12 indicator	<u>FALSE</u>
FACH measurement occasion info	Not Present
Measurement control system information	
 Intra-frequency measurement system 	
information	
- Intra-frequency measurement identity-number	1
- Intra-frequency cell info list	
- Removed CHOICE intra-frequency info listcell	Not PresentRemove no intra-frequency cells
removal	
- New intra-frequency info list	Set to id of cell 2
 Intra-frequency cell id Cell info 	0.40
- Cell inio - Cell individual offset	0 dB
- Reference time difference to cell	0 chips
- Primary CPICH Info	Set to same code as used for cell 2
- Primary Scrambling Code	Not Present
- Primary CPICH TX power	FALSE
- Read SFN Indicator	FALSE
- TX Diversity Indicator	
- Cell selection and Re-selection info for	
SIB11/12	0dB
- Qoffset _{s,n}	0dBm
- Maximum allowed UL TX power	Not Present
- HCS neighbouring cell information	-115dB
- Qqualmin	-20dB
- Qrxlevmin	
- Intra-frequency measurement quantity	0
- Filter Coefficient	CPICH RSCP
- Measurement quantity	
- Intra-frequency measurement for RACH	
reporting	No report
 SFN-SFN observed time difference 	No report
- Reporting quantity	No report
 Maximum number of reported cells on RACH 	
 Reporting information for state CELL_DCH 	
 Intra-frequency reporting quantity 	
 Reporting quantities for active set cells 	No report
 SFN-SFN observed time difference reporting 	
indicator	<u>FALSE</u>
 Cell synchronisation information reporting 	
indicator	FALSE
- Cell identity reporting indicator	FALSE
- CPICH Ec/No reporting indicator	FALSE
- CPICH RSCP reporting indicator	FALSE
- Pathloss reporting indicator	FALSE
- CFN-SFN observed time difference	No report
- Reporting quantities for monitored set cells	No report
- SFN-SFN observed time difference reporting	ENISE
indicatorCell synchronisation information reporting	<u>FALSE</u>
indicator	TRUE
<u>indicator</u> - Cell identity <u>reporting indicator</u>	FALSE
- CPICH Ec/No <u>reporting indicator</u>	TRUE
- CPICH EC/No reporting indicator - CPICH RSCP reporting indicator	FALSE
- Pathloss reporting indicator	FALSE
- CFN-SFN observed time difference	Not present
- Reporting quantities for detected cells	Periodic reporting criteria
- CHOICE report criteria	Infinity
- Amount of reporting	12 seconds
- Reporting interval	Not present
- Inter-frequency measurement system	1.5. \$100011
information	Not Present
- Inter- system - <u>RAT</u> measurement system	Not Present
information	
	Not Present
 Traffic volume measurement system 	I INULFIESEIIL

 UE internal measurement system information 	

RRC CONNECTION REQUEST (Step 3)

Information Element	Value/Remarks
Establishment cause	Check to see if set to originating call of a
	supported traffic class
Measured results on RACH	Check to see if this IE is absent

RRC CONNECTION SETUP (Step 4)

Use the message sub-type in clause 9 of TS 34.108, which is marked as "Transition to CELL_DCH".

MEASUREMENT REPORT (Step 6)

Information Element	Value/Remarks
Measurement identity-number	Check to see if set to 1
Measured Results	
- CHOICE measurement	Check to see if set to "Intra-frequency measured results list"
 Intra-frequency measured results 	
- Cell measured results	
- Cell Identity	Check to see if it is absent
- CFN-SFN observed time difference	Check to see if this IE is absent
 SFN-SFN observed time difference 	Check to see if this IE is absent
 Cell synchronisation information 	Check to see if this IE is absent
- Primary CPICH Info	
- Primary Scrambling Code	Check to see if it's the same code for cell 2
- CPICH Ec/No	Check to see if this IE is absent
- CPICH RSCP	Check value to see if within acceptable range
- Pathloss	Check to see if this IE is absent
Measured Results on RACH	Check to see if this IE is absent
Event Results	Check to see if this IE is absent

MEASUREMENT CONTROL (Step 7)

Information Element	Value/Remark
information Element	Value/Itelliank

Measurement Identity Number Measurement Command Setup Measurement Reporting Mode Not Present Additional measurements list Not Present CHOICE measurement type Intra-frequency measurement - Intra-frequency cell info list - CHOICE Removed-intra-frequency info listcell Not PresentRemove no intra-frequency cells removal - New intra-frequency info list Set to id of cell 2 - Intra-frequency cell id - Cell info 0 dB - Cell individual offset 0 chips - Reference time difference to cell - Primary CPICH Info Set to same code as used for cell 2 - Primary Scrambling Code Not Present - Primary CPICH TX power **FALSE** - Read SFN Indicator **FALSE** - TX Diversity Indicator - Intra-frequency measurement quantity - Filter Coefficient **CPICH RSCP** - Measurement quantity - Intra-frequency reporting quantity Same as in default message content - Reporting quantities for active set cells No report - SFN-SFN observed time difference reporting indicator **FALSE** Cell synchronisation information reporting **FALSE** indicator - Cell identity reporting indicator **FALSE FALSE** - CPICH Ec/No reporting indicator - CPICH RSCP reporting indicator **FALSE** - Pathloss reporting indicator **FALSE** CFN-SFN observed time difference - Reporting quantities for monitored set cells No report - SFN-SFN observed time difference reporting **FALSE** indicator - Cell synchronisation information reporting indicator **TRUE** Cell identity reporting indicator **FALSE** - CPICH Ec/No reporting indicator TRUE - CPICH RSCP reporting indicator **FALSE** - Pathloss reporting indicator **FALSE** CFN-SFN observed time difference Not present - Reporting quantities for detected cells - Reporting cell status Report cells wWithin active and/or monitored - CHOICE reporteding cell frequency - Maximum number of reporteding cells-type2 - Measurement validity Not present

- CHOICE report criteria

- Parameters required for each events

- Intra-frequency event identity

- Triggering conditions 1

- Triggering condition 2

- Reporting range

- Cells forbidden to affect reporting range

- \//

- Hysteresis

- Threshold used frequency

- Reporting deactivation threshold

- Replacement activation threshold

- Time to trigger

- Amount of reporting

- Reporting interval

- Reporting cell status

DPCH compressed mode status info

cells set on used frequency and or within active and/or monitored cells set on non-used

Intra-frequency measurement reporting criteria

Monitored set cells

Not Present

Not Present

Not Present Not Present

1 dB

-675 dBm

Not Present Not Present

0 msec

500 msec

Not Present Not Present

CR page 450

MEASUREMENT REPORT (Step 10)

Information Element	Value/Remarks
Measurement identity-number	Check to see if set to 2
Measured Results	
- CHOICE measurement	Check to see if set to "Intra-frequency measured results list"
 Intra-frequency measured results 	
 Cell measured results 	
- Cell Identity	Check to see if it is absent
- CFN-SFN observed time difference	Check to see if this IE is absent
 SFN-SFN observed time difference 	Check to see if this IE is absent
 Cell synchronisation information 	Check to see if this IE is absent
- Primary CPICH Info	
- Primary Scrambling Code	Check to see if it's the same code for cell 2
- CPICH Ec/No	Check to see if this IE is absent
- CPICH RSCP	Check to see if this IE is present
- Pathloss	Check to see if this IE is absent
Measured Results on RACH	Check to see if this IE is absent
Event Results	Check to see if this IE is absent

8.4.1.1.5 Test Requirement

After step 5 the UE shall start to transmit MEASUREMENT REPORT message. The measurement quantity "CPICH RSCP" of cell 2 shall be reported to the SS at 12 seconds interval.

After step 7 the UE shall not transmit any MEASUREMENT REPORT messages within 15 seconds after SS has transmitted the MEASUREMENT CONTROL message.

After step 9 the UE shall transmit 2 MEASUREMENT REPORT messages on the uplink DCCH, to report that the <u>CPICH</u> RSCP value for cell 2 has dropped below the <u>specified</u>-threshold <u>stated in the MEASUREMENT CONTROL</u> message transmitted by the SS in step 7.

8.4.1.2 Measurement Control and Report: Inter-frequency measurement for transition from idle mode to CELL DCH state

8.4.1.2.1 Definition

8.4.1.2.2 Conformance requirement

After entering CELL_DCH state from idle mode, the UE shall discontinue the monitoring of the list of neighbouring cells assigned in the SYSTEM INFORMATION BLOCK TYPESystem Information Block 11 or 12 messages on BCCH. In CELL_DCH state, when the UE receives a MESUREMENT CONTROL message requesting for interfrequency measurement to be setup, it shall start inter-frequency measurement and the associated reporting activities if "DPCH compressed mode status info" IE in the message simultaneously activates at least one compressed mode pattern sequence. When the UE receives a MEASUREMENT CONTROL message with "Reporting cell status" IE omitted, it shall not include "Cell measured results" IE for any cells in MEASUREMENT REPORT messages sent on uplink DCCH.

Reference

3GPP TS 25.331 clause 8.4.1.<u>8.29</u>, 8.6.7.9

8.4.1.2.3 Test Purpose

To confirm that the UE terminates stops the monitoring activities of the list of neighbouring cell assigned in the IE "inter-frequency cell info" in SYSTEM INFORMATION BLOCK TYPESystem Information Block type 11 messages, after it enters CELL_DCH state from idle mode. To confirm that the UE starts to perform inter-frequency measurement

and related reporting activities, when it receives a MEASUREMENT CONTROL message with the "DPCH compress mode status info" IE indicating that an existing dormant stored compressed mode pattern sequence be simultaneously activated. To confirm that the UE exclude the IE "cell measured results" for any cells in the MEASUREMENT REPORT messages, after it receives a MEASUREMENT CONTROL message with "Reporting cell status" IE omitted.

8.4.1.2.4 Method of test

Initial Condition

System Simulator: 2 cells – The initial configurations of the 2 cells in the SS should follow the values indicated in table 8.4.1.2-1. The table is found in "Test Procedure" sub-clause. System Simulator: 2 cells — Cell 1 is active and has a transmission level (CPICH RSCP) of –50 dBm for P-CPICH, cell 4 is also active but with a transmission power 20dB below cell 1.

UE:CS-registered idle mode (state 2) or PS-registered idle mode (state 3) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE.

Test Procedure

<u>Table 8.4.1.2-1 illustrates the downlink power to be applied for the 2 cells.</u>

Unit Cell 1 <u>Parameter</u> Cell 4 UTRA RF <u>Ch. 1</u> Ch. 2 **Channel Number** <u>dB</u> 5.7 2.3 I_{or}/I_{oc} CPICH Ec/lo dB CPICH RSCP dBm

Table 8.4.1.2-1

The UE is initially at idle mode and has selected cell 1 for camping. The <u>SYSTEM BLOCK INFORMATION TYPESystem Information Block type</u> 11 messages is are modified with respect to the default settings to prevent reporting of CFN-SFN difference, and also to include cell 4 into the monitored neighbour cells list. The key measurement parameters are as follow: measurement type = "inter frequency measurement", measurement quantity = "CPICH RSCP".

SS prompts the operator to make an outgoing call for one of the traffic classes supported by the UE. The UE shall transmit a RRC CONNECTION REQUEST message on the CCCH, SS replies with RRC CONNECTION SETUP message and allocates uplink and downlink dedicated physical resources to the UE. The RRC CONNECTION SETUP message contains IE "DPCH compressed mode info", signifying the establishment of a transmission pattern gap sequence with TGPSI=1. UE shall sends RRC CONNECTION SETUP COMPLETE on the uplink DCCH and then moves to CELL_DCH state. The UE shall not transmit any MEASUREMENT REPORT messages, which pertain to measurement readings from cells belonging to the monitored set.

SS sends PHYSICAL CHANNEL RECONFIGURATION message on the downlink DCCH, specifying that compressed mode sequence pattern with TGPSI=1 be deactivated. The UE shall reply with PHYSICAL CHANNEL RECONFIGURATION COMPLETE message on the uplink DCCH. It shall stop compressed mode operations at the activation time stated in PHYSICAL CHANNEL RECONFIGURATION message. After the designated activation time has elapsed, SS sends MEASUREMENT CONTROL message on the downlink DCCH. In this message, SS requests UE to perform inter-frequency measurement with periodic reporting of CPICH RSCP values for cell 24. The "DPCH compressed status info" IE in this message activates the compressed mode transmission gap pattern sequence with TGPSI = 1. The UE shall starts inter-frequency measurement for cell 4's CPICH RSCP values. It shall report this measurement result by transmitting MEASUREMENT REPORT messages on uplink DCCH periodically at 16 seconds time interval.

In the next sequence, SS sends another MEASUREMENT CONTROL message on the downlink DCCH. In this message, the IE "Reporting cell status" is not included. The UE shall send a-MEASUREMENT REPORT messages on the uplink DCCH, excluding with the IE "Cell measured results" excluded from these messages.

Expected Sequence

Step	Direction UE SS	Message	Comment
1	<u> </u>	SYSTEM INFORMATION BLOCK TYPE 11System Information Block type 11	The UE is idle mode and camped onto cell 1.—Cell 2 is also active but has transmission power 20dB below that of cell 1.—SYSTEM INFORMATION BLOCK TYPE System Information Block Type 11 to be transmitted is different from the default settings (see specific message contents)
2			SS prompts the operator to make an outgoing call.
3	\rightarrow	RRC CONNECTION REQUEST	
4	<u>←</u>	RRC CONNECTION SETUP	SS allocates dedicated physical channels to UE, <u>as</u> well <u>as</u> specifying the use activation of compressed mode format behaviour with TGCPSI = 1.
5)	RRC CONNECTION SETUP COMPLETE	UE transmits this message to acknowledge the RRC CONNECTION SETUP message and moves to CELL_DCH state.
6			SS raises the CPICH RSCP of cell 2 by 10dB.—SS checks to see that no MEASUREMENT REPORT messages are sent from UE to cell 1.
7	+	PHYSICAL CHANNEL RECONFIGURATION	Existing compressed mode sequence pattern is deactivated in this message.
8	→	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	UE shall remain in CELL_DCH state.
9	+	MEASUREMENT CONTROL	SS requests UE to start inter- frequency measurement for cell 24 and performing periodic reporting for cell ² 2 4 CPICH RSCP. "DPCH compressed mode status info" IE is set to simultaneously activate compressed mode pattern.
10	\rightarrow	MEASUREMENT REPORT	UE shall report cell 2's 4's CPICH RSCP reading periodically.
11	+	MEASUREMENT CONTROL	SS changes the reporting criteria of cell 24 to 'event 2c'. "Reporting cell status" IE in this message is omitted.

12	→	MEASUREMENT REPORT	SS monitors the uplink DCCH to make sure that only 1 such message is received 32 seconds after step 11. This message shall not contain IE "Cell measured results"
----	----------	--------------------	---

Specific Message Content

All messages indicated below shall use the same content as described in default message content, with the following exceptions:

Information Element	Value/Remark
Reference to other system information blocks	Not Present
SIB12 indicator	FALSE
FACH measurement occasion info	Not Present
Measurement control system information	
- Intra-frequency measurement system	Not Present
information	
- Inter-frequency measurement system	
information	
- Inter-frequency measurement identity-number	1
- Inter-frequency cell info list	'
- Removed-CHOICE intraer-frequency info	Not PresentNo inter-frequency cells removed
listcell removal	Not i resemino inter-frequency cens femoved
- New inter-frequency info list	Set to id of cell 24
	Set to id of cell 24
- Inter-frequency cell id	Cot to the unlink LIADECN of call 24
- Frequency info	Set to the uplink UARFCN of cell 24
- UARFCN uplink (Nu)	Set to the downlink UARFCN of cell 24
- UARFCN downlink (Nd)	0.15
- Cell info	0 dB
- Cell individual offset	0 chips
- Reference time difference to cell	
- Primary CPICH Info	Set to same code as used for cell <u>2</u> 4
 Primary Scrambling Code 	Not Present
 Primary CPICH TX power 	FALSE
 Read SFN Indicator 	FALSE
- TX Diversity Indicator	
 Cell selection and re-selection info-for SIB 	
11/12	0 dB
- Qoffset _{s.n}	0 dBm
- Maximum allowed UL TX power	Not Present
 HCS neighbouring cell information 	-20 dB, -115dBm
- Qqualmin, Qrxlevmin	,
- Inter-frequency measurement quantity	Inter-frequency reporting criteria
— CHOICE Reporting criteria	θ ΄ ΄ ΄ ΄ ΄ ΄ ΄ ΄ ΄ ΄ ΄ ΄ ΄ ΄ ΄ ΄ ΄ ΄ ΄
Filter coefficient	CPICH RSCP
- Measurement quantity for frequency quality	
estimate	
- Inter-frequency measurement reporting criteria	
- Parameters required for each event	2c
- Inter-frequency event identity	Not Present
- Threshold used frequency	Not Present
- W used frequency	2.0dB
	0 sec
	Infinity
	0 Not Present
- Reporting interval	I INUL FIESTIL
- Parameters required for each non used	CEdD _m
frequency	-65dBm
- Threshold non-used frequency	O Not Broad and
	Not Present
- Inter- system - <u>RAT</u> measurement system	Not Present
information	
 Traffic volume measurement system 	Not Present
	I .
information	

RRC CONNECTION REQUEST (Step 3)

Information Element	Value/Remarks
Establishment cause	Check to see if set to originating call of a
	supported traffic class
Measured results on RACH	Check to see if this IE is absent

RRC CONNECTION SETUP (Step 4)

Use the message sub-type found in TS 34.108 clause 9 which is marked as "Transition to CELL_DCH", with the following exceptions:

Information Element	Value/Remarks
Downlink information common for all radio links	
- Downlink DPCH info common for all RL	
- CHOICE Mode	FDD
 Downlink DPCH power control information 	0 (Single)
- DPC mode	Refer to the parameter set in TS 34.108
- DL rate matching restriction information	Not Present
- Spreading factor	
- Fixed or flexible position	Flexible
- TFCI existence	FALSE
- Number of bits for Pilot bits (SF=128, 256)	Not Present
- DPCH compressed mode info	
- TGPSI	1
- TGPS Status Flag	Active
- TGCFN	<u>255</u>
- Transmission gap pattern sequence	
configuration parameters	
- TGMP	FDD Measurement
- TGPRC	62
- TGCFN	(Current CFN + (256 - TTI/10msec)) mod 256
- TGSN	8
- TGL1	10
- TGL2	5
- TGD	15
- TGPL1	35
- TGPL2	35
- RPP	Mode 1
- ITP	Mode 1
- UL/DL Mode	DL
 Downlink compressed mode method 	SF/2
 Uplink compressed mode method 	Not Present
 Downlink frame type 	A
- DeltaSIR1	2.0
- DeltaSIRAfter1	1.0
- DeltaSIR2	Not Present
- DeltaSIR2After2	Not Present
- TX Diversity Mode	None
- SSDT information	Not Present
- S field	
- Code Word Set	
 Default DPCH Offset Value 	0

Information Element	Value/remark
Message Type	
Integrity check info	Not Present
Integrity protection mode info	Not Present
Ciphering mode info	Not Present
Activation time	(256+CFN - (CFN MOD 8 + 8))MOD 256
New U-RNTI	Not Present
New C-RNTI	Not Present
DRX_RRC State indicator	No DRXCELL DCH
UTRAN DRX cycle length coefficient	Not Present –use default
CN information info	Not Present
URA identity	Not Present
RB with PDCP information	Not Present
Frequency info	Not Present – use default
Maximum allowed UL TX power	33dBm
CHOICE channel requirement	Uplink DPCH info
- Uplink DPCH power control info	Opinik Bi Oli ililo
- DPCCH power offset	-6dB
- PC Preamble	8slot
- Power Control Algorithm	Algorithm1
- TPC step size	1 dB
- Scrambling code type	Long
- Scrambling code type - Scrambling code number	0
- Number of DPDCH	Not Present (Use default value of 1)
	SF is reference to TS34.108 clause 6.10
- Spreading factor	Parameter Set
- TFCI existence	TRUE
- Number of FBI bit	
	Not Present (Use default value of 0) Reference to TS34.108 clause 6.10 Parameter
- Puncturing Limit	
CHOICE Mada	Set
CHOICE Mode	FDD Not Present
- Downlink PDSCH information	Not Present
Downlink information common for all radio links	
- Downlink DPCH info common for all RL	FDD
- CHOICE mode	FDD
- Downlink DPCH power control information	0 (-i)
- DPC mode	0 (single)
- DL rate matching restriction information	Not Present
- Spreading factor	Reference to TS34.108 clause 6.10 Parameter
Fixed on Flexible Desiries	Set
- Fixed or Flexible Position	Flexible
- TFCI existence	TRUE
- Number of bits for Pilot bits (SF=128,256)	Not Present
- DPCH compressed mode info	
- Transmission gap pattern sequence	
- TGPSI	1
- TPGS status Flag	Inactive
- TGCFN	<u>255</u>
- Transmission gap pattern sequence	Not Present
configuration parameters	
- TX Diversity mode	None
- SSDT information	Not Present
- S field	
- Code Word Set	
- Default DPCH Offset Value	0
Downlink information per radio link list	Not Present

Information Floment	Value/Remark
Information Element	
Measurement Identity-Number	1
Measurement Command	Setup
Measurement Reporting Mode	Not Present
Additional measurements list	Not Present
CHOICE measurement type	Inter-frequency measurement
- Inter-frequency cell info list	
 CHOICE Removed inter-frequency info listcell 	Not PresentNo inter-frequency cells removed
removal	
 New inter-frequency info list 	Set to id of cell <u>24</u>
- Inter-frequency cell id	
- Frequency info	UARFCN of the uplink frequency for cell 24
- UARFCN uplink (Nu)	UARFCN of the downlink frequency for cell 24
- UARFCN downlink (Nd)	
- Cell info	0 dB
- Cell individual offset	0 chips
- Reference time difference to cell	
- Primary CPICH Info	Set to same code as used for cell 24
- Primary Scrambling Code	Not Present
- Primary CPICH TX power	FALSE
- Read SFN Indicator	FALSE
- TX Diversity Indicator	TALOE
- Inter-frequency measurement quantity	Inter-frequency reporting criteria
- CHOICE reporting criteria	0
- Filter Coefficient	CPICH RSCP
	CPICH ROCP
- Measurement quantity for frequency quality	
estimate	FALSE
- Inter-frequency reporting quantity	_
- UTRA Carrier RSSI	TRUE
- Frequency quality estimate	
- Non frequency related cell reporting quantities	
- SFN-SFN observed time difference reporting	No report
<u>indicator</u>	
 Cell synchronisation information reporting 	<u>FALSE</u>
indicator	
 Cell Identity reporting indicator 	FALSE
 CPICH Ec/No reporting indicator 	FALSE
 CPICH RSCP reporting indicator 	TRUE
 Pathloss reporting indicator 	FALSE
	FALSE
- Reporting cell status	
 CHOICE reporteding cell 	Report cell Wwithin active and/or monitored
	cells set on used frequency and or within
	active and/or monitored cells set on non-used
 Maximum number of reporteding cells type2 	frequency
- Measurement validity	2
- Inter-frequency set update	Not present
- CHOICE report criteria	Not present
- Amount of reporting	Periodic reporting criteria
- Reporting interval	Infinity
DPCH compressed mode status info	16 seconds
- TGPSI	
- TGPS Status Flag	1
- TGCFN	Active
	255

MEASUREMENT REPORT (Step 10)

Information Element	Value/Remarks
Measurement identity-number	Check to see if set to 1
Measured Results	
- CHOICE measurement	Check to see if set to "Inter-frequency measured results list"
 Inter-frequency measurement results 	
- Frequency info	
- UARFCN (uplink)	Check to see if set to the UARFCN of the uplink frequency for cell 24
- UARFCN (downlink)	Check to see if set to the UARFCN of the downlink frequency for cell 24
- UTRA carrier RSSI	Check to see if it is absent
 Inter-frequency cell measurement results 	
 Cell measured results 	
- Cell Identity	Check to see if it is absent
 SFN-SFN observed time difference 	Check to see if it is absent
	Check to see if it is absent
 Cell synchronisation information 	Check to see if it is absent
- Primary CPICH Info	
- Primary Scrambling Code	Check to see if set to the same code for cell 24
- CPICH Ec/No	Check to see if it is absent
- CPICH RSCP	Check to see if it is present
- Pathloss	Check to see if it is absent
Measured Results on RACH	Check to see if it is absent
Event Results	Check to see if it is absent

Information Element	Value/Remark
Measurement Identity-Number	1
Measurement Command	Modify
Measurement Reporting Mode	Not Present
Additional measurements list	Not Present
CHOICE measurement type	Inter-frequency measurement
- Inter-frequency cell info list	
- Removed-CHOICE inter-frequency info listcell	Not PresentNo inter-frequency cells removed
removal	The tribution of the tr
- New inter-frequency info list	Set to id of cell 24
- Inter-frequency cell id	
- Frequency info	UARFCN of the uplink frequency for cell 24
- UARFCN uplink (Nu)	UARFCN of the downink frequency for cell 24
- UARFCN downlink (Nd)	
- Cell info	0 dB
- Cell individual offset	0 chips
- Reference time difference to cell	S 5PS
- Primary CPICH Info	Set to same code as used for cell 24
- Primary Scrambling Code	Not Present
- Primary CPICH TX power	FALSE
- Read SFN Indicator	FALSE
- TX Diversity Indicator	TALGE
- Inter-frequency measurement quantity	Inter-frequency reporting criteria
- CHOICE reporting criteria	0
- Filter Coefficient	CPICH RSCP
Measurement quantity for frequency quality	or for Roof
estimate	
- Inter-frequency reporting quantity	FALSE
- UTRA Carrier RSSI	TRUE
- Frequency quality estimate	TROL
Non frequency related cell reporting quantities	
- SFN-SFN observed time difference reporting	No report
indicator	The report
- Cell synchronisation information reporting	FALSE
indicator	TALGE
- Cell Identity reporting indicator	FALSE
- CPICH Ec/No_reporting indicator	FALSE
- CPICH RSCP reporting indicator	TRUE
- Pathloss reporting indicator	FALSE
- CFN-SFN observed time difference	FALSE
- Reporting cell status	Not Present
- Measurement validity	Not present
- Inter-frequency set update	Not present
- CHOICE report criteria	Inter-frequency measurement reporting criteria
Griolog roport sintend	Inter requestey measurement reporting enterta
- Parameters required for each event	Event 2c
- Inter-frequency event identity	Not Present
- Threshold used frequency	Not Present
- W used frequency	0.5 dB
- Hysteresis	0 milliseconds
- Time to trigger	1
- Amount of reporting	0
- Reporting interval	Not Present
- Reporting cell status	
- Parameters required for each non-used	
frequency	-85 dBm
- Threshold non used frequency	0
- W non used frequency	Not Present
DPCH compressed mode status info	THOU TOUGHT
	1

MEASUREMENT REPORT (Step 12)

Information Element	Value/Remarks
Measurement identity number	Check to see if set to 1
Measured Results	
- CHOICE measurement	Check to see if set to "Inter-frequency measured results list"
 Inter-frequency measurement results 	
- Frequency info	
- UARFCN (uplink)	Check to see if set to the UARFCN of the
	uplink frequency for cell 24
- UARFCN (downlink)	Check to see if set to the UARFCN of the
	downlink frequency for cell 24
- UTRA carrier RSSI	Check to see if it is absent
 Inter-frequency cell measurement results 	Check to see if it is absent
Measured Results on RACH	Check to see if it is absent
Event Results	Check to see if it is absent

8.4.1.2.5 Test Requirement

After step 5 the UE shall not transmit any MEASUREMENT REPORT messages pertaining to the measurement of CPICH RSCP quantity of cell 4.

After step 9 the UE shall transmit MEASUREMENT REPORT messages on uplink DCCH, reporting cell <u>24</u>'s CPICH RSCP value at periodic time interval of 16 seconds.

After step 11 the UE shall transmit only 1 MEASUREMENT REPORT message on the uplink DCCH. In this message, no<u>IE</u> "<u>inter-frequency Ccell measured results</u>" <u>IEs</u> should be <u>found preab</u>sent.

8.4.1.3 Measurement Control and Report: Intra-frequency measurement for transition from idle mode to CELL FACH state

8.4.1.3.1 Definition

8.4.1.3.2 Conformance requirement

During a transition from idle mode to CELL_FACH state, the UE shall start to monitor neighbouring cells listed in the IE "Intra-frequency cell info" received in SYSTEM INFORMATION BLOCK TYPE 11 System Information Block type 11 or TYPE 12 messages. If an "intra-frequency measurement reporting criteria" is also specified in these messages, the UE shall store this information and apply these reporting rules when deciding to transmit MEASUREMENT REPORT messages after following a subsequent transition to CELL_DCH state. If reporting during RACH transmissions is dictated by the UTRAN, the UE shall append the relevant measurement information when sending messages on RACH (e.g. RRC CONNECTION REQUEST and CELL UPDATE).

Reference

3GPP TS 25.331, clause 8.4.1.<u>9</u>10

8.4.1.3.3 Test Purpose

To confirm that the UE begins to monitor the neighbouring cells in the monitored listed in IE "intra-frequency cell info" of System Information Block type 11 or 12 messages, after it has entered the CELL_FACH state.—The list of neighbouring cell can be known from SYSTEM INFORMATION BLOCK TYPE 11 or TYPE 12 messages. If information regarding the intra-frequency measurement reporting criteria is also broadcasted, the UE shall save store this information and apply the reporting criteria after a subsequent transition to CELL_DCH state. If RACH measurement reporting is dictated in SYSTEM INFORMATION BLOCK TYPE System Information Block type 11 or TYPE 12 messages, the UE shall include these measurements when transmitting on the RACH channel.

8.4.1.3.4 Method of test

Initial Condition

System Simulator: 2 cells – both cell 1 and cell 2 are active. Cell 2 is transmitting at 15 dB below cell 1.See Table 8.4.1.3-1 for the power settings.

UE: CS-registered idle mode (state 2) or PS-registered idle mode (state 3) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE.

Test Procedure

Table 8.4.1.3-1 illustrates the downlink power to be applied for the 2 cells in this test case.

Table 8.4.1.3-1

<u>Parameter</u>	<u>Unit</u>	Cell 1	Cell 2
<u>UTRA RF</u>		<u>Ch. 1</u>	<u>Ch. 2</u>
Channel Number			
\hat{I}_{or}/I_{oc}	<u>dB</u>	<u>5.7</u>	<u>2.3</u>
CPICH Ec/lo	<u>dB</u>	<u>-12.4</u>	<u>-15.8</u>
CPICH RSCP	dBm	-74.3	-77.7

The UE is initially at idle mode and has selected cell 1 for camping. The SYSTEM BLOCK TYPESystem Information Block type 11 messages is are modified with respect to the default settings to prevent reporting of CFN-SFN difference and also to include cell 2 into the monitored neighbour cell list. The key measurement parameters are as follow: measurement type = "intra-frequency measurement", measurement quantity = "CPICH Ec/No", report criteria = "periodic reporting criteria", reporting interval = "12 seconds". In the sSystem iInformation type 11 messages, reporting of CPICH Ec/No is also required for intra-frequency reporting when transmitting RACH messages to cell 1.

SS prompts the operator to make an outgoing call of one of the traffic classes supported by the UE. The UE shall transmit a RRC CONNECTION REQUEST message on the uplink CCCH, SS replies with RRC CONNECTION SETUP message and allocates PRACH and S-CCPCH-physical channels for uplink and downlink usage. UE shall then enter CELL_FACH state. SS starts timer T305 and waits until timer T305 expires, the UE shall send a CELL UPDATE message which includesing the measurement reading of cell 2's CPICH Ec/No values. SS then reply with CELL UPDATE CONFIRM message on the downlink DCCH, without changing the physical channel resources.

In the next sequence, SS transmits PHYSICAL CHANNEL RECONFIGURATION message, and allocates DPCH channels to the UE. The UE shall transit to CELL_DCH state and then send a MEASUREMENT REPORT message, correctly stating the measurement identity. The measurement identity indicated shall match the value that was previously broadcasted on SYSTEM INFORMATION BLOCK TYPE 11 System Information Block type 11 messages when the UE was still in idle mode. The IE "Measured results" in the MEASUREMENT REPORT messages shall contain measured values of cell 2's CPICH Ec/No.

Expected Sequence

Step	Direction UE SS	Message	Comment
1	<u>∪E 33</u> ←	SYSTEM INFORMATION BLOCK TYPE 2, SYSTEM INFORMATION BLOCK TYPE 11-System Information Block type 2, System Information Block type 11	The UE is idle mode and camped onto cell 1. Cell 2 is also active but has transmission power 15dB below that of cell 1. SYSTEM INFORMATION TYPE 2 and 11 System Information Block type 2 and 11 to be transmitted are different from the default settings (see specific message contents)
2			SS prompts the test operator to make an outgoing call.
3	>	RRC CONNECTION REQUEST	The Ec/No value of P-CPICH of Cell 1 should be reported.
4	+	RRC CONNECTION SETUP	SS allocates common physical channels to UE.
5	\rightarrow	RRC CONNECTION COMPLETE	UE shall enter CELL_FACH state, and transmit this message to acknowledge the RRC CONNECTION SETUP message.
6			SS monitors the uplink DCCH to confirm that no MEASURENENT REPORT messages are detected. SS waits for 5 minutes (for the expiry of T305 timer), so that UE will initiate a periodic cell updating procedure.
7	→	CELL UPDATE	This message shall contain measurement readings of CPICH Ec/No for cell 1.
8	+	CELL UPDATE CONFIRM	SS does not change the physical channel configurationsallocation.
9	+	PHYSICAL CHANNEL RECONFIGURATION	SS assigns DPCH physical resources to the UE, but keeps the parameters for transport channels and RBs unchanged.
10	→	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	UE shall transit to CELL_DCH state after sending this message.
11	→	MEASUREMENT REPORT	UE shall begin to report cell 2's CPICH Ec/No value periodically at 12 seconds interval. The measurement identity shall match that broadcasted measurement contexts transmitted on BCCH in step 1

Specific Message Content

SYSTEM INFORMATION BLOCK TYPE System Information Block type 2 (Step 1)

Information Element	Value/Remarks
UE Timers and constants in connected mode	
- T305	5 minutes.

Information Element	Value/Remark
Reference to other system information blocks	value/Remark
SIB12 indicator	FALSE
FACH measurement occasion info	TALOE
- FACH Measurement occasion cycle length	2
coefficient	=
- Inter-frequency FDD measurement indicator	FALSE
- Inter-frequency TDD measurement indicator	FALSE
- Inter-RAT measurement indicators	Not Present
Measurement control system information	
 Intra-frequency measurement system 	
information	
 Intra-frequency measurement identity-number 	5
- Intra-frequency cell info list	
- <u>CHOICE</u> Removed intra-frequency info listcell	Not PresentRemove no intra-frequency cells
removal	
- New intra-frequency info list	Set to id of cell 2
- Intra-frequency cell id	0.40
- Cell info	0 dB
Cell individual offset Reference time difference to cell	0 chips
- Primary CPICH Info	Set to same code as used for cell 2
- Primary Scrambling Code	Not Present
- Primary CPICH TX power	FALSE
- Read SFN Indicator	FALSE
- TX Diversity Indicator	17/202
- Cell selection and Re-selection info-for	
SIB11/12	0 dB
- Qoffset _{s,n}	0 dBm
- Maximum allowed UL TX power	Not Present
 HCS neighbouring cell information 	-20dB, -115dBm
- Qqualmin, Qrxlevmin	
Intra-frequency Measurement quantity	0
- Filter Coefficient	CPICH Ec/No
- Measurement quantity	
- Intra-frequency measurement for RACH	Newspert
reporting - SFN-SFN observed time difference	No report CPICH Ec/No
- Reporting quantity	Current cell
- Maximum number of reported cells on RACH	Current cen
- Reporting information for state CELL_DCH	
- Intra-frequency reporting quantity	
- Reporting quantities for active set cells	No report
- SFN-SFN observed time difference reporting	·
<u>indicator</u>	FALSE
- Cell synchronisation information reporting	
indicator	FALSE
- Cell identity reporting indicator	FALSE
- CPICH Ec/No reporting indicator	FALSE
- CPICH RSCP reporting indicator	FALSE
- Pathloss reporting indicator - CFN-SFN observed time difference	FALSE
- Reporting quantities for monitored set cells	No report
- SFN-SFN observed time difference reporting	I NO TEPOIL
indicator	FALSE
- Cell synchronisation information reporting	· / 1202
indicator	TRUE
- Cell identity reporting indicator	TRUE
- CPICH Ec/No reporting indicator	FALSE
- CPICH RSCP reporting indicator	FALSE
- Pathloss reporting indicator	FALSE
- CFN-SFN observed time difference	Not present
 Reporting quantities for detected set cells 	
- Measurement Reporting Mode	Acknowledged mode RLC
- Measurement Reporting Transfer Mode	Periodic Reporting
 Periodic Reporting/Event Trigger Reporting 	

<u>Mode</u>	Periodical reporting criteria
- CHOICE report criteria	Infinity
- Amount of reporting	12 seconds
- Reporting interval	
- Reporting Cell Status	Report cells \within active and/or monitored
- CHOICE report <u>eding</u> cell	cells set on used frequency and or within
	active and/or monitored cells set on non-used
	frequency
 Maximum number of reporteding cells type2 	2
- Inter-frequency measurement system	Not Present
information	
- Traffic volume measurement system	Not Present
information	
- UE internal measurement system information	Not Present

RRC CONNECTION REQUEST (Step 3)

Information Element	Value/Remarks
Initial UE Identity	Check to see if it is the same as the IMSI in
	USIM card, TMSI or P-TMSI previously
	allocated.
Establishment cause	Check to see if set to originating call of the
	compatible traffic classes supported by the UE
Measured results on RACH	
 Measured result for current cell 	
 CHOICE measurement quantity 	Check to see if value is present
- CPICH Ec/No	Check to see if this IE is absent
 Measured results for monitored cells 	

RRC CONNECTION SETUP (Step 4)

Use the same message sub-type found in Annex A, which is entitled "Transition to CELL_FACH".

CELL UPDATE (Step 7)

Information Element	Value/Remarks
U-RNTI	Check to see if set to same U-RNTI value
	assigned in RRC CONNECTION SETUP
	message.
Integrity check info	Not checked
AM_RLC error indicator (for C-plane)	Check to see if set to 'FALSE'
AM_RLC error indicator (for U-plane)	Check to see if set to 'FALSE'
START List	Checked to see if the 'CN domain identity' and
	'START' IEs are present for all CN domains
	supported by the UE
Cell update cause	Check to see if set to 'Periodic cell updating'
Protocol error indicator	Check to see if set to 'FALSE'
Measured results on RACH	
 Measurement result for current cell 	
- CHOICE measurement quantity	CPICH Ec/No
- CPICH Ec/No	Checked to see if set to within an acceptable
	range.
 Measurement results for monitored cells 	Checked to see if this IE is not present.
Protocol error information	Check to see if set to 'FALSE'

PHYSICAL CHANNEL RECONFIGURATION (Step 9)

Use the same message sub-type found in Annex A, which is entitled "Packet to CELL_DCH from CELL_FACH".

MEASUREMENT REPORT (Step 11)

Information Element	Value/Remarks
Measurement identity-number	Check to see if set to 5
Measured Results	
- CHOICE measurement	Check to see if set to "Intra-frequency measured results list"
 Intra-frequency measured results 	
 Cell measured results 	
- Cell Identity	Check to see if it is absent
 SFN-SFN observed time difference 	Check to see if this IE is absent
- CFN-SFN observed time difference	Check to see if this IE is absent
 Cell synchronisation information 	Check to see if this IE is absent
- Primary CPICH Info	
- Primary Scrambling Code	Check to see if it's the same code for cell 2
- CPICH Ec/No	Check to see if this IE is within acceptable
	range
- CPICH RSCP	Check to see if this IE is absent
- Pathloss	Check to see if this IE is absent
Measured Results on RACH	Check to see if this IE is absent
Event Results	Check to see if this IE is absent

8.4.1.3.5 Test Requirement

After step 3 the UE shall send RRC CONNECTION REQUEST message-in response to the paging message sent by the SS, which includes measured value reading of the cell 1's CPICH Ec/No value.

After step 5 the UE shall not transmit any MEASUREMENT REPORT messages on the uplink DCCH.

After step 6 the UE shall initiate cell update procedure by transmitting CELL UPDATE message. In this message, the cause shall be set to "periodic cell updating". It shall include measurement measured readings value for cell 1's CPICH Ec/No measurement quantity.

After step 10 the UE shall move to CELL_DCH state and transmit MEASUREMENT REPORT messages at 12 seconds interval. In these messages, neighbouring cell 2's CPICH Ec/No value shall be reported. The IE "measurement identity" in this message shall match the IE "Intra-frequency measurement identity" found in SYSTEM INFORMATION BLCOK TYPESystem Information Block type 11 messages transmitted in step 1.

8.4.1.4 Measurement Control and Report: Inter-frequency measurement for transition from idle mode to CELL_FACH state

8.4.1.4.1 Definition

8.4.1.4.2 Conformance requirement

After entering CELL_FACH state from idle mode, the UE shall start to monitor the list of "inter-frequency" neighbouring cells assigned in the <u>SYSTEM INFORMATION BLOCK TYPESystem Information Block type</u> 11 or 12 messages on <u>transmitted on the BCCH/FACH</u>.

Reference

3GPP TS 25.331, clause 8.4.1.9.210

8.4.1.4.3 Test Purpose

To confirm that the UE begins to monitor the list of neighbouring cell assigned in the IE "inter-frequency cell info" in SYSTEM INFORMATION BLOCK TYPESystem Information Block type 11 or 12 messages, after it enters

CELL_FACH state from idle mode. However, it shall not transmit any MEASUREMENT REPORT messages to report measured results for inter-frequency cells.

8.4.1.4.3 Method of test

Initial Condition

System Simulator: 2 cells – The initial configurations of the 2 cells in the SS should follow the values indicated in the columns marked "T0" in table 8.4.1.4-1. The table is found in "Test Procedure" sub-clause. System Simulator: 2 cells – Cell 1 is active and has a downlink transmission level for CPICH RSCP at 40 dBm, cell 4 is also active but with a transmission power 10dB below cell 1.

UE: CS-registered idle mode (state 2) or PS-registered idle mode (state 3) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE.

Test Procedure

<u>Table 8.4.1.4-1 illustrates the downlink power to be applied for the 2 cells at various time instants of the test execution.</u>

<u>Columns marked "T0" denote the initial conditions, while columns marked "T1 are to be applied subsequently. The exact instants on which these values shall be applied are described in the texts in this sub-clause.</u>

Unit **Parameter** Cell 1 Cell 4 T0 T0 **UTRA RF** Ch. 2 <u>Ch. 1</u> **Channel** Number dΒ 9.7 0.5 0.5 9.7 \hat{I}_{or}/I_{oc} CPICH -12.8 -10.4 -12.8 -10.4 dB Ec/lo CPICH dBm -70.3 -79.5 -79.5 -70.3 **RSCP**

Table 8.4.1.4-1

The UE is initially at idle mode and has selected cell 1 for camping. The SYSTEM BLOCK TYPESystem Information Block type 12 messages is are modified with respect to the default settings to prevent reporting of CFN-SFN difference and also to include cell 42 into the monitored neighbour cell list for inter-frequency measurement type. The key measurement parameters are as follow: measurement type = "inter-frequency measurement", measurement quantity = "CPICH Ec/No".

SS prompts the operator to make an outgoing call of a supported traffic class. The UE shall transmit a RRC CONNECTION REQUEST message on the CCCH, SS replies with RRC CONNECTION SETUP message and allocates PRACH channel ion the uplink and S-CCPCH channel on the downlink to the UE. UE then moves to CELL_FACH_state. The UE shall not transmit any MEASUREMENT REPORT messages, which pertain to measurement readings from inter-frequency cells belonging to the monitored set. SS decreases the transmission power of CPICH in cell 1 by 10dB. At the same time, it raises the corresponding downlink transmission power of cell 4 by 10dB. SS re-adjusts its downlink power settings according to columns marked "T1" in Table 8.4.1.4-1. This is expected to trigger a cell reselection in the UE. The UE shall send CELL UPDATE message to cell 4 in order to report this event. Upon receiving this message, SS replies with the default CELL UPDATE CONFIRM message on the downlink DCCH.

Expected Sequence

Step	Dire	ction	Message	Comment
	UE	SS		
1		←	SYSTEM INFORMATION BLOCK TYPE 11System Information Block type 11	The UE is idle mode and camped onto cell 1. Cell 4 is also active but has transmission power 10dB below that of cell 1. SYSTEM INFORMATION TYPE 11-System Information Block type 11 to be transmitted is different from the default settings (see specific message contents)
2				SS prompts the test operator to make an outgoing call.
3		\rightarrow	RRC CONNECTION REQUEST	
4		←	RRC CONNECTION SETUP	SS allocates PRACH and S- CCPCH resources to UE
5		→	RRC CONNECTION COMPLETE	UE transmits this message to acknowledge the SETUP message and moves to CELL_FACH state.
6				SS checks to see that no MEASUREMENT REPORT messages are sent from UE to cell 1.
7				SS decreases the transmission power of cell 1 by 10dB and increases cell 4's downlink power by 10dB-reconfigures the downlink transmission power, according to columns "T1" of Table 8.4.1.4-1.
8		→	CELL UPDATE	UE shall detect that cell 4 has become stronger than cell 1. It sends this message after reselecting to cell 4
9		←	CELL UPDATE CONFIRM	Use default message.

Specific Message Content

All messages indicated below shall use the same content as described in default message content, with the following exceptions:

Information Element	Value/Remark
SIB12 Indicator	FALSE
Reference to other system information blocks	Not Present
FACH measurement occasion info	
 <u>k_UTRAFACH Measurement occasion cycle</u> 	2
length coefficient	
 Inter-frequency FDD measurement indicator 	TRUE
 Inter-frequency TDD measurement indicator 	<u>FALSE</u>
 Inter-RAT measurement indicators Other RAT 	Not Present
present in inter-system cell info	
Measurement control system information	Not Present
- Intra-frequency measurement system	
information	
- Inter-frequency measurement system	
information	1
- Inter-frequency measurement identity-number	
- Inter-frequency cell info list	Not PresentNo inter-frequency cells removed
- <u>CHOICE</u> Removed interra-frequency info	Set to id of cell 4
listcell removal - New inter-frequency info list	Set to id of cell 4
- New Inter-frequency fino list - Inter-frequency cell id	Set to uplink UARFCN of cell 4
- Frequency info	Set to the downlink UARFCN of cell 4
- UARFCN uplink (Nu)	Cot to the downlink OAKI ON OI Cell 4
- UARFCN downlink (Nd)	0 dB
- Cell info	0 chips
- Cell individual offset	Compo
- Reference time difference to cell	Set to same code as used for cell 4
- Primary CPICH Info	Not Present
- Primary Scrambling Code	FALSE
- Primary CPICH TX power	FALSE
- Read SFN Indicator	
- TX Diversity Indicator	
- Cell selection and re-selection info for SIB	0 dB
11/12	0 dBm
- Qoffset _{s,n}	Not Present
- Maximum allowed UL TX power	-20dB, -115dBm
- HCS neighbouring cell information	
- Qqualmin, Qrxlevmin	Inter-frequency reporting criteria
Inter-frequency measurement quantity	0
— CHOICE Reporting criteria	CPICH Ec/No
- Filter coefficient	
— - Measurement quantity for frequency quality estimate	
- Inter-frequency measurement reporting criteria	2c
- Parameters required for each event	Not Present
- Inter-frequency event identity	1 VOC 1 TOSCITE
- Threshold used frequency	2.0dB
- W used frequency	0 sec
- Hysteresis	Infinity
- Time to trigger	0
- Amount of reporting	
- Reporting interval	
- Parameters required for each non used	-55dBm
frequency	0
- Threshold non-used frequency	Not Present
- W non-used frequency	Not Present
- Inter- system - <u>RAT</u> measurement system	
information	Not Present
- Traffic volume measurement system	
information	
- UE internal measurement system information	

RRC CONNECTION REQUEST (Step 3)

Information Element	Value/Remarks
Establishment cause	Check to see if set to originating call of the
	compatible traffic class supported by the UE
	Check to see if this IE is absent
Measured results on RACH	

RRC CONNECTION SETUP (Step 4)

Use the message sub-type in default message content defined in Annex A, which is marked as "Transition to CELL FACH".

CELL UPDATE (Step 8)

Information Element	Value/Remarks
U-RNTI	Check to see if same to value as in RRC
	CONNECTION SETUP message
Integrity check info	Check to see if it is absent
AM_RLC error indication (for C-plane)	Check to see if it is set to FALSE
AM_RLC error indication (for U-plane)	Check to see if it is set to FALSE
START List	Checked to see if the 'CN domain identity' and
	'START' IEs are present for all CN domains
	supported by the UE
Cell update cause	Check to see if it is set to "Cell Reselection"
Protocol error info	Check to see if it is absent or set to FALSE
Measured results on RACH	Check to see if it is absent
Protocol error information	Check to see if it is absent

CELL UPDATE CONFIRM (Step 9)

Use the message sub-type in default message content defined in Annex A.

8.4.1.4.5 Test Requirement

After step 5 the UE shall not transmit any MEASUREMENT REPORT messages pertaining to the measurement of CPICH Ec/No quantity for cell 4.

After step 7 the UE shall reselect to cell 4 and transmit a CELL UPDATE message on the uplink CCCH of cell 42.

8.4.1.5 Measurement Control and Report: Intra-frequency measurement for transition from CELL DCH to CELL FACH state

8.4.1.5.1 Definition

8.4.1.5.2 Conformance requirement

After entering CELL_FACH state from CELL_DCH state, the UE shall terminate-stop intra-frequency type measurement reporting originating from assigned in a previous MEASUREMENT CONTROL message. After transition to CELL_FACH state, the UE shall start to monitor neighbouring cells listed in the "intra-frequency cell info" received in SYSTEM INFORMATION TYPESystem Information Block type 11 or 12. The UE shall apply the reporting criteria specified in sSystem iInformation Block type 11 or 12 messages, after a subsequent return to CELL_DCH state. If requested to perform measurement reporting on RACH channels, the UE shall append the measured results when transmitting uplink RACH messages.

Reference

3GPP TS 25.331, clause 8.4.1.76.1

8.4.1.5.3 Test Purpose

To confirm that the UE <u>ceases-stops</u> to perform intra-frequency measurement <u>reporting</u> specified in a previously received MEASUREMENT CONTROL message, when it moves from CELL_DCH state to CELL_FACH state. To confirm that the UE read the <u>sSystem iInformation Block type 11 or 12 messages</u> when in CELL_FACH state and starts to monitor the neighbouring cells as indicated in <u>these system information</u> messages. To confirm that the UE performs

measurements on uplink RACH channel and appends the measured results in RACH messages, when requested in the system information messages.

8.4.1.5.4 Method of test

Initial Condition

System Simulator: 3 cells – Cell 1 and cell 2 are active, but the transmitter of cell 3 is switched off. Cell 1 is transmitting its CPICH (RSCP) at 20 dB above cell 2. The initial configurations of the 3 cells in the SS should follow the values indicated in the column marked "T0" in table 8.4.1.14-1. The table is found in "Test Procedure" sub-clause.

UE: CS-registered idle mode (state 2) or PS-registered idle mode (state 3) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE.

Test Procedure

Table 8.4.1.5-1 illustrates the downlink power to be applied for the 3 cells at various time instants of the test execution. Column marked "T0" denotes the initial conditions, while columns marked "T1 are to be applied subsequently. The exact instants on which these values shall be applied are described in the texts in this sub-clause.

Parameter Unit Cell 1 Cell 2 Cell 3 T0 T0 T1 T0 UTRA RF Ch. 1 <u>Ch. 1</u> Ch. 1 Channel Number 0.5 dΒ <u>5.3</u> <u>5.3</u> 0.5 3.3 \hat{I}_{or}/I_{oc} Cell 3 is switched off CPICH -12.1 -13.5 -16.9 -18.3 -15.5 dB Ec/lo CPICH -74.7 -74.7 -79.5 -79.5 -76.7 dBm **RSCP**

Table 8.4.1.5-1

The UE is initially in idle mode and has selected cell 1 as the current cell. The SYSTEMSystem INFORMATION Information BLOCK-Block TYPE-type 11 message is modified from its default message contents, in order to prevent the UE's reporting of CFN-SFN difference. No measurement requirements are specified for the UE in any of the sSystem iInformation bBlock (type 11 and or 12) messages.

SS then prompts the test operator to initiate an outgoing call of a supported traffic class. When UE transmits a RRC CONNECTION REQUEST message on RACH, SS replies with RRC CONNECTION SETUP message. Uplink and downlink DPCH physical channels are allocated. Upon receiving RRC CONNECTION SETUP message, the UE shall transmit RRC CONNECTION SETUP COMPLETE message on DCCH and then moves to CELL_DCH state. SS then sends a MEASUREMENT CONTROL message to UE. In this message, the SS requests the establishment of an intrafrequency measurement task for the measurement based on an estimation of cell 2's CPICH Ec/No. At the same time, reporting of cell 2's CPICH Ec/No is commanded with the reporting criteria set to "periodic reporting" and reporting interval set to 16 seconds. The UE shall start transmitting MEASUREMENT REPORT messages at 16 seconds interval corresponding to the requested reporting event.

SS transmits PHYSICAL CHANNEL RECONFIGURATION message, triggering a switch of transport channels from DCH_(UL)/DCH_(DL) to RACH_(UL)/FACH_(DL). After receiving this message, the UE shall reconfigure itself and repliesy with a PHYSICAL CHANNEL RECONFIGURATION COMPLETE message on RACH. SS acknowledges this message and the UE shall move to CELL_FACH state and read the Seystem iInformation Block messages. SS then monitors the uplink channels to verify that no MEASUREMENT REPORT messages are received.

SS reconfigures itself according to the settings in columns marked "T1" in table 8.4.1.5-1. It then transmits In SYSTEM-System INFORMATION-Information BLOCK-Block TYPE type 12 messages in cell 1, SS-which indicates to includes cell 3 into the neighbour cells monitoring list. IEs "Intra-frequency reporting quantity for RACH Reporting" and IE "Maximum number of Reported cells on RACH" are also specified in this these messages. An event-triggered (event type 1a) reporting criterion is specified for intra-frequency measurements.—Cell 3 is turned on and its transmission strength is set to exceed the lower bound of the reporting range specified. SS then pages for the UE using PAGING TYPE 1 message. The UE shall respond with a CELL UPDATE message, which comprises measurement readings of CPICH Ec/No measurements for cell 1 and cell 3. Upon the receipt of CELL UPDATE message, SS replies with CELL UPDATE CONFIRM message on the downlink DCCH. This message does not change

the physical resources nor allocate any new RNTI identities.—The UE shall remain in CELL_FACH state. SS transmits PHYSICAL CHANNEL RECONFIGURATION message again, this time specifying allocating dedicated physical resources for both uplink and downlink directions to the UE. The UE shall then send PHYSICAL CHANNEL RECONFIGURATION COMPLETE message and return to CELL_DCH state. SS listens to the uplink DCCH for MEASUREMENT REPORT messages. SS verifies that all messages received pertain to the periodic measured value of cell 2's CPICH Ec/No value. UE shall not send any reports containing the measured values of cell 3.

Expected Sequence

Step	Direction	Message	Comment
	UE SS		
1	←	SYSTEM INFORMATION BLOCK TYPESystem Information Block type 11	UE is initially in idle mode in cell 1, and test operator is asked to make an outgoing call.
2	\rightarrow	RRC CONNECTION REQUEST	
3	→ ←	RRC CONNECTION SETUP	Uplink and downlink DPCH resources are allocated.
4	\rightarrow	RRC CONNECTION SETUP COMPLETE	UE shall move to CELL_DCH state.
5	←	MEASUREMENT CONTROL	SS requests for measurement and reporting of cell 2's CPICH Ec/No value.
6	→	MEASUREMENT REPORT	UE shall send periodic report at 16 seconds interval. SS waits for 2 consecutive reports before proceeding to step 7.
7	+	PHYSICAL CHANNEL RECONFIGURATION	SS switches the physical resources to common physical channels.
8	→	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	UE shall reconfigure its uplink and downlink channels before transiting to CELL_FACH state.
9	+	SYSTEM INFORMATION BLOCK TYPESystem Information Block type 12	SS reconfigures itself according to the settings stated in column "T1" of table 8.4.1.5-1. SIB type 12 indicates UE SS-includes cell 3 into the monitored neighbour cell list. SS waits for 1 minutes and verifies that no MEASURE-MENT REPORT messages are receiveddetected on the uplink. Cell 3 is switched on, and SS adjusts the transmission power for CPICH such that its Ec/No falls into the report range specified.
10	←	PAGING TYPE 1	SS pages for UE using the assigned U-RNTI identity.
11	→	CELL UPDATE	UE shall transmit this message with measured results on RACH channels for cell 1 and cell 3 appended present in this message.
12	+	CELL UPDATE CONFIRM	No changes in physical resource allocation and RNTI identities.
13	+	PHYSICAL CHANNEL RECONFIGURATION	Allocates dedicated physical channels.
14	\rightarrow	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	UE shall transit to CELL_DCH state.
15	+	MEASUREMENT REPORT	UE shall continue to report cell 2's CPICH Ec/No value on a periodic basis.

Specific Message Content

SYSTEM INFORMATION BLOCK TYPESystem Information Block type 11 (Step 1)

Information Element	Value/Remark
Reference to other system information blocks	Not Present
SIB12 indicator	<u>FALSE</u>
FACH measurement occasion info	Not Present
Measurement control system information	
Intra-frequency measurement system information	Not Present
Inter-frequency measurement system information	Not Present
- Inter-system-RAT measurement system	Not Present
information	Not Present
- Traffic volume measurement system	
information	Not Present
- UE internal measurement system information	

RRC CONNECTION SETUP (Step 3)

Use the same message sub-type found in Annex A, which is entitled "Transition to CELL_DCH".

Information Element	Value/Remark	
Measurement Identity-Number	5	
Measurement Command	Setup	
Measurement Reporting Mode	Not Present	
Additional measurements list	Not Present	
CHOICE measurement type	Intra-frequency measurement	
- Intra-frequency cell info list		
- Removed-CHOICE intra-frequency info listcell	Not PresentRemove no intra-frequency cells	
removal	<u> </u>	
- New intra-frequency info list	Set to id of cell 2	
- Intra-frequency cell id		
- Cell info	0 dB	
- Cell individual offset	0 chips	
- Reference time difference to cell	'	
- Primary CPICH Info	Set to same code as used for cell 2	
- Primary Scrambling Code	Not Present	
- Primary CPICH TX power	FALSE	
- Read SFN Indicator	FALSE	
- TX Diversity Indicator		
- Intra-frequency measurement quantity	0	
- Filter Coefficient	CPICH Ec/No	
- Measurement quantity		
- Intra-frequency reporting quantity		
- Reporting quantities for active set cells	No report	
- SFN-SFN observed time difference reporting		
indicator	<u>FALSE</u>	
- Cell synchronisation information reporting		
indicator	TRUE	
 Cell identity reporting indicator 	TRUE	
 CPICH Ec/No reporting indicator 	FALSE	
 CPICH RSCP reporting indicator 	FALSE	
 Pathloss reporting indicator 	FALSE	
- CFN-SFN observed time difference		
 Reporting quantities for monitored set cells 	No report	
- SFN-SFN observed time difference reporting		
indicator	<u>FALSE</u>	
- Cell synchronisation information reporting		
indicator	TRUE	
- Cell identity reporting indicator	TRUE	
- CPICH Ec/No reporting indicator	FALSE	
- CPICH RSCP reporting indicator	FALSE	
- Pathloss reporting indicator	FALSE Not prepart	
- CFN-SFN observed time difference	Not present	
Reporting quantities for detected cells Reporting cell status	Report cells \(\forall \) within active and/or monitored	
- CHOICE reporteding cell	cells-set on used frequency and or within	
- Of IOTOL Teport <u>eumi</u> cell	active and/or monitored eells-set on non-used	
	frequency	
- Maximum number of reporteding cells type2	2	
- Measurement validity	Not present	
- CHOICE report criteria	Periodical reporting criteria	
- Amount of reporting	Infinity	
- Reporting interval	16 seconds	
DPCH compressed mode status info	Not Present	

MEASUREMENT REPORT (Step 6)

Information Element	Value/Remarks
Measurement identity number	Check to see if set to 5
Measured Results	
- CHOICE measurement	Check to see if set to "Intra-frequency
	measured results list"
 Intra-frequency measured results 	
 Cell measured results 	
- Cell Identity	Check to see if it is absent
 SFN-SFN observed time difference 	Check to see if this IE is absent
 - CFN-SFN observed time difference 	Check to see if this IE is absent
 Cell synchronisation information 	Check to see if this IE is absent
- Primary CPICH Info	
- Primary Scrambling Code	Check to see if it's the same code for cell 2
- CPICH Ec/No	Check to see if this IE is present
- CPICH RSCP	Check to see if this IE is absent
- Pathloss	Check to see if this IE is absent
Measured Results on RACH	Check to see if this IE is absent
Event results	Check to see if this IE is absent

PHYSICAL CHANNEL RECONFIGURATION (Step 7)

Use the same message sub-type found in Annex A, which is entitled "(Packet to CELL_FACH from CELL_DCH in PS)"

Information Element	Value/Remark	
Reference to other system information blocks	Not Present	
FACH measurement occasion info	Not Present	
Measurement control system information		
- Intra-frequency measurement system		
information		
- Intra-frequency measurement identity number	6	
 Intra-frequency cell info list 		
 Removed CHOICE intra-frequency cell 	Not PresentRemove no intra-frequency cells	
removalinfo list		
- New intra-frequency info list	Set to id of cell 3	
- Intra-frequency cell id		
- Cell info	0 dB	
- Cell individual offset	0 chips	
- Reference time difference to cell	Cat to same and an used for call 2	
- Primary CPICH Info	Set to same code as used for cell 3	
 Primary Scrambling Code Primary CPICH TX power 	Not Present FALSE	
- Read SFN Indicator	FALSE	
- TX Diversity Indicator	I ALGE	
- Cell selection and Re-selection info-for		
SIB11/12	0dB	
- Qoffset _{s.n}	0dBm	
- Maximum allowed UL TX power	Not Present	
- HCS neighbouring cell information	-20dB, -115dBm	
- Qqualmin, Qrxlevmin	,	
- Intra-frequency measurement quantity	0	
- Filter Coefficient	CPICH Ec/No	
 Measurement quantity 		
 Intra-frequency measurement for RACH 		
reporting	No report	
- SFN-SFN observed time difference	CPICH Ec/No	
- Reporting quantity	Current cell + best neighbour	
 Maximum number of reported cells on RACH Reporting information for state CELL_DCH 		
Intra-frequency reporting quantity		
- Reporting quantities for active set cells	No report	
- SFN-SFN observed time difference reporting	Two report	
indicator	FALSE	
- Cell synchronisation information reporting	<u></u>	
indicator	TRUE	
- Cell identity reporting indicator	TRUE	
 CPICH Ec/No reporting indicator 	FALSE	
 CPICH RSCP reporting indicator 	FALSE	
 Pathloss reporting indicator 	FALSE	
- CFN-SFN observed time difference		
- Reporting quantities for monitored set cells	No report	
- SFN-SFN observed time difference reporting	FALCE	
indicator	<u>FALSE</u>	
 Cell synchronisation information reporting indicator 	TRUE	
- Cell identity reporting indicator	TRUE	
- CPICH Ec/No reporting indicator	FALSE	
- CPICH RSCP reporting indicator	FALSE	
- Pathloss reporting indicator	FALSE	
- CFN-SFN observed time difference	Not present	
- Reporting quantities for detected cells	Intra-frequency measurement reporting criteria	
- CHOICE report criteria	Only 1 reporting event is defined	
·	1a Ta	
- Parameter required for each event	Not Present	
 Intra-frequency event identity 	Monitored set cells	
- Triggering condition 1	10.0 3.0 dB	
- Triggering condition 2	Not present	
- Reporting range	0.0	
- Cells forbidden to affect reporting	<u>FDD</u>	
- CHOICE Mode		

- Primary CPICH info	Set to scrambling code for cell 3
- Primary scrambling code	1.0 dB
- W	Not Present
- Hysteresis	7
- Threshold used frequency	Not Present
 Reporting deactivation threshold 	0
- Replacement activation threshold	Infinity
- Time to trigger	8 seconds
- Amount of reporting	Not present
- Reporting Interval	
 Inter-frequency measurement system 	Not present
information	Not present
 Inter-system-RAT measurement system 	
information	Not present
 Traffic volume measurement system 	
information	
 UE internal measurement system information 	

PAGING TYPE 1 (Step 10)

Information Element	Value/Remarks
Page Record List	
- Page Record	
- CHOICE Paging originator	UTRAN Originator
- U-RNTI	Set to same U-RNTI value as assigned in the
	RRC CONNECTION SETUP message
BCCH modification info	Not Present

CELL UPDATE (Step 11)

Information Element	Value/Remarks
U-RNTI	Check to see if same to value as in PAGING
	TYPE 1 message
Integrity check info	Check to see if it is absent
AM_RLC error indication (for C-plane)	Check to see if it is set to FALSE
AM_RLC error indication (for U-plane)	Check to see if it is set to FALSE
START List	Checked to see if the 'CN domain identity' and
	'START' IEs are present for all CN domains
	supported by the UE
Cell update cause	Check to see if it is set to "Paging Response"
Protocol error info	Check to see if it is absent or set to FALSE
Measured results on RACH	
 Measurement result for current cell 	
 CHOICE measurement quantity 	Check to see if set to CPICH Ec/No
- CPICH Ec/No	Check to see if it is within acceptable range
 Measurement results for monitored cells 	·
 SFN-SFN observed time difference 	Not Checked
 Primary CPICH info 	
- Primary scrambling code	Check to see if the same as cell 3's code.
- CHOICE measurement quantity	Check to see if set to CPICH Ec/No
- CPICH Ec/No	Check to see if it is within acceptable range
Protocol error information	Check to see if it is absent

PHYSICAL CHANNEL RECONFIGURATION (Step 13)

Use the same message sub-type found in Annex A, which is entitled "(Packet to CELL_DCH from CELL_FACH in PS)"

MEASUREMENT REPORT (Step 15)

The contents of this message should be the same as in step 6.

Measurement identity	Check to see if set to 6
Measured Results	
- CHOICE measurement	Check to see if set to "Intra-frequency
	measured results list"
 Intra-frequency measured results 	
- Cell measured results	
- Cell Identity	Check to see if it is absent
 SFN-SFN observed time difference 	Check to see if this IE is absent
 Cell synchronisation information 	Check to see if this IE is absent
- Primary CPICH Info	
 Primary Scrambling Code 	Check to see if it's the same code for cell 2
- CPICH Ec/No	Check to see if this IE is present
- CPICH RSCP	Check to see if this IE is absent
- Pathloss	Check to see if this IE is absent
Measured Results on RACH	Check to see if this IE is absent
Event results	Check to see if this IE is absent

8.4.1.5.5 Test Requirement

After step 5, the UE shall start to transmit MEASUREMENT REPORT messages at 16 seconds interval. The message shall contain the measured result of cell 2's CPICH Ec/No value.

After step 8, the UE shall not send any MEASUREMENT REPORT messages which containing reporting quantities requested in MEASUREMENT CONTROL messages in step 5.

After step 10, the UE shall respond to the paging and transmit a CELL UPDATE message. In this message, the measured values CPICH Ec/No for cell 1 and cell 3 shall be included.

After step 14, the UE shall revert to the original follow measurement reporting mechanism specified in System Information Block type 12 messages sent in step 9. and It shall send MEASUREMENT REPORT messages containing estimates for cell 2's CPICH Ec/No value. The UE shall not transmit any MEASUREMENT REPORT messages, which indicate report measurement quantities of cell 3.

8.4.1.6 Measurement Control and Report: Inter-frequency measurement for transition from CELL_DCH to CELL_FACH state

8.4.1.6.1 Definition

8.4.1.6.2 Conformance requirement

When transiting from CELL_DCH state to CELL_FACH state, the UE shall terminate stop all measurement reporting activities related to inter-frequency measurements assigned in a MEASUREMENT CONTROL message. After reaching CELL_FACH state, the UE shall begin to monitor neighbouring cells listed in the IE "inter-frequency cell info" of specified in the sSystem iInformation Block type 11 or 12 messages.

Reference

3GPP TS 25.331, clause 8.4.1.7<u>6.2</u>

8.4.1.6.3 Test Purpose

To verify that UE ceases to transmit MEASUREMENT REPORT messages to report inter-frequency type measurements when moving from CELL_DCH state to CELL_FACH. This requirement shall be observed even if it the UE has detected that inter-frequency type measurement reporting criteria have been satisfied in CELL_FACH state. To verify that the UE extracts information for inter-frequency measurement from sSystem information Block type 11 or 12 messages after reaching CELL_FACH state.

8.4.1.6.4 Method of test

Initial Condition

System Simulator: 2 cells – The initial configurations of the 2 cells in the SS should follow the values indicated in the columns marked "T0" in table 8.4.1.6-1. The table is found in "Test Procedure" sub-clause.

SS: 2 cells — Both cell 1 and cell 4 are active. The applicable radio conditions for cell 1 and cell 4 are as follow:

	Cell 1	Cell 4
UARFCN (N _u and N _d)	Channel 1	Channel 2
CPICH RSCP	-75 dBm	-85 dBm
Q _{rxlevmin}	-90 dBm	-90 dBm
S	15	5

UE: CS-registered idle mode (state 2) or PS-registered idle mode (state 3) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE.

Test Procedure

Table 8.4.1.6-1 illustrates the downlink power to be applied for the 2 cells at various time instants of the test execution. Columns marked "T0" denote the initial conditions, while columns marked "T1 are to be applied subsequently. The exact instants on which these values shall be applied are described in the texts in this sub-clause.

<u>Parameter</u>	<u>Unit</u>	<u>Ce</u>	<u>II 1</u>	<u>Ce</u>	<u>II 4</u>
		<u>T0</u>	<u>T1</u>	<u>T0</u>	<u>T1</u>
UTRA RF Channel Number		<u>Ch</u>	<u>. 1</u>	<u>Ch</u>	<u>ı. 2</u>
\hat{I}_{or}/I_{oc}	<u>dB</u>	<u>9.5</u>	<u>-1.5</u>	<u>-1.5</u>	<u>9.5</u>
CPICH Ec/lo	<u>dB</u>	<u>-10.5</u>	<u>-13.8</u>	<u>-13.8</u>	<u>-10.5</u>
CPICH RSCP	<u>dBm</u>	<u>-70.5</u>	<u>-81.5</u>	<u>-81.5</u>	<u>-70.5</u>

Table 8.4.1.6-1

The UE is initially in idle mode, after camping on cell 1. SS prompts the test operator to initial an outgoing call for one of the supported traffic classes. The SYSTEM System INFORMATION Information Block TYPE-type 11 message is modified so that no measurement tasks are required to be performed byof the UE. The UE shall respond to the paging by sending a RRC CONNECTION REQUEST message on the uplink CCCH-carried by RACH. Upon receiving this message, SS allocates dedicated DPCH physical channels to the UE by transmitting RRC CONNECTION SETUP message. The UE shall reply by transmitting a RRC CONNECTION SETUP COMPLETE message. SS then checks the IE "Measurement Capability" of this message and verifies that the UE is capable of performing inter-frequency measurements under FDD mode. After the confirmation of the UE inter-frequency measurement capability, SS transmits PHYSICAL CHANNEL RECONFIGURATION message. In this message, information related to the radio bearers and uplink/downlink radio resources remain unchanged as compared to the equivalent IEs found in RRC CONNECTION SETUP message. The only difference concerns IE "DPCH compressed mode info" is present, which indicates that the UE shall apply the given parameters for compressed mode operations. The UE shall return a PHYSICAL CHANNEL RECONFIGURATION COMPLETE message to acknowledge that compressed mode mechanism can be exercised.

SS then sends a MEASUREMENT CONTROL message to the UE, specifying that cell 4 be the measurement object for inter-frequency type measurement. The periodic reporting criterion is selected for this measurement. SS waits for 8 seconds to allow the periodic timer to expire. The UE shall send a MEASUREMENT REPORT message containing measured result of cell 4's measurement reporting quantity (CPICH Ec/No). SS transmits PHYSICAL CHANNEL RECONFIGURATION message again, requesting the UE to switch from uplink and downlink DPCH to common physical channels. The UE shall return a PHYSICAL CHANNEL RECONFIGURATION COMPLETE message and then move to CELL_FACH state.

SS modifies the contents of Master Information Block (MIB) and System Information Block (SIB) type 11. In SIB 11, cell 4 is added to the neighbouring cell list in the "inter-frequency cell info" IE. SS waits for another-8 seconds to detect any possible uplink MEASUREMENT REPORT messages as a result of inter-frequency measurements. SS then increases the transmission power of cell 4 by 20 dBreconfigures the downlink transmission power settings of cell 1 and cell 4 according to the values stated in columns "T1" of Table 8.4.1.6-1. SS waits for [x] seconds to allow the UE to perform cell re-selection. The UE shall transmit a CELL UPDATE message on the uplink CCCH of cell 4, specifying the cause as "cell re-selection". SS replies with CELL UPDATE CONFIRM message on the downlink DCCH to complete the cell update procedure.

Step	Direction UE SS	Message	Comment
1		System Information Block type 11	UE is initially in idle mode and camped onto cell 1. SYSTEM INFORMATION BLOCK TYPESystem Information Block type 11 is redefined, modified with respect to the default settings in order to disabling disable a measurement and reporting
2			activities. SS prompts the test operator to trigger an outgoing call for
			supported traffic class
3	→ ←	RRC CONNECTION REQUEST RRC CONNECTION SETUP	Unlink and describe DDCII
•			Uplink and downlink DPCH resources are allocated.
5	→	RRC CONNECTION SETUP COMPLETE	UE shall indicate that it's capable of performing interfrequency measurement for FDD mode.
6	+	PHYSICAL CHANNEL RECONFIGURATION	SS instructs UE to begin compressed mode operation.
7	\rightarrow	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	UE shall remain in CELL_DC state.
8	+	MEASUREMENT CONTROL	SS indicates that the CPICH Ec/No of cell 4 shall be monitored and reported. SS waits for 8 seconds.
9	→	MEASUREMENT REPORT	UE shall transmit this message to report cell 4's CPICH Ec/No value.
10	+	PHYSICAL CHANNEL RECONFIGURATION	SS changes the physical channel allocation to common channel configuration.
11	\rightarrow	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	UE shall moves to CELL_FACH state.
<u>12</u>	<u></u>	Master Information Block, System Information Block type 11	SS modifies MIB and SIB 11. Cell 4 is included in the neighbouring cells list for inte frequency measurement
1 <u>3</u> 2			SS waits for 8 seconds to verify that no MEASUREMENT REPORT messages are detected.
13			SS changes SYSTEM INFORMATION BLOCK TYP 12 message to request cell 4 to be included into neighbour cell list for inter-frequency measurements.
14			SS changes the power settings for cell 1 and cell 4 according to columns marked "T1" of Table 8.4.1.6- <u>1increases the transmission power of cell 4 by 20 dB</u> , and then waits for [x] seconds to allow the UE to re-select to a new cell.
15	\rightarrow	CELL UPDATE	UE shall detect determine the cell 4 has become the best cound then perform cell reselection procedure.
		CELL UPDATE CONFIRM	

Notes:

The value [x] seconds is TBD, after the cell re-selection duration is determined from TS 25.304.

Specific Message Content

SYSTEM System INFORMATION Information BLOCK Block TYPE Type 112 (Step 1)

Information Element	Value/Remark
References to other system information blocks	Not Present
FACH measurement occasion info	
 - k_UTRAFACH Measurement occasion cycle 	2
length coefficient	
 Inter-frequency FDD measurement indicator 	<u>FALSE</u>
 Inter-frequency TDD measurement indicator 	<u>FALSE</u>
- Other RAT present in inter-system cell	Not Present
infoInter-RAT measurement indicators	
Measurement control system information	Not Present
 Intra-frequency measurement system 	
information	Not Present
 Inter-frequency measurement system 	
information	Not Present
 Inter-system-<u>RAT</u> measurement system 	
information	Not Present
 Traffic volume measurement system 	
information	Not Present
 UE Internal measurement system information 	

RRC CONNECTION SETUP (Step 4)

Use the same message sub-type found in Clause 9 of TS 34.108, which is entitled "Transition to CELL_DCH"

RRC CONNECTION SETUP COMPLETE (Step 5)

Information Element	Value/Remarks		
START List			
- CN Domain Identity	Check to see if it is present for all supported		
•	CN domains		
- START	Check to see if it is present for all supported		
	CN domains		
UE Radio access capability			
- ICS Version	Checked to see if set to 'R99'		
- PDCP capability	Not checked.		
- RLC capability	Not checked.		
- Transport channel capability	Not checked.		
- RF capability	Not checked.		
- Physical channel capability	Not checked.		
 UE multi-mode/multi-RAT capability 	Not checked.		
- Security capability	Not checked.		
- LCS capability	Not checked.		
 Measurement capability 			
 FDD measurements DL 	Checked to see if set to 'TRUE'		
 TDD measurements DL 	Not checked.		
 GSM measurements DL 	Not checked.		
- GSM 900 DL	Not checked.		
- DCS 1800 DL	Not checked.		
- GSM 1900 DL	Not checked.		
 Multi-carrier measurement DL 	Not checked.		
 FDD measurements UL 	Checked to see if set to 'TRUE'		
- TDD measurements UL	Not checked.		
 GSM measurements UL 	Not checked.		
- GSM 900 UL	Not checked.		
- DCS 1800 UL	Not checked.		
- GSM 1900 UL	Not checked.		
 Multi-carrier measurement UL 	Not checked.		
UE system specific capability	Not checked.		

PHYSICAL CHANNEL RECONFIGURATION (Step 6)

Use the same message sub-type found in Annex A, which is entitled "(Packet to CELL_DCH from CELL_DCH in PS)", with the following exceptions in the IE(s) concerned:

Information Element	Value/Remarks		
Downlink information common for all radio links			
 Downlink DPCH info common for all RL 			
- CHOICE Mode	FDD		
 Downlink DPCH power control information 			
- DPC mode	0 (Single)		
 DL rate matching restriction information 	Not Present		
- Spreading factor	Refer to the parameter set in TS 34.108		
- Fixed or flexible position	Flexible		
- TFCI existence	FALSE		
 Number of bits for Pilot bits (SF=128, 256) 	Not Present		
- DPCH compressed mode info			
- TGPSI	1		
- TGPS Status Flag	Active		
- TGCFN	<u>255</u>		
- Transmission gap pattern sequence			
configuration parameters			
- TGMP	FDD Measurement		
- TGPRC	62		
- TGCFN	(Current CFN + (256 - TTI/10msec)) mod 256		
- TGSN	8		
- TGL1	10		
- TGL2	5		
- TGD	15		
- TGPL1	35		
- TGPL2	35		
- RPP	Mode 1		
- ITP	Mode 1		
- UL/DL Mode	DL		
 Downlink compressed mode method 	SF/2		
 Uplink compressed mode method 	Not Present		
 Downlink frame type 	A		
- DeltaSIR1	2.0		
- DeltaSIRAfter1	1.0		
- DeltaSIR2	Not Present		
- DeltaSIRAfter2	Not Present		
- TX Diversity Mode	None		
- SSDT information	Not Present		
- S field			
- Code Word Set			
- Default DPCH Offset Value	0		

L.C	V-1 - /D 1
Information Element	Value/Remark
Measurement Identity Number	15
Measurement Command	Setup
Measurement Reporting Mode	Not Present
Additional measurements list	Not Present
CHOICE measurement type	Inter-frequency measurement
- Inter-frequency cell info list	
 Removed CHOICE inter-frequency cell 	Not PresentNo inter-frequency cells removed
removalinfo list	
 New inter-frequency info list 	Set to id of cell 4
- Inter-frequency cell id	
- Frequency info	UARFCN of the uplink frequency for cell 4
- UARFCN uplink (Nu)	UARFCN of the downlink frequency for cell 4
- UARFCN downlink (Nd)	
- Cell info	0 dB
 Cell individual offset 	0 chips
 Reference time difference to cell 	
- Primary CPICH Info	Set to same code as used for cell 4
 Primary Scrambling Code 	Not Present
- Primary CPICH TX power	FALSE
- Read SFN Indicator	FALSE
- TX Diversity Indicator	
 Inter-frequency measurement quantity 	Inter-frequency reporting criteria
 CHOICE reporting criteria 	0
- Filter Coefficient	CPICH Ec/No
 Measurement quantity for frequency quality 	
estimate	
 Inter-frequency reporting quantity 	FALSE
- UTRA Carrier RSSI	TRUE
- Frequency quality estimate	
- Non frequency related cell reporting quantities	
- SFN-SFN observed time difference reporting	No report
<u>indicator</u>	
 Cell synchronisation information reporting 	<u>FALSE</u>
indicator	541.05
- Cell Identity reporting indicator	FALSE
- CPICH Ec/No reporting indicator	TRUE
- CPICH RSCP reporting indicator	FALSE
- Pathloss reporting indicator	FALSE
	FALSE
- Reporting cell status	Not present
 CHOICE reporteding cell 	B 4 11 W 30 1 1 1 1 1
	Report cells \within active and/or monitored
	cells set on used frequency and or within active
- Maximum number of report <u>ed</u> ing cells type2	and/or monitored cells set on non-used
- Measurement validity	frequency
- Inter-frequency set update	2
- CHOICE report criteria	Not present
- Amount of reporting	Not Present
- Reporting interval	Periodic reporting criteria
DPCH compressed mode status info	Infinity
	8 seconds
	Not Present

MEASUREMENT REPORT (Step 9)

Information Element	Value/Remarks
Measurement identity-number	Check to see if set to 15
Measured Results	
- CHOICE measurement	Check to see if set to "Inter-frequency measured results list"
 Inter-frequency measurement results 	
- Frequency info	
- UARFCN (uplink)	Check to see if set to the UARFCN of the
	uplink frequency for cell 4
- UARFCN (downlink)	Check to see if set to the UARFCN of the
Or it it is it (downline)	downlink frequency for cell 4
- UTRA carrier RSSI	Check to see if it is absent
	Check to see if it is absent
- Inter-frequency cell measurement results	
- Cell measured results	
- Cell Identity	Check to see if it is absent
 SFN-SFN observed time difference 	Check to see if it is absent
	Check to see if it is absent
 Cell synchronisation information 	Check to see if it is absent
- Primary CPICH Info	
- Primary Scrambling Code	Check to see if set to the same code for cell 4
- CPICH Ec/No	Check to see if it is present
- CPICH RSCP	Check to see if it is absent
- Pathloss	Check to see if it is absent
Measured Results on RACH	Check to see if it is absent
Event Results	Check to see if it is absent

PHYSICAL CHANNEL RECONFIGURATION (Step 10)

Use the same message sub-type found in Annex A, which is entitled "(Packet to CELL_FACH from CELL_DCH in PS)". $\underline{2}$

Master Information Block (Step 12)

Information Element	<u>Value/Remarks</u>
MIB value tag	<u>2</u>

System Information Block type 11 (Step 12)

Information Element	<u>Value/Remarks</u>
Measurement control system information	
- Use of HCS	Not used
- Cell selection and reselection -	CPICH_Ec/No
quality_measure	
 Intra-frequency measurement system 	Not Present
<u>information</u>	
 Inter-frequency measurement system 	
<u>information</u>	
 Inter-frequency cell info list 	
 CHOICE Inter-frequency cell removal 	No inter-frequency cells removed
 New inter-frequency cells 	
- Inter-frequency cell id	<u>0</u>
- Frequency info	
- CHOICE mode	<u>FDD</u>
- UARFCN uplink (Nu)	Set to uplink UARFCN for cell 4
- UARFCN downlink (Nd)	Set to downlink UARFCN for cell 4
- Cell info	
- Cell individual offset	Not Present – use default of 0 dB
- Reference time difference to cell	Not Present
- CHOICE Mode	<u>FDD</u>
- Primary CPICH info	
- Primary scrambling code	Set to the scrambling code of cell 4
- Primary CPICH Tx power	Not Present
- Read SFN indicator	FALSE
- Cell selection and re-selection info	Not Present
- Inter-RAT measurement system information	Not Present
- Traffic volume measurement system	Not Present
information	
- UE Internal measurement system information	Not Present

CELL UPDATE (Step 15)

Information Element	Value/Remarks
U-RNTI	Check to see if same to value as in RRC
	CONNECTION SETUP message
Integrity check info	Check to see if it is absent
AM_RLC error indication (for C-plane)	Check to see if it is set to FALSE
AM_RLC error indication (for U-plane)	Check to see if it is set to FALSE
START List	Checked to see if the 'CN domain identity' and
	'START' IEs are present for all CN domains
	supported by the UE
Cell update cause	Check to see if it is set to "Cell Reselection"
Protocol error info	Check to see if it is absent or set to FALSE
Measured results on RACH	Check to see if it is absent
Protocol error information	Check to see if it is absent

CELL UPDATE CONFIRM (Step 16)

Use the same message sub-type found in Annex A.

8.4.1.6.5 Test Requirement

After step 4 the UE shall transmit RRC CONNECTION SETUP COMPLETE message with the IE "Measurement capability", indicating that both uplink and downlink inter-frequency measurements for FDD mode are supported.

After step 8 the UE shall transmit MEASUREMENT REPORT message to report cell 4's CPICH Ec/No value.

After step 11 the UE shall stop sending MEASUREMENT REPORT messages, which contain inter-frequency measured results for cell 4's CPICH Ec/No value.

After step 14 the UE shall transmit CELL UPDATE message to inform that a cell reselection to cell 4 has occurred.

8.4.1.7 Measurement Control and Report: Intra-frequency measurement for transition from CELL FACH to CELL DCH state

8.4.1.7.1 Definition

8.4.1.7.2 Conformance requirement

When transiting from CELL_FACH state to CELL_DCH state, the UE shall resume intra-frequency measurement if it has previously stored such a measurement context in CELL_DCH state, and Hift thise measurement context indicates reporting in is indicated to be resume in CELL_DCH state, the UE shall also re-start the associated reporting activities for the resumed intra-frequency measurement. If the UE has performed a cell reselection whilst out of CELL_DCH state, the UE shall not re-start intra-frequency measurement previously designated to be resumed in CELL_DCH state.

-In the case when the UE is not assigned any measurement tasks, it shall <u>continue to</u> monitor the list of <u>intra-frequency</u> neighbouring cells stated in <u>sSystem iInformation Block type 11 or 12</u> messages. It shall transmit MEASUREMENT REPORT messages when the reporting criteria (if specified in <u>sSystem Iinformation Block type 11 or 12</u> messages) are met. When in CELL_DCH state, the UE shall override existing measurement and reporting contexts <u>extracted-obtained</u> from <u>listening to <u>sSystem iInformation Block type 11 or 12</u> messages, if a MEASUREMENT CONTROL message is received. The UE shall then apply the new measurement and reporting parameters <u>received in the MEASUREMENT CONTROL message.accordingly</u>. If the UE has performed a cell reselection whilst out of CELL_DCH state, the UE shall not re start intra frequency measurement previously designated to be resumed in CELL_DCH state.</u>

Reference

3GPP TS 25.331, clause 8.4.1.87.1

8.4.1.7.3 Test Purpose

To confirm that UE resumes intra-frequency measurements and the associated reporting when it enters CELL_DCH state from CELL_FACH state, and that if-such a-measurement contexts (and optionally, the reporting context) haves been stored for resumption in before entering CELL_DCH state. To confirm that the UE continues to monitor the intra-frequency neighbour cells intra-frequency measurement listed found in the sSystem iInformation Block type 11 or 12 messages, if no previously assigned measurements is are present. To confirm that the UE transmits MEASUREMENT REPORT messages if reporting conditions stated in sSystem iInformation Block type 11 or 12 messages have been satisfied. To confirm that a MEASUREMENT CONTROL message received in CELL_DCH state results in the overriding of current overrides the measurement and associated reporting contexts maintained by in the UE by virtue of System Information Block type 11 or 12 messages.

8.4.1.7.4 Method of test

Initial Condition

System Simulator: 3 cells – The initial configurations of the 3 cells in the SS should follow the values indicated in the column marked "TO" in table 8.4.1.7-1. The table is found in "Test Procedure" sub-clause.

System Simulator: 3 cells—All 3 are all active. The transmission power (RSCP) for CPICH of the 3 cells are as follow: cell 1 (40dBm), cell 2 (55dBm) and cell 3 (70dBm).

UE: CS-CELL_FACH_Initial (state 6-2) or PS-CELL_FACH_Initial (state 6-4) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE.

Test Procedure

Table 8.4.1.7-1 illustrates the downlink power to be applied for the 3 cells at various time instants of the test execution. Column marked "T0" denotes the initial conditions, while columns marked "T1" are to be applied subsequently. The exact instants on which these values shall be applied are described in the text in this sub-clause.

Table	8.4	.1.	.7-1
--------------	-----	-----	------

Para-meter	<u>Unit</u>	Ce	<u>II 1</u>	Ce	<u>II 2</u>	Ce	II <u>3</u>
		<u>T0</u>	<u>T1</u>	<u>T0</u>	<u>T1</u>	<u>T0</u>	<u>T1</u>
UTRA RF		<u>Ch. 1</u>		<u>Ch. 1</u>		<u>Ch. 1</u>	
<u>Channel</u>							
<u>Number</u>							
\hat{I}_{or}/I_{oc}	<u>dB</u>	<u>7.9</u>	<u>-5.0</u>	7.3	<u>7.3</u>	<u>5.7</u>	<u>5.7</u>
CPICH	<u>dB</u>	=	=	=	=	=	=
Ec/lo		<u>14.2</u>	<u>25.2</u>	<u>14.8</u>	<u>12.9</u>	<u>16.4</u>	<u>14.5</u>
CPICH RSCP	<u>dBm</u>	<u>-</u> 72.1	<u>-</u> 85.0	<u>-</u> 72.7	<u>-</u> 72.7	<u>-</u> 74.3	<u>-</u> 74.3

The UE is brought to CELL_FACH state <u>in cell 1</u>, after -it has successfully executed procedure P4 or P6 (depending on the CN domain supported by the UE) as specified in clause 7.4 of TS 34.108. The SYSTEM INFORMATION BLOCK TYPE-System Information Block type 12 message is changed with respect to the default message contents, specifying that cell 2 is to be included in the neighbouring cell list for intra-frequency measurement. Event 1e is selected in IE "Intra-frequency measurement received measurement quantity" is set to CPICH Ec/No.

SS send a PHYSICAL CHANNEL RECONFIGURATION message to UE, allocating dedicated physical channels on both uplink and downlink directions. Upon receiving such a message, the UE shall return PHYSICAL CHANNEL RECONFIGURATION COMPLETE message and then move to CELL DCH state. The UE shall send MEASUREMENT REPORT messages to indicate that the measured result of cell 2's CPICH Ec/No value, as the measurement quantity has exceeded the threshold value in sSystem iInformation Block type 12 messages. After receiving theis MEASUREMENT REPORT messages, SS transmits a MEASUREMENT CONTROL message in which it specifies that only intra-frequency measurement and periodic reporting for cell 3 's CPICH Ec/No shall be performed. The reporting criteria are the same as in SYSTEM INFORMATION BLOCK TYPE 12 message, except that the reporting threshold is lowered. After receiving such a message, the UE shall transmit another set of MEASUREMENT REPORT messages. SS verifies that only measurement readings for cell 3 's CPICH Ec/No is-are included in these messages. and no IEs shall be present, which pertain to measurement quantity in cell 2. After receiving the MEASUREMENT REPORT message reporting readings for cell 3, SS transmits another MEASUREMENT CONTROL message. The key parameters specified in this message are: measurement command = 'setup', measurement type = 'intra frequency measurement', measurement object = 'cell 2', reporting criteria = 'periodic reporting', measurement validity = 'resume', UE state for reporting = 'CELL_DCH'. SS verifies that the UE continue to send MEASUREMENT REPORT messages at 16 seconds interval, and that the contents of these messages are similar to those received prior to the transmission of the second downlink MEASUREMENT CONTROL messages.

Next, SS sends PHYSICAL CHANNEL RECONFIGURATION message to UE. In this message, the physical channel resources are switched to common physical channels - PRACH for the uplink and S-CCPCH for the downlink. The UE shall reply with a PHYSICAL CHANNEL RECONFIGURATION COMPLETE and transits to CELL_FACH. SS waits for 16 seconds and checks the uplink RACH to confirm that no MEASUREMENT REPORT messages are received. SS transmits MEASUREMENT CONTROL message on the downlink DCCH. The key parameters specified in this message are: measurement command = 'setup', measurement type = 'intra-frequency measurement', measurement object = 'cell 2', reporting criteria = 'periodic reporting', measurement validity IE is present and "UE state" = "CELL DCH". SS waits for 160 seconds, verifies that no MEASUREMENT REPORT messages are detected on the uplink DCCH, before sending another PHYSICAL CHANNEL RECONFIGURATION message, comprising allocating DPCH physical channels to the UE. UE shall then return to CELL_DCH state, start to monitor the neighbour cell specified by the SS in the second-latest MEASUREMENT CONTROL message while the UE was previously in CELL DFACH state. The UE shall resume periodic reporting of cell 2's CPICH RSCP measured results by sending MEASUREMENT REPORT messages. Following the reception of the MEASUREMENT REPORT message, SS commands the UE to stop performing measurements and generation of reports for cell 2 CPICH RSCP. Thereafter, SS then verifies that no MEASUREMENT REPORT messages are detected. After this requirement is satisfied, SS sends MEASUREMENT CONTROL on the downlink DCCH once more. This message is identical to the one sent in step 710 (see specific message content).

In the next sequence, SS dispatches a PHYSICAL CHANNEL RECONFIGURATION message on the downlink DCCH. In this message, common physical channel resources are assigned to the UE. The UE shall respond with a

PHYSICAL CHANNEL RECONFIGURATION COMPLETE message and then transit to CELL_FACH state. SS monitor the uplink DCCH once more to verify that no MEASUREMENT REPORT messages are detected. Following this step, SS increases the P CPICH RSCP of cell 3 to 45dBm and then stops the transmission of all downlink physical channels of cell 1.SS modifies the contents of Master Information Block and System Information Block type 12 messages, followed by a reconfiguration of the downlink transmission power of the respect cells according to the settings in columns "T1" in Table 8.4.1.7-1. SS starts timer T305 and then waits for timer T305it to expire. The UE shall discover an "out-of-service" condition and initiate a cell re-selection procedure. This is verified in the SS when a CELL UPDATE message is received on the uplink CCCH with the "cell update cause" IE set to "cell reselection". SS transmits a CELL UPDATE CONFIRM message on the DCCH to end the cell update procedure. Next, SS sends a PHYSICAL CHANNEL RECONFIGURATION message on the downlink DCCH, assigning DPCH channel in both uplink and downlink directions. The UE shall respond with a PHYSICAL CHANNEL RECONFIGURATION COMPLETE message and then return to CELL_FADCH state. SS checks that the UE does not generate any MEASUREMENT REPORT messages on the uplink DCCH.

Expected Sequence

Step	Direction	Message	Comment
	UE SS		
1	←	SYSTEM INFORMATION BLOCK	UE is initially in CELL_FACH
		TYPESystem Information Block type 12	in cell1, after having
			successfully executed
			procedure P4 or P6,
			depending on the supported
			CN domain. Refer to clause
			7.4 of TS 34.108 for details.
			TYPE BLOCK TYPESystem Information Block type 12
			messages is are changed
			according to the descriptions
			in <u>"Specific Message</u>
			Contents" sub-clause.
2	←	PHYSICAL CHANNEL	Allocates DPCH physical
_	`	RECONFIGURATION	channels.
3	\rightarrow	PHYSICAL CHANNEL	UE shall move to CELL_DCH
3	,	RECONFIGURATION COMPLETE	state.
4	\rightarrow	MEASUREMENT REPORT	Reports cell 2's RSCP_CPICH
7	,	WEAGONEWENT REFORT	Ec/No measurement value.
5	+	MEASUREMENT CONTROL	Specifies cell 3 as the
5	`		measurement object for intra-
			frequency measurement. The
			reporting criteria is the same
			as defined in SYSTEM
			INFORMATION BLOCK TYPE
			12 message, with the
			exception that threshold for
			event 1e is lowered.
6	\rightarrow	MEASUREMENT REPORT	UE shall report the estimated
			value for cell 3's CPICH RSCP
			Ec/No reading only.
7	- ←	MEASUREMENT CONTROL	SS instructs the UE to perform
			intra-frequency measurement
			and reporting for cell 2. These
			activities should be resumed if
			the UE subsequently transits
		MEANUEL PERSON	to CELL_DCH state again.
8	- →	MEASUREMENT REPORT	SS checks that the content of
			this message is the same as
07	←	DUVEICAL CHANNEL	that received in step 6.
9 7	~	PHYSICAL CHANNEL	Allocates PRACH and S-
100	\rightarrow	RECONFIGURATION PHYSICAL CHANNEL	CCPCH physical channels. UE shall move to CELL_FACH
10 8	7	RECONFIGURATION COMPLETE	state.
11 9		INCOUNT IOUNATION COMPLETE	SS waits for 16 seconds and
++ 3			checks that no
			MEASUREMENT REPORT
			messages are sent by UE.
10		MEASUREMENT CONTROL	SS instructs the UE to perform
10		WILAGUILINI CONTROL	intra-frequency measurement
			and reporting for cell 2. These
			activities should be resumed if
			the UE subsequently transits
		1	OF I DOI I I I I I I I I I I I I I I I I I
			I to CELL DCH state again
11			to CELL DCH state again. SS once again waits for 16
<u>11</u>			SS once again waits for 16
<u>11</u>			SS once again waits for 16 seconds and verifies that no
<u>11</u>			SS once again waits for 16 seconds and verifies that no MEASUREMENT REPORT
	←	PHYSICAL CHANNEL	SS once again waits for 16 seconds and verifies that no MEASUREMENT REPORT messages are sent by UE.
<u>11</u>	-	PHYSICAL CHANNEL RECONFIGURATION	SS once again waits for 16 seconds and verifies that no MEASUREMENT REPORT messages are sent by UE. SS waits for 10 seconds
	←	PHYSICAL CHANNEL RECONFIGURATION	SS once again waits for 16 seconds and verifies that no MEASUREMENT REPORT messages are sent by UE. SS waits for 10 seconds before sending this message.
	+		SS once again waits for 16 seconds and verifies that no MEASUREMENT REPORT messages are sent by UE. SS waits for 10 seconds

	13	\rightarrow	PHYSICAL CHANNEL	UE shall return to CELL_DCH
			RECONFIGURATION COMPLETE	state.
	14	\rightarrow	MEASUREMENT REPORT	UE begins to report cell 2's
				measured results for CPICH
				RSCP again.
	15	←	MEASUREMENT CONTROL	Terminate all the intra-
				frequency measurement
				activity related to cell 2.—SS
				checks that no more
				MEASUREMENT REPORT
				messages are sent by the UE.
	16			SS waits for 16 seconds and
				verifies that UE stop
				transmitting MEASUREMENT
				REPORT messages.
	17	←	MEASUREMENT CONTROL	This Mmessage is the same
				as in step 7 10
•	18	+	PHYSICAL CHANNEL	Allocates PRACH and S-
	-		RECONFIGURATION	CCPCH physical channels.
ļ	19	\rightarrow	PHYSICAL CHANNEL	UE shall move to CELL_FACH
		-	RECONFIGURATION COMPLETE	state.
•	20			SS checks that no
				MEASUREMENT REPORT
				messages are received.
1 -	21	_ ←	Master Information Block	SS increases the transmission
	21		System Information Block type 12	power of cell 3 (P-CPICH
			System mornation block type 12	RSCP is raised to -45dBm)
				and switches off all downlink
				transmissions for cell 1.
				System Information Block type
				12 messages are modified to
				include cell 2 and cell 3 into
				neighbouring cells monitored
				list for intra-frequency type
				measurements. SS
				reconfigures the downlink
				transmission power settings
				for cell 1 to cell 3 according to
				columns "T1" in Table 8.4.1.7-
				1, runs timer T305, and then
			OF LUNDATE	waits until T305 expires.
	22	\rightarrow	CELL UPDATE	UE shall re-selects to cell 2
				and then perform a cell update
				procedure.
	23	←	CELL UPDATE CONFIRM	UE shall stay in CELL_FACH
ļ				state.
	24	\rightarrow	PHYSICAL CHANNEL	DPCH physical channels are
			RECONFIGURATION	assigned to the UE in this
				message.
Ī	25	+	PHYSICAL CHANNEL	UE shall return to CELL_DCH
			RECONFIGURATION COMPLETE	state.
Ī	26			SS checks that no
				MEASUREMENT REPORT
				messages are received on
				uplink DCCH.
L				

Specific Message Content

SYSTEM INFORMATION BLOCK TYPE-System Information Block type 12 (Step 1)

Information Element	Value/Remark
Reference to other system information blocks	Not Present
FACH measurement occasion info	Not Present
Measurement control system information	
- Intra-frequency measurement system	
information	
- Intra-frequency measurement identity-number	10
- Intra-frequency cell info list	

- CHOICE Removed intra-frequency info listcell Not PresentRemove no intra-frequency cells removal - New intra-frequency info list Set to id of cell 2 - Intra-frequency cell id - Cell info 0 dB - Cell individual offset 0 chips - Reference time difference to cell - Primary CPICH Info Set to same code as used for cell 2 - Primary Scrambling Code Not Present - Primary CPICH TX power **FALSE** - Read SFN Indicator **FALSE** - TX Diversity Indicator Not Present - use default values - Cell selection and Re-selection info-for SIB11/12 OdB Qoffset_{s.n} 0dBm Maximum allowed UL TX power Not Present HCS neighbouring cell information -20dB, -115dBm **Qqualmin**, Qrxlevmin - Intra-frequency measurement quantity - Filter Coefficient CPICH Ec/No - Measurement quantity Not Present - Intra-frequency measurement for RACH reporting No report - Maximum number of reported cells on RACH - Reporting information for state CELL_DCH - Intra-frequency reporting quantity - Reporting quantities for active set cells No report - SFN-SFN observed time difference reporting **FALSE** - Cell synchronisation information reporting **FALSE** indicator - Cell identity reporting indicator **FALSE** - CPICH Ec/No reporting indicator **FALSE** - CPICH RSCP reporting indicator **FALSE** - Pathloss reporting indicator **FALSE** CFN-SFN observed time difference - Reporting quantities for monitored set cells No report - SFN-SFN observed time difference reporting **FALSE** indicator - Cell synchronisation information reporting **TRUE** indicator **FALSE**TRUE - Cell identity reporting indicator - CPICH Ec/No reporting indicator **TRUE**FALSE - CPICH RSCP reporting indicator **FALSE** - Pathloss reporting indicator **FALSE** CFN-SFN observed time difference Not present - Reporting quantities for detected cells Intra-frequency measurement reporting criteria - CHOICE report criteria - Parameter required for each event Not Present - Intra-frequency event identity Monitored set cells Triggering condition 1 Not present - Triggering condition 2 Not present - Reporting range FDD - Cells forbidden to affect reporting Set to the scrambling code of cell 2 - CHOICE Mode - Primary CPICH Info Not present Not Present0 dB - Primary scrambling code -60dBm18 dB - W - Hysteresis Not present - Threshold used frequency Not present - Reporting deactivation threshold - Replacement activation threshold Infinity 16 seconds - Time to trigger - Amount of reporting - Reporting Interval Report cells within monitored set cells on used - Reporting cell status frequency

- CHOICE reported cells

	Not Present
 Maximum number of reported cells 	
- Inter-frequency measurement system	Not Present
information	Not Present
- Inter- system- RAT measurement system	
information	Not Present
- Traffic volume measurement system	
information	
- UE internal measurement system information	

PHYSICAL CHANNEL RECONFIGURATION (Step 2, Step 12 and Step 24)

Use the same message sub-type found in Annex A, which is entitled "Packet to CELL_DCH from CELL_FACH in PS"

MEASUREMENT REPORT (Step 4)

Information Element	Value/Remarks
Measurement identity-number	Check to see if set to 10
Measured Results	
- CHOICE measurement	Check to see if set to "Intra-frequency measured results list"
 Intra-frequency measured results 	
 Cell measured results 	
- Cell Identity	Check to see if it is absent
 SFN-SFN observed time difference 	Check to see if this IE is absent
- CFN-SFN observed time difference	Check to see if this IE is absent
 Cell synchronisation information 	Check to see if this IE is absent
- Primary CPICH Info	
- Primary Scrambling Code	Check to see if it's the same code for cell 2
- CPICH Ec/No	Check to see if this IE is absentpresent
- CPICH RSCP	Check to see if this IE is presentabsent
- Pathloss	Check to see if this IE is absent
Measured Results on RACH	Check to see if this IE is absent
Event Results	
- CHOICE event result	Check to see if it's set to 'Intra-frequency
	measurement event results'
 Intra-frequency event identity 	Check to see if this IE is set to '1e'
- Cell measured event results	
- Primary CPICH info	
- Primary scrambling code	Check to see if it's the same code for cell 2

MEASUREMENT CONTROL (Step 5)

Information Element	Value/Remark
inioiniation Lienent	value/itelliai k

Measurement Identity Number Measurement Command Setup Measurement Reporting Mode Not Present Additional measurements list Not Present CHOICE measurement type Intra-frequency measurement - Intra-frequency cell info list - Removed-CHOICE intra-frequency cell Not PresentRemove no intra-frequency cells removalinfo list - New intra-frequency info list Set to id of cell 3 - Intra-frequency cell id - Cell info 0 dB - Cell individual offset 0 chips - Reference time difference to cell - Primary CPICH Info Set to same code as used for cell 3 - Primary Scrambling Code Not Present - Primary CPICH TX power **FALSE** - Read SFN Indicator **FALSE** - TX Diversity Indicator Not Present Cell selection and Re-selection info - Intra-frequency measurement quantity - Filter Coefficient CPICH RSCPEc/No - Measurement quantity - Intra-frequency reporting quantity - Reporting quantities for active set cells No report - SFN-SFN observed time difference reporting indicator **FALSE** - Cell synchronisation information reporting indicator **FALSE FALSE** - Cell identity reporting indicator - CPICH Ec/No reporting indicator **FALSE** - CPICH RSCP reporting indicator **FALSE** - Pathloss reporting indicator FALSE CFN-SFN observed time difference - Reporting quantities for monitored set cells No report - SFN-SFN observed time difference reporting indicator **FALSE** Cell synchronisation information reporting **TRUE** indicator **FALSE**TRUE -- Cell identity reporting indicator -- CPICH Ec/No reporting indicator **TRUE**FALSE -- CPICH RSCP reporting indicator **FALSE EALSE** -- Pathloss reporting indicator CFN-SFN observed time difference Not present - Reporting quantities for detected cells Not present - Reporting cell status Not present - Measurement validity Intra-frequency measurement criteria - CHOICE report criteria - Parameters required for each event 1e - Intra-frequency event identity Not Present - Triggering condition 1 Monitored set cells - Triggering condition 2 Not Present Not Present - Reporting Range - Cells forbidden to affect Reporting range FDD CHOICE Mode _- Primary CPICH Info Set to the same scrambling code for cell 3 - Primary Scrambling Code Not Present Not Present0 dB Not Present - Hysteresis - Reporting deactivation threshold Not Present - Replacement activation threshold -90dBm-19.5 dB - Reporting Threshold 0 - Time to Trigger Infinity - Amount of reporting 16 seconds - Reporting interval - Reporting cell status Report cells within monitored set cells on used

frequency 1 Not Present

- CHOICE reported cells

DPCH compressed mode status info

- Maximum number of reported cells

MEASUREMENT REPORT (Step 6-and step 8)

Information Element	Value/Remarks
Measurement identity number	Check to see if set to 11
Measured Results	
- CHOICE measurement	Check to see if set to "Intra-frequency measured results list"
 Intra-frequency measured results 	
 Cell measured results 	
- Cell Identity	Check to see if it is absent
 SFN-SFN observed time difference 	Check to see if this IE is absent
- CFN-SFN observed time difference	Check to see if this IE is absent
 Cell synchronisation information 	Check to see if this IE is absent
- Primary CPICH Info	
- Primary Scrambling Code	Check to see if it's the same code for cell 3
- CPICH Ec/No	Check to see if this IE is absentpresent
- CPICH RSCP	Check to see if this IE is presentabsent
- Pathloss	Check to see if this IE is absent
Measured Results on RACH	Check to see if this IE is absent
Event Results	
- CHOICE event result	Check to see if it's set to 'Intra-frequency measurement event results'
 Intra-frequency event identity 	Check to see if this IE is set to '1e'
- Cell measured event results	
- Primary CPICH info	
- Primary scrambling code	Check to see if it's the same code for cell 3

PHYSICAL CHANNEL RECONFIGURATION (Step 97 and 18)

Use the same message sub-type found in Annex A, which is entitled "Packet to CELL_FACH from CELL_DCH in PS"

MEASUREMENT CONTROL (Step 7-10 and 17)

Information Floment	Value/Pemark
Information Element	Value/Remark

Measurement Identity Number 12 Measurement Command Setup Measurement Reporting Mode Not Present Additional measurements list Not Present CHOICE measurement type Intra-frequency measurement - Intra-frequency cell info list - CHOICE intra- frequency cell Rremovaled Remove no intra-frequency cellsNot Present intra-frequency info list - New intra-frequency info list Set to id of cell 2 - Intra-frequency cell id - Cell info 0 dB - Cell individual offset 0 chips - Reference time difference to cell - Primary CPICH Info Set to same code as used for cell 2 - Primary Scrambling Code Not Present - Primary CPICH TX power **FALSE** - Read SFN Indicator **FALSE** - TX Diversity Indicator Not Present Cell selection and Re-selection info - Intra-frequency measurement quantity - Filter Coefficient **CPICH RSCP** - Measurement quantity - Intra-frequency reporting quantity - Reporting quantities for active set cells No report - SFN-SFN observed time difference reporting indicator **FALSE** - Cell synchronisation information reporting **FALSE** indicator **FALSE** - Cell identity reporting indicator - CPICH Ec/No reporting indicator **FALSE** - CPICH RSCP reporting indicator **FALSE FALSE** - Pathloss reporting indicator CFN-SFN observed time difference - Reporting quantities for monitored set cells No report - SFN-SFN observed time difference reporting indicator **FALSE** Cell synchronisation information reporting **TRUE** indicator **FALSE** - Cell identity reporting indicator - CPICH Ec/No reporting indicator **TRUE** - CPICH RSCP reporting indicator **FALSE** - Pathloss reporting indicator FALSE CFN-SFN observed time difference Not present - Reporting quantities for detected cells Not present - Reporting cell status Resume - Measurement validity CELL DCH - UE state Intra-frequency measurement criteria - CHOICE report criteria - Parameters required for each event - Intra-frequency event identity Not Present - Triggering condition 1 Monitored set cells - Triggering condition 2 Not Present - Reporting Range Not Present - Cells forbidden to affect Reporting range - Primary CPICH Info Set to the same scrambling code for cell 2 - Primary Scrambling Code Not Present - W Not Present0 dB Not Present - Hysteresis - Reporting deactivation threshold Not Present - Replacement activation threshold -6580 dBm - Reporting Threshold 0 - Time to Trigger Infinity - Amount of reporting 16 seconds - Reporting interval - Reporting cell status Report cells within monitored set cells on used - CHOICE reporteding cell frequencyWithin monitored cells on used frequency and within monitored cells on non-- Maximum number of reported cells Maximum used frequency number of reporting cells type2

DPCH compressed mode status info

Not Present

PHYSICAL CHANNEL RECONFIGURATION (Step 12 and 24)

Same as in step 2

MEASUREMENT REPORT (Step 14)

Information Element	Value/Remarks
Measurement identity-number	Check to see if set to 12
Measured Results	
- CHOICE measurement	Check to see if set to "Intra-frequency measured results list"
 Intra-frequency measured results 	
 Cell measured results 	
- Cell Identity	Check to see if it is absent
 SFN-SFN observed time difference 	Check to see if this IE is absent
- CFN-SFN observed time difference	Check to see if this IE is absent
 Cell synchronisation information 	Check to see if this IE is absent
- Primary CPICH Info	
- Primary Scrambling Code	Check to see if it's the same code for cell 2
- CPICH Ec/No	Check to see if this IE is absent
- CPICH RSCP	Check to see if this IE is present
- Pathloss	Check to see if this IE is absent
Measured Results on RACH	Check to see if this IE is absent
Event Results	
- CHOICE event result	Check to see if it's set to 'Intra-frequency measurement event results'
- Intra-frequency event identity	Check to see if this IE is set to '1e'
- Cell measured event results	
- Primary CPICH info	
- Primary scrambling code	Check to see if it's the same code for cell 2

MEASUREMENT CONTROL (Step 15)

Information Element	Value/Remarks
Measurement Identity-Number	12
Measurement Command	Release
Measurement Reporting Mode	Not Present
Additional measurements list	Not Present
CHOICE Measurement type	Not Present
DPCH compressed mode status info	Not Present

Master Information Block (Step 21)

Information Element	<u>Value/Remarks</u>
MIB Value Tag	2

System Information Block type 12 (Step 21)

Information Element	<u>Value/Remark</u>
FACH measurement occasion info	Not Present
Measurement control system information	
- Use of HCS	Not used
- Cell_selection_and_reselection	CPICH_Ec/No
quality measure	
- Intra-frequency measurement system	13
information	
- Intra-frequency measurement identity	Not Present
- Intra-frequency cell info list	Cell 2 and Cell 3 are added
- CHOICE intra-frequency cell removal	Remove no intra-frequency cells
- New intra-frequency info list	<u> </u>
- Intra-frequency cell id	Set to id of cell 2
- Cell info	
- Cell individual offset	0 dB
- Reference time difference to cell	0 chips
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as used for cell 2
- Primary CPICH TX power	Not Present
- Read SFN Indicator	FALSE
- TX Diversity Indicator	FALSE
- Cell selection and Re-selection info	Not Present – use default values
- Intra-frequency cell id	Set to id of cell 3
- Cell info	
- Cell individual offset	<u>0 dB</u>
- Reference time difference to cell	0 chips
- Primary CPICH Info	
 Primary Scrambling Code 	Set to same code as used for cell 3
- Primary CPICH TX power	Not Present
- Read SFN Indicator	<u>FALSE</u>
 TX Diversity Indicator 	<u>FALSE</u>
 Cell selection and Re-selection info 	Not Present – use default values
 Intra-frequency measurement quantity 	
- Filter Coefficient	<u>0</u>
 Measurement quantity 	CPICH Ec/No
 Intra-frequency measurement for RACH 	Not Present
reporting	
 Maximum number of reported cells on RACH 	No report
 Reporting information for state CELL_DCH 	
 Inter-frequency measurement system 	Not Present
<u>information</u>	
 Inter-RAT measurement system information 	Not Present
- Traffic volume measurement system	Not Present
<u>information</u>	
 UE internal measurement system information 	Not Present

CELL UPDATE (Step 22)

Information Element	Value/Remarks	
U-RNTI		
- SRNC Identity	Check to see if set to '0000 0000 0000 0001'	
- S-RNTI	Check to see if set to '0000 0000 0000 0000	
	0000 0001'	
Integrity check info	Not checked	
START List	Checked to see if the 'CN domain identity' and	
	'START' IEs are present for all CN domains	
	supported by the UE	
AM_RLC error indicator (for C-plane)	Check to see if set to 'FALSE'	
AM_RLC error indicator (for U-plane)	Check to see if set to 'FALSE'	
Cell Update Cause	Check to see if set to 'Cell Re-selection'	
Protocol error indicator	Check to see if it is absent or set to 'FALSE'	
Measured results on RACH	Check to see if it is absent	
Protocol error information	Check to see if it is absent	

CELL UPDATE CONFIRM (Step 23)

Use the default message content of the same message type in Annex A.

8.4.1.7.5 Test Requirement

After step 3 the UE shall report cell 2's CPICH RSCP Ec/No value by transmitting a-MEASUREMENT REPORT messages.

After step 5 the UE shall delete all measurement and reporting contexts obtained from <u>listening to sSystem iInformation</u> messages. <u>However, iIt</u> shall transmit a-MEASUREMENT REPORT messages containing measured results of cell 3's CPICH RSCP value only, in accordance to the contents of MEASUREMENT CONTROL message transmitted by SS in step 5.

After step 7 the UE shall continue to transmit MEASUREMENT REPORT messages, which contain identical contents to that sent in step 6.

After <u>step 9 and step 11</u> the UE shall <u>stop-not transmit MEASUREMENT REPORT messages, which pertain to intra-frequency <u>type measurement reporting.</u></u>

After step 13 the UE shall resume the measurement and reporting activities as specified in MEASUREMENT CONTROL message received in step 7<u>10</u>. The UE shall transmit MEASUREMENT REPORT messages, containing measured results of cell 2's CPICH RSCP value.

After step 15 the UE shall stop all-measurement activities pertaining to periodic reporting of cell 2's CPICH RSCP, no MEASUREMENT REPORT messages shall be detectable by the SS in-on_the uplink DCCH.

After step 21 the UE shall re-select to cell 32 and initiate a cell update procedure. SS shall receive a CELL UPDATE message on the uplink CCCH of cell 32, with the "cell update cause" IE stated as "cell re-selection".

After step 25 the UE shall not start to performresume measurements and any associated reporting activities for cell 2's CPICH RSCP, no MEASUREMENT REPORT messages shall be detectable by the SS in the uplink DCCH.

8.4.1.8 Measurement Control and Report: Inter-frequency measurement for transition from CELL_FACH to CELL_DCH state

8.4.1.8.1 Definition

8.4.1.8.2 Conformance requirement

When transiting from CELL_FACH state to CELL_DCH state, the UE shall stop monitoring the list of inter-frequency neighbour cells indicated in <u>sSystem iInformation Block type 11 or 12</u> messages. If the UE has a previously stored inter-frequency measurement context marked as 'resume' <u>and for which the IE "UE state for reporting" has been assigned to "CELL DCH"</u>, it shall reinstate the suspended measurement and associated reporting activities after it has re-entered CELL_DCH state. The UE shall be able to start <u>for terminate inter-frequency measurements</u> by decoding the "DPCH compressed mode status info" IE in MEASUREMENT CONTROL messages.

Reference

3GPP TS 25.331 clause 8.4.1.8<u>7.2, 8.5.7.6.13</u>

8.4.1.8.3 Test Purpose

To confirm that the UE erases all inter-frequency measurement contexts <u>used-received from System Information Block type 11 or 12 while in CELL_FACH</u> state, when it moves to CELL_DCH. To confirm that the UE resumes inter-frequency measurements and reporting, <u>which was stored previously stored in the UE</u>, after it moves to CELL_DCH state. To confirm that the UE resumes inter-frequency measurement and reporting activities after it has received a MEASUREMENT CONTROL message specifying that a stored compressed mode pattern sequence be re-activated.

8.4.1.8.4 Method of test

Initial Condition

SS: 3 cells - All 3 cells are active and the applicable radio conditions are stated in the table below:

System Simulator: 3 cells – The initial configurations of the 3 cells in the SS should follow the values indicated in table 8.4.1.8-1. The table is found in "Test Procedure" sub-clause.

	Cell 1	Cell 4	Cell 5
UARFCN (N _u and N _d)	Channel 1	Channel 2	Channel 2
CPICH RSCP	-60 dBm	-65 dBm	-75 dBm
Q _{rxlevmin}	-90 dBm	-90 dBm	-90 dBm
\$	30	25	15

UE: CS-DCCH+DTCH_DCH (State 6-9) or PS-DCCH+DTCH_DCH (State 6-10) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE.

Test Procedure

Table 8.4.1.8-1 illustrates the downlink power to be applied for the 3 cells in this test.

Table 8.4.1.8-1

Para- meter	<u>Unit</u>	Cell 1	Cell 4	Cell 5
UTRA RF Channel Number		<u>Ch. 1</u>	<u>Ch. 2</u>	<u>Ch. 2</u>
\hat{I}_{or}/I_{oc}	<u>dB</u>	<u>9.7</u>	<u>7.3</u>	<u>5.7</u>
CPICH Ec/lo	<u>dB</u>	<u>-10.4</u>	<u>-12.7</u>	<u>-14.3</u>
CPICH RSCP	<u>dBm</u>	<u>-70.3</u>	<u>-72.7</u>	<u>-74.3</u>

The UE is in CELL_DCH state in cell 1, after successfully executing procedures P11 or P13 as specified in clause 7.4 of TS 34.108. Next, SS transmits MEASUREMENT CONTROL message to request the UE to execute an interfrequency measurement for cell 5. The parameters of the reporting criteria are as follow: event-triggered with event identity = '2c', reporting quantity = "CPICH RSCP", threshold for non-used frequency = '-850 dBm', hysteresis = '1.0dB', time to trigger = '10 seconds', amount of reporting = '1' and reporting interval = '0'. In the same message, IE "Measurement validity" is present set to 'resume' and "UE state-for reporting' is assigned the value 'CELL_DCH'. SS checks that no MEASUREMENT REPORT messages are detected on the uplink DCCH after it has transmitted the MEASUREMENT CONTROL message.

SS modifies the content of SYSTEM INFORMATION BLOCK TYPE 12 message, which indicates that cell 4 shall be included in the list of neighbouring cells to be monitored for inter frequency measurements.—Following this action, SS sends a PHYSICAL CHANNEL RECONFIGURATION message on the downlink DCCH and commands the UE to change its physical channel to PRACH and S-CCPCH. SS modifies the content of System Information Block type 12 messages, such that cell 4 is included in the list of neighbouring cells to be monitored for inter-frequency measurements. Once again, SS verifies that the UE does not transmit MEASUREMENT REPORT messages in the uplink direction.

SS sends PHYSICAL CHANNEL RECONFIGURATION message to allocate dedicated physical channels to the UE. SimultaneouslyIn this message, SS uses this message to commands the UE to start applying compressed mode mechanism for DPCH. The UE shall reply with PHYSICAL CHANNEL RECONFIGURATION COMPLETE message and then move to CELL_DCH state. SS waits for 10 seconds. The UE shall transmit MEASUREMENT REPORT message, containing the selected frequency quality estimate (in this case CPICH Ec/No) of cell 5. SS verifies that this message does not contain measured results for cell 4. After sending this message, the UE shall not transmit any more MEASUREMENT REPORT messages.

SS modifies the reporting criteria by transmitting a MEASUREMENT CONTROL message on the downlink DCCH using AM-RLC. In this message, SS commands the UE to perform inter-frequency measurement and reporting for cell 5 using periodic reporting mechanism. Upon receiving this message, the UE shall transmit MEASUREMENT REPORT message at 2 seconds interval. In the next sequence, SS transmits a PHYSICAL CHANNEL RECONFIGURATION message and deactivates the compressed mode pattern sequence with "TGPSI" IE set to 1. The UE shall respond by sending PHYSICAL CHANNEL RECONFIGURATION COMPLETE message and also stop the periodic reporting activities. Following this, SS sends a MEASUREMENT CONTROL message and re-activates the compressed mode pattern sequence by using the "DPCH compressed mode status" IE. SS confirms that the UE has reconfigured itself to start measurement reporting again. The SS shall be able to receive MEASUREMENT REPORT messages continuously at 2 seconds interval.

Expected Sequence

Step	Direction UE SS	Message	Comment
1			The initial state of UE is in CELL_DCH state of cell 1, after executing procedure P11 or P13, depending on the supported CN domain. Refer to clause 7.4 of TS 34.108.
2	←	MEASUREMENT CONTROL	SS specifies inter-frequency measurement and reporting parameters for cell 5, with "measurement validity" IE present and "UE state" set to "CELL DCH".
3			SS checks that no MEASUREMENT REPORT messages are detected on the uplink DCCH.
4			SS modifies SYSTEM INFORMATION BLOCK TYPE 12 to include cell 4 into the neighbour cell list for inter- frequency measurements.
<u>4</u> 5	+	PHYSICAL CHANNEL RECONFIGURATION	SS allocates PRACH and S-CCPCH physical resources.
<u>5</u> 6	\rightarrow	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	UE shall moves to CELL_FACH state.
<u>6</u>	<u> </u>	System Information Block type 12	SS modifies SIB 12 in order to include cell 4 into the neighbour cell list for interfrequency measurements.
7			SS confirms that there are no transmissions of MEASUREMENT REPORT message in the uplink direction.
8	←	PHYSICAL CHANNEL RECONFIGURATION	SS allocates DPCH physical channels and specifies compressed mode parameters
9	\rightarrow	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	UE moves to CELL_DCH state.
10	→	MEASUREMENT REPORT	UE shall resume inter- frequency measurement task for cell 5 and transmit this message to report the measured CPICH Ec/No value
11	←	MEASUREMENT CONTROL	SS changes the reporting criteria for cell 5 to 'periodic reporting'
12	→	MEASUREMENT REPORT	UE shall begin to transmit this message at 2 seconds interval.
13	←	PHYSICAL CHANNEL RECONFIGURATION	SS deactivates the currently used pattern sequence for compressed mode operation.
14	→	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	UE stays in CELL_DCH state. SS waits for 5 seconds and verifies that no MEASUREMENT REPORT messages are received.
15	←	MEASUREMENT CONTROL	SS activates the pattern sequence stored by the UE.
16	÷	MEASUREMENT REPORT	SS checks that MEASURE- MENT REPORT messages are received at 2 seconds interval.

SYSTEM INFORMATION BLOCK TYPE 12 (Step 1)

Information Element	Value/Remark
References to other system information blocks	Not Present
FACH measurement occasion info	
— -k UTRA	2
- Other RAT present in inter-system cell info	Not Present
Measurement control system information	
- Intra-frequency measurement system	Not Present
information	
- Inter-frequency measurement system	
information	
- Inter-frequency measurement identity number	1 5
- Inter-frequency cell info list	
- Removed inter-frequency info list	Not Present
- New inter-frequency info list	THOU TOOGHT
- Inter-frequency cell id	Set to id of cell 4
	Got to la or con 1
- Cell individual offset	0 dB
- Reference time difference to cell	0 chips
	o ompo
- Primary Scrambling Code	Set to same code as used for cell 4
- Primary CPICH TX power	Not Present
- Read SFN Indicator	FALSE
- TX Diversity Indicator	FALSE
- Cell selection and Re-selection info for	I ALGE
SIB11/12	
	OdB
	OdBm
- HCS neighbouring cell information	Not Present
——————————————————————————————————————	-20dB, -90dBm
- Inter-frequency measurement quantity	-2005, -3005111
	Inter-frequency reporting criteria
- Filter Coefficient	0
- Measurement quantity for frequency quality	CPICH Ec/No
estimate	OFIGH EU/NO
- Inter-frequency measurement reporting criteria	
- Parameters required for each event	
- Parameters required for each event - Inter-frequency event identity	2c
- Threshold used frequency	Not Present
	Not Present
	Not Present 1.0dB
——————————————————————————————————————	1.005 10 sec
	10 sec 4
	+ 0
——- Reporting Interval ——- Parameters required for each non used	∀
•	
frequency Throshold non-used frequency	- 80dBm
- Threshold non-used frequency	1
	0.0
Inter-system measurement system information	Not Present
Traffic volume measurement system	INOL FICSCIII
information	Not Propert
- UE internal measurement system information	Not Present

Information Florent	Volum /D a monte	
Information Element	Value/Remark	
Measurement Identity-Number Measurement Command	• •	
	Setup Not Present	
Measurement Reporting Mode Additional measurements list	Not Present	
CHOICE measurement type	Inter-frequency measurement	
- Inter-frequency cell info list	Net Dresenthie inter frequency cells removed	
- <u>CHOICE Removed inter-frequency info listcell</u>	Not PresentNo inter-frequency cells removed	
removal - New inter-frequency info list	Set to id of cell 5	
	Set to id of cell 5	
 Inter-frequency cell id Frequency info 	UARFCN of the uplink frequency for cell 5	
- Prequency into - UARFCN uplink (Nu)	UARFCN of the downlink frequency for cell 5	
- UARFCN downlink (Nd)	OART CIT OF the downlink frequency for cell 5	
- Cell info	0 dB	
- Cell individual offset	0 chips	
- Reference time difference to cell	o criips	
- Primary CPICH Info	Set to same code as used for cell 5	
- Primary Scrambling Code	Not Present	
- Primary CPICH TX power	FALSE	
- Read SFN Indicator	FALSE	
- TX Diversity Indicator	I ALSL	
- Inter-frequency measurement quantity	Inter-frequency reporting criteria	
- CHOICE reporting criteria	0	
- Filter Coefficient	CPICH Ec/No	
Measurement quantity for frequency quality	011011120/110	
estimate		
- Inter-frequency reporting quantity	FALSE	
- UTRA Carrier RSSI	TRUE	
- Frequency quality estimate	11102	
Non frequency related cell reporting quantities		
- SFN-SFN observed time difference reporting	No report	
indicator	The report	
- Cell synchronisation information reporting	FALSE	
indicator		
- Cell Identity reporting indicator	TRUE	
- CPICH Ec/No reporting indicator	FALSE	
- CPICH RSCP reporting indicator	TRUE	
- Pathloss reporting indicator	FALSE	
- CFN-SFN observed time difference	FALSE	
- Reporting cell status	Not present	
- Measurement validity	·	
- Resume/Release	Resume	
- UE State	CELL_DCH	
- Inter-frequency set update	Not Present	
- CHOICE report criteria	Inter-frequency measurement reporting criteria	
 Parameters required for each event 	2c	
 Inter-frequency event identity 	Not Present	
- Threshold used frequency	Not Present	
- W used frequency	1.0 dB	
- Hysteresis	10 seconds	
- Time to trigger	4	
- Amount of reporting	0	
	Not Present	
- Reporting cell status		
- Parameters required for each non-used	050 15	
frequency	-8 <u>5</u> 0 dBm	
- Threshold non used frequency	0.0	
- W non-used frequency	Not Present	
DPCH compressed mode status info		

PHYSICAL CHANNEL RECONFIGURATION (Step 45)

Use the same message sub-type found in Annex A titled "(Packet to CELL_FACH from CELL_DCH in PS)".

System Information Block type 12 (Step 6)

Information Element	<u>Value/Remark</u>
FACH measurement occasion info	
- FACH Measurement occasion cycle length	<u>2</u>
coefficient	
 Inter-frequency FDD measurement indicator 	TRUE
 Inter-frequency TDD measurement indicator 	<u>FALSE</u>
 Inter-RAT measurement indicators 	Not Present
Measurement control system information	
 Intra-frequency measurement system 	Not Present
<u>information</u>	
 Inter-frequency measurement system 	
<u>information</u>	
 Inter-frequency cell info list 	
 CHOICE inter-frequency cells removal 	No inter-frequency cells removed
 New inter-frequency info list 	
 Inter-frequency cell id 	Set to id of cell 4
- Cell info	
 Cell individual offset 	<u>0 dB</u>
- Reference time difference to cell	0 chips
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as used for cell 4
 Primary CPICH TX power 	Not Present
- Read SFN Indicator	<u>FALSE</u>
- TX Diversity Indicator	<u>FALSE</u>
 Cell selection and Re-selection info 	Not Present – use default values
- Inter-RAT measurement system information	Not Present
 Traffic volume measurement system 	Not Present
<u>information</u>	
 UE internal measurement system information 	Not Present

PHYSICAL CHANNEL RECONFIGURATION (Step 8)

Use the same message sub-type found in Annex A, which is entitled "(Packet to CELL_DCH from CELL_FACH in PS)", with the following exceptions in the IE(s) concerned:

Information Element	Value/Remarks
Downlink information common for all radio links	
 Downlink DPCH info common for all RL 	
- CHOICE Mode	FDD
 Downlink DPCH power control information 	
- DPC mode	0 (Single)
 DL rate matching restriction information 	Not Present
- Spreading factor	Refer to the parameter set in TS 34.108
- Fixed or flexible position	Flexible
- TFCI existence	FALSE
 Number of bits for Pilot bits (SF=128, 256) 	Not Present
 DPCH compressed mode info 	
- TGPSI	1
- TGPS Status Flag	Active
- TGCFN	<u>255</u>
 Transmission gap pattern sequence 	
configuration parameters	
- TGMP	FDD Measurement
- TGPRC	62
- TGCFN	(Current CFN + (256 – TTI/10msec)) mod 256
- TGSN	8
- TGL1	10
- TGL2	5
- TGD	15
- TGPL1	35
- TGPL2	35
- RPP	Mode 1
- ITP	Mode 1
- UL/DL Mode	DL
 Downlink compressed mode method 	SF/2
 Uplink compressed mode method 	Not Present
 Downlink frame type 	A
- DeltaSIR1	2.0
- DeltaSIRAfter1	1.0
- DeltaSIR2	Not Present
- DeltaSIRAfter2	Not Present
- TX Diversity Mode	None
- SSDT information	Not Present
- S field	
- Code Word Set	
- Default DPCH Offset Value	0

MEASUREMENT REPORT (Step 10)

Information Element	Value/Remarks
Measurement identity-number	Check to see if set to 14
Measured Results	
- CHOICE measurement	Check to see if set to "Inter-frequency
	measured results list"
 Inter-frequency measurement results 	
- Frequency info	
- UARFCN (uplink)	Check to see if set to the UARFCN of the
,	uplink frequency for cell 5
- UARFCN (downlink)	Check to see if set to the UARFCN of the
,	downlink frequency for cell 5
- UTRA carrier RSSI	Check to see if it is absent
 Inter-frequency cell measurement results 	
- Cell measured results	
- Cell Identity	Check to see if it is absent
- SFN-SFN observed time difference	Check to see if it is absent
- CFN-SFN observed time difference	Check to see if it is absent
 Cell synchronisation information 	Check to see if it is absent
- Primary CPICH Info	
- Primary Scrambling Code	Check to see if set to the same code for cell 5
- CPICH Ec/No	Check to see if it is present
- CPICH RSCP	Check to see if it is absent
- Pathloss	Check to see if it is absent
Measured Results on RACH	Check to see if it is absent
Event Results	
- CHOICE event result	Inter-frequency event results
 Inter-frequency event identity 	Check to see if it's set to '2c'
 Inter-frequency cells 	
- Frequency Info	
- UARFCN (uplink)	Check to see if set to the UARFCN of the
	uplink frequency for cell 5
- UARFCN (downlink)	Check to see if set to the UARFCN of the
	downlink frequency for cell 5
 Non frequency related measurement event 	
results	
- Primary CPICH Info	
- Primary Scrambling Code	Check to see if set to the same code for cell 5

Information Element	Value/Remark	
Measurement Identity-Number	14	
Measurement Command	Modify	
Measurement Reporting Mode	Not Present	
Additional measurements list	Not Present	
CHOICE measurement type	Inter-frequency measurement	
- Inter-frequency cell info list	men modeling modelinem	
- CHOICE Removed-inter-frequency info listcell	Not PresentNo inter-frequency cells removed	
removal		
- New inter-frequency info list	Set to id of cell 5	
- Inter-frequency cell id		
- Frequency info	UARFCN of the uplink frequency for cell 5	
- UARFCN uplink (Nu)	UARFCN of the downlink frequency for cell 5	
- UARFCN downlink (Nd)		
- Cell info	0 dB	
- Cell individual offset	0 chips	
- Reference time difference to cell	o ompo	
- Primary CPICH Info	Set to same code as used for cell 5	
- Primary Scrambling Code	Not Present	
- Primary CPICH TX power	FALSE	
- Read SFN Indicator	FALSE	
- TX Diversity Indicator	I ALOL	
- Inter-frequency measurement quantity	Inter-frequency reporting criteria	
- CHOICE reporting criteria	1	
- Filter Coefficient	CPICH Ec/No	
- Measurement quantity for frequency quality	OF IOTI EC/NO	
estimate		
- Inter-frequency reporting quantity	FALSE	
- UTRA Carrier RSSI	TRUE	
- Frequency quality estimate	TROL	
Non frequency related cell reporting quantities		
- SFN-SFN observed time difference reporting	No report	
indicator	ino report	
- Cell synchronisation information reporting	FALSE	
indicator	FALSE	
- Cell Identity reporting indicator	TRUE	
- CPICH Ec/No reporting indicator	FALSE	
- CPICH EC/No reporting indicator - CPICH RSCP reporting indicator	TRUE	
- Pathloss reporting indicator	FALSE	
- Fatriloss reporting indicator - CFN-SFN observed time difference	FALSE	
	I -	
- Reporting cell status	Not present	
- CHOICE report <u>ed</u> ing cell	Papart calls Wwithin active and/or manitored	
	Report cells \(\frac{\pi}{\pi}\) within active and/or monitored	
Maximum number of repetteding cells to = 0	cells set on used frequency and or within	
- Maximum number of report <u>ed</u> ing cells type2	active and/or monitored cells set on non-used	
- Measurement validity	frequency	
- Inter-frequency set update	2 Not Present	
- CHOICE report criteria	Not Present	
- Amount of reporting	Not Present	
- Reporting interval	Periodic reporting criteria	
DPCH compressed mode status info	Infinity	
	2000 milliseconds	
	Not Present	

MEASUREMENT REPORT (Step 12, 16)

Information Element	Value/Remarks
Measurement identity-number	Check to see if set to 14
Measured Results	
- CHOICE measurement	Check to see if set to "Inter-frequency measured results list"
 Inter-frequency measurement results 	
- Frequency info	
- UARFCN (uplink)	Check to see if set to the UARFCN of the uplink frequency for cell 5
- UARFCN (downlink)	Check to see if set to the UARFCN of the downlink frequency for cell 5
- UTRA carrier RSSI	Check to see if it is absent
 Inter-frequency cell measurement results 	
- Cell measured results	
- Cell Identity	Check to see if set to the id of cell 5
 SFN-SFN observed time difference 	Check to see if it is absent
 Cell synchronisation information 	Check to see if it is absent
- Primary CPICH Info	
- Primary Scrambling Code	Check to see if set to the same code for cell 5
- CPICH Ec/No	Check to see if it is present
- CPICH RSCP	Check to see if it is absent
- Pathloss	Check to see if it is absent
 CFN-SFN observed time difference 	Check to see if it is absent
Measured Results on RACH	Check to see if it is absent
Event Results	Check to see if it is absent

PHYSICAL CHANNEL RECONFIGURATION (Step 13)

Use the same message transmitted in step 8 with the following modifications:

Information Element	Value/Remarks
Downlink information common for all radio links	
 DPCH compressed mode info 	
- TGPSI	1
- TGPS Status Flag	Inactive
- TGCFN	<u>255</u>
 Transmission gap pattern sequence 	Not Present
configuration parameters	

MEASUREMENT CONTROL (Step 15)

Information Element	Value/Remark
Measurement Identity-Number	Any number except 14
Measurement Command	Release Modify
Measurement Reporting Mode	Not Present
Additional measurements list	Not Present
CHOICE measurement type	Not Present
DPCH compressed mode status info	
 Transmission gap pattern sequence 	
- TGPSI	1
- TGPS Flag	Active
- TGCFN	255

8.4.1.8.5 Test Requirement

After step 2 the UE shall not send any MEASUREMENT REPORT messages on the uplink DCCH of cell 1.

After step 9 the UE shall transmit a MEASUREMENT REPORT message, containing the measured results for cell 5's CPICH Ec/No value. The UE shall not transmit any messages pertaining to cell 4's measurements.

After step 11 the UE shall send MEASUREMENT REPORT messages, which comprises cell 5's CPICH Ec/No measured value at 2 seconds interval. The "Event results" IE shall be omitted in these messages.

After step 14 the UE shall not transmit any MEASUREMENT REPORT messages.

After step 15 the UE shall resume the transmission of MEASUREMENT REPORT messages with identical contents as in those received after step $\underline{69}$.

8.4.1.9 Measurement Control and Report: Unsupported measurement in the UE

8.4.1.9.1 Definition

8.4.1.9.2 Conformance requirement

If the UTRAN indicates the UE to perform a measurement that is not supported in the UE, the UE shall keep the measurement configuration. Then the UE shall transmit a MEASUREMENT CONTROL FAILURE message on the DCCH using AM RLC_within 8 frames (excluding the effect of TTI mis-alignment).

Reference

3GPP TS 25.331 clause 8.4.1

8.4.1.9.3 Test purpose

To confirm that the UE transmits a MEASUREMENT CONTROL FAILURE message, with the value "unsupported measurement" specified in IE "failure cause" when the SS commanded the UE to perform an unsupported measurement by sending a MEASUREMENT CONTROL message.

8.4.1.9.4 Method of test

Initial Condition

System Simulator: 1cell

UE: CS-DCCH_DCH (State 6-5) or PS-DCCH_DCH (State 6-7) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE.

[Editor's note: It is assumed in this test that the UE under test does not possess any inter-system-RAT measurement capability. The mandatory type(s) of measurement capability that should be implemented by the UE is to be discussed]

Test Procedure

The UE is in the CELL_DCH state. The SS transmits a MEASUREMEN CONTROL message which includes parameters (e.g. Measurement identity number: 2, measurement command: Setup, measurement type: inter-system_RAT measurements, measurement reporting mode: unacknowledged and periodical reporting, measurement object: inter-system_RAT_cell information, measurement quantity: Signal strength, reporting quantity: RSSI on BCCH carrier). As the UE under test does not support inter-system_RAT_measurement, it shall transmit a MEASUREMENT CONTROL FAILURE message on the uplink DCCH using AM RLC.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1				The UE is in the CELL_DCH state.
2		←	MEASUREMENT CONTROL	Including the parameters (e.g. Measurement identity number: 2, measurement command: Setup, measurement type: inter-system-RAT measurements, measurement reporting mode: unacknowledged and periodical reporting, measurement object: inter-system-RAT cell information, measurement quantity: Signal strength, reporting-quantity: RSSI on BCCH carrier).
3		→	MEASUREMENT CONTROL FAILURE	Which is set to "unsupported measurement" in IE "failure cause". SS calculates the time interval between the transmissions of MEASUREMENT CONTROL message on the downlink DCCH to the uplink reception of MEASUREMENT CONTROL FAILURE on the uplink DCCH. SS verifies that the calculated time interval is within (8 frames + TTI)

MEASUREMENT CONTROL (Step 2)

Information Element	Value/Remark	
RRC transaction identifier	Select an arbitrary an integer between 0 and 3	
Measurement Identity-Number	2	
Measurement Command	Setup	
Measurement Reporting Mode	Unacknowledged mode	
Additional measurements list	Not Present	
CHOICE measurement type	Inter-system-RAT measurement	
- Inter-system-RAT cell info list	micr dystem <u>raxa medearement</u>	
- <u>CHOICE Removed inter-system RAT</u> cells	Not PresentRemove no inter-RAT cells	
removal	1	
- New inter- system -RAT cells	1	
- Inter- system-IXAT cells - Inter- system- RAT cell id	GSM	
- CHOICE Radio Access Technology	Not Present	
	Not Present	
- Cell selection and re-selection info		
- Qoffset _{s,n}	-20dB, -90dBm	
	Set to the maximum allowed by UE RF power	
- Qqualmin, Qrxlevmin	class	
- Maximum allowed UL TX power	Set to the BSIC code of cell 2	
	Set to the ARFCN assigned to cell 2	
- BSIC	Not Present	
- BSIC ARFCN		
- Output power	Inter-frequency reporting criteria GSM	
 Inter-system RAT measurement quantity 	GSM Carrier RSSI	
- CHOICE system reporting criteria	0	
 Measurement quantity 	CPICH Ec/NoNot required	
- Filter Coefficient		
- Measurement quantity for frequency quality		
estimateBSIC verification required	<u>FALSE</u>	
- Inter-system-RAT reporting quantity	GSM	
- UTRAN estimate quantity	FALSE	
- CHOICE system	FALSE	
- Pathloss	TRUE	
- Observed time difference to GSM cell	Not Present	
- GSM Carrier RSSI	No reporting	
- Reporting cell status	TRUE	
- CHOICE report criteria	FALSE	
- UTRA Carrier RSSI		
- Frequency quality estimate	FALSE	
Non frequency related cell reporting quantities	FALSE	
- SFN-SFN observed time difference	FALSE	
- CPICH Ec/NO		
	FALSE	
- oriorricor	FALSE	
	Not Present	
- CFN-SFN observed time difference	Not i rescrit	
	Within manitored calls on used frequency and	
- Reporting Cell Status	Within monitored cells on used frequency and	
	within monitored cells on non-used frequency	
Martin and Co. Co. Co. Co.	2 Desiredical consenting positionic	
- Maximum number of reporting cells type2	Periodical reporting criteria	
CHOICE Report Criteria	Infinity	
- Amount of reporting	32 seconds	
- Reporting Interval	Not Present	
DPCH compressed mode status info		

MEASUREMENT CONTROL FAILURE (Step 3)

Information Element	Value/Remarks
RRC transaction identifier	Check if it is set to the same value of the same
	IE in the MEASUREMENT CONTROL
	message sent in Step 2.
Failure cause	To cCheck if it is set to "Unsupported
	measurement"

8.4.1.9.<u>5</u>4 Test requirement

After step 2 the UE shall identify the unsupported measurement element in the MEASUREMENT CONTROL message and transmit a MEASUREMENT CONTROL FAILURE. In this message, the value "unsupported measurement" shall be specified in IE "failure cause". SS shall be able to receive the MEASUREMENT CONTROL FAILURE message on the uplink DCCH, within a time interval of (8 frames + Transmission Time Interval) from the sending of the last transport block that contains the downlink MEASUREMENT CONTROL message.

8.4.1.10 Measurement Control and Report: Failure (Invalid Message Reception)

8.4.1.10.1 Definition

8.4.1.10.2 Conformance requirement

When the UE received a MEASUREMENT CONTROL message containing an unexpected conditional IE, it shall reply with a MEASUREMENT CONTROL FAILURE message stating the appropriate protocol error information. The UE shall transmit the MEASUREMENT CONTROL FAILURE message at the latest of 8 frames (excluding the effect of TTI mis-alignment), after SS has sent the last transport block containing the downlink MEASUREMENT CONTROL message. It shall maintain the monitoring and measurement reporting mechanism as in before the MEASUREMENT CONTROL message has been received.

Reference

3GPP TS 25.331 clause 8.4.1.9

8.4.1.10.3 Test Purpose

To confirm that the UE does not change its current monitoring and measurement settings after it has received an illegal MEASUREMENT CONTROL message, which contains an unexpected IE error. To confirm that the UE resume its normal measurement reporting operations after transmitting MEASUREMENT CONTROL FAILURE message to the SS.

8.4.1.10.4 Method of test

Initial Condition

System Simulator: 1 cell.

UE: CS-DCCH_DCH (State 6-5) or PS-DCCH_DCH (State 6-7) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE.

Test Procedure

The UE is initially brought to CELL_DCH. SS transmits a MEASUREMENT CONTROL message to the UE, commanding it to start transmitting report messages for the reporting quantity "UE Transmitted Power". SS then waits for the MEASUREMENT RERORT message with the allocated measurement identity to arrive. Then it transmits the MEASUREMENT CONTROL message again. In this message, SS requests that the reporting activities for "UE Transmitted Power" be stopped. At the end of this message, SS appends an unknown information element. When the

UE receives this message, it shall reply with MEASURMENT CONTROL FAILURE message as it has detected a protocol error. It shall not cease to report its own transmission power level using MEASUREMENT REPORT messages.

Expected Sequence

Step	Direction		Message	Comment
	UE	SS	-	
1				The UE is CELL_DCH state in cell 1.
2	←		MEASUREMENT CONTROL	SS transmits this message on downlink DCCH to instruct UE to start reporting the quantity "UE transmit power".
3	-)	MEASUREMENT REPORT	UE shall send this message periodically at 32 seconds interval
4		(MEASURMENT CONTROL	SS sends a MEASUREMENT CONTROL message to request that UE stop the reporting activity.
5	-)	MEASUREMENT CONTROL FAILURE	UE shall maintain its current measurement context and send this message. SS calculates the time interval between the transmissions of MEASUREMENT CONTROL message in step 4 to the uplink reception of MEASUREMENT CONTROL FAILURE message. SS verifies that the calculated time interval is within (8 frames + TTI)
6		→	MEASUREMENT REPORT	32 seconds after step 3, UE shall continue to transmit this message to the SS.

Specific Message Content

MEASUREMENT CONTROL (Step 2)

Information Element	Value/Remark
Measurement Identity-Number	3
Measurement Command	Setup
Measurement Reporting Mode	Not Present
Additional measurements list	Not Present
CHOICE measurement type	UE internal measurement
- UE internal measurement quantity	
- Measurement quantity	UE Transmitted Power
- Filter coefficient	0
 UE internal reporting quantity 	
- UE Transmitted Power	TRUE
 UE Rx-Tx time difference 	FALSE
CHOICE report criteria	Periodical reporting criteria
- Amount of reporting	Infinity
- Reporting interval	32 seconds
DPCH compressed mode status info	Not Present

MEASUREMENT REPORT (Step 3)

Information Element	Value/Remarks
Measurement identity-number	Check to see if set to 3
Measured Results	
CHOICE measurement	Check to see if set to "UE internal measurement"
- UE Transmitted Power	Check to see if the reported power is compatible with RF class
- UE Rx-Tx report entries	Check to see if this IE is absent
Measured Results on RACH	Check to see if this IE is absent
Event Results	Check to see if this IE is absent

MEASUREMENT CONTROL (Step 4)

Information Element	Value/Remark
RRC transaction identifier	Selects an arbitrary integer between 0 and 3
Measurement Identity-Number	3
Measurement Command	Stop
Measurement Reporting Mode	Not Present
Additional measurements list	Not Present
CHOICE measurement type	Contains an arbitrary value
DPCH compressed mode status info	Not Present

MEASUREMENT CONTROL FAILURE (Step 5)

Information Element	Value/Remark
RRC transaction identifier	Check if it is set to the same value of the same
	IE in the MEASUREMENT CONTROL
	message sent in Step 4.
Failure cause	Check to see if set to "protocol error"
Protocol error information	Check to see if set to "Message extension not
	comprehended"

MEASUREMENT REPORT (Step 6)

Same as in the requirement for step 3

8.4.1.10.5 Test Requirement

After step 4 the UE shall transmit MEASUREMENT <u>CONTROL</u> FAILURE message, stating the IE "failure cause" as "protocol error" and IE "protocol error information" as "message extension not comprehended". <u>SS shall be able to receive the MEASUREMENT CONTROL FAILURE message on the uplink DCCH, within a time interval of (8 frames + Transmission Time Interval) from the sending of the last transport block that contains the downlink <u>MEASUREMENT CONTROL message in step 4.</u> <u>It-The UE</u> shall continue to send MEASUREMENT REPORT with the correct identity number and measurement result entries at approximately 32 seconds interval.</u>

8.4.1.11 Measurement Control and Report: Compressed Mode Configuration Failure during radio bearer reconfiguration procedure

8.4.1.11.1	Definition

8.4.1.11.2 Conformance requirement

During a radio bearer reconfiguration procedure, the UTRAN might request the activation of a new transmission gap pattern sequence configuration. If the UE detects a runtime error due to overlapping compressed mode configuration (when transmission gap pattern sequences create transmission gaps in the same frame), it shall delete the transmission

gap pattern sequence configuration associated with highest value of TGPSI. The UE shall also terminate any inter-frequency measurements corresponding to the deleted transmission gap pattern sequence. Finally, the UE shall transmit a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM-RLC, with the cause value in IE "failure cause" set to "compressed mode runtime error".

Reference

3GPP TS 25.331 clause 8.2.2, clause 8.2.11, clause 8.6.6.1<u>5</u>4

8.4.1.11.3 Test purpose

To confirm that the UE transmits a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the uplink DCCH using AM RLC, if the it receives a RADIO BEARER RECONFIGURATION message which includes IE "DPCH compressed mode info" that causes an illegal overlap involving more than one parallel transmission gap pattern sequences. To confirm that the UE terminate any inter-frequency measurements corresponding to the deleted transmission gap pattern sequence

8.4.1.11.4 Method of test

Initial Condition

System Simulator: 2 cells – both cell 1 and cell 4 are active. <u>See Table 8.4.1.11-1 for the power settings. The downlink power level of cell 4 is 10dB below cell 1.</u>

UE: CS-DCCH+DTCH_DCH (State 6-9) or PS-DCCH+DTCH_DCH (State 6-10) as specified in TS34.108 clause 7.4, depending on the CN domain supported.

Test Procedure

Table 8.4.1.11-1 illustrates the downlink power to be applied for the 2 cells in this test case.

Table 8.4.1.11-1

<u>Parameter</u>	<u>Unit</u>	Cell 1	Cell 4
UTRA RF		<u>Ch. 1</u>	Ch. 2
Channel Number			
\hat{I}_{or}/I_{oc}	<u>dB</u>	<u>5.7</u>	<u>2.3</u>
CPICH Ec/lo	dB	<u>-11.0</u>	<u>-12.0</u>
CPICH RSCP	dBm	<u>-74.3</u>	<u>-77.7</u>

The UE is in the CELL_DCH state in cell 1. SS sends a MEASUREMENT CONTROL message on the downlink DCCH to request the UE to start inter-frequency measurement for cell 24's CPICH Ec/No value. Simultaneously, the stored transmission gap pattern sequence configuration associated with TGPSI=1 is indicated to be activated in this message. Upon the reception of this message, the UE shall transmit MEASUREMENT REPORT messages periodically at 16 seconds interval to report cell 24's measurement results. Next, SS sends a second MEASUREMENT CONTROL message. In this message, a new measurement task is to be established for the measurement and reporting of cell 24's CPICH RSCP value on a periodic basis. A deactivated transmission pattern gap sequence configuration (with TGPSI=2) is associated with this new measurement task.

The SS transmits a RADIO BEARER RECONFIGURATION message and commands the activation of transmission gap pattern sequence with TGPSI=2. This is expected to result in the detection of a runtime error due to overlapping compressed mode configuration. The UE then shall transmit a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM-RLC. In this message, the value of IE "failure cause" shall be set to "compressed mode runtime error". The UE shall terminate all inter-frequency measurement tasks associated with TGPSI=2. However, the UE shall continue to send MEASUREMENT REPORT messages to report cell 24's CPICH Ec/No value, which is measured during the transmission gap created by compressed mode configuration corresponding to TGPSI=1.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1			UE is initially in CELL_DCH state.
2	←	MEASUREMENT CONTROL	Start inter-frequency measurements for cell 24's CPICH Ec/No using transmission gap pattern sequence with TGPSI=1.
3	\rightarrow	MEASUREMENT REPORT	UE reports cell 24's CPICH Ec/No readings periodically.
4	←	MEASUREMENT CONTROL	Assign inter-frequency measurements for cell 24's CPICH RSCP. This measurement task is associated with transmission gap pattern sequence with TGPSI=2, which has not been activated yet.
5	←	RADIO BEARER RECONFIGURATION	SS specifies the parameters for transmission gap pattern sequence with TGPSI=2 and activate it simultaneously
6			UE shall delete transmission gap pattern sequence configuration associated with TGPSI=2.
7	→	PHYSICAL CHANNEL RECONFIGURATION FAILURE	Failure cause shall be set to "Compressed mode runtime error"
8	\rightarrow	MEASUREMENT REPORT	The contents shall be the same as that in step 3.

MEASUREMENT CONTROL (Step 2)

Information Element	Value/Remark
Measurement Identity-Number	1
Measurement Command	Setup
Measurement Reporting Mode	Not Present
Additional measurements list	Not Present
CHOICE measurement type	Inter-frequency measurement
- Inter-frequency cell info list	mer mequency measurement
- Removed CHOICE inter-frequency info listcell	Not PresentNo inter-frequency cells removed
removal	The trace in the street in equation y come to interest
- New inter-frequency info list	Set to id of cell 4
- Inter-frequency cell id	
- Frequency info	UARFCN of the uplink frequency for cell 4
- UARFCN uplink (Nu)	UARFCN of the downink frequency for cell 4
- UARFCN downlink (Nd)	Oracle of the downlink frequency for oon 1
- Cell info	0 dB
- Cell individual offset	0 chips
- Reference time difference to cell	o onipo
- Primary CPICH Info	Set to same code as used for cell 4
- Primary Scrambling Code	Not Present
- Primary CPICH TX power	FALSE
- Read SFN Indicator	FALSE
- TX Diversity Indicator	TALOL
- Inter-frequency measurement quantity	Inter-frequency reporting criteria
- CHOICE reporting criteria	0
- Filter Coefficient	CPICH Ec/No
- Measurement quantity for frequency quality	CFICIT EC/NO
estimate	
- Inter-frequency reporting quantity	FALSE
- UTRA Carrier RSSI	TRUE
- Frequency quality estimate	INOL
Non frequency related cell reporting quantities	
- SFN-SFN observed time difference reporting	No report
indicator	No report
- Cell synchronisation reporting indicator	FALSE
- Cell Identity reporting indicator	FALSE
- CPICH Ec/No_reporting indicator	TRUE
- CPICH RSCP_reporting indicator	FALSE
	FALSE
- Pathloss reporting indicator - CFN-SFN observed time difference	FALSE
- Reporting cell status	TALOE
- Reporting cell status - CHOICE report <u>eding</u> cell	Penort cells Wwithin active and/or manitored
- OF OTOE TEPOTI <u>eumi</u> cell	Report cells \(\psi_w\) ithin active and/or monitored cells-set on used frequency \(\frac{and}{and}\) or within active
Maximum number of reporteding cells time?	and/or monitored cells-set on non-used
Maximum number of reporteding cells type2 Massurement validity	frequency 2
- Measurement validity	
- Inter-frequency set update	Not present
- CHOICE report criteria - Amount of reporting	Not present Periodic reporting criteria
- Reporting interval	Infinity 16 seconds
DPCH compressed mode status info	To Seconds
- TGPSI	1
- TGPS Status Flag	1 Active
- TGCFN	Active
	<u>255</u>

MEASUREMENT REPORT (Step 3 and 8)

Information Element	Value/Remarks
Integrity check info	If integrity protection is activated, this IE shall be present and SS checks that the MAC-I value matches with the calculated X-MAC value. Else, this IE shall be absent.

Measurement identity-number Measured Results

- CHOICE measurement
- Inter-frequency measurement results
- Frequency info
- UARFCN (uplink)
- UARFCN (downlink)
- UTRA carrier RSSI
- Inter-frequency cell measurement results
- Cell measured results
- Cell Identity
- SFN-SFN observed time difference
- CFN-SFN observed time difference
- Cell synchronisation information
- Primary CPICH Info
- Primary Scrambling Code
- CPICH Ec/No
- CPICH RSCP
- Pathloss

Measured Results on RACH

Event Results

Check to see if set to "1"

Check to see if set to "Inter-frequency measured results list"

Check to see if set to the UARFCN of the uplink frequency for cell 4
Check to see if set to the UARFCN of the downlink frequency for cell 4
Check to see if it is absent

Check to see if it is absent Check to see if it is absent Check to see if it is absent Check to see if it is absent

Check to see if set to the same code for cell 4 Check to see if it is present

Check to see if it is absent Check to see if it is absent Check to see if it is absent Check to see if it is absent

Information Element	Value/Remark
Measurement Identity-Number	2
Measurement Command	Setup
Measurement Reporting Mode	Not Present
Additional measurements list	Not Present
CHOICE measurement type	Inter-frequency measurement
- Inter-frequency cell info list	inter-frequency measurement
- Removed-CHOICE inter-frequency cell	Not PresentNo inter-frequency cells removed
removal info list	ivot Fresentino inter-frequency cells femoved
- New inter-frequency info list	Set to id of cell 4
• •	Set to id of cell 4
 Inter-frequency cell id Frequency info 	LIABECN of the uplink frequency for cell 4
- UARFCN uplink (Nu)	UARFCN of the uplink frequency for cell 4 UARFCN of the downink frequency for cell 4
- UARFCN downlink (Nd)	OARPON of the downlink frequency for cell 4
- Cell info	0 dB
- Cell individual offset	
- Reference time difference to cell	0 chips
	Set to same code as used for cell 4
 Primary CPICH Info Primary Scrambling Code 	Not Present
- Primary Scrambling Code - Primary CPICH TX power	FALSE
- Read SFN Indicator	FALSE
	FALSE
- TX Diversity Indicator	Inter frequency reporting criteria
- Inter-frequency measurement quantity	Inter-frequency reporting criteria
- CHOICE reporting criteria	0
- Filter Coefficient	CPICH RSCP
- Measurement quantity for frequency quality	
estimate	FALCE
- Inter-frequency reporting quantity	FALSE
- UTRA Carrier RSSI	TRUE
- Frequency quality estimate	
- Non frequency related cell reporting quantities	No veneut
- SFN-SFN observed time difference reporting	No report
indicator	FALCE
- Cell synchronisation information reporting	<u>FALSE</u>
indicator	FALCE
- Cell Identity reporting indicator	FALSE
- CPICH Ec/No reporting indicator	TRUEFALSE
- CPICH RSCP reporting indicator	FALSETRUE
- Pathloss reporting indicator	FALSE
	FALSE
- Reporting cell status	Depart calls We sith in a stire and law reconstants
- CHOICE report <u>eding</u> cell	Report cells \(\psi_{\text{w}}\) ithin active and/or monitored
	cells set on used frequency and or within
Maximum number of new arts discussed to the control of the control	active and/or monitored cells-set on non-used
- Maximum number of report <u>eding</u> cells type2	frequency
- Measurement validity	2
- Inter-frequency set update	Not present
- CHOICE report criteria	Not present
- Amount of reporting	Periodic reporting criteria
- Reporting interval	Infinity
DPCH compressed mode status info	16 seconds
- TGPSI	
- TGPS Status Flag	2
TGCFN	Inactive
	255

RADIO BEARER RECONFIGURATION

The contents of RADIO BEARER RECONFIGURATION message in this test case is identical to the message sub-type title "Packet to CELL_DCH from CELL_DCH in PS" found in Annex A with the following exceptions:

Information Element	Value/remark
- DPCH compressed mode info	
- TGPSI	2
- TGPS Status Flag	Active
- TGCFN	<u>255</u>
- Transmission gap pattern sequence configuration	
parameters	
- TGMP	FDD Measurement
- TGPRC	62
- TGCFN	(Current CFN + (256 – TTI/10msec)) mod 256
- TGSN	8
- TGL1	10
- TGL2	5
- TGD	15
- TGPL1	35
- TGPL2	35
- RPP	Mode 1
- ITP	Mode 1
- UL/DL Mode	DL
 Downlink compressed mode method 	SF/2
 Uplink compressed mode method 	Not Present
- Downlink frame type	A
- DeltaSIR1	2.0
- DeltaSIRafter1	1.0
- DeltaSIR2	Not Present
- DeltaSIRafter2	Not Present

PHYSICAL CHANNEL RECONFIGURATION FAILURE

Information Element	Value/remark
Integrity check info	If integrity protection is activated, this IE shall be present and SS checks that the MAC-I value matches with the calculated X-MAC value. Else, this IE shall be absent.
Failure cause	Checked to see if set to "compressed mode runtime error"
Protocol error informationDeleted TGPSI	Checked to see if it is absent Checked to see if it is set to "2"

8.4.1.11.5 Test requirement

After step 6 the UE shall keep transmission gap pattern sequence configuration associated with TGPSI=1. It shall delete the transmission gap pattern sequence configuration associated with TGPSI=2, and delete the inter-frequency measurements corresponding to it. It shall transmit PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the uplink DCCH, with the IE "Failure cause" set to "Compressed mode runtime error".

After step 7 the UE shall continue to send MEASUREMENT REPORT messages periodically, to report the CPICH Ec/No readings for cell <u>42</u>. However, no MEASUREMENT REPORT messages containing the <u>CPICH RSCP</u> readings for cell <u>42</u> shall be sent by the UE.

8.4.1.12 Measurement Control and Report: Compressed Mode Configuration Failure during transport channel reconfiguration procedure

8.4.1.12.1 ———Definition

8.4.1.12.2 Conformance requirement

During a transport channel reconfiguration procedure, the UTRAN might request the activation of a new transmission gap pattern sequence configuration. If the UE detects a runtime error due to overlapping compressed mode

configuration (when transmission gap pattern sequences create transmission gaps in the same frame), it shall delete the transmission gap pattern sequence configuration associated with highest value of TGPSI. The UE shall also terminate any inter-frequency measurements corresponding to the deleted transmission gap pattern sequence. Finally, the UE shall transmit a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM-RLC, with the cause value in IE "failure cause" set to "compressed mode runtime error".

Reference

3GPP TS 25.331 clause 8.2.2, clause 8.2.11, clause 8.6.6.145

8.4.1.12.3 Test purpose

To confirm that the UE transmits a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the uplink DCCH using AM RLC, if the it receives a TRANSPORT CHANNEL RECONFIGURATION message which includes IE "DPCH compressed mode info" that causes an illegal overlap involving more than one parallel transmission gap pattern sequences. To confirm that the UE terminate any inter-frequency measurements corresponding to the deleted transmission gap pattern sequence

8.4.1.12.4 Method of test

Initial Condition

System Simulator: 2 cells – both cell 1 and cell 4 are active. <u>See Table 8.4.1.11-1 in clause 8.4.1.11.4 for the power settings. The downlink power level of cell 4 is 10dB below cell 1.</u>

UE: CS-DCCH+DTCH_DCH (State 6-9) or PS-DCCH+DTCH_DCH (State 6-10) as specified in TS34.108 clause 7.4, depending on the CN domain supported.

Test Procedure

For this test case, the downlink transmission power settings should follow that specified in Table 8.4.1.11-1 in clause 8.4.1.11.4.

The UE is in the CELL_DCH state in cell 1. SS sends a MEASUREMENT CONTROL message on the downlink DCCH to request the UE to start inter-frequency measurement for cell 24's CPICH Ec/No value. Simultaneously, the stored transmission gap pattern sequence configuration associated with TGPSI=1 is indicated to be activated in this message. Upon the reception of this message, the UE shall transmit MEASUREMENT REPORT messages periodically at 16 seconds interval to report cell 24's measurement results. Next, SS sends a second MEASUREMENT CONTROL message. In this message, a new measurement task is to be established for the measurement and reporting of cell 24's CPICH RSCP value on a periodic basis. A deactivated transmission pattern gap sequence configuration (with TGPSI=2) is associated with this new measurement task.

The SS transmits a TRANSPORT CHANNEL RECONFIGURATION message and commands the activation of transmission gap pattern sequence with TGPSI=2. This is expected to result in the detection of a runtime error due to overlapping compressed mode configuration. The UE then shall transmit a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM-RLC. In this message, the value of IE "failure cause" shall be set to "compressed mode runtime error". The UE shall terminate all inter-frequency measurement tasks associated with TGPSI=2. However, the UE shall continue to send MEASUREMENT REPORT messages to report cell $\underline{24}$'s CPICH Ec/No value, which is measured during the transmission gap created by compressed mode configuration corresponding to TGPSI=1.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1			UE is initially in CELL_DCH state.
2	←	MEASUREMENT CONTROL	Start inter-frequency
			measurements for cell 24's CPICH Ec/No using transmission gap pattern sequence with TGPSI=1.
3	→	MEASUREMENT REPORT	UE reports cell 24's CPICH Ec/No readings periodically.
4	+	MEASUREMENT CONTROL	Assign inter-frequency measurements for cell 24's CPICH RSCP. This measurement task is associated with transmission gap pattern sequence with TGPSI=2, which has not been activated yet.
5	+	TRANSPORT CHANNEL RECONFIGURATION	SS specifies the parameters for transmission gap pattern sequence with TGPSI=2 and activate it simultaneously
6			UE shall delete transmission gap pattern sequence configuration associated with TGPSI=2.
7	→	PHYSICAL CHANNEL RECONFIGURATION FAILURE	Failure cause shall be set to "Compressed mode runtime error"
8	\rightarrow	MEASUREMENT REPORT	The contents shall be the same as that in step 3.

MEASUREMENT CONTROL (Step 2)

Measurement Identity-Number Measurement Reporting Mode Additional measurements list CHOICE measurement type - Inter-frequency cell info list - Removed-CHOICE inter-frequency cell removalirefo list - New inter-frequency info list - Inter-frequency cell id - Frequency cell id - Frequency uplink (Nu) - UARFCN uplink (Nu) - UARFCN uplink (Nu) - Cell info - Cell individual offset - Reference time difference to cell - Primary CPICH Info - Primary CPICH Info - Primary CPICH TX power - Read SFN Indicator - TX Diversity Indicator - Inter-frequency measurement quantity - CHOICE reporting criteria - Fitter Coefficient - Measurement quantity for frequency quality estimate - Inter-frequency reporting quantities - SFN-SFN observed time difference reporting indicator - Cell Identity reporting indicator - CPICH RSCP reporting ind		
Measurement Reporting Mode Additional measurements list CHOICE measurement type Inter-frequency cell info list Removed CHOICE inter-frequency cell removalinfo list Inter-frequency cell id Frequency (cell id Frequency (cell info list UARFCN uplink (Nu) UARFCN uplink (Nu) UARFCN uplink (Nu) Cell info Cell individual offset Reference time difference to cell Primary CPICH Info Primary CPICH Info Not Present Not Pre	Information Element	Value/Remark
Measurement Reporting Mode Additional measurements list CHOICE measurement type Inter-frequency cell info list Removal_CHOICE inter-frequency cell removalinfo list New inter-frequency info list Inter-frequency cell id Frequency info UARFCN uplink (Nu) UARFCN uplink (Nu) Cell info Cell individual offset Reference time difference to cell Primary CPICH Info Primary CPICH Info Primary CPICH TX power Read SFN Indicator Inter-frequency measurement quantity CHOICE reporting criteria Filter Coefficient Measurement quantity for frequency quality estimate Inter-frequency reporting quantities SFN-SFN observed time difference reporting indicator CPICH Ec/No reporting indicator CPICH Ec/No reporting indicator CPICH RSCP of the uplink frequency for cell 4 UARFCN of the uplink frequency for cell 4 Not Present		-
Additional measurement slist CHOICE measurement type Inter-frequency cell info list - Remeved_CHOICE_inter-frequency cell removalinfe-list - New inter-frequency or cell id - Frequency info - UARFCN uplink (Nu) - UARFCN downlink (Nd) - Cell info - Cell info - Cell info - Cell info - Primary CPICH Info -		
CHOICE measurement type Inter-frequency cell info list Removal-CHOICE inter-frequency cell removalinfe-list New inter-frequency info list Inter-frequency cell id Frequency cell id Frequency cell id Frequency info UARFCN uplink (Nu) UARFCN downlink (Nd) Cell info Cell individual offset Reference time difference to cell Primary CPICH Info Primary CPICH TX power Read SFN Indicator TX Diversity Indicator Inter-frequency measurement quantity CHOICE reporting criteria Filter Coefficient Measurement quantity of frequency quality estimate Inter-frequency reporting quantities SFN-SFN observed time difference reporting indicator CPICH Ec/No reporting indicator Pathloss reporting indicator CPICH Ec/No reporting indicator CPICH Ec/No reporting indicator Pathloss repured v cells to id of cell 4 UARFCN of the uplink frequency for cell 4 UARFCN of the downink frequency for cell 4 UARFCN of the downink frequency for cell 4 UARFCN of the downink frequency for cell 4 UARFCN of the uplink freq		
- Inter-frequency cell info list - Removed childing list - New inter-frequency info list - Inter-frequency cell id - Frequency info - UARFCN uplink (Nu) - UARFCN uplink (Nu) - Cell info - Cell info - Cell individual offset - Primary CPICH Info - Primary CPICH Info - Primary CPICH Info - Primary CPICH Info - Primary CPICH TX power - Read SFN Indicator - TX Diversity Indicator - Inter-frequency neasurement quantity - CHOICE reporting criteria - Filter Coefficient - Measurement quantity of frequency quality estimate - Inter-frequency reporting quantities - SFN-SFN observed time difference reporting indicator - Cell Jennicy reporting indicator - CPICH Ec/No reporting indicator - CPICH Ec/No reporting indicator - CPICH SCP reporting indicator - CPICH SSP observed time difference - Reporting cell status - CHOICE reporteding cells - Maximum number of reporteding cells-type2 - Measurement validity - Maximum number of reporteding cells-type2 - Measurement validity - Inter-frequency cells removed - Var Present No inter-frequency cells removed - Val Cell 4 - UARFCN of the uplink frequency for cell 4 - UARFCN of the downink frequency for cell 4 - Var Present No inter-frequency for cell 4 - Var Present No inter-frequency for cell 4 - VARFCN of the uplink frequency for cell 4 - VARFCN of the downink frequency for cell 4 - Var Present No inter-frequency for cell 4 - Var Present		
- Removed CHOICE inter-frequency cell removalinfo list - New inter-frequency info list - New inter-frequency cell id - Frequency info - UARFCN uplink (Nu) - UARFCN downlink (Nd) - Cell individual offset - Reference time difference to cell - Primary CPICH Info - Primary CPICH TX power - Read SFN Indicator - Inter-frequency measurement quantity - CHOICE reporting criteria - Filter Coefficient - Measurement quantity for frequency quality estimate - Inter-frequency reporting quantity - UTRA Carrier RSSI - Frequency quality estimate - Non frequency related cell reporting quantities - SFN-SFN observed time difference reporting indicator - Cell Identity reporting indicator - CPICH Ec/No reporting indicator - CPICH Ec/No reporting indicator - CPICH SCP reporting indi		Inter-frequency measurement
removalinfo-liet - New inter-frequency info list - Inter-frequency cell id - Frequency info - UARFCN uplink (Nu) - UARFCN downlink (Nd) - Cell info - Cell info - Cell info - Cell info - Primary CPICH Info - Primary CPICH Tx power - Primary Scrambling Code - Primary CPICH Tx power - Read SFN Indicator - TX Diversity Indicator - Inter-frequency measurement quantity - CHOICE reporting criteria - Filter Coefficient - Measurement quantity for frequency quality estimate - Inter-frequency reporting quantities - SFN-SFN observed time difference reporting indicator - Cell synchronisation information reporting indicator - CPICH Ec/No reporting indicator - CPICH Ec/No reporting indicator - CPICH Ec/No - CPICH RosCP reporting indicator - CPICH RosCP reporting indicator - CFN-SFN observed time difference - Reporting cell status - CHOICE reporteding cells - Maximum number of reporteding cells-type2 - Measurement validity - Maximum number of reporteding cells-type2 - Measurement validity - Set to id of cell 4 - UARFCN of the uplink frequency for cell 4 UARFCN of the uplink frequency for cell 4 UARFCN of the uplink frequency for cell 4 UARFCN of the downink frequency for cell 4 UARFCN of the uplink frequency for cell 4 UARFCN of the uplink frequency for cell 4 UARFCN of the uplink frequency for cell 4 UARFCN of the downink frequency for cell 4 UARFCN of the uplink frequency for cell 4 UARFCN of the uplink frequency for cell 4 UARFCN of the uplink frequency for cell 4 UARFCN of the downink frequency for cell 4 UARFCN of the uplink frequency for cell 4 UARFCN of the uplink frequency for cell 4 UARFCN of the uplink frequency for cell 4 UARFCN of the downink frequenc		
- New inter-frequency info list - Inter-frequency cell id - Frequency info - UARFCN uplink (Nu) - UARFCN downlink (Nd) - Cell info - Cell individual offset - Reference time difference to cell - Primary CPICH Info - Primary Scrambling Code - Primary CPICH TX power - Read SFN Indicator - TX Diversity Indicator - Inter-frequency measurement quantity - CHOICE reporting criteria - Filter Coefficient - Measurement quantity for frequency quality estimate - Inter-frequency reporting quantity - UTRA Carrier RSSI - Frequency quality estimate - Non frequency related cell reporting quantities - SFN-SFN observed time difference reporting indicator - Cell Identity reporting indicator - CPICH Ec/No reporting indicator - CPICH SSCP		Not PresentNo inter-frequency cells removed
- Inter-frequency cell id - Frequency info - UARFCN uplink (Nu) - UARFCN downlink (Nd) - Cell info - Cell individual offset - Reference time difference to cell - Primary CPICH Info - Primary CPICH Info - Primary CPICH ITX power - Read SFN Indicator - TX Diversity Indicator - TX Diversity Indicator - Inter-frequency measurement quantity - CHOICE reporting criteria - Filter Coefficient - Measurement quantity for frequency quality estimate - Inter-frequency reporting quantities - SFN-SFN observed time difference reporting indicator - Cell synchronisation information reporting indicator - CPICH RSCP reporting indicator - CPICH RSCP reporting indicator - CFN-SFN observed time difference - Reporting cell status - CHOICE reporteding cell - Maximum number of reporteding cells-type2 - Measurement validity UARFCN of the uplink frequency for cell 4 UARFCN of the downink frequency for cell 4 UARFCN of the downink frequency for cell 4 UARFCN of the downink frequency for cell 4 UARFCN of the uplink frequency for cell 4 UARFCN of the downink frequency for cell 4 Not Present FALSE FALSE TRUE FALSE		
- Frequency info - UARFCN uplink (Nu) - UARFCN downlink (Nd) - Cell info - Cell info - Cell info - Cell info - Primary CPICH Info - Primary Scrambling Code - Primary CPICH TX power - Read SFN Indicator - TX Diversity Indicator - Inter-frequency measurement quantity - CHOICE reporting criteria - Filter Coefficient - Inter-frequency reporting quantity - UTRA Carrier RSSI - Frequency quality estimate - Non frequency related cell reporting indicator - Cell synchronisation information reporting indicator - Cell Identity reporting indicator - CPICH Ec/No reporting indicator - CPICH E		Set to id of cell 4
- UARFCN downlink (Nd) - UARFCN downlink (Nd) - Cell info - Cell info - Cell individual offset - Reference time difference to cell - Primary CPICH Info - Primary Scrambling Code - Primary CPICH TX power - Read SFN Indicator - TX Diversity Indicator - Inter-frequency measurement quantity - CHOICE reporting criteria - Filter Coefficient - Measurement quantity for frequency quality estimate - Inter-frequency reporting quantity - UTRA Carrier RSSI - Frequency quality estimate - Non frequency related cell reporting quantities - SFN-SFN observed time difference reporting indicator - Cell synchronisation information reporting indicator - CPICH Ec/No reporting indicator - CPICH RSCP reporting indicator - CPICH RSCP reporting indicator - CFN-SFN observed time difference - Reporting cell status - CHOICE reporteding cell - Maximum number of reporteding cells-type2 - Measurement validity UARFCN of the downink frequency for cell 4 No dB 0 chips Set to same code as used for cell 4 Not Present FALSE FALSE FALSE TRUE FALSE		
- UARFCN downlink (Nd) - Cell info - Cell infof - Cell individual offset - Reference time difference to cell - Primary CPICH Info - Primary CPICH Info - Primary CPICH TX power - Read SFN Indicator - TX Diversity Indicator - Inter-frequency measurement quantity - CHOICE reporting criteria - Filter Coefficient - Measurement quantity for frequency quality estimate - Inter-frequency reporting quantity - UTRA Carrier RSSI - Frequency quality estimate - Non frequency related cell reporting quantities - SFN-SFN observed time difference reporting indicator - Cell synchronisation information reporting indicator - CPICH Ec/No reporting indicator - CPICH Ec/No no reporting indicator - CPICH Ec/No no reporting indicator - CPICH Ec/No no reporting indicator - Cell Identity reporting indicator - CPICH Ec/No no reporting indicat		
- Cell individual offset - Cell individual offset - Reference time difference to cell - Primary CPICH Info - Primary Scrambling Code - Primary CPICH TX power - Read SFN Indicator - TX Diversity Indicator - Inter-frequency measurement quantity - CHOICE reporting criteria - Filter Coefficient - Measurement quantity for frequency quality estimate - Inter-frequency reporting quantity - UTRA Carrier RSSI - Frequency quality estimate - Non frequency quality estimate - Non frequency related cell reporting quantities - SFN-SFN observed time difference reporting indicator - Cell Identity reporting indicator - CPICH Ec/No reporting indicator - Pathloss reporting indicator - Pathloss reporting indicator - Pathloss reporting indicator - CFN-SFN observed time difference - Reporting criteria 0 CPICH Ec/No FALSE TRUE FALSE	UARFCN of the downink frequency for cell 4	
- Cell individual offset - Reference time difference to cell - Primary CPICH Info - Primary CPICH TX power - Read SFN Indicator - TX Diversity Indicator - Inter-frequency measurement quantity - CHOICE reporting criteria - Filter Coefficient - Measurement quantity for frequency quality estimate - Inter-frequency reporting quantity - UTRA Carrier RSSI - Frequency quality estimate - Non frequency related cell reporting quantities - SFN-SFN observed time difference reporting indicator - Cell synchronisation information reporting indicator - CPICH Ec/No FALSE TRUE FALSE TRUE FALSE TRUE FALSE TRUE FALSE FALSE TRUE FALSE FAL		
- Reference time difference to cell - Primary CPICH Info - Primary Scrambling Code - Primary CPICH TX power - Read SFN Indicator - TX Diversity Indicator - Inter-frequency measurement quantity - CHOICE reporting criteria - Filter Coefficient - Measurement quantity for frequency quality estimate - Inter-frequency reporting quantity - UTRA Carrier RSSI - Frequency quality estimate - Non frequency related cell reporting quantities - SFN-SFN observed time difference reporting indicator - Cell synchronisation information reporting indicator - CPICH Ec/No reporting indicator - CPICH RSCP reporting indicator - CFN-SFN observed time difference - Reporting criteria 0 CPICH Ec/No FALSE TRUE FALSE FAL		
- Primary CPICH Info - Primary Scrambling Code - Primary CPICH TX power - Read SFN Indicator - TX Diversity Indicator - Inter-frequency measurement quantity - CHOICE reporting criteria - Filter Coefficient - Measurement quantity for frequency quality estimate - Inter-frequency reporting quantity - UTRA Carrier RSSI - Frequency quality estimate - Non frequency related cell reporting quantities - SFN-SFN observed time difference reporting indicator - Cell synchronisation information reporting indicator - Cell Identity reporting indicator - CPICH Ec/No FALSE TRUE FALSE F		0 chips
- Primary Scrambling Code - Primary CPICH TX power - Read SFN Indicator - TX Diversity Indicator - Inter-frequency measurement quantity - CHOICE reporting criteria - Filter Coefficient - Measurement quantity for frequency quality estimate - Inter-frequency reporting quantity - UTRA Carrier RSSI - Frequency quality estimate - Non frequency related cell reporting quantities - SFN-SFN observed time difference reporting indicator - Cell synchronisation information reporting indicator - CPICH Ec/No reporting indicator - CPICH Ec/No reporting indicator - CPICH RSCP reporting indicator - CPICH RSCP reporting indicator - CFN-SFN observed time difference - Reporting criteria 0 CPICH Ec/No FALSE TRUE FALSE		
- Primary CPICH TX power - Read SFN Indicator - TX Diversity Indicator - Inter-frequency measurement quantity - CHOICE reporting criteria - Filter Coefficient - Measurement quantity for frequency quality estimate - Inter-frequency reporting quantity - UTRA Carrier RSSI - Frequency quality estimate - Non frequency related cell reporting quantities - SFN-SFN observed time difference reporting indicator - Cell synchronisation information reporting indicator - CPICH Ec/No reporting indicator - CPICH Ec/No reporting indicator - CPICH RSCP reporting indicator -		
- Read SFN Indicator - TX Diversity Indicator - Inter-frequency measurement quantity - CHOICE reporting criteria - Filter Coefficient - Measurement quantity for frequency quality estimate - Inter-frequency reporting quantity - UTRA Carrier RSSI - Frequency quality estimate - Non frequency related cell reporting quantities - SFN-SFN observed time difference reporting indicator - Cell synchronisation information reporting indicator - Cell Identity reporting indicator - CPICH Ec/No reporting indicator - CPICH RSCP reporting indicator - CPICH RSCP reporting indicator - Pathloss reporting indicator - CFN-SFN observed time difference - Reporting cell status - CHOICE reporteding cell - Maximum number of reporteding cells-type2 - Measurement validity - Inter-frequency reporting of CPICH Ec/No - CPICH Ec/No - CPICH Ec/No - FALSE - TRUE - FALSE - TRUE - FALSE - TRUE - FALSE - TRUE - FALSE -		
- TX Diversity Indicator - Inter-frequency measurement quantity - CHOICE reporting criteria - Filter Coefficient - Measurement quantity for frequency quality estimate - Inter-frequency reporting quantity - UTRA Carrier RSSI - Frequency quality estimate - Non frequency related cell reporting quantities - SFN-SFN observed time difference reporting indicator - Cell synchronisation information reporting indicator - Cell Identity reporting indicator - CPICH Ec/No FALSE TRUE FALSE TRUE FALSE TRUE FALSE TRUE FALSE FALSE FALSE TRUE FALSE TRUE FALSE TRUE FALSE TRUE FALSE FALSE TRUE FALSE FALSE TRUE FALSE TRUE FALSE FALSE TRUE FALSE TRUE FALSE Report cells \(\frac{\pmathrm{W}}{\pmathrm{W}}\) in active and/or monitored cells set on non-used frequency andor within active and/or monitored cells set on non-used frequency Amaximum number of reporteding cells type2 Measurement validity		_
- Inter-frequency measurement quantity - CHOICE reporting criteria - Filter Coefficient - Measurement quantity for frequency quality estimate - Inter-frequency reporting quantity - UTRA Carrier RSSI - Frequency quality estimate - Non frequency related cell reporting quantities - SFN-SFN observed time difference reporting indicator - Cell synchronisation information reporting indicator - Cell Identity reporting indicator - CPICH Ec/No reporting indicator - CPICH RSCP reporting indicator - Pathloss reporting indicator - CFN-SFN observed time difference - Reporting cell status - CHOICE reporteding cell - Maximum number of reporteding cells type2 - Measurement validity Inter-frequency reporting of CPICH Ec/No CPICH Ec/No - CPICH Ec/No - FALSE - Report cells \(\frac{\pmax}{\pmax}\) within active and/or monitored seteolls on used frequency and or within active and/or monitored eells-set on non-used frequency - Measurement validity		FALSE
- CHOICE reporting criteria - Filter Coefficient - Measurement quantity for frequency quality estimate - Inter-frequency reporting quantity - UTRA Carrier RSSI - Frequency quality estimate - Non frequency related cell reporting quantities - SFN-SFN observed time difference reporting indicator - Cell synchronisation information reporting indicator - Cell Identity reporting indicator - CPICH RSCP reporting indicator - CPICH RSCP reporting indicator - Pathloss reporting indicator - CFN-SFN observed time difference - Reporting cell status - CHOICE reporteding cell - Maximum number of reporteding cells type2 - Measurement validity - Measurement validity - Measurement validity - CPICH Ec/No TRUE FALSE FAL		
- Filter Coefficient - Measurement quantity for frequency quality estimate - Inter-frequency reporting quantity - UTRA Carrier RSSI - Frequency quality estimate - Non frequency related cell reporting quantities - SFN-SFN observed time difference reporting indicator - Cell synchronisation information reporting indicator - Cell Identity reporting indicator - CPICH Ec/No reporting indicator - CPICH RSCP reporting indicator - Pathloss reporting indicator - CFN-SFN observed time difference - Reporting cell status - CHOICE reporteding cell Report cells \(\frac{\pmax}{\pmax}\) within active and/or monitored setcells on used frequency andor within active and/or monitored cells set on non-used frequency - Measurement validity		, , ,
- Measurement quantity for frequency quality estimate - Inter-frequency reporting quantity - UTRA Carrier RSSI - Frequency quality estimate - Non frequency related cell reporting quantities - SFN-SFN observed time difference reporting indicator - Cell synchronisation information reporting indicator - Cell Identity reporting indicator - CPICH Ec/No reporting indicator - CPICH RSCP reporting indicator - Pathloss reporting indicator - Pathloss reporting indicator - Reporting cell status - CHOICE reporteding cell - Maximum number of reporteding cells type2 - Measurement validity FALSE TRUE FALSE FALSE FALSE FALSE Report cells \(\frac{\psi_w}{\psi}\) ithin active and/or monitored set cells on used frequency andor within active and/or monitored eells set on non-used frequency 2		
estimate - Inter-frequency reporting quantity - UTRA Carrier RSSI - Frequency quality estimate - Non frequency related cell reporting quantities - SFN-SFN observed time difference reporting indicator - Cell synchronisation information reporting indicator - Cell Identity reporting indicator - CPICH Ec/No reporting indicator - CPICH RSCP reporting indicator - Pathloss reporting indicator - Pathloss reporting indicator - CFN-SFN observed time difference - Reporting cell status - CHOICE reporteding cell - Maximum number of reporteding cells type2 - Measurement validity FALSE TRUE FALSE FAL		CPICH Ec/No
- Inter-frequency reporting quantity - UTRA Carrier RSSI - Frequency quality estimate - Non frequency related cell reporting quantities - SFN-SFN observed time difference reporting indicator - Cell synchronisation information reporting indicator - Cell Identity reporting indicator - CPICH Ec/No reporting indicator - CPICH RSCP reporting indicator - Pathloss reporting indicator - CFN-SFN observed time difference - Reporting cell status - CHOICE reporteding cell - Maximum number of reporteding cells-type2 - Measurement validity FALSE TRUE FALSE TRUE No report FALSE FALSE FALSE FALSE Report cells \(\frac{\pmathcal{W}}{\pmathcal{W}}\) ithin active and/or monitored set set cells on used frequency and/or monitored dells-set on non-used frequency} 1 PALSE TRUE FALSE TRUE		
- UTRA Carrier RSSI - Frequency quality estimate - Non frequency related cell reporting quantities - SFN-SFN observed time difference reporting indicator - Cell synchronisation information reporting indicator - Cell Identity reporting indicator - CPICH Ec/No reporting indicator - CPICH RSCP reporting indicator - Pathloss reporting indicator - Pathloss reporting indicator - CFN-SFN observed time difference - Reporting cell status - CHOICE reporteding cell - Maximum number of reporteding cells-type2 - Measurement validity TRUE TRUE FALSE FALSE FALSE FALSE Report cells \(\frac{\pmathcal{W}}{\pmathcal{W}}\) ithin active and/or monitored setcells on used frequency andor within active and/or monitored dells-set on non-used frequency 1		
- Frequency quality estimate - Non frequency related cell reporting quantities - SFN-SFN observed time difference reporting indicator - Cell synchronisation information reporting indicator - Cell Identity reporting indicator - CPICH Ec/No reporting indicator - CPICH RSCP reporting indicator - Pathloss reporting indicator - Pathloss reporting indicator - CFN-SFN observed time difference - Reporting cell status - CHOICE reporteding cell - Maximum number of reporteding cells type2 - Measurement validity No report FALSE FA		_
- Non frequency related cell reporting quantities - SFN-SFN observed time difference reporting indicator - Cell synchronisation information reporting indicator - Cell Identity reporting indicator - CPICH Ec/No reporting indicator - CPICH RSCP reporting indicator - Pathloss reporting indicator - Pathloss reporting indicator - CFN-SFN observed time difference - Reporting cell status - CHOICE reporteding cell - Maximum number of reporteding cells type2 - Measurement validity No report FALSE FALSE FALSE FALSE FALSE FALSE Report cells \(\Psi\) within active and/or monitored setcells on used frequency and/or monitored delle-set on non-used frequency 2		TRUE
- SFN-SFN observed time difference reporting indicator - Cell synchronisation information reporting indicator - Cell Identity reporting indicator - CPICH Ec/No reporting indicator - CPICH RSCP reporting indicator - Pathloss reporting indicator - Pathloss reporting indicator - CFN-SFN observed time difference - Reporting cell status - CHOICE reporteding cell - Maximum number of reporteding cells type2 - Measurement validity No report FALSE FALSE FALSE FALSE FALSE Report cells \(\frac{\text{Wwithin active and/or monitored setcells}}{\text{on non-used frequency and or within active and/or monitored cells-set on non-used frequency}} - Measurement validity		
indicator - Cell synchronisation information reporting indicator - Cell Identity reporting indicator - CPICH Ec/No reporting indicator - CPICH RSCP reporting indicator - Pathloss reporting indicator - Pathloss reporting indicator - CFN-SFN observed time difference - Reporting cell status - CHOICE reporteding cell - Maximum number of reporteding cells type2 - Measurement validity - Cell Identity reporting indicator - FALSE - FA		l N
- Cell synchronisation information reporting indicator - Cell Identity reporting indicator - CPICH Ec/No reporting indicator - CPICH RSCP reporting indicator - CPICH RSCP reporting indicator - Pathloss reporting indicator - Pathloss reporting indicator - CFN-SFN observed time difference - Reporting cell status - CHOICE reporteding cell - Maximum number of reporteding cells-type2 - Measurement validity FALSE FALSE FALSE FRUE FALSE FALS		No report
indicator - Cell Identity reporting indicator - CPICH Ec/No reporting indicator - CPICH RSCP reporting indicator - Pathloss reporting indicator - Pathloss reporting indicator - CFN-SFN observed time difference - Reporting cell status - CHOICE reporteding cell - Maximum number of reporteding cells type2 - Measurement validity FALSE		EAL 05
- Cell Identity reporting indicator - CPICH Ec/No reporting indicator - CPICH RSCP reporting indicator - Pathloss reporting indicator - Pathloss reporting indicator - CFN-SFN observed time difference - Reporting cell status - CHOICE reporteding cell - Maximum number of reporteding cells-type2 - Measurement validity FALSE FALSE FALSE		FALSE
- CPICH Ec/No reporting indicator - CPICH RSCP reporting indicator - Pathloss reporting indicator - CFN-SFN observed time difference - Reporting cell status - CHOICE reporteding cell - Maximum number of reporteding cells type2 - Measurement validity TRUE FALSE FALSE FALSE Report cells \(\frac{\pmaintmax}{\pmaintmax}\) within active and/or monitored setcells on used frequency and/or monitored cells set on non-used frequency 2		EAL 05
- CPICH RSCP reporting indicator - Pathloss reporting indicator - CFN-SFN observed time difference - Reporting cell status - CHOICE reporteding cell - Maximum number of reporteding cells-type2 - Measurement validity FALSE FALSE FAL		_
- Pathloss reporting indicator - CFN-SFN observed time difference - Reporting cell status - CHOICE reporteding cell - Maximum number of reporteding cells-type2 - Measurement validity FALSE FALSE Report cells Wwithin active and/or monitored seteells on used frequency andor within active and/or monitored cells-set on non-used frequency 2		_
- CFN-SFN observed time difference - Reporting cell status - CHOICE reporteding cell - Maximum number of reporteding cells type2 - Measurement validity FALSE Report cells Wwithin active and/or monitored setcells on used frequency and/or monitored eells set on non-used frequency 2		_
- Reporting cell status - CHOICE reporteding cell - Maximum number of reporteding cells type2 - Measurement validity - Report cells \(\frac{\pmathbb{W}}{\pmathbb{W}}\) ithin active and/or monitored setcells on used frequency and/or monitored eells set on non-used frequency 2	- Patnioss reporting indicator	
- CHOICE reporteding cell Report cells \(\frac{\text{\phi}}{\text{w}}\) ithin active and/or monitored setcells on used frequency and/or within active and/or monitored eells-set on non-used frequency - Maximum number of reporteding cells type2 - Measurement validity Report cells \(\frac{\text{\phi}}{\text{w}}\) ithin active and/or monitored setcells on used frequency frequency 2		FALSE
- Maximum number of reporteding cells type2 - Measurement validity - Setcells on used frequency andor within active and/or monitored cells set on non-used frequency 2		Descrit celle Wesithin active and/on accritental
- Maximum number of reporteding cells type2 - Measurement validity and/or monitored cells set on non-used frequency 2	- CHOICE report <u>eding</u> cell	
- Maximum number of report <u>eding</u> cells type2 - Measurement validity frequency 2		
- Measurement validity 2	Marrian con a contact of managers discussed by a collection of	
		•
Inter frequency set undete		
- Inter-frequency set update Not present - CHOICE report criteria Not present		
'		·
- Amount of reporting Periodic reporting criteria		
- Reporting interval Infinity		
DPCH compressed mode status info - TGPSI 16 seconds		TO SECONUS
		1
- TGPS Status Flag 1 Active		
255	- 1001 N	

MEASUREMENT REPORT (Step 3 and 8)

Information Element	Value/Remarks
Integrity check info	If integrity protection is activated, this IE shall
	be present and SS checks that the MAC-I
	value matches with the calculated X-MAC
	value. Else, this IE shall be absent.
Measurement identity-number	Check to see if set to "1"
Measured Results	
- CHOICE measurement	Check to see if set to "Inter-frequency
	measured results list"
 Inter-frequency measurement results 	
- Frequency info	
- UARFCN (uplink)	Check to see if set to the UARFCN of the
	uplink frequency for cell 4
- UARFCN (downlink)	Check to see if set to the UARFCN of the
	downlink frequency for cell 4
- UTRA carrier RSSI	Check to see if it is absent
 Inter-frequency cell measurement results 	
- Cell measured results	
- Cell Identity	Check to see if it is absent
 SFN-SFN observed time difference 	Check to see if it is absent
- CFN-SFN observed time difference	Check to see if it is absent
 Cell synchronisation information 	Check to see if it is absent
- Primary CPICH Info	
- Primary Scrambling Code	Check to see if set to the same code for cell 4
- CPICH Ec/No	Check to see if it is present
- CPICH RSCP	Check to see if it is absent
- Pathloss	Check to see if it is absent
Measured Results on RACH	Check to see if it is absent
Event Results	Check to see if it is absent

Information Element	Value/Remark
Measurement Identity-Number	2
Measurement Command	Setup
Measurement Reporting Mode	Not Present
Additional measurements list	Not Present
CHOICE measurement type	Inter-frequency measurement
- Inter-frequency cell info list	' '
 Removed CHOICE inter-frequency info listcell 	Not PresentNo inter-frequency cells removed
removal	
- New inter-frequency info list	Set to id of cell 4
- Inter-frequency cell id	
- Frequency info	UARFCN of the uplink frequency for cell 4
- UARFCN uplink (Nu)	UARFCN of the downink frequency for cell 4
 - UARFCN downlink (Nd) 	
- Cell info	0 dB
 Cell individual offset 	0 chips
 Reference time difference to cell 	
- Primary CPICH Info	Set to same code as used for cell 4
 Primary Scrambling Code 	Not Present
 Primary CPICH TX power 	FALSE
- Read SFN Indicator	FALSE
- TX Diversity Indicator	
 Inter-frequency measurement quantity 	Inter-frequency reporting criteria
- CHOICE reporting criteria	0
- Filter Coefficient	CPICH RSCP
 Measurement quantity for frequency quality 	
estimate	544.05
- Inter-frequency reporting quantity	FALSE
- UTRA Carrier RSSI	TRUE
- Frequency quality estimate	
- Non frequency related cell reporting quantities	No report
- SFN-SFN observed time difference reporting	No report
indicator Call avachranication information reporting	EALCE
 Cell synchronisation information reporting indicator 	FALSE
- Cell Identity reporting indicator	FALSE
- CPICH Ec/No reporting indicator	TRUEFALSE
- CPICH RSCP reporting indicator	FALSETRUE
- Pathloss reporting indicator	FALSE
	FALSE
- Reporting cell status	TALGE
- CHOICE report <u>eding</u> cell	Report cells Wwithin active and/or monitored
or roto 2 roport <u>oa</u> mig oon	cells set on used frequency andor within active
	and/or monitored cells-set on non-used
 Maximum number of reporteding cells-type2 	frequency
- Measurement validity	2
- Inter-frequency set update	Not present
- CHOICE report criteria	Not present
- Amount of reporting	Periodic reporting criteria
- Reporting interval	Infinity
DPCH compressed mode status info	16 seconds
- TGPSi	
- TGPS Status Flag	2
- TGCFN	Inactive
	<u>255</u>

TRANSPORT CHANNEL RECONFIGURATION

The contents of TRANSPORT CHANNEL RECONFIGURATION message in this test case is identical to the message sub-type title "Packet to CELL_DCH from CELL_DCH in PS" found in Annex A with the following exceptions:

ſ	Information Element	Value/remark
	- DPCH compressed mode info	
	- TGPSI	2
	- TGPS Status Flag	Active
	- TGCFN	<u>255</u>
	- Transmission gap pattern sequence configuration	
	parameters	
	- TGMP	FDD Measurement
	- TGPRC	62
	- TGCFN	(Current CFN + (256 – TTI/10msec)) mod 256
	- TGSN	8
	- TGL1	10
	- TGL2	5
	- TGD	15
	- TGPL1	35
	- TGPL2	35
	- RPP	Mode 1
	- ITP	Mode 1
	- UL/DL Mode	DL
	 Downlink compressed mode method 	SF/2
	 Uplink compressed mode method 	Not Present
	- Downlink frame type	A
	- DeltaSIR1	2.0
	- DeltaSIRafter1	1.0
	- DeltaSIR2	Not Present
	- DeltaSIRafter2	Not Present

PHYSICAL CHANNEL RECONFIGURATION FAILURE

Information Element	Value/remark
Integrity check info	If integrity protection is activated, this IE shall be present and SS checks that the MAC-I value matches with the calculated X-MAC value. Else, this IE shall be absent.
Failure cause	Checked to see if set to "compressed mode runtime error"
Protocol error informationDeleted TGPSI	Checked to see if it is absent Checked to see if it is set to "2"

8.4.1.12.5 Test requirement

After step 6 the UE shall keep transmission gap pattern sequence configuration associated with TGPSI=1. It shall delete the transmission gap pattern sequence configuration associated with TGPSI=2, and delete the inter-frequency measurements corresponding to it. It shall transmit PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the uplink DCCH, with the IE "Failure cause" set to "Compressed mode runtime error".

After step 7 the UE shall continue to send MEASUREMENT REPORT messages periodically, to report the CPICH Ec/No readings for cell <u>24</u>. However, no MEASUREMENT REPORT messages containing the <u>CPICH</u> RSCP readings for cell <u>24</u> shall be sent by the UE.

8.4.1.13 Measurement Control and Report: Compressed Mode Configuration Failure during physical channel reconfiguration procedure

8.4.1.13.1 ———Definition

8.4.1.13.2 Conformance requirement

During a physical channel reconfiguration procedure, the UTRAN might request the activation of a new transmission gap pattern sequence configuration. If the UE detects a runtime error due to overlapping compressed mode configuration (when transmission gap pattern sequences create transmission gaps in the same frame), it shall delete the transmission gap pattern sequence configuration associated with highest value of TGPSI. The UE shall also terminate

any inter-frequency measurements corresponding to the deleted transmission gap pattern sequence. Finally, the UE shall transmit a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM-RLC, with the cause value in IE "failure cause" set to "compressed mode runtime error".

Reference

3GPP TS 25.331 clause 8.2.2, clause 8.2.11, clause 8.6.6.14

8.4.1.13.3 Test purpose

To confirm that the UE transmits a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the uplink DCCH using AM RLC, if the-it-receivesed a PHYSICAL CHANNEL RECONFIGURATION message which-includes IE "DPCH compressed mode info" that causes an illegal overlap involving more than one parallel transmission gap pattern sequences. To confirm that the UE terminate any inter-frequency measurements corresponding to the deleted transmission gap pattern sequence

8.4.1.13.4 Method of test

Initial Condition

System Simulator: 2 cells – both cell 1 and cell 4 are active. <u>See Table 8.4.1.11-1 in clause 8.4.1.11.4 for the power settings.</u> The downlink power level of cell 4 is 10dB below cell 1.

UE: CS-DCCH+DTCH_DCH (State 6-9) or PS-DCCH+DTCH_DCH (State 6-10) as specified in TS34.108 clause 7.4, depending on the CN domain supported.

Test Procedure

For this test case, the downlink transmission power settings should follow that specified in Table 8.4.1.11-1 in clause 8.4.1.11.4.

The UE is in the CELL_DCH state in cell 1. SS sends a MEASUREMENT CONTROL message on the downlink DCCH to request the UE to start inter-frequency measurement for cell 24's CPICH Ec/No value. Simultaneously, the stored transmission gap pattern sequence configuration associated with TGPSI=1 is indicated to be activated in this message. Upon the reception of this message, the UE shall transmit MEASUREMENT REPORT messages periodically at 16 seconds interval to report cell 24's measurement results. Next, SS sends a second MEASUREMENT CONTROL message. In this message, a new measurement task is to be established for the measurement and reporting of cell 24's CPICH RSCP value on a periodic basis. A deactivated transmission pattern gap sequence configuration (with TGPSI=2) is associated with this new measurement task.

The SS transmits a PHYSICAL CHANNEL RECONFIGURATION message and commands the activation of transmission gap pattern sequence with TGPSI=2. This is expected to result in the detection of a runtime error due to overlapping compressed mode configuration. The UE then shall transmit a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM-RLC. In this message, the value of IE "failure cause" shall be set to "compressed mode runtime error". The UE shall terminate all inter-frequency measurement tasks associated with TGPSI=2. However, the UE shall continue to send MEASUREMENT REPORT messages to report cell $2\underline{4}$'s CPICH Ec/No value, which is measured during the transmission gap created by compressed mode configuration corresponding to TGPSI=1.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1			UE is initially in CELL_DCH state.
2	←	MEASUREMENT CONTROL	Start inter-frequency
			measurements for cell 24's CPICH Ec/No using transmission gap pattern sequence with TGPSI=1.
3	→	MEASUREMENT REPORT	UE reports cell 24's CPICH Ec/No readings periodically.
4	+	MEASUREMENT CONTROL	Assign inter-frequency measurements for cell 24's CPICH RSCP. This measurement task is associated with transmission gap pattern sequence with TGPSI=2, which has not been activated yet.
5	←	PHYSICAL CHANNEL RECONFIGURATION	SS specifies the parameters for transmission gap pattern sequence with TGPSI=2 and activate it simultaneously
6			UE shall delete transmission gap pattern sequence configuration associated with TGPSI=2.
7	→	PHYSICAL CHANNEL RECONFIGURATION FAILURE	Failure cause shall be set to "Compressed mode runtime error"
8	\rightarrow	MEASUREMENT REPORT	The contents shall be the same as that in step 3.

MEASUREMENT CONTROL (Step 2)

Information Flament	Value ID
Information Element	Value/Remark
Measurement Identity-Number	1 Setus
Measurement Command	Setup
Measurement Reporting Mode	Not Present
Additional measurements list	Not Present
CHOICE measurement type	Inter-frequency measurement
- Inter-frequency cell info list	N C N C
- Removed CHOICE inter-frequency info listcell	Not PresentNo inter-frequency cells removed
removal	
- New inter-frequency info list	Set to id of cell 4
- Inter-frequency cell id	LIADEON of the continue for over a set of
- Frequency info	UARFCN of the uplink frequency for cell 4
- UARFON uplink (Nu)	UARFCN of the downink frequency for cell 4
- UARFCN downlink (Nd)	0.40
- Cell info	0 dB
- Cell individual offset	0 chips
- Reference time difference to cell	
- Primary CPICH Info	Set to same code as used for cell 4
- Primary Scrambling Code	Not Present
- Primary CPICH TX power	FALSE
- Read SFN Indicator	FALSE
- TX Diversity Indicator	
- Inter-frequency measurement quantity	Inter-frequency reporting criteria
- CHOICE reporting criteria	0
- Filter Coefficient	CPICH Ec/No
Measurement quantity for frequency quality	
estimate	54.05
- Inter-frequency reporting quantity	FALSE
- UTRA Carrier RSSI	TRUE
- Frequency quality estimate	
- Non frequency related cell reporting quantities	
- SFN-SFN observed time difference reporting	No report
indicator	54.05
- Cell synchronisation information reporting	<u>FALSE</u>
indicator	54.05
- Cell Identity reporting indicator	FALSE
- CPICH Ec/No reporting indicator	TRUE
- CPICH RSCP reporting indicator	FALSE
- Pathloss reporting indicator	FALSE
- CFN-SFN observed time difference	FALSE
- Reporting cell status	B
- CHOICE report <u>eding</u> cell	Report cells Wwithin active and/or monitored
	setcells on used frequency andor within active
	and/or monitored setcells on non-used
- Maximum number of reporteding cells type2	frequency
- Measurement validity	2
- Inter-frequency set update	Not present
- CHOICE report criteria	Not present
- Amount of reporting	Periodic reporting criteria
- Reporting interval	Infinity
DPCH compressed mode status info	16 seconds
- TGPSI	
- TGPS Status Flag	1
- TGCFN	Active
	<u>255</u>

MEASUREMENT REPORT (Step 3 and 8)

Information Element	Value/Remarks
Integrity check info	If integrity protection is activated, this IE shall
	be present and SS checks that the MAC-I
	value matches with the calculated X-MAC
	value. Else, this IE shall be absent.
Measurement identity-number	Check to see if set to "1"
Measured Results	
- CHOICE measurement	Check to see if set to "Inter-frequency
	measured results list"
 Inter-frequency measurement results 	
- Frequency info	
- UARFCN (uplink)	Check to see if set to the UARFCN of the
	uplink frequency for cell 4
- UARFCN (downlink)	Check to see if set to the UARFCN of the
	downlink frequency for cell 4
- UTRA carrier RSSI	Check to see if it is absent
 Inter-frequency cell measurement results 	
- Cell measured results	
- Cell Identity	Check to see if it is absent
 SFN-SFN observed time difference 	Check to see if it is absent
- CFN-SFN observed time difference	Check to see if it is absent
 Cell synchronisation information 	Check to see if it is absent
- Primary CPICH Info	
- Primary Scrambling Code	Check to see if set to the same code for cell 4
- CPICH Ec/No	Check to see if it is present
- CPICH RSCP	Check to see if it is absent
- Pathloss	Check to see if it is absent
Measured Results on RACH	Check to see if it is absent
Event Results	Check to see if it is absent

	V
Information Element	Value/Remark
Measurement Identity-Number	2
Measurement Command	Setup
Measurement Reporting Mode	Not Present
Additional measurements list	Not Present
CHOICE measurement type	Inter-frequency measurement
- Inter-frequency cell info list	
 Removed CHOICE inter-frequency cell 	Not PresentNo inter-frequency cells removed
removalinfo list	
 New inter-frequency info list 	Set to id of cell 4
- Inter-frequency cell id	
- Frequency info	UARFCN of the uplink frequency for cell 4
- UARFCN uplink (Nu)	UARFCN of the downink frequency for cell 4
- UARFCN downlink (Nd)	-
- Cell info	0 dB
- Cell individual offset	0 chips
- Reference time difference to cell	
- Primary CPICH Info	Set to same code as used for cell 4
- Primary Scrambling Code	Not Present
- Primary CPICH TX power	FALSE
- Read SFN Indicator	FALSE
- TX Diversity Indicator	
 Inter-frequency measurement quantity 	Inter-frequency reporting criteria
 CHOICE reporting criteria 	0
- Filter Coefficient	CPICH RSCP
 Measurement quantity for frequency quality 	
estimate	
- Inter-frequency reporting quantity	FALSE
- UTRA Carrier RSSI	TRUE
- Frequency quality estimate	
- Non frequency related cell reporting quantities	
- SFN-SFN observed time difference reporting	No report
<u>indicator</u>	
- Cell synchronisation information reporting	<u>FALSE</u>
indicator	
- Cell Identity reporting indicator	FALSE
- CPICH Ec/No reporting indicator	TRUE FALSE
- CPICH RSCP reporting indicator	<u>FALSETRUE</u>
- Pathloss reporting indicator	FALSE
	FALSE
- Reporting cell status	
- CHOICE report <u>eding</u> cell	Report cells Wwithin active and/or monitored
	setcells on used frequency andor within active
	and/or monitored cells on non-used frequency
- Maximum number of reporteding cells type2	2
- Measurement validity	Not present
- Inter-frequency set update	Not present
- CHOICE report criteria	Periodic reporting criteria
- Amount of reporting	Infinity
- Reporting interval	16 seconds
DPCH compressed mode status info	
- TGPSI	2
- TGPS Status Flag	Inactive
- TGCFN	<u>255</u>

PHYSICAL CHANNEL RECONFIGURATION

The contents of PHYSICAL CHANNEL RECONFIGURATION message in this test case is identical to the message sub-type title "Packet to CELL_DCH from CELL_DCH in PS" found in Annex A with the following exceptions:

Information Element	Value/remark
- DPCH compressed mode info	
- TGPSI	2
- TGPS Status Flag	Active
- TGCFN	<u>255</u>
 Transmission gap pattern sequence configuration 	
parameters	
TGMP	FDD Measurement
TGPRC	62
TGCFN	(Current CFN + (256 – TTI/10msec)) mod 256
TGSN	8
TGL1	10
TGL2	5
TGD	15
TGPL1	35
TGPL2	35
RPP	Mode 1
ITP	Mode 1
UL/DL Mode	DL
Downlink compressed mode method	SF/2
Uplink compressed mode method	Not Present
Downlink frame type	A
DeltaSIR1	2.0
DeltaSIRafter1	1.0
DeltaSIR2	Not Present
DeltaSIRafter2	Not Present

PHYSICAL CHANNEL RECONFIGURATION FAILURE

Information Element	Value/remark
Integrity check info	If integrity protection is activated, this IE shall be present and SS checks that the MAC-I value matches with the calculated X-MAC value. Else, this IE shall be absent.
Failure cause	Checked to see if set to "compressed mode runtime error"
Protocol error informationDeleted TGPSI	Checked to see if it is absent Checked to see if it is set to "2"

8.4.1.13.5 Test requirement

After step 6 the UE shall keep transmission gap pattern sequence configuration associated with TGPSI=1. It shall delete the transmission gap pattern sequence configuration associated with TGPSI=2, and delete the inter-frequency measurements corresponding to it. It shall transmit PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the uplink DCCH, with the IE "Failure cause" set to "Compressed mode runtime error".

After step 7 the UE shall continue to send MEASUREMENT REPORT messages periodically, to report the CPICH Ec/No readings for cell <u>24</u>. However, no MEASUREMENT REPORT messages containing the <u>CPICH RSCP</u> readings for cell <u>24</u> shall be sent by the UE.

8.4.1.14 Measurement Control and Report: Cell forbidden to affect reporting range

8.4.1.14.1 Definition

8.4.1.14.2 Conformance requirement

When event 1A is ordered by the UTRAN in a MEASUREMENT CONTROL message, the UE shall send a MEASUREMENT REPORT message when a primary CPICH measured has entered the specified reporting range. The UTRAN can request that a certain primary CPICH be forbidden to affect the reporting range used for event 1A measurement reporting. However, the UE shall ignore such request from the UTRAN if two conditions are satisfied –

(a) the primary CPICH concerned is included in the active set, and (b) all cells in the active set are defined as primary CPICH forbidden to affect the reporting range.

Reference

3GPP TS 25.331 clause 14.1.2.1, clause 14.1.5.4

8.4.1.14.3 Test Purpose

To confirm that the UE reports to the SS, if a primary CPICH currently measured by the UE enters the reporting range (event 1A). The reporting range was specified in a MEASUREMENT CONTROL message received earlier. To confirm that the UE ignores SS's request to forbid the updating of reporting range, when (a) the primary CPICH concerned is one of the cells currently in active set and (b) all cells in the active sets are marked as primary CPICH forbidden to affect the reporting range.

8.4.1.14.4 Method of test

Initial Condition

System Simulator: 3 cells – The initial configurations of the 3 cells in the SS should follow the values indicated in the column marked "TO" in table 8.4.1.14-1. The table is found in "Test Procedure" sub-clause.

<u>UE: CS-DCCH DCH (State 6-5) or PS-DCCH DCH (State 6-7) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE.</u>

Test Procedure

Table 8.4.1.14-1 illustrates the downlink power to be applied for the 3 cells at various time instants of the test execution. Column marked "T0" denotes the initial conditions, while columns marked "T1", "T2", "T3" and "T4" are to be applied subsequently. The exact instants on which these values shall be applied are described in the text in this sub-clause.

Table 8.4.1.14-1

Para- meter	<u>Unit</u>	Cell 1				Cell 2				Cell 3						
<u> </u>		<u>T0</u>	<u>T1</u>	T2	<u>T3</u>	<u>T4</u>	<u>T0</u>	<u>T1</u>	<u>T2</u>	<u>T3</u>	<u>T4</u>	<u>T0</u>	<u>T1</u>	T2	<u>T3</u>	<u>T4</u>
UTRA RF Channel Number				<u>Ch. 1</u>					<u>Ch. 1</u>					<u>Ch. 1</u>		
$\frac{\hat{I}_{or}/I_{oc}}{}$	<u>dB</u>	<u>9.5</u>	<u>9.5</u>	<u>9.5</u>	<u>9.5</u>	<u>9.5</u>	4.0	4.0	4.0	<u>14.5</u>	<u>14.5</u>	is d off	8.0	<u>1.5</u>	<u>1.5</u>	8.0
CPICH Ec/lo	<u>dB</u>	<u>-</u> 11.4	<u>-</u> 13.2	<u>-</u> 11.9	<u>-</u> 16.5	<u>-</u> 17.0	<u>-</u> 16.9	<u>-</u> 18.7	<u>-</u> 17.4	<u>-</u> 11.5	<u>-</u> 12.0	Cell 3 is	<u>-</u> 14.7	<u>-</u> 19.9	<u>-</u> 24.5	<u>-</u> 18.5
CPICH RSCP	<u>dBm</u>	<u>-</u> 70.5	<u>-</u> 70.5	<u>-</u> 70.5	<u>-</u> 70.5	<u>-</u> 70.5	<u>-</u> 76.0	<u>-</u> 76.0	<u>-</u> 76.0	<u>-</u> 66.5	<u>-</u> 66.5	\S	<u>-</u> 72.0	<u>-</u> 78.5	<u>-</u> 78.5	<u>-</u> 72.0

The UE is initially in CELL DCH state of cell 1. SS then performs an active set update procedure by sending ACTIVE SET UPDATE REQUEST message on the downlink DCCH. Cell 2 is to be added to the active set, according to the content of this downlink message. The UE shall reply with an ACTIVE SET UPDATE COMPLETE message on the uplink DCCH, and include cell 2 to the active set when the activation time specified has elapsed.

SS configures itself according to the values in columns "T1" shown above. SS then sends a MEASUREMENT CONTROL message to the UE, commanding the start of intra-frequency measurement for all 3 cells. The intra-frequency measurement report criteria is set to event-triggered using event 1A. The reporting range is set to 5 dB in the MEASUREMENT CONTROL message. The UE shall send a MEASUREMENT REPORT on the uplink DCCH, which contains the CPICH RSCP reading for cell 3.

SS executes the active set update procedure again, but requesting that cell 3 be added to the active set this time. The UE shall respond with ACTIVE SET UPDATE message on the uplink DCCH and then includes cell 3 into its current active

set. Following this, SS configures itself according to the values in columns "T2" shown above. The UE should detect that CPICH RSCP of cell 3 has dropped out of the reporting range.

Next, SS configures itself according to the values in columns "T3" shown above. SS then sends a MEASUREMENT CONTROL message to command that all cells in the active set are forbidden to update the reporting range for event 1A.

Finally, SS configures itself according to the values in columns "T4" shown above. The UE shall proceed to update the reporting range as cell 2 has become the strongest cell. Although the CPICH RSCP value of cell 3 has been restored, this value still falls outside the new reporting range. Therefore, the UE shall not transmit a MEASUREMENT REPORT message on the uplink to report the triggering of event 1A.

Expected sequence

Step	Direction UE SS	<u>Message</u>	Comment
1	<u>~</u>	System Information Block type 11	UE is initially in CELL DCH state in cell 1. SIB 11 is modified to include both cell 2 and 5 into the monitored cell list.
<u>2</u>	<u></u>	ACTIVE SET UPDATE	SS asks UE to add cell 2 into the active set
3	<u>→</u>	ACTIVE SET UPDATE COMPLETE	
4			SS configures itself according to the settings stated in column "T1" of Table 8.4.1.14-1.
<u>5</u>	<u></u> ←	MEASUREMENT CONTROL	SS commands the start of measurement tasks for CPICH RSCP of cell 1, cell 2 and cell 3. All 3 cells are added under IE "Intra-frequency cell info list". The reporting criteria is set to event-triggered using event type 1A, with reporting range = 5 dB.
<u>6</u>	<u> </u>	MEASUREMENT REPORT	UE shall report that cell 3 has entered the reporting range for event 1A.
<u>7</u>	<u></u>	ACTIVE SET UPDATE	SS asks UE to add cell 3 into the active set
8	<u>→</u>	ACTIVE SET UPDATE COMPLETE	
9			SS configures itself according to the settings stated in column "T2" of Table 8.4.1.14-1.
<u>10</u>			SS configures itself according to the settings stated in column "T3" of Table 8.4.1.14-1.
<u>11</u>	<u></u>	MEASUREMENT CONTROL	SS forbids all cells in active list to affect the reporting range
12			SS configures itself according to the settings stated in column "T4" of Table 8.4.1.14-1.
<u>13</u>			UE shall ignore the restrictions imposed by the messages received in step 11. It shall update the reporting range. SS verifies that no MEASUREMENT REPORT messages are received in the uplink direction

Specific Message Contents

ACTIVE SET UPDATE (Step 2)

The contents of ACTIVE SET UPDATE message for this test step is identical to the same message found in Annex A with the following exceptions:

Information Element	<u>Value/remark</u>
RRC transaction identifier	<u>0</u>
Radio link addition information	
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as assigned for cell 2
- Downlink DPCH info for each RL	
- Primary CPICH usage for channel estimation	P-CPICH can be used.
 DPCH frame offset 	0 chips
 Secondary CPICH info 	Not Present
 DL channelisation code 	This IE is repeated for all existing downlink
	DPCHs allocated to the UE
 Secondary scrambling code 	Not Present
 CHOICE Spreading factor 	<u>512</u>
- Code Number	For each DPCH, assign the same code
	number in the current code given in cell 1.
 Scrambling code change 	Not Present
- TPC Combination Index	Not Present
- SSDT Cell Identity	Not Present
 Close loop timing adjustment mode 	Not Present
- TFCI Combining Indicator	Not Present
- SCCPCH information for FACH	Not Present
Radio link removal information	Not Present

ACTIVE SET UPDATE COMPLETE (Step 3 and Step 8)

Information Element	<u>Value/remark</u>			
RRC transaction identifier	Check to see if it is set to 0			

MEASUREMENT CONTROL (Step 5)

The contents of MEASUREMENT CONTROL message for this test step is identical to the same message found in Annex A with the following exceptions:

Information Element	<u>Value/Remark</u>
RRC transaction identifier	1
Measurement Identity	<u> 1</u>
Measurement Command	<u>Setup</u>
Measurement Reporting Mode	Not Present
Additional measurements list	Not Present
CHOICE measurement type	Intra-frequency measurement
- Intra-frequency cell info list	
- CHOICE intra-frequency cell removal	Remove no intra-frequency
- New intra-frequency info list	3 cells are specified – cell 1, cell 2 and cell 3
- Intra-frequency cell id	<u>0</u>
- Cell info	0.40
- Cell individual offset	0 dB
- Reference time difference to cell - Primary CPICH Info	0 chips
- Primary Scrambling Code	Set to same code as used for cell 1
- Primary CPICH TX power	Not Present
- Read SFN Indicator	FALSE
- TX Diversity Indicator	FALSE
- Intra-frequency cell id	1
- Cell info	
- Cell individual offset	0 dB
- Reference time difference to cell	0 chips
- Primary CPICH Info	<u> </u>
- Primary Scrambling Code	Set to same code as used for cell 2
- Primary CPICH TX power	Not Present
- Read SFN Indicator	FALSE
- TX Diversity Indicator	FALSE
- Intra-frequency cell id	2
- Cell info	_
- Cell individual offset	0 dB
- Reference time difference to cell	0 chips
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as used for cell 3
- Primary CPICH TX power	Not Present
- Read SFN Indicator	FALSE
- TX Diversity Indicator	FALSE
- Intra-frequency measurement quantity	
- Filter Coefficient	<u>0</u>
- Measurement quantity	<u>CPICH RSCP</u>
- Intra-frequency reporting quantity	
- Reporting quantities for active set cells	
- SFN-SFN observed time difference reporting	No report
indicator	54.05
- Cell synchronisation information reporting	<u>FALSE</u>
indicator	E41.0E
- Cell identity reporting indicator	FALSE
- CPICH Ec/No reporting indicator	FALSE
- CPICH RSCP reporting indicator	TRUE
- Pathloss reporting indicator	FALSE
- Reporting quantities for monitored set cells	No report
- SFN-SFN observed time difference reporting	No report
indicator - Cell synchronisation information reporting	ENISE
 Cell synchronisation information reporting indicator 	FALSE
- Cell identity reporting indicator	ENISE
- Cell identity reporting indicator - CPICH Ec/No reporting indicator	FALSE FALSE
- CPICH EC/NO reporting indicator - CPICH RSCP reporting indicator	FALSE
- Pathloss reporting indicator	FALSE
- Reporting quantities for detected cells	Not present
- Reporting quantities for detected cells - Reporting cell status	Not present
- Reporting cell status - Measurement validity	Not present
- CHOICE report criteria	Intra-frequency measurement reporting criteria
OHOIOL TOPOIL GIRCHA	Only 1 event is specified
- Parameters required for each events	1a
- Intra-frequency event identity	Not Present
- Triggering conditions 1	Active set cells
- Triggering conditions 2	5.0 dB
ringgoring conditions E	<u> </u>

- Reporting range	Not Present
 Cells forbidden to affect reporting range 	<u>0</u>
W	<u>0 dB</u>
- Hysteresis	Not Present
- Threshold used frequency	<u>3</u>
- Reporting deactivation threshold	Not Present
- Replacement activation threshold	<u>0 msec</u>
- Time to trigger	<u>1</u>
- Amount of reporting	<u>0</u>
- Reporting interval	Not Present
- Reporting cell status	Not Present
DPCH compressed mode status info	

MEASUREMENT REPORT (Step 6)

Information Element	Value/Remarks
RRC transaction identifier	Check to see if set to 1
Measurement identity	Check to see if set to 1
Measured Results	
- CHOICE measurement	Check to see if set to "Intra-frequency
	measured results list"
 Intra-frequency measured results 	
- Cell measured results	
- Cell Identity	Check to see if it is absent
 SFN-SFN observed time difference 	Check to see if this IE is absent
 Cell synchronisation information 	Check to see if this IE is absent
- Primary CPICH Info	
- Primary Scrambling Code	Check to see if it's the same code for cell 3
- CPICH Ec/No	Check to see if this IE is absent
- CPICH RSCP	Check to see if this IE is present
- Pathloss	Check to see if this IE is absent
Measured Results on RACH	Check to see if this IE is absent
Additional Measured Results	Check to see if this IE is absent
Event Results	Check to see if this IE is absent

ACTIVE SET UPDATE (Step 7)

 $\frac{\text{The contents of ACTIVE SET UPDATE message for this test step is identical to the same message found in Annex A}{\text{with the following exceptions:}}$

Information Element	<u>Value/remark</u>
RRC transaction identifier	<u>0</u>
Radio link addition information	
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as assigned for cell 3
 Downlink DPCH info for each RL 	
 Primary CPICH usage for channel estimation 	P-CPICH can be used.
- DPCH frame offset	0 chips
- Secondary CPICH info	Not Present
- DL channelisation code	This IE is repeated for all existing downlink
	DPCHs allocated to the UE
 Secondary scrambling code 	Not Present
 CHOICE Spreading factor 	<u>512</u>
- Code Number	For each DPCH, assign the same code
	number in the current code given in cell 1.
 Scrambling code change 	Not Present
- TPC Combination Index	Not Present
- SSDT Cell Identity	Not Present
 Close loop timing adjustment mode 	Not Present
- TFCI Combining Indicator	Not Present
- SCCPCH information for FACH	Not Present
Radio link removal information	Not Present

MEASUREMENT CONTROL (Step 11)

The contents of MEASUREMENT CONTROL message for this test step is identical to the same message found in Annex A with the following exceptions:

Information Element	Value/Remark
RRC transaction identifier	
Measurement Identity	1 1
Measurement Command	Modify
Measurement Reporting Mode	Not Present
Additional measurements list	Not Present
CHOICE measurement type	Intra-frequency measurement
- Intra-frequency cell info list	Not Present
- Intra-frequency measurement quantity	Not Present
- Intra-frequency reporting quantity	Not Present
- CHOICE report criteria	Intra-frequency measurement reporting criteria
	Only 1 event is specified
- Parameters required for each events	<u>1a</u>
- Intra-frequency event identity	Not Present
- Triggering conditions 1	Active set cells
- Triggering conditions 2	5.0 dB
- Reporting range	3 cells - cell 1, cell 2 and cell 3
 Cells forbidden to affect reporting range 	<u>FDD</u>
- CHOICE Mode	
- Primary CPICH info	Set to the same code as in cell 1
- Primary scrambling code	<u>FDD</u>
- CHOICE Mode	
- Primary CPICH info	Set to the same code as in cell 2
- Primary scrambling code	<u>FDD</u>
- CHOICE Mode	
- Primary CPICH info	Set to the same code as in cell 3
 Primary scrambling code 	<u>0</u>
<u>- W</u>	<u>0 dB</u>
- Hysteresis	Not Present
 Threshold used frequency 	<u>3</u>
 Reporting deactivation threshold 	Not Present
 Replacement activation threshold 	<u>0 msec</u>
- Time to trigger	<u>1</u>
 Amount of reporting 	<u>0</u>
- Reporting interval	Not Present
- Reporting cell status	Not Present
DPCH compressed mode status info	

8.4.1.14.5 Test requirement

After step 2, the UE shall transmit ACTIVE SET UPDATE COMPLETE message on the uplink DCCH to acknowledge the successful addition of cell 2 into the active set. The UE shall be able to communicate with cell 2 in both the uplink and downlink directions after this step.

After step 5, the UE shall send a MEASUREMENT REPORT message on the uplink DCCH. The message shall contain the measurement reading for cell 3's CPICH RSCP.

After step 7, the UE shall transmit ACTIVE SET UPDATE COMPLETE message on the uplink DCCH to acknowledge the successful addition of cell 3 into the active set. The UE shall be able to communicate with cell 23 in both the uplink and downlink directions after this step.

After step 12, the UE shall ignore the previous restriction imposed for the updating of reporting range. It shall determine that cell 3's RSCP value is not within the updated reporting range. SS verifies that the UE does not send a MEASUREMENT REPORT message on the uplink DCCH to report cell 3's RSCP value.

Annex A: Default RRC Message Contents

This clause contains the default values of RRC messages, other than those specified in TS 34.108 clauses 6 and 9. Unless indicated otherwise in specific test cases, they shall be transmitted by the system simulator in RRC messages, and which are required to be received from the UE under test.

The necessary L3 messages are listed in alphabetic order, with the exception of the SYSTEM INFORMATION messages, where it is the information elements which are listed in alphabetic order (this is because some information elements occur in several SYSTEM INFORMATION types).

In this clause, decimal values are normally used. However, sometimes a hexadecimal value, indicated by an "H", or a binary value, indicated by a "B" is used.

Default SYSTEM INFORMATION:

NOTE 1: SYSTEM INFORMATION BLOCK TYPE 1 (except for PLMN type "GSM-MAP"), SYSTEM INFORMATION BLOCK TYPE 8, SYSTEM INFORMATION BLOCK TYPE 9, SYSTEM INFORMATION BLOCK TYPE 10, SYSTEM INFORMATION BLOCK TYPE 14, SYSTEM INFORMATION BLOCK TYPE 16 messages are not

used.

Information Element	Value/remark
Message Type	
RRC transaction identifier	Arbitrarily selects one integer between 0 to 3
Integrity check info	The presence of this IE is dependent on IXIT statements
	in TS 34.123-2. If integrity protection is indicated to be
	active, this IE is present with the values of the sub IEs as
	stated below. Else, this IE and the sub-IEs are omitted.
- message authentication code	SS calculates the value of MAC-I for this message and
•	writes to this IE.
 RRC message sequence number 	SS provides the value of this IE, from its internal counter.
Integrity protection mode info	Not Present
Ciphering mode info	Not Present
Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
New U-RNTI	Not Present
CN information info	Not Present
RB with PDCP information list	Not Present
Maximum allowed UL TX power	33dBm
Radio link addition information	(This IE is repeated for addition RL number.)
- Primary CPICH info	
- Primary scrambling code	The value is for additional cell
- Downlink DPCH info for each RL	
- Primary CPICH usage for channel estimation	Primary CPICH may be used
- DPCH frame offset	0 chips
- Secondary CPICH info	Not Present
- DL channelisation code	
- Secondary scrambling code	1
- Code number	SF-1(SF is reference to TS34.108 clause 6.10 Parameter
Carambling and about	Set)
 Scrambling code change TPC combination index 	No change
	0
SSDT Cell Identity Closed loop timing adjustment mode	l -a Not Present
- TFCI combining indicator	TRUE
- SCCPCH Information for FACH	INOL
- Secondary CCPCH info	
- Selection Indicator	Not Present
- Primary CPICH usage for channel estimation	Primary CPICH may be used
- Secondary CPICH info	Not Present
- Secondary scrambling code	1
- SSDT Indicator	FALSE
- Spreading factor	Reference to TS34.108 clause 6.10 Parameter Set
- Code number	SF-1(SF is reference to TS34.108 clause 6.10 Parameter
	Set)
 Pilot symbol existence 	FALSE
- TFCI existence	TRUE
- Fixed or Flexible Position	Flexible
- Timing offset	0
- TFCS	(This IE is repeated for TFC number for PCH and FACH.)
- Normal	
- TFCI Field 1 information	A LPC
 CHOICE TFCS representation TFCS addition information 	Addition
	Number of hits used must be enough to sever all
- CHOICE CTFC Size	Number of bits used must be enough to cover all combinations of CTFC from clause 6.10.
- CTFC information	Refer to TS34.108 clause 6.10 Parameter Set.
- Power offset information	Not Present
- FACH/PCH information	NOUTESCH
- TFS	(PCH)
- CHOICE Transport channel type	Common transport channels
- Dynamic Transport format information	(This IE is repeated for TFI number.)
- RLC Size	Reference to TS34.108 clause 6.10 Parameter Set
- Number of TBs and TTI List	(This IE is repeated for TFI number.)
- Number of Transport blocks	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
- CHOICE mode	FDD
- CHOICE Logical Channel List	ALL

- Semi-static Transport Format information	
- Transmission time interval	Reference to TS34.108 clause 6.10 Parameter Set
- Type of channel coding	Reference to TS34.108 clause 6.10 Parameter Set
- Coding Rate	Reference to TS34.108 clause 6.10 Parameter Set
- Rate matching attribute	Reference to TS34.108 clause 6.10 Parameter Set
- CRC size	Reference to TS34.108 clause 6.10 Parameter Set
- TFS	(FACH)
- CHOICE Transport channel type	Common transport channels
 Dynamic Transport format information 	(This IE is repeated for TFI number.)
- RLC Size	Reference to TS34.108 clause 6.10 Parameter Set
 Number of TBs and TTI List 	(This IE is repeated for TFI number.)
 Number of Transport blocks 	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
- CHOICE mode	<u>FDD</u>
- CHOICE Logical Channel List	<u>ALL</u>
 Semi-static Transport Format information 	
 Transmission time interval 	Reference to TS34.108 clause 6.10 Parameter Set
 Type of channel coding 	Reference to TS34.108 clause 6.10 Parameter Set
- Coding Rate	Reference to TS34.108 clause 6.10 Parameter Set
 Rate matching attribute 	Reference to TS34.108 clause 6.10 Parameter Set
- CRC size	Reference to TS34.108 clause 6.10 Parameter Set
 References to system information blocks 	Not Present
 Scheduling information 	
Radio link removal information	(This IE is repeated for removal RL number.)
- Primary CPICH info	
- Primary scrambling code	The value is for removal cell
TX Diversity Mode	None
SSDT information	Not Present

Contents of ACTIVE SET UPDATE COMPLETE message: AM

	Information Element	Value/remark
	Message Type	
l	RRC transaction identifier	Checked to see if it matches the same value used in the
		corresponding downlink ACTIVE SET UPDATE message
	Integrity check info	The presence of this IE is dependent on IXIT statements
		in TS 34.123-2. If integrity protection is indicated to be
		active, this IE shall be present with the values of the sub
		IEs as stated below. Else, this IE and the sub-IEs shall be
		absent.
	 Message authentication code 	This IE is checked to see if it is present. The value is
		compared against the XMAC-I value computed by SS.
	- RRC Message sequence number	This IE is checked to see if it is present. The value is
		used by SS to compute the XMAC-I value.
	Uplink integrity protection activation info	Not checked
	Radio bearer uplink ciphering activation time info	Not checked
I	RB with PDCP information list	Not checked

Contents of ACTIVE SET UPDATE FAILURE message: AM

Information Element	Value/remark
Message Type	
RRC transaction identifier	Checked to see if it matches the same value used in the
	corresponding downlink ACTIVE SET UPDATE message
Integrity check info	The presence of this IE is dependent on IXIT statements
	in TS 34.123-2. If integrity protection is indicated to be
	active, this IE shall be present with the values of the sub
	IEs as stated below. Else, this IE and the sub-IEs shall be
	absent.
- Message authentication code	This IE is checked to see if it is present. The value is
	compared against the XMAC-I value computed by SS.
- RRC Message sequence number	This IE is checked to see if it is present. The value is
	used by SS to compute the XMAC-I value.
Failure cause	Value will be checked

Contents of CELL UPDATE message: TM

Information Element	Value/remark
Message Type	
U-RNTI	Checked to see if it is set to the following values
- SRNC identity	0000 0000 0001B
- S-RNTI	0000 0000 0000 0000 0001B
RRC transaction identifier	Checked to see if it is absent
Integrity check info	The presence of this IE is dependent on IXIT statements
	in TS 34.123-2. If integrity protection is indicated to be
	active, this IE shall be present with the values of the sub
	IEs as stated below. Else, this IE and the sub-IEs shall be absent.
- Message authentication code	This IE is checked to see if it is present. The value is
	compared against the XMAC-I value computed by SS.
- RRC Message sequence number	This IE is checked to see if it is present. The value is
	used by SS to compute the XMAC-I value.
START List	Checked to see if the 'CN domain identity' and 'START'
	IEs are present for all CN domains supported by the UE
- CN domain identity	Checked to see if it is one of the supported CN domains
- START	Checked to see if it is present
AM_RLC error indication_(for c-plane)	Not cChecked to see if it is set to 'FALSE'
AM_RLC error indication (for u-plane)	Not cChecked to see if it is set to 'FALSE'
Cell update cause	See the test content
Failure cause	Checked to see if it is absent
RB timer indicator	
- T314 expired	Checked to see if it is set to 'FALSE'
- T315 expired	Checked to see if it is set to 'FALSE'
Protocol error indicator	Not checked
Measured results on RACH	Not checked
Protocol error information	Not checked

Information Element	Value/remark
Message Type	
U-RNTI	If this message is sent on CCCH, use the following
	values. Else, this IE is absent.
- SRNC identity	0000 0000 0001B
- S-RNTI	0000 0000 0000 0000 0001B
RRC transaction identifier	Selects an arbitrary integer between 0 to 3
Integrity check info	The presence of this IE is dependent on IXIT statements
	in TS 34.123-2. If integrity protection is indicated to be
	active, this IE is present with the values of the sub IEs as
	stated below. Else, this IE and the sub-IEs are omitted.
- message authentication code	SS calculates the value of MAC-I for this message and
mossage admonitodation code	writes to this IE.
- RRC message sequence number	SS provides the value of this IE, from its internal counter.
Integrity protection mode info	Not Present
Ciphering mode info	Not Present
Activation time	Not Present – use default value
New U-RNTI	Not Present
New C-RNTI	Not Present
DRX-RRC State indicator	NoDRXCELL FACH
UTRAN DRX cycle length coefficient	Not Present
RLC reset indicator_(for C-plane)	FALSE
RLC reset indicator (for U-plane)	FALSE
CN information info	Not Present
URA identity	0000 0000 0001B
RB information to release list	Not Present
RB information to reconfigure list	Not Present
RB information to be affected list	Not Present
RB with PDCP information list	Not Present
UL Transport channel information common for all	Not Present
transport channels	
Deleted TrCH information list	Not Present
- Deleted UL TrCH information	
Added or Reconfigured TrCH information list	Not Present
- Added or Reconfigured UL TrCH information	
CHOICE Mode	Not Present
DL Transport channel information common for all	Not Present
transport channels	
Deleted TrCH information list	Not Present
- Deleted DL TrCH information	
Added or Reconfigured TrCH information list	Not Present
Added or Reconfigured DL TrCH information	
Frequency info	Not Present
Maximum allowed UL TX power	33dBm
CHOICE channel requirement	Not Present
——-CHOICE mode	<u>FDD</u>
- Downlink PDSCH information	Not Present
Downlink information common for ene all radio links	Not Present
Downlink information per radio link list	Not Present

Ţ	Information Element	Value/remark
Ţ	Message Type	
$\ $	RRC transaction identifier	Arbitrarily selects an unused integer between 0 to 3
	Integrity check info	The presence of this IE is dependent on IXIT statements
J		in TS 34.123-2. If integrity protection is indicated to be
		active, this IE is present with the values of the sub IEs as stated below. Else, this IE and the sub-IEs are omitted.
ıl	- m Message authentication code	SS calculates the value of MAC-I for this message and
۱	miniossage admicritication code	writes to this IE.
	- RRC message sequence number	SS provides the value of this IE, from its internal counter.
IJ	Measurement Identity-Number	1
	Measurement Command	Setup
	Measurement Reporting Mode	
	- Measurement Report Transfer Mode	Acknowledged mode RLC
	- Measurement Reporting/Event Trigger Reporting Mode	Event Trigger
	Additional measurement list	
	- CHOICE Measurement type	
	- Intra-frequency measurement	
	- Intra-frequency cell info	
J	- New intra-frequency cell	
	 Intra-frequency cell-id cell info 	0
	- Cell info - Cell individual offset	0dB
	- Reference time difference to cell	Not Present
	- Primary CPICH info	
	- Primary scrambling code	150
	- Primary CPICH Tx power	Not Present
	- Read SFN number	FALSE
	 TX Diversity indicator Intra-frequency measurement quantity 	FALSE
	- Filter coefficient	0
	- Measurement quantity	CPICH RSCP
	- Intra-frequency reporting quantity	
	- Reporting quantities for active set cells	
IJ	- SFN-SFN observed time difference reporting	No report
$\ $	indicator - Cell synchronisation information reporting	FALSE
$\ $	indicator	IALUL
II	- Cell Identity reporting indicator	TRUE
II	 CPICH Ec/N0 reporting indicator 	FALSE
$\ $	- CPICH RSCP reporting indicator	TRUE
$\ $	- Pathloss reporting indicator	FALSE
1	- CFN-SFN observed time difference	FALSE
ıl	 Reporting quantities for monitored cells SFN-SFN observed time difference reporting 	No report
	indicator	The report
I	- Cell Identity reporting indicator	TRUE
I	 CPICH Ec/No reporting indicator 	FALSE
IJ	- CPICH RSCP reporting indicator	TRUE
II	- Pathloss <u>reporting indicator</u> - CFN-CFN observed time difference	FALSE
I	- CFN-CFN observed time difference - Reporting quantities for detected set cells	FALSE Not Present
	- Reporting quantities for detected set cells - Reporting cell status	NOTE I GOGIIL
IJ	- CHOICE report <u>ed</u> ing cell	Within-Report cell within active set and/or monitored cells
IJ	· _ ~	on used frequency and within monitored cells on non-
II		used frequency
IJ	- Maximum number of report <u>eding</u> cells type 2	Not Dropper
ıl	 Measurement validity Periodical CHOICE reporting criteria 	Not Present
۱	- Periodical <u>CHOICE</u> reporting criteria - Amount of reporting	Periodic reporting criteria Infinity
	- Reporting interval	64 sec
I	- Inter- system-RAT measurement	Not Present
I	- LCS - <u>UP</u> measurement	Not Present
	- Traffic Volume measurement	Not Present
L	- Quality measurement	Not Present

 UE internal measurement 	Not Present
CHOICE Measurement type	Intra-frequency measurement
- Intra-frequency cell info list	Not Present
 Intra-frequency measurement quantity 	Not Present
 Intra-frequency cell reporting quantity 	Not Present
- Report cell status	Not Present
- Measurement validity	Not Present
- CHOICE report criteria	No reporting
DPCH Compressed mode status info	Not Present

Contents of MEASUREMENT CONTROL FAILURE message: AM

Information Element	Value/remark
Message Type	
RRC transaction identifier	Checked to see if it's set to the identical value for the same IE in the downlink MEASUREMENT CONTROL
	<u>message</u>
Integrity check info	The presence of this IE is dependent on IXIT statements
	in TS 34.123-2. If integrity protection is indicated to be
	active, this IE shall be present with the values of the sub
	IEs as stated below. Else, this IE and the sub-IEs shall be
	absent.
- Message authentication code	This IE is checked to see if it is present. The value is
	compared against the XMAC-I value computed by SS.
- RRC Message sequence number	This IE is checked to see if it is present. The value is
	used by SS to compute the XMAC-I value.
Failure cause	See the test content

Contents of MEASUREMENT REPORT message: AM or UM

Information Element	Value/remark
Message Type	
Integrity check info	The presence of this IE is dependent on IXIT statements in TS 34.123-2. If integrity protection is indicated to be active, this IE shall be present with the values of the sub IEs as stated below. Else, this IE and the sub-IEs shall be absent.
- Message authentication code	This IE is checked to see if it is present. The value is compared against the XMAC-I value computed by SS.
- RRC Message sequence number	This IE is checked to see if it is present. The value is used by SS to compute the XMAC-I value.
Measurement identity number	1
Measured Results	
 Intra-frequency measured results 	
- Cell measured results	
- Cell Identity	0000 0000 0000 0000 0000 0000 0010B
 SFN-SFN observed time difference 	Not checked
 CFN-SFN observed time differenceCell 	Not checked
synchronisation information	
- Primary CPICH info	
- Primary scrambling code	150
- CPICH Ec/N0	Not checked
- CPICH RSCP	The presence should be checked
- Pathloss	Not checked
CN domain identity	Not checked
NAS message	Not checked
Measured results on RACH	Not checked
Additional measured results	Not checked
Event results	Not checked

Contents of PAGING TYPE 1 message: TM (SMS in CS)

Information Element	Value/remark
Message Type	
Paging record	
- CHOICE Used paging identity	CN identity
- Paging cause	Low Priority SignallingSMS
- CN domain identity	CS domain
- CHOICE UE identity	
- IMSI (GSM-MAP)	Set to the same octeted string as in the IMSI stored in the
	USIM card
BCCH modification info	Not Present

Contents of PAGING TYPE 1 message: TM (SMS in PS)

Information Element	Value/remark
Message Type	
Paging record	
 CHOICE Used paging identity 	<u>CN identity</u>
- Paging cause	Low Priority SignallingSMS
- CN domain identity	PS domain
- CHOICE UE identity	
- IMSI <u>(GSM-MAP)</u>	Set to the same octeted string as in the IMSI stored in the
	USIM card
BCCH modification info	Not Present

Contents of PAGING TYPE 2 message: AM (Speech in CS)

Information Element	Value/remark
Message Type	
RRC transaction identifier	Arbitrarily selects an integer between 0 and 3
Integrity check info	The presence of this IE is dependent on IXIT statements
	in TS 34.123-2. If integrity protection is indicated to be
	active, this IE is present with the values of the sub IEs as
	stated below. Else, this IE and the sub-IEs are omitted.
- message authentication code	SS calculates the value of MAC-I for this message and writes to this IE.
 RRC message sequence number 	SS provides the value of this IE, from its internal counter.
Paging cause	Terminating Conversational Call
CN domain identity	CS domain
Paging rRecord tType identifier	Select the same type as in the IE "Initial UE Identity" in
	RRC CONNECTION REQUEST" message.

Arbitrarily selects an integer between 0 and 3 The presence of this IE is dependent on IXIT statements in TS 34.1 present 23-2. If integrity protection is indicated to be active, this IE is present-with the values of the sub IEs as stated below. Else, this IE and the sub-IEs are omitted. - message authentication code - message sequence number Integrity protection mode info Ciphering mode info Not Present Not Pres	Information Element	Value/remark
Integrity check into The presence of this IE is dependent on IXIT statements in TS 34.1 present. Set Integrity protection is indicated to be active, this IE is present-with the values of the sub IEs as stated below. Else, this IE and the sub-IEs are omitted. Scalculates the value of MAC-I for this message and writes to this IE. Schower U-RNTI New C-RNTI Not Present Not P	Message Type	
in TS 34.1 present 23-2. If integrity protection is indicated to be active, this IE is present-with the values of the sub IEs as stated below. Else, this IE and the sub-IEs are omitted. Sc acloulates the value of MAC-I for this message and writes to this IE. Sc acloulates the value of MAC-I for this message and writes to this IE. Sc acloulates the value of MAC-I for this message and writes to this IE. Sc acloulates the value of MAC-I for this message and writes to this IE. Sc acloulates the value of MAC-I for this message and writes to this IE. Sc provides the value of this IE, from its internal counter. Not Present		
to be active, this IE is present-with the values of the sub lEs as stated below. Else, this IE and the sub-lEs are omitted. - RRC message sequence number integrity protection mode info Ciphering mode info Activation time New U-RNTI New C-RNTI RRC State DRX-Indicator UTRAN DRX cycle length coefficient CINTAN DRX cycl	Integrity check info	
- message authentication code - RRC message sequence number Integrity protection mode info Ciphering mode info Ciphering mode info Rec State DRX-indicator UTRAN DRX cycle length coefficient CN information info URA identity NOt Present		
- message authentication code - RRC message sequence number Integrity protection mode info Cliphering mode info Activation time - Activation time - New U-RNTI - New C-RNTI - Not Present - Not Pr		
- message authentication code - RRC message sequence number integrity protection mode info Ciphering mode info Ciphering mode info Activation time New U-RNTI New C-RNTI New C-RNTI New C-RNTI New C-RNTI New C-RNTI NEW State DRX-indicator UTRAN DRX cycle length coefficient UTRAN DRX cycle length coefficient UTRAN DRX cycle length coefficient URA Identity URA Identity U-NARFCN uplink(Nu) - UARFCN uplink(Nu) - UARFCN uplink(Nu) - UARFCN downlink(Nd) Maximum allowed UL TX power Uplink DPCH power control info - DPCCH power offset - P-C Preamble - Power Control Algorithm - TFC step size - Scrambling code number - Number of DPDCH - Spreading code umber - Number of DPDCH - Spreading factor - TFCI existence - Number of Filb bit - Puncturing Limit - CHOICE-mode - Timting indicator - DPC mode - DP mode - Timting indicator - PIFC I existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPS] - TGPS - TGPS - TGPS - TGPS - TGCFN - TGSN - TGSN - TGSN - TGL1 - TGSC - TGCFN - TGSC - TGCFN - TGSC - TGCFN - TGSC - TGCFN - TGSC - TGCL1 - TGCL2 - TGD		IEs as stated below. Else, this IE and the sub-IEs are
*** *** *** *** *** *** *** *** *** **		omitted.
*** *** *** *** *** *** *** *** *** **	- message authentication code	SS calculates the value of MAC-I for this message and
Integrity protection mode info Ciphering mode info Activation time New U-RNTT New C-RNTT New C-RNT Not Present Not P	•	——————————————————————————————————————
Integrity protection mode info Ciphering mode info Activation time New U-RNTT New C-RNTT New C-RNT Not Present Not P	- RRC message sequence number	SS provides the value of this IE. from its internal counter.
Ciphering mode into Activation time New U-RNTI New C-RNTI RRC State DRX-indicator UTRAN DRX cycle length coefficient Oli Information info URA identity URA identi		
Activation time New U-RNTI New C-RNTI RC State DRA-indicator UTRAN DRX cycle length coefficient URA identity URA		
New U-RNTI NRC State DRX-indicator UTRAND RX cycle length coefficient CN information info URA identity Not Present Reference to Ts34.108 clause 6.10 Parameter Set Reference to Ts34.108 c		
Now C-RNTI RRC State DRX-indicator UTRAN DRX cycle length coefficient CN information info URA identify RB with PDCP information Frequency info - UARFCN uplink(Nu) - UARFCN downlink(Nd) Maximum allowed UL TX power Uplink DPCH power control info - DPCCH power offset - PC Preamble - Power Control Algorithm - TPC step size - Scrambling code number - Number of DPDCH - spreading factor - TFCI existence - Number of FBI bit - Downlink DPCH information Downlink information common for all radio links - Downlink DPCH prode - DI rate matching restriction information - DPC mode - DI rate matching restriction information - DPC mode - DI rate matching restriction information - DPC mode - TTPC sistence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TGPC - TGCFN - TGSN - TGL1 - TGL2 - TGD		
RRC State DRX-indicator UTRAN DRX cycle length coefficient ON information info URA identity RR with PDCP information Frequency info		
UTRAN DRX cycle length coefficient Chi information info URA identity RB with PDCP information Frequency info - UARFCN downlink(Nd) Asximum allowed UL TX power Uplink DPCH power control info - DPCCH power offset - PC Preamble - Power Control Algorithm - TTC step size - Scrambling code number - Number of DPDCH - spreading factor - TFCI existence - Number of FBI bit - Puncturing Limit CHOICE Mode - Downlink PDSCH information Downlink information common for all RL - CHOICE mode - Timing indicator - CFN-targetSFN frame offset - Downlink DPCH hor common for all RL - CHOICE mode - Timing indicator - Fixed or Flexible Position - TFCI existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TGPSI - TGPSI - TGPC - TGGFN - TGRPC - TGCFN - TGSN - TGSN - TGSL - TGL1 - TGL2 - TGD		
CN information into UNRA identity RB with PDCP information Frequency info		
URA identity B with PDCP information Frequency info - UARFCN uplink(Nu) - UARFCN downlink(Nd) Maximum allowed UL TX power Uplink DPCH pinfor - Uplink DPCH power control info - DPCCH power offset - PC Preamble - PC Preamble - Power Control Algorithm - TPC step size - Scrambling code type - Scrambling code number - Number of DPDCH - spreading factor - TFC! existence - Number of FBI bit - Puncturing Limit CHOICE Mode - Timing indicator - CFN-LategelSPN frame offset - Downlink DPCH power control information - DPC mode - Timing indicator - CFN-LategelSPN frame offset - Downlink DPCH power control information - Spreading factor - TFC! existence - Number of FISI bit Pilot bits (SF=128,256) - DPCH compressed mode info - TGPS! - TGPS - TGPC - TGPR - T		
RB with PDCP Information Frequency info		
Frequency info		
- UARFCN downlink(Nd) Maximum allowed UL TX power Uplink DPCH power control info - Uplink DPCH power control info - Uplink DPCH power control info - DPCCH power offset - PC Preamble - Power Control Algorithm - TPC step size - Scrambling code type - Scrambling code number - Number of DPDCH - spreading factor - TFC lexistence - Number of FBI bit - Puncturing Limit CHOICE Mode - Downlink DPCH information Downlink information common for all radio links - Downlink DPCH price common for all RL - CHOICE mode - Diming indicator - CFN-targetSFN frame offset - Downlink DPCH power control information - DPC mode - DL rate matching restriction information - Spreading factor - Fixed or Flexible Position - TFCI existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TGPSI - TGPR - TGBFN - TGBFN - TGBFN - TGBR - TGBR - TGBC - TGCFN - TGSN - TGL1 - TGL2 - TGD - TGCFN - TGL1 - TGL2 - TGD - TGCFN - TGL3 - TGL1 - TGL2 - TGD - Uplink DPCH power control information - Uplink DPCH power control information - TGCFN - TGSN - TGL1 - TGL2 - TGD - TGCFN - TGL3 - TGL1 - TGL2 - TGD - TGCFN - TGL3 - TGL5 - TGL5 - TGD - TGCFN - TGL5 - TGCFN - TGL1 - TGL2 - TGD - TGCFN - TGL5 - TGCFN - TGL5 - TGCFN - TGL5 - TGCFN - TGL5 - TGCFN - TGL5 - TGCFN - T		INOUT TESETIL
- UARFCN downlink(Nd) Maximum allowed UL TX power Uplink DPCH info - Uplink DPCH power control info - DPCCH power offset - PC Preamble - Power Control Algorithm - TPC step size - Scrambling code type - Scrambling code number - Number of DPDCH - spreading factor - TFCI existence - Number of FBI bit - Puncturing Limit CHOICE Mode - Downlink PDSCH information Downlink information common for all radio links - Downlink DPCH power control information - DPC mode - Timing indicator - CFN-targetiSFN frame offset - Downlink DPCH power control information - DPC mode - DL rate matching restriction information - DPC mode - DL rate matching restriction information - Spreading factor - Fixed or Flexible Position - TFCI existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TGCFN - Transmission gap pattern sequence configuration parameters - TGMP - TOPRC - TGSN - TGL1 - TGL2 - TGD - TGD - TGCFN - TGD - TGCFN - TGSN - TGL1 - TGL2 - TGD - TGD - TGCFN - TGD - TGCFN - TGD - TGCFN - TGD - TGCFN - TGSN - TGL1 - TGL2 - TGD - TGD - TGD - TGCFN - TGD - TGCFN - TGD - TGCFN - TGD - TGCFN - TGCFN - TGSN - TGL1 - TGL2 - TGD - TGD - TGD - TGCFN - TGD - TGCFN - TGD - TGCFN - TGC		Potoronoo to TS34 109 alauga 6 10 Parameter Cat
Maximum allowed UL TX power Uplink DPCH info - Uplink DPCH power control info - DPCCH power offset - PC Preamble - Power Control Algorithm - TPC step size - Scrambling code type - Scrambling code number - Number of DPDCH - spreading factor - TFCI existence - Number of FBI bit - Downlink PDSCH information Downlink information common for all RL - CHOICE mode - Downlink DPCH power control information - DPC mode - Diming indicator - CFN-targetSFN frame offset - Downlink DPCH power control information - DPC mode - DL rate matching restriction information - DPC mode - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TGEFN - Transmission gap pattern sequence configuration parameters - TGMP - TGPRC - TGCFN - TGSN - TGL1 - TGL2 - TGD - TGDE - TGCS - TGCL1 - TGDS - TGCL1 - TGDS - TGDS - TGCL1 - TGDS - TGCL1 - TGDS - TGCL2 - TGCC - TGCCA - T		
Uplink DPCH power control info - Uplink DPCH power offset - PC Preamble - Power Control Algorithm - TPC step size - Scrambling code type - Scrambling code number - Number of DPDCH - spreading factor - TFC l existence - Number of FBI bit - Puncturing Limit CHOICE Mode - Downlink DPCH info common for all radio links - Downlink DPCH info common for all RL - CHOICE mode - Timing indicator - CFN-targetSFN frame offset - Downlink DPCH power control information - DPC mode - DL rate matching restriction information - DPC mode - DL rate matching restriction information - Spreading factor - Fixed or Flexible Position - TFC existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TGPSI - TGBPS - TGBPS - TGBPR - TGBP		
- Uplink DPCH power control info - DPCCH power offset - PC Preamble - Power Control Algorithm - TPC step size - Scrambling code type - Scrambling code number - Number of DPDCH - spreading factor - TFCl existence - Number of BPCH information Downlink information common for all radio links - Downlink DPCH info common for all RL - CHOICE mode - Timing indicator - CFN-targetSFN frame offset - Downlink DPCH power control information - DPC mode - DL rate matching restriction information - Spreading factor - TFCl existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TTGPS - TGRS - TGMP - TGRRC - TGSN - TGL1 - TGL2 - TGD - TTGD - TTGD - TTGD - TTGD - TTGD - TTGD - TGCFN - TGSN - TGL1 - TGD - TGCFN - TGD - TGCFN - TGCFN - TGD - TGCFN -		330BM
- DPCCH power offset - PC Preamble - Power Control Algorithm - TPC step size - Scrambling code type - Scrambling code number - Number of DPDCH - spreading factor - TFCI existence - Number of FBI bit - Puncturing Limit - CHOICE Mode - Downlink PDSCH information Downlink information common for all radio links - Downlink DPCH info common for all RL - CHOICE mode - Timing indicator - CFN-targetSFN frame offset - Downlink DPCH power control information - DPC mode - DL rate matching restriction information - Spreading factor - TFCI existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGCFN - TGGFN - TGGFN - TGGFN - TGGFN - TGGFN - TGGFN - TGGPR - TGGFN - TGGN - TGCFN - TGSN - TGL1 - TGL2 - TGD - TGD - TGCPC - TGCFN - TGD - TGCPC - TGCFN - TGSN - TGL1 - TGL2 - TGD - TGCPC - TGCFN - TGCPC		
- PC Preamble - Power Control Algorithm - TPC step size - Scrambling code type - Scrambling code number - Number of DPDCH - spreading factor - TFCI existence - Number of FBI bit - Puncturing Limit CHOICE Mode - Downlink DPSCH information Downlink information common for all radio links - Downlink DPCH grower control information - DPC mode - Timing indicator - CFN-targetSFN frame offset - Downlink DPCH power control information - DPC mode - DL rate matching restriction information - Spreading factor - Fixed or Flexible Position - TFCI existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - Treasmission gap pattern sequence configuration parameters - TGMP - TGSN - TGSN - TGL1 - TGL2 - TGD - TGSN - TGL1 - TGL2 - TGD - TGSN - TGL1 - TGL2 - TGD -		2.15
- Power Control Algorithm - TPC step size - Scrambling code type - Scrambling code number - Number of DPDCH - spreading factor - TFCI existence - Number of FBI bit - Puncturing Limit CHOICE Mode - Downlink PDCH information Downlink information common for all radio links - Downlink DPCH info common for all RL - CHOICE mode - Timing indicator - CFN-targetSFN frame offset - Downlink DPCH power control information - DPC mode - DL rate matching restriction information - Spreading factor - Fixed or Flexible Position - TFCI existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TGGFN - Transmission gap pattern sequence configuration parameters - TGMP - TGSN - TGSN - TGL1 - TGL2 - TGD - Strambling code type - O (0 to 16777215) Not Present(1) Se is reference to TS34.108 clause 6.10 Parameter Set FDD Not Present Not Present O (single) Not Present O (single) Not Present Not Present Not Present PDD Maintain Not Present Not Present Not Present O (single) Not Present O (current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256		
- TFC step size - Scrambling code type - Scrambling code number - Number of DPDCH - spreading factor - TFCI existence - Number of FBI bit - Puncturing Limit CHOICE Mode - Downlink DPSCH information Downlink information common for all radio links - Downlink DPCH info common for all RL - CHOICE mode - Timing indicator - CFN-targetSFN frame offset - Downlink DPCH power control information - DPC mode - DL rate matching restriction information - Spreading factor - Fixed or Flexible Position - TFCI existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TGSPS - TGMP - TGSPN - TGSN - TGL1 - TGL1 - TGL2 - TGD		
- Scrambling code type - Scrambling code number - Number of DPDCH - spreading factor - TFCI existence - Number of FBI bit - Puncturing Limit CHOICE Mode - Downlink PDSCH information Downlink information common for all radio links - Downlink DPCH info common for all RL - CHOICE mode - Timing indicator - CFN-targetSFN frame offset - Downlink DPCH power control information - DPC mode - DL rate matching restriction information - Spreading factor - Fixed or Flexible Position - TFCI existence - Number of bits for Pilot bits (SF=128,256) - DPCH compressed mode info - TGPSI - TGGFN - TGGRC - TGGFN - TGSN - TGL1 - TGL2 - TGD Long 0 (0 to 16777215) Not Present(1) SF is reference to TS34.108 clause 6.10 Parameter Set TRUE Not Present 0 (single) Not Present 0 (single) Not Present 1 (current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 10 - TGSN - TGL1 - TGL2 - TGD - TGDF - TGSN - TGDF - TG		
- Scrambling code number - Number of DPDCH - spreading factor - TFCI existence - Number of FBI bit - Puncturing Limit CHOICE Mode - Downlink DPSCH information Downlink information common for all RL - CHOICE mode - Timing indicator - CFN-targetSFN frame offset - Downlink DPCH power control information - DPC mode - DL rate matching restriction information - Spreading factor - Fixed or Flexible Position - TFCI existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TGPSI - TGRPS - TGRPC - TGCFN - TGSN - TGSN - TGL1 - TGL2 - TGD 0 (5 16777215) Not Present(1) Reference to TS34.108 clause 6.10 Parameter Set FDD Not Present 0 (5 ingle) Not Present 0 (single) Not Present - O (single) Not Present - O (single) Not Present - TGSA-108 clause 6.10 Parameter Set FDD Not Present - TGSA-108 clause 6.10 Parameter Set FDD Not Present - TGSA-108 clause 6.10 Parameter Set FDD Not Present - TGSA-108 clause 6.10 Parameter Set FDD Not Present		
- Number of DPDCH - spreading factor - TFCI existence - Number of FBI bit - Puncturing Limit CHOICE Mode - Downlink PDSCH information Downlink information common for all radio links - Downlink DPCH info common for all RL - CHOICE mode - Timing indicator - CFN-targetSFN frame offset - Downlink DPCH power control information - DPC mode - DL rate matching restriction information - Spreading factor - Fixed or Flexible Position - TFCI existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TROS status Flag - TGCFN - Transmission gap pattern sequence configuration parameters - TGMP - TGPRC - TGSN - TGL1 - TGL2 - TGD		
- spreading factor - TFCI existence - Number of FBI bit - Puncturing Limit CHOICE Mode - Downlink PDSCH information Downlink information common for all radio links - Downlink DPCH info common for all RL - CHOICE mode - Timing indicator - CFN-targetSFN frame offset - Downlink DPCH power control information - DPC mode - DL rate matching restriction information - Spreading factor - Fixed or Flexible Position - TFCI existence - Number of FBI bit - TGPSI - TGBSI - TGGFN - TGSN - TGSN - TGL - TGCFN - TGSN - TGL - TGCL - Number of FBI bit - Not Present - TS34.108 clause 6.10 Parameter Set - FDD Not Present - Not Present - FDD Maintain Not Present - O (single) Not Present - Not Present - TGS4.108 clause 6.10 Parameter Set - FDD Maintain - Not Present - TGS4.108 clause 6.10 Parameter Set - FDD - Maintain - Not Present - Not Present - TGS4.108 clause 6.10 Parameter Set - FDD - Maintain - Not Present - Not Present - Not Present - TGS4.108 clause 6.10 Parameter Set - FDD - Maintain - Not Present - Not Prese		
TRUE Not Present(0) Reference to TS34.108 clause 6.10 Parameter Set FDD Not Present CHOICE Mode Downlink PDSCH information Downlink pDSCH information common for all radio links Downlink pDSCH info common for all RL CHOICE mode Timing indicator CFN-targetSFN frame offset Downlink DPCH power control information DPC mode DL rate matching restriction information Spreading factor Fixed or Flexible Position TFCI existence Number of bits for Pilot bits(SF=128,256) DPCH compressed mode info TGPSI TPGS status Flag TGCFN Transmission gap pattern sequence configuration parameters TGMP TGGFN TGSN TGL	- Number of DPDCH	
- Number of FBI bit - Puncturing Limit CHOICE Mode - Downlink PDSCH information Downlink information common for all radio links - Downlink DPCH info common for all RL - CHOICE mode - Timing indicator - CFN-targetSFN frame offset - Downlink DPCH power control information - DPC mode - DL rate matching restriction information - Spreading factor - Fixed or Flexible Position - TFCI existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TPGS status Flag - TGCFN - Transmission gap pattern sequence configuration parameters - TGMP - TGPRC - TGCFN - TGSN - TGL1 - TGL2 - TGD Not Present Not Present Not Present Not Present Very Course to TS34.108 clause 6.10 Parameter Set FDD Maintain Not Present - FDD Maintain Not Present 1 (single) Not Present (current CTN + (256 – TTI/10msec)) mod 256 - TGCFN - TGSN - TGCFN - TGSN - TGL1 - TGL2 - TGD - TGCFN - TGSN - TGSN - TGL2 - TGD - TGCFN - TGSN - TGCFN - TGSN - TGCL2 - TGD - TGD - TGD - TGD - TGCFN - TGCL2 - TGD - TGCFN - TGD - TGCFN - TGCL2 - TGD - TGCFN - TGD - TGCFN - TGCFN - TGCL2 - TGD - TGCFN - TGD - TGCFN - TGCL2 - TGD - TGCFN - TGCL2 - TGD - TGCL3 - TGD - TGCL3 - TGCL3 - TGCL4 - TGCL5 - TGCL5 - TGCL6 - TGCCCC - TGCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCC	- spreading factor	SF is reference to TS34.108 clause 6.10 Parameter Set
- Puncturing Limit CHOICE Mode - Downlink PDSCH information Downlink information common for all radio links - Downlink DPCH info common for all RL - CHOICE mode - Timing indicator - CFN-targetSFN frame offset - Downlink DPCH power control information - DPC mode - DL rate matching restriction information - Spreading factor - Fixed or Flexible Position - TFCI existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TGSRS - TGMP - TGPRC - TGCFN - TGSN - TGSN - TGL1 - TGL2 - TGD	- TFCI existence	TRUE
CHOICE Mode - Downlink PDSCH information Downlink information common for all radio links - Downlink DPCH info common for all RL - CHOICE mode - Timing indicator - CFN-targetSFN frame offset - Downlink DPCH power control information - DPC mode - DL rate matching restriction information - Spreading factor - Fixed or Flexible Position - TFCI existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TGGFN - Transmission gap pattern sequence configuration parameters - TGMP - TGCFN - TGSN - TGL1 - TGL2 - TGD Dwit Present FDD Maintain Not Present O (single) Not Present Reference to TS34.108 clause 6.10 Parameter Set Flexible TRUE Not Present 1 inactive (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256	- Number of FBI bit	
- Downlink PDSCH information Downlink information common for all radio links - Downlink DPCH info common for all RL - CHOICE mode - Timing indicator - CFN-targetSFN frame offset - Downlink DPCH power control information - DPC mode - DL rate matching restriction information - Spreading factor - Fixed or Flexible Position - TFCI existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TGGFN - Transmission gap pattern sequence configuration parameters - TGMP - TGPRC - TGCFN - TGSN - TGL1 - TGL2 - TGD Not Present Maintain Not Present O (single) Not Present Not Present 1 (single) Not Present Not Present C (single) Not Present O (single) Not Present I (single) Not Present O (single) Not Present O (single) Not Present Reference to TS34.108 clause 6.10 Parameter Set Flexible TRUE Not Present O (current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256	- Puncturing Limit	Reference to TS34.108 clause 6.10 Parameter Set
Downlink information common for all radio links - Downlink DPCH info common for all RL	CHOICE Mode	FDD
- Downlink DPCH info common for all RL - CHOICE mode - Timing indicator - CFN-targetSFN frame offset - Downlink DPCH power control information - DPC mode - DL rate matching restriction information - Spreading factor - Fixed or Flexible Position - TFCI existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TPGS status Flag - TGCFN - Transmission gap pattern sequence configuration parameters - TGMP - TGCFN - TGSN - TGL1 - TGL2 - TGD - TIGL2 - TGD - TIGL2 - TGD - TIMING Maintain Not Present - Not Present - Not Present - TS34.108 clause 6.10 Parameter Set - FLexible - TRUE - Not Present - TGSA - Inactive - (Current CFN + (256 – TTI/10msec)) mod 256 - TTI/10msec)) mod 256	- Downlink PDSCH information	Not Present
- CHOICE mode - Timing indicator - CFN-targetSFN frame offset - Downlink DPCH power control information - DPC mode - DL rate matching restriction information - Spreading factor - Fixed or Flexible Position - TFCI existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TPGS status Flag - TGCFN - Transmission gap pattern sequence configuration parameters - TGMP - TGPRC - TGSN - TGL1 - TGL2 - TGD - TROB - TGL2 - TGD - TIME - Maintain - Not Present - 0 (single) - Not Present - Reference to TS34.108 clause 6.10 Parameter Set - Flexible - TRUE - Not Present - TGS4.108 clause 6.10 Parameter Set -	Downlink information common for all radio links	
- CHOICE mode - Timing indicator - CFN-targetSFN frame offset - Downlink DPCH power control information - DPC mode - DL rate matching restriction information - Spreading factor - Fixed or Flexible Position - TFCI existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TPGS status Flag - TGCFN - Transmission gap pattern sequence configuration parameters - TGMP - TGPRC - TGSN - TGL1 - TGL2 - TGD - TROB - TGL2 - TGD - TIME - Maintain - Not Present - 0 (single) - Not Present - Reference to TS34.108 clause 6.10 Parameter Set - Flexible - TRUE - Not Present - TGS4.108 clause 6.10 Parameter Set -		
- Timing indicator - CFN-targetSFN frame offset - Downlink DPCH power control information - DPC mode - DL rate matching restriction information - Spreading factor - Fixed or Flexible Position - TFCI existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TPGS status Flag - TGCFN - Transmission gap pattern sequence configuration parameters - TGMP - TGPRC - TGSN - TGL1 - TGL2 - TGD - TGL2 - TGD - TGD - TGD - TGL2 - TGD - TGL2 - TGD - TGL2 - TGD - TGD - TGL2 - TGD - TGL2 - TGD - TGR - TGL2 - TGD - TGR - TGL2 - TGD - TGR - TGR - TGR - TGL2 - TGD - TGR - TGR - TGL2 - TGD - TGR - TGR - TGR - TGR - TGL2 - TGD - TGR - TGL2 - TGD - TGR		FDD
- CFN-targetSFN frame offset - Downlink DPCH power control information - DPC mode - DL rate matching restriction information - Spreading factor - Fixed or Flexible Position - TFCI existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TPGS status Flag - TGCFN - Transmission gap pattern sequence configuration parameters - TGMP - TGPRC - TGSN - TGSN - TGL1 - TGL2 - TGD - TGD - DV Present 0 (single) Not Present Reference to TS34.108 clause 6.10 Parameter Set Flexible TRUE Not Present 1 inactive (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 - TGL1 - TGL2 - TGD		
- Downlink DPCH power control information - DPC mode - DL rate matching restriction information - Spreading factor - Fixed or Flexible Position - TFCI existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TPGS status Flag - TGCFN - Transmission gap pattern sequence configuration parameters - TGMP - TGPRC - TGCFN - TGSN - TGL1 - TGL2 - TGD - DPCH power control information 0 (single) Not Present Reference to TS34.108 clause 6.10 Parameter Set Flexible TRUE Not Present - TGUE (Current CFN + (256 – TTI/10msec)) mod 256 - TDD Measurement - CQUITENT CFN + (256 – TTI/10msec)) mod 256 - TGL1 - TGL2 - TGD - T		
- DPC mode - DL rate matching restriction information - Spreading factor - Fixed or Flexible Position - TFCI existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TPGS status Flag - TGCFN - Transmission gap pattern sequence configuration parameters - TGMP - TGPRC - TGSN - TGL1 - TGL2 - TGD O (single) Not Present Reference to TS34.108 clause 6.10 Parameter Set Flexible TRUE Not Present (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 8 10 10 10 10 15		<u></u>
- DL rate matching restriction information - Spreading factor - Fixed or Flexible Position - TFCI existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TPGS status Flag - TGCFN - Transmission gap pattern sequence configuration parameters - TGMP - TGPRC - TGCFN - TGSN - TGL1 - TGL2 - TGD Not Present Reference to TS34.108 clause 6.10 Parameter Set Flexible TRUE Not Present		(single)
- Spreading factor - Fixed or Flexible Position - TFCI existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TPGS status Flag - TGCFN - Transmission gap pattern sequence configuration parameters - TGMP - TGPRC - TGCFN - TGSN - TGL1 - TGL2 - TGD Reference to TS34.108 clause 6.10 Parameter Set Flexible TRUE Not Present 1 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 8 (Current CFN + (256 – TTI/10msec)) mod 256 10 11 12 13 14 15 15 15 15 15 15 16 17 18 18 19 19 19 19 19 19 19 19		
- Fixed or Flexible Position - TFCI existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TPGS status Flag - TGCFN - Transmission gap pattern sequence configuration parameters - TGMP - TGPRC - TGSN - TGL1 - TGL2 - TGD - TFCI existence TRUE Not Present (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 8 10 5 10 5 15		
- TFCI existence - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TPGS status Flag - TGCFN - Transmission gap pattern sequence configuration parameters - TGMP - TGPRC - TGSN - TGL1 - TGL2 - TGD - TRUE Not Present TRUE Not Present TRUE Not Present 1 (Current CFN + (256 – TTI/10msec)) mod 256 FDD Measurement 62 (Current CFN + (256 – TTI/10msec)) mod 256 (Current CFN + (256 – TTI/10msec)) mod 256 8 1 1 1 1 1 1 1 1 1 1 1 1		
- Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TPGS status Flag - TGCFN - Transmission gap pattern sequence configuration parameters - TGMP - TGPRC - TGSN - TGSN - TGL1 - TGL2 - TGD - TGD - TGD - TGD - TGD - TGCFN - TGD - TGSN - TGL1 - TGL2 - TGD - TGCFN - TGL2 - TGD - TGCFN - TGD -		
- DPCH compressed mode info - TGPSI - TPGS status Flag - TGCFN - Transmission gap pattern sequence configuration parameters - TGMP - TGCFN - TGCFN - TGCFN - TGSN - TGL1 - TGL2 - TGD - TGL2 - TGD - T		1110
- TGPSI - TPGS status Flag - TGCFN - Transmission gap pattern sequence configuration parameters - TGMP - TGCFN - TGCFN - TGPRC - TGCFN - TGSN - TGL1 - TGL2 - TGD - TGD - TGD - TGD - TGD - TGC - TTI/10msec)) mod 256 - TGC -		INOUTIESEIIL
- TPGS status Flag - TGCFN - Transmission gap pattern sequence configuration parameters - TGMP - TGCFN - TGCFN - TGSN - TGL1 - TGL2 - TGD - TGD - TGD - TGD - TGD - TGL2 - TGD - TG		1
- TGCFN - Transmission gap pattern sequence configuration parameters - TGMP - TGPRC - TGCFN - TGSN - TGL1 - TGL2 - TGD - TGD (Current CFN + (256 – TTI/10msec)) mod 256 100 5 15		
- Transmission gap pattern sequence configuration parameters - TGMP - TGPRC - TGCFN - TGSN - TGL1 - TGL2 - TGD - TGD - TGD - TGL2 - TGD -	- TPGS status Flag	inactive
- Transmission gap pattern sequence configuration parameters - TGMP - TGPRC - TGCFN - TGSN - TGL1 - TGL2 - TGD - TGD - TGD - TGL2 - TGD -	- TGCFN	(Current CFN + (256 - TTI/10msec)) mod 256
parameters TGMP	- Transmission gap pattern sequence configuration	
- TGMP - TGPRC - TGCFN - TGSN - TGL1 - TGL2 - TGD - TGD - TGD - TGD - TGD - TGL2 - TGD - T	- · · · · · · · · · · · · · · · · · · ·	
- TGPRC 62 - TGCFN (Current CFN + (256 - TTI/10msec)) mod 256 - TGSN 8 - TGL1 10 - TGL2 5 - TGD 15		FDD Measurement
- TGCFN - TGSN - TGL1 - TGL2 - TGD (Current CFN + (256 - TTI/10msec)) mod 256 8 10 5 15		
- TGSN 8 10 10 - TGL2 5 - TGD 15		
TGL1		, , , , , , , , , , , , , , , , , , , ,
- TGL2 5 - TGD 15		
TGD 15		

- TGPL2 35 - RPP Mode 1 _- ITPITP Mode 1 _- UL/DL Mode DL _- Downlink compressed mode method SF/2 - Uplink compressed mode method Not Present - Downlink frame type _- DeltaSIR1 2.0 _- DeltaSIRafter1 1.0 - DeltaSIR2 Not Present - DeltaSIRafter2 Not Present - TX Diversity mode None - SSDT information Not Present - Default DPCH Offset Value Downlink information for each radio links - Primary CPICH info - Primary scrambling code 100 - PDSCH with SHO DCH info Not Present Not Present - PDSCH code mapping - Downlink DPCH info for each RL - Primary CPICH usage for channel estimation Primary CPICH may be used - DPCH frame offset 0 chips - Secondary CPICH info Not Present - DL channelisation code - Secondary scrambling code Reference to TS34.108 clause 6.10 Parameter Set - Spreading factor SF-1(SF is reference to TS34.108 clause 6.10 Parameter - Code number - Scrambling code change No change - TPC combination index 0 - SSDT Cell Identity -a - Closed loop timing adjustment mode Not Present - Secondary CCPCH info Not Present - TFCS Not Present Not Present - FACH/PCH information - References to system information blocks Not Present

Scheduling information

Information Element	Value/remark
Message Type	Value/Tellial N
RRC transaction identfier	Arbitrarily selects an integer between 0 and 3
Integrity check info	The presence of this IE is dependent on IXIT statements
and girly chock and	in TS 34.123-2. If integrity protection is indicated to be
	active, this IE is present with the values of the sub IEs as
	stated below. Else, this IE and the sub-IEs are omitted.
- message authentication code	SS calculates the value of MAC-I for this message and
g	writes to this IE.
- RRC message sequence number	SS provides the value of this IE, from its internal counter.
Integrity protection mode info	Not Present
Ciphering mode info	Not Present
Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
New U-RNTI	Not Present
New C-RNTI	Not Present
RRC State DRX indicator	NoDRXCELL_DCH
UTRAN DRX cycle length coefficient	Not Present
CN information info	Not Present
URA identity	Not Present
RB with PDCP information	Not Present
Frequency info	Deference to TOO 4 400 sleves 0 40 D
- UARFCN uplink(Nu)	Reference to TS34.108 clause 6.10 Parameter Set
- UARFCN downlink(Nd) Maximum allowed UL TX power	Reference to TS34.108 clause 6.10 Parameter Set
Uplink DPCH info	JJUDIII
- Uplink DPCH into	
- DPCCH power offset	-6dB
- PC Preamble	8slot
- Power Control Algorithm	Algorithm1
- TPC step size	1dB
- Scrambling code type	Long
- Scrambling code number	0 (0 to 16777215)
- Number of DPDCH	Not Present(1)
- spreading factor	SF is reference to TS34.108 clause 6.10 Parameter Set
- TFCI existence	TRUE
- Number of FBI bit	Not Present(0)
- Puncturing Limit	Reference to TS34.108 clause 6.10 Parameter Set
CHOICE Mode	FDD
- Downlink PDSCH information Downlink information common for all radio links	Not Present
- Downlink DPCH info common for all RL	
- DOWNINK DPCH INIO COMMON TOT All RE - CHOICE mode	FDD
- Timing indicator	Maintain
- CFN-targetSFN frame offset	Not Present
- Downlink DPCH power control information	
- DPC mode	0 (single)
- DL rate matching restriction information	Not Present
- Spreading factor	Reference to TS34.108 clause 6.10 Parameter Set
- Fixed or Flexible Position	Flexible
- TFCI existence	FALSE
 Number of bits for Pilot bits(SF=128,256) 	Not Present
- DPCH compressed mode info	
- TGPSI	1
TGPS Status Flag	inactive
TGCFN	(Current CFN + (256 - TTI/10msec)) mod 256
- Transmission gap pattern sequence configuration	
parameters	
TGMP	FDD Measurement
- TGPRC	62
	(Current CFN + (256 - TTI/10msec)) mod 256
TGSN	8
TGL1	10
TGL2	5
TGD	15
- TGPL1 - TGPL2	35 35
- IUFLZ	JU

- RPP Mode 1 - ITP _- UL/DL Mode DL _- Downlink compressed mode method SF/2 - Uplink compressed mode method Not Present _- Downlink frame type _- DeltaSIR1 2.0 _- DeltaSIRafter1 1.0 - DeltaSIR2 Not Present Not Present _- DeltaSIRafter2 - TX Diversity mode None - SSDT information Not Present - Default DPCH Offset Value Downlink information for each radio links - Primary CPICH info - Primary scrambling code 100 - PDSCH with SHO DCH info Not Present - PDSCH code mapping Not Present - Downlink DPCH info for each RL - Primary CPICH usage for channel estimation Primary CPICH may be used - DPCH frame offset 0 chips - Secondary CPICH info Not Present - DL channelisation code - Secondary scrambling code - Spreading factor Reference to TS34.108 clause 6.10 Parameter Set - Code number SF-1(SF is reference to TS34.108 clause 6.10 Parameter Set) - Scrambling code change No change - TPC combination index 0 - SSDT Cell Identity -a - Closed loop timing adjustment mode Not Present - Secondary CCPCH info Not Present Not Present - TFCS - FACH/PCH information Not Present - References to system information blocks Not Present

- Scheduling information

Information Element	Value/remark
	value/remark
Message Type RRC transaction identifier	Arbitrarily selects an integer between 0 and 3
Integrity check info	The presence of this IE is dependent on IXIT statements in
intogrity official fills	TS 34.123-2. If integrity protection is indicated to be active,
	this IE is present with the values of the sub IEs as stated
	below. Else, this IE and the sub-IEs are omitted.
- message authentication code	SS calculates the value of MAC-I for this message and
, and the second	writes to this IE.
- RRC message sequence number	SS provides the value of this IE, from its internal counter.
Integrity protection mode info	Not Present
Ciphering mode info	Not Present
Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
New U-RNTI	Not Present
New C-RNTI	Not Present
RRC State DRX indicator	Not Present
UTRAN DRX cycle length coefficient CN information info	Not Present Not Present
URA identity	Not Present
RB with PDCP information	Not Present
Frequency info	
- UARFCN uplink(Nu)	Reference to TS34.108 clause 6.10 Parameter Set
- UARFCN downlink(Nd)	Reference to TS34.108 clause 6.10 Parameter Set
Maximum allowed UL TX power	33dBm
Uplink DPCH info	
- Uplink DPCH power control info	
- DPCCH power offset	-6dB
- PC Preamble	8slot
- Power Control Algorithm	Algorithm1
- TPC step size - Scrambling code type	1dB Long
- Scrambling code type - Scrambling code number	0 (0 to 16777215)
- Number of DPDCH	Not Present(1)
- spreading factor	SF is reference to TS34.108 clause 6.10 Parameter Set
- TFCI existence	TRUE
- Number of FBI bit	Not Present(0)
- Puncturing Limit	Reference to TS34.108 clause 6.10 Parameter Set
CHOICE Mode	FDD
- Downlink PDSCH information	Not Present
Downlink information common for all radio links	
- Downlink DPCH info common for all RL	FDD
- Timing indicator	Maintain
- CFN-targetSFN frame offset	Not Present
- Downlink DPCH power control information	
- DPC mode	0 (single)
- DL rate matching restriction information	Not Present
- Spreading factor	Reference to TS34.108 clause 6.10 Parameter Set
- Fixed or Flexible Position	Flexible
- TFCI existence	TRUE
- Number of bits for Pilot bits(SF=128,256)	Not Present
- DPCH compressed mode info - TGPSI	1
- TGPSI - TGPS Status Flag	Inactive
- TGCFN	(Current CFN + (256 – TTI/10msec)) mod 256
- Transmission gap pattern sequence configuration	The second secon
parameters	
TGMP	FDD Measurement
- TGPRC	62
	(Current CFN + (256 - TTI/10msec)) mod 256
- TGSN	8
TGL1	10
TGL2	5
- TGD - TGPL1	15 35
<u>- IUFLI</u>	J J J

- TGPL2 35 - RPP Mode 1 _- ITP Mode 1 _- UL/DL Mode DL _- Downlink compressed mode method SF/2 - Uplink compressed mode method Not Present - Downlink frame type - DeltaSIR1 2.0 _- DeltaSIRafter1 1.0 - DeltaSIR2 Not Present - DeltaSIRafter2 Not Present - TX Diversity mode None - SSDT information Not Present - Default DPCH Offset Value Downlink information for each radio links - Primary CPICH info - Primary scrambling code 100 - PDSCH with SHO DCH info Not Present - PDSCH code mapping Not Present - Downlink DPCH info for each RL - Primary CPICH usage for channel estimation Primary CPICH may be used - DPCH frame offset 0 chips - Secondary CPICH info Not Present - DL channelisation code - Secondary scrambling code - Spreading factor Reference to TS34.108 clause 6.10 Parameter Set SF-1(SF is reference to TS34.108 clause 6.10 Parameter - Code number - Scrambling code change No change - TPC combination index 0 - SSDT Cell Identity -a - Closed loop timing adjustment mode Not Present - Secondary CCPCH info Not Present - TFCS Not Present Not Present - FACH/PCH information - References to system information blocks Not Present

Scheduling information

Information Element	Value/remark
Message Type	raidoji oliidi K
RRC transaction identfier	Arbitrarily selects an integer between 0 and 3
Integrity check info	The presence of this IE is dependent on IXIT statements
Integrity check into	in TS 34.123-2. If integrity protection is indicated to be
	active, this IE is present with the values of the sub IEs as
manage authoritication and	stated below. Else, this IE and the sub-IEs are omitted.
- message authentication code	SS calculates the value of MAC-I for this message and
DDC	writes to this IE.
- RRC message sequence number	SS provides the value of this IE, from its internal counter.
Integrity protection mode info	Not Present
Ciphering mode info	Not Present
Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
New U-RNTI	Not Present
New C-RNTI	Not Present
RRC State DRX indicator	NoDRXCELL_DCH
UTRAN DRX cycle length coefficient	Not Present
CN information info	Not Present
URA identity	Not Present
RB with PDCP information	Not Present
Frequency info	
- UARFCN uplink(Nu)	Reference to TS34.108 clause 6.10 Parameter Set
- UARFCN downlink(Nd)	Reference to TS34.108 clause 6.10 Parameter Set
Maximum allowed UL TX power	33dBm
Uplink DPCH info	
- Uplink DPCH power control info	
- DPCCH power offset	-6dB
- PC Preamble	8slot
- Power Control Algorithm	Algorithm1
- TPC step size	1dB
- Scrambling code type	Long
- Scrambling code number	0 (0 to 16777215)
- Number of DPDCH	Not Present(1)
- spreading factor	SF is reference to TS34.108 clause 6.10 Parameter Set
- TFCI existence	TRUE
- Number of FBI bit	Not Present(0)
- Puncturing Limit	Reference to TS34.108 clause 6.10 Parameter Set
CHOICE Mode	FDD
- Downlink PDSCH information	Not Present
Downlink information common for all radio links	
- Downlink DPCH info common for all RL	500
	FDD
- Timing indicator	Maintain
- CFN-targetSFN frame offset	Not Present
- Downlink DPCH power control information	O (circula)
- DPC mode	0 (single)
- DL rate matching restriction information	Not Present
- Spreading factor	Reference to TS34.108 clause 6.10 Parameter Set
- Fixed or Flexible Position	Flexible
- TFCI existence	TRUE
- Number of bits for Pilot bits(SF=128,256)	Not Present
- DPCH compressed mode info	
- TGPSI	1 Inactive
- TGPS Status Flag	Inactive
- TGCFN Transmission gap pattern sequence configuration	(Current CFN + (256 – TTI/10msec)) mod 256
- Transmission gap pattern sequence configuration	
parameters	EDD Magaurament
- TGMP	FDD Measurement
- TGPRC	62 (Current CEN + (256 TTI/(10mage)) mod 256
	(Current CFN + (256 – TTI/10msec)) mod 256
- TGSN	8
- TGL1	10
- TGL2	5
- TGD	15
- TGPL1	35

- TGPL2 35 - RPP Mode 1 - ITP Mode 1 - UL/DL Mode DL - Downlink compressed mode method SF/2 - Uplink compressed mode method Not Present - Downlink frame type - DeltaSIR1 2.0 - DeltaSIRafter1 1.0 - DeltaSIR2 Not Present - DeltaSIRafter2 Not Present - TX Diversity mode None - SSDT information Not Present - Default DPCH Offset Value Downlink information for each radio links - Primary CPICH info - Primary scrambling code 100 - PDSCH with SHO DCH info Not Present Not Present - PDSCH code mapping - Downlink DPCH info for each RL - Primary CPICH usage for channel estimation Primary CPICH may be used - DPCH frame offset 0 chips - Secondary CPICH info Not Present - DL channelisation code - Secondary scrambling code Reference to TS34.108 clause 6.10 Parameter Set - Spreading factor SF-1(SF is reference to TS34.108 clause 6.10 Parameter - Code number No change - Scrambling code change - TPC combination index 0 - SSDT Cell Identity -a - Closed loop timing adjustment mode Not Present - Secondary CCPCH info Not Present - TFCS Not Present Not Present - FACH/PCH information - References to system information blocks Not Present

Scheduling information

Contents of PHYSICAL CHANNEL RECONFIGURATION message: AM or UM (Packet to CELL_FACH from CELL_DCH in PS)

Ī	Information Element	Value/remark
ľ	Message Type	
I	RRC transaction identfier	Arbitrarily selects an integer between 0 and 3
•	Integrity check info	The presence of this IE is dependent on IXIT statements
	• ,	in TS 34.123-2. If integrity protection is indicated to be
		active, this IE is present with the values of the sub IEs as
		stated below. Else, this IE and the sub-IEs are omitted.
	 message authentication code 	SS calculates the value of MAC-I for this message and
		writes to this IE.
	 RRC message sequence number 	SS provides the value of this IE, from its internal counter.
	Integrity protection mode info	Not Present
	Ciphering mode info	Not Present
	Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
	New U-RNTI	Not Present
.	New C-RNTI	Not Present
Ш	RRC State DRX indicator	NoDRXCELL_FACH
	UTRAN DRX cycle length coefficient	Not Present
	CN information info	Not Present
	URA identity	Not Present
	RB with PDCP information	Not Present
	Frequency info	Defense to T004 400 eleves 0.40 Dementer 0.4
	- UARFCN uplink(Nu)	Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set
	- UARFCN downlink(Nd) Maximum allowed UL TX power	33dBm
	Uplink DPCH power control info	Not Present
ıl	CHOICE channel requirement	PRACH info (for RACH)Not Present
	- CHOICE mode	FDD
	- Available Signature	'0000 0000 1111 1111'B
	- Available SF	Reference to TS34.108 clause 6.10 Parameter Set
	- Scrambling code number	θ
	- Puncturing Limit	Reference to TS34.108 clause 6.10 Parameter Set
Ш	- Available Sub Channel number	'1111 1111 1111'B
	CHOICE Mode	FDD
	- Downlink PDSCH information	Not Present
.	Downlink information common for all radio links	Not Present
	Downlink information for each radio links list	Not Present
	- Primary CPICH info - Primary scrambling code	100
	- PDSCH with SHO DCH info	Not Present
	- PDSCH code mapping	Not Present
$\ $	- Downlink DPCH info for each RL	Not Present
$\ $	- Secondary CCPCH info	
	- Primary CPICH usage for channel estimation	Primary CPICH may be used
	- Secondary CPICH info	Not Present
	- Secondary scrambling code	<mark>4</mark>
	- SSDT Indicator	FALSE
	- Spreading factor	Reference to TS34.108 clause 6.10 Parameter Set
	- Code number	SF-1(SF is reference to TS34.108 clause 6.10 Parameter
$\ $	Dilet ayımbal ayistanas	Set)
$\ $	- Pilot symbol existence - TFCI existence	FALSE TRUE
$\ $	- Fixed or Flexible Position	Flexible
$\ $	- Timing offset	Clexible 0
$\ $	- TFCS	Not Present
$\ $	FACH/PCH information	Not Present
$\ $	- References to system information blocks	Net Present
$\ $	- Scheduling information	
L		

Information Element	Value/remark
Message Type	
RRC transaction identfier	Arbitrarily selects an integer between 0 and 3
Integrity check info	The presence of this IE is dependent on IXIT statements
	in TS 34.123-2. If integrity protection is indicated to be
	active, this IE is present with the values of the sub IEs as
	stated below. Else, this IE and the sub-IEs are omitted.
- message authentication code	SS calculates the value of MAC-I for this message and
-	writes to this IE.
 RRC message sequence number 	SS provides the value of this IE, from its internal counter.
Integrity protection mode info	Not Present
Ciphering mode info	Not Present
Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
New U-RNTI	Not Present
New C-RNTI	0000 0000 0000 0010B
RRC State DRX indicator	NoDRXCELL_FACH
UTRAN DRX cycle length coefficient	Not Present
CN information info	Not Present
URA identity	Not Present
RB with PDCP information	Not Present
Frequency info	
- UARFCN uplink(Nu)	Reference to TS34.108 clause 6.10 Parameter Set
- UARFCN downlink(Nd)	Reference to TS34.108 clause 6.10 Parameter Set
Maximum allowed UL TX power	33dBm
Uplink DPCH power control info	Not Present
CHOICE channel requirement	PRACH info (for RACH)Not Present
— - CHOICE mode — - Available Signature	FDD '0000 0000 1111 1111'B
- Available SF	Reference to TS34.108 clause 6.10 Parameter Set
Scrambling code number	
- Puncturing Limit	Reference to TS34.108 clause 6.10 Parameter Set
- Available Sub Channel number	'1111 1111 1111'B
CHOICE Mode	FDD
- Downlink PDSCH information	Not Present
Downlink information common for all radio links	Not Present
Downlink information for each radio links list	Not Present
- Primary CPICH info	
- Primary scrambling code	<mark>150</mark>
- PDSCH with SHO DCH info	Not Present
- PDSCH code mapping	Not Present
- Downlink DPCH info for each RL	<mark>Not Present</mark>
- Secondary CCPCH info	Not Present
- Primary CPICH usage for channel estimation	Primary CPICH may be used
- Secondary CPICH usage for channel estimation - Secondary CPICH info	Not Present
- Secondary scrambling code	1
- SSDT Indicator	FALSE
- Spreading factor	Reference to TS34.108 clause 6.10 Parameter Set
- Code number	SF-1(SF is reference to TS34.108 clause 6.10 Parameter
	Set)
- Pilot symbol existence	FALSE
- TFCI existence	TRUE
Fixed or Flexible Position	Flexible
- Timing offset	0
	Not Present
- FACH/PCH information	Not Present
- References to system information blocks	Not Present
- Scheduling information	

Contents of PHYSICAL CHANNEL RECONFIGURATION COMPLETE message: AM

Message Type	
RRC transaction identifier	Checked to see if it's set to identical value of the same IE
	in the downlink PHYSICAL CHANNEL
	RECONFIGURATION message
Integrity check info	The presence of this IE is dependent on IXIT statements
	in TS 34.123-2. If integrity protection is indicated to be
	active, this IE shall be present with the values of the sub
	IEs as stated below. Else, this IE and the sub-IEs shall be
	absent.
 Message authentication code 	This IE is checked to see if it is present. The value is
	compared against the XMAC-I value computed by SS.
- RRC Message sequence number	This IE is checked to see if it is present. The value is
	used by SS to compute the XMAC-I value.
Uplink integrity protection activation info	Not checked
COUNT-C activation time	The presence of this IE depends on the following 2
	factors: (a) There exists RB(s) mapped to RLC-TM, (b)
	UE is transiting to CELL_DCH state after the
	reconfiguration procedure. Else, this IE is absent.
CHOICE mode	<u>FDD</u>
Radio bearer uplink ciphering activation time info	Not checked
RB with PDCP information list	Not checked
Other information element	Not checked

Information Element	Value/remark
Message Type	
RRC transaction identifier	Arbitrarily selects an integer between 0 and 3
Integrity check info	The presence of this IE is dependent on IXIT statements
	in TS 34.123-2. If integrity protection is indicated to be
	active, this IE is present with the values of the sub IEs as
	stated below. Else, this IE and the sub-IEs are omitted.
- message authentication code	SS calculates the value of MAC-I for this message and
message authentication code	writes to this IE.
- RRC message sequence number	SS provides the value of this IE, from its internal counter.
	Not Present
Integrity protection mode info	
Ciphering mode info	The presence of this IE is dependent on IXIT statements
	in TS 34.123-2. If ciphering is indicated to be active, this
	IE present with the values of the sub IEs as stated below.
	Else, this IE is omitted.
- Ciphering mode command	Start
- Ciphering algorithm	Use one of the supported ciphering algorithms
 Ciphering Aactivation time for DPCH 	(256+CFN-(CFN MOD 8 + 8))MOD 256
 Radio bearer downlink ciphering activation time 	Not Present
info	
- Radio bearer identity	
- RLC sequence number	
Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
New U-RNTI	Not Present
New C-RNTI	Not Present
RRC State DRX indicator	Not resent NoDRXCELL DCH
UTRAN DRX cycle length coefficient	Not Present
CN information info	Not Present
URA identity	Not Present
Signalling RB information to setup	Not Present
RAB information for setup	
- RAB info	
- RAB identity	0000 0001B
- CN domain identity	CS domain
 NAS Synchronization Indicator 	Not Present
- Re-establishment timer	
- T314	20 seconds
- RB information to setup	
- RB identity	10 5
- PDCP info	Not Present
- CHOICE RLC info type	RLC info
- CHOICE Uplink RLC mode	TM RLC
- Transmission RLC discard	Not Present
- Segmentation indication	TRUE
- CHOICE Downlink RLC mode	TM RLC
- Segmentation indication	TRUE
- RB mapping info	I TOL
- Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
Number of unlink DLC legical shapped	Not Present
- Number of <u>uplink</u> RLC logical channels	DCH
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	12
- Logical channel identity	74
- CHOICE RLC size list	All
- MAC logical channel priority	1
- Downlink RLC logical channel info	
- Logical channel max loss	0
 Number of <u>downlink</u> RLC logical channels 	1
- Downlink transport channel type	DCH
- DL Transport channel identity	<u>62</u>
- Logical channel identity	<u>7</u> 4
RB information to be affected	(UM DCCH for RRC)
- RB identity	1
- RB mapping info	·
- Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
- NEC logical charmer mapping mulcator	INOUT TESEUL

.,	
- Number of <u>uplink</u> RLC logical channels	1
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	<u>5</u> 4
- Logical channel identity	1
- CHOICE RLC size list	All
- MAC logical channel priority	1
- Logical channel max loss	0
- Downlink RLC logical channel info	
- Number of downlink RLC logical channels	1
- Downlink transport channel type	DCH
- <u>DL</u> Transport channel identity	10
- Logical channel identity	(AM DOCK! for DDC)
RB information to be affected	(AM DCCH for RRC)
- RB identity	²
- RB mapping info	
- Information for each multiplexing option	Not Present
- RLC logical channel mapping indicator - Number of <u>uplink</u> RLC logical channels	Not Present 1
- Uplink transport channel type	DCH
- UL Transport channel identity	DCH <u>5</u> 4
- Logical channel identity	2
- CHOICE RLC size list	<u> </u>
- MAC logical channel priority	2
- logical channel max loss	2 0
- Downlink RLC logical channel info	
- Number of <u>downlink</u> RLC logical channels	1
- Downlink transport channel type	DCH
- <u>DL</u> Transport channel identity	1 <u>0</u>
- Logical channel identity	2
RB information to be affected	(AM DCCH for NAS_DT High priority)
- RB identity	3
- RB mapping info	-
- Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
- Number of <u>uplink</u> RLC logical channels	1
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	<u>5</u> 4
- Logical channel identity	3
- CHOICE RLC size list	All
- MAC logical channel priority	3
	θ
- Downlink RLC logical channel info	
- Number of <u>downlink</u> RLC logical channels	1
- Downlink transport channel type	DCH
- <u>DL</u> Transport channel identity	1 <u>0</u>
- Logical channel identity	3
RB information to be affected	(AM DCCH for NAS_DT Low priority)
- RB identity	4
- RB mapping info	
- Information for each multiplexing option	N C C
- RLC logical channel mapping indicator	Not Present
- Number of <u>uplink</u> RLC logical channels	1
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	54
- Logical channel identity	4
- CHOICE RLC size list	All
- MAC logical channel priority	4
- Logical channel max loss Downlink PLC logical channel info	0
- Downlink RLC logical channel info	1
- Number of <u>downlink</u> RLC logical channels	1 DCH
- Downlink transport channel type	
- <u>DL</u> Transport channel identity - Logical channel identity	1 <u>0</u> 4
RB with PDCP information list	Not Present
UL Transport channel information for all transport	INOUT LESCHE
channels	
- TFC subset	(This IE is repeated for TFC number.)
- Allowed Transport Format combination	0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108
/ mowed Transport i office combination	1 0 10 Max 11 0 Value 1 (Max 11 0 Value 15 16161 to 1004.100

- PRACH TFCS
- CHOICE mode
- UL DCH TFCS
- Normal
- TFCI Field 1 information
- CHOICE TFCS representation
- TFCS addition information
- CHOICE CTFC Size
- CTFC information
- Power offset information
- CHOICE Gain Factors
- Gain factor •c
- Gain factor •d
- Reference TFC ID
- Power offset Pp-m

Added or Reconfigured UL TrCH information

- Transport channel identity
- TFS
- CHOICE Transport channel type
 - Dynamic Transport format information
- RLC Size
 - Number of TBs and TTI List
 - Transmission Time Interval
 - _- Number of Transport blocks
 - RLC size
 - CHOICE Logical Channel list
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size

Added or Reconfigured UL TrCH information

- Transport channel identity
- TFS
- CHOICE Transport channel type
 - Dynamic Transport format information
- RLC Size
 - Number of TBs and TTI List
 - Transmission Time Interval
 - Number of Transport blocks
- RLC size
 - CHOICE Logical Channel list
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size

DRAC static information

DL Transport channel information common for all transport channel

- SCCPCH TFCS
- CHOICE DL parameters
- DL DCH TFCS
- Normal
- TFCI Field 1 information
- CHOICE TFCS representation
- TFCS addition information
- CHOICE CTFC Size
- CTFC information
- Power offset information

clause 6.10 Parameter Set.)

Not Present

FDD

(This IE is repeated for TFC number.)

Addition

Number of bits used must be enough to cover all combinations of CTFC from clause 6.10. Refer to TS34.108 clause 6.10 Parameter Set

Signalled Gain Factor

Ü

O

Not Present

0dB

12

Dedicated transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set If TrCH reconfiguration is executed then this is needed(e.g The rate of SRB for DCCH is changed.). 54

Dedicated transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Not Present

Not Present Independent

(This IE is repeated for TFC number.)

Addition

Number of bits used must be enough to cover all combinations of CTFC from clause 6.10. Refer to TS34.108 clause 6.10 Parameter Set Not Present

Added or Reconfigured DL TrCH information - Transport channel identity - CHOICE DL parameters SameAsUL **UL TrCH identity** <u>12</u> - DCH quality target - BLER Quality value -6.30.00- Transparent mode signalling info Not Present Added or Reconfigured DL TrCH information If TrCH reconfiguration is executed then this is needed(e.g The rate of SRB for DCCH is changed.). - Transport channel identity - CHOICE DL parameters Independent - UL TrCH identity <u>5</u>1 - TFS - CHOICE Transport channel type **Dedicated transport channels** - Dynamic Transport format information (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set RLC Size - Number of TBs and TTI List (This IE is repeated for TFI number.) - Transmission Time Interval Not Present - Number of Transport blocks Reference to TS34.108 clause 6.10 Parameter Set RI C size Reference to TS34.108 clause 6.10 Parameter Set - CHOICE Logical Channel list ALL - Semi-static Transport Format information - Transmission time interval Reference to TS34.108 clause 6.10 Parameter Set - Type of channel coding Reference to TS34.108 clause 6.10 Parameter Set - Coding Rate Reference to TS34.108 clause 6.10 Parameter Set - Rate matching attribute Reference to TS34.108 clause 6.10 Parameter Set - CRC size Reference to TS34.108 clause 6.10 Parameter Set - DCH quality target - BLER Quality value <u>-6.3</u>0.00 Not Present - Transparent mode signalling info Frequency info - UARFCN uplink(Nu) Reference to TS34.108 clause 6.10 Parameter Set - UARFCN downlink(Nd) Reference to TS34.108 clause 6.10 Parameter Set Maximum allowed UL TX power 33dBm Uplink DPCH info - Uplink DPCH power control info - DPCCH power offset -6dB - PC Preamble 8slot - Power Control Algorithm Algorithm1 - TPC step size 1dB - Scrambling code type Long - Scrambling code number 0 (0 to 16777215) - Number of DPDCH Not Present(1) - spreading factor SF is reference to TS34.108 clause 6.10 Parameter Set - TFCI existence **TRUE** - Number of FBI bit Not Present(0) - Puncturing Limit Reference to TS34.108 clause 6.10 Parameter Set **CHOICE Mode FDD** - Downlink PDSCH information Not Present Downlink information common for all radio links Downlink DPCH info common for all RL CHOICE mode **FDD** - Timing indicator Maintain - CFN-targetSFN frame offset **Not Present** - Downlink DPCH power control information - DPC mode 0 (single) - DL rate matching restriction information Not Present - Spreading factor Reference to TS34.108 clause 6.10 Parameter Set - Fixed or Flexible Position Flexible - TFCI existence **TRUE** - Number of bits for Pilot bits(SF=128,256) Not Present

(Current CFN + (256 - TTI/10msec)) mod 256

inactive

- DPCH compressed mode info

- Transmission gap pattern sequence configuration

- TGPS Status Flag

parameters - TGMP

- TGPSI

- TGCFN

- TGPRC 62 **TGCFN** (Current CFN + (256 - TTI/10msec)) mod 256 - TGSN 8 - TGL1 10 - TGL2 5 - TGD 15 - TGPL1 35 - TGPL2 35 - RPP Mode 1 Mode 1 - ITP - UL/DL Mode DL - Downlink compressed mode method SF/2 - Uplink compressed mode method Not Present - Downlink frame type - DeltaSIR1 2.0 - DeltaSIRafter1 1.0 Not Present - DeltaSIR2 - DeltaSIRafter2 Not Present - TX Diversity mode None - SSDT information Not Present - Default DPCH Offset Value Downlink information for each radio links - Primary CPICH info - Primary scrambling code 100 - PDSCH with SHO DCH info Not Present - PDSCH code mapping Not Present - Downlink DPCH info for each RL - Primary CPICH usage for channel estimation Primary CPICH may be used - DPCH frame offset 0 chips - Secondary CPICH info Not Present - DL channelisation code - Secondary scrambling code - Spreading factor Reference to TS34.108 clause 6.10 Parameter Set - Code number SF-1(SF is reference to TS34.108 clause 6.10 Parameter Set) - Scrambling code change No change - TPC combination index 0 - SSDT Cell Identity -a - Closed loop timing adjustment mode Not Present - Secondary CCPCH info Not Present - TFCS Not Present - FACH/PCH information Not Present - References to system information blocks Not Present Scheduling information

Information Element	Value/remark
Message Type	
RRC transaction identifier	Arbitrarily selects an integer between 0 and 3
Integrity check info	The presence of this IE is dependent on IXIT statements
	in TS 34.123-2. If integrity protection is indicated to be active, this IE is present with the values of the sub IEs as
	stated below. Else, this IE and the sub-IEs are omitted.
- message authentication code	SS calculates the value of MAC-I for this message and
	writes to this IE.
- RRC message sequence number	SS provides the value of this IE, from its internal counter.
Integrity protection mode info	Not Present
Ciphering mode info	The presence of this IE is dependent on IXIT statements
	in TS 34.123-2. If ciphering is indicated to be active, this IE present with the values of the sub IEs as stated
	below. Else, this IE is omitted
- Ciphering mode command	Start
- Ciphering algorithm	Use one of the supported ciphering algorithms
 Ciphering Aactivation time for DPCH 	Not Present
- Radio bearer downlink ciphering activation time	
info	105
- Radio bearer identity - RLC sequence number	105 Current RLC SN + 2
Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
New U-RNTI	Not Present
New C-RNTI	Not Present
RRC State DRX indicator	NoDRXCELL_DCH
UTRAN DRX cycle length coefficient	Not Present
CN information info URA identity	Not Present Not Present
Signalling RB information to setup	Not Present
RAB information for setup	TVOCT TOOGHT
- RAB info	
- RAB identity	0000 0001B
- CN domain identity	PS domain
- NAS Synchronization Indicator - Re-establishment timer	Not Present
- T314	20 seconds
- RB information to setup	20 00001140
- RB identity	<u>20</u> 5
- PDCP info	Not Present
- CHOICE RLC info type	RLC info
- CHOICE Uplink RLC mode - Transmission RLC discard	AM RLC
- SDU discard mode	Max DAT retransmissions
- MAX_DAT	4
- Timer_MRW	100
- MaxMRW	4
- Transmission window size	8
- Timer_RST - Max_RST	500
- Polling info	7
- Timer_poll_prohibit	200
- Timer_poll	200
- Poll_SDU	1
- Last transmission PU poll	TRUE
- Last retransmission PU poll- Poll_Windows	TRUE 99
- CHOICE Downlink RLC mode	AM RLC
- In-sequence delivery	TRUE
- Receiving window size	8
- Downlink RLC status info	
- Timer_status_prohibit	200
- Timer_EPC	200 TRUE
- Missing PU indicator - RB mapping info	TRUE
- Information for each multiplexing option	
information for each multiplexiting option	

- RLC logical channel mapping indicator	Not Present
- Number of uplink RLC logical channels	1
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	<u>1</u> 2
- Logical channel identity	<u>7</u> 4
- CHOICE RLC size list	All
- MAC logical channel priority	1
- Logical channel max loss	0
- Downlink RLC logical channel info	♥
- Number of <u>downlink</u> RLC logical channels	1
- Downlink transport channel type	DCH
- <u>DL</u> Transport channel identity	<u>62</u>
- Logical channel identity	74
RB information to be affected	(UM DCCH for RRC)
- RB identity	1
- RB mapping info	
- Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
- Number of <u>uplink</u> RLC logical channels	1
- Uplink transport channel type	DCH
- Transport channel identity	1
- <u>UL</u> Logical channel identity	<u>5</u> 4
- CHOICE RLC size list	Äll
- MAC logical channel priority	1
- Logical channel max loss	0
- Downlink RLC logical channel info	
- Number of <u>downlink</u> RLC logical channels	1
- Downlink transport channel type	DCH
- <u>DL</u> Transport channel identity	10
- Logical channel identity	1 <u>0</u> 1
RB information to be affected	(AM DCCH for RRC)
- RB identity	2
	2
- RB mapping info	
- Information for each multiplexing option	Not Present
- RLC logical channel mapping indicator	Not Present 1
- Number of <u>uplink</u> RLC logical channels	
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	<u>5</u> 4
- Logical channel identity	2
- MAC logical channel priority	$\frac{1}{2}$
- MAC logical channel priority	
- Logical channel max loss	0
- Downlink RLC logical channel info	
- Number of <u>downlink</u> RLC logical channels	1
- Downlink transport channel type	DCH
- <u>DL</u> Transport channel identity	10
- Logical channel identity	2
RB information to be affected	(AM DCCH for NAS_DT High priority)
- RB identity	3
- RB mapping info	
- Information for each multiplexing option	
 - RLC logical channel mapping indicator 	Not Present
- Number of <u>uplink</u> RLC logical channels	1
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	<u>5</u> 4
- Logical channel identity	3
- CHOICE RLC size list	All
- MAC logical channel priority	3
	0
- Downlink RLC logical channel info	
- Number of downlink RLC logical channels	1
- Downlink transport channel type	DCH
- <u>DL</u> Transport channel identity	1 <u>0</u>
- Logical channel identity	3
RB information to be affected	(AM DCCH for NAS_DT Low priority)
- RB identity	4
- RB mapping info	
- Information for each multiplexing option	

- RLC logical channel mapping indicator
- Number of uplink RLC logical channels
- Uplink transport channel type
- UL Transport channel identity
- Logical channel identity
- CHOICE RLC size list
- MAC logical channel priority
- Logical channel max loss
- Downlink RLC logical channel info
- Number of downlink RLC logical channels
- Downlink transport channel type
- DL Transport channel identity
- Logical channel identity

RB with PDCP information list

UL Transport channel information for all transport channels

- TFC subset
- Allowed Transport Format combination
- PRACH TECS
- CHOICE mode
- UL DCH TFCS
- Normal
- TFCI Field 1 information
- CHOICE TFCS representation
- TFCS addition information
- CHOICE CTFC Size
- CTFC information
- Power offset information

Added or Reconfigured UL TrCH information

- Transport channel identity
- TFS
- Dynamic Transport format information
- Number of Transport blocks
- RLC size
- Semi-static Transport Format information
- Transmission time interval
- Type of channel coding
- Coding Rate
- Rate matching attribute
- CRC size

Added or Reconfigured UL TrCH information

- Transport channel identity
- TFS
- CHOICE Transport channel type
 - Dynamic Transport format information
- RLC Size
 - Number of TBs and TTI List
 - Transmission Time Interval
 - _- Number of Transport blocks
- **RLC size**
 - CHOICE Logical Channel list
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size

DRAC static information

DL Transport channel information common for all transport channel

- SCCPCH TFCS
- CHOICE DL parameters
- DL DCH TFCS
- Normal
- TFCI Field 1 information

Not Present

DCH

<u>5</u>1

ΑII

4

DCH

10

Not Present

(This IE is repeated for TFC number.)

0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108 clause 6.10 Parameter Set.)

Not Present

FDD

(This IE is repeated for TFC number.)

Addition

Number of bits used must be enough to cover all combinations of CTFC from clause 6.10. Refer to TS34.108 clause 6.10 Parameter Set Not Present

12

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

Reference to TS34.108 clause 6.10 Parameter Set

Reference to TS34.108 clause 6.10 Parameter Set

Reference to TS34.108 clause 6.10 Parameter Set

Reference to TS34.108 clause 6.10 Parameter Set

Reference to TS34.108 clause 6.10 Parameter Set

If TrCH reconfiguration is executed then this is

needed(e.g The rate of SRB for DCCH is changed.). 51

Dedicated transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Not Present

Not Present Independent

(This IE is repeated for TFC number.)

- CHOICE CTFC Size
- TFCS addition information
- CHOICE CTFC Size
- CTFC information
- Power offset information

Added or Reconfigured DL TrCH information

- Transport channel identity
- CHOICE DL parameters
- TFS
- CHOICE Transport channel type
 - Dynamic Transport format information
- RLC Size
 - Number of TBs and TTI List
 - Transmission Time Interval
 - Number of Transport blocks
- RLC size
 - CHOICE Logical Channel list
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size
 - DCH quality target
 - BLER Quality value
- Transparent mode signalling info

Added or Reconfigured DL TrCH information

- Transport channel identity
- TFS
- CHOICE Transport channel type
- CHOICE DL parameters
 - Dynamic Transport format information
- RLC Size
 - Number of TBs and TTI List
 - Transmission Time Interval
 - _- Number of Transport blocks
- RLC size
 - CHOICE Logical Channel list
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size
 - DCH quality target
 - BLER Quality value
 - Transparent mode signalling info

Frequency info

- UARFCN uplink(Nu)
- UARFCN downlink(Nd)

Maximum allowed UL TX power

Uplink DPCH info

- Uplink DPCH power control info
- DPCCH power offset
- PC Preamble
- Power Control Algorithm
- TPC step size
- Scrambling code type
- Scrambling code number
- Number of DPDCH
- spreading factor
- TFCI existence
- Number of FBI bit
- Puncturing Limit
- **CHOICE Mode**
 - Downlink PDSCH information

Addition

Number of bits used must be enough to cover all combinations of CTFC from clause 6.10. Refer to TS34.108 clause 6.10 Parameter Set Not Present

62

Independent

Dedicated transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

-6.30.00

Not Present

If TrCH reconfiguration is executed then this is needed(e.g The rate of SRB for DCCH is changed.). 10

Dedicated transport channels

Independent

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

<u>-6.3</u>0.00

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set 33dBm

-6dB 8slot Algorithm1 1dB Long

0 (0 to 16777215)

Not Present(1)

SF is reference to TS34.108 clause 6.10 Parameter Set

TRUE

Not Present(0)

Reference to TS34.108 clause 6.10 Parameter Set

FDD

Not Present

Downlink information common for all radio links - Downlink DPCH info common for all RL **CHOICE** mode **EDD** - Timing indicator <u>Maintain</u> - CFN-targetSFN frame offset Not Present - Downlink DPCH power control information - DPC mode 0 (single) Not Present - DL rate matching restriction information Reference to TS34.108 clause 6.10 Parameter Set - Spreading factor - Fixed or Flexible Position Flexible - TFCI existence **TRUE** - Number of bits for Pilot bits(SF=128,256) Reference to TS34.108 clause 6.10 Parameter Set - DPCH compressed mode info - TGPSI - TGPS Status Flag (Current CFN + (256 - TTI/10msec)) mod 256 TGCFN - Transmission gap pattern sequence configuration parameters - TGMP **FDD Measurement** - TGPRC **TGCFN** (Current CFN + (256 - TTI/10msec)) mod 256 - TGSN 10 - TGL1 - TGL2 5 15 - TGD - TGPL1 35 - TGPL2 35 - RPP Mode 1 - ITP Mode 1 - UL/DL Mode DL - Downlink compressed mode method SF/2 - Uplink compressed mode method Not Present - Downlink frame type - DeltaSIR1 2.0 - DeltaSIRafter1 1.0 - DeltaSIR2 Not Present - DeltaSIRafter2 Not Present - TX Diversity mode None - SSDT information Not Present - Default DPCH Offset Value Downlink information for each radio links - Primary CPICH info - Primary scrambling code - PDSCH with SHO DCH info Not Present - PDSCH code mapping Not Present - Downlink DPCH info for each RL - Primary CPICH usage for channel estimation Primary CPICH may be used - DPCH frame offset 0 chips - Secondary CPICH info Not Present - DL channelisation code - Secondary scrambling code Reference to TS34.108 clause 6.10 Parameter Set - Spreading factor - Code number SF-1(SF is reference to TS34.108 clause 6.10 Parameter Set) - Scrambling code change No change - TPC combination index - SSDT Cell Identity -a Not Present - Closed loop timing adjustment mode - Secondary CCPCH info Not Present Not Present - FACH/PCH information Not Present - References to system information blocks Not Present Scheduling information

Information Element	Value/remark
Message Type	
RRC transaction identifier	Arbitrarily selects an integer between 0 and 3
Integrity check info	The presence of this IE is dependent on IXIT statements in
	TS 34.123-2. If integrity protection is indicated to be active, this IE is present with the values of the sub IEs as stated
	below. Else, this IE and the sub-IEs are omitted.
- message authentication code	SS calculates the value of MAC-I for this message and
J	writes to this IE.
- RRC message sequence number	SS provides the value of this IE, from its internal counter.
Integrity protection mode info	Not Present
Ciphering mode info	The presence of this IE is dependent on IXIT statements in
	TS 34.123-2. If ciphering is indicated to be active, this IE present with the values of the sub IEs as stated below.
	Else, this IE is omitted.
- Ciphering mode command	Start
- Ciphering algorithm	Use one of the supported ciphering algorithms.
 Ciphering Aactivation time for DPCH 	Not Present
- Radio bearer downlink ciphering activation time	
info	205
- Radio bearer identity - RLC sequence number	205 Current RLC SN + 2
Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
New U-RNTI	Not Present
New C-RNTI	Not Present
RRC State DRX indicator	NoDRXCELL_DCH
UTRAN DRX cycle length coefficient CN information info	Not Present Not Present
URA identity	Not Present
Signalling RB information to setup	Not Present
RAB information for setup	
- RAB info	
- RAB identity	0000 0001B
- CN domain identity	PS domain
- NAS Synchronization Indicator - Re-establishment timer	Not Present
- T314	20 seconds
- RB information to setup	
- RB identity	<u>20</u> 5
- PDCP info	Not Present
- CHOICE RLC info type - CHOICE Uplink RLC mode	RLC info
- Transmission RLC discard	AW REC
- CHOICE SDU discard mode	Max DAT retransmissions
- MAX_DAT	4
- Timer_MRW	100
- MaxMRW	4
- Transmission window size - Timer_RST	8 500
- Max_RST	4
- Polling info	
- Timer_poll_prohibit	200
- Timer_poll	200
- Poll_SDU - Last transmission PU poll	1 TRUE
- Last transmission PU poll - Last retransmission PU poll	TRUE
- Poll_Windows	99
- CHOICE Downlink RLC mode	AM RLC
- In-sequence delivery	TRUE
- Receiving window size	8
- Downlink RLC status info	200
- Timer_status_prohibit - Timer_EPC	200
- Missing PU indicator	TRUE
- RB mapping info	
- Information for each multiplexing option	

- RLC logical channel mapping indicator	Not Present
 Number of <u>uplink</u> RLC logical channels 	1
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	12
- Logical channel identity	74
- CHOICE RLC size list	All
- MAC logical channel priority	1
- Logical channel max loss	θ
- Downlink RLC logical channel info - Number of downlink RLC logical channels	1
- Downlink transport channel type	DCH
- <u>DL</u> Transport channel identity	62
- Logical channel identity	74
RB information to be affected	(UM DCCH for RRC)
- RB identity	1
- RB mapping info	
- RLC logical channel mapping indicator	Not Present
- Information for each multiplexing option	
- Number of uplink RLC logical channels	1
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	1
- Logical channel identity	<u>5</u> 4
- CHOICE RLC size list	All
- MAC logical channel priority	1
- Logical channel max loss	0
- Downlink RLC logical channel info	
- Number of downlink RLC logical channels	1
- Downlink transport channel type	DCH
- <u>UL</u> Transport channel identity - Logical channel identity	104 1
RB information to be affected	(AM DCCH for RRC)
- RB identity	2
- RB mapping info	-
- Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
- Number of uplink RLC logical channels	1
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	<u>5</u> 4
- Logical channel identity	2
- CHOICE RLC size list	$\begin{vmatrix} AII \\ 2 \end{vmatrix}$
- MAC logical channel priority	2
- Logical channel max loss	0
- Downlink RLC logical channel info	
- Number of downlink RLC logical channels	1 PCH
- Downlink transport channel type	DCH
- <u>UL</u> Transport channel identity - Logical channel identity	104
RB information to be affected	(AM DCCH for NAS_DT High priority)
- RB identity	3
- RB mapping info	Ĭ
- Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
- Number of uplink RLC logical channels	1
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	<u>5</u> 4
- Logical channel identity	3
- CHOICE RLC size list	All
- MAC logical channel priority	3
- Logical channel max loss	θ
- Downlink RLC logical channel info	
- Number of <u>downlink</u> RLC logical channels	1
- Downlink transport channel type	DCH
- <u>DL</u> Transport channel identity	10
- Logical channel identity	(AM DCCH for NAS DT Low priority)
RB information to be affected - RB identity	(AM DCCH for NAS_DT Low priority)
- RB mapping info	7
- Information for each multiplexing option	
information for each multiplexing option	<u>l</u>

- RLC logical channel mapping indicator Not Present - Number of uplink RLC logical channels - Uplink transport channel type DCH - UL Transport channel identity <u>5</u>1 - Logical channel identity - CHOICE RLC size list ΑII - MAC logical channel priority 4 Logical channel max loss - Downlink RLC logical channel info - Number of downlink RLC logical channels - Downlink transport channel type DCH - DL Transport channel identity 10 - Logical channel identity RB with PDCP information list Not Present UL Transport channel information for all transport channels (This IE is repeated for TFC number.) - TFC subset - Allowed Transport Format combination 0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108 clause 6.10 Parameter Set.) - PRACH TECS Not Present - CHOICE mode FDD - UL DCH TFCS (This IE is repeated for TFC number.) - Normal - TFCI Field 1 information - CHOICE TFCS representation Addition - TFCS addition information - CHOICE CTFC Size Number of bits used must be enough to cover all combinations of CTFC from clause 6.10. Refer to TS34.108 clause 6.10 Parameter Set - CTFC information - Power offset information - CHOICE Gain Factors Signalled Gain Factor - Gain factor •c - Gain factor •d n - Reference TFC ID Not Present - Power offset Pp-m 0dB Deleted UL TrCH information - Transport channel identity <u>15</u> Added or Reconfigured UL TrCH information - Transport channel identity 12 - TFS - CHOICE Transport channel type Dedicated transport channels - Dynamic Transport format information (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set - RLC Size - Number of TBs and TTI List (This IE is repeated for TFI number.) - Transmission Time Interval Not Present - Number of Transport blocks Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set - RLC size - CHOICE Logical Channel list ALL - Semi-static Transport Format information - Transmission time interval Reference to TS34.108 clause 6.10 Parameter Set - Type of channel coding Reference to TS34.108 clause 6.10 Parameter Set - Coding Rate Reference to TS34.108 clause 6.10 Parameter Set - Rate matching attribute Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set - CRC size Added or Reconfigured UL TrCH information - Transport channel identity 54 - TFS - CHOICE Transport channel type **Dedicated transport channels** - Dynamic Transport format information (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set - RLC Size - Number of TBs and TTI List (This IE is repeated for TFI number.) - Transmission Time Interval Not Present - Number of Transport blocks Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set - CHOICE Logical Channel list ALL - Semi-static Transport Format information - Transmission time interval Reference to TS34.108 clause 6.10 Parameter Set - Type of channel coding Reference to TS34.108 clause 6.10 Parameter Set

- Coding Rate
- Rate matching attribute
- CRC size

DRAC static information

DL Transport channel information common for all transport channel

- SCCPCH TFCS
- CHOICE DL parameters
- DL DCH TFCS
- Normal
- TFCI Field 1 information
- CHOICE TFCS representation
- TFCS addition information
- CHOICE CTFC Size
- CTFC information
- Power offset information

Deleted DL TrCH information

- Transport channel identity
- Transport channel identity
- Transport channel identity

Added or Reconfigured DL TrCH information

- Transport channel identity
- CHOICE DL parameters
- TFS
- CHOICE Transport channel type
- Dynamic Transport format information
- RLC Size
- Number of TBs and TTI List
 - Transmission Time Interval
 - Number of Transport blocks
 - RLC size
- CHOICE Logical Channel list
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size
 - DCH quality target
 - BLER Quality value
 - Transparent mode signalling info

Added or Reconfigured DL TrCH information

- Transport channel identity
- CHOICE DL parameters
- CHOICE Transport channel type
 - Dynamic Transport format information
- RLC Size
 - Number of TBs and TTI List
 - Transmission Time Interval
 - _- Number of Transport blocks
 - **RLC size**
 - CHOICE Logical Channel list
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size
 - DCH quality target
 - BLER Quality value
 - Transparent mode signalling info

Frequency info

- UARFCN uplink(Nu)
- UARFCN downlink(Nd)

Maximum allowed UL TX power

Uplink DPCH info

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Not Present

Not Present

Independent

(This IE is repeated for TFC number.)

Addition

Number of bits used must be enough to cover all combinations of CTFC from clause 6.10. Refer to TS34.108 clause 6.10 Parameter Set Not Present

<u>12</u> <u>13</u> <u>14</u>

62

Independent

Dedicated transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set **ALL**

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

-6.30.00

Not Present

10

Independent

Dedicated transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

-6.30.00

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set 33dBm

- Uplink DPCH power control info - DPCCH power offset -6dB - PC Preamble 8slot Algorithm1 - Power Control Algorithm - TPC step size 1dB - Scrambling code type Long - Scrambling code number 0 (0 to 16777215) - Number of DPDCH Not Present(1) - spreading factor SF is reference to TS34.108 clause 6.10 Parameter Set - TFCI existence **TRUE** - Number of FBI bit Not Present(0) - Puncturing Limit Reference to TS34.108 clause 6.10 Parameter Set **CHOICE Mode FDD** - Downlink PDSCH information Not Present Downlink information common for all radio links - Downlink DPCH info common for all RL **CHOICE** mode **EDD** - Timing indicator **Maintain** - CFN-targetSFN frame offset Not Present - Downlink DPCH power control information - DPC mode 0 (single) - DL rate matching restriction information Not Present - Spreading factor Reference to TS34.108 clause 6.10 Parameter Set - Fixed or Flexible Position Flexible - TFCI existence TRUE - Number of bits for Pilot bits(SF=128,256) Reference to TS34.108 clause 6.10 Parameter Set - DPCH compressed mode info - TGPSI - TGPS Status Flag inactive - TGCFN (Current CFN + (256 - TTI/10msec)) mod 256 - Transmission gap pattern sequence configuration parameters - TGMP **FDD Measurement** - TGPRC **TGCFN** (Current CFN + (256 - TTI/10msec)) mod 256 - TGSN 8 - TGL1 10 - TGL2 5 15 - TGD - TGPL1 35 - TGPL2 35 - RPP Mode 1 - ITP Mode 1 - UL/DL Mode DL SF/2 - Downlink compressed mode method - Uplink compressed mode method Not Present - Downlink frame type - DeltaSIR1 2.0 - DeltaSIRafter1 1.0 - DeltaSIR2 Not Present - DeltaSIRafter2 Not Present - TX Diversity mode None - SSDT information Not Present - Default DPCH Offset Value 0 Downlink information for each radio links - Primary CPICH info - Primary scrambling code 100 - PDSCH with SHO DCH info Not Present - PDSCH code mapping Not Present - Downlink DPCH info for each RL - Primary CPICH usage for channel estimation Primary CPICH may be used - DPCH frame offset - Secondary CPICH info Not Present - DL channelisation code - Secondary scrambling code - Spreading factor Reference to TS34.108 clause 6.10 Parameter Set - Code number SF-1(SF is reference to TS34.108 clause 6.10 Parameter Set)

- Scrambling code change	No change
- TPC combination index	0
- SSDT Cell Identity	-a
 Closed loop timing adjustment mode 	Not Present
- Secondary CCPCH info	Not Present
- TFCS	Not Present
- FACH/PCH information	Not Present
- References to system information blocks	Not Present
- Scheduling information	

Message Type RRC transaction identifier Arbitrarily selects an integer between 0 and 3 The property of this IS is dependent on IVIT attached.	
RRC transaction identifier Arbitrarily selects an integer between 0 and 3 The presence of this IF is dependent on IVIT statements.	
Integrity check info The presence of this IE is dependent on IXIT statements and indicated to be	
TS 34.123-2. If integrity protection is indicated to be this IE is present with the values of the sub IEs as sta	
below. Else, this IE and the sub-IEs are omitted.	ieu
- message authentication code SS calculates the value of MAC-I for this message ar	d
writes to this IE.	
- RRC message sequence number SS provides the value of this IE, from its internal cour	ter.
Integrity protection mode info Not Present	
Ciphering mode info The presence of this IE is dependent on IXIT statements and the presence of this IE is dependent on IXIT statements.	
TS 34.123-2. If ciphering is indicated to be active, this present with the values of the sub IEs as stated below	
Else, this IE is omitted	, .
- Ciphering mode command start	
- Ciphering algorithm Use one of the supported ciphering algorithms	
- <u>Ciphering Aa</u> ctivation time for DPCH Not Present	
- Radio bearer downlink ciphering activation time	
info Padio boarer identity 205	
- Radio bearer identity - RLC sequence number - RLC SN + 2	
Activation time (256+CFN-(CFN MOD 8 + 8))MOD 256	
New U-RNTI Not Present	
New C-RNTI Not Present	
RRC State DRX indicator NoDRXCELL_FACH	
UTRAN DRX cycle length coefficient Not Present	
CN information info URA identity Not Present Not Present	
Signalling RB information to setup Not Present	
RAB information for setup	
- RAB info (AM DTCH for PS domain)	
- RAB identity 0000 0001B	
- CN domain identity PS domain NAS Symphysization Indicator	
- NAS Synchronization Indicator - Re-establishment timer	
- T314 20 seconds	
- RB information to setup	
- RB identity <u>20</u> 5	
- PDCP info Not Present	
- CHOICE RLC info type - CHOICE Uplink RLC mode RLC info AM RLC	
- Transmission RLC discard	
- CHOICE SDU discard mode Max DAT retransmissions	
- MAX_DAT 4	
- Timer_MRW 100	
- MaxMRW - Transmission window size 8	
- Transmission window size 8 - Timer_RST 500	
- Max_RST 4	
- Polling info	
- Timer_poll_prohibit 200	
- Timer_poll 200	
- Poll_SDU - Last transmission PU poll TRUE	
- Last retransmission PU poll TRUE	
- Poll_Windows 99	
- CHOICE Downlink RLC mode AM RLC	
- In-sequence delivery TRUE	
- Receiving window size 8	
- Downlink RLC status info - Timer_status_prohibit 200	
- Timer_status_profilbit 200 - Timer_EPC 200	
- Missing PU indicator TRUE	
- RB mapping info	
- Information for each multiplexing option	

 RLC logical channel mapping indicator 	Not Present
- Number of <u>uplink</u> RLC logical channels	1
- Uplink transport channel type	RACH
- Logical channel identity	<u>7</u> 6
- CHOICE RLC size list	<u>All</u>
- MAC logical channel priority	6
- Logical channel max loss	0
- Downlink RLC logical channel info	
- Number of downlink RLC logical channels	1
- Downlink transport channel type	FACH
- Logical channel identity	<u>6</u> 4
RB information to be affected	(UM DCCH for RRC)
- RB identity	1
- RB mapping info	
- Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
- Number of uplink RLC logical channels	1
- Uplink transport channel type	RACH
- Logical channel identity	1
- CHOICE RLC size list	
- MAC logical channel priority	<u>All</u> 2
- Logical channel max loss	0
- Downlink RLC logical channel info	
- Number of <u>downlink</u> RLC logical channels	1
- Downlink transport channel type	FACH
- Logical channel identity	1
RB information to be affected	(AM DCCH for RRC)
- RB identity	2
- RB mapping info	
- Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
- Number of <u>uplink</u> RLC logical channels	1
- Uplink transport channel type	RACH
- Logical channel identity	2
- CHOICE RLC size list	
	<u>All</u> 3
- MAC logical channel priority - Logical channel max loss	$\frac{3}{\theta}$
- Downlink RLC logical channel info	♥
	4
- Number of <u>downlink</u> RLC logical channels	1 FACH
- Downlink transport channel type	
- Logical channel identity	2 (AM DCCH for NAS DT High priority)
RB information to be affected	(AM DCCH for NAS_DT High priority)
- RB identity	3
- RB mapping info	
- Information for each multiplexing option	Not Drogent
- RLC logical channel mapping indicator	Not Present
- Number of <u>uplink</u> RLC logical channels	1 DACH
- Uplink transport channel type	RACH
- Logical channel identity	3
- CHOICE RLC size list	<u>All</u> 4
- MAC logical channel priority	
- Logical channel max loss	0
- Downlink RLC logical channel info	
- Number of <u>downlink</u> RLC logical channels	1
- Downlink transport channel type	FACH
- Logical channel identity	3
RB information to be affected	(AM DCCH for NAS_DT Low priority)
- RB identity	4
- RB mapping info	
- Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
- Number of <u>uplink</u> RLC logical channels	1
- Uplink transport channel type	RACH
- Logical channel identity	4
- CHOICE RLC size list	<u>All</u>
- MAC logical channel priority	5
- Logical channel max loss	0
- Downlink RLC logical channel info	

- Number of downlink RLC logical channels - Downlink transport channel type **FACH** - Logical channel identity RB information to be affected (TM CCCH uplink for RRC) - RB identity - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator Not Present - Number of uplink RLC logical channels **RACH** - Uplink transport channel type - Logical channel identity 5 Logical channel max loss 0 - CHOICE RLC size list <u>All</u> - MAC logical channel priority Downlink RLC logical channel info Not Present RB information to be affected (UM CCCH for RRC) - RB identity - RB mapping info - Information for each multiplexing option - Number of downlink RLC logical channels - Downlink transport channel type **FACH** - Logical channel identity - Downlink RLC logical channel info Not Present RB information to be affected (TM BCCH for RRC) - RB identity - RB mapping info - Information for each multiplexing option - Number of downlink RLC logical channels - Downlink transport channel type **FACH** - Logical channel identity 5 - Downlink RLC logical channel info Not Present RB information to be affected (TM PCCH for RRC) - RB identity - RB mapping info - Information for each multiplexing option - Number of downlink RLC logical channels - Downlink transport channel type **PCH** - Logical channel identity - Downlink RLC logical channel info Not Present RB with PDCP information list Not Present UL Transport channel information for all transport channels - TFC subset (This IE is repeated for TFC number.) - Allowed Transport Format combination 0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108 clause 6.10 Parameter Set.) - PRACH TFCS Not Present - CHOICE mode **FDD** - UL DCH TFCS Not Present **Deleted UL TrCH information** - Transport channel identity - Transport channel identity 5 Added or Reconfigured UL TrCH information - Transport channel identity 15 - CHOICE Transport channel type Common transport channels - Dynamic Transport format information (This IE is repeated for TFI number) - RLC Size Reference to TS34.108 clause 6.10 Parameter Set - Number of TBs and TTI List (This IE is repeated for TFI number.) - Number of Transport blocks Reference to TS34.108 clause 6.10 Parameter Set **RLC size** Reference to TS34.108 clause 6.10 Parameter Set - CHOICE mode FDD - CHOICE Logical Channel List ALL - Semi-static Transport Format information - Transmission time interval Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set - Type of channel coding - Coding Rate Reference to TS34.108 clause 6.10 Parameter Set - Rate matching attribute Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set - CRC size

DRAC static information DL Transport channel information common for all transport channel - SCCPCH TFCS - Normal - TFCI Field 1 information - CHOICE TFCS representation - TFCS addition information - CHOICE CTFC Size - CTFC information - Power offset information - CHOICE DL parameters - DL DCH TFCS Deleted DL TrCH information - Transport channel identity - Transport channel identity Added or Reconfigured DL TrCH information - Transport channel identity - CHOICE DL parameters - CHOICE Transport channel type - Dynamic Transport format information - RLC Size - Number of TBs and TTI List - Number of Transport blocks RLC size - CHOICE mode - CHOICE Logical Channel List - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size - DCH quality target - Transparent mode signalling info Added or Reconfigured DL TrCH information - Transport channel identity - CHOICE DL parameters - CHOICE Transport channel type - Dynamic Transport format information - RLC Size - Number of TBs and TTI List - Number of Transport blocks - RLC size - CHOICE mode - CHOICE Logical Channel List - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size

- DCH quality target

- TFS

- RLC Size

RLC size

- CHOICE mode

- CHOICE DL parameters

- Transparent mode signalling info

Added or Reconfigured DL TrCH information - Transport channel identity

- CHOICE Transport channel type

- Number of TBs and TTI List

- Number of Transport blocks

- Dynamic Transport format information

Not Present

(This IE is repeated for TFC number.)

Addition

Number of bits used must be enough to cover all combinations of CTFC from clause 6.10. Refer to TS34.108 clause 6.10 Parameter Set

Independent Not Present

6 10

Independent

Common transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set **FDD**

<u>ALL</u>

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Not Present Not Present

<u>13</u>2

Independent

Common transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set FDD

ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Not Present

143

Independent

Not Present

Common transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set FDD

- CHOICE Logical Channel List	ALL
- Semi-static Transport Format information	D (
- Transmission time interval	Reference to TS34.108 clause 6.10 Parameter Set
- Type of channel coding	Reference to TS34.108 clause 6.10 Parameter Set
- Coding Rate	Reference to TS34.108 clause 6.10 Parameter Set
 Rate matching attribute 	Reference to TS34.108 clause 6.10 Parameter Set
- CRC size	Reference to TS34.108 clause 6.10 Parameter Set
- DCH quality target	Not Present
- Transparent mode signalling info	Not Present
Added or Reconfigured DL TrCH information	
- Transport channel identity	4
- CHOICE DL parameters	Independent
——————————————————————————————————————	independent
- Dynamic Transport format information	(This IE is repeated for TFI number)
- Number of Transport blocks	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
'	Reference to 1004.100 diause 0.101 diameter oct
- Semi-static Transport Format information	Peterance to TC24 100 clause 6.40 December Cot
- Transmission time interval	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
- Coding Rate	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
— - DCH quality target	Not Present
	Not Present
Frequency info	
- UARFCN uplink(Nu)	Reference to TS34.108 clause 6.10 Parameter Set
- UARFCN downlink(Nd)	Reference to TS34.108 clause 6.10 Parameter Set
Maximum allowed UL TX power	33dBm
CHOICE channel requirement	PRACH info (for RACH)Not Present
- CHOICE mode	FDD
- Available Signature	'0000 0000 1111 1111'B
	Reference to TS34.108 clause 6.10 Parameter Set
- Scrambling code number	0
	Reference to TS34.108 clause 6.10 Parameter Set
- Available Sub Channel number	'1111 1111 1111'B
CHOICE Mode	FDD
- Downlink PDSCH information	Not Present
Downlink information common for all radio links	Not Present
Downlink information for each radio links	Not Present
- Primary CPICH info	
- Primary scrambling code	100
- PDSCH with SHO DCH info	Not Present
- PDSCH code mapping	Not Present
——————————————————————————————————————	Not Present
- Secondary CCPCH info	Not I Tosum
- Selection Indicator	Not Present
- Primary CPICH usage for channel estimation	Primary CPICH may be used
- Secondary CPICH info	Not Present
	INOL FIEDHIL
- Secondary scrambling code - channelisation code	
- Secondary scrambling code	1
- SSDT Indicator	FALSE
- Spreading factor	Reference to TS34.108 clause 6.10 Parameter Set
——- Code number	SF-1(SF is reference to TS34.108 clause 6.10 Parameter
Bill 1 1 1 1 1 1 1 1 1	Set)
- Pilot symbol existence	FALSE
	TRUE
- Fixed or Flexible Position	Flexible
	0
——- TFCS	Not Present
	Not Present
	Not Present
<u> </u>	

Information Element	Value/remark
Message Type	Talaan Talaan
RRC transaction identifier	Arbitrarily selects an integer between 0 and 3
Integrity check info	The presence of this IE is dependent on IXIT statements in
	TS 34.123-2. If integrity protection is indicated to be active,
	this IE is present with the values of the sub IEs as stated
	below. Else, this IE and the sub-IEs are omitted.
- message authentication code	SS calculates the value of MAC-I for this message and
	writes to this IE.
- RRC message sequence number	SS provides the value of this IE, from its internal counter.
Integrity protection mode info	Not Present
Ciphering mode info	The presence of this IE is dependent on IXIT statements in
	TS 34.123-2. If ciphering is indicated to be active, this IE
	present with the values of the sub IEs as stated below.
	Else, this IE is omitted
- Ciphering mode command	start
- Ciphering algorithm	Use one of the supported ciphering algorithms
- <u>Ciphering Aa</u> ctivation time for DPCH	Not Present
- Radio bearer downlink ciphering activation time	
info	005
- Radio bearer identity	205
- RLC sequence number	Current RLC SN
Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
New U-RNTI	Not Present
New C-RNTI	Not Present
RRC State DRX-indicator UTRAN DRX cycle length coefficient	NeDRXCELL_FACH Not Present
CN information info	Not Present
URA identity	Not Present
Signalling RB information to setup	Not Present
RAB information for setup	(AM DTCH for PS domain)
- RAB info	(AM DICITION 3 domain)
- RAB identity	0000 0001B
- CN domain identity	PS domain
- NAS Synchronization Indicator	Not Present
- Re-establishment timer	
- T314	20 seconds
- RB information to setup	
- RB identity	<u>20</u> 5
- PDCP info	Not Present
- CHOICE RLC info type	RLC info
- CHOICE Uplink RLC mode	AM RLC
- Transmission RLC discard	
- CHOICE SDU discard mode	Max DAT retransmissions
- MAX_DAT	4
- Timer_MRW	100
- MaxMRW	4
- Transmission window size	8 500
- Timer_RST - Max_RST	500
- Max_RST - Polling info	*
- Folling Into - Timer_poll_prohibit	200
- Timer_poil_profiloit - Timer_poll	200
- Poll_SDU	1
- Last transmission PU poll	TRUE
- Last retransmission PU poll	TRUE
- Poll_Windows	99
- CHOICE Downlink RLC mode	AM RLC
- In-sequence delivery	TRUE
- Downlink RLC status info	
- Timer_status_prohibit	200
- Timer_EPC	200
- Missing PU indicator	TRUE
- RB mapping info	
- Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present

- Number of <u>uplink</u> RLC logical channels	1
- Uplink transport channel type	RACH
- Logical channel identity	
	76
- CHOICE RLC size list	All
- MAC logical channel priority	6
- Logical channel max loss	0
- Downlink RLC logical channel info	
- Number of downlink RLC logical channels	1
- Downlink transport channel type	FACH
- Logical channel identity	<u>6</u> 4
RB information to be affected	(UM DCCH for RRC)
- RB identity	1
- RB mapping info	·
- Information for each multiplexing option	N . B
 RLC logical channel mapping indicator 	Not Present
 Number of <u>uplink</u> RLC logical channels 	1
- Uplink transport channel type	RACH
- Logical channel identity	1
- CHOICE RLC size list	1 -
	All
- MAC logical channel priority	2
	θ
 Downlink RLC logical channel info 	
- Number of downlink RLC logical channels	1
- Downlink transport channel type	FACH
	1
- Logical channel identity	1 -
RB information to be affected	(AM DCCH for RRC)
- RB identity	2
- RB mapping info	
 Information for each multiplexing option 	
- RLC logical channel mapping indicator	Not Present
- Number of <u>uplink</u> RLC logical channels	1
	1 -
 Uplink transport channel type 	RACH
- Logical channel identity	2
- CHOICE RLC size list	All
- MAC logical channel priority	$\frac{1}{3}$
- Logical channel max loss	0
- Downlink RLC logical channel info	8
- Number of <u>downlink</u> RLC logical channels	1
 Downlink transport channel type 	FACH
 Logical channel identity 	2
RB information to be affected	(AM DCCH for NAS_DT High priority)
- RB identity	3
	•
- RB mapping info	
- Information for each multiplexing option	l N / B
- RLC logical channel mapping indicator	Not Present
- Number of <u>uplink</u> RLC logical channels	1
- Uplink transport channel type	RACH
- Logical channel identity	3
- CHOICE RLC size list	All
- MAC logical channel priority	4
- Logical channel max loss	θ
 - Downlink RLC logical channel info 	
 Number of <u>downlink</u> RLC logical channels 	1
- Downlink transport channel type	FACH
- Logical channel identity	3
RB information to be affected	(AM DCCH for NAS_DT Low priority)
- RB identity	4
- RB mapping info	
- Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
- Number of uplink RLC logical channels	1
- Uplink transport channel type	RACH
- Logical channel identity	4
	1 -
- CHOICE RLC size list	All
- MAC logical channel priority	5
- Logical channel max loss	0
- Downlink RLC logical channel info	
- Number of downlink RLC logical channels	1

- Logical channel identity RB information to be affected (TM CCCH uplink for RRC) - RB identity - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator Not Present - Number of uplink RLC logical channels **RACH** - Uplink transport channel type - Logical channel identity 5 - CHOICE RLC size list <u>All</u> - MAC logical channel priority - Logical channel max loss 0 RB information to be affected (UM CCCH for downlink RRC) - RB identity - RB mapping info - Information for each multiplexing option - Downlink RLC logical channel info - Number of downlink RLC logical channels - Downlink transport channel type **FACH** - Logical channel identity RB information to be affected (TM BCCH for RRC) - RB identity - RB mapping info - Information for each multiplexing option - Downlink RLC logical channel info - Number of downlink RLC logical channels - Downlink transport channel type **FACH** - Logical channel identity (TM PCCH for RRC) RB information to be affected - RB identity - RB mapping info - Information for each multiplexing option - Downlink RLC logical channel info - Number of downlink RLC logical channels - Downlink transport channel type **PCH** - Logical channel identity **RB with PDCP information list** Not Present UL Transport channel information for all transport channels - TFC subset (This IE is repeated for TFC number.) 0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108 - Allowed Transport Format combination clause 6.10 Parameter Set.) - PRACH TFCS Not Present - CHOICE mode FDD - UL DCH TFCS Not Present Added or Reconfigured UL TrCH information - Transport channel identity 1<u>5</u> - CHOICE Transport channel type Common transport channels - Dynamic Transport format information (This IE is repeated for TFI number) - RLC Size Reference to TS34.108 clause 6.10 Parameter Set - Number of TBs and TTI List (This IE is repeated for TFI number.) - Number of Transport blocks Reference to TS34.108 clause 6.10 Parameter Set **RLC size** Reference to TS34.108 clause 6.10 Parameter Set - CHOICE mode FDD - CHOICE Logical Channel List ALL - Semi-static Transport Format information - Transmission time interval Reference to TS34.108 clause 6.10 Parameter Set - Type of channel coding Reference to TS34.108 clause 6.10 Parameter Set - Coding Rate Reference to TS34.108 clause 6.10 Parameter Set - Rate matching attribute Reference to TS34.108 clause 6.10 Parameter Set - CRC size Reference to TS34.108 clause 6.10 Parameter Set DRAC static information Not Present DL Transport channel information common for all transport channel - SCCPCH TFCS (This IE is repeated for TFC number.) - Normal

FACH

- Downlink transport channel type

- TFCI Field 1 information
- CHOICE TFCS representation
- TFCS addition information
- CHOICE CTFC Size
- CTFC information
- Power offset information
- CHOICE DL parameters
- DL DCH TFCS

Added or Reconfigured DL TrCH information

- Transport channel identity
- CHOICE DL parameters
- TFS
- CHOICE Transport channel type
- Dynamic Transport format information
- RLC Size
- Number of TBs and TTI List
 - Number of Transport blocks
- RLC size
 - CHOICE mode
- CHOICE Logical Channel List
 - Transport block size
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size
 - DCH quality target
 - Transparent mode signalling info

Added or Reconfigured DL TrCH information

- Transport channel identity
- CHOICE DL parameters
- TFS
- CHOICE Transport channel type
 - Dynamic Transport format information
- ŔLC Size
 - Number of TBs and TTI List
 - Number of Transport blocks
 - RLC size
- CHOICE mode
 - CHOICE Logical Channel List
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size
 - DCH quality target
 - Transparent mode signalling info

Added or Reconfigured DL TrCH information

- Transport channel identity
- CHOICE DL parameters
- · TFS
- CHOICE Transport channel type
- Dynamic Transport format information
- RLC Size
 - Number of TBs and TTI List
 - Number of Transport blocks
 - RLC size
- CHOICE mode
- CHOICE Logical Channel List
- Semi-static Transport Format information
- Transmission time interval
- Type of channel coding
- Coding Rate
- Rate matching attribute
- CRC size

Addition

Number of bits used must be enough to cover all combinations of CTFC from clause 6.10.

Refer to TS34.108 clause 6.10 Parameter Set

Independent

Not Present

112

Independent

Common transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set FDD

ALL

Reference to TS34.108 clause 6.10 Parameter Set

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Not Present

Not Present Not Present

132

Independent

Common transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set FDD

ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

Not Present Not Present

<u>14</u>3

Independent

Common transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set FDD

ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

Reference to TS34.108 clause 6.10 Parameter Set

Reference to TS34.108 clause 6.10 Parameter Set

Reference to TS34.108 clause 6.10 Parameter Set

Not Present - DCH quality target - Transparent mode signalling info Not Present Added or Reconfigured DL TrCH information Transport channel identity CHOICE DL parameters Independent TES - CHOICE Transport channel type Common transport channels **Dynamic Transport format information** (This IE is repeated for TFI number) RLC Size Reference to TS34.108 clause 6.10 Parameter Set Number of TBs and TTI List (This IE is repeated for TFI number.) Number of Transport blocks Reference to TS34.108 clause 6.10 Parameter Set RLC size Reference to TS34.108 clause 6.10 Parameter Set - CHOICE mode FDD - CHOICE Logical Channel List ALL Semi-static Transport Format information Transmission time interval Reference to TS34.108 clause 6.10 Parameter Set Type of channel coding Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Coding Rate Rate matching attribute Reference to TS34.108 clause 6.10 Parameter Set - CRC size Reference to TS34.108 clause 6.10 Parameter Set DCH quality target Not Present Transparent mode signalling info Not Present Frequency info - UARFCN uplink(Nu) Reference to TS34.108 clause 6.10 Parameter Set - UARFCN downlink(Nd) Reference to TS34.108 clause 6.10 Parameter Set Maximum allowed UL TX power 33dBm PRACH info (for RACH)Not Present CHOICE channel requirement **CHOICE** mode **FDD Available Signature** '0000 0000 1111 1111'B Available SF Reference to TS34.108 clause 6.10 Parameter Set Scrambling code number **Puncturing Limit** Reference to TS34.108 clause 6.10 Parameter Set - Available Sub Channel number '1111 1111 1111'B **CHOICE Mode FDD** - Downlink PDSCH information Not Present Downlink information common for all radio links Not Present Downlink information for each radio links Not Present Primary CPICH info Primary scrambling code PDSCH with SHO DCH info Not Present PDSCH code mapping Not Present Downlink DPCH info for each RL Not Present Secondary CCPCH info Selection Indicator Not Present Primary CPICH may be used Primary CPICH usage for channel estimation Secondary CPICH info Not Present Secondary scrambling code channelisation code Secondary scrambling code SSDT Indicator Spreading factor Reference to TS34.108 clause 6.10 Parameter Set SF-1(SF is reference to TS34.108 clause 6.10 Parameter Code number Set) Pilot symbol existence **FALSE** TRUE TFCI existence Fixed or Flexible Position **Flexible** Timing offset **TFCS** Not Present **FACH/PCH** information Not Present Not Present References to system information blocks Scheduling information

Information Element	Value/remark
Message Type	
RRC transaction identifier	Arbitrarily selects an integer between 0 and 3
Integrity check info	The presence of this IE is dependent on IXIT statements in
	TS 34.123-2. If integrity protection is indicated to be active,
	this IE is present with the values of the sub IEs as stated
	below. Else, this IE and the sub-IEs are omitted.
- message authentication code	SS calculates the value of MAC-I for this message and
-	writes to this IE.
 RRC message sequence number 	SS provides the value of this IE, from its internal counter.
Integrity protection mode info	Not Present
Ciphering mode info	Not Present
Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
New U-RNTI	Not Present
New C-RNTI	Not Present
RRC State DRX-indicator	NoDRXCELL_DCH
UTRAN DRX cycle length coefficient	Not Present
CN information info	Not Present
URA identity	Not Present
RAB information to reconfigure list	Not Present
RB information to reconfigure list	Not Present
RB information to be affected	(UM DCCH for RRC)
- RB identity	1
- RB mapping info	
- Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
 Number of <u>uplink</u> RLC logical channels 	1
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	<u>5</u> 4
- Logical channel identity	1
- CHOICE RLC size list	All
- MAC logical channel priority	1
- Logical channel max loss	0
- Number of downlink RLC logical channels	1
- Downlink transport channel type	DCH
- <u>DL</u> Transport channel identity	10
- Logical channel identity	1 (AM DOCH for BBC)
RB information to be affected	(AM DCCH for RRC)
- RB identity	2
 RB mapping info Information for each multiplexing option 	
- Information for each multiplexing option - RLC logical channel mapping indicator	Not Present
- Number of <u>uplink</u> RLC logical channels	1
- Number of <u>uplink RLC logical charmers</u> - Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	5 <u>1</u>
- Logical channel identity	2
- CHOICE RLC size list	
- MAC logical channel priority	<u>All</u> 2
- Logical channel max loss	$\frac{2}{\theta}$
- Downlink RLC logical channel info	
- Number of downlink RLC logical channels	1 <u>0</u>
- Downlink transport channel type	DCH
- <u>DL</u> Transport channel identity	1
- Logical channel identity	2
RB information to be affected	(AM DCCH for NAS_DT High priority)
- RB identity	3
- RB mapping info	
- Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
Number of <u>uplink</u> RLC logical channels	1
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	<u>5</u> 4
- Logical channel identity	$\left \frac{3}{3}\right $
- CHOICE RLC size list	
- MAC logical channel priority	<u>All</u> 3
- Logical channel max loss	$\frac{3}{9}$
	I - 7

- Downlink RLC logical channel info	
- Number of <u>downlink</u> RLC logical channels	1
- Downlink transport channel type	DCH
- <u>DL</u> Transport channel identity	10
- Logical channel identity RB information to be affected	(AM DCCH for NAS, DT Low priority)
- RB identity	(AM DCCH for NAS_DT Low priority)
- RB mapping info	7
- Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
- Number of uplink RLC logical channels	1
- Uplink transport channel type	DCH
 <u>UL</u>Transport channel identity 	<u>5</u> 4
- Logical channel identity	4
- CHOICE RLC size list	All
- MAC logical channel priority	4
- Logical channel max loss - Downlink RLC logical channel info	0
- Number of downlink RLC logical channels	1
- Downlink transport channel type	DCH
- <u>DL</u> Transport channel identity	1 <u>0</u>
- Logical channel identity	4
RB information to be affected	(TM DTCH)
- RB identity	<u>10</u> 5
- RB mapping info	
 Information for each multiplexing option 	
- RLC logical channel mapping indicator	Not Present
- Number of <u>uplink</u> RLC logical channels	1
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity - Logical channel identity	<u>12</u> <u>7</u> 4
- CHOICE RLC size list	<u>7+</u> <u>All</u>
- MAC logical channel priority	1
- Logical channel max loss	0
- Downlink RLC logical channel info	
- Number of downlink RLC logical channels	1
- Downlink transport channel type	DCH
- <u>DL</u> Transport channel identity	<u>62</u>
- Logical channel identity	<u>7</u> 4
UL Transport channel information for all transport	
channels	(This IC is reported for TCC number)
- TFC subset - Allowed Transport Format combination	(This IE is repeated for TFC number.) 0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108
- Allowed Transport Format Combination	clause 6.10 Parameter Set.)
- PRACH TFCS	Not Present
- CHOICE mode	FDD
- UL DCH TFCS	(This IE is repeated for TFC number.)
- Normal	,
- TFCI Field 1 information	
- CHOICE TFCS representation	Addition
- TFCS addition information	
- CHOICE CTFC Size	
- CTFC information	Refer to TS34.108 clause 6.10 Parameter Set
- Power offset information	Signallad Cain Factor
- CHOICE Gain Factor	Signalled Gain Factor
- Gain factor •c - Gain factor •d	0
- Gain factor •d - Reference TFC ID	Not Present
- Power offset Pp-m	0dB
Deleted UL TrCH information	Not Present
- Transport channel identity	
Added or Reconfigured UL TrCH information	
- Transport channel identity	<u>5</u> 4
- TFS	
- CHOICE Transport channel type	Dedicated transport channels
- Dynamic Transport format information	(This IE is repeated for TFI number)
- RLC Size	Reference to TS34.108 clause 6.10 Parameter Set
- Number of TBs and TTI List	(This IE is repeated for TFI number.)

- Transmission Time Interval
- Number of Transport blocks
- RLC size
- CHOICE Logical Channel list
- Semi-static Transport Format information
- Transmission time interval
- Type of channel coding
- Coding Rate
- Rate matching attribute
- CRC size

DRAC static information

DL Transport channel information common for all transport channel

- SCCPCH TFCS
- CHOICE DL parameters
- DL DCH TFCS
- Normal
- TFCI Field 1 information
- CHOICE TFCS representation
- TFCS addition information
- CHOICE CTFC Size
- CTFC information
- Power offset information
- CHOICE Gain Factors
- Gain factor •c
- Gain factor •d
- Reference TFC ID
- Power offset Pp-m

Deleted DL TrCH information

Added or Reconfigured DL TrCH information

- Transport channel identity
- CHOICE DL parameters
- UL TrCH Identity
- DCH quality target
- BLER Quality value
- Transparent mode signalling info

Frequency info

- UARFCN uplink(Nu)
- UARFCN downlink(Nd)

Maximum allowed UL TX power

Uplink DPCH info

- -Uplink DPCH power control info
- DPCCH power offset
- PC Preamble
- Power Control Algorithm
- TPC step size
- Scrambling code type
- Scrambling code number
- Number of DPDCH
- spreading factor
- TFCI existence
- Number of FBI bit
- Puncturing Limit

CHOICE Mode

- Downlink PDSCH information

Downlink information common for all radio links

- Downlink DPCH info common for all RL
- -- CHOICE mode
- Timing indicator
- CFN-targetSFN frame offset
- Downlink DPCH power control information
- DPC mode
- DL rate matching restriction information
- Spreading factor
- Fixed or Flexible Position
- TFCI existence

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Not Present

Not Present Independent

(This IE is repeated for TFC number.)

Addition

Number of bits used must be enough to cover all combinations of CTFC from clause 6.10. Refer to TS34.108 clause 6.10 Parameter Set

Signalled Gain Factor

0

0

Not Present

0dB

Not Present

101

SameAsUL

<u>5</u>1

<u>-6.3</u>0.00

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set 33dBm

-6dB

8slot

Algorithm1

1dB

Long

0 (0 to 16777215)

Not Present(1)

SF is reference to TS34.108 clause 6.10 Parameter Set TRUE

Not Droppint/O

Not Present(0)

Reference to TS34.108 clause 6.10 Parameter Set

FDD

Not Present

Maintain Not Present

0 (single)

Not Present

Reference to TS34.108 clause 6.10 Parameter Set

Flexible TRUE

Not Present - Number of bits for Pilot bits(SF=128,256) - DPCH compressed mode info - TGPSI - TGPS Status Flag Inactive (Current CFN + (256 - TTI/10msec)) mod 256 - TGCFN - Transmission gap pattern sequence configuration parameters - TGMP **FDD Measurement** - TGPRC **TGCFN** (Current CFN + (256 - TTI/10msec)) mod 256 - TGSN 8 - TGL1 10 - TGL2 5 15 - TGD - TGPL1 35 35 - TGPL2 - RPP Mode 1 - ITP Mode 1 - UL/DL Mode DL SF/2 - Downlink compressed mode method - Uplink compressed mode method Not Present - Downlink frame type - DeltaSIR1 2.0 - DeltaSIRafter1 1.0 Not Present - DeltaSIR2 Not Present - DeltaSIRafter2 - TX Diversity mode None - SSDT information Not Present - Default DPCH Offset Value 0 Downlink information for each radio links - Primary CPICH info 100 - Primary scrambling code 100 - PDSCH with SHO DCH info Not Present - PDSCH code mapping Not Present - Downlink DPCH info for each RL - Primary CPICH usage for channel estimation Primary CPICH may be used - DPCH frame offset 0 chips - Secondary CPICH info Not Present - Secondary scrambling code - channelisation code - DL channelisation code - Secondary scrambling code - Spreading factor Reference to TS34.108 clause 6.10 Parameter Set - Code number SF-1(SF is reference to TS34.108 clause 6.10 Parameter Set) - Scrambling code change No change - TPC combination index - SSDT Cell Identity -a - Closed loop timing adjustment mode Not Present - Secondary CCPCH info Not Present - TFCS Not Present - FACH/PCH information Not Present - References to system information blocks Not Present Scheduling information

	Information Element	Value/remark
Н	Message Type	Valueremark
	RRC transaction identifier	Arbitrarily selects an integer between 0 and 3
	ntegrity check info	The presence of this IE is dependent on IXIT statements in
		TS 34.123-2. If integrity protection is indicated to be active,
		this IE is present with the values of the sub IEs as stated
		below. Else, this IE and the sub-IEs are omitted.
	- message authentication code	SS calculates the value of MAC-I for this message and
	550	writes to this IE.
١,	- RRC message sequence number ntegrity protection mode info	SS provides the value of this IE, from its internal counter. Not Present
	Ciphering mode info	Not Present
	Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
	New U-RNTI	Not Present
l	New C-RNTI	Not Present
ا ا	RRC State DRX indicator	NoDRXCELL_DCH
	JTRAN DRX cycle length coefficient	Not Present
	CN information info	Not Present
	JRA identity	Not Present
	RAB information to reconfigure list RB information to reconfigure list	Not Present Not Present
	RB information to be affected	(UM DCCH for RRC)
1 '	- RB information to be affected	(ON BOOT TO TATO)
1	- RB identity	1
	- RB mapping info	
	- Information for each multiplexing option	
	 RLC logical channel mapping indicator 	Not Present
	- Number of <u>uplink</u> RLC logical channels	1
	- Uplink transport channel type	DCH
!	 <u>UL</u> Transport channel identity Logical channel identity 	<u>5</u> 1 1
ı	- CHOICE RLC size list	AII
! -	- MAC logical channel priority	<u>/···</u> 1
-	- Logical channel max loss	0
-	- Downlink RLC logical channel info	
	- Number of <u>downlink</u> RLC logical channels	1
ı	Downlink transport channel type DL Transport channel identity	DCH
	- <u>DL</u> transport channel identity - Logical channel identity	1 <u>0</u> 1
П	RB information to be affected	(AM DCCH for RRC)
	- RB identity	2
	- RB mapping info	
	- Information for each multiplexing option	
-	- RLC logical channel mapping indicator	Not Present
I	- Number of <u>uplink</u> RLC logical channels	1 DCH
ıl	- Uplink transport channel type - UL Transport channel identity	
'	- Logical channel identity	<u>5</u> 4 2
	- CHOICE RLC size list	
. []	- MAC logical channel priority	<u>All</u> 2
-	- Logical channel max loss	θ
-	- Downlink RLC logical channel info	1
1	 Number of <u>downlink</u> RLC logical channels Downlink transport channel type 	1 DCH
ı	- DL Transport channel identity	1 <u>0</u>
1	- Logical channel identity	2
	RB information to be affected	(AM DCCH for NAS_DT High priority)
	- RB identity	3
	- RB mapping info	
	 Information for each multiplexing option RLC logical channel mapping indicator 	Not Present
-	- Number of <u>uplink</u> RLC logical channels	1
1	- Uplink transport channel type	DCH
	- <u>UL</u> Transport channel identity	<u>5</u> 4
	- Logical channel identity	3
IL.	- CHOICE RLC size list	<u>All</u>

- MAC logical channel priority	3
- Logical channel max loss	0
- Downlink RLC logical channel info	
- Number of <u>downlink</u> RLC logical channels	1
- Downlink transport channel type	DCH
- <u>DL</u> Transport channel identity	10
- Logical channel identity	(AM DOOLL for NACE DT Love princity)
RB information to be affected	(AM DCCH for NAS_DT Low priority)
- RB identity	4
- RB mapping info	
- Information for each multiplexing option	Not Droppet
- RLC logical channel mapping indicator	Not Present
- Number of <u>uplink</u> RLC logical channels	DCH
- Uplink transport channel type	
- <u>UL</u> Transport channel identity	5 <u>4</u> 4
- Logical channel identity	All
- CHOICE RLC size list - MAC logical channel priority	All 4
- Logical channel max loss	0
- Downlink RLC logical channel info	9
- Number of <u>downlink</u> RLC logical channels	1
- Downlink transport channel type	DCH
- DL Transport channel identity	1 <u>0</u>
- Logical channel identity	10/4
RB information to be affected	(DTCH TM)
- RB identity	10 5
- RB mapping info	100
- Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
- Number of <u>uplink</u> RLC logical channels	1
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	12
- Logical channel identity	<u>7</u> 4
- CHOICE RLC size list	
- MAC logical channel priority	1
- Logical channel max loss	0
 Downlink RLC logical channel info 	
- Number of downlink RLC logical channels	1
- Downlink transport channel type	DCH
- <u>DL</u> Transport channel identity	<u>62</u>
- Logical channel identity	<u>7</u> 4
RB information to be affected	(DTCH TM)
- RB identity	<u>11</u> 6
- RB mapping info	
- Information for each multiplexing option	Not Droppet
- RLC logical channel mapping indicator	Not Present
- Number of <u>uplink</u> RLC logical channels	1 DCH
- Uplink transport channel type	
- <u>UL_</u> Transport channel identity - Logical channel identity	<u>2</u> 3 <u>8</u> 4
- CHOICE RLC size list	<u>0+</u> <u>All</u>
- MAC logical channel priority	<u>All</u> 1
- logical channel max loss	l d
- Downlink RLC logical channel info	Ĭ
- Number of downlink RLC logical channels	1
- Downlink transport channel type	DCH
- <u>DL</u> Transport channel identity	73
- Logical channel identity	84
RB information to be affected	(This IE is needed for 12.2 kbps and 10.2 kbps)
- RB identity	127
- RB mapping info	_
- Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
- Number of <u>uplink</u> RLC logical channels	1
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	<u>3</u> 4
- Logical channel identity	<u>9</u> 4
- CHOICE RLC size list	All

- MAC logical channel priority
 Logical channel max loss
- Downlink RLC logical channel info
- Number of downlink RLC logical channels
- Downlink transport channel type
- DL Transport channel identity
- Logical channel identity

UL Transport channel information for all transport channels

- TFC subset
- Allowed Transport Format combination
- PRACH TFCS
- CHOICE mode
- UL DCH TFCS
- Normal
- TFCI Field 1 information
- CHOICE TFCS representation
- TFCS addition information
- CHOICE CTFC Size
- CTFC information
- Power offset information
- CHOICE Gain Factors
- Gain factor •c
- Gain factor •d
- Reference TFC ID
- Power offset Pp-m

Deleted UL TrCH information

- Transport channel identity

Added or Reconfigured UL TrCH information

- Transport channel identity
- TFS
- CHOICE Transport channel type
- Dynamic Transport format information
- RLC Size
- Number of TBs and TTI List
 - Transmission Time Interval
 - Number of Transport blocks
 - RLC size
 - CHOICE Logical Channel list
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size

DRAC static information

DL Transport channel information common for all transport channel

- SCCPCH TFCS
- CHOICE DL parameters
- DL DCH TFCS
- Normal
- TFCI Field 1 information
- CHOICE TFCS representation
- TFCS addition information
- CHOICE CTFC Size
- CTFC information
- Power offset information

Deleted DL TrCH information

Added or Reconfigured DL TrCH information

- Transport channel identity
- CHOICE DL parameters
- UL TrCH Identity
- DCH quality target
- BLER Quality value

0

1

DCH

<u>8</u>4 91

(This IE is repeated for TFC number.)

0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108 clause 6.10 Parameter Set.)

Not Present

FDD

(This IE is repeated for TFC number.)

Addition

Number of bits used must be enough to cover all combinations of CTFC from clause 6.10. Refer to TS34.108 clause 6.10 Parameter Set

Signalled Gain Factor

0

0

Not Present

0dB

Not Present

54

Dedicated transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Not Present

Not Present Independent

(This IE is repeated for TFC number.)

Addition

Number of bits used must be enough to cover all combinations of CTFC from clause 6.10. Refer to TS34.108 clause 6.10 Parameter Set

Not Present Not Present

<u>10</u>1

SameAsUL

51

<u>-6.3</u>0.00

- Transparent mode signalling info Not Present Frequency info - UARFCN uplink(Nu) Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set - UARFCN downlink(Nd) Maximum allowed UL TX power 33dBm Uplink DPCH info - Uplink DPCH power control info - DPCCH power offset -6dB - PC Preamble 8slot - Power Control Algorithm Algorithm1 - TPC step size 1dB - Scrambling code type Long - Scrambling code number 0 (0 to 16777215) - Number of DPDCH Not Present(1) - spreading factor SF is reference to TS34.108 clause 6.10 Parameter Set - TFCI existence **TRUE** - Number of FBI bit Not Present(0) - Puncturing Limit Reference to TS34.108 clause 6.10 Parameter Set **CHOICE Mode** FDD - Downlink PDSCH information Not Present Downlink information common for all radio links - Downlink DPCH info common for all RL CHOICE mode **EDD** - Timing indicator **Maintain** - CFN-targetSFN frame offset Not Present - Downlink DPCH power control information - DPC mode 0 (single) - DL rate matching restriction information Not Present - Spreading factor Reference to TS34.108 clause 6.10 Parameter Set - Fixed or Flexible Position Flexible - TFCI existence TRUE - Number of bits for Pilot bits(SF=128,256) Not Present - DPCH compressed mode info - TGPSI - TGPS Status Flag Inactive - TGCFN (Current CFN + (256 - TTI/10msec)) mod 256 - Transmission gap pattern sequence configuration parameters - TGMP **FDD Measurement** - TGPRC **TGCFN** (Current CFN + (256 - TTI/10msec)) mod 256 - TGSN - TGL1 10 - TGL2 5 15 - TGD - TGPL1 35 - TGPL2 35 - RPP Mode 1 - ITP Mode 1 - UL/DL Mode - Downlink compressed mode method SF/2 Not Present - Uplink compressed mode method - Downlink frame type - DeltaSIR1 2.0 - DeltaSIRafter1 1.0 Not Present - DeltaSIR2 - DeltaSIRafter2 Not Present - TX Diversity mode None - SSDT information Not Present - S field - Code Word Set - Default DPCH Offset Value 0 Downlink information for each radio links - Primary CPICH info - Primary scrambling code 100 - PDSCH with SHO DCH info Not Present - PDSCH code mapping Not Present - Downlink DPCH info for each RL

- Primary CPICH usage for channel estimation - DPCH frame offset - Secondary CPICH info - DL channelisation code

Secondary scrambling codeSpreading factor

- Code number

- Scrambling code change - TPC combination index

- SSDT Cell Identity

- Closed loop timing adjustment mode

- Secondary CCPCH info

- TFCS

- FACH/PCH information

- References to system information blocks

Scheduling information

Primary CPICH may be used

0 chips Not Present

Reference to TS34.108 clause 6.10 Parameter Set SF-1(SF is reference to TS34.108 clause 6.10 Parameter

No change

0

-a

Not Present Not Present

Not Present Not Present

Not Present

Information Element	Value/remark
Message Type	
RRC transaction identifier	Arbitrarily selects an integer between 0 and 3
Integrity check info	The presence of this IE is dependent on IXIT statements in
	TS 34.123-2. If integrity protection is indicated to be active,
	this IE is present with the values of the sub IEs as stated
	below. Else, this IE and the sub-IEs are omitted.
 message authentication code 	SS calculates the value of MAC-I for this message and
	writes to this IE.
- RRC message sequence number	SS provides the value of this IE, from its internal counter.
Integrity protection mode info	Not Present
Ciphering mode info	Not Present
Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
New U-RNTI	Not Present
New C-RNTI	Not Present
RRC State DRX indicator	NoDRXCELL_DCH
UTRAN DRX cycle length coefficient	Not Present
CN information info	Not Present
URA identity	Not Present
RAB information to reconfigure list	Not Present
RB information to reconfigure list	Not Present
RB information to be affected	(UM DCCH for RRC)
- RB identity	1
 RB mapping info Information for each multiplexing option 	
- RLC logical channel mapping indicator - Number of uplink RLC logical channels	
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	<u>5</u> 4
- Logical channel identity	<u>3</u> + 1
- CHOICE RLC size list	All
- MAC logical channel priority	1
- Logical channel max loss	Û
- Downlink RLC logical channel info	
- Number of downlink RLC logical chann	els 1
- Downlink transport channel type	DCH
- <u>DL</u> Transport channel identity	1 <u>0</u>
- Logical channel identity	1
RB information to be affected	(AM DCCH for RRC)
- RB identity	2
- RB mapping info	
- Information for each multiplexing option	
- RLC logical channel mapping indicator	<u> </u>
- Number of <u>uplink</u> RLC logical channels	
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	<u>5</u> 4 2
- Logical channel identity	
- CHOICE RLC size list	All 2
- MAC logical channel priority	
- Logical channel max loss	0
- Downlink RLC logical channel info	
- Number of <u>downlink</u> RLC logical chann	els 1 DCH
Downlink transport channel type DL Transport channel identity	1 <u>0</u>
- Logical channel identity	2
RB information to be affected	(AM DCCH for NAS_DT High priority)
- RB identity	3
- RB mapping info	Ŭ
- Information for each multiplexing option	
- RLC logical channel mapping indicator	
Number of <u>uplink</u> RLC logical channels	
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	<u>5</u> 4
- Logical channel identity	$\frac{\omega}{3}$
- CHOICE RLC size list	All
	1

- MAC logical channel priority	3
- Logical channel max loss	θ
- <u>Downlink RLC logical channel info</u> - Number of <u>downlink RLC logical channels</u>	1
- Downlink transport channel type	DCH
- <u>DL</u> Transport channel identity	1 <u>0</u>
- Logical channel identity	3
RB information to be affected	(AM DCCH for NAS_DT Low priority)
- RB identity	4
- RB mapping info	
 Information for each multiplexing option 	
- RLC logical channel mapping indicator	Not Present
- Number of <u>uplink</u> RLC logical channels	1
Uplink transport channel type UL Transport channel identity	DCH
- Logical channel identity	<u>5</u> 4 4
- CHOICE RLC size list	All
- MAC logical channel priority	4
- Logical channel max loss	θ
- Downlink RLC logical channel info	
- Number of <u>downlink</u> RLC logical channels	1
- Downlink transport channel type	DCH
- <u>DL</u> Transport channel identity	10
- Logical channel identity RB information to be affected	4 (AM DTCH)
- RB identity	20 5
- RB mapping info	==-
- Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
 Number of <u>uplink</u> RLC logical channels 	1
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	$\frac{12}{74}$
- Logical channel identity - CHOICE RLC size list	74 All
- MAC logical channel priority	1
- Logical channel max loss	ů
- Downlink RLC logical channel info	
- Number of downlink RLC logical channels	1
- Downlink transport channel type	DCH
- <u>DL</u> Transport channel identity	<u>62</u>
- Logical channel identity	<u>7</u> 4
UL Transport channel information for all transport	
channels - TFC subset	(This IE is repeated for TFC number.)
- Allowed Transport Format combination	0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108
7 monou Transport Format combination	clause 6.10 Parameter Set.)
- PRACH TFCS	Not Present
- CHOICE mode	FDD
- UL DCH TFCS	(This IE is repeated for TFC number.)
- Normal	
- TFCI Field 1 information	Addition
- CHOICE TFCS representation - TFCS addition information	Addition
- CHOICE CTFC Size	Number of bits used must be enough to cover all
GHOIGE OTH G GIZE	combinations of CTFC from clause 6.10.
- CTFC information	Refer to TS34.108 clause 6.10 Parameter Set
- Power offset information	
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor ∙c	0
- Gain factor •d	0
- Reference TFC ID	Not Present
- Power offset Pp-m	0dB
Deleted UL TrCH information - Transport channel identity	Not Present
Added or Reconfigured UL TrCH information	I NOT I JESEUL
- Transport channel identity	<u>5</u> 4
- TFS	
- CHOICE Transport channel type	Dedicated transport channels

- Dynamic Transport format information
- RLC Size
- Number of TBs and TTI List
 - Transmission Time Interval
 - Number of Transport blocks
- **RLC size**
- CHOICE Logical Channel list
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size

DRAC static information

DL Transport channel information common for all transport channel

- SCCPCH TFCS
- CHOICE DL parameters
- DL DCH TFCS
- Normal
- TFCI Field 1 information
- CHOICE TFCS representation
- TFCS addition information
- CHOICE CTFC Size
- CTFC information
- Power offset information
- CHOICE Gain Factors
- Gain factor •c
- Gain factor •d
- Reference TFC ID
- Power offset Pp-m

Deleted DL TrCH information

- Transport channel identity Added or Reconfigured DL TrCH information

- Transport channel identity
- CHOICE DL parameters
- CHOICE Transport channel type
- Dynamic Transport format information
- RLC Size
 - Number of TBs and TTI List
 - Transmission Time Interval
 - Number of Transport blocks
- RLC size
 - CHOICE Logical Channel list
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size
 - DCH quality target
 - BLER Quality value
 - Transparent mode signalling info

Frequency info

- UARFCN uplink(Nu)
- UARFCN downlink(Nd)

Maximum allowed UL TX power Uplink DPCH info

- -Uplink DPCH power control info
- -DPCCH Power offset
- -PC Preamble
- -Power Control Algorithm
- -TPC step size
- Scrambling code type
- Scrambling code number
- Number of DPDCH

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Not Present

Not Present

Independent

(This IE is repeated for TFC number.)

Additon

Number of bits used must be enough to cover all combinations of CTFC from clause 6.10. Refer to TS34.108 clause 6.10 Parameter Set

Signalled Gain Factor

0

0

Not Present

0dB

Not Present

101

Independent

Dedicated transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set <u>AL</u>L

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

-6.30.00Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set 33dBm

-6dB 8slot Algorithm1 1dB Long

0 (0 to 16777215) Not Present(1)

- spreading factor

- TFCI existence

- Number of FBI bits

- Puncturing Limit

CHOICE Mode

- Downlink PDSCH information

Downlink information common for all radio links Downlink information for each radio links

Williak Illioitiation for each facili

- Primary CPICH info

- Primary scrambling code

- PDSCH with SHO DCH info

- PDSCH code mapping

- Downlink DPCH info for each RL

- Primary CPICH usage for channel estimation

- DPCH frame offset

- Secondary CPICH info

- DL channelisation code

- Secondary scrambling code

- Spreading factor

- Code number

- Scrambling code change

- TPC combination index

- SSDT Cell Identity

- Closed loop timing adjustment mode

- Secondary CCPCH info

- TFCS

- FACH/PCH information

- References to system information blocks

Scheduling information

SF is reference to TS34.108 clause 6.10 Parameter Set

TRUE

Not Present(0)

Reference to TS34.108 clause 6.10 Parameter Set

FDD

Not Present

Not Present

100

Not Present

Not Present

Primary CPICH may be used

0 chips

Not Present

1

Reference to TS34.108 clause 6.10 Parameter Set

SF-1(SF is reference to TS34.108 clause 6.10 Parameter

Set)

No change

0

-a

Not Present

Not Present Not Present

Not Present

Not Present

Contents of RADIO BEARER RECONFIGURATION message: AM or UM (Packet to CELL_DCH from CELL_FACH in PS)

Ī	Information Element	Value/remark
Ī	Message Type	
	RRC transaction identifier	Arbitrarily selects an integer between 0 and 3
	Integrity check info	The presence of this IE is dependent on IXIT statements in
		TS 34.123-2. If integrity protection is indicated to be active,
		this IE is present with the values of the sub IEs as stated
		below. Else, this IE and the sub-IEs are omitted.
	- message authentication code	SS calculates the value of MAC-I for this message and
		writes to this IE.
	- RRC message sequence number	SS provides the value of this IE, from its internal counter.
	Integrity protection mode info	Not Present
	Ciphering mode info	Not Present
	Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
	New U-RNTI	Not Present
	New C-RNTI	Not Present
į	RRC State DRX indicator	NoDRXCELL_DCH
	UTRAN DRX cycle length coefficient	Not Present
	CN information info	Not Present
	URA identity	Not Present
	RAB information to reconfigure list	Not Present
	RB information to reconfigure list	(LIM DOCK for DDC)
	- RB information to reconfigure	(UM DCCH for RRC)
	- RB identity	1 Not Present
П	- PDCP info	Not Present Not Present
I	- <u>CHOICE</u> RLC info <u>type</u>	Not Present
	 RB mapping info Information for each multiplexing option 	
ıl	- RLC logical channel mapping indicator	Not Present
	- Number of <u>uplink</u> RLC logical channels	1
1	- Uplink transport channel type	DCH
ıl	- <u>UL</u> Transport channel identity	<u>5</u> 4
1	- Logical channel identity	<u>⊻</u> ⊤ 1
ıl	- CHOICE RLC size list	AII
!	- MAC logical channel priority	7.11 1
I	- Logical channel max loss	0
	- Downlink RLC logical channel info	
	- Number of downlink RLC logical channels	1
	 Downlink transport channel type 	DCH
	 <u>DL</u>Transport channel identity 	1 <u>0</u>
	 Logical channel identity 	1
	- RB stop/continue	Not Present
	- RB information to reconfigure	(AM DCCH for RRC)
	- RB identity	2
۱	- PDCP info	Not Present
Ц	- <u>CHOICE</u> RLC info <u>type</u>	Not Present
	- RB mapping info	
d	- Information for each multiplexing option	Not Propert
II	- RLC logical channel mapping indicator	Not Present
Ц	 Number of <u>uplink</u> RLC logical channels Uplink transport channel type 	<u>5</u> 4 DCH
ıl	- UL Transport channel identity	1
4	- <u>oc</u> transport channel identity - Logical channel identity	2
П	- CHOICE RLC size list	<u>AII</u>
1	- MAC logical channel priority	2
ıl	- Logical channel max loss	0
II	- Downlink RLC logical channel info	-
	- Number of downlink RLC logical channels	1
1	- Downlink transport channel type	DCH
Il	- <u>DL_Transport channel identity</u>	<u>104</u>
1	- Logical channel identity	2
H	- RB stop/continue	Not Present
1	- RB information to reconfigure	(AM DCCH for NAS_DT High priority)
	- RB identity	3
	- PDCP info	Not Present
$\ $	- CHOICE RLC info type	Not Present
-		

- RB mapping info	
- Information for each multiplexing option	
	Not Dropout
- RLC logical channel mapping indicator	Not Present
 Number of <u>uplink</u> RLC logical channels 	1
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	<u>5</u> 4
- Logical channel identity	3
- CHOICE RLC size list	All
- MAC logical channel priority	3
	θ
 Number of <u>downlink</u> RLC logical channels 	1
- Downlink transport channel type	DCH
- <u>DL</u> Transport channel identity	1 <u>0</u>
- Logical channel identity	3
- RB stop/continue	Not Present
- RB information to reconfigure	(AM DCCH for NAS_DT Low priority)
- RB identity	4
- PDCP info	Not Present
- RLC info	Not Present
- RB mapping info	
- Information for each multiplexing option	
	Not Droppet
- RLC logical channel mapping indicator	Not Present
 Number of <u>uplink</u> RLC logical channels 	1
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	<u>5</u> 4
 Logical channel identity 	4
- CHOICE RLC size list	All
- MAC logical channel priority	4
- Logical channel max loss	0
- Downlink RLC logical channel info	
	4
- Number of downlink RLC logical channels	1
 Downlink transport channel type 	DCH
- <u>DL</u> Transport channel identity	1 <u>0</u>
	1 · <u>· ·</u> 4
- Logical channel identity	· ·
- RB information to reconfigure	(AM DTCH)
- RB identity	20 5
- PDCP info	Not Present
- CHOICE RLC info type	Not Present
- RB mapping info	
- Information for each multiplexing option	
	Not Dropout
- RLC logical channel mapping indicator	Not Present
 Number of <u>uplink</u> RLC logical channels 	1
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	12
- Logical channel identity	<u>7</u> 4
- CHOICE RLC size list	All
- MAC logical channel priority	1
- Logical channel max loss	0
- Downlink RLC logical channel info	
- Number of downlink RLC logical channels	1
- Downlink transport channel type	DCH
- <u>DL</u> Transport channel identity	<u>62</u>
- Logical channel identity	74
- RB stop/continue	Not Present
RB information to be affected	Not Present
UL Transport channel information for all transport	
channels	
- TFC subset	(This IE is repeated for TFC number.)
- Allowed Transport Format combination	0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108
/ morrow franciport i office combination	
	clause 6.10 Parameter Set.)
- PRACH TFCS	Not Present
- CHOICE mode	FDD
- UL DCH TFCS	
	(This IE is repeated for TFC number.)
- Normal	
- TFCI Field 1 information	
- CHOICE TFCS representation	
- TFCS addition information	
- I FL > 3000100 INTORMATION	
- CHOICE CTFC Size	Number of bits used must be enough to cover all

- CTFC information
- Power offset information
- CHOICE Gain Factors
- Gain factor •c
- Gain factor •d
- Reference TFC ID
- Power offset Pp-m

Deleted UL TrCH information

- Transport channel identity

Added or Reconfigured UL TrCH information

- Transport channel identity
- TFS
- CHOICE Transport channel type
- Dynamic Transport format information
- RLC Size
- Number of TBs and TTI List
 - Transmission Time Interval
 - Number of Transport blocks
 - RLC size
- CHOICE Logical Channel list
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size

Added or Reconfigured UL TrCH information

- Transport channel identity
- TFS
- CHOICE Transport channel type
 - Dynamic Transport format information
- RLC Size
 - Number of TBs and TTI List
 - Transmission Time Interval
 - _- Number of Transport blocks
- RLC size
 - CHOICE Logical Channel list
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size

DRAC static information

DL Transport channel information common for all transport channel

- SCCPCH TFCS
- CHOICE DL parameters
- DL DCH TFCS
- Normal
- TFCI Field 1 information
- CHOICE TFCS representation
- TFCS addition information
- CHOICE CTFC Size
- CTFC information
- Power offset information

Deleted DL TrCH information

- Transport channel identity
- Deleted DL TrCH information
 - Transport channel identity
 - Transport channel identity

Added or Reconfigured DL TrCH information

- Transport channel identity
- CHOICE DL parameters
- TFS
- CHOICE Transport channel type

combinations of CTFC from clause 6.10.

Refer to TS34.108 clause 6.10 Parameter Set

Signalled Gain Factor

0

Not Present

0dB

15Not Present

51

Dedicated transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

<u>ALL</u>

Reference to TS34.108 clause 6.10 Parameter Set

12

Dedicated transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

Reference to TS34.108 clause 6.10 Parameter Set

Reference to TS34.108 clause 6.10 Parameter Set

Reference to TS34.108 clause 6.10 Parameter Set

Reference to TS34.108 clause 6.10 Parameter Set

Reference to TS34.108 clause 6.10 Parameter Set

Not Present

Not Present

Independent

(This IE is repeated for TFC number.)

Addition

Number of bits used must be enough to cover all combinations of CTFC from clause 6.10.

Refer to TS34.108 clause 6.10 Parameter Set Not Present

123

<u>13</u>4

<u>14</u>

101

Independent

Dedicated transport channels

- Dynamic Transport format information
- RLC Size
- Number of TBs and TTI List
- Transmission Time Interval
- Number of Transport blocks
- RLC size
- CHOICE Logical Channel list
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size
 - DCH quality target
 - BLER Quality value
- Transparent mode signalling info

Added or Reconfigured DL TrCH information

- Transport channel identity
- CHOICE DL parameters
- TFS
- CHOICE Transport channel type
 - Dynamic Transport format information
- RLC Size
 - Number of TBs and TTI List
 - Transmission Time Interval
 - Number of Transport blocks
 - RLC size
 - CHOICE Logical Channel list
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size
 - DCH quality target
 - BLER Quality value
 - Transparent mode signalling info

Frequency info

- UARFCN uplink(Nu)
- UARFCN downlink(Nd)

Maximum allowed UL TX power

Uplink DPCH info

- -Uplink DPCH power control info
- -DPCCH Power offset
- -PC Preamble
- -Power Control Algorithm
- -TPC step size
- Scrambling code type
- Scrambling code number
- Number of DPDCH
- spreading factor
- TFCI existence
- Number of FBI bits
- Puncturing Limit

CHOICE Mode

- Downlink PDSCH information

Downlink information common for all radio links

- Downlink DPCH info common for all RL
- CHOICE mode
 - Timing indicator
 - CFN-targetSFN frame offset
 - Downlink DPCH power control information
 - DPC mode
 - DL rate matching restriction information
 - Spreading factor
 - Fixed or Flexible Position
 - TFCI existence
 - Number of bits for Pilot bits(SF=128,256)

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

ALL

Reference to TS34.108 clause 6.10 Parameter Set

-6.30.00

Not Present

62

Independent

Dedicated transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

Reference to TS34.108 clause 6.10 Parameter Set

-6.30.00

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set 33dBm

-6dB

8slot

Algorithm1

1dB

Long

0 (0 to 16777215)

Not Present(1)

SF is reference to TS34.108 clause 6.10 Parameter Set

TRUE

Not Present(0)

Reference to TS34.108 clause 6.10 Parameter Set

FDD

Not Present

FDD

<u>Maintain</u>

Not Present

0 (single) Not Present

Reference to TS34.108 clause 6.10 Parameter Set

Flexible

TRUE

Reference to TS34.108 clause 6.10 Parameter Set

- DPCH compressed mode info - TGPSI - TGPS Status Flag Inactive - TGCFN (Current CFN + (256 - TTI/10msec)) mod 256 - Transmission gap pattern sequence configuration parameters - TGMP **FDD Measurement** - TGPRC 62 - TGCFN (Current CFN + (256 - TTI/10msec)) mod 256 - TGSN 8 - TGL1 10 - TGL2 5 - TGD 15 - TGPL1 35 - TGPL2 35 - RPP Mode 1 - ITP Mode 1 - UL/DL Mode DL SF/2 - Downlink compressed mode method - Uplink compressed mode method Not Present - Downlink frame type 2.0 - DeltaSIR1 - DeltaSIRafter1 1.0 - DeltaSIR2 Not Present Not Present - DeltaSIRafter2 - TX Diversity mode None - SSDT information Not Present - Default DPCH Offset Value 0 Downlink information for each radio links - Primary CPICH info - Primary scrambling code 100 - PDSCH with SHO DCH info Not Present - PDSCH code mapping Not Present - Downlink DPCH info for each RL - Primary CPICH usage for channel estimation Primary CPICH may be used - DPCH frame offset 0 chips - Secondary CPICH info Not Present - DL channelisation code - Secondary scrambling code - Spreading factor Reference to TS34.108 clause 6.10 Parameter Set - Code number SF-1(SF is reference to TS34.108 clause 6.10 Parameter Set) - Scrambling code change No change - TPC combination index 0 - SSDT Cell Identity -a - Closed loop timing adjustment mode - Secondary CCPCH info Not Present Not Present - TFCS Not Present - FACH/PCH information Not Present - References to system information blocks Not Present

Scheduling information

[Information Element	Value/remark
Ţ	Message Type	
	RRC transaction identifier	Arbitrarily selects an integer between 0 and 3
	Integrity check info	The presence of this IE is dependent on IXIT statements
		in TS 34.123-2. If integrity protection is indicated to be
		active, this IE is present with the values of the sub IEs as
		stated below. Else, this IE and the sub-IEs are omitted.
	- message authentication code	SS calculates the value of MAC-I for this message and
	DDC magazaga aaguanaa numbar	writes to this IE.
	- RRC message sequence number Integrity protection mode info	SS provides the value of this IE, from its internal counter. Not Present
	Ciphering mode info	Not Present
	Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
	New U-RNTI	Not Present
	New C-RNTI	Not Present
	RRC State DRX-indicator	NoDRXCELL_FACH
	UTRAN DRX cycle length coefficient	Not Present
	CN information info	Not Present
	URA identity	Not Present
	RAB information to reconfigure list	Not Present
	RB information to reconfigure list	
	- RB information to reconfigure	(UM DCCH for RRC)
	- RB identity	1 Not Droppet
ı	- PDCP info	Not Present
	- <u>CHOICE</u> RLC info <u>type</u>	Not Present
	 RB mapping info Information for each multiplexing option 	
ı	- RLC logical channel mapping indicator	Not Present
	- Number of <u>uplink</u> RLC logical channels	1
	- Uplink transport channel type	RACH
	- Logical channel identity	1
i	- CHOICE RLC size list	<u>All</u>
	- MAC logical channel priority	2
	- Logical channel max loss	θ
	- Downlink RLC logical channel info	
	- Number of downlink RLC logical channels	1
	- Downlink transport channel type	FACH
ı	Logical channel identityRB stop/continue	1 Not Present
ı	- RB information to reconfigure	(AM DCCH for RRC)
	- RB identity	2
	- PDCP info	Not Present
	- CHOICE RLC info type	Not Present
1	- RB mapping info	
	 Information for each multiplexing option 	
	- RLC logical channel mapping indicator	Not Present
	 Number of <u>uplink</u> RLC logical channels 	1
	- Uplink transport channel type	RACH
١	- Logical channel identity	2
۱	- CHOICE RLC size list	<u>All</u> 3
١	- MAC logical channel priority	0
	- Downlink RLC logical channel info	∀
	- Number of downlink RLC logical channels	1
	- Downlink transport channel type	FACH
	- Logical channel identity	2
	- RB stop/continue	Not Present
	- RB information to reconfigure	(AM DCCH for NAS_DT High priority)
	- RB identity	3
	- PDCP info	Not Present
	- <u>CHOICE</u> RLC info type	Not Present
	- RB mapping info	
ı	- Information for each multiplexing option	Not Dropont
	- RLC logical channel mapping indicator	Not Present
L	 Number of <u>uplink</u> RLC logical channels 	1

- Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel max-lose - Downlink RLC logical channels - PDCP linfo - RLC logical channel spriority - Logical channel dentity - RS stock-continue - RS mapping info - Information for each multiplexing option - RLC logical channel rappe in logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channels - Downlink RLC logical channels - RS information to reconfigure - RB information to reconfigure - RB information for each multiplexing option - RLC logical channel mapping indicator		
- Logical channel identity - CHOICE RLC size ilist - MAC logical channel priority - Logical channel mapsing indicator - Runber of juglink RLC logical channels - Downlink transport channel type - Runber of juglink RLC logical channels - PDCP info - RLC logical channel mapping indicator - Number of juglink RLC logical channels - Downlink transport channel type - Logical channel mapping indicator - Number of juglink RLC logical channels - Downlink RLC logical channels - Downlink transport channel priority - Logical channel mapping indicator - Number of juglink RLC logical channels - Downlink RLC logical channels - RB identity - POCP info - RLC logical channel with process - RB mapping info - Information for each multiplexing option - RLC logical channel for - Uplick transport channel fyre - Logical channel mapping indicator - RB identity - POCP info - CHOICE RLC info type - Logical channel mapping indicator - RB mormation for each multiplexing option - RLC logical channel mapping indicator - RB dentity - CHOICE RLC info type - Logical channel dentity - RB stopicontinue - RB information to reconfigure - RB information to reconfigure - RB information to reconfigure - RB information in reach multiplexing option - RLC logical channel mapping indicator - RB mapping info - Information in reach multiplexing option - RLC logical channel mapping indicator - RB mapping info - Information in reach multiplexing option - RLC logical channel mapping indicator - RB information in reach multiplexing option - RLC logical channel mapping indicator - RB mapping info - Information in reach multiplexing option - RLC logical channel mapping indicator - RB mapping info - Information in reach multiplexing option - RLC logical channel mapping indicator - RB mapping info - Information in reach multiplexing	- Uplink transport channel type	RACH
- C-HOICE RLC isze list - MAC logical channel priority - Logical channel max-bes - Downlink RLC logical channel so - Downlink RLC logical channel so - Downlink RLC logical channel so - RB information to reconfigure - RB information for each multiplexing option - RLC logical channel sope - Logical channel sope - RD downlink RLC logical channels - Number of logical channel sope - Logical channel mapping information for each multiplexing option - RLC logical channel mapping information for sech multiplexing option - RLC logical channel mapping information for sech multiplexing option - RLC logical channel sope - RB stop/continue - RB information to reach multiplexing option - RLC logical channel sope - Log	- Logical channel identity	
- MAC logical channel priority - Logical channel mapping - Number of downlink RLC logical channels - Downlink transport channel type - Logical channel information to reconfigure - RB information to reconfigure - RB promitik transport channel type - Cholice RLC info type - RB mapping info - Information for each multiplexing option - RLC logical channel information to reconfigure - RB mapping info - Cholice RLC size ist - Downlink transport channel type - Logical channel mapping indicator - Number of typink RLC logical channels - Downlink transport channel type - Logical channel info - Number of typink RLC logical channels - Downlink RLC logical channels - Downlink TRLC logical channel info - RLC logical channel mapping indicator - RB information to reconfigure - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of typink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of typink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of typink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of typink RLC logical channels - Uplink transport channel type - Logical channel more of typink RLC logical channels - Downlink RLC logical channel info - Number of dypink RLC mode - Segmentation indicator - RB information to reconfigure - RB information to reconfig		
Legieal-channel max-less Downlink RLC logical channels Downlink RLC logical channel info		
Downlink RLC logical channels - Downlink transport channel type - Logical channel information to reconfigure - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of upilink RLC logical channels - Logical channel mapping indicator - Number of upilink RLC logical channels - Logical channel priority - Logical channel reconfigure - RB information to reconfigure - RB information to reconfigure - RB mapping info - Information for ach multiplexing option - RLC logical channel mapping indicator - Number of upilink RLC logical channels - Logical channel mapping indicator - Number of upilink RLC logical channels - Logical channel mapping indicator - Number of upilink RLC logical channels - Logical channel mapping indicator - Number of upilink RLC logical channels - Logical channel mapping indicator - Number of upilink RLC logical channels - Logical channel mode - Logical channel midication - RB mapping info - Information for each multiplexing option - RLC logical channel info - Number of upilink RLC logical channel info - RB mapping info - Information for each multiplexing option - RLC logical channel mode - CHOICE RLC info type - Logical channel mapping indicator - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of upilink RLC logical channel info - RLC logical channel mapping indicator - Number of upilink RLC logical channel info - RLC logical channel mapping indicator - Number of upilink RLC logical channel info - RLC logical channel mapping indicator - RLC logical channel map		4
Number of downlink RLC logical channels Downlink transport channel type	- Logical channel max loss	0
Number of downlink RLC logical channels Downlink transport channel type	- Downlink RLC logical channel info	
- Downlink transport channel type - Logical channel identity - RB stop/continue - RB information to reconfigure - RB identity - PDCP info - CHOICE RLC info type - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - RL information for each multiplexing option - RLC logical channel mapping indicator - RLC logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel rinority - Logical channel rinority - Logical channel rinority - Logical channel rinority - RB stop/continue - RB identity - PDCP info - RLC logical channel info - RLC logical channel priority - RB mapping info - Information for each multiplexing option - RLC logical channel priority - Logical channel information for each multiplexing option - RLC logical channel priority - Logical channel mapping indicator - RL information to reconfigure - RB stop/continue - RB identity - RB stop/continue - RB mapping info - Information for each multiplexing option - RLC logical channel priority - Logical channel mapping indicator - RLC info type - CHOICE RLC size list - MAC logical channel priority - Logical channel mapping indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indication - RLC logical channel mapping indication - RLC logical channel mapping indicator - RLC more of updire RLC logical channels - Logical channel mapping indicator - RLC more of updire RLC logical channels - Logical channel mapping indicator - RLC more of updire RLC logical channels - Logical ch		1
Logical channel identity RB stop/continue RB identity PDCP info Not Present		FACH
RB information to reconfigure RB identity PDCP info CHOICE RLC info type RB mapping info Information for each multiplexing option RLC logical channel mapping indicator RLC logical channel mapping indicator Logical channel mapping indicator RCC logical channel reconfigure CHOICE RLC size list CHOICE RLC size list AMAC logical channel priority Logical channel max less Downlink RLC logical channels Number of gownlink RLC logical channels Not Present Not Present Not Present Not Present ALI RACH ALI RACH ALI		
RB information to reconfigure RB identity PDCP info CHOICE RLC info type RB mapping info Information for each multiplexing option RLC logical channel mapping indicator Number of uplink RLC logical channels Uplink transport channel type Logical channel max-lose Downlink RLC logical channel info Number of downlink RLC logical channels Downlink RLC logical channel info RB information to reconfigure RB mapping info Information for each multiplexing option RB mapping info Information for each multiplexing option RB clogical channel mapping indicator Number of uplink RLC logical channels Uplink transport channel type RB mapping info Information for each multiplexing option RCC logical channel mapping indicator Number of uplink RLC logical channels Downlink transport channel type CHOICE RLC size isi ARC logical channel mapping indicator Number of uplink RLC logical channels Downlink transport channel type Committed RLC logical channels Downlink transport channel type Committed RLC logical channels Downlink RLC logical channels CHOICE RLC size isis RB information to reconfigure RB information to reconfigure RB stop/continue RCC force RLC size isis RCC december of uplink RLC logical channels RB information to reconfigure RB information to reconfigure RCC december of uplink RLC logical channels RCC december of uplink RLC logical c	.1	
RB identity PDCP info CHOICE RLC info type RB mapping info Information for each multiplexing option RLC logical channel mapping indicator - Number of uplink RLC logical channels Uplink transport channel type Logical channel mapping indicator - Logical channel mapping indicator - Logical channel dentity - CHOICE RLC size list - MAC logical channel priority - Logical channel maxibes - Downlink RLC logical channels - Downlink RLC logical channels - Downlink transport channel type - Logical channel mapping indicator - RB information to reconfigure - RB indentity - RB stoppontinue - RB indentity - CHOICE RLC size list - MAC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channels - Downlink RLC logical channels - RB information to reconfigure - RB information to reconfigure - RB information for each multiplexing option - RLC logical channel mapping indication - CHOICE RLC info type - CHOICE Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Logical channel mapping indicator - Number of uplink RLC logical channels - Segmentation indication - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Logical channel mapping indicator - Number of uplink RLC logical channels - Logical channel mapping indicator - Number of uplink RLC logical channels - Logical channel mapping indicator - Number of		
RB identity PDCP info CHOICE RLC info type RB mapping info Information for each multiplexing option RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel priority - CHOICE RLC size list - MAC logical channel priority - Logical channel maxing - Number of downlink RLC logical channels - Downlink RLC logical channels - Downlink transport channel type - Logical channel maxing - RB information to reconfigure - RB information for each multiplexing option - RLC logical channel mapping indicator - Number of downlink RLC logical channels - Holling RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel maxines - Downlink RLC logical channels - Downlink RLC logical channel info - Number of downlink RLC logical channels - Downlink RLC logical channels - Downlink RLC logical channels - CHOICE RLC info type - RB information to reconfigure - RB information to reconfigure - RB information for each multiplexing option - RLC logical channel mapping indication - CHOICE RLC info type - CHOICE R	- RB information to reconfigure	(AM DCCH for NAS_DT Low priority)
PDCP info O-HOLDE RLC info type RB mapping info Information for each multiplexing option RLC logical channel mapping indicator Number of uplink RLC logical channels Uplink transport channel type Logical channel mapping indicator Number of uplink RLC logical channels Downlink RLC logical channel priority Legical channel indentity RB stopocontinue RB information to reconfigure RB information to reconfigure RB mapping info Information to reach multiplexing option RLC logical channel mapping indicator Number of uplink RLC logical channels Uplink transport channel type Logical channel mapping indicator Number of uplink RLC logical channels Uplink transport channel type Logical channel mapping indicator Number of uplink RLC logical channels Uplink transport channel type Logical channel mapping indicator Number of uplink RLC logical channels Uplink transport channel type Logical channel mapping indicator Number of uplink RLC logical channels Uplink transport channel type Logical channel mapping indication RB indomation to reconfigure RB information to reconfigure RB information to reach multiplexing option RCC logical channel mapping indicator Number of uplink RLC logical channels Downlink RLC logical channels Downlink RLC logical channels CHOICE RLC size list RB information to reconfigure RB information to reach multiplexing option RCC HOICE plink RLC mode Segmentation indication RCC HOICE plink RLC mode Segmentation indication RCC HOICE RLC size list Logical channel mapping indicator Not Present RCC HOICE RLC info type CHOICE Info type CHOICE SIZE list All CHOICE RLC size list Logical channel mapping indicator Not Present RCC HOICE RLC size list Logical channel mapping indicator Not Present RCC HOICE RCC size list All CHOICE RCC size list All CHO	- RB identity	
CHOICE RLC info type		Not Present
RE mapping info Information for each multiplexing option RLC logical channel mapping indicator Number of uplink RLC logical channels Uplink transport channel type Logical channel marw isss Downlink RLC logical channels Downlink RLC logical channels RB information to reconfigure RB information for each multiplexing prior RB information to reconfigure RB information for each multiplexing option LOGICE RLC info type Logical channel mapping info Not Present 1 All All All All All All All All All A		
- Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel information for each multiplexing option - RDC logical channel information for each multiplexing option - RDC logical channel information for each multiplexing option - RDC logical channel information for each multiplexing option - RDC logical channel information for each multiplexing option - RCC		Not riesent
- RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel mapping indicator - RB information for each multiplexing option - RLC logical channel priority - RB stop/continue - RB information for each multiplexing option - RLC logical channel priority - Logical channel inde - Uplink RLC logical channels - Downlink RLC logical channels - Logical channel mapping indicator - RL indicator - RLC logical channel mapping indicator - RLC logical channel priority - Logical channel indentity - CHOICE RLC size list - MAC logical channel priority - Logical channel mapping indicator - RLC logical channel priority - Logical channel mapping indicator - RLC logical channel priority - Logical channel mapping indicator - RLC logical channel priority - Logical channel mapping indicator - RLC logical channel priority - RB stop/continue - RB information for each multiplexing option - RR poping info - CHOICE RLC size list - RB stop/continue - RB information for each multiplexing option - RR poping info - CHOICE RLC info type - CHOICE PLC info type - CHOICE Qublink RLC mode - Segmentation indication - RR mapping info - RR papping inf		
- Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel priority - Logical channel priority - Logical channel riax-tesse - Downlink RLC logical channels - Downlink RLC logical channels - Downlink ransport channel type - Logical channel identity - RB stop/continue - RB information to reconfigure - RB information to reconfigure - RB information to reconfigure - RB information to reach multiplexing option - CHOICE RLC info type - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel max less - Downlink RLC logical channels - Downlink transport channel type - Logical channel identity - RB stop/continue - RB information to reconfigure - RB information for each multiplexing option - RLC logical channel info - Number of downlink RLC logical channels - Downlink RLC logical channels - Downlink RLC logical channels - Segmentation indication - RB apping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Segmentation indication - RB apping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channel - RB sitop/continue - RB information to reconfigure - Not Present - N		
- Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Legical channel max less - Downlink RLC logical channels - Downlink RLC logical channel so - Downlink RLC logical channel so - Downlink RLC logical channel so - Downlink transport channel type - Logical channel identity - RB stop/continue - RB information to reconfigure - RB information to reconfigure - RB information to reconfigure - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel info - Unimber of uplink RLC logical channels - Downlink RLC logical channel info - Number of uplink RLC logical channels - Downlink transport channel type - Logical channel max less - Downlink RLC logical channels - Downlink RLC logical channels - Downlink transport channel type - Logical channel info - Number of downlink RLC logical channels - RB information to reconfigure - RB identity - PDCP info - RB identity - PDCP info - CHOICE RLC info type - CHOICE BLC info type - CHOICE Uplink RLC mode - Transmission RLC discard - Segmentation indication - RB information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink RLC logical channels - Downlink RLC logical channel info - Number of uplink RLC logical channels - Uplink transport channel reportiv - Legical channel max less - Downlink RLC logical channel info - Number of uplink	 RLC logical channel mapping indicator 	Not Present
- Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Legical channel max less - Downlink RLC logical channels - Downlink RLC logical channel so - Downlink RLC logical channel so - Downlink RLC logical channel so - Downlink transport channel type - Logical channel identity - RB stop/continue - RB information to reconfigure - RB information to reconfigure - RB information to reconfigure - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel info - Unimber of uplink RLC logical channels - Downlink RLC logical channel info - Number of uplink RLC logical channels - Downlink transport channel type - Logical channel max less - Downlink RLC logical channels - Downlink RLC logical channels - Downlink transport channel type - Logical channel info - Number of downlink RLC logical channels - RB information to reconfigure - RB identity - PDCP info - RB identity - PDCP info - CHOICE RLC info type - CHOICE BLC info type - CHOICE Uplink RLC mode - Transmission RLC discard - Segmentation indication - RB information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink RLC logical channels - Downlink RLC logical channel info - Number of uplink RLC logical channels - Uplink transport channel reportiv - Legical channel max less - Downlink RLC logical channel info - Number of uplink	- Number of uplink RLC logical channels	1
- Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel max-toses - Downlink RLC logical channels - Downlink RLC logical channels - Downlink RLC logical channel type - Logical channel identity - RB stop/continue - RB information to reconfigure - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Rumber of uplink RLC logical channels - Uplink transport channel type - Logical channel max-toses - Downlink RLC logical channels - Logical channel identity - RB information to reconfigure - RB information to reconfigure - CHOICE RLC info type - CHOICE Uplink RLC mode - Transmission RLC discard - Segmentation indication - RLC logical channel mapping indicator - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Logical channel mapping indicator - RB information to reconfigure - CHOICE RLC size list - MAC logical channel priority - Logical channel mapping indicator - RB information to reconfigure - Mot Present - Mot		RACH
- CHOICE RLC size list - MAC logical channel priority - Legical-channel max-lose - Downlink RLC logical channels - Downlink RLC logical channels - Downlink RLC logical channels - Logical channel identity - RB stop/continue - RB information to reconfigure - RB information to reconfigure - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel max-lose - Downlink RLC logical channels - Logical channel info - RB information to reconfigure - RB information for each multiplexing option - RLC logical channel mapping indicator - RD mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Logical channel mapping indicator - Number of uplink RLC logical channels - Logical channel mapping indicator - Number of uplink RLC logical channels - Logical channel mapping indicator - Number of uplink RLC logical channels - Logical channel mapping indicator - Number of uplink RLC logical channels - Logical channel mapping indicator - Number of uplink RLC logical channels - Logical channel mapping indicator - Number of uplink RLC logical channels - Logical channel mapping indicator - Number of uplink RLC logical channels - Logical channel mapping indicator - Number of		
- MAC logical channel priority - Logical channel max-lose - Downlink RLC logical channels - Downlink RLC logical channels - Downlink transport channel type - Logical channel identity - RB stop/continue - RB information to reconfigure - RB information to reconfigure - RB information to reconfigure - RB mapping info - CHOICE RLC info type - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - RUmber of uplink RLC logical channels - Uplink transport channel type - Logical channel priority - CHOICE RLC size list - MAC logical channel priority - RB information to reconfigure - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC mode - Transmission RLC discard - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical chan		
- Logical Channel max-lose - Downlink RLC logical channels - Number of downlink RLC logical channels - Downlink transport channel type - Logical channel identity - RB stop/continue - RB information to reconfigure - RB information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel information for each multiplexing - Downlink RLC logical channels - RB information to reconfigure - CHOICE RLC info type - CHOICE LUplink RLC mode - Transmission RLC discard - Segmentation indication - RB mapping info - Information to reach multiplexing option - RLC logical channel identity - CHOICE RLC info type - Logical channel identity - RB mapping info - Information to reach multiplexing option - RLC logical channel max-lose - Downlink RLC logical channels - Logical channel max-lose - Downlink RLC logical channels - Logical channel max-lose - Downlink RLC logical channels - Logical channel max-lose - Downlink RLC logical channels - Logical channel max-lose - Downlink RLC logical channels - Logical channel max-lose - Downlink RLC logical channels - RB information to reconfigure		
- Downlink RLC logical channels - Downlink transport channel type - Logical channel identity - RB stop/continue - RB information to reconfigure - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC info type - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel info - Number of downlink RLC logical channels - Downlink RLC mode - RB information to reconfigure - CHOICE Downlink RLC mode - Transmission RLC discard - Segmentation indication - CHOICE Downlink RLC mode - Transmission RLC discard - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC mode - Townlink RLC mode - Townlink RLC mode - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel max less - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure - RB information to reconfigure		
- Number of downlink RLC logical channels - Downlink transport channel type - Logical channel identity - RB stop/continue - RB information to reconfigure - RB information to reach multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Logical channel identity - Logical channel identity - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - Logical channel mapping indicator - Number of uplink RLC logical channels - Downlink RLC logical channels - Logical channel identity - RB information to reconfigure - CHOICE RLC info type - CHOICE Downlink RLC mode - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC mode - Segmentation indication - RB mapping info - RLC logical channel mapping indicator - Number of uplink RLC mode - Segmentation indication - RB mapping info - RLC logical channel mapping indicator - Number of uplink RLC mode - Segmentation indication - RB mapping info - RLC logical channel mapping indicator - Number of uplink RLC mode - Segmentation indication - RB mapping info - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - All - MC Logical channel identity - CHOICE RLC size list - All - RB stop/continue - RB information to reconfigure - RB information to reconfigure		0
- Number of downlink RLC logical channels - Downlink transport channel type - Logical channel identity - RB stop/continue - RB information to reconfigure - RB information to reach multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Logical channel identity - Logical channel identity - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - Logical channel mapping indicator - Number of uplink RLC logical channels - Downlink RLC logical channels - Logical channel identity - RB information to reconfigure - CHOICE RLC info type - CHOICE Downlink RLC mode - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC mode - Segmentation indication - RB mapping info - RLC logical channel mapping indicator - Number of uplink RLC mode - Segmentation indication - RB mapping info - RLC logical channel mapping indicator - Number of uplink RLC mode - Segmentation indication - RB mapping info - RLC logical channel mapping indicator - Number of uplink RLC mode - Segmentation indication - RB mapping info - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - All - MC Logical channel identity - CHOICE RLC size list - All - RB stop/continue - RB information to reconfigure - RB information to reconfigure	- Downlink RLC logical channel info	
- Downlink transport channel type - Logical channel identity - RB information to reconfigure - RB identity - PDCP info - CHOICE RLC info type - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Logical channel identity - Logical channel identity - Logical channel max-lose - Downlink RLC logical channels - Downlink transport channel type - Logical channel identity - RB stop/continue - RB identity - RB identity - PDCP into - CHOICE RLC info type - CHOICE Duplink RLC mode - Transmission RLC discard - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel identity - Logical channel identity - Logical channel identity - Logical channel max-lose - Downlink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel mapping indicator - Number of uplink RLC logical channel mapping indicator - Number of uplink RLC logical channel info - Not Present - Logical channel mapping indicator - Not Present - Logical channel mapping indicator - Not Present - RB information for econfigure - RB information for RRC) - Downlink RLC logical channel info - Not Present - Not Pre		1
- Logical channel identity - RB stop/continue - RB information to reconfigure - RB information to reconfigure - RB identity - PDCP info - CHOICE RLC info type - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel max loss - Downlink RLC logical channels - Downlink RLC logical channels - Downlink RLC logical channel type - RB stop/continue - RB identity - PDCP info - CHOICE RLC info type - CHOICE Downlink RLC mode - Transmission RLC discard - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel type - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - Logical channel identity - Logical channel identity - Logical channel priority - Logical channel identity - Logical channel priority - Logical channel max loss - Downlink RLC logical channel info - MB stop/continue - RB information to reconfigure - RB stop/continue		FACH
RB stop/continue RB information to reconfigure RB information to reconfigure RB identity PDCP info CHOICE RLC info type RB mapping info Information for each multiplexing option RLC logical channel mapping indicator Number of uplink RLC logical channels Downlink RLC logical channel info Not Present RACH All RACH RACH RACH RACH RACH RACH RACH RACH		
RB information to reconfigure RB identity PDCP info Not Present		
RB identity - PDCP info - CHOICE RLC info type - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel mation - Number of downlink RLC logical channels - Downlink RLC logical channel info - Number of downlink RLC logical channels - Downlink RLC logical channel info - Number of downlink RLC logical channels - Downlink RLC logical channel info - Number of downlink RLC logical channels - Downlink RLC logical channel info - Number of downlink RLC logical channels - Downlink RLC logical channel info - RB information to reconfigure - RB information to reconfigure - CHOICE RLC info type - CHOICE Uplink RLC mode - Transmission RLC discard - Segmentation indication - RLC logical channel mapping indicator - Number of uplink RLC mode - Information for each multiplexing option - RLC logical channel identity - Logical channel identity - Logical channel identity - Logical channel identity - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink RLC logical channels - Uplink RLC logical channels - Uplink RLC logical channel info - RB stop/continue - RB information to reconfigure RB information to reconfigure RB information to reconfigure ROT Present Not Present - Not		
PDCP info CHOICE RLC info type RB mapping info Information for each multiplexing option RLC logical channel mapping indicator Number of uplink RLC logical channels Uplink transport channel type Logical channel identity CHOICE RLC size list MAC logical channel priority Logical channel max loss Downlink RLC logical channels info Number of downlink RLC logical channels Downlink transport channel type Logical channel identity RB stop/continue RB information to reconfigure RB information to reconfigure RB information to reconfigure CHOICE RLC info type CHOICE RLC info type CHOICE RLC info type CHOICE RLC info type CHOICE Downlink RLC mode Transmission RLC discard Segmentation indication RB mapping info Information for each multiplexing option RR mapping info Information for each multiplexing option RLC logical channel mapping indicator Number of uplink RLC logical channels Uplink transport channel type Logical channel identity CHOICE RLC isree israely information for each multiplexing option RR mapping info Nut Present TM RLC TRUE All TACH		
- CHOICE RLC info type - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel priority - CHOICE RLC size list - MAC logical channel mapping indicator - Number of downlink RLC logical channels - Downlink RLC logical channel info - Number of downlink RLC logical channels - Downlink transport channel type - Logical channel identity - RB stop/continue - RB information to reconfigure - RB identity - PDCP info - CHOICE RLC info type - CHOICE Uplink RLC mode - Transmission RLC discard - Segmentation indication - RLC logical channel identity - RB mapping info - Information for each multiplexing option - RLC logical channel identity - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - Logical	- RB identity	<u>20</u> 5
- CHOICE RLC info type - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel priority - CHOICE RLC size list - MAC logical channel mapping indicator - Number of downlink RLC logical channels - Downlink RLC logical channel info - Number of downlink RLC logical channels - Downlink transport channel type - Logical channel identity - RB stop/continue - RB information to reconfigure - RB identity - PDCP info - CHOICE RLC info type - CHOICE Uplink RLC mode - Transmission RLC discard - Segmentation indication - RLC logical channel identity - RB mapping info - Information for each multiplexing option - RLC logical channel identity - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - Logical	- PDCP info	Not Present
- RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Legical channel max loss - Downlink RLC logical channels - Downlink transport channel type - Logical channel identity - RB stop/continue - RB information to reconfigure - RB information to reconfigure - RB information indication - CHOICE RLC info type - CHOICE Uplink RLC mode - Transmission RLC discard - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel impority - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure Not Present - RACH - All - ALI - ACH - CHOICE RLC size list - MAC logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure Not Present - Number of uplink RLC mode - RB stop/continue - RB stop/continue - RB information to reconfigure Not Present - Number of uplink RLC mode - Segmentation to reconfigure - Not Present - Not Present - Number of uplink RLC mode - Number of uplink RLC mode - Number of uplink RLC mode - Segmentation indication - RLC info - TM RLC - Stantamental type - Logical channel max loss - Downlink RLC mode - RL Stop/continue - RL Stop/continue - RL Stop/continue - RL Stop/continue - RL Stop/contin		Not Present
- Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel priority - CHOICE RLC size list - MAC logical channel priority - Logical channel priority - Logical channel priority - Logical channel mak loss - Downlink RLC logical channels - Downlink RLC logical channel type - Logical channel identity - RB stop/continue - RB information to reconfigure - CHOICE RLC info type - CHOICE Uplink RLC mode - Transmission RLC discard - Segmentation indication - RLC logical channel septon - RLC logical channel mapping indicator - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channels - Uplink RLC logical channels - Uplink RLC logical channel info - Legical channel mapping indicator - Number of uplink RLC logical channel info - Legical channel mapping indicator - Not Present - Legical channel mapping indicato		
- RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Legical channel max-lose - Downlink RLC logical channels - Downlink RLC logical channels - Downlink transport channel type - Logical channel identity - RB stop/continue - RB information to reconfigure - CHOICE RLC info type - CHOICE RLC info type - CHOICE Uplink RLC mode - Transmission RLC discard - Segmentation indication - RLC logical channel mapping indicator - RLC logical channel mapping indicator - RLC logical channel identity - CHOICE RLC size list - MAC logical channel max lose - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure Not Present - RACH - All - All - CHOICE RLC size list - MAC logical channel info - RB stop/continue - RB information to reconfigure Not Present - Nut Present -		
- Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel max loss - Downlink RLC logical channels - Downlink transport channel type - Logical channel identity - RB stop/continue - RB information to reconfigure - RB identity - PDCP info - CHOICE RLC info type - CHOICE RLC info type - CHOICE Uplink RLC mode - Transmission RLC discard - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC mode - Uplink transport channel type - Logical channel mapping indicator - Logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure - Logical channel max loss - Downlink RLC logical channel info - RB information to reconfigure - RB information to reconfigure - RB information to reconfigure		Not Dropont
- Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel max lose - Downlink RLC logical channels - Downlink transport channel type - Logical channel identity - RB stop/continue - RB information to reconfigure - RB mapping info - RC logical channel mapping indicator - Number of uplink RLC logical channels - Logical channel identity - Logical channel max lose - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure - RACH -		
- Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel max lose - Downlink RLC logical channels - Number of downlink RLC logical channels - Logical channel identity - RB stop/continue - RB information to reconfigure - RB identity - PDCP info - CHOICE RLC info type - CHOICE Uplink RLC mode - Transmission RLC discard - Segmentation indication - CHOICE Downlink RLC mode - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of upplink RLC logical channels - Uplink transport channel type - Logical channel priority - CHOICE RLC size list - MAC logical channel max lose - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure - RB stop/continue - RB information to reconfigure - Logical channel max lose - Downlink RLC logical channel info - RB information to reconfigure - RB information to reconfigure - RB information to reconfigure		· ·
- CHOICE RLC size list - MAC logical channel priority - Logical channel max lose - Downlink RLC logical channels - Downlink transport channel type - Logical channel identity - RB stop/continue - RB information to reconfigure - RB identity - PDCP info - CHOICE RLC info type - CHOICE Uplink RLC mode - Transmission RLC discard - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel priority - CHOICE RLC size list - MAC logical channel max lose - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure - RB stop/continue - RB information to reconfigure - RIII - ACH - CHOICE RLC size list - AII - Not Present - No		RACH
- MAC logical channel priority - Logical channel max loss - Downlink RLC logical channels - Number of downlink RLC logical channels - Downlink transport channel type - Logical channel info - RB istop/continue - RB information to reconfigure - RB identity - PDCP info - CHOICE RLC info type - CHOICE Uplink RLC mode - Transmission RLC discard - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channel type - Logical channel priority - Logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure - RB information to reconfigure 64 - Mot Present (TM CCCH uplink for RRC) 0 Not Present - TM RLC - Not Present - TRUE - Not Present	- Logical channel identity	<u>7</u> 6
- MAC logical channel priority - Logical channel max loss - Downlink RLC logical channels - Number of downlink RLC logical channels - Downlink transport channel type - Logical channel info - RB istop/continue - RB information to reconfigure - RB identity - PDCP info - CHOICE RLC info type - CHOICE Uplink RLC mode - Transmission RLC discard - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel mapping indicator - Number of uplink RLC logical channel type - Logical channel priority - Logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure - RB information to reconfigure 64 - Mot Present (TM CCCH uplink for RRC) 0 Not Present - TM RLC - Not Present - TRUE - Not Present	- CHOICE RLC size list	All
- Legical channel max loss - Downlink RLC logical channel info - Number of downlink RLC logical channels - Downlink transport channel type - Logical channel identity - RB stop/continue - RB information to reconfigure - RB identity - PDCP info - CHOICE RLC info type - CHOICE Uplink RLC mode - Transmission RLC discard - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB stop/continue - RB information to reconfigure - RB information to reconfigure ### ACCH Outpublic RLC logical channel info Not Present		
- Downlink RLC logical channel info - Number of downlink RLC logical channels - Downlink transport channel type - Logical channel identity - RB stop/continue - RB information to reconfigure - RB identity - PDCP info - CHOICE RLC info type - CHOICE Uplink RLC mode - Transmission RLC discard - Segmentation indication - CHOICE Downlink RLC mode - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure 1 1 FACH 64 Not Present (TM CCCH uplink for RRC) Not Present TRUE 1 TRUE 1 Not Present 1 RACH 5 All 1 Not Present Not Present 1 Not Present		
- Number of downlink RLC logical channels - Downlink transport channel type - Logical channel identity - RB stop/continue - RB information to reconfigure - RB information to reconfigure - RB identity - PDCP info - CHOICE RLC info type - CHOICE Uplink RLC mode - Transmission RLC discard - Segmentation indication - CHOICE Downlink RLC mode - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB stop/continue - RB stop/continue - RB information to reconfigure 1 FACH 64 Not Present (TM CCCH uplink for RRC) 0 Not Present TRUE TRUE - TRUE - NAC Present 1 RACH - Segmentation indicator - Not Present 1 - Not Present Not Present - Not Present		∀
- Downlink transport channel type - Logical channel identity - RB stop/continue - RB information to reconfigure - RB identity - PDCP info - CHOICE RLC info type - CHOICE Uplink RLC mode - Transmission RLC discard - Segmentation indication - CHOICE Downlink RLC mode - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel max loes - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure FACH 64 Not Present (TM CCCH uplink for RRC) Not Present RLC info TM RLC TRUE TRUE TRUE Not Present 1 RACH 5 4 All Not Present 1 Not Present Not Present 1 Not Present		4
- Logical channel identity - RB stop/continue - RB information to reconfigure - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel priority - Logical channel mapping info - RB stop/continue - RB stop/continue - RB information to reconfigure Not Present -		1
- RB stop/continue - RB information to reconfigure - RB identity - PDCP info - CHOICE RLC info type - CHOICE Uplink RLC mode - Transmission RLC discard - Segmentation indication - CHOICE Downlink RLC mode - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Legical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure Not Present - Not Present - All - MAC logical channel priority - Legical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure Not Present - Not Pres		
- RB information to reconfigure - RB identity - PDCP info - CHOICE RLC info type - CHOICE Uplink RLC mode - Transmission RLC discard - Segmentation indication - CHOICE Downlink RLC mode - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB stop/continue - RB information to reconfigure (TM CCCH uplink for RRC) 0 Not Present TRUE TRUE TM RLC TRUE TM RLC TRUE TRUE TRUE ANOT Present 1 RACH 5 All 1 Not Present Not Present Not Present 1 Not Present Not Present 1 Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present		<u>6</u> 4
- RB information to reconfigure - RB identity - PDCP info - CHOICE RLC info type - CHOICE Uplink RLC mode - Transmission RLC discard - Segmentation indication - CHOICE Downlink RLC mode - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB stop/continue - RB information to reconfigure (TM CCCH uplink for RRC) 0 Not Present TRUE TRUE TM RLC TRUE TM RLC TRUE TRUE TRUE ANOT Present 1 RACH 5 All 1 Not Present Not Present Not Present 1 Not Present Not Present 1 Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present Not Present	- RB stop/continue	Not Present
- RB identity - PDCP info - CHOICE RLC info type - CHOICE Uplink RLC mode - Transmission RLC discard - Segmentation indication - CHOICE Downlink RLC mode - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure		
- PDCP info - CHOICE RLC info type - CHOICE Uplink RLC mode - Transmission RLC discard - Segmentation indication - CHOICE Downlink RLC mode - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure Not Present RLC info TM RLC TRUE TM RLC TRUE TM RLC TRUE		
- CHOICE RLC info type - CHOICE Uplink RLC mode - Transmission RLC discard - Segmentation indication - CHOICE Downlink RLC mode - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure RLC info TM RLC Not Present TRUE Not Present Not Present Not Present		
- CHOICE Uplink RLC mode - Transmission RLC discard - Segmentation indication - CHOICE Downlink RLC mode - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure TM RLC Not Present Not Present Not Present All Not Present Not Present Not Present (UM CCCH downlink for RRC)		
- Transmission RLC discard - Segmentation indication - CHOICE Downlink RLC mode - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure Not Present TRUE TM RLC TRUE Mot Present 1 RACH 5 All Not Present Not Present Not Present Not Present (UM CCCH downlink for RRC)		
- Segmentation indication - CHOICE Downlink RLC mode - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure TRUE TM RLC TRUE Not Present ALI TRUE TM RLC TRUE Not Present ALI Not Present	· ·	
- CHOICE Downlink RLC mode - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure TM RLC TRUE Not Present All All Not Present		
- CHOICE Downlink RLC mode - Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure TM RLC TRUE Not Present Not Present All Not Present	- Segmentation indication	TRUE
- Segmentation indication - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure TRUE Not Present Not Present All Not Present		TM RLC
- RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure - Not Present Not Present Not Present Not Present (UM CCCH downlink for RRC)		
- Information for each multiplexing option - RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure Not Present		
- RLC logical channel mapping indicator - Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure Not Present 1 RACH 5 All 1 Not Present		
- Number of uplink RLC logical channels - Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure 1 RACH 5 All 1 RACH 5 Not Present Not Present Not Present (UM CCCH downlink for RRC)		Not Droppet
- Uplink transport channel type - Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure RACH 5 All Not Present Not Present (UM CCCH downlink for RRC)		
- Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure - Logical channel info - Not Present Not Present (UM CCCH downlink for RRC)		•
- Logical channel identity - CHOICE RLC size list - MAC logical channel priority - Logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure - Logical channel info - Not Present Not Present (UM CCCH downlink for RRC)		RACH
- CHOICE RLC size list - MAC logical channel priority - Logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure All 1 Not Present Not Present (UM CCCH downlink for RRC)		5
- MAC logical channel priority - Logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure 1 0 Not Present Not Present (UM CCCH downlink for RRC)		
- Logical channel max loss - Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure - Logical channel max loss - Not Present Not Present (UM CCCH downlink for RRC)		
- Downlink RLC logical channel info - RB stop/continue - RB information to reconfigure Not Present Not Present (UM CCCH downlink for RRC)		
- RB stop/continue - RB information to reconfigure Not Present (UM CCCH downlink for RRC)		
- RB information to reconfigure (UM CCCH downlink for RRC)		
- RB identity 0		1 2
	- RB identity	0

- CHOICE RLC info type - CHOICE Uplink RLC mode Not Present - CHOICE Downlink RLC mode **UM RLC** - RB mapping info - Information for each multiplexing option - Downlink RLC logical channel info - Number of downlink RLC logical channels **FACH** - Downlink transport channel type - Logical channel identity - RB stop/continue Not Present (TM BCCH for RRC) - RB information to reconfigure - RB identity - PDCP info Not Present - CHOICE RLC info type RLC info - CHOICE Uplink RLC mode Not Present - CHOICE Downlink RLC mode TM RLC - Segmentation Indication **TRUE** - RB mapping info - Information for each multiplexing option - Downlink RLC logical channel info - Number of downlink RLC logical channels - Downlink transport channel type **FACH** - Logical channel identity - RB stop/continue Not Present - RB information to reconfigure (TM PCCH for RRC) - RB identity - PDCP info Not Present - CHOICE RLC info type **RLC** info - CHOICE Uplink RLC mode Not Present - CHOICE Downlink RLC mode TM RLC - Segmentation Indication **TRUE** - RB mapping info - Information for each multiplexing option - Downlink RLC logical channel info - Number of downlink RLC logical channels - Downlink transport channel type **PCH** - Logical channel identity - RB stop/continue Not Present RB information to be affected Not Present UL Transport channel information for all transport channels - TFC subset (This IE is repeated for TFC number.) - Allowed Transport Format combination 0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108 clause 6.10 Parameter Set.) - PRACH TFCS Not Present - CHOICE mode **FDD** - UL DCH TFCS Not Present Deleted UL TrCH information - Transport channel identity 1Not Present - Transport channel identity Added or Reconfigured UL TrCH information - Transport channel identity 154 - CHOICE Transport channel type Common transport channels - Dynamic Transport format information (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set - RLC Size (This IE is repeated for TFI number.) - Number of TBs and TTI List _- Number of Transport blocks Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set - CHOICE mode FDD - CHOICE Logical Channel List ALL - Semi-static Transport Format information - Transmission time interval Reference to TS34.108 clause 6.10 Parameter Set - Type of channel coding Reference to TS34.108 clause 6.10 Parameter Set - Coding Rate Reference to TS34.108 clause 6.10 Parameter Set - Rate matching attribute Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set - CRC size

Not Present

- PDCP info

DRAC static information

DL Transport channel information common for all transport channel

- SCCPCH TFCS
- Normal
- TFCI Field 1 information
- CHOICE TFCS representation
- TFCS addition information
- CHOICE CTFC Size
- CTFC information
- Power offset information
- CHOICE DL parameters
- DL DCH TFCS

Deleted DL TrCH information

- Transport channel identity
- Transport channel identity

Added or Reconfigured DL TrCH information

- Transport channel identity
- CHOICE DL parameters
- TFS
- CHOICE Transport channel type
- Dynamic Transport format information
- RLC Size
- Number of TBs and TTI List
 - Number of Transport blocks
- RLC size
 - CHOICE mode
- CHOICE Logical Channel List
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size
 - DCH quality target
 - Transparent mode signalling info

Added or Reconfigured DL TrCH information

- Transport channel identity
- CHOICE DL parameters
- TFS
 - CHOICE Transport channel type
 - Dynamic Transport format information
 - RLC Size
- Number of TBs and TTI List
 - Number of Transport blocks
- RLC size
 - CHOICE mode
- CHOICE Logical Channel List
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size
 - DCH quality target
- Transparent mode signalling info

Added or Reconfigured DL TrCH information

- Transport channel identity
- CHOICE DL parameters
- TFS
- CHOICE Transport channel type
 - Dynamic Transport format information
- RLC Size
- Number of TBs and TTI List
 - Number of Transport blocks
- RLC size
 - CHOICE mode

Not Present

(This IE is repeated for TFC number.)

Addition

Number of bits used must be enough to cover all combinations of CTFC from clause 6.10. Refer to TS34.108 clause 6.10 Parameter Set

Not Present Independent Not Present

Not Present6

10

121

Independent

Common transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set FDD

ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Not Present Not Present

<u>13</u>2

Independent

Common transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set FDD

ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Not Present

<u>14</u>3

Independent

Not Present

Common transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

FDD

- CHOICE Logical Channel List - Semi-static Transport Format information	ALL
- Transmission time interval	Reference to TS34.108 clause 6.10 Parameter Set
- Type of channel coding	Reference to TS34.108 clause 6.10 Parameter Set
- Coding Rate	Reference to TS34.108 clause 6.10 Parameter Set
- Rate matching attribute	Reference to TS34.108 clause 6.10 Parameter Set
- CRC size	Reference to TS34.108 clause 6.10 Parameter Set
- DCH quality target	Not Present
- Transparent mode signalling info	Not Present
Added or Reconfigured DL TrCH information	
	4
	Independent
 TFS	
Dynamic Transport format information	(This IE is repeated for TFI number)
Number of Transport blocks	Reference to TS34.108 clause 6.10 Parameter Set
- Semi-static Transport Format information	
- Transmission time interval	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
——————————————————————————————————————	Reference to TS34.108 clause 6.10 Parameter Set
- Rate matching attribute	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
- DCH quality target	Not Present
- Transparent mode signalling info	Not Present
Frequency info	
- UARFCN uplink(Nu)	Reference to TS34.108 clause 6.10 Parameter Set
- UARFCN downlink(Nd)	Reference to TS34.108 clause 6.10 Parameter Set
Maximum allowed UL TX power	33dBm
CHOICE channel requirement	PRACH info (for RACH)Not Present
— - CHOICE mode	FDD
- Available Signature	'0000 0000 1111 1111'B
- Available SF	Reference to TS34.108 clause 6.10 Parameter Set
- Scrambling code number	0
- Puncturing Limit	Reference to TS34.108 clause 6.10 Parameter Set
Available Sub Channel number	'1111 1111 1111'B
CHOICE Mode	FDD
- Downlink PDSCH information	Not Present
Downlink information common for all radio links	Not Present
Downlink information for each radio links	Not Present
- Primary CPICH info	<u>Not i resent</u>
- Primary scrambling code	100
	Not Present
PDSCH with SHO DCH fillo - PDSCH code mapping	Not Present
 PDSCH code mapping Downlink DPCH info for each RL	Not Present
——————————————————————————————————————	HOLT FOODIN
- Selection Indicator	Not Present
- Primary CPICH usage for channel estimation	Primary CPICH may be used
- Secondary CPICH info	Not Present
- Secondary CFICH IIIIC - Secondary scrambling code	1
- Secondary scrambling code - SSDT Indicator	+ FALSE
- Spreading factor	Reference to TS34.108 clause 6.10 Parameter Set
——————————————————————————————————————	SF-1(SF is reference to TS34.108 clause 6.10 Parameter
— — — — — — — — — — — — — — — — — — —	Set)
- Pilot symbol existence	FALSE
TFCI existence	TRUE
- Fixed or Flexible Position	Flexible
Timing offset TFCS	0 Not Propert
- FACH/PCH information	Not Present Not Present
- References to system information blocks	Not Present
- Scheduling information	

Ī	Information Element	Value/remark
ŀ	Message Type	Valaditomant
IJ	RRC transaction identifier	Arbitrarily selects an integer between 0 and 3
•	Integrity check info	The presence of this IE is dependent on IXIT statements in
		TS 34.123-2. If integrity protection is indicated to be active,
		this IE is present with the values of the sub IEs as stated
		below. Else, this IE and the sub-IEs are omitted.
	- message authentication code	SS calculates the value of MAC-I for this message and
	DD0	writes to this IE.
	- RRC message sequence number	SS provides the value of this IE, from its internal counter.
	Integrity protection mode info Ciphering mode info	Not Present Not Present
	Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
	New U-RNTI	Not Present
	New C-RNTI	0000 0000 0000 0010B
П	RRC State DRX-indicator	NoDRXCELL_FACH
	UTRAN DRX cycle length coefficient	Not Present
	CN information info	Not Present
	URA identity	Not Present
	RAB information to reconfigure list	Not Present
	RB information to reconfigure list	(UNA DOOLL (DDO)
	- RB information to reconfigure	(UM DCCH for RRC)
	- RB identity - PDCP info	1 Not Present
١١	- PDCP IIII0 - <u>CHOICE</u> RLC info <u>type</u>	Not Present
ı	- RB mapping info	Not i resent
	- Information for each multiplexing option	
l	- RLC logical channel mapping indicator	Not Present
	- Number of uplink RLC logical channels	1
	- Uplink transport channel type	RACH
	- Logical channel identity	1
l	- CHOICE RLC size list	All
ı	- MAC logical channel priority	2 0
	- Logical channel max loss - Downlink RLC logical channel info	♥
	- Number of downlink RLC logical channels	1
ı	- Downlink transport channel type	FACH
	- Logical channel identity	1
	- RB information to reconfigure	(AM DCCH for RRC)
	- RB identity	2
ı	- PDCP info	Not Present
IJ	- <u>CHOICE_</u> RLC info <u>type</u> - RB mapping info	Not Present
	- No mapping into - Information for each multiplexing option	
П	- RLC logical channel mapping indicator	Not Present
IJ	- Number of <u>uplink</u> RLC logical channels	1
	- Uplink transport channel type	RACH
,	- Logical channel identity	2
IJ	- CHOICE RLC size list	All
١١	- MAC logical channel priority	3 0
Ш	- Downlink RLC logical channel info	♥
IJ	- Number of <u>downlink</u> RLC logical channels	1
1	- Downlink transport channel type	FACH
	- Logical channel identity	2
IJ	- RB stop/continue	Not Present
	- RB information to reconfigure	(AM DCCH for NAS_DT High priority)
	- RB identity	3 Not Propert
ار	- PDCP info - <u>CHOICE</u> RLC info <u>type</u>	Not Present Not Present
1	- <u>CHOICE</u> REC IIIIO <u>type</u> - RB mapping info	NOTE TO COLL
	- Information for each multiplexing option	
H	- RLC logical channel mapping indicator	Not Present
$\ \ $	- Number of uplink RLC logical channels	1
Ĺ	- Uplink transport channel type	RACH

- Logical channel identity	3
- CHOICE RLC size list	<u>All</u>
 MAC logical channel priority 	4
- Logical channel max loss	0
- Downlink RLC logical channel info	
- Number of downlink RLC logical channels	1
- Downlink transport channel type	FACH
- Logical channel identity	3
- RB stop/continue	Not Present
- RB information to reconfigure	(AM DCCH for NAS_DT Low priority)
- RB identity	4
- PDCP info	Not Present
- CHOICE RLC info type	Not Present
- RB mapping info	
 Information for each multiplexing option 	
 - RLC logical channel mapping indicator 	Not Present
- Number of <u>uplink</u> RLC logical channels	1
 Uplink transport channel type 	RACH
- Logical channel identity	4
- CHOICE RLC size list	<u>All</u>
- MAC logical channel priority	5
- Logical channel max loss	0
- Downlink RLC logical channel info	
- Number of <u>downlink</u> RLC logical channels	1
- Downlink transport channel type	FACH
- Logical channel identity	4
- RB stop/continue	Not Present
- RB information to reconfigure	(AM DTCH)
- RB identity	20 5
- PDCP info	Not Present
- CHOICE RLC info type	Not Present
- RB mapping info	
- Information for each multiplexing option	Not Propert
- RLC logical channel mapping indicator	Not Present
- Number of <u>uplink</u> RLC logical channels	RACH
Uplink transport channel type Logical channel identity	6 6
- CHOICE RLC size list	<u>8</u>
- MAC logical channel priority	<u>All</u> 6
- Logical channel max loss	$\frac{1}{\theta}$
- Downlink RLC logical channel info	₩ ₩
- Number of downlink RLC logical channels	1
- Downlink transport channel type	FACH
- Logical channel identity	64
- RB stop/continue	Not Present
- RB information to reconfigure	(TM CCCH uplink for RRC)
- RB identity	0
- PDCP info	Not Present
- CHOICE RLC info	RLC info
- CHOICE Uplink RLC mode	TM RLC
- Transmission RLC Discard	Not Present
- Segmentation Indication	TRUE
- CHOICE Downlink RLC mode	Not Present
- Segmentation Indication	
- RB mapping info	
- Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
- Number of uplink RLC logical channels	1
- Uplink transport channel type	RACH
- Logical channel identity	5
- CHOICE RLC size list	<u>All</u>
- MAC logical channel priority	1
- Logical channel max loss	θ
 Downlink RLC logical channel info 	Not Present
- RB stop/continue	Not Present
- RB information to reconfigure	(UM CCCH for RRC)
- RB identity	0 Not Droppet
- PDCP info	Not Present

- RB mapping info - Information for each multiplexing option - Downlink RLC logical channel info - Number of downlink RLC logical channels - Downlink transport channel type **FACH** - Logical channel identity - RB stop/continue Not Present - RB information to reconfigure (TM BCCH for RRC) - RB identity - PDCP info Not Present - CHOICE RLC info type RLC info - CHOICE Uplink RLC mode Not Present - CHOICE Downlink RLC mode TM RLC - Segmentation Indication **TRUE** - RB mapping info - Information for each multiplexing option - Downlink RLC logical channel info - Number of downlink RLC logical channels **FACH** - Downlink transport channel type - Logical channel identity - RB stop/continue **Not Present** - RB information to reconfigure (TM PCCH for RRC) - RB identity - PDCP info Not Present - CHOICE RLC info type **RLC** info - CHOICE Uplink RLC mode Not Present - CHOICE Downlink RLC mode TM RLC - Segmentation indication **TRUE** - RB mapping info - Information for each multiplexing option - Downlink RLC logical channel info - Number of downlink RLC logical channels - Downlink transport channel type **PCH** - Logical channel identity - RB stop/continue Not Present RB information to be affected Not Present UL Transport channel information for all transport channels (This IE is repeated for TFC number.) - TFC subset - Allowed Transport Format combination 0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108 clause 6.10 Parameter Set.) - PRACH TFCS Not Present - CHOICE mode **FDD** - UL DCH TFCS Not Present Deleted UL TrCH information - Transport channel identity Not Present Added or Reconfigured UL TrCH information - Transport channel identity <u>15</u>1 - CHOICE Transport channel type Common transport channels - Dynamic Transport format information (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set - RLC Size (This IE is repeated for TFI number.)
Reference to TS34.108 clause 6.10 Parameter Set - Number of TBs and TTI List - Number of Transport blocks Reference to TS34.108 clause 6.10 Parameter Set RLC size - CHOICE mode FDD - CHOICE Logical Channel List ALL - Semi-static Transport Format information - Transmission time interval Reference to TS34.108 clause 6.10 Parameter Set - Type of channel coding Reference to TS34.108 clause 6.10 Parameter Set - Coding Rate Reference to TS34.108 clause 6.10 Parameter Set - Rate matching attribute Reference to TS34.108 clause 6.10 Parameter Set - CRC size Reference to TS34.108 clause 6.10 Parameter Set DRAC static information Not Present DL Transport channel information common for all

RLC info

UM RLC

Not Present

- CHOICE RLC info type - CHOICE Uplink RLC mode

- CHOICE Downlink RLC mode

transport channel

- SCCPCH TFCS
- Normal
- TFCI Field 1 information
- CHOICE TFCS representation
- TFCS addition information
- CHOICE CTFC Size
- CTFC information
- Power offset information
- CHOICE DL parameters
- DL DCH TFCS

Deleted DL TrCH information

- Transport channel identity

Added or Reconfigured DL TrCH information

- Transport channel identity
- CHOICE DL parameters
- TFS
- CHOICE Transport channel type
 - Dynamic Transport format information
- RLC Size
 - Number of TBs and TTI List
 - Number of Transport blocks
 - RLC size
 - CHOICE mode
 - CHOICE Logical Channel List
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size
 - DCH quality target
 - Transparent mode signalling info

Added or Reconfigured DL TrCH information

- Transport channel identity
- CHOICE DL parameters
- TFS
- CHOICE Transport channel type
 - Dynamic Transport format information
- RLC Size
 - Number of TBs and TTI List
 - Number of Transport blocks
 - -- RLC size
 - CHOICE Logical Channel List
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size
 - DCH quality target
 - Transparent mode signalling info

Added or Reconfigured DL TrCH information

- Transport channel identity
- CHOICE DL parameters
- TFS
- CHOICE Transport channel type
- Dynamic Transport format information
- RLC Size
- Number of TBs and TTI List
 - Number of Transport blocks
- RLC size
 - CHOICE mode
- CHOICE Logical Channel List
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding

(This IE is repeated for TFC number.)

Addition

Number of bits used must be enough to cover all combinations of CTFC from clause 6.10.

Refer to TS34.108 clause 6.10 Parameter Set

Not Present Independent

Not Present

Not Present

121

Independent

Common transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set FDD

ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Not Present

132

Independent

Not Present

Common transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Not Present

143

Independent

Not Present

Common transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

FDD ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

- Rate matching attribute Reference to TS34.108 clause 6.10 Parameter Set - CRC size Reference to TS34.108 clause 6.10 Parameter Set Not Present - DCH quality target - Transparent mode signalling info Not Present Added or Reconfigured DL TrCH information Transport channel identity CHOICE DL parameters Independent TFS **Dynamic Transport format information** (This IE is repeated for TFI number) Number of Transport blocks Reference to TS34.108 clause 6.10 Parameter Set **RLC size** Reference to TS34.108 clause 6.10 Parameter Set Semi-static Transport Format information Transmission time interval Reference to TS34.108 clause 6.10 Parameter Set Type of channel coding Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Coding Rate Rate matching attribute Reference to TS34.108 clause 6.10 Parameter Set **CRC** size Reference to TS34.108 clause 6.10 Parameter Set DCH quality target Not Present Not Present Transparent mode signalling info Frequency info - UARFCN uplink(Nu) Reference to TS34.108 clause 6.10 Parameter Set - UARFCN downlink(Nd) Reference to TS34.108 clause 6.10 Parameter Set Maximum allowed UL TX power 33dBm PRACH info (for RACH)Not Present CHOICE channel requirement CHOICE mode EDD Available Signature '0000 0000 1111 1111'B Available SF Reference to TS34.108 clause 6.10 Parameter Set Scrambling code number Puncturing Limit Reference to TS34.108 clause 6.10 Parameter Set Available Sub Channel number '1111 1111 1111'B CHOICE Mode FDD - Downlink PDSCH information Not Present Downlink information common for all radio links Not Present Downlink information for each radio links Not Present Primary CPICH info Primary scrambling code 150 PDSCH with SHO DCH info Not Present PDSCH code mapping Not Present Downlink DPCH info for each RL Not Present Secondary CCPCH info Selection Indicator Not Present Primary CPICH may be used Primary CPICH usage for channel estimation Secondary CPICH info Not Present Secondary scrambling code SSDT Indicator **FALSE** Spreading factor Reference to TS34.108 clause 6.10 Parameter Set SF-1(SF is reference to TS34.108 clause 6.10 Parameter Code number Set) FALSE Pilot symbol existence **TRUE** TFCI existence Fixed or Flexible Position **Flexible** Timing offset **TFCS** Not Present **FACH/PCH** information Not Present References to system information blocks Not Present Scheduling information

Reference to TS34.108 clause 6.10 Parameter Set

- Coding Rate

Contents of RADIO BEARER RECONFIGURATION COMPLETE message: AM

Message Type	
RRC transaction identifier	Checked to see if the value is identical to the same IE in
	the downlink RADIO BEARER RECONFIGURATION
	COMPLETE message
Integrity check info	The presence of this IE is dependent on IXIT statements
	in TS 34.123-2. If integrity protection is indicated to be
	active, this IE shall be present with the values of the sub
	IEs as stated below. Else, this IE and the sub-IEs shall be
	absent.
- Message authentication code	This IE is checked to see if it is present. The value is
moodago aanomoanon oodo	compared against the XMAC-I value computed by SS.
- RRC Message sequence number	This IE is checked to see if it is present. The value is
Titto Message sequence number	used by SS to compute the XMAC-I value.
Uplink integrity protection activation info	Not checked
COUNT-C activation time	The presence of this IE depends on the following 2
COONT-C activation time	
	factors: (a) There exists RB(s) mapped to RLC-TM and
	(b) UE is transiting to CELL DCH state after the
04	reconfiguration procedure. Else, this IE is absent.
Other information elementCHOICE mode	Not checked FDD
Radio bearer uplink ciphering activation time info	Not checked
RB with PDCP information list	Not checked

Information Element	Value/remark
Message Type	
Integrity check info	The presence of this IE is dependent on IXIT statements in TS 34.123-2. If integrity protection is indicated to be active, this IE is present with the values of the sub IEs as stated below. Else, this IE and the sub-IEs are omitted.
- message authentication code	SS calculates the value of MAC-I for this message and writes to this IE.
- RRC message sequence number	SS provides the value of this IE, from its internal counter.
Integrity protection mode info	Not Present
Ciphering mode info	Not Present
Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
New U-RNTI	Not Present
New C-RNTI RRC State DRX-indicator	Not Present
UTRAN DRX cycle length coefficient	Not Present
CN information info	Not Present
URA identity	Not Present
RAB information to reconfigure list	Not Present
RB information to release	
- RB identity	<u>510</u>
RB information to be affected	(UM DCCH for RRC)
- RB identity	1
RB mapping info Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
- Number of <u>uplink</u> RLC logical channels	1
- Uplink transport channel type	DCH
 <u>UL</u> Transport channel identity 	<u>5</u> 4
- Logical channel identity	1
- CHOICE RLC size list	All
- MAC logical channel priority	1
- Logical channel max loss - Downlink RLC logical channel info	θ
- Number of <u>downlink</u> RLC logical channels	1
- Downlink transport channel type	DCH
 <u>DL</u> Transport channel identity 	<u>10</u> 4
- Logical channel identity	1
RB information to be affected	(AM DCCH for RRC)
- RB identity - RB mapping info	2
- Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
 Number of <u>uplink</u> RLC logical channels 	1
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	<u>5</u> 4 2
Logical channel identity CHOICE RLC size list	
- CHOICE RLC SIZE IIST - MAC logical channel priority	All 2
- Logical channel max loss	0
- Downlink RLC logical channel info	
- Number of downlink RLC logical channels	1
- Downlink transport channel type	DCH
- <u>DL</u> Transport channel identity	1 <u>0</u> 2
- Logical channel identity RB information to be affected	AM DCCH for NAS_DT High priority)
- RB identity	3
- RB mapping info	
- Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
- Number of <u>uplink</u> RLC logical channels	1
 Uplink transport channel type <u>UL</u> Transport channel identity 	DCH <u>5</u> 4
- Logical channel identity	3
- CHOICE RLC size list	All
- MAC logical channel priority	3

Logical channel max loss 0 - Downlink RLC logical channel info - Number of downlink RLC logical channels DCH - Downlink transport channel type - DL Transport channel identity 10 - Logical channel identity RB information to be affected (AM DCCH for NAS_DT Low priority) - RB identity - RB mapping info - RLC logical channel mapping indicator Not Present - Information for each multiplexing option - Number of uplink RLC logical channels - Uplink transport channel type DCH - UL Transport channel identity <u>5</u>1 - Logical channel identity - CHOICE RLC size list <u>All</u> - MAC logical channel priority 4 Logical channel max loss 0 - Downlink RLC logical channel info - Number of downlink RLC logical channels - Downlink transport channel type DCH - DL Transport channel identity <u>10</u>4 - Logical channel identity UL Transport channel information for all transport channels - TFC subset (This IE is repeated for TFC number.) 0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108 - Allowed Transport Format combination clause 6.10 Parameter Set.) - PRACH TFCS Not Present - CHOICE mode - UL DCH TFCS (This IE is repeated for TFC number.) - Normal - TFCI Field 1 information - CHOICE CTFC representation Addition - TFCS addition information - CHOICE CTFC Size Number of bits used must be enough to cover all combinations of CTFC from clause 6.10. Refer to TS34.108 clause 6.10 Parameter Set - CTFC information - Power offset information - CHOICE Gain Factor Signalled Gain Factor - Gain factor •c 0 - Gain factor •d 0 Not Present - Reference TFC ID - Power offset Pp-m 0dB Deleted UL TrCH Information - Transport channel identity Added or Reconfigured UL TrCH information If TrCH reconfiguration is executed then this is needed(e.g. The rate of SRB for DCCH is changed.). - Transport channel identity - TFS - CHOICE Transport channel type Dedicated transport channels - Dynamic Transport format information (This IE is repeated for TFI number) - RLC Size Reference to TS34.108 clause 6.10 Parameter Set - Number of TBs and TTI List (This IE is repeated for TFI number.) - Transmission Time Interval Not Present - Number of Transport blocks Reference to TS34.108 clause 6.10 Parameter Set **RLC size** Reference to TS34.108 clause 6.10 Parameter Set - CHOICE Logical Channel list - Semi-static Transport Format information - Transmission time interval Reference to TS34.108 clause 6.10 Parameter Set - Type of channel coding Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set - Coding Rate - Rate matching attribute Reference to TS34.108 clause 6.10 Parameter Set - CRC size Reference to TS34.108 clause 6.10 Parameter Set CPCH set ID Not Present DRAC static information Not Preaent DL Transport channel information common for all

transport channel

- SCCPCH TFCS
- CHOICE DL parameters
- DL DCH TFCS
- Normal
- TFCI Field 1 information
- CHOICE CTFC representation
- TFCS addition information
- CHOICE CTFC Size
- CTFC information
- Power offset information

Deleted DL TrCH Information

- Transport channel identity

Added or Reconfigured DL TrCH information

- Transport channel identity
- CHOICE DL parameters
- TFS
- CHOICE Transport channel type
 - Dynamic Transport format information
- RLC Size
 - Number of TBs and TTI List
 - Transmission Time Interval
 - Number of Transport blocks
- **RLC size**
 - CHOICE Logical Channel list
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size
 - DCH quality target
 - BLER Quality value
 - Transparent mode signalling info

Frequency info

- UARFCN uplink(Nu)
- UARFCN downlink(Nd)

Maximum allowed UL TX power

Uplink DPCH info

- Uplink DPCH power control info
- DPCCH power offset
 - PC Preamble
 - Power Control Algorithm
 - TPC step size
- Scrambling code type
- Scrambling code number
- Number of DPDCH
- spreading factor
- TFCI existence
- Number of FBI bit
- Puncturing Limit

CHOICE Mode

- Downlink PDSCH information

Downlink information common for all radio links

- Downlink DPCH info common for all RL
- CHOICE mode
 - Timing indicator
 - CFN-targetSFN frame offset
 - Downlink DPCH power control information
 - DPC mode
 - DL rate matching restriction information
 - Spreading factor
 - Fixed or Flexible Position
 - TFCI existence
 - Number of bits for Pilot bits(SF=128,256)
 - DPCH compressed mode info
 - TGPSI

Not Present

Independent

(This IE is repeated for TFC number.)

Addition

Number of bits used must be enough to cover all combinations of CTFC from clause 6.10.

Refer to TS34.108 clause 6.10 Parameter Set

Not Present

If TrCH reconfiguration is executed then this is needed(e.g The rate of SRB for DCCH is changed.).

101

Independent

Dedicated transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

Reference to TS34.108 clause 6.10 Parameter Set

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

-6.30.00

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set 33dBm

-6dB

8slot

Algorithm1

1dB

Long

0 (0 to 16777215)

Not Present(1)

SF is reference to TS34.108 clause 6.10 Parameter Set

TRUE

Not Present(0)

Reference to TS34.108 clause 6.10 Parameter Set

Not Present

FDD

Maintain

Not Present

0 (single)

Not Present

Reference to TS34.108 clause 6.10 Parameter Set

FALSE

Reference to TS34.108 clause 6.10 Parameter Set

- TGPS Status Flag inactive - TGCFN (Current CFN + (256 - TTI/10msec)) mod 256 - Transmission gap pattern sequence configuration parameters - TGMP **FDD** Measurement - TGPRC - TGCFN (Current CFN + (256 - TTI/10msec)) mod 256 - TGSN - TGL1 10 - TGL2 5 - TGD 15 - TGPL1 35 - TGPL2 35 - RPP Mode 1 - ITP Mode 1 - UL/DL Mode DL - Downlink compressed mode method SF/2 - Uplink compressed mode method Not Present - Downlink frame type - DeltaSIR1 2.0 - DeltaSIRafter1 1.0 - DeltaSIR2 Not Present - DeltaSIRafter2 Not Present - TX Diversity mode None - SSDT information Not Present - Default DPCH Offset Value Downlink information for each radio links - Primary CPICH info - Primary scrambling code 100 - PDSCH with SHO DCH info Not Present - PDSCH code mapping Not Present - Downlink DPCH info for each RL - Primary CPICH usage for channel estimation Primary CPICH may be used - DPCH frame offset 0 chips - Secondary CPICH info Not Present - DL channelisation code - Secondary scrambling code - Spreading factor Reference to TS34.108 clause 6.10 Parameter Set - Code number SF-1(SF is reference to TS34.108 clause 6.10 Parameter Set) - Scrambling code change No change - TPC combination index 0 - SSDT Cell Identity -a - Closed loop timing adjustment mode Not Present - Secondary CCPCH info Not Present - TFCS Not Present - FACH/PCH information Not Present - References to system information blocks Not Present

Scheduling information

Information Element	Value/remark
Message Type	
RRC transaction identifier	Arbitrarily selects an integer between 0 and 3
Integrity check info	The presence of this IE is dependent on IXIT statements
	in TS 34.123-2. If integrity protection is indicated to be
	active, this IE is present with the values of the sub IEs as
manage and provide a stigent and a	stated below. Else, this IE and the sub-IEs are omitted.
- message authentication code	SS calculates the value of MAC-I for this message and writes to this IE.
- RRC message sequence number	SS provides the value of this IE, from its internal counter.
Integrity protection mode info	Not Present
Ciphering mode info	Not Present
Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
New U-RNTI	Not Present
New C-RNTI	Not Present
RRC State DRX-indicator	NoDRXCELL_DCH
UTRAN DRX cycle length coefficient	Not Present
CN information info	Not Present
Signalling Connection release indication	Not Present
URA identity RAB information to reconfigure list	Not Present Not Present
RB information to reconligure list	NOU I IGOGIIL
- RB identity	5 10
RB information to release	<u> </u>
- RB identity	<u>116</u>
RB information to release	
- RB identity	<u>12</u> 7
RB information to be affected	(UM DCCH for RRC)
- RB identity	1
- RB mapping info	
- Information for each multiplexing option	Not Dropont
 - RLC logical channel mapping indicator - Number of <u>uplink RLC logical channels</u> 	Not Present 1
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	<u>5</u> 4
- Logical channel identity	1
- CHOICE RLC size list	<u>All</u>
- MAC logical channel priority	1
- Logical channel max loss	θ
- Downlink RLC logical channel info	
- Number of <u>downlink</u> RLC logical channels	1
 Downlink transport channel type <u>DL</u> Transport channel identity 	DCH 10
- <u>DL</u> transport channel identity - Logical channel identity	1 <u>0</u> 1
RB information to be affected	(AM DCCH for RRC)
- RB identity	2
- RB mapping info	
 Information for each multiplexing option 	
- RLC logical channel mapping indicator	Not Present
- Number of <u>uplink</u> RLC logical channels	1
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	<u>5</u> 4 2
Logical channel identity CHOICE RLC size list	
- MAC logical channel priority	<u>All</u> 2
- Logical channel max loss	$\frac{2}{\theta}$
- Downlink RLC logical channel info	-
- Number of downlink RLC logical channels	1
 Downlink transport channel type 	DCH
- <u>DL</u> Transport channel identity	1 <u>0</u>
- Logical channel identity	2
RB information to be affected	(AM DCCH for NAS_DT High priority)
- RB identity	3
RB mapping info Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
- NEO logical channel mapping mulcator	<u>INOCLI 16361IL</u>

N. J. C. P. D. O. J. J. J.	
 Number of <u>uplink</u> RLC logical channels 	1
 Uplink transport channel type 	DCH
- <u>UL</u> Transport channel identity	<u>5</u> 4
 Logical channel identity 	3
- CHOICE RLC size list	<u>All</u>
 MAC logical channel priority 	3
- Logical channel max loss	θ
 Downlink RLC logical channel info 	
 Number of <u>downlink</u> RLC logical channels 	1
- Downlink transport channel type	DCH
 DL Transport channel identity 	1 <u>0</u>
 Logical channel identity 	3
RB information to be affected	
	(AM DCCH for NAS_DT Low priority)
- RB identity	4
- RB mapping info	
 Information for each multiplexing option 	
 RLC logical channel mapping indicator 	Not Present
- Number of uplink RLC logical channels	1
	·
 Uplink transport channel type 	DCH
- UL Transport channel identity	<u>5</u> 4
- Logical channel identity	$\frac{3}{4}$
- CHOICE RLC size list	<u>All</u>
- MAC logical channel priority	4
- Logical channel max loss	θ
 Downlink RLC logical channel info 	
- Number of downlink RLC logical channels	1
 Downlink transport channel type 	DCH
 DL Transport channel identity 	1 <u>0</u>
- Logical channel identity	4
RB with PDCP information list	Not Present
UL Transport channel information for all transport	
channels	
- TFC subset	(This IE is repeated for TFC number.)
 Allowed Transport Format combination 	0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108
7 moneta Transport i emilia combination	
	clause 6.10 Parameter Set.)
- PRACH TFCS	Not Present
- CHOICE mode	FDD
- UL DCH TFCS	(This IE is repeated for TFC number.)
- Normal	
- TFCI Field 1 information	
	A Little
 CHOICE CTFC representation 	Addition
 TFCS addition information 	
- CHOICE CTFC Size	Number of bits used must be enough to cover all
- CHOICE CIFC Size	9
	combinations of CTFC from clause 6.10.
 CTFC information 	Refer to TS34.108 clause 6.10 Parameter Set
- Power offset information	
- CHOICE Gain Factor	Signalled Gain Factor
- Gain factor •c	0
- Gain factor •d	
	0
- Reference TFC ID	Not Present
- Power offset Pp-m	0dB
	000
Deleted UL TrCH Information	
- Transport channel identity	<u>1</u> 2
Deleted UL TrCH Information	-
	00
- Transport channel identity	<u>2</u> 3
Deleted UL TrCH Information	
- Transport channel identity	<u>3</u> 4
Added or Reconfigured UL TrCH information	If TrCH reconfiguration is executed then this is
	needed(e.g The rate of SRB for DCCH is changed.).
- Transport channel identity	<u>5</u> 4
	≚'
- TFS	
 CHOICE Transport channel type 	Dedicated transport channels
- Dynamic Transport format information	(This IE is repeated for TFI number)
- RLC Size	Reference to TS34.108 clause 6.10 Parameter Set
 Number of TBs and TTI List 	(This IE is repeated for TFI number.)
- Transmission Time Interval	Not Present
Number of Transport blocks	Reference to TS34.108 clause 6.10 Parameter Set
- RLC size	Reference to TS34.108 clause 6.10 Parameter Set
·	

- CHOICE Logical Channel list
- Semi-static Transport Format information
- Transmission time interval
- Type of channel coding
- Coding Rate
- Rate matching attribute
- CRC size

CPCH set ID

DRAC static information

DL Transport channel information common for all transport channel

- SCCPCH TFCS
- CHOICE DL parameters
- DL DCH TFCS
- Normal
- TFCI Field 1 information
- CHOICE CTFC representation
- TFCS addition information
- CHOICE CTFC Size
- CTFC information
- Power offset information

Deleted DL TrCH Information

- Transport channel identity

Deleted DL TrCH Information

- Transport channel identity

Deleted DL TrCH Information

Transport channel identity

Added or Reconfigured DL TrCH information

- Transport channel identity
- CHOICE DL parameters
- UL TrCH Identity
- TFS
- CHOICE Transport channel type
 - Dynamic Transport format information
- ŔLC Size
 - Number of TBs and TTI List
 - Transmission Time Interval
 - Number of Transport blocks
 - Bit mode RLC size info
 - Transport block size
- CHOICE Logical Channel list
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size
 - DCH quality target
 - BLER Quality value
- Transparent mode signalling info

Frequency info

- UARFCN uplink(Nu)
- UARFCN downlink(Nd)

Maximum allowed UL TX power

Uplink DPCH info

- Uplink DPCH power control info
- DPCCH power offset
- PC Preamble
- Power Control Algorithm
- TPC step size
- Scrambling code type
- Scrambling code number
- Number of DPDCH
- spreading factorTFCI existence
- Number of FBI bit

ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Not Present

Not Present Independent

Not Preaent

(This IE is repeated for TFC number.)

Addition

Number of bits used must be enough to cover all combinations of CTFC from clause 6.10. Refer to TS34.108 clause 6.10 Parameter Set Not Present

62

73

84

If TrCH reconfiguration is executed then this is needed(e.g The rate of SRB for DCCH is changed.).

10

Independent

<u>5</u>1

Dedicated transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Not Present

Reference to TS34.108 clause 6.10 Parameter Set

Reference to TS34.108 clause 6.10 Parameter Set ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

<u>-6.3</u>0.00

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set 33dBm

-6dB 8slot

Algorithm1 1dB

Long

0 (0 to 16777215)

Not Present(1)

SF is reference to TS34.108 clause 6.10 Parameter Set

TRUE

Not Present(0)

- Puncturing Limit Reference to TS34.108 clause 6.10 Parameter Set **CHOICE Mode** - Downlink PDSCH information Not Present Downlink information common for all radio links - Downlink DPCH info common for all RL FDD CHOICE mode - Timing indicator Maintain - CFN-targetSFN frame offset Not Present - Downlink DPCH power control information - DPC mode 0 (single) - DL rate matching restriction information Not Present - Spreading factor Reference to TS34.108 clause 6.10 Parameter Set - Fixed or Flexible Position N/A - TFCI existence **FALSE** - Number of bits for Pilot bits(SF=128,256) Reference to TS34.108 clause 6.10 Parameter Set - DPCH compressed mode info - TGPSI - TGPS Status Flag inactive - TGCFN (Current CFN + (256 - TTI/10msec)) mod 256 - Transmission gap pattern sequence configuration parameters - TGMP **FDD Measurement** - TGPRC 62 **TGCFN** (Current CFN + (256 - TTI/10msec)) mod 256 - TGSN - TGL1 10 - TGL2 5 - TGD 15 - TGPL1 35 - TGPL2 35 - RPP Mode 1 - ITP Mode 1 - UL/DL Mode DL - Downlink compressed mode method SF/2 - Uplink compressed mode method Not Present - Downlink frame type - DeltaSIR1 2.0 - DeltaSIRafter1 1.0 - DeltaSIR2 Not Present - DeltaSIRafter2 Not Present - TX Diversity mode None - SSDT information Not Present - Default DPCH Offset Value 0 Downlink information for each radio links - Primary CPICH info - Primary scrambling code 100 - PDSCH with SHO DCH info Not Present - PDSCH code mapping Not Present - Downlink DPCH info for each RL - Primary CPICH usage for channel estimation Primary CPICH may be used - DPCH frame offset 0 chips - Secondary CPICH info Not Present - DL channelisation code - Secondary scrambling code - Spreading factor Reference to TS34.108 clause 6.10 Parameter Set - Code number SF-1(SF is reference to TS34.108 clause 6.10 Parameter Set) - Scrambling code change No change - TPC combination index 0 - SSDT Cell Identity - Closed loop timing adjustment mode Not Present - Secondary CCPCH info Not Present - TFCS Not Present - FACH/PCH information Not Present - References to system information blocks Not Present Scheduling information

Contents of RADIO BEARER RELEASE message: AM or UM (Packet to CELL_DCH from CELL_DCH in PS)

Information Element	Value/remark
Message Type	
RRC transaction identifier	Arbitrarily selects an integer between 0 and 3
Integrity check info	The presence of this IE is dependent on IXIT statements in
	TS 34.123-2. If integrity protection is indicated to be active,
	this IE is present with the values of the sub IEs as stated
and the section of the section and the section of t	below. Else, this IE and the sub-IEs are omitted.
- message authentication code	SS calculates the value of MAC-I for this message and
- RRC message sequence number	writes to this IE. SS provides the value of this IE, from its internal counter.
Integrity protection mode info	Not Present
Ciphering mode info	Not Present
Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
New U-RNTI	Not Present
New C-RNTI	Not Present
RRC State DRX indicator	NoDRXCELL_DCH
UTRAN DRX cycle length coefficient	Not Present
CN information info	Not Present
Signalling Connection release indication	Not Present
URA identity	Not Present
RAB information to reconfigure list	Not Present
RB information to release	005
- RB identity RB information to be affected	205 (UM DOCH for BBC)
	(UM DCCH for RRC)
- RB identity - RB mapping info	1
- Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
Number of <u>uplink</u> RLC logical channels	1
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	<u>5</u> 4
- Logical channel identity	1
- CHOICE RLC size list	All
- MAC logical channel priority	1
- Logical channel max loss	0
- Downlink RLC logical channel info	
- Number of <u>downlink</u> RLC logical channels	1
- Downlink transport channel type	DCH
- <u>DL</u> Transport channel identity	10
- Logical channel identity RB information to be affected	1 (AM DCCH for RRC)
- RB identity	2
- RB mapping info	
- RLC logical channel mapping indicator	Not Present
- Information for each multiplexing option	
Number of <u>uplink</u> RLC logical channels	1
- Uplink transport channel type	DCH
 <u>UL</u>Transport channel identity 	<u>5</u> 4
- Logical channel identity	2
- CHOICE RLC size list	All
- MAC logical channel priority	$\left \frac{\overline{2}}{2} \right $
	θ
- Downlink RLC logical channel info	
Number of <u>downlink</u> RLC logical channels Downlink transport channel type	1 DCH
- DL Transport channel identity	1 <u>0</u>
- Logical channel identity	2
RB information to be affected	(AM DCCH for NAS_DT High priority)
- RB identity	3
- RB mapping info	
- Information for each multiplexing option	
 RLC logical channel mapping indicator 	Not Present
 Number of <u>uplink</u> RLC logical channels 	1
- Uplink transport channel type	DCH
 <u>UL</u>Transport channel identity 	<u>5</u> 4

- Logical channel identity <u>All</u> 3 - CHOICE RLC size list - MAC logical channel priority 0 Logical channel max loss - Downlink RLC logical channel info - Number of downlink RLC logical channels - Downlink transport channel type DCH - DL Transport channel identity 10 - Logical channel identity 3 RB information to be affected (AM DCCH for NAS_DT Low priority) - RB identity - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator Not Present - Number of uplink RLC logical channels - Uplink transport channel type DCH - UL Transport channel identity <u>5</u>1 - Logical channel identity 4 - CHOICE RLC size list <u>All</u> - MAC logical channel priority 4 Logical channel max loss 0 - Downlink RLC logical channel info - Number of downlink RLC logical channels - Downlink transport channel type DCH - DL Transport channel identity 10 - Logical channel identity RB with PDCP information list Not Present UL Transport channel information for all transport channels - TFC subset (This IE is repeated for TFC number.) - Allowed Transport Format combination 0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108 clause 6.10 Parameter Set.) Not Present - PRACH TFCS - CHOICE mode - UL DCH TFCS (This IE is repeated for TFC number.) - Normal - TFCI Field 1 information - CHOICE CTFC representation Addition - TFCS addition information - CHOICE CTFC Size Number of bits used must be enough to cover all combinations of CTFC from clause 6.10. Refer to TS34.108 clause 6.10 Parameter Set - CTFC information - Power offset information Not Present Deleted UL TrCH Information - Transport channel identity Added or Reconfigured UL TrCH information If TrCH reconfiguration is executed then this is needed(e.g. The rate of SRB for DCCH is changed.). - Transport channel identity 51 - CHOICE DL parameters Independent - CHOICE Transport channel type **Dedicated transport channels** - Dynamic Transport format information (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set - RLC Size - Number of TBs and TTI List (This IE is repeated for TFI number.) - Transmission Time Interval Not Present - Number of Transport blocks Reference to TS34.108 clause 6.10 Parameter Set **RLC size** Reference to TS34.108 clause 6.10 Parameter Set - CHOICE Logical Channel list ALL - Semi-static Transport Format information - Transmission time interval Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set - Type of channel coding - Coding Rate Reference to TS34.108 clause 6.10 Parameter Set - Rate matching attribute Reference to TS34.108 clause 6.10 Parameter Set - CRC size Reference to TS34.108 clause 6.10 Parameter Set Not Present CPCH set ID DRAC static information Not Preaent

DL Transport channel information common for all

transport channel

- SCCPCH TFCS
- CHOICE DL parameters
- DL DCH TFCS
- Normal
- TFCI Field 1 information
- CHOICE CTFC representation
- TFCS addition information
- CHOICE CTFC Size
- CTFC information
- Power offset information

Deleted DL TrCH Information

- Transport channel identity

Added or Reconfigured DL TrCH information

- Transport channel identity
- CHOICE DL parameters
- TFS
- CHOICE Transport channel type
- Dynamic Transport format information
- RLC Size
 - Number of TBs and TTI List
 - Transmission Time Interval
 - Number of Transport blocks
- **RLC size**
 - CHOICE Logical Channel list
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size
 - DCH quality target
 - BLER Quality value
- Transparent mode signalling info

Frequency info

- UARFCN uplink(Nu)
- UARFCN downlink(Nd)

Maximum allowed UL TX power

Uplink DPCH info

- Uplink DPCH power control info
- DPCCH power offset
- PC Preamble
- Power Control Algorithm
- TPC step size
- Scrambling code type
- Scrambling code number
- Number of DPDCH
- spreading factor
- TFCI existence
- Number of FBI bit
- Puncturing Limit

CHOICE Mode

- Downlink PDSCH information

Downlink information common for all radio links

- Downlink DPCH info common for all RL
- CHOICE mode
 - Timing indicator
 - CFN-targetSFN frame offset
 - Downlink DPCH power control information
 - DPC mode
 - DL rate matching restriction information
 - Spreading factor
 - Fixed or Flexible Position
 - TFCI existence
 - Number of bits for Pilot bits(SF=128,256)
 - DPCH compressed mode info
 - TGPSI

Not Present

(This IE is repeated for TFC number.)

Addition

Number of bits used must be enough to cover all combinations of CTFC from clause 6.10. Refer to TS34.108 clause 6.10 Parameter Set Not Present

If TrCH reconfiguration is executed then this is needed(e.g. The rate of SRB for DCCH is changed.).

Independent

Dedicated transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

-6.30.00Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

33dBm

-6dB 8slot Algorithm1

1dB Long

0 (0 to 16777215)

Not Present(1)

SF is reference to TS34.108 clause 6.10 Parameter Set **TRUE**

Not Present(0)

Reference to TS34.108 clause 6.10 Parameter Set

Not Present

FDD

Maintain Not Present

0 (single) Not Present

Reference to TS34.108 clause 6.10 Parameter Set

N/A

FALSE

Reference to TS34.108 clause 6.10 Parameter Set

inactive - TGPS Status Flag - TGCFN (Current CFN + (256 - TTI/10msec)) mod 256 - Transmission gap pattern sequence configuration parameters - TGMP **FDD** Measurement - TGPRC - TGCFN (Current CFN + (256 - TTI/10msec)) mod 256 - TGSN - TGL1 10 - TGL2 5 - TGD 15 - TGPL1 35 - TGPL2 35 - RPP Mode 1 - ITP Mode 1 - UL/DL Mode DL - Downlink compressed mode method SF/2 - Uplink compressed mode method Not Present - Downlink frame type - DeltaSIR1 2.0 - DeltaSIRafter1 1.0 - DeltaSIR2 Not Present - DeltaSIRafter2 Not Present - TX Diversity mode None - SSDT information Not Present - Default DPCH Offset Value Downlink information for each radio links - Primary CPICH info - Primary scrambling code 100 - PDSCH with SHO DCH info Not Present - PDSCH code mapping Not Present - Downlink DPCH info for each RL - Primary CPICH usage for channel estimation Primary CPICH may be used - DPCH frame offset 0 chips - Secondary CPICH info Not Present - DL channelisation code - Secondary scrambling code - Spreading factor Reference to TS34.108 clause 6.10 Parameter Set - Code number SF-1(SF is reference to TS34.108 clause 6.10 Parameter Set) - Scrambling code change No change - TPC combination index 0 - SSDT Cell Identity -a - Closed loop timing adjustment mode Not Present - Secondary CCPCH info Not Present - TFCS Not Present - FACH/PCH information Not Present - References to system information blocks Not Present

Scheduling information

Contents of RADIO BEARER RELEASE message: AM or UM (Packet to CELL_DCH from CELL_FACH in PS)

Information Element	Value/remark
Message Type	
RRC transaction identifier	Arbitrarily selects an integer between 0 and 3
Integrity check info	The presence of this IE is dependent on IXIT statements in
	TS 34.123-2. If integrity protection is indicated to be active,
	this IE is present with the values of the sub IEs as stated
	below. Else, this IE and the sub-IEs are omitted.
- message authentication code	SS calculates the value of MAC-I for this message and
DDO	writes to this IE.
- RRC message sequence number	SS provides the value of this IE, from its internal counter.
Integrity protection mode info Ciphering mode info	Not Present Not Present
Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
New U-RNTI	Not Present
New C-RNTI	Not Present
RRC State DRX indicator	NoDRXCELL_DCH
UTRAN DRX cycle length coefficient	Not Present
CN information info	Not Present
Signalling Connection release indication	Not Present
URA identity	Not Present
RAB information to reconfigure list	Not Present
RB information to release	205
- RB identity RB information to release	<u>20</u> 5
- RB identity	6
RB information to release	
- RB identity	7
RB information to release	
- RB identity	0
RB information to be affected	(UM DCCH for RRC)
- RB identity	1
- RB mapping info	
- Information for each multiplexing option	N / P
- RLC logical channel mapping indicator	Not Present
Number of <u>uplink</u> RLC logical channels Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	<u>5</u> 4
- Logical channel identity	1
- CHOICE RLC size list	All
- MAC logical channel priority	1
- Logical channel max loss	0
- Downlink RLC logical channel info	
- Number of <u>downlink</u> RLC logical channels	1
- Downlink transport channel type	DCH
 <u>DL</u> Transport channel identity Logical channel identity 	104 1
RB information to be affected	(AM DCCH for RRC)
- RB identity	2
- RB mapping info	-
- Information for each multiplexing option	
 RLC logical channel mapping indicator 	Not Present
 Number of <u>uplink</u> RLC logical channels 	1
- Uplink transport channel type	DCH
- <u>UL</u> Transport channel identity	54
- Logical channel identity	2
- CHOICE RLC size list - MAC logical channel priority	<u>AII</u> 2
- MAC logical channel priority	² 0
- Downlink RLC logical channel info	~
- Number of <u>downlink</u> RLC logical channels	1
- Downlink transport channel type	DCH
 <u>DL</u>Transport channel identity 	1 <u>0</u>
- Logical channel identity	2
RB information to be affected	(AM DCCH for NAS_DT High priority)
- RB identity	3

- RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator **Not Present** - Number of uplink RLC logical channels - Uplink transport channel type DCH - UL Transport channel identity 51 - Logical channel identity 3 - CHOICE RLC size list <u>All</u> - MAC logical channel priority 3 0 Logical channel max loss - Downlink RLC logical channel info - Number of downlink RLC logical channels - Downlink transport channel type DCH - DL Transport channel identity <u>100</u>4 - Logical channel identity RB information to be affected (AM DCCH for NAS_DT Low priority) - RB identity - RB mapping info - Information for each multiplexing option - RLC logical channel mapping indicator Not Present - Number of uplink RLC logical channels DCH - Uplink transport channel type - UL Transport channel identity <u>5</u>1 - Logical channel identity 4 - CHOICE RLC size list ΑII - MAC logical channel priority 4 Logical channel max loss 0 - Downlink RLC logical channel info - Number of downlink RLC logical channels - Downlink transport channel type DCH - DL Transport channel identity 10 - Logical channel identity 4 **RB with PDCP information list Not Present** UL Transport channel information for all transport channels (This IE is repeated for TFC number.) - TFC subset - Allowed Transport Format combination 0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108 clause 6.10 Parameter Set.) Not Present - PRACH TFCS - CHOICE mode FDD - UL DCH TFCS (This IE is repeated for TFC number.) - Normal - TFCI Field 1 information - CHOICE CTFC representation Addition - TFCS addition information - CHOICE CTFC Siz Number of bits used must be enough to cover all combinations of CTFC from clause 6.10. - CTFC Refer to TS34.108 clause 6.10 Parameter Set - Power offset information - CHOICE Gain Factors Signalled Gain Factor - Gain factor •c 0 - Gain factor •d 0 Not Present - Reference TFC ID - Power offset Pp-m 0dB Deleted UL TrCH Information 15Not Present Added or Reconfigured UL TrCH information - Transport channel identity <u>5</u>1 - CHOICE Transport channel type Dedicated transport channels - Dynamic Transport format information (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set - RLC Size - Number of TBs and TTI List (This IE is repeated for TFI number.) - Transmission Time Interval Not Present - Number of Transport blocks Reference to TS34.108 clause 6.10 Parameter Set **RLC size** Reference to TS34.108 clause 6.10 Parameter Set

ALL

- CHOICE Logical Channel list

- Semi-static Transport Format information

- Transmission time interval
- Type of channel coding
- Coding Rate
- Rate matching attribute
- CRC size

CPCH set ID

DRAC static information

DL Transport channel information common for all transport channel

- SCCPCH TFCS
- CHOICE DL parameters
- DL DCH TFCS
- Normal
- TFCI Field 1 information
- CHOICE CTFC representation
- TFCS addition information
- CHOICE CTFC Size
- CTFC information
- Power offset information

Deleted DL TrCH Information

- Transport channel identity

Deleted DL TrCH Information

- Transport channel identity

Deleted DL TrCH Information

Transport channel identity

Added or Reconfigured DL TrCH information

- Transport channel identity
- CHOICE DL parameters
- TFS
- CHOICE Transport channel type
- Dynamic Transport format information
- RLC Size
- Number of TBs and TTI List
 - Transmission Time Interval
 - _- Number of Transport blocks
 - RLC size
- CHOICE Logical Channel list
- Semi-static Transport Format information
- Transmission time interval
- Type of channel coding
- Coding Rate
- Rate matching attribute
- CRC size
- DCH quality target
- BLER Quality value
- Transparent mode signalling info

Frequency info

- UARFCN uplink(Nu)
- UARFCN downlink(Nd)

Maximum allowed UL TX power

Uplink DPCH info

- Uplink DPCH power control info
- DPCCH power offset
- PC Preamble
- Power Control Algorithm
- TPC step size
- Scrambling code type
- Scrambling code number
- Number of DPDCH
- spreading factor
- TFCI existence
- Number of FBI bit
- Puncturina Limit

CHOICE Mode

- Downlink PDSCH information

Downlink information common for all radio links

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Not Present

Not Present Independent

Not Preaent

(This IE is repeated for TFC number.)

Addition

Number of bits used must be enough to cover all combinations of CTFC from clause 6.10. Refer to TS34.108 clause 6.10 Parameter Set Not Present

<u>1</u>2

13

14

101

Independent

Dedicated transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

<u>-6.3</u>0.00 Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set 33dBm

-6dB 8slot Algorithm1 1dB Long

0 (0 to 16777215)

Not Present(1)

SF is reference to TS34.108 clause 6.10 Parameter Set

TRUE

Not Present(0)

Reference to TS34.108 clause 6.10 Parameter Set

FDD

Not Present

- Downlink DPCH info common for all RI **CHOICE** mode **EDD** - Timing indicator **Maintain** - CFN-targetSFN frame offset **Not Present** - Downlink DPCH power control information - DPC mode 0 (single) - DL rate matching restriction information Not Present - Spreading factor Reference to TS34.108 clause 6.10 Parameter Set - Fixed or Flexible Position N/A - TFCI existence **FALSE** - Number of bits for Pilot bits(SF=128,256) Reference to TS34.108 clause 6.10 Parameter Set - DPCH compressed mode info - TGPSI - TGPS Status Flag inactive - TGCFN (Current CFN + (256 - TTI/10msec)) mod 256 - Transmission gap pattern sequence configuration parameters - TGMP **FDD Measurement** - TGPRC - TGCEN (Current CFN + (256 - TTI/10msec)) mod 256 - TGSN 10 - TGL1 - TGL2 5 - TGD 15 - TGPL1 35 - TGPL2 35 - RPP Mode 1 - ITP Mode 1 - UL/DL Mode DL - Downlink compressed mode method SF/2 - Uplink compressed mode method Not Present - Downlink frame type - DeltaSIR1 2.0 - DeltaSIRafter1 1.0 - DeltaSIR2 Not Present - DeltaSIRafter2 Not Present - TX Diversity mode None - SSDT information Not Present - Default DPCH Offset Value Downlink information for each radio links - Primary CPICH info - Primary scrambling code 100 - PDSCH with SHO DCH info Not Present - PDSCH code mapping Not Present - Downlink DPCH info for each RL - Primary CPICH usage for channel estimation Primary CPICH may be used - DPCH frame offset 0 chips - Secondary CPICH info Not Present - DL channelisation code - Secondary scrambling code - Spreading factor Reference to TS34.108 clause 6.10 Parameter Set SF-1(SF is reference to TS34.108 clause 6.10 Parameter - Code number Set) - Scrambling code change No change - TPC combination index n - SSDT Cell Identity -a - Closed loop timing adjustment mode Not Present - Secondary CCPCH info Not Present Not Present - TFCS - FACH/PCH information Not Present - References to system information blocks Not Present Scheduling information

Contents of RADIO BEARER RELEASE message: AM or UM (Packet to CELL_FACH from CELL_DCH in PS)

Information Element	Value/remark
Message Type	
RRC transaction identifier	Arbitrarily selects an integer between 0 and 3
Integrity check info	The presence of this IE is dependent on IXIT statements
	in TS 34.123-2. If integrity protection is indicated to be active, this IE is present with the values of the sub IEs as
	stated below. Else, this IE and the sub-IEs are omitted.
- message authentication code	SS calculates the value of MAC-I for this message and
moodge admonitodation code	writes to this IE.
- RRC message sequence number	SS provides the value of this IE, from its internal counter.
Integrity protection mode info	Not Present
Ciphering mode info	Not Present
Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
New U-RNTI	Not Present
New C-RNTI RRC State DRX -indicator	Not Present NeDRXCELL_FACH
UTRAN DRX cycle length coefficient	Not Present
CN information info	Not Present
Signalling Connection release indication	Not Present
URA identity	Not Present
RAB information to reconfigure list	Not Present
RB information to release	
- RB identity	20 5
RB information to be affected	(UM DCCH for RRC)
- RB identity - RB mapping info	1
- Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
- Number of uplink RLC logical channels	1
- Uplink transport channel type	RACH
- Logical channel identity	1
- CHOICE RLC size list	All
- MAC logical channel priority	2
- Logical channel max loss	θ
- <u>Downlink RLC logical channel info</u> - Number of <u>downlink RLC logical channels</u>	1
- Downlink transport channel type	FACH
- Logical channel identity	1
RB information to be affected	(AM DCCH for RRC)
- RB identity	2
- RB mapping info	
- Information for each multiplexing option	l N + B
- RLC logical channel mapping indicator	Not Present
 Number of <u>uplink</u> RLC logical channels Uplink transport channel type 	RACH
- Logical channel identity	2
- CHOICE RLC size list	All
- MAC logical channel priority	3
- Logical channel max loss	θ
- Downlink RLC logical channel info	
- Number of <u>downlink</u> RLC logical channels	1
- Downlink transport channel type	FACH
- Logical channel identity RB information to be affected	2 (AM DCCH for NAS_DT High priority)
- RB identity	(AN DECH OF NAS_DT High phonty)
- RB mapping info	~
- Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
 Number of <u>uplink</u> RLC logical channels 	1
- Uplink transport channel type	RACH
- Logical channel identity	3
- CHOICE RLC size list	All
- MAC logical channel priority	4
- Logical channel max loss - Downlink RLC logical channel info	0
- DOWNINK KEC IOGICAI CHANNEI INIO	

- Number of downlink RLC logical channels	1
 Downlink transport channel type 	FACH
- Logical channel identity	3
RB information to be affected	(AM DCCH for NAS_DT Low priority)
- RB identity	4
- RB mapping info	
- Information for each multiplexing option	Not Decorat
- RLC logical channel mapping indicator	Not Present
- Number of <u>uplink</u> RLC logical channels	1 RACH
 Uplink transport channel type Logical channel identity 	4
- CHOICE RLC size list	All
- MAC logical channel priority	5
- Logical channel max loss	θ
- Downlink RLC logical channel info	8
- Number of downlink RLC logical channels	1
- Downlink transport channel type	FACH
- <u>DL</u> Transport channel identity	1
- Logical channel identity	4
RB information to be affected	(TM CCCH uplink for RRC)
- RB identity) o
- RB mapping info	
- Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
- Number of uplink RLC logical channels	1
- Uplink transport channel type	RACH
- Logical channel identity	5
- CHOICE RLC size list	<u>All</u>
- MAC logical channel priority	1
- Logical channel max loss	θ
- Downlink RLC logical channel info	Not Present
RB information to be affected	(UM CCCH downlink for RRC)
- RB identity	0
- RB mapping info	
- Information for each multiplexing option	
- Downlink RLC logical channel info	
- Number of downlink RLC logical channels	1
- Downlink transport channel type	FACH
- Logical channel identity RB information to be affected	(TM BCCH for BBC)
- RB identity	(TM BCCH for RRC)
- RB mapping info	O
- Information for each multiplexing option	
- Downlink RLC logical channel info	
- Number of <u>downlink</u> RLC logical channels	1
- Downlink transport channel type	FACH
- Logical channel identity	5
RB information to be affected	(TM PCCH for RRC)
- RB identity	7
- RB mapping info	
- Information for each multiplexing option	
- Downlink RLC logical channel info	
- Number of downlink RLC logical channels	1
 Downlink transport channel type 	PCH
- Logical channel identity	1
RB with PDCP information list	Not Present
UL Transport channel information for all transport	
channels	
- TFC subset	(This IE is repeated for TFC number.)
- Allowed Transport Format combination	0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108
DD A OU TEOO	clause 6.10 Parameter Set.)
- PRACH TFCS	Not Present
- CHOICE mode	FDD Not Present
- UL DCH TFCS	Not Present
Deleted UL TrCH Information	12
- Transport channel identity Added or Reconfigured UL TrCH information	12
- Transport channel identity	154
- Hansport Ghalliel Identity	<u>10</u> T

- TFS
- CHOICE Transport channel type
- Dynamic Transport format information
- RLC Size
- Number of TBs and TTI List
 - _- Number of Transport blocks
- RLC size
 - CHOICE mode
 - CHOICE Logical Channel List
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size

CPCH set ID

DRAC static information

DL Transport channel information common for all transport channel

- SCCPCH TFCS
- Normal
- TFCI Field 1 information
- CHOICE CTFC representation
- TFCS addition information
- CHOICE CTFC Size
- CTFC information
- Power offset information
- CHOICE DL parameters
- DL DCH TFCS

Deleted DL TrCH Information

Added or Reconfigured DL TrCH information

- Transport channel identity
- CHOICE DL parameters
- TFS
- CHOICE Transport channel type
- Dynamic Transport format information
- <mark>ŔLC Size</mark>
- Number of TBs and TTI List
 - _- Number of Transport blocks
- RLC size
 - CHOICE mode
 - CHOICE Logical Channel List
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size
 - DCH quality target
 - Transparent mode signalling info

Added or Reconfigured DL TrCH information

- Transport channel identity
- CHOICE DL parameters
- TFS
- CHOICE Transport channel type
- Dynamic Transport format information
- RLC Size
- Number of TBs and TTI List
- Number of Transport blocks
- RLC size
 - CHOICE mode
 - CHOICE Logical Channel List
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate

Common transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set FDD

ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Not Present

Not Preaent

(This IE is repeated for TFC number.)

Addition

Number of bits used must be enough to cover all combinations of CTFC from clause 6.10.
Refer to TS34.108 clause 6.10 Parameter Set
Not Present
Independent

Independent Not Present 6Not Present

134 Independent

Common transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set (This IE is repeated for TFI number.)

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set FDD

ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Not Present Not Present

<u>132</u>

Independent

Common transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set (This IE is repeated for TFI number.)

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set FDD

ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

- Rate matching attribute Reference to TS34.108 clause 6.10 Parameter Set - CRC size Reference to TS34.108 clause 6.10 Parameter Set - DCH quality target Not Present Not Present - Transparent mode signalling info Added or Reconfigured DL TrCH information Transport channel identity - CHOICE DL parameters Independent TES **Dynamic Transport format information** (This IE is repeated for TFI number) - Number of Transport blocks Reference to TS34.108 clause 6.10 Parameter Set **RLC size** Reference to TS34.108 clause 6.10 Parameter Set Semi-static Transport Format information Transmission time interval Reference to TS34.108 clause 6.10 Parameter Set Type of channel coding Reference to TS34.108 clause 6.10 Parameter Set Coding Rate Reference to TS34.108 clause 6.10 Parameter Set Rate matching attribute Reference to TS34.108 clause 6.10 Parameter Set -CRC size Reference to TS34.108 clause 6.10 Parameter Set DCH quality target Not Present Not Present Transparent mode signalling info Frequency info - UARFCN uplink(Nu) Reference to TS34.108 clause 6.10 Parameter Set - UARFCN downlink(Nd) Reference to TS34.108 clause 6.10 Parameter Set Maximum allowed UL TX power 33dBm CHOICE channel requirement PRACH info (for RACH)Not Present CHOICE mode **EDD** Available Signature '0000 0000 1111 1111'B Available SF Reference to TS34.108 clause 6.10 Parameter Set Scrambling code number **Puncturing Limit** Reference to TS34.108 clause 6.10 Parameter Set Available Sub Channel number '1111 1111 1111'B **CHOICE Mode** FDD - Downlink PDSCH information Not Present Downlink information common for all radio links Not Present Downlink information for each radio links Not Present Primary CPICH info Primary scrambling code 100 PDSCH with SHO DCH info Not Present PDSCH code mapping Not Present Downlink DPCH info for each RL Not Present Secondary CCPCH info Selection Indicator Not Present Primary CPICH may be used Primary CPICH usage for channel estimation Secondary CPICH info Not Present Secondary scrambling code channelisation code Secondary scrambling code SSDT Indicator **FALSE** Reference to TS34.108 clause 6.10 Parameter Set Spreading factor Code number SF-1(SF is reference to TS34.108 clause 6.10 Parameter Set) Pilot symbol existence **FALSE** TRUE TFCI existence Fixed or Flexible Position **Flexible** Timing offset **TFCS** Not Present **FACH/PCH** information Not Present References to system information blocks Not Present Scheduling information

Contents of RADIO BEARER RELEASE message: AM or UM (Packet to CELL_FACH from CELL_FACH in PS)

Information Element	Value/remark
Message Type	
RRC transaction identifier	Arbitrarily selects an integer between 0 and 3
Integrity check info	The presence of this IE is dependent on IXIT statements in
	TS 34.123-2. If integrity protection is indicated to be active, this IE is present with the values of the sub IEs as stated
	below. Else, this IE and the sub-IEs are omitted.
- message authentication code	SS calculates the value of MAC-I for this message and
message authentication code	writes to this IE.
- RRC message sequence number	SS provides the value of this IE, from its internal counter.
Integrity protection mode info	Not Present
Ciphering mode info	Not Present
Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
New U-RNTI	Not Present
New C-RNTI	Not Present
RRC State DRX indicator	NoDRXCELL_FACH
UTRAN DRX cycle length coefficient	Not Present
CN information info Signalling Connection release indication	Not Present Not Present
URA identity	Not Present
RAB information to be reconfigured list	Not Present
RB information to release	
- RB identity	20 5
RB information to be affected	(UM DCCH for RRC)
- RB identity	1
- RB mapping info	
- Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
- Number of <u>uplink</u> RLC logical channels	1 RACH
Uplink transport channel type Logical channel identity	1
- CHOICE RLC size list	All
- MAC logical channel priority	2
- Logical channel max loss	0
- Downlink RLC logical channel info	
- Number of <u>downlink</u> RLC logical channels	1
- Downlink transport channel type	FACH
- Logical channel identity RB information to be affected	1 (AM DCCH for RRC)
- RB identity	2
- RB mapping info	
- Information for each multiplexing option	
 RLC logical channel mapping indicator 	Not Present
 Number of <u>uplink</u> RLC logical channels 	1
- <u>UL</u> Transport channel identity	
- Logical channel identity	2
- CHOICE RLC size list - MAC logical channel priority	<u>All</u> 3
- MAC logical channel priority - Logical channel max loss	3 0
- Downlink RLC logical channel info	Ĭ
Number of <u>downlink</u> RLC logical channels	1
- Downlink transport channel type	FACH
- Logical channel identity	2
RB information to be affected	(AM DCCH for NAS_DT High priority)
- RB identity	3
- RB mapping info	
Information for each multiplexing option RLC logical channel mapping indicator	Not Present
- RLC logical channel mapping indicator - Number of <u>uplink</u> RLC logical channels	Not Present 1
- Uplink transport channel type	RACH
- Logical channel identity	3
- CHOICE RLC size list	All
- MAC logical channel priority	4
- Logical channel max loss	0
 - Downlink RLC logical channel info 	

- Number of <u>downlink</u> RLC logical channels	1 FACH
 Downlink transport channel type Logical channel identity 	3
RB information to be affected	(AM DCCH for NAS_DT Low priority)
- RB identity	4
- RB mapping info	
 Information for each multiplexing option 	
 RLC logical channel mapping indicator 	Not Present
 Number of <u>uplink</u> RLC logical channels 	1
- Uplink transport channel type	RACH
- Logical channel identity	4
- CHOICE RLC size list - MAC logical channel priority	<u>All</u> 5
- MAC logical channel phonty - Logical channel max loss	δ θ
- Downlink RLC logical channel info	•
- Number of downlink RLC logical channels	1
- Downlink transport channel type	FACH
- Logical channel identity	4
RB information to be affected	(TM CCCH uplink for RRC)
- RB identity	0
- RB mapping info	
- Information for each multiplexing option	Not Droppet
- RLC logical channel mapping indicator - Number of <u>uplink</u> RLC logical channels	Not Present 1
- Uplink transport channel type	RACH
- Logical channel identity	5
- CHOICE RLC size list	All
- MAC logical channel priority	1
- Logical channel max loss	0
 Downlink RLC logical channel info 	Not Present
RB information to be affected	(UM CCCH downlink for RRC)
- RB identity	0
- RB mapping info	
 Information for each multiplexing option Downlink RLC logical channel info 	
- Number of <u>downlink</u> RLC logical channels	1
- Downlink transport channel type	FACH
- Logical channel identity	6
RB information to be affected	(TM BCCH for RRC)
- RB identity	6
- RB mapping info	
- Information for each multiplexing option	
- Downlink RLC logical channel info	
 Number of <u>downlink</u> RLC logical channels Downlink transport channel type 	1 FACH
Logical channel identity	5
RB information to be affected	(TM PCCH for RRC)
- RB identity	7
- RB mapping info	
- Information for each multiplexing option	
 Downlink RLC logical channel info 	
 Number of <u>downlink</u> RLC logical channels 	1
- Downlink transport channel type	PCH
- Logical channel identity	1 Not Broomt
RB with PDCP information list UL Transport channel information for all transport	Not Present
channels	
- TFC subset	(This IE is repeated for TFC number.)
- Allowed Transport Format combination	0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108
	clause 6.10 Parameter Set.)
- PRACH TFCS	Not Present
- CHOICE mode	FDD
- UL DCH TFCS	Not Present
Deleted UL TrCH Information	15Not Present
Added or Reconfigured UL TrCH information	454
 Transport channel identity TFS 	<u>15</u> 4
- TFS - CHOICE Transport channel type	Common transport channels
OHOIOL Hansport channel type	Common transport oriallicis

(This IE is repeated for TFI number) - Dynamic Transport format information Reference to TS34.108 clause 6.10 Parameter Set - RLC Size - Number of TBs and TTI List (This IE is repeated for TFI number.) - Number of Transport blocks Reference to TS34.108 clause 6.10 Parameter Set RLC size Reference to TS34.108 clause 6.10 Parameter Set - CHOICE mode - CHOICE Logical Channel List ALL - Semi-static Transport Format information - Transmission time interval Reference to TS34.108 clause 6.10 Parameter Set - Type of channel coding Reference to TS34.108 clause 6.10 Parameter Set - Coding Rate Reference to TS34.108 clause 6.10 Parameter Set - Rate matching attribute Reference to TS34.108 clause 6.10 Parameter Set - CRC size Reference to TS34.108 clause 6.10 Parameter Set CPCH set ID Not Present DRAC static information Not Preaent DL Transport channel information common for all transport channel - SCCPCH TFCS (This IE is repeated for TFC number.) - Normal - TFCI Field 1 information - CHOICE CTFC representation Addition - TFCS addition information - CHOICE CTFC Size Number of bits used must be enough to cover all combinations of CTFC from clause 6.10. - CTFC information Refer to TS34.108 clause 6.10 Parameter Set - Power offset information Not Present - CHOICE DL parameters Independent - DL DCH TFCS Not Present Deleted DL TrCH Information - Transport channel identity <u>124</u> Added or Reconfigured DL TrCH information Not Present Frequency info - UARFCN uplink(Nu) Reference to TS34.108 clause 6.10 Parameter Set - UARFCN downlink(Nd) Reference to TS34.108 clause 6.10 Parameter Set Maximum allowed UL TX power 33dBm CHOICE channel requirement PRACH info (for RACH) Not Present CHOICE mode **EDD** Available Signature '0000 0000 1111 1111'B Available SF Reference to TS34.108 clause 6.10 Parameter Set Scrambling code number **Puncturing Limit** Reference to TS34.108 clause 6.10 Parameter Set Available Sub Channel number '1111 1111 1111'B **CHOICE Mode** FDD - Downlink PDSCH information Not Present Downlink information common for all radio links Not Present Downlink information for each radio links Not Present - Primary CPICH info Primary CPICH may be used 100 Primary scrambling code PDSCH with SHO DCH info Not PresentNot Present PDSCH code mapping Not Present Downlink DPCH info for each RL Not Present Secondary CCPCH info Selection Indicator Not Present Primary CPICH usage for channel estimation Primary CPICH may be used Secondary CPICH info Not Present Secondary scrambling code channelisation code Secondary scrambling code SSDT Indicator **FALSE** Spreading factor Reference to TS34.108 clause 6.10 Parameter Set SF-1(SF is reference to TS34.108 clause 6.10 Parameter Code number Set) Pilot symbol existence **FALSE TRUE** TFCI existence Fixed or Flexible Position **Flexible** Timing offset

Not Present

TFCS

	Not Present
- References to system information blocks	Not Present
- Scheduling information	

Contents of RRC CONNECTION RE-ESTABLISHMENT message: UM (The others of speech in CS)

Information Element	Value/remark
Message Type	
U-RNTI	Not Present
Integrity check info	The presence of this IE is dependent on IXIT statements
	in TS 34.123-2. If integrity protection is indicated to be
	active, this IE is present with the values of the sub IEs as
	stated below. Else, this IE and the sub-IEs are omitted.
- message authentication code	SS calculates the value of MAC-I for this message and
mossage aumenmeaner eeue	writes to this IE.
- RRC message sequence number	SS provides the value of this IE, from its internal counter.
Integrity protection mode info	Not Present
Ciphering mode info	Not Present
Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
New U-RNTI	Not Present
New C-RNTI	Not Present
DRX indicator	NoDRX
UTRAN DRX cycle length coefficient	Not Present
CN information info	Not Present
Signalling RB information to setup	(UM DCCH for RRC)
- RB identity	4
- CHOICE RLC info type	
- RLC info	
	UM RLC
- Transmission RLC discard	
- SDU discard mode	Max DAT retransmissions
- MAX_DAT	4
	100
	4
- CHOICE Downlink RLC mode	UM RLC
- Information for each multiplexing option	
- Number of RLC logical channels	1
- Uplink transport channel type	DCH
- Transport channel identity	1
	1
	1
- Logical channel max loss	θ
- Number of RLC logical channels	1
	DCH
- Transport channel identity	1
- Logical channel identity	1
Signalling RB information to setup	(AM DCCH for RRC)
- RB identity	2
- CHOICE RLC info type	
- RLC info	
	AM RLC
- Transmission RLC discard	L BAT ()
- SDU discard mode	Max DAT retransmissions
- MAX_DAT	4
- Timer_MRW	100
- MaxMRW	4
- Transmission window size	8
	500
- Max_RST	4
- Polling info	200
- Timer_poll_prohibit	200 200
	1 TDIJE
- Last transmission PU poll	TRUE
- Last retransmission PU poll	TRUE
	99 AM PL C
	AM RLC
	TRUE
- Receiving window size	8
- Downlink RLC status info	200
	200

1	
——————————————————————————————————————	200
- Missing PU indicator	TRUE
- RB mapping info	
- Information for each multiplexing option	
- Number of RLC logical channels	1
- Uplink transport channel type	DCH
- Transport channel identity	1
- Logical channel identity	2
	2
- MAC logical channel priority	
- Logical channel max loss	0
- Number of RLC logical channels	4
Downlink transport channel type	DCH
	1
Logical channel identity	2
Signalling RB information to setup	(AM DCCH for NAS_DT High priority)
- RB identity	3
- CHOICE RLC info type	
	AM RLC
- Transmission RLC discard	AWINEO
SDU discard mode	Max DAT retransmissions
	Max DAT retransmissions 4
- MAX_DAT	·
- Timer_MRW	100
——————————————————————————————————————	4
- Transmission window size	8
——————————————————————————————————————	500
——————————————————————————————————————	4
- Polling info	
	200
Timer_poll	200
Poll_SDU	1
- Last transmission PU poll	TRUE
- Last retransmission PU poll	TRUE
	99
Poll_Windows - CHOICE Downlink RLC mode	AM RLC
- In-sequence delivery	TRUE
- Receiving window size	8
- Downlink RLC status info	
	200
- Timer_EPC	200
——————————————————————————————————————	TRUE
- RB mapping info	
- Information for each multiplexing option	
- Number of RLC logical channels	4
- Uplink transport channel type	DCH
- Transport channel identity	1
- Logical channel identity	3
- MAC logical channel priority	3
- Logical channel max loss	0
- Number of RLC logical channels	1
- Downlink transport channel type	DCH
	1
- Logical channel identity	3
Signalling RB information to setup	(AM DCCH for NAS_DT Low priority)
RB identity	4
- CHOICE RLC info type	
- CHOICE Uplink RLC mode	AM RLC
- Transmission RLC discard	
- SDU discard mode	Max DAT retransmissions
- SDO diseale mode - MAX_DAT	4
	-
	100
- MaxMRW	4
- Transmission window size	8
	500
——————————————————————————————————————	4
Polling info	
- Timer_poll_prohibit	200

——————————————————————————————————————	200
Poll_SDU	1
- Last transmission PU poll	TRUE
	TRUE
	99
	AM RLC
	TRUE
- Receiving window size	8
- Downlink RLC status info	
	200
——————————————————————————————————————	200
- Missing PU indicator	TRUE
- Information for each multiplexing option	
- Number of RLC logical channels	1
	DCH
Transport channel identity	1
	4
——————————————————————————————————————	4
- Logical channel max loss	θ
	1
Downlink transport channel type	DCH
Transport channel identity	1
	4
RAB information for setup	
——- RAB info	
——- RAB identity	0000 0001B
	CS domain
	20 seconds
- RB information to setup	
RB identity	5
— - PDCP info	Not Present
——- RLC info	
- CHOICE Uplink RLC mode	TM RLC
- Transmission RLC Discard	Not Present
- Segmentation Indication	TRUE
	TM RLC
	TRUE
- Number of RLC logical channels	1
Uplink transport channel type	DCH
	2
- Logical channel identity	1
	1
- Logical channel max loss	0
- Number of RLC logical channels	1
- Downlink transport channel type	DCH
- Transport channel identity	2
	1
RB information to release	Not Present
RB information to reconfigure	Not Present
RB information to be affected	Not Present
UL Transport channel information for all transport	
channels	
	(This IE is repeated for TFC number.)
- Allowed Transport Format combination	0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108 clause 6.10 Parameter Set.)
	Not Present
- CHOICE mode	FDD
— - UL DCH TFCS	(This IE is repeated for TFC number.)
——- Normal	,
- TFCI Field 1 information	
- CHOICE CTFC representation	Addition
- TFCS addition information	
- CHOICE CTFC Size	Number of bits used must be enough to cover all

	combinations of CTFC from clause 6.10.
- CTFC information	Refer to TS34.108 clause 6.10 Parameter Set
- Power offset information	
	Signalled Gain Factor
- Gain factor •c	O O
- Gain factor •d	$\frac{\theta}{\theta}$
- Reference TFC ID	Not Present
	0dB
Deleted UL TrCH information	Not Present
Added or Reconfigured UL TrCH information	
- Transport channel identity	1
- TFS	·
- Dynamic Transport format information	(This IE is repeated for TFI number)
	Reference to TS34.108 clause 6.10 Parameter Set
- Number of Transport blocks	
- RLC size	Reference to TS34.108 clause 6.10 Parameter Set
 Semi-static Transport Format information 	
- Transmission time interval	Reference to TS34.108 clause 6.10 Parameter Set
- Type of channel coding	Reference to TS34.108 clause 6.10 Parameter Set
- Coding Rate	Reference to TS34.108 clause 6.10 Parameter Set
- Rate matching attribute	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
Added or Reconfigured UL TrCH information	
- Transport channel identity	2
	
- Dynamic Transport format information	(This IE is repeated for TFI number)
- Number of Transport blocks	Reference to TS34.108 clause 6.10 Parameter Set
- RI C size	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to 1994. Too Gause 0. TO Farameter Set
- Semi-static Transport Format information	
- Transmission time interval	Reference to TS34.108 clause 6.10 Parameter Set
- Type of channel coding	Reference to TS34.108 clause 6.10 Parameter Set
- Coding Rate	Reference to TS34.108 clause 6.10 Parameter Set
- Rate matching attribute	Reference to TS34.108 clause 6.10 Parameter Set
- CRC size	Reference to TS34.108 clause 6.10 Parameter Set
0.1000=0	
DRAC static information	Not Present
DL Transport channel information common for all	
transport channel	
	Not Present
- CHOICE DL parameters	Independent
— - DL DCH TFCS	(This IE is repeated for TFC number.)
- Normal	(This is to repeated for the Hamber.)
- TFCI Field 1 information	
- CHOICE CTFC representation	Addition
- CHOICE CTFC Size	Number of bits used must be enough to cover all
OTTOTOL OTTO OILO	combinations of CTFC from clause 6.10.
0.750 : 4:	
- CTFC information	Refer to TS34.108 clause 6.10 Parameter Set
	Not Present
Deleted DL TrCH information	Not Present
Added or Reconfigured DL TrCH information	
	1
Added or Reconfigured DL TrCH information - Transport channel identity CHOICE DI parameters	4 Independent
- Transport channel identity - CHOICE DL parameters	4 Independent
- Transport channel identity - CHOICE DL parameters - TFS	Independent
- Transport channel identity - CHOICE DL parameters - TFS - Dynamic Transport format information	Independent (This IE is repeated for TFI number)
- Transport channel identity - CHOICE DL parameters - TFS	Independent
Transport channel identity CHOICE DL parameters TES Dynamic Transport format information Number of Transport blocks	Independent (This IE is repeated for TFI number)
- Transport channel identity - CHOICE DL parameters - TFS - Dynamic Transport format information - Number of Transport blocks - RLC size	(This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set
Transport channel identity CHOICE DL parameters TFS Dynamic Transport format information Number of Transport blocks RLC size Semi-static Transport Format information	Independent (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set
Transport channel identity CHOICE DL parameters TFS Dynamic Transport format information Number of Transport blocks RLC size Semi-static Transport Format information Transmission time interval	Independent (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set
Transport channel identity CHOICE DL parameters TFS Dynamic Transport format information Number of Transport blocks RLC size Semi-static Transport Format information Transmission time interval Type of channel coding	Independent (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set
Transport channel identity CHOICE DL parameters TFS Dynamic Transport format information Number of Transport blocks RLC size Semi-static Transport Format information Transmission time interval Type of channel coding Coding Rate	Independent (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set
Transport channel identity CHOICE DL parameters TFS Dynamic Transport format information Number of Transport blocks RLC size Semi-static Transport Format information Transmission time interval Type of channel coding	Independent (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set
Transport channel identity CHOICE DL parameters TFS Dynamic Transport format information Number of Transport blocks RLC size Semi-static Transport Format information Transmission time interval Type of channel coding Coding Rate Rate matching attribute	Independent (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set
Transport channel identity CHOICE DL parameters TFS Dynamic Transport format information Number of Transport blocks RLC size Semi-static Transport Format information Transmission time interval Type of channel coding Coding Rate Rate matching attribute CRC size	Independent (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set
Transport channel identity CHOICE DL parameters TFS Dynamic Transport format information Number of Transport blocks RLC size Semi-static Transport Format information Transmission time interval Type of channel coding Coding Rate Rate matching attribute CRC size DCH quality target	Independent (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set
Transport channel identity CHOICE DL parameters TFS Dynamic Transport format information Number of Transport blocks RLC size Semi-static Transport Format information Transmission time interval Type of channel coding Coding Rate Rate matching attribute CRC size DCH quality target BLER Quality value	Independent (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set
Transport channel identity CHOICE DL parameters TFS Dynamic Transport format information Number of Transport blocks RLC size Semi-static Transport Format information Transmission time interval Type of channel coding Coding Rate Rate matching attribute CRC size DCH quality target BLER Quality value Transparent mode signalling info	Independent (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set
- Transport channel identity - CHOICE DL parameters - TES - Dynamic Transport format information - Number of Transport blocks - RLC size - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size - DCH quality target - BLER Quality value - Transparent mode signalling info Added or Reconfigured DL TrCH information	Independent (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set
- Transport channel identity - CHOICE DL parameters - TES - Dynamic Transport format information - Number of Transport blocks - RLC size - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size - DCH quality target - BLER Quality value - Transparent mode signalling info	Independent (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set

——- TES	
Dynamic Transport format information	(This IE is repeated for TFI number)
- Number of Transport blocks	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
- Semi-static Transport Format information	
- Transmission time interval	Reference to TS34.108 clause 6.10 Parameter Set
- Type of channel coding	Reference to TS34.108 clause 6.10 Parameter Set
- Coding Rate	Reference to TS34.108 clause 6.10 Parameter Set
- Rate matching attribute	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
— - DCH quality target	
- BLER Quality value	0.00
	Not Present
Frequency info	
- UARFCN uplink(Nu)	Reference to TS34.108 clause 6.10 Parameter Set
——- UARFCN downlink(Nd)	Reference to TS34.108 clause 6.10 Parameter Set
Maximum allowed UL TX power	33dBm
Uplink DPCH info	
Uplink DPCH power control info	
- DPCCH power offset	-6dB
	8slot
- Power Control Algorithm	Algorithm1
- TPC step size	1dB
	Long
- Scrambling code number	0 (0 to 16777215)
	Not Present(1)
	SF-1(SF is reference to TS34.108 clause 6.10 Parameter
	Set)
	TRÚE
- Number of FBI bit	Not Present(0)
- Puncturing Limit	Reference to TS34.108 clause 6.10 Parameter Set
Downlink PDSCH information	Not Present
CPCH SET info	Not Present

Information Element	Value/remark
Message Type	
Integrity check info	The presence of this IE is dependent on IXIT statements in TS 34.123-2. If integrity protection is indicated to be
	active, this IE is present with the values of the sub IEs as
manage of the entire tion and a	stated below. Else, this IE and the sub-IEs are omitted.
- message authentication code	SS calculates the value of MAC-I for this message and writes to this IE.
- RRC message sequence number	SS provides the value of this IE, from its internal counter.
RRC transaction identifier	Arbitrarily selects an integer between 0 and 3
Integrity protection mode info	Not Present
Ciphering mode info	Not Present
New U-RNTI	See the test content
- SRNC identity	Goo the tool contoni
New C-RNTI	See the test content
DRX indicator	NoDRX
UTRAN DRX cycle length coefficient	Not Present
UE Timers and constants in connected mode	
<u>T301</u>	2000 milliseconds
<u>- N301</u>	<u>2</u>
<u>T302</u>	4000 milliseconds
<u>- N302</u>	3
<u>- T304</u>	1000 milliseconds
- N304 T205	3
<u>- T305</u> - T307	60 minutes
- 1307 - T308	50 seconds 320 milliseconds
- T306 - T309	8 seconds
- T310	320 milliseconds
- N310	5
- T311	500 milliseconds
- T312	5 seconds
- N312	200
- T313	10 seconds
- N313	200
<u>- T314</u>	20 seconds
<u>T315</u>	30 seconds
<u>- N315</u>	<u>200</u>
<u>- T316</u>	50 seconds
<u>- T317</u>	1800 seconds
CN information info	Not Present
URA identity	Not present
RB with PDCP information list	Not Present

Contents of UTRAN MOBILITY INFORMATION CONFIRM message: AM

Information Element	Value/remark
Message Type	
RRC transaction identifier	Checked to see if it matches the value of the same IE in
	downlink UTRAN MOBILITY INFORMATION message
Integrity check info	The presence of this IE is dependent on IXIT statements
	in TS 34.123-2. If integrity protection is indicated to be
	active, this IE shall be present with the values of the sub
	IEs as stated below. Else, this IE and the sub-IEs shall be
	absent.
- Message authentication code	This IE is checked to see if it is present. The value is
	compared against the XMAC-I value computed by SS.
- RRC Message sequence number	This IE is checked to see if it is present. The value is
	used by SS to compute the XMAC-I value.
COUNT-C activation time	The presence of this IE depends on the following 2
	factors: (a) There exists RB(s) mapped to RLC-TM, (b)
	UE is transiting to CELL_DCH state after the
	reconfiguration procedure. Else, this IE is absent.
Uplink integrity protection activation info	Not checked
Radio bearer uplink ciphering activation time info	Not checked
RB with PDCP information list	Not checked

Contents of RRC CONNECTION RE-ESTABLISHMENT message: UM (Speech in CS)

Information Element	Value/remark
Message Type	varao/romani
U-RNTI	Not Present
Integrity check info	The presence of this IE is dependent on IXIT statements in
Integrity official into	TS 34.123-2. If integrity protection is indicated to be
	active, this IE is present with the values of the sub IEs as
	stated below. Else, this IE and the sub-IEs are omitted.
- message authentication code	SS calculates the value of MAC-I for this message and
message authentication code	writes to this IE.
- RRC message sequence number	SS provides the value of this IE, from its internal counter.
Integrity protection mode info	Not Present
Ciphering mode info	The presence of this IE is dependent on IXIT statements in
Ophoning mode into	TS 34.123-2. If ciphering is indicated to be active, this IE
	present with the values of the sub IEs as stated below.
	Else, this IE is omitted from this message
- Ciphering mode command	start
- Ciphering algorithm	Use one of the supported ciphering algorithms
	(256+CFN-(CFN MOD 8 + 8))MOD 256
- Radio bearer downlink ciphering activation time	(20010114 (0114 MOD 0 1 0))MOD 200
info	
- Radio bearer identity	0
- RLC sequence number	Current RLC SN + 2
- Radio bearer identity	4
- RLC sequence number	Current RLC SN+2
- Radio bearer identity	2
- RLC sequence number	Current RLC SN+2
- Radio bearer identity	3
- RLC sequence number	Current RLC SN+2
- Radio bearer identity	4
- RLC sequence number	Current RLC SN+2
Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
New U-RNTI	Not Present
New C-RNTI	Not Present
DRX indicator	noDRX
UTRAN DRX cycle length coefficient	Not Present
RLC reset indicator (for C-plane)	FALSE
RLC reset indicator (for U-plane)	FALSE
CN information info	Not Present
Signalling RB information to setup	(UM DCCH for RRC)
- RB identity	1 1
- RLC info	
- CHOICE Uplink RLC mode	UM RLC
- Transmission RLC discard	
- SDU discard mode	Max DAT retransmissions
——————————————————————————————————————	4
- Timer_MRW	100
	4
- CHOICE Downlink RLC mode	UM RLC
- RB mapping info	
 Information for each multiplexing option 	
- Number of RLC logical channels	4
- Uplink transport channel type	DCH
- Transport channel identity	4
- Logical channel identity	4
- MAC logical channel priority	4
- Logical channel max loss	0
- Number of RLC logical channels	1
- Downlink transport channel type	DCH
- Transport channel identity	1
- Logical channel identity	1
Signalling RB information to setup	(AM DCCH for RRC)
	2
- CHOICE Uplink RLC mode	AM RLC

- Transmission RLC discard	1
- SDU discard mode	Max DAT retransmissions
- MAX DAT	4
Timer MRW	100
- MaxMRW	4
- Transmission window size	8
- Timer_RST	500
- Max RST	4
- Polling info	
- Timer_poll_prohibit	200
- Timer_poll	200
- Poll_SDU	1
- Last transmission PU poll	TRUE
- Last retransmission PU poll	TRUE
	99
	AM RLC
- In-sequence delivery	TRUE
- Receiving window size	8
- Downlink RLC status info	
- Timer_status_prohibit	200
Timer_EPC	200
- Missing PU indicator	TRUE
- RB mapping info	
Information for each multiplexing option	
- Number of RLC logical channels	1
- Uplink transport channel type	DCH
	1
- Logical channel identity	2
	2
- Logical channel max loss	0
- Number of RLC logical channels	1
	DCH
- Transport channel identity	4
- Logical channel identity	(AM DOOLL for NAC DT High and often)
Signalling RB information to setup	(AM DCCH for NAS_DT High priority)
- RB identity - CHOICE RLC info type	3
RLC info	
	AM RLC
- Transmission RLC discard	TWINEO
- SDU discard mode	Max DAT retransmissions
- MAX_DAT	4
- Timer_MRW	100
	4
- Transmission window size	8
- Timer_RST	500
- Max_RST	4
- Polling info	
- Timer_poll_prohibit	200
- Timer_poll	200
Poll_SDU	4
- Last transmission PU poll	TRUE
- Last retransmission PU poll	TRUE
	99
- CHOICE Downlink RLC mode	AM RLC
	TRUE
- Timer_status_prohibit	200
	200
	TRUE
- RB mapping info	
- Information for each multiplexing option	
- Number of RLC logical channels	1
- Uplink transport channel type	DCH
I repend the channel identity	1
- Transport channel identity	0
- Logical channel identity	3
	3 3 0

- Number of RLC logical channels	4
- Downlink transport channel type	DCH
- Transport channel identity	4
	3
Signalling RB information to setup	(AM DCCH for NAS_DT Low priority)
- RB identity	4
- CHOICE RLC info type	
- RLC info	
- CHOICE Uplink RLC mode	AM RLC
- Transmission RLC discard	
- SDU discard mode	Max DAT retransmissions
	4
	100
- MaxMRW	4
- Transmission window size	8
	500
	4
	200
- Timer_poll	200
	4
- Last transmission PU poll	TRUE
- Last retransmission PU poll	TRUE
	99
	AM RLC
- In-sequence delivery	TRUE
- Receiving window size	8
- Downlink RLC status info	
- Timer_status_prohibit	200
- Timer_EPC	200
- Missing PU indicator	TRUE
- RB mapping info	
- Information for each multiplexing option	
- Number of RLC logical channels	4
- Uplink transport channel type	DCH
- Transport channel identity	4
- Logical channel identity	4
- MAC logical channel priority	4
- Logical channel max loss	θ
- Number of RLC logical channels	4
- Downlink transport channel type	DCH
- Transport channel identity	4
- Logical channel identity	4
RAB information for setup	
	0000 0001B
	CS domain
- Re-establishment timer	
	20 seconds
- RB information to setup	
	5
	Not Present
	TM RLC
- Transmission RLC Discard	Not Present
- Segmentation Indication	TRUE
	TM RLC
- Segmentation Indication	TRUE
- RB mapping info	
 Information for each multiplexing option 	
- Number of RLC logical channels	4
- Uplink transport channel type	DCH
- Transport channel identity	2
- Logical channel identity	4
- MAC logical channel priority	4
l catalian i i i	
- Logical channel max loss	0
- Logical channel max loss - Number of RLC logical channels	4 DCH

- Transport channel identity	2
- Logical channel identity	1
- RB information to setup	
	6
- PDCP info	Not Present
	Not Present
'	
- CHOICE Uplink RLC mode	TM RLC
- Transmission RLC Discard	Not Present
- Segmentation Indication	TRUE
- CHOICE Downlink RLC mode	TM RLC
	TRUE
- RB mapping info	
- Information for each multiplexing option	
- Number of RLC logical channels	1
- Uplink transport channel type	DCH
- Transport channel identity	3
- Logical channel identity	1
——————————————————————————————————————	1
	θ
- Number of RLC logical channels	1
- Downlink transport channel type	DCH
- Transport channel identity	3
- Logical channel identity	1
	·
- RB information to setup	(This IE is needed for 12.2 kbps and 10.2 kbps)
- RB identity	7
— - PDCP info	Not Present
RLC info	
- CHOICE Uplink RLC mode	TM RLC
- Transmission RLC Discard	Not Present
- Segmentation Indication	TRUE
- CHOICE Downlink RLC mode	TM RI C
- Segmentation Indication	TRUE
RB mapping info	
- Number of RLC logical channels	4
- Uplink transport channel type	DCH
- Transport channel identity	4
- Logical channel identity	1
- MAC logical channel priority	1
- Logical channel max loss	0
- Number of RLC logical channels	1
- Downlink transport channel type	DCH
- Transport channel identity	4
	1
RB information to release	Not Present
RB information to reconfigure	Not Present
RB information to be affected	Not Present
	INOLFIESSIII
UL Transport channel information for all transport	
channels	
——- TFC subset	(This IE is repeated for TFC number.)
- Allowed Transport Format combination	0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108
'	clause 6.10 Parameter Set.)
- PRACH TECS	Not Present
——————————————————————————————————————	FDD
	' = =
— - UL DCH TFCS	(This IE is repeated for TFC number.)
- CHOICE CTFC representation	Addition
- TFCS addition information	
CHOICE CTEC Size	Number of hits used must be enough to sever all
- CHOICE CTFC Size	Number of bits used must be enough to cover all
	combinations of CTFC from clause 6.10.
	Refer to TS34.108 clause 6.10 Parameter Set
- Power offset information	
	Signalled Gain Factor
- Gain factor +c	0
- Gain factor •d	0
Reference TFC ID	Not Present

- Power offset Pp-m	0dB
Deleted UL TrCH information	Not Present
Added or Reconfigured UL TrCH information	
	1
TFS	
- Dynamic Transport format information	(This IE is repeated for TFI number)
- Number of Transport blocks	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
- Semi-static Transport Format information	D (T004 400 L
- Transmission time interval	Reference to TS34.108 clause 6.10 Parameter Set
- Type of channel coding	Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set
- Coding Rate	
	Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set
Added or Reconfigured UL TrCH information	THE PROPERTY OF THE PROPERTY O
	2
Transport charmer actuary	=
——————————————————————————————————————	(This IE is repeated for TFI number)
- Number of Transport blocks	Reference to TS34.108 clause 6.10 Parameter Set
- RI C size	Reference to TS34.108 clause 6.10 Parameter Set
- Semi-static Transport Format information	Treatment to 100 i. 100 diadec c. 10 1 diameter cot
- Transmission time interval	Reference to TS34.108 clause 6.10 Parameter Set
- Type of channel coding	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
- Rate matching attribute	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
Added or Reconfigured UL TrCH information	
- Transport channel identity	3
- Dynamic Transport format information	(This IE is repeated for TFI number)
- Number of Transport blocks	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
- Semi-static Transport Format information	
	Reference to TS34.108 clause 6.10 Parameter Set
- Type of channel coding	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
- Rate matching attribute	Reference to TS34.108 clause 6.10 Parameter Set
- CRC size	Reference to TS34.108 clause 6.10 Parameter Set
Added or Reconfigured UL TrCH information	(This IE is needed for 12.2 kbps and 10.2 kbps)
- Transport channel identity	4
——— TFS ——— Dynamic Transport format information	(This IE is repeated for TFI number)
- Dynamic Transport format information - Number of Transport blocks	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
- Semi-static Transport Format information	Reference to 1934, 100 clause 6, 10 Fatameter set
- Transmission time interval	Reference to TS34.108 clause 6.10 Parameter Set
- Type of channel coding	Reference to TS34.108 clause 6.10 Parameter Set
- Coding Rate	Reference to TS34.108 clause 6.10 Parameter Set
- Rate matching attribute	Reference to TS34.108 clause 6.10 Parameter Set
- CRC size	Reference to TS34.108 clause 6.10 Parameter Set
DRAC static information	Not Present
DL Transport channel information common for all	Not i room.
transport channel	
— SCCPCH TFCS	Not Present
	Independent
— - DL DCH TFCS	(This IE is repeated for TFC number.)
	(
- TFCI Field 1 information	
- CHOICE CTFC representation	Addition
- TFCS addition information	
OHOLOE OTTO O	No. 1 and the second second
	Number of bits used must be enough to cover all
OTEO intra di	combinations of CTFC from clause 6.10.
	Refer to TS34.108 clause 6.10 Parameter Set
— Power offset information Deleted DL TrCH information	Not Present
	Not Present
Added or Reconfigured DL TrCH information	1
- Transport channel identity	4

	SameAsUL
- UL TrCH Identity	1
- DCH quality target	
BLER Quality value	0.00
- Transparent mode signalling info	Not Present
Added or Reconfigured DL TrCH information	HOLL LOOPIN
	2
- Transport channel identity	
CHOICE DL parameters	SameAsUL
- UL TrCH Identity	2
— - DCH quality target	
	0.00
- Transparent mode signalling info	Not Present
Added or Reconfigured DL TrCH information	
- Transport channel identity	3
- CHOICE DL parameters	SameAsUL
- UL TrCH Identity	3
	- 3
DCH quality target	
BLER Quality value	0.00
	Not Present
Added or Reconfigured DL TrCH information	(This IE is needed for 12.2 kbps and 10.2 kbps)
- Transport channel identity	4
- CHOICE DL parameters	SameAsUL
— - UL TrCH Identity	4
- DCH quality target	
BLER Quality value	0.00
- Transparent mode signalling info	Not Present
	Not Present
Frequency info	
- UARFCN uplink(Nu)	Reference to TS34.108 clause 6.10 Parameter Set
——- UARFCN downlink(Nd)	Reference to TS34.108 clause 6.10 Parameter Set
Maximum allowed UL TX power	33dBm
Uplink DPCH info	
- Uplink DPCH power control info	
- DPCCH power offset	-6dB
	8slot
- Power Control Algorithm	Algorithm1
- TPC step size	1 1dB
- Scrambling code type	Long
- Scrambling code number	0 (0 to 16777215)
	Not Present(1)
	SF is reference to TS34.108 clause 6.10 Parameter Set
- TFCI existence	TRUE
- Number of FBI bit	Not Present(0)
- Puncturing Limit	Reference to TS34.108 clause 6.10 Parameter Set
Downlink PDSCH information	Not Present
CPCH SET info	Not Present
Downlink information for each radio links	11011100011
- Primary CPICH info	100
- Primary scrambling code	100
	Not Present
- PDSCH code mapping	Not Present
— - Downlink DPCH info for each RL	
- Primary CPICH usage for channel estimation	Primary CPICH may be used
- DPCH frame offset	0 chips
- Secondary CPICH info	Not Present
- DL channelisation code	
- Secondary scrambling code	4
Spreading factor	Reference to TS34.108 clause 6.10 Parameter Set
- Code number	SF-1(SF is reference to TS34.108 clause 6.10 Parameter
	Set)
	No change
- TPC combination index	0
	-a
- Closed loop timing adjustment mode	Not Present
- Secondary CCPCH info	Not Present
——————————————————————————————————————	Not Present
FACH/PCH information	Not Present
11	
- References to system information blocks	Not Present

Contents of RRC CONNECTION RE-ESTABLISHMENT message: UM (Packet to CELL_DCH from CELL_DCH in PS)

Information Element	Value/remark
Message Type	
U-RNTI	Not Present
Integrity check info	The presence of this IE is dependent on IXIT statements in
	TS 34.123-2. If integrity protection is indicated to be active,
	this IE is present with the values of the sub IEs as stated
	below. Else, this IE and the sub-IEs are omitted.
- message authentication code	SS calculates the value of MAC-I for this message and
g	writes to this IE.
- RRC message sequence number	SS provides the value of this IE, from its internal counter.
Integrity protection mode info	Not Present
Ciphering mode info	The presence of this IE is dependent on IXIT statements in
- Free mag	TS 34.123-2. If ciphering is indicated to be active, this IE
	present with the values of the sub IEs as stated below.
	Else, this IE is omitted from this message.
- Ciphering mode command	start
	Use one of the supported ciphering algorithms
- Activation time for DPCH	Not Present
- Radio bearer downlink ciphering activation time	
info	
- Radio bearer identity	0
	Current RLC SN + 2
- Radio bearer identity	4
- RLC sequence number	Current RLC SN+2
	2
- RLC sequence number	Current RLC SN+2
- Radio bearer identity	3
- RLC sequence number	Current RLC SN+2
- Radio bearer identity	4
	Current RLC SN+2
Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
New U-RNTI	Not Present
New C-RNTI	Not Present
DRX indicator	noDRX
UTRAN DRX cycle length coefficient	Not Present
RLC reset indicator (for C-plane)	FALSE
RLC reset indicator (for U-plane)	FALSE
CN information info	Not Present
Signalling RB information to setup	(UM DCCH for RRC)
	4
- CHOICE RLC info type	
	UM RLC
	UM RLU
- Transmission RLC discard - SDU discard mode	Max DAT retransmissions
	Wax DAT retransmissions 4
Timer MRW	4 1 00
- MaxMRW	4
- CHOICE Downlink RLC mode	UM RLC
- RB mapping info	OM NEO
- Information for each multiplexing option	
- Number of RLC logical channels	4
- Uplink transport channel type	DCH
- Transport channel identity	4
- Logical channel identity	1
- MAC logical channel priority	1
- Logical channel max loss	θ
- Number of RLC logical channels	1
- Downlink transport channel type	DCH
- Transport channel identity	4
- Logical channel identity	4
Signalling RB information to setup	(AM DCCH for RRC)
- RB identity	2
- CHOICE ŘLC info type	
•	·

	AM RLC
- Transmission RLC discard	
- SDU discard mode	Max DAT retransmissions
- MAX_DAT	4
- Timer_MRW - MaxMRW	100
	4 8
	0 500
	200 4
——————————————————————————————————————	4
Timer_poll_prohibit	200
Timer_poll	200
Poll_SDU	1
Last transmission PU poll	TRUE
- Last retransmission PU poll	TRUE
- Poll Windows	99
- CHOICE Downlink RLC mode	AMRIC
- In-sequence delivery	TRUE
- Receiving window size	8
- Downlink RLC status info	
- Timer_status_prohibit	200
- Timer_EPC	200
- Missing PU indicator	TRUE
- RB mapping info	
- Information for each multiplexing option	
- Number of RLC logical channels	4
- Uplink transport channel type	DCH
- Transport channel identity	1
Logical channel identity	2
- MAC logical channel priority	2
- Logical channel max loss	0
	4
- Downlink transport channel type	DCH
	4
- Logical channel identity	2
Signalling RB information to setup	(AM DCCH for NAS_DT High priority)
RB identity	3
- CHOICE Uplink RLC mode	AM RLC
- Transmission RLC discard	N DAT A SECOND
- SDU discard mode	Max DAT retransmissions
MAX_DAT	4
- Timer_MRW	100
- MaxMRW	4
- Transmission window size	8
- Timer_RST	500
- Max_RST	4
- Polling info	200
Timer_poll_prohibit	200 200
— Timer_poll Poll_SDU	
	1 TRUE
Last transmission PU poll Last retransmission PU poll	TRUE
	99
——————————————————————————————————————	AM RLC
- In-sequence delivery	TRUE
Receiving window size	1KUE 8
- Downlink RLC status info	Ĭ
- Timer_status_prohibit	200
Timer_EPC	200
- Missing PU indicator	TRUE
- RB mapping info	
- Information for each multiplexing option	
- Number of RLC logical channels	4
- Uplink transport channel type	DCH
- Transport channel identity	1
- Logical channel identity	3

- MAC logical channel priority	3
- Logical channel max loss	0
- Number of RLC logical channels	1
 Downlink transport channel type 	DCH
- Transport channel identity	1
- Logical channel identity	3
Signalling RB information to setup	(AM DCCH for NAS_DT Low priority)
- RB identity	4
- CHOICE RLC info type	
- RLC info	
	AM RLC
- Transmission RLC discard	M DAT ()
	Max DAT retransmissions
- MAX_DAT	4 100
- Timer_MRW	1
- MaxMRW	4
- Transmission window size	8 500
	4
- Polling info	200
- Timer_pell_prehibit	200 200
	200 1
	+ TRUE
- Last transmission PU poll - Last retransmission PU poll	TRUE
	1RUE 99
	AM RLC
	TRUE
- In-sequence delivery - Receiving window size	180E 8
- Downlink RLC status info	8
- Timer_status_prohibit	200
Timer_EPC	200 200
- Missing PU indicator	TRUE
	TROE
- Information for each multiplexing option	
- Number of RLC logical channels	4
- Uplink transport channel type	DCH
- Transport channel identity	4
- Logical channel identity	4
- MAC logical channel priority	4
- Logical channel max loss	$\dot{\theta}$
- Number of RLC logical channels	1
- Downlink transport channel type	DCH
- Transport channel identity	1
- Logical channel identity	4
RAB information for setup	
- RAB identity	0000 0001B
	PS domain
	20 seconds
	5
PDCP info	Not Present
- CHOICE Uplink RLC mode	AM RLC
- Transmission RLC discard	
- SDU discard mode	Max DAT retransmissions
	4
Timer_MRW	100
- MaxMRW	4
- Transmission window size	8
- Timer_RST	500
	4
- Timer_poll_prohibit	200
	200
- Poll SDU	4

- Last transmission PU poll	TRUE
- Last retransmission PU poll	TRUE
- Poll Windows	99
	AM RLC
- In-sequence delivery	TRUE
- Receiving window size	8
- Downlink RLC status info	
- Timer_status_prohibit	200
Timer EPC	200
- Missing PU indicator	TRUE
RB mapping info	TRUE
- Information for each multiplexing option	
- Number of RLC logical channels	4
- Uplink transport channel type	DCH
- Transport channel identity	
	2
Logical channel identity	1
- MAC logical channel priority	1
- Logical channel max loss	0
- Number of RLC logical channels	1
- Downlink transport channel type	DCH
- Transport channel identity	2
- Logical channel identity	1
RB information to release	Net Present
RB information to reconfigure	Not Present
RB information to be affected	Not Present
UL Transport channel information for all transport	
channels	
- TFC subset	(This IE is repeated for TFC number.)
	0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108
	clause 6.10 Parameter Set.)
	Not Present
	FDD
UL DCH TFCS	(This IE is repeated for TFC number.)
- TFCI Field 1 information	
- CHOICE CTFC representation	Addition
- TFCS addition information	
	Number of bits used must be enough to cover all
0110102 011 0 0120	combinations of CTFC from clause 6.10.
- CTFC information	Refer to TS34.108 clause 6.10 Parameter Set
- Power offset information	Rolor to 100 1.100 diadoo 0.101 diamotor oot
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor •c	0
- Gain factor •d	Alat Draggert
- Reference TFC ID	Not Present
- Power offset Pp-m	OdB
Deleted UL TrCH information	Not Present
Added or Reconfigured UL TrCH information	
- Transport channel identity	1
TFS	
- Dynamic Transport format information	(This IE is repeated for TFI number)
- Number of Transport blocks	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
- Semi-static Transport Format information	
	Reference to TS34.108 clause 6.10 Parameter Set
- Type of channel coding	Reference to TS34.108 clause 6.10 Parameter Set
- Coding Rate	Reference to TS34.108 clause 6.10 Parameter Set
- Rate matching attribute	Reference to TS34.108 clause 6.10 Parameter Set
- CRC size	Reference to TS34.108 clause 6.10 Parameter Set
Added or Reconfigured UL TrCH information	
- Transport channel identity	2
- TFS	
- Dynamic Transport format information	(This IE is repeated for TFI number)
- Number of Transport blocks	Reference to TS34.108 clause 6.10 Parameter Set
- RLC size	Reference to TS34.108 clause 6.10 Parameter Set
Semi-static Transport Format information	1 totolonoo to 100 i. 100 olaabo o. 10 i alamotol oot
- Transmission time interval	Reference to TS34.108 clause 6.10 Parameter Set
	THOROTORIO TO TOO TO TO CONTROL TO THE TOTAL THE TOTAL TO AL TO THE TO

- Type of channel coding	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
- Rate matching attribute	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
DRAC static information	Not Present
DL Transport channel information common for all	
transport channel	
- SCCPCH TFCS	Not Present
- CHOICE DL parameters	Independent
- DL DCH TFCS	(This IE is repeated for TFC number.)
- TFCI Field 1 information	A dalision
- CHOICE CTFC representation	Addition
- TFCS addition information	
OLIGIOE OTEO O:	
	Number of bits used must be enough to cover all
CTEC information	combinations of CTFC from clause 6.10
- CTFC information - Power offset information	Refer to TS34.108 clause 6.10 Parameter Set
	Not Present
Deleted DL TrCH information	Not Present
Added or Reconfigured DL TrCH information	4
- Transport channel identity	•
——- CHOICE DL parameters	Independent
——————————————————————————————————————	(This IE is repeated for TFI number)
——————————————————————————————————————	Reference to TS34.108 clause 6.10 Parameter Set
RLC size	Reference to TS34.108 clause 6.10 Parameter Set
- Semi-static Transport Format information	Treference to 1901. 100 clause 0. 10 1 arameter get
- Transmission time interval	Reference to TS34.108 clause 6.10 Parameter Set
- Type of channel coding	Reference to TS34.108 clause 6.10 Parameter Set
- Coding Rate	Reference to TS34.108 clause 6.10 Parameter Set
- Rate matching attribute	Reference to TS34.108 clause 6.10 Parameter Set
- CRC size	Reference to TS34.108 clause 6.10 Parameter Set
— - DCH quality target	1.5.5151165 to 1.55 1.165 diadeo 6.16 1 diameter oot
- BLER Quality value	0.00
- Transparent mode signalling info	Not Present
Added or Reconfigured DL TrCH information	
- Transport channel identity	2
	Independent
——- TFS	
- Dynamic Transport format information	(This IE is repeated for TFI number)
	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
- Semi-static Transport Format information	
	Reference to TS34.108 clause 6.10 Parameter Set
- Type of channel coding	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
- Rate matching attribute	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
- DCH quality target	
-BLER Quality value	0.00
- Transparent mode signalling info	Not Present
Frequency info	
UARFCN uplink(Nu)	Reference to TS34.108 clause 6.10 Parameter Set
- UARFCN downlink(Nd)	Reference to TS34.108 clause 6.10 Parameter Set
Maximum allowed UL TX power	33dBm
Uplink DPCH info	
- Uplink DPCH power control info	O.ID
	- 6dB
- PC Preamble	8slot
- Power Control Algorithm	Algorithm1
- TPC step size	1 dB
- Scrambling code type	Long 0 (0 to 16777215)
- Scrambling code number	0 (0 to 16777215)
- Number of DPDCH	Not Present(1) SE is reference to TS34.108 clause 6.10 Parameter Set
	TRUE
	Not Present(0)
- Number of FDI DIL	1401 F1858HR(U)

Puncturing Limit Reference to TS34.108 clause 6.10 Parameter Set **Downlink PDSCH information** Not Present **CPCH SET info** Not Present Downlink information for each radio links Primary CPICH info - Primary scrambling code 100 Not Present - PDSCH with SHO DCH info PDSCH code mapping Not Present Downlink DPCH info for each RL Primary CPICH usage for channel estimation Primary CPICH may be used DPCH frame offset 0 chips Secondary CPICH info Not Present - DL channelisation code - Secondary scrambling code Spreading factor Reference to TS34.108 clause 6.10 Parameter Set Code number SF-1(SF is reference to TS34.108 clause 6.10 Parameter Set) Scrambling code change No change TPC combination index 0 SSDT Cell Identity -2 Closed loop timing adjustment mode Not Present Secondary CCPCH info Not Present **TFCS** Not Present **FACH/PCH information** Not Present References to system information blocks Not Present Scheduling information

Contents of RRC CONNECTION RE-ESTABLISHMENT message: UM (Packet to CELL_DCH from CELL_FACH in PS)

Information Element	Value/remark
Message Type	
U-RNTI	Not Present
Integrity check info	The presence of this IE is dependent on IXIT statements in
	TS 34.123-2. If integrity protection is indicated to be active,
	this IE is present with the values of the sub IEs as stated
	below. Else, this IE and the sub-IEs are omitted.
- message authentication code	SS calculates the value of MAC-I for this message and
· ·	writes to this IE.
- RRC message sequence number	SS provides the value of this IE, from its internal counter.
Integrity protection mode info	Not Present
Ciphering mode info	The presence of this IE is dependent on IXIT statements in
	TS 34.123-2. If ciphering is indicated to be active, this IE
	present with the values of the sub IEs as stated below.
	Else, this IE is omitted.
- Ciphering mode command	start
——- Ciphering algorithm	Use one of the supported ciphering algorithms
- Activation time for DPCH	Not Present
 Radio bearer downlink ciphering activation time 	
info	
- Radio bearer identity	0
	Current RLC SN + 2
	4
- RLC sequence number	Current RLC SN+2
	2
- RLC sequence number	Current RLC SN+2
- Radio bearer identity	3
	Current RLC SN+2
	4
	Current RLC SN+2
Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
New U-RNTI	Not Present
New C-RNTI	Not Present
DRX indicator	noDRX
UTRAN DRX cycle length coefficient	Not Present
RLC reset indicator (for C-plane)	FALSE
RLC reset indicator (for U-plane)	FALSE
CN information info	Not Present
Signalling RB information to setup	(UM DCCH for RRC)
	1
- CHOICE RLC info type	
- RLC info	LIMBLO
	UM RLC
- Transmission RLC discard	M DAT (· ·
	Max DAT retransmissions
	4
Timer_MRW	100 4
——————————————————————————————————————	4
	UM RLC
——————————————————————————————————————	
- Information for each multiplexing option - Number of RLC logical channels	1
- Number of RLC logical channels	1 DCH
	DGR 4
	+ 1
- MAC logical channel priority	1 4
- Logical channel max loss	0
- Number of RLC logical channels	1 1
- Normber of KEC logical charmers - Downlink transport channel type	DCH
- Transport channel identity	1 4
	1 4
Signalling RB information to setup	(AM DCCH for RRC)
Signaling RB information to setup	(AW) DECT 101 RRS)
——————————————————————————————————————	-
TREO IIIIO	

1	
	AM RLC
- Transmission RLC discard	M DAT (: :
- SDU discard mode	Max DAT retransmissions
	4 100
	100 4
- Transmission window size	8
- Timer RST	500
	4
- Polling info	4
Timer_poll_prohibit	200
- Timer_poll	200
	1
- Last transmission PU poll	TRUE
- Last retransmission PU poll	TRUE
	99
- CHOICE Downlink RLC mode	AM RLC
- In-sequence delivery	TRUE
- Receiving window size	8
- Downlink RLC status info	
- Timer_status_prohibit	200
- Timer EPC	200
- Missing PU indicator	TRUE
- RB mapping info	
- Information for each multiplexing option	
- Number of RLC logical channels	4
- Uplink transport channel type	DCH
- Transport channel identity	4
- Logical channel identity	2
- MAC logical channel priority	2
- Logical channel max loss	0
- Number of RLC logical channels	4
- Downlink transport channel type	DCH
- Transport channel identity	4
- Logical channel identity	2
Signalling RB information to setup	(AM DCCH for NAS_DT High priority)
- RB identity	3
	AM RLC
- SDU discard mode	Max DAT retransmissions
	4
	100
	4
- Transmission window size	8
- Timer_RST	500
	4
	000
- Timer_poll_prohibit	200
- Timer_poll	200
	4
- Last transmission PU poll	TRUE
- Last retransmission PU poll	TRUE
- Poll_Windows	99 AM BL C
- CHOICE Downlink RLC mode	AM RLC
- In-sequence delivery	TRUE
- Receiving window size	8
	200
- Timer_status_prohibit	200
- Timer_EPC Missing BL indicator	200 TBUE
- Missing PU indicator	TRUE
Number of PLC legical changes	1
- Number of RLC logical channels	1 DCH
- Uplink transport channel type	
- Transport channel identity	4 3
- Logical channel identity	3

MAC la via al alcanda la via vita.	
- MAC logical channel priority	3
- Logical channel max loss	0
- Number of RLC logical channels	1
Downlink transport channel type Transport channel identity	DCH 1
	+ 3
Signalling RB information to setup	(AM DCCH for NAS_DT Low priority)
- RB identity	4
——————————————————————————————————————	*
- RLC info	
	AM RLC
- Transmission RLC discard	AWI NEO
- SDU discard mode	Max DAT retransmissions
- MAX_DAT	4
- Timer_MRW	100
	4
- Transmission window size	g
- Timer_RST	500
	4
- Timer_poll_prohibit	200
Timer_poll	200
	4
- Last transmission PU poll	TRUE
- Last retransmission PU poll	TRUE
	99
- CHOICE Downlink RLC mode	AM RLC
- In-sequence delivery	TRUE
- Receiving window size	8
- Downlink RLC status info	♥
- Timer_status_prohibit	200
Timer_EPG	200
Missing PU indicator	TRUE
	TXOE
- Information for each multiplexing option	
- Number of RLC logical channels	4
	DCH
- Transport channel identity	1
	4
- MAC logical channel priority	4
	$\frac{1}{\theta}$
- Number of RLC logical channels	4
n er er er er	DCH
- Downlink transport channel type - Transport channel identity	1
Logical channel identity	4
RAB information for setup	7
——- RAB info	
	0000 0001B
——————————————————————————————————————	PS domain
Re-establishment timer	ro uumam
——————————————————————————————————————	20 seconds
	20 seconds
	_
- RB identity	5 Not Present
——————————————————————————————————————	NOT Present
	AM RLC
	AW RLU
- Transmission RLC discard	May DAT retronomissions
	Max DAT retransmissions
	4
	100 4
- MaxMRW	4
- Transmission window size	8
- Timer_RST	500
	4
	000
	200
	200
	4

- Last transmission PU poll	TRUE
- Last retransmission PU poll	TRUE
- Poll Windows	99
	AMR RLC
- In-sequence delivery	TRUE
- Receiving window size	8
- Downlink RLC status info	
- Timer_status_prohibit	200
- Timer EPC	200
- Missing PU indicator	TRUE
	1 NOE
- Information for each multiplexing option	
	4
- Number of RLC logical channels	1
- Uplink transport channel type	DCH
- Transport channel identity	$\frac{2}{\cdot}$
	1
- MAC logical channel priority	1
- Logical channel max loss	θ
	1
- Downlink transport channel type	DCH
- Transport channel identity	2
- Logical channel identity	4
RB information to release	Not Present
RB information to reconfigure	Not Present
RB information to be affected	Not Present
UL Transport channel information for all transport	THOU THOUSING
channels	
- TFC subset	(This IE is repeated for TFC number.)
- Allowed Transport Format combination	0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108
- Allowed Hansport Format combination	clause 6.10 Parameter Set.)
DDACH TECC	
- PRACH TECS	Not Present
	FDD
— - UL DCH TFCS	(This IE is repeated for TFC number.)
- TFCI Field 1 information	
- CHOICE CTFC representation	Addition
- TFCS addition information	
	Number of bits used must be enough to cover all
	combinations of CTFC from clause 6.10.
- CTFC information	Refer to TS34.108 clause 6.10 Parameter Set
- Power offset information	
- CHOICE Gain Factors	Signalled Gain Factor
- Gain factor •c	θ
- Gain factor •d	$\frac{1}{\theta}$
- Reference TFC ID	Not Present
- Power offset Pp-m	OdB
Deleted UL TrCH information	Not Present
Added or Reconfigured UL TrCH information	THOSE TOUBLINE
Added of Reconligured OL Tron Information	4
ll ·	+
——————————————————————————————————————	/This IC is reported for TCI normalism?
- Dynamic Transport format information	(This IE is repeated for TFI number)
- Number of Transport blocks	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
- Semi-static Transport Format information	
	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
- Coding Rate	Reference to TS34.108 clause 6.10 Parameter Set
- Rate matching attribute	Reference to TS34.108 clause 6.10 Parameter Set
- CRC size	Reference to TS34.108 clause 6.10 Parameter Set
Added or Reconfigured UL TrCH information	
- Transport channel identity	2
- TES	_
- Dynamic Transport format information	(This IE is repeated for TFI number)
- Number of Transport blocks	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
Semi-static Transport Format information	Tronoronoo to 100 i. 100 biadoo 0. 10 if didilibibi 35t
- Semi-static transport Format information - Transmission time interval	Reference to TS34.108 clause 6.10 Parameter Set
 i ranomiooiun iimo mildi vai	Traigrands to 1334.100 diauss 0.10 Falamster 381

	Reference to TS34.108 clause 6.10 Parameter Set
- Coding Rate	Reference to TS34.108 clause 6.10 Parameter Set
- Rate matching attribute	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
DRAC static information	Not Present
DL Transport channel information common for all	
transport channel	
- SCCPCH TFCS	Not Present
	Independent
— - DL DCH TFCS	(This IE is repeated for TFC number.)
- TFCI Field 1 information	
	Addition
- TFCS addition information	
	Number of bits used must be enough to cover all
GHOIGE OTT O GIZE	combinations of CTFC from clause 6.10.
	Refer to TS34.108 clause 6.10 Parameter Set
- Power offset information	Not Present
Deleted DL TrCH information	Not Present
Added or Reconfigured DL TrCH information	11001-1000110
- Transport channel identity	4
CHOICE DL parameters	Independent
TES	ппасропион:
- Dynamic Transport format information	(This IE is repeated for TFI number)
- Dynamic Transport format information - Number of Transport blocks	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
- Semi-static Transport Format information	Reference to 1331.100 Glause 0.10 Fatameter 39t
- Semi-static Hansport Format Information - Transmission time interval	Deference to TC24 100 elevine 6 10 December Cet
- Type of channel coding	Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
- Rate matching attribute	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
- DCH quality target	0.00
- BLER Quality value	0.00
- Transparent mode signalling info	Not Present
Added or Reconfigured DL TrCH information	2
- Transport channel identity	
	Independent
TFS	(This IC is reported for TCI number)
- Dynamic Transport format information	(This IE is repeated for TFI number)
- Number of Transport blocks	Reference to TS34.108 clause 6.10 Parameter Set
- RLC size	Reference to TS34.108 clause 6.10 Parameter Set
- Semi-static Transport Format information - Transmission time interval	Deference to TCO4 400 elevine C40 Deverantes Cet
Transferring interval	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
- Rate matching attribute	Reference to TS34.108 clause 6.10 Parameter Set
	Reference to TS34.108 clause 6.10 Parameter Set
- DCH quality target	0.00
- BLER Quality value	0.00
- Transparent mode signalling info	Not Present
Frequency info	D (T004 (00)
UARFCN uplink(Nu)	Reference to TS34.108 clause 6.10 Parameter Set
- UARECN downlink(Nd)	Reference to TS34.108 clause 6.10 Parameter Set
Maximum allowed UL TX power	33dBm
Uplink DPCH info	
- Uplink DPCH power control info	0.15
——————————————————————————————————————	-6dB
	8slot
- Power Control Algorithm	Algorithm1
	1dB
Scrambling code type	Long
	0 (0 to 16777215)
Number of DPDCH	Not Present(1)
	SF is reference to TS34.108 clause 6.10 Parameter Set
II TEOL : (
——————————————————————————————————————	TRUE Not Present(0)

	Reference to TS34.108 clause 6.10 Parameter Set
Downlink PDSCH information	Not Present
CPCH SET info	Not Present
Downlink information for each radio links	
- Primary CPICH info	
- Primary scrambling code	100
- PDSCH with SHO DCH info	Not Present
- PDSCH code mapping	Not Present
- Downlink DPCH info for each RL	
- Primary CPICH usage for channel estimation	Primary CPICH may be used
	0 chips
- Secondary CPICH info	Not Present
- Secondary scrambling code	1
- Spreading factor	Reference to TS34.108 clause 6.10 Parameter Set
	SF-1(SF is reference to TS34.108 clause 6.10 Parameter
	Set)
	No change
	0
	-a
	Not Present
	Not Present
——- TFCS	Not Present
——- FACH/PCH information	Not Present
- References to system information blocks	Not Present

Contents of RRC CONNECTION RE-ESTABLISHMENT COMPLETE message: AM

Information Element	Value/remark
Message Type	
Integrity check info	The presence of this IE is dependent on IXIT statements
	in TS 34.123-2. If integrity protection is indicated to be
	active, this IE shall be present with the values of the sub
	IEs as stated below. Else, this IE and the sub-IEs shall
	be absent.
- Message authentication code	This IE is checked to see if it is present. The value is
	compared against the XMAC-I value computed by SS.
RRC Message sequence number	This IE is checked to see if it is present. The value is
	used by SS to compute the XMAC-I value.
Uplink integrity protection activation info	Not checked
START	Not checked
Radio bearer uplink ciphering activation time info	If ciphering is not activated in RRC CONNECTION RE-
	ESTABLISHMENT message, this IE must be absent.
	Else, SS checks this IE for the presence of activation
	times of all ciphered uplink RLC-UM and RLC-AM RBs
RB with PDCP information list	Not checked
- RB with PDCP information	

Contents of RRC CONNECTION RE-ESTABLISHMENT REQUEST message: TM

Information Element	Value/remark
Message Type	
U-RNTI	Value is checked to see if it is equal to the previously
	assigned U-RNTI
Integrity check info	The presence of this IE is dependent on IXIT statements
	in TS 34.123-2. If integrity protection is indicated to be
	active, this IE shall be present with the values of the sub
	IEs as stated below. Else, this IE and the sub-IEs shall
	be absent.
Message authentication code	This IE is checked to see if it is present. The value is
	compared against the XMAC-I value computed by SS.
RRC Message sequence number	This IE is checked to see if it is present. The value is
	used by SS to compute the XMAC-I value.
START list	Checked to see if the 'CN domain identity' and 'START'
	IEs are present for all CN domains supported by the UE
	Checked to see if it is one of the supported CN domains
- START	Checked to see if present
AM_RLC error indication (for C-plane)	Not checked
AM_RLC error indication (for U-plane)	Not checked
Protocol error indicator	Value is checked to see if it is set to FALSE
Measured results on RACH	Not checked
Protocol error information	Should be absent as "Protocol error indicator" = FALSE

Contents of RRC CONNECTION REJECT message: UM

Information Element	Value/remark
Message Type	
Initial UE identity	Set to the UE's IMSI (GSM-MAP) or TMSI.
RRC transaction identifier	Arbitrarily selects an integer between 0 and 3
Rejection cause	Unspecified
Wait Time	0
Redirection info	Not Present

Information Element	Value/remark
Message Type	
RRC transaction identifier	Arbitrarily select a integer between 0 and 3
Initial UE identity	Reference to TS34.108 clause 6.10 Parameter Set
RRC transaction identifier	Arbitrarily select a integer between 0 and 3
Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
New U-RNTI	
- SRNC identity	0000 0000 0001B
- S-RNTI	0000 0000 0000 0000 0001B
New C-RNTI	0000 0000 0000 0001B
RRC state indicator	CELL_FACH
UTRAN DRX cycle length coefficient	5 (2 to 12)
Capability update requirement	
 UE radio access capability update requirement 	FALSE
 System specific capability update requirement 	Not Present
Signalling RB information to setup	(UM DCCH for RRC)
- RB identity	1
- CHOICE RLC info type	RLC info
RLC info	
- CHOICE Uplink RLC mode	UM RLC
- Transmission RLC discard	
- SDU discard mode	Max DAT retransmissions
- MAX_DAT	4
- Timer_MRW	100
- MaxMRW	4
- CHOICE Downlink RLC mode	UM RLC
- RB mapping info	
- Information for each multiplexing option	N · P
- RLC logical channel mapping indicator	Not Present
- Number of <u>uplink</u> RLC logical channels	1
- Uplink transport channel type	RACH
- Logical channel identity	1
- CHOICE RLC size list	All 2
- MAC logical channel priority	2 0
- Logical channel max loss - Downlink RLC logical channel info	₩
- Number of downlink RLC logical channels	1
- Downlink transport channel type	FACH
- Logical channel identity	1
Signalling RB information to setup	(AM DCCH for RRC)
- RB identity	2
- CHOICE RLC info type	RLC info
- RLC info	INCO IIIIO
- CHOICE Uplink RLC mode	AM RLC
- Transmission RLC discard	
- SDU discard mode	Max DAT retransmissions
- MAX_DAT	4
- Timer_MRW	100
- MaxMRW	4
- Transmission window size	8
- Timer_RST	500
- Max_RST	4
- Polling info	
- Timer_poll_prohibit	200
- Timer_poll	200
- Poll_SDU	1
- Last transmission PU poll	TRUE
- Last retransmission PU poll	TRUE
- Poll_Windows	99
- CHOICE Downlink RLC mode	AM RLC
- In-sequence delivery	TRUE
- Receiving window size	8
- Downlink RLC status info	
- Timer_status_prohibit	200
- Timer_EPC	200
- Missing PU indicator	TRUE

- RB mapping info	
 Information for each multiplexing option 	
- RLC logical channel mapping indicator	Not Present
- Number of uplink RLC logical channels	1
- Uplink transport channel type	RACH
- Logical channel identity	2
- CHOICE RLC size list	All
- MAC logical channel priority	$\frac{70}{3}$
- Logical channel max loss	$\frac{3}{9}$
	♥
- Downlink RLC logical channel info	
- Number of <u>downlink</u> RLC logical channels	1
- Downlink transport channel type	FACH
- Logical channel identity	2
Signalling RB information to setup	(AM DCCH for NAS_DT High priority)
- RB identity	3
- CHOICE RLC info type	RLC info
- CHOICE Uplink RLC mode	AM RLC
- Transmission RLC discard	7.111.11.20
- SDU discard mode	Max DAT retransmissions
	Wax DAT Tetransmissions
- MAX_DAT	'
- Timer_MRW	100
- MaxMRW	4
- Transmission window size	8
- Timer_RST	500
- Max_RST	4
- Polling info	
- Timer_poll_prohibit	200
- Timer_poll	200
- Poll_SDU	1
- Last transmission PU poll	TRUE
- Last retransmission PU poll	TRUE
- Poll_Windows	99
- CHOICE Downlink RLC mode	AM RLC
	TRUE
- In-sequence delivery	
- Receiving window size	8
- Downlink RLC status info	
- Timer_status_prohibit	200
- Timer_EPC	200
- Missing PU indicator	TRUE
- RB mapping info	
 Information for each multiplexing option 	
- RLC logical channel mapping indicator	Not Present
- Number of <u>uplink</u> RLC logical channels	1
- Uplink transport channel type	RACH
- Logical channel identity	3
- CHOICE RLC size list	All
- MAC logical channel priority	4
Logical channel max loss	$\frac{1}{\theta}$
- Downlink RLC logical channel info	
- Number of <u>downlink</u> RLC logical channels	1
	FACH
- Downlink transport channel type	
- Logical channel identity	(AMA DOCUL for NAC DT Low priority)
Signalling RB information to setup	(AM DCCH for NAS_DT Low priority)
- RB identity	4
- CHOICE RLC info type	RLC info
RLC info	
- CHOICE Uplink RLC mode	AM RLC
- Transmission RLC discard	
- SDU discard mode	Max DAT retransmissions
- MAX_DAT	4
- Timer_MRW	100
- MaxMRW	4
- Transmission window size	8
- Timer_RST	500
- Max_RST	4
- Polling info	
- Timer_poll_prohibit	200
Timor_poii_profiloit	

- Timer_poll	200
- Poll_SDU	1
- Last transmission PU poll	TRUE
- Last retransmission PU poll	TRUE
- Poll_Windows	99
- CHOICE Downlink RLC mode	AM RLC
- In-sequence delivery	TRUE
- Receiving window size	8
- Downlink RLC status info	
- Timer_status_prohibit	200
- Timer_EPC	200
- Missing PU indicator	TRUE
- RB mapping info	
- Information for each multiplexing option	
- RLC logical channel mapping indicator	Not Present
- Number of uplink RLC logical channels	1
	RACH
- Uplink transport channel type	
- Logical channel identity	4
- CHOICE RLC size list	All
- MAC logical channel priority	5
	θ
- Downlink RLC logical channel info	
- Number of downlink RLC logical channels	1
- Downlink transport channel type	FACH
- Logical channel identity	4
Signalling RB information to setup	(TM CCCH for RRC)
RB identity	0 0 0 0 1 10 1 1 1 1 1 1 1 1 1 1 1 1 1
- CHOICE RLC info type	♥
	TMDLO
- CHOICE Uplink RLC mode	TM RLC
- Transmission RLC Discard	Not Present
	TRUE
CHOICE Downlink RLC mode	Not Present
- Segmentation indication	TRUE
- RB mapping info	
- Information for each multiplexing option	
- Number of RLC logical channels	1
- Uplink transport channel type	RACH
- Logical channel identity	5
MAC logical channel priority	1
- Logical channel max loss	(IM CCCH for BBC)
Signalling RB information to setup	(UM CCCH for RRC)
- RB identity	 0
——————————————————————————————————————	
- RLC info	
	UM RLC
- Transmission RLC discard	
- SDU discard mode	Max DAT retransmissions
——————————————————————————————————————	4
- Timer MRW	100
	4
- RB mapping info	'
- RB mapping into - Information for each multiplexing option	
- Number of RLC logical channels	
Downlink transport channel to the	4 FACH
- Downlink transport channel type	
- Logical channel identity	(TM B0011 (BB0)
Signalling RB information to setup	(TM BCCH for RRC)
- RB identity	6
- CHOICE Uplink RLC mode	Not Present
- CHOICE Downlink RLC mode	TM RLC
- Segmentation indication	TRUE
- RB mapping info	
- Information for each multiplexing option	
- Number of RLC logical channels	4
- Downlink transport channel type	FACH
- Logical channel identity	
	5

(TM PCCH for RRC) Signalling RB information to setup RB identity CHOICE RLC info type **RLC** info CHOICE Uplink RLC mode Not Present TM RLC **CHOICE Downlink RLC mode** - Segmentation Indication TRUE RB mapping info Information for each multiplexing option Number of RLC logical channels Downlink transport channel type **FACH** Logical channel identity UL Transport channel information for all transport channels (This IE is repeated for TFC number.) - Allowed Transport Format combination 0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108 clause 6.10 Parameter Set.) Not Present - PRACH TFCS - CHOICE mode **FDD** Not Present - UL DCH TFCS Added or Reconfigured UL TrCH information - Transport channel identity 151 - CHOICE Transport channel type Common transport channels - Dynamic Transport format information (This IE is repeated for TFI number) - RLC Size Reference to TS34.108 clause 6.10 Parameter Set - Number of TBs and TTI List (This IE is repeated for TFI number.) - Number of Transport blocks Reference to TS34.108 clause 6.10 Parameter Set RLC size Reference to TS34.108 clause 6.10 Parameter Set - CHOICE mode FDD - CHOICE Logical Channel List ALL - Semi-static Transport Format information - Transmission time interval Reference to TS34.108 clause 6.10 Parameter Set - Type of channel coding Reference to TS34.108 clause 6.10 Parameter Set - Coding Rate Reference to TS34.108 clause 6.10 Parameter Set - Rate matching attribute Reference to TS34.108 clause 6.10 Parameter Set - CRC size Reference to TS34.108 clause 6.10 Parameter Set DL Transport channel information common for all transport channel - SCCPCH TFCS (This IE is repeated for TFC number.) - Normal - TFCI Field 1 information - CHOICE CTFC representation Addition - TFCS addition information - CHOICE CTFC Size Number of bits used must be enough to cover all combinations of CTFC from clause 6.10. - CTFC information Refer to TS34.108 clause 6.10 Parameter Set - Power offset information Not Present - CHOICE DL parameters Independent - DL DCH TFCS Not Present Added or Reconfigured DL TrCH information - Transport channel identity 141 - CHOICE DL parameters Independent - CHOICE Transport channel type Common transport channels - Dynamic Transport format information (This IE is repeated for TFI number) - RLC Size Reference to TS34.108 clause 6.10 Parameter Set - Number of TBs and TTI List (This IE is repeated for TFI number.) Reference to TS34.108 clause 6.10 Parameter Set - Number of Transport blocks - RLC size Reference to TS34.108 clause 6.10 Parameter Set - CHOICE mode **FDD** - CHOICE Logical Channel List ALL - Semi-static Transport Format information - Transmission time interval Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set - Type of channel coding - Coding Rate Reference to TS34.108 clause 6.10 Parameter Set - Rate matching attribute Reference to TS34.108 clause 6.10 Parameter Set

- CRC size Reference to TS34.108 clause 6.10 Parameter Set - DCH quality target - BLER Quality value <u>-6.3</u>0.00 - Transparent mode signalling info Not Present Frequency info - UARFCN uplink(Nu) Reference to TS34.108 clause 6.10 Parameter Set - UARFCN downlink(Nd) Reference to TS34.108 clause 6.10 Parameter Set Maximum allowed UL TX power CHOICE channel requirement 33dBm PRACH info (for RACH)Not Present - CHOICE mode **FDD** Available Signature '0000 0000 1111 1111'B - Available SĔ Reference to TS34.108 clause 6.10 Parameter Set - Scrambling code number Reference to TS34.108 clause 6.10 Parameter Set - Puncturing Limit Available Sub Channel number '1111 1111 1111'B Downlink information common for all radio links Not Present Downlink information for each radio links Not Present

Contents of RRC STATUS message: AM

Information Element	Value/remark
Message Type	
Integrity check info	The presence of this IE is dependent on IXIT statements in TS 34.123-2. If integrity protection is indicated to be active, this IE shall be present with the values of the sub IEs as stated below. Else, this IE and the sub-IEs shall be absent.
- Message authentication code	This IE is checked to see if it is present. The value is compared against the XMAC-I value computed by SS.
- RRC Message sequence number	This IE is checked to see if it is present. The value is used by SS to compute the XMAC-I value.
Identification of received message	Not Present
- Received message type	
- RRC transaction identifier	
Protocol error information	
- Protocol error cause	Value will be checked.

Contents of SECURITY MODE FAILURE message: AM

Information Element	Value/remark
Message Type	
RRC transaction identifier	Checked to see if the value is the identical to the same IE in the downlink SECURITY MODE COMMAND message.
Integrity check info	The presence of this IE is dependent on IXIT statements in TS 34.123-2. If integrity protection is indicated to be active, this IE shall be present with the values of the sub
	IEs as stated below. Else, this IE and the sub-IEs shall be absent.
- Message authentication code	This IE is checked to see if it is present. The value is compared against the XMAC-I value computed by SS.
- RRC Message sequence number	This IE is checked to see if it is present. The value is used by SS to compute the XMAC-I value.
Failure cause	Value will be checked

Contents of TRANSPORT CHANNEL RECONFIGURATION message: AM or UM (The others of speech in CS)

Message Type RRC transaction identifier Integrity check info Integrity protection mode info Ciphering mode info		
Abitiarily selects an integer between 0 and 3 Integrity check info - message authentication code - RRC message sequence number Integrity protection mode info Ciphering mode info Activation time New U-RNTI New C-RNTI Not Present Not P	Information Element	Value/remark
Integrity check info Integrity check info In message authentication code In message authentication code RRC message sequence number integrity protection mode info Ciphering mode info Not Present Cin Info Ciphering mode info Not Present Ciphering mode info Not Pr		
in TS 34.123-2. If integrity protection is indicated to be active, this IE is present with the values of the sub IEs as stated below. Else, this IE and the sub-IEs are omitted. Sc calculates the value of MAC-I for this message and writes to this IE. SC active the value of this IE, from its internal counter. Not Present Not Prese		
- message authentication code - RRC message sequence number Integrity protection mode info Ciphering mode	Integrity check info	
- message authentication code - RRC message sequence number Integrity protection mode info Ciphering mode info Activation time New U-RNTI New C-RNTI Not Present Not Pres		
- message authentication code - RRC message sequence number Integrity protection mode info Ciphering mode info Aditation time New L-RNTI New C-RNTI RRC State DRX-indicator UTRAN DRX cycle length coefficient CN information info UR A identity RW the PDCP info UL Transport channel information or LPC CHOICE CTFC representation - FTCS subset - CHOICE mode - UL DCH TCS - Normal - TFCI Field 1 information - CHOICE CTFC representation - FCW or Gain factor vo - Reference to FC ID - Power offset Pp-m Added or Reconfigured UL TrCH information - Transport channel identity - Transport channel identity - Transport channel identity - Transmission trime interval - Number of Transport blocks - RLC Size - Number of Transport bromat information - Transmission trime interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC Size DRAC static information contained information - CRC Size DRAC static information common for all transport channel of CRC Size DRAC static information common for all transport channel formation common for all transport channel formati		
- RRC message sequence number Integrity protection mode info Ciphering mode info Activation time New U-RNTI New C-RNTI Ne	manage of the action time to	
RRC message sequence number Integrity protection mode info Ciphering mode info Ciphering mode info Activation time New U.RNTI RRC State DRX-indicator UTRAN DRX cycle length coefficient CN information info URA identity RRC State DRX-indicator Not Present Not Pr	- message authentication code	
Integrity protection mode info Ciphering mode info Ciphering mode info Activation time New U-RNTI New C-RNTI Not Present Not Presen	PPC massage acquence number	
Ciphering mode into Activation time New U-RNT RRC State DRX-indicator UTRAN DRX cycle length coefficient Not Present Not Presen		
Activation time New U-RNTI New C-RNTI Not Present Not Pre		
New C-RNTI RRC State DRX-indicator UTRAN DRX cycle length coefficient CN information info URA identity URA identity URA identity TFS subset - Allowed Transport Format combination - PRACH TFCS - CHOICE mode - UL DCH TFCS - OHOICE mode - UL DCH TFCS - Normal - TFCI Field 1 information - CHOICE CTFC representation - TFCS addition information - CHOICE CTFC representation - TFCS addition information - CHOICE GTFC fize - Gain factor **e - CHOICE Transport channel information - RLC Size - Number of Transport blocks - RLC-size - Semi-static Transport format information - Transmission time interval - Number of Transport channel information - Transmission time interval - Number of Transport blocks - RLC-size - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - Rate matching attribute - Rate matching attribute - Rate matching attribute - SCCPCH TFCS - Not Present -		
New C-RNTI UTRAN DRX cycle length coefficient ON information info URA identity RB with PDCP info UL Transport channel information for all transport channels - TFC subset - Allowed Transport Format combination - PRACH TFCS - CHOICE mode - UL DCH TFCS - Normal - TFCI Field 1 information - CHOICE CFTC representation - TFCS addition information - Power offset information - Power offset information - CHOICE Gain Factors - Gain factor •		
RRC State DRX-indicator UTRAN DRX cycle length coefficient CN information info URA identity R8 with PDCP info UL Transport channel information for all transport channels		
UTRAN DRX cycle length coefficient CN information info URA identity RB with PDCP info UL Transport channel information for all transport channels - TFC subset - Allowed Transport Format combination - PRACH TFCS - CHOICE mode - UL DCH TFCS - Normal - TFCI Field 1 information - CHOICE CTFC representation - TFCs addition information - CHOICE CTFC Size - CHOICE Gain Factors - Gain factor • c - Gain factor • c - Gain factor • c - Reference TFC ID - Power offset Pp-m Added or Reconfigured UL TrCH information - Transport channel identity - TFS - CHOICE Transport format information - RLC Size - Number of Transport format information - RLC Size - Number of Transport format information - RLC Size - Semi-static Transport format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size DRAC static information - SCCPCH TFCS Not Present Not Pres		
CRI information info URA identity RB with PDCP info UL Transport channel information for all transport channels - TFC subset - Allowed Transport Format combination - PRACH TFCS - CHOICE mode - UL DCH TFCS - Normal - TFCI Field 1 information - CHOICE CTFC representation - TFCS addition information - CHOICE CTFC Size - CTFC information - Power offset information - Power offset information - CHOICE Gain factor *c - CHOICE Transport channel identity - TFS - CHOICE Transport format information - RLC Size - Number of Transport format information - Power offset Pp-m Added or Reconfigured UL TrCH information - RLC Size - Number of Transport format information - RLC Size - Number of Transport format information - RLC Size - Number of Transport format information - RLC Size - Number of Transport format information - RLC Size - Number of Transport format information - RLC Size - Semi-static Transport format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size DRAC Static information - SCCPCH TFCS Not Present Not Present Not Pr		
RB with PDCP info UL Transport channel information for all transport channels - TFC subset - Allowed Transport Format combination - PRACH TFCS - CHOICE mode - UL DCH TFCS - Normal - TFCI Field 1 information - CHOICE CTFC representation - TFCS addition information - CHOICE CTFC representation - TFCS addition information - CHOICE CTFC Size - CTFC information - Power offset information - Power offset information - CHOICE GTFC ID - Power offset Pp-m Added or Reconfigured UL TrCH information - Transport channel identity - TFS - CHOICE Transport format information - RLC Size - Number of Transport format information - Transport channel identity - Transmission Time Interval - Number of Transport format information - RLC Size - Semi-static Transport format information - Transmission Time Interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size - DRAC static information DL Transport channel - SCCPCH TFCS Not Present		Not Present
RB with PDCP info UL Transport channel information for all transport channels - TFC subset - Allowed Transport Format combination - PRACH TFCS - CHOICE mode - UL DCH TFCS - Normal - TFCI Field 1 information - CHOICE CTFC representation - TFCS addition information - CHOICE CTFC representation - TFCS addition information - CHOICE CTFC Size - CTFC information - Power offset information - Power offset information - CHOICE GTFC ID - Power offset Pp-m Added or Reconfigured UL TrCH information - Transport channel identity - TFS - CHOICE Transport format information - RLC Size - Number of Transport format information - Transport channel identity - Transmission Time Interval - Number of Transport format information - RLC Size - Semi-static Transport format information - Transmission Time Interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size - DRAC static information DL Transport channel - SCCPCH TFCS Not Present	URA identity	Not Present
channels - TFC subset - Allowed Transport Format combination - PRACH TFCS - CHOICE mode - UL DCH TFCS - Normal - TFCI Field 1 information - CHOICE CTFC representation - TFCS addition information - CHOICE CTFC regresentation - TFCS addition information - CHOICE Gain Factors - Gain factor *c - Gain factor *c - Gain factor *c - Gain factor *c - Reference TFC ID - Power offset Pp-m Added or Reconfigured UL TrCH information - Transport channel identity - TFS - CHOICE Transport channel type - Dynamic Transport format information - Transmission Time Interval - Number of Transport sport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size DRAC static information DL Transport channel - SCCPCH TFCS (This IE is repeated for TFC number.) Addition (This IE is repeated for TFC number.) Addition		Not Present
- TFC subset - Allowed Transport Format combination - PRACH TFCS - C-HOICE mode - UL DCH TFCS - Normal - TFCI Field 1 information - CHOICE CTFC representation - TFCS addition information - CHOICE CTFC Size - CTFC information - CHOICE CTFC Size - CTFC information - Power offset information - Power offset information - CHOICE Gain Factors - Gain factor • c - Reference TFC ID - Power offset Pp-m Added or Reconfigured UL TrCH information - Transport channel identity - TFS - CHOICE Transport format information - RLC Size - Number of TBs and TTI List - Transmission Time Interval - Number of Transport blocks - RLC eize - Semi-static Transport format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size DRAC static information DL Transport channel - SCCPCH TFCS (This IE is repeated for TFC number.) Addition (This IE is repeated for TFC number.) Addition Addition Addition Addition Addition Addition Addition Addition Addition Number of bits used must be enough to cover all combinations of CTFC from clause 6.10 Parameter Set Signalled Gain Factors 0 0 0 Not Present Dedicated transport channels (This IE is repeated for TFI number.) Not Present Dedicated transport channels (This IE is repeated for TFI number.) Not Present Reference to TS34.108 clause 6.10 Parameter Set Reference	UL Transport channel information for all transport	
- Allowed Transport Format combination - PRACH TFCS - CHOICE mode - UL DCH TFCS - Normal - TFCI Field 1 information - CHOICE CTFC representation - TFCS addition information - CHOICE CTFC Size - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor • C - Gain factor • C - Gain factor • C - Reference TFC ID - Power offset Pp-m Added or Reconfigured UL TrCH information - Transport channel identity - TFS - CHOICE Transport channel type - Dynamic Transport format information - RLC Size - Number of TBs and TTI List - Transmission Time Interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size DRAC static information DL Transport channel information common for all transport channel - SCCPCH TFCS O MaxTFCValue-1 (MaxTFCValue is refer to TS34.108 clause 6.10 Parameter Set.) Not Present FDD (This IE is repeated for TFC number.) Addition		
- PRACH TFCS - CHOICE mode - UL DCH TFCS - Normal - TFCI Field 1 information - CHOICE CTFC representation - TFCS addition information - CHOICE CTFC Size - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor • c - CHOICE Transport channel identity - TFS - CHOICE Transport channel type - Dynamic Transport format information - RLC Size - Number of TBs and TTI List - Transmission Time Interval - Number of Transport blocks - RLC-size - Semi-static Transport format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size - DRAC static information DL Transport channel information common for all transport channel - SCCPCH TFCS Clause 6.10 Parameter Set.) Not Present		
- PRACH TFCS - CHOICE mode - UL DCH TFCS - Normal - TFCI Field 1 information - CHOICE CTFC representation - TFCS addition information - CHOICE CTFC Size - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor •c - Gain factor •c - Gain factor •c - Power offset Pp-m Added or Reconfigured UL TrCH information - Transport channel identity - TFS - CHOICE Transport channel type - Dynamic Transport format information - RLC Size - Number of TBs and TTI List - Transmission time interval - Number of Transport blocks - RLC size - Semi-static Transport Format information - Transmission time interval - Coding Rate - Rate matching attribute - CRC size - DRAC static information DL Transport channel information common for all transport channel - SCCPCH TFCS Not Present Addition	 Allowed Transport Format combination 	
- CHOICE mode - UL DCH TFCS - Normal - TFCI Field 1 information - CHOICE CTFC representation - TFCS addition information - CHOICE CTFC Size - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor •c - Gain factor •c - Gain factor •d - Reference TFC ID - Power offset Pp-m Added or Reconfigured UL TrCH information - Transport channel identity - TFS - CHOICE Transport format information - RLC Size - Number of TBs and TTI List - Transmission Time Interval - Number of Transport blocks - RLC size - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC Size DRAC static information DL Transport channel information common for all transport channel - SCCPCH TFCS (This IE is repeated for TFC number.) Addition		· ·
- UL DCH TFCS - Normal - TFCI Field 1 information - CHOICE CTFC representation - TFCS addition information - CHOICE CTFC Size - CHOICE CTFC Size - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor *c - Gain factor *c - Gain factor *d - Reference TFC ID - Power offset Pp-m Added or Reconfigured UL TrCH information - Transport channel identity - TFS - CHOICE Transport channel type - Dynamic Transport format information - RLC Size - Number of TBs and TTI List - Transmission Time Interval - Number of Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC Size DRAC static information DL Transport channel - SCCPCH TFCS (This IE is repeated for TFC number.) Addition		
- Normal - TFCI Field 1 information - CHOICE CTFC representation - TFCS addition information - CHOICE CTFC Size - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor • ○ - Gain factor • ○ - Gain factor • ○ - Reference TFC ID - Power offset Pp-m Added or Reconfigured UL TrCH information - Transport channel identity - TFS - CHOICE Transport channel type - Dynamic Transport format information - RLC Size - Number of TBs and TTI List - Transmission Time Interval - Number of Transport blocks - RLC size - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size DRAC static information DL Transport channel - SCCPCH TFCS - SCCPCH TFCS Addition		1 ·
- TFCI Field 1 information - CHOICE CTFC representation - TFCS addition information - CHOICE CTFC Size - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor •c - Gain factor •c - Gain factor •d - Reference TFC ID - Power offset Pp-m Added or Reconfigured UL TrCH information - Transport channel identity - TFS - CHOICE Transport format information - RLC Size - Number of TBs and TTI List - Transmission Time Interval - Number of Transport blocks - RLC size - Semi-static Transport Format information - Transmission time interval - Coding Rate - Rate matching attribute - CRC size DRAC static information DL Transport channel - SCCPCH TFCS Addition		(This IE is repeated for TFC number.)
- CHOICE CTFC representation - TFCS addition information - CHOICE CTFC Size - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor •c - Gain factor •c - Gain factor •d - Reference TFC ID - Power offset Pp-m Added or Reconfigured UL TrCH information - Transport channel identity - TFS - CHOICE Transport channel type - Dynamic Transport format information - RLC Size - Number of TBs and TTI List - Transmission Time Interval - Number of Transport blocks - RLC size - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size DRAC static information DL Transport channel - SCCPCH TFCS Addition Number of bits used must be enough to cover all combinations OTFC from clause 6.10 Parameter Set Number of bits used must be enough to cover all combinations OTFC from clause 6.10 Parameter Set Signalled Gain Factors 0 Not Present Dedicated transport channels (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clau		
- CHOICE CTFC Size - CTFC information - CHOICE Gain Factors - CHOICE Gain Factors - Gain factor •c - Gain factor •d - Power offset Pp-m Added or Reconfigured UL TrCH information - RLC Size - Number of TBs and TTI List - Transmission Time Interval - Number of Transport blocks - RLC cize - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size DRAC static information DL Transport channel - SCCPCH TFCS Number of this used must be enough to cover all combinations of CTFC from clause 6.10 Refer to TS34.108 clause 6.10 Parameter Set 0 Not Present 0dB Not Present 0dB Sd Sd Lodicated transport channels (This IE is repeated for TF1 number) Reference to TS34.108 clause 6.10 Parameter Set Reference to		Addition
- CHOICE CTFC Size - CTFC information - Power offset information - CHOICE Gain Factors - Gain factor •c - Gain factor •c - Reference TFC ID - Power offset Pp-m Added or Reconfigured UL TrCH information - Transport channel identity - TFS - CHOICE Transport format information - RLC Size - Number of TBs and TTI List - Transmission Time Interval - Number of Transport blocks - RLC size - Semi-static Transport blocks - RLC size - Semi-static Transport format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size DRAC static information DL Transport channel information common for all transport channel - SCCPCH TFCS Number of bits used must be enough to cover all combinations of CTFC from clause 6.10. Refer to TS34.108 clause 6.10 Parameter Set Signalled Gain Factors 0 0 Not Present Dedicated transport channels (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108		7 (dalilot)
combinations of CTFC from clause 6.10. Refer to TS34.108 clause 6.10 Parameter Set CHOICE Gain Factors Cain factor •c Cacholice Transport clause 6.10 Parameter Set CHOICE Transport channel identity CHOICE Transport channel type Cacholice Transport format information Cacholice Transport channel type Cacholice Transport channels Cacho	Tr oo addition information	
combinations of CTFC from clause 6.10. Refer to TS34.108 clause 6.10 Parameter Set CHOICE Gain Factors Cain factor •c Cacholice Transport clause 6.10 Parameter Set CHOICE Transport channel identity CHOICE Transport channel type Cacholice Transport format information Cacholice Transport channel type Cacholice Transport channels Cacho	- CHOICE CTFC Size	Number of bits used must be enough to cover all
- CTFC information - Power offset information - CHOICE Gain Factors - Gain factor •c - Gain factor •c - Gain factor •d - Reference TFC ID - Power offset Pp-m Added or Reconfigured UL TrCH information - Transport channel identity - TFS - CHOICE Transport channel type - Dynamic Transport format information - RLC Size - Number of TBs and TTI List - Transmission Time Interval - Number of Transport blocks - RLC size - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size DRAC static information DL Transport channel information common for all transport channel cannot common for all cannot c		
- CHOICE Gain Factors - Gain factor • C - Gain factor • C - Reference TFC ID - Power offset Pp-m Added or Reconfigured UL TrCH information - Transport channel identity - TFS - CHOICE Transport channel type - Dynamic Transport format information - RLC Size - Number of TBs and TTI List - Transmission Time Interval - Number of Transport blocks - RLC size - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size DRAC static information DL Transport channel - SCCPCH TFCS Signalled Gain Factors 0 Not Present Dedicated transport channels (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10	- CTFC information	Refer to TS34.108 clause 6.10 Parameter Set
- Gain factor •c - Gain factor •d - Reference TFC ID - Power offset Pp-m Added or Reconfigured UL TrCH information - Transport channel identity - TFS - CHOICE Transport channel type - Dynamic Transport format information - RLC Size - Number of TBs and TTI List - Transmission Time Interval - Number of Transport blocks - RLC size - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size DRAC static information DL Transport channel information common for all transport channel - SCCPCH TFCS 0 Not Present Dedicated transport channels (This IE is repeated for TFI number.) Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Pa	 Power offset information 	
- Gain factor •d - Reference TFC ID - Power offset Pp-m Added or Reconfigured UL TrCH information - Transport channel identity - TFS - CHOICE Transport channel type - Dynamic Transport format information - RLC Size - Number of TBs and TTI List - Transmission Time Interval - Number of Transport blocks - RLC size - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size DRAC static information DL Transport channel - SCCPCH TFCS ONot Present Dedicated transport channels (This IE is repeated for TFI number.) Reference to TS34.108 clause 6.10 Parameter Set Ref		Signalled Gain Factors
- Reference TFC ID - Power offset Pp-m Added or Reconfigured UL TrCH information - Transport channel identity - TFS - CHOICE Transport channel type - Dynamic Transport format information - RLC Size - Number of TBs and TTI List - Transmission Time Interval - Number of Transport blocks - RLC size - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size DRAC static information DL Transport channel - SCCPCH TFCS Not Present DdB Not Present Dedicated transport channels (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set (This IE is repeated for TFI number.) Not Present Reference to TS34.108 clause 6.10 Parameter Set		
- Power offset Pp-m Added or Reconfigured UL TrCH information - Transport channel identity - TFS - CHOICE Transport channel type - Dynamic Transport format information - RLC Size - Number of TBs and TTI List - Transmission Time Interval - Number of Transport blocks - RLC size - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size DRAC static information DL Transport channel - SCCPCH TFCS OdB 54 Dedicated transport channels (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter S		
Added or Reconfigured UL TrCH information - Transport channel identity - TFS - CHOICE Transport channel type - Dynamic Transport format information - RLC Size - Number of TBs and TTI List - Transmission Time Interval - Number of Transport blocks - RLC size - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size DRAC static information DL Transport channel - SCCPCH TFCS Dedicated transport channels (This IE is repeated for TFI number.) Reference to TS34.108 clause 6.10 Parameter Set (This IE is repeated for TFI number.) Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 P		
- Transport channel identity - TFS - CHOICE Transport channel type - Dynamic Transport format information - RLC Size - Number of TBs and TTI List - Transmission Time Interval - Number of Transport blocks - RLC size - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size DRAC static information DL Transport channel - SCCPCH TFCS - CHOICE Transport channel type - Dedicated transport channels (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set		nan
- TFS - CHOICE Transport channel type - Dynamic Transport format information - RLC Size - Number of TBs and TTI List - Transmission Time Interval - Number of Transport blocks - RLC size - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size Dedicated transport channels (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set		51
- CHOICE Transport channel type - Dynamic Transport format information - RLC Size - Number of TBs and TTI List - Transmission Time Interval - Number of Transport blocks - RLC size - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size Dedicated transport channels (This IE is repeated for TFI number.) Not Present Reference to TS34.108 clause 6.10 Parameter Set	: 	
- Dynamic Transport format information - RLC Size - Number of TBs and TTI List - Transmission Time Interval - Number of Transport blocks - RLC size - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size DRAC static information DL Transport channel - SCCPCH TFCS Time IE is repeated for TFI number.) Reference to TS34.108 clause 6.10 Parameter Set	: 1	Dedicated transport channels
- RLC Size - Number of TBs and TTI List - Transmission Time Interval - Number of Transport blocks - RLC size - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size DRAC static information DL Transport channel - SCCPCH TFCS Reference to TS34.108 clause 6.10 Parameter Set (This IE is repeated for TFI number.) Not Present Reference to TS34.108 clause 6.10 Parameter Set		
- Number of TBs and TTI List - Transmission Time Interval - Number of Transport blocks - RLC size - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size DRAC static information DL Transport channel - SCCPCH TFCS Tumber of TBs and TTI List - Transmission Time Interval Reference to TS34.108 clause 6.10 Parameter Set		
Number of Transport blocks RLC size - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size DRAC static information DL Transport channel - SCCPCH TFCS Reference to TS34.108 clause 6.10 Parameter Set	- Number of TBs and TTI List	
- RLC size - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size DRAC static information DL Transport channel - SCCPCH TFCS Reference to TS34.108 clause 6.10 Parameter Set	- Transmission Time Interval	Not Present
- Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size DRAC static information DL Transport channel - SCCPCH TFCS Reference to TS34.108 clause 6.10 Parameter Set Not Present		
- Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size DRAC static information DL Transport channel - SCCPCH TFCS Reference to TS34.108 clause 6.10 Parameter Set	1	Reference to TS34.108 clause 6.10 Parameter Set
- Type of channel coding - Coding Rate - Rate matching attribute - CRC size DRAC static information DL Transport channel - SCCPCH TFCS Reference to TS34.108 clause 6.10 Parameter Set Not Present		D (
- Coding Rate - Rate matching attribute - CRC size DRAC static information DL Transport channel information common for all transport channel - SCCPCH TFCS Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Not Present Not Present Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Not Present		
- Rate matching attribute - CRC size DRAC static information DL Transport channel information common for all transport channel - SCCPCH TFCS Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Not Present Not Present Not Present		
- CRC size DRAC static information DL Transport channel information common for all transport channel - SCCPCH TFCS Reference to TS34.108 clause 6.10 Parameter Set Not Present Not Present Not Present		
DRAC static information DL Transport channel information common for all transport channel - SCCPCH TFCS Not Present Not Present		
DL Transport channel information common for all transport channel - SCCPCH TFCS Not Present		
transport channel - SCCPCH TFCS Not Present		
- SCCPCH TFCS Not Present		
		Not Present
- CHOICE DL parameters Independent	- CHOICE DL parameters	Independent
- DL DCH TFCS (This IE is repeated for TFC number.)	- DL DCH TFCS	(This IE is repeated for TFC number.)
- Normal	- Normal	

- TFCI Field 1 information - CHOICE CTFC representation Addition - TFCS addition information - CHOICE CTFC Size Number of bits used must be enough to cover all combinations of CTFC from clause 6.10. - CTFC information Refer to TS34.108 clause 6.10 Parameter Set - Power offset information Not Present Added or Reconfigured DL TrCH information - Transport channel identity SameAsUL - CHOICE DL parameters - UL TrCH Identity 51 - DCH quality target - BLER Quality value -6.30.00 - Transparent mode signalling info Not Present Frequency info - UARFCN uplink(Nu) Reference to TS34.108 clause 6.10 Parameter Set - UARFCN downlink(Nd) Reference to TS34.108 clause 6.10 Parameter Set Maximum allowed UL TX power 33dBm Uplink DPCH info -Uplink DPCH power control info - DPCCH power offset -6dB - PC Preamble 8slot Algorithm1 - Power Control Algorithm - TPC step size 1dB - Scrambling code type Long - Scrambling code number 0 (0 to 16777215) - Number of DPDCH Not Present(1) - spreading factor SF is reference to TS34.108 clause 6.10 Parameter Set - TFCI existence **TRUE** - Number of FBI bit Not Present(0) - Puncturing Limit Reference to TS34.108 clause 6.10 Parameter Set **CHOICE Mode** FDD - Downlink PDSCH information Not Present Downlink information common for all radio links - Downlink DPCH info common for all RL **FDD** CHOICE mode - Timing indicator Maintain - CFN-targetSFN frame offset Not Present - Downlink DPCH power control information - DPC mode 0 (single) - DL rate matching restriction information Not Present - Spreading factor Reference to TS34.108 clause 6.10 Parameter Set - Fixed or Flexible Position Flexible - TFCI existence **TRUE** - Number of bits for Pilot bits(SF=128,256) Not Present - DPCH compressed mode info - TGPSI - TGPS Status Flag inactive - TGCFN (Current CFN + (256 - TTI/10msec)) mod 256 - Transmission gap pattern sequence configuration parameters - TGMP **FDD Measurement** - TGPRC **TGCFN** (Current CFN + (256 - TTI/10msec)) mod 256 - TGSN 8 - TGL1 10 - TGL2 5 - TGD 15 35 - TGPL1 - TGPL2 35 - RPP Mode 1 Mode 1 - ITP - UL/DL Mode DΙ - Downlink compressed mode method SF/2 - Uplink compressed mode method Not Present - Downlink frame type

2.0

- DeltaSIR1

- DeltaSIRafter1 1.0 - DeltaSIR2 Not Present - DeltaSIRafter2 Not Present - TX Diversity mode None - SSDT information Not Present - Default DPCH Offset Value Downlink information for each radio links - Primary CPICH info - Primary scrambling code 100 - PDSCH with SHO DCH info Not Present - PDSCH code mapping Not Present Downlink DPCH info for each RLPrimary CPICH usage for channel estimation Primary CPICH may be used - DPCH frame offset 0 chips - Secondary CPICH info Not Present - DL channelisation code - Secondary scrambling code - Spreading factor Reference to TS34.108 clause 6.10 Parameter Set - Code number SF-1(SF is reference to TS34.108 clause 6.10 Parameter Set) - Scrambling code change - TPC combination index No change 0 - SSDT Cell Identity -a - Closed loop timing adjustment mode - Secondary CCPCH info Not Present Not Present Not Present - TFCS - FACH/PCH information Not Present TFS

- References to system information blocks

Scheduling information

Not Present

Information Element	Value/remark
Message Type	
RRC transaction identifier	Arbitrarily selects an integer between 0 and 3
Integrity check info	The presence of this IE is dependent on IXIT statements
	in TS 34.123-2. If integrity protection is indicated to be
	active, this IE is present with the values of the sub IEs as
	stated below. Else, this IE and the sub-IEs are omitted.
- message authentication code	SS calculates the value of MAC-I for this message and
mossage aumenmeaner cour	writes to this IE.
- RRC message sequence number	SS provides the value of this IE, from its internal counter.
Integrity protection mode info	Not Present
Ciphering mode info	Not Present
Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
New U-RNTI	Not Present "
New C-RNTI	Not Present
RRC State DRX-indicator	NoDRXCELL_DCH
UTRAN DRX cycle length coefficient	Not Present
CN information info	Not Present
URA identity	Not Present
RB with PDCP information	Not Present
UL Transport channel information for all transport	
channels	
- TFC subset	(This IE is repeated for TFC number.)
 Allowed Transport Format combination 	0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108
	clause 6.10 Parameter Set.)
- PRACH TFCS	Not Present
- CHOICE mode	FDD
- UL DCH TFCS	(This IE is repeated for TFC number.)
- Normal	
- TFCI Field 1 information	
- CHOICE CTFC representation	Addition
- TFCS addition information	
0110105 0750 0	
- CHOICE CTFC Size	Number of bits used must be enough to cover all
0750 : (combinations of CTFC from clause 6.10.
- CTFC information	Refer to TS34.108 clause 6.10 Parameter Set
- Power offset information	Cian alla d Onin Frantana
- CHOICE Gain Factors	Signalled Gain Factors 0
- Gain factor •c	0
- Gain factor •d	
- Reference TFC ID	Not Present 0dB
- Power offset Pp-m	OUB
Added or Reconfigured UL TrCH information - Transport channel identity	<u>5</u> 4
- Transport charmer identity - TFS	= '
- CHOICE Transport channel type	Dedicated transport channels
- Dynamic Transport format information	(This IE is repeated for TFI number)
- RLC Size	Reference to TS34.108 clause 6.10 Parameter Set
- Number of TBs and TTI List	(This IE is repeated for TFI number.)
- Transmission Time Interval	Not Present
Number of Transport blocks	Reference to TS34.108 clause 6.10 Parameter Set
- RLC size	Reference to TS34.108 clause 6.10 Parameter Set
- CHOICE Logical Channel list	<u>ALL</u>
- Semi-static Transport Format information	
- Transmission time interval	Reference to TS34.108 clause 6.10 Parameter Set
- Type of channel coding	Reference to TS34.108 clause 6.10 Parameter Set
- Coding Rate	Reference to TS34.108 clause 6.10 Parameter Set
- Rate matching attribute	Reference to TS34.108 clause 6.10 Parameter Set
- CRC size	Reference to TS34.108 clause 6.10 Parameter Set
DRAC static information	Not Present
DL Transport channel information common for all	
transport channel	[-
- SCCPCH TFCS	Not Present
- CHOICE DL parameters	Independent
- DL DCH TFCS	(This IE is repeated for TFC number.)
- Normal	

- TFCI Field 1 information - CHOICE CTFC representation Addition - TFCS addition information - CHOICE CTFC Size Number of bits used must be enough to cover all combinations of CTFC from clause 6.10. Refer to TS34.108 clause 6.10 Parameter Set - CTFC information - Power offset information Not Present Added or Reconfigured DL TrCH information - Transport channel identity - CHOICE DL parameters SameAsUL - UL TrCH Identity 51 - DCH quality target - BLER Quality value -6.30.00 - Transparent mode signalling info Not Present Frequency info - UARFCN uplink(Nu) Reference to TS34.108 clause 6.10 Parameter Set - UARFCN downlink(Nd) Reference to TS34.108 clause 6.10 Parameter Set Maximum allowed UL TX power 33dBm Uplink DPCH info -Uplink DPCH power control info - DPCCH power offset -6dB - PC Preamble 8slot - Power Control Algorithm Algorithm1 - TPC step size 1dB - Scrambling code type Long - Scrambling code number 0 (0 to 16777215) - Number of DPDCH Not Present(1) - spreading factor SF is reference to TS34.108 clause 6.10 Parameter Set - TFCI existence **TRUE** - Number of FBI bit Not Present(0) - Puncturing Limit Reference to TS34.108 clause 6.10 Parameter Set **CHOICE Mode** FDD - Downlink PDSCH information Not Present Downlink information common for all radio links - Downlink DPCH info common for all RL **FDD** CHOICE mode - Timing indicator **Maintain** - CFN-targetSFN frame offset Not Present - Downlink DPCH power control information - DPC mode 0 (single) - DL rate matching restriction information Not Present - Spreading factor Reference to TS34.108 clause 6.10 Parameter Set - Fixed or Flexible Position Flexible - TFCI existence **TRUE** - Number of bits for Pilot bits(SF=128,256) Not Present - DPCH compressed mode info - TGPSI - TGPS Status Flag inactive - TGCFN (Current CFN + (256 - TTI/10msec)) mod 256 - Transmission gap pattern sequence configuration parameters - TGMP **FDD Measurement** - TGPRC **TGCFN** (Current CFN + (256 - TTI/10msec)) mod 256 - TGSN 8 - TGL1 10 - TGL2 5 - TGD 15 - TGPL1 35 - TGPL2 35 - RPP Mode 1 Mode 1 - ITP - UL/DL Mode DΙ - Downlink compressed mode method SF/2 - Uplink compressed mode method Not Present - Downlink frame type

2.0

- DeltaSIR1

- DeltaSIRafter1 1.0 - DeltaSIR2 Not Present - DeltaSIRafter2 Not Present - TX Diversity mode None - SSDT information Not Present - Default DPCH Offset Value Downlink information for each radio links - Primary CPICH info - Primary scrambling code 100 - PDSCH with SHO DCH info Not Present - PDSCH code mapping Not Present Downlink DPCH info for each RLPrimary CPICH usage for channel estimation Primary CPICH may be used - DPCH frame offset 0 chips - Secondary CPICH info Not Present - DL channelisation code - Secondary scrambling code - Spreading factor Reference to TS34.108 clause 6.10 Parameter Set - Code number SF-1(SF is reference to TS34.108 clause 6.10 Parameter Set) - Scrambling code change - TPC combination index No change 0 - SSDT Cell Identity -a - Closed loop timing adjustment mode - Secondary CCPCH info Not Present Not Present Not Present - TFCS - FACH/PCH information Not Present

Not Present

- References to system information blocks

Scheduling information

г	Information Florent	M.L. Johnson
L	Information Element	Value/remark
1	Message Type RRC transaction identifier	Arbitrarily colocts on integer between 0 and 2
	Integrity check info	Arbitrarily selects an integer between 0 and 3 The presence of this IE is dependent on IXIT statements
	integrity check into	in TS 34.123-2. If integrity protection is indicated to be
		active, this IE is present with the values of the sub IEs as
		stated below. Else, this IE and the sub-IEs are omitted.
	- message authentication code	SS calculates the value of MAC-I for this message and
	message admentioation code	writes to this IE.
	- RRC message sequence number	SS provides the value of this IE, from its internal counter.
	Integrity protection mode info	Not Present
	Ciphering mode info	Not Present
	Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
	New U-RNTI	Not Present
	New C-RNTI	Not Present
	RRC State DRX indicator	NoDRXCELL_DCH
•	UTRAN DRX cycle length coefficient	Not Present
	CN information info	Not Present
	URA identity	Not Present
	RB with PDCP information	Not Present
	UL Transport channel information for all transport	
	channels	
	- TFC subset	(This IE is repeated for TFC number.)
	- Allowed Transport Format combination	0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108
	DD A OLL TECC	clause 6.10 Parameter Set.)
	- PRACH TFCS	Not Present FDD
	- CHOICE mode - UL DCH TFCS	(This IE is repeated for TFC number.)
	- Normal	(This IL is repeated for Tr & Humber.)
	- TFCI Field 1 information	
	- CHOICE CTFC representation	Addition
	- TFCS addition information	
	- CHOICE CTFC Size	Number of bits used must be enough to cover all
	0750 : (combinations of CTFC from clause 6.10.
	- CTFC information	Refer to TS34.108 clause 6.10 Parameter Set
	- Power offset information - CHOICE Gain Factors	Signalled Gain Factors
	- Griolog Gain Factors - Gain factor •c	0
	- Gain factor •d	0
	- Reference TFC ID	Not Present
	- Power offset Pp-m	0dB
	Added or Reconfigured UL TrCH information	
	- Transport channel identity	<u>5</u> 4
	- TFS	
l	- CHOICE Transport channel type	Dedicated transport channels
l	- Dynamic Transport format information	(This IE is repeated for TFI number)
	- RLC Size	Reference to TS34.108 clause 6.10 Parameter Set
	- Number of TBs and TTI List	(This IE is repeated for TFI number.) Not Present
	- Transmission Time Interval - Number of Transport blocks	Reference to TS34.108 clause 6.10 Parameter Set
	- Number of Transport blocks	Reference to TS34.108 clause 6.10 Parameter Set
	- CHOICE Logical Channel list	ALL
1	- Semi-static Transport Format information	
	- Transmission time interval	Reference to TS34.108 clause 6.10 Parameter Set
	- Type of channel coding	Reference to TS34.108 clause 6.10 Parameter Set
	- Coding Rate	Reference to TS34.108 clause 6.10 Parameter Set
	- Rate matching attribute	Reference to TS34.108 clause 6.10 Parameter Set
	- CRC size	Reference to TS34.108 clause 6.10 Parameter Set
	DRAC static information	Not Present
	DL Transport channel information common for all	
	transport channel	
	- SCCPCH TFCS	Not Present
	- CHOICE DL parameters	Independent (This IE is reported for TEC number)
L	- DL DCH TFCS	(This IE is repeated for TFC number.)

- Normal
- TFCI Field 1 information
- CHOICE TFC representation
- TFCS addition information
- CHOICE CTFC Size
- CTFC information
- Power offset information

Added or Reconfigured DL TrCH information

- Transport channel identity
- CHOICE DL parameters
- CHOICE Transport channel type
- TFS
- Dynamic Transport format information
- RLC Size
- Number of TBs and TTI List
 - Transmission Time Interval
 - Number of Transport blocks
 - RLC size
- CHOICE Logical Channel list
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size
 - DCH quality target
 - BLER Quality value
 - Transparent mode signalling info

Frequency info

- UARFCN uplink(Nu)
- UARFCN downlink(Nd)

Maximum allowed UL TX power

Uplink DPCH info

- -Uplink DPCH power control info
- -DPCCH Power offset
- -PC Preamble
- -Power Control Algorithm
- -TPC step size
- Scrambling code type
- Scrambling code number
- Number of DPDCH
- spreading factor
- TFCI existence
- Number of FBI bits
- Puncturing Limit

CHOICE Mode

- Downlink PDSCH information

Downlink information common for all radio links

- Downlink DPCH info common for all RL
- -- CHOICE mode
- Timing indicator
- CFN-targetSFN frame offset
- Downlink DPCH power control information
- DPC mode
- DL rate matching restriction information
- Spreading factor
- Fixed or Flexible Position
- TFCI existence
- Number of bits for Pilot bits(SF=128,256)
- DPCH compressed mode info
- TGPSI
- TGPS Status Flag
- TGCFN
- Transmission gap pattern sequence configuration parameters
- TGMP

Addition

Number of bits used must be enough to cover all combinations of CTFC from clause 6.10. Refer to TS34.108 clause 6.10 Parameter Set Not Present

10

Independent

Dedicated transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

-6.30.00

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set 33dBm

-6dB

8slot

Algorithm1

1dB

Long

0 (0 to 16777215)

Not Present(1)

SF is reference to TS34.108 clause 6.10 Parameter Set

TRUE

Not Present(0)

Reference to TS34.108 clause 6.10 Parameter Set

FDD

Not Present

Not Present

Maintain Not Present

0 (single)

Not Present

Reference to TS34.108 clause 6.10 Parameter Set

Flexible TRUE Not Present

.

inactive

(Current CFN + (256 - TTI/10msec)) mod 256

FDD Measurement

- TGPRC **TGCFN** (Current CFN + (256 - TTI/10msec)) mod 256 - TGSN 8 - TGL1 10 - TGL2 5 - TGD 15 - TGPL1 35 - TGPL2 35 - RPP Mode 1 Mode 1 - ITP - UL/DL Mode DL - Downlink compressed mode method SF/2 - Uplink compressed mode method Not Present - Downlink frame type - DeltaSIR1 2.0 - DeltaSIRafter1 1.0 Not Present - DeltaSIR2 - DeltaSIRafter2 Not Present - TX Diversity mode None - SSDT information Not Present - S field - Code Word Set - Default DPCH Offset Value 0 Downlink information for each radio links - Primary CPICH info - Primary scrambling code 100 - PDSCH with SHO DCH info Not Present - PDSCH code mapping Not Present - Downlink DPCH info for each RL - Primary CPICH usage for channel estimation Primary CPICH may be used - DPCH frame offset 0 chips - Secondary CPICH info Not Present - DL channelisation code - Secondary scrambling code Reference to TS34.108 clause 6.10 Parameter Set - Spreading factor - Code number SF-1(SF is reference to TS34.108 clause 6.10 Parameter Set) - Scrambling code change No change - TPC combination index 0 - SSDT Cell Identity -a - Closed loop timing adjustment mode Not Present Not Present - Secondary CCPCH info - TFCS Not Present - FACH/PCH information Not Present - References to system information blocks Not Present Scheduling information

Contents of TRANSPORT CHANNEL RECONFIGURATION message: AM or UM (Packet to CELL_DCH from CELL_FACH in PS)

ſ	Information Flores	VI-1
ļ	Information Element	Value/remark
ا	Message Type RRC transaction identifier	Arbitrarily selects an integer between 0 and 3
I	Integrity check info	The presence of this IE is dependent on IXIT statements
	magny sheak and	in TS 34.123-2. If integrity protection is indicated to be
		active, this IE is present with the values of the sub IEs as
		stated below. Else, this IE and the sub-IEs are omitted.
	- message authentication code	SS calculates the value of MAC-I for this message and
	550	writes to this IE.
	- RRC message sequence number	SS provides the value of this IE, from its internal counter.
	Integrity protection mode info Ciphering mode info	Not Present Not Present
	Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
	New U-RNTI	Not Present
	New C-RNTI	Not Present
	RRC State DRX-indicator	NoDRXCELL_DCH
	UTRAN DRX cycle length coefficient	Not Present
	CN information info	Not Present
	URA identity RB with PDCP information	Not Present Not Present
	UL Transport channel information for all transport	INOLI IGOGIIL
	channels	
	- TFC subset	(This IE is repeated for TFC number.)
	 Allowed Transport Format combination 	0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108
	DD 4 OU TEOO	clause 6.10 Parameter Set.)
	- PRACH TFCS - CHOICE mode	Not Present FDD
	- UL DCH TFCS	(This IE is repeated for TFC number.)
	- Normal	(This is is repeated for it o hamber.)
	- TFCI Field 1 information	
	- CHOICE CTFC representation	Addition
	- TFCS addition information	
	CHOICE CTEC Size	Number of hits used must be enough to sever all
	- CHOICE CTFC Size	Number of bits used must be enough to cover all combinations of CTFC from clause 6.10.
	- CTFC information	Refer to TS34.108 clause 6.10 Parameter Set
	- Power offset information	
	- CHOICE Gain Factors	Signalled Gain Factors
	- Gain factor •c	0
	- Gain factor •d	0
	- Reference TFC ID	Not Present
	- Power offset Pp-m Added or Reconfigured UL TrCH information	OdB
	- Transport channel identity	1
	- TFS	
	- CHOICE Transport channel type	Dedicated transport channels
ا،	- Dynamic Transport format information	(This IE is repeated for TFI number)
$\ $	- RLC Size- Number of TBs and TTI List	Reference to TS34.108 clause 6.10 Parameter Set (This IE is repeated for TFI number.)
	- Transmission Time Interval	Not Present
	- Number of Transport blocks	Reference to TS34.108 clause 6.10 Parameter Set
$\ $		Reference to TS34.108 clause 6.10 Parameter Set
	- CHOICE Logical Channel list	<u>ALL</u>
	- Semi-static Transport Format information	Deference to TC24 100 clause C 40 Deverter Cc4
	- Transmission time interval - Type of channel coding	Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set
	- Coding Rate	Reference to TS34.108 clause 6.10 Parameter Set
	- Rate matching attribute	Reference to TS34.108 clause 6.10 Parameter Set
	- CRC size	Reference to TS34.108 clause 6.10 Parameter Set
	Added or Reconfigured UL TrCH information	
IJ	- Transport channel identity	<u>5</u> 2
ا	- TFS	Dedicated transport channels
IJ	- CHOICE Transport channel type - Dynamic Transport format information	Dedicated transport channels (This IE is repeated for TFI number)
I	- RLC Size	Reference to TS34.108 clause 6.10 Parameter Set
۱L		THE PROPERTY OF THE PROPERTY O

- Number of TBs and TTI List
- Transmission Time Interval
- Number of Transport blocks
- RLC size
- CHOICE Logical Channel list
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size

DRAC static information

DL Transport channel information common for all transport channel

- SCCPCH TFCS
- CHOICE DL parameters
- DL DCH TFCS
- Normal
- TFCI Field 1 information
- CHOICE CTFC representation
- TFCS addition information
- CHOICE CTFC Size
- CTFC information
- Power offset information

Added or Reconfigured DL TrCH information

- Transport channel identity
- CHOICE DL parameters
- · TFS
- CHOICE Transport channel type
- Dynamic Transport format information
- RLC Size
- Number of TBs and TTI List
- Transmission Time Interval
- _- Number of Transport blocks
- RLC size
 - CHOICE Logical Channel list
 - Semi-static Transport Format information
 - Transmission time interval
 - Type of channel coding
 - Coding Rate
 - Rate matching attribute
 - CRC size
 - DCH quality target
 - BLER Quality value
 - Transparent mode signalling info

Added or Reconfigured DL TrCH information

- Transport channel identity
- CHOICE DL parameters
- TFS
- CHOICE Transport channel type
- Dynamic Transport format information
- RLC Size
- Number of TBs and TTI List
- Transmission Time Interval
- Number of Transport blocks
- RLC size
- CHOICE Logical Channel list
- Semi-static Transport Format information
- Transmission time interval
- Type of channel coding
- Coding Rate
- Rate matching attribute
- CRC size
- DCH quality target
- BLER Quality value
- Transparent mode signalling info

(This IE is repeated for TFI number.)

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Not Present

Not Present

(This IE is repeated for TFC number.)

Addition

Number of bits used must be enough to cover all combinations of CTFC from clause 6.10.
Refer to TS34.108 clause 6.10 Parameter Set Not Present

64

Independent

Dedicated transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set

(This IE is repeated for TFI number.)

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

<u>-6.3</u>0.00 Not Present

<u>102</u>

Independent

Dedicated transport channels

(This IE is repeated for TFI number)

Reference to TS34.108 clause 6.10 Parameter Set (This IE is repeated for TFI number.)

Not Present

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set ALL

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

<u>-6.3</u>0.00 Not Present Frequency info - UARFCN uplink(Nu) Reference to TS34.108 clause 6.10 Parameter Set - UARFCN downlink(Nd) Reference to TS34.108 clause 6.10 Parameter Set Maximum allowed UL TX power 33dBm Uplink DPCH info -Uplink DPCH power control info -DPCCH Power offset -6dB -PC Preamble 8slot Algorithm1 -Power Control Algorithm -TPC step size 1dB - Scrambling code type Long - Scrambling code number 0 (0 to 16777215) - Number of DPDCH Not Present(1) - spreading factor SF is reference to TS34.108 clause 6.10 Parameter Set - TFCI existence - Number of FBI bits Not Present(0) - Puncturing Limit Reference to TS34.108 clause 6.10 Parameter Set **CHOICE Mode FDD** - Downlink PDSCH information Not Present Downlink information common for all radio links - Downlink DPCH info common for all RL **EDD** CHOICE mode - Timing indicator Maintain - CFN-targetSFN frame offset Not Present - Downlink DPCH power control information - DPC mode 0 (single) - DL rate matching restriction information Not Present - Spreading factor Reference to TS34.108 clause 6.10 Parameter Set - Fixed or Flexible Position Flexible - TFCI existence **TRUE** - Number of bits for Pilot bits(SF=128,256) Reference to TS34.108 clause 6.10 Parameter Set - DPCH compressed mode info - TGPSI - TGPS Status Flag inactive - TGCFN (Current CFN + (256 - TTI/10msec)) mod 256 - Transmission gap pattern sequence configuration parameters - TGMP **FDD Measurement** - TGPRC **TGCFN** (Current CFN + (256 - TTI/10msec)) mod 256 - TGSN 8 10 - TGL1 - TGL2 5 - TGD 15 - TGPL1 35 - TGPL2 35 - RPP Mode 1 - ITP Mode 1 - UL/DL Mode DΙ - Downlink compressed mode method SF/2 - Uplink compressed mode method Not Present - Downlink frame type 2.0 - DeltaSIR1 - DeltaSIRafter1 1.0 Not Present - DeltaSIR2 - DeltaSIRafter2 Not Present - TX Diversity mode None Not Present - SSDT information - Default DPCH Offset Value Downlink information for each radio links - Primary CPICH info - Primary scrambling code 100 - PDSCH with SHO DCH info Not Present - PDSCH code mapping Not Present - Downlink DPCH info for each RL - Primary CPICH usage for channel estimation Primary CPICH may be used - DPCH frame offset 0 chips - Secondary CPICH info Not Present

- DL channelisation code - Secondary scrambling code - Spreading factor Reference to TS34.108 clause 6.10 Parameter Set - Code number SF-1(SF is reference to TS34.108 clause 6.10 Parameter Set) - Scrambling code change No change - TPC combination index 0 - SSDT Cell Identity
- Closed loop timing adjustment mode
- Secondary CCPCH info -a Not Present Not Present - TFCS Not Present - FACH/PCH information Not Present - References to system information blocks Not Present Scheduling information

Γ	Information Element	Value/remark
ŀ	Message Type	Valdorionian
I	RRC transaction identifier	Arbitrarily selects an integer between 0 and 3
	Integrity check info	The presence of this IE is dependent on IXIT statements
		in TS 34.123-2. If integrity protection is indicated to be
		active, this IE is present with the values of the sub IEs as
	manage and the settlem and a	stated below. Else, this IE and the sub-IEs are omitted.
	- message authentication code	SS calculates the value of MAC-I for this message and writes to this IE.
	- RRC message sequence number	SS provides the value of this IE, from its internal counter.
	Integrity protection mode info	Not Present
	Ciphering mode info	Not Present
	Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
	New U-RNTI	Not Present
	New C-RNTI	Not Present
	RRC State DRX-indicator	NoDRXCELL_FACH
	UTRAN DRX cycle length coefficient CN information info	Not Present Not Present
	URA identity	Not Present
	RB with PDCP information	Not Present
	UL Transport channel information for all transport	
	channels	
	- TFC subset	(This IE is repeated for TFC number.)
	- Allowed Transport Format combination	0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108 clause 6.10 Parameter Set.)
	- PRACH TFCS	Not Present
	- CHOICE mode	FDD
	- UL DCH TFCS	Not Present
	Added or Reconfigured UL TrCH information	
	- Transport channel identity	<u>5</u> 4
ı	- TFS	Common transport channels
l	- CHOICE Transport channel type - Dynamic Transport format information	Common transport channels (This IE is repeated for TFI number)
	- RLC Size	Reference to TS34.108 clause 6.10 Parameter Set
	- Number of TBs and TTI List	(This IE is repeated for TFI number.)
	 Number of Transport blocks 	Reference to TS34.108 clause 6.10 Parameter Set
		Reference to TS34.108 clause 6.10 Parameter Set
	- CHOICE mode - CHOICE Logical Channel list	FDD ALL
l	- Semi-static Transport Format information	ALL
	- Transmission time interval	Reference to TS34.108 clause 6.10 Parameter Set
	- Type of channel coding	Reference to TS34.108 clause 6.10 Parameter Set
	- Coding Rate	Reference to TS34.108 clause 6.10 Parameter Set
	- Rate matching attribute	Reference to TS34.108 clause 6.10 Parameter Set
	- CRC size DRAC static information	Reference to TS34.108 clause 6.10 Parameter Set Not Present
	DL Transport channel information common for all	NOTE TESENT
	transport channel	
	- SCCPCH TFCS	(This IE is repeated for TFC number.)
	- Normal	
	- TFCI Field 1 information	A dallei
	CHOICE CTFC representation TFCS addition information	Addition
	- TPGS addition information	
	- CHOICE CTFC Size	
	- CTFC information	Refer to TS34.108 clause 6.10 Parameter Set
	- Power offset information	Not Present
	- CHOICE DL parameters	Independent
	- DL DCH TFCS	Not Present
ıl	Added or Reconfigured DL TrCH information - Transport channel identity	104
'	- CHOICE DL parameters	Independent
	- TFS	
	- CHOICE Transport channel type	Common transport channels
L	- Dynamic Transport format information	(This IE is repeated for TFI number)

- RLC Size - Number of TBs and TTI List _- Number of Transport blocks **RLC size** - CHOICE mode - CHOICE Logical Channel list - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size - DCH quality target BLER Quality value - Transparent mode signalling info Added or Reconfigured DL TrCH information - Transport channel identity - CHOICE DL parameters - CHOICE Transport channel type - Dynamic Transport format information - RLC Size - Number of TBs and TTI List - Number of Transport blocks **RLC size** - CHOICE mode - CHOICE Logical Channel list - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size - DCH quality target BLER Quality value - Transparent mode signalling info Added or Reconfigured DL TrCH information - Transport channel identity - CHOICE DL parameters - TFS - CHOICE Transport channel type - Dynamic Transport format information - RLC Size - Number of TBs and TTI List - Number of Transport blocks - RLC size - CHOICE mode - CHOICE Logical Channel list - Semi-static Transport Format information - Transmission time interval - Type of channel coding - Coding Rate - Rate matching attribute - CRC size

- DCH quality target **BLER Quality value** - Transparent mode signalling info Added or Reconfigured DL TrCH information - Transport channel identity - CHOICE DL parameters - TFS

- CHOICE Transport channel type

- Number of TBs and TTI List - Number of Transport blocks

- RLC Size

RLC size

- CHOICE mode

- Dynamic Transport format information

Reference to TS34.108 clause 6.10 Parameter Set (This IE is repeated for TFI number.) Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set FDD ALL Reference to TS34.108 clause 6.10 Parameter Set Not Present Not Present Independent Common transport channels (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set (This IE is repeated for TFI number.) Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set FDD ALL Reference to TS34.108 clause 6.10 Parameter Set Not Present Not Present Independent Common transport channels (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set (This IE is repeated for TFI number.) Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set FDD <u>ALL</u>

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

Not Present

Not Present

Independent

Common transport channels (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set (This IE is repeated for TFI number.) Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set FDD

- CHOICE Logical Channel list

- Semi-static Transport Format information

- Transmission time interval

- Type of channel coding

- Coding Rate

- Rate matching attribute

- CRC size

- DCH quality target

- Transparent mode signalling info

Frequency info

- UARFCN uplink(Nu)

- UARFCN downlink(Nd)

Maximum allowed UL TX power

CHOICE channel requirement

- CHOICE mode

—- Available Signature

- Available SF

Scrambling code number

- Puncturing Limit

- Available Sub Channel number

CHOICE Mode

- Downlink PDSCH information

Downlink information common for all radio links

Downlink information for each radio links

- Primary CPICH info

-- Primary scrambling code

-- PDSCH with SHO DCH info

- PDSCH code mapping

- Downlink DPCH info for each RL

- Secondary CCPCH info

Selection Indicator

- Primary CPICH usage for channel estimation

- Secondary CPICH info

-- Secondary scrambling code

SSDT Indicator

Spreading factor

- Code number

- Pilot symbol existence

- TFCI existence

Fixed or Flexible Position

- Timing offset

- TFCS

- FACH/PCH information

- References to system information blocks

- Scheduling information

<u>ALL</u>

Reference to TS34.108 clause 6.10 Parameter Set

Not Present

Not Present

Reference to TS34.108 clause 6.10 Parameter Set

Reference to TS34.108 clause 6.10 Parameter Set

33dBm

PRACH info (for RACH)Not Present

FDD

'0000 0000 1111 1111'B

Reference to TS34.108 clause 6.10 Parameter Set

θ

Reference to TS34.108 clause 6.10 Parameter Set

'1111 1111 1111'B

FDD

Not Present

Not Present

100

Not Present

Not Present

Not Present

Not Present

Primary CPICH may be used

Not Present

4

FALSE

Reference to TS34.108 clause 6.10 Parameter Set

SF-1(SF is reference to TS34.108 clause 6.10 Parameter

Set)

FALSE

TRUE

Flexible

0

Not Present

Not Present

Not Present

Contents of TRANSPORT CHANNEL RECONFIGURATION message: AM or UM (Packet to CELL_FACH from CELL_FACH in PS)

r	Holl GLLL_I ACITILITY 3)	
ļ	Information Element	Value/remark
ا،	Message Type RRC transaction identifier	Arbitrarily selects an integer between 0 and 3
1	Integrity check info	The presence of this IE is dependent on IXIT statements
	mognly oncon the	in TS 34.123-2. If integrity protection is indicated to be
		active, this IE is present with the values of the sub IEs as
		stated below. Else, this IE and the sub-IEs are omitted.
	- message authentication code	SS calculates the value of MAC-I for this message and
	DDC massage as a superior a superior	writes to this IE.
	- RRC message sequence number Integrity protection mode info	SS provides the value of this IE, from its internal counter. Not Present
	Ciphering mode info	Not Present
	Activation time	(256+CFN-(CFN MOD 8 + 8))MOD 256
	New U-RNTI	Not Present
	New C-RNTI	0000 0000 0000 0010B
	RRC State DRX-indicator	NoDRXCELL_FACH
	UTRAN DRX cycle length coefficient	Not Present
	CN information info	Not Present
	URA identity RB with PDCP information	Not Present Not Present
	UL Transport channel information for all transport	Not i lesent
	channels	
	- TFC subset	(This IE is repeated for TFC number.)
	 Allowed Transport Format combination 	0 to MaxTFCValue-1 (MaxTFCValue is refer to TS34.108
		clause 6.10 Parameter Set.)
	- PRACH TFCS	Not Present
	- CHOICE mode - UL DCH TFCS	FDD Not Present
	Added or Reconfigured UL TrCH information	Not Fresent
I	- Transport channel identity	154
	- TFS	<u></u> -
	 CHOICE Transport channel type 	Common transport channels
.	- Dynamic Transport format information	(This IE is repeated for TFI number)
	- RLC Size	Reference to TS34.108 clause 6.10 Parameter Set
	 Number of TBs and TTI List Number of Transport blocks 	(This IE is repeated for TFI number.) Reference to TS34.108 clause 6.10 Parameter Set
	- RLC size	Reference to TS34.108 clause 6.10 Parameter Set
	- CHOICE mode	FDD
	- CHOICE Logical Channel List	ALL
	 Semi-static Transport Format information 	
	- Transmission time interval	Reference to TS34.108 clause 6.10 Parameter Set
	- Type of channel coding - Coding Rate	Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set
	- Rate matching attribute	Reference to TS34.108 clause 6.10 Parameter Set
	- CRC size	Reference to TS34.108 clause 6.10 Parameter Set
	DRAC static information	Not Present
	DL Transport channel information common for all	
	transport channel	(TI: 15:
	- SCCPCH TFCS - Normal	(This IE is repeated for TFC number.)
	- TFCI Field 1 information	
	- CHOICE CTFC representation	Addition
	- TFCS addition information	, 133.113.1
	- CHOICE CTFC Size	Number of bits used must be enough to cover all
		combinations of CTFC from clause 6.10.
	- CTFC information	Refer to TS34.108 clause 6.10 Parameter Set
	- Power offset information	Independent
	 CHOICE DL parameters DL DCH TFCS 	Independent Not Present
	Added or Reconfigured DL TrCH information	Hot i loodit
П	- Transport channel identity	4 <u>12</u>
	- CHOICE DL parameters	Independent
	- TFS	
 [- CHOICE Transport channel type	Common transport channels

- Dynamic Transport format information
- RLC Size - Number of TBs and TTI List
- Number of Transport blocks
- RLC size
- CHOICE mode
- CHOICE Logical Channel List - Semi-static Transport Format information
- Transmission time interval
- Type of channel coding
- Coding Rate
- Rate matching attribute - CRC size
- DCH quality target
- Transparent mode signalling info
Added or Reconfigured DL TrCH information - Transport channel identity
- CHOICE DL parameters
- TFS
- CHOICE Transport channel type
- Dynamic Transport format information
- RLC Size - Number of TBs and TTI List
- Number of Transport blocks
- RLC size
- CHOICE mode - CHOICE Logical Channel List
- Semi-static Transport Format information
- Transmission time interval
- Type of channel coding
- Coding Rate - Rate matching attribute
- CRC size
- DCH quality target
- Transparent mode signalling info
Added or Reconfigured DL TrCH information - Transport channel identity
- CHOICE DL parameters
- TFS
- CHOICE Transport channel type - Dynamic Transport format information
- RLC Size
- Number of TBs and TTI List
Number of Transport blocks
- CHOICE mode
- CHOICE Logical Channel List
- Semi-static Transport Format information
- Transmission time interval - Type of channel coding
- Coding Rate
- Rate matching attribute
- CRC size
DCH quality target Transparent mode signalling info
Added or Reconfigured DL TrCH information
——————————————————————————————————————
Dynamic Transport format information
Number of Transport blocks
Transmission time interval
- Type of channel coding
- Rate matching attribute

(This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set (This IE is repeated for TFI number.) Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set FDD ALL Reference to TS34.108 clause 6.10 Parameter Set Not Present Not Present 132 Independent Common transport channels (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set (This IE is repeated for TFI number.) Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set FDD <u>ALL</u> Reference to TS34.108 clause 6.10 Parameter Set Not Present Not Present 143 Independent Common transport channels (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set (This IE is repeated for TFI number.) Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set FDD ALL Reference to TS34.108 clause 6.10 Parameter Set Not Present Not Present Independent Common transport channels (This IE is repeated for TFI number) Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set Reference to TS34.108 clause 6.10 Parameter Set

	Reference to TS34.108 clause 6.10 Parameter Set
- DCH quality target	Not Present
- Transparent mode signalling info	Not Present
Frequency info	
- UARFCN uplink(Nu)	Reference to TS34.108 clause 6.10 Parameter Set
- UARFCN downlink(Nd)	Reference to TS34.108 clause 6.10 Parameter Set
Maximum allowed UL TX power	33dBm
CHOICE channel requirement	PRACH info (for RACH)Not Present
CHOICE mode	FDD
- Available Signature	'0000 0000 1111 1111'B
— - Available SF	Reference to TS34.108 clause 6.10 Parameter Set
- Scrambling code number	0
	Reference to TS34.108 clause 6.10 Parameter Set
- Available Sub Channel number	'1111 1111 1111'B
CHOICE Mode	FDD
- Downlink PDSCH information	Not Present
Downlink information common for all radio links	Not Present
Downlink information for each radio links	Not Present
- Primary scrambling code	150
	Not Present
- PDSCH code mapping	Not Present
- Downlink DPCH info for each RL	Not Present
- Selection Indicator	Not Present
- Primary CPICH usage for channel estimation	Primary CPICH may be used
	Not Present
- Secondary scrambling code	1
	FALSE
- Spreading factor	Reference to TS34.108 clause 6.10 Parameter Set
- Code number	SF-1(SF is reference to TS34.108 clause 6.10 Parameter
	Set)
- Pilot symbol existence	FALSE
	TRUE
- Fixed or Flexible Position	Flexible
- Timing offset	0
— - TFCS	Not Present
- FACH/PCH information	Not Present
- References to system information blocks	Not Present
- Scheduling information	

Contents of TRANSPORT CHANNEL RECONFIGURATION COMPLETE message: AM

Information Element	Value/remark
Message Type	
RRC transaction identifier	Checked to see if the value is identical to the same IE in
	the downlink TRANSPORT CHANNEL
	RECONFIGURATION message
Integrity check info	The presence of this IE is dependent on IXIT statements
	in TS 34.123-2. If integrity protection is indicated to be
	active, this IE shall be present with the values of the sub
	IEs as stated below. Else, this IE and the sub-IEs shall
	be absent.
- Message authentication code	This IE is checked to see if it is present. The value is
	compared against the XMAC-I value computed by SS.
- RRC Message sequence number	This IE is checked to see if it is present. The value is
	used by SS to compute the XMAC-I value.
Uplink integrity protection activation info	Not checked
Other information element	Not checked
COUNT-C activation time	The presence of this IE depends on the following 2
	factors: (a) There exists RB(s) mapped to RLC-TM and
	(b) UE is transiting to CELL_DCH state after the
	reconfiguration procedure. Else, this IE is absent.
Radio bearer uplink ciphering activation time info	Not checked
RB with PDCP information list	Not checked

Contents of TRANSPORT FORMAT COMBINATION CONTROL message: AM or UM (in CELL_DCH)

Information Element	Value/remark
Message Type	
RRC transaction identifier	Arbitrarily selects an integer between 0 and 3
Integrity check info	The presence of this IE is dependent on IXIT statements
	in TS 34.123-2. If integrity protection is indicated to be
	active, this IE is present with the values of the sub IEs as
	stated below. Else, this IE and the sub-IEs are omitted.
- Message authentication code	SS calculates the value of MAC-I for this message and writes to this IE.
- RRC Message sequence number	SS provides the value of this IE, from its internal counter.
DPCH TFCS in Uplink	
 Allowed Transport format combination 	0 (The TFC is constructed from ALL TF0)
—index	

Contents of UE CAPABILITY ENQUIRY message: [TBD]AM or UM

Information Element	Value/remark
Message Type	
RRC transaction identifier	Arbitrarily selects an integer between 0 and 3
Integrity check info	The presence of this IE is dependent on IXIT statements
	in TS 34.123-2. If integrity protection is indicated to be
	active, this IE is present with the values of the sub IEs as
	stated below. Else, this IE and the sub-IEs are omitted.
- Message authentication code	SS calculates the value of MAC-I for this message and writes to this IE.
- RRC Message sequence number	SS provides the value of this IE, from its internal counter.
Capability update requirement	, , , , , , , , , , , , , , , , , , , ,
- UE radio access capability update requirement	TRUE
- System specific capability update requirement	UE only supports 1 system
list	
- System specific capability update requirement	GSM

Contents of UE CAPABILITY INFORMATION message: AM-or-UM

Information Element	Value/remark
Message Type	
RRC transaction identifier	Checked to see if the value is identical to the same IE in
	the downlink UE CAPABILITY ENQUIRY message.
Integrity check info	The presence of this IE is dependent on IXIT statements
	in TS 34.123-2. If integrity protection is indicated to be
	active, this IE shall be present with the values of the sub
	IEs as stated below. Else, this IE and the sub-IEs shall be
	absent.
- Message authentication code	This IE is checked to see if it is present. The value is
	compared against the XMAC-I value computed by SS.
- RRC Message sequence number	This IE is checked to see if it is present. The value is
	used by SS to compute the XMAC-I value.
UE radio access capability	Value will be checked. Stated capability must be
	compatible with 34.123-2 (ICS statements) and the user
	settings
- ICS Version	
- PDCP Capability	
- RLC Capability	
- Transport channel capability	
- RF Capability	
- Physical channel capability	
- UE multi-mode/multi-RAT capability	
- Security Capability	
- LCS Capability	
- Measurement capability UE system specific capability	Value will be checked. UE must include the classmark
OE System specific capability	
	information for the supported systemRAT

Contents of UE CAPABILITY INFORMATION CONFIRM message: UM

Information Element	Value/remark
Message Type	
Integrity check info	The presence of this IE is dependent on IXIT statements in TS 34.123-2. If integrity protection is indicated to be active, this IE is present with the values of the sub IEs as stated below. Else, this IE and the sub-IEs are omitted.
- Message authentication code	SS calculates the value of MAC-I for this message and writes to this IE.
- RRC Message sequence number	SS provides the value of this IE, from its internal counter.

Contents of URA UPDATE message: TM

Information Element	Value/remark
Message Type	
U-RNTI	
- SRNC identity	0000 0000 0001B
- S-RNTI	0000 0000 0000 0000 0001B
RRC transaction identifier	Checked to see if it is absent
Integrity check info	The presence of this IE is dependent on IXIT statements
	in TS 34.123-2. If integrity protection is indicated to be
	active, this IE shall be present with the values of the sub
	IEs as stated below. Else, this IE and the sub-IEs shall be
	absent.
- Message authentication code	This IE is checked to see if it is present. The value is
	compared against the XMAC-I value computed by SS.
- RRC Message sequence number	This IE is checked to see if it is present. The value is
	used by SS to compute the XMAC-I value.
AM_RLC error indication	Not checked
URA update cause	See the test content
Protocol error indicator	Not cChecked to see if it is absent or set to 'FALSE'
Protocol error information	Not cChecked to see if it is absent

Contents of URA UPDATE CONFIRM message: UM

Information Element	Value/remark
Message Type	
U-RNTI	If this message is sent on CCCH, use the following
	values. Else, this IE is absent.
- SRNC identity	0000 0000 0001B
- S-RNTI	0000 0000 0000 0000 0001B
RRC transaction identifier	Arbitrarily selects and integer between 0 and 3
Integrity check info	The presence of this IE is dependent on IXIT statements
	in TS 34.123-2. If integrity protection is indicated to be
	active, this IE is present with the values of the sub IEs as
	stated below. Else, this IE and the sub-IEs are omitted.
- message authentication code	SS calculates the value of MAC-I for this message and
	writes to this IE.
- RRC message sequence number	SS provides the value of this IE, from its internal counter.
Integrity protection mode info	Not Present
Ciphering mode info	Not Present
New U-RNTI	Not Present
New C-RNTI	Not Present
DRX RRC state indicator	NoDRXURA PCH
UTRAN DRX cycle length coefficient	Not Present
CN information info	Not Present
URA identity	See the test content
RB with PDCP information list	Not Present